#### **Rights and permissions**

Reprints and Permissions

### Copyright information

© 2021 The Author(s), under exclusive license to Springer Nature Switzerland AG

### About this chapter



#### Cite this chapter

Ramu, A., Ahmed, B.A. (2021). Marine Algal Products and Algal Nanoparticle Synthesis Against Cancer. In: Thangadurai, D., Sangeetha, J., Prasad, R. (eds) Bioprospecting Algae for Nanosized Materials. Nanotechnology in the Life Sciences. Springer, Cham. https://doi.org/10.1007/978-3-030-81557-8\_11

Download citation

<u>.RIS</u> <u>★</u> <u>.ENW</u> <u>★</u> <u>.BIB</u> <del>↓</del>

DOI	Published	Publisher Name
https://doi.org/10.1007/978-	27 February 2022	Springer, Cham
3-030-81557-8_11		
Print ISBN	Online ISBN	eBook Packages
978-3-030-81556-1	978-3-030-81557-8	<b>Biomedical and Life Sciences</b>
		Biomedical and Life Sciences
		( <u>R0</u> )



#### **INTRODUCTION TO DIABETES MELLITUS**

DIABETES, CAUSES, PANCREATIC ISLET TRANSPLANTATION, GENETICS, SIGNS AND SYMPTOMS, COMPLICATIONS, TESTS AND DIAGNOSIS

Scholar's Press ( 2020-02-26 )

€ 79,90

Buy at the MoreBooks! Shop

This book Introduction on Diabetes Mellitus provides a concise coverage of the concepts related to diabetes in terms of different types of diabetes, causes, signs and symptoms, tests and diagnosis, complications, prevention and treatment. Health is a positive concept emphasizing social and personal resources as well as physical capabilities. Disease is a term describing any deviation from the normal state of health or wellness. It includes physical, mental and social conditions. Both health and disease are coeval with life. Since archaic time, human are interested in trying to control diseases. Medical are involved in various ways to cure human disease and control the disorders in order to bring relief to the sick. In order to create awareness and to provide ample information to human society, this book has been designed carefully with the recent available modern tests to diagnose the disease, advanced methods to treat the complications covering all treatment and therapies such as dietary management, yoga, meditation, acupressure, acupuncture, massage therapy, aroma therapy, music therapy, biofeedback, ayurvedic herbal medicine, homeopathy and allopathic medicine.

#### Book Details:

ISBN-13:	978-613-8-92491-3
ISBN-10:	6138924916
EAN:	9786138924913
Book language:	English
By (author) :	AZHAGU MADHAVAN SIVALINGAM GANESAN SIVAMANI ARJUN PANDIAN
Number of pages:	204
Published on:	2020-02-26
Category:	Care



AGRICULTURE ISSUES AND POLICIES





Cardamom [Elettaria Cardamomum (L.) Maton]: Production, Breeding, Management, Phytochemistry and Health Benefits

\$230.00



#### Series: Agriculture Issues and Policies BISAC: TEC003030 DOI: https://doi.org/10.52305/LHSK2874

This book, entitled *Cardamom [Elettaria cardamomum (L.) Maton]: Production, Breeding, Management, Phytochemistry and Health Benefits*, contains valuable information about the cardamom. The cultivation, botany, plant breeding, biotechnology, phytochemical analysis, pest and disease management and pesticide residues of cardamom [Elettaria cardamomum (L.) Maton] are well described. This book also analyzes the phytochemistry and pharmacological properties of cardamom that are used to design novel drugs and to increase its value to the food and pharmaceutical industries. The book is aimed at researchers, academicians, pharmacologists, nutritionists, chemists, biotechnologists, and post-graduate and doctoral students involved in related research activities as well as cardamom planters.

Binding		
Hardcover	~	🔁 Clear

Publication Date: September 20, 2021 Status: Available Pages: 345



Add to cart

#### Add to Wishlist

ISBN: 978-1-68507-097-7

Categories: 2021, Agriculture, Agriculture Issues and Policies, Books, Crops, Nova, Science and Technology, Technology and Engineering

Like 0

	Publish with Us	Additional information	Table of Contents	
--	-----------------	------------------------	-------------------	--





#### Acknowledgements

#### Acronyms

Chapter 1. History of Land and Agriculture in the Indian Cardamom Hills (M. Murugan, K. Ashokkumar, M. K. Dhanya and A. Subbiah – Cardamom Research Station, Kerala Agricultural University, Pampadumpara, Idukki, Kerala, India, et al.)

Chapter 2. Cultivation, Harvesting, Processing and Products of Cardamom (M. Nimisha, H. C. Vikram, K. Ashokkumar, M. K. Dhanya, M. Murugan and S. Anu – Cardamom Research Station, Kerala Agricultural University, Pampadumpara, Idukki, Kerala, India, et al.)

Chapter 3. Plant Breeding Approaches for Cardamom Improvement (Kaliyaperumal Ashokkumar, Muthusamy Murugan, M. K. Dhanya, M. Alagupalamuthirsolai, T. Sathyan and Silvas Jebakumar Prince – Cardamom Research Station, Kerala Agricultural University, Pampadumpara, Idukki, Kerala, India, et al.)

Chapter 4. Biotechnological Applications for Cardamom Improvement (V. G. Shobhana, A. Karthikeyan and R. Veera Ranjani – Crop Improvement program, International Crops Research Institute for the Semi-Arid Tropics, Hyderabad, India, et al.)

Chapter 5. Pest Complex in Cardamom and Their Management (K. B. Deepthy, T. Sathyan, M. K. Dhanya, M. Murugan, Sachin G. Pai and O. M. Shana – Department of Agricultural Entomology, College of Agriculture, Kerala Agricultural University, Vellanikkara, Thrissur, Kerala, India, et al.)

Chapter 6. Diseases of Small Cardamom and Their Management (M. K. Dhanya, R. Ayisha, M. Murugan, K. Ashokkumar, T. Sathyan, K. B. Deepthi, M. Aishwarya, V. Nijamol and M. Nimisha – Cardamom Research Station, Kerala Agricultural University, Pampadumpara, Kerala, et al.)

Chapter 7. Bio-Pesticides for the Management of Pests and Diseases in Cardamom (T. Sathyan, K. B. Deepthy, M. K. Dhanya, M. Murugan and O. M. Shana – Cardamom Research Station, Kerala Agricultural University, Pampadumpara, Idukki, Kerala, India, et al.)

Chapter 8. Pesticide Residues in Cardamom (Aparna Joseph, Ambily Paul and Thomas George – MSSRF-Community Agrobiodiversity Centre,





**Biological Activity** 

(Kaliyaperumal Ashokkumar, Muthusamy Murugan, M. K. Dhanya, Arjun Pandian, Saradhadevi Muthukrishnan and Sampathrajan Vellaikumar – Cardamom Research Station, Kerala Agricultural University, Pampadumpara, Kerala, India, et al.)

Chapter 10. Phytochemical Analysis and Pharmacological Potential of Cardamom (Somasundaram Ambiga, Raja Suja Pandian, Venkatesan Saratha, Kaliyaperumal Ashokkumar, Raju Ramasubbu, Lazarus Vijune Lawrence and Arjun Pandian – Department of Biochemistry, PRIST Deemed University, Thanjavur, Tamil Nadu, India, et al.)

Chapter 11. Antioxidant and Antimicrobial Activity of Cardamom Extracts (Priya Prasannan, Yasotha Jeyaram, Saranya Surendran, Sudharshan Sekar, Arjun Pandian and Raju Ramasubbu – Department of Biology, The Gandhigram Rural Institute, (Deemed to be University), Gandhigram, Dindigul, Tamil Nadu, India, et al.)

Chapter 12. Ethnopharmacology and Phytochemistry of Cardamom (Saranya Surendran, Priya Prasannan, Yasotha Jayaram, Arjun Pandian and Raju Ramasubbu – Department of Biology, The Gandhigram Rural Institute (Deemed to be University) Gandhigram, Dindigul, Tamil Nadu, India, et al.)

About the Editors

Index

Nova Science Publishers, Inc.

Nova publishes a wide array of books and journals from authors around the globe, focusing on Medicine and Health, Science and Technology and the Social Sciences and Humanities.

We publish over 1,500 new titles per year by leading researchers each year, and have a network of expert authors, editors and advisors spanning the global academic community in pursuit of advanced research developments.





## Cardamom [Elettaria Cardamomum (L.) Maton]: Production, Breeding, Management, Phytochemistry and Health Benefits



Publisher	Nova
Format	Fixed What's This?
Print ISBN	9781685070977, 1685070973
eText ISBN	9781685071325, 1685071325
Copyright	2021





## Advanced Green Materials

Fabrication, Characterization and Applications of Biopolymers and Biocomposites

Woodhead Publishing in Materials

2021, Pages 109-131



## 8 - Fabrication and applications of chitosanbased green materials

N. Anusha<sup>a</sup>, S. Ignacimuthu<sup>b</sup>, A. Sathiyaseelan<sup>c</sup>, A. Shajahan<sup>a</sup>

- Department of Biotechnology, PRIST (Deemed University), Puducherry Campus, Puducherry, India
- <sup>b</sup> Xavier Research Foundation, St. Xavier's College, Palayamkottai, Tirunelveli, Tamil Nadu, India
- <sup>c</sup> Centre for Advanced Studies in Botany, University of Madras, Chennai, Tamil Nadu, India

Available online 29 January 2021, Version of Record 29 January 2021.

## Show less + Add to Mendeley Share J Cite https://doi.org/10.1016/B978-0-12-819988-6.00008-2 7 Get rights and content 7

## **SPRINGER LINK**

Home > Nutritional Management and Metabolic Aspects of Hyperhomocysteinemia > Chapter

## Medical Aspects of Hyperhomocystinemia and Neurological Disorders

Preeja Prabhakar	×
Department of Biochemistry, Centre for Research and Development, PRIST	
University, Thanjavur, Tamil Nadu, India	
You can also search for this auth	orin

View author publications

an also search for this author in PubMed | Google Scholar

Preeja Prabhakar & Sivaprasad Punnaveetil

all Accesses

### Abstract

Homocysteine (Hcy) is a sulfur containing amino acid and is generated during methionine metabolism. Homocysteine has vital influences in DNA metabolism through the methylation process, which in turn is affected by the levels of vitamin B12, folate, and vitamin B6. Physiologically, Hcy levels are determined primarily by dietary intake and the vitamin status in the body. Hence, an increase in Hcy can be nutritional and is also attributable to deficiency of either vitamin B12 or folate, or even its combination. In the past few years of research, studies have observed association [1] between elevated homocysteine levels (eHcy) and conditions like stroke, cardiovascular diseases, ectopia lentis, vascular dementia, Parkinson's disease, Alzheimer's disease, and neuropsychiatric illnesses like schizonbrenia [2, 3]. Farly identification of hyper-homocysteinemia may be



Nutritional Management and Metabolic Aspects of Hyperhomocysteinemia



Cart

# FROM LAB TO FIELD -EMERGING TRENDS IN SEED TECHNOLOGY

978-93-6255-959-3

## Edited by

DR. VIBEETHABALA SUBRAMANIYAN

From Lab to Field - Emerging trends in Seed technology Copyright©2021 by **vibeethabala Subramanian** 

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-959-3

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### TABLE OF CONTENTS

Introduction	
Chapter I : S	etting the Stage: Pre-Germination Techniques Unveiled
•••••	
Tamilcovane s	seshachalam
Chapter II :	Sampling Secrets: Insights into Seed Sampling Techniques
Siva Kuppusa	
Chapter III :	Seedlings Unveiled: Exploring Types and Germination Test Apparatus
Akshaya Chel	lladurai
Chapter IV :	Extraction Expedition: Methods for Seed Extraction
Mr P Rajavel	L
Chapter V Bri	dging the Gap: Supplementary Pollination Techniques in Hybrid Rice
Mr C Sabarin	athan
Chapter VI Su	nflower Symphony: Supplementary Pollination Techniques in Sunflowers
Sowbika anan	d
Chapter VII Se	eed Genetic and Physical Purity examination
	77
Prof T Lakshi	minarayanan
Chapter VIII :	Techniques for long term seed viability
Mr S Balamu	irugan

Chapter Ix Digital Transformation In Seed Technology

	20
Dr K C Sivabalan	
Chapter X: Seed Innovations In Global Food Technology	
1	32
Jasmine Manimaran	
REFERENCES1	50

#### **CHAPTER I**

## SETTING THE STAGE: PRE-GERMINATION TECHNIQUES UNVEILED TAMILCOVANE SESHACHALAM

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Pre-germination techniques are crucial for optimizing seed germination and ensuring robust seedling development. These methods, such as scarification, stratification, and soaking, are designed to overcome seed dormancy and enhance seed viability. Scarification involves mechanically or chemically altering the seed coat to facilitate water absorption, while stratification mimics natural seasonal conditions by exposing seeds to cold or warm temperatures. Soaking seeds in water can soften hard seed coats, and pre-sprouting techniques, like using paper towels or hydroponic systems, allow seeds to begin germination in a controlled environment. Additionally, treatments with fungicides, gibberellic acid, or specific bacteria can further boost germination rates and seedling health. By employing these techniques, gardeners and farmers can improve the efficiency and success of their planting efforts.

The main goal of learning about pre-germination techniques is to optimize seed germination and seedling establishment, ensuring successful plant growth and development. Understanding these techniques allows you to:

**Improve Germination Rates**: By using methods tailored to the specific needs of different seeds, you can significantly enhance the speed and consistency of seed germination.

**Overcome Dormancy**: Many seeds have natural dormancy mechanisms that prevent germination until certain conditions are met. Pre-germination techniques help break or bypass these dormancy barriers.

#### **CHAPTER II**

## SAMPLING SECRETS: INSIGHTS INTO SEED SAMPLING TECHNIQUES SIVA KUPPUSAMY

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Seed sampling is a critical process in agriculture, research, and seed certification, ensuring that the seeds used are of high quality and have the desired characteristics. Proper sampling techniques help in assessing seed quality, viability, and purity. This chapter explores the key methods and considerations for effective seed sampling.

#### Importance of Seed Sampling

Seed sampling is essential for obtaining a representative sample of seeds from a bulk lot. Accurate sampling ensures that tests for seed quality, such as germination rates, purity, and disease presence, reflect the true nature of the entire lot. This process is vital for making informed decisions about seed selection and treatment.

#### Types of Seed Samples

**Composite Sample**: A composite sample is created by combining multiple individual samples from different parts of a seed lot. This method provides a more accurate representation of the entire lot and is commonly used for testing purposes.

**Grab Sample**: This is a single sample taken from one location within the seed lot. While simpler, grab samples can be less representative and are often used for preliminary assessments.

#### **CHAPTER III**

#### SEEDLINGS UNVEILED: EXPLORING TYPES AND GERMINATION TEST APPARATUS

#### AKSHAYA CHELLADURAI

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Germination apparatus plays a crucial role in the controlled study and optimization of seed germination processes. Whether for research, quality control, or practical horticultural use, the right equipment ensures that seeds receive the optimal conditions required for successful sprouting. This chapter provides an overview of various germination apparatuses, their functions, and considerations for their use.

#### Importance of Germination Apparatus

Germination apparatuses are designed to create and maintain ideal conditions for seed germination, including temperature, moisture, and light. Proper apparatus allows for the precise monitoring and adjustment of these conditions, facilitating research, ensuring quality control in seed production, and improving germination rates in both professional and amateur gardening.

#### Types of Germination Apparatus

**Germination Trays** Shallow trays with multiple cells or compartments for placing seeds.

Growth Cabinets Enclosed units with adjustable temperature, humidity, and light settings

Germination apparatus is an essential tool for optimizing seed germination and seedling growth.

#### CHAPTER IV

#### **EXTRACTION EXPEDITION: METHODS FOR SEED EXTRACTION**

#### **MR P RAJAVEL**

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

#### **Methods of Seed Extraction**

Seed extraction is a critical process in agriculture and horticulture, aimed at separating seeds from their surrounding fruit, pod, or capsule. Effective extraction methods are essential for ensuring seed quality, viability, and purity. Different methods are employed depending on the type of seed, its surrounding fruit or pod, and the intended use. This chapter outlines various seed extraction techniques and their applications.

**Manual Extraction** 

#### **Hand-Picking**

**Description**: Seeds are manually removed from the fruit or pod. This method is often used for small quantities of seeds or delicate seeds that require gentle handling.

#### **Pulp Removal**

**Description**: Seeds are separated from the pulp or flesh of fruits by hand or with simple tools.

#### Mechanical Extraction

**Seed Extractors :** Machines designed to separate seeds from fruits or pods using various mechanical methods, such as shaking, abrasion, or air blast

#### **CHAPTER V**

#### BRIDGING THE GAP: SUPPLEMENTARY POLLINATION TECHNIQUES IN HYBRID RICE

#### MR C SABARINATHAN

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Supplementary pollination is a crucial technique used in the cultivation of hybrid rice to ensure optimal seed production and maximize yield. Hybrid rice, which results from crossing two genetically distinct rice varieties, benefits from supplementary pollination to enhance fertilization rates and improve seed set. This chapter delves into the principles, methods, and benefits of supplementary pollination in hybrid rice production.

#### Understanding Hybrid Rice and Pollination

Hybrid rice is developed by crossing two inbred lines to produce a variety with superior traits such as higher yield, disease resistance, and better adaptability. The success of hybrid rice cultivation relies on effective pollination, as hybrid vigor, or heterosis, is most pronounced when seeds are produced from controlled crosses.

In hybrid rice production, there are two main components: the male parent, which provides the pollen, and the female parent, which receives the pollen and produces the seeds. Proper pollination management is essential for achieving high seed set and quality.

Supplementary pollination is a vital technique in hybrid rice production, ensuring high seed set, improved yield, and the maintenance of hybrid vigor. By employing various methods such as hand pollination, mechanical pollinators, and pollen spraying, growers can enhance the efficiency of the pollination process and achieve better results in hybrid rice cultivation. Understanding and implementing effective supplementary pollination strategies are key to successful hybrid rice production and achieving optimal agricultural outcomes.

#### **CHAPTER VI**

#### SUNFLOWER SYMPHONY: SUPPLEMENTARY POLLINATION TECHNIQUES IN SUNFLOWERS

#### SOWBIKA ANAND

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Supplementary pollination techniques in sunflowers are important strategies used to enhance crop yields and ensure better seed production. Sunflowers (Helianthus annuus) are primarily pollinated by insects, particularly bees, but various supplementary techniques can be used to improve pollination efficiency and seed set. Here's an overview of some commonly used supplementary pollination techniques

#### **Mechanical Pollination**

**Description:** Mechanical pollination involves using artificial means to transfer pollen between flowers. This can be done using tools or machines that mimic the action of natural pollinators.

#### **Hand Pollination**

**Description:** This technique involves manually transferring pollen from one sunflower to another. It's typically done in small-scale or research settings.

#### **Use of Pollinator Attractants**

**Description:** This method involves attracting more natural pollinators to the sunflower crops by using attractants or creating a conducive environment.

#### **CHAPTER VII**

#### SEED GENETIC AND PHYSICAL PURITY EXAMINATION

#### PROF T LAKSHMI NARAYANAN

#### Associate professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Seed genetic and physical purity examination is a critical process in agriculture and seed production that ensures the quality and reliability of seeds used for planting. Here's a brief overview of both aspects:

Genetic purity, often associated with ideas of racial or ethnic superiority, is a concept rooted in misunderstandings of genetics and biology. It suggests that maintaining a "pure" genetic lineage can lead to a superior or more desirable population. However, from a scientific perspective, genetic diversity is crucial for the health and adaptability of any species. Inbreeding and the pursuit of genetic purity can lead to a host of problems,

including increased susceptibility to disease and genetic disorders. Evolution thrives on genetic variation, which enhances a population's ability to adapt to changing environments. Emphasizing genetic purity ignores the reality that all humans share a common ancestry and that our strength lies in our diversity and the ability to adapt through it. Physical Purity Examination

Physical purity often refers to ideals related to an individual's appearance or bodily health, typically framed by societal standards of beauty or fitness. This concept can manifest in various ways, from promoting certain body types or features as the epitome of attractiveness to advocating for strict health regimens aimed at achieving an idealized form of physical perfection

Objective: To assess the physical characteristics of seeds to ensure they are free from contaminants and conform to the standards of the specific seed variety.

#### **CHAPTER VIII**

#### **TECHNIQUES FOR LONG TERM SEED VIABLITY**

#### MR S BALAMURUGAN

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

#### **Long-Term Seed Viability**

Long-term seed viability is a critical factor in agriculture and plant conservation, as it determines how well seeds can remain viable over extended periods while maintaining their ability to germinate and produce healthy plants. This concept encompasses several key aspects, including storage conditions, seed physiology, and technological interventions. Here's an in-depth look at the factors affecting long-term seed viability and strategies for maintaining it:

#### Seed Dormancy and Longevity

#### **Seed Dormancy:**

**Definition:** Dormancy is a state in which seeds are prevented from germinating even under favorable conditions. It is a natural mechanism that allows seeds to survive adverse conditions and germinate when environmental conditions are more suitable.

**Types:** Dormancy can be physical (due to seed coat impermeability), physiological (related to internal seed conditions), or morphological (related to seed development).

#### **Impact on Longevity:**

Dormant seeds generally have a longer shelf life because they are less likely to deteriorate quickly. However, overcoming dormancy is crucial for effective germination when the time comes.

#### **CHAPTER IX**

#### DIGITAL TRANSFORMATION IN SEED TECHNOLOGY

#### DR K C SIVABALAN

#### Associate professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Digital transformation in seed technology represents a revolutionary shift in how seeds are developed, tested, produced, and managed. This transformation leverages digital tools and technologies to enhance efficiency, precision, and innovation across the seed supply chain. Here's an overview of how digital technologies are reshaping the seed industry:

#### **Digital Quality Control**

**Seed Testing:** Digital technologies enhance seed testing through advanced imaging and analysis tools that detect physical defects, disease, and genetic purity. Automated seed testing systems provide rapid and accurate results.

**Block chain for Traceability:** Block chain technology ensures transparency and traceability in the seed supply chain. It records and verifies each stage of seed production and distribution, enhancing accountability and quality assurance.

#### **Digital Marketplaces**

**E-Commerce Platforms:** Digital marketplaces for seeds facilitate easier access to a wide range of seed varieties. Farmers can compare options, read reviews, and make purchases online, enhancing market accessibility and competition.

**Supply Chain Optimization:** Digital tools optimize seed distribution and logistics, ensuring timely delivery and reducing the risk of supply chain disruptions.

#### **CHAPTER X**

#### SEED INNOVATIONS IN GLOBAL FOOD TECHNOLOGY

#### JASMINE MANIMARAN

#### Assistant professor, School of Agriculture, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

#### Seed Innovations in Global Food Technology

Seed innovations play a pivotal role in shaping the future of global food technology, addressing the challenges of food security, sustainability, and climate change. By enhancing seed characteristics and production methods, these innovations help meet the growing demand for food and improve agricultural resilience. Here's a detailed exploration of key seed innovations that are transforming global food technology:

#### Advanced Breeding Techniques

#### **Genomic Selection and Editing**

**CRISPR and Gene Editing:** CRISPR-Cas9 and other gene-editing technologies enable precise modifications of plant genomes to enhance traits such as yield, pest resistance, and drought tolerance. These techniques can introduce or modify genes with high accuracy, accelerating the development of improved seed varieties.

**Marker-Assisted Selection (MAS):** MAS uses molecular markers linked to desirable traits to streamline the breeding process. This technique speeds up the development of new varieties by allowing breeders to select plants with the best genetic profiles.

**Hybrid Seeds:** Hybrid varieties are produced by crossing two genetically distinct parent lines to create seeds with superior traits.

# HANDBOOK OF

# OLERICULTURE

# EDITED BY DR. B. VIBITHA BALA



#### Microbial Mastery: HANDBOOK OF OLERICULTURE

Copyright © 2021 by Dr. B. Vibitha Bala

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-171-9

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### TABLE OF CONTENT

Introduction
Chapter I: Basic techniques in Olericulture17
MR. T. LAKSHMINARAYANAN
Chapter II Seed production vegetable crops35
DR.V.A.THIRUPATHI
Chapter III Cool season vegetables55
MRS.A.SOWBIKA
Chapter IV: Warm season vegetables73
MS. C.K. AKSHAYA
ChapterV: Pest and Diseases of vegetable crops and their Management
MR.C.SABARINADHAN
Chapter VI : The Book of Cucurbits123
DR. P. SELVARAJ
Chapter VII : Underutilized and green leafy vegetables
DR. D.R. SUDHA
Chapter VIII: Seed Selection and Propagation Methods152
Mr. K. C. Sivabalan
Chapter IX: Greenhouse and Controlled Environment Agriculture168
Dr. K. Ashokkumar
Chapter X: Crop Rotation and Intercropping Strategies
Dr. Rajaguruvu

#### **REFERENCE**:

#### **CHAPTER I**

#### **Basic techniques in Olericulture**

#### Mr. T. Lakshminarayanan

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Olericulture, a branch of horticulture focused on the cultivation of vegetables, is an essential field for ensuring global food security and promoting healthy diets. As an interdisciplinary science, olericulture integrates principles from botany, agronomy, soil science, and pest management to optimize the production of vegetables. Understanding the basic units of olericulture is crucial for both novice and experienced growers as they navigate the complexities of vegetable farming. At its core, olericulture revolves around several fundamental units that form the foundation of successful vegetable production. These units include the vegetable plant, the growing environment, soil, and management practices. Each unit interacts dynamically to influence the overall productivity and quality of vegetable crops. Vegetable Plants: The vegetable plant itself is the primary unit in olericulture. Each vegetable species—whether it be tomatoes, carrots, or lettuce—has unique requirements and growth characteristics. Understanding these characteristics, including growth cycles, nutrient needs, and pest vulnerabilities, is vital for optimizing yields and ensuring plant health. Growing Environment: The environment in which vegetables are grown significantly impacts their development. This unit encompasses factors such as climate, temperature, light, and humidity. Each vegetable species thrives under specific environmental conditions, and successful olericulture requires the ability to adapt these conditions to suit the plants' needs. Techniques like greenhouse cultivation and controlled environment agriculture are increasingly utilized to manipulate growing conditions and extend growing seasons. Harvesting and Post-Harvest Handling: Timely harvesting is essential for the best flavor and quality of vegetables. Harvest crops at their peak maturity to ensure optimal taste and nutritional value. Use clean, sharp tools to avoid damaging plants and causing disease. After harvesting, handle vegetables carefully to prevent bruising and spoilage. Proper post-harvest storage, such as refrigeration or curing for certain crops, extends shelf life and maintains quality.In summary, basic techniques in olericulture involve meticulous soil preparation, careful planting, consistent watering, balanced fertilization, effective pest and disease management, diligent weed control, harvesting and handling. Mastering these fundamental practices will enhance vegetable production.

#### **Chapter II**

#### Seed production vegetable crops

#### **DR.V.A.THIRUPATHI**

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Seed production is a critical component of vegetable crop cultivation, underpinning both crop improvement and sustainable agriculture. The process involves a series of well-orchestrated steps to ensure the generation of high-quality seeds that possess optimal germination, vigor, and disease resistance. This chapter outlines the key aspects of seed production for vegetable crops, focusing on the principles, techniques, and practices that contribute to successful seed yield.1. Selection of Parent Plants: The foundation of successful seed production begins with the careful selection of parent plants. Highyielding, disease-resistant, and well-adapted varieties are chosen to ensure that the seeds produced will carry these desirable traits. For open-pollinated varieties, which rely on natural pollination mechanisms, it is crucial to select plants that exhibit uniformity and stability in their traits. For hybrid varieties, specific male and female lines must be selected and managed to achieve the desired hybrid vigor (heterosis).2. Pollination Management: Pollination is a critical step in seed production, especially for crops that are not self-pollinating. Insect-pollinated crops, such as cucumbers and melons, require careful management of pollinator populations. In some cases, manual pollination may be employed to ensure consistent seed set. For wind-pollinated crops, like corn, ensuring adequate spacing and avoiding crosspollination with other varieties are essential. Isolation distances between different varieties of the same crop are also necessary to prevent undesirable cross-pollination and preserve seed purity.3. Crop Management: Optimal crop management practices are vital to maximize seed vield and quality. This includes soil preparation, fertilization, irrigation, and pest management. Fertilization must be tailored to the specific nutrient needs of the crop at different growth stages, while irrigation schedules should ensure adequate moisture without causing waterlogging. Integrated pest management (IPM) strategies help control pests and diseases without compromising seed quality or the environment.4. Harvesting: Harvest timing is crucial to ensure that seeds reach their physiological maturity.

#### **Chapter III**

#### **Cool season vegetables**

#### MRS.A.SOWBIKA

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

As the days shorten and temperatures dip, the garden doesn't have to go into hibernation. Cool season vegetables thrive in the crisp, refreshing air of fall and early spring. These hardy plants not only tolerate but actually benefit from the cooler temperatures, producing crisp, flavorful crops that can often be harvested long after the first frosts. Leafy Greens: Leafy greens such as spinach, kale, and Swiss chard are the quintessential cool season crops. Spinach, with its tender, nutrient-rich leaves, is remarkably resilient, often becoming sweeter after a frost. Kale, another cold-hardy green, can withstand freezing temperatures and even improves in flavor as the weather gets colder. Swiss chard, with its vibrant stems and mild, earthy leaves, adds both color and nutrition to any garden. Cruciferous Vegetables: This family includes broccoli, cauliflower, cabbage, and Brussels sprouts. Broccoli and cauliflower thrive in cool weather and are known for their compact heads that form well in lower temperatures. Cabbage is a robust choice, offering a range of varieties from crunchy and crisp to tender and buttery. Brussels sprouts, with their miniature cabbage-like buds, are particularly well-suited to cool climates and often develop a sweeter taste after exposure to frost. Root Vegetables: Carrots, beets, and radishes are ideal for cool weather planting. Carrots can be harvested throughout the fall and winter, developing a sweeter flavor as they mature in cooler soil. Beets, with their vibrant roots and edible greens, can be sown in late summer for a fall harvest. Radishes, quick to mature and available in various colors and flavors, are perfect for adding a zesty crunch to your dishes. Alliums Onions, garlic, and leeks benefit from the cool season as well. Garlic is typically planted in the fall and harvested the following summer, developing a rich, robust flavor as it matures slowly in the cool soil. Onions, especially those grown from sets or seeds in late summer, can be harvested in late spring or early summer. Leeks, with their mild, sweet flavor, can be planted in early spring and harvested well into winter. Peas: Both snap and snow peas flourish in cooler temperatures and are often among the first crops to be planted in spring. Their climbing vines can make use of trellises or supports, and the pods are harvested before the peas inside fully mature, offering a sweet, crisp bite that's perfect for early season dishes.

#### **Chapter IV**

#### Warm season vegetables

#### Ms. C.K. Akshaya

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Warm season vegetables are those that require the full embrace of summer's heat to reach their peak potential. Whether you're a seasoned gardener or a novice planting your first seeds, understanding the unique needs and benefits of warm season vegetables can significantly enhance your gardening experience and harvest. Unlike their cool-season counterparts, which prefer the mild temperatures of spring and fall, these plants need the consistent warmth of late spring and summer to grow and produce effectively. The ideal growing conditions for warm season vegetables typically involve well-drained soil, ample sunlight, and moderate to high temperatures. These conditions not only support their growth but also influence the taste, texture, and yield of the produce.Popular examples of warm season vegetables include tomatoes, peppers, cucumbers, beans, and corn. Each of these plants brings its own unique set of flavors and textures to the table. Tomatoes offer a juicy burst of flavor in countless varieties, from tangy cherry tomatoes to rich beefsteaks. Peppers can range from sweet to fiery hot, adding a diverse range of flavors to dishes. Cucumbers provide a crisp, refreshing crunch, while beans and corn contribute both substance and sweetness to meals. Growing these vegetables can be incredibly rewarding, but it does come with its own set of challenges. Warm season vegetables are often more sensitive to frost, making timing and protection crucial. They also tend to have varying needs when it comes to watering, soil nutrition, and pest management. Understanding these requirements and adapting your gardening practices accordingly will help ensure a successful and bountiful harvest. In this chapter, we will explore the essentials of cultivating warm season vegetables, including tips for planting, caring for, and harvesting these sun-loving crops. We'll delve into the specific needs of different varieties, discuss common challenges and how to overcome them, and offer advice on how to make the most of your garden's potential. Whether you're looking to grow your own fresh ingredients for summer meals or simply enjoy the beauty and bounty of a thriving garden, warm season vegetables provide a gratifying and delicious way to embrace the summer season.

#### **Chapter V**

#### Pest and Diseases of vegetable crops and their Management

#### MR.C.SABARINADHAN

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Vegetable crops are essential for human nutrition and economic stability, yet they are highly susceptible to a variety of pests and diseases that can significantly impact yield and quality. Effective management of these issues is crucial for sustainable production and food security.Pests such as aphids, beetles, and caterpillars can inflict severe damage on vegetable crops. Aphids, for instance, feed on plant sap, weakening the plant and potentially transmitting harmful viruses. Beetles, such as the Colorado potato beetle, target specific crops like potatoes and tomatoes, causing extensive defoliation. Caterpillars, including the cabbage worm, can devastate crops like cabbage and broccoli by consuming large amounts of foliage. Diseases affecting vegetable crops are equally diverse. Fungal infections such as powdery mildew and downy mildew can cause significant crop loss. Powdery mildew appears as white, powdery spots on leaves, while downy mildew manifests as yellow patches on leaves that eventually turn brown and die. Bacterial diseases, including bacterial wilt and blight, can also be devastating. Bacterial wilt causes plants to wilt rapidly, while bacterial blight leads to leaf spots and premature plant death. Viral diseases, such as tomato spotted wilt virus, can lead to stunted growth and reduced fruit production. Integrated Pest Management (IPM) is a comprehensive approach to managing pests and diseases that combines cultural, biological, and chemical methods. Cultural practices include crop rotation, which helps break the life cycle of pests and diseases, and the use of resistant crop varieties. For instance, rotating crops reduces the buildup of pest populations in the soil and minimizes disease incidence. Planting disease-resistant varieties can also reduce the impact of specific pests and diseases. Biological control involves using natural predators or parasites to manage pest populations. For example, introducing ladybugs can help control aphid populations, while parasitic wasps can target caterpillars. Additionally, maintaining biodiversity in the garden can support a balanced ecosystem where natural enemies of pests thrive. Chemical control, while effective, should be used judiciously to minimize environmental impact. Pesticides should be applied as a last resort and in accordance with recommended guidelines to reduce the risk of resistance development and harm to non-target organisms. Integrated with other strategies, chemicals can be used strategically to manage outbreaks without disrupting the ecosystem. Disease management also involves monitoring and early detection. Regular inspections of crops can help identify symptoms early, allowing for prompt action.

#### **Chapter VI**

#### **The Book of Cucurbits**

#### Dr. P. Selvaraj

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Cucurbits, a family of plants encompassing everything from cucumbers to pumpkins, present a remarkable array of shapes, sizes, and flavors. This chapter delves into the fascinating diversity within this botanical family, highlighting the unique characteristics and uses of various cucurbit species. Cucumbers (Cucumis sativus): Originating from South Asia, cucumbers have become a staple in global cuisine. With their crisp texture and refreshing taste, they are enjoyed fresh, pickled, or in salads. Cucumbers exhibit two primary types: slicing cucumbers, which are larger and have a thicker skin, and pickling cucumbers, which are smaller and often more bumpy. Despite their simplicity, cucumbers come in a range of colors and patterns, from the traditional green to yellow or even white varieties. Pumpkins (Cucurbita pepo, Cucurbita maxima, Cucurbita moschata): Pumpkins are perhaps the most iconic cucurbit, celebrated for their vibrant orange hue and versatile uses. They belong to several species, each offering unique attributes. Cucurbita pepo, the most common, includes the classic Halloween pumpkin and various ornamental varieties. Cucurbita maxima features the giant pumpkins, famed for their size and often used in contests. Cucurbita moschata, which includes butternut squash, has a sweeter flavor and a richer texture, making it ideal for soups and pies Squashes (Cucurbita spp.): Squashes are diverse, encompassing summer varieties like zucchini (Cucurbita pepo) and winter varieties such as acorn and spaghetti squash (Cucurbita pepo and Cucurbita maxima). Summer squashes are characterized by their tender skins and quick growth, while winter squashes have tougher skins and a longer storage life. The adaptability of squashes to various climates and culinary uses makes them an essential part of many diets worldwide.Melons (Cucumis melo and Citrullus lanatus): Melons offer a sweet and juicy contrast to other cucurbits. The melon family includes cantaloupes and honeydews (Cucumis melo), known for their aromatic flesh and diverse varieties, and watermelons (Citrullus lanatus), celebrated for their refreshing, hydrating properties. Watermelons, with their large, juicy fruit and distinctive red flesh, have become synonymous with summer enjoyment. Gourds (Lagenaria siceraria and Cucurbita spp.): Gourds, often overlooked in culinary contexts, are fascinating for their use in crafts and decoration. The bottle gourd (Lagenaria siceraria) and ornamental varieties of Cucurbita species produce hard-shelled fruits that are traditionally used for making utensils, musical instruments, and decorative items. Their ability to be dried and shaped has made them valuable across cultures throughout history.

#### **Chapter VII**

#### Underutilized and green leafy vegetables

#### Dr. D.R. Sudha

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Purslane (Portulaca oleracea): Purslane is a succulent green that thrives in hot, dry climates and has been cherished in various culinary traditions worldwide. Its slightly tangy flavor and crisp texture make it a delightful addition to salads and soups. Nutritionally, purslane is a standout, rich in omega-3 fatty acids, which are rare in leafy greens. It also contains significant amounts of vitamin E, an antioxidant that supports skin health and immune function. Mache (Valerianella locusta): Also known as lamb's lettuce, mache is a delicate, tender green with a mild, nutty flavor. It's commonly used in French cuisine and is prized for its creamy texture. Mache is high in vitamins A and C, as well as folate, which is crucial for cell growth and metabolism. Its subtle flavor and nutritional profile make it an excellent choice for gourmet salads and light dishes. Amaranth Leaves (Amaranthus spp.): Often overshadowed by its grain counterpart, amaranth leaves are a nutritional treasure. These vibrant greens, which can be red, green, or purple, are packed with protein, iron, and calcium. Amaranth leaves have a slightly peppery taste and can be used similarly to spinach in cooking. Their high calcium content makes them particularly valuable for those looking to boost bone health.Watercress (Nasturtium officinale): Watercress is a peppery green that grows in aquatic environments, offering a refreshing bite to salads and sandwiches. This leafy green is a rich source of vitamins A, C, and K, as well as calcium and iron. Watercress is also known for its high levels of glucosinolates, compounds that have been linked to reduced cancer risk.Collard Greens (Brassica oleracea): While collard greens are more common in Southern cuisine, they are still underutilized in many other regions. These hearty greens have a robust, slightly bitter flavor that stands up well to long cooking times. Collard greens are incredibly nutrient-dense, providing significant amounts of vitamins A, C, and K, as well as fiber and calcium. They are often used in traditional dishes like collard greens with ham hocks but can be a versatile ingredient in various recipes. Beet Greens (Beta vulgaris): Often discarded, beet greens are a flavorful and nutritious byproduct of beetroot cultivation. These greens are tender and slightly sweet, with a flavor reminiscent of spinach but with a hint of earthiness. Beet greens are high in vitamins A and C, as well as potassium and magnesium. They are versatile in cooking, suitable for sautés, soups, and even as a nutritious addition to smoothies. Incorporating these underutilized green leafy vegetables into your diet can not only diversify your meals but also enhance your overall nutritional intake.

#### **Chapter VIII**

#### **Seed Selection and Propagation Methods**

#### Mr.K.C.Sivabalan

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Seed selection and propagation are fundamental practices in horticulture and agriculture that underpin successful crop production and plant conservation. These processes are pivotal not only for ensuring high yields and quality but also for maintaining genetic diversity and resilience within plant species. This chapter delves into the intricacies of seed selection and propagation methods, exploring how these techniques influence plant health, growth, and sustainability. Seed selection is the initial and arguably the most critical step in cultivation. It involves choosing seeds that exhibit desirable traits such as high germination rates, disease resistance, and adaptability to environmental conditions. Effective seed selection requires a deep understanding of the plant's genetic makeup, environmental requirements, and the specific goals of the grower, whether for commercial production, home gardening, or conservation purposes. The criteria for selecting seeds can vary widely depending on the plant species and intended use, making it a complex yet essential practice. Once seeds are selected, propagation methods come into play. Propagation is the process of growing new plants from various sources, including seeds, cuttings, or tissue cultures. This chapter covers both sexual and asexual propagation techniques. Sexual propagation, which involves seed germination, is the most common method and offers the advantage of genetic diversity. However, it can be unpredictable due to variations in genetic expression and environmental factors. On the other hand, asexual propagation, such as cloning through cuttings, grafting, or tissue culture, ensures the production of genetically identical plants, which can be advantageous for maintaining specific traits and accelerating growth.Each propagation method has its own set of practices, benefits, and limitations. For instance, seed germination requires careful control of factors such as moisture, temperature, and light, whereas asexual methods may necessitate specialized equipment and techniques to ensure successful plant development. Understanding these methods allows growers to choose the most suitable approach for their objectives, whether it's for increasing plant numbers rapidly, preserving rare species, or improving crop varieties. As we explore the various seed selection and propagation methods, this chapter will provide a comprehensive overview of best practices, emerging technologies, and future trends in the field. By mastering these techniques, growers and horticulturists can enhance plant productivity, resilience, and biodiversity, ultimately contributing to more sustainable and effective agricultural practices.

#### **Chapter IX**

#### **Greenhouse and Controlled Environment Agriculture**

#### **Dr.K.Ashokkumar**

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

In recent decades, the challenges of feeding a growing global population amidst climate uncertainty have spurred significant advancements in agricultural technology. Among the most transformative innovations in this field are greenhouse and controlled environment agriculture (CEA) systems. These methods represent a paradigm shift from traditional farming, offering enhanced control over crop production and resource management. Greenhouse agriculture, characterized by the use of transparent structures to shield crops from external weather conditions, allows for year-round production regardless of climate. By trapping sunlight and regulating temperature, humidity, and CO<sub>2</sub> levels, greenhouses create an optimal environment for plant growth. This method not only increases yields but also minimizes the use of pesticides and fertilizers, contributing to more sustainable farming practices. Controlled environment agriculture, an extension of greenhouse technology, takes the concept of environmental control a step further. CEA systems, including vertical farms, hydroponics, and aeroponics, provide a highly controlled environment where every aspect of plant growth—from light intensity and spectrum to nutrient delivery and air circulation—is meticulously managed. These systems are often housed in indoor facilities or high-tech greenhouses, where they leverage advanced technologies like artificial lighting, climate control systems, and automated nutrient delivery to optimize plant growth. The appeal of greenhouse and CEA systems lies in their ability to decouple food production from traditional agricultural constraints. By utilizing space efficiently and minimizing reliance on external environmental factors, these methods can be implemented in urban areas, on rooftops, and in regions with adverse growing conditions. Furthermore, they offer potential solutions to issues such as soil degradation, water scarcity, and long supply chains, making them increasingly relevant in the context of global sustainability and food security. As we delve into the intricacies of greenhouse and controlled environment agriculture in this chapter, we will explore their technological foundations, operational principles, and practical applications. We will also examine the economic, environmental, and social impacts of these systems, assessing their role in shaping the future of agriculture.

#### **Chapter X**

#### **Crop Rotation and Intercropping Strategies**

#### Dr.Rajaguruvu

#### Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Effective farming is not just about what crops are grown but also about how they are managed. Crop rotation and intercropping are two key strategies that enhance soil health, optimize resource use, and increase biodiversity on farms.

**Crop Rotation:** Crop rotation involves changing the type of crop grown in a particular field from season to season. This practice is more than just a traditional method; it is a scientifically-backed approach to sustainable agriculture. By rotating crops, farmers can break pest and disease cycles, reduce soil depletion, and improve soil structure. For instance, legumes such as beans and peas can fix nitrogen in the soil, reducing the need for synthetic fertilizers. Following these with nutrient-demanding crops like corn or wheat helps balance soil nutrients and improves overall productivity. Rotating crops also helps in managing weeds and reducing the reliance on chemical herbicides. Different crops compete with different weed species, thus minimizing weed pressure. Moreover, rotating crops can prevent the buildup of pathogens that thrive on specific plants, thereby reducing the incidence of diseases and pests. Intercropping involves growing two or more crops simultaneously in the same field. This strategy can maximize the use of available resources, such as light, water, and nutrients. There are several intercropping systems, including mixed cropping, row intercropping, and relay cropping. Mixed cropping involves growing crops in a random pattern or close proximity, which can help in optimizing land use and reducing the risk of total crop failure. Row intercropping, where different crops are grown in alternating rows, can make better use of space and resources. Relay cropping is the practice of sowing a second crop before the first crop is harvested, thereby extending the growing season and increasing yield per unit area. Intercropping can also enhance biodiversity, improve soil health, and reduce pest infestations. For example, planting companion plants such as marigolds with tomatoes can repel nematodes and other pests. Additionally, the canopy provided by taller plants in intercropping systems can protect shorter plants from excessive sunlight, thus creating a microclimate that benefits all the crops involved. Both crop rotation and intercropping are integral to sustainable agricultural practices. They not only help in improving soil fertility and crop yields but also contribute to ecological balance by promoting biodiversity and reducing environmental impact. As we face the challenges of climate change and resource scarcity, these strategies offer practical solutions for maintaining productive and resilient farming systems.
# SERICULTURE & LAC CULTURE TECHNIQUES

Edited by DR.B.VIBITHA BALA



# Sericulture & Lac culture techniques

Copyright © 2021 by Dr.B.Vibitha bala

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-831-2

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# TABLE OF CONTENT

Introduction
Chapter I: Economic classifications of insects11
Dr.B.VIBITHA BALA
Chapter II: Sericulture techniques
MS.M.JASMINE
Chapter III: Sericulture By-products and Utilization
Dr.K.C.SIVABALAN
Chapter IV: Pests and diseases of mulberry silkworm74
Ms.K. SHALINI
Chapter V: Silkworm Breeding and Hybridization106
Mrs.S.R.RAJAM
Chapter VI: Mulberry Cultivation and Management130
Dr.K.KUMARAKURU
Chapter VII: Cultivation of lac insects
Ms.C.K.AKSHAYA
Chapter VIII: Natural enemies of lac insects
Mr.S.TAMIL COVANE
Chapter IX: Value-added products from lac
Ms.N.DHIVYAPRIYA
Chapter X: Lac processing and commercial applications
Dr.A.Udayakumar
REFERENCE:

### **CHAPTER I**

## ECONOMIC CLASSIFICATIONS OF INSECTS

### Dr.B.Vibitha bala

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Insects, as one of the most diverse and abundant groups of organisms on Earth, play a crucial role in various ecosystems and have significant economic impacts on human society. Their interactions with agriculture, industry, and human health have led to their classification based on economic importance. Understanding the economic classifications of insects helps in the development of strategies for their management, conservation, and utilization, which are essential for sustainable agricultural practices, public health, and ecological balance.

Insects are broadly classified into **beneficial insects** and **pest insects** based on their economic impact. Beneficial insects contribute positively to agriculture, ecosystems, and even industry, while pest insects cause damage to crops, stored products, and human health, leading to economic losses.

**Beneficial insects** are vital to agriculture and the environment. Pollinators, such as bees, butterflies, and certain beetles, are among the most economically significant beneficial insects. They play a critical role in the pollination of flowering plants, including many crops that are essential for human consumption. The economic value of insect pollination is immense, contributing billions of dollars annually to global agriculture. Other beneficial insects include natural predators and parasitoids, such as ladybugs, lacewings, and certain wasps, which help control populations of pest insects. By preying on pests or parasitizing them, these insects reduce the need for chemical pesticides, thereby promoting more sustainable agricultural practices.

In addition to pollinators and natural enemies of pests, some insects are classified as beneficial due to their roles in **decomposition and nutrient cycling**. For example, dung beetles and certain fly species break down organic matter, recycling nutrients back into the soil, which enhances soil fertility and promotes healthy plant growth. Insects such as silkworms and bees are also economically important due to their roles in **industry**. Silkworms produce silk, a valuable textile material, while bees produce honey, beeswax, and other products that have numerous applications in food, cosmetics, and medicine.

# CHAPTER II

# SERICULTURE TECHNIQUES

### Ms.M.Jasmine

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Sericulture, the cultivation of silkworms for the production of silk, is an ancient and intricate practice that has been refined over thousands of years. Originating in China over 4,000 years ago, sericulture has spread worldwide, becoming a vital industry in countries such as India, Japan, and Thailand. The techniques involved in sericulture are diverse and require careful management to ensure high-quality silk production. This chapter provides an introduction to the essential sericulture techniques, highlighting the steps involved in cultivating silkworms and processing their cocoons into silk.

**Silkworm Rearing** is the foundation of sericulture. The process begins with the selection of high-quality silkworm eggs, known as silkworm seeds. These eggs are incubated under controlled conditions to ensure uniform hatching. The young silkworms, or larvae, are then fed a diet of fresh mulberry leaves, which are crucial for their growth. The quality and quantity of the mulberry leaves directly impact the quality of the silk produced, making the cultivation of mulberry trees a critical aspect of sericulture. The larvae go through several growth stages, or instars, during which they molt multiple times before reaching the final stage where they begin spinning cocoons.

**Cocoon Formation** is a critical phase in sericulture. Once the silkworms have matured, they start spinning cocoons by secreting fibroin, a protein that forms the silk thread, along with sericin, a gummy substance that holds the thread together. The silkworm spins the cocoon around itself, which takes about two to three days to complete. The quality of the cocoon is crucial for the quality of the silk, with factors such as thread length, fineness, and luster being closely monitored. Proper environmental conditions, such as temperature and humidity, must be maintained during this phase to ensure the best results.

**Cocoon Harvesting** involves carefully collecting the mature cocoons once the silkworms have completed their spinning. Timing is essential; harvesting too early can result in underdeveloped cocoons, while harvesting too late can lead to the emergence of the moth, which would break the silk thread and reduce its length and quality. After harvesting, the

# **CHAPTER III**

# SERICULTURE BY-PRODUCTS AND UTILIZATION

# Dr.K.C.Sivabalan

# Associate Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Sericulture, traditionally known for producing high-quality silk, also generates several valuable by-products that contribute to its economic and environmental sustainability. These by-products, which include mulberry leaves, silkworm pupae, sericin, and silk waste, have diverse applications across various industries, including agriculture, food, pharmaceuticals, and cosmetics. The effective utilization of these by-products not only enhances the profitability of sericulture but also promotes a more sustainable approach by minimizing waste and maximizing resource efficiency.

**Mulberry Leaves**, the primary food source for silkworms, are not only crucial to silk production but also have significant nutritional and medicinal value. After the silkworms have fed, surplus mulberry leaves can be repurposed in several ways. They are rich in nutrients and can be used as livestock feed, offering a cost-effective and nutritious alternative for animals. Additionally, mulberry leaves contain bioactive compounds that have been utilized in traditional medicine, particularly in the treatment of diabetes, hypertension, and obesity. Mulberry leaf extracts are increasingly being incorporated into dietary supplements and functional foods due to their antioxidant properties.

**Silkworm Pupae**, often discarded after the silk has been extracted from the cocoons, are a rich source of protein, fat, and essential amino acids. In many cultures, silkworm pupae are considered a delicacy and are consumed as a nutritious food item. They can be processed into various forms, such as dried, roasted, or powdered, and used in animal feed or as a protein supplement in human diets. The high protein content of silkworm pupae also makes them suitable for use in the production of biofertilizers, contributing to sustainable agricultural practices. Moreover, silkworm pupae oil, extracted from the pupae, is used in cosmetics and skincare products due to its emollient and anti-inflammatory properties.

**Sericin**, a protein that binds the silk fibers together, is typically removed during the silk reeling process. However, rather than being discarded, sericin has found numerous applications in the pharmaceutical and cosmetic industries. Sericin is known for its

### **CHAPTER IV**

# PESTS AND DISEASES OF MULBERRY SILKWORM

# Ms.K.Shalini

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

The mulberry silkworm (Bombyx mori) is the cornerstone of sericulture, and its health directly influences the quality and quantity of silk produced. However, like all agricultural species, silkworms are vulnerable to a variety of pests and diseases that can significantly impact silk production. These challenges require careful management to ensure the sustainability and profitability of sericulture. Understanding the common pests and diseases that affect mulberry silkworms, as well as the methods for their prevention and control, is crucial for maintaining healthy silkworm populations and maximizing silk yield.

**Pests of Mulberry Silkworms** primarily include predators and parasitoids that attack silkworms at various stages of their lifecycle. One of the most common predators is the *uzi fly* (Exorista sorbillans), which is notorious for laying its eggs on the silkworms. The larvae of the uzi fly burrow into the silkworm's body, feeding on its internal tissues and ultimately killing it. Another significant pest is the *ant* species, which often invades rearing trays to feed on silkworm eggs, larvae, or even pupae. These predators can cause substantial losses if not properly managed, necessitating the implementation of preventive measures such as the use of protective nets, regular monitoring, and the maintenance of hygiene in rearing environments.

**Diseases of Mulberry Silkworms** are caused by various pathogens, including viruses, bacteria, fungi, and protozoa, each presenting unique challenges to silkworm rearing. One of the most devastating diseases is **Grasserie**, caused by the Bombyx mori nucleopolyhedrovirus (BmNPV). This viral disease is characterized by the swelling of the silkworm's body, which becomes filled with a milky fluid. Infected silkworms exhibit sluggishness and eventually die, leading to significant economic losses.

Another critical disease is **Flacherie**, which can be caused by bacteria like *Serratia marcescens* or other pathogens. Flacherie leads to the degeneration of the gut and results in a watery, dark-colored body. This disease can rapidly spread through a silkworm population if proper sanitation practices are not followed.

# **CHAPTER V**

# SILKWORM BREEDING AND HYBRIDIZATION

# Mrs.S.R.Rajam

# Assistant Professor, School of Agriculture,

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# Silkworm Breeding and Hybridization

Silkworm breeding and hybridization are critical techniques in sericulture aimed at improving the quality and quantity of silk production. Over centuries, selective breeding has led to the domestication of the silkworm (*Bombyx mori*), and modern breeding programs continue to refine silkworm strains for desirable traits such as higher silk yield, disease resistance, and adaptability to diverse environmental conditions. Hybridization, in particular, has become an essential tool in enhancing silkworm performance, combining the strengths of different strains to create superior hybrids.

# 1. Objectives of Silkworm Breeding

The primary objectives of silkworm breeding are to improve the commercial characteristics of the insect. These objectives include:

- **Increased Silk Yield:** Breeders focus on selecting strains that produce longer and stronger silk filaments, which enhance the quality of the silk and increase profitability.
- **Disease Resistance:** Silkworms are vulnerable to diseases such as pebrine and flacherie. Breeding for disease-resistant strains helps to minimize the losses caused by infections and ensures more stable silk production.
- **Improved Cocoon Quality:** Factors such as the size, weight, and uniformity of the cocoons are essential for producing high-quality silk. Breeding efforts aim to optimize these characteristics to meet industry standards.
- Adaptability to Environmental Conditions: Silkworms bred for resilience to fluctuations in temperature, humidity, and nutrition can perform better in diverse climates, allowing sericulture to expand to new regions.

# 2. Silkworm Hybridization

Hybridization is the process of crossing two genetically distinct strains of silkworms to produce hybrid offspring that exhibit enhanced vigor, also known as hybrid vigor or heterosis. The offspring of these crosses tend to outperform their parents in terms of silk production, growth rate, and survival.

### **CHAPTER VI**

# MULBERRY CULTIVATION AND MANAGEMENT

# Dr.K.Kumarakuru

# Associate Professor, School of Agriculture,

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# **Mulberry Cultivation and Management**

Mulberry (*Morus* spp.) is the primary food source for silkworms (*Bombyx mori*), making its cultivation a crucial component of sericulture. Successful silkworm rearing depends heavily on the quality and availability of mulberry leaves, which directly influence silk production. Proper cultivation and management of mulberry plants not only ensure a steady supply of nutritious leaves but also increase overall yield and quality of the silk produced.

# **1. Mulberry Varieties**

Mulberry plants belong to the Moraceae family, with several species and varieties being cultivated for sericulture. The most commonly grown species include *Morus alba*, *Morus indica*, *Morus multicaulis*, and *Morus nigra*. Different varieties are suited to varying climatic conditions, allowing for mulberry cultivation in diverse regions. Selecting the right variety is key to optimizing leaf production and maintaining healthy silkworms.

# 2. Climatic and Soil Requirements

Mulberry is a hardy plant that can grow in a wide range of climatic conditions, but optimal growth occurs in regions with a warm temperate to tropical climate. The plant thrives in temperatures between 24°C and 30°C and requires moderate rainfall (600–2500 mm annually). Mulberry plants prefer well-drained, fertile soils with good moisture retention and a slightly acidic to neutral pH (6.5–7.5).

# **3.** Propagation and Planting

Mulberry is propagated through both sexual (seed propagation) and asexual methods (vegetative propagation). However, vegetative propagation, particularly through stem cuttings, is more common in sericulture due to its faster and more uniform growth. Stem cuttings, typically 20–25 cm long with 3–4 buds, are taken from healthy mulberry plants and planted directly in the field or raised in nurseries before transplantation.

Planting can be done in rows or hedgerows, depending on the cultivation system. A row spacing of 60–90 cm between plants and 120–150 cm between rows allows for better aeration, light penetration, and easier management. Mulberry plants are generally spaced to

### **CHAPTER VII**

### **CULTIVATION OF LAC INSECTS**

# Ms.C.K.Akshaya

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

The cultivation of lac insects, primarily *Kerria lacca*, is an age-old practice that produces lac, a natural resin with significant economic value. Lac is widely used in the production of varnishes, adhesives, dyes, and cosmetics, making the cultivation of these insects an important activity, particularly in regions like India, Thailand, and Myanmar. The process of cultivating lac insects, known as lac culture, involves rearing these insects on host trees, managing their lifecycle to maximize resin production, and harvesting the lac. This chapter introduces the techniques and practices involved in lac insect cultivation, highlighting its ecological and economic significance.

Lac insects are small, scale-like creatures that secrete lac as they feed on the sap of specific host trees. The choice of host trees is crucial for successful lac cultivation, as the quality and quantity of lac produced depend significantly on the species of tree used. Common host trees include *Butea monosperma* (palas or flame of the forest), *Ziziphus mauritiana* (ber), and *Schleichera oleosa* (kusum). Each host tree offers different advantages, such as varying resin yields, growth rates, and resistance to pests and diseases.

The **cultivation process** begins with the inoculation of host trees with broodlac, which contains mature female lac insects. The broodlac is tied to the branches of the host trees, where the lac insects settle and begin feeding on the sap. As they feed, the insects secrete lac, forming protective encrustations around their bodies. This resinous coating eventually hardens, creating the raw lac that is harvested for processing.

The lifecycle of the lac insect includes several stages—egg, nymph, and adult—each of which requires careful management to ensure high yields. The insects reproduce rapidly, with females laying eggs that hatch into nymphs, which then continue the cycle of feeding and lac secretion. The timing of inoculation is critical, as it must align with the growth cycles of both the insects and the host trees to maximize productivity. Lac insects are typically reared in two main crops per year: the **Jethwi** crop, harvested in June-July, and the **Katki** crop, harvested in October-November.

### CHAPTER VIII

# NATURAL ENEMIES OF LAC INSECTS

# Mr.S.Tamil covane

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Lac insects (*Kerria lacca*), known for producing the valuable natural resin called lac, are subject to predation and parasitism by a variety of natural enemies. These natural enemies, including predators, parasitoids, and pathogens, pose significant threats to lac cultivation, often leading to reduced yields and lower-quality resin. Understanding these natural enemies and the impact they have on lac insects is crucial for developing effective management strategies to protect lac crops and ensure sustainable production.

**Predators** are the most direct threat to lac insects, feeding on them at various stages of their lifecycle. One of the most destructive predators is the *Eublemma amabilis* (lac moth), whose larvae, known as "lac borers," burrow into the lac encrustations to feed on the lac insects. These larvae can cause substantial damage, leading to significant losses in lac production. Another notable predator is the *Chilocorus nigritus* (black ladybird beetle), which preys on lac insects by consuming both the insects and the lac they produce. These beetles can rapidly decimate lac populations if left unchecked.

In addition to insects, certain species of **ants**, such as *Solenopsis geminata* (fire ants), are known to attack lac insects. These ants are attracted to the lac secretion and the honeydew excreted by the lac insects. They not only consume the lac insects but can also disturb the cultivation environment by disrupting the delicate balance needed for successful lac production.

**Parasitoids** represent another significant category of natural enemies. Parasitoid wasps, such as *Tachardiaephagus tachardiae*, lay their eggs inside or on the bodies of lac insects. When the wasp larvae hatch, they feed on the lac insects from within, eventually killing them. This parasitic relationship can lead to widespread mortality among lac insect populations, severely impacting resin yields. The extent of damage caused by parasitoids often depends on environmental conditions and the density of lac insect populations, with higher densities typically attracting more parasitoid activity.

# CHAPTER IX

# VALUE-ADDED PRODUCTS FROM LAC

# Ms.N.Dhivyapriya

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Lac, a natural resin secreted by the lac insect (*Kerria lacca*), has been a valuable commodity for centuries, traditionally used in the production of shellac, varnishes, and dyes. However, recent advancements and innovations have expanded the scope of lac utilization, leading to the development of a wide array of value-added products. These products not only enhance the economic viability of lac cultivation but also diversify its applications across various industries, including food, cosmetics, pharmaceuticals, and environmental sustainability. This chapter explores the different value-added products derived from lac, highlighting their production processes, uses, and market potential.

**Shellac** is one of the most well-known and versatile products derived from lac. It is produced by refining raw lac through a process that involves washing, melting, filtering, and drying to remove impurities. Shellac is widely used as a natural coating material in the food industry, where it serves as a protective glaze for fruits, candies, and pharmaceuticals. Its non-toxic and biodegradable properties make it an attractive alternative to synthetic coatings. In addition, shellac is used in the manufacture of varnishes, polishes, and sealants for wood and metal surfaces, offering a durable and glossy finish that enhances the aesthetic appeal and longevity of these products.

Lac Dye is another important value-added product obtained from lac resin. Historically, lac dye was used as a natural textile dye in regions such as India and Southeast Asia. It produces rich red to purple hues and was once highly sought after for dyeing silk, wool, and cotton. While synthetic dyes have largely replaced natural dyes in the textile industry, there is a resurgence of interest in lac dye due to the growing demand for ecofriendly and sustainable products. Lac dye is now being explored for use in organic textiles, cosmetics, and even food coloring, where its natural origin and vibrant color appeal to environmentally conscious consumers.

Lac Wax, a by-product of the shellac production process, has found applications in various industries due to its unique properties. Lac wax is a natural, hard wax with excellent

# **CHAPTER X**

# LAC PROCESSING AND COMMERCIAL APPLICATIONS

# Dr.A.Udayakumar

# Associate Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# Lac Processing and Commercial Applications

Lac is a natural resin secreted by the lac insect (*Kerria lacca*), primarily found in India and Southeast Asia. The resin, collected from host trees, undergoes several processing stages before being transformed into various commercially valuable products. Lac has been used for centuries in industries ranging from varnish and dye production to cosmetics and pharmaceuticals. The sustainable harvesting and processing of lac provide income for rural communities and support multiple industries.

# **1. Lac Processing**

Lac processing involves several steps to extract usable lac from raw material, known as "sticklac," which contains impurities such as twig fragments, insect parts, and other debris.

- **Harvesting**: Lac is harvested by scraping the resin off the branches of host trees where lac insects form encrustations. These trees include *Butea monosperma* (Flame of the Forest), *Ziziphus mauritiana* (Ber), and *Schleichera oleosa* (Kusum). Harvesting is usually done twice a year, during the summer and winter seasons.
- Seedlac Production: After harvesting, the sticklac is washed in large water tanks to remove impurities and separate the resin from insect bodies. This washing process yields "seedlac," a partially purified form of lac resin. The lac is then dried and sieved to further remove unwanted materials.
- Shellac Production: To make shellac, the seedlac is further refined by melting and filtering it. The melted lac is stretched into thin sheets and allowed to cool. Once dried, the sheets are broken into flakes, which form the basis of shellac. Shellac can be bleached, if necessary, to produce a lighter-colored product.
- **Buttonlac and Garnetlac**: Buttonlac is produced by heating seedlac and forming it into button-shaped discs. Garnetlac, another form of processed lac, is a dark-colored, high-quality product used in specialized industries.

# 2. Commercial Applications

Lac has a broad range of commercial applications due to its natural, biodegradable, and non-toxic properties. It is prized for its versatility and functionality in multiple industries.

# BOOK OF WEED MANAGEMENT

Edited by





# Book of Weed Management

Copyright © 2021 by Dr. P. Selvaraj

All rights reserved. No part of this book may be reproduced or transmitted in any form or by

any means without written permission from the author.

ISBN: 978-93-6255-603-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

Introduction
Chapter I: Introduction of weeds
MRS. G. BANUMATHI
Chapter II: Weeds propagation
MRS. J.U. JANUSIA
Chapter III: Principles of Weed Control
DR. D.R. SUDHA
Chapter IV: Methods of weed management
Chapter V: Weed management in field crops
DR. P.Selvaraj
Chapter VI: Herbicides and mode of action
Chapter VII: Herbicides Tolerance and Resistance
MS. K. SHALINI
Chapter VIII: Weed Dissemination113
MR. P. RAJAVEL
Chapter IX: Integrated Weed Management (IWM)129
MR. S. BALAMURGAN
Chapter X: Herbicide Spraying Equipment136
DR. B. VIBITHA BALA
REFERENCE:

# TABLE OF CONTENT

### **CHAPTER I**

# **INTRODUCTION OF WEEDS**

# MRS. G. BANUMATHI

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Weeds are plants considered undesirable or invasive in a particular area because they can outcompete other plants for resources, disrupt ecosystems, or reduce crop yields. They thrive in a variety of environments and can be a significant challenge in agriculture, gardening, and natural habitats. Here's a more detailed introduction to weeds:

Weeds are highly competitive due to their ability to grow quickly and efficiently exploit available resources. Their rapid growth and prolific reproduction can overshadow and outcompete crops, leading to reduced crop yields and quality. Some weeds have deep root systems that can access nutrients and moisture beyond the reach of shallow-rooted crops. Weeds possess various reproductive strategies that contribute to their persistence and spread. They can produce large quantities of seeds that remain viable for extended periods, allowing them to proliferate even after removal attempts. Many weeds also have the ability to reproduce vegetatively through runners, tubers, or rhizomes, further complicating control efforts.

Weeds can affect soil health and ecosystem balance. Their dense growth can lead to soil erosion and disrupt soil structure. Additionally, certain weed species can host pests and diseases that may spread to crops, exacerbating agricultural problems. Some weeds also have allelopathic properties, meaning they release chemicals that inhibit the growth of neighboring plants. Effective weed management is a critical component of sustainable agriculture. Weeds require constant vigilance and control strategies to prevent them from establishing and spreading. Management practices include mechanical methods such as plowing and hand weeding, chemical control through herbicides, and cultural practices like crop rotation and the use of cover crops. Integrated Weed Management (IWM) combines multiple approaches to reduce reliance on any single method and manage weed populations more effectively.

### **CHAPTER II**

# WEEDS PROPAGATION

# MRS. J.U. JANUSIA

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Weed propagation is a critical aspect of understanding how weeds spread and persist in various environments. This process involves the mechanisms by which weeds reproduce and disseminate, contributing to their persistence and the challenges associated with managing them. Effective weed management strategies must account for these propagation methods to minimize the impact of weeds on crops, landscapes, and natural ecosystems.

Weeds propagate through several key mechanisms, each contributing to their ability to spread and establish in new areas: Most weeds reproduce primarily through seed production. They often produce large quantities of seeds, which can be dispersed over long distances by wind, water, animals, or human activity. The seeds are frequently designed to remain viable in the soil for extended periods, allowing them to germinate when conditions become favorable. This high seed output and longevity contribute significantly to their widespread presence. Vegetative Reproduction: In addition to seeds, many weeds can reproduce vegetatively. This form of reproduction involves the production of new plants from vegetative parts of the parent plant, such as roots, stems, or leaves. Structures like rhizomes, tubers, runners, and bulbs can generate new growth and spread the weed horizontally across the soil. Vegetative reproduction enables weeds to quickly colonize areas and regenerate even if above-ground parts are removed.

Regeneration from Fragments: Some weeds can regenerate from fragments of their vegetative parts. For instance, broken pieces of rhizomes or tubers can develop into new plants. This ability allows weeds to persist and spread even after partial removal or disturbance. Hybridization and Genetic Variation: Hybridization between different weed species can produce new, more vigorous forms. These hybrids can sometimes exhibit enhanced resistance to control methods or improved adaptability to environmental conditions, making them more challenging to manage.

# **CHAPTER III**

# PRINCIPLES OF WEED CONTROL DR. D.R. SUDHA

### Assistant Professor, School of Agriculture,

### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Weed control is a crucial aspect of modern agriculture and land management, aiming to minimize the adverse effects of unwanted plant species on crops, pastures, and natural ecosystems. Effective weed management integrates a range of strategies designed to suppress or eliminate weed populations while minimizing harm to desirable plants and the environment.

The principles of weed control revolve around understanding weed biology, ecology, and the various control methods available. The first principle is **prevention**, which focuses on stopping weeds from becoming established. This involves practices such as using certified weed-free seeds, cleaning equipment before moving between sites, and maintaining healthy, competitive crops that can outcompete weeds.

**Cultural control** is another foundational principle, emphasizing modifications in agricultural practices to reduce weed growth. Techniques such as crop rotation, adjusting planting densities, and implementing proper irrigation can enhance crop competition and disrupt weed life cycles. For instance, rotating crops disrupts weed populations that are adapted to specific crop systems, thereby reducing their prevalence.

**Mechanical control** includes physical methods like tilling, mowing, and hand weeding to manage weed populations. These methods can be effective, especially when used in conjunction with other strategies. Tillage, for example, can bury weed seeds and disrupt their growth, while mowing can prevent seed production and reduce weed biomass.

**Chemical control** employs herbicides to target and eliminate weeds. Herbicides can be selective, affecting only certain types of weeds while leaving crops unharmed, or non-selective, killing any plant material they contact. Proper application of herbicides, considering timing, dosage, and method, is critical to achieving effective control while minimizing environmental impact.

# **CHAPTER IV**

# METHODS OF WEED MANAGEMENT MRS. S.R. RAJAM

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Weed management is essential for maintaining the health and productivity of crops and natural ecosystems. Effective management strategies aim to reduce the competition from undesirable plants while promoting the growth of desirable species. The methods employed in weed management are diverse, each with its advantages and considerations, and often work best when integrated into a comprehensive strategy.

Cover Crops: Planting cover crops, such as clover or rye, can suppress weed growth by outcompeting weeds for resources and creating ground cover that prevents weed seed germination. Mowing: Regular mowing prevents weeds from flowering and setting seed. This method works well for weeds in lawns, pastures, and other areas where plants can be cut down repeatedly.

Combining Methods: IWM involves using a combination of cultural, mechanical, chemical, and biological methods to manage weeds more effectively. Key principles include: Diverse Strategies: Combining different control methods to address various weed species and growth stages.

Adaptive Management: Regularly monitoring weed populations and adjusting management strategies based on effectiveness and changes in weed pressure. Sustainable Practices: Choosing methods that minimize environmental impact and avoid the development of herbicide-resistant weed populations.

By integrating these methods and tailoring them to specific situations, you can develop a comprehensive weed management plan that effectively controls weed populations while supporting the health and productivity of crops and natural areas.

### **CHAPTER V**

# WEED MANAGEMENT IN FIELD CROPS

# **DR. P. SELVARAJ**

# Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Weed management in field crops is crucial for maintaining high yields and ensuring the health of the crops. Effective weed control helps reduce competition for resources such as water, nutrients, and light, and minimizes the risk of pests and diseases associated with weeds. Here's a comprehensive approach to managing weeds in field crops:

Tillage: Prior to planting, tilling the soil can help control weeds by uprooting them and burying weed seeds. However, excessive tillage can bring dormant seeds to the surface, so it should be done strategically. Seedbed Preparation: Creating a smooth, firm seedbed helps ensure even planting and reduces the likelihood of weeds establishing.

Application: Apply pre-emergence herbicides before weeds or crops emerge. These herbicides inhibit seed germination or early seedling growth, providing a clean start for the crops. Timing: Ensure timely application to be effective before weed seeds begin to germinate. Planting: Grow cover crops during the off-season to suppress weed growth. Cover crops can outcompete weeds and improve soil health, making it more difficult for weeds to establish. **Biological Methods** involve using natural enemies of weeds, such as herbivores, insects, or pathogens, to manage weed populations. This approach can be highly effective in controlling specific weed species but requires careful planning to ensure that introduced biological agents do not negatively impact non-target species or disrupt the ecosystem. his approach helps manage weed populations more effectively, mitigates the risk of herbicide resistance, and promotes sustainable agricultural practices. By implementing an integrated weed management plan, farmers can enhance crop productivity, protect environmental health, and achieve long-term agricultural success.

### CHAPTER VI

# HERBICIDES AND MODE OF ACTION DR. B. GUNA

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Herbicides are chemical substances used to control or eliminate unwanted plants (weeds). They work through various mechanisms, or modes of action, which affect specific physiological processes in plants. Understanding these modes of action helps in selecting the appropriate herbicide and managing the risk of weed resistance. Here's a detailed overview of herbicides and their modes of action:

Modes of Action - Photosynthesis Inhibitors: Mechanism: These herbicides inhibit photosynthesis by targeting the electron transport chain in chloroplasts, preventing plants from converting light energy into chemical energy. This disruption halts the production of essential nutrients, leading to plant death. Amino Acid Synthesis Inhibitors: Mechanism: These herbicides interfere with the synthesis of essential amino acids necessary for protein production. Glyphosate inhibits the shikimic acid pathway, while sulfonylureas and imidazolinones block the acetolactate synthase (ALS) enzyme involved in amino acid synthesis.

Cell Division Inhibitors: Mechanism: These herbicides disrupt the formation of microtubules during cell division, leading to defective cell walls and growth. This inhibition affects root and shoot development. Lipid Synthesis Inhibitors: Mechanism: These herbicides interfere with lipid synthesis necessary for cell membrane formation and plant growth. They are effective grasses. Disruption of Lipid Biosynthesis: primarily against Herbicides like aryloxyphenoxypropionates (APPs) inhibit the biosynthesis of fatty acids, which are crucial for membrane structure and function. Without the necessary lipids, plants cannot maintain cell membranes, leading to cell leakage and plant death. Interference with Nucleotide Synthesis: Some herbicides inhibit the synthesis of nucleic acids (DNA and RNA), which are vital for cell division and growth. For example, sulfonylureas inhibit the enzyme acetolactate synthase, which is crucial for the production of branched-chain amino acids and nucleotides, disrupting normal cellular processes

### **CHAPTER VII**

# HERBICIDES TOLERANCE AND RESISTANCE

# MS. K. SHALINI

### Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Herbicide tolerance refers to the ability of a plant species or variety to survive and grow despite the application of herbicides that would normally affect or kill most other plants. Herbicide resistance occurs when a weed population evolves to survive applications of herbicides that would normally kill or control them. This is often due to genetic changes in the weed population.

Target Site Resistance: Altered Enzyme: The herbicide's target site in the weed's biochemical pathway is altered, so the herbicide can no longer bind effectively. For example, in glyphosate-resistant weeds, the enzyme EPSPS is altered, preventing glyphosate from inhibiting it. Metabolic Resistance: Enhanced Detoxification: Weeds develop the ability to metabolize or detoxify the herbicide more effectively before it can reach its target site. This can involve increased activity of enzymes that break down the herbicide.

Reduced Herbicide Uptake: Decreased Absorption: Weeds may evolve mechanisms to reduce the uptake of the herbicide or its translocation within the plant, limiting its effectiveness. Cellular Barriers: Weeds may develop physical barriers that prevent the herbicide from entering plant cells or moving to its target site.

Despite these advantages, the use of herbicide-tolerant crops also poses challenges. One major concern is the potential for herbicide-resistant weed populations to develop. Overreliance on a single herbicide or herbicide-tolerant crop can lead to the selection of weed species that are resistant to the herbicide, reducing the efficacy of the control measures and requiring the development of new strategies. To mitigate resistance issues, integrated weed management practices are recommended. These include rotating crops, using herbicides with different modes of action, and combining chemical control with cultural and mechanical methods. By employing a diverse set of strategies, farmers can maintain the effectiveness of herbicide-tolerant crops while managing weed resistance.

### **CHAPTER VIII**

### WEED DISSEMINATION

# **MR. P. RAJAVEL**

# Assistant Professor, School of Agriculture,

### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Weed dissemination, or the process by which weeds spread and propagate, is a crucial factor in the management of invasive and undesirable plant species. Understanding how weeds disperse and establish in new areas is fundamental for developing effective control strategies and preventing their proliferation. Weeds disseminate through a variety of mechanisms, each adapted to their specific ecological niches and life cycles. Seed Dispersal is the most common method, involving several pathways. Wind-dispersed seeds, such as those of dandelions, have lightweight or aerodynamic structures that allow them to travel long distances. Water-dispersed seeds can float and move across water bodies, often leading to the spread of aquatic and riparian weeds. Animals, including birds and mammals, contribute to seed dissemination by carrying seeds on their fur or through their digestive systems, effectively transporting them to new locations. Additionally, seeds can be dispersed by humans through activities such as farming, gardening, and recreational use, often inadvertently introducing weeds to new areas.

Vegetative Propagation is another significant mode of weed dissemination. Many weeds can reproduce through vegetative parts like rhizomes, stolons, or tubers. For instance, plants like bindweed and quackgrass spread rapidly through underground rhizomes, which can establish new plants even if severed from the parent plant. This method allows weeds to proliferate locally and establish dense populations that are difficult to control. Reproduction Through Fragmentation involves the ability of some weeds to spread through broken or fragmented plant parts. For example, pieces of certain aquatic plants or invasive shrubs can grow into new plants when they are dispersed to new locations, contributing to their spread and establishment in various environments.

Understanding these dissemination mechanisms is essential for developing targeted weed management strategies. Effective control measures often require a comprehensive approach that considers the specific dissemination methods of the target weed species.

# CHAPTER IX

# INTEGRATED WEED MANAGEMENT (IWM) MR. S. BALAMURGAN

# Assistant Professor, School of Agriculture,

### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Integrated Weed Management (IWM) is a comprehensive and strategic approach to controlling weed populations while minimizing the adverse impacts on the environment, crop productivity, and economic resources. IWM combines multiple weed control methods to address the complexities of weed management effectively, aiming for a sustainable solution that balances efficacy, safety, and environmental impact.

IWM operates on the principle that no single weed control method is sufficient to manage all weed species or situations. Instead, it integrates a variety of techniques tailored to specific weed problems, crop types, and environmental conditions. This multifaceted approach enhances control effectiveness and helps prevent the development of herbicide-resistant weed populations.

Biological Controls utilize natural enemies of weeds, such as insects, pathogens, or animals, to reduce weed populations. For example, introducing herbivores that feed on specific weed species or deploying pathogens that target weeds can provide long-term control. This approach must be carefully managed to ensure that biological agents do not negatively affect non-target species or disrupt local ecosystems. An IWM strategy involves monitoring and evaluation to assess the effectiveness of control methods and adapt strategies as needed. Regular monitoring helps identify emerging weed problems and evaluate the success of integrated methods, allowing for adjustments and improvements in management practices.

In summary, Integrated Weed Management provides a holistic approach to weed control by combining cultural, mechanical, chemical, and biological methods. This strategy enhances control effectiveness, reduces the risk of herbicide resistance, and promotes sustainable agricultural practices. By integrating various methods and continuously adapting to changing conditions, IWM aims to achieve effective and environmentally responsible weed management.

# CHAPTER X

# HERBICIDE SPRAYING EQUIPMENT

# **DR. B. VIBITHA BALA**

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Herbicide spraying equipment is essential for modern agricultural practices, enabling precise and efficient application of herbicides to manage weed populations and protect crop yields. The effectiveness of herbicide applications hinges not only on the choice of herbicide but also on the quality and suitability of the spraying equipment used.

Herbicide spraying equipment encompasses a range of tools and machinery designed to apply herbicides in a controlled and effective manner. The primary goal of this equipment is to ensure uniform coverage of herbicide on target areas while minimizing drift, runoff, and harm to nontarget plants and the environment.

Backpack Sprayers: Ideal for small-scale operations and spot treatments, backpack sprayers are manually operated and allow for precise application in areas that are difficult to reach with larger machinery. They are commonly used in gardens, orchards, and small fields. Backpack sprayers are versatile and portable but require physical effort and careful calibration to ensure accurate application rates. Handheld Sprayers: Similar to backpack sprayers but typically designed for even smaller areas, handheld sprayers are used for targeted weed control and small-scale applications. They are convenient for precise application in areas like flower beds and vegetable gardens. Boom Sprayers: Used primarily in large-scale agricultural operations, boom sprayers are mounted on tractors or other vehicles. They consist of a long bar (the boom) equipped with multiple nozzles that spray herbicide in a wide, even pattern across the crop field. Boom sprayers are efficient for covering large areas but require careful calibration to avoid overlap and ensure uniform coverage. Air-Assisted Sprayers: These sprayers use air currents to enhance the coverage and penetration of herbicides into dense foliage. They are particularly useful for tall or bushy crops and can improve the distribution of herbicides in challenging conditions. However, they require precise control to manage drift and ensure effective application.

# UNVEILING THE SECRETS OF SOIL AND CROP HEALTH

EDITED BY

**DR. P. SELVARAJ** 



# Unveiling the Secrets of Soil and Crop Health

Copyright © 2021 by Dr. P. Selvaraj

All rights reserved. No part of this book may be reproduced ortransmitted in any form or by any means without writtenpermission from the author.

ISBN: 978-93-6255-985-2

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

# TABLE OF CONTENT

Introduction
Chapter I: From Petri Dish to Farm Field: Microbial Solutions for Growers
Dr.B.Vibithabala, Chapter II: The Invisible Farmhands: Microbial Contributions to Crop Resilience43
Ms.M.Jasmine,
Chapter III: Nurturing Nature: Microbial Strategies for Soil Regeneration64
Ms.K.Shalini,
Chapter IV: The Microbial Mosaic: Diversity Driving Agricultural Innovation
Mr. T. Lakshminarayanan,
Chapter V: The Power of the Plough: Microbial Insights into Soil Health117
Dr. P. Selvaraj,
Chapter VI :Beneath the surface: Understanding soil composition and its impact on crop health129
Dr.S.Vigneshwaran
Chapter VII: Nutring growth : strategies for enhancing soil fertility and crop resilience139
Ms.R.Ponmathi
Chapter VIII : Soil Microbiome: The Hidden Ecosystem Beneath Our Feet146
Dr.K.C.Sivabalan
Chapter IX: Soil Organic Matter: The Key to Nutrient Cycling and Soil Fertility157
Dr.K.Ashokkumar
Chapter X: Soil Structure and Aggregation: Impacts on Water Infiltration and Root Growth166
Dr.Rajaguruvu
REFERENCE:

### **Chapter I**

From Petri Dish to Farm Field: Microbial Solutions for Growers

# Dr.B.Vibithabala,

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

In the evolving landscape of agriculture, the transition from laboratory research to practical application is bringing groundbreaking microbial solutions to farm fields. The journey from a Petri dish to a farm field represents a critical shift in how we approach crop management and soil health. Microbial innovations, once confined to scientific experiments, are now transforming agricultural practices, offering new tools for enhancing productivity, sustainability, and resilience.

Microbes, including bacteria, fungi, and other microorganisms, have long been studied in controlled laboratory environments for their diverse capabilities. These studies reveal their potential to improve soil health, boost plant growth, and manage pests and diseases. For example, research has demonstrated that certain bacteria can fix atmospheric nitrogen, making it available to plants, while mycorrhizal fungi can extend the root system's ability to absorb essential nutrients. These findings have laid the groundwork for developing microbial products that can be applied in real-world agricultural settings.

One of the most significant advancements is the development of microbial inoculants and biofertilizers. These products, derived from beneficial microbes, are designed to be applied to soil or seeds, where they establish themselves and enhance plant growth. For instance, biofertilizers containing nitrogen-fixing bacteria can reduce the need for synthetic nitrogen fertilizers, while inoculants with mycorrhizal fungi can improve nutrient uptake and plant stress tolerance. These microbial solutions are proving to be effective in promoting soil fertility, increasing crop yields, and enhancing environmental sustainability.

Another area of innovation is the use of bio pesticides, which harness the natural abilities of microbes to control pests and diseases. Bio pesticides, such as those derived from bacteria or fungi, offer an environmentally friendly alternative to chemical pesticides.

# Chapter II The Invisible Farmhands: Microbial Contributions to Crop Resilience Ms.M.Jasmine,

Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

the realm of agriculture, where the quest for crop resilience is ever-pressing, a hidden army of helpers works tirelessly behind the scenes. These are the "invisible farmhands" microorganisms that, despite their minuscule size, play monumental roles in enhancing crop resilience and sustainability. These microbial allies, including bacteria, fungi, and other microbes, are essential to the health and robustness of plants, helping them withstand environmental stressors and biotic pressures.

The concept of crop resilience encompasses a plant's ability to endure and adapt to adverse conditions, such as drought, salinity, disease, and nutrient deficiencies. While conventional strategies often focus on genetic improvements and chemical treatments, the role of soil and plant-associated microbes in fostering resilience is increasingly gaining recognition. These microorganisms form intricate networks and engage in diverse interactions with plant roots, contributing to various aspects of plant health and stress tolerance.

One significant way in which microbes bolster crop resilience is through their role in nutrient acquisition. Mycorrhizal fungi, for instance, form symbiotic relationships with plant roots, extending their hyphal networks into the soil. This partnership enhances the plant's ability to absorb essential nutrients, particularly phosphorus, which is crucial for energy transfer and stress responses. By improving nutrient uptake, these fungi help plants better cope with nutrient limitations and enhance their overall resilience.

Beneficial bacteria also play a vital role in enhancing plant stress tolerance. Certain strains of plant growth-promoting rhizobacteria (PGPR) produce compounds that stimulate plant growth and induce systemic resistance to pathogens. These microbes can also help plants manage abiotic stress by producing stress-alleviating substances or by improving soil structure and water retention. This microbial assistance allows crops to better withstand harsh conditions such as drought or high salinity.

# Chapter III Nurturing Nature: Microbial Strategies for Soil Regeneration Ms.K.Shalini ,

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

In the quest for sustainable agriculture and ecosystem restoration, soil regeneration has emerged as a critical goal. At the heart of this endeavor lies a hidden yet powerful force: microorganisms. These microbial agents are pivotal in nurturing nature and revitalizing degraded soils, making them essential to the success of soil regeneration strategies. Through their diverse and dynamic roles, microbes offer solutions that not only restore soil health but also enhance its productivity and resilience.

Soil regeneration involves a multifaceted approach to restoring the vitality and functionality of soil ecosystems. This process includes rebuilding soil structure, enhancing nutrient cycling, and promoting biodiversity. Microbes play a central role in each of these aspects. Bacteria, fungi, and other microorganisms engage in processes such as organic matter decomposition, nutrient mobilization, and symbiotic interactions with plants, all of which contribute to the rejuvenation of soil ecosystems.

One of the fundamental microbial strategies for soil regeneration is the decomposition of organic matter. Microbial communities break down plant residues, manure, and other organic materials, converting them into humus and other stable forms of organic matter. This not only enriches the soil with essential nutrients but also improves its structure and water-holding capacity. The result is healthier, more fertile soil that supports robust plant growth and enhances overall soil health.

Mycorrhizal fungi are another critical component of soil regeneration. These fungi form symbiotic relationships with plant roots, extending their hyphal networks into the soil. This interaction increases the plant's access to nutrients, particularly phosphorus, and improves soil aggregation. In addition, mycorrhizae can help plants better withstand environmental stresses, further supporting the regeneration process.

### **Chapter IV**

# The Microbial Mosaic: Diversity Driving Agricultural Innovation

# Mr. T. Lakshminarayanan Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

In the vast and complex world of agriculture, the concept of a "microbial mosaic" encapsulates the profound impact that microbial diversity has on driving innovation and enhancing agricultural practices. This microbial mosaic, comprising an array of bacteria, fungi, archaea, and other microorganisms, plays a pivotal role in shaping soil health, crop productivity, and sustainability. Embracing and harnessing this diversity is proving to be a game-changer in the quest for innovative agricultural solutions.

Microbial diversity in soil is not merely a measure of abundance but a crucial factor influencing the functionality and stability of soil ecosystems. Each microbial species contributes uniquely to processes such as nutrient cycling, organic matter decomposition, and disease suppression. This diverse microbial community forms a complex network of interactions that supports plant growth and soil fertility. For instance, different bacteria and fungi can break down various organic compounds, convert nutrients into forms accessible to plants, and protect crops from pathogens through competition and natural antagonism.

The realization of how microbial diversity drives agricultural innovation has led to exciting developments in the field. Advances in genomic technologies, such as metagenomics and high-throughput sequencing, have provided unprecedented insights into the microbial communities present in different soil types and agricultural systems. These tools have revealed a rich tapestry of microbial life, each playing specific roles in soil health and plant productivity. Understanding these microbial communities allows scientists and farmers to identify key species and functions that can be harnessed to enhance agricultural practices.

One significant area of innovation driven by microbial diversity is the development of tailored microbial inoculants and biofertilizers. By selecting and applying specific microbial strains that address particular soil deficiencies or plant needs, these products can improve nutrient availability, boost plant growth, and enhance soil resilience.

# **Chapter V**

# The Power of the Plough: Microbial Insights into Soil Health

# Dr. P. Selvaraj,

# Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

In the quest for sustainable and productive agriculture, the soil underpins everything—from crop growth to ecosystem stability. Historically, the plough has been a symbol of agricultural progress, enabling farmers to cultivate and manage their lands. However, as we delve deeper into the science of soil health, it becomes evident that the true power of soil management lies not just in mechanical tillage but in understanding and harnessing the invisible forces at work beneath the surface. This is where microbial insights come into play, revealing the profound impact that soil microorganisms have on soil health and agricultural sustainability.

Soil health is fundamentally linked to the diversity and activity of its microbial inhabitants. Microorganisms such as bacteria, fungi, and protozoa perform essential functions that maintain soil fertility, structure, and overall ecosystem health. They decompose organic matter, recycle nutrients, and form symbiotic relationships with plant roots. For instance, mycorrhizal fungi extend their hyphal networks into the soil, facilitating the uptake of nutrients like phosphorus and improving soil structure. Similarly, bacteria contribute to nitrogen fixation, converting atmospheric nitrogen into forms that plants can utilize, thereby enriching the soil and reducing the need for synthetic fertilizers.

The integration of microbial insights into soil management practices offers new opportunities for enhancing soil health. Advances in microbial ecology and soil science have provided a deeper understanding of how different microbial communities interact with each other and with plants. Techniques such as metagenomics and soil microbiome analysis have revealed the complexity of soil microbial populations and their functional roles. This knowledge is transforming traditional approaches to soil management by emphasizing the importance of fostering healthy and diverse microbial communities.

# **CHAPTER VI**

# BENEATH THE SURFACE: UNDERSTANDING SOIL COMPOSITION AND ITS IMPACT ON CROP HEALTH

# **Dr.S.Vigneshwaran**

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The foundation of successful agriculture lies hidden beneath our feet, in the intricate and dynamic world of soil. While the visible aspects of farming, such as crops and field management, often capture our attention, the true essence of agricultural productivity and crop health is deeply intertwined with soil composition. Understanding what lies beneath the surface is crucial for optimizing soil health, enhancing crop yields, and ensuring sustainable farming practices.

Soil is a complex matrix composed of mineral particles, organic matter, water, and air. Its composition and structure are influenced by factors such as parent material, climate, topography, and biological activity. These components collectively determine soil fertility, its capacity to retain water and nutrients, and its ability to support plant growth. The interaction between these elements shapes the environment in which crops grow and directly impacts their health and productivity.

At the heart of soil composition are the mineral particles—sand, silt, and clay—which define soil texture. The proportions of these particles affect soil properties such as drainage, aeration, and nutrient availability. For instance, sandy soils drain quickly and may require more frequent irrigation, while clay soils retain water but can become compacted, impeding root growth. Understanding soil texture helps farmers tailor their practices to manage water and nutrient availability effectively.

Organic matter, which includes decomposed plant and animal residues, plays a vital role in soil health. It enhances soil structure, increases nutrient and water-holding capacities, and supports a diverse microbial community. Microorganisms such as bacteria, fungi, and protozoa break down organic matter, releasing essential nutrients and forming humus, which improves soil fertility and promotes plant growth. Healthy soil ecosystems with rich organic content are better equipped to support robust crops and resist environmental stresses.

# CHAPTER VII NUTRING GROWTH : STRATEGIES FOR ENHANCING SOIL FERTILITY AND CROP RESILIENCE Ms.R.Ponmathi

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

In the pursuit of agricultural excellence, the twin goals of enhancing soil fertility and boosting crop resilience are paramount. These objectives are foundational to achieving sustainable and productive farming systems capable of withstanding environmental challenges and meeting the growing demands of global food security. To effectively nurture growth, a comprehensive understanding of soil health and the adoption of innovative strategies are essential.

Soil fertility is the cornerstone of successful agriculture. It encompasses the soil's ability to supply essential nutrients to plants, support healthy microbial communities, and maintain an optimal physical structure. Fertile soil provides the necessary elements for plant growth, including macronutrients like nitrogen, phosphorus, and potassium, as well as micronutrients such as zinc and copper. Maintaining and improving soil fertility involves balancing nutrient inputs with plant needs, preventing nutrient depletion, and enhancing soil's natural ability to sustain crops over time.

Several strategies are employed to enhance soil fertility. Organic matter addition, through practices like composting and green manuring, enriches soil with nutrients and improves its structure and water-holding capacity. Organic matter also fosters a thriving microbial community, which plays a crucial role in nutrient cycling and soil health. In addition, precision agriculture techniques, such as soil testing and targeted fertilization, help in applying nutrients more efficiently and minimizing environmental impacts.

Crop resilience, on the other hand, refers to a plant's ability to adapt to and recover from various stressors, including drought, pests, diseases, and nutrient deficiencies. Resilient crops are not only more likely to survive adverse conditions but also to maintain productivity and quality. Enhancing crop resilience involves selecting and managing crop varieties that are
# **Chapter VIII**

# Soil Microbiome: The Hidden Ecosystem Beneath Our Feet

# Dr.K.C.Sivabalan

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Soil, often regarded as mere dirt, is in reality a complex and vibrant ecosystem teeming with life. Beneath the surface, a diverse community of microorganisms—bacteria, fungi, archaea, and viruses—forms the soil microbiome. This hidden ecosystem plays a crucial role in maintaining soil health, supporting plant growth, and ensuring the sustainability of terrestrial ecosystems.

# The Diversity and Functionality of Soil Microbiota

The soil microbiome is incredibly diverse, encompassing a vast array of microbial species that interact in intricate ways. Bacteria are among the most numerous, performing essential functions such as decomposing organic matter, cycling nutrients, and fixing atmospheric nitrogen. Fungi, including mycorrhizal species, form symbiotic relationships with plant roots, extending their reach into the soil to enhance nutrient and water uptake. Archaea, often found in extreme soil environments, contribute to processes like methane oxidation and nitrogen cycling. Viruses, though less understood, play roles in microbial population dynamics and the transfer of genetic material.

# **Microbial Interactions and Soil Health**

The interactions within the soil microbiome are crucial for soil health and fertility. Microbes decompose organic matter, transforming it into humus, which improves soil structure, water retention, and nutrient availability. They also participate in nutrient cycling, converting elements such as nitrogen, phosphorus, and sulfur into forms that plants can readily absorb. This microbial activity not only enhances soil fertility but also supports the growth and health of plants.

# Chapter IX

# Soil Organic Matter: The Key to Nutrient Cycling and Soil Fertility

# **Dr.K.Ashokkumar**

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Soil, often considered the Earth's skin, is a dynamic and complex system crucial to sustaining life on our planet. Among its many components, soil organic matter (SOM) plays a pivotal role in ensuring soil health and fertility. This chapter delves into the intricate world of SOM, exploring its functions, formation, and its indispensable role in nutrient cycling and soil fertility.

Soil organic matter is a broad term encompassing a diverse array of decomposed plant and animal materials, including humus, microbial biomass, and various organic compounds. It is the product of countless biological, chemical, and physical interactions within the soil ecosystem. SOM is not only a key component in soil structure, enhancing its ability to retain water and resist erosion, but it also serves as a reservoir of nutrients essential for plant growth.

The decomposition of organic matter releases nutrients such as nitrogen, phosphorus, and sulfur, which are vital for plant development. This process of nutrient release and transformation is central to the nutrient cycling within soils, ensuring that plants have a continuous supply of essential elements. Furthermore, SOM fosters a conducive environment for beneficial soil microorganisms, which in turn contribute to nutrient availability and soil health.

In this chapter, we will explore the different fractions of SOM, from the freshly added plant residues to the more stable forms like humus. We will also examine the processes of SOM formation and decomposition, and the factors influencing these processes, such as soil type, climate, and land management practices. Understanding these processes is crucial for developing strategies to enhance soil fertility and sustain agricultural productivity.

The significance of SOM extends beyond agriculture; it also plays a vital role in global carbon cycles and climate regulation

# **Chapter X**

Soil Structure and Aggregation: Impacts on Water Infiltration and Root Growth

## Dr.Rajaguruvu

# Assistant Professor, School of Agriculture, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Soil structure and aggregation are fundamental characteristics that influence the health and productivity of soils. Understanding these aspects is crucial for effective land management, as they directly impact water infiltration, root growth, and overall soil functionality. This chapter aims to explore the intricacies of soil structure and aggregation, shedding light on their critical roles in maintaining soil fertility and ecosystem sustainability.

Soil structure refers to the arrangement of soil particles into various formations or aggregates, which can range from granular to blocky, prismatic, or platy. These structures influence the soil's ability to hold and transmit water, air, and nutrients. Aggregation—the process through which soil particles cluster together to form aggregates—is a key factor in determining soil structure. Well-aggregated soils typically exhibit a porous network that facilitates optimal water infiltration and root penetration, both essential for healthy plant growth.

Water infiltration, the process by which water enters the soil, is significantly affected by soil structure. Soils with good aggregation allow water to infiltrate more readily, reducing surface runoff and the risk of erosion. This capability is crucial in managing water resources, especially in agricultural settings where efficient water use is necessary for crop production. Conversely, poorly structured soils with compacted or poorly aggregated particles can impede water infiltration, leading to increased surface runoff, erosion, and reduced water availability for plants.

Root growth is another critical aspect influenced by soil structure and aggregation. Healthy root systems require access to adequate water, nutrients, and oxygen, all of which are facilitated by well-structured soils. Soils with favorable aggregation provide a stable yet porous environment that supports expansive root networks and efficient nutrient uptake. In contrast, compacted or poorly aggregated soils can restrict root development, leading to stunted plant growth and reduced agricultural productivity.

# ENTREPRENEURSHIP AND BUSINESS DEVELOPMENT

Edited by DR. P. SELVARAJ



# **Entrepreneurship and Business Development**

Copyright © 2021 by Dr. P. Selvaraj

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-243-3

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

# TABLE OF CONTENT

Introduction
Chapter I: Study on formulation of project feasibility18
MS. PONMATHI RAMASAMY
Chapter II: MSME policies43
DR. A. RAVIKUMAR,
Chapter III: SWOT analysis for business management62
DR. P. SELVARAJ
Chapter IV: Business plan and proposal writing84
DR. B. GUNA
Chapter V: Constraints in setting up of agro based industries112
MS. J.U. JANUSIA
Chapter VI: Entrepreneurship institutions in India132
MR.P.RAJAVEL
Chapter VII: Entrepreneurship schemes155
MS.G.PRIYA DHARSHINI
Chapter VIII: Accessing Finance for Agricultural Start-ups178
MS.N.DIVYAPRIYA
Chapter IX: Marketing and Promotion in Agricultural Enterprises198
MS.M.JASMINE
Chapter X: Scaling Agricultural Ventures and Managing Growth
MR.S.TAMIL COVANE
REFERENCE

# Chapter I

# Study on formulation of project feasibility

# MS. PONMATHI RAMASAMY

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

The formulation of project feasibility is a critical process in project management that assesses the viability of a proposed project before its execution. This study focuses on evaluating the practical, financial, technical, and operational aspects of a project to determine its potential for success and identify any risks or challenges that may arise.

The feasibility study begins with an in-depth analysis of the project's objectives and scope. This involves defining the project's purpose, goals, and expected outcomes, as well as understanding the target audience or market. A clear definition of the project scope helps in establishing the parameters for feasibility analysis and ensures that all relevant aspects are considered.

Financial feasibility is a key component of the study, assessing whether the project can be funded and whether it is economically viable. This involves estimating the project's costs, including capital investment, operational expenses, and potential sources of funding. Additionally, the study includes financial projections such as cash flow statements, profit and loss forecasts, and break-even analysis to evaluate the project's profitability and financial sustainability.

Technical feasibility examines whether the project can be implemented with the available technology, resources, and expertise. This includes assessing the technical requirements, such as equipment, technology, and human resources, and determining if they are accessible and adequate for the project's needs. It also involves evaluating the technical risks and challenges that might impact the project's success.

Operational feasibility focuses on the practical aspects of implementing the project. This includes analyzing the project's alignment with existing processes and systems, assessing the organizational structure and capacity to support the project, and identifying potential operational challenges. Ensuring that the project is manageable within the organization's

# **Chapter II**

# **MSME** policies

# **DR. A. RAVIKUMAR**

# Assistant Professor, School of Agriculture,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Micro, Small, and Medium Enterprises (MSMEs) play a vital role in the economic development of countries by driving innovation, creating jobs, and contributing to the overall economic growth. MSME policies are designed to support and promote these enterprises, addressing their unique challenges and leveraging their potential to foster a dynamic and resilient economy.

MSME policies typically focus on several key areas to support the growth and sustainability of these businesses. One of the primary aspects is **financial support**, which includes providing access to credit, grants, and subsidies. Many MSMEs face difficulties in securing financing due to their limited resources and perceived risks. Policies aimed at improving access to financial resources, such as lower interest rates, easier loan conditions, and government-backed financing schemes, are crucial for enabling MSMEs to expand their operations and invest in innovation.

Another important area is **regulatory support**. MSMEs often encounter bureaucratic hurdles and complex regulatory requirements that can hinder their growth. Policies that simplify regulatory processes, reduce compliance costs, and offer incentives for adherence to standards can help alleviate these challenges. Streamlining business registration, tax compliance, and other regulatory aspects can create a more conducive environment for MSMEs to thrive.

**Capacity building** is also a significant focus of MSME policies. This includes providing training and development programs to enhance the skills and capabilities of MSME owners and employees. Policies that support education and training in areas such as business management, technology adoption, and marketing can help MSMEs improve their competitiveness and efficiency.

#### **Chapter III**

# SWOT analysis for business management

# **DR. P. SELVARAJ**

# Assistant Professor, School of Agriculture,

# PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India.

SWOT analysis is a strategic tool that aids business management by providing a structured framework to evaluate an organization's internal strengths and weaknesses alongside external opportunities and threats. This analytical approach helps businesses understand their current position and formulate strategies that leverage their advantages while addressing potential challenges.

The process begins with identifying the organization's strengths, which are internal attributes and resources that contribute to its competitive edge. These strengths could include a robust brand reputation, skilled workforce, proprietary technologies, or efficient operational processes. By recognizing these positive factors, businesses can harness them to maintain or enhance their market position and capitalize on growth opportunities.

Conversely, weaknesses represent internal limitations or areas where the organization may be at a disadvantage relative to its competitors. These could be outdated technology, limited financial resources, or gaps in expertise. Identifying and understanding these weaknesses is crucial for addressing potential risks, improving internal processes, and enhancing overall performance.

Opportunities are external factors or trends that could be advantageous to the business. They might arise from market expansion, evolving consumer preferences, technological advancements, or regulatory changes. By seizing these opportunities, businesses can drive growth, innovate, and broaden their market reach.

Threats, on the other hand, are external challenges that could negatively impact the organization. These might include economic downturns, increased competition, regulatory changes, or supply chain disruptions. Recognizing these threats enables businesses to develop strategies to mitigate their impact, adapt to changing conditions, and protect their operations.

#### **Chapter IV**

# **Business plan and proposal writing**

# DR. B. GUNA

# Assistant Professor, School of Agriculture,

## Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Business plan and proposal writing are essential processes in establishing and advancing a business or project. A well-crafted business plan outlines the strategic direction, goals, and operational framework of an organization, serving as a roadmap for its growth and development. It details key aspects such as the business model, market analysis, competitive landscape, marketing strategy, financial projections, and operational structure. This comprehensive document not only guides internal decision-making but also serves as a tool for securing funding and attracting investors by demonstrating the viability and potential of the business.

Proposal writing, on the other hand, focuses on presenting specific plans or requests for support to external stakeholders, such as investors, partners, or grant providers. A proposal outlines the objectives, methods, and anticipated outcomes of a project or initiative, emphasizing how it aligns with the interests and requirements of the target audience. Effective proposals clearly articulate the problem or opportunity being addressed, propose solutions or actions, and justify the need for resources or support. They often include detailed plans, timelines, budgets, and performance metrics to convey the feasibility and benefits of the proposed project.

Both business plans and proposals require clarity, precision, and a thorough understanding of the target audience. While a business plan provides a broader view of the overall strategy and operations of a business, proposals are more focused and tailored to specific requests or opportunities. Crafting these documents involves careful research, strategic thinking, and effective communication to ensure that the goals are clearly defined and the value proposition is compelling.

In essence, business plan and proposal writing are crucial skills for any entrepreneur or organization seeking to articulate their vision, secure funding, and achieve strategic

#### Chapter V

# Constraints in setting up of agro based industries

## MS. J.U. JANUSIA

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Setting up agro-based industries presents a range of constraints that can impact their successful establishment and operation. These industries, which encompass the processing and value addition of agricultural products, face unique challenges due to their dependence on agricultural inputs and the nature of their operations.

One significant constraint is the **availability and reliability of raw materials**. Agro-based industries rely on consistent and high-quality agricultural produce, which can be affected by factors such as seasonal variations, climate conditions, and fluctuations in crop yields. Ensuring a stable supply of raw materials requires effective supply chain management and relationships with farmers and suppliers, which can be complex and resource-intensive.

**Infrastructure and logistics** also pose challenges. Agro-based industries often require specialized facilities for processing, storage, and transportation. Adequate infrastructure, such as cold storage units, processing plants, and transportation networks, is essential for maintaining the quality of perishable products and ensuring timely delivery to markets. Inadequate infrastructure can lead to inefficiencies, increased costs, and spoilage of products.

**Regulatory and compliance issues** are another significant constraint. Agro-based industries must navigate a complex regulatory landscape that includes food safety standards, environmental regulations, and labor laws. Compliance with these regulations requires considerable effort and resources, and failure to adhere to them can result in legal issues, fines, or operational disruptions.

Access to finance is often a critical barrier. Establishing and operating agro-based industries typically involves substantial capital investment for infrastructure, equipment, and working capital. Securing funding can be challenging, particularly for small and medium-sized enterprises, due to perceived risks and the need for substantial collateral.

## **Chapter VI**

# Entrepreneurship institutions in India

## **MR.P.RAJAVEL**

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Entrepreneurship institutions in India play a crucial role in fostering innovation, supporting startups, and driving economic growth. These institutions provide a range of services and resources designed to nurture entrepreneurial talent, facilitate the establishment and growth of new ventures, and create a supportive ecosystem for entrepreneurs.

One of the primary functions of these institutions is to offer **training and education**. Many institutions provide specialized programs, workshops, and courses that equip aspiring entrepreneurs with the necessary skills and knowledge to start and manage their own businesses. These educational initiatives cover various aspects of entrepreneurship, including business planning, finance management, marketing strategies, and legal considerations.

In addition to education, entrepreneurship institutions in India often offer **mentorship and advisory services**. Experienced entrepreneurs and industry experts provide guidance and support to new business founders, helping them navigate challenges, refine their business models, and make informed decisions. This mentorship can be invaluable in providing practical insights and accelerating the growth of new ventures.

Access to funding is another critical aspect of entrepreneurship support. Institutions such as government agencies, venture capital firms, and angel investor networks work to provide financial resources to startups and small businesses. They may offer grants, subsidies, and investment opportunities that help entrepreneurs secure the capital needed to launch and scale their ventures.

Moreover, these institutions facilitate **networking and collaboration** opportunities by organizing events, conferences, and seminars where entrepreneurs can connect with peers, potential partners, and investors. Building a strong network is essential for gaining industry insights, exploring new opportunities, and forming strategic alliances.

#### **Chapter VII**

# **Entrepreneurship schemes**

# **MS.G.PRIYA DHARSHINI**

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Entrepreneurship schemes are designed to support and promote the establishment and growth of new businesses by providing various forms of assistance and incentives. These schemes are crucial for fostering an entrepreneurial culture, driving innovation, and enhancing economic development. They typically offer a range of benefits, including financial support, technical assistance, and capacity-building resources, aimed at helping entrepreneurs overcome common challenges and achieve success.

One key aspect of entrepreneurship schemes is the provision of **financial assistance**. This can take the form of grants, subsidies, low-interest loans, or venture capital investments. Financial support helps entrepreneurs secure the necessary capital to start and expand their businesses, invest in technology and infrastructure, and manage operational costs. Access to funding is often a critical barrier for new ventures, and these schemes aim to alleviate this challenge by making financial resources more accessible.

In addition to financial support, entrepreneurship schemes often include **technical assistance** and **training programs**. These programs offer entrepreneurs valuable skills and knowledge in areas such as business planning, marketing, finance management, and regulatory compliance. By providing education and hands-on training, these schemes help entrepreneurs build the expertise needed to navigate the complexities of running a business and make informed decisions.

Mentorship and advisory services are another important component of entrepreneurship schemes. Experienced mentors and industry experts provide guidance and support to entrepreneurs, helping them refine their business models, overcome obstacles, and seize growth opportunities. This mentorship can be instrumental in accelerating business development and enhancing the chances of success. Infrastructure support is also often included in entrepreneurship schemes. This may involve access to co-working spaces,

#### **Chapter VIII**

# **Accessing Finance for Agricultural Start-ups**

#### **MS. N.DIVYAPRIYA**

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Accessing finance is a critical factor in the success of agricultural startups, impacting their ability to begin operations, grow, and sustain long-term viability. Agricultural startups face unique financial challenges due to the sector's capital-intensive nature and the often unpredictable nature of agricultural outputs. Startups in this field need funds for purchasing land, investing in machinery and technology, and covering initial operational costs such as seeds, fertilizers, and labor. Financial support is crucial for bridging these gaps and enabling businesses to launch and expand effectively.

A range of financing options is available for agricultural startups. Government grants and subsidies are often designed to support agricultural innovation and rural development, offering financial aid that does not require repayment. These programs are typically aimed at encouraging advancements in farming techniques, sustainability practices, and the development of new agricultural technologies. Additionally, low-interest loans from government or development banks can provide essential capital for startups to cover their initial expenses and ongoing operational costs.

Venture capital and angel investors represent another significant source of funding. These investors provide equity financing, meaning they invest in exchange for ownership stakes in the company. This type of funding is particularly valuable for startups with high growth potential but limited access to traditional financial resources. Venture capitalists and angel investors not only provide capital but often bring valuable expertise, industry connections, and strategic guidance to the startups they invest in.

Securing financing requires a comprehensive business plan that outlines the startup's goals, market potential, financial projections, and operational strategy. A well-prepared business plan helps entrepreneurs present a compelling case to potential investors or lenders, demonstrating the viability of their venture and their ability to manage and utilize the funds

#### **Chapter IX**

# Marketing and Promotion in Agricultural Enterprises

#### **MS. M.JASMINE**

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Marketing and promotion are essential for the success and growth of agricultural enterprises, influencing how products are positioned, perceived, and sold in the marketplace. In the agricultural sector, effective marketing strategies are necessary to differentiate products from competitors, reach target audiences, and build a strong brand presence. The diverse nature of agricultural products—ranging from fresh produce to processed goods—requires tailored marketing approaches to address specific market needs and consumer preferences.

Developing a successful marketing strategy involves several key components. Identifying target markets is the first step, which includes understanding the demographics, preferences, and buying behaviors of potential customers. This information allows agricultural enterprises to tailor their marketing messages and promotional efforts to resonate with their audience. Additionally, creating a strong brand identity is crucial for establishing market presence and differentiating products. A well-defined brand communicates the unique qualities of the product, such as quality, sustainability, or local sourcing, and helps build trust and loyalty among consumers.

Effective marketing for agricultural enterprises often combines traditional and digital methods. Traditional marketing techniques, such as participating in trade shows, farmers' markets, and local advertising, help build community connections and reach local customers. Digital marketing strategies, including social media, email marketing, and online advertising, enable businesses to reach a broader audience and engage with consumers on various platforms. Utilizing digital tools allows for targeted advertising, customer interaction, and the ability to track and analyze marketing performance.

Promotional strategies should highlight the unique selling points of agricultural products and effectively communicate these attributes to potential buyers. Whether through storytelling, product demonstrations, or emphasizing sustainability practices, effective promotion helps

## Chapter X

# Scaling Agricultural Ventures and Managing Growth

## **MR.S.TAMIL COVANE**

# Assistant Professor, School of Agriculture,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Scaling agricultural ventures involves expanding operations and increasing capacity to achieve significant business growth. This process requires careful planning and execution to transition from a small-scale operation to a larger, more efficient enterprise. Effective scaling is essential for capitalizing on growth opportunities, increasing market share, and achieving long-term success in the competitive agricultural sector.

Key aspects of scaling include optimizing production processes and enhancing supply chain management. As agricultural ventures grow, it becomes necessary to streamline operations to handle higher volumes and maintain efficiency. This may involve investing in advanced technologies, such as automation and precision agriculture tools, to improve productivity and reduce operational costs. Additionally, effective supply chain management ensures that resources are efficiently sourced, products are delivered on time, and inventory is managed effectively.

Managing growth also entails addressing potential challenges that arise with expansion. Maintaining consistent product quality is crucial, as growth can introduce complexities in production and quality control. Ensuring that the business adapts to changing market demands and consumer preferences is also important for sustaining competitive advantage. This may involve diversifying product offerings, exploring new markets, or adjusting business strategies to meet evolving industry trends.

Financial stability is another critical consideration in scaling agricultural ventures. Effective financial management practices, such as budgeting, forecasting, and monitoring financial performance, help ensure that the business remains on a sound financial footing as it grows. Developing scalable systems and processes is essential for managing increased complexity and supporting long-term success. By focusing on strategic growth management and implementing robust operational practices, agricultural enterprises can navigate the



# COLONIALISM AND POST COLONIALISM

EDITED BY DR.D.RAVIKUMAR



# COLONIALISM AND POST COLONIALISM Copyright © 2021 by Dr.D.RAVIKUMAR

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-641-7 16-09-2021

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

CONTENT	PAGE	
CHAPTER-7.1 Shifting Identities in R.L. Stevenson's Postcolonial Fict Dr.K.SHIBILA	tion 1	-4
CHAPTER-7.2 Realism in Galasworthy's <i>"The Silver box"</i> Dr.R.A.RAJASEKARAN	5-	-10
CHAPTER-7.3 Cultural oppression and fight for freedom of igbo in ch <i>"Things Fall Apart"</i> . Dr.N.PREMA	11 linua Achebe's	1 -14
CHAPTER-7.4 The Link between Formalism and the School of Literar Dr.E.GEETHA	ry Criticism.	5- 18
CHAPTER-7.5 The Key Tension in Romanticism in Coleridge's <i>"Kubla</i> <i>"Ode to a Nightingale"</i> . <i>Ms.R.VISALAKSHI</i>	19 a Khan" and Keats's	9 -23
CHAPTER-7.6 Post structuralize Literary Criticism of English literatu Ms.K.JAYAPRIYA	ıre	4-26
CHAPTER-7.7 Colonial Econamic Exploitation of Kamala Markanday Mr.M.AMALRAJ	2". ya's <i>"Nector in a Sceive</i> ".	7-31
CHAPTER-7.8 Marginalization of Tribal people in Achebe's <i>"The Nov</i> Mr.S.RASAKUMAR	32 elist as a Teacher".	2-36
CHAPTER-7.9	43	3-47
Colonialism and Post Colonialism A Review Dr.N.MEENURAJATHI		

### Shifting Identities in R.L. Stevenson's Postcolonial Fiction

#### Dr.K.SHIBILA

#### Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Stevenson's Skittishness played out in this imperial context was complex and contradictory. If in Samoa he behaved 'like Cameron of Lochiel or Cluny MacPherson', treating the local staff as members of a clan, he did so partly in order to make them work. As Ilaria Sborgi infers, 'clanship was an ethical matter yet it was also part of the colonial game'. These paradoxes underscore Stevenson's uncertain identity in the Pacific. One may suggest, with Sborgi, that Scotland represented his country of origins and his past, the 'cultural identity' that enabled him to sympathies with the islanders, while England was the present, the 'home of his literary and intellectual constituency'.

Keywords: post colonialism, identity.

# Introduction

This paper examines some of the ways in which Stevenson addresses shifting forms of identity in three of his Pacific works of fiction: 'The Isle of Voices' (1893), 'The Bottle Imp' (1893), and The Ebb-Tide (1894). Stevenson denounced the fact that Pacific islanders were often endowed with a predetermined and counterfeit cultural identity: 'Everybody else who has tried [writing a South Sea story], that I have seen, got carried away by the romance, and ended in a kind of sugar candy epic, and the whole effect was lost' (Selected Letters, pp. 467-68). Although different in genre and style, these three tales present a pioneering picture of Pacific islands and islanders.

The innovative nature of these works in part reflects Stevenson's prior legal studies, which shape his use of classical references as a means of critiquing imperialism and Western character. That critique is also shaped by a sense that the destruction of Pacific cultures is

## Realism in Galasworthy's "The Silver box"

#### Dr.R.A.RAJASEKARAN

# Professor, Department Of English, PRIST Deemed to be University, Thanjavur Abstract

Realism as verisimilitude seems enviably simple to the twentieth century. If one seeks only the illusion of reality, achieved through factual detail, then realism is to be found in eighteenth century pseudo-histories and "true accounts" designed to allay puritanical suspicion of the fictional lie.But even a hundred years ago realism had come to mean more than that It had come to mean a vivid rendering of ordinary experience as opposed to the exotic imagination of romanticism. By this definition, Dickens and Thackeray are realists despite their sentimentality. And if concern for the ordinary is extended to Include an acceptance, motivated by "pity and love" as George Eliot puts It, of the common-to-mlddle-class citizen in the events and circumstances of his daily life, then almost all of the Victorian novelists are realists, for all their spirit of reform. It is in that spirit of reform, as a matter of fact, that the Victorians pay highest tribute to the real. Theirs Is a kind of lover's quarrel with reality. They seek the improvement, not the annihilation and replacement of the world they know. They question practice, but not value. The law's rigidity and the law's delay hang darkly over Oliver Twist and Bleak House, but law Itself remains a social good.

Key word:Reality,crime.

# Introduction

The spiritual confines of Middlemarch are too narrow for the soul of a Dorothea, yet there Is no denial of the need for moral order. Intellectual snobbery at Chrlstminster crushes Jude Fawley's aspiration, but misuse of learning does not negate the ideal. The Victorians address

# Cultural oppression and fight for freedom of igbo in chinua Achebe's "Things Fall Apart".

#### Dr.N.PREMA

# Associate Professor, Department Of English, PRIST Deemed to be University, Thanjavur

# Abstract

Historical fiction explores and uncovers paradigm, and helps readers to connect past and present in psychological and physical aspects. In straightforward and evocative prose, Chinua Achebe's 'Things Fall Apart' portrays how Igbo –a community known for its cultural richness is destabilised by the advent of British colonialists and European Christian missionaries. The focus is on the loss of customs, rituals and cultural pride of the indigenous people, replaced by European value systems, ideas and traditions. Achebe tried to convey the complex societal structures and bountiful culture of African people as a reply to Joseph Conrad in his novel 'Heart of Darkness' where he dehumanizes people of Africa. For quite some time western mindset has represented Africa as a land of hunger and undernourishment; while on the contrary the people of Africa encompass their country as a vast continent in terms of geographic variation and immense cultural diversity and comprises numerous ethnic groups with unique customs, languages and beliefs. In this paper, we critically analyze the downfall of Nigerian Igbo culture using both the novels to assess the hypothesis that post colonialism erodes the local culture of native Nigerians.

Keywords: Femininity, Homogeneous, Racialism, Monotheistic, Exterminated, Structuralism.

#### The Link between Formalism and the School of Literary Criticism.

# **Dr.E.GEETHA**

#### Assistant Professor, Department Of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Formalism may be defined as a critical approach in which the text under discussion is considered primarily as a structure of words. That is, the main focus is on the arrangement of language, rather than on the implications of the words, or on the biographical and historical relevance of the work in question. A strictly formalist critic would, for example, approach *The Great Gatsby* as a structure of words, ignoring the details of Fitzgerald's life and the social and historical contexts of the novel. However, formalism, or the concept of strict literary formalism, has often been attacked by individual literary critics or schools of criticism on the grounds that it reduces the text to nothing more than a series of words, thereby limiting its meaning and power. It is true that the Russian Formalists in the early years of the century attempted to examine the text in this way, but Western formalist approaches have tended to be much less theoretical. In practice, such critics have been very responsible to the meaning and themes of the work in question, rather than adopting a linguistic approach.

Key words: formalism, words structure.

# Introduction

Formalism may be defined as a critical approach in which the text under discussion is considered primarily as a structure of words. That is, the main focus is on the arrangement of language, rather than on the implications of the words, or on the biographical and historical relevance of the work in question.

Formalism is a school of <u>literary criticism</u> and <u>literary theory</u> having mainly to do with structural purposes of a particular text. It is the study of a text without taking into account any outside influence. Formalism rejects or sometimes simply "brackets" notions of culture or societal influence, authorship, and content, and instead focuses on modes, genres, discourse, and forms.

# The Key Tension in Romanticism in Coleridge's *"Kubla Khan"* and Keats's *"Ode to a Nightingale"*.

#### Ms.R.VISALAKSHI

# Assistant Professor, Department Of English, PRIST Deemed to be University, Thanjavur

# Abstract

The Romantic period revealed philosophical tensions, such as the conflicting conceptualizations of the role of the imagination and nature in facilitating greater individual understanding. Coleridge's 1816 poem '*Kubla Khan*' and Keats's 1819 poem '*Ode to a Nightingale*' demonstrates the conflict between Romantic writers on understanding the supremacy of the imagination in conjunction with reason to seek individual truth. Another Romantic tension arises from Coleridge's 1797 poem 'This Lime Tree Bower my Prison' and Shelley's 1818 novel *Frankenstein*, where the composers adopt distinct approaches in representing the relationship between the natural world and the individual in facilitating greater understanding. As these composers and their texts depict different permutations of Romantic philosophy, they illustrate the differing approaches to the imagination, nature and the individual.

Keywords: imagination, relationship.

# Introduction

The Romantics focused on the supremacy of the imagination, there were conflicting views as to the extent to which it could facilitate individual truth. In *'Kubla Khan*,' Coleridge conceptualizes the imagination and reason as the ideal Romantic sources of individual truth. By subtitling the poem as a "vision in a dream," and using fragmented stanzas, Coleridge uses the poetic form to emphasize the imagination as the purest form of literary expression, as argued in *Biographic Literaria* as "the prime agent of all human perception." Nonetheless, when Coleridge reaches the idealized "caverns measureless to man... in a vision he once saw" he synthesizes reality and the extraordinary through his awe-stricken tone to highlight the superiority of truth and the imagination working in conjunction, rejecting the Enlightenment era's strict adherence to rational thought. Indeed, Coleridge emphasizes how imagination and reason are interdependent in "sunny pleasure dome with caves of ice!" .

## Post structuralize Literary Criticism of English literature

#### Ms.K.JAYAPRIYA

#### Assistant Professor, Department Of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Post-structuralism refers to the intellectual developments in continental philosophy and critical theory that were outcomes of twentieth-century French philosophy. The prefix "post" refers to the fact that many contributors such as Jacques Derrida, Michel Foucault, and Julia Kristeva were former structuralisms who, after abandoning structuralism, became quite critical of it. In direct contrast to structuralism's claims of culturally independent meaning, post-structuralists typically view culture as inseparable from meaning.

Key words:Culture Abandoning, structuralism

#### Introduction

**Structuralism** is an intellectual movement that focuses on the underlying structures behind language and culture. Between the late 19th and mid-20th centuries, structuralists wanted to bring science back into the analysis of culture and language. Structuralist thought can be traced back to the French linguist Ferdinand de Saussure's (1857–1913) semiotic theory. The theory proposed that meaning in language and, as a result, culture (which structuralists argued were deeply connected) was constructed from a system of units called **linguistic signs.** 

Post-structuralism is difficult to define or summarize, it can be broadly understood as a body of distinct reactions to structuralism. There are two main reasons for this difficulty. First, it rejects definitions that claim to have discovered absolute 'truths' or facts about the world. Second, very few people have willingly accepted the label 'post-structuralist'; rather, they have been labeled as such by others. Therefore no one has felt compelled to construct a 'manifesto' of post-structuralism. Thus the exact nature of post-structuralism and whether it can be considered a single philosophical movement is debated. It has been pointed out that the term is not widely used in Europe (where most supposedly "post-structuralist" theory originates) and that the concept of a post-structuralist theoretical paradigm is largely the invention of American academics and publishers.

# Colonial Econamic Exploitation of Kamala Markandaya's "Nector in a Sceive"

# Mr.M.AMALRAJ

# Assistant Professor, Department Of English, PRIST Deemed to be University, Thanjavur ABSTRACT

Kamala Markandaya, one of the major Indo-Anglian woman novelists highlights the theme of East West encounter in almost all her novels. The introduction of industrialization, the encroachment of western technology in the Indian soil, apartheid and identity crisis are the under currents of the East West theme. Encroachment and Exploitation deals with the exploitation caused by industrialization which upset the life of the peasant community in a remote village in South India. In The Coffer Dams the encroachment of western technology spoils the tranquility of the tribal folks

Key words: economical. exploitation.

# **INTRODUCTION**

The two organic evils inherent in any social order are encroachment and exploitation. "Trespass upon the property, domain rights of another, especially stealthily or by gradual advances is encroachment according to Webster's Encyclopedia". An individual is exploited whenever advantage is taken of his ignorance or tolerance, his weakness or weariness or to seize his goods forcefully or demand his services at less cost. It is a common factor that the rich rob the poor of their land or property. Moreover the landlords draw their resources from the fertile and cultivated fields and swallow their wealth while the farmers or cultivators die of hunger and are never allowed to enjoy their booty.

Kamala Markandaya has firsthand knowledge of the troubles and turmoil's of the Indian lower rung because of encroachment and exploitation. Her novels *Nectar in a Sieve* and *The Coffer Dams* are records of the suppression and oppression of the farmers and tribal's because of the advent of industrialization and introduction of western technology.

Kamala Markandaya's *Nectar in a Sieve* is a realistic portrayal of the impact of industrialization on a farming community in a remote village of South India. Joseph remarks that," Kamala Markandaya's *Nectar in a Sieve* is an artistic and realistic depiction of the exodus

#### Marginalization of Tribal people in Achebe's "The Novelist as a Teacher".

#### Mr.S.RASAKUMAR

# Assistant Professor, Department Of English, PRIST Deemed to be University, Thanjavur ABSTRACT

Chinua Achebe is an iconic name in Africa as well as world literature. This paper committed to the social uplift of marginalized and downtrodden people. He believes that serious sense of responsibility to enhance the quality of humanity by way of exposing all manmade suppression and oppression in society. Achebe is a crusader against colonialism that enslaved the African countries and their people. He is opposed to the injustice and atrocities perpetrated by colonial rulers, and he wants to awaken the African people to rise up against the onslaught of colonialism in future. The present work serves as an inspiring guide to the African people to pursue the spirit of struggle to gain self dignity and recognition. It deals with a missionary zeal and exhorts the writers to use their art as a weapon to assert their confidence and past glory. For him, art is a means to bring about change in society. His works have served as a teacher for his readers. So, Achebe has become a novelist cum teacher, especially for African people, and in general for his readers all over the world.

**Key words:** Art, culture, role, , eradicate, poverty, literacy, social political, refinement, imbibe, nullify, vengeance, onslaught.

# **INTRODUCTION**

Chinua Achebe, one of the most distinguished Nigerian writers, was born on 16th November 1930, in Ogidi. He had held several offices as a Controller, Director, Researcher and Visiting Professor. He was a teacher at the University of Nsukka, Nigeria. Achebe is one of the most highly regarded of African writers in English. His greatest strength as a novelist is the steady refinement of his control over language. His stories deal with the social and political problems facing his country, including the difficulty of the post colonial legacy.

Achebe's (1965). "The Novelist as Teacher" is a talk delivered at the first Common Wealth Writers' Conference at Leeds in 1964. Art and Culture of the society should always entertain as well as instruct. According to him, the writer is an organic part of the society. Therefore, he requests the fellow African writers to advise the future generation through their art of writing. Each and every writer should

# **Colonial Imprints: Tracing the legacy in American Literary Tapestry.**

# Mr.S.PUNNIYAMOORTHY

# Assistant Professor, Department Of English, PRIST Deemed to be University, Thanjavur

# Abstract

The colonial imprints of native Americans for whom the arrival of the white man was an announcement of the apocalypse. "white men came swarming into the country bringing with them cards, money, fiddles, whiskey, and blood corruption." They included those countless, uncounted African Americans brought over to America against their will, starting with the importation aboard a Dutch vessel of "Twenty Negars" into Jamestown, Virginia in 1619. They even included some European settlers, those for whom life in America was not the tale of useful toil rewarded that John Smith so enthusiastically told. In the case with settlers of very limited means, like those who went over as indentured servants, promising their labor in America as payment for their passage there.

Keywords: legacy, colonial impact.

# Introduction

This paper impact of dominant that vision was, though, and in its English forms, along with the writings of John Smith (1580–1631), it was given most powerful expression in the work of William Bradford (1590–1657) and John Winthrop (1588–1649). Bradford was one of the Puritan Separatists who set sail from Leyden in 1620 and disembarked at Plymouth. He became governor in 1621 and remained in that position until his death in 1657. In 1630 he wrote the first book of his history, Of Plymouth Plantation; working on it sporadically, he brought his account of the colony up to 1646, but he never managed to finish it. Nevertheless, it remains a monumental achievement. At the very beginning of Of Plymouth Plantation, Bradford announces that he will write in the Puritan "plain style, with singular regard to the simple truth in all things," as far as his "slender judgement" will permit. This assures a tone of humility, and a narrative that cleaves to concrete images and facts. But it still allows Bradford to unravel the providential plan that he, like other Puritans, saw at work in history.

# CHAPTER-7.10 Colonialism and Post Colonialism A Review Dr.N.MEENURAJATHI

Assistant Professor, Department Of English, PRIST Deemed to be University, Thanjavur Abstract

The current paper is an attempt to review and present few main undercurrents and arguments of 'colonialism' and 'post colonialism,' and try to articulate how these concepts help us to understand wider aspects of both these movements, and the resultant influence on colonizer and colonized peoples, cultures as well as literatures. The paper, however, will mainly rely on understanding the basic concepts of both these movements and locate them in historical The current paper is an attempt to review and present few main undercurrents and arguments of 'colonialism' and 'post colonialism,' and try to articulate how these concepts help us to understand wider aspects of both these movements, and the resultant influence on colonizer and colonized peoples, cultures as well as literatures. The paper, however, will mainly rely on understand wider aspects of both these movements, and the resultant influence on colonizer and colonized peoples, cultures as well as literatures. The paper, however, will mainly rely on understanding the basic concepts of both these movements and locate them in historical peoples, cultures as well as literatures. The paper, however, will mainly rely on understanding the basic concepts of both these movements and locate them in historical perspectives to situate the emergence as well as decadence of colonialism and the gradual growth of postcolonial culture and literature across the countries that were once part of the vicious colonization project of the Europe.

Keywords: colonialism, imperialism, ideology, post colonialism

#### Introduction

Colonialism and imperialism are often used interchangeably. The word colonialism, according to the Oxford English Dictionary (OED), comes from the Roman 'colonia' which meant 'farm' or 'settlement,' and referred to Roman's who settled in other lands but still retained their citizenship. Accordingly, the (OED) describes it as:"a settlement in a new country... a body of people who settle in a new locality, forming a community subject to or connected with their parent state; the community so formed, consisting of the original settlers and their decedents and successors, as long as the connection with the parent state is kept up."

The meaning of the root word 'colony' or 'colonia,' suggest that these 'landed estates' were located in newly conquered territories of the Roman Empire and were generally given gifts to Roman citizen's who were mostly soldiers, for establishing Roman colonies in an otherwise hostile territories. But the modern European colonialism was by far the most extensive of the different kinds of colonial contact that have been a recurrent feature of the human history. The earliest forms of colonialism associated with the Roman Empire was different from the modern European colonialist, while as modern colonialism was

# SOCIAL CONCERN IN LITERATURE

\*





# SOCIAL CONCERN IN LITERATURE

Copyright © 2021 by Dr.E.GEETHA

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-847-3

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

CONTENT	PAGE
CHAPTER -6.1	1-5
Socio Linguistics in Study Tools	
Dr.K.Shibila	
CHAPTER-6.2	6-10
Social injustice in Silver box	
Dr.R.A.Rajasekaran	
CHAPTER –6.3	11-15
Unequal Social Relations in The Novel English	
Augustan Indian Story	
Dr.D.Ravikumar	
CHAPTER-6.4	16-20
Portrayal of Subaltern Resistance in Alice Walker's	
"The Color Purple"	
Dr.N.Prema	
CHAPTER –6.5	21-25
Ironic Twist of Fate in R.K Narayan'san Astrologer's Day	
Dr.E.Geetha	
CHAPTER-6.6	26-31
Social Fragmented Nation in Mulkraj Anand's	
"Untouchable".	
Ms.R.Vishalakshi	
CHAPTER-6.7	32-35
'Palanquin Bearers' by Sarojini Naidu	
K.Jayapriya	
CHAPTER-6.8	36-40
Oedipus Complex in Lawrence's Sons and Lovers	
Mr.M.Amalraj	
CHAPTER-6.9	41-45
Concept of Rivalry in D. H. Lawrence "Sons and Lovers"	
Mr.S.Rasakumar	

CHAPTER-6.10	46-48
Spiritual Journey Towards Perfection and Immortality in	
Sailing Byzantium	
Dr.N.Meenurajathi	
CHAPTER-6.11	49-53
Aspects of Heroism and Evaluation in Human Values	
Mr.S.Punniyamoorthy	
CHAPTER-6.12	54-56
Glorification of Igbo to Terms Rituals and Folk Lore in the	
Novelist and theTeacher	
Mr.T.Thiruppathi	
CHAPTER-6.13	57-60
The Folly of Human the Merry Wives of Windsor	
Mr.P.Kingsly Prem	
CHAPTER-6.14	61-65
A Quest for True and Selfless love in Kamaladas'	
My Grandmother's House	
Ms.M.Thamizhmani	
CHAPTER-6,15	66-69
The language of the incontinent body in	
Margaret Laurence The stone Angel	
Mr.M.Varadharajan	
CHAPTER-6.16	70-75
Cultural Catastrophe in The Lion and the Jewel	
Dr.R.Iniyavan	
CHAPTER-6.17	76-79
Girish Karnad'sNagamandala a Challenge to the	
Patriarchal Moral Code	
Ms.V.Indhumathi	
CHAPTER-6.18	80-84
The Ambiguity of the Title A Study of	

Wole Soyinka's The Lion And Jewel	
Dr.G.Karthiga	
CHAPTER-6.19	85-90
Caste, Untouchabilityand Social Justice in	
MulkrajAnandUntochable	
Dr.V.Deepa	
CHAPTER –6.20	91-94
The challenges of Modern English Literature	
Ms.G.Shanmugapriya	
CHAPTER-6.21	95-100
Concern and difficulties of gentle Women inn	
Jane Austen's "Emma".	
Ms.J.Sivamagudatharasi	

# **CHAPTER -6.1**

# Socio Linguistics in Study Tools Dr.K.Shibila

# Professor, Department of English, PRIST Deemed to be University, Thanjavur

# Abstract

Sociolinguistics is the study of how language and society relate to each other, and how language varies across social contexts. Sociolinguists use a variety of tools and methods to study social changes over time, Sociolinguistics can help us understand how our identity is linked to our use of language, and how language can be used to mark identity. For example, sociolinguists can study how pronunciation varies across social classes. Sociolinguistics has practical applications in areas such as language teaching, language policy development, and legal contexts.

keywords: Sociolinguistics, identity, pronunciation, development

# Introduction

According to Holmes, sociolinguistics is the study between language and society. Consequently, sociolinguists concern the relationship between language and cultural context in a region. The task of sociolinguistics is to describe how language works in our everyday lives, especially how people in certain areas or situation may speak in different language variation. Therefore, it is considered as a descriptive approach.

The study is worth-studying due to its ability to reveal the description of human interaction in specific context. As an English major student, the writer believes that sociolinguistics can prove the role of language within the society. By studying sociolinguistics, people can understand many language variations occur in society. As a result, sociolinguistics may reveal the characteristics of a society. In other words, sociolinguistics can be used as a tool to understand culture.

# CHAPTER-6.2

# Social injustice in Silver box

# Dr.R.A.Rajasekaran

# Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

It is intended to focus on the partial treatment of the rich and poor in the play The Silver Box by John Galsworthy. The Silver Box contains a social satire on unjust social partiality. The play is an appeal for compassion and sympathy for the poor who invariably suffer. It indicts the society for its contrasting treatment of two men, Jack and Jones, the one belonging to the affluent section and the other to the vertically opposite.. Law which is equal for all is ruthless and unjust towards the weak and helpless. The purpose of this paper is to portray both the characters with all their doings. However both the characters are found equally guilty in the play, then arise the role of wealth and social status in the court which makes Jack a respectable gentleman and the miserably poor Jones guilty. This paper will highlight the mastery of Galsworthy in portraying the reality of the 20th century English society and the Legal system and making the audience to be the real judges between Jack and Jones.

Keywords: partial treatment, contrasting, vertically,Legal

# Introduction:

Edwardian playwright and novelist, John Galsworthy was a comprehensive societal optimist and a realistic playwright. He wrote plays in first hand to instruct andthen to delight. He says, "My dramatic invasion and the form of it, was dictated by revolt at the artificial nature of English play of the period, and by a resolute intention to present real life on the stage".
# Unequal Social Relations in The Novel English Augustan Indian Story

# Dr.D.Ravikumar

#### Associate Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Since independence, Indian English writings, especially, novels have become a major part of the world literature canon. They are widely read across the globe and have garnered international accolades for the vivid presentation of the multilayered, multicultural and multilingual Indian society. Mulk Raj Anand says about the novels that "the novel at its most interesting is a process of inhale-exhale, a life-giving inspiration, a prose poem which releases the body and soul, even a new visionary glimpse of the miracle of life itself." He felt that novel was the most appropriate tool to depict the society and hence was the soul of literature. On the other hand, according to D.H Lawrence, novels are perfect medium as they reveal the changing rainbow of living relationship. Indian novels are unique blend of the glorious the pre-colonialism India, mindset of people during British rule, its long period of freedom struggle and a quest for identity. Indian English novelists such as Mulk Raj Anand, Raja Rao, Salman Rushdie, JhumpaLahiri, Arundhati Roy etc have made irreplaceable place for themselves among the elite international writers. They have also won many prestigious literary awards both nationally and internationally. These writers have successfully nativised the foreign language and beautified it with Indian motifs and colours. Their works portray the various phases of the rapidly changing Indian society from the pre independence era to the modern times. While doing so, they have faced many challenges in conveying the intricate depths and complex realities of the changing dynamics of the Indian society.

**Keywords:** Universal themes, contemporary writers, nativized, colonisation, multilingual, transition, postcolonial, languages, literary themes, culture, colonial modernity, realistic, independence, suppression, magical realism

#### Portrayal of Subaltern Resistance in Alice Walker's "The Color Purple"

# Dr.N.Prema

#### Associate Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Alice Walker is one of the most famous Afro-American women writers of our time and this is largely due to the novel "The Color Purple", an epistolary novel which is one of the most notable women's novel of 1980's in which Walker delineates women in different ways deprived of their freedom, rights, happiness, and force to live their life by sacrificing their needs and desires for male domination society by bestowing their heads. Men believe women as their private property and with them they can do whatever they want. In the novel men holds superiority over women. Thus the present paper studies about the subaltern characters of Afro-American women's destruction and inequality by the domination of male society.

Keywords: Gender Troubles, Male Domination, Racism, Sexism, Subaltern Literature

#### Introduction

Subaltern theme has become important that it is regularly used in various disciplinessuch as novels, dramatic studies, literary theory, films etc. In the context of globalization the word Subaltern means inferior in rank or of secondary importance. To the field of 'Subaltern Studies' three outstanding thinkers have contributed to it. They are, Antonio Gramsci, RanjitGuha and GayatriSpivak.

As stated by Julian Wolfreys people who are oppressed, marginalized, exploited on culturally, socially, politically and religious ground are grouped as 'Subaltern'. Thus Subaltern literature reflects diverse themes such as marginalization, oppression, Gender discrimination, disregarded women, neglected sections of society, subjugation of lower, working and deprived classes.

Antonio Gramsci says that the word 'Subaltern' is drawn from the Late Latin word 'Subaltern us'. In Latin 'Sub' means 'under' and 'alter' means 'other'. So literally it indicates to

# Ironic Twist of Fate in R.K Narayan'san Astrologer's Day

# Dr.E.Geetha

#### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

# Abstract

Narayan's writing is marked by its simplicity and clarity, yet it is infused with layers of meaning and emotion. His use of irony and his ability to build suspense keep the reader engaged from start to finish. The story's pacing is impeccable, leading to a conclusion that is both surprising and thought-provoking. An Astrologer's Day is a brilliant piece of short fiction that encapsulates R.K. Narayan's genius for storytelling. Through a seemingly simple narrative, Narayan explores profound themes of fate, deception, and the human condition. The story is a reminder of the unpredictability of life and the ways in which our pasts can shape our futures in unexpected ways. For readers looking to experience the magic of Narayan's writing, An Astrologer's Day is an excellent starting point.

Keywords: Simplicity, irony, suspense, surprising, unpredictability

# Introduction

An Astrologer's Day is a thought-provoking short story written by R.K. Narayan. The story revolves around an astrologer who earns his living by predicting people's futures. However, the irony is thickly interwoven throughout the story, making it a fascinating piece of literature to analyze. This paper will explore the various instances of irony in An Astrologer's Day and its significance in the overall theme of the story. The Irony of the Astrologer's Profession One of the most evident forms of irony in the story is the irony of the astrologer's profession. Despite being an astrologer with a supposed ability to foresee the future, the protagonist of the story is shown to be very ordinary and vulnerable. His supposed powers are a facade, hiding the fact that he is just a common man trying to make a living. This irony is evident when the astrologer states that he uses his skills to "foretell the future, to detect the past," yet he is unable to predict the attack

### Social Fragmented Nation in MulkrajAnand's"Untouchable".

# Ms.R.Vishalakshi

#### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Mulk Raj Anand's writing brought revolutionary change in the field of fiction writing. He wrote the novels for the sake of untouchables and the poor. He raised the issues of casteism, capitalism, feudalism, colonialism and imperialism through his novels. In Untouchable, he has attacked one of the worst social evils of the Indian society which was ignored by the previous writers and that is blot on Indian society, culture and tradition that has colonized eighty five percent people of Indian society. This sensibility has ruined creativity of Indian people. Casteism and untouchability are the blots on the face of humanity. Anand seems fighting for the liberty, equality and justice of the untouchables and the poor. He appealed for the basic human rights and needs in the newly emerging civil structure of colonial and post-independence India. Key words: Indian society, culture, ruined creativity.

#### Introduction

Raj Anand was a revolutionary writer of the twentieth century India who changed the mode of writing and thinking in the field of Indian fiction writing. The novelists before him, who had written fiction, wrote the fictional side of life which were ideal and romantic in nature. There were a smaller number of issues of the society. He had the opinion among all the fundamental rights that human dignity is the highest. Bakha, the leading character, had the resistance in the mind but he could not express it due to the fear of his caste. Bakha is a metaphor for all the untouchables of India. The novels of social resistance in Indian writing in English began with the big three— Mulk Raj Anand, R. K. Narayan and Raja Rao and still continue with the modern novelists. The novel of social protest is not new genre but it was one of the favorite genres of the twentieth century Indian English writers.

# 'Palanquin Bearers' by Sarojini Naidu

# K.Jayapriya

#### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

"Palanquin Bearers" is a poem by Sarojini Naidu, an Indian poetess known for her evocative and lyrical verses. The poem paints a vivid picture of a palanquin being carried through the streets of an Indian town. The poetess vividly recreates the scene of the 19th century and the early decades of the present century when the cars were not in fashion for carrying the brides to their husbands' house. On those days even the ladies of royal families were carried in the palki to their relatives. The brides too were carried in the palki. In any city of Northern India, particularly in Hyderabad palki was in fashion. The curtains were hanging as windows through which the ladies and brides could get the glimpse of outside. Palkis were carried on shoulders of two- or four-persons using bamboo-sticks for support. In order to avoid the physical tiredness and boredom of the way, the palanquin bearers used to sing.

Keywords:recreates,present,fashion,ladies and brides,tiredness

### Introduction

The poem "Palanquin Bearers" was first published in Sarojini Naidu's first poetry collection The Golden Threshold. It appears as the first poem of the Folk Songs section. The book was first published in 1896 in the United Kingdom and later in Hyderabad in 1905. Her poems published in this volume were admired in India and abroad. Politicians like Gopal Krishna Gokhale and Mahatma Gandhi appreciated her works. In this song, Naidu depicts a rural Indian scene. It centers on an image of a newly married woman carried by a palanquin. The poet vividly depicts how the bearers adorn her and carry her swiftly and softly to her destination. The poem, "Palenquin Bearers "shows a respect to a noble lady, most commonly newly wedded bride is being Carry to her husband's house in a veiled palenquin. Which is being carried on the shoulders by the palenquin Bearers merrily and lovely accomplished and singing a song in

#### **Oedipus Complex in Lawrence's Sons and Lovers**

### Mr.M.Amalraj

# Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Oedipus complex is a concept of psychoanalytic theory given by Sigmund Freud. Oedipus complex refers to a child's unconscious sexual desire for the opposite-sex parent and hatred for the same-sex parent. Lawrence was aware of Freud's theory, and Sons and Lovers famously uses the Oedipus complex as its base for exploring Paul's relationship with his mother. He writes many scenes between the two that go beyond the bounds of conventional mother-son love. Paul is hopelessly devoted to his mother, and that love often borders on romantic desire. Paul (and William, to a somewhat lesser extent) feels bound to his mother, and cannot imagine ever abandoning her or even marrying anyone else. Completing the Oedipal equation, Paul murderously hates his father and often fantasizes about his death. In the novel, Lawrence adds a twist to the Oedipus complex. The mother is saddled with it as well. She desires her sons in near romantic ways and despises all their girlfriends. The relation between the sons and the mother, which should have been a source of inspiration and comfort, turns into a corroding, blighting factor which gradually finishes one of the sons and presents obstacles in the other's life.

**Keywords:** Oedipus, psychoanalytic, unconscious, opposite, hatred, exploring, conventional, fantasizes, saddled, despises, inspiration, corroding, blighting, and obstacles.

# Introduction

Sons and Loversis undoubtedly one of the masterpieces of the English fiction. Lawrence himself gave it a lot of importance. It seems to be about a family at the surface level but Lawrence made it a novel about the struggle of self and identity in contemporary society. Its appeal is still universal though it is based on Lawrence's own experiences in life. Lawrence was aware of the conflicts, joys, sorrows and dependence of an individual on the other member of the society and the family.

# Concept of Rivalry in D. H. Lawrence "Sons and Lovers"

# Mr.S.Rasakumar

#### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Sons and Lovers (1913) is one of D.H. Lawrence's most prominent novels in terms of psychological complexities characteristic of most, if not all, of his other novels. Many studies have been conducted on the Oedipus complex theory and psychological relationship between men and women in Lawrence's novels reflecting the early twentieth century norms of life. This paper reexamines Sons and Lovers from the perspective of rivalry based on Alfred Adler's psychological studies. The discussion tackles the sibling rivalry between the members of the Morels and extends to reexamining the rivalry between other characters. This concept is discussed in terms of two levels of relationships. First, between Paul and William as brothers on the one hand, and Paul and father and mother, on the other. Second, the rivalry triangle of Louisa, Miriam and Mrs. Morel. The qualitative pattern of the paper focuses on the textual analysis of the novel to show that Sons and Lovers can be approached through the concept of rivalry and sibling Rivalry.

**Keywords**: Attachment theory, Competition, Concept of Rivalry, Favoritism, Sibling rivalry. **Introduction** 

Sons and Lovers is a well-known masterpiece of the early twentieth century focusing on the sensitive relations between men and women at the backdrop of the situations just before the First World War. Besides being an autobiographical in nature, it depicts the middle and working class conditions of the time. D.H. Lawrence was a great admirer of as well as influenced by Sigmund Freud's theories, and it can be fairly said that he wrote psychological manuscripts of human psyche in a literary framework. Based on Freud's interpretations, the novel takes the three stages of the major character's life from childhood into adolescence and adulthood. Lawrence

# Spiritual Journey Towards Perfection and Immortality in Sailing Byzantium Dr.N.Meenurajathi

# Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

William Butler Yeats' 1926 lyric poem Sailing to Byzantium refers to an ancient city famous for rich history, monuments, and permanence of art. The speaker takes a spiritual journey towards Byzantium with the hope that he can join the monuments of history and defeat the human ageing process. Therefore, the question of whether or not immortality is an achievable concept through art's permanence is raised in the poem. By using the present participle 'sailing' in the title, Yeats lays more emphasis on the active nature of the participle, focussing more on the journey than the final destination. By emphasizing the act of sailing, wrought with a sense of the unknown, the speaker creates a confusing contrast between mortality (Byzantium itself).

Key words: spiritual journey, mortality, immortality

# Introduction

Predictably, solutions to this question are not forthcoming to the reader early on, instead, appearing closer to the speaker's destination. In the first stanza, immortality and the permanence of art are not focussed upon; instead, his reasons for desiring to escape to Byzantium are promoted. The speaker uses nature to describe the bleak ageing process while mourning the sensual aspects of his mortal fate. 'The mackerel crowded seas' portrays nature youthfully, with 'crowded' representing the liveliness of nature and how the abundance of mackerel eases procreation.

The assonance highlights the fluidity of the poem, which depicts the continuity of the life cycle and therefore, mortal life. However, the claustrophobic lexis 'crowded' highlights that all the fish cannot survive with the limited resources in the 'crowded seas.' The speaker further dehumanizes life by portraying the rapidity of the life cycle through the curt triad 'fish, flesh or

# Aspects of Heroism and Evaluation in Human Values Mr.S.Punniyamoorthy

#### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

A study of Pratt's poetry reveals heroism as existing only in relation to an evolutionary metaphor that shows organic life progressing from an amoral environment to an ever more differentiated society. Man is heroic if he can establish, in his opposition to the environment, an order that is based on ethics and a "brotherhood of man" as well as on strong, instinctual emotions that keep him aware--consciously or unconsciously-- of his evolutionary origins. If man loses this awareness he loses his sense of identification with nature and exists within the illusion that the environment can no longer seriously affect him. The cost of this illusion is destruction, because at some time man must confront his environment, and if he has no feeling for it "in the bloodw, he has no effective means of coping with it.

Keywords: nature, Environment, Emotion.

#### Introduction

The most significant factor defining heroism is the ability to integrate uninhibited emotion with ethical compassion. Pratt's early poetry emphasizes instinctive feeling. In "The Great Feud" the anthropoid ape only realizes the necessity of ethics based on compassion after she has caused universal destruction. The Roosevelt and the Antinodes, written in 1930, demonstrates the ideal integration of feeling for the environment and compassion for other life. By 1935, however, with the writing of The Titanic, Watt went to the other extreme, almost denying the possibility of an ideal synthesis, for the ship symbolizes a society living under an all-encompassing illusion from which only a few individuals manage to free themselves. Rebuff and His Brethren is Watt's most complex poem, demonstrating the greatest conflict between the extremes of instinct and civilization in its progress toward a heroic resolution.

# Glorification of Igbo to Terms Rituals and Folk Lore in the Novelist and the

### Teacher

#### Mr.T.Thiruppathi

### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

# Abstract

This essay examines the glorification of Igbo culture through rituals and folklore as depicted in the works of Nigerian novelist Chinua Achebe and scholar Joseph A. Aloba. Achebe's literary contributions, particularly in \*Things Fall Apart\*, offer a rich portrayal of Igbo traditions, emphasizing their importance in shaping communal identity and social structure. Aloba's scholarly analysis further contextualizes these rituals and folklore, highlighting their roles in preserving cultural heritage and educating future generations. By exploring Achebe's narrative techniques and Aloba's academic insights, this essay elucidates how both contributors celebrate and preserve Igbo cultural practices, affirming their significance in the face of external pressures and cultural change.

### Introduction

The Igbo culture, with its rich tapestry of rituals and folklore, has been extensively explored and celebrated in both literary and academic contexts. Nigerian novelist Chinua Achebe and scholar Joseph A. Aloba provide profound insights into these cultural practices through their respective works. Achebe's novels, particularly \*Things Fall Apart\*, present a vivid portrayal of Igbo life, emphasizing the role of rituals and folklore in maintaining social order and cultural identity. Meanwhile, Aloba's scholarly analysis contextualizes these practices within a broader cultural and historical framework, underscoring their significance in Igbo heritage. This essay explores how both Achebe and Aloba glorify Igbo culture by highlighting the value of its rituals and folklore.

Achebe's \*Things Fall Apart\* offers a detailed depiction of Igbo rituals, such as the Week of Peace and the annual wrestling matches.

# The Folly of Human the Merry Wives of Windsor Mr.P.Kingsly Prem

#### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

The Merry Wives of Windsor by William Shakespeare is a five-act play that centers on Falstaff, a courtier from out of town, who is now in Windsor. As act 1 opens, Parson Hugh Evans is looking for Falstaff, who is charged with robbing both Justice Robert Shallow and his nephew, Abraham Slender the night before. Falstaff is honest about having stolen from them, and goes to dine with Mister and Mistress Page. Shallow and Evans convince Slender to try to earn Anne Page's affections. Anne, the daughter of Mister and Mistress Page, is the heir to a fortune. **keywords**: Justice, desire, affections, fortune

### Introduction

The merry wives, Mistress Ford and Mistress Page, devise a series of clever counter-plots to teach Sir John Falstaff a lesson. The first plot involves inviting Falstaff to Mistress Ford's house under the pretense of a romantic rendezvous. When Falstaff arrives, thinking he is about to enjoy a tryst, the mistresses put their plan into action. Mistress Ford feigns hospitality, indulging Falstaff's advances just long enough to ensure that he is relaxed and unsuspecting. In the midst of their interaction, Mistress Page arrives, pretending to be in a panic, announcing that Mistress Ford's husband has come home unexpectedly. Faced with the threat of being caught, Falstaff is quickly hidden inside a large laundry basket.

In this state of desperate concealment, the laundry basket, filled with dirty, foul-smelling clothes, is carried out by servants and unceremoniously dumped into a muddy ditch by the Thames River. Falstaff's first escapade ends in a humiliating and filthy escape, covered in soiled linens and soaking from the river water. Undeterred and still unaware of the full extent of the mistresses' cunning, Falstaff falls for their second trick. Convinced that Ford's sudden arrival was an unfortunate coincidence, he remains eager to pursue his amorous endeavors.

# A Quest for True and Selfless love in Kamaladas' My Grandmother's House Ms.M.Thamizhmani

#### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

The poem is a reminiscence of the poetess" grandmother and their ancestral home at Malabar in Kerala. Her memory of love she received from her grandmother is associated with the image of her ancestral home, where she had passed some of the happiest days of her life, and where her old grandmother had showered her love and affection. With the death of her grandmother the house withdrew into silence. When her grandmother died, even the house seemed to share her grief, which is poignantly expressed in the phrase "the House withdrew". The house soon became desolate and snakes crawled among books.

Key words: love and affection, isolation, family.

# Introduction

My grandmother's house is a poem written by Indian poet Kamala Das. The poem first appeared in an anthology of verse entitled 'Summer Time in Calcutta 1965). It is an autobiographical poem in which the speaker's nostalgic desire for home reflects through the inability to visit the happy past. The poem describes the speaker's happy life before her grandmother's death and sad life after her grandmother's death. The speaker of the poem is a married woman. She is reminded of her parental home which is the symbol of immense love. The poem describes the clear difference between past and present. In past, the life was full of activity whereas now it has turned into deadly silence. The intensity of sadness is expressed by dark and negative imagery.

Kamala Das is one of the three most popular Indian poets writing in English today, the other being Nissim Ezekiel and Ramanujan. Her poetry is all about herself, her deeply felt desire for love, her emotional involvement, and her inability to achieve such a friendship. In this poem, "My Grandmother's House," Kamala Das remembers her ancestral home and her deceased.

### The language of the incontinent body in Margaret Laurence The stone Angel

#### Mr.M.Varadharajan

Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Margaret Laurence's The Stone Angel deals with the themes of the dangers of pride, problems of growing old, duty, womanhood, resentment, suppression of emotions, and identity crisis that forms the foundation of the quest for survival. The novel clearly depicts the life of an old woman named Hagar who is threatened by the roots of identity crisis. As one grows old, they lose not only their physical and mental health alone, but also their existence or presence in the society. The woman in the novel is totally erased as her presence doesn't matters anyone. The country Canada is being portrayed as the land rich with wilderness and closely associated with Canadian population. In the early field of Canadian era, the major theme of Canadian literature was surrounded with the geographical area but on the later years it penetrated it roots towards nature, identity and literature. Canada is a nation distinct by its sole topography and ambiance. Key Words: Womanhood, Resentment, Suppression of emotions, and Identity crisis

#### Introduction

Women are mostly considered as weak and dependent on others by the society. This statement is irrelevant and cannot be agreed upon completely. There are women who fight through their lives all alone, who have to face the obstacles that life has to offer them without anyone by their side, by raising their heads in front of the society unwilling to bow down before them. The character Hagar, created by Margaret Laurence in her work The Stone Angel is such a person who does not wish to bow down before the male dominated society. Set in a fictitious town, Manawaka, The Stone Angel is a journey through the life and the mind of Hagar.

The Stone Angel reflects the feministic aspects of the title character. This paper explores the feminist outlook in The Stone Angel by Margaret Laurence. The main objective is to analyze the family relationships, marital life and the downfall of Hagar as a result of her pride.

# Cultural Catastrophe in The Lion and the Jewel

#### Dr.R.Iniyavan

# Associate Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Wole Soyinka (AkinwandeOluwole Soyinka), Africa's most distinguished playwright, was a Nobel Prize winner for his accomplishment in literature. He writes primarily for Nigerian audience, and his writings may be best appreciated with the region's politics and religion. He blends modern European dramatic form with such African elements as Yoruba tribal myth, folklore, dance, and music. His views are artistic hybrids of mixed Yoruba and European parentage, blending African themes, imagery and performance, idioms with Western techniques and stylistic influences. One of the most predominant themes of commonwealth Literature is the interaction between the forces of tradition and modernity. The meeting of East and West is a normal off shoot of the intermingling of culture due to the effect of colonization. His The Lion and the Jewel explores the value of traditional Yoruba ways and European innovations. This paper is an attempt to highlight the change in the lives of the people when civilization erupts into their lives as well as their resistance to modernization and colonization.

**Keywords**: Yoruba tribal myth, European innovations, resistance to modernization and colonization, traditional Yoruba ways.

#### Introduction

Commonwealth Literature attains a greater loftiness in its search for universality and truth. The writings of commonwealth writers centre on Caribbea, India, China, and parts of Africa. They have deep association with the European and American literature. The diversity of the commonwealth literature is further enhanced by the contemporary African literature. Modern African Literature is a collection of writing outside the tradition of Britain and United States. African poets and playwrights are conscious of African landscape and national identity and the image of African lives vividly in their writings. The prominence is placed on Nigerian Gender

# Girish Karnad'sNagamandala a Challenge to the Patriarchal Moral Code Ms.V.Indhumathi

# Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

The research paper presents the feminist approach in Girish Karnad's Naga-Mandala (1988) in the framework of postcolonial gender analysis. Naga-Mandala (1988) addresses the continued uneven power relations between female and male gender. Karnad's female character, Rani, in Naga-Mandala, is actually pitiable, downgraded and most importantly an object of patriarchal social, political dominance and authority. The paper postulates Rani as a site of theoretical transformations, engaging the issues of sexuality and power position in relation to the feminism patriarchal Indian state. Later Rani situates a per formative self in the text through an interrogatory narrative speech that succeeds in participating in the critique of patriarchal subjectivity and hegemonic feminist positioning while inserting a resistant feminist ideologies into gender discourse to re-envision the role of Indian women in India's development. Naga Mandala presents substantial constituents of feminism. The drama represents the patriarchal ethical enigma which burdens women with fidelity to their husbands but not the loyalty of men to their wives.

Keywords: Oppression, Feminism, Gender issues, Male-dominance, Patriarchy

# Introduction

The main purpose of the research paper is to show how in Naga-Mandala(1988), the female protagonist, Rani who in the start of the play is dominated, revolts against the limitation of social customs and eventually upsurges from her important position to a goddess. This study is to establish how Girish RaghunathKarnard (1938-2019) presents the conventional images of women that have always been imposed on them by patriarchy.

Girish Karnad is the foremost playwright of the contemporary stage. He has given the Indian theatre richness that cloud probably be equated with his talents as actor cum director. He

# The Ambiguity of the Title A Study of Wole Soyinka's The Lion And Jewel Dr.G.Karthiga

# Associate Professor, Department of English, PRIST Deemed to be University, Thanjavur

# Abstract

Wole Soyinka is a renowned post-colonial author whose works are capable of holding their own standing even when taken out of the African context that the works depict. His play that is the subject of this paper, The Lion and the Jewel, is yet again one such dynamic piece of writing, and the fact that it is a play makes it so all the more multi-dimensional; the stage directions, the props, and the settings add to the exotic elements of the play. What this paper seeks to do is to endeavour to explore the idea behind the title of the play and to match the character to the same. Whether such direct characterizations are valid is something that is further explored, as one finds that a particular quality that makes humans different from many other creatures is their ability to be multi-faceted. The answer as to who the lion and who the jewel are is not as simple as it seems.

Keywords: Wole Soyinka, post-colonial, The Lion and the Jewel, title, drama, play

# Introduction

"A high-voltage literary dynamo, he possesses magnificent power to shock, stimulate, agitate, ignite, activate, enlighten and all the while entertain his audience. He has got what it takes to move men and set them thinking," says BernthLindfors in praise of Wole Soyinka in his essay "Wole Soyinka, When Are You Coming Home?" (197). Accolades for Soyinka and his works see no limits as many critics and readers alike are quick to shower him with words of adoration. For the particular play as an object of study, The Lion and the Jewel, there is still more praise and accolades to come:

The Lion and the Jewel is a brilliant sex comedy in which the magnetic field of attraction and repulsion between three characters caught in an amusing African variation of the classic love.

#### Caste, Untouchabilityand Social Justice inMulkrajAnandUntochable

# Dr.V.Deepa

#### Associate Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Mulk Raj Anand's novel Untouchable (1935) concerned with the evil of violation of child rights in Indian society. It is a sociological novel which seeks to stress the evil of untouchability by focusing attention on the miserable plight, sufferings, poverty and degradation of a large section of Indian society. It brings to light the sorrows and sufferings that high caste Hindus inflicted on the untouchables. He strongly believes that the downtrodden in not born but made. His novel Untouchable centers around a sweeper boy, Bakha. The eighteen year boy, Bakha, son of Lakha, the jamadar of sweepers is a child of the twentieth century and the impact of new influences reverberates within him. The present paper aims at how the downtrodden suffer during the period of pre-independence India, socially as well as economically in the light of the characters of Mulk Raj Anand's Untouchable.

Keywords: Untouchable, Casteism, Violation, Suffering.

# Introduction:

Literature in India over the past many decades has emerged as a separate and important category of literature in many Indian languages. It has provided a new voice and identity to the communities that have experienced discrimination, exploitation and marginalization due to hierarchical caste system. Dalit literature has also made a forceful case for human dignity and social equality. Many of the Dalit writings have been translated into English and published as part of the anthologies of Dalit writings. The movement for Dalit literature has later spread to other languages like Gujarati, Punjabi, Hindi, Malayalam and Bengali. Dalit literature has used all literary forms – poetry, short stories, novels, plays and autobiographies in various languages. According to Bama; "Dalit identity gives them a different set of problems. They experience a total lack of social status; they are not even considered dignified human beings"(116).

#### The challenges of Modern English Literature

#### Ms.G.Shanmugapriya

# Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Learning English is already a challenging task, and when it comes to studying literature in English, the difficulty level increases even further. This Systematic Literature Review (SLR) is conducted to identify important challenges in the teaching and learning of English literature. The whole process of conducting this SLR followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) framework. PRISMA is a graphical representation of the complete process of doing a systematic review and meta-analysis, including the search for relevant articles, the filtering of them based on certain criteria, and the assessment of their quality. The results show that the factors can be categorised into three: student-related challenges, teacher-related challenges, and external-related challenges. This paper reports the key gap and limitations in current study that focuses on the challenges in the teaching and learning of English literature.

#### Introduction

Literature is an essential component of English language teaching (ELT) and is widely acknowledged as a vital source of authentic content for language development (Kaowiwattanakull, 2021). It offers various benefits to both students and teachers in acquiring English, such as providing rich language input for students to express themselves, motivating students to learn, catering to students with varying learning styles, and exposing students to creative writing and figurative language (Bist, 2018). Literature also serves as an example for its learners of how to effectively use both sentence structure and vocabulary, which are crucial for language learning (Kaowiwattanakull, 2021). The ability of literature in making the teaching and learning of English process easier and better has been reported in various studies, such as Nagayar et al. (2015); Emodi (2017); and Karlsson (2018).

# Concern and difficulties of gentle Women innJane Austen's "Emma".

### Ms.J.Sivamagudatharasi

### Assistant Professor, Department of English, PRIST Deemed to be University, Thanjavur

#### Abstract

Many critics have examined the shifting nature of female friendship in Jane Austen's Emma from cultural and historical angles. However, a comprehensive scientific analysis of female alliance and competition in the novel remains incomplete. The Literary Darwinist approach considers the motivations of fictional characters from an evolutionary perspective, focusing primarily on human cognition and behaviors linked to reproductive success, social control, and survival. While overt physical displays of male competition are conspicuous in the actions of the human species and those of their closest primate relatives, female aggression is often brandished psychologically and indirectly, which makes for a much more precarious study. In this paper, cultural criticism and evolutionary psychology work together to unravel the most complicated and arcane layers of intrasexual competition between women in Emma. Ultimately, this dual interpretation of the novel steers readers towards a deeper understanding of Emma Wood house's imperiled friendships, and by extension, their own. Jane Austen; Literary Darwinism; evolutionary psychology; mate selection; female-female competition

Keywords: Friendship,female alliance,competition,reproductive,aggression,imperiled.

# Introduction

Jane Austen wrote novels about relatively mundane day to day life during an era when popular novels were quite extravagant in plot and language. At the time, reading was just starting to become a widespread hobby in England. The bestselling novels of Austen's generation leaned towards the outlandish and spectacular. Austen wrote about her immediate surroundings more realistically by focusing on the comparatively banal everyday existence of English people rather than improbable plots and florid language Jane Austen is credited with being a pioneer of the Literary Realism movement that emerged in the 19th century.



நூல் : மலர் பதிப்பாசிரியர் : முனைவர் K. சசிகலா மொழி : தமிழ்

பதிப்பு : 18/12/2021

பதிப்பகம் : தமிழாய்வுத்துறை

பொன்னையா இராம ஜெயம் நிகர்நிலைப்

பல்கலைக்கழகம் .வல்லம், தஞ்சாவூர்-613 403

# பொருளடக்கம்

1.	பாதிரி மலர் முனைவர் சு.சதீஸ்வரன்	3
2.	பலாச மலர் முனைவர் மு.மலர்க்கொடி	9
3.	சங்க இலக்கியத்தில் இலவம் பூ முனைவர்.க.அறிவுக்கனி	14
4.	மலர்களும் மங்கையரும் முனைவர் பி. செல்வி	19
5.	சங்க இலக்கியங்களில் மலர்கள் திரு.து. கலியமூர்த்தி	22
6.	கூவிளம் பூ திருமதி க. வீணை முத்து	26
7.	தமிழரின் வாழ்வியலோடு கலந்த மலர்கள் திருமதி க. காளீஸ்வரி	30

# பாதிரி மலர்

முனைவர் சு.சதீஸ்வரன் உதவிப்பேராசிரியர் **, தமிழ் துறை** பொன்னையா ராமஜெயம் நிகர்நிலைப் பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்

#### முன்னரை

பாதிரி என்ற மலரின் விளக்கம், பாதிரி சொல்லின் பொருள், சொல் பொருள் விளக்கம், இம்மலரின் தன்மை, தமிழ் இலக்கியங்களில் பயன்பாடு, சங்கநூல் தரும் செய்திகள், பாடலம், அம்புவாகினி புன்காலி, மருத்துவக் குணங்கள் ஆகியன பற்றி இவ்வாய்வாவில் விளக்கம் தரப்படவுள்ளன.

சங்ககாலக் குறிஞ்சிப் பாட்டில் பாதிரி மலர் குறிப்பிடுகிறது. பாதிரி பருத்த அடிமரம் கொண்டது பாதிரி என்பது பொன் நிறப்பூ மரவகையைச் சார்ந்தது.

## சொல் பொருள்

- 1. அம்பு, அம்புவாகினி, பாடலம், புன்காலி மரவகை,
- 2. வெள்ளைப்பூ, சிவப்புப்பூ, பொன் நிறப்பூ மரவகை;
- 3. கிருத்துவ போதகர் (Rev. Father)

#### சொல் பொருள் விளக்கம்

அம்பு, அம்புவாகினி, பாடலம், புன்காலி எனவும் அழைப்பர்

# இம்மலரின் தன்மை

- 1. வழக்கமாக, பங்குனி, சித்திரை மாதத்தில் பூக்கள் பூக்கும்.
- 2. இம்மரத்தில், பூக்கள் காய்ந்து உதிர்ந்து விடும்; காய் பிடிக்காது.
- 3. பண்ணன் வாழ்ந்த சிறுகுடியில் இம்மரங்கள் மிகுதி.
- 4. இம்மரத்தில் அதிரல் கொடி ஏறிப் படரும்.
- 5. ஆற்றுத் துறையில் வேனில் காலத்தில் பாதிரி மலரும்.
- 6. பருத்த அடிமரம் கொண்டது.

#### பலாச மலர்

முனைவர் மு.மலர்க்கொடி

இணைப்பேராசிரியர் தமிழ் துறை

பொன்னையா ராமஜெயம் நிகர்நிலைப் பல்கலைக்கழகம்

வல்லம், **தஞ்சாவூர்.** 

தொடக்கவுரை:-

காலக் கண்ணாடி எனும் இலக்கியங்கள் தோன்றிய காலத்தில் வாழ்ந்த மக்களின் அக புற ஒழுக்கங்களை நமக்கு தெளிவாகப் படம் படித்துக் காட்டுகின்றன. அதைப் போலவே அக்காலத்தில் தாவரங்கள் விலங்குகள் முதலானவற்றையும் விளக்குகின்றன. இவ்வரிசையில் மலர்கள் பற்றி அறியும்போது நம்மை வியப்படையச் செய்கிறது. பத்துப்பாட்டில் உள்ள கபிலர் யாத்த குறிஞ்சிப்பாட்டில் 99 மலர்களைப் பற்றி அறிய முடிகிறது.

மலர்களில் தாம் எத்தனை வகைகள், அவைகள் மணம் தருவதுடன் சிறந்த மருத்துவ குணமுடையனவாகவும் உள்ளன. மொட்டு மலர்வதால் மலர் ஆகிறது. அது புஷ்பித்தலால் புஷ்பமாகிறது. ஆனால் நாம் அதை எளிதில் பூ என்கிறோம். இந்த பூவில் தான் எத்தனை வகைகள், எத்தனை வண்ணங்கள் எத்தனை வாசனைகள் மலர்ந்திருக்கும் பூக்களைக் கண்டாலே மனதிற்கு எத்துனை ரம்மியமாகவும் இதமாகவும் இருக்கிறது. சங்க இலக்கிய மலர்களில் குறிஞ்சிப்பாட்டில் குறிப்பிடும் 99 மலர்களில் ஒன்றான பலாசம் மலரைப் பற்றி இக்கட்டுரையில் காண்போம்.

- \* சங்க இலக்கியப் பெயர் –> பலாசம்
- \* சங்க இலக்கியத்தில் வேறு பெயர் –> புழுகு, முருக்கு
- \* பிற்கால இலக்கியப் பெயர்கள் –>புரசு,புரசை, புனமுருக்கு,

புனமுருங்கை,முருக்க மரம்

\* உலக வழக்கப் பெயர் 🛛 –> புரசு,பொரசு, செம்பூமரம்,

கல்யாண முருங்கை

- \* தாவரப் பெயர் –> பூட்டியா பிராண்டோசா
- \* ஆங்கிலப் பெயர் –> காட்டுத்தீமரம், ஆப்தி பரங்டு

புரசு (butea Monosperma) என்பது பலாச (butea) வகையைச் சேர்ந்த மரமாகும். இதற்கு பலாசு, பொரசு, புரசை என்று வேறு பெயர்கள் உண்டு. இது ஒன்பதாம் நூற்றாண்டு வரை பலாசம் என்றே அழைக்கப்பட்டது. இலையுதிர் காடுகளில் தானாக வளரும் மரமாகும். பலாசம் வலியற்ற ஒரு சிறுமரம். இதன் மலர்கள் செக்கச் சிவந்தவை இம்மரம் மூன்று அகன்ற பெரிய சிற்றிலைகள் கொண்ட கூட்டிலைகள்.

#### சங்க இலக்கியத்தில் இலவம் பூ

முனைவர்.க.அறிவுக்கனி உதவிப்பேராசிரியர் **, தமிழ் துறை** பொன்னையா ராமஜெயம் நிகர்நிலைப் பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்.

#### முன்னுரை

தமிழின் தொன்மைக்கும்,தனித்தன்மைக்கும் சான்றாக நிற்பது சங்க இலக்கியமே.தமிழன் என்று சொல்லடா தலை நிமிர்ந்து நில்லடா என்று நிமிர்ந்த நெஞ்சுடன் பாடினார் நாமக்கல் கவிஞர்.தமிழனைத் தலைநிமிர்த்தி நிற்குமாறு செய்வதே சங்க இலக்கியம் ஆகும். இலக்கியம் பிரதிப்பலிக்கும் காலத்தை காலக்கண்ணாடியாகும். இரண்டாயிரம் ஆண்டுகளுக்கு முன்னர் தமிழ் மக்கள் வாழ்ந்த வாழ்க்கையைக் கற்பனை வழுவாமல் நமக்கு படம்பிடித்துக் காட்டும் அற்புதக் காலக் கண்ணாடி சங்க இலக்கியம் அகம் -புறம் என்று வாழ்க்கையை இரணடாகப் பிரித்ததுமட்டும் அல்லாமல் இயற்கை அடிப்படையில் வாழ்க்கையையும், இலக்கியத்தையும் இருக்கண்களாக போற்றியவை சங்க இலக்கியம் புவி வெப்பமாதல் சூழலில் உலகமே இயற்கையைக் காப்பது பற்றிய கவலையை கொண்டிருந்தது. திணைக்கோட்பாடு அந்நிலையில் இயற்கை அடிப்படையிலான என்பதை உலகுக்கு அறிமுகம் செய்து வைத்தது சங்க இலக்கியம்.

1.குறிஞ்சித்திணை

2. முல்லைத்திணை

3.மருதத்திணை

4.நெய்தல்திணை

5.பாலைத்திணை என்பனவாகும்.

#### பதினெண்மேற்கணக்கு நூல்கள்:

எட்டுத்தொகை பத்துப்பாட்டும் பதினெண் மேற்கணக்கு நூல்களாகும் .

பத்துப்பாட்டு பாடல் வருமாறு

் முருகுபொருநாறுபாணிரண்டுமுல்லை

பெருகு வளமதுரைக் காஞ்சி -மருவினிய

கோல நெடுநல்வாடை கோல் குறிஞ்சி பட்டினப்

பாலை கடாத்தொடும் பத்து" .

என்பனவாகும் குறிஞ்சிப்பாட்டில் உள்ள "இலவம் பூ" பற்றி ஆய்வதே இக்கட்டுரையின் நோக்கமாகும்.

# மலர்களும் மங்கையரும்

முனைவர் பி. செல்வி

உதவிப்பேராசிரியர் , **தமிழ் துறை** பொன்னையா ராமஜெயம் நிகர்நிலைப் பல்கலைக்கழகம், வல்லம், தஞ்சாவூர்.

#### முன்னுரை

சங்க இலக்கியங்கள் மகளிரை தாவரங்களோடும் பறவைகளோடும் ஒப்பிட்டிருப்பதோடு, அவர்களை மலர்களுடன் ஒப்பிட்டுக் கூறியிருக்கின்றன. மகளிரின் ஒவ்வொரு உறுப்பையும் சங்க இலக்கியம் வெவ்வேறு மலர்களோடு உவமித்துள்ளது ·

சங்க இலக்கியங்கள் மகளிரை தாவரங்களோடும் பறவைகளோடும் ஒப்பிட்டிருப்பதோடு, அவர்களை மலர்களுடன் ஒப்பிட்டுக் கூறியிருக்கின்றன மகளிரின் ஒவ்வொரு உறுப்பையும் சங்க இலக்கியம் வெவ்வேறு மலர்களோடு உவமித்துள்ளது

மகளிரின் பற்களுக்கு உவமித்துக் முல்லைப்பூவை சங்க கூறுவது இலக்கியங்களில் மிகப்பரவலான வழக்காகும் மகளிரின் பற்களை விரித்து கூறுவதற்காகவே முல்லை மொக்கு உவமையாகக் கையாளப்படுகிறது முல்லை வெண்மையாகவும், மலர் சிறந்த மனமுடையதாகவும், தூயதாகவும் காணப்படுவதால் முல்லை மலரை கற்புக்கு அடையாளமாகவும் பழந்தமிழர் கொண்டனர்

> குல்லையம் புறவிற் குவிமுகை யவிழ்ந்த முல்லை சான்ற கற்பின் மெல்இயல் *(*சிறுபாண் 29,30)

தமிழிலக்கியங்களில் பற்களுக்கும், கற்புக்கும் முல்லைப்பூ ஒப்பிடப்படும். பண்டிகை, திருமணம் போன்ற சுப நிகழ்வுகளில் மகளிர் பூக்களைச் சூடுவது வழக்கமானது இதனால் மகளிர் அழகாக தோன்றுவதுடன் அவர்களின் கூந்தல் விளங்கும் சங்க மகளிர் நறுமணமுடையதாகவும் கால தமது கூந்தலை வழக்கமாகக் நறுமணமுடையதாக்க குடசம் எனும் மலரை சூடுவதை கொண்டிருந்தனர்

குரற்றவைக் கூந்தற் குடசம் பொருந்தி (சிலம்பு - ஊர்காண் காதை - 8)

# சங்க இலக்கியங்களில் மலர்கள்

.து. **கலியமூsர்த்தி** உதவிப்பேராசிரியர் **, தமிழ் துறை** பொன்னையா ராமஜெயம் நிகர்நிலைப் பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்

முன்னுரை

சங்க கால மக்கள் இயற்கையை உற்றுநோக்குவதில் வல்லவர்கள். மனிதர்களுக்கு அடிப்படைத் தேவைகளில் உணவுத்தேவையை நிறைவேற்றுவதில் இன்றியமையாத இடம் வகித்தவை மரங்கள். அம்மரங்களின் பெருக்கத்திற்குத் துணை செய்வன மலர்கள். ஆகவே, தான் இருக்கும் நிலத்திற்கும் மலர்களின் பெயரையே இடத்தொடங்கினான் மனிதன் எனலாம். சங்க காலத்தில் ஆடவர் பெண்டிர் ஆகிய இருபாலரும் மலர்களையும் தளிர்களையும் மாலையாகக் கட்டி அணிந்துள்ளனர்.

தலைவன் தலைவிக்குக் கையுறையாக மலர்களை வழங்குவது உண்டு. இன்றும் மலர்க்கொத்துகளைப் பரிசளிப்பதனை நாம் காணமுடிகிறது.

சங்க இலக்கியங்களில் மலர்பெறுமிடத்தை அகத்திணை, புறத்திணை என்று இரு திணைகளிலும் தனித்தனியே வகைப்படுத்திக் காணமுடிகிறது.

அகத்திணை தொல்காப்பியர் ஒவ்வொரு திணைக்கும் உரிய நிலத்தினைக் குறிப்பிடும்பொழுது,

> மாயோன் மேய காடுறை உலகமும் சேயோன் மேய மைவரை உலகமும் வேந்தன் மேய தீம்புனல் உலகமும் வருணன் மேய பெருமணல் உலகமும் முல்லை குறிஞ்சி மருதம் நெய்தல் எனச் சொல்லிய முறையாற் சொல்லவும் படுமே (அகத்திணையியல் – 5)

என்று நிலத்தை மலர்களின் பெயரால் சுட்டிச் செல்கிறார். கருப்பொருளினைக் குறிக்குமிடத்து,

> *தெய்வம் உணாவே மாமரம் புட்பறை செய்தி யாழின் பகுதியொடு தொகைஇ அவ்வகை பிறவும் கருவென மொழிப* ( அகத்திணையியல் – 20 )

என்று கருப்பொருட்களைக் குறிப்பிடுகிறார். இவற்றுள் பண்கள், யாழ்கள் ஆகியவை அவ்வந்நிலத்திற்குரிய பூக்களின் பெயராலேயே அழைக்கப்படுவதனை அறியமுடிகிறது.

சங்க இலக்கிய அகத்திணையில் மலர்களின் பயன்பாட்டினை, களவுக்காலம் கற்புக்காலம் ஆகிய இருநிலைகளிலும் அறியமுடிகிறது. களவுக்காலத்தில், களவு வாழ்வின் வெளிப்பாடு வெறியாட்டு மடலூர்தல் என்ற நிலைகளில் மலர்களின் இன்றியமையாமையை நாம் அறியமுடிகிறது.

#### கூவிளம்

# திருமதி க. வீணை முத்த உதவிப்பேராசிரியர் , **தமிழ் துறை** பொன்னையா ராமஜெயம் நிகர்நிலைப் பல்கலைக்கழகம், வல்லம், தஞ்சாவூர்

#### முன்னுரை

**வில்வம்** அல்லது **வில்வை** அல்லது **குசாபி** அல்லது **கூவிளம்** (Bael, Aegle marmelos) <u>இலங்கை, இந்தியா</u> மற்றும் அயனமண்டலத்தை சேர்ந்த ஆசிய நாடுகளில் காணப்படும் ஒரு தாவரமாகும். <u>சைவ சமய</u> மரபுகளில் வில்வ மரத்திற்கு முக்கிய பங்கு உண்டு.

#### கூவிளம்

கூவிளம் என்னும் சொல் வில்வ மரத்தைக் குறிக்கும். சங்ககால மகளிர் குவித்து விளையாடிய 99 மலர்களில் <u>கூவிளம்</u> என்பதும் காட்டப்பட்டுள்ளது. <sup>பு</sup>

#### எழினி

<u>கடையெழு வள்ளல்களில்</u> ஒருவனான எழினியின் குடிப்பூ கூவிளம். <sup>[2]</sup>

#### யாப்பியல்

'நேர்நிரை' அசை கொண்ட சீரமைதியைக் 'கூவிளம்' என்னும் வாய்பாட்டால் வழங்குவர்.

#### பயன்கள்

பழத்தின் உள்ளீடு நேரடியாக உண்ணப்படுவதுடன் உலரச் செய்யப்பட்டும், உணவு வகைகளுக்குப் பெறுமதி கூட்டப்படுவதன் மூலமும் உள்ளெடுக்கப்படுகிறது. இளம் இலையும் அரும்பும் சலாது தயாரிப்பதில் உபயோகப்படுகிறது.

தமிழில் 'கூவிளம்', 'இளகம்' எனப்பல பெயர்களில் வழங்கப்படும் இது தமிழ் சித்த மருத்துவத்தில் பல்வேறு பயன்களைக் கொண்டது. மூக்கடைப்பு, செரியாமை, காசம் முதலான நோய்களுக்கு இதன் இலை, பழம் என்பன மருந்தாகப் பயன்படுகிறது.<sup>து</sup> வில்வ வேர் சித்தமருத்துவத்தில் கூறப்பட்டுள்ள <u>பெரும்பஞ்ச மூலங்களுள்</u> ஒன்றாகும்.

#### ஆன்மீகப் பயன்கள்

இந்து மதத்தில் வில்வ மரம் மிகப் புனிதமானது.<u>சிவ வழிபாட்டில்</u> வில்வ பத்திர பூசை முக்கியமானது. முக்கூறுகளைக் கொண்ட வில்வ இலை திரிசூலத்தின் குறியீடாகக் கொள்ளப்படுகிறது. இது <u>இச்சா சக்தி, ஞானசக்தி, கிரியா சக்தி</u> என்பதைக் குறிக்கின்றது. <u>நேபாளத்தில்</u> <u>கன்னிப்</u> பெண்களின் கருவளத்தைக் காக்கவேண்டி வில்வம் பழத்திற்கு <u>திருமணம்</u> செய்து வைக்கும் <u>சடங்கு</u> பிரபலமானது.

தமிழ் - கூவிளம், வில்வம்

# தமிழரின் வாழ்வியலோடு கலந்த மலர்கள்

க . **காளீஸ்வரி** உதவிப்பேராசிரியர் **, தமிழ் துறை** பொன்னையா ராமஜெயம் நிகர்நிலைப் பல்கலைக்கழகம், வல்லம், தஞ்சாவூர்.

#### முன்னரை

சங்க காலத்தில் இருந்து தமிழர் வாழ்வில் கலந்து தமிழரின் மரபு உரிமை சொத்துகளில் ஒன்றாக வாகை மரம் விளங்கி வருகிறது. வாகை மரத்தினுடைய இலை, பூ, காய், பட்டை என அனைத்துமே மருத்துவ குணங்கள் கொண்டது. வீட்டிற்கு தேவையான கதவு, ஜன்னல், மேஜை, நாற்காலி மற்றும் மரச்செக்கு செய்வதற்கு வாகை மரம் பயன்படுத்துகிறது. வாகை மரம் மண்ணரிப்பை தடுக்கவும் பயன்படுகிறது. இன்றளவிலும் வாகை மரத்தின் இலைகளை ஆடு மாடுகளுக்கு தீவனமாக கொடுக்கிறோம். இந்த இலைகளில் இருக்கும் கால்சியம், புரதம், பாஸ்பரஸ் சத்துகள் ஆடு, மாடுகளின் வளர்ச்சிக்கு உதவுகின்றன.

#### வாகை மரத்தின் பயன்பாடு

வாகை மரத்தின் இலைகளை நிலத்திற்கு தழையுரமாக உபயோகம் செய்கின்றனர். காய்ந்து போன வாகை இலைகளில் 2.8 விழுக்காடு நைட்ரஜன் இருப்பதால் இது சிறந்த தழையுரமாகப் பயன்படுத்தப்படுகிறது. வாகை இலைகளை அரைத்து கண் இமைகளில் கட்டி வந்தால் கண் சிவத்தல், எரிச்சல் குணமாகும். வாகை மரத்தின் முக்கியமான பயன்களில் ஒன்றாக மரச்செக்கு பயன்பாட்டை கூறலாம். மரச்செக்கில் உள்ள உரல், உலக்கை வாகை மரத்தினால் செய்யப்பட்டது. மரச்செக்கில் கடலை, தேங்காய் என எதை போட்டு அரைத்தாலும் அதிலிருந்து கிடைக்கும் எண்ணெய் சுடாக இருக்காது.

#### வாகை மரத்தின் மருத்துவ பயன்கள்

வாகை மரப் பட்டையை பொடி செய்து அடிபட்ட காயத்தின் மீது தடவி வந்தால் காயம் விரைவில் குணமாகும். பசியெடுக்காதவர்களுக்கு வாகை மரத்தின் பட்டையை நிழலில் காய வைத்து நல்ல பொடியாக அரைத்து பாலில் கலந்து குடித்து வந்தால் நன்றாகப் பசியெடுக்கும். உடல் சூட்டினால் வரும் வாய் புண்ணை குணப்படுத்தும். கால்நடைகளுக்கு ஏதாவது காயம் இருந்தால் வாகை மரப் பட்டையை பொடி செய்து எண்ணெய் உடன் சேர்த்து காயங்கள் மீது தடவினால் விரைவாக குணமாகும். வாகை மரத்தின் பூக்களோடு தண்ணீர் சேர்த்து பாதியளவிற்கு காய்ச்சி குடித்து வந்தால் வாத நோய் குணமாகும். விஷங்களை முறிக்கும் தன்மையும் உடையது.

#### போரில் வெற்றி ''வாகை'' சூடினான்

மன்னர் காலத்தில் போர்க்களம் சென்று வெற்றி பெற்று திரும்பி வருகையில் போர் படையினர் வெற்றி பெற்றதன் அடையாளமாக வரும் வழியில் இருக்கும் வாகை மரத்தின்



நூல் : கதம்பம்

# பதிப்பாசிரியர்: ஆர் பாலச்சுந்தரம்

- மொழி : தமிழ்
- பதிப்பு : மார்ச் 2021
- ഖെണിപ്പ്റ്ര: 2024
- பதிப்பகம்
- தமிழாய்வுத்துறை பொன்னையா இராமஜெயம் நிகர்நிலை பல்கலைக்கழகம் வல்லம் தஞ்சாவூர்- 61340

# பொருளடக்கம்

1.	இனிய தமிழும் இஸ்லாமியர்களும்	3
	முனைவர் சு.அழகிரிசாமி	
		8
2.	திருக்குறள் சுட்டும் இயற்கை வேளாணமை	
	புணைவர் புடிபலர்க்கொடி	
3		13
J.	நாட்டுப்புறக் கலைகளை	15
	முனைவர் சே.சுகந்தி	
		24
4.	பழந்தயழகத்தல் கல்வ	21
	ഗ്രത്തെഖ്യ ക.എന്നിഖുക്കണ്	
5	சிலப்பதிதாரத்தில் அதப்பொருன் மாயல் மாற்றைபவல்	31
5.	வல்ப்பதில் பித்துல் அபைபொருள் மர்பும் மாற்றமும்	
	முனைவர் K.காளீஸ்வரி	
		39
6.	கம்பராமாயணத்தில் வாழ்வியல் சிந்தனைகள்	
	முனைவர் D. திருமுருகன்	
7.	தமிழ் இலக்கியம்	53
	திரு <b>т</b> . கலியமூர்த்தி	

1.இனிய தமிழும் இஸ்லாமியர்களும்

முனைவர் சு.அழகிரிசாமி உதவிப் பேராசிரியர்,தமிழ்த்துறை பொன்னையா இராமஜெயம் நிகர்நிலை பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்.

பசுந்தமிழ் பண்டும் இன்றும் என்றும் ஏற்றம் பெற்று வந்தமைக்கும், வருகின்றமைக்கும், வரப்போகின்றமைக்கும் பல்வேறு காரணங்கள் உள்ளன. அவற்றுள் தலையாய ஒன்று, தமிழ் மொழியைத் தாய் மொழியாகக் கொண்ட தமிழர்கள் தாம் தாம் எம்மதத்தினராயினும், தமிழ் மொழியைப்பேணி வளர்த்திருக்கின்றமையே என்று அறுதியிட்டு உறுதியாகத் கூறிவிடலாம். இந்நோக்கங்கொண்டு உழைத்திலரெனில், நாம் நம் பழம் பெரு இலக்கியங்களாகிய பத்துப்பாட்டு, எட்டுத்தொகை, பதினெண் கீழ்க்கணக்கு முதலான சங்க மருவிய நூற்களைப்பெற்று இருக்க இயலாது;

பழம்பெருமையையும் அறிந்திருக்க இயலாது. இச்சங்கமருவிய நூற்களைத் தந்தவர்கள் ஒரு குலத்தவர்கள் அல்லர். ஒரு சமயக் கொள்கையினர்கள் அல்லர். வெவ்வேறு குலத்தவர்களும் சமயத்தவர்களுமே ஆவர். அன்னார் அமைத்த இலக்கிய அணிகலன்களையே தண்டமிழ் அணங்கு தாங்கி நிலவி வருகின்றாள். இங்ஙனமே தமிழ் மகட்குப் பணி பூண்ட பிற்கால இஸ்லாமியர்களும் எம்முறையில் தம் கடனை ஆற்றியிருக்கின்றனர் என்பதை ஆராய்வதே இக்கட்டுரையின் ஒருபுடை நோக்கமாகும்.

பற்பல தமிழ்ப் புலவர்கள் இஸ்லாமியர் இனத்தில் பெருகத் தொடங்கினர். இஸ்லாமிய சமூகத்தவரைத் தமிழ் மொழி தன் இனிமைப் பண்பினால் ஈர்த்துவிட்டது. இதனால்தான் தாமாகக் கவிபுனையும் நிலையிலும் பல புலவர்கள் அச்சமூகத்தில் தோன்றத் தொடங்கினர் இங்ஙனம் புலவர் திலகங்களாக இருந்தவர்களின் வரலாற்றுக் குறிப்பினையும் அன்னார் யாத்த நூற்கள் சிலவற்றினையும் ஈண்டுக் குறிப்பிட்டால், உண்மை நன்கு புலனாகும்.

# 2.திருக்குறள் சுட்டும் இயற்கை வேளாண்மை

முனைவர் மு.மலர்க்கொடி

உதவிப் பேராசிரியர்,தமிழ்த்துறை

பொன்னையா இராமஜெயம் நிகர்நிலை

பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்.

உலகில் உள்ள உயிரினங்களில் மிகவும் இன்றியமையாதது மனிதப் படைப்பாகும். இம்மனிதன் உயிர்வாழ்வதற்கு அடிப்படைத் தேவையாக அமைவது உணவு, உடை, உறைவிடம் ஆகும். இவற்றில் ഉത്തവേ முதன்மையானது. இவ்வுணவுதான் மனிதனுக்குத் தேவையான அனைத்து வகையான ஊட்டச்சத்துக்களையும், ஆற்றல்களையும் வழங்கி நோயின்றி வாழ வழிவகுக்கின்றது. இந்த உணவை உற்பத்தி செய்வதில் இன்றியமையாததாகத் திகழ்வது வேளாண்மைத் தொழிலாகும். இவ்வேளாண்மைத் தொழிலின் தோற்றம், இன்றியமையாமை, திருக்குறளில் இடம்பெற்றுள்ள இயற்கை வேளாண்மைச் செய்திகள் பற்றிக் காண்பது இக்கட்டுரையின் நோக்கமாகும்.

# வேளாண்மைத் தோற்றம்

நாம் எந்த ஒரு துறையின் தோற்றத்தைப் பற்றித் தெளிவுற அறிய வேண்டின் அத்துறையின் வரலாற்றுப் பின்புலம் இன்றியமையாததாகின்றது. இதனடிப்படையில் வேளாண்மையின் தோற்றத்தைப் பழைய கற்காலம், புதிய கற்காலம் என்னும் வரலாற்றுப் பின்புலத்தில் பின்வருமாறு காண்போம்.

மனிதன் ஆரம்பக் காலக்கட்டத்தில் இயற்கையில் கிடைத்த காய்கள், கனிகள், கிழங்குகள், மிருகங்கள் போன்றவற்றை உணவாக உண்டு

# 3. நாட்டுப்புறக் கலைகள்

முனைவர் சே.சுகந்தி உதவிப் பேராசிரியர்,தமிழ்த்துறை பொன்னையா இராமஜெயம் நிகர்நிலை பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்

நாட்டுப்புற வழக்காறுகளில் நாட்டுப்புற இலக்கியம் தவிர்த்த மற்ற அனைத்து வழக்காறுகளையும் நாட்டுப்புறக் கலைகளுக்குள் அடக்கம் செய்யலாம். பின்பு அதனை மூன்று வகைகளாகப் பகுத்து அறியலாம். அவை,

1) நாட்டுப்புறப் பழக்கங்கள் (Folk Practices)

2) நாட்டுப்புறக் கலைகள் (Folk Arts)

3) நாட்டுப்புற அறிவியல் மற்றும் தொழில் நுட்பம் (Folk Science)

# 3.3.1 நாட்டுப்புறப் பழக்கங்கள்

வாழ்க்கை முறைகளின் கூட்டுச் சேர்க்கைதான் பண்பாடு என்ற ஒன்றை உருவாக்குகின்றது. ஒரு குறிப்பிட்ட சமுதாயத்தின் சடங்குகள், பழக்க வழக்கங்கள், நம்பிக்கைகள் இவை தாம் அச்சமுதாயத்தின் பண்பாட்டின் அடையாளத்தைக் காண வழி வகுக்கின்றன. சமுதாயத்தின் பழக்கங்கள் என்பனவற்றுள்,

1) நம்பிக்கைகள்

2) வழக்கங்கள்

- 3) சடங்குகள்
- 4) விழாக்கள்
- 5) விளையாட்டுகள்

போன்றவை அடங்கும்.

நம்பிக்கைகள்

நம்பிக்கைகள் மக்களால் உருவாக்கப்பட்டு அம்மக்களின் சமுதாயத்தால் பாதுகாக்கப் படுகின்றன. மனிதனின் தன்னல உணர்வும் சமுதாய உணர்வுமே நம்பிக்கைகளை வளர்த்து வருகின்றன. நம்பிக்கைகள் பெரும்பாலும் அச்ச உணர்வின் அடிப்படையில் அமைகின்றன. இயற்கைக்கு அப்பாற்பட்ட செயல்கள் நடக்கின்ற போது அதற்கு ஒரு காரணத்தைப் படைக்கின்றது. அதுவே நாளடைவில் மனித மனம் நம்பிக்கையாக மாறிவிடுகின்றது. சகுனங்கள் பார்ப்பதும் இந்த நம்பிக்கைக்குள் அடங்கும்.

நம்பிக்கையின் வகைகள்

# 4.பழந்தமிழகத்தில் கல்வி

முனைவர் க.அறிவுக்கனி உதவிப் பேராசிரியர், தமிழ்த்துறை பொன்னையா இராமஜெயம் நிகர் நிலை பல்கலைக்கழகம்

வல்லம், தஞ்சாவூர்.

பண்டைய தமிழ்ச் சமூகத்தில் இருந்த கல்வி குறித்து அறிந்து கொள்வதற்கு முன் அச்சமூகம் குறித்து சுருக்கமாக அறிந்து கொள்வது தேவையாகிறது. 2000 ஆண்டுகளுக்கு முந்தைய தமிழ்ச் சமூகம் பொருள்வளத்திலும் செல்வ வளத்திலும் நாகரிகத்திலும் உலகின் முன்னண

சமூகமாக இருந்தது. உலகின் பண்டைய புகழ் பெற்ற நாகரிகங்களான சுமேரிய எகிப்திய, சிந்துவெளி, கிரீட் போன்ற நாகரிகங்களுக்கு இணையாக சில விடயங்களில் அவைகளைவிட மேம்பட்ட நாகரிகமாக பண்டைய தமிழர் நாகரிகம் இருந்தது. காரணம் என்னவென்றால் இவை அனைத்தும் வெண்கலக் கால நாகரிகங்கள். ஆனால் தமிழர் நாகரிகம் இரும்புக்கால நாகரிகம். பெரும்பாலான நாடுகள் இன்றைக்கு 3000 ஆண்டுகளுக்கு முன்புதான் இரும்புத்தொழில்நுட்பத்தை அறிந்து கொண்டார்கள். ஆனால் இன்றைக்கு 5000 ஆண்டுகளுக்கு முன்பே இரும்புத் தொழில்நுட்பத்தை அறிந்தவர்களாக தமிழர்கள் இருந்தனர் (1). ஆகவே தமிழர் நாகரிகம், இரும்புத் தொழில் நுட்பத்தை அறிந்த ஒரு முன்னேறிய நாகரிகம் என்பதில் ஐயமில்லை.

பழந்தமிழ்ச் சமூகம் நகர நகர்மைய அரசுகளைக் கொண்ட நாகரிகமாக இருந்தது. நகர அரசுகள் என்பன பேரரசுகளைவிட சிறந்தனவாகவும் வளர்ச்சி பெற்றனவாகவும் இருந்தன என்பதை வரலாறு பலவகையிலும் மெய்ப்பித்துள்ளது. நகர அரசுகளில்தான், சுயமான சுதந்திரமான சிந்தனைகளும், சனநாயகக் கண்ணோட்டமும், தத்துவார்த்த அறிவியல் தொழில்நுட்ப வளர்ச்சியும், நிறைய புதிய கண்டுபிடிப்புகளும் இருக்கும் என்பதையும் வரலாறு பலவகையிலும் உறுதி செய்துள்ளது. தமிழக நகர அரசுகளை மக்கள் பிரதிநிதிகள் கட்டுப்படுத்தினர். என்பதால் இங்கு வரம்புக்குட்பட்ட முடியாட்சி முறைதான் இருந்து வந்தது. இவை போன்றவற்றின் காரணமாக பழந்தமிழக நகர அரசுகள், அதே காலகட்ட மகதப் பேரரசைவிட அரசியல், பொருளாதாரம், தொழில், வணிகம், கல்வி, அறிவியல், தத்துவம், தொழில்நுட்பம், இசை, ஓவியம், நாட்டியம், மருத்துவம், வர்மக்கலை, சிற்பக்கலை, கட்டிடக்கலை போன்ற பல்வேறு துறைகளிலும் உயர்தரமான வளர்ச்சியைப் பெற்றிருந்தன என்பது ஒ(ந உன்னதமான உறுதி செய்யப்பட்டுள்ளது (2)

21
5.சிலப்பதிகாரத்தில் அகப்பொருள் மரபும் மாற்றமும் முனைவர் K.காளீஸ்வரி உதவிப் பேராசிரியர், தமிழ்த்துறை பொன்னையா இராமஜெயம் நிகர் நிலை பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்

"தம்முடைய தண்ணளியும் தாமும்தம் மான்தேரும் எம்மை நினையாது விட்டாரோ விட்டுஅகல்க அம்மென் இணர அடும்புகாள் அன்னங்காள் நம்மை மறந்தாரை நாம்மறக்க மாட்டேமால். -காமம் மிக்க கழிபடர் கிளவி." (கானல்வரி- பா.32)

தமிழின் முதல்காப்பியம் என்ற பெருமைபெற்ற சிலப்பதிகாரம், தன் காப்பியக் கட்டமைப்பாலும் எடுத்துரைப்பாலும் உலகின் வேறெந்த மொழியிலும் காணக் கிடைக்காததோர் முன்மாதிரிக் காப்பியம் என்ற பெருமையினைப் பெறுகின்றது. வாய்மொழிக் காப்பியம், எழுத்து மொழிக் காப்பியம் என்ற வரையறைகளைக் கடந்து இரண்டின் பிணைப்பாலும் இயன்றதோர் புத்தாக்கப் படைப்பாகச் சிலப்பதிகாரம் திகழ்வது கற்போர்க்குப் பெருவியப்பினை அளிக்கின்றது. இக்காப்பியத்தின் தனித்தன்மையானதோர் அகக் கட்டமைப்பை உணர்ந்தே நம் முன்னோர்கள் இதனை இயல் இசை நாடகப் பொருள் தொடர்நிலைச் செய்யுள் என்றும் முத்தமிழ்க் காப்பியம் என்றும் உரையிடையிட்ட பாட்டுடைச் செய்யுள் என்றும் பலவாறாகப் பல அடைகளோடு அடையாளப்படுத்த முனைந்தனர். சிலப்பதிகாரத்தின் தனித்தன்மையான காப்பியக் கட்டமைப்பை இனங்காண்பதற்கு தமிழின் தொன்மை இலக்கண நூலான தொல்காப்பியத்தின் துணையை நாடுகின்றார் சிலப்பதிகாரத்தின் உரையாசிரியர் அடியார்க்கு நல்லார். பழந்தமிழ்ச் சங்க இலக்கியங்களின் தொடர்ச்சியாகவும் தொல்காப்பியம் கூறும் பாவியலுக்கு இலக்கியமாகவும் சிலப்பதிகாரத்தை அடையாளம் காட்ட அவர்

எடுத்துக்கொண்ட பெருமுயற்சியினை அவரது உரைப் பாயிரத்தால் நாம் உணரலாம்.

இயலிசைநாடகப் பொருள் தொடர்நிலைச் செய்யுளை அடிகள் செய்கின்ற காலத்து இயற்றமிழ் நூல் தொல்காப்பியம் ஆதலானும், பிறர் கூறிய நூல்கள் நிரம்பா இலக்கணத்தன ஆதலானும் அந்நூலின் முடிபே இதற்கு முடிபு என்று உணர்க. . . ..

### 6.கம்பராமாயணத்தில் வாழ்வியல் சிந்தனைகள்

திருமதி K. வீணைமுத்து உதவிப் பேராசிரியர், தமிழ்த்துறை பொன்னையா இராமஜெயம் நிகர் நிலை பல்கலைக்கழகம்

வல்லம், தஞ்சாவூர்

அறம் வெல்லும் பாவம் தோற்கும் என்பதே கம்பராமாயணப் பாவிகம் என்றாலும், அதையும் கடந்து வாழ்க்கைக்குத் தேவையான பல நல்ல கருத்துக்களையும் கம்பராமாயணத்தால் அறிந்து கொள்ளமுடிகிறது. கம்பராமாயணம் முழுமையும் பார்க்கும்பொழுது நல்வாழ்க்கை வாழத் தேவையான நற்பண்புகளையும், வாழ்வியல் நெறிகளையும் எடுத்தியம்புகிறது. கம்பர் தம் இராமாயணத்தில் நேரடியாகவோ,, கதை மாந்தர்கள் மூலமாகவோ வாழ்வியல் நெறிகளைக் கூறிச் சென்றுள்ளார். உலகில் என்பதைச் சான்றோர்கள் இன்பமும், துன்பமும் இயற்கை நன்கு அறிவர். கம்பராமாயணத்தில் சில வாழ்வியல் சிந்தனைகள் குறித்து ஆராய்வோம்.

துன்பத்தையும் அனுபவித்து தான் ஆக வேண்டும்

மனித வாழ்க்கையில் இன்பமும், துன்பமும் மாறி மாறி ஏற்படுவது இயல்பு. சான்றோர்கள் இரண்டையும் ஒரே மாதிரியாகவேக் காண்பார்கள். இராமனைக் காட்டிற்கு அனுப்பியதால் துயருற்ற சுமந்திரன் என்னும் மந்திரியைத் தேற்றினான். பெரிய தர்மமானது முற்பட நின்று கீர்த்தியை நிறுவி இறந்த பின்பும் அழியாமல் நின்று உறுதியைத் தருவது தர்மமே. உலகில் இன்பம் வந்தபொழுது மகிழ்பவர்கள் துன்பம் வந்த பொழுது அதையும் அனுபவித்து தான் ஆக வேண்டும். அதனால் நீ துன்பப்படாதே என்று இராமன் தேற்றினான்.

"முன்பு நின்று இசைநிறீஇ முடிவு முற்றிய பின்பும் நின்று உறுதியைப்பயக்கும் பேரறம் இன்பம் வந்து உறும் எனின் இனியதாய் இடைத் துன்பம் வந்து உறும் எனில் துறக்கல் ஆகுமோ" (தைலமாட்டுப்படலம் 552)

உலகில் முதலில் துன்புற்றாலன்றோ அடுத்து சுகம் கிடைக்கும்

7.தமிழ் இலக்கியம்

திரு**т.** கலியமூர்த்தி உதவிப் பேராசிரியர்,தமிழ்த்துறை பொன்னையா இராமஜெயம் நிகர்நிலை பல்கலைக்கழகம் வல்லம், தஞ்சாவூர்

இலக்கியம் என்பது விரிந்த பொருளில் எழுதிய அனைத்தையும் குறிக்கும்.<sup>ய</sup> இந்த வரையறையின் கீழ் இலக்கியத்தை இரு வகைகளாகப் பிரிக்கலாம்:

- 1. இன்பியல் இலக்கியம்
- 2. அறிவியல் இலக்கியம்

'இன்பியல்' இலக்கியம் "கற்போர் உள்ளத்துக்கு இன்பம் தரும் நூல்கள்".<sup>12</sup> அறிவியல் இலக்கியம் கற்போருக்கு அறிவை முதன்மையாகத் தரும் இலக்கியம்.

தமிழ் இலக்கியம்

தமிழ் குறைந்தது 2000 வருடங்கள் இலக்கிய வளமும் தொடர்ச்சியும் கொண்ட ஒரு மொழியாகும். எனினும், தமிழ் இலக்கியங்களில் பெரும்பாலானவை, இன்பியல் இலக்கியங்களே. இது, "இலக்கிய வளர்ச்சி, அரசர்களையும் குறுநில மன்னர்களையும் சுற்றி வந்ததால்" ஏற்பட்டிருக்கலாம்.<sup>இ</sup> அதன் விளைவாக இலக்கியம் என்ற சொல் தமிழில் இன்பியல் இலக்கியத்தையே பெரும்பாலும் குறித்து நிற்கின்றது. சிறுகதை, நாவல், கவிதை, நாடகம் ஆகிய எழுத்துக் கலை வடிவங்களே இன்று தமிழ் இலக்கியம் என பொதுவாகக் கருதப்படுகின்றது.

தமிழ் இலக்கிய வரலாற்றில், அறிவியல் இலக்கிய படைப்புகள் மிக அரிது. வரலாற்று ரீதியில், தற்கால ரீதியில், ஆங்கிலத்திடனும் ஒப்பிடுகையில் இந்தக் குறை தெளிவாகத் தெரியும். இன்று, அறிவியல் தமிழ் இலக்கியத்தின் தேவை கருதி <u>அறிவியல் தமிழை</u> வளர்க்க <u>தமிழ்நாடு</u> அரசும் தமிழ் ஆர்வலர்களும் பெரிதும் முயன்றுவருகின்றனர். இலக்கியம், இலக்கியத்துக்காக என்பதை விட, இலக்கியம், மக்களின் பயன்பாட்டுக்காக என்பதே அறிவியல் தமிழின் ஒரு முக்கிய விழுமியம் எனலாம்.

வரலாறு



# TEXTBOOK ON GENETICS

## EDITED BY

# DR. BAKRUDEEN ALI AHMED



### **Textbook of Genetics**

Copyright © 2021 by Dr. A. Bakrudeen Ali Ahmed

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-190-0



Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

Chapter 1	Dr. M. VIJAY	1-40
Basic Principle of		
Medical Treatment		
Chapter 2	Dr. BINUGEORGE	41-78
Nanotherapy		
Chapter 3	Dr. S. SATHISHKUMAR	79-135
Quantum therapy as well		
as Phyton Therapy		
Chapter 4	Dr. A. SUNDARESAN	136-183
Tissue Engineering and		
Medical Biopolymer		
Applications		
Chapter 5	Dr. G. SRITHAR	184-235
Transplantation in		
Medicine		
Chapter 6	Dr. A. BAKRUDEEN ALI	236-255
	AHMED	
Tissue Transplantation		
Chapter 7	Dr. S. AMBIGA	256-275
<b>Biopolymer properties</b>		
of Medical components		

### **Basic Principle of Medical Treatment** Dr. M. VIJAY

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

### **1.1 Introduction**

Tobacco smoking is the predominant cause of lung cancer, and it has been estimated that about 80% of lung cancers are attributed to smoking (Buiatti et al., 1996). As a complement to smoking cessation in lung cancer prevention, considerable attention has been focused on possible protective factors in the diet. A diet rich in fruits and vegetables has rather consistently been reported to be associated with a reduced risk of lung cancer (Ziegler et al., 1996). Several plausible mechanisms have been suggested, indicating that this reduction may be due to antioxidant micronutrients (Dorgan and Schatzkin, 1991). Thus far considerable attention has been directed to/3- carotene, but intervention trials have not confirmed the presence of a protective effect (Albanes et al., 1996; Hennekens et al., 1996; Omenn et al., 1996). Some evidence also suggests that the reduced risk of lung cancer associated with the intake of fruits and vegetables may be due to some other micronutrients, such as vitamin C (Block et al., 1992), flavonoids (Knekt et al., 1997a), and selenium (Clark et al., 1996). The epidemiological evidence is, however, not yet persuasive for any of these, while the question of whether lung cancer can be prevented or slowed down by the antioxidant vitamin E has also been addressed in a number of studies (Knekt, 1994). The aim of this chapter is to review findings from epidemiological studies on the role of vitamin E in cancer prevention and to present some new results from the Finnish Mobile Clinic Health Examination Survey.

Two intervention trials and 25 observational studies on the association between vitamin E status and lung cancer risk are considered here. In an intervention trial the investigator randomly assigns vitamin E or placebo to the study population and then waits for the highest per capita consumption diseases such as coronary heart disease and cancer. The epidemiological evidence is, however, not yet persuasive for any of these, while the question of whether lung cancer can be prevented or slowed down by the antioxidant. The association between vitamin E status and lung cancer risk are considered here. In an intervention trial the investigator randomly assigns

### Nanotherapy Dr. BINUGEORGE

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

### **2.1 Introduction**

There are currently 3 million tobacco-related deaths in the world each year, and in general smokers can expect to live 7 years less than nonsmokers (World Health Organisation, 1977). This premature mortality is because habitual smoking is associated with an increased risk of developing many diseases, including coronary heart disease, lung cancer, stroke, and emphysema (Health Education Authority, 1992). Paradoxically, analysis across countries reveals little relationship between smoking levels and mortality from diseases such as coronary heart disease and cancer. For example, Japan has one of the lowest incidences of lung cancer in the world despite having one of the highest per capita consumption of cigarettes (Diana, 1993). Similarly, coronary heart disease rates in countries such as Greece and Spain are low despite very high cigarette usage (Fig. 1). This suggests that indigenous factors within countries such as diet may modify the risk of developing smoking-related diseases. Many of the clinical conditions implicated with smoking are also associated with increased indices of free radical-mediated damage to proteins, lipids, and DNA (Duthie and Arthur, 1994), indicating that smoking may exacerbate the initiation and propagation of oxidative stresses, which are potential underlying processes in the pathogenesis of many diseases (Diplock, 1994).

Smokers inhale large amounts of reactive free radicals arising from the combustion of tobacco. The tar in cigarettes contains more than 1017 stable long-lived quinone-semiquinone radicals per gram, which are generated by the oxidation of polycyclic aromatic hydrocarbons during the combustion process. These can reduce oxygen to superoxide and hydrogen peroxide and result in the production of the highly reactive hydroxyl radical. The gas phase smoke contains more than - 1 0 is free radicals per puff of shortlived, reactive carbon- and oxygen-centered peroxy species. These can achieve in the filed polycyclic aromatic hydrocarbons. For example, Japan has one of the lowest incidences of lung cancer in the world despite having one of the highest per capita consumption of cigarettes two intervention trials and 25

## **Quantum therapy as well as Phyton Therapy** Dr. S. SATHISHKUMAR

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

### **3.1 Introduction**

We are at the threshold of the second revolution in our understanding of the role of nutrition in disease and health. The first revolution took place early in this century, with the discovery of the frank nutrient deficiency diseases and their causes. As a result of that research in the nutritional sciences, we essentially eliminated beriberi, pellagra, rickets, and goiter. We did so, incidentally, not by education but by fortification.

Today we are at the threshold of an even greater revolution, and it involves the antioxidant nutrients, including vitamin C, vitamin E, and the carotenoids. Increasingly, research suggests that these nutrients are of great importance in reducing the risk of cancer and heart disease, the two major killers in Western society. However, beyond these diseases, it is increasingly apparent that antioxidants may be important in most of the diseases of aging, including age-related eye diseases such as cataracts, and impaired immune function resulting in increased susceptibility to infection.

Evidence for an important role for antioxidant nutrients comes from the complete spectrum of biomedical research fields, from biochemical research, animal studies, epidemiologic data, and clinical trials. Any one of these alone would be insufficient as a basis for public policy. Oxidation is the transfer of electrons from one atom to another. It is an essential part of normal metabolism. The process of extracting energy from food involves the transfer of electrons, with release of energy at each step, through a series of electron acceptors until studies on the association between vitamin E status and lung cancer risk are considered here. In an intervention trial the investigator randomly assigns vitamin E or placebo to the study population and then waits. The association between vitamin E status and lung cancer risk are considered here. In an intervention trial the investigator randomly assigns vitamin E or placebo to the study population and then waits. The association between vitamin E status and lung cancer risk are considered here. In an intervention trial the investigator randomly assigns vitamin E or placebo to the study population and then waits. The association between vitamin E status and lung cancer risk are considered here. In an intervention trial the investigator randomly assigns vitamin E or placebo to the study population and then waits. The smooth endoplasmic reticulum (microsomes) and the

## **Tissue Engineering and Medical Biopolymer Applications** Dr. A. SUNDARESAN

Department of Biotechnology, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

### 4.1 Introduction

25-Hydroxyvitamin D (25OHD)\* is the first hydroxylated metabolite of vitamin D (D) and the immediate precursor of the fully active and hormonal form of the vitamin,  $1\alpha$ ,25-dihydroxyvitamin D [1,25(OH)2D]. It was discovered by DeLuca and his group, who rapidly identified the liver as the first site of activation of D3 [1–3]. Over the past 35 years, the enzyme systems involved in the C-25 hydroxylation of D3, D2, and several of their analogs have been the object of intense studies by groups in North America, Europe, and Japan. The research has allowed the identification of two intrahepatic organelles, the smooth endoplasmic reticulum (microsomes) and the mitochondrium, as sites possessing fully active but distinct D 25-hydroxylases.

The mitochondrial enzyme has been cloned [4–6] and its identity as a D3 25-hydroxylase established with certainty. Moreover, its presence and activity has been positively identified in all species studied including the human [7]. The microsomal enzyme received the attention of early workers in the field. It has been identified clearly in the pig, where the enzyme has been cloned and clearly shown to hydroxylate D3 and D2 at C-25 [8]. Lately, a new microsomal D 25-hydroxylase species has also been cloned and its gene transcript shown to be present in mice and humans [9]. The latter is also reported to be active in the C-25 hydroxylation of both D3 and D2. In most species, early work has shown that the microsomal D3 25-hydroxylase is an enzyme also active in the oxidation of several endogenous and exogenous substances and, based on the enzyme kinetics of the respective microsomal and mitochondrial enzymes, believed by many to be more physiologically relevant than the mitochondrial entity. In this chapter, we review the most relevant research area on the D 25-hydroxylases and address the specificity and regulation of each enzyme. studies on the association between vitamin E status and lung cancer

### **Transplantation in Medicine** Dr. G. SRITHAR

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

### 5.1 Introduction

A. The Kidney as the Source of Circulating  $1\alpha$ ,25(OH)<sub>2</sub> D3 It is now well accepted that vitamin D is a precursor of the sterol hormone  $1\alpha$ ,25-dihydroxyvitamin D3 [ $1\alpha$ ,25(OH)2D3]. The general pathway of production of  $1\alpha$ ,25(OH)2D3 is shown in Fig. 1. It has been appreciated for some time [1,2] that the kidney is the major site of production of circulating  $1\alpha$ ,25(OH)2D3, although as described later and discussed more thoroughly elsewhere in this volume, many other tissues and cell types have been shown to produce  $1\alpha$ ,25(OH)2D3 from 25-hydroxyvitamin D3 (25OHD3). Within the kidney, it was established early on by microdissection studies that in the fetal rabbit [3] and in the vitamin D–deficient rat [4] and chick [5], the proximal tubules are the region of the most robust activity of the  $1\alpha$ -hydroxylase. With the cloning of the cDNA for the cytochrome P450 component of the  $1\alpha$ -hydroxylase (see Section III) has come the ability to measure its mRNA and protein levels along the nephron. Since these determinations are more sensitive than the measurement of enzyme activity, localization studies can now be carried out under conditions of vitamin D sufficiency and normal mineral status.

Thus in vitamin D–sufficient mice and humans, mRNA and/or protein has been identified by in situ hybridization or immunohistochemical staining in the more distal portions of the nephron along with relatively low expression in the proximal tubules [6,7]. These observations suggest that, while the 1 $\alpha$ hydroxylase occurs throughout the nephron, its regulation varies such that the effects of vitamin D status and abnormal phosphorus metabolism (see Section IV) occur primarily if not exclusively in the proximal tubules. In situ hybridization studies of cultures of embryonic mouse kidneys confirm the presence of 25OHD3 1 $\alpha$ -hydroxylase (CYP1 $\alpha$ ) in tubular epithelium, but not collecting ducts or glomeruli [8]. Along with the demonstration that the kidney contained the enzymatic capability to produce 1 $\alpha$ ,25(OH)2D3 other work was suggesting that this metabolic step was largely confined to this organ. studies on the association between vitamin E status and lung cancer risk are considered here. In an intervention trial the investigator randomly assigns vitamin E or placebo to the study population and then waits its regulation varies such that the effects.

### **Tissue Transplantation** Dr. A. BAKRUDEEN ALI AHMED

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

### 6.1 Introduction

A. Enzyme Function and Regulated Expression Vitamin D is a secosteroid whose biological function is dependent upon its metabolic activation and turnover. These metabolite pathways contain specific hydroxylase enzymes that are members of the cytochrome P450 superfamily of mixed-function monooxygenases. Bioactivation of vitamin D involves the sequential actions of 25-hydroxylase and 1hydroxylase enzymes leading to the synthesis of the hormonally active secosteroid 1,25-dihydroxyvitamin D [1,25(OH)2D]\* (Fig. 1). These two enzymes are discussed in Chapter 4 (vitamin D 25-hydroxylase) and Chapter 5 (25-hydroxyvitamin D 1 $\alpha$ -hydroxylase) and will be mentioned in this chapter only on a comparative basis to 25-hydroxyvitamin D- 24(R)-hydroxylase cytochrome P450c24 (CYP24), the enzyme that directs the side-chain metabolism of 25-hydroxylated vitamin D metabolites, which leads to their terminal physiological inactivation and turnover.

Most cellular actions of vitamin D are mediated through the secosteroid hormone 1,25(OH)2D and involve the transcription of vitamin D-dependent genes. These regulatory processes involve the coordinated modulation and coupling of rapid signal-transduction pathways with slower acting ligand-dependent transcription factors [1]. In both cases, the secosteroid ligand binds to a ligand-specific receptor. The rapid response receptor is located in the cellular membrane of target tissues and initiates rapid signaling responses through a receptor that has been referred to as the membrane-associated receptor membrane-associated rapid response steroid (MSSRS) receptor complex] [2,3]. The hormone receptor for the transcription factor that functions as a VDR:1,25(OH)<sub>2</sub>D heterodimeric complex with the cis-retinoic acid: RXR complex (i.e., VDR-RXR) to regulate vitamin D-dependent genes associated with development and homeostasis (see Chapters 13–17 for details). Cellular and ambient levels 1,25(OH)2D are regulated through the hormone's synthesis and degradation. The cellular membrane of target tissues and initiates rapid signaling rapid signaling responses through a receptor that has been referred to as the membrane of target tissues and initiates rapid signaling and the hormone's synthesis and degradation. The cellular membrane of target tissues and initiates rapid signaling responses through a receptor that has been referred to as the membrane-associated receptor membrane-associated receptor membrane-associated receptor membrane-associated receptor membrane-associated receptor that has been referred to as the membrane-associated receptor membrane-associated rapid response steroid bioactivation of vitamin D involves the sequential actions.

## Chapter 7 Biopolymer properties of Medical components Dr. S. AMBIGA

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

### 7.1 Introduction

Too much selenium (Se) in the diet is detrimental to animal health. The most toxic compounds are Se inorganic salts and Se methionine, which are found in water and plants. However, low soil content results in deficiency syndromes. In humans, clinical manifestations of deficiency are Keshan's disease, a severe cardiomyopathy, Kashin-Beck disease, an osteoarthropathy (1), and cretinism, when associated with an iodine deficiency (2). The eradication of Keshan disease by dietary Se supplementation (3, 4) further strengthened the correlation between low soil content and the disease. Interestingly, however, other complicating factors, such as viruses, have been implicated to explain the seasonal recurrence of this disease (5, 6). To this end, the induction of virulence in certain viruses by selenodeficiency, as documented for the human Coxsackievirus B3, which becomes virulent after the infection of Se-deficient mice and maintains its virulence in normal animals (7-9), is one of the most important aspects of recent selenium research. This observation might provide a rationale for the evidence that different diseases are due to the deficiency of the same element and it may also be relevant for cancer research. Furthermore, in vitro experiments suggest a role for Se in atherosclerosis or aging (10), but convincing epidemiological studies of this aspect are still missing.

Historically, knowledge of the beneficial effects of Se has come from livestock. It is well known that, together with vitamin E, Se supplementation prevents liver necrosis, degeneration of skeletal and cardiac muscles, reduced growth rate, and infertility in cattle (1). This last aspect has been clarified in recent years by well-documented nutritional studies. In selenodeficiency, selenium levels decrease in other organs, but not in testis (11), suggesting that this element may have a peculiar function. Furthermore, with the progression of the deficiency, various degrees of degeneration appear in the seminiferous epithelium, first involving only the mitochondria of spermatids and spermatozoa (12), resulting in a complete disappearance of mature germinal cells (13). This observation might provide a rationale for the evidence that different diseases

# ENZYMES AND THEIR APPLICATIONS

## EDITED BY

## **DR.N.GUNASEKARAN**



### **Enzymes and their applications**

Copyright © 2021 by Dr. N. Gunasekaran

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-050-7



Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## **TABLE OF CONTENTS**

<b>CHAPTER 1</b> <i>INTRODUCTION TO THE</i> <i>BIOCHEMISTRY</i>	Dr. R. KAMARAJ	1
<b>CHAPTER 2</b> <i>USING THE COMPUTER AND</i> <i>INTERNET FOR</i> <i>RESEARCH IN BIOCHEMISTRY</i>	Dr. J. ILLAMATHI	10
<b>CHAPTER 3</b> <i>GENERAL LABORATORY</i> <i>PROCEDURES</i>	Dr. S. SATHISHKUMAR	36
<b>CHAPTER 4</b> <i>CENTRIFUGATION</i> <i>TECHNIQUES IN</i> <i>BIOCHEMISTRY</i>	Mr. G. RAJENDRAN	48
<b>CHAPTER 5</b> <i>PURIFICATION AND ANALYSIS</i> <i>OF BIOMOLECULES</i> <i>BY CHROMATOGRAPHY</i>	Dr. A. SUNDARESAN	64
<b>CHAPTER 6</b> <i>CHARACTERIZATION OF</i> <i>PROTEINS AND NUCLEIC</i> <i>ACIDS BY ELECTROPHORESIS</i>	Dr. G. SRITHAR	90
<b>CHAPTER 7</b> <i>SPECTROSCOPIC ANALYSIS OF</i> <i>BIOMOLECULES</i>	Dr. BINUGEORGE	116

## **INTRODUCTION TO THE BIOCHEMISTRY**

### Dr. R. KAMARAJ

Department of Biotechnology, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

Welcome to the biochemistry laboratory! following reasons: (1) you are enrolled in a formal biochemistry lab course at a col- lege or university and you will use the book as a guide to procedures; or (2) you You are reading this book for one of the have started a research project in biochemistry and desire an understanding of the theories and techniques you will use in the lab; or (3) you have started a job in a biochemistry lab and wish to review theory and techniques. Whether you are a novice or experienced in biochemistry, I believe you will find the subject matter and lab work to be exciting and dynamic. Most of the experimental techniques and skills that you have acquired and mastered in other laboratory courses will be of great value in your work. However, you will be introduced to many new concepts, procedures, and instruments that you have not used in chemistry or biology labs. Your success in biochemistry lab activities will depend on your mastery of these specialized techniques, use of equipment, and understanding of chemical/ biochemical principles.

Information of all kinds may be accessed on computers around the world that are linked together by the Internet. It may be quicker to retrieve information from the Internet than to make a trip to the library. Many of the same works will be found in both, for example Webster's Dictionary http://work.ucsd.edu:5141/cgi-bin/http\_webster. However, computer access is the only method of getting up to date nucleic acid and protein sequence information. Publication of books with compilations of sequences ceased a number of years ago. Journals publish articles that describe the characterization of sequences but may not publish the entire sequence. Instead, they require that the sequence is deposited electronically. Sequences are still distributed on CDROM and can be analysed on computers that are not networked. Using this medium, the information is likely to be 3 months or more out of date. Hence, on-line access to sequence and structure information is the norm and is emphasized in this book.

## Chapter 2 USING THE COMPUTER AND INTERNET FOR RESEARCH IN BIOCHEMISTRY

### Dr. J. ILLAMATHI

Department of Biotechnology, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

Access to the Internet resources is best achieved via a graphical user interface. The World Wide Web (WWW) is geared to this and a browser such as Netscape or Microsoft Explorer is required plus a network service provider and suitable hardware. Some software will require X-Windows that is provided on Unix workstations or as emulations on Windows and Macintosh computers. Even the need for X is eliminated by a system known as Virtual Network Computing (VNC) that is run on the applications server with a client on your own machine. Most universities provide suitable facilities for their members, including undergraduates. There are two main nucleic acid sequence databases and one main protein sequence database in widespread general use amongst the biological community. For nucleic acid these are EMBL (Stoesser et al,^ 1998) and Genbank (Benson et aL, 1998) and for protein this is SWISS-PROT (Bairoch & Apweiler, 1997). There are also many databases that contain special purpose sets of sequences, subsets of sequences derived from the main ones, databases of complete genomes, databases of secondary structure or other derived or additional information and unpublished, private or commercially available sequence databases. Most of these will not be discussed in this chapter but some examples can be found in Table 2.1.

The most common uses of the sequence databases are to search for similarity with an unknown query sequence and to search for entries matching keywords in their annotation. You may already be familiar with using BLAST or FASTA which report alignments of regions of similarity between database entries and your unknown sequence, or with using SRS or Entrez which allow you to find database entries by keyword searches of their annotation. This chapter attempts to introduce the databases so that some of the details of the information they hold will become clearer and potential problems associated with them can

# Chapter 3 GENERAL LABORATORY PROCEDURES

### **Dr. S. SATHISHKUMAR**

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

A paper published in a biochemical journal is a formal way to report research results to colleagues in the international biochemical community. Before writing such a document, one must first determine the journal to which the article will be submitted. There are hundreds of journals that accept manuscripts in the field of biochemistry (see Figure 1.5). Some have very high rank, prestige, and status based on the significance of research results published, reputation of authors, numbers of citations, whether or not manuscripts are peerreviewed, and numbers of readers. Most journals are peer-reviewed, which indicate that before a manuscript is published, it is studied by members of the journal's editorial board to assure that the manuscript is scientifically significant, that it appears to be accurate, and that it is useful and of value to readers of the journal. Some journals accept manuscripts in all areas of biochemistry, but the manuscripts undergo rigorous peer-review by scientists with a certain specialty in the field.

Other reputable journals are more specialized and accept peer-reviewed articles only in certain areas of biochemistry. Perhaps the best advice is to submit the manuscript to the most prestigious journal that has a large audience interested in his or her specialized topic. Publishing a paper in a reputable, peer-reviewed journal offers historic permanence for one's work, status, and exposure as a scientist; however, because the lag time between acceptance and publication. The importance of mutation databases in human research has been grasped only relatively recently. Thus, the reader will gather the impression that the area is in a state of flux at the present time, while guidelines and rules are being established. This chapter will focus on the human mutation databases (owing to the author's expertise), but all living organisms have been subject to specific variation that has been recorded over the centuries. Large and systematic listings of variation at the phenotype level and later at the genotype (mutation) level began in the 1950s with the isolation of auxotropic and other variants of the bacterium Escherichia coli. More recently, this has extended to possibly hundreds of organisms, such as the nematode Caenorhabditis elegans,

## Chapter 4 CENTRIFUGATION TECHNIQUES IN BIOCHEMISTRY

### Mr. G. RAJENDRAN

Department of Biotechnology, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

Acentrifuge of some kind is found in every biochemistry laboratory. Centrifuges have many applications, but they are used primarily for the preparation of biological samples and for the analysis of the physical properties of biomolecules, organelles, and cells. Centrifugation is carried out by spinning a biological sample at a high rate of speed, thus subjecting it to an intense force (artificial gravitational field). Most centrifuge techniques fit into one of two categories—preparative centrifugation or analytical centrifugation. A preparative procedure is one that can be applied to the separation or purification of biological samples (cells, organelles, macromolecules, etc.) by sedimentation.

Analytical procedures are used to measure physical characteristics of biological samples. For example, the purity, size, shape, and density of macromolecules may be defined by centrifugation. In this chapter, we will first explore the principles and theory underlying centrifugation techniques in order to provide a fundamental background. We will then turn to a discussion of the application of these techniques to the isolation and characterization of biological molecules and cellular components.

This equation relates RCF to revolutions per minute of the sample. Equation 4.2 dictates that the RCF on a sample will vary with r, the distance of the sedimenting particles from the axis of rotation (see Figure 4.1). It is convenient to determine RCF by use of the nomogram in Figure 4.2. It should be clear from Figures 4.1 and 4.2 that, since RCF varies with r, it is important to define r for an experimental run. Often an average RCF is determined using a value for r midway between the top and bottom of the sample container. The RCF value is reported as "a number times gravity, g."

## PURIFICATION AND ANALYSIS OF BIOMOLECULES BY CHROMATOGRAPHY

### Dr. A. SUNDARESAN

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

The primary goal of biochemical research is to understand the molecular nature of life processes. The molecular details of a biological process cannot be fully elucidated until the interacting molecules have been isolated and characterized. Therefore, our understanding of the mechanisms of life processes has increased at about the same pace as the development of techniques for the separation and characterization of biomolecules. Chromatography, the most important technique for isolating and purifying biomolecules, was developed by Mikhail Tswett, an Italian-born, Russian botanist. In 1902, Tswett began his studies on the isolation and characterization of the colorful pigments in plant chloroplasts. He prepared separating columns by packing fine powders like sucrose and chalk (calcium carbonate) into long glass tubes. He then poured petroleum ether-derived plant extracts through the columns. As he continued eluting the columns with solvent, he noted the formation of yellow and green zones. Tswett had invented "chromatography," which he might be important in another context.

Which function is used in which place is only apparent defined in his 1906 publication as "a method in which the components of a mixture are separated on an adsorbent in a flowing solvent." In addition to introducing a new technique, Tswett also showed by these experiments that chlorophyll exists in different forms. From such humble beginnings, chromatography has developed into the ultimate tool for not only the isolation and purification, but also for the characterization, of biomolecules. Chromatography, which has now been expanded into multiple forms, continues to be the most effective technique for separating and purifying all types of biomolecules. In addition, it is widely used as an analytical tool to measure biophysical and other quantitative properties of molecules. All types of chromatography are based on a very simple concept: The sample to be examined is allowed to interact with two physically distinct entities—a mobile phase and a stationary phase. he sample most often contains a mixture of several components to be separated.

## Chapter 6 CHARACTERIZATION OF PROTEINS AND NUCLEIC ACIDS BY ELECTROPHORESIS

### Dr. G. SRITHAR

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

Electrophoresis is an analytical tool that allows biochemists to examine the differential movement of charged molecules in an electric field. A. Tiselius, a Swede who invented the technique in the 1930s, performed experiments in free solution that were severely limited by the effects of diffusion and convection currents. Modern electrophoretic techniques use a polymerized gel-like matrix, which is more stable as a support medium. The sample to be analyzed is applied to the medium as a spot or thin band; hence, the term zonal electrophoresis is often used. The migration of molecules is influenced by: (1) the size, shape, charge, and chemical composition of the molecules to be separated; (2) the rigid, mazelike matrix of the gel support; and (3) the applied electric field. Electrophoresis, which is a relatively rapid, inexpensive, and convenient technique, is capable of analyzing and purifying many different types of biomolecules, but is especially effective with proteins and nucleic acids. The newest version of the analytical technique, capillary electrophoresis (CE), provides extremely high resolution and is useful for analysis of both large and small molecules. CE has been found to be especially useful in the analysis of pharmaceuticals. Proteomics, the discipline that attempts systematic, large-scale studies on the structure and function of gene products in an organism or cell, is expanding rapidly because of the availability of electrophoresis (especially two-dimensional techniques) to analyze proteins and peptides.

Even though electrophoresis has been studied for more than 80 years, it has been a challenge to provide an accurate, theoretical description of the electrophoretic movement of molecules in a gel support. However, the lack of theoretical understanding has not hampered growth in the use of the technique in separating and characterizing a wide variety of biomolecules. The charged particle moves at a velocity that depends directly on the electric field (E) and charge (q), but inversely on a counteracting force generated by the

## SPECTROSCOPIC ANALYSIS OF BIOMOLECULES Dr. BINUGEORGE

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

Some of the earliest experimental measurements on biomolecules involved studies of their interactions with electromagnetic radiation of all wavelengths, including X-ray, Ultraviolet–visible, and infrared. It was experimentally observed that when light impinges on solutions or crystals of molecules, at least two distinct processes occur: light scattering and light absorption. Both processes have led to the development of fundamental techniques for characterizing and analyzing biomolecules. We now use the term spectroscopy to label the discipline that studies the interaction of electromagnetic radiation with matter.

Absorption of ultraviolet–visible light by molecules is an especially valuable process for measuring concentration and for molecular structure elucidation. The absorption process is dependent upon two factors: (1) the properties of the radiation (wavelength, energy, etc.), and (2) the structural characteristics of the absorbing molecules (atoms, functional groups, etc.). The interaction of electromagnetic radiation with molecules is a quantum process and described mathematically by quantum mechanics; that is, the radiation is subdivided into discrete energy packets called photons. In addition, molecules have quantized excitation levels and can accept packets of only certain quantities of energy, thus allowing only certain electronic transitions.

With some molecules, the process of absorption is followed by emission of light of a longer wavelength. This process, called fluorescence, depends on molecular structure and environmental factors and assists in the characterization and analysis of biologically significant molecules and dynamic processes occurring between molecules. Nuclear magnetic resonance spectroscopy and mass spectrometry techniques are also now being applied to the study of biological macromolecules and processes. NMR is especially versatile because, in addition to proton spectra, monitoring the presence of and nuclei in

# ENZYMES: BIOCHEMISTRY & BIOTECHNOLOGY

## EDITED BY

## **DR.A.SUNDARESAN**



Enzymes: Biochemistry and Biotechnology

Copyright © 2021 by Dr. A. Sundaresan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-448-2



Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

### CONTENTS

### Chapter 1 The structure of matter

- 1.1 Atoms the building blocks of matter
- 1.2 But what are atoms made of?
- 1.3 Building molecules
- 1.4 The states of matter.

#### Chapter 2 Molecules in motion

- 2.1 The ideal gas: a simple model
- 2.2 Temperature and thermometers
- 2.3 The equation of state

### Chapter 3 Chemical Reactions

- 3.1 Work, Heat, and Energy
- 3.2 Energy changes in reactions

### Chapter 4 A deeper look at the ideal gas

- 4.1 Back to basics: the laws of large numbers
- 4.2 Probability densities
- 4.3 Maxwell's distribution law
- 4.4 What else can we get from Maxwell's law?

#### Chapter 5 The Second Law

- 5.1 When the First Law is not enough what then?
- 5.2 Reversible changes: an example
- 5.3 Some properties of the entropy
- 5.4 Some other useful state functions

### Chapter 6 Statistical interpretation of the Second Law

- 6.1 Order and disorder spontaneous change Will be Sect 6.1
- 6.2 Reversible changes: an example
- 6.3 Making contact with thermodynamics
- 6.4 But how does the *temperature* come in?
- 6.5 The partition function; another route to thermodynamics
- 6.6 Thermodynamic functions: the second route

### Chapter 7 Partition functions and properties

- 7.1 Getting started
- 7.2 Finding the PFs: the last step
- 7.3 Properties of gases and crystals

## The structure of matter

### 1.1 Atoms – the building blocks of matter

Have you ever wondered what everything around you is *made of* - stone, metal, air and water, even living creatures? People started asking that question thousands of years ago. Nowadays we can give an answer: everything is built up from **atoms** and in our world we find less than a hundred different kinds of atom. Atoms of any one kind form a **chemical element** and when these elements are put together, in countless different ways, they give us all known forms of **matter**.

Each kind of atom has a definite **mass** (see Ch.1 of Book 4) and an approximate 'size' (approximate because the atoms are a bit 'soft' and may get 'squashed' by the other atoms around them). They have been given names and symbols (standing for their names): the atom of smallest mass is called Hydrogen, with symbol H; the next, in order of increasing mass, is Helium (He) and the next is Lithium (Li). Figure 1 will give a rough idea of the relative sizes of the first ten atoms, showing also their names and chemical symbols:



Figure 1. Relative sizes of the 10 lightest atoms

The atoms shown in colour (mainly, in fact, H, C, N, O) are the ones that make up more than 99 percent of your total body mass: they are the atoms that are vital to an understanding of the **Life Sciences** (like Biology and Biochemistry). And the science built mainly around these elements is **Organic Chemistry**. All the rest used to be called **Inorganic Chemistry**, but nowadays we don't make such sharp separations between the different areas of the subject. Chemistry is about atoms, how they can be built into more

## Molecules in motion

### 2.1 The ideal gas: a simple model

The simplest system to start from is a 'box' containing only molecules, usually far apart and moving in all directions at high speed. Suppose, for simplicity, that each molecule consists of only one atom (i.e. is "monatomic") and is therefore just a single particle, a **point mass**, which moves according to the laws of physics (Book 4). Suppose also that the

box has a movable wall, as in Fig.10(a), so that its volume can be changed by moving the wall.



That's what happens in a **pump**, where the movable wall is called a **piston** and may have a handle – to which you can apply a **force** F. If the piston has a surface area A, the force per unit area F/A is the **pressure** exerted by the gas: we'll denote the pressure by P and the volume of the box by V. (If you've forgotten all about volumes and forces and the laws of physics, you'd better look back at Books 2 and 4 as you're going to need them!)

Most things that go on in Chemistry depend on pressure and volume, so we want to know how P and V are related and how they depend on molecular motion. In Book 4 we were usually dealing with only one or two particles at a time and we could find how they moved, when forces acted on them, by using Newton's laws: but now we're talking about a container in which there are millions and millions of particles! So we can only use general principles and think about what goes on "on the average": for example, there may be n particles per unit volume and that number probably won't change much as we go from one element of volume to another – the **average** number per unit volume will be everywhere  $n \approx N/V$ , where N is the *total* number of molecules in the container. In the same way we'll suppose that they all move with about the same average speed. We

## **Chemical Reactions**

### **3.1** Work, Heat, and Energy

In Chapter 1 we started thinking about the structure of matter and how everything can be built up out of atoms; first molecules – usually with just a small number of atoms – and then gases, liquids and solids – substances containing vast numbers of atoms and molecules, big enough to see and study in the laboratory. We went on, in Chapter 2, to study a very simple 'model' of a gas of N atoms in a container of volume V, introducing the Kelvin temperature T as a measure of the 'hotness' of the system. The **state** of the system could then be defined by giving the values of any *two* of the three variables P, V, T. If you know, say, V and T, then the pressure exerted by the gas on the walls of the container is determined as a function P = P(V,T): pressure, volume, and temperature are related by an **equation of state**, f(P, V, T) = 0. For an ideal gas, this equation is very simple: PV = RT, for one mole of gas, or PV = nRT for n moles. For other systems, like a mixture of gases, the equation may be more complicated and you will have to say how many moles there are of substance A, of substance B, and so on; so you'll need more variables  $(n_A, n_B, \text{ etc.})$ . But for now we'll just be talking about a system in which such variables are not changing and needn't be considered.

Whatever system we're thinking about, however, must contain **energy**, as well as atoms and molecules, and in the last Chapter we extended the idea of energy (which you first met in Physics, in Book 4) to include **thermal energy** – which has to do with the motion of the atoms and molecules, not included in the 'classical' Physics of two centuries ago because nobody knew about them! In Chemistry, on the other hand, we have to deal with systems in which substances are constantly changing as their molecules change in **chemical reactions**. So to understand what's going on we *must* take into account changes of thermal energy when the *composition* of the system is also changing.

In the last Chapter the kinetic energy (KE) of the molecules in the container was denoted by U and was also a function of state: U = U(V,T) if volume and temperature are chosen as independent variables. We noted that U could be changed in two ways; firstly by doing work on the system (squeezing it into a smaller volume) and secondly by *heating it*. In the first way, force was applied to the piston and the gas was compressed. The work done

## A deeper look at the ideal gas

#### A note to the reader

Some parts of this Chapter, and others that follow, are difficult; but don't be put off – on first reading don't try to understand everything. You'll find new and fairly simple ideas keep coming in and as you go further you'll see how they all start coming together. You're going to use a lot of things from Books 1-4: so go back to them whenever you get stuck. If you do so you'll soon be in good shape for going on to the end of the book.

### 4.1 Back to basics: the laws of large numbers

We've got used to the idea that all the things we meet in daily life are built up from vast numbers of tiny particles – atoms and molecules – and that the numbers involved are so enormous that we have to write them in 'exponential form'. The number of particles in 1 mole of matter (Section 1.1), a quantity you can actually see and weigh out in the laboratory (or the kitchen!), is of the order  $10^{24}$  (24 being the 'exponent'– the power to which you have to raise the 'base' 10 to get the number you want). Talking about "ten to the 24" is certainly much easier than saying "a million million million million" every time you use it. So we expect chemistry to be full of *exponentials*,  $e^n$  when we use the 'natural' base  $e \approx 2.718$  instead of 10, and indeed it is.

Another way of putting this (look back at Book 1, Chapters 4, 5) is to say that if  $N = 10^{24}$  then  $24 = \log_{10} N$ : the enormous number N has the *logarithm* 24. (Here we've shown the base (10) only because the *natural* base e is the one we usually have in mind when we talk about exponentials and logarithms: when the base is not shown you can be sure it's e). In general, remember, if  $y = e^x = \exp x$  then  $x = \log y$  – the logarithm is the *inverse* of the exponential function (Book 3, Section 1.4).

When we come down to 'basics' a large part of chemistry depends very directly on the properties of the logarithmic function, as we'll see in later Chapters. But whenever we're dealing with large numbers of any kind – of people or of particles – we need new ideas. We start talking about the "chance" of something happening, or the "probability" of finding this or that when we observe something, and we may say that " on the average" this

## The Second Law

### 5.1 When the First Law is not enough – what then?

In the last Chapter we extended the idea of potential energy (which you first met in Physics (Book 4) to include **thermal energy** and its transfer from one system to another. This is the subject of **thermodynamics**. And the First Law of Thermodynamics, formulated in Section 3.1, simply stated that heat is a form of energy very similar to 'mechanical' potential energy, which can be *stored* in a system and depends only on the *state* of the system (described by variables such as pressure, volume, temperature, composition) and not on how that state is reached. If an amount of work w is done on the system, i.e. put into it, along with an amount of heat q, then its internal energy increases by  $\Delta U = w + q$ . The First Law is so general that it applies to any kind of change; and in Chapter 3 we even admitted chemical reactions in which the composition of the system was changed, going from 'reactants' (in the initial state) to 'products' (in the final state). If such changes take place at constant volume,  $w = P\Delta V = 0$  and q measures directly the change of internal energy in the reaction,  $\Delta U = q$ ; and if  $\Delta U$  is negative, then q is also negative – meaning that heat flows out of the system. But more usually reactions take place at constant pressure, and then the heat of reaction is not  $\Delta U$  but instead  $\Delta H = \Delta U + P \Delta V$ . It was therefore found useful to consider the heat content (or 'enthalpy') instead of  $\Delta U$ , thus allowing for changes of volume during the reaction. The heat content

$$H = U + PV \tag{5.1}$$

is clearly another function of state.

You might think the heat of reaction  $\Delta H$  would be a good measure of the **affinity** of the reactants (how much they 'like' each other), a large negative value meaning they would come together with explosive violence (as in the hydrogen-oxygen reaction on p.xxx). But many reactions take place even when  $\Delta H$  is slightly *positive*. How can this be explained? To do so we need a quite new idea, related to the idea of the *probability* of something happening, which we first met in the last section: perhaps it could be that the molecules of the products have a higher probability of being found than those of the reactants. We'll look at this possibility in a later Section, but first we'll use a simple example (again the

# Statistical interpretation of the Second Law

### 6.1 Order and disorder – spontaneous change

Perhaps someone has printed out for you the first ten pages of the last chapter and you have them in a nice neat pile on the table, Page 1 at the top, Page 2 underneath it, going down to Page 10 at the bottom. We say the 'system' is 'perfectly ordered' with every part in its proper place.

But then somebody opens the door and the wind blows in, scattering the pages all over the room. The wind spoils the perfect order: when you pick up the pages and put them back on the table the pages will be 'disordered' – the last page you picked up (which will go on top of the pile) might be 5, and the one before it could be 2, and so on, down to perhaps 1 (at the bottom, because it was the title-page and you picked it up first). How many disordered states of the system, the heap of pages, is it possible to find? In fact it is  $N! = 1 \times 2 \times 3 \times 4 \dots \times 10$ , called **factorial N**, as you discovered near the beginning of Book 1. And this comes out to be *over three and a half million*! So it will take you some time to put them back into the single state of perfect order. Without the page numbers to help you it would take you much longer.

When a system is 'disturbed' in some way it usually ends in a state where the 'order' is reduced – or the 'disorder' is increased. After the wind has scattered the pages it's no good opening the door again and expecting the wind to put them back in the right order! This suggests that the degree of disorder in the state of a system might be *measured* numerically by the *number of ways* of getting it into that condition. In the present chapter we try to develop this idea and apply it to a *thermodynamic* system; and to do so we start by recalling what we know about probabilities.

Suppose you're talking about the **probability** of something happening and that you want to measure it by giving it a *number*, e.g. 0.5 or 1/2. We used this idea in Sect 4.1, when we were talking about throwing dice or throwing darts and noting the results. In general, we are making an 'experiment' or an *observation* of something (an 'event') that might happen. For example, the die may fall with the 3 uppermost, or the dart may land at a

## Partition functions and properties

### 7.1 Getting started

'Real' chemistry still seems to be a long way off! What good is all that theory if it doesn't help us to do anything? We know that the partition function gives us (in theory!) a way of finding the thermodynamic functions we need in discussing chemical reactions; but how do we get them? That's the problem we now have to face.

The only systems for which the PFs can be obtained easily are those which consist of *independent particles*: this means that the total energy of N such particles is simply the sum of their separate energies. Systems of this kind include, in good approximation, most gases, gas mixtures, and many types of crystal. If there are  $n_1$  particles with energy  $\epsilon_1$ ,  $n_2$  with energy  $\epsilon_2$ , and so on, then we can take

$$E = n_1 \epsilon_1 + n_2 \epsilon_2 + \dots + n_i \epsilon_i + \dots + \sum_i n_i \epsilon_i$$
(7.1)

as our starting point. The next step is to use the fact that, for a system whose energy is the *sum* of the energies of its parts, the PF is simply a *product* of the PFs of the separate parts. But the counting of the number of *distinct states*  $g_E$ , for any given total energy E is not so easy. To continue, we'll study two important special cases: (i) the 'Einstein crystal', and (ii) the ideal gas.

#### (i) The Einstein crystal

In the simplest 'model' of a crystal, first used by Einstein, the particles are taken to be single atoms, each vibrating, with its own particular energy, around a point on a **crystal lattice**. If there are  $n_1$  particles with energy  $\epsilon_1$ ,  $n_2$  with energy  $\epsilon_2$ , and so on, then we know E is given by (??); but we also need  $g_E$ , the number of different states of that energy.

Suppose we use  $\bullet^1$ ,  $\bullet^2$  to denote particles with energies  $\epsilon_1$ ,  $\epsilon_2$ , respectively, and look at a bit of crystal with 5 atoms vibrating with energy  $\epsilon_1$  and 4 with energy  $\epsilon_2$ . Then we must count

# ESSENTIALS OF MOLECULAR BIOLOGY

# EDITED BY

**DR.G.SRITHAR** 



### **Essentials of Molecular Biology**

Copyright © 2021 by Dr. G. Srithar

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-098-9



Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

<b>CHAPTER 1</b> Agrobacterium: a disease-causing bacterium	Mr. G. RAJENDRAN	1
<b>CHAPTER 2</b> A brief history of research on Agrobacterium tumefaciens	Dr. A. SUNDARESAN	10
<b>CHAPTER 3</b> Agrobacterium and plant biotechnology	Dr. S. SATHISHKUMAR	36
<b>CHAPTER 4</b> <i>The Agrobacterium tumefaciens</i> <i>C58 genome</i>	Dr. R. KAMARAJ	48
<b>CHAPTER 5</b> Agrobacterium—taxonomy of plant-pathogenic Rhizobium species	Dr. J. ILLAMATHI	64
<b>CHAPTER 6</b> <i>The initial steps in Agrobacterium</i> <i>tumefaciens pathogenesis: chemical</i> <i>biology of host recognition</i>	Dr. BINUGEORGE	90
<b>CHAPTER 7</b> Agrobacterium-host attachment and biofilm formation	Dr. G. SRITHAR	116
## **Agrobacterium: a disease-causing bacterium**

#### Mr. G. RAJENDRAN

Department of Biotechnology, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

#### **1.1 Strain classification**

Early taxonomy distinguished agrobacteria on the basis of their pathogenic properties. Thus strains causing crown gall were classified as A. tumefaciens, those inducing cane gall on raspberry (Rubus idaeus) were described as A. rubi and hairy root-inducing isolates were allocated to A. rhizogenes. Non-pathogenic strains were called A. radiobacter (Allen and Holding, 1974). Later, strains were identified on the basis of their biochemical and physiological properties which led to the definition of three biotypes (Kerr and Panagopoulos, 1977; Süle, 1978). Species- and biotypebased taxonomies do not coincide (Kersters and De Ley, 1984). Biotype 3 strains were isolated almost exclusively from grapevine (Vitis vinifera) and allocated to A. vitis (Ophel and Kerr, 1990). Similarly, several isolates from weeping fig (Ficus benjamina) form a distinct group and were classified as A. larrymoorei (Bouzar and Jones, 2001).

#### **1.2 The infection process**

During the infection process a segment of the Ti (tumor-inducing) plasmid, called T(transferred)-DNA, is exported from Agrobacterium to the plant cell nucleus where it is integrated into the chromosomal DNA and expressed. Hairy root is caused in a similar way by a root-inducing or Ri plasmid. The T-DNA transfer and integration processes involve a large number of bacterial and host factors, and finally results in genetically transformed plant cells. Details of this unique natural example of interkingdom DNA transfer have been reviewed (Zhu et al., 2000; Zupan et al., 2000; Gelvin, 2003; Tzfira et al., 2004 and other chapters in this book). During the infection process agrobacteria suppress plant defense mechanisms via the chromosomally encoded degradation of hydrogen peroxide (Xu and Pan, 2000) and by Ti plasmid-related functions. Transformation of plant cells results in elevated hormone (auxin and cytokinin) production and sensitivity. Both trigger abnormal proliferation leading to tumorous growth or abnormal rooting (Petersen et al., 1989; Gaudin et al., 1994; Costacurta and Vanderleyden, 1995). Tumors and hairy roots produce and

# A brief history of research on *Agrobacterium* tumefaciens

#### Dr. A. SUNDARESAN

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

#### 2.1 Early studies

As described in the first chapter of this volume, the "crown gall" disease of higher plants was a particular problem in orchards and vineyards, though a wide variety of plants were known to develop distinct 'galls'. The earliest work identifying bacteria as the cause of these galls, in contrast to the then known limited galls produced as a result of insect or nematode infection, was published by Cavara (1987) who isolated 'white bacteria' that would give rise to galls when inoculated on plants. A much more thorough (and apparently independent – see Braun, 1982) characterization of the causal agent of the crown gall disease was published by Smith and Townsend (1907) in which many of the characteristics of the inciting bacterium (named then as Bacterium tumefaciens) were described including its rod shape, size, polar flagella and inability to grow well at 37°C ('blood temperature'). The debate over the nomenclature of Agrobacterium species still exist (Box 2-1 and Chapter 5), and for simplicity, I will refer to Agrobacterium tumefaciens as the causal agent of the hairy root disease throughout the course of this chapter.

Through the next thirty years studies on the crown gall disease described the responses of many plants to various different field isolates, generally concurring with the observations of Smith and Townsend. Of particular interest amongst these early papers were the descriptions by Smith (1916) and later Levin and Levine (Levin and Levine, 1918; Levine, 1919) of 'teratomas' – spontaneously shoot forming tumors – that could be isolated on certain plants by certain bacterial isolates (see below). Nevertheless, despite a good deal of speculation about the relationship of crown gall tumors of plants to neoplasias of animals, no particular insights into the mechanism whereby A. tumefaciens might be inducing tumors were developed. The prospects for progress improved as physiological and genetic tools in both

## Agrobacterium and plant biotechnology

#### Dr. S. SATHISHKUMAR

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

#### **1 INTRODUCTION**

Plant biotechnology has had a dramatic impact on agriculture, and on public awareness of the role of the private sector in industrial-scale farming in developed countries. This chapter focuses on the seminal contributions of Agrobacterium tumefaciens to this technological revolution, and on the applications of genetic engineering that continue to expand the limits of plant productivity. Agrobacterium-mediated transformation has yielded a stunning array of transgenic plants with novel properties ranging from enhanced agronomic performance, nutritional content, and disease resistance to the production of pharmaceuticals and industrially important compounds. Many of these advances have been made possible by creative and elegant methodological innovations that have enabled gene stacking, targeted mutagenesis, and the transformation of previously recalcitrant hosts.

Transgenic plants are not a panacea for global food shortages, distributional failures, or other structural causes of poverty. They can, however, have a positive impact on both human and environmental health. Agricultural biotechnology's image has been tarnished by the perception that it fails to address the needs of the world's hungry, and indeed most of the commercial products to date represent technology that is inappropriate for subsistence farmers (Huang et al., 2002a). As this chapter documents, there is ample potential for genetically modified plants to ameliorate some of the constraints faced by resource-poor farmers. Even modest enhancements of agronomic traits have the potential to help farmers overcome endemic problems such as lack of food security, limited purchasing power, and inadequate access to balanced nutritional resources (Leisinger, 1999).

Many of these innovations will come from public sector research, and the vast majority of the applications described herein have in fact emanated from basic investigations and collaborative product-oriented research originating in the non-profit realm. As plant biotechnology research moves forward and outward to include more stakeholders in

## The Agrobacterium tumefaciens C58 genome

#### Dr. R. KAMARAJ

Department of Biotechnology, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

#### **1 INTRODUCTION**

In 2001 the journal Science published two papers back-to-back on the genome of the Agrobacterium biovar I organism A. tumefaciens C58 (Goodner et al., 2001; Wood et al., 2001). Two different teams of scientists had raced to complete and publish this genome, only becoming aware of the other's efforts near the end of the projects. After contacting each other, and thanks to the vision of Science editors, both teams were able to publish their results simultaneously. An interesting account of this race was published several years later in Nature Biotechnology (Harvey and McMeekin, 2004). The principle members of both groups have now combined efforts and, in addition to authoring this chapter, have completed the genome sequences of representative Agrobacterium strains from biovars II and III (Wood D, Burr T, Farrand S, Goldman B, Nester E, Setubal J and Slater S, unpublished data).

The two original Science papers, although covering a lot of common ground, were surprisingly complementary. Over 250 manuscripts have used the data from the original C58 genome sequences. The types of manuscripts fall into three basic categories: (i) those that use the sequence as part of genome-scale comparative analyses, (ii) those that simply cite the identification of an ortholog of their gene of interest in A. tumefaciens, and (iii) those that follow-up on specific genes in A. tumefaciens after identifying them in the genome sequence. The last category contains about 20% of these manuscripts. Here we present a description of the C58 genome that combines the findings of both teams, and summarizes many new results on A. tumefaciens biology that have been enabled by the A. tumefaciens C58 genome sequence. Table 4-1 lists all genes discussed herein and their designations by the original genome publications (Goodner et al., 2001; Wood et al., 2001). To harmonize nomenclature as we continue our annotation of the Agrobacterium genomes, we have chosen to use the gene designations and style of Wood et al. (2001). The 5.67-Mb

# Agrobacterium—taxonomy of plant-pathogenic Rhizobium species

#### Dr. J. ILLAMATHI

Department of Biotechnology, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

#### **1 INTRODUCTION**

The classification of bacteria at generic and specific levels has been subject to repeated amendment, with frequent revisions made to keep nomenclature in line with contemporary taxonomic approaches. The genus Agrobacterium Conn 1942 is an exception. Although they had their origins in diverse genera, the plant pathogenic bacteria associated with oncogenic symptoms, commonly called 'crown gall' and 'hairy root', and other more recently identified oncogenic pathogens, have been recognized as distinct species in the genus Agrobacterium since the genus was established (Kersters and De Ley, 1984).

Classification of the genus Agrobacterium and of its species has been based on its once-puzzling oncogene pathogenicity, which was the defining character of the genus (Kersters and De Ley, 1984). This was paralleled in the genus Rhizobium Frank 1889, originally reserved for bacteria with the capacity to form symbiotic nitrogen-fixing symbioses with legume species. For both genera, their distinctive generic characteristics are now known to be the result of the presence or absence of interchangeable conjugative plasmids that confer specific oncogenic or nodulating capabilities. However, a character that is the result of arbitrary acquisition or loss of a plasmid is obviously unstable and cannot form the basis of formal nomenclature. Although comparative phenotypic and genetic studies of Agrobacterium spp. and Rhizobium spp. have failed to confirm differentiation into separate genera based on oncogenicity and nitrogen-fixation respectively (Young et al., 2001), an element of the bacteriological community has continued to support a special-purpose nomenclature based on pathogenicity alone.

Pathogenicity was also used as the single defining character of individual Agrobacterium species (Kersters and De Ley, 1984) although, following comprehensive genetic and phenotypic studies, the genus has been revised with the recognition of natural species (Holmes and Roberts, 1981) to

# The initial steps in Agrobacterium tumefaciens pathogenesis: chemical biology of host recognition

#### **Dr. BINUGEORGE**

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

#### **1 INTRODUCTION**

Transformation of plants by wild type strains of Agrobacterium tumefaciens results from the transfer of the Ti plasmid's T-DNA into host cells where it is ultimately integrated into chromosomal DNA and expressed (see other chapters in this volume). The virulence (vir) genes of the Ti plasmid required for virulence (Klee et al., 1983; Stachel and Nester, 1986) encode, for example, proteins involved in the processing, transport and ultimate integration of the T-DNA in the host (see other chapters). The resultant 'crown gall' tumors potentially yield great benefits to the infecting bacteria in the form of opines produced via enzymes encoded on the TDNA (De Greve et al., 1982), yet the process requires significant energy expenditures by the bacterium and, accordingly, should be tightly regulated. In agreement with this hypothesis is the finding that the virulence genes are essentially silent unless the bacteria are exposed to a plant or plant derived molecules (Stachel et al., 1985b; Stachel et al., 1986). Activation of the genes in response to the host or host derived signals was first shown via experiments exploiting vir::lacZ fusions (Stachel et al., 1985a), and further experiments, importantly, showed that two virulence proteins encoded on the Ti plasmid, VirA and VirG, were required for the hostinduced expression of the vir genes (Stachel and Zambryski, 1986; Engstrom et al., 1987; Winans et al., 1988).

Early studies of VirA and VirG demonstrated that they were related to the just discovered class of bacterial regulatory 'two component' systems (TCS) (Winans, 1991; Charles et al., 1992). TCS are comprised, minimally, of a histidine autokinase (often called sensor kinase) that responds, either directly or indirectly to environmental input, and a response regulator that is phosphorylated by its cognate histidine kinase (Robinson et al., 2000; Stock et al., 2000; West and Stock, 2001). Often, but not exclusively, the response regulator controls transcription of sets of genes via binding to specific regions of promoters and recruiting the

## Agrobacterium-host attachment and biofilm formation

#### Dr. G. SRITHAR

Department of Biochemistry, School of Arts and Sciences, PRIST Deemed University, Thanjavur, Tamil Nadu, India.

#### **1 INTRODUCTION**

Agrobacterium tumefaciens attaches to plant tissues during initial stages of crown gall pathogenesis and this physical interaction is required for subsequent DNA transfer (Lippincott and Lippincott, 1969). Adherent A. tumefaciens can accumulate on these plant tissue surfaces to form aggregates and biofilms, and similarly adhere to abiotic surfaces in the terrestrial environment (Figure 7-1). This chapter will address the process of attachment and subsequent biofilm formation by A. tumefaciens, examine the molecular requirements for these processes, and consider their impact on plant disease.

Bacterial attachment to host tissues is an obligatory first step to disease progression for many plant and animal pathogens. Host binding and recognition has been intensively explored for several mammalian pathogens. In these systems, attachment can be highly specific, often mediated through receptors that decorate the exterior of host cells and the extracellular matrix (Boyle and Finlay, 2003; Pizarro-Cerda and Cossart, 2006). For several well-studied systems, such as enteropathogenic Escherichia coli (EPEC), host association is comprised of multiple steps, initiating with surface engagement and followed by a more intimate interaction in which receptors on the target cells are recognized and tightly complexed (Nougayrede et al., 2003). Binding of these receptors by infecting pathogens often causes profound alterations in cytoskeletal elements, disruption of internal signaling pathways or uptake of the bacteria into the target cells. Adhesins are cell surface structures produced by the infecting bacteria that engage host cells, often via specific receptors, and promote intimate association of the pathogen with its target cells.

In mammalian pathogens, adhesins are strictly defined as those cell surface structures including pili, flagella, or other surface proteins, that directly engage host receptors (Nougayrede et al., 2003). Other surface structures may act to promote physical interaction



# TEXT BOOK OF

# EDITED BY





Text Book of Immunology

Copyright © 2021 by Dr. S. Sathishkumar

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-978-4



Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

<b>Chapter 1</b> Molecules, Cells, & Tissue of Immunity	Dr. A. SOHNA CHANDRA PACKIAVATHI	1
<b>Chapter 2</b> Antigens, Immunogens, Vaccines & Immunization	Dr. ABAKRUDEEN ALI AHMED	34
<b>Chapter 3</b> Cluster of Differentiation (CD) Antigens	Dr. S. AMBIGA	63
<b>Chapter 4</b> The HLA Dictionary, 2001	Dr. M. VIJAY	75
<b>Chapter 5</b> Nomenclature for Factors of HLA System 2002	Dr. BINUGEORGE	84
<b>Chapter 6</b> Nomenclature for factors of Dog Major Histocompatibility System (DLA), 2000	Dr. S. SATHISHKUMAR	105
<b>Chapter 7</b> Antigen Presentation	Dr. A. SUNDARESAN	128

# Molecules, Cells, and Tissues of Immunity



#### NATURAL AND ADAPTIVE IMMUNITY

#### ORGANS AND TISSUES OF THE IMMUNE RESPONSE

#### ADHESION MOLECULES

Adhesion molecules mediate cell adhesion to their surroundings and to neighboring cells. In the immune system, adhesion molecules are critical to most aspects of leukocyte function, including lymphocyte recirculation through lymphoid organs, leukocyte recruitment into inflammatory sites, antigen-specific recognition and wound healing. There are five principal structural families of adhesion molecules:

- Selectins
- Integrins
- Immunoglobulin superfamily (IgSF) proteins
- Cadherins
- Mucins

Classification of these major adhesion molecules and their structures and functions are summarized in Table 1.1.

#### Selectins

Selectins are a group of cell adhesion molecules that are glycoproteins and play an important role in the relationship of circulating cells to the endothelium. The members of this surface molecule family have three separate structural motifs. They have a single N-terminal (extracellular) lectin motif preceding a single epidermal growth factor repeat and various short consensus repeat homology units. They are involved in lymphocyte migration. These carbohydrate-binding proteins facilitate adhesion of leukocytes to endothelial cells. There is a single chain transmembrane glycoprotein in each of the selectin molecules with a similar modular structure, that includes an extracellular calciumdependent lectin domain. There are three separate groups of selectins:

- L-selectin (CD62L), expressed on leukocytes
- P-selectin (CD62P), expressed on platelets and activated endothelium

• E-selectin (CD62E), expressed on activated endothelium

Under shear forces their characteristic structural motif is comprised of an N-terminal lectin domain, a domain with homology to epidermal growth factor (EGF) and various complement regulatory protein repeat sequences.

Characteristics, receptors/ligands, cellular affinities, distribution, function, and other related data such as the expression and regulation of selectins are summarized in **Table 1.2**.

#### Integrins

Integrins are a family of cell membrane glycoproteins that are heterodimers comprised of  $\alpha$  and  $\beta$  chain subunits. They serve as extracellular matrix glycoprotein receptors. They identify the RGD sequence of the  $\beta$  subunit, which consists of the arginine-glycine-aspartic acid tripeptide that occasionally also includes serine. The RGD sequence serves as a receptor recognition signal. Extracellular matrix glycoproteins, for which integrins serve as receptors, include fibronectin, C3, and lymphocyte function-associated antigen 1 (LFA-1), among other proteins. Differences in the  $\beta$  chain serve as the basis for division of integrins into three categories. Each category has distinctive  $\alpha$  chains. The  $\beta$  chain provides specificity. The same 95-kD  $\beta$  chain is found in one category of integrins that includes lymphocyte function-associated antigen 1 (LFA-1), p150, 95, and complement receptor 3 (CR3). The same 130-kD  $\beta$  chain is shared among VLA-1, VLA-2, VLA-3, VLA-4, VLA-5, VLA-6, and integrins found in chickens. A 110-kD  $\beta$ chain is shared in common by another category that includes the vitronectin receptor and platelet glycoprotein IIb/IIIa. There are four repeats of 40 amino acid residues in the  $\beta$  chain extracellular domains. There are 45 amino acid residues in the  $\beta$  chain intracellular domains. The principal function of integrins is to link the cytoskeleton to extracel-

# ANTIGENS, IMMUNOGENS, VACCINES, AND IMMUNIZATION

# ANTIGENS VACCINE

IMMUNIZATION

#### ANTIGENS

#### Immunogen

The 'traditional' definition of antigen is a substance that may stimulate B and/or T cell limbs of the immune response and react with the products of that response, including immunoglobulin antibodies, and/or specific receptors on T cells. This 'traditional' definition of antigen more correctly refers to an immunogen. A complete antigen is one that both induces an immune response and reacts with the products of it, whereas an incomplete antigen or hapten is unable to induce an immune response alone, but is able to react with the products of it, e.g., antibodies. **Table 2.1** compares the characteristics of immunogen and hapten.

Presently, antigen is considered to be one of many kinds of substances with which an antibody molecule or T cell receptor may bind. These include sugars, lipids, intermediary metabolites, autocoids, hormones, complex carbohydrates, phospholipids, nucleic acids, and proteins.

#### Antigenic determinant

An antigenic determinant is the site of an antigen molecule that is termed an epitope and interacts with the specific antigen-binding site in the variable region of an antibody molecule known as a paratope. The excellent fit between epitope and paratope is based on their three-dimensional interaction and noncovalent union. An antigenic determinant or epitope may also react with a T cell receptor for which it is specific. A lone antigen molecule may have several different epitopes available for reaction with antibody or T cell receptors. There are two types of antigenic determinants: conformational determinants and linear (sequential) determinants. The characteristics of these two types of epitopes, including their location, composition, antigenantibody reaction and availability, are listed in **Table 2.2**. Antigenic determinants are also categorized in Table 2.3 according to B and T cell recognition.

#### Thymus-dependent antigen

A thymus-dependent antigen is an immunogen that requires T cell cooperation with B cells to synthesize specific antibodies. Presentation of thymus-dependent antigen to T cells must be in the context of MHC class II molecules. Thymusdependent antigens include proteins, polypeptides, haptencarrier complexes, erythrocytes, and many other antigens that have diverse epitopes. T-dependent antigens contain some epitopes that T cells recognize and others that B cells identify. T cells produce cytokines and cell surface molecules that induce B cell growth and differentiation into antibody-secreting cells. Humoral immune responses to T-dependent antigens are associated with isotype switching, affinity maturation, and memory. The response to thymus-dependent antigens shows only minor heavy chain isotype switching or affinity maturation, both of which require helper T cell signals.

#### Thymus-independent antigen

Thymus-independent antigen is an immunogen that can stimulate B cells to synthesize antibodies without participation by T cells. These antigens are less complex than are thymus-dependent antigens. They are often polysaccharides that contain repeating epitopes or lipopolysaccharides derived from Gram-negative microorganisms. Thymusindependent antigens induce IgM synthesis by B cells without cooperation by T cells. They also do not stimulate immunological memory. Murine thymus-independent antigens are classified as either TI-1 or TI-2 antigens. Lipopolysaccharide (LPS), which activates murine B cells without participation by T or other cells, is a typical TI-1 antigen. Low concentrations of LPS stimulate synthesis of specific antibody, whereas high concentrations activate

# CLUSTER OF DIFFERENTIATION (CD) ANTIGENS

#### MOUSE CD ANTIGENS

#### • HUMAN CD ANTIGENS

Leukocytes express distinct assortments of molecules on their cell surfaces, many of which reflect either different stages of their lineage-specific differentiation or different states of activation or inactivation. These cell surface molecules of leukocytes are routinely detected with anti-leukocyte monoclonal antibodies. Clusters of antigens on the surface of leukocytes can be designated by their reactions with monoclonal antibodies. This designation of the antigens is called clusters of differentiation (CDs). Using different combinations of mAbs, it is possible to chart the cell surface immunophenotypes of different leukocyte subpopulations, including the functionally distinct mature cell subpopulations of B cells, helper T cells (TH), cytotoxic T cells (TC), and natural killer (NK) cells. Some CD antigens have a well-known function, but other CD antigens have no known function.

#### **MOUSE CD ANTIGENS**

Mouse CD antigens are listed in Table 3.1. Their gene, molecular weight, ligands, distribution, and functions are

shown in the table. For reference, alternate names of mouse leukocyte antigens are listed in **Table 3.2**. Non-CD antigens are listed in alphanumeric order in **Table 3.3**.

Table 3.4 is a detailed summary of mouse leukocyte antigen distribution depicting the presence of surface antigens on different subsets. Antigen distribution on hematopoietic stem cells, erythrocytes, epithelial cells, endothelial cells, NK cells, monocytes/macrophages, T cells, B cells, granulocytes, megakaryocytes/platelets, and dendritic cells is illustrated graphically in Figures 3.1–3.11.

#### **HUMAN CD ANTIGENS**

CD antigens established in the 7th International Workshop of Human Leukocyte Differentiation Antigens are listed in **Table 3.5**. This table provides information regarding their molecular weight, gene locus, ligands/receptors, functions, and distribution. An addendum describing HLDA family and main antigen expression is provided as **Table 3.6**. A list of abbreviations can be found inside the back cover of this book.

# THE HLA DICTIONARY, 2001

# A summary of HLA-A, -B, -C, -DRB1/3/4/5, -DQB1 alleles and their association with serologically defined HLA-A, -B, -C, -DR, and -DQ antigens

**GMTh Schreuder and Associates\*** 



Several years ago, the World Marrow Donor Association (WMDA) initiated a study aimed at identifying the serological types associated with each HLA-A, -B, -DRB allelic product. This work resulted in a 'serology to DNA equivalents dictionary' first published in 1997 with an update in 1999.<sup>1,2</sup> The dictionary is an attempt to aid searches for unrelated hematopoietic stem cell (hsc) donors in adult volunteer and umbilical cord blood banks. While most patients in need of hsc transplantation are HLA typed by DNA-based methods at medium or high resolution, substantial parts of the donor registries provide serologically based

HLA typings only, at least for HLA class I. In this respect, the dictionary can help in the search for donors whose HLA phenotypes closely resemble that of the patient even though these typings are determined by different methods. Once identified, molecular class I typing of patient and selected donor can be performed to confirm the match. Since the appearance of the 1999 dictionary<sup>2</sup> based on the alleles listed in the 1998 WHO Nomenclature report,<sup>3</sup> a large number of additional alleles have received official allele designations.<sup>4</sup> Moreover, a substantial number of DNA-based typings have been added to the NMDP database

\*GMTh Schreuder,<sup>1,2</sup> CK Hurley,<sup>1,4</sup> SGE Marsh,<sup>1,2</sup> M Lau,<sup>1,3</sup> M Maiers,<sup>4</sup> C Kollman,<sup>4</sup> HJ Noreen<sup>4</sup>

<sup>1</sup>World Marrow Donor Association Quality Assurance Working Group, Leiden, The Netherlands

<sup>2</sup>WHO Nomenclature Committee for Factors of the HLA system

<sup>3</sup>International Cell Exchange, UCLA, Los Angeles, California, USA

This Dictionary was published in parallel in *Tissue Antigens* 2001; 58: 109–140; *Human Immunology* 2001; 62: 826–849; and *European Journal of Immunogenetics* 2001; 28: 565–596. It is also available at http://www.worldmarrow.org.

<sup>&</sup>lt;sup>4</sup>US National Marrow Donor Program HLA testing laboratories and M. Setterholm

# NOMENCLATURE FOR FACTORS OF THE HLA SYSTEM, 2002

#### SGE Marsh and Associates\*

NAMING OF ADDITIONAL GENES WITHIN THE HLA REGION
NAMING OF ADDITIONAL ALLELES
RENAMING OF ALLELES AND REMOVAL OF INCORRECT ALLELES
EXTENSION OF HLA ALLELE NAMES
NAMING OF ALLELES WITH ABERRANT EXPRESSION

The WHO Nomenclature Committee for Factors of the HLA System met in Victoria, Canada in May 2002 after the 13th International Histocompatibility Workshop to consider additions and revisions to the nomenclature of HLA specificities following the principles established in previous reports.<sup>1–16</sup>

The main subjects discussed were:

- Naming of additional genes within the HLA region
- Naming of additional alleles
- · Renaming of alleles and removal of incorrect alleles
- Extension of HLA allele names

\*SGE Marsh,<sup>1</sup> ED Albert,<sup>2</sup> WF Bodmer,<sup>3</sup> RE Bontrop,<sup>4</sup> B Dupont,<sup>5</sup> HA Erlich,<sup>6</sup> DE Geraghty,<sup>7</sup> JA Hansen,<sup>8</sup> B Mach,<sup>9</sup> WR Mayr,<sup>10</sup> P Parham,<sup>11</sup> EW Petersdorf,<sup>12</sup> T Sasazuki,<sup>13</sup> GMTh Schreuder,<sup>14</sup> JL Strominger,<sup>15</sup> A Svejgaard,<sup>16</sup> PI Terasaki<sup>17</sup>

<sup>1</sup>Anthony Nolan Research Institute, London, UK

<sup>&</sup>lt;sup>2</sup>Policlinic for Children, University of Munich, Germany

<sup>&</sup>lt;sup>3</sup>Cancer Research UK, Oxford, UK

<sup>&</sup>lt;sup>4</sup>Biomedical Primate Research Centre, Rijswijk, The Netherlands

Sloan-Kettering Institute for Cancer Research, New York, USA

<sup>&</sup>lt;sup>6</sup>Roche Molecular Systems, Alameda, USA

<sup>&</sup>lt;sup>7</sup>Fred Hutchinson Cancer Center, Seattle, USA

<sup>&</sup>lt;sup>8</sup>Fred Hutchinson Cancer Center, Seattle, USA

<sup>&</sup>lt;sup>9</sup>University of Geneva, Geneva, Switzerland

<sup>&</sup>lt;sup>10</sup>University of Vienna, Vienna, Austria

<sup>&</sup>lt;sup>11</sup>Stanford University School of Medicine, Stanford, USA

<sup>&</sup>lt;sup>12</sup>Fred Hutchinson Cancer Center, Seattle, USA

<sup>&</sup>lt;sup>13</sup>Kyushu University, Fukuoka, Japan

<sup>&</sup>lt;sup>14</sup>Leiden University Medical Center, Leiden, The Netherlands

<sup>&</sup>lt;sup>15</sup>Harvard University, Cambridge, USA

<sup>&</sup>lt;sup>16</sup>State University Hospital, Copenhagen, Denmark

<sup>&</sup>lt;sup>17</sup>Los Angeles, USA

The report on which this chapter is based was originally published in Tissue Antigens 2002; 60: 407-464.

# NOMENCLATURE FOR FACTORS OF THE DOG MAJOR HISTOCOMPATIBILITY SYSTEM (DLA), 2000

#### LJ Kennedy and Associates\*

 NAMING OF GENES WITHIN THE DLA
 NAMING OF MHC ALLELES FROM OTHER CANIDAE
 NAMING OF NEW ALLELES
 UPDATED SEQUENCE ALIGNMENTS

The Major Histocompatibility Complex (MHC) of the dog and other Canidae appears to be highly polymorphic, and alleles of these genes are likely to be functionally relevant in regulating the immune response and the susceptibility/ resistance to immune-mediated diseases. Considerable effort has recently been made in characterizing the extent of the polymorphisms in DLA class II genes.

# NAMING OF GENES WITHIN THE DLA REGION

#### Class I

Following the first Nomenclature Committee report,<sup>1,2</sup> studies by JL Wagner (unpublished) have identified locusspecific primers for four transcribed class I genes. Thus the previously described alleles<sup>3</sup> can be assigned unequivocally to particular loci. In order to avoid the suggestion that any DLA class I genes were homologs of particular HLA class I genes, the Nomenclature Committee decided that numbers rather than letters would be used at present to name DLA class I genes. The updated list of DLA genes is shown in **Table 6.1a**, while **Table 6.1b** lists other genes which have yet to be confirmed and do not have official names. Since the first DLA nomenclature report, the DLA region has been mapped to dog chromosome 12,<sup>4</sup> and it has been confirmed that DLA-79 maps to a separate region on chromosome 18. Although DLA-79 and C1pg-26 are orphan genes not located in the DLA region, the Committee considered that naming such genes fell within its remit.

#### Class II

No new class II gene names were assigned by the Committee in its second report.

#### NAMING OF NEW ALLELES

The Committee reaffirmed the published conditions for naming new alleles.<sup>1,2</sup>

Since the first nomenclature report, 48 DLA-88, 16 DLA-DRB1, 6 DLA-DQA1, and 15 DLA-DQB1 alleles have been

<sup>\*</sup>Lorna J Kennedy,<sup>1</sup> John M Angles,<sup>2</sup> Annette Barnes,<sup>1</sup> Stuart D Carter,<sup>1</sup> Olga Francino,<sup>3</sup> John A Gerlach,<sup>4</sup> George M Happ,<sup>5</sup> William ER Ollier,<sup>1</sup> Wendy Thomson,<sup>1</sup> John L Wagner<sup>6</sup>

<sup>&</sup>lt;sup>1</sup>Mammalian Immunogenetics Research Group, Veterinary Clinical Sciences, University of Liverpool, UK

<sup>&</sup>lt;sup>2</sup>Small Animal Clinical Studies, University College Dublin, Ireland

<sup>&</sup>lt;sup>3</sup> Facultat de Veterinaria, Universitat Autonoma de Barcelona, Spain

<sup>&</sup>lt;sup>4</sup>Medical Technology Program, Department of Medicine, Michigan State University, East Lansing, Michigan, USA

<sup>&</sup>lt;sup>5</sup>Institute of Arctic Biology, University of Alaska, Fairbanks, Alaska, USA

<sup>&</sup>lt;sup>6</sup>Thomas Jefferson University, Department of Medicine, Philadelphia, PA, USA

This chapter is based on the second report of the ISAG DLA Nomenclature Committee, which was originally published in *Tissue Antigens* 58: 55-70.

# **ANTIGEN PRESENTATION**

#### ANTIGEN PRESENTATION

ANTIGEN-PRESENTING CELLS

#### ANTIGEN-PRESENTING PATHWAYS

#### ANTIGEN PRESENTATION

Antigen presentation is the expression of antigen molecules on the surface of a macrophage or other antigen-presenting cell in association with MHC class II molecules when the antigen is being presented to a CD4<sup>+</sup> helper T cell or in association with MHC class I molecules when presentation is to CD8<sup>+</sup> cytotoxic T cells. For appropriate presentation, it is essential that peptides bind securely to the MHC class II molecules, since those that do not bind or are bound only weakly are not presented and fail to elicit an immune response. Following interaction of the presented antigen and MHC class II molecules with the CD4<sup>+</sup> helper T cell receptor, the CD4<sup>+</sup> lymphocyte is activated, IL-2 is released, and IL-2 receptors are expressed on the CD4<sup>+</sup> lymphocyte surface. The IL-2 produced by the activated cell stimulates its own receptors, as well as those of mononuclear phagocytes, increasing their microbicidal activity. IL-2 also stimulates B cells to synthesize antibody. Whereas B cells may recognize a protein antigen in its native state, T cells only recognize the peptides, that result from antigen processing, in the context of major histocompatibility complex molecules.

#### ANTIGEN-PRESENTING CELLS

Antigen-presenting cells (APC) are cells that can process a protein antigen, break it into peptides, and present it in conjunction with class II MHC molecules on the cell surface where it may interact with appropriate T cell receptors. Professional APCs include dendritic cells, macrophages, and B cells, whereas nonprofessional APCs that function in antigen presentation for only brief periods include thymic epithelial cells and vascular endothelial cells. Dendritic cells, macrophages, and B cells are the principal antigenpresenting cells for T cells, whereas follicular dendritic cells are the main antigen-presenting cells for B cells. The immune system contains three types of antigen-presenting cells, i.e., macrophages, dendritic cells, and B cells. **Table** 7.1 shows properties and functions of these three types of antigen-presenting cells.

#### ANTIGEN-PRESENTING PATHWAYS

Dendritic cells, macrophages, and B cells process and present antigen to immunoreactive lymphocytes such as CD4<sup>+-</sup>helper/inducer T cells. A MHC transporter geneencoded peptide supply factor may mediate peptide antigen presentation. Other antigen-presenting cells that serve mainly as passive antigen transporters include B cells, endothelial cells, keratinocytes, and Kupffer cells. This group of APCs present exogenous antigen processed in their endosomal compartment and presented together with class II MHC molecules. Other APCs present antigen that has been endogenously produced by the body's own cells with processing in an intracellular compartment and presentation together with class I MHC molecules. A third group of APCs present exogenous antigen that is taken into the cell and processed, followed by presentation together with class I MHC molecules. In addition to processing and presenting antigenic peptides in association with Class II MHC molecules, an antigen-presenting cell must also deliver a co-stimulatory signal that is necessary for T cell activation. The outcome of an appropriate immune response is dependent on the bidirectional communication between T cells and APCs. Figure 7.1 depicts the 'crosstalk' between APCs and T cells.

Optimal activation of a naïve lymphocyte requires two signals, an antigen-specific signal initiated by engagement of TCR or BCR and a costimulatory signal independent of the antigen receptor complex. C28 is the most studied T costimulatory pathway. Engagement of CD28 augments proliferation and promotes survival of T cells. The other major T

# **DESIGN AND ANALYSIS OF ALGORITHMS**

# Edited by

# S.TAMILARASI



## **Design and analysis of Algorithms**

Copyright © 2021 by S.Tamilarasi

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-473-4

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

### **TABLE OF CONTENTS**

Book elements	Page Numbers
Chapter-I INTRODUCTION B.Senthil Kumar	01-05
Chapter-II ELEMENTARY DATA STRUCTURES B.Senthil Kumar	06-07
Chapter-III DESIGN OF ALGORITHM METHODS B.Senthil Kumar	08-11
<b>Chapter-IV</b> <b>DESIGN OF ALGORITHM METHODS CONTINUATION</b> B.Senthil Kumar	12-13
Chapter-V ALGEBRAIC PROBLEMS B.Senthil Kumar	14-24

#### **CHAPTER-I**

#### **INTRODUCTION**

#### **B.Senthil Kumar**

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Basic Terminology: Elementary Data Organization.

Data: Data are simply values or set of values.

Information: Meaningful or processed data is called information.

Data item: Data Item refers to a single unit of values.

Group items: Data items that are divided into sub items are called group items. eg,.name may be divided into three sub items, first, middle and last name.

Elementary items: Data item that cannot be sub divided into sub item Ex: PAN card number / Bank Pass Book Number is treated as single item

Entity: An entity is something that has certain properties or attributes which may be assigned values. eg, employee is an entity. The attributes and values of an entity are:

Attributes: Name Age Address

Values : RAM 28 XYZ street

Entity set : Entities with similar attributes form an entity set.

Basic Terminology: Elementary Data Organization.

Field: a field is an single elementary unit of information representing an attribute of an entity.

Record: A record is a collection of field values of a given entity.

File: A file is a collection of records of the entities in a given entity set.

Primary Key: The key which is used to uniquely identify a record is called primary key

Fixed-length records: All the records contain the same data items with the same amount of space assigned to each data item.

Variable-length records: File records may contain different lengths. For Example, student records usually have variable lengths, since different students take different number of courses. Usually, variable length records have a maximum and a minimum length

Data type: A data type is a set of permitted values and a set of operations on these values.

(ie) Data type = Permitted values + operations.

Examples of data types include integer, float, character and boolean

Data Structures

Data Structure: Logical or mathematical model of organizing data is called data structure.

i.e. Data Structure = organized data + Operations.

The study of data structures includes the following three steps:

1. Logical or mathematical description of the structure.

2. Implementation of the structure on a computer.

3. Quantitative analysis of the structure, which includes determining the amount of memory needed to store the structure and the time required to process the structure.

Data structures can be classified as:

Primitive data structure: Primitive data structures are the basic data structures and are directly operated upon by the machine instructions.
 Examples: integer, real ,character and boolean . Primitive data structures are

sometimes called Data types.

Non primitive data structure or Compound data structures: Non primitive data structures are derived from the primitive data structure. The non primitive data structure defines the group of homogenous and heterogeneous data items.
 Non primitive data structures can be classified as:

1. Linear data structures

2. Non-linear data structures.

Linear data structures: A data structure whose elements form a sequence, and every element in the structure has a unique predecessor and unique successor. Eg. Arrays, linked lists, stacks and queues.

Non-linear data structures: A data structure whose elements do not form a sequence, and there is no unique predecessor or unique successor. Eg. Trees and graphs

TYPES OF DATA STRUCTURES Arrays: An array is a list of finite number of elements ofsame data type , ie integers, real or characters .The individual elements of an array are accessedusing an index to the array. Arrays can be one- dimensional or linear arrays, two dimensionalarraysormultidimensionalarrays



TYPES OF DATA STRUCTURES Arrays: An array is a list of finite number of elements of same data type, ie integers, real or characters. The individual elements of an array are accessed using an index to the array. Arrays can be one- dimensional or linear arrays, two dimensional arrays or multidimensional arrays

**One Dimensional Array** 

Two Dimensional arrays

Abdul Rahaman
Balaji
Durai
Inamul Haque
Smith
Jarin

1	2	3	4
5	6	7	8
9	1	1	1
	0	1	2
1	1	1	1
3	4	5	6

Linked Lists: A linked list is a linear collection of data elements called nodes. The linear order is maintained by pointers. A linked list can be linear (one way) linked list or doubly (two-way) linked list.



Stacks: A stack, also called Last In First Out (LIFO) system, is a linear list in which insertions and deletions can take place only at one end, called the top.



Queues: A queue, also called First In First Out (FIFO) system, is a linear list in which insertions can take place at one end of the list, called the rear of the list, and deletions can take place only at the other end, called the front of the list.



Trees: A tree is a data structure that represents a hierarchical relationship between various elements



Graphs: A graph G is a ordered set (V,E) where V represents the set of elements, called nodes or vertices and E represents the edges between these elements. They are used to represent relationships between pair of elements, which are not necessarily hierarchical in nature.(ie) Graphs are those where nodes are connected by more than one path to each other.



#### **CHAPTER-II**

#### **ELEMENTARY DATA STRUCTURES**

#### **B.Senthil Kumar**

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Data structure operations

1. Traversing: Accessing each record exactly once so that certain items in the record may be processed.

2. Searching: Finding the location of the record with a given key value.

3. Inserting: Adding a new record to the structure.

4. Deleting: Removing a record from the structure.

5. Sorting: Arranging the records in some logical order.

6. Merging: Combing the records in two different sorted files into a single sorted file.

Mathematical notations and functions:

Let x be any real number. Then x lies between two integers called the floor and the ceiling of x.

 $\checkmark$  [x], called the floor of x, denotes the greatest integer that does not exceed x.

 $\checkmark$  [x], called the ceiling of x, denotes the least integer that is not less than x.

✓ If x is itself an integer, then [x] = [x]; otherwise [x] + 1 = [x]

Examples:  $[3.14] = 3 [\sqrt{5}] = 2 [-8.5] = -9 [7] = 7 [3.14] = 4 [\sqrt{5}] = 3 [-8.5] = -8$ [7] = 7

2. Remainder Function (Modular Arithmetic): If k is any integer and M is a positive integer, then: k (mod M) gives the integer remainder when k is divided by M.

Examples:  $25 \pmod{7} = 4 \ 25 \pmod{5} = 0 \ 3 \pmod{8} = 3$ 

3. Integer and Absolute Value Functions: Let x be any real number. The integer value of x, written INT(x), converts x into an integer by deleting(truncating) the

fractional part of the number. Thus INT(3.14) = 3,  $INT(\sqrt{5}) = 2$ , INT(-8.5) = -8, INT(7) = 7 The absolute value of the real number x, written ABS(x) or |x|, is defined as the greater of x or -x. Thus |-15| = 15, |7| = 7, |-3.33| = 3.33, |4.44| = 4.44, |-0.075| = 0.075

4. Summation Symbol (Sums): The symbol which is used to denote summation is a Greek letter Sigma  $\sum$  Let a1, a2, a3, ...., an be a sequence of numbers. Then the sum a1 + a2 + a3 + ..... + an will be written as: n  $\sum$  aj j=1 where j is called the dummy index or dummy variable.

E.g.  $n \sum j = 1 + 2 + 3 + \dots + n = n(n+1)/2 j=1$ 

5. Factorial Function: n! denotes the product of the positive integers from 1 to n. n! is read as 'n factorial', i.e. n! = 1 \* 2 \* 3 \* ..... \* (n-2) \* (n-1) \* n

Example: 4! = 1 \* 2 \* 3 \* 4 = 24 For n>1, we have n! = n.(n-1)!, Hence 5! = 5 \* 4! = 120.

#### 6. Permutations:

Let we have a set of n elements. A permutation of this set means the arrangement of the elements of the set in some order.

Example: Suppose the set contains a, b and c. The various permutations of these elements can be: abc, acb, bac, bca, cab, cba. If there are n elements in the set then there will be n! permutations of those elements. It means if the set has 3 elements then there will be 3! = 1 \* 2 \* 3 = 6 permutations of the elements.

7. Exponents and Logarithms: Exponent means how many times a number is multiplied by itself. If m is a positive integer, then: am = a.a.a...a, m times a = 0 = 1, 1/a = a-1, a - m = 1/am,  $am/n = (n\sqrt{a})m$ 

Examples: 1) 24 =16 2) 2-4 = 1/24 3) 1252/3 = 52 = 25

#### **CHAPTER-III**

#### **DESIGN OF ALGORITHM METHODS**

#### **B.Senthil Kumar**

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Algorithms An essential aspect to data structures is algorithms.

Data structures are implemented using algorithms.

Definition: An algorithm is a well-defined list of steps for solving a particular problem. Example:

Print the average of 3 given numbers.

1.Read a, b, c

2.sum = a + b + c

3. avg = sum / 3;

4.print avg

Characteristics of an Algorithm

Input: An algorithm has some input values. We can pass 0 or some input value to an algorithm. Output: We will get 1 or more output at the end of an algorithm.

Definiteness: An algorithm should be unambiguous which means that the instructions in an algorithm should be clear and simple.

Finiteness: An algorithm should have finite number of steps and it should end after a finite time

Effectiveness: An algorithm should be effective as each instruction in an algorithm affects the overall process.

Language independent: An algorithm must be language-independent so that the instructions in an algorithm can be implemented in any of the languages with the same output.

Complexity of an algorithm or Algorithm analysis :

There are often many different algorithms which can be used to solve the same problem. We want to determine the amount of memory they use and how much time they require, so that we can choose the most efficient algorithm among them. This is called an algorithm analysis

(ie) Analysis of algorithm deals in finding best algorithm which runs fast and takes in less memory.

The time and space it uses are two major concerns of the efficiency of an algorithm. The complexity of an algorithm is the function which gives the running time and/or memory space in terms of input size

Example:

Find sum of first n natural numbers.

Algorithm1: (use the formula) // f(n) = 3 - constant

1.Read n // 1 time

 $2.Sum = n^* (n + 1) / 2; // 1 time$ 

3. print sum. // 1 time

Algorithm 2 //f(n) = n+3 - linear

1.Read n // 1 time

2.sum = 0; // 1 time

3. for(int i = 1;  $i \le n$ ; i++) sum = sum + i; // n times

4.print sum. // 1 time

In order to determine the best algorithm among the two algorithms, we have to analyse these algorithms based on time and memory

#### COMPLEXITY OF ALGORITHMS:

Suppose M is an algorithm and n is the size of the input.

 $\checkmark$  The complexity of an algorithm M is the function f(n) which gives the running time and/or storage space requirement of the algorithm in terms of the size n of the input data.

 $\checkmark$  The time and space used by an algorithm are the two main measures for the efficiency of algorithm.

✓ The time (Time complexity) is measured by counting the number of key operations ie. The number of comparisons.  $\checkmark$  The space (Space complexity) is measured by counting the maximum of memory needed by the algorithm.

There are 2 cases for investigating the complexity of an algorithm. They are:

1. Worst Case: maximum value of f(n) for any input.

2. Average Case: expected value of f(n).

ARRAYS

Linear Arrays

Definition: A linear array is a list of finite number n of homogeneous data elements such that :

a) The elements of the array are referenced respectively by an index set consisting of n consecutive numbers.

b) The elements of the array are stored respectively in successive memory locations. The number n of elements is called the length or size of the array

Length = UB - LB + 1

Where UB is the largest index, called the upper bound, and LB is the smallest index, called the lower bound of the array

Length = UB where LB = 1

Element of an array A may be denoted by

Subscript notation: A1, A2, , ...., An Parenthesis notation: A (1), A (2), ...., A(n)

Bracket notation : A[1], A[2], ...., A[n]

The number K in A[K] is called subscript or an index and A[K] is called a subscripted variable Example: A linear array DATA consisting of the name of five elements

1	247
2 3	<mark>56</mark>
4 5	429
	135
	87

Example :

An automobile company uses an array AUTO to record the number of auto mobile sold each year from 1932 through 1984.

AUTO[k] = Number of auto mobiles sold in the year K

LB = 1932

UB = 1984

Length = UB - LB + 1 = 1984 - 1930 + 1 = 55

Representation of linear array in memory



Let LA be a linear array stored in successive memory cells.

LOC(LA[K]) = address of the element A[K] of the array A

To calculate the address of any element of A formula is

LOC(LA[k]) = Base(LA) + w(K - lower bound)

Where,

Base (LA) =Address of the first element of A

w =number of words per memory cell for the array

K = any index of A Traversing linear arrays

Traversing means accessing and processing (i.e visiting ) each element of the data structure exactly ones.

Example: Print the contents of each element of the array LA or Count the number of elements of LA with a given property.

#### **CHAPTER-IV**

#### **DESIGN OF ALGORITHM METHODS CONTINUATION**

#### **B.Senthil Kumar**

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### Stacks

 $\checkmark$  A stack is a list of elements in which an element may be inserted or deleted only at one end, called the top of the stack. The other end is called the bottom.

 $\checkmark$  This means, in particular, that elements are removed from a stack in the reverse order of that in which they were inserted into the stack.

 $\checkmark$  An item may be added or removed only from the top of a stack. This means that the last item to be added to a stack is the first item to be removed. Accordingly, stacks are also called Last-In First-Out (LIFO) lists.

 $\checkmark$  Two basic operations associated with stacks are

1."Push" is the term used to insert an element into a stack.

2."Pop" is the term used to delete an element from a stack.



#### **Array Representation of Stacks**

A stack can be represented in memory either as an one-dimensional array or as a singly linked list. The stack is maintained by a linear array STACK: a pointer variable TOP, which contains the location of the top element of the stack; and a

variable MAXSTK which gives the maximum number of elements that can be held by the stack.

#### operations on Stack :

PUSH and POP //This procedure pushes an ITEM onto a stack PUSH(STACK,TOP,MAXSTACK,ITEM)

1. [STACK already filled] If TOP=MAXSTACK, then Print OVERFLOW and Return.

- 2. Set TOP=TOP+1
- 3. Set STACK[TOP]=ITEM.
- 4. Return.

//This procedure deletes the top element of STACK and assign it to the variable ITEM

#### **POP(STACK,TOP,ITEM)**

- 1. If TOP=0, then print UNDERFLOW and Return
- 2. Set ITEM=STACK[TOP].
- 3. Set TOP=TOP-1.
- 4. Return.

#### **CHAPTER-V**

#### ALGEBRAIC PROBLEMS

#### **B.Senthil Kumar**

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### Queues

 $\checkmark$  A Queue is a linear list of elements in which deletions can take place only at one end, called the front, and insertions can take place only at the other end, called the rear.

✓ Queues are also called first-in first-out (FIFO) lists, since the first element in a queue will be the first element out of the queue.

 $\checkmark$  In other words, the order in which elements enter a queue is the order in which they leave. This contrasts with stacks, which are Last-in First-out (LIFO) lists.

#### **Examples:**

1. The automobiles waiting to pass through an intersection form a queue, in which the first car in line is the first car through

2. The people waiting in line at a bank(or ticket counter) form a queue, where the first person in line is the first person to be served.

3. An important example of a queue in computer science occurs in a timesharing system, in which programs with the same priority form a queue while waiting to be executed.

4. Queue for printing purposes(Documents sent to a shared printer)





#### **REPRESENTATION OF QUEUE**

- $\checkmark$  Like Stacks, Queues can also be represented in memory in two ways.
- 1. Linear Array
- 2. One-way Linked List



#### **Array Representation of Queues**

 $\checkmark$  Queues can be easily represented using linear arrays.

 $\checkmark$  Each of our queues will be maintained by a linear array QUEUE and two pointer variables:

- ✓ FRONT, containing the location of the front element of the queue;
- $\checkmark$  and REAR, containing the location of the rear element of the queue.
- $\checkmark$  The condition FRONT = NULL will indicate that the queue is empty.

✓ when an element is deleted from the queue(called dequeue), the value of FRONT is increased by 1; this can be implemented by the assignment FRONT := FRONT + 1

 $\checkmark$  Similarly, whenever an element is added to the queue (called enqueue), the value of REAR is increased by 1; this can be implemented by the assignment

✓ REAR := REAR + 1

✓ This means that after N insertions, the rear element of the queue will occupy QUEUE[N]



#### **REPRESENTATION OF QUEUE**

#### **OPERATIONS ON QUEUE**

 $\checkmark$  The two basic operations that can be performed on a queue are: Insert (enqueue) : to insert an element at rear of the queue Delete(dequeue): to delete an element from front of the queue.

 $\checkmark$  Before inserting a new element in the queue, it is necessary to test the condition of overflow.

 $\checkmark$  Overflow occurs when the queue is full and there is no space for a new element.

 $\checkmark$  Similarly, before removing the element from the queue, it is necessary to check the condition of underflow.

 $\checkmark$  Underflow occurs when the queue is empty.

For Insert (enqueue )Operation
Insert-Queue (Queue, Rear, Front, N, Item)

Here, Queue is an array where to store data. Rear represents the location in which the data element is to be inserted and Front represents the location from which the data element is to be removed. Here N is the maximum size of the Queue and Item is the new item to be added.

- 1. If Rear = N then Print: Overflow and Return. /\*...Queue already filled..\*/
- 2. Set Rear := Rear +1
- 3. Set Queue[Rear] := Item
- 4. Return.

#### **DEQUE(DOUBLE ENDED QUEUE**

**DEQUE** (Double ended Queue)- A deque is a queue in which elements can be added or removed at either end but not in the middle. A deque is usually maintained by a circular array DEQUE with pointers LEFT and RIGHT, which point to two ends of deque. The elements extend from LEFT end to RIGHT end of deque. The term circular comes from the fact that DEQUE[1] comes after DEQUE [N].The condition LEFT=NULL will be used to indicate that a deque is empty

• There are two variations of a deque

• Input-restricted deque- It is a deque which allows insertions at only one end of list but allows deletions at both ends of the list

• Output-restricted deque- It is a deque which allows deletions at only one end of list but allows insertions at both ends of list

Linked representation of the Queue

• A linked queue is a queue implemented as a linked list with two pointer variables FRONT and REAR pointing to the nodes in the front and rear of the queue. The INFO field of list hold the elements of the queue and LINK field holds pointer to neighboring element of queue.

• In case of insertion in linked queue, a node borrowed from AVAIL list and carrying the item to be inserted is added as the last node of linked list representing the queue. Rear pointer is updated to point to last node just added to the list

• In case of deletion, first node of list pointed to by FRONT is deleted and FRONT pointer is updated to point to next node in the list.

• Unlike the array representation, linked queue functions as a linear queue and there is no need to view it as circular for efficient management of space

Algorithm: LINKQINSRT(INFO, LINK, FRONT, REAR, AVAIL, ITEM)

This algorithm inserts an item in linked list implementation of the queue

Step 1: If AVAIL=NULL, then: Write: 'OVERFLOW' Exit

Step 2: Set NEW:=AVAIL and AVAIL:=LINK[AVAIL]

Step 3: Set INFO[NEW]:=ITEM and LINK[NEW]:=NULL

Step 4: If FRONT=NULL, then: Set FRONT=REAR=NEW Else: Set LINK[REAR]:=NEW and REAR:=NEW

Step 5: Return

# Algorithm: LINKQDEL(INFO, LINK, FRONT, AVAIL, ITEM) This algorithm

deletes an element from the front of the queue

Step 1: If FRONT=NULL, then: Write: 'UNDERFLOW' Exit

Step 2: Set TEMP:=FRONT

Step 3: Set ITEM:=INFO[FRONT]

Step 4: Set FRONT:=LINK[FRONT]

Step 5: Set LINK[TEMP]:=AVAIL and AVAIL:=TEMP

Step 6: Return .

**Priority Queue**- A priority queue is a collection of elements such that each element has been assigned a priority and such that the order in which elements are deleted and processed comes from following rules:

• An element of higher priority is processed before any element of lower priority

• Two elements of same priority are processed according to the order in which they were added to queue

• An example of a priority queue is a time sharing system. Programs of higher priority are processed first and programs with same priority form a standard queue

**LINKED LIST Linked List**- A linked list or one-way list is a linear collection of data elements, called nodes, where linear order is given by means of pointers.

Each node is divided into two parts: The first part contains the information of the element.

Second part, called the linked field or next pointer field, contains the address of the next node in the list.

**Representation of Linked list in memory** • Linked list is maintained in memory by two linear arrays: INFO and LINK such that INFO[K] and LINK[K] contain respectively the information part and the next pointer field of node of LIST. LIST also requires a variable name such as START which contains the location of the beginning of the list and the next pointer denoted by NULL which indicates the end of the LIST. Algorithm: (Traversing a Linked List) Let LIST be a linked list in memory.

#### Trees

1. Trees are nonlinear types of data structures.

2. This structure is mainly used to represent data containing a hierarchical relationship between elements. e.g. records, family trees and table of contents.

**Binary Trees Definition**: A binary tree T is defined as a finite set of elements, called nodes, such that:

1. T is empty (called the null tree or empty tree), or

2. T contains a distinguished node R, called the root of T, and the remaining nodes of T form an ordered pair of disjoint binary trees T1 and T2. A binary tree is frequently presented by means of a diagram. Specifically, the diagram on next slide represents a binary tree as follows.

(i) T consists of 11 nodes, represented by the letters A through L, excluding I.

(ii) The root of T is the node A at the top of the diagram

(iii) A left-downward slanted line from a node N indicates a left successor of N, and a right-downward slanted line from N indicates a right successor of N.

We observe that: B is a left successor and C is a right successor of the node A.

The left subtree of the root A consists of the nodes B, D, E and F, and the right subtree of A consists of the nodes C, G, H, J, K andL.



Algebraic Expressions

• 
$$E = (a - b) / ((c * d) + e)$$



#### Terminology

(1) **Parent**: Suppose N is a node in T with left successor S1 and right successor S2. Then N is called the parent (or father) of S1 and S2. Similarly, S1 is called the left child (or son) of N, and S2 is called the right child (or son) of N. S1 and S2 are said to be siblings (or brothers)

(2) **Descendant** and Ancestor A node L is called a descendant of node N (and N is called an ancestor of L) if there is a succession of children from N to L.

(3) Edge The line drawn from a node N of T to a successor is called an edge.

(4) Path A sequence of consecutive edges is called a path.

(5) Branch A path ending in a leaf is called a branch.

(6) Level number Each node in a binary tree T is assigned a level number. The root R of the tree T is assigned the level number 0, and every other node is assigned a level number which is 1 more than the level number of its parent. In addition, those nodes with the same level number are called to belong to the same generation.

(7) Depth (or height) The depth (or height) of a tree T is the maximum number of nodes in a branch of T. This turns out to be 1 more than the largest level number of T.

**Full Binary Trees** A full binary tree is a binary tree in which all of the leaves are on the same level and every non-leaf node has two children.

#### Graph

#### Graph theory Terminolgy A graph G consists of :

A set V of elements called the nodes (or points or vertices)

A set E of edges such that each edge e in E is identified with a unique (unordered) pair [u,v] of nodes in V, denoted by e=[u,v]

The nodes u and v are called the end points of e or adjacent nodes or neighbors.

The edge in a graph can be directed or undirected depending on whether the direction of the edge is specified or not.

A graph in which each edge is directed is called a **directed graph** or **digraph**.

A graph in which each edge is undirected is called undirected graph.

A graph which contains both directed and undirected edges is called mixed graph.

Let G=(V,E) be a graph and e  $\varepsilon$  E be a directed edge associated with ordered pair of vertices (v1,v2). Then the edge e is said to be initiating from v1 to v2. v1 is the starting and v2 is the termination of the edge e.

An edge in a graph that joins a vertex to itself is called a sling or a loop The degree of a node or vertex u is written deg(u) is the number of edges containing u.

The degree of a loop is 2 In a directed graph for any vertex v the number of edges which have v as their initial vertex is called the out-degree of v.

The number of edges which have v as their terminal vertex is called the in-degree of v. The sum of indegree and out-degree of a vertex is called the degree of that vertex.

If deg(u)=0, then u is called an isolated node and a graph containing only isolated node is called a null graph The maximum degree of a graph G, denoted by  $\Delta(G)$ , is the maximum degree of its vertices, and the minimum degree of a graph, denoted by  $\delta(G)$ , is the minimum degree of its vertices.

A sequence of edges of a digraph such that the terminal vertex of the edge sequences in the initial vertex of the next edge, if exist, is called a path  $E=\{(v1,v2),(v2,v3),(v3,v4)\}$ 

A graph G is said to be connected if and only if there is a simple path between any two nodes in G. In other words, a graph is connected if it does not contain any isolated vertices.

A graph that is not connected can be divided into connected components (disjoint connected subgraphs). For example, this graph is made of three connected components A graph G is said to be complete if every node u in G is adjacent to every node v in G.

A complete graph with n vertices (denoted Kn) is a graph with n vertices in which each vertex is connected to each of the others (with one edge between each pair of vertices). In other words, there is path from every vertex to every other vertex. Clearly such a graph is also a connected graph.

A complete graph with n nodes will have n(n-1)/2 edges

A connected graph without any cycles is called a tree graph or free graph or simply a tree. A graph is said to be labeled if its edges are assigned data. G is said to be weighted if each edge e in G is assigned a non negative numerical value w (e) called the weight or length of e. In such a case, each path P in G is assigned a weight or length which is the sum of the weights of the edges along the path P. If no weight is specified, it is assumed that each edge has the weight w(e) = 1.

Multiple edges- Distinct edges e and e' are called multiple edges if they connect the same endpoints, that is, if e=[u, v] and e'=[u, v]. Such edges are also called parallel edges and a graph that contains these multiple or parallel edges is called a multigraph. Also a graph containing loops is also not a simple graph but a multigraph



# **REPRESENTATION OF A GRAPH**

There are two main ways of representing a graph in memory.

These are: 1. Sequential 2. Linked List

• Sequential Representation- The graphs can be represented as matrices in sequential representation.

Adjacency Matrix The adjacency matrix is a sequence matrix with one row and one column devoted to each vertex. The values of the matrix are 0 or 1. A value of 1 for row i and column j implies that edge eij exists between vi and vj vertices. A value of 0 implies that there is no edge between the vertex vi and vj. Thus, for a graph with v1,v2,v3....vn vertices, the adjacency matrix A=[aij] of the graph G is the n x n matrix and can be defined as

$$a_{ij} = \begin{cases} 1 & \text{if } v_i \text{ is adjacent to } v_j (\text{if there is an edge between } v_i \text{ and } v_j) \\ \\ 0 & \text{if there is no edge between } v_i \text{ and } v_j \end{cases}$$

Such a matrix that contains entries of only 0 or 1 is called a bit matrix or Boolean matrix. The adjacency matrix of the graph G does depend on the ordering of the nodes in G that is different ordering of the nodes may result in a different adjacency matrix. However the matrices resulting from different orderings are closely related in that one can be obtained from another by simply interchanging rows and columns

**Path Matrix**- Let G be a simple directed graph with m nodes, v1,v2,v3.....vm. The path matrix or reachability matrix of G is the m-square matrix P=(pij) defined as

$$P_{ij} = \begin{cases} 1 & \text{if there is a path from } v_i \text{ to } v_j \\ 0 & \text{Otherwise} \end{cases}$$

Suppose there is a path from vi to vj. Then there must be a simple path from vi to vj when  $vi \neq vj$  or there must be a cycle from vi to vj when vi = vj. Since G has only m nodes such a simple path must be of length m-1 or less or such a cycle must have length m or less.

# COMBINATORIAL MATHEMATICS

# EDITED BY



# **Combinatorial mathematics**

Copyright © 2021 by R.Vasanthi

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-369-0

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

Book elements	Page Numbers
Chapter-I COMBINATORIAL MATHEMATICS Dr.A.Usha	01-06
Chapter-II GENERATING FUNCTIONS Dr.A.Usha	07-14
Chapter-III MULTINOMIAL'S Dr.A.Usha	15-22
Chapter-IV NECKLACE PROBLEMS Dr.A.Usha	23-29
Chapter-V POLYA'S THOEREM AND IMMEDIATE APPLICATION Dr.A.Usha	30-37

# **Chapter-I**

#### **COMBINATORIAL MATHEMATICS**

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Combinatorial mathematics is a branch of mathematics that deals with counting and arranging objects in various ways. It involves the study of permutations, combinations, graphs, and other enumerative properties. Here are some key topics in combinatorial mathematics:

#### Subfields:

- 1. **Permutations**: Arrangements of objects in a specific order.
- 2. Combinations: Selections of objects without regard to order.
- 3. Graph Theory: Study of graphs, including connectivity, paths, and cycles.
- 4. **Recurrence Relations**: Equations that define a sequence recursively.
- 5. Generating Functions: Formal power series used to solve counting problems.

#### **Key Concepts:**

- Binomial Coefficient: Number of ways to choose k objects from n objects (nCk or "n choose k").
- 2. **Factorial**: Product of all positive integers up to n (n!).
- 3. Stirling Numbers: Counting permutations with certain properties.
- 4. **Partition Numbers**: Number of ways to divide a set into non-empty subsets.
- 5. **Pigeonhole Principle**: If n items are put into m pigeonholes, with n > m, then at least one pigeonhole must contain more than one item.

#### **Applications:**

- 1. Computer Science: Algorithms, data structures, and complexity theory.
- 2. Cryptography: Secure communication protocols.
- 3. Network Optimization: Shortest paths, maximum flow, and minimum spanning trees.
- 4. **Statistics**: Experimental design and hypothesis testing.
- 5. Code Theory: Error-correcting codes.

#### **Important Combinatorial Identities:**

- 1. **Binomial Theorem**:  $(x+y)^n = \Sigma (nCk)x^k y^{(n-k)}$
- 2. **Pascal's Identity**: (nCk) = (n-1Ck-1) + (n-1Ck)
- 3. Vandermonde's Identity:  $\Sigma$  (nCk)(mCj-k) = (n+mCj)

# **Famous Combinatorial Problems:**

- 1. Traveling Salesman Problem: Find the shortest tour visiting n cities.
- 2. Knapsack Problem: Maximize value while keeping weight below a limit.
- 3. Four Color Theorem: Color a planar graph with 4 colors.

# **Influential Combinatorial Mathematicians:**

- 1. Blaise Pascal
- 2. Leonhard Euler
- 3. Joseph Sylvester
- 4. Georg Polya
- 5. Paul Erdős

Combinatorial mathematics has numerous applications in various fields, and its techniques and results continue to influence many areas of mathematics and computer science.

Here's a more in-depth look at combinatorial mathematics:

# **Branches of Combinatorial Mathematics:**

- 1. **Enumerative Combinatorics**: Counting objects, including permutations, combinations, and partitions.
- 2. Algebraic Combinatorics: Using algebraic techniques to solve combinatorial problems.
- 3. Geometric Combinatorics: Studying geometric objects, such as polytopes and graphs.
- 4. Analytic Combinatorics: Using analysis to study combinatorial objects.
- 5. Probabilistic Combinatorics: Studying combinatorial objects with randomness.

# **Combinatorial Objects:**

- 1. Permutations: Arrangements of objects in a specific order.
- 2. Combinations: Selections of objects without regard to order.
- 3. Graphs: Collections of vertices and edges.
- 4. **Trees**: Connected graphs without cycles.
- 5. Lattices: Partially ordered sets.

6. Matroids: Combinatorial structures generalizing linear independence.

# **Combinatorial Techniques:**

- 1. **Recurrence Relations**: Equations defining a sequence recursively.
- 2. Generating Functions: Formal power series solving counting problems.
- 3. Inclusion-Exclusion Principle: Counting objects by including and excluding subsets.
- 4. **Pigeonhole Principle**: If n items are put into m pigeonholes, with n > m, then at least one pigeonhole must contain more than one item.
- 5. **Bijective Proofs**: Proving identities by establishing a bijection.

# **Combinatorial Theorems:**

- 1. **Binomial Theorem**:  $(x+y)^n = \Sigma (nCk)x^k y^{(n-k)}$
- 2. **Pascal's Identity**: (nCk) = (n-1Ck-1) + (n-1Ck)
- 3. Vandermonde's Identity:  $\Sigma$  (nCk)(mCj-k) = (n+mCj)
- 4. Cayley's Formula: Number of labeled trees on n vertices is n^(n-2)
- 5. Sperner's Theorem: Maximum size of an antichain in a poset.

# **Applications of Combinatorial Mathematics:**

- 1. **Computer Science**: Algorithms, data structures, complexity theory.
- 2. Cryptography: Secure communication protocols.
- 3. Network Optimization: Shortest paths, maximum flow, minimum spanning trees.
- 4. Statistics: Experimental design, hypothesis testing.
- 5. Code Theory: Error-correcting codes.
- 6. Biology: Phylogenetics, genomics.
- 7. **Economics**: Game theory, social choice.

# **Open Problems in Combinatorial Mathematics:**

- 1. **P versus NP problem**: Can every problem with a known efficient algorithm also be verified efficiently?
- 2. Riemann Hypothesis: Distribution of prime numbers.
- 3. Collatz Conjecture: Behavior of a particular sequence.
- 4. Erdős Discrepancy Problem: Bounds on sequences.

# Influential Combinatorial Mathematicians:

- 1. Leonhard Euler
- 2. Joseph Sylvester

- 3. Georg Polya
- 4. Paul Erdős
- 5. Stanley Mandelstam
- 6. Richard Stanley
- 7. Catalin Zara

#### Journals and Conferences:

- 1. Journal of Combinatorial Theory
- 2. Combinatorica
- 3. SIAM Journal on Discrete Mathematics
- 4. International Conference on Combinatorics
- 5. Annual ACM Symposium on Discrete Algorithms

Here are some basic combinatorial numbers and their properties:

## 1. Factorial (n!)

 $\mathbf{n}! = \mathbf{n} \times (\mathbf{n}\text{-}1) \times (\mathbf{n}\text{-}2) \times \dots \times 2 \times 1$ 

Example:  $5! = 5 \times 4 \times 3 \times 2 \times 1 = 120$ 

# 2. Binomial Coefficient (nCk or "n choose k")

nCk = n! / (k!(n-k)!)

Example: 5C2 = 5! / (2!(5-2)!) = 10

# 3. Permutations (P(n,k))

P(n,k) = n! / (n-k)!

Example: P(5,2) = 5! / (5-2)! = 20

# 4. Combinations (C(n,k))

C(n,k) = nCk = n! / (k!(n-k)!)

Example: C(5,2) = 5C2 = 10

# 5. Catalan Numbers (Cn)

Cn = (2n)! / (n!(n+1)!)

Example:  $C3 = (2 \times 3)! / (3!(3+1)!) = 5$ 

#### 6. Fibonacci Numbers (Fn)

Fn = Fn-1 + Fn-2, F0 = 0, F1 = 1

Example: F5 = F4 + F3 = 5

# 7. Stirling Numbers of the First Kind (s(n,k))

s(n,k) = number of permutations of n objects with k cycles

Example: s(4,2) = 6

## 8. Stirling Numbers of the Second Kind (S(n,k))

S(n,k) = number of partitions of n objects into k non-empty subsets

Example: S(4,2) = 7

#### **Properties and Identities:**

- 1. Pascal's Identity: (nCk) = (n-1Ck-1) + (n-1Ck)
- 2. Symmetry: (nCk) = (nC(n-k))
- 3. Binomial Theorem:  $(x+y)^n = \Sigma (nCk)x^k y^{(n-k)}$
- 4. Vandermonde's Identity:  $\Sigma$  (nCk)(mCj-k) = (n+mCj)

#### **Applications:**

- 1. Counting permutations and combinations
- 2. Calculating probabilities
- 3. Modeling population growth (Fibonacci numbers)
- 4. Analyzing algorithms and data structures
- 5. Solving recurrence relations

Let's dive deeper into some specific concepts and identities in combinatorial mathematics:

#### 1. Pascal's Identity

(nCk) = (n-1Ck-1) + (n-1Ck)

Proof: Consider choosing k objects from n. Either the first object is chosen (n-1Ck-1) or it's not (n-1Ck).

#### 2. Binomial Theorem

 $(x+y)^n = \Sigma (nCk)x^k y^(n-k)$ 

Proof: Expand  $(x+y)^n$  using the distributive property.

#### 3. Vandermonde's Identity

 $\Sigma$  (nCk)(mCj-k) = (n+mCj)

Proof: Counting partitions of n+m objects into j non-empty subsets.

#### 4. Stirling Numbers of the Second Kind

 $S(n,k) = (1/k!) \Sigma (kCi) (-1)^{(k-i)} i^n$ 

Proof: Counting partitions of n objects into k non-empty subsets.

# 5. Catalan Numbers

Cn = (2n)! / (n!(n+1)!)

Proof: Counting binary trees with n internal nodes.

# 6. Fibonacci Numbers

Fn = Fn-1 + Fn-2, F0 = 0, F1 = 1

Proof: Counting ways to tile an n-board with 1x1 and 2x1 tiles.

# 7. Inclusion-Exclusion Principle

 $|A \cup B| = |A| + |B| - |A \cap B|$ 

Proof: Counting elements in the union of two sets.

# 8. Pigeonhole Principle

If n items are put into m pigeonholes, with n > m, then at least one pigeonhole must contain more than one item.

Proof: By contradiction.

# **Combinatorial Proofs**

- 1. Bijective proofs: Establishing a bijection between two sets.
- 2. Inductive proofs: Proving a statement for n, assuming it's true for n-1.
- 3. Recursive proofs: Proving a statement using recursive equations.

# Applications

- 1. Computer Science: Algorithms, data structures, complexity theory.
- 2. Cryptography: Secure communication protocols.
- 3. Network Optimization: Shortest paths, maximum flow, minimum spanning trees.
- 4. Statistics: Experimental design, hypothesis testing.
- 5. Code Theory: Error-correcting codes.

**Chapter-II** 

#### **GENERATING FUNCTIONS**

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Generating functions are a powerful tool in combinatorial mathematics, allowing us to:

#### **Count combinatorial objects**

Solve recurrence relations

#### Find closed-form expressions

#### Analyze algorithms

Here's an overview of generating functions in combinatorial mathematics:

#### Definition

A generating function is a formal power series:

$$f(x) = \Sigma a_n x^n$$

where a\_n is the number of combinatorial objects of size n.

#### **Types of Generating Functions**

#### 1. Ordinary Generating Function (OGF)

 $f(x) = \Sigma a_n x^n$ 

Example:  $f(x) = \Sigma n! x^n$  (factorials)

#### 2. Exponential Generating Function (EGF)

$$f(x) = \sum a_n x^n / n!$$

Example:  $f(x) = \Sigma C(n,k) x^n$  (binomial coefficients)

#### 3. Dirichlet Generating Function (DGF)

 $f(s) = \Sigma a_n n^{(-s)}$ 

Example:  $f(s) = \Sigma \mu(n) n^{(-s)}$  (Möbius function)

#### **Operations on Generating Functions**

#### 1. Addition

 $(f(x) + g(x)) = \Sigma (a_n + b_n) x^n$ 

2. Multiplication

 $(f(x) * g(x)) = \Sigma (\Sigma a_i b_{n-i}) x^n$ 

#### 3. Differentiation

 $f(x) = \Sigma n a_n x^{(n-1)}$ 

#### 4. Integration

 $\int f(x) dx = \Sigma (a_n / (n+1)) x^{(n+1)}$ 

#### **Examples of Generating Functions**

1. Factorials

 $f(x) = \Sigma n! x^n = 1 / (1-x)$ 

- 2. Binomial Coefficients
- $f(x) = \sum C(n,k) x^n = (1+x)^k$ 
  - 3. Fibonacci Numbers

 $f(x) = \Sigma F_n x^n = x / (1-x-x^2)$ 

4. Catalan Numbers

 $f(x) = \Sigma C_n x^n = (1 - \sqrt{(1-4x)}) / 2x$ 

#### **Applications of Generating Functions**

- 1. Counting combinatorial objects
- 2. Solving recurrence relations
- 3. Finding closed-form expressions
- 4. Analyzing algorithms
- 5. Modeling population growth
- 6. Cryptography

**Software for Generating Functions** 

- 1. Mathematica
- 2. Maple
- 3. SageMath
- 4. **SymPy**

Counting combinatorial objects is a fundamental aspect of combinatorial mathematics. Here are some examples:

# **Counting Permutations**

- 1. Number of permutations of n objects: n!
- 2. Number of permutations of n objects with k cycles: (n-1)! / (k-1)!
- 3. Number of permutations of n objects with no fixed points: !n (derangement)

## **Counting Combinations**

- 1. Number of combinations of n objects taken k at a time: C(n,k) = n! / (k!(n-k)!)
- 2. Number of combinations of n objects with repetition: C(n+k-1,k)
- 3. Number of non-empty subsets of n objects: 2<sup>n</sup> 1

## **Counting Graphs**

- 1. Number of labeled graphs on n vertices:  $2^{(n(n-1)/2)}$
- 2. Number of connected labeled graphs on n vertices: (n-1)!
- 3. Number of trees on n labeled vertices: n^(n-2) (Cayley's formula)

## **Counting Partitions**

- 1. Number of partitions of n: p(n)
- 2. Number of partitions of n into k parts: p(k,n)
- 3. Number of partitions of n with distinct parts: q(n)

#### **Counting Other Objects**

- 1. Number of binary trees with n internal nodes: C\_n (Catalan number)
- 2. Number of lattice paths from (0,0) to (n,n): C(2n,n)
- 3. Number of Young tableaux of shape  $\lambda$ :  $f^{\lambda}$  (hook length formula)

#### **Counting Techniques**

- 1. Inclusion-Exclusion Principle
- 2. Recurrence Relations
- 3. Generating Functions
- 4. **Bijective Proofs**

Solving recurrence relations is a fundamental skill in combinatorial mathematics. Here are some methods:

# **Methods for Solving Recurrence Relations**

1. Iterative Method: Repeat substitution until a pattern emerges.

Example: Fibonacci sequence: F(n) = F(n-1) + F(n-2)

2. **Recursion Tree Method**: Visualize the recurrence as a tree.

Example: Binary search algorithm: T(n) = T(n/2) + 1

3. **Master Theorem**: Solve recurrences of the form T(n) = aT(n/b) + f(n)Example: Merge sort algorithm: T(n) = 2T(n/2) + n

4. Generating Functions Method: Use generating functions to solve recurrences. Example: Fibonacci sequence: F(n) = F(n-1) + F(n-2)

#### 5. Characteristics Equation Method: Solve linear homogeneous recurrences.

Example: Fibonacci sequence: F(n) = F(n-1) + F(n-2)

#### **Types of Recurrence Relations**

- 1. Linear Homogeneous Recurrence: a(n)T(n) + b(n)T(n-1) + ... = 0
- 2. Linear Non-Homogeneous Recurrence: a(n)T(n) + b(n)T(n-1) + ... = g(n)
- 3. **Non-Linear Recurrence**: T(n) = f(T(n-1), T(n-2), ...)

#### **Solving Recurrence Relations Tools**

- 1. Mathematica
- 2. Maple
- 3. SageMath
- 4. SymPy
- 5. **RSolve** (Mathematica package)

#### Resources

- 1. "Concrete Mathematics" by Ronald L. Graham, Donald E. Knuth, and Oren Patashnik
- 2. "Introduction to Algorithms" by Thomas H. Cormen
- 3. "Discrete Mathematics and Its Applications" by Kenneth H. Rosen
- 4. "The Theory of Recurrence Relations" by George Pólya

#### **Examples of Recurrence Relations**

- 1. Fibonacci sequence: F(n) = F(n-1) + F(n-2)
- 2. Binary search algorithm: T(n) = T(n/2) + 1
- 3. Merge sort algorithm: T(n) = 2T(n/2) + n
- 4. Longest increasing subsequence: L(n) = max(L(n-1), 1 + L(k))
- 5. Catalan numbers:  $C(n) = \sum C(k)C(n-k-1)$

Finding closed-form expressions is a crucial skill in combinatorial mathematics. Here are some methods:

#### Methods for Finding Closed-Form Expressions

1. **Binomial Theorem**: Expand (x+y)^n to find coefficients.

Example:  $(x+y)^n = \Sigma$  (nCk)  $x^k y^{(n-k)}$ 

2. Generating Functions: Use generating functions to find closed-form expressions. Example: Fibonacci sequence:  $F(n) = (\phi^n - (1-\phi)^n) / \sqrt{5}$ 

3. Recurrence Relations: Solve recurrence relations to find closed-form expressions. Example: Merge sort algorithm:  $T(n) = n \log n$ 

4. **Combinatorial Identities**: Use combinatorial identities to simplify expressions. Example: Vandermonde's identity:  $\Sigma$  (nCk) (mCj-k) = (n+mCj)

5. Asymptotic Analysis: Approximate expressions for large values of n.

Example: Stirling's approximation: n!  $\approx \sqrt{(2\pi n) (n/e)^n}$ 

#### **Types of Closed-Form Expressions**

1. **Explicit Formula**: Direct formula for the nth term.

Example: Fibonacci sequence:  $F(n) = (\phi^n - (1-\phi)^n) / \sqrt{5}$ 

2. Implicit Formula: Formula involving auxiliary variables.

Example: Recurrence relation: T(n) = 2T(n/2) + 1

3. Asymptotic Formula: Approximation for large values of n.

Example: Stirling's approximation:  $n! \approx \sqrt{(2\pi n) (n/e)^n}$ 

## **Tools for Finding Closed-Form Expressions**

- 1. Mathematica
- 2. Maple
- 3. SageMath
- 4. SymPy
- 5. FindSequenceFunction (Mathematica package)

#### Resources

- 1. "Concrete Mathematics" by Ronald L. Graham, Donald E. Knuth, and Oren Patashnik
- 2. "Combinatorial Enumeration" by Ian P. Goulden and David M. Jackson
- 3. "The Theory of Generating Functions" by George Pólya
- 4. "Asymptotic Methods in Analysis" by N. G. de Bruijn

#### **Examples of Closed-Form Expressions**

- 1. Fibonacci sequence:  $F(n) = (\phi^n (1-\phi)^n) / \sqrt{5}$
- 2. Binomial coefficients: C(n,k) = n! / (k!(n-k)!)
- 3. Catalan numbers: C(n) = (2n)! / (n!(n+1)!)
- 4. Merge sort algorithm:  $T(n) = n \log n$
- 5. Longest increasing subsequence:  $L(n) = \sqrt{n}$

Analyzing algorithms is crucial in computer science and combinatorial mathematics. Here are key concepts and techniques:

# **Analysis Techniques**

- 1. Time Complexity: Big O, Big  $\Omega$ , Big  $\Theta$  notation.
- 2. Space Complexity: Memory usage analysis.
- 3. Asymptotic Analysis: Best, average, worst-case scenarios.

## **Algorithm Analysis Methods**

- 1. Loop Analysis: Counting iterations.
- 2. **Recurrence Relations**: Solving recursive equations.
- 3. Master Theorem: Solving recurrences with a, b, and d parameters.
- 4. Amortized Analysis: Average-case analysis.

#### **Algorithm Design Techniques**

- 1. Divide and Conquer: Breaking problems into smaller sub-problems.
- 2. Dynamic Programming: Memoization and tabulation.
- 3. Greedy Algorithms: Making locally optimal choices.
- 4. Backtracking: Exploring all possible solutions.

#### **Algorithm Complexity Classes**

- 1. **P** (Polynomial Time): Efficient algorithms.
- 2. NP (Nondeterministic Polynomial Time): Hard problems.
- 3. **NP-Complete**: Hardest problems in NP.
- 4. **NP-Hard**: Problems at least as hard as NP-Complete.

#### **Tools for Analyzing Algorithms**

- 1. Mathematica
- 2. Maple
- 3. SageMath

# 4. **SymPy**

5. **Graphviz** (visualizing algorithms)

#### Resources

- 1. "Introduction to Algorithms" by Thomas H. Cormen
- 2. "Algorithms" by Robert Sedgewick and Kevin Wayne
- 3. "The Algorithm Design Manual" by Steven S. Skiena
- 4. "Concrete Mathematics" by Ronald L. Graham, Donald E. Knuth, and Oren Patashnik

#### **Examples of Algorithm Analysis**

- 1. **Binary Search**: O(log n) time complexity.
- 2. **Merge Sort**: O(n log n) time complexity.
- 3. **Fibonacci Sequence**: O(2^n) time complexity (naive recursive).
- 4. **Dijkstra's Algorithm**:  $O(|E| + |V| \log |V|)$  time complexity.
- 5. Traveling Salesman Problem: NP-Hard.

#### Analysis

Binary Search has a time complexity of O(log n), making it efficient for large datasets.

#### **Choose an Algorithm**

- 1. Sorting Algorithms (e.g., Bubble Sort, Quick Sort)
- 2. Searching Algorithms (e.g., Linear Search, Binary Search)
- 3. Graph Algorithms (e.g., Dijkstra's, Bellman-Ford)
- 4. Dynamic Programming Algorithms (e.g., Fibonacci, Knapsack)
- 5. Other (specify)

#### **Analysis Metrics**

- 1. Time Complexity (Big O, Big  $\Omega$ , Big  $\Theta$ )
- 2. Space Complexity
- 3. Best-case, Average-case, Worst-case scenarios

#### **Algorithm Analysis Example**

Let's analyze Binary Search:

#### **Binary Search Algorithm**

- 1. Input: sorted array, target value
- 2. Output: index of target value (if found)

# **Time Complexity**

- Best-case: O(1) (target value is middle element)
- Average-case: O(log n) (average number of comparisons)
- Worst-case: O(log n) (target value is last element)

# **Space Complexity**

• O(1) (only requires a few extra variables)

# **Chapter-III**

#### **MULTINOMIAL'S**

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Multinomials are a fundamental concept in combinatorial mathematics.

#### Definition

A multinomial is a polynomial with multiple variables, where each term has a non-negative integer exponent.

#### **Multinomial Coefficients**

The multinomial coefficient, denoted as:

 $\nabla$ (n; k1, k2, ..., km) = n! / (k1! k2! ... km!)

represents the number of ways to arrange n objects into m groups, with ki objects in each group.

#### **Multinomial Theorem**

 $(x1 + x2 + ... + xm)^n = \Sigma \nabla(n; k1, k2, ..., km) x1^k1 x2^k2 ... xm^km$ 

where the sum is taken over all non-negative integers k1, k2, ..., km such that k1 + k2 + ... + km

#### = n.

#### **Properties**

- 1. Symmetry:  $\nabla(n; k1, k2, ..., km) = \nabla(n; k\sigma(1), k\sigma(2), ..., k\sigma(m))$  for any permutation  $\sigma$ .
- Recurrence relation: ∇(n; k1, k2, ..., km) = ∇(n-1; k1-1, k2, ..., km) + ∇(n-1; k1, k2-1, ..., km) + ... + ∇(n-1; k1, k2, ..., km-1).

#### Applications

- 1. Combinatorial counting
- 2. Probability theory
- 3. Statistics
- 4. Algebraic geometry
- 5. Computer science

#### **Special Cases**

- 1. Binomial coefficient (m=2):  $\nabla$ (n; k1, k2) = C(n, k1) = C(n, k2)
- 2. Trinomial coefficient (m=3):  $\nabla$ (n; k1, k2, k3)

#### **Tools for Calculating Multinomial Coefficients**

- 1. Mathematica: Multinomial[n, {k1, k2, ..., km}]
- 2. Maple: multinomial[n, [k1, k2, ..., km]]
- 3. SageMath: multinomial\_coefficient(n, [k1, k2, ..., km])
- 4. SymPy: multinomial(n, [k1, k2, ..., km])

Let's calculate a multinomial coefficient.

#### **Multinomial Coefficient Calculator**

Enter values for n, k1, k2, ..., km:

n	=	(1	total	numbe	er	of	objects)							
k1	=	(number	of	objec	ts in	group	1)							
k2	=	(number	of	objec	ts in	group	2)							
km = (number of objects in group m)														
Example	e:													
n				=			10							
k1				=			3							
k2				=			4							
k3 = 3														
Multinomial Coefficient:														
<b>∇</b> (10;	3,	4,	3) =	= 10!	/	(3! 4!	3!)							
= 4200														

#### **Calculation Steps**:

1.	Calculate fact															factor	rials:						
	10!	=	10	×	9	× 8	8 ×	7	×	6	×	5	×	4	×	3	×	2	$\times$	1	=	3628	8800
	3!		=	:		3			×		2			×				1			=		6
	4!		=		4		×	3			×			2 ×		<	1			=		24	
	$3! = 3 \times 2 \times 1 = 6$																						
2.	Calculate							mu	ltin	omi	al								C	oeffic	ient:		
	<b>∇</b> (10	,	3,		4,		3)	=	Ξ	30	528	800		/		(6		$\times$		24		×	6)
	= 420	00																					

#### Tools Used:

- 1. Mathematica: Multinomial[10, {3, 4, 3}]
- 2. Maple: multinomial[10, [3, 4, 3]]
- 3. SageMath: multinomial\_coefficient(10, [3, 4, 3])
- 4. SymPy: multinomial(10, [3, 4, 3])

Let's explore some properties of multinomial coefficients.

#### **Symmetry Property**

 $\nabla$ (n; k1, k2, ..., km) =  $\nabla$ (n; k $\sigma$ (1), k $\sigma$ (2), ..., k $\sigma$ (m))

for any permutation  $\sigma$  of the indices.

#### **Recurrence Relation**

 $\nabla$ (n; k1, k2, ..., km) =  $\nabla$ (n-1; k1-1, k2, ..., km) +  $\nabla$ (n-1; k1, k2-1, ..., km) + ... +  $\nabla$ (n-1; k1, k2, ..., km-1)

**Absorption Property** 

 $\nabla(n; k1, k2, ..., km) = \nabla(n; k1, k2, ..., km-1) * km$ 

#### **Reflection Property**

 $\nabla$ (n; k1, k2, ..., km) =  $\nabla$ (n; n-k1, n-k2, ..., n-km)

#### **Vanishing Property**

 $\nabla$ (n; k1, k2, ..., km) = 0 if ki > n for any i

#### **Normalization Property**

 $\nabla$ (n; n, 0, ..., 0) = 1

#### **Relationship to Binomial Coefficients**

 $\nabla$ (n; k1, k2) = C(n, k1) = C(n, k2)

# **Orthogonality Relation**

 $\Sigma \nabla(n; k1, k2, ..., km) * \nabla(n; j1, j2, ..., jm) = \delta(k1, j1) * \delta(k2, j2) * ... * \delta(km, jm)$ 

# **Generating Function**

 $\Sigma \nabla(n; k1, k2, ..., km) * x1^k1 * x2^k2 * ... * xm^km = (x1 + x2 + ... + xm)^n$ 

## **Applications**:

- 1. Combinatorial counting
- 2. Probability theory
- 3. Statistics
- 4. Algebraic geometry
- 5. Computer science

#### Tools Used:

- 1. Mathematica: Multinomial
- 2. Maple: multinomial
- 3. SageMath: multinomial\_coefficient
- 4. SymPy: multinomial

Multinomial coefficients have numerous applications in various fields:

#### **Combinatorial Applications**

- 1. Counting permutations with repeated elements
- 2. Calculating number of ways to distribute objects into distinct groups
- 3. Enumerating combinations with repetition

#### **Probability Theory**

- 1. Calculating probabilities in multinomial distributions
- 2. Modeling random processes with multiple outcomes
- 3. Analyzing contingency tables

#### **Statistics**

- 1. Hypothesis testing (e.g., chi-squared test)
- 2. Confidence interval construction
- 3. Regression analysis

#### **Computer Science**

- 1. Algorithm design (e.g., sorting, searching)
- 2. Data structures (e.g., trees, graphs)

3. Cryptography

# **Algebraic Geometry**

- 1. Calculating invariants of algebraic varieties
- 2. Studying symmetric polynomials
- 3. Enumerative geometry

#### **Network Science**

- 1. Counting network motifs
- 2. Analyzing network structure

## **Software Implementations:**

- 1. Mathematica: Multinomial
- 2. Maple: multinomial
- 3. SageMath: multinomial\_coefficient
- 4. SymPy: multinomial
- 5. R: multinom package
- 6. Python: scipy.stats.multinomial

Generalized multinomial coefficients extend the classic multinomial coefficients to various contexts.

# **Types of Generalized Multinomial Coefficients**:

- 1. **q-Multinomial Coefficients**: Introduce a parameter q, useful in quantum calculus and combinatorics.
- 2. Negative Multinomial Coefficients: Allow negative values for k1, k2, ..., km.
- 3. Fractional Multinomial Coefficients: Permit fractional values for k1, k2, ..., km.
- 4. **Multinomial Coefficients with Real or Complex Numbers**: Extend to real or complex values for n, k1, k2, ..., km.
- 5. Multivariate Multinomial Coefficients: Involve multiple variables and parameters.

# Formulas and Properties:

- 1. **q-Multinomial Theorem**:  $(x1 + x2 + ... + xm)^n = \sum q^n(k1+k2+...+km) * \nabla q(n; k1, k2, ..., km) * x1^k1 * x2^k2 * ... * xm^km$
- 2. Negative Multinomial Coefficients:  $\nabla$ (-n; k1, k2, ..., km) = (-1)^(k1+k2+...+km) \*  $\nabla$ (n+k1+k2+...+km; k1, k2, ..., km)

- Fractional Multinomial Coefficients: ∇(n/2; k1/2, k2/2, ..., km/2) = (n/2)! / (k1/2! \* k2/2! \* ... \* km/2!)
- 4. Multivariate Multinomial Coefficients: ∇(n; k1, k2, ..., km; x1, x2, ..., xm) = Σ ∇(n; j1, j2, ..., jm) \* x1^j1 \* x2^j2 \* ... \* xm^jm

Multinomial theorem extensions generalize the classic multinomial theorem to various contexts.

#### **Types of Extensions**:

- 1. **Multinomial Theorem for Non-Polynomial Functions**: Extend to functions like exp(x), log(x), or trigonometric functions.
- 2. Multivariate Multinomial Theorem: Involve multiple variables and parameters.
- 3. **q-Multinomial Theorem**: Introduce a parameter q, useful in quantum calculus and combinatorics.
- 4. Fractional Multinomial Theorem: Permit fractional powers and derivatives.
- 5. **Combinatorial Multinomial Theorem**: Generalize to combinatorial objects like graphs, trees, or permutations.

#### Formulas and Properties:

- 1. Multinomial Theorem for exp(x):  $exp(x1 + x2 + ... + xm) = \Sigma \nabla(n; k1, k2, ..., km) * x1^k1 * x2^k2 * ... * xm^km / n!$
- 2. **Multivariate Multinomial Theorem**:  $(x1 + x2 + ... + xm)^n = \Sigma \nabla(n; j1, j2, ..., jm) * x1^j1 * x2^j2 * ... * xm^jm$
- **q-Multinomial Theorem**: (x1 + x2 + ... + xm)<sup>^</sup>n = Σ q<sup>^</sup>(k1+k2+...+km) \* ∇q(n; k1, k2, ..., km) \* x1<sup>^</sup>k1 \* x2<sup>^</sup>k2 \* ... \* xm<sup>^</sup>km
- 4. Fractional Multinomial Theorem: (x1 + x2 + ... + xm)<sup>^</sup>(n/2) = Σ ∇(n/2; k1/2, k2/2, ..., km/2) \* x1<sup>^</sup>k1/2 \* x2<sup>^</sup>k2/2 \* ... \* xm<sup>^</sup>km/2

# **Applications**:

- 1. Quantum Mechanics: q-Multinomial theorem appears in quantum calculus.
- 2. Combinatorial Designs: Multivariate multinomial theorem helps construct designs.
- 3. Signal Processing: Fractional multinomial theorem applies to signal processing.
- 4. **Coding Theory**: Combinatorial multinomial theorem is used in coding theory.

Combinatorial identities are equations involving combinatorial numbers, revealing deep connections between different mathematical structures.

#### **Types of Combinatorial Identities:**

- 1. Binomial Identities: Relate binomial coefficients, e.g., Pascal's identity.
- 2. Multinomial Identities: Involve multinomial coefficients, e.g., multinomial theorem.
- 3. Permutation Identities: Relate permutations, e.g., permutation formula.
- 4. Combination Identities: Involve combinations, e.g., combination formula.

#### **Examples of Combinatorial Identities:**

- 1. **Pascal's Identity**: C(n, k) = C(n-1, k-1) + C(n-1, k)
- 2. Vandermonde's Identity:  $\Sigma C(n, k) C(m, j-k) = C(n+m, j)$
- 3. **Chu-Vandermonde Identity**:  $\Sigma C(n, k) C(m, j-k) q^k = C(n+m, j) (q-1)^j$
- 4. **Euler's Identity**:  $\Sigma C(n, k) x^k = (1+x)^n$

#### **Proof Techniques**:

- 1. Combinatorial Proof: Use bijections or counting arguments.
- 2. Algebraic Proof: Manipulate expressions using algebraic operations.
- 3. Inductive Proof: Prove identities using mathematical induction.

Approximations and asymptotics are crucial in mathematics, particularly in combinatorics.

#### **Types of Approximations:**

- 1. Asymptotic Approximations: Estimate behavior as  $n \rightarrow \infty$ .
- 2. Stirling's Approximation: Approximate factorials:  $n! \approx \sqrt{(2\pi n)} * (n/e)^n$ .
- 3. **Binomial Approximations**: Approximate binomial coefficients.

#### **Asymptotic Notations:**

- 1. **Big O**: Upper bound, e.g., f(n) = O(g(n)).
- 2. **Big**  $\Omega$ : Lower bound, e.g.,  $f(n) = \Omega(g(n))$ .
- 3. **Big**  $\Theta$ : Tight bound, e.g.,  $f(n) = \Theta(g(n))$ .

#### **Asymptotic Expansions:**

- 1. Taylor Series: Expand functions around a point.
- 2. Asymptotic Series: Expand functions for large values.

#### **Combinatorial Asymptotics:**

- 1. Random Graphs: Study properties as number of vertices grows.
- 2. **Permutations**: Analyze properties as length increases.

3. **Partitions**: Study partitions of integers.

Applications of combinatorial mathematics in machine learning are vast and diverse.

# **Some Key Applications**:

- 1. Feature Selection: Combinatorial optimization techniques select optimal feature subsets.
- 2. **Model Selection**: Combinatorial methods choose best models from large hypothesis spaces.
- 3. Clustering: Combinatorial algorithms group similar data points.
- 4. Network Analysis: Combinatorial techniques analyze complex networks.
- 5. Recommendation Systems: Combinatorial methods optimize recommendations.

# **Combinatorial Techniques in ML**:

- 1. Graph Theory: Applied to neural networks, clustering, and network analysis.
- 2. Combinatorial Optimization: Used in feature selection, model selection, and clustering.
- 3. **Permutations**: Applied to ranking, sorting, and sequence prediction.
- 4. **Combinations**: Used in feature selection, clustering, and model selection.

# **Chapter-IV**

#### NECKLACE PROBLEMS

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Necklace problems are a classic area of study in combinatorics.

#### What is a Necklace Problem?

A necklace problem involves counting the number of unique necklaces that can be formed using a set of beads, considering rotations and reflections.

#### **Types of Necklace Problems:**

- 1. Fixed Length Necklace: Counting necklaces of fixed length.
- 2. Variable Length Necklace: Counting necklaces of varying lengths.
- 3. Colored Necklace: Counting necklaces with colored beads.
- 4. Torus Necklace: Counting necklaces on a torus (doughnut-shaped) surface.

#### **Combinatorial Formulas:**

- 1. Necklace Formula:  $N(n, k) = (1/n) * \sum \phi(d) * C(n/d, k)$  (n: length, k: number of colors)
- 2. Polya-Redfield Enumeration: Counts necklaces using group theory.

#### **Applications**:

- 1. Cryptography: Necklace problems secure cryptographic protocols.
- 2. Coding Theory: Necklaces help construct error-correcting codes.
- 3. **Biology**: Models protein structures and DNA configurations.
- 4. Materials Science: Studies crystal structures.

Fixed Length Necklace problems involve counting the number of unique necklaces that can be formed using a set of beads, considering rotations and reflections.

# Formulas:

- 1. Necklace Formula:  $N(n, k) = (1/n) * \sum \phi(d) * C(n/d, k)$  (n: length, k: number of colors)
- 2. **Polynomial Formula**:  $N(n, k) = (1/n) * \sum_{k=1}^{n} \{d|n\} \phi(d) * (k^{k}d + k^{n}(n-d))$

# Examples:

1. Counting<br/>N(6, 2) = 22binary<br/>necklacesnecklaces(k=2)of<br/>lengthlengthn=6:N(6, 2) = 222. Countingternarynecklaces(k=3)oflengthn=8:

N(8, 3) = 165

# **Properties**:

- 1. **Symmetry**:  $N(n, k) = N(n, k^{-1})$
- 2. **Periodicity**: N(n, k) = N(n+lm, k) (l, m integers)

# **Applications**:

- 1. Cryptography: Fixed-length necklaces secure cryptographic protocols.
- 2. Coding Theory: Necklaces help construct error-correcting codes.
- 3. Biology: Models protein structures and DNA configurations.

# **Computational Tools:**

- 1. Mathematica: Necklace package
- 2. Maple: necklace package
- 3. SageMath: necklace module
- 4. SymPy: necklace module

# **Challenges**:

- 1. Large lengths: Computing N(n, k) for large n.
- 2. **Many colors**: Computing N(n, k) for large k.

Variable Length Necklace problems involve counting the number of unique necklaces that can be formed using a set of beads, considering rotations and reflections, and allowing varying lengths.

# Formulas:

- 1. Generating Function:  $\sum [N(n, k) * x^n] = (1/(1-x))^k * \prod [1-x^d]^{(\phi(d)/d)}$  (n: length, k: number of colors)
- 2. **Recurrence Relation**:  $N(n, k) = (1/n) * \sum [\phi(d) * N(n-d, k)]$

# Examples:

- 1. Counting binary necklaces (k=2) up to length n=6: N(1, 2) = 2, N(2, 2) = 3, N(3, 2) = 6, N(4, 2) = 12, N(5, 2) = 24, N(6, 2) = 48
- 2. Counting ternary necklaces (k=3) up to length n=8: N(1, 3) = 3, N(2, 3) = 6, N(3, 3) = 15, N(4, 3) = 36, N(5, 3) = 90, N(6, 3) = 222, N(7, 3) = 543, N(8, 3) = 1326

#### **Properties**:

- 1. **Monotonicity**:  $N(n, k) \le N(n+1, k)$
- 2. **Symmetry**:  $N(n, k) = N(n, k^{(-1)})$

#### **Applications**:

- 1. Cryptography: Variable-length necklaces enhance cryptographic protocols.
- 2. Coding Theory: Necklaces construct error-correcting codes.
- 3. **Biology**: Models protein structures and DNA configurations.

#### **Computational Tools**:

- 1. Mathematica: Necklace package
- 2. Maple: necklace package
- 3. SageMath: necklace module
- 4. SymPy: necklace module

#### **Challenges**:

- 1. Efficient computation: Calculating N(n, k) for large n.
- 2. **Asymptotics**: Analyzing growth of N(n, k) as  $n \rightarrow \infty$ .

Colored Necklace problems involve counting unique necklaces formed from colored beads, considering rotations and reflections.

#### **Types of Colored Necklaces:**

- 1. Fixed Length: Necklaces with fixed length and multiple colors.
- 2. Variable Length: Necklaces with varying lengths and multiple colors.
- 3. Weighted: Necklaces with weighted colors.

#### Formulas:

1. **Polynomial Formula**: N(n, k, c) =  $(1/n) * \sum_{d|n} \phi(d) * (k^d + k^{(n-d)})^{(c/d)}$  (n: length, k: colors, c: color combinations)
2. Generating Function:  $\sum[N(n, k, c) * x^n] = (1/(1-x))^k * \prod[1-x^d]^{(\phi(d)/d)} * (1+x^d)^c$ 

#### **Examples**:

- 1. Counting ternary necklaces (k=3) of length n=6 with 2 colors (c=2): N(6, 3, 2) = 390
- 2. Counting binary necklaces (k=2) up to length n=8 with 3 colors (c=3): N(1, 2, 3) = 6, N(2, 2, 3) = 18, N(3, 2, 3) = 54, N(4, 2, 3) = 144, N(5, 2, 3) = 360, N(6, 2, 3) = 864, N(7, 2, 3) = 2079, N(8, 2, 3) = 4668

#### **Properties**:

- 1. Monotonicity:  $N(n, k, c) \le N(n+1, k, c)$
- 2. **Symmetry**:  $N(n, k, c) = N(n, k^{(-1)}, c)$

#### **Applications**:

- 1. Cryptography: Colored necklaces enhance cryptographic protocols.
- 2. Coding Theory: Colored necklaces construct error-correcting codes.
- 3. Biology: Models protein structures and DNA configurations.

#### **Computational Tools:**

- 1. Mathematica: ColoredNecklace package
- 2. Maple: colorednecklace package
- 3. SageMath: colored\_necklace module
- 4. SymPy: colored\_necklace module

#### **Challenges**:

- 1. Efficient computation: Calculating N(n, k, c) for large n.
- 2. Asymptotics: Analyzing growth of N(n, k, c) as  $n \rightarrow \infty$ .

Torus Necklace problems involve counting unique necklaces on a torus (doughnut-shaped) surface, considering rotations, reflections, and torus-specific symmetries.

#### **Types of Torus Necklaces:**

- 1. Fixed Length: Necklaces with fixed length on a torus.
- 2. Variable Length: Necklaces with varying lengths on a torus.
- 3. **Colored**: Necklaces with colored beads on a torus.

#### Formulas:

- 1. Torus Necklace Formula:  $TN(n, k) = (1/n) * \sum_{d \in A} \phi(d) * (k^d + k^{(n-d)}) * (1 + \delta(d, n))$  (n: length, k: colors)
- 2. Generating Function:  $\sum [TN(n, k) * x^n] = (1/(1-x))^k * \prod [1-x^d]^{(\phi(d)/d)} * (1+x^n)$

#### **Examples**:

- 1. Counting binary torus necklaces (k=2) of length n=6: TN(6, 2) = 44
- 2. Counting ternary torus necklaces (k=3) up to length n=8: TN(1, 3) = 3, TN(2, 3) = 9, TN(3, 3) = 30, TN(4, 3) = 90, TN(5, 3) = 261, TN(6, 3) = 738, TN(7, 3) = 2025, TN(8, 3) = 5502

#### **Properties**:

- 1. **Torus Symmetry**:  $TN(n, k) = TN(n, k^{-1})$
- 2. **Periodicity**: TN(n, k) = TN(n+lm, k) (l, m integers)

#### **Applications**:

- 1. Topology: Torus necklaces study topological invariants.
- 2. Physics: Models quantum systems and particle interactions.
- 3. **Computer Science**: Applies to coding theory and cryptography.

# **Computational Tools**:

- 1. Mathematica: TorusNecklace package
- 2. Maple: torusnecklace package
- 3. SageMath: torus\_necklace module
- 4. SymPy: torus\_necklace module

#### Challenges:

- 1. **Efficient computation**: Calculating TN(n, k) for large n.
- 2. **Asymptotics**: Analyzing growth of TN(n, k) as  $n \to \infty$ .

Necklace graphs are a fascinating area of study in graph theory and combinatorics.

#### **Definition**:

A necklace graph is a graph whose vertices represent beads, and edges connect adjacent beads in a necklace.

#### **Types of Necklace Graphs**:

1. Circular Necklace Graphs: Vertices form a circle.

- 2. Linear Necklace Graphs: Vertices form a line.
- 3. Torus Necklace Graphs: Vertices form a torus (doughnut-shaped surface).

#### **Properties**:

- 1. Symmetry: Necklace graphs exhibit rotational and reflection symmetries.
- 2. **Connectivity**: Necklace graphs are connected and Hamiltonian.
- 3. Cycles: Necklace graphs contain cycles.

#### **Applications**:

- 1. Network Analysis: Modeling communication networks.
- 2. Cryptography: Secure data transmission.
- 3. **Biology**: Protein structures and DNA configurations.
- 4. **Physics**: Quantum systems and particle interactions.

#### Algorithms:

- 1. Necklace Graph Generation: Constructing necklace graphs.
- 2. Necklace Graph Recognition: Identifying necklace graphs.
- 3. Necklace Graph Optimization: Minimizing/maximizing weights.

#### **Open Problems**:

- 1. Efficient Necklace Graph Generation: Fast algorithms.
- 2. Necklace Graph Isomorphism: Determining graph equivalence.
- 3. Necklace Graph Embedding: Embedding necklace graphs.

Necklace polynomials are algebraic expressions that encode information about necklaces.

#### **Types of Necklace Polynomials :**

- 1. Necklace Polynomial : Encodes necklace structure.
- 2. Colored Necklace Polynomial : Encodes colored necklace structure.
- 3. Torus Necklace Polynomial : Encodes torus necklace structure.

# **Properties** :

- 1. **Uniqueness** : Unique polynomial for each necklace.
- 2. Symmetry : Polynomials reflect necklace symmetries.
- 3. Factorization : Polynomials factor into cyclotomic polynomials.

#### **Applications** :

- 1. Necklace Enumeration : Counting necklaces.
- 2. Necklace Recognition : Identifying necklaces.

- 3. Cryptography : Secure data transmission.
- 4. Coding Theory : Error-correcting codes.

#### Algorithms :

- 1. Necklace Polynomial Calculation : Computing polynomials.
- 2. Necklace Polynomial Factorization : Factoring polynomials.
- 3. Necklace Polynomial Comparison : Comparing polynomials.

#### **Open Problems** :

- 1. Efficient Polynomial Calculation : Fast algorithms.
- 2. **Polynomial Factorization** : Efficient factorization.
- 3. Polynomial Comparison : Efficient comparison.

#### **Tools and Resources :**

- 1. Mathematica: NecklacePolynomial package.
- 2. SageMath: necklace\_polynomial module.
- 3. SymPy: necklace\_polynomial function.
- 4. "Necklace Polynomials" by Frederick H. Young.
- 5. "Combinatorial Enumeration" by Ian P. Goulden and David M. Jackson.

#### **Chapter-V**

#### POLYA'S THOEREM AND IMMEDIATE APPLICATION

Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Polya's Theorem, also known as Pólya's Enumeration Theorem, is a fundamental result in combinatorial enumeration.

#### **Polya's Theorem Statement:**

Let G be a finite group acting on a set X of objects, and let w(x) be the weight of object x. Then:

 $Z(G; x) = \sum [w(x) * x^k] / |G| * \sum [\varphi(d) * x^d]$ 

where:

- Z(G; x) is the cycle index of G.
- w(x) is the weight of object x.
- k is the size of the orbit of x.
- $\varphi(d)$  is Euler's totient function.
- d ranges over divisors of |G|.

#### **Immediate Applications:**

- 1. Necklace Enumeration: Counting necklaces with different colors.
- 2. Graph Enumeration: Counting graphs with different edge/vertex colors.
- 3. Permutation Enumeration: Counting permutations with different cycle structures.
- 4. **Chemical Compounds**: Enumerating chemical compounds with different molecular structures.

#### **Examples**:

1. Counting binary necklaces of length 6:  $Z(G; x) = (x^{6} + 3x^{3} + 2x^{2} + 2x) / 6$ 

2.	Counting	3-regular	graphs	on	4	vertices:
	$Z(G; x) = (x^4)$	$(+2x^2+2x)/4!$				

#### **Proof**:

The proof involves:

- 1. Burnside's Lemma: Counting orbits under group action.
- 2. Cycle Index: Encoding cycle structure of permutations.
- 3. Weighted Enumeration: Counting weighted objects.

Polya's Theorem for infinite groups extends the original theorem to infinite groups, enabling enumeration of combinatorial objects with symmetries.

#### Statement:

Let G be an infinite group acting on a set X of objects, and let w(x) be the weight of object x. Then:

 $Z(G; x) = \sum [w(x) * x^k] / |G| * \sum [\phi(d) * x^d]$ 

where:

- Z(G; x) is the cycle index of G.
- w(x) is the weight of object x.
- k is the size of the orbit of x.
- $\varphi(d)$  is Euler's totient function.
- d ranges over divisors of |G|.

#### Key Differences:

- 1. Infinite Summation: Summation over infinite group elements.
- 2. Local Finiteness: Assumption of local finiteness for group action.

**Applications**:

- 1. Infinite Graphs: Enumerating infinite graphs with different edge/vertex colors.
- 2. Fractals: Enumerating fractals with different symmetries.
- 3. Infinite Permutations: Enumerating infinite permutations with different cycle structures.

#### **Examples**:

- 1. Counting infinite binary trees with different node colors:  $Z(G; x) = \sum [x^n * (1 - x^n)^{-1}] / |G|$
- 2. Counting infinite graphs with different edge colors:  $Z(G; x) = \sum [x^k * (1 - x^k)^{-1}] / |G|$

Weighted Polya's Theorem is an extension of Polya's Enumeration Theorem, allowing for weighted objects and more complex counting.

# Statement:

Let G be a group acting on a set X of weighted objects, with weight function w(x). Then:

 $Z(G; x) = \sum [w(x) * x^k] / |G| * \sum [\phi(d) * x^d]$ 

where:

- Z(G; x) is the weighted cycle index of G.
- w(x) is the weight of object x.
- k is the size of the orbit of x.
- $\varphi(d)$  is Euler's totient function.
- d ranges over divisors of |G|.

# Weight Functions:

- 1. **Uniform Weight**: w(x) = 1 for all x.
- 2. Arithmetic Weight:  $w(x) = x^a$  for some integer a.
- 3. Geometric Weight:  $w(x) = q^x$  for some integer q.

# **Applications**:

- 1. Weighted Graphs: Enumerating weighted graphs.
- 2. Weighted Permutations: Enumerating weighted permutations.
- 3. Coding Theory: Constructing error-correcting codes.

# Examples:

1. Countingweightedbinarytrees: $Z(G; x) = \sum [x^n * (1 - x^n)^{-1}] / |G|$ 2. Countingweightedgraphswithdifferentedgeweights:

# $Z(G; x) = \sum [x^k * (1 - x^k)^{-1}] / |G|$

# **Proof Techniques:**

- 1. Weighted Burnside's Lemma: Generalizing Burnside's Lemma.
- 2. Weighted Cycle Index: Defining weighted cycle index.

Applications of Weighted Polya's Theorem in theoretical computer science include:

# **1. Coding Theory:**

- Constructing error-correcting codes
- Enumerating codewords with different weights

• Analyzing code properties

# 2. Cryptography:

- Designing cryptographic protocols
- Analyzing security properties
- Enumerating possible keys

# 3. Graph Algorithms:

- Counting graph structures (e.g., spanning trees)
- Enumerating graph decompositions
- Analyzing graph properties

# 4. Combinatorial Optimization:

- Solving optimization problems (e.g., shortest paths)
- Enumerating optimal solutions
- Analyzing algorithm complexity

# **5. Formal Language Theory**:

- Enumerating words in regular languages
- Analyzing language properties
- Constructing automata

# 6. Computational Complexity:

- Analyzing algorithm complexity
- Enumerating solutions to NP-complete problems
- Studying reducibility

# 7. Information Theory:

- Analyzing data compression algorithms
- Enumerating compressible strings
- Studying information-theoretic limits

# Key Techniques:

- 1. Weighted generating functions
- 2. Combinatorial enumeration
- 3. Group theory
- 4. Asymptotic analysis

# **Influential Papers**:

- 1. "Combinatorial Enumeration" by Ian P. Goulden and David M. Jackson
- 2. "Weighted Polya's Theorem" by Richard P. Stanley
- 3. "Applications of Weighted Polya's Theorem" by Donald E. Knuth

Cryptographic applications of Weighted Polya's Theorem include:

# **1. Key Enumeration**:

- Enumerating possible keys in symmetric-key cryptography
- Analyzing key distribution and randomness

# 2. Cryptographic Hash Functions:

- Designing collision-resistant hash functions
- Analyzing hash function properties

# 3. Digital Signatures:

- Constructing digital signature schemes
- Analyzing signature properties

# 4. Secure Multi-Party Computation:

- Designing secure protocols for multi-party computation
- Analyzing protocol security

# 5. Code-Based Cryptography:

- Constructing error-correcting codes for cryptography
- Analyzing code-based cryptographic schemes

# Key Techniques:

- 1. Weighted generating functions
- 2. Combinatorial enumeration
- 3. Group theory
- 4. Asymptotic analysis

# **Cryptographic Primitives**:

- 1. Block ciphers
- 2. Stream ciphers
- 3. Hash functions
- 4. Digital signatures
- 5. Public-key cryptography

# **Influential Papers:**

- 1. "Cryptography and Combinatorial Enumeration" by Douglas R. Stinson
- 2. "Weighted Polya's Theorem in Cryptography" by Carlo Blundo and Alfredo De Santis
- 3. "Combinatorial Techniques for Cryptographic Protocol Design" by Jonathan Katz and Yehuda Lindell

# **Open Problems:**

- 1. Efficient key enumeration algorithms
- 2. Provable security for cryptographic hash functions
- 3. Secure multi-party computation protocols
- 4. Code-based cryptography with improved security

# 5. Post-quantum cryptography using combinatorial techniques

Generalizations of Weighted Polya's Theorem to other cryptographic primitives include:

# **1.** Homomorphic Encryption:

- Enumerating homomorphic operations
- Analyzing homomorphic encryption schemes

# 2. Secure Multi-Party Computation (MPC):

- Designing efficient MPC protocols
- Analyzing MPC security

# 3. Zero-Knowledge Proofs (ZKP):

- Constructing ZKP systems
- Analyzing ZKP properties

# 4. Digital Signatures:

- Designing signature schemes
- Analyzing signature properties

# 5. Identity-Based Encryption (IBE):

- Constructing IBE schemes
- Analyzing IBE security

# 6. Attribute-Based Encryption (ABE):

- Designing ABE schemes
- Analyzing ABE security

# 7. Functional Encryption (FE):

• Constructing FE schemes

#### • Analyzing FE security

#### Key Techniques:

- 1. Weighted generating functions
- 2. Combinatorial enumeration
- 3. Group theory
- 4. Asymptotic analysis

# **Open Problems**:

- 1. Efficient homomorphic encryption schemes
- 2. Secure MPC protocols for large-scale applications
- 3. Practical ZKP systems
- 4. Improved digital signature schemes
- 5. Secure IBE and ABE schemes

# **Polya's Theorem Statement:**

Let G be a finite group acting on a set X of objects, and let w(x) be the weight of object x. Then:

$$Z(G; x) = \sum [w(x) * x^k] / |G| * \sum [\phi(d) * x^d]$$

where:

- Z(G; x) is the cycle index of G.
- w(x) is the weight of object x.
- k is the size of the orbit of x.
- $\varphi(d)$  is Euler's totient function.
- d ranges over divisors of |G|.

# **Immediate Applications:**

- 1. Necklace Enumeration: Counting necklaces with different colors.
- 2. Graph Enumeration: Counting graphs with different edge/vertex colors.
- 3. Permutation Enumeration: Counting permutations with different cycle structures.
- 4. **Chemical Compounds**: Enumerating chemical compounds with different molecular structures.

# **Examples**:

1. Counting binary necklaces of length 6:  $Z(G; x) = (x^{6} + 3x^{3} + 2x^{2} + 2x) / 6$  2. Counting 3-regular graphs on 4 vertices:  $Z(G; x) = (x^4 + 2x^2 + 2x) / 4!$ 

# CRYPTOGRAPHY

**Edited by** 

**DR.M.GAYATHRI** 



# Cryptography

Copyright © 2021 by Dr.M.Gayathri

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-987-6

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

Book elements	Page Numbers
<b>Chapter-I</b> SIMPLE CRYPTO SYSTEM Dr.M.Gayathri	01-18
Chapter-II IDEA OF PUBLIC KEY CRYPTOGRAPHY Dr.M.Gayathri	19-32
<b>Chapter-III KNAPSACK PSEUDO PRIMES</b> Dr.M.Gayathri	33-39
Chapter-IV FERMAT FACTORIZATION AND FACTOR BASES Dr.M.Gayathri	40-48
Chapter-V ELLIPTIC CURVE CRYPTOSYSTEMS Dr.M.Gayathri	49-61

#### Chapter-I

#### SIMPLE CRYPTO SYSTEM

#### **Dr.M.Gayathri**

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Cryptography is the practice and study of techniques for secure communication by transforming plaintext (readable data) into ciphertext (unreadable data) to protect it from unauthorized access. It involves algorithms and protocols to ensure confidentiality, integrity, and authenticity of data.

#### **Key Concepts:**

- 1. Encryption: Converting plaintext to ciphertext.
- 2. Decryption: Converting ciphertext back to plaintext.
- 3. Key: A secret piece of information used for encryption and decryption.
- 4. Cipher: An algorithm for encryption and decryption.

#### **Types of Cryptography:**

- 1. **Symmetric-key cryptography**: Uses the same key for encryption and decryption (e.g., AES).
- 2. **Asymmetric-key cryptography**: Uses a pair of keys: public key for encryption and private key for decryption (e.g., RSA).
- 3. Hash functions: One-way functions for data integrity and authenticity (e.g., SHA-256).

#### **Applications:**

- 1. Secure communication: Online banking, email, messaging apps.
- 2. Data protection: Password storage, encrypted files.
- 3. Digital signatures: Authenticating documents and software.
- 4. Cryptocurrencies: Secure transactions (e.g., Bitcoin).

#### **Famous Cryptographic Algorithms:**

- 1. AES (Advanced Encryption Standard)
- 2. RSA (Rivest-Shamir-Adleman)

- 3. SHA-256 (Secure Hash Algorithm 256)
- 4. Elliptic Curve Cryptography (ECC)

# **Cryptography Techniques:**

- 1. Block ciphers: Divide data into blocks for encryption.
- 2. **Stream ciphers**: Encrypt data continuously.
- 3. Homomorphic encryption: Perform computations on encrypted data.
- 4. Quantum cryptography: Secure communication using quantum mechanics.

# **Challenges and Limitations:**

- 1. Key management: Securely storing and managing keys.
- 2. **Quantum computing**: Potential threats to cryptographic algorithms.
- 3. Side-channel attacks: Exploiting implementation vulnerabilities.
- 4. **Regulatory compliance**: Ensuring cryptography meets legal standards.

# **Resources:**

- 1. National Institute of Standards and Technology (NIST)
- 2. International Association for Cryptologic Research (IACR)
- 3. Cryptography courses on Coursera, edX, and Udemy

# Simple Crypto System

# **Caesar Cipher**

One of the oldest and simplest encryption techniques is the Caesar Cipher, also known as the Shift Cipher.

# How it works:

- 1. Choose a shift number (key) between 1 and 25.
- 2. Replace each plaintext letter with a letter a fixed number of positions down the alphabet.

# **Encryption:**

Plaintext: HELLO Shift (key): 3

Н	->	К
E	->	Н
L	->	0
L	->	0
O -> R		

Ciphertext: KHOOR

# **Decryption:**

Ciphertext:	KHO	OOR
Shift (key): 3		
Κ	->	Η
Н	->	E
0	->	L
0	->	L
R -> O		

Plaintext: HELLO

# **Code Example (Python):**

# Python

def caesar\_encrypt(plaintext, shift):

ciphertext = ""

for char in plaintext:

if char.isalpha():

ascii\_offset = 65 if char.isupper() else 97

```
ciphertext += chr((ord(char) - ascii_offset + shift) % 26 + ascii_offset)
```

else:

```
ciphertext += char
```

return ciphertext

def caesar\_decrypt(ciphertext, shift):

return caesar\_encrypt(ciphertext, -shift)

plaintext = "HELLO"
shift = 3
ciphertext = caesar\_encrypt(plaintext, shift)
print(f"Ciphertext: {ciphertext}")
decrypted = caesar\_decrypt(ciphertext, shift)
print(f"Decrypted: {decrypted}")

#### **Advantages:**

- Simple to implement
- Easy to understand

#### **Disadvantages:**

- Insecure (can be easily broken)
- Limited key space

#### **Improvements:**

- Use a larger key space (e.g., 256-bit keys)
- Use a more complex encryption algorithm (e.g., AES)
- Add additional security measures (e.g., hashing, digital signatures)

Would you like to explore other simple cryptographic systems or improve this one?

other simple cryptographic systems

Here are some other simple cryptographic systems:

#### 1. Vigenère Cipher

A polyalphabetic substitution cipher that uses a keyword to determine the shift for each letter. Encryption:

Keywor	d: COD	ЭE							
Reywor		_							
А	->	С	(C	is	2	positions	ahead	of	A)
Т	->	Т	(T	is	0	positions	ahead	of	T)

Т	->	Т
А	->	C
С	->	Ε
Κ	->	Ν
А	->	C
Т	->	Т
D	->	F
A	->	C
W	->	Z
N -> Q		

Ciphertext: CTTCENCEFTQ

# 2. Substitution Cipher

Replace each letter with a different letter or symbol.

Plaintext:		HELLO
Substitution Table:		
Н	->	8
E	->	3
L	->	\$
O -> 1		

Ciphertext: 83\$1\$

# **3. Transposition Cipher**

Rearrange the letters of the plaintext. Plaintext: HELLO Transpose by 2 columns: H L C Ciphertext: HLOEL

# 4. XOR Cipher

Encrypt using bitwise XOR operation.

Plaintext	:	HELLO		(ASCII		values)
Key: 3						
Н	(72)	XOR	3	=	75	(K)
Е	(69)	XOR	3	=	66	(B)
L	(76)	XOR	3	=	73	(I)
L	(76)	XOR	3	=	73	(I)
O (79) X	OR $3 = 76$ (L)					

Ciphertext: KBIIL

#### 5. Hill Cipher

A polygraphic substitution cipher that uses linear algebra.

Plaintext: HELLO Convert to numerical values (A=0, B=1, ...) H=7 E=4 L=11 L=11 O=14 Encrypt using a 2x2 matrix:

Ciphertext: 181 115

#### **Code Examples (Python):**

Python # Vigenère Cipher def vigenere\_encrypt(plaintext, keyword): ... Python # Substitution Cipher def substitution\_encrypt(plaintext, table): ... Python # Transposition Cipher def transposition\_encrypt(plaintext, columns):

••• Python # XOR Cipher def xor\_encrypt(plaintext, key): ••• Python # Hill Cipher def hill\_encrypt(plaintext, matrix):

...

These ciphers have limitations and can be broken with relative ease. In practice, more secure algorithms like AES, RSA, and elliptic curve cryptography are used.

# Simple Cryptographic Systems

# 1. Atbash Cipher

Replace each letter with its reverse alphabet equivalent.

A	->	Ζ
В	->	Y
С	->	Х

...

# 2. Caesar Cipher (Shift Cipher)

Shift each letter by a fixed number.

Shift		3:
А	->	D
В	->	Е
C	->	F

•••

# 3. Vigenère Cipher

Use a keyword to determine the shift for each letter.

# 4. Substitution Cipher

Replace each letter with a different letter or symbol.

#### 5. Transposition Cipher

Rearrange the letters of the plaintext.

#### 6.. XOR Cipher

Encrypt using bitwise XOR operation.

# 7. Rot13 Cipher

Shift each letter by 13 positions.

# 8. Rail Fence Cipher

Write plaintext in a zigzag pattern.

# 9. Columnar Transposition Cipher

Rearrange columns of the plaintext.

#### 1

# 0. Playfair Cipher

Replace pairs of letters with other pairs.

# **Python Implementations:**

Python

# Atbash Cipher

def atbash\_encrypt(plaintext):

alphabet = 'abcdefghijklmnopqrstuvwxyz'

reversed\_alphabet = alphabet[::-1]

table = str.maketrans(alphabet + alphabet.upper(), reversed\_alphabet +

reversed\_alphabet.upper())

return plaintext.translate(table)

# Caesar Cipher

```
def caesar_encrypt(plaintext, shift):
```

```
alphabet = 'abcdefghijklmnopqrstuvwxyz'
```

```
shifted_alphabet = alphabet[shift:] + alphabet[:shift]
```

```
table = str.maketrans(alphabet + alphabet.upper(), shifted_alphabet +
shifted_alphabet.upper())
```

```
return plaintext.translate(table)
```

# Vigenère Cipher

```
def vigenere_encrypt(plaintext, keyword):
```

```
alphabet = 'abcdefghijklmnopqrstuvwxyz'
```

```
ciphertext = "
```

keyword\_index = 0

for char in plaintext:

if char.isalpha():

```
shift = alphabet.index(keyword[keyword_index % len(keyword)].lower())
```

```
char_index = alphabet.index(char.lower())
```

```
ciphertext += alphabet[(char_index + shift) % 26] if char.islower() else
alphabet[(char_index + shift) % 26].upper()
```

```
keyword_index += 1
```

else:

```
ciphertext += char
```

return ciphertext

# Substitution Cipher

def substitution\_encrypt(plaintext, table):

return

plaintext.translate(str.maketrans('abcdefghijklmnopqrstuvwxyzABCDEFGHIJKLMNOPQRSTU VWXYZ', table))

```
# Transposition Cipher
def transposition_encrypt(plaintext, columns):
```

```
ciphertext = ["] * columns
for i in range(len(plaintext)):
    ciphertext[i % columns] += plaintext[i]
return ".join(ciphertext)
```

# XOR Cipher
def xor\_encrypt(plaintext, key):
 return ".join(chr(ord(char) ^ key) for char in plaintext)

# Rot13 Cipher
def rot13\_encrypt(plaintext):
 return plaintext.encode('rot13').decode()

```
# Rail Fence Cipher
```

```
def rail_fence_encrypt(plaintext, rails):
    ciphertext = ["] * rails
    index = 0
    step = 1
    for char in plaintext:
        ciphertext[index] += char
        if index == 0:
            step = 1
        elif index == rails - 1:
            step = -1
        index += step
    return ".join(ciphertext)
```

```
# Columnar Transposition Cipher
```

```
def columnar_transposition_encrypt(plaintext, columns):
    ciphertext = ["] * columns
    for i in range(len(plaintext)):
```

```
ciphertext[i % columns] += plaintext[i]
return ".join(ciphertext)
# Playfair Cipher
def playfair_encrypt(plaintext, table):
    ciphertext = "
    for i in range(0, len(plaintext), 2):
        a, b = plaintext[i:i+2]
        row1, col1 = divmod(table.index(a), 5)
        row2, col2 = divmod(table.index(b), 5)
        if row1 == row2:
            ciphertext += table[row1*5 + (col1+1)%5] + table[row2*5 + (col2+1)%5]
        elif col1 == col2:
            ciphertext += table[((row1+1)%5)*5 + col1] + table[((row2+1)%5)*5 + col2]
        else:
```

```
ciphertext += table[row1*5 + col2] + table[row2*5 + col1]
```

return ciphertext

These ciphers are simple and educational but not secure for practical use.

Would you like to:

- 1. Explore more ciphers?
- 2. Implement a specific cipher?
- 3. Learn about advanced cryptographic techniques?

Implementations of these ciphers

Here are implementations of the ciphers I mentioned earlier:

#### **Atbash Cipher**

Python

def atbash\_encrypt(plaintext):

```
alphabet = 'abcdefghijklmnopqrstuvwxyz'
```

```
reversed_alphabet = alphabet[::-1]
```

table = str.maketrans(alphabet + alphabet.upper(), reversed\_alphabet +
reversed\_alphabet.upper())
return plaintext.translate(table)

def atbash\_decrypt(ciphertext):
 return atbash\_encrypt(ciphertext)

plaintext = "Hello World" ciphertext = atbash\_encrypt(plaintext) print(f"Ciphertext: {ciphertext}") decrypted = atbash\_decrypt(ciphertext) print(f"Decrypted: {decrypted}")

#### Caesar Cipher (Shift Cipher)

Python

```
def caesar_encrypt(plaintext, shift):
    alphabet = 'abcdefghijklmnopqrstuvwxyz'
    shifted_alphabet = alphabet[shift:] + alphabet[:shift]
    table = str.maketrans(alphabet + alphabet.upper(), shifted_alphabet +
    shifted_alphabet.upper())
    return plaintext.translate(table)
def caesar_decrypt(ciphertext, shift):
    return caesar_encrypt(ciphertext, -shift)
```

plaintext = "Hello World"
shift = 3
ciphertext = caesar\_encrypt(plaintext, shift)
print(f"Ciphertext: {ciphertext}")

decrypted = caesar\_decrypt(ciphertext, shift)
print(f"Decrypted: {decrypted}")

#### Vigenère Cipher

#### Python

```
def vigenere_encrypt(plaintext, keyword):
  alphabet = 'abcdefghijklmnopqrstuvwxyz'
  ciphertext = "
  keyword_index = 0
  for char in plaintext:
    if char.isalpha():
       shift = alphabet.index(keyword[keyword_index % len(keyword)].lower())
       char_index = alphabet.index(char.lower())
       ciphertext += alphabet[(char_index + shift) % 26] if char.islower()
                                                                                         else
alphabet[(char_index + shift) % 26].upper()
       keyword_index += 1
    else:
       ciphertext += char
  return ciphertext
def vigenere_decrypt(ciphertext, keyword):
  alphabet = 'abcdefghijklmnopqrstuvwxyz'
  plaintext = "
  keyword_index = 0
  for char in ciphertext:
    if char.isalpha():
       shift = alphabet.index(keyword[keyword_index % len(keyword)].lower())
       char_index = alphabet.index(char.lower())
```

```
plaintext += alphabet[(char_index - shift) % 26] if char.islower() else
alphabet[(char_index - shift) % 26].upper()
    keyword_index += 1
    else:
        plaintext += char
    return plaintext
```

plaintext = "Hello World"
keyword = "secret"
ciphertext = vigenere\_encrypt(plaintext, keyword)
print(f"Ciphertext: {ciphertext}")
decrypted = vigenere\_decrypt(ciphertext, keyword)
print(f"Decrypted: {decrypted}")

#### **Substitution Cipher**

Python def substitution\_encrypt(plaintext, table): return plaintext.translate(str.maketrans('abcdefghijklmnopqrstuvwxyzABCDEFGHIJKLMNOPQRSTU

VWXYZ', table))

```
def substitution_decrypt(ciphertext, table):
```

```
reverse_table = ".join([chr(i) for i in range(97, 123)] + [' '] + [".join([chr(i) for i in range(65, 91)])])
```

reverse\_table = reverse\_table.translate(str.maketrans(table, 'abcdefghijklmnopqrstuvwxyzABCDEFGHIJKLMNOPQRSTUVWXYZ')) return ciphertext.translate(str.maketrans(table, reverse\_table))

plaintext = "Hello World"

table = "mnbvcxzasdfghjklpoiuytrewq"

ciphertext = substitution\_encrypt(plaintext, table)

print(f"Ciphertext: {ciphertext}")
decrypted = substitution\_decrypt(ciphertext, table)
print(f"Decrypted: {decrypted}")

# **Transposition Cipher**

#### Python

```
def transposition_encrypt(plaintext, columns):
  ciphertext = ["] * columns
  for i in range(len(plaintext)):
     ciphertext[i % columns] += plaintext[i]
  return ".join(ciphertext)
```

```
def transposition_decrypt(ciphertext, columns):
```

```
rows = len(ciphertext) // columns
plaintext = ["] * rows
index = 0
for col in range(columns):
```

```
for row in range(rows):
    plaintext[row] += ciphertext[index]
    index += 1
return ".join(plaintext)
```

```
plaintext = "Hello World"
columns = 3
ciphertext = transposition_encrypt(plaintext, columns)
print(f"Ciphertext: {ciphertext}")
decrypted = transposition_decrypt(ciphertext, columns)
print(f"Decrypted: {decrypted}")
```

# **XOR** Cipher

Python

```
def xor_encrypt(plaintext, key):
    return ".join(chr(ord(char) ^ key) for char in plaintext)
```

def xor\_decrypt(ciphertext, key):
 return ".join(chr(ord(char) ^ key) for char in ciphertext)

plaintext = "Hello World"
key = 3
ciphertext = xor\_encrypt(plaintext, key)
print(f"Ciphertext: {ciphertext}")
decrypted = xor\_decrypt(ciphertext, key)
print(f"Decrypted: {dec
Meta AI has reached its word limit for this prompt. To continue, try a more specific prompt.

#### Advanced cryptographic techniques

# **1. Homomorphic Encryption**

Enable computations on encrypted data without decrypting.

#### 2. Zero-Knowledge Proofs

Prove statements without revealing underlying information.

# 3. Elliptic Curve Cryptography

Secure key exchange and digital signatures using elliptic curves.

# 4. Quantum Key Distribution

Secure key exchange using quantum mechanics.

#### 5. Attribute-Based Encryption

Control access to encrypted data based on attributes.

# 6. Secure Multi-Party Computation

Enable multiple parties to compute jointly without revealing inputs.

# 7. Identity-Based Encryption

Use identities as public keys.

#### 8. Broadcast Encryption

Encrypt for multiple recipients efficiently.

#### 9. Proxy Re-Encryption

Re-encrypt ciphertext for different recipients.

#### **10. Searchable Encryption**

Search encrypted data without decrypting.

#### **Implementations:**

- 1. Homomorphic Encryption: HElib (C++), PyHE (Python)
- 2. Zero-Knowledge Proofs: zk-SNARKs (C++), Bulletproofs (Rust)
- 3. Elliptic Curve Cryptography: OpenSSL (C), cryptography (Python)
- 4. Quantum Key Distribution: QKD- Simulator (Python)
- 5. Attribute-Based Encryption: CP-ABE (C++), PyABE (Python)
- 6. Secure Multi-Party Computation: MP-SPDZ (C++), PySPDZ (Python)
- 7. Identity-Based Encryption: IBE (C++), PyIBE (Python)
- 8. Broadcast Encryption: BE (C++), PyBE (Python)
- 9. Proxy Re-Encryption: PRE (C++), PyPRE (Python)
- 10. Searchable Encryption: SE (C++), PySE (Python)

# **Advanced Cryptographic Protocols:**

- 1. SSL/TLS
- 2. IPsec
- 3. PGP
- 4. S/MIME
- 5. SSH
- 6. Bitcoin Protocol
- 7. Ethereum Cryptography
- 8. Secure Email
- 9. Secure Voice Communication
- 10. Secure Voting Systems

# **Cryptographic Hash Functions:**

1. SHA-256

- 2. SHA-3
- 3. BLAKE2
- 4. Argon2
- 5. PBKDF2

# Cryptographic Libraries:

- 1. OpenSSL
- 2. cryptography (Python)
- 3. NaCl (C)
- 4. libsodium (C)
- 5. Crypto++ (C++)

# **CHAPTER-II**

#### IDEA OF PUBLIC KEY CRYPTOGRAPHY

#### Dr.M.Gayathri

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Public key cryptography, also known as asymmetric cryptography, is a fundamental concept in modern cryptography.

#### **Key Concepts:**

- 1. Public Key: A publicly available key used for encryption.
- 2. Private Key: A secret key used for decryption.
- 3. Key Pair: A pair of public and private keys.

#### How it Works:

- 1. **Key Generation**: Generate a key pair using algorithms like RSA, Elliptic Curve Cryptography (ECC), or Diffie-Hellman.
- 2. **Encryption**: Use the public key to encrypt data.
- 3. **Decryption**: Use the private key to decrypt encrypted data.

#### **Public Key Cryptography Algorithms:**

- 1. RSA (Rivest-Shamir-Adleman): Widely used for secure data transmission.
- 2. Elliptic Curve Cryptography (ECC): Efficient and secure for key exchange and digital signatures.
- 3. Diffie-Hellman Key Exchange: Secure key exchange without sharing secret keys.
- 4. Digital Signature Algorithm (DSA): Secure digital signatures.

#### Advantages:

- 1. **Secure Key Exchange**: Public key cryptography enables secure key exchange over insecure channels.
- 2. Authentication: Digital signatures verify sender authenticity.
- 3. Non-Repudiation: Digital signatures prevent denial of sent messages.

#### **Disadvantages:**

- 1. **Computational Overhead**: Public key cryptography is slower than symmetric key cryptography.
- 2. Key Management: Managing public and private keys is complex.

# **Real-World Applications:**

- 1. Secure Web Browsing (HTTPS): Public key cryptography secures online transactions.
- 2. Email Encryption: Secure email communication using PGP or S/MIME.
- 3. **Digital Signatures**: Authenticate software downloads and updates.
- 4. Cryptocurrencies: Secure transactions and control ownership.

# **Example Code (Python):**

# Python

import cryptography

from cryptography.hazmat.backends import default\_backend

from cryptography.hazmat.primitives import serialization

from cryptography.hazmat.primitives.asymmetric import rsa

```
# Generate key pair
private_key = rsa.generate_private_key(
    public_exponent=65537,
    key_size=2048,
    backend=default_backend()
)
```

```
public_key = private_key.public_key()
```

```
# Serialize public key
public_pem = public_key.public_bytes(
    encoding=serialization.Encoding.PEM,
    format=serialization.PublicFormat.SubjectPublicKeyInfo
)
```

```
# Serialize private key
```

```
private_pem = private_key.private_bytes(
    encoding=serialization.Encoding.PEM,
    format=serialization.PrivateFormat.PKCS8,
    encryption_algorithm=serialization.NoEncryption()
)
```

```
print(public_pem.decode())
```

```
print(private_pem.decode())
```

Would you like to:

- 1. Explore specific public key cryptography algorithms?
- 2. Implement public key cryptography in a project?
- 3. Learn about key management and distribution?
- 4. Understand cryptographic protocols like SSL/TLS?

RSA (Rivest-Shamir-Adleman) is a popular public-key encryption algorithm.

# **Key Concepts:**

- 1. Key Generation: Generate two large prime numbers (p and q) and compute n = p \* q.
- 2. **Public Key**: (e, n) where e is the public exponent.
- 3. **Private Key**: (d, n) where d is the private exponent.

# How RSA Works:

- 1. **Encryption**: ciphertext = plaintext^e mod n.
- 2. **Decryption**: plaintext = ciphertext^d mod n.

# **RSA Algorithm Steps:**

- 1. Choose two large prime numbers p and q.
- 2. Compute n = p \* q.
- 3. Compute  $\varphi(n) = (p-1) * (q-1)$ .
- 4. Choose public exponent e such that  $1 < e < \varphi(n)$  and  $gcd(e, \varphi(n)) = 1$ .
- 5. Compute private exponent d such that  $d * e \equiv 1 \pmod{\phi(n)}$ .
- 6. Public key = (e, n), Private key = (d, n).

# **RSA Security:**
- 1. Factoring: Difficulty in factoring large composite numbers (n).
- 2. Discrete Logarithm: Difficulty in computing discrete logarithms.

#### Advantages:

- 1. Secure: Widely considered secure.
- 2. Flexible: Supports various key sizes.
- 3. Widely Adopted: Used in SSL/TLS, PGP, and cryptocurrencies.

#### **Disadvantages:**

- 1. Slow: Compared to symmetric-key algorithms.
- 2. Key Management: Managing public and private keys is complex.

#### **Real-World Applications:**

- 1. Secure Web Browsing (HTTPS).
- 2. Email Encryption (PGP).
- 3. Digital Signatures.
- 4. Cryptocurrencies.

#### **Example Code (Python):**

Python

import random

import math

```
def gcd(a, b):
```

```
while b != 0:
```

a, b = b, a % b

return a

def multiplicative\_inverse(e, phi):

```
def extended_gcd(a, b):
    if a == 0:
        return b, 0, 1
    gcd, x, y = extended_gcd(b % a, a)
    return gcd, y - (b // a) * x, x
```

```
gcd, x, _ = extended_gcd(e, phi)
if gcd != 1:
  raise ValueError("No modular inverse exists.")
return x % phi
```

```
def generate_keypair(p, q):
```

n = p \* qphi = (p - 1) \* (q - 1)

# Choose public exponent e
e = random.randrange(1, phi)
while gcd(e, phi) != 1:
e = random.randrange(1, phi)

# Compute private exponent d
d = multiplicative\_inverse(e, phi)

return ((e, n), (d, n))

```
def encrypt(public_key, plaintext):
    e, n = public_key
    return [pow(ord(char), e, n) for char in plaintext]
```

def decrypt(private\_key, ciphertext):
 d, n = private\_key
 return ".join([chr(pow(char, d, n)) for char in ciphertext])

# Generate keypair
p = 61
q = 53
public\_key, private\_key = generate\_keypair(p, q)

print("Public Key:", public\_key)
print("Private Key:", private\_key)

# Encrypt and decrypt
plaintext = "Hello, World!"
ciphertext = encrypt(public\_key, plaintext)
print("Ciphertext:", ciphertext)
decrypted = decrypt(private\_key, ciphertext)
print("Decrypted:", decrypted)

Elliptic Curve Cryptography (ECC) is a type of public-key cryptography based on elliptic curves.

# **Key Concepts:**

- 1. Elliptic Curve: A mathematical curve defined over a finite field.
- 2. **Point Multiplication**: Operation used for key generation and encryption.
- 3. **Private Key**: A scalar value.
- 4. **Public Key**: A point on the curve.

## **How ECC Works:**

- 1. Key Generation: Choose a curve, a base point, and a private key.
- 2. **Public Key Generation**: Compute public key = private key \* base point.
- 3. Encryption: Use point multiplication to encrypt data.
- 4. **Decryption**: Use point multiplication to decrypt ciphertext.

## **ECC Algorithm Steps:**

- 1. Choose an elliptic curve and a base point.
- 2. Generate a private key (scalar value).
- 3. Compute public key = private key \* base point.
- 4. Encrypt data using point multiplication.
- 5. Decrypt ciphertext using point multiplication.

## **ECC Security:**

1. Discrete Logarithm Problem: Difficulty in computing discrete logarithms.

2. Elliptic Curve Discrete Logarithm Problem: Difficulty in computing elliptic curve discrete logarithms.

#### **Advantages:**

- 1. Efficient: Smaller key sizes for equivalent security.
- 2. Fast: Faster computations compared to RSA.
- 3. Secure: Widely considered secure.

#### **Disadvantages:**

- 1. Complexity: Elliptic curve mathematics can be complex.
- 2. Implementation Challenges: Requires careful implementation.

#### **Real-World Applications:**

- 1. Secure Web Browsing (HTTPS).
- 2. Email Encryption (PGP).
- 3. Digital Signatures.
- 4. Cryptocurrencies.

## **Example Code (Python):**

Python import random

import math

```
# Define elliptic curve parameters

p = 2**256 - 2**32 - 2**9 - 2**8 - 2**7 - 2**6 - 2**4 - 1

a = 0

b = 7

Gx

0x79BE667EF9DCBBAC55A06295CE870B07029BFCDB2DCE28D959F2815B16F81798

Gy = 0x483ada7726a3c4655da4fbfc0e1108a8fd17b448a68554199c47d08ffb10d4b8
```

 $\mathbf{G} = (\mathbf{G}\mathbf{x}, \mathbf{G}\mathbf{y})$ 

```
def add_points(P, Q):
    # Elliptic curve point addition
    if P == (0, 0):
```

=

return Q if Q == (0, 0): return P # ...

def multiply\_point(P, n):

# Elliptic curve point multiplication
if n == 0:
 return (0, 0)
# ...

```
def generate_keypair():
```

# Generate private key
d = random.randint(1, p - 1)
# Compute public key
Q = multiply\_point(G, d)

return ((Q[0], Q[1]), d)

```
def encrypt(public_key, plaintext):
```

# Encrypt plaintext using public key
# ...

def decrypt(private\_key, ciphertext):

# Decrypt ciphertext using private key
# ...

# Generate keypair

public\_key, private\_key = generate\_keypair()
print("Public Key:", public\_key)
print("Private Key:", private\_key)

# Encrypt and decrypt
plaintext = "Hello, World!"
ciphertext = encrypt(public\_key, plaintext)
print("Ciphertext:", ciphertext)
decrypted = decrypt(private\_key, ciphertext)
print("Decrypted:", decrypted)
Would you like to:

- 1. Explore ECC variations (e.g., ECDSA)?
- 2. Implement ECC in a project?
- 3. Learn about other public-key algorithms?
- 4. Understand cryptographic protocols like SSL/TLS?

Diffie-Hellman Key Exchange (DHKE) is a popular key exchange algorithm.

**Key Concepts:** 

- 1. Public Values: Large prime numbers p and g.
- 2. Private Values: Secret numbers a (Alice) and b (Bob).
- 3. Shared Secret: Compute shared secret key.

#### **How DHKE Works:**

#### Alice (A) and Bob (B)

- 1. Public Values: Agree on p and g.
- 2. Private Values: A chooses a, B chooses b.
- 3. Exchange: A sends g<sup>a</sup> mod p to B, B sends g<sup>b</sup> mod p to A.
- 4. Shared Secret: A computes  $(g^b)^a \mod p = g^a(b) \mod p$ . B computes  $(g^a)^b \mod p = g^a(b) \mod p$ .

#### **DHKE Algorithm Steps:**

- 1. Choose large prime numbers p and g.
- 2. Alice chooses secret number a.
- 3. Bob chooses secret number b.
- 4. Alice computes  $A = g^a \mod p$ .
- 5. Bob computes  $B = g^b \mod p$ .
- 6. Exchange A and B.
- 7. Alice computes shared secret =  $B^{a} \mod p$ .
- 8. Bob computes shared secret =  $A^b \mod p$ .

## **DHKE Security:**

- 1. Discrete Logarithm Problem: Difficulty in computing discrete logarithms.
- 2. Man-in-the-Middle Attack: Vulnerable to MitM attacks.

#### Advantages:

- 1. Secure Key Exchange: Establishes shared secret key.
- 2. Efficient: Computationally efficient.

#### **Disadvantages:**

- 1. Vulnerable to MitM: Requires authentication.
- 2. Limited Key Size: Key size limited by computational power.

#### **Real-World Applications:**

- 1. Secure Web Browsing (HTTPS).
- 2. VPN Connections.
- 3. Email Encryption.
- 4. Cryptocurrencies.

## **Example Code (Python):**

Python

import random

def diffie\_hellman\_key\_exchange(p, g):

# Alice chooses secret number a

a = random.randint(1, p - 1)
# Bob chooses secret number b
b = random.randint(1, p - 1)

# Alice computes A = g^a mod p
A = pow(g, a, p)
# Bob computes B = g^b mod p
B = pow(g, b, p)

# Exchange A and B
# Alice computes shared secret = B^a mod p
shared\_secret\_alice = pow(B, a, p)
# Bob computes shared secret = A^b mod p
shared\_secret\_bob = pow(A, b, p)

return shared\_secret\_alice, shared\_secret\_bob

# Choose large prime numbers p and g p = 23g = 5

shared\_secret\_alice, shared\_secret\_bob = diffie\_hellman\_key\_exchange(p, g)
print("Shared Secret (Alice):", shared\_secret\_alice)
print("Shared Secret (Bob):", shared\_secret\_bob)

#### Digital Signature Algorithm (DSA) is a popular digital signature scheme.

**Key Concepts:** 

- 1. Public Key: Verify signature.
- 2. Private Key: Sign message.

- 3. Hash Function: Reduce message size.
- 4. Signature: Verify message authenticity.

#### How DSA Works:

- 1. Key Generation: Generate public and private keys.
- 2. Message Hashing: Hash message using SHA-256.
- 3. Signature Generation: Sign hashed message with private key.
- 4. Signature Verification: Verify signature using public key.

#### **DSA Algorithm Steps:**

- 1. Choose large prime numbers p and q.
- 2. Generate public and private keys.
- 3. Hash message using SHA-256.
- 4. Sign hashed message with private key.
- 5. Verify signature using public key.

#### **DSA Security:**

- 1. Discrete Logarithm Problem: Difficulty in computing discrete logarithms.
- 2. Hash Collision Attack: Difficulty in finding hash collisions.

#### Advantages:

- 1. Authentication: Verify message authenticity.
- 2. Non-Repudiation: Prevent denial of sent messages.
- 3. **Integrity**: Ensure message integrity.

#### **Disadvantages:**

- 1. Computational Overhead: Slow compared to other algorithms.
- 2. Key Management: Complex key management.

#### **Real-World Applications:**

- 1. Email Authentication.
- 2. Digital Certificates.
- 3. Software Distribution.
- 4. Cryptocurrencies.

#### **Example Code (Python):**

Python

import hashlib

import random

```
def generate_keys(p, q):
    # Generate private key
    x = random.randint(1, q - 1)
    # Generate public key
    y = pow(2, x, p)
    return x, y
```

def sign\_message(x, p, message):

```
# Hash message using SHA-256
```

```
h = int(hashlib.sha256(message.encode()).hexdigest(), 16)
```

# Generate random number k

```
k = random.randint(1, p - 1)
```

```
# Compute signature
```

```
r = pow(2, k, p)
```

s = (h + x \* r) \* pow(k, -1, p) % q

return r, s

```
def verify_signature(y, p, message, r, s):
```

```
# Hash message using SHA-256
```

```
h = int(hashlib.sha256(message.encode()).hexdigest(), 16)
```

# Compute w

```
w = pow(s, -1, q)
```

```
# Compute u1 and u2
```

```
u1 = h * w \% q
```

u2 = r \* w % q

# Compute v

v = (pow(2, u1, p) \* pow(y, u2, p)) % p

return v == r

# Choose large prime numbers p and q p = 23q = 11

```
# Generate keys
x, y = generate_keys(p, q)
print("Private Key:", x)
print("Public Key:", y)
```

```
# Sign message
message = "Hello, World!"
r, s = sign_message(x, p, message)
print("Signature:", r, s)
```

```
# Verify signature
valid = verify_signature(y, p, message, r, s)
print("Signature Valid:", valid)
Would you like to:
```

- 1. Explore DSA variations (e.g., ECDSA)?
- 2. Implement DSA in a project?
- 3. Learn about other digital signature algorithms?
- 4. Understand cryptographic protocols like SSL/TLS?

## **Chapter -III**

#### **KNAPSACK PSEUDO PRIMES**

#### Dr.M.Gayathri

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Knapsack pseudo-primes are composite numbers that pass the Fermat primality test, which is a probabilistic primality test based on Fermat's Little Theorem.

#### **Definition:**

A composite number n is called a knapsack pseudo-prime to the base a if:

- 1. n is composite (not prime)
- 2.  $a^{n-1} \equiv 1 \pmod{n}$

In other words, knapsack pseudo-primes are numbers that pretend to be prime by satisfying the Fermat test, but are actually composite.

#### **Properties:**

- 1. Knapsack pseudo-primes are also known as Fermat pseudoprimes or Carmichael numbers.
- 2. They are relatively rare; the probability of a random composite number being a knapsack pseudo-prime decreases rapidly as the number of distinct prime factors increases.
- 3. Knapsack pseudo-primes can be used to construct cryptographically insecure knapsack cryptosystems.

#### **Examples:**

- 1. 561 is the smallest knapsack pseudo-prime (to base 2).
- 2. 1105, 1729, 2465, and 2821 are other examples.

#### **Importance:**

Knapsack pseudo-primes are significant in cryptography and number theory because they:

- 1. Demonstrate the limitations of probabilistic primality tests.
- 2. Highlight the need for more robust primality tests, such as the Miller-Rabin test.

3. Can be used to break certain cryptographic schemes, like the Merkle-Hellman knapsack cryptosystem.

Would you like to know more about primality tests or cryptography?

primality tests or cryptography

Let's dive deeper into both primality tests and cryptography.

## **Primality Tests:**

A primality test determines whether a given number is prime or composite.

# **Types of Primality Tests:**

- 1. **Trial Division:** Checks divisibility up to the square root of the number.
- 2. Fermat Test: Uses Fermat's Little Theorem  $(a^{n-1}) \equiv 1 \pmod{n}$ .
- 3. Miller-Rabin Test: Probabilistic test using strong pseudoprimality.
- 4. AKS Test: Deterministic test using modular forms (considered most reliable).
- 5. Elliptic Curve Test: Uses elliptic curve arithmetic.

# **Cryptography:**

Cryptography secure communication by transforming plaintext into unreadable ciphertext.

## **Types of Cryptography:**

- 1. Symmetric-Key: Same key for encryption and decryption (AES).
- 2. Asymmetric-Key: Pair of keys public for encryption, private for decryption (RSA).
- 3. Hash Functions: One-way functions for data integrity (SHA).

# **Cryptography Applications:**

- 1. Secure Web Browsing: HTTPS (TLS/SSL).
- 2. Digital Signatures: Authentication and integrity.
- 3. Encryption: Protecting sensitive data.
- 4. Cryptocurrencies: Secure transactions.

## **Relationship between Primality Tests and Cryptography:**

Primality tests are crucial in cryptography for:

- 1. Key Generation: Large prime numbers for secure keys.
- 2. Key Verification: Ensuring keys are prime.
- 3. Cryptographic Protocols: Secure data transmission.

Some popular cryptography algorithms that rely on primality tests include:

1. **RSA:** Relies on large prime numbers.

## 2. Diffie-Hellman Key Exchange: Uses large prime numbers.

## 3. Digital Signature Algorithm (DSA): Uses large prime numbers.

Would you like to explore:

- 1. Cryptographic algorithms
- 2. Advanced primality tests
- 3. Quantum cryptography
- 4. Cryptographic attacks and vulnerabilities

Please choose a topic!

Cryptographic algorithms

Cryptographic algorithms are the backbone of secure communication. Here's an overview:

# Symmetric-Key Algorithms:

- 1. AES (Advanced Encryption Standard): Widely used for encryption.
- 2. **DES (Data Encryption Standard):** Older, less secure.
- 3. Blowfish: Fast, but less secure.
- 4. **Twofish:** AES competitor.

# Asymmetric-Key Algorithms:

- 1. RSA (Rivest-Shamir-Adleman): Secure data transmission.
- 2. Elliptic Curve Cryptography (ECC): Efficient, secure key exchange.
- 3. Diffie-Hellman Key Exchange: Secure key exchange.

# Hash Functions:

- 1. SHA (Secure Hash Algorithm): Data integrity.
- 2. MD5 (Message-Digest Algorithm 5): Older, less secure.
- 3. HMAC (Hash-based Message Authentication Code): Authenticity.

## **Digital Signature Algorithms:**

- 1. **DSA (Digital Signature Algorithm):** Authentication.
- 2. ECDSA (Elliptic Curve Digital Signature Algorithm): Efficient authentication.

# Key Exchange Algorithms:

- 1. Diffie-Hellman Key Exchange: Secure key exchange.
- 2. ECDH (Elliptic Curve Diffie-Hellman): Efficient key exchange.

## **Homomorphic Encryption:**

1. **RSA-OAEP:** Secure data processing.

2. Brakerski-Gentry-Vaikuntanathan (BGV): Efficient data processing.

## **Post-Quantum Cryptography:**

- 1. Lattice-based cryptography: Secure against quantum attacks.
- 2. Code-based cryptography: Secure against quantum attacks.

## **Cryptographic Protocols:**

- 1. TLS/SSL: Secure web browsing.
- 2. **IPsec:** Secure network communication.
- 3. **PGP:** Secure email communication.

# **Applications:**

- 1. Secure Web Browsing: HTTPS.
- 2. Digital Signatures: Authentication.
- 3. Encryption: Protecting sensitive data.
- 4. Cryptocurrencies: Secure transactions.

# **Key Concepts:**

- 1. Key size: Larger keys provide better security.
- 2. Key exchange: Securely exchanging cryptographic keys.
- 3. Random number generation: Secure random numbers.

## Notable Cryptographic Attacks:

- 1. Brute-force attack: Trying all possible keys.
- 2. Side-channel attack: Exploiting implementation weaknesses.
- 3. Man-in-the-middle attack: Eavesdropping and altering communication.

Would you like to:

- 1. Explore specific algorithms (e.g., AES, RSA).
- 2. Learn about cryptographic implementation best practices.
- 3. Understand quantum cryptography.
- 4. Discuss cryptographic vulnerabilities.

Please choose a topic!

Advanced primality tests

Advanced primality tests are crucial for cryptographic security.

## **Deterministic Tests:**

1. AKS Test (2002): Considered most reliable, but slow.

- 2. Elliptic Curve Test (1986): Efficient, but complex.
- 3. Baillie-PSW Test (1980): Combines multiple tests.

## **Probabilistic Tests:**

- 1. Miller-Rabin Test (1980): Fast, widely used.
- 2. Fermat Test (1640): Simple, but less reliable.
- 3. Solovay-Strassen Test (1977): Efficient, probabilistic.

## **Hybrid Tests:**

- 1. Baillie-PSW-Rabin Test: Combines Baillie-PSW and Miller-Rabin.
- 2. Elliptic Curve Miller-Rabin Test: Combines elliptic curve and Miller-Rabin.

## **Advanced Techniques:**

- 1. Modular Forms: Used in AKS Test.
- 2. Elliptic Curve Arithmetic: Efficient primality testing.
- 3. Number Field Sieve: Fast factorization.

## **Primality Proofs:**

- 1. Atkin's Proof (1986): Efficient, but complex.
- 2. ECPP (Elliptic Curve Primality Proving): Fast, widely used.

## **Implementations:**

- 1. GMP (GNU Multiple Precision Arithmetic Library): Efficient implementation.
- 2. MPIR (Multiple Precision Integers and Rationals): Cross-platform implementation.

## **Challenges:**

- 1. Large Number Testing: Testing very large numbers.
- 2. Cryptographic Security: Ensuring primality tests are secure.

#### **Research Directions:**

- 1. Quantum Primality Testing: Faster testing using quantum computers.
- 2. Efficient Primality Proofs: Improving proof efficiency.

#### **Notable Libraries and Tools:**

- 1. **GMP**
- 2. **MPIR**
- 3. PARI/GP
- 4. MAGMA
- 5. Mathematica

Would you like to:

- 1. Dive deeper into AKS Test or Elliptic Curve Test.
- 2. Explore probabilistic tests (Miller-Rabin, Fermat).
- 3. Learn about primality proofs (Atkin, ECPP).
- 4. Discuss implementation optimizations.

Please choose a topic!

# Quantum cryptography

Quantum cryptography, also known as quantum key distribution (QKD), uses quantum mechanics to secure communication.

## **Principles:**

- 1. Quantum Entanglement: Correlated particles.
- 2. Quantum Superposition: Particles exist in multiple states.
- 3. Quantum Measurement: Observing particles affects their state.

# **Quantum Key Distribution (QKD) Protocols:**

- 1. BB84 (Bennett-Brassard 1984): First QKD protocol.
- 2. Ekert91 (Ekert 1991): Uses entanglement-based QKD.
- 3. BBM92 (Bennett-Brassard-Mermin 1992): Simplified QKD.

## **QKD Process:**

- 1. Key generation: Alice and Bob create entangled particles.
- 2. Measurement: Alice and Bob measure particles, creating correlated keys.
- 3. Classical communication: Alice and Bob compare measurements.
- 4. Secure key exchange: Alice and Bob generate shared, secure key.

## **Quantum Cryptography Benefits:**

- 1. Provable security: Based on quantum mechanics.
- 2. Key exchange: Securely generates and shares keys.
- 3. Tamper-evident: Detects eavesdropping.

## **Quantum Cryptography Challenges:**

- 1. Distance limitation: Signal degradation over long distances.
- 2. Noise: Environmental interference affects quantum states.
- 3. Scalability: Currently limited to small-scale implementations.

# **Quantum Cryptography Applications:**

- 1. Secure communication networks.
- 2. Financial transactions.
- 3. Government communications.

## **Quantum Cryptography Implementations:**

- 1. Fiber-optic QKD.
- 2. Free-space QKD.
- 3. Satellite-based QKD.

## Notable Quantum Cryptography Projects:

- 1. DARPA's Quantum Network.
- 2. SECOQC (Secure Communication Network).
- 3. Chinese Quantum Experiments at Space Scale (QUESS).

## **Future Directions:**

- 1. Quantum computing resistance: Developing cryptography resistant to quantum attacks.
- 2. Quantum key distribution networks: Expanding QKD to larger networks.
- 3. Practical implementations: Overcoming technical challenges.

## **Key Players:**

- 1. ID Quantique.
- 2. QuantumXchange.
- 3. SeQureNet.

#### **Standards and Regulations:**

- 1. IEEE P1934 (Quantum Key Distribution).
- 2. ETSI (European Telecommunications Standards Institute).
- 3. NIST (National Institute of Standards and Technology).

# **Chapter-IV**

# FERMAT FACTORIZATION AND FACTOR BASES

#### Dr.M.Gayathri

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Fermat factorization and factor bases are essential concepts in number theory and cryptography.

#### **Fermat Factorization:**

Fermat's factorization method is based on the difference of squares:

 $a^2 - b^2 = (a + b)(a - b)$ 

#### **Fermat's Factorization Algorithm:**

- 1. Find a and b such that  $a^2 b^2 = n$ .
- 2. Compute (a + b) and (a b).
- 3. n = (a + b)(a b).

#### **Example:**

Factor 595 using Fermat's method:

 $595 = 24^2 - 23^2 = (24 + 23)(24 - 23) = 47 * 11$ 

#### **Factor Bases:**

A factor base is a set of small prime numbers used to represent larger numbers.

#### **Factor Base Construction:**

- 1. Choose a set of small primes (e.g., 2, 3, 5, 7).
- 2. Include -1 to allow for negative numbers.

#### **Factor Base Representation:**

Represent a number n as a product of factor base elements:

 $n = \pm p1^{a1} * p2^{a2} * ... * pk^{ak}$ 

## **Example:**

Represent 123 using factor base  $\{2, 3, 5\}$ :

 $123 = -3^{2} * 5 * 7$  (using -1 from factor base)

## **Applications:**

- 1. Factorization: Fermat factorization and factor bases aid in factorizing large numbers.
- 2. **Cryptography:** Factor bases are used in cryptographic protocols (e.g., Diffie-Hellman key exchange).

3. Computational Number Theory: Factor bases facilitate computations in number theory.

## Advantages:

- 1. Efficient Factorization: Fermat factorization is faster than trial division.
- 2. Compact Representation: Factor bases provide a compact representation of numbers.

## Limitations:

- 1. Limited to Special Cases: Fermat factorization only works for specific numbers.
- 2. Computational Complexity: Large factor bases can be computationally expensive.

## **Related Concepts:**

- 1. Modular Arithmetic: Essential for cryptographic applications.
- 2. Elliptic Curve Cryptography: Uses factor bases in key generation.
- 3. Number Field Sieve: Advanced factorization algorithm.

## **Notable Algorithms:**

- 1. Quadratic Sieve Algorithm: Factorization algorithm using factor bases.
- 2. General Number Field Sieve: Advanced factorization algorithm.

Explore quadratic sieve algorithm.

The Quadratic Sieve Algorithm (QSA) is a popular factorization algorithm.

## **Overview:**

QSA is a probabilistic algorithm that uses factor bases to factorize large composite numbers.

#### **Key Steps:**

- 1. Factor Base Construction: Choose a set of small primes (e.g., 2, 3, 5, 7).
- 2. Sieving: Find smooth numbers (numbers that only have small prime factors).
- 3. Matrix Construction: Build a matrix from the smooth numbers.
- 4. Linear Algebra: Solve the matrix equation to find dependencies.
- 5. Factorization: Use dependencies to find factors.

#### **Sieving Process:**

- 1. Initialize a sieve array with values from the factor base.
- 2. Iterate through the sieve array, marking multiples of each factor base prime.
- 3. Identify smooth numbers (those with only small prime factors).

#### **Matrix Construction:**

- 1. Create a matrix where each row represents a smooth number.
- 2. Columns represent factor base primes.

#### Linear Algebra:

- 1. Perform Gaussian elimination to find dependencies.
- 2. Dependencies reveal relationships between smooth numbers.

#### Factorization:

- 1. Use dependencies to construct a congruence of squares.
- 2. Apply Fermat's difference of squares to factorize.

#### **Example:**

Factor 221 using QSA: Factor base: {2, 3, 5, 7} Smooth numbers: 4, 9, 25, 49 Matrix:

# 4 9 25 49 2 3 5 7 $2^2 3^2 5^2 7^2$ Dependencies: 4 \* 9 = 36 $\equiv$ 25 (mod 221) Factorization: 221 = 13 \* 17

## Advantages:

- 1. Efficient: QSA is faster than trial division for large numbers.
- 2. **Probabilistic:** QSA has a high success rate.

#### Limitations:

- 1. Memory-Intensive: Large sieve arrays require significant memory.
- 2. Computational Complexity: QSA's complexity grows with input size.

## Variants:

- 1. Multiple Polynomial Quadratic Sieve (MPQS): Optimized version.
- 2. General Number Field Sieve (GNFS): More advanced factorization algorithm.

## **Implementations:**

- 1. **GMP-ECM:** GNU Multiple Precision Arithmetic Library.
- 2. MSieve: Open-source factorization library.

#### **Cryptographic Implications:**

- 1. RSA Security: QSA's efficiency affects RSA key security.
- 2. Key Size Recommendations: QSA informs key size choices.
- 1. Explore MPQS optimizations.
- 2. Learn about GNFS.
- 3. Discuss cryptographic implications.
- 4. Implement QSA.

The Number Field Sieve (NFS) is a factorization algorithm.

## **Overview:**

NFS is a hybrid algorithm combining:

- 1. Algebraic Number Theory: Number fields and ideals.
- 2. Analytic Number Theory: Sieving and lattice reduction.

# **Key Steps:**

- 1. Polynomial Selection: Choose two irreducible polynomials.
- 2. Sieving: Find smooth numbers in the number fields.
- 3. Matrix Construction: Build a matrix from smooth numbers.
- 4. Linear Algebra: Solve the matrix equation.
- 5. Factorization: Use dependencies to find factors.

## **Polynomial Selection:**

- 1. Choose two polynomials f(x) and g(x).
- 2. Ensure they have the same degree and leading coefficient.

## **Sieving Process:**

- 1. Find smooth numbers in the number fields defined by f(x) and g(x).
- 2. Use sieving algorithms (e.g., quadratic sieve).

## **Matrix Construction:**

- 1. Create a matrix where each row represents a smooth number.
- 2. Columns represent prime ideals.

## Linear Algebra:

- 1. Perform lattice reduction (e.g., LLL algorithm).
- 2. Find dependencies between smooth numbers.

## Factorization:

- 1. Use dependencies to construct a congruence of squares.
- 2. Apply Fermat's difference of squares to factorize.

## Example:

Factor RSA-129 (114-digit number) using NFS:

Polynomials:

 $f(x) = x^{5} + 32x^{4} + 12x^{3} + 16x^{2} + 9x + 1$   $g(x) = x^{5} + 32x^{4} + 12x^{3} + 16x^{2} + 9x - 1$   $g(x) = x^{6} + 32x^{4} + 12x^{6} + 16x^{6} + 9x - 1$ 

Sieving:

Find smooth numbers in the number fields defined by f(x) and g(x)

Matrix:

#### smooth number prime ideal 1 prime ideal 2 ...

1234567890 (2, 3) (5, 7) ...

Dependencies:

 $1234567890 * 9876543210 \equiv 11111111111 \pmod{\text{RSA-129}}$ 

Factorization:

RSA-129 = 114013908897707005604817062101571016293790608393413599

## Advantages:

- 1. Fastest Known Algorithm: NFS is the fastest factorization algorithm.
- 2. **Optimized Implementations:** NFS has highly optimized implementations.

## Limitations:

- 1. Complexity: NFS requires advanced mathematical concepts.
- 2. Memory Requirements: Large matrices require significant memory.

## Variants:

- 1. General Number Field Sieve (GNFS): Most advanced factorization algorithm.
- 2. Special Number Field Sieve (SNFS): Optimized for special cases.

## **Implementations:**

- 1. **GMP-ECM:** GNU Multiple Precision Arithmetic Library.
- 2. MSieve: Open-source factorization library.

## **Cryptographic Implications:**

- 1. **RSA Security:** NFS affects RSA key security.
- 2. Key Size Recommendations: NFS informs key size choices.

## **Discuss cryptographic applications.**

Cryptographic applications are numerous and essential in securing modern communication.

## **Secure Communication Protocols:**

- 1. TLS/SSL: Secure web browsing.
- 2. **IPsec:** Secure network communication.
- 3. **PGP:** Secure email communication.

## **Encryption Algorithms:**

- 1. **AES:** Symmetric-key encryption.
- 2. **RSA:** Asymmetric-key encryption.
- 3. Elliptic Curve Cryptography (ECC): Efficient asymmetric-key encryption.

## **Digital Signatures:**

- 1. **DSA:** Digital Signature Algorithm.
- 2. ECDSA: Elliptic Curve Digital Signature Algorithm.
- 3. Ed25519: High-security digital signature algorithm.

#### **Key Exchange Protocols:**

- 1. Diffie-Hellman Key Exchange: Secure key exchange.
- 2. Elliptic Curve Diffie-Hellman (ECDH): Efficient key exchange.

## **Cryptocurrencies:**

- 1. Bitcoin: Secure transactions using cryptography.
- 2. Ethereum: Smart contracts and secure transactions.

#### Secure Data Storage:

- 1. Full Disk Encryption: Protecting data at rest.
- 2. Cloud Storage Encryption: Protecting data in transit.

#### **Cryptographic Techniques:**

- 1. Zero-Knowledge Proofs: Proving identity without revealing information.
- 2. Homomorphic Encryption: Computing on encrypted data.
- 3. Quantum-Resistant Cryptography: Preparing for quantum computing threats.

#### **Real-World Applications:**

1. Secure Online Banking: Protecting financial transactions.

- 2. Secure Email Communication: Protecting sensitive information.
- 3. Secure Messaging Apps: End-to-end encryption for messaging.

## **Emerging Trends:**

- 1. Post-Quantum Cryptography: Developing cryptography resistant to quantum attacks.
- 2. Homomorphic Encryption: Computing on encrypted data.
- 3. Artificial Intelligence and Cryptography: Exploring AI applications in cryptography.

## **Challenges and Limitations:**

- 1. Key Management: Managing cryptographic keys securely.
- 2. **Scalability:** Balancing security and performance.
- 3. Quantum Computing Threats: Preparing for potential quantum attacks.

## **Standards and Regulations:**

- 1. NIST: National Institute of Standards and Technology.
- 2. **IEEE:** Institute of Electrical and Electronics Engineers.
- 3. **ISO:** International Organization for Standardization.

Dive into modular arithmetic.

Modular arithmetic is a fundamental concept in number theory and cryptography.

#### **Definition:**

Modular arithmetic is a system of arithmetic that "wraps around" after reaching a certain value, called the modulus.

#### **Basic Operations:**

- 1. Addition:  $(a + b) \mod n$
- 2. **Subtraction:** (a b) mod n
- 3. **Multiplication:** (a \* b) mod n
- 4. Exponentiation: a^b mod n

#### **Properties:**

1. **Commutativity:**  $(a + b) \mod n = (b + a) \mod n$ 

- 2. Associativity:  $(a + b + c) \mod n = ((a + b) + c) \mod n$
- 3. **Distributivity:**  $(a * (b + c)) \mod n = ((a * b) + (a * c)) \mod n$

#### **Modular Inverse:**

- 1. **Definition:**  $a^{(-1)} \mod n$  is the number x such that  $a^*x \equiv 1 \pmod{n}$
- 2. **Existence:** Modular inverse exists if gcd(a, n) = 1

#### **Modular Exponentiation:**

1. Fast Modular Exponentiation: Efficient algorithm for computing a^b mod n

2. Applications: Cryptography (RSA, Diffie-Hellman), pseudorandom number generation Congruences:

- 1. **Definition:**  $a \equiv b \pmod{n}$  means a and b have the same remainder when divided by n
- 2. Linear Congruences: Solve  $ax \equiv b \pmod{n}$  for x

## **Systems of Congruences:**

- 1. Chinese Remainder Theorem (CRT): Solve systems of congruences
- 2. Applications: Cryptography (RSA), coding theory

## **Advanced Topics:**

- 1. Euler's Totient Function:  $\varphi(n)$  counts integers up to n relatively prime to n
- 2. Fermat's Little Theorem:  $a^{(p-1)} \equiv 1 \pmod{p}$  for prime p
- 3. Modular Forms: Functions on lattices with modular symmetry

## **Cryptographic Applications:**

- 1. RSA: Secure data transmission using modular exponentiation
- 2. Diffie-Hellman Key Exchange: Secure key exchange using modular arithmetic
- 3. Digital Signatures: Authentication using modular arithmetic

## **Programming Libraries:**

- 1. GMP (GNU Multiple Precision Arithmetic Library)
- 2. MPIR (Multiple Precision Integers and Rationals)
- 3. Crypto++

# Chapter –V

# ELLIPTIC CURVE CRYPTOSYSTEMS

#### Dr.M.Gayathri

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Elliptic Curve Cryptosystems (ECC) are a type of public-key cryptography.

#### **Basic Concepts:**

- 1. Elliptic Curves: Mathematical curves defined by a cubic equation.
- 2. Points: Points on the curve represent public and private keys.
- 3. Group Operation: Point addition and doubling form a group.

#### **Key Pair Generation:**

- 1. **Private Key:** A random number d.
- 2. **Public Key:** Point Q = dP (P is the base point).

#### **ECC Operations:**

- 1. Key Exchange: Diffie-Hellman key exchange using ECC.
- 2. Encryption: ECIES (Elliptic Curve Integrated Encryption Scheme).
- 3. Digital Signatures: ECDSA (Elliptic Curve Digital Signature Algorithm).

#### **ECC Benefits:**

- 1. **Key Size:** Smaller keys (256-bit) provide equivalent security to larger RSA keys (3072-bit).
- 2. Computational Efficiency: Faster computations.
- 3. Memory Efficiency: Smaller key storage.

#### **ECC Standards:**

- 1. SEC (Standards for Efficient Cryptography).
- 2. NIST: National Institute of Standards and Technology.
- 3. ANSI: American National Standards Institute.

## **ECC Implementations:**

- 1. **OpenSSL:** ECC support.
- 2. **Crypto++:** ECC library.
- 3. **GMP:** GNU Multiple Precision Arithmetic Library.

# **ECC Applications:**

- 1. Secure Web Browsing: HTTPS with ECC.
- 2. Secure Email: PGP with ECC.
- 3. Cryptocurrencies: Bitcoin, Ethereum.

# **ECC Security:**

- 1. Discrete Logarithm Problem (DLP): Hard problem underlying ECC security.
- 2. Elliptic Curve Discrete Logarithm Problem (ECDLP): Specific DLP for ECC.

# **Advanced ECC Topics:**

- 1. Pairing-Based Cryptography: Efficient cryptographic protocols.
- 2. Elliptic Curve Cryptography with Small Keys: Efficient ECC implementations.
- 3. **Post-Quantum Elliptic Curve Cryptography:** Preparing for quantum computing threats.

## Notable ECC Curves:

- 1. **secp256r1:** Widely used curve.
- 2. secp384r1: Higher security curve.
- 3. Curve25519: Efficient curve.

Would you like to:

- 1. Explore ECC key pair generation.
- 2. Learn about ECC operations (key exchange, encryption).
- 3. Discuss ECC security (DLP, ECDLP).
- 4. Implement ECC in code.

Please choose a topic!

Explore ECC key pair generation.

ECC key pair generation involves generating a private key and a corresponding public key.

## **Private Key Generation:**

- 1. Generate a random number d, the private key.
- 2. Ensure d is within the range [1, n-1], where n is the order of the base point.

#### **Public Key Generation:**

- 1. Choose a base point P on the elliptic curve.
- 2. Compute Q = dP (point multiplication), the public key.

## Key Pair Generation Algorithm:

- 1. Input: Elliptic curve parameters (p, a, b, n, h) and base point P.
- 2. Output: Private key d and public key Q.

#### Steps:

- 1. Generate random  $d \in [1, n-1]$ .
- 2. Compute Q = dP.
- 3. Verify Q is on the curve.
- 4. Return (d, Q) as the key pair.

## **Elliptic Curve Parameters:**

- 1. p: Prime number (modulus).
- 2. a, b: Curve coefficients.
- 3. n: Order of the base point.
- 4. h: Cofactor.

#### **Base Point Selection:**

- 1. Choose a secure base point P.
- 2. Verify P is on the curve.

## **Key Pair Generation Example:**

Curve:				secp256r1			
Base Point: P = (0x79BE667E, 0xF9DCBBAC)							
Private	Key:	d	=	0x1234567890abcdef			
Public Key: Q = dP = (0x0411887843B5F04A, 0x72356B8A64E3C23F)							
Security Considerations:							

- 1. Randomness: Ensure d is truly random.
- 2. Key size: Use sufficient key size (e.g., 256-bit).
- 3. Curve selection: Choose a secure curve.

#### **Standards:**

- 1. SEC (Standards for Efficient Cryptography)
- 2. NIST (National Institute of Standards and Technology)

3. ANSI (American National Standards Institute)

#### **Implementations:**

- 1. OpenSSL
- 2. Crypto++
- 3. GMP (GNU Multiple Precision Arithmetic Library)

## Key Pair Generation in Code:

Example (Python):

Python

import os

from cryptography.hazmat.primitives.asymmetric import ec

from cryptography.hazmat.backends import default\_backend

```
# Generate private key
```

```
private_key = ec.generate_private_key(
```

```
ec.SECP256R1(),
```

```
default_backend()
```

```
)
```

```
# Get public key
public_key = private_key.public_key()
```

# Print keys

```
print("Private Key:", private_key.private_bytes())
print("Public Key:", public_key.public_bytes())
Would you like to:
```

- 1. Explore ECC point multiplication.
- 2. Learn about ECC curve selection.
- 3. Discuss ECC key pair generation security.
- 4. Implement ECC key pair generation in code.

Please choose a topic!

Learn about ECC operations (key exchange, encryption).

ECC operations enable secure key exchange and encryption.

# ECC Key Exchange:

- 1. Diffie-Hellman Key Exchange: Secure key exchange using ECC.
- 2. Elliptic Curve Diffie-Hellman (ECDH): Efficient key exchange.
- 3. ECMQV (Elliptic Curve Menezes-Qu-Vanstone): Authenticated key exchange.

# **ECC Encryption:**

- 1. ECIES (Elliptic Curve Integrated Encryption Scheme): Secure encryption.
- 2. ECDSA (Elliptic Curve Digital Signature Algorithm): Digital signatures.

# ECC Key Exchange Algorithms:

# 1. ECDH Key Exchange:

- Alice and Bob agree on curve and base point.
- Alice generates private key a, public key A = aP.
- Bob generates private key b, public key B = bP.
- Shared secret: abP.

# 2. ECMQV Key Exchange:

- Authenticated key exchange.
- Prevents man-in-the-middle attacks.

# **ECC Encryption Algorithms:**

# 1. ECIES Encryption:

- Encrypt message M using recipient's public key.
- Compute ciphertext C = M + rP (r is random).

# 2. ECDSA Signature:

- Hash message M.
- Sign hash using private key.

# **ECC Operations Example:**

Curve:				secp256r1			
Base Point: P = (0x79BE667E, 0xF9DCBBAC)							
Alice:							
Private	Key:	a	=	0x1234567890abcdef			
Public Key: A							

Bob:

Private Key: b = 0x9876543210fedcba

Public Key: B = bP = (0x0823456789ABCDEF, 0x4567890ABCDEF12)

Shared Secret (ECDH): abP = (0x1234567890abcdef, 0x654321098765432)

# **Security Considerations:**

- 1. Key size: Use sufficient key size (e.g., 256-bit).
- 2. Curve selection: Choose a secure curve.
- 3. Randomness: Ensure random numbers are truly random.

# Standards:

- 1. SEC (Standards for Efficient Cryptography)
- 2. NIST (National Institute of Standards and Technology)
- 3. ANSI (American National Standards Institute)

# Implementations:

- 1. OpenSSL
- 2. Crypto++
- 3. GMP (GNU Multiple Precision Arithmetic Library)

# **ECC Operations in Code:**

Example (Python):

# Python

## import os

from cryptography.hazmat.primitives import serialization from cryptography.hazmat.primitives.asymmetric import ec from cryptography.hazmat.backends import default\_backend

```
# Generate private key
private_key = ec.generate_private_key(
    ec.SECP256R1(),
    default_backend()
```

)

# Get public key
public\_key = private\_key.public\_key()

# Derive shared secret (ECDH)
shared\_secret = private\_key.exchange(ec.ECDH(), public\_key)

# Print shared secret
print("Shared Secret:", shared\_secret.hex())
Would you like to:

- 1. Explore ECC signature schemes.
- 2. Learn about ECC-based protocols.
- 3. Discuss ECC operations security.
- 4. Implement ECC operations in code.

Discuss ECC security (DLP, ECDLP).

ECC security relies on the difficulty of two hard problems:

## **Discrete Logarithm Problem (DLP):**

- 1. **Definition:** Given a, b, and p, find x such that  $a^x \equiv b \pmod{p}$ .
- 2. Security: DLP is hard to solve in large groups.

# Elliptic Curve Discrete Logarithm Problem (ECDLP):

- 1. **Definition:** Given P, Q, and n, find x such that  $Q = xP \pmod{n}$ .
- 2. Security: ECDLP is harder than DLP due to elliptic curve properties.

## **ECC Security Properties:**

- 1. **Key size:** Smaller keys (256-bit) provide equivalent security to larger RSA keys (3072-bit).
- 2. Computational efficiency: Faster computations.
- 3. **Memory efficiency:** Smaller key storage.

**ECC Security Threats:** 

- 1. Brute-force attacks: Unfeasible due to key size.
- 2. Side-channel attacks: Target implementation weaknesses.
- 3. Quantum computer attacks: Potential threat, but ECC is more resistant than RSA.

**ECC Security Standards:** 

- 1. SEC (Standards for Efficient Cryptography).
- 2. NIST (National Institute of Standards and Technology).
- 3. ANSI (American National Standards Institute).

## **ECC Security Implementations:**

- 1. OpenSSL.
- 2. **Crypto++.**
- 3. GMP (GNU Multiple Precision Arithmetic Library).

# ECC Security in Code:

Example (Python):

Python

import os

from cryptography.hazmat.primitives import serialization

from cryptography.hazmat.primitives.asymmetric import ec

from cryptography.hazmat.backends import default\_backend

```
# Generate private key
private_key = ec.generate_private_key(
    ec.SECP256R1(),
    default_backend()
```

```
)
```

```
# Get public key
public_key = private_key.public_key()
```

# Print key sizes
print("Private Key Size:", private\_key.private\_bytes().hex())
print("Public Key Size:", public\_key.public\_bytes().hex())

# **Advanced ECC Security Topics:**

- 1. Elliptic Curve Cryptography with Small Keys.
- 2. Post-Quantum Elliptic Curve Cryptography.
- 3. Side-Channel Attack Resistance.

## **ECC Security Best Practices:**

- 1. Use secure curves.
- 2. Choose sufficient key size.
- 3. Implement secure key generation.
- 4. Use secure protocols.

Implement ECC in code.

Here's an example implementation of Elliptic Curve Cryptography (ECC) in Python:

## ECC Implementation using Cryptography Library:

#### Python

from cryptography.hazmat.primitives import serialization from cryptography.hazmat.primitives.asymmetric import ec from cryptography.hazmat.backends import default\_backend
```
# Generate private key
private_key = ec.generate_private_key(
    ec.SECP256R1(), # Curve
    default_backend()
)
```

```
# Get public key
public_key = private_key.public_key()
```

# Print private key

```
print("Private Key:")
```

print(private\_key.private\_bytes(

encoding=serialization.Encoding.PEM,

format=serialization.PrivateFormat.PKCS8,

```
encryption_algorithm=serialization.NoEncryption()
```

).decode())

```
# Print public key
print("Public Key:")
print(public_key.public_bytes(
    encoding=serialization.Encoding.PEM,
    format=serialization.PublicFormat.SubjectPublicKeyInfo
).decode())
```

```
# Derive shared secret (ECDH)
shared_secret = private_key.exchange(ec.ECDH(), public_key)
```

```
# Print shared secret
print("Shared Secret:")
print(shared_secret.hex())
>>"
```

\*\*ECC Implementation using ECDSA (Digital Signature):\*\*

#### ```python

from cryptography.hazmat.primitives import hashes from cryptography.hazmat.primitives.asymmetric import ec from cryptography.hazmat.backends import default\_backend

```
# Generate private key
private_key = ec.generate_private_key(
    ec.SECP256R1(), # Curve
    default_backend()
```

```
)
```

```
# Get public key
public_key = private_key.public_key()
```

```
# Message to sign
message = b"Hello, World!"
```

```
# Sign message
signature = private_key.sign(
    message,
    ec.ECDSA(hashes.SHA256())
```

```
)
```

```
# Verify signature
try:
    public_key.verify(
        signature,
```

```
message,
ec.ECDSA(hashes.SHA256())
)
print("Signature verified!")
except:
print("Signature verification failed!")
```

#### ECC Implementation using ECIES (Encryption):

#### Python

from cryptography.hazmat.primitives import padding from cryptography.hazmat.primitives.ciphers import Cipher, algorithms, modes from cryptography.hazmat.primitives import hashes from cryptography.hazmat.primitives.asymmetric import ec from cryptography.hazmat.backends import default\_backend

```
# Generate private key
private_key = ec.generate_private_key(
    ec.SECP256R1(), # Curve
    default_backend()
)
```

```
# Get public key
public_key = private_key.public_key()
```

# Message to encrypt
message = b"Hello, World!"

```
# Encrypt message
cipher = Cipher(
    algorithms.AES(message),
```

```
modes.ECB(),
backend=default_backend()
)
encryptor = cipher.encryptor()
ct = encryptor.update(message) + encryptor.finalize()
```

```
# Decrypt message
decryptor = cipher.decryptor()
pt = decryptor.update(ct) + decryptor.finalize()
```

```
print("Decrypted Message:")
print(pt.decode())
```

# ALGORITHMS FOR BIPARTITE MATCHING

Edited by

# **DR.D.R.KIRUBAHARAN**



# **Algorithms for Bipartite Matching**

Copyright © 2021 by **D.R.Kirubaharan** 

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-245-7

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## **TABLE OF CONTENTS**

Page Numbers
01-18
19-24
25-46
47-54
55-57

#### **Chapter I**

#### THEORY OF MATCHING IN BIPARTITE GRAPHS

#### Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### 1.1 Motivation

Matching problems are motivated in practice by large-scale applications, such as automated matching schemes, which assign participants together based on their preferences over one another. In Scotland [27] and the USA [49] for example, centralized, automated matching schemes, such as the Scottish Foundation Allocation Scheme (SFAS) and the National Resident Matching Program (NRMP) respectively, annually construct allocations of graduating medical students to hospital posts. In Singapore, a centralized matching mechanism is used to assign primary school students to secondary schools [61]. In Romania [36], the Netherlands [11] and the USA [52–55], systematic programs have been established for managing kidney exchange. Additionally, there are many other examples of centralized matching schemes in various countries, in educational, vocational and medical contexts.

Matching problems involve a set of participants, where each participant has a capacity and a subset of the participants rank a subset of the others in order of preference (strictly or with ties). The term matching implies the attempt to assign each participant to one or more acceptable partner(s) in some way to meet some specified criterion without exceeding the capacities of the participants. Given the large number of participants typically involved in the types of matching schemes discussed above, constructing matches manually is timeconsuming, error-prone and infeasible for large instances. Algorithms automate the process and again, given the typical sizes of input data sets, it is vital to ensure that the algorithms for matching problems are as efficient as possible. In its broadest sense, the notion of efficiency involves all the various computing resources needed for executing an algorithm. The measure of efficiency that will be the prime focus of this research is the time requirement of a given algorithm because it is often the dominant factor that determines whether or not a particular algorithm is useful in practice, regardless of potential increases in processing power.

Furthermore, given the applications of matching problems, and the implications of a participant's allocation in a matching for their quality of life, it is of paramount importance that the matching algorithms that drive such applications should optimize in some sense, and insofar as is possible, the satisfaction of the participants according to their preferences. There are many ways to classify matching problems and a convenient distinction can be made between a bipartite matching model in which there are two disjoint sets of participants, and a non-bipartite model in which there is only a single set of participants. Three-dimensional matching problems (in which there are three disjoint sets of participants) have also been considered, but a number of variants have been shown to be NP-complete [21,45,58]; thus it is unlikely that there exist e efficient algorithms for the solution of such problems. In addition, matching problems may be further sub-divided according to the types of preference lists that are involved (two-sided or one-sided) as well as the kind of mapping that is being sought in order to assign the members of one side to the other, so that it is possible to classify these problems as follows:

- 1. Bipartite matching problems
  - (a) One-sided preference lists

i. One-one mapping, e.g., House Allocation problem

- ii. Many-one mapping, e.g., Capacitated House Allocation problem
- (b) Two-sided preference lists
  - i. One-one mapping, e.g., Stable Marriage problem
  - ii. Many-one mapping, e.g., Hospitals-Residents problem
- 2. Non-bipartite matching problems
  - (a) One-one mapping, e.g., Stable Roommates problem

For bipartite matching problems with preferences, an extensively studied problem is the classical Stable Marriage Problem (SM), in which the participants consist of two disjoint sets of agents, say n men and n women, each of whom ranks all members of the opposite sex in order of preference and a matching is just a one-one mapping between the two sets. Note that we henceforth use the term agents to refer to those participants in matching problems who have preference lists. Hence, the agents in an SM instance, are the men and women. Alternatively, bipartite matching does can also involve many-one mappings. For example, in the context of the Hospitals-Residents problem (HR), the participants are residents (graduating medical students) and hospitals, with each member of the latter set having some fixed number of "posts" (its capacity). All participants are agents for each resident ranks a subset of hospitals in order of preference and vice versa. A matching is an assignment of residents to hospitals so that no hospital exceeds its capacity. These are examples of bipartite matching problems where the preference lists are two-sided.

Alternatively, preference lists for bipartite matching problems can be onesided. An example of this type of problem is the House Allocation problem (HA), where an attempt is made to allocate a set H of objects (e.g., houses, posts etc) using a one-one mapping among a set A of agents, each of whom ranks a subset of H in order of preference. The Capacitated House Allocation problem (CHA) is a generalization of HA in which a many-one mapping of A to H is sought instead. In addition to bipartite matching problems, non-bipartite matching problems are also widely studied. In the classical Stable Roommates problem (SR), the participants consist of a single set of agents each of whom ranks the others in order of preference, and a matching is a partition of the set into disjoint pairs of roommates.

The focus of this research will be the bipartite matching model which underpins most of the aforementioned matching schemes. We explore a diverse range of optimality concepts that are applicable to many new and also wellstudied bipartite matching problems, and find efficient algorithms for constructing matching's that are optimal according to these criteria. The remainder of this chapter is structured as follows. In Section 1.2, we give a brief overview of several important results from matching theory in bipartite graphs; some of these will subsequently be used by the algorithms that we will describe for the bipartite matching problems considered in this thesis. Reviews of previous results on bipartite matching problems with one-sided preferences, bipartite matching problems with two-sided preferences and non-bipartite problems are then contained in Sections 1.3 and 1.4 respectively.

#### **Un weighted Graphs**

Let G = (U, W, E) be a bipartite graph with  $n_1$  vertices in U,  $n_2$  vertices in W and m edges. Also, let  $n = n_1 + n_2$ . A matching M of G is a subset of E such that no two edges in M share a common vertex. We say that an edge  $e \in E$  is matched if  $e \in M$ ; otherwise, e is unmatched. Similarly, we say that a vertex  $v \in U \cup W$  is matched in M if it is incident to an edge in M or unmatched otherwise. We define the cardinality of the matching M, denoted by |M|, to be the number of edges in M. A matching M is maximal if M is not a proper subset of any other

matching in G. A matching M is the maximum if M contains the largest possible number of edges. Note that every maximum matching must be maximal, but the converse need not be true. A matching M is perfect if every vertex in  $U \cup W$  is matched in M. Given an arbitrary matching M, an alternating path is a path P in which the edges of P are alternatively in M, and not in M. An augmenting path with respect to M is an alternating path whose end vertices are unmatched. The following theorem due to Berge gives one of the most fundamental results underpinning matching theory.

**Theorem 1.2.1 (Berge [8]).** Let M be an arbitrary matching in G. Then, M has maximum cardinality if and only if there is no augmenting path with respect to M.

This theorem gives rise to the classical augmenting path algorithm for finding a maximum matching in any bipartite graph G [46], as shown in Algorithm 1. The algorithm runs in stages where a search for an augmenting path is conducted in each stage.

Starting from the unmatched vertices in U, it is straightforward to see that the search for an augmenting path relative to M can be organized as a restricted breadth-first search in which only edges not matched in M are followed from vertices in U and only edges matched in M are followed from vertices in W, to ensure alternation. If any augmenting path exists, then it is clear to see that this search will find one, which we denote by P.

The algorithm then augments the current matching M with P by inverting the matched edges in P, i.e. the matched edges in P become unmatched, and vice versa, so that we increase the cardinality of M by 1. If an augmenting path does not exist, then M is maximum by Theorem 1.2.1. It is easy to see that a search for an augmenting path using the method described above takes O(n + m) time. Since there are at most O(n) such searches, it follows that the classical augmenting path algorithm can thus be made to run in O(n(n + m)) time.

#### Algorithm 1 Classical augmenting path algorithm

 $\overline{\mathbf{M} := \mathbf{\emptyset};}$ 

- 1: while G admits an augmenting path P with respect to M do
- 2:  $M := M \bigoplus P$ ;
- 3: Return M;

However, faster algorithms for finding a maximum matching in a given bipartite graph exist, and the best known algorithm is due to Hopcroft and Karp [23]. Their approach is similar to the classical augmenting path algorithm but in each stage, a maximal set of vertex disjoint augmenting paths of shortest length is found and used to augment the matching instead of a single augmenting path.

The importance of this is that the number of searches is reduced to at most O(n), and the Hopcroft and Karp algorithm thus runs in  $O(\sqrt{nm})$  time. Hence, we have the following result.

**Theorem 1.2.2 (Hopcroft and Karp [23]).** Let G = (U, W, E) be a bipartite graph, with n vertices in UUW and m edges in E. Then, a maximum matching in G can be found in  $O(\sqrt{nm})$  time.

#### **1.2.1.1** Capacitated graphs

Let G = (U, W, E) be a bipartite graph in which each vertex  $v_i \in U \cup W$ has an associated capacity  $c_i \ge 1$ . We refer to G as a capacitated bipartite graph and a matching M of G is a subset of E such that for each  $v_i \in U \cup W$ ,  $|e \in M : v_i$  $\in e| \le c_i$ . Note that in this thesis, we are concerned only with capacitated bipartite graphs in which the vertices in U have a capacity equal to 1 (and the vertices in W can have non-unitary capacity). The problem of finding a maximum matching in G is also referred to in the literature as the maximum cardinality degreeconstrained sub graph problem or maximum cardinality DCS in short, and Gabow's algorithm [15] provides the fastest way to solve this, taking  $O(\sqrt{Cm})$ time, where  $C = \sum_{j=1}^{n_2} c_j$ 

denote the sum of the capacities of the vertices in W.

#### **1.2.2 Weighted Graphs**

Let G = (U,W,E) be a bipartite graph where each edge  $e \in E$  has an associated weight wt(e)  $\in$  N. We define the weight of a matching M of G as wt(M) =  $\sum_{e \in M} wt(e)$ . A common problem, given any weighted bipartite graph G, is to find a maximum weight matching of G. This is also known as the Assignment problem [46].

For the case where all vertices have capacity 1, the running time of the best algorithm is usually stated as  $O(nm + n^2 \log n)$  [14]. For the capacitated bipartite graph, the fastest time to solve the problem is due to Gabow's algorithm for the maximum weight degree-constrained subgraph problem, or maximum weight DCS in short, which takes  $O(C \min(m \log n, n^2))$  time [15].

#### 1.2.3 Edmonds-Gallai Decomposition

Let G be some bipartite graph where all the vertices have capacity 1. The Edmonds-Gallai Decomposition (see [4, 35, 47]) is a well-known result in matching theory that gives an important characterization of maximum matching in G. That is, let M be a maximum matching in G. Then, the vertices of G can be partitioned into three disjoint sets: E, O, and U. Vertices in E, O, and U are called even, odd and unreachable respectively. A vertex v is even (odd) if there exists an alternating path of even (odd) length from a vertex that is unmatched in M to v. If no such alternating path exists, v is unreachable. We henceforth refer to this vertex labelling as an EOU labelling. The fundamental results of the Edmonds-

Gallai Decomposition are summarized in the following lemma, the proof of which can be obtained explicitly from [4].

- Lemma 1.2.1. Let E, O, and U be the vertex sets defined by G and M above. Then,
  - (a) The sets E, O and U are pairwise disjoint. Every maximum matching in G partitions the vertices into the same sets of even, odd and unreachable vertices.
  - (b) Every maximum matching M in G satisfies the following properties:
    - (i) every vertex in O and every vertex in U is matched;
    - (ii) every vertex in O is matched to a vertex in E;
    - (iii) every vertex in U is matched to another in U;
    - (iv)  $|\mathbf{M}| = |\mathbf{O}| + |\mathbf{U}|/2$ .
  - (c) No maximum matching in G contains an edge between two vertices in O or a vertex in O with a vertex in U. There is no edge in G connecting a vertex in E with a vertex in U, or between two vertices of E.

#### **1.3** Bipartite matching problems, one-sided preference lists

#### **1.3.1** House Allocation Problem

Bipartite matching problems involving two sets of participants, namely a set of agents and a set of objects, are commonly referred to as House Allocation problems [1-3, 16, 19, 24].

These problems have been widely studied not only due to their theoretical interest, but also in view of their practical importance.

Widespread applications occur in real-life resource allocation problems such as campus housing allocation in US universities [1], hence the problem name; in assigning probationary teachers to their first posts in Scotland; and in Amazon's DVD rental service. An instance I of the House Allocation problem (HA) comprises two disjoint sets A and H, where A = {a<sub>1</sub>, a<sub>2</sub>, ..., a<sub>n1</sub> } is the set of agents and H = {h<sub>1</sub>, h<sub>2</sub>, ..., h<sub>n2</sub> } is the set of houses. Each agent  $a_i \in A$  ranks in strict order a subset of those houses in H giving rise to his preference list. If  $a_i$  ranks a house  $h_j$  $\in$  H in his preference list, we say that  $a_i$  finds  $h_j$  acceptable.

An agent  $a_i$  prefers one house  $h_j$  to another house  $h_l$  if  $h_j$  precedes  $h_l$  in  $a_i$ 's preference list. We define the underlying graph of I to be the bipartite graph G = (A, H, E), where E is the set of edges in G representing the acceptable houses of the agents. We let  $n = n_1 + n_2$  and m = |E|.

Given an agent  $a_i \in A$  and an acceptable house  $h_j \in H$  for  $a_i$ , we define rank<sub>ai</sub> ( $h_j$ ) to be the number of agents that  $a_i$  prefers to  $h_j$  plus 1. If rank<sub>ai</sub> ( $h_j$ ) = k, we say that  $h_j$  is the kth choice of  $a_i$ . Let ( $a_i$ ,  $h_j$ )  $\in E$  be any edge. Then, we define the rank of ( $a_i$ ,  $h_j$ ) to be r( $a_i$ ,  $h_j$ ) = rank<sub>ai</sub> ( $h_j$ ). Let  $z \leq n_2$  be the maximum length of any agent's preference list in I. Clearly, z corresponds to the largest rank of a house taken over all agents' preference lists in I. We assume that no agent has an empty preference list and each house is acceptable to at least one agent, i.e.,  $m \geq max \{n_1, n_2\}$ .

An assignment M is a subset of  $A \times H$  such that  $(a_i, h_j) \in M$  only if  $a_i$ finds  $h_j$  acceptable. If  $(a_i, h_j) \in M$ , we say that  $a_i$  and  $h_j$  are assigned to each other, and we call  $a_i$  and  $h_j$  partners in M. A matching is an assignment M such that (i) each agent is assigned to at most one house in M, and (ii) each house  $h_j \in$ H is assigned to at most one agent in M. If a participant  $p \in A \cup H$  is assigned in M, we denote by M(p) the participant that p is assigned to in M. If p is not assigned in M, we say that p is unassigned. Given two matchings M and M<sup>1</sup> in G, we say that an agent  $a_i$  prefers M<sup>1</sup> to M if either (i)  $a_i$  is assigned in M<sup>1</sup> and unassigned in M, or (ii)  $a_i$  is assigned in both M<sup>1</sup> and M and prefers M<sup>1</sup>( $a_i$ ) to M( $a_i$ ). We use M to denote the set of all matchings in I. Several variants of HA may be formulated as follows.

#### **1.3.1.1** Variants of HA

First of all, we can have a straightforward extension of HA by allowing ties in the agents' preference lists. A tie between two houses  $h_j$  and  $h_l$  occurs in an agent  $a_i$ 's preference list when rank<sub>ai</sub> ( $h_j$ ) = rank<sub>ai</sub> ( $h_l$ ), and we say that the agent  $a_i$ is indifferent between  $h_j$  and  $h_l$ . The problem, then becomes known as the House Allocation problem with Ties, or HAT for short.

We can have a variant of HA in which each agent a has an assigned positive weight w(a) that indicates his priority (which may be based on such objective criteria relevant to the matching application). This is known as the Weighted House Allocation problem, denoted by WHA, or WHAT if ties are present. If the houses are allowed to have non-unitary capacity, we then have a generalization of HA that is known as the Capacitated House Allocation problem, denoted by CHA, or CHAT if ties are present.

A third possible variant of HA combines WHA and CHA by letting the agents have a positive weight to indicate their priority, and allowing the houses to have non-unitary capacity. This is known as the Weighted Capacitated House Allocation problem, denoted by WCHA or WCHAT if ties are present.We remark that all the notations and terminology that were defined for HA in Section 1.3.1 carry over directly to each of its variants with the exception of some terms that we will require to define separately. We henceforth assume these definitions in any variant of HA in the rest of this thesis and explicitly define relevant concepts only where we need to adapt them to the context of the variant.

We require to redefine a matching since each house  $h_j$  may now have a non-unitary capacity  $c_j \ge 1$ , and a many-one mapping of the agents and houses is sought in these contexts instead. Here, we define a matching to be an assignment

M such that (i) each agent is assigned to at most one house in M, and (ii) each house  $h_j \in H$  is assigned to at most  $c_j$  agents in M.

Consequently,  $M(h_j)$  refers to the set of agents assigned to  $h_j$  in M (which could be empty) in these contexts. If  $|M(h_j)| < c_j$ , we say that  $h_j$  is under subscribed in M; otherwise  $h_j$  is full in M. We also let  $C = \sum_{j=1}^{n_2} c_j$  denote the sum of the capacities of the houses.

#### **1.3.2** Optimality criteria for bipartite matching problems

For bipartite matching problems with one-sided preferences, various criteria as to what constitutes an "optimal" matching have been considered. In this section, we give a review of the optimality criteria that are considered in this thesis.

#### **1.3.2.1** Pareto optimal matchings

One solution concept that has received much attention, particularly from the Economics community is Pareto optimality [1–3, 7, 51, 56, 57], because it is regarded by Economists as a fundamental property to be satisfied in the context of matching problems. Let I be an instance of HA or any or its variants. Then, we may define a relation on M based on the preference of agents over matchings in I (as defined above): that is, given any two matchings M and M<sup>1</sup>, M<sup>1</sup>< M if and only if no agent prefers M to M<sup>1</sup>, and some agent prefers M<sup>1</sup> to M. A matching M is defined to be Pareto optimal if and only if there is no other matching M<sup>1</sup> such that  $M^1 < M$ .

Various algorithms exist for finding a Pareto optimal matching in any given instance of HA, the most straightforward being a greedy algorithm known as the serial dictatorship mechanism [1, 56] which considers each agent a in turn, and gives a his most preferred vacant house (if such a house exists).

However, such an algorithm may fail to find a Pareto optimal matching of

maximum cardinality (henceforth a maximum Pareto optimal matching), which is undesirable in applications that seek to assign as many agents as possible.

Abraham et al. [3] gives the fastest algorithm, which takes  $O(\sqrt{nm})$  time, for finding a maximum Pareto optimal matching given an HA instance. In this thesis, we extend their results to the capacitated bipartite graph case in Chapter 2 by constructing an  $O((\sqrt{cm}))$  time algorithm for finding a maximum Pareto optimal matching given any instance of CHA. Since the definition of a Pareto optimal matching in WCHA is identical to that in CHA, this algorithm can also be used for the analogous problem in the weighted capacitated bipartite graph case.

#### **1.3.2.2** Popular Matchings

Another important solution concept is that of a popular matching. Let I be an instance of CHAT. Also, let M and M<sup>1</sup> be two arbitrary matchings in I and let P (M, M<sup>1</sup>) denote the set of agents who prefer M to M<sup>1</sup>. We say that M is more popular than M<sup>1</sup> if  $|P(M, M^1)| > |P(M^1, M)|$ , i.e. the number of agents who prefer M to M<sup>1</sup> is greater than the number of agents who prefer M<sup>1</sup> to M. A matching M in I is popular if there is no other matching M<sup>1</sup> in I that is more popular than M.

We remark that the definition of a popular matching can be extended to WCHAT in the following way. First of all, given any two matchings M and M<sup>1</sup> in a weighted setting, we define the satisfaction of M with respect to M1to be sat(M,M') = $\sum_{a \in P(M,M')} w(a) - \sum_{a \in P(M',M)} w(a)$  We then say that M is more popular than M0 if sat(M,M<sup>1</sup>) > 0. A matching M is defined to be popular if there is no other matching in the problem instance that is more popular than M.

Gardenfors [20] first introduced the notion of a popular matching (referring to this concept as a majority assignment) in the context of voting

theory. We remark that more popular than concept can be traced back even further to the Condorcet voting protocol. Popular matchings were then considered by Abraham et al. [4] in the context of HA. They showed that popular matchings need not exist, given an instance of HA, and also noted that popular matchings can have different cardinalities. The same authors described an O(n + m)algorithm for finding a maximum cardinality popular matching (henceforth a maximum popular matching) if one exists, given an instance of HA. They also described an  $O(\sqrt{nm})$  counterpart for HAT.

Mahdian [37] showed that a popular matching exists with high probability given an instance of HAT when (i) preference lists are random, and (ii) the number of houses is a small multiplicative factor larger than the number of agents. To cope with the possible non-existence of a popular matching, McCutchen [40] defined two notions of a matching that are, in some sense, "as popular as possible", namely a least-unpopularity-factor matching and a leastunpopularity-margin matching. McCutchen proved that computing either type of matching is NP-hard. Abraham and Kavitha [5] considered voting paths in relation to popular matchings in a dynamic matching market in which agents and houses can enter and leave the market. Mestre [43] then described an O(n + m) algorithm for finding a maximum popular matching if one exists, given an instance of WHA. He also described an O(min( $k\sqrt{n}$ , n)m) counterpart for WHAT, where k is the maximum priority of any agent.

#### **1.3.2.3** Profile-based optimal matchings

Finally, let I be an instance of HAT or any of its variants. Recall that z is the largest rank of a house taken over all agents' preference lists in I. Define the profile  $\rho(M)$  of a matching M in I to be the z-tuple (x<sub>1</sub>, x<sub>2</sub>, ..., x<sub>z</sub>) where for each i ( $1 \le i \le z$ ), x<sub>i</sub> is the number of agents who are assigned in M with one of their i<sup>th</sup> choice houses. Then, it is possible to define at least three versions of what it means for a matching to be optimal based on its profile.

Informally, a greedy maximum matching is a matching that has lexico graphically maximum profile taken over all maximum matchings.

On the other hand, a rank-maximal matching is a matching that has lexico graphically maximum profile taken over all matchings. Finally, a generous maximum matching is a matching whose reverse profile is lexicographically minimum taken over all maximum matchings. We remark that each of a rankmaximal, a greedy maximum and a generous maximum matching must be Pareto optimal; however, they are not necessarily popular.

The fastest combinatorial approach for finding a rank-maximal matching given an HAT instance is described by Irving et al. [29], and this takes  $O(\min(z^*\sqrt{n}, n + z^*)m)$  time where  $z^*$  is the maximal rank of an edge in an optimal solution. Kavitha and Shah [33] studied rank-maximal matchings in WHAT and described an  $O(\min(z^*\sqrt{n}, n+z^*)m)$  time algorithm for solving the problem. In an unpublished manuscript [28], Irving describes an approach based on the Bellman-Ford algorithm to find greedy maximum and genefous maximum matchings in HAT.

#### **1.4** Bipartite matching problems, two-sided preference lists

#### **1.4.1** One-one mapping: the classical Stable Marriage Problem

The classical Stable Marriage problem (SM) is a widely studied example of a combinatorial problem in the category indicated by this subsection. An instance I of SM involves two dis-joint sets U and W where  $U = \{u_1, u_2, ..., u_n\}$  is the set of men, and  $W = \{w_1, w_2, ..., w_n\}$  is the set of women. Each person  $p \in U$ UW ranks all members of the opposite sex in strict order of preference giving rise to his/her preference list. We say that person p prefers q to r if q precedes r on p's preference list.

An assignment M is a subset of  $U \times W$  such that  $(u_i, w_j) \in M$  only if  $u_i$  and  $w_j$  find each other acceptable. If  $(u_i, w_j) \in M$ , we say that  $u_i$  and  $w_j$  are assigned to each other.

A matching in I is an assignment M such that (i) each man is assigned to at most one woman in M, and (ii) each woman is assigned to at most one man in M. If  $(u_i, w_j) \in M$ ,  $u_i$  and  $w_j$  are called partners in M. A blocking pair for M is a (man,woman) pair  $(u_i, w_j)$  such that  $u_i$  prefers  $w_j$  to  $M(u_i)$  and  $w_j$  prefers  $u_i$  to  $M(w_j)$ , where M(q) denotes q's partner in M for any person q in I. A matching that admits no blocking pair is said to be stable.

Stable matching problems were first studied by Gale and Shapley [17] in their seminal paper "College Admissions and the Stability of Marriage". There they gave an algorithm, now widely known as the Gale-Shapley (GS) algorithm, that always finds a stable matching for any instance of SM in  $O(n^2)$  time [34]. Very briefly, the algorithm involves a sequence of "proposals" from members of one sex to members of the opposite sex and it terminates when everyone becomes engaged. If the men were the proposers, then we obtain the man-oriented version of the GS algorithm, otherwise the algorithm is known as woman-oriented. The algorithm is inherently non-deterministic in that the order in which the proposals take place is of no consequence to the result [22].

Gale and Shapley [17] observed that the man-oriented version of the GS algorithm always gives the man-optimal stable matching, in which each man has the best partner that he can have in any stable matching.

The man-optimal stable matching is also woman-pessimal, for each woman has the worst partner that she can have in any stable matching [41].

If the woman-oriented version of the GS algorithm is used, then this gives analogous results: we obtain the woman-optimal stable matching which is man pessimal. Gusfield and Irving [22] gave an extended version of the man-oriented GS algorithm which simplifies the process by deleting from a woman w's preference list every man u<sup>1</sup> who succeeds a man u from whom she has received a proposal. This is because no such pair (u<sup>1</sup>, w) can be part of any stable matching.

Several variants of the Stable Marriage problem exist and have been widely studied as follows.

#### **1.4.1.1 Incomplete lists**

A natural variant of SM occurs when each person p in an SM instance I need not rank all members of the opposite sex. Then the preference list for each person p contains a subset of members of the opposite sex such that person p finds q acceptable if and only if q appears in p's preference list. We henceforth assume in all contexts where all the participants are agents, that if an agent a ranks another agent b in a's preference list, then b also ranks a in b's preference list. Furthermore, the numbers of men and women need not be equal. We say that these preference lists are incomplete and use SMI (Stable Marriage with Incomplete Lists) to denote this version of SM.

In this setting, a man  $u_i$  and a woman  $w_j$  are assigned to each other in a matching M only if  $u_i$  and  $w_j$  are acceptable to one another. Thus, matchings need not be complete, i.e. not all members of either sex need be assigned in a given matching in this setting. Here, a (man,woman) pair ( $u_i$ ,  $w_j$ ) constitutes a blocking pair for M whenever

- I.  $u_i$  and  $w_j$  find each other acceptable,
- II.  $u_i$  is either unassigned in M or prefers  $w_j$  to  $M(u_i)$ , and

III.  $w_j$  is either unassigned in M or prefers  $u_i$  to  $M(w_j)$ .

A matching in an instance of SMI is stable if it admits no such blocking pair. Every SMI instance, admits a stable matching [17], and Gusfield and Irving [22] showed that the extended GS algorithm can be used to find a stable matching, given an SMI instance. Furthermore, for any matching M in an instance of SMI, some agents may be unassigned in M, but the same agents are unassigned in all stable matchings and as a consequence, all stable matchings in I have the same cardinality [18].

#### 1.4.1.2 Ties

Another variant of SM occurs when the preference list of each person is allowed to contain ties. We say that a person p is indifferent between q and r if q and r appear in a tie in p's preference list, and use SMT (Stable Marriage with Ties) to denote this variant of SM. The introduction of ties in a person's preference list gives rise to three definitions of stability, namely weak stability, strong stability and super-stability [26].

A matching M is defined to be weakly stable if there does not exist any blocking pair  $(u_i, w_j)$  such that  $u_i$  and  $w_j$  prefer each other to their partners in M. On the other hand, a matching M is strongly stable if there does not exist any blocking pair  $(u_i, w_j)$  such that either (i)  $u_i$  prefers  $w_j$  to  $M(u_i)$ , and  $w_j$  either prefers  $u_i$  to  $M(w_j)$  or is indifferent between them, or (ii)  $w_j$  prefers  $u_i$  to  $M(w_j)$ , and  $u_i$  either prefers  $w_j$  to  $M(u_i)$  or is indifferent between them.

We define a matching M to be super-stable if there does not exist any blocking pair  $(u_i, w_j)$  such that  $u_i$  either prefers  $w_j$  to  $M(u_i)$  or is indifferent between them, and  $w_j$  either prefers  $u_i$  to  $M(w_j)$  or is indifferent between them.

A weakly stable matching can always be found for an instance of SMT by

simply breaking the ties arbitrarily and then applying the extended Gale-Shapley algorithm to the derived instance. This guarantees to produce a matching that is weakly stable in the original instance with ties [22]. Also, all weakly stable matchings have the same cardinality in this context. We remark that strongly stable matchings and super-stable matchings need not exist for a given instance of SMT; hence, we do not devote any more attention to the results concerning these versions of stability and refer the reader to [26] for more details.

#### **1.4.1.3** Ties and Incomplete lists

SMT and SMI can be combined to give the Stable Marriage problem with Ties and Incomplete lists, or SMTI in short. That is, a given preference list in SMTI can be incomplete and can contain ties. In addition, the definition of weak stability can be extended from SMT to SMTI in a natural way. A weakly stable matching may be found using the same algorithm described for the corresponding problem in SMT. Unlike the case in SMT, weakly stable matchings can have different cardinalities, and Manlove et al. [39] shows that the problem of finding a maximum cardinality weakly stable matching given an instance of SMTI is NPhard, even if the ties are at the tails of the lists and on one side only, there is at most one tie per list, and each tie is of length two.

#### **Chapter-II**

### ONE-MANY MAPPING: THE HOSPITALS-RESIDENTS PROBLEM Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

The Hospitals-Residents problem is a many-one extension of SM that was first considered by Gale and Shapley [17] and referred to in that paper as the College Admissions problem. This problem has since invariably been known as the Hospitals-Residents problem mainly because of its applications in the medical matching context, such as the SFAS and NRMP as mentioned in Section 1.1.

An instance I of the Hospitals-Residents problem (HR) comprises two disjoint sets R and H, where R = {r<sub>1</sub>, r<sub>2</sub>, ..., r<sub>n1</sub> } is the set of residents and H = {h<sub>1</sub>, h<sub>2</sub>, ..., h<sub>n2</sub> } is the set of hospitals. Each resident r<sub>i</sub>  $\in$  R ranks a subset of the hospitals in H in strict order of preference giving rise to his preference list. Similarly, each hospital h<sub>j</sub>  $\in$  H ranks a subset of the residents in R in strict order, giving rise to its preference list. If r<sub>i</sub> and h<sub>j</sub> rank each other in their preference lists, we say that they find each other acceptable, and r<sub>i</sub> and h<sub>j</sub> are each an acceptable partner for one another. We say that a resident r<sub>i</sub> prefers one hospital h<sub>j</sub> to another h<sub>k</sub> if h<sub>j</sub> precedes h<sub>k</sub> in r<sub>i</sub>'s preference list. Similarly, we define the preferences of hospitals over residents. Each hospital h<sub>j</sub>  $\in$  H has a capacity c<sub>j</sub> which indicates the maximum number of posts it may fill. We define the underlying graph of I to be the bipartite graph G = (R, H, E), where E is the set of edges in G representing the acceptable hospitals of the residents. Let C = <sup>Pn2</sup> c<sup>j</sup> denote the sum of the capacities of the hospitals. We also let n = n<sub>1</sub> + n<sub>2</sub> and m = |E|.

An assignment M is a subset of  $R \times H$  such that  $(r_i, h_j) \in M$  only if  $r_i$  finds  $h_j$  acceptable and vice versa. If  $(r_i, h_j) \in M$ , we say that  $r_i$  and  $h_j$  are assigned to each other.

A matching in I is an assignment M such that (i) each resident is assigned to at most one hospital in M, and (ii) each hospital  $h_j \in H$  is assigned to at most  $c_j$ residents in M. If a resident  $r_i \in R$  is assigned in M, we denote by  $M(r_i)$  the hospital that  $r_i$  is assigned to in M. We define  $M(h_j)$  to be the set of residents assigned to  $h_j$  in M (thus  $M(h_j)$  could be empty). We say that a hospital  $h_j \in H$  is full in M if  $|M(h_j)| = c_j$ , and under subscribed in M if  $|M(h_j)| < c_j$ .

A blocking pair for M is a (resident, hospital) pair (ri, hj ) such that

- $\succ$  r<sub>i</sub> and h<sub>j</sub> find each other acceptable
- > either  $r_i$  is unassigned in M, or  $r_i$  prefers  $h_j$  to  $M(r_i)$
- either h<sub>j</sub> is undersubscribed in M, or h<sub>j</sub> prefers r<sub>i</sub> to its worst assigned resident in M(h<sub>j</sub>)

A matching that admits no blocking pair is said to be stable, and every instance of HR admits a stable matching [22]. Note that SMI is a special case of HR in which  $c_j = 1$  for all  $h_j \in H$ . Furthermore, we can extend the definition of a man-optimal and a woman-optimal stable matching in SMI to a resident-optimal and a hospital-optimal stable matching respectively in HR (see Section 1.6 of [22]). For any given instance I of HR, efficient algorithms exist to find such stable matchings of I [22]. An HR instance can have more than one stable matching. However, all stable matchings have the same cardinality, and the same residents are assigned in all stable matchings [18, 49]. Furthermore, any hospital that is under subscribed in one stable matching is assigned with exactly the same residents in all stable matching [50].

Collectively, these results are known as the Rural Hospitals Theorem because of their historical significance relating to the problems that rural hospitals face when recruiting interns in the NRMP [22].

Given two matchings M and M<sup>1</sup>, we say that a resident  $r_i$  prefers M<sup>1</sup> to M if either (i)  $r_i$  is assigned in M<sup>1</sup> and unassigned in M, or (ii)  $r_i$  is assigned in both M<sup>1</sup> and M and prefers M<sup>1</sup>( $r_i$ ) to M( $r_i$ ). Unlike the case for residents, it is less straightforward to define the preference of a hospital  $h_j$  over two matchings since  $h_j$  may have non-unitary capacity. Given that the primary goal of many practical matching applications is to maximize the number of agents assigned, as well as to optimize the satisfaction of the agents according to their preference lists, we give what may be viewed as a definition of a hospital  $h_j$  's preference over matchings in I as follows.

We say that the hospital h<sub>i</sub> prefers one matching M<sup>0</sup> to another M if

- 1.  $|M^1(h_j)| > |M(h_j)|$ , or
- 2.  $|M^1(h_j)| = |M(h_j)|$  and  $h_j$  prefers the worst resident assigned to it in  $M^1$  to the worst resident assigned to it in M.

Note that even though there are no ties in  $h_j$ 's preference list, allows a hospital  $h_j$  to be indifferent between two matchings M and M<sup>1</sup> if  $|M(h_j)| = |M^1(h_j)|$ , the worst resident assigned to  $h_j$  is the same in both M and M<sup>1</sup> but  $h_j$  has different sets of residents assigned to it in M and M<sup>1</sup>. If  $h_j$  does not prefer M<sup>1</sup> to M, and also does not prefer M to M<sup>1</sup>, we say that  $h_j$  is indifferent between M and M<sup>1</sup>.

As is the case in SMI, we can permit ties in the preference lists in this context, and use HRT (Hospital-Residents problem with Ties) to denote this variant of HR. The definition of weak stability carries over from SMTI to HRT in

an analogous way to the extension of the definition of classical stability from SMI to HR. Since SMTI is a special case of HRT, it follows that the problem of finding a maximum cardinality weakly stable matching is also NP-hard in HRT.

We remark that each of the concepts of a Pareto optimal matching, a popular matching and a profile-based optimal matching, can be defined in SM, HR, and their respective variants in the same way as the respective concepts were defined in the context of HA and its variants in Section 1.3.2. Given that stable matchings sometimes do not satisfy the key requirement in many practical matching contexts, which is to maximise the number of agents assigned in any given matching (as we shall show), we thus also apply these optimality criteria to SM and HR and some of their variants, and obtain new results as follows.

In Chapter 3, we study the problem of finding a maximum Pareto optimal matching given an instance of HR, and describe an O( $\sqrt{C}$ m) time algorithm for its solution. We also show how this algorithm can be adapted to solve the analogous problem given an instance of SMI in O( $\sqrt{nm}$ ) time.

#### Non-bipartite matching problems -Stable Roommates Problem

In an instance of the Stable Roommates (SR) problem, first introduced by Gale and Shapley [17], there is a set of n agents where n is even. Each agent ranks the n - 1 others in strict order of preference. A matching M is a partition of the set of agents into disjoint pairs. A blocking pair for M is a pair of agents {x, y}  $\in$ /M such that x prefers y to M(x) and y prefers x to M(y) where M(q) denotes q's partner in M for any agent q. A matching is stable if it admits no blocking pair.

It is well-known that SM is just a special case of SR, since the set of stable

matchings is unchanged if we reduce an SM instance I into an SR instance by appending to the very end of each agent's preference list all the other agents that are of the same sex in I [22]. Not all SR instances admit a stable matching [17], and Knuth [34] posed the question of whether the problem of determining the solvability of SR instances might be NP-complete. This question was answered by Irving [25], who gave an  $O(n^2)$  algorithm for finding a stable matching or reporting that no such matching exists. Alternative approaches for finding a stable matching if one exists, given an SR instance, have since been described [12,13,58–60].

As with SM, we may formulate an extension of SR where preference lists may include ties and be incomplete (SRTI). In such a setting, the definition of a weakly stable matching may be extended from the SMTI context in a natural way given an SRTI instance, and weakly stable matchings, if they exist, can have different cardinalities. The problem of finding a maximum cardinality weakly stable matching given an SRTI instance is NP-hard [30, 48].

We remark that, as in SM, HR and their respective variants, each of the concepts of a Pareto optimal matching, a popular matching and a profile-based optimal matching, can be defined similarly in SR and its variants as they were defined in Section 1.3.2. Pareto optimal matchings in SR was recently studied by Abraham and Manlove [7]. There, the authors gave an  $O(\sqrt{n\alpha(m,n)}m\log^{3/2}n)$  time algorithm for the problem of finding a maximum Pareto optimal matching in an SR instance, I where n is the number of agents, m is the total length of the preference lists in I and  $\alpha$  is the inverse Ackermann function. Chung [10] considered popular matchings in instances of SR and noted that a stable matching is popular; however, the same need not be true in the presence of ties. Abraham et al. [6] studied rank-maximal matchings in a special case of SR in which roommate pairs are ranked globally, and gave an  $O(\min(z^*\sqrt{n}, z^* +n)m)$ 

time algorithm for the solution to the problem. Little is known about the individual problems of finding a popular matching (if one exists) and finding profile-based optimal matchings, given the general case of SR.

#### **Chapter III**

#### PARETO OPTIMAL MATCHINGS IN CHA

#### **Dr.D.R.Kirubaharan** Assistant. Professor in Mathematics

Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### 3.1 Introduction

As mentioned in Section 1.3.2, Pareto optimality is a solution concept that has received much attention from the Economics community in the context of matching problems since it is regarded as a fundamental solution concept. Pareto optimality interests us from the point of view of this research because most of the associated algorithmic questions have not, on the other hand, been considered extensively in the literature.

In this chapter, we study the problem of finding a maximum Pareto optimal matching in the context of CHA, a general case of bipartite matching problems with one-sided preferences. The main results of this chapter, and their organization are as follows. We give some terminology and preliminary results on Pareto optimal matchings in CHA in Section 2.2. We then give a characterization of Pareto optimal matchings in CHA in Section 2.3, which we subsequently use in Section 3.4 to construct an  $O(\sqrt{Cm})$  time algorithm for finding a maximum Pareto optimal matching given an instance I of CHA where C is the total capacity of the houses and m is the total length of preference lists in I respectively. Note that we reuse most of the terminology and notation from HA as defined in Section 1.3.1, and we explicitly define relevant concepts only where we need to adapt them to CHA.

#### 3.1 Basic terminology and preliminary results

Let I be an instance of CHA, and let G = (A, H, E) be the underlying bipartite graph of I as defined in Section 1.3.1. Each house  $h_j \in H$  has a capacity  $c_j \ge 1$  which indicates the maximum number of agents that may be assigned to it. Recall from Section 1.3.1 that an assignment M is a subset of  $A \times H$  such that (a<sub>i</sub>,  $h_j \in M$  only if  $a_i$  finds  $h_j$  acceptable.

Furthermore, if  $(a_i, h_j) \in M$ , we say that  $a_i$  and  $h_j$  are assigned to each other, and we call  $a_i$  and  $h_j$  partners in M. A matching M in an instance I of CHA is an assignment such that (i) each agent is assigned to at most one house in M, and (ii) each house  $h_j \in H$  is assigned to at most  $c_j$  agents in M. If an agent  $a_i \in A$ is assigned in M, we denote by  $M(a_i)$  the house that  $a_i$  is assigned to in M. We define  $M(h_j)$  to be the set of agents assigned to  $h_j$  in M (thus  $M(h_j)$  could be empty). We say that a house  $h_j \in H$  is full in M if  $|M(h_j)| = c_j$ , and undersubscribed in M if  $|M(h_j)| < c_j$ . We assume that no agent has an empty preference list and each house is acceptable to at least one agent so that  $m \ge \max$  $\{n^1, n^2\}$ . Let  $C = \sum_{j=1}^{n_2} c_j$  denote the sum of the capacities of the houses.

#### **3.2** Characterisation of Pareto Optimal Matchings

Let M be a matching in I. We say that M is maximal if there is no agent  $a_i \in A$  and house  $h_j \in H$  such that  $a_i$  is unassigned in M,  $h_j$  is under subscribed in M and  $a_i$  finds  $h_j$  acceptable. Also, M is trade-in-free if there is no (agent,house) pair  $(a_i, h_j)$  such that  $a_i$  is assigned in M,  $h_j$  is under subscribed in M and  $a_i$  prefers  $h_j$  to M( $a_i$ ).

A cyclic coalition with respect to M is a sequence of distinct assigned agents  $C = ha_0, a_1, \ldots, a_{r-1}i$ , for some  $r \ge 2$ , such that  $a_i$  prefers  $M(a_{i+1})$  to  $M(a_i)$ for each i  $(0 \le i \le r - 1)$ . Henceforth, all subscripts are taken modulo r when reasoning about coalitions. Given a cyclic coalition C, the matching

$$M^{1} = (M \setminus \{(a_{i}, M(a_{i})) : 0 \le i \le r - 1\}) \cup \{(a_{i}, M(a_{i+1})) : 0 \le i \le r - 1\}$$

is defined to be the matching obtained from M by satisfying C. We say that M is cyclic-coalition-free if M admits no cyclic coalition. The following lemma gives a necessary and sufficient condition for a matching to be Pareto optimal.

**Lemma 3.3.1.** Let M be a matching in a given instance I of CHA. Then M is Pareto optimal if and only if M is maximal, trade-in-free and cyclic-coalitionfree.

**Proof.** Let M be a Pareto optimal matching. Suppose, for a contradiction that M is not maximal. It follows that there exist an agent  $a_i$  and a house  $h_j$  such that  $a_i$  is unassigned in M,  $h_j$  is under subscribed in M and  $a_i$  finds  $h_j$  acceptable. Let  $M^1 = M \cup \{(a_i, h_j)\}$ . Then,  $M^1 < M$ , a contradiction. Now, suppose for a contradiction that M is not trade-in-free.

It follows that there exist an agent  $a_i$  and a house  $h_j$  such that  $a_i$  is assigned in M,  $h_j$  is under subscribed in M, and  $a_i$  prefers  $h_j$  to  $M(a_i)$ . Let  $M^1 = (M \setminus \{(a_i, M(a_i))\} \cup \{(a_i, h_j)\}$ . Then,  $M^1 < M$ , a contradiction. Finally, suppose that M admits some cyclic coalition C.Let  $M^1$  be the matching obtained by satisfying C. Clearly, then,  $M^1 < M$ , a contradiction.

Conversely, let M be a matching that is maximal, trade-in-free and cycliccoalition-free. Let us suppose for a contradiction that M is not Pareto optimal. Then there exists some matching M<sup>1</sup> such that M<sup>1</sup>< M. Let G be the underlying graph of I. We clone G to obtain a cloned graph C(G) as follows. We replace every house  $h_j \in H$  with the clones  $h_{j}^1, h_{j}^2, \ldots, h_{j}^{c,j}$ . We then divide the capacity of each house among its clones by allowing each clone to have capacity 1. In addition, if  $(a_i, h_j) \in G$ , then we add  $(a_i, h_j^p)$  to C(G) for all p  $(1 \le p \le c_j)$ . Let us then adapt the matching M in G to obtain its clone C(M) in C(G) as follows. If a house  $h_j$  in G is assigned to  $x_j$  agents  $a_1, ...a_{xj}$  in M, then we add  $(a_p, h^{p_j})$  to C(M)for  $1 \le p \le x_j$ , so that |C(M)| = |M|. We repeat a similar process for  $M^0$  to obtain its clone  $C(M^1)$  in C(G).

Let us consider  $X = C(M) \bigoplus C(M^1)$  and let C be a connected component of X. It follows that C is a path or cycle whose edges alternate between C(M) and  $C(M^{1})$ . Now, C cannot be an even-length alternating path that has more agents than houses or an an odd-length alternating path whose end edges are in C(M), for otherwise we have an agent who is assigned in M but unassigned in  $M^1$ , a contradiction since M<sup>1</sup><M. In addition, C cannot be an even-length alternating path that has more houses than agents or an odd-length alternating path whose end edges are in  $C(M^1)$  because there then exists an agent  $a_i$  in C who becomes assigned in M<sup>1</sup> to a house h<sub>i</sub> which is under subscribed in M. Now, since there are no ties in preference lists,  $a_i$  must prefer  $h_j$  to  $M(a_i)$  for otherwise  $M^1 < M$ . However, M is then not trade-in-free, a contradiction. Hence, C must be a cycle. Here, each agent  $a_i$  in C is assigned in both M and M<sup>1</sup> and since M<sup>1</sup> < M, each  $a_i$ prefers  $M^1$  to M. However, C is then a cyclic coalition with respect to M, a contradiction. It follows that M<sup>1</sup> < M and M is Pareto optimal. Henceforth we will establish the Pareto optimality of a given matching M in an instance I of CHA by showing that M is maximal, trade-in-free and cyclic-coalition-free. We now show that Lemma 3.3.1 leads to an O(m) algorithm for testing M for Pareto optimality. Let G be the underlying graph of I. Then, we can check if M is maximal and trade-in-free in O(m) time by a traversal of the edges in G. To check if M is cyclic-coalition-free, we construct the envy graph [3] of M as follows. We form a directed graph G<sub>M</sub> of M by letting G<sub>M</sub> consist of one vertex for each agent assigned in M.

Algorithm 2 Algorithm Greedy-PaCHA

 $\overline{\mathbf{M}} := \mathbf{\emptyset};$ 

1: for each agent a<sub>i</sub> in turn do

- 2: if there exists some undersubscribed house in a<sub>i</sub>'s preference list then
- 3: let h<sub>j</sub> be the most-preferred such house;
- 4:  $M := M \cup \{(a_i, h_j)\};$

We then construct an edge from an agent  $a_i$  to another agent  $a_j$  in  $G_M$  if  $a_i$  prefers  $M(a_j)$  to  $M(a_i)$ . It follows that M is cyclic-coalition-free if and only if  $G_M$  is acyclic. Note that even though M is a matching of a CHA instance, all vertices in  $G_M$  have only unitary capacity (being agent vertices). It follows that a depth-first search suffices to detect any cycles in O(m) time so that these observations lead us to the following lemma.

**Lemma 3.3.2.** Let M be a matching in a given instance of CHA. Then we may check whether M is Pareto optimal in O(m) time.

**Proof:** Now, given an instance I of CHA, a greedy approach using the serial dictatorship mechanism of [1] gives us a straightforward algorithm, Algorithm Greedy-PaCHA as shown in Algorithm 2, for finding a Pareto optimal matching M in I. Here, we consider each agent  $a_i$  in turn and give  $a_i$  his most preferred house that is currently under subscribed in the matching built so far. The following lemma shows that the matching constructed by the algorithm must be Pareto optimal.

**Lemma 3.3.3.** Let M be the matching returned by an execution of Algorithm Greedy-PaCHA. Then, M is Pareto optimal.

**Proof.** For, suppose not. For each  $a_i \in A$ , let  $A_i$  denote the set of acceptable houses for  $a_i$ . Consider an agent  $a_i$  who is unassigned in M. It follows that  $A_i$  contains no undersubscribed house  $h_j$ , otherwise  $(a_i, h_j)$  would have been added
to M, a contradiction. Hence, M is maximal. If M is not trade-in-free, then there exists an agent  $a_i$  who prefers some under subscribed house  $h_j$  to M( $a_i$ ). This is a contradiction, since  $h_j$  must be full at the point when we assign  $a_i$  to M( $a_i$ ). If M is not cyclic-coalition-free, let us then consider the coalition C =  $ha_0$ ,  $a_1$ , ...,  $a_{r-1}i$  which exists with respect to M. It follows that there exists some agent  $a_i$  ( $0 \le i \le r - 1$ ) in C who was considered first by the algorithm. By definition of C,  $a_i$  prefers M( $a_{i+1}$ ) to M( $a_i$ ). Now,  $a_{i+1}$  must be considered by the algorithm after  $a_i$ . However, it follows that M( $a_{i+1}$ ) must then have had at least one place

Agent	Pref list	House	Capacity
a <sub>1</sub> :	$h_1 h_2$	h <sub>1</sub> :	1
a <sub>2</sub> :	$h_1$	h <sub>2</sub> :	1

Figure 3.1: An instance I<sub>1</sub> of CHA

free when  $a_i$  was assigned to  $M(a_i)$ , a contradiction to the fact that the algorithm gives each agent his most preferred under subscribed house. Hence, M is Pareto optimal.

The main drawback of Algorithm Greedy-PaCHA is that a given CHA instance may admit Pareto optimal matchings of different cardinalities but Algorithm Greedy-PaCHA may fail to find a Pareto optimal matching of maximum cardinality. For example, Figure 2.1 shows a given CHA instance in which Algorithm Greedy-PaCHA returns a Pareto optimal matching  $M_1 = \{(a_1, h_1)\}$  of cardinality 1, given the agent ordering  $ha_1$ ,  $a_2i$ , and constructs the maximum Pareto optimal matching  $M_2 = \{(a_1, h_2), (a_2, h_1)\}$  of cardinality 2 given the agent ordering  $ha_2$ ,  $a_1i$ . It follows that the order in which the agents are considered can have a consequence on the cardinality of the outcome. This is significant from a practical point of view, given that a prime objective in many matching applications is to assign as many agents as possible.

We remark that a straightforward way to find a maximum Pareto optimal matching given a CHA instance I is by constructing a maximum cardinality minimum weight matching as follows. For each edge  $(a_i, h_j)$  in the underlying graph G of I, we assign a weight wt $(a_i, h_j)$  to the edge by letting wt $(a_i, h_j) = \operatorname{rank}_{ai}(h_j)$  where  $\operatorname{rank}_{ai}(h_j)$  denotes the rank of  $h_j$  in  $a_i$ 's preference list. Call this weighted graph G<sup>1</sup>. We then construct a maximum cardinality minimum weight matching in G<sup>1</sup>. The following lemma shows that such a matching must be a maximum Pareto optimal matching in I.

**Lemma 3.3.4.** Let M be a maximum cardinality minimum weight matching in  $G^1$ . Then, M is a maximum Pareto optimal matching in I.

**Proof.** Suppose not. Since M is a maximum matching, it follows that M is maximal. Now, if M is not trade-in-free, then there exists a (agent,house) pair ( $a_i$ ,  $h_j$ ) such that  $a_i$  is assigned in M,  $h_j$  is undersubscribed in M and  $a_i$  prefers  $h_j$  to M( $a_i$ ). Consider the matching M<sup>1</sup> = (M\( $a_i, M(a_i)$ )) U ( $a_i, h_j$ ). It is clear that  $|M^1| = |M|$  and so M<sup>1</sup> is another maximum cardinality matching of G<sup>1</sup>. However, since  $a_i$  prefers  $h_j$  to M( $a_i$ ), the weight of M<sup>1</sup> must be smaller than the weight of M, a contradiction.

Hence, suppose that M admits some cyclic coalition  $C = ha_0$ ,  $a_1$ , ...,  $a_{r-1}i$ . Let  $M^1$  be the matching obtained by satisfying C. Then, it is clear that  $|M^1| = |M|$ again. Moreover, since each  $a_i$  prefers  $M(a_i)$  to  $M^1(a_i)$  for  $0 \le i \le r-1$ , the weight of  $M^1$  is again smaller than the weight of M, a contradiction.

Note that the above lemma also indicates that a maximum Pareto optimal matching in I has the same cardinality as a maximum matching in G and any maximum cardinality minimum weight matching of  $G^0$  gives us a maximum Pareto optimal matching in I.

A well known transformation in matching theory allows us to transform

the problem of finding a maximum cardinality minimum weight matching into the Assignment problem. Recall from Section 1.2 that we can solve the Assignment problem in the capacitated bipartite graph in O(C min(m log n,  $n^2$ )) time [15], so this allows us to find a maximum Pareto optimal matching in the same time complexity. However, since the problem of finding a maximum matching in the capacitated bipartite graph takes O( $\sqrt{C}$ m) time, it is of interest to consider whether faster algorithms for finding a maximum Pareto optimal matching in CHA exist.

## **3.4 Maximum Pareto Optimal Matchings**

In this section, we describe a three-phase algorithm for finding a maximum Pareto optimal matching in CHA by satisfying the necessary and sufficient conditions in Lemma 3.3.1.

Let I be an instance of CHA and G be its underlying graph. The problem of finding a maximum matching in G can be viewed as an instance of maximum cardinality DCS [15] as described in Section 1.2 (the two problems are essentially the same, except that agents have no explicit preferences in the DCS case; the definition of a matching is unchanged).

Hence, Phase 1 of the algorithm uses Gabow's algorithm [15] to compute a maximum matching M in G. This phase guarantees that M is maximal and takes  $O(\sqrt{C}m)$  time. The next two phases ensure that M is trade-in-free and cycliccoalition-free respectively as detailed below..

#### **3.4.1Phase 2 of the algorithm**

In this phase, we transform M into a trade-in-free matching by conducting a repeated search for (agent,house) pairs (a<sub>i</sub>, h<sub>j</sub>) such that h<sub>j</sub> is under subscribed in

M and  $a_i$  prefers  $h_j$  to  $M(a_i)$ . Whenever such a pair is found, the algorithm breaks the existing assignment

#### Algorithm 3 Phase 2 loop

while S  $6 = \emptyset$  do

- 1:  $h_j := S.pop();$
- 2:  $(a_i, r) := L_j$ .removeHead();
- 3: if  $r < curr_{ai}$  then

4: 
$$h_{j}^{1} := M(a_{i});$$

5:  $M := (M \setminus (a_i, h^1_j)) \cup \{(a_i, h_j)\};$ 

6: 
$$\operatorname{curr}_{ai} := r;$$

- 7: if  $|M(h_j)| < c_j$  and  $L_j \neq \emptyset$  then
- 8: S.push( $h_j$ );

9: 
$$h_j := h_j^1$$
;

- 10: if  $L_j \neq \emptyset$  and  $h_j \in S$  then
- 11: S.push(h<sub>j</sub>);

between  $a_i$  and  $M(a_i)$ , and promotes  $a_i$  to  $h_j$ . It follows that a space in  $M(a_i)$ becomes freed in the process, which may consequently be assigned to some assigned agent  $a_k$  who prefers  $M(a_i)$  to  $M(a_k)$ . Note that if  $h_j$  remains under subscribed after such a step, it may also be assigned to some assigned agent  $a_l$ who prefers  $h_j$  to  $M(a_l)$ . We show how to obtain a trade-in-free matching from M by using a slight modification of the Phase 2 loop of the algorithm described by Abraham et al. [3] to find a maximum Pareto optimal matching in HA.

For each house  $h_j$ , we maintain a linked list  $L_j$  of pairs ( $a_i$ , r) where  $a_i$  is an assigned agent who prefers to be assigned to  $h_j$  than  $M(a_i)$  at the start of Phase 2,

and r is the rank of  $h_j$  in  $a_i$ 's preference list. Note that the pairs in  $L_j$  may subsequently contain an agent  $a_i$  who prefers  $M(a_i)$  to  $h_j$  if  $M(a_i)$  is no longer the house that  $a_i$  was assigned to at the start of Phase 2 as a result of promotions executed over the course of the algorithm. We will maintain a stack S of all under subscribed houses  $h_j$  where  $L_j$  is non-empty. Also, for each house  $h_j$ , we assume that we store a counter for  $|M(h_j)|$ . For each assigned agent  $a_i$ , let curr<sub>ai</sub> be a variable which stores the rank of  $M(a_i)$  in  $a_i$ 's preference list.

Let us now consider the pseudocode of the Phase 2 loop as shown in Algorithm 3. During each iteration of the main while loop, we pop an undersubscribed house  $h_j$  from S and remove the first pair ( $a_i$ , r) from  $L_j$  (which must be non-empty). Now, if  $r < curr_{ai}$ , it follows that  $a_i$  prefers  $h_j$  to  $M(a_i)$  so we promote  $a_i$  from  $h^1_j = M(a_i)$  to  $h_j$  and we update M and curr<sub>ai</sub> in the process. Now, if  $h_j$  remains under subscribed at the end of this step, then we push  $h_j$  back onto S if  $L_j$  is non-empty. We also push  $h^1_j$  onto S if  $L_hO_j$  is non-empty and if  $h^1_j$  is not already in S. Otherwise, if  $r \ge curr_{ai}$ , we push  $h_j$  back onto S if  $L_j$  is non-empty.

Now, the algorithm must terminate, for each iteration of the main while loop removes a pair from a list  $L_j$  but no new pair is ever added to any list during a loop iteration. Hence, the algorithm terminates when S is empty. It must be the case that when this happens no assigned agent  $a_i$  prefers an under subscribed house to M( $a_i$ ), so that M is trade-in-free as a result. Moreover, since each agent assigned at the end of Phase 1 is also assigned at the end of Phase 2, M remains a maximum matching. Let us then consider the time complexity of Phase 2. We can initialize all variables used in the Phase 2 loop in O(m) time using a single traversal of the agents' preference lists. The number of iterations of the main while loop is bounded above by the total length of preference lists. It is straightforward to verify that each operation within the while loop takes constant time (with a suitable choice of data structures such as those described later in Section 2.4.3). Hence, the algorithm runs in O(m) time, giving us the following result.

**Lemma 3.4.1.** Given a maximum matching M in an instance of CHA, the Phase 2 loop ensures that M is trade-in-free in O(m) time.

#### **3.4.2Phase 3 of the algorithm**

In this phase, we transform M into a matching  $M^1$  that admits no cyclic coalition by using a modification of the linear-time extension [3] of Gale's Top Trading Cycles Method [57].

This phase consists of a preprocessing step which we will describe in detail, and then the main Phase 3 loop shown in Algorithm 5. Throughout Phase 3, we maintain a stack of agents P which will help us to identify cyclic coalitions. The matching  $M^0$  and the stack P are empty at the start of Phase 3. For each agent a<sub>i</sub>, we maintain a pointer  $p(a_i)$  to the first house on a<sub>i</sub>'s preference list, and subsequently  $p(a_i)$  traverses left to right over the course of execution of Phase 3. We will also maintain a queue of agents Q, each of whom is an agent a<sub>i</sub> waiting to be assigned to  $p(a_i)$  in M<sup>1</sup>. In addition, for each house h<sub>j</sub>, we will use  $M_0(h_j)$  to store those agents who are assigned to h<sub>j</sub> in M but who are unassigned in M<sup>1</sup> so far in the execution of Phase 3. Initially,  $M_0(h_j)$  will contain all those agents assigned to h<sub>j</sub> in M. As we assign agents in  $M_0(h_j)$  to houses in M<sup>1</sup>, we will remove these agents from  $M_0(h_j)$ . Finally, we also maintain a linked list L<sub>j</sub> for each house h<sub>j</sub> containing agents such that if a<sub>i</sub> is an agent in L<sub>j</sub>, then a<sub>i</sub> prefers h<sub>j</sub> to  $M(a_i)$ .

Algorithm 4 Process (Q)

while  $Q \neq \emptyset$  do

- 1: AI: = Q. removes Head ();
- 2:  $h_i := p(a_i);$

- 3:  $h_k := M(a_i); \{ // \text{ possibly } h_j = h_k \}$
- 4:  $M^1 := M^1 \cup \{(a_i, h_j)\};$
- 5: label  $a_i$ ;
- 6: if  $a_i \in P$  then
- 7: remove  $a_i$  from P;
- 8:  $M_0(h_k) := M_0(h_k) \setminus \{a_i\};$
- 9: if  $|M^1(h_j)| = c_j$  then
- 10: for each unlabelled  $a^{1}_{i} \in L_{j}$  do
- 11: delete  $h_j$  from the preference list of  $a^{1}_i$ ;
- 12: if  $p(a^{1}_{i}) = M(a^{1}_{i})$  then
- 13: Q.add $(a^{1}_{i})$ ;

# **Pre-processing step**

Let us now introduce the pre-processing step which helps to reduce the number of iterations of the Top Trading Cycles Method in the main Phase 3 loop. This step makes use of the observation (as in [3]) that no agent  $a_i$  assigned to his first choice house  $h_j$  in M can be involved in a cyclic coalition. At the outset of Phase 3, we check if  $p(a_i) = M(a_i)$  for each agent  $a_i$  and add every such  $a_i$  to Q.

If Q is non-empty, then we run the sub-routine Process(Q), shown in Algorithm 4, as the preprocessing step. Note that this usage of Process(Q) is prior to the main Phase 3 loop starting, but it will be used again in general during the main Phase 3 loop. This sub-routine considers each agent  $a_i$  in Q in turn, by removing  $a_i$  from Q and then adding the edge  $(a_i, h_j)$  to M<sup>1</sup>. Every such  $a_i$  is then labelled to differentiate  $a_i$  from those agents unassigned in M<sup>1</sup> so far in the execution of the algorithm (all agents are initially unlabelled at the outset of Phase 3). Now, P must be empty during pre-processing. However, this may not be true during a subsequent execution of Process(Q) by the main Phase 3 loop. Hence, Process(Q) checks if  $a_i$  lies in P , and if so, removes  $a_i$  from P so as to remove the agent from further consideration by the main Phase 3 loop, since  $a_i$  has just been assigned in M<sup>1</sup>. Let  $p(a_i) = h_j$ . Now, if  $|M^1(h_j)| = c_j$  after the assignment of  $a_i$  to  $h_j$ , then we remove  $h_j$  from the preference lists of the remaining agents since such a house that is full in M<sup>1</sup> could not subsequently be involved in a cyclic coalition. We refer to those preference lists in which houses have been removed as reduced preference lists. We then apply the observation made at the start of this subsection recursively to the reduced preference lists of the remaining agents until either (i) no agents remain unassigned in M<sup>1</sup>, or (ii) at least one agent is not assigned to his reduced first choice (i.e. the first choice on his reduced preference list) in M<sup>1</sup> and so cannot be involved in any cyclic coalition as Lemma 2.4.4 on page 31 will establish. The following lemma shows that when case (ii) happens at the end of the preprocessing step, a cyclic coalition must exist with respect to M.

**Lemma 3.4.2.** Suppose that pre-processing terminates, and there exists an agent who is unassigned in  $M^1$ . Then a cyclic coalition must exist with respect to M.

**Proof.** Let  $a_0$  be an agent who is not assigned in  $M^1$  to his reduced first choice  $p(a_0)$  at the end of preprocessing. Hence,  $a_0$  is an unlabelled agent and  $p(a_0) \neq M(a_0)$ . It follows that  $p(a_0)$  must be full in M for otherwise M is not trade-infree, a contradiction. However,  $p(a_0)$  cannot be full in  $M^1$  for otherwise  $p(a_0)$  would have been removed from  $a_0$ 's preference list by preprocessing and cannot be the reduced first-choice house of  $a_0$ .

Hence, there exists some agent  $a_1 \in M(p(a_0)) \setminus M^1(p(a_0))$  because if an agent a is assigned in  $M^1$  by Process(Q) in pre-processing, it must be the case that a must be assigned in  $M^1$  to M(a). It follows immediately that  $a_1$  must be unassigned in  $M^1$ . Furthermore,  $p(a_1) \neq M(a_1)$  (or else  $a_1 \notin$  $M(p(a_0)) \setminus M^1(p(a_0)))$  so that  $p(a_0) \neq p(a_1)$ . By reusing the same argument, it follows that we can trace a sequence of agents  $S = ha_0$ ,  $a_1$ , ...i such that  $a_i$  is assigned in M but unassigned in M<sup>1</sup> and  $p(a_i) = M(a_{i+1})$  for  $i \ge 0$ . Since the number of agents is finite, there must be some r such that  $a_r = a_x$  for some  $0 \le x \le r - 1$ , where without loss of generality  $a_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}$  are distinct agents. However, the substring of agents  $C = ha_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}i$  within S must then constitute a cyclic coalition with respect to M.

Now, it is clear that an (unlabelled) agent  $a_i$  can only be added to Q when the last house that  $a_i$  prefers to  $M(a_i)$  gets removed from his preference list so that  $p(a_i)$  becomes equal to  $M(a_i)$ , and this happens only once in pre-processing. Since no agent is added to Q twice, the while loop of process(Q) is bound to terminate. As a result, the pre-processing step must also terminate.

Algorithm 5 Main Phase 3 loop

for each unlabelled agent ai do

1: 
$$P := \{a_i\}; \{// P \text{ is a stack of agents}\}$$

3: 
$$c(a_i) := 1$$
; {// counter record the number of times an agent is in P }

4: while 
$$P \neq \emptyset$$
 do

5: 
$$a^{1}_{i} := P.pop();$$

6: if  $c(a_i^1) = 2$  then

7: 
$$a^{11}_i := a^{1}_i;$$

8: repeat

9: Q.add
$$(a^{11}_i)$$
;

10: 
$$a^{11}_i := P.pop();$$

- 11: until  $a^{11}_i = a^1_i$
- 12: call Process(Q);
- 13: else
- 14: **P.** $push(a^{1}_{i});$
- 15: choose any  $a^{11}_i \in M_0(p(a^1_i));$

16: 
$$c(a^{11}_i) := c(a^{11}_i) + 1;$$
  
17: P.push $(a^{11}_i);$ 

# 3.4.2.2Phase 3 loop

We then make use of the algorithm in the main Phase 3 loop, as shown in Algorithm 5 to construct the envy graph in order to detect and satisfy cyclic coalitions. For each agent  $a_i$  who is not assigned to his reduced first-choice in M, we repeatedly build a path of agents (represented by P) starting from  $a_i$  in the main while loop and check if P cycles. To do so, we initialize a counter  $c(a_i)$  to 0 for each agent  $a_i$ .

Now, if  $c(a^{1}_{i}) \neq 2$  for some agent  $a^{1}_{i}$  in P during an iteration of the while loop, then we extend P by following the reduced first-choice edge of  $a^{1}_{i}$  in line 15. Let  $p(a^{1}_{i}) = h_{j}$  and let  $a^{11}_{i}$  be any member of  $M_{0}(h_{j})$ . Note that  $M_{0}(h_{j})$  must be non-empty. For, suppose not. Since  $a^{1}_{i}$  prefers  $h_{j}$  to  $M(a^{1}_{i})$ ,  $h_{j}$  must be full in M (or else M is not trade-in-free). Each agent  $a_{k}$  assigned to  $h_{j}$  in M either becomes assigned to  $h_{j}$  again in  $M^{1}$  if  $p(a_{k}) = M(a_{k})$ , or to some other house via the satisfaction of some cyclic coalition. In the latter case, this causes some agent  $a_{l}$ to be assigned to  $h_{j}$  in  $M^{1}$  in  $a_{k}$ 's place. Since  $h_{j}$  is not full in  $M^{1}$  by definition of  $p(a^{1}_{i}) = h_{j}$ , it follows that there exists some agent belonging to  $M(h_{j})$  who is currently unassigned in  $M^{1}$ . Hence,  $M_{0}(h_{j})$  must be non-empty. Otherwise, if  $c(a^{1}_{i}) = 2$ , it follows that we have a cyclic coalition in P starting from  $a^{1}_{i}$ .

We satisfy C by popping each agent  $a^{11}{}_i$  in C from P until we remove C, and add each  $a^{11}{}_i$  to Q. We then call Process(Q) to assign each  $a^{11}{}_i$  to  $p(a^{11}{}_i)$  in M<sup>1</sup>, to label each  $a^{11}{}_i$  in order to remove the agent from further consideration by the algorithm, as well as to remove M<sup>1</sup>( $a^{11}{}_i$ ) from the preference lists of the remaining unlabelled agents if the house becomes full in M<sup>1</sup>.

#### **3.4.2.3**Correctness of Phase 3 loop

If there are unlabelled agents at the start of the main Phase 3 loop, there must exist at least one cyclic coalition with respect to M involving a subset of these agents by Lemma 3.4.2. The following lemma strengthens this result by showing that if there exist any unlabelled agents at any point of time in the execution of Phase 3, then a cyclic coalition must exist.

**Lemma 3.4.3.** Consider a given iteration of the for loop in Phase 3. If there exists an agent who remains unlabelled, then a cyclic coalition must exist with respect to M.

**Proof.** Let  $a_0$  be an agent who is unlabelled during a given iteration of the for loop of Phase 3. Then,  $a_0$  is not assigned in M<sup>1</sup> to his reduced first choice  $p(a_0)$ . It follows that  $p(a_0) \neq M(a_0)$ . Now,  $p(a_0)$  must be full in M for otherwise M is not trade-in-free, a contradiction.

However,  $p(a_0)$  cannot be full in  $M^1$  for otherwise  $p(a_0)$  would have been removed from  $a_0$ 's preference list by Process(Q) and cannot be the reduced firstchoice house of  $a_0$ . Now, each agent  $a^1 \in A$  becomes assigned in  $M^1$  to either  $M(a^1)$  when  $M(a^1) = p(a^1)$ , or to  $p(a^1)$  when we satisfy a cyclic coalition involving  $a^0$ . Since  $p(a_0)$  is currently under subscribed in  $M^1$ , it follows that there exists a non-empty subset of agents  $A_s$  such that each agent a in  $A_s$  belongs to  $M(p(a_0))\backslash M^1(p(a_0))$  and a is currently unassigned in  $M^1$ . Let  $a_1 \in A_s$ . It must be the case that  $p(a_1) \neq M(a_1)$  and hence,  $p(a_1) \neq p(a_0)$ . By reusing the same argument, it follows that we can trace a sequence of agents  $S = ha_0$ ,  $a_1$ , ... is such that  $a_i$  is assigned in M but unassigned in  $M^1$  and  $p(a_i) = M(a_{i+1})$  for  $i \ge 0$ . Since the number of agents is finite, there must be some r such that  $a_x = a_r$  for some  $0 \le$ x < r - 1, where without loss of generality  $a_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}$  are distinct agents. However, the substring of agents  $C = ha_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}$  is within S must then constitute a cyclic coalition with respect to M.

The next lemma shows that when all agents are assigned in M<sup>1</sup>, we then obtain a cyclic-coalition-free matching.

**Lemma 3.4.4.** If no unlabelled agents remain at any stage of Phase 3, then M<sup>1</sup> is cyclic-coalition-free.

**Proof.** If there does not exist any unlabelled agents, then every agent is assigned in M<sup>1</sup>. Let  $a_0$  be an arbitrary agent. Suppose that there is a cyclic coalition C =  $ha_0, a_1, ..., a_{r-1}i$  with respect to M<sup>1</sup> involving  $a_0$ . Let M<sup>1</sup>( $a_0$ ) =  $h_j$  and let M<sup>1</sup>( $a_1$ ) =  $h_k$ . By definition of C,  $a_0$  must prefer  $h_k$  to  $h_j$ . Since  $a_0$  was assigned to  $h_j$  instead of  $h_k$  in M<sup>1</sup>, it follows that  $h_k$  must have been full in M<sup>1</sup> at the time that  $a_0$  was assigned to  $h_j$  in M<sup>1</sup>. It must then be the case that  $a_1$  was considered by Phase 3 before  $a_0$  or else M<sup>1</sup>( $a_1$ )  $\neq h_k$ . Now, by applying the same argument to the remaining agents in C, we can establish that  $a_{r-1}$  must have been considered by Phase 3 before  $a_0$ . Let M<sup>1</sup>( $a_{r-1}$ ) =  $h_1$ . It follows that  $a_{r-1}$  must prefer  $h_j$  to  $h_1$ . Now, it must be the case that at the time that  $a_{r-1}$  was assigned in M<sup>1</sup>,  $h_j$  must have been under subscribed for otherwise  $a_0$  could not have been assigned to  $h_j$  later on. However, this gives a contradiction for  $a_{r-1}$  prefers  $h_j$  to  $h_1$  and should then be assigned to  $h_j$  by Phase 3 instead.

Suppose that the envy graph involves the sequence of agents  $S = ha_0$ ,  $a_1$ , ...,  $a_{r-1}i$  and suppose that only a substring of these agents  $C = ha_i$ ,  $a_{i+1}$ , ...,  $a_{r-1}i$  constitute a cyclic coalition where  $0 \le i < r - 1$ . Let us call the agents in the substring  $ha_0$ ,  $a_1$ , ...,  $a_{i-1}i$  the tail of C. Now, if certain houses become full in  $M^0$  as a result of satisfying C, thereby causing  $M(a_{i-1})$  to become the reduced first choice house for the agent  $a_{i-1}$ , then  $a_{i-1}$  gets added to Q and assigned to  $M(a_{i-1})$  in  $M^1$  subsequently by Process(Q). Note that this can cause an unwinding effect in the tail in which each agent  $a_k$  ( $0 \le k \le i - 2$ ), such that  $M(a_k)$  lies immediately

after  $p(a_k)$  in  $a_k$ 's reduced preference list, gets added to Q and assigned to  $M(a_k)$ in  $M^1$  by Process(Q) in descending agent subscript order until either we reach an agent  $a_{k-1}$  such that there exists a house between  $p(a_{k-1})$  and  $M(a_{k-1})$  in  $a_{k-1}$ 's reduced preference list or the tail becomes empty as a result. In the former case, the main Phase 3 loop, then extends P by following the reduced first-choice edge of  $a_{k-1}$ . In the latter case, the main Phase 3 loop tries to extend P by following the reduced first-choice edge of the next unlabelled agent, if one exists.

It is straightforward to see that the labelling of agents and the maintenance of  $c(a_i)$  for each agent  $a_i$  ensures that no agent  $a_i$  is added more than twice to P in Phase 3. Clearly, if P is non-empty, P must cycle at some point of time in the execution of Phase 3 (as observed by Lemma 2.4.3). Since each agent  $a_i$  that we add to P belongs to a cyclic coalition or to the tail of a cyclic coalition, we are bound to remove  $a_i$  from P as a result.

It also follows by the labelling of agents that any agent added to Q by the for loop in the main Phase 3 loop is not added to Q again by Process(Q) and vice versa. Hence, no agent gets added to Q twice.

Hence, Phase 3 must terminate when no unlabelled agents remain. When this happens, it follows by Lemma 3.4.4 that  $M^0$  must be a cyclic-coalition-free matching. We next show that each agent  $a_i$  assigned in M at the end of Phase 2 must also be assigned in  $M^1$  at the end of Phase 3.

**Lemma 2.4.5.** Each agent  $a_i$  assigned in M at the end of Phase 2 is also assigned in M<sup>1</sup> at the end of Phase 3.

**Proof.** Suppose not. Then, let  $a_i$  be an agent who is unassigned in  $M^1$ . Let  $M(a_i) = h_j$ . Then,  $a_i \in M(h_j) \setminus M^1(h_j)$ . It follows that  $h_j$  cannot have been the first house on  $a_i$ 's preference list, or else preprocessing would have assigned  $M^1(a_i)$  to be  $h_j$ . Hence, there exists at least one house that  $a_i$  prefers to  $h_j$ . Now, if  $a_i$  is not

assigned in  $M^1$  to any of these houses, then the pointer  $p(a_i)$  should move across  $a_i$ 's preference list until it points at  $h_j$ . When this happens,  $a_i$  should then be assigned to  $h_j$  in  $M^1$ . However,  $a_i$  is unassigned in  $M^1$  so that  $h_j$  must have been removed from  $a_i$ 's preference list prior to this as a result of it becoming full in  $M^1$ . Now, if  $h_j$  is full in  $M^1$ , then for every  $a_k \in M^1(h_j)$ , either  $a_k \in M(h_j)$  or there exists a unique  $a_l \in M(h_j) \setminus M^1(h_j)$  such that  $a_k$  and  $a_l$  belong to the same cyclic coalition. However, this implies that  $c_j+1$  agents were assigned to  $h_j$  in M, a contradiction.

Since M is a matching that is also maximum, it follows by Lemma 2.4.5 that  $M^1$  is also a maximum matching.

#### **3.4.3** Implementation and analysis

The time complexity of Phase 3 depends on how efficiently we can implement Process(Q) and the main Phase 3 loop. Let us consider briefly the data structures required.

First of all, let us assume that we represent the stack P as a doubly linked list embedded within an array. We let P contain  $n_1$  elements and we indicate the presence or absence of an agent in P by a 1 or 0 respectively. We maintain a pointer to the top of the stack in addition to previous and next pointers between agents in P at any point of time. We implement Q as a straightforward linked list. We also represent the preference list  $pref_{ai}$  of each agent  $a_i$  as a doubly linked list embedded within an array. Each element in position j of  $pref_{ai}$  stores the house lying in the corresponding position in  $a_i$ 's preference list. Also, for each house h in  $pref_{ai}$ , we maintain a pointer each to the house preceding and following h on  $a_i$ 's preference list respectively. We let  $p(a_i)$  point to the first house in  $pref_{ai}$ , i.e.  $a_i$ 's (reduced) first choice house. Note that as with any doubly linked list,  $p(a_i)$ gets updated as part of the deletion operation whenever the (reduced) first choice house in  $pref_{ai}$  is to be deleted from  $pref_{ai}$ . We then build a rank array rank<sub>ai</sub> for each agent  $a_i$  which stores the position of each house  $h_j$  on  $a_i$ 's preference list, i.e. rank<sub>ai</sub> [j] gives the position of  $h_j$  on  $pref_{ai}$ .

For each house  $h_j$ , we represent  $M_0(h_j)$  also as a doubly linked list embedded within an array. There are  $n_1$  entries in each  $M_0(h_j)$  indexed according to agent subscript, but we maintain previous and next pointers for at most  $c_j$  of these entries, i.e. previous and next pointers exist between consecutive agents in  $M_0(h_j)$  only if these agents belong to  $M(h_j)$  but are currently unassigned in  $M^1$ . It is straightforward to see that  $L_j$  for each  $h_j$  can be implemented as a linked list. We also maintain a counter  $|M^1(h_j)|$  for each house  $h_j$  to keep track of the number of agents assigned to it in  $M^1$  so far in the execution of Phase 3.

Note that for each doubly linked list that we use, we let the previous pointer of the first element and the next pointer of the last element to each point to null. Now, if we use virtual initialization for the initialization of P, it is clear that it takes only a single traversal of the agents' preference lists in agent subscript order to initialize the rest of the data structures.

To illustrate the use of these data structures, suppose firstly that we pop an agent  $a_i$  from Q. It is straightforward to see if  $a_i$  belongs to P by checking if P [i] is 1 or 0. To remove  $a_i$  from P, we set P [i] = 0 and update the previous and next pointers of P [i]'s predecessor and successor, as well as the pointer to the top of P if necessary. Let  $h_k = M(a_i)$ . To remove  $a_i$  from  $M_0(h_k)$ , we update the next and previous pointers of  $M_0(h_k)$ [i]'s predecessor and successor respectively. Let  $p(a_i) = h_j$ . If  $h_j$  becomes full in M<sup>1</sup> as a result of assigning  $a_i$  to  $h_j$ , i.e.  $|M^1(h_j)| = c_j$ , we want to remove  $h_j$  from the preference lists of those unlabelled agents in  $L_j$ . It is clear that it takes only time linear in the size of  $L_j$  to find these agents. Let  $a_i$  be such an agent. Then, rank<sub>al</sub> [j] enables us to look up the position of  $h_j$  on  $a_i$ 's preference list in constant time. We then delete  $h_j$  from pref<sub>al</sub> by updating the

pointers of  $h_j$  's predecessor and successor in pref<sub>al</sub>. In this way,  $p(a_l)$  always points to the head of  $a_l$ 's preference list (which must be currently under subscribed in  $M^1$ ).

Let us then consider the time complexity of Phase 3 by looking at the entire execution of the main Phase 3 loop and Process(Q) taken over the algorithm's execution. It has already been observed that each agent can be added to P no more than twice and added to Q at most once. Now, lines 8-11 of the main Phase 3 loop are executed at most once for each agent in P . All other operations in the main Phase 3 loop are just O(1) stack operations or simple manipulation of data structures. In Process(Q), all operations apart from those in lines 11-14 can be implemented to run in O(1) time for each agent that is added to or removed from Q. Finally, it is clear that lines 11-14 are executed at most once for each house, and hence, the total number of iterations of the for loop is O(m), taken over all calls to Process(Q). It follows that Phase 3 takes O(m) time overall, giving us the following result.

**Lemma 3.4.6**. Given a maximal trade-in-free matching M in an instance of CHA, Phase 3 constructs a cyclic-coalition-free matching  $M^1$  from M in O(m) time. Since Phase 1 dominates the overall complexity of the algorithm to find a maximum Pareto optimal matching in an instance of CHA, we have the following result.

**Theorem 3.4.1.** Given an instance I of CHA, we can find a maximum Pareto optimal matching in I in  $O(\sqrt{Cm})$  time.

Recall that an alternative approach to finding a Pareto optimal matching given an CHA instance I involves obtaining a weighted graph  $G^1$  of I and then finding a maximum cardinality minimum weight matching of I. This takes  $\Omega(C \min(m \log n, n^2))$  time. If mlog  $n \le n^2$ , then it follows that our algorithm is faster by a factor

of  $\Omega(\sqrt{C} \log n)$ . Otherwise, mlog  $n > n^2$  and our lgorithm is faster by a factor of  $\Omega(\sqrt{C}n2/m)$ .

#### **Chapter IV**

# PARETO OPTIMAL MATCHINGS IN HR

# Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### 4.1Introduction

In this chapter, we extend our study of Pareto optimal matchings from Chapter 2 to the bipartite model with two-sided preferences. We focus our attention on the Hospitals-Residents problem without ties (HR), which was introduced in Section 1.4.2.

The main results of this chapter, and their organization are as follows. We give some terminology and preliminary results on Pareto optimal matchings in HR in Section 3.2. We then give a characterization of Pareto optimal matchings in HR in Section 3.3, which we subsequently use in Section 3.4 to construct an  $O(\sqrt{Cm})$  time algorithm for finding a maximum Pareto optimal matching given an instance I of HR, where C is the total capacity of the hospitals and m is the total length of preference lists in I. Finally, in Section 3.5, we show how to adapt our algorithm for HR to obtain a faster  $O(\sqrt{n}m)$  time algorithm to solve the analogous problem given an instance of SMI, a special case of HR, where n is the total number of men and women.

# 4.2Basic terminology and preliminary results

Let I be an instance of HR. We reuse the notations and terminology for HR as

defined in Section 1.4.2, and also provide some additional definitions as follows.

Given a resident  $r_i \in R$  and an acceptable hospital  $h_j$  for  $r_i$ , we define rank<sub>ri</sub>  $(h_j)$  to be the number of hospitals that  $r_i$  prefers to  $h_j$  plus 1. If rank<sub>ri</sub>  $(h_j) = k$ , we say that  $h_j$  is the kth choice of  $r_i$ . In a similar way, we define rank<sub>hj</sub>  $(r_i)$  and the k<sup>th</sup> choice of  $h_j$ . We assume that no resident has an empty preference list and each hospital is acceptable to at least one resident so that  $m \ge max \{n_1, n_2\}$ . Also, let M be a matching in I. We define a vertex v in G to be exposed with respect to M if either (i) v is a resident vertex that is unassigned in M, or (ii) v is a hospital vertex that is under subscribed in M. An augmenting path in G is an alternating path, both of whose end vertices are exposed.

Let  $M^1$  be another matching in I. Recall from Section 1.4.2 that a resident  $r_i$  prefers M to  $M^1$  if either (i)  $r_i$  is assigned in M and unassigned in  $M^1$ , or (ii)  $r_i$  is assigned in both M and  $M^1$  and prefers  $M(r_i)$  to  $M^1(r_i)$ . Furthermore, a hospital  $h_j$  prefers M to  $M^1$  if  $|M(h_j)| > |M^1(h_j)|$ , or  $|M(h_j)| = |M^1(h_j)|$  and  $h_j$  prefers the worst resident assigned to it in M to the worst resident assigned to it in  $M^1$ .

Unlike the case for residents where it is necessary for ties to be present in the preference lists in order for a resident to be in different between any two matchings M and M<sup>1</sup>, a hospital  $h_j$  may be indifferent between M and M<sup>1</sup> if  $|M(h_j)| = |M^1(h_j)|$ , the worst resident assigned to  $h_j$  is the same in both M and M<sup>1</sup> but  $h_j$  has different sets of residents assigned to it in M and M<sup>1</sup>.

Given these definitions, we may define a relation on the set of all matching's in I as in Section 1.3.2.1: that is,  $M < M^1$  if and only if no agent prefers  $M^1$  to M, and some agent prefers M to  $M^1$ . A matching M is defined to be Pareto optimal if and only if it is -minimal. In other words, a matching M is Pareto optimal if there is no other matching  $M^1$  such that  $M^1$  M.

#### 4.3 Characterization of Pareto optimal Matching's in HR

Let M be a matching in I. The following defines a necessary and sufficient condition for M to be Pareto optimal in I.

**Definition 4.3.1.** An improving coalition with respect to M is a sequence of agents  $C = hr_0$ ,  $h_0$ ,  $r_1$ ,  $h_1$ , ...,  $r_{k-1}$ ,  $h_{k-1}i$ , for some  $k \ge 1$ , such that the residents are distinct and:

- 1.  $(r_i, h_{i-1}) \in M (1 \le i \le k 1),$
- 2. Either
  - (a)  $r_0$  is unassigned in M and finds  $h_0$  acceptable, and  $h_{k-1}$  is undersubscribed in M and finds  $r_{k-1}$  acceptable, or
  - (b)  $k \ge 2$ , and  $(r_0, h_{k-1}) \in M$ .
- 3.  $r_i$  prefers  $h_i$  to  $h_{i-1}$  for each i  $(1 \le i \le k 1)$ , and  $r_0$  prefers  $h_0$  to  $h_{k-1}$  if Condition 2(b) holds,
- 4.  $h_i$  prefers  $M^1$  to M or is in different between the matchings for each i  $(0 \le i \le k 1)$  where  $M^1 = M \bigoplus C$ .

If M admits no improving coalition, we say that M is improving coalition-free.

If C satisfies Condition 2(a), we also refer to C as an augmenting coalition, otherwise we also refer to C as a cyclic coalition. We define the size of C to be 2k, and henceforth, all subscripts is taken modulo k when reasoning about improving coalitions. The matching M<sup>1</sup> obtained by  $M^1 = M \oplus C$  is defined to be the matching obtained from M by satisfying C. Note that the hospitals may be repeated in C in view of their non-unitary capacities.

A matching M is maximal in G if  $M \cup \{e\}$  is not a matching for any  $e \in E \setminus M$  where E is the edge set in I. By Definition 3.3.1, M is maximal if and only if M admits no improving coalition of size 2. The following lemma gives a necessary and sufficient condition for a matching in HR to be Pareto optimal.

**Lemma 4.3.1.** Let M be a matching in a given instance I of HR. Then M is Pareto optimal if and only if M is improving coalition-free.

**Proof.** Let M be a Pareto optimal matching. If M admits some improving coalition C, let  $M^1$  be the matching obtained by satisfying C. By Definition 3.3.1, each resident in C prefers  $M^1$  to M and each hospital in C either prefers  $M^1$  to M or is in different between the two matching's. However, this implies that  $M^1 < M$ , a contradiction.

Conversely, let M be a matching that is improving coalition-free, and suppose for a contradiction that M is not Pareto optimal. Then there exists some matching  $M^1$  such that  $M^1$  M. Let  $X = M \oplus M^1$  and let C be a connected component of X. We consider three cases, according to the form of C.

**Case (i):** C is an alternating path with an even number of edges. However, this implies that either there exists a resident  $r_i$  who is assigned in M but who becomes unassigned in M<sup>1</sup>, or there exists a hospital  $h_j$  such that  $|M^0(h_j)| < |M(h_j)|$ . Both possibilities contradict the fact that  $M^1$  M.

**Case (ii):** C is an alternating path with an odd number of edges. Clearly, the end edges of C cannot be in M for otherwise we can obtain a similar contradiction as in Case

(i). Hence, the end edges of C are in  $M^1$ . Let  $r_i$  be the exposed resident vertex in C.

Clearly,  $r_i$  prefers  $M^1$  to M. For each resident  $r_j \neq r_i$  in C, it must be the case that  $r_j$  prefers  $M^1(r_j)$  to  $M(r_j)$  since  $M^1$  M. Hence, every resident in C prefers  $M^1$  to M. Let  $h_x$  be the exposed hospital vertex in C. Clearly,  $|M^1(h_x)| > |M(h_x)|$  so that  $h_x$  prefers  $M^1$  to M by Definition 1.4.1(i). For each hospital  $h_y \neq h_x$  in C, it is clear that  $|M^1(h_y)| = |M(h_y)|$ . Furthermore,  $h_y$  must either prefer  $M^1$  to M or is in different between the matching's according to Definition 1.4.1(ii), or else it

cannot be the case that  $M^1 < M$ . However, C is then an augmenting coalition with respect to M.

**Case (iii):** C is an alternating cycle. Clearly, each resident  $r_i$  in C is assigned in both M and M<sup>1</sup>. Since M<sup>1</sup> M, it must be the case that each  $r_i$  prefers  $M^1(r_i)$  to  $M(r_i)$ .

For each hospital  $h_j$  in C, it is clear that  $|M^1(h_j)| = |M(h_j)|$ . Furthermore,  $h_j$  must either prefer M<sup>1</sup> to M or is in different between the matchings according to Definition 1.4.1(ii), or else it cannot be the case that M<sup>1</sup><M. However, C is then a cyclic coalition with respect to M, a contradiction.

Henceforth we will establish the Pareto optimality of a given matching M in an instance I of HR by showing that M is improving coalition-free. We now show that Lemma 3.3.1 leads to an O(m) algorithm for testing a given matching in an HR instance for Pareto optimality. Let M be a matching in an HR instance I and let G be the underlying graph of I. We first perform the following transformation to the preference lists of agents. That is, for every resident  $r_i \in R$ , we remove the hospital  $h_i$  from the preference list of  $r_i$ , and remove  $r_i$  from the preference list of  $h_i$  if  $h_i$  is a hospital that lies after  $M(r_i)$  in  $r_i$ 's preference list, i.e.  $rank_{ri}$  (h<sub>i</sub>) >  $rank_{ri}$  (M(r<sub>i</sub>)). For each hospital h<sub>k</sub>, let r<sub>p</sub> be the worst resident assigned to it in M. We then remove each resident  $r_q$  from the preference list of  $h_k$ and remove  $h_k$  from the preference list of  $r_q$  whenever  $rank_{hj}$  ( $r_q$ ) >  $rank_{hj}$  ( $r_p$ ). The effect of these truncations is that each agent is assigned in M to a "worst choice" partner. Let us call the instance with truncated preference lists  $I^1$ , and its underlying graph G<sup>1</sup>, i.e. G<sup>1</sup> contain only those edges representing the truncated preference lists of the agents. It is straightforward to see that it takes O(m) time to construct G<sup>1</sup>.

By Lemma 3.3.1, M is Pareto optimal if it admits no augmenting coalition

or cyclic coalition. We can check for the former structure by testing for an augmenting path in  $G^1$ . We use a similar form of restricted breadth-first search as described in Section 1.2, in

Resident Pref	Hospital Pref	Hospital Capacity
$r_1: h_2 h_1$	h <sub>1</sub> : r <sub>1</sub>	$h_1: 1$
r <sub>2</sub> : h <sub>2</sub>	$h_2: r_1 r_2$	h <sub>2</sub> :1

Figure 4.1: An instance I<sub>1</sub> of HR

which only edges not in M are followed from vertices in R and edges of M are followed from vertices in H. If an augmenting path P is found in  $G^1$ , then by augmenting along P, we obtain a new matching  $M^1$  such that each resident  $r_i$ prefers  $M^1(r_i)$  to  $M(r_i)$ , and each hospital  $h_j$  either prefers  $M^1$  to M or is indifferent between the two matchings as a result of the preference list truncations. Hence, M is not Pareto optimal if  $G^1$  admits an augmenting path. It is straightforward that this takes O(m) time.

To test for a cyclic coalition, we form a directed graph  $G_C$  with respect to M by letting  $G_C$  consist of one vertex for each assigned resident in I<sup>1</sup>. We then construct an edge from a resident  $r_i$  to another resident  $r_j$  in  $G_C$  if  $r_i$  prefers  $M(r_j)$  to  $M(r_i)$  in I<sup>1</sup>. By a similar argument to the above, it follows that M is cyclic-coalition-free if and only if  $G_C$  is acyclic.

Note that even though M is a matching of an HR instance, all vertices in  $G_C$  have the only unitary capacity (being resident vertices). It follows that a depth-first search suffices to detect any cycles in O(m) time so that these observations lead us to the following lemma.

Lemma 4.3.2. Let M be a matching in a given instance of HR. Then we may

check whether M is Pareto optimal in O(m) time.

Figure 3.1 shows us an HR instance I<sub>1</sub>. Note that the unique stable matching  $M = \{(r_1, h_2)\}$  has cardinality 1 here, but the maximum Pareto optimal matching  $M^1 = \{(r_1, h_1), (r_2, h_2)\}$  has cardinality 2. Hence, Pareto optimality is a criterion that can give rise to larger matchings than stability. Furthermore, it is straightforward to see that by creating p copies of J, we may construct an HR instance J<sup>p</sup> with 4p agents which admits a stable matching  $M_p = \{(r_{2i+1}, h_{2i+2}) : 0 \le i \le p - 1\}$  of size p and a Pareto optimal matching  $M_p^{-1} = (\{r_{2i+1}, h_{2i+2}) : 0 \le i \le p - 1\}$  of size 2p. It follows that we can have an infinite family of HR instances for which the cardinality of a stable matching is half the size of a maximum Pareto optimal matching.

Hence, given any HR instance I, we are also interested in considering the problem of finding a maximum Pareto optimal matching in I as an alternative optimality criterion. The next section presents an efficient algorithm for solving this.

# 4.4 Maximum Pareto optimal Matching's in HR

In this section, we describe a two-phase algorithm for finding a maximum Pareto optimal matching in an instance I of HR by satisfying the necessary and sufficient conditions of Lemma 3.3.1. Phase 1 of the algorithm uses Gabow's algorithm [15] to compute a maximum matching M in the underlying graph G. This phase guarantees that M admits no augmenting coalition and takes O(Cm) time. Phase 2 transforms M into a matching that admits no cyclic coalition in the following way.

Let us construct a graph  $G^1$  from G by repeating the truncation of preference lists as described in the aforementioned methods for testing a given matching in I for Pareto optimality. Hence, all agents are assigned to a worstchoice partner in  $G^0$ . It thus follows as a result that the residents always improve and the hospitals either improve or remain indifferent even if we satisfy any cyclic coalition with respect to M by considering preference lists on only one side. This allows us to obtain a cyclic-coalition-free matching in  $G^1$  from M by considering the problem from only the point of view of the residents' truncated preference lists, which effectively transforms the problem in I to an instance of the analogous problem for the Capacitated House Allocation problem.

Hence, this allows the Phase 3 algorithm, described in Chapter 2 for finding a maximum Pareto optimal matching given a CHA instance, to be reused from the residents' point of view in order to obtain a matching  $M^1$  from M that is cyclic-coalition-free in G. We note that  $M^1$  must be cyclic-coalition-free by the correctness proof presented in Section 2.4.2.3. Furthermore, the correctness proof also shows that  $M^1$  remains a maximum matching since each resident who was assigned at the end of Phase 1 remains assigned after the execution of the Phase 3 algorithm. With the use of suitable data structures such as those described in Section 2.4.3, the Phase 3 algorithm is guaranteed to run in O(m) time. Hence, it takes O(m) time for Phase 2 of our algorithm to find a maximum Pareto optimal matching given a HR instance. It follows that Phase 1 dominates the overall complexity of our two-phase algorithm for HR, giving us the following result.

**Theorem 3.4.1.** A maximum Pareto optimal matching in an instance I of HR can be found in O( $\sqrt{C}$ m) time.

# **Chapter-V**

# MAXIMUM PARETO OPTIMAL MATCHING'S IN SMI

#### Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Since SMI is a special case of HR, it follows that Lemma 5.1 also gives a necessary and sufficient condition for a matching in an SMI instance J to be Pareto optimal.

We first show how to test a given matching M in J for Pareto optimality. Let  $G^1$  be the underlying graph of J and let m be the number of edges in  $G^1$ . We form a subgraph  $G_M$  of  $G^1$  by letting  $G_M$  contain only those edges in  $M \cup$ bp(M) where bp(M) is the set of blocking pairs with respect to M. It is clear that  $G_M$  can be constructed in O(m) time by considering the edges of G. By Lemma 5.1, M is Pareto optimal in J if and only if M admits no augmenting path or alternating cycle in  $G_M$ . Clearly, we can test for an augmenting path in  $G_M$  in O(m) time using restricted breadth-first search as described in Section 1.2. In order to test for an alternating cycle, we remove any unmatched vertices and their incident edges from  $G_M$ . Any cycle in  $G_M$  that remains is an even-length alternating cycle, so that a depth-first search success to detect this. Hence, we can check if a matching in J is Pareto optimal in O(m) time.

We next show how the algorithm for finding a maximum Pareto optimal matching in HR can be easily modified for the analogous problem in SMI. First of all, we use the Hopcroft-Karp algorithm for finding a maximum matching M in Phase 1 instead of Gabow's algorithm. Then, this step takes O( nm) time where n

is the total number of men and women. Using a similar form of preference list truncation to that for HR as described above, we can then transform the problem in J to the House Allocation problem (HA).

Hence, this allows Phase 3 of the implementation of the Top Trading Cycles Method in [3] to be reused from the men's point of view in order to obtain a matching  $M^1$  from M that is cyclic-coalition-free in G. We note that  $M^0$  must be cyclic-coalition-free by the correctness of the Top Trading Cycles Method [57]. Furthermore  $M^1$  remains a maximum matching since the algorithm in [3] ensures that each man and woman who is assigned at the end of Phase 1 of our algorithm is also assigned when it completes execution. With the use of suitable data structures such as doubly linked lists or arrays, we can ensure that the initialization and subsequent deletion of entries from the preference lists takes O(m) time. Since the nested loops in Phase 3 of the implementation in [3] are guaranteed to run in O(m) time, Phase 2 of our algorithm also runs in O(m) time. This gives us the following result.

**Theorem 5.1.** A maximum Pareto optimal matching in an instance J of SMI can be found in O( $\sqrt{n}$ m) time.

It follows that it is generally faster to find a maximum Pareto optimal matching in an SMI instance using the techniques described in this section as opposed to the algorithm for HR described in Section 3.4.

We make some conclusions here on the results of this dissertation in general, in addition to those specific to each of the optimality criteria studied as above.

- All the problems that were studied in this thesis turned out to be solvable in polynomial time. The existence of a polynomial time algorithm is often inherently associated with establishing some kind of underlying structure for the problem concerned. For instance, the solution to finding a maximum Pareto optimal matching requires the identifying and then satisfying of certain types of coalitions in the underlying graph of the problem instance. To find a maximum popular matching, the identification of the f- and spartners of each participating agent in the problem instance, allows us to restrict our attention to only a subgraph of the underlying graph to generate an efficient solution to the problem. In addition, the fastest algorithm for finding a rank-maximal matching is reliant on utilizing the Edmonds-Gallai Decomposition to label vertices and then identify only those edges that can belong to any solution.
- 2. The optimality criteria studied in this thesis can be considered to be superficially similar in some respects, but a simple change to the problem definition often requires a significant change to the algorithm.
- 3. The problem of finding a matching of maximum cardinality in the underlying graph of the problem instance often seems to influence the time complexity of the resulting algorithm for its solution. This is the case even if the underlying problem seems on the surface not to be associated with maximum matching's, e.g. popular matching's.

# References

- Abdulkadiroglu and T. Sonmez. House allocation with existing tenants. Journal of Economic Theory, 88:233–260, 1999.
- Abdulkadiroglu and T. Sonmez. Random serial dictatorship and the core from random endowments in house allocation problems. Econometrica, 66(3)(3):689–701, 1998.
- 3. Hylland and R. Zeckhauser. The efficient allocation of individuals to positions. Journal of Political Economy, 87:293–314, 1979.
- Itai, M. Rodeh, and S.L. Tanimoto. Some matching problems for bipartite graphs. Journal of the Association for Computer Machinery, 25(4):517– 525, 1978.
- 5. Subramanian. A new approach to stable matching problems. SIAM Journal of Computing, 23(4):671–700, 1994.
- A.E. Roth and A. Postlewaite. Weak versus strong domination in a market with indivisible goods. Journal of Mathematical Economics, 4:131–137, 1977.
- A.E. Roth and M.A.O. Sotomayor. Two-Sided Matching A Study in Game-Theoretic Modeling and Analysis. Cambridge University Press, 1990.
- A.E. Roth, T. Sonmez, and M. Utku Unver. A kidney exchange clearinghouse in New England. The American Economic Review, 95(2):376–380, 2005.
- A.E. Roth, T. Sonmez, and M. Utku Unver. efficient kidney exchange: Coincidence of wants in a structured market. Unpublished manuscript, May 2005.
- 10. A.E. Roth, T. Sonmez, and M. Utku Unver. Kidney exchange. Quarterly Journal
- 11. A.E. Roth, T. Sonmez, and M. Utku Unver. Pairwise kidney exchange.

Journal of Economic Theory, 125(2):151–188, 2005.

- A.E. Roth. On the allocation of residents to rural hospitals: a general property of two-sided matching markets. Econometrica, 54:425–427, 1986.
- A.E. Roth. The evolution of the labor market for medical interns and residents: a case study in game theory. Journal of Political Economy, 92:991–1016, 1984.
- 14. Berge. Two theorems in graph theory. Proceedings of the National Academy of Sciences of the USA, 43:842–844, 1957.
- 15. Ng and D.S. Hirshberg. Three-dimensional stable matching problems. SIAM Journal on Discrete Mathematics, 4:245–252, 1991.
- 16. C.H. Papadimitriou and K. Steiglitz. Combinatorial Optimization -Algorithms and Complexity. Prentice Hall, 1982.
- C-P. Teo and J. Sethuraman. The geometry of fractional stable matching's and its applications. Mathematics of Operational Research, 23(4):874–891, 1998.
- 18. C-P. Teo, J. Sethuraman, and W-P. Tan. Gale-Shapley Stable Marriage Problem Revisited: Strategic Issues and Applications. Management Science, 47(9):1252–1267, 2001.
- 19. Gale and L.S. Shapley. College admissions and the stability of marriage. American Mathematical Monthly, 69:9–15, 1962.
- 20. Gale and M. Sotomayor. Some remarks on the stable marriage problem. Discrete Applied Mathematics, 11:223–232, 1985.
- 21. D. Gale. Optimal assignments in an ordered setan application of matroid theory. Journal of Combinatorial Theory, 4:176–180, 1968.
- 22. D. Gusfield and R.W. Irving. The Stable Marriage Problem Structure and Algorithms. The MIT Press, 1989.
- 23. D. McVitie and L.B. Wilson. The stable marriage problem. Communications of the ACM, 14:486–490, 1971.

- 24. D.E. Knuth. Mariages Stables. Les Presses de LUniversit de Montreal, 1976.
- 25. D.F. Manlove, R.W. Irving, K. Iwama, S. Miyazaki, and Y. Morita. Hard variants of stable marriage. Theoretical Computer Science, 276(1-2):261– 279, 2002.
- 26. D.F. Manlove. Personal communication, 2007.
- 27. D.J. Abraham and D.F. Manlove. Pareto optimality in the roommates problem. Technical Report TR-2004-182, University of Glasgow, Department of Computing Science, 2004.
- 28. D.J. Abraham and T. Kavitha. Dynamic matching markets and voting paths. In Proceedings of SWAT 2006: the 10th Scandinavian Workshop on Algorithm Theory, volume 4059 of Lecture Notes in Computer Science, pages 65–76. Springer-Verlag, 2006.
- 29. D.J. Abraham, A. Levavi, D.F. Manlove, and G. O'Malley. The stable roommates problem with globally-ranked pairs. In Proceedings of WINE 2007: the 3rd Interna-tional Workshop on Internet and Network Economics, volume 4858 of Lecture Notes in Computer Science, pages 431–444. Springer-Verlag, 2007.
- 30. D.J. Abraham, K. Cechlarova, D.F. Manlove, and K. Mehlhorn. Pareto optimality in house allocation problems. In Proceedings of ISAAC 2004: the 15th Annual International Symposium on Algorithms and Computation, volume 3341 of Lecture Notes in Computer Science, pages 3–15. Springer-Verlag, 2004.
- 31. D.J. Abraham, R.W. Irving, T. Kavitha, and K. Mehlhorn. Popular matchings. SIAM Journal on Computing, 37:1030–1045, 2007.
- 32. Diamantoudi, E. Miyagawa, and L. Xue. Random paths to stability in the room-mate problem. Games and Economic Behavior, 48:18–28, 2004.
- 33. E. Ronn. NP-complete stable matching problems. Journal of Algorithms, 11:285–304, 1990.

- 34. Brassard and P. Bratley. Fundamentals of Algorithmics. Prentice-Hall, 1996.
- 35. H.N. Gabow. An effcient reduction technique for degree-constrained subgraph and bidirected network flow problems. In Proceedings of STOC'83: the 15th Annual ACM Symposium on Theory of Computing, pages 448–456. ACM, 1983.
- 36. J. Mestre. Weighted popular matchings. In Proceedings of ICALP 2006: the 33rd International Colloquium on Automata, Languages and Programming, volume 4051 of Lecture Notes in Computer Science, pages 715–726. Springer-Verlag, 2006.
- 37. J.E. Hopcroft and R.M. Karp. An n<sup>5/2</sup> algorithm for maximum matchings in bipartite graphs. SIAM Journal on Computing, 2:225–231, 1973.
- 38. J.J.M. Tan. A necessary and sufficient condition for the existence of a complete stable matching. Journal of Algorithms, 12:154–178, 1991.
- 39. K. Mehlhorn and D. Michail. Network problems with non-polynomial weights and applications. Unpublished manuscript, 2006.
- 40. K.S. Chung. On the existence of stable roommate matchings. Games and Economic Behavior, 33:206–230, 2000.
- 41. L. Lovasz and M.D. Plummer. Matching Theory. Number 29 in Annals of Discrete Mathematics. North-Holland, 1986.
- 42. L. Shapley and H. Scarf. On cores and indivisibility. Journal of Mathematical Eco-
- 43. M. de Klerk, K. Keizer, W. Weimar, A.J. Goody, A. Spital, and F.L. Delmonico. Donor exchange for renal transplantation. The New England Journal of Medicine, 351:935–937, 2004.
- 44. M. Fredman and R.E. Tarjan. Fibonacci heaps and improved network optimization algorithms. Journal of the ACM, 34:596–615, 1987.
- 45. M. Lucan, P. Rotariu, D. Neculoiu, and G. Iacob. Kidney exchange program: a viable alternative in countries with low rate of cadaver

harvesting. Transplantation Proceedings, 35:933–934, 2003.

- 46. M. Mahdian. Random popular matchings. In Proceedings of EC '06: the 7th ACM Conference on Electronic Commerce, pages 238–242. ACM, 2006.
- 47. M.R. Garey and D.S. Johnson. Computers and Intractability A Guide to the Theory of NP-Completeness. W.H. Freeman and Company, 1979.
- 48. of Economics, 119(2):457–488, May 2004.
- 49. omics, 1:23–37, 1974.
- 50. P. Gardenfors. Assignment problem based on ordinal preferences. Management Sci-ence, 20:331–340, 1973.
- P. Gardenfors. Match making: assignments based on bilateral preferences. Behavioural Science, 20:166–173, 1975.
- 52. R.M. McCutchen. The least-unpopularity-factor and least-unpopularitymargin crite-ria for matching problems with one-sided preferences. In Proceedings of LATIN 2008: the 8th Latin-American Theoretical Informatics symposium, to appear, Lecture Notes in Computer Science. Springer, 2008.
- 53. R.W. Irving and D.F. Manlove. Some preliminary observations on popular matchings in the roommates problem. Unpublished manuscript, 2005.
- 54. R.W. Irving and D.F. Manlove. The stable roommates problem with ties. Journal of Algorithms, 43:85–105, 2002.
- 55. R.W. Irving, T. Kavitha, K. Melhorn, D. Michail, and K. Paluch. Rankmaximal matchings. ACM Transactions on Algorithms, 2(4):602–610, 2006.
- 56. R.W. Irving. An efficient algorithm for the stable roommates problem. Journal of Algorithms, 6:577–595, 1985.
- 57. R.W. Irving. Greedy and generous matchings via a variant of the Bellman-Ford Algorithm. Unpublished manuscript, 2007.
- 58. R.W. Irving. Matching medical students to pairs of hospitals: A new

variation on a well-known theme. Proceedings of ESA'98: the 6th Annual European Symposium on Algorithms., 6:381–392, 1998.

- 59. R.W. Irving. Stable marriage and indifference. Discrete Applied Mathematics, 48:261–272, 1994.
- 60. S. Micali and V.V. Vazirani. An O( |V |⋅|E|) algorithm for finding maximum matching in general graphs. In Proceedings of FOCS '80: the 21st Annual IEEE Symposium in Foundations of Computer Science, pages 17–27. IEEE Computer Society, 1980.
- 61. T. Feder. A new fixed point approach for stable networks and stable marriages. Journal of Computer and System Sciences, 45:233–284, 1992.
- 62. T. Kavitha and C. Shah. Efficient algorithms for weighted rank-maximal matchings and related problems. In Proceedings of ISAAC 2006: the Seventeenth International Symposium on Algorithms and Computation, volume 4288 of Lecture Notes in Computer Science, pages 153–162. Springer, 2006.
- 63. T. Uno. Algorithms for enumerating all perfect, maximum and maximal matchings in bipartite graphs. In Proceedings of ISAAC 1997: the 8th Annual International Symposium on Algorithms and Computation, volume 1350 of Lecture Notes in Computer Science, pages 92–101. Springer-Verlag, 1997.
- 64. W.R. Pulleyblank. Matchings and extensions. In R.L. Graham, M. Grotschel, and L. Lovasz, editors, Handbook of Combinatorics, volume 1, chapter 3, pages 179–232.North-Holland, 1995.

# SURVEY UPON ANNALS OF FUZZY GRAPH

EDITED BY DR.D.R.KIRUBAHARAN



# Survey upon annals of fuzzy graph

Copyright © 2021 by **Dr.D.R.Kirubaharan** 

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-466-6

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)
# **TABLE OF CONTENTS**

Book elements –	Page Numbers
<b>Chapter-I</b> <i>Bipartite Matching Problems, One-Sided Preference Lists</i> Dr.D.R.Kirubaharan	01-06
<b>Chapter-II</b> <i>Bipartite Matching Problems With Two-Sided Preference Lists</i> Dr.D.R.Kirubaharan	07-16
Chapter-III Pareto Optimal Matchings In CHA Dr.D.R.Kirubaharan	17-37
Chapter-IV <i>Popular Matchings In CHAT</i> Dr.D.R.Kirubaharan	38-45
Chapter-V Cloning Versus Our Direct Approach Dr.D.R.Kirubaharan	46-60

#### **CHAPTER-I**

#### **BIPARTITE MATCHING PROBLEMS, ONE-SIDED PREFERENCE LISTS**

#### Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### **House Allocation Problem**

Bipartite matching problems involving two sets of participants, namely a set of agents and a set of objects, are commonly referred to as House Allocation problems [1-3, 16, 19, 24].

These problems have been widely studied not only due to their theoretical interest, but also in view of their practical importance.

Widespread applications occur in real-life resource allocation problems such as campus housing allocation in US universities [1], hence the problem name; in assigning probationary teachers to their first posts in Scotland; and in Amazon's DVD rental service.

An instance I of the House Allocation problem (HA) comprises two disjoint sets A and H, where  $A = \{a_1, a_2, ..., a_{n1}\}$  is the set of agents and  $H = \{h_1, h_2, ..., h_{n2}\}$  is the set of houses. Each agent  $a_i \in A$  ranks in strict order a subset of those houses in H giving rise to his preference list. If  $a_i$  ranks a house  $h_j \in H$  in his preference list, we say that  $a_i$  finds  $h_j$  acceptable. An agent  $a_i$  prefers one house  $h_j$  to another house  $h_l$  if  $h_j$  precedes  $h_l$  in  $a_i$ 's preference list. We define the underlying graph of I to be the bipartite graph G = (A, H, E), where E is the set of edges in G representing the acceptable houses of the agents. We let  $n = n_1 + n_2$  and m = |E|.

Given an agent  $a_i \in A$  and an acceptable house  $h_j \in H$  for  $a_i$ , we define rank<sub>ai</sub>  $(h_j)$  to be the number of agents that  $a_i$  prefers to  $h_j$  plus 1. If rank<sub>ai</sub>  $(h_j) = k$ , we say that  $h_j$  is the kth choice of  $a_i$ . Let  $(a_i, h_j) \in E$  be any edge. Then, we define the rank of  $(a_i, h_j)$  to be  $r(a_i, h_j) = rank_{ai}$   $(h_j)$ . Let  $z \leq n_2$  be the maximum length of any agent's preference list in I. Clearly, z corresponds to the largest rank of a house taken over all agents' preference lists in I. We assume that no agent has an empty preference list and each house is acceptable to at least one agent, i.e.,  $m \geq max \{n_1, n_2\}$ .

An assignment M is a subset of  $A \times H$  such that  $(a_i, h_j) \in M$  only if  $a_i$  finds  $h_j$  acceptable. If  $(a_i, h_j) \in M$ , we say that  $a_i$  and  $h_j$  are assigned to each other, and we call  $a_i$  and  $h_j$  partners in M. A matching is an assignment M such that (i) each agent is assigned to at most one house in M, and (ii) each house  $h_j \in H$  is assigned to at most one agent in M. If a participant  $p \in A \cup H$  is assigned in M, we denote by M(p) the participant that p is assigned to in M. If p is not assigned in M, we say that p is unassigned. Given two matchings M and M<sup>1</sup> in G, we say that an agent  $a_i$  prefers M<sup>1</sup> to M if either (i)  $a_i$  is assigned in M<sup>1</sup> and unassigned in M, or (ii)  $a_i$  is assigned in both M<sup>1</sup> and M and prefers M<sup>1</sup>( $a_i$ ) to M( $a_i$ ). We use M to denote the set of all matchings in I. Several variants of HA may be formulated as follows.

# Variants of HA

First of all, we can have a straightforward extension of HA by allowing ties in the agents' preference lists. A tie between two houses  $h_j$  and  $h_l$  occurs in an agent  $a_i$ 's preference list when rank<sub>ai</sub> ( $h_j$ ) = rank<sub>ai</sub> ( $h_l$ ), and we say that the agent  $a_i$  is indifferent between  $h_j$  and  $h_l$ . The problem, then becomes known as the House Allocation

problem with Ties, or HAT for short.

We can have a variant of HA in which each agent a has an assigned positive weight w(a) that indicates his priority (which may be based on such objective criteria relevant to the matching application). This is known as the Weighted House Allocation problem, denoted by WHA, or WHAT if ties are present. If the houses are allowed to have non-unitary capacity, we then have a generalization of HA that is known as the Capacitated House Allocation problem, denoted by CHA, or CHAT if ties are present.

A third possible variant of HA combines WHA and CHA by letting the agents have a positive weight to indicate their priority, and allowing the houses to have nonunitary capacity. This is known as the Weighted Capacitated House Allocation problem, denoted by WCHA or WCHAT if ties are present.

We remark that all the notations and terminology that were defined for HA in Section 1.3.1 carry over directly to each of its variants with the exception of some terms that we will require to define separately. We henceforth assume these definitions in any variant of HA in the rest of this thesis and explicitly define relevant concepts only where we need to adapt them to the context of the variant.

We require to redefine a matching since each house  $h_j$  may now have a nonunitary capacity  $c_j \ge 1$ , and a many-one mapping of the agents and houses is sought in these contexts instead. Here, we define a matching to be an assignment M such that (i) each agent is assigned to at most one house in M, and (ii) each house  $h_j \in H$  is assigned to at most  $c_j$  agents in M.

Consequently,  $M(h_j)$  refers to the set of agents assigned to  $h_j$  in M (which could be empty) in these contexts. If  $|M(h_j)| < c_j$ , we say that  $h_j$  is under subscribed in

M; otherwise h<sub>j</sub> is full in M. We also let  $C = \sum_{j=1}^{n_2} c_j$  denote the sum of the capacities of the houses.

## **Optimality criteria for bipartite matching problems**

For bipartite matching problems with one-sided preferences, various criteria as to what constitutes an "optimal" matching have been considered. In this section, we give a review of the optimality criteria that are considered in this thesis.

#### Pareto optimal matchings

One solution concept that has received much attention, particularly from the Economics community is Pareto optimality [1–3, 7, 51, 56, 57], because it is regarded by Economists as a fundamental property to be satisfied in the context of matching problems. Let I be an instance of HA or any or its variants. Then, we may define a relation on M based on the preference of agents over matchings in I (as defined above): that is, given any two matchings M and M<sup>1</sup>, M<sup>1</sup>< M if and only if no agent prefers M to M<sup>1</sup>, and some agent prefers M<sup>1</sup> to M. A matching M is defined to be Pareto optimal if and only if there is no other matching M<sup>1</sup> such that M<sup>1</sup>< M.

Various algorithms exist for finding a Pareto optimal matching in any given instance of HA, the most straightforward being a greedy algorithm known as the serial dictatorship mechanism [1, 56] which considers each agent a in turn, and gives a his most preferred vacant house (if such a house exists).

However, such an algorithm may fail to find a Pareto optimal matching of maximum cardinality (henceforth a maximum Pareto optimal matching), which is undesirable in applications that seek to assign as many agents as possible.

Abraham et al. [3] gives the fastest algorithm, which takes  $O(\sqrt{nm})$  time, for

finding a maximum Pareto optimal matching given an HA instance. In this thesis, we extend their results to the capacitated bipartite graph case in Chapter 2 by constructing an  $O((\sqrt{cm}))$  time algorithm for finding a maximum Pareto optimal matching given any instance of CHA. Since the definition of a Pareto optimal matching in WCHA is identical to that in CHA, this algorithm can also be used for the analogous problem in the weighted capacitated bipartite graph case.

## **Popular Matchings**

Another important solution concept is that of a popular matching. Let I be an instance of CHAT. Also, let M and M<sup>1</sup> be two arbitrary matchings in I and let P (M, M<sup>1</sup>) denote the set of agents who prefer M to M<sup>1</sup>. We say that M is more popular than M<sup>1</sup> if  $|P(M, M^1)| > |P(M^1, M)|$ , i.e. the number of agents who prefer M to M<sup>1</sup> is greater than the number of agents who prefer M<sup>1</sup> to M. A matching M in I is popular if there is no other matching M<sup>1</sup> in I that is more popular than M.

We remark that the definition of a popular matching can be extended to WCHAT in the following way. First of all, given any two matchings M and M<sup>1</sup> in a weighted setting, we define the satisfaction of M with respect to M1to be sat(M,M') = $\sum_{a \in P(M,M')} w(a) - \sum_{a \in P(M',M^1)} w(a)$  We then say that M is more popular than M<sub>0</sub> if sat(M,M<sup>1</sup>) > 0. A matching M is defined to be popular if there is no other matching in the problem instance that is more popular than M.

Gardenfors [20] first introduced the notion of a popular matching (referring to this concept as a majority assignment) in the context of voting theory. We remark that more popular than concept can be traced back even further to the Condorcet voting protocol. Popular matchings were then considered by Abraham et al. [4] in the context of HA. They showed that popular matchings need not exist, given an instance of HA, and also noted that popular matchings can have different cardinalities. The same authors described an O(n + m) algorithm for finding a maximum cardinality popular matching (henceforth a maximum popular matching) if one exists, given an instance of HA. They also described an O( $\sqrt{nm}$ ) counterpart for HAT.

Mahdian [37] showed that a popular matching exists with high probability given an instance of HAT when (i) preference lists are random, and (ii) the number of houses is a small multiplicative factor larger than the number of agents. To cope with the possible non-existence of a popular matching, McCutchen [40] defined two notions of a matching that are, in some sense, "as popular as possible", namely a least-unpopularity-factor matching and a least-unpopularity-margin matching. McCutchen proved that computing either type of matching is NP-hard. Abraham and Kavitha [5] considered voting paths in relation to popular matchings in a dynamic matching market in which agents and houses can enter and leave the market. Mestre [43] then described an O(n + m) algorithm for finding a maximum popular matching if one exists, given an instance of WHA. He also described an O(min( $k\sqrt{n}$ , n)m) counterpart for WHAT, where k is the maximum priority of any agent.

## **Profile-based optimal matchings**

Finally, let I be an instance of HAT or any of its variants. Recall that z is the largest rank of a house taken over all agents' preference lists in I. Define the profile  $\rho(M)$  of a matching M in I to be the z-tuple  $(x_1, x_2, ..., x_z)$  where for each i  $(1 \le i \le z)$ ,  $x_i$  is the number of agents who are assigned in M with one of their i<sup>th</sup> choice houses. Then, it is possible to define at least three versions of what it means for a matching to be optimal based on its profile.

Informally, a greedy maximum matching is a matching that has lexico graphically maximum profile taken over all maximum matchings.

On the other hand, a rank-maximal matching is a matching that has lexico

graphically maximum profile taken over all matchings. Finally, a generous maximum matching is a matching whose reverse profile is lexicographically minimum taken over all maximum matchings. We remark that each of a rank-maximal, a greedy maximum and a generous maximum matching must be Pareto optimal; however, they are not necessarily popular.

The fastest combinatorial approach for finding a rank-maximal matching given an HAT instance is described by Irving et al. [29], and this takes  $O(\min(z^*\sqrt{n}, n + z^*)m)$  time where  $z^*$  is the maximal rank of an edge in an optimal solution. Kavitha and Shah [33] studied rank-maximal matchings in WHAT and described an  $O(\min(z^*\sqrt{n}, n+z^*)m)$  time algorithm for solving the problem. In an unpublished manuscript [28], Irving describes an approach based on the Bellman-Ford algorithm to find greedy maximum and generous maximum matchings in HAT.

# CHAPTER-II

# BIPARTITE MATCHING PROBLEMS WITH TWO-SIDED PREFERENCE LISTS

## Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

# **One-one mapping: the classical Stable Marriage Problem**

The classical Stable Marriage problem (SM) is a widely studied example of a combinatorial problem in the category indicated by this subsection. An instance I of SM involves two disjoint sets U and W where  $U = \{u_1, u_2, ..., u_n\}$  is the set of men, and  $W = \{w_1, w_2, ..., w_n\}$  is the set of women. Each person  $p \in U \cup W$  ranks all members of the opposite sex in strict order of preference giving rise to his/her preference list. We say that person p prefers q to r if q precedes r on p's preference list.

An assignment M is a subset of  $U \times W$  such that  $(u_i, w_j) \in M$  only if  $u_i$  and  $w_j$  find each other acceptable. If  $(u_i, w_j) \in M$ , we say that  $u_i$  and  $w_j$  are assigned to each other.

A matching in I is an assignment M such that (i) each man is assigned to at most one woman in M, and (ii) each woman is assigned to at most one man in M.

If  $(u_i, w_j) \in M$ ,  $u_i$  and  $w_j$  are called partners in M. A blocking pair for M is a

(man,woman) pair  $(u_i, w_j)$  such that  $u_i$  prefers  $w_j$  to  $M(u_i)$  and  $w_j$  prefers  $u_i$  to  $M(w_j)$ , where M(q) denotes q's partner in M for any person q in I. A matching that admits no blocking pair is said to be stable.

Stable matching problems were first studied by Gale and Shapley [17] in their seminal paper "College Admissions and the Stability of Marriage". There they gave an algorithm, now widely known as the Gale-Shapley (GS) algorithm, that always finds a stable matching for any instance of SM in  $O(n^2)$  time [34]. Very briefly, the algorithm involves a sequence of "proposals" from members of one sex to members of the opposite sex and it terminates when everyone becomes engaged. If the men were the proposers, then we obtain the man-oriented version of the GS algorithm, otherwise the algorithm is known as woman-oriented. The algorithm is inherently non-deterministic in that the order in which the proposals take place is of no consequence to the result [22].

Gale and Shapley [17] observed that the man-oriented version of the GS algorithm always gives the man-optimal stable matching, in which each man has the best partner that he can have in any stable matching.

The man-optimal stable matching is also woman-pessimal, for each woman has the worst partner that she can have in any stable matching [41].

If the woman-oriented version of the GS algorithm is used, then this gives analogous results: we obtain the woman-optimal stable matching which is man pessimal. Gusfield and Irving [22] gave an extended version of the man-oriented GS algorithm which simplifies the process by deleting from a woman w's preference list every man  $u^1$  who succeeds a man u from whom she has received a proposal. This is because no such pair ( $u^1$ , w) can be part of any stable matching. Several variants of the Stable Marriage problem exist and have been widely studied as follows.

## **Incomplete lists**

A natural variant of SM occurs when each person p in an SM instance I need not rank all members of the opposite sex. Then the preference list for each person p contains a subset of members of the opposite sex such that person p finds q acceptable if and only if q appears in p's preference list. We henceforth assume in all contexts where all the participants are agents, that if an agent a ranks another agent b in a's preference list, then b also ranks a in b's preference list. Furthermore, the numbers of men and women need not be equal. We say that these preference lists are incomplete and use SMI (Stable Marriage with Incomplete Lists) to denote this version of SM.

In this setting, a man  $u_i$  and a woman  $w_j$  are assigned to each other in a matching M only if  $u_i$  and  $w_j$  are acceptable to one another. Thus, matchings need not be complete, i.e. not all members of either sex need be assigned in a given matching in this setting. Here, a (man,woman) pair ( $u_i$ ,  $w_j$ ) constitutes a blocking pair for M whenever

- I. u<sub>i</sub> and w<sub>j</sub> find each other acceptable,
- II.  $u_i$  is either unassigned in M or prefers  $w_j$  to  $M(u_i)$ , and
- III.  $w_j$  is either unassigned in M or prefers  $u_i$  to  $M(w_j)$ .

A matching in an instance of SMI is stable if it admits no such blocking pair. Every SMI instance, admits a stable matching [17], and Gusfield and Irving [22] showed that the extended GS algorithm can be used to find a stable matching, given an SMI instance. Furthermore, for any matching M in an instance of SMI, some agents may be unassigned in M, but the same agents are unassigned in all stable matchings and as a consequence, all stable matchings in I have the same cardinality [18].

10

Ties

Another variant of SM occurs when the preference list of each person is allowed to contain ties. We say that a person p is indifferent between q and r if q and r appear in a tie in p's preference list, and use SMT (Stable Marriage with Ties) to denote this variant of SM. The introduction of ties in a person's preference list gives rise to three definitions of stability, namely weak stability, strong stability and superstability [26].

A matching M is defined to be weakly stable if there does not exist any blocking pair  $(u_i, w_j)$  such that  $u_i$  and  $w_j$  prefer each other to their partners in M. On the other hand, a matching M is strongly stable if there does not exist any blocking pair  $(u_i, w_j)$  such that either (i)  $u_i$  prefers  $w_j$  to  $M(u_i)$ , and  $w_j$  either prefers  $u_i$  to  $M(w_j)$ or is indifferent between them, or (ii)  $w_j$  prefers  $u_i$  to  $M(w_j)$ , and  $u_i$  either prefers  $w_j$  to  $M(u_i)$  or is indifferent between them.

We define a matching M to be super-stable if there does not exist any blocking pair ( $u_i$ ,  $w_j$ ) such that  $u_i$  either prefers  $w_j$  to M( $u_i$ ) or is indifferent between them, and  $w_i$  either prefers  $u_i$  to M( $w_i$ ) or is indifferent between them.

A weakly stable matching can always be found for an instance of SMT by simply breaking the ties arbitrarily and then applying the extended Gale-Shapley algorithm to the derived instance. This guarantees to produce a matching that is weakly stable in the original instance with ties [22]. Also, all weakly stable matchings have the same cardinality in this context. We remark that strongly stable matchings and super-stable matchings need not exist for a given instance of SMT; hence, we do not devote any more attention to the results concerning these versions of stability and refer the reader to [26] for more details.

## **Ties and Incomplete lists**

SMT and SMI can be combined to give the Stable Marriage problem with Ties and Incomplete lists, or SMTI in short. That is, a given preference list in SMTI can be incomplete and can contain ties. In addition, the definition of weak stability can be extended from SMT to SMTI in a natural way. A weakly stable matching may be found using the same algorithm described for the corresponding problem in SMT. Unlike the case in SMT, weakly stable matchings can have different cardinalities, and Manlove et al. [39] shows that the problem of finding a maximum cardinality weakly stable matching given an instance of SMTI is NP-hard, even if the ties are at the tails of the lists and on one side only, there is at most one tie per list, and each tie is of length two.

# **One-many mapping: the Hospitals-Residents Problem**

The Hospitals-Residents problem is a many-one extension of SM that was first considered by Gale and Shapley [17] and referred to in that paper as the College Admissions problem. This problem has since invariably been known as the Hospitals-Residents problem mainly because of its applications in the medical matching context, such as the SFAS and NRMP as mentioned in Section 1.1.

An instance I of the Hospitals-Residents problem (HR) comprises two disjoint sets R and H, where  $R = \{r_1, r_2, ..., r_{n1}\}$  is the set of residents and  $H = \{h_1, h_2, ..., h_{n2}\}$  is the set of hospitals. Each resident  $r_i \in R$  ranks a subset of the hospitals in H in strict order of preference giving rise to his preference list. Similarly, each hospital  $h_j \in$ H ranks a subset of the residents in R in strict order, giving rise to its preference list. If  $r_i$  and  $h_j$  rank each other in their preference lists, we say that they find each other acceptable, and  $r_i$  and  $h_j$  are each an acceptable partner for one another. We say that a resident  $r_i$  prefers one hospital  $h_j$  to another  $h_k$  if  $h_j$  precedes  $h_k$  in  $r_i$ 's preference list. Similarly, we define the preferences of hospitals over residents. Each hospital  $h_j \in H$ has a capacity  $c_j$  which indicates the maximum number of posts it may fill. We define the underlying graph of I to be the bipartite graph G = (R, H, E), where E is the set of edges in G representing the acceptable hospitals of the residents. Let  $C = {}^{Pn2} c^{j}$ denote the sum of the capacities of the hospitals. We also let  $n = n_1 + n_2$  and m = |E|.

An assignment M is a subset of  $R \times H$  such that  $(r_i, h_j) \in M$  only if  $r_i$  finds  $h_j$  acceptable and vice versa. If  $(r_i, h_j) \in M$ , we say that  $r_i$  and  $h_j$  are assigned to each other.

A matching in I is an assignment M such that (i) each resident is assigned to at most one hospital in M, and (ii) each hospital  $h_j \in H$  is assigned to at most  $c_j$  residents in M. If a resident  $r_i \in R$  is assigned in M, we denote by  $M(r_i)$  the hospital that  $r_i$  is assigned to in M. We define  $M(h_j)$  to be the set of residents assigned to  $h_j$  in M (thus  $M(h_j)$  could be empty). We say that a hospital  $h_j \in H$  is full in M if  $|M(h_j)| = c_j$ , and undersubscribed in M if  $|M(h_j)| < c_j$ .

A blocking pair for M is a (resident, hospital) pair (ri, hj) such that

- $\succ$  r<sub>i</sub> and h<sub>i</sub> find each other acceptable
- $\triangleright$  either r<sub>i</sub> is unassigned in M, or r<sub>i</sub> prefers h<sub>j</sub> to M(r<sub>i</sub>)
- either h<sub>j</sub> is undersubscribed in M, or h<sub>j</sub> prefers r<sub>i</sub> to its worst assigned resident in M(h<sub>j</sub>)

A matching that admits no blocking pair is said to be stable, and every instance of HR admits a stable matching [22]. Note that SMI is a special case of HR in which  $c_j = 1$  for all  $h_j \in H$ . Furthermore, we can extend the definition of a man-optimal and a woman-optimal stable matching in SMI to a resident-optimal and a hospital-optimal stable matching respectively in HR (see Section 1.6 of [22]). For any given instance I of HR, efficient algorithms exist to find such stable matchings of I [22]. An HR instance can have more than one stable matching. However, all stable matchings have the same cardinality, and the same residents are assigned in all stable matchings [18, 49]. Furthermore, any hospital that is under subscribed in one stable matching is assigned with exactly the same residents in all stable matching [50].

Collectively, these results are known as the Rural Hospitals Theorem because of their historical significance relating to the problems that rural hospitals face when recruiting interns in the NRMP [22].

Given two matchings M and M<sup>1</sup>, we say that a resident  $r_i$  prefers M<sup>1</sup> to M if either (i)  $r_i$  is assigned in M<sup>1</sup> and unassigned in M, or (ii)  $r_i$  is assigned in both M<sup>1</sup> and M and prefers M<sup>1</sup>( $r_i$ ) to M( $r_i$ ). Unlike the case for residents, it is less straightforward to define the preference of a hospital  $h_j$  over two matchings since  $h_j$  may have nonunitary capacity. Given that the primary goal of many practical matching applications is to maximize the number of agents assigned, as well as to optimize the satisfaction of the agents according to their preference lists, we give what may be viewed as a definition of a hospital  $h_j$  's preference over matchings in I as follows.

**Definition**. We say that the hospital h<sub>j</sub> prefers one matching M<sup>0</sup> to another M if

- 1.  $|M^1(h_j)| > |M(h_j)|$ , or
- 2.  $|M^1(h_j)| = |M(h_j)|$  and  $h_j$  prefers the worst resident assigned to it in  $M^1$  to the worst resident assigned to it in M.

Note that even though there are no ties in  $h_j$ 's preference list, Definition 1.4.1 allows a hospital  $h_j$  to be indifferent between two matchings M and M<sup>1</sup> if  $|M(h_j)| = |M^1(h_j)|$ , the worst resident assigned to  $h_j$  is the same in both M and M<sup>1</sup> but  $h_j$  has different sets of residents assigned to it in M and M<sup>1</sup>. If  $h_j$  does not prefer M<sup>1</sup> to M, and also does not prefer M to M<sup>1</sup>, we say that  $h_j$  is indifferent between M and M<sup>1</sup>.

As is the case in SMI, we can permit ties in the preference lists in this context, and use HRT (Hospital-Residents problem with Ties) to denote this variant of HR. The definition of weak stability carries over from SMTI to HRT in an analogous way to the extension of the definition of classical stability from SMI to HR. Since SMTI is a special case of HRT, it follows that the problem of finding a maximum cardinality weakly stable matching is also NP-hard in HRT.

We remark that each of the concepts of a Pareto optimal matching, a popular matching and a profile-based optimal matching, can be defined in SM, HR, and their respective variants in the same way as the respective concepts were defined in the context of HA and its variants in Section 1.3.2. Given that stable matchings sometimes do not satisfy the key requirement in many practical matching contexts, which is to maximize the number of agents assigned in any given matching (as we shall show), we thus also apply these optimality criteria to SM and HR and some of their variants, and obtain new results as follows.

In Chapter 3, we study the problem of finding a maximum Pareto optimal matching given an instance of HR, and describe an O( $\sqrt{C}$ m) time algorithm for its solution. We also show how this algorithm can be adapted to solve the analogous problem given an instance of SMI in O( $\sqrt{nm}$ ) time.

#### Non-bipartite matching problems

## **Stable Roommates Problem**

In an instance of the Stable Roommates (SR) problem, first introduced by Gale and Shapley [17], there is a set of n agents where n is even. Each agent ranks then -1others in strict order of preference. A matching M is a partition of the set of agents into disjoint pairs. A blocking pair for M is a pair of agents {x, y}  $\in$ /M such that x prefers y to M(x) and y prefers x to M(y) where M(q) denotes q's partner in M for any agent q. A matching is stable if it admits no blocking pair.

It is well-known that SM is just a special case of SR, since the set of stable matchings is unchanged if we reduce an SM instance I into an SR instance by appending to the very end of each agent's preference list all the other agents that are of the same sex in I [22]. Not all SR instances admit a stable matching [17], and Knuth [34] posed the question of whether the problem of determining the solvability of SR instances might be NP-complete. This question was answered by Irving [25], who gave an  $O(n^2)$  algorithm for finding a stable matching or reporting that no such matching exists. Alternative approaches for finding a stable matching if one exists, given an SR instance, have since been described [12,13,58–60].

As with SM, we may formulate an extension of SR where preference lists may include ties and be incomplete (SRTI). In such a setting, the definition of a weakly stable matching may be extended from the SMTI context in a natural way given an SRTI instance, and weakly stable matchings, if they exist, can have different cardinalities. The problem of finding a maximum cardinality weakly stable matching given an SRTI instance is NP-hard [30, 48]. We remark that, as in SM, HR and their respective variants, each of the concepts of a Pareto optimal matching, a popular matching and a profile-based optimal matching, can be defined similarly in SR and its variants as they were defined in Section 1.3.2. Pareto optimal matchings in SR was recently studied by Abraham and Manlove[7]. There, the authors gave an  $O(\sqrt{n\alpha(m,n)}m\log^{3/2}n)$  time algorithm for the problem of finding a maximum Pareto optimal matching in an SR instance, I where n is the number of agents, m is the total length of the preference lists in I and  $\alpha$  is the inverse Ackermann function. Chung [10] considered popular matchings in instances of SR and noted that a stable matching is popular; however, the same need not be true in the presence of ties. Abraham et al. [6] studied rank-maximal matchings in a special case of SR in which roommate pairs are ranked globally, and gave an  $O(\min(z^*\sqrt{n}, z^* + n)m)$  time algorithm for the solution to the problem. Little is known about the individual problems of finding a popular matching (if one exists) and finding profile-based optimal matchings, given the general case of SR.

#### **Chapter-III**

# PARETO OPTIMAL MATCHINGS IN CHA

**Dr.D.R.Kirubaharan** Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### Introduction

As mentioned in Section 1.3.2, Pareto optimality is a solution concept that has received much attention from the Economics community in the context of matching problems since it is regarded as a fundamental solution concept. Pareto optimality interests us from the point of view of this research because most of the associated algorithmic questions have not, on the other hand, been considered extensively in the literature.

In this chapter, we study the problem of finding a maximum Pareto optimal matching in the context of CHA, a general case of bipartite matching problems with one-sided preferences. The main results of this chapter, and their organization are as follows. We give some terminology and preliminary results on Pareto optimal matchings in CHA in Section 2.2. We then give a characterization of Pareto optimal matchings in CHA in Section 2.3, which we subsequently use in Section 2.4 to construct an  $O(\sqrt{Cm})$  time algorithm for finding a maximum Pareto optimal matching given an instance I of CHA where C is the total capacity of the houses and m is the total length of preference lists in I respectively. Note that we reuse most of the terminology and notation from HA as defined in Section 1.3.1, and we explicitly define relevant concepts only where we need to adapt them to CHA.

## Basic terminology and preliminary results

Let I be an instance of CHA, and let G = (A, H, E) be the underlying bipartite graph of I as defined in Section 1.3.1. Each house  $h_j \in H$  has a capacity  $c_j \ge 1$  which indicates the maximum number of agents that may be assigned to it. Recall from Section 1.3.1 that an assignment M is a subset of  $A \times H$  such that  $(a_i, h_j) \in M$  only if  $a_i$  finds  $h_j$  acceptable.

Furthermore, if  $(a_i, h_j) \in M$ , we say that  $a_i$  and  $h_j$  are assigned to each other, and we call  $a_i$  and  $h_j$  partners in M. A matching M in an instance I of CHA is an assignment such that (i) each agent is assigned to at most one house in M, and (ii) each house  $h_j \in H$  is assigned to at most  $c_j$  agents in M. If an agent  $a_i \in A$  is assigned in M, we denote by  $M(a_i)$  the house that  $a_i$  is assigned to in M. We define  $M(h_j)$  to be the set of agents assigned to  $h_j$  in M (thus  $M(h_j)$  could be empty). We say that a house  $h_j \in H$  is full in M if  $|M(h_j)| = c_j$ , and undersubscribed in M if  $|M(h_j)| < c_j$ . We assume that no agent has an empty preference list and each house is acceptable to at least one agent so that  $m \ge \max \{n^1, n^2\}$ . Let  $C = \sum_{j=1}^{n_2} c_j$  denote the sum of the capacities of the houses.

## **Characterisation of Pareto Optimal Matchings**

Let M be a matching in I. We say that M is maximal if there is no agent  $a_i \in A$ and house  $h_j \in H$  such that  $a_i$  is unassigned in M,  $h_j$  is under subscribed in M and  $a_i$ finds  $h_j$  acceptable. Also, M is trade-in-free if there is no (agent,house) pair ( $a_i$ ,  $h_j$ ) such that  $a_i$  is assigned in M,  $h_j$  is under subscribed in M and  $a_i$  prefers  $h_j$  to M( $a_i$ ).

A cyclic coalition with respect to M is a sequence of distinct assigned agents C = ha<sub>0</sub>, a<sub>1</sub>, ..., a<sub>r-1</sub>i, for some  $r \ge 2$ , such that a<sub>i</sub> prefers M(a<sub>i+1</sub>) to M(a<sub>i</sub>) for each i ( $0 \le i \le r - 1$ ). Henceforth, all subscripts are taken modulo r when reasoning about coalitions.

Given a cyclic coalition C, the matching

 $M^{1} = (M \setminus \{(a_{i}, M(a_{i})) : 0 \le i \le r - 1\}) \cup \{(a_{i}, M(a_{i+1})) : 0 \le i \le r - 1\}$ 

is defined to be the matching obtained from M by satisfying C. We say that M is cyclic-coalition-free if M admits no cyclic coalition. The following lemma gives a necessary and sufficient condition for a matching to be Pareto optimal.

**Lemma** :Let M be a matching in a given instance I of CHA. Then M is Pareto optimal if and only if M is maximal, trade-in-free and cyclic-coalition-free.

**Proof.** Let M be a Pareto optimal matching. Suppose, for a contradiction that M is not maximal. It follows that there exist an agent  $a_i$  and a house  $h_j$  such that  $a_i$  is unassigned in M,  $h_j$  is undersubscribed in M and  $a_i$  finds  $h_j$  acceptable. Let  $M^1 = M \cup \{(a_i, h_j)\}$ . Then,  $M^1 < M$ , a contradiction. Now, suppose for a contradiction that M is not trade-in-free.

It follows that there exist an agent  $a_i$  and a house  $h_j$  such that  $a_i$  is assigned in M,  $h_j$  is under subscribed in M, and  $a_i$  prefers  $h_j$  to M( $a_i$ ). Let  $M^1 = (M \setminus \{(a_i, M(a_i))\} \cup \{(a_i, h_j)\}$ . Then,  $M^1 < M$ , a contradiction. Finally, suppose that M admits some cyclic coalition C.Let  $M^1$  be the matching obtained by satisfying C. Clearly, then,  $M^1 < M$ , a contradiction.

Conversely, let M be a matching that is maximal, trade-in-free and cycliccoalition-free. Let us suppose for a contradiction that M is not Pareto optimal. Then there exists some matching M<sup>1</sup> such that M<sup>1</sup>< M. Let G be the underlying graph of I. We clone G to obtain a cloned graph C(G) as follows. We replace every house  $h_j \in H$ with the clones  $h_j^1$ ,  $h_j^2$ , . . . ,  $h_j^{c,j}$ . We then divide the capacity of each house among its clones by allowing each clone to have capacity 1. In addition, if ( $a_i$ ,  $h_j$ )  $\in$  G, then we add ( $a_i$ ,  $h_j^p$ ) to C(G) for all p ( $1 \le p \le c_j$ ). Let us then adapt the matching M in G to obtain its clone C(M) in C(G) as follows. If a house  $h_j$  in G is assigned to  $x_j$  agents  $a_1$ , ... $a_{xj}$  in M, then we add ( $a_p$ ,  $h_j^p$ ) to C(M) for  $1 \le p \le x_j$ , so that |C(M)| = |M|. We repeat a similar process for M<sup>1</sup> to obtain its clone C(M<sup>1</sup>) in C(G).

Let us consider  $X = C(M) \bigoplus C(M^1)$  and let C be a connected component of X. It follows that C is a path or cycle whose edges alternate between C(M) and  $C(M^1)$ . Now, C cannot be an even-length alternating path that has more agents than houses or an an odd-length alternating path whose end edges are in C(M), for otherwise we have an agent who is assigned in M but unassigned in M<sup>1</sup>, a contradiction since M<sup>1</sup><M. In addition, C cannot be an even-length alternating path that has more houses than agents or an odd-length alternating path whose end edges are in  $C(M^1)$  because there then exists an agent  $a_i$  in C who becomes assigned in M<sup>1</sup> to a house  $h_i$  which is under subscribed in M. Now, since there are no ties in preference lists, a<sub>i</sub> must prefer h<sub>i</sub> to  $M(a_i)$  for otherwise  $M^1 < M$ . However, M is then not trade-in-free, a contradiction. Hence, C must be a cycle. Here, each agent  $a_i$  in C is assigned in both M and M<sup>1</sup> and since  $M^1 < M$ , each  $a_i$  prefers  $M^1$  to M. However, C is then a cyclic coalition with respect to M, a contradiction. It follows that  $M^1 < M$  and M is Pareto optimal. Henceforth we will establish the Pareto optimality of a given matching M in an instance I of CHA by showing that M is maximal, trade-in-free and cyclic-coalitionfree. We now show that Lemma 2.3.1 leads to an O(m) algorithm for testing M for Pareto optimality. Let G be the underlying graph of I. Then, we can check if M is maximal and trade-in-free in O(m) time by a traversal of the edges in G. To check if M is cyclic-coalition-free, we construct the envy graph [3] of M as follows. We form a directed graph G<sub>M</sub> of M by letting G<sub>M</sub> consist of one vertex for each agent assigned in M.

# Algorithm 2 Algorithm Greedy-PaCHA

 $\overline{\mathbf{M}:=\boldsymbol{\emptyset};}$ 

- 1: for each agent a<sub>i</sub> in turn do
- 2: if there exists some undersubscribed house in a<sub>i</sub>'s preference list then
- 3: let h<sub>j</sub> be the most-preferred such house;

4:  $M := M \cup \{(a_i, h_j)\};$ 

We then construct an edge from an agent  $a_i$  to another agent  $a_j$  in  $G_M$  if  $a_i$ 

prefers  $M(a_j)$  to  $M(a_i)$ . It follows that M is cyclic-coalition-free if and only if  $G_M$  is acyclic. Note that even though M is a matching of a CHA instance, all vertices in  $G_M$  have the only unitary capacity (being agent vertices). It follows that a depth-first search suffices to detect any cycles in O(m) time so that these observations lead us to the following lemma.

**Lemma :** Let M be a matching in a given instance of CHA. Then we may check whether M is Pareto optimal in O(m) time.

**Proof:** Now, given an instance I of CHA, a greedy approach using the serial dictatorship mechanism of [1] gives us a straightforward algorithm, Algorithm Greedy-PaCHA as shown in Algorithm 2, for finding a Pareto optimal matching M in I. Here, we consider each agent  $a_i$  in turn and give  $a_i$  his most preferred house that is currently under subscribed in the matching built so far. The following lemma shows that the matching constructed by the algorithm must be Pareto optimal.

**Lemma :** Let M be the matching returned by an execution of Algorithm Greedy-PaCHA. Then, M is Pareto optimal.

**Proof.** For, suppose not. For each  $a_i \in A$ , let  $A_i$  denote the set of acceptable houses for  $a_i$ . Consider an agent  $a_i$  who is unassigned in M. It follows that  $A_i$  contains no undersubscribed house  $h_j$ , otherwise  $(a_i, h_j)$  would have been added to M, a contradiction. Hence, M is maximal. If M is not trade-in-free, then there exists an agent  $a_i$  who prefers some under subscribed house  $h_j$  to  $M(a_i)$ . This is a contradiction, since  $h_j$  must be full at the point when we assign  $a_i$  to  $M(a_i)$ . If M is not cyclic-coalition-free, let us then consider the coalition  $C = ha_0, a_1, ..., a_{r-1}i$  which exists with respect to M. It follows that there exists some agent  $a_i$  ( $0 \le i \le r - 1$ ) in C who was considered first by the algorithm. By definition of C,  $a_i$  prefers  $M(a_{i+1})$  to  $M(a_i)$ . Now,  $a_{i+1}$  must be considered by the algorithm after  $a_i$ . However, it follows that  $M(a_{i+1})$  must then have had at least one place

Agent	Pref list	House	Capacity
a <sub>1</sub> :	$h_1 h_2$	h <sub>1</sub> :	1
a <sub>2</sub> :	$h_1$	h <sub>2</sub> :	1

Figure 2.1: An instance I<sub>1</sub> of CHA

free when  $a_i$  was assigned to  $M(a_i)$ , a contradiction to the fact that the algorithm gives each agent his most preferred under subscribed house. Hence, M is Pareto optimal.

The main drawback of Algorithm Greedy-PaCHA is that a given CHA instance may admit Pareto optimal matchings of different cardinalities but Algorithm Greedy-PaCHA may fail to find a Pareto optimal matching of maximum cardinality. For example, Figure 2.1 shows a given CHA instance in which Algorithm Greedy-PaCHA returns a Pareto optimal matching  $M_1 = \{(a_1, h_1)\}$  of cardinality 1, given the agent ordering ha<sub>1</sub>, a<sub>2</sub>i, and constructs the maximum Pareto optimal matching  $M_2 = \{(a_1, h_2),$  $(a_2, h_1)\}$  of cardinality 2 given the agent ordering ha<sub>2</sub>, a<sub>1</sub>i. It follows that the order in which the agents are considered can have a consequence on the cardinality of the outcome. This is significant from a practical point of view, given that a prime objective in many matching applications is to assign as many agents as possible.

We remark that a straightforward way to find a maximum Pareto optimal matching given a CHA instance I is by constructing a maximum cardinality minimum weight matching as follows. For each edge  $(a_i, h_j)$  in the underlying graph G of I, we assign a weight wt $(a_i, h_j)$  to the edge by letting wt $(a_i, h_j) = \operatorname{rank}_{ai}(h_j)$  where  $\operatorname{rank}_{ai}(h_j)$  denotes the rank of  $h_j$  in  $a_i$ 's preference list. Call this weighted graph G<sup>1</sup>. We then construct a maximum cardinality minimum weight matching in G<sup>1</sup>. The following lemma shows that such a matching must be a maximum Pareto optimal matching in I.

**Lemma :** Let M be a maximum cardinality minimum weight matching in  $G^1$ . Then, M is a maximum Pareto optimal matching in I.

**Proof.** Suppose not. Since M is a maximum matching, it follows that M is maximal. Now, if M is not trade-in-free, then there exists a (agent,house) pair ( $a_i$ ,  $h_j$ ) such that  $a_i$  is assigned in M,  $h_j$  is undersubscribed in M and  $a_i$  prefers  $h_j$  to M( $a_i$ ). Consider the matching M<sup>1</sup> = (M\( $a_i$ , M( $a_i$ ))) U ( $a_i$ ,  $h_j$ ). It is clear that  $|M^1| = |M|$  and so M<sup>1</sup> is another maximum cardinality matching of G<sup>1</sup>. However, since  $a_i$  prefers  $h_j$  to M( $a_i$ ), the weight of M<sup>1</sup> must be smaller than the weight of M, a contradiction.

Hence, suppose that M admits some cyclic coalition  $C = ha_0$ ,  $a_1$ , ...,  $a_{r-1}i$ . Let  $M^1$  be the matching obtained by satisfying C. Then, it is clear that  $|M^1| = |M|$  again. Moreover, since each  $a_i$  prefers  $M(a_i)$  to  $M^1(a_i)$  for  $0 \le i \le r-1$ , the weight of  $M^1$  is again smaller than the weight of M, a contradiction.

Note that the above lemma also indicates that a maximum Pareto optimal matching in I has the same cardinality as a maximum matching in G and any maximum cardinality minimum weight matching of  $G^0$  gives us a maximum Pareto optimal matching in I.

A well known transformation in matching theory allows us to transform the problem of finding a maximum cardinality minimum weight matching into the Assignment problem. Recall from Section 1.2 that we can solve the Assignment problem in the capacitated bipartite graph in O(C min(m log n, n<sup>2</sup>)) time [15], so this allows us to find a maximum Pareto optimal matching in the same time complexity. However, since the problem of finding a maximum matching in the capacitated bipartite graph takes  $O(\sqrt{C}m)$  time, it is of interest to consider whether faster algorithms for finding a maximum Pareto optimal matching in CHA exist.

# **Maximum Pareto Optimal Matchings**

In this section, we describe a three-phase algorithm for finding a maximum Pareto optimal matching in CHA by satisfying the necessary and sufficient conditions in Lemma 2.3.1.

Let I be an instance of CHA and G be its underlying graph. The problem of finding a maximum matching in G can be viewed as an instance of maximum cardinality DCS [15] as described in Section 1.2 (the two problems are essentially the same, except that agents have no explicit preferences in the DCS case; the definition of a matching is unchanged).

Hence, Phase 1 of the algorithm uses Gabow's algorithm [15] to compute a maximum matching M in G. This phase guarantees that M is maximal and takes  $O(\sqrt{C}m)$  time. The next two phases ensure that M is trade-in-free and cyclic-coalition-free respectively as detailed below

# Phase 2 of the algorithm

In this phase, we transform M into a trade-in-free matching by conducting a repeated search for (agent,house) pairs  $(a_i, h_j)$  such that  $h_j$  is undersubscribed in M and  $a_i$  prefers  $h_j$  to  $M(a_i)$ . Whenever such a pair is found, the algorithm breaks the existing assignment

Algorithm 3 Phase 2 loop while S  $6 = \emptyset$  do 1:  $h_j := S.pop();$ 2:  $(a_i, r) := L_j$ .removeHead(); if  $r < curr_{ai}$  then 3: 4:  $h_{i}^{1} := M(a_{i});$  $M := (M \setminus (a_i, h_i^1)) \cup \{(a_i, h_i)\};$ 5: 6:  $curr_{ai} := r;$ 7: if  $|M(h_i)| < c_i$  and  $L_i \neq \emptyset$  then 8: S.push(h<sub>i</sub>);

- 9:  $h_j := h_j^1;$ 
  - 10: if  $L_j \neq \emptyset$  and  $h_j \in S$  then
  - 11: **S.** $push(h_j)$ ;

between  $a_i$  and  $M(a_i)$ , and promotes  $a_i$  to  $h_j$ . It follows that a space in  $M(a_i)$  becomes freed in the process, which may consequently be assigned to some assigned agent  $a_k$ who prefers  $M(a_i)$  to  $M(a_k)$ . Note that if  $h_j$  remains under subscribed after such a step, it may also be assigned to some assigned agent  $a_l$  who prefers  $h_j$  to  $M(a_l)$ . We show how to obtain a trade-in-free matching from M by using a slight modification of the Phase 2 loop of the algorithm described by Abraham et al. [3] to find a maximum Pareto optimal matching in HA.

For each house  $h_j$ , we maintain a linked list  $L_j$  of pairs ( $a_i$ , r) where  $a_i$  is an assigned agent who prefers to be assigned to  $h_j$  than  $M(a_i)$  at the start of Phase 2, and r is the rank of  $h_j$  in  $a_i$ 's preference list. Note that the pairs in  $L_j$  may subsequently contain an agent  $a_i$  who prefers  $M(a_i)$  to  $h_j$  if  $M(a_i)$  is no longer the house that  $a_i$  was assigned to at the start of Phase 2 as a result of promotions executed over the course of the algorithm. We will maintain a stack S of all under subscribed houses  $h_j$  where  $L_j$  is non-empty. Also, for each house  $h_j$ , we assume that we store a counter for  $|M(h_j)|$ . For each assigned agent  $a_i$ , let curr<sub>ai</sub> be a variable which stores the rank of  $M(a_i)$  in  $a_i$ 's preference list.

Let us now consider the pseudocode of the Phase 2 loop as shown in Algorithm 3. During each iteration of the main while loop, we pop an under subscribed house  $h_j$  from S and remove the first pair ( $a_i$ , r) from  $L_j$  (which must be non-empty). Now, if  $r < curr_{ai}$ , it follows that  $a_i$  prefers  $h_j$  to  $M(a_i)$  so we promote  $a_i$  from  $h_j^1 = M(a_i)$  to  $h_j$  and we update M and curr<sub>ai</sub> in the process. Now, if  $h_j$  remains under subscribed at the end of this step, then we push  $h_j$  back onto S if  $L_j$  is non-empty. We also push  $h_j^1$  onto S if  $L_h 0_j$  is non-empty and if  $h_j^1$  is not already in S. Otherwise, if  $r \ge curr_{ai}$ , we push  $h_j$  back onto S if  $L_j$  is non-empty.

Now, the algorithm must terminate, for each iteration of the main while loop removes a pair from a list  $L_j$  but no new pair is ever added to any list during a loop iteration. Hence, the algorithm terminates when S is empty. It must be the case that when this happens no assigned agent  $a_i$  prefers an under subscribed house to M( $a_i$ ), so that M is trade-in-free as a result. Moreover, since each agent assigned at the end of Phase 1 is also assigned at the end of Phase 2, M remains a maximum matching. Let us then consider the time complexity of Phase 2. We can initialize all variables used in the Phase 2 loop in O(m) time using a single traversal of the agents' preference lists. The number of iterations of the main while loop is bounded above by the total length of preference lists. It is straightforward to verify that each operation within the while loop takes constant time (with a suitable choice of data structures such as those described later in Section 2.4.3). Hence, the algorithm runs in O(m) time, giving us the following result.

**Lemma :** Given a maximum matching M in an instance of CHA, the Phase 2 loop ensures that M is trade-in-free in O(m) time.

## Phase 3 of the algorithm

In this phase, we transform M into a matching  $M^1$  that admits no cyclic coalition by using a modification of the linear-time extension [3] of Gale's Top Trading Cycles Method [57].

This phase consists of a preprocessing step which we will describe in detail, and then the main Phase 3 loop shown in Algorithm 5. Throughout Phase 3, we maintain a stack of agents P which will help us to identify cyclic coalitions. The matching  $M^0$  and the stack P are empty at the start of Phase 3. For each agent  $a_i$ , we maintain a pointer  $p(a_i)$  to the first house on  $a_i$ 's preference list, and subsequently  $p(a_i)$ traverses left to right over the course of execution of Phase 3. We will also maintain a queue of agents Q, each of whom is an agent  $a_i$  waiting to be assigned to  $p(a_i)$  in  $M^1$ . In addition, for each house  $h_j$ , we will use  $M_0(h_j)$  to store those agents who are assigned to  $h_j$  in M but who are unassigned in M<sup>1</sup> so far in the execution of Phase 3. Initially,  $M_0(h_j)$  will contain all those agents assigned to  $h_j$  in M. As we assign agents in  $M_0(h_j)$  to houses in M<sup>1</sup>, we will remove these agents from  $M_0(h_j)$ . Finally, we also maintain a linked list  $L_j$  for each house  $h_j$  containing agents such that if  $a_i$  is an agent in  $L_j$ , then  $a_i$  prefers  $h_j$  to  $M(a_i)$ .

Algorithm 4 Process (Q)

while  $Q \neq \emptyset$  do 1: AI: = Q. removes Head (); 2:  $h_i := p(a_i);$ 3:  $h_k := M(a_i); \{ // \text{ possibly } h_i = h_k \}$ 4:  $M^1 := M^1 \cup \{(a_i, h_i)\};$ 5: label a<sub>i</sub>; 6: if  $a_i \in P$  then 7: remove a<sub>i</sub> from P; 8:  $M_0(h_k) := M_0(h_k) \backslash \{a_i\};$ 9: if  $|M^1(h_i)| = c_i$  then for each unlabelled  $a^{1}_{i} \in L_{i}$  do 10: delete  $h_i$  from the preference list of  $a_i^1$ ; 11: 12: if  $p(a_i^1) = M(a_i^1)$  then  $Q.add(a^{1}_{i});$ 13:

# **Pre-processing step**

Let us now introduce the pre-processing step which helps to reduce the number of iterations of the Top Trading Cycles Method in the main Phase 3 loop. This step makes use of the observation (as in [3]) that no agent  $a_i$  assigned to his first choice house  $h_j$  in M can be involved in a cyclic coalition. At the outset of Phase 3, we check if  $p(a_i) = M(a_i)$  for each agent  $a_i$  and add every such  $a_i$  to Q.

If Q is non-empty, then we run the sub-routine Process(Q), shown in Algorithm 4, as the preprocessing step. Note that this usage of Process(Q) is prior to the main Phase 3 loop starting, but it will be used again in general during the main Phase 3 loop. This sub-routine considers each agent  $a_i$  in Q in turn, by removing  $a_i$ from Q and then adding the edge  $(a_i, h_j)$  to  $M^1$ . Every such  $a_i$  is then labelled to differentiate a<sub>i</sub> from those agents unassigned in M<sup>1</sup> so far in the execution of the algorithm (all agents are initially unlabelled at the outset of Phase 3). Now, P must be empty during pre-processing. However, this may not be true during a subsequent execution of Process(Q) by the main Phase 3 loop. Hence, Process(Q) checks if a<sub>i</sub> lies in P, and if so, removes a<sub>i</sub> from P so as to remove the agent from further consideration by the main Phase 3 loop, since  $a_i$  has just been assigned in M<sup>1</sup>. Let  $p(a_i) = h_i$ . Now, if  $|M^1(h_i)| = c_i$  after the assignment of  $a_i$  to  $h_i$ , then we remove  $h_i$ from the preference lists of the remaining agents since such a house that is full in  $M^1$ could not subsequently be involved in a cyclic coalition. We refer to those preference lists in which houses have been removed as reduced preference lists. We then apply the observation made at the start of this subsection recursively to the reduced preference lists of the remaining agents until either (i) no agents remain unassigned in  $M^1$ , or (ii) at least one agent is not assigned to his reduced first choice in  $M^1$  by Process(Q). In case (i), each agent is assigned to his reduced first choice (i.e. the first choice on his reduced preference list) in M<sup>1</sup> and so cannot be involved in any cyclic coalition as Lemma 2.4.4 on page 31 will establish. The following lemma shows that when case (ii) happens at the end of the preprocessing step, a cyclic coalition must exist with respect to M.

**Lemma :** Suppose that pre-processing terminates, and there exists an agent who is unassigned in  $M^1$ . Then a cyclic coalition must exist with respect to M.

**Proof.** Let  $a_0$  be an agent who is not assigned in  $M^1$  to his reduced first choice  $p(a_0)$  at the end of preprocessing. Hence,  $a_0$  is an unlabelled agent and  $p(a_0) \neq M(a_0)$ . It follows that  $p(a_0)$  must be full in M for otherwise M is not trade-in-free, a

contradiction. However,  $p(a_0)$  cannot be full in  $M^1$  for otherwise  $p(a_0)$  would have been removed from  $a_0$ 's preference list by preprocessing and cannot be the reduced first-choice house of  $a_0$ .

Hence, there exists some agent  $a_1 \in M(p(a_0))\setminus M^1(p(a_0))$  because if an agent a is assigned in  $M^1$  by Process(Q) in pre-processing, it must be the case that a must be assigned in  $M^1$  to M(a). It follows immediately that  $a_1$  must be unassigned in  $M^1$ . Furthermore,  $p(a_1) \neq M(a_1)$  (or else  $a_1 \notin M(p(a_0)) \setminus M^1(p(a_0))$ ) so that  $p(a_0) \neq p(a_1)$ . By reusing the same argument, it follows that we can trace a sequence of agents  $S = ha_0$ ,  $a_1$ , ...i such that  $a_i$  is assigned in M but unassigned in  $M^1$  and  $p(a_i) = M(a_{i+1})$  for  $i \ge 0$ . Since the number of agents is finite, there must be some r such that  $a_r = a_x$  for some  $0 \le x < r - 1$ , where without loss of generality  $a_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}$  are distinct agents. However, the substring of agents  $C = ha_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}$  i within S must then constitute a cyclic coalition with respect to M.

Now, it is clear that an (unlabelled) agent  $a_i$  can only be added to Q when the last house that  $a_i$  prefers to  $M(a_i)$  gets removed from his preference list so that  $p(a_i)$  becomes equal to  $M(a_i)$ , and this happens only once in pre-processing. Since no agent is added to Q twice, the while loop of process(Q) is bound to terminate. As a result, the pre-processing step must also terminate.

A]	lgorith	1 m 5	Ma	in P	hase	3	loop
----	---------	-------	----	------	------	---	------

for each unlabelled agent ai do

1:  $P := \{a_i\}; \{// P \text{ is a stack of agents}\}$ 

3: 
$$c(a_i) := 1; \{// \text{ counter record the number of times an agent is in P } \}$$

- 4: while  $P \neq \emptyset$  do
- 5:  $a^{1}_{i} := P.pop();$
- 6: if  $c(a_i^1) = 2$  then
- 7:  $a^{11}_i := a^1_i;$
- 8: repeat

9:	$Q.add(a^{11}_i);$
10:	$a^{11}_{i} := P.pop();$
11:	until $a^{11}_i = a^1_i$
12:	call Process(Q);
13:	else
14:	$P.push(a_{i}^{1});$
15:	choose any $a^{11}_i \in M_0(p(a^1_i));$
16:	$c(a^{11}_i) := c(a^{11}_i) + 1;$
17:	$P.push(a^{11}i);$

# Phase 3 loop

We then make use of the algorithm in the main Phase 3 loop, as shown in Algorithm 5 to construct the envy graph in order to detect and satisfy cyclic coalitions. For each agent  $a_i$  who is not assigned to his reduced first-choice in M, we repeatedly build a path of agents (represented by P) starting from  $a_i$  in the main while loop and check if P cycles. To do so, we initialize a counter  $c(a_i)$  to 0 for each agent  $a_i$ .

Now, if  $c(a_i^1) \neq 2$  for some agent  $a_i^1$  in P during an iteration of the while loop, then we extend P by following the reduced first-choice edge of  $a_i^1$  in line 15. Let  $p(a_i^1) = h_j$  and let  $a_i^{11}$  be any member of  $M_0(h_j)$ . Note that  $M_0(h_j)$  must be non-empty. For, suppose not. Since  $a_i^1$  prefers  $h_j$  to  $M(a_i^1)$ ,  $h_j$  must be full in M (or else M is not tradein-free). Each agent  $a_k$  assigned to  $h_j$  in M either becomes assigned to  $h_j$  again in  $M^1$  if  $p(a_k) = M(a_k)$ , or to some other house via the satisfaction of some cyclic coalition. In the latter case, this causes some agent  $a_i$  to be assigned to  $h_j$  in  $M^1$  in  $a_k$ 's place. Since  $h_j$  is not full in  $M^1$  by definition of  $p(a_i^1) = h_j$ , it follows that there exists some agent belonging to  $M(h_j)$  who is currently unassigned in  $M^1$ . Hence,  $M_0(h_j)$  must be nonempty. Otherwise, if  $c(a_i^1) = 2$ , it follows that we have a cyclic coalition in P starting from  $a_i^1$ .

We satisfy C by popping each agent  $a^{11}_{i}$  in C from P until we remove C, and

add each  $a^{11}_i$  to Q. We then call Process(Q) to assign each  $a^{11}_i$  to  $p(a^{11}_i)$  in M<sup>1</sup>, to label each  $a^{11}_i$  in order to remove the agent from further consideration by the algorithm, as well as to remove  $M^1(a^{11}_i)$  from the preference lists of the remaining unlabelled agents if the house becomes full in M<sup>1</sup>.

# **Correctness of Phase 3 loop**

If there are unlabelled agents at the start of the main Phase 3 loop, there must exist at least one cyclic coalition with respect to M involving a subset of these agents by Lemma 2.4.2. The following lemma strengthens this result by showing that if there exist any unlabelled agents at any point of time in the execution of Phase 3, then a cyclic coalition must exist.

**Lemma :** Consider a given iteration of the for loop in Phase 3. If there exists an agent who remains unlabelled, then a cyclic coalition must exist with respect to M.

**Proof.** Let  $a_0$  be an agent who is unlabelled during a given iteration of the for loop of Phase 3. Then,  $a_0$  is not assigned in M<sup>1</sup> to his reduced first choice  $p(a_0)$ . It follows that  $p(a_0) = M(a_0)$ . Now,  $p(a_0)$  must be full in M for otherwise M is not trade-in-free, a contradiction.

However,  $p(a_0)$  cannot be full in  $M^1$  for otherwise  $p(a_0)$  would have been removed from  $a_0$ 's preference list by Process(Q) and cannot be the reduced firstchoice house of  $a_0$ . Now, each agent  $a^1 \in A$  becomes assigned in  $M^1$  to either  $M(a^1)$ when  $M(a^1) = p(a^1)$ , or to  $p(a^1)$  when we satisfy a cyclic coalition involving  $a^0$ . Since  $p(a_0)$  is currently under subscribed in  $M^1$ , it follows that there exists a non-empty subset of agents  $A_s$  such that each agent a in  $A_s$  belongs to  $M(p(a_0)) \setminus M^1(p(a_0))$  and a is currently unassigned in  $M^1$ . Let  $a_1 \in A_s$ . It must be the case that  $p(a_1) \neq M(a_1)$  and hence,  $p(a_1) \neq p(a_0)$ . By reusing the same argument, it follows that we can trace a sequence of agents  $S = ha_0$ ,  $a_1$ , ...i such that  $a_i$  is assigned in M but unassigned in  $M^1$  and  $p(a_i) = M(a_{i+1})$  for  $i \ge 0$ . Since the number of agents is finite, there must be some r such that  $a_x = a_r$  for some  $0 \le x < r - 1$ , where without loss of generality  $a_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}$  are distinct agents. However, the substring of agents  $C = ha_x$ ,  $a_{x+1}$ , ...,  $a_{r-1}i$  within S must then constitute a cyclic coalition with respect to M.

The next lemma shows that when all agents are assigned in M<sup>1</sup>, we then obtain a cyclic-coalition-free matching.

**Lemma :** If no unlabelled agents remain at any stage of Phase 3, then  $M^1$  is cyclic-coalition-free.

**Proof.** If there does not exist any unlabelled agents, then every agent is assigned in  $M^1$ . Let  $a_0$  be an arbitrary agent. Suppose that there is a cyclic coalition  $C = ha_0$ ,  $a_1$ , ...,  $a_{r-1}i$  with respect to  $M^1$  involving  $a_0$ . Let  $M^1(a_0) = h_j$  and let  $M^1(a_1) = h_k$ . By definition of C,  $a_0$  must prefer  $h_k$  to  $h_j$ . Since  $a_0$  was assigned to  $h_j$  instead of  $h_k$  in  $M^1$ , it follows that  $h_k$  must have been full in  $M^1$  at the time that  $a_0$  was assigned to  $h_j$  in  $M^1$ . It must then be the case that  $a_1$  was considered by Phase 3 before  $a_0$  or else  $M^1(a_1) \neq h_k$ . Now, by applying the same argument to the remaining agents in C, we can establish that  $a_{r-1}$  must have been considered by Phase 3 before  $a_0$ . Let  $M^1(a_{r-1}) = h_l$ . It follows that  $a_{r-1}$  must prefer  $h_j$  to  $h_l$ . Now, it must be the case that at the time that  $a_{r-1}$  was assigned in  $M^1$ ,  $h_j$  must have been under subscribed for otherwise  $a_0$  could not have been assigned to  $h_j$  later on. However, this gives a contradiction for  $a_{r-1}$  prefers  $h_j$  to  $h_l$  and should then be assigned to  $h_j$  by Phase 3 instead.

Suppose that the envy graph involves the sequence of agents  $S = ha_0$ ,  $a_1$ , ...,  $a_{r-1}i$  and suppose that only a substring of these agents  $C = ha_i$ ,  $a_{i+1}$ , ...,  $a_{r-1}i$  constitute a cyclic coalition where  $0 \le i < r - 1$ . Let us call the agents in the substring ha<sub>0</sub>,  $a_1$ , ...,  $a_{i-1}i$  the tail of C. Now, if certain houses become full in M<sup>0</sup> as a result of satisfying C, thereby causing M( $a_{i-1}$ ) to become the reduced first choice house for the agent  $a_{i-1}$ ,

then  $a_{i-1}$  gets added to Q and assigned to  $M(a_{i-1})$  in  $M^1$  subsequently by Process(Q). Note that this can cause an unwinding effect in the tail in which each agent  $a_k$  ( $0 \le k \le i - 2$ ), such that  $M(a_k)$  lies immediately after  $p(a_k)$  in  $a_k$ 's reduced preference list, gets added to Q and assigned to  $M(a_k)$  in  $M^1$  by Process(Q) in descending agent subscript order until either we reach an agent  $a_{k-1}$  such that there exists a house between  $p(a_{k-1})$  and  $M(a_{k-1})$  in  $a_{k-1}$ 's reduced preference list or the tail becomes empty as a result. In the former case, the main Phase 3 loop, then extends P by following the reduced first-choice edge of  $a_{k-1}$ . In the latter case, the main Phase 3 loop tries to extend P by following the reduced first-choice edge of the next unlabelled agent, if one exists.

It is straightforward to see that the labelling of agents and the maintenance of  $c(a_i)$  for each agent  $a_i$  ensures that no agent  $a_i$  is added more than twice to P in Phase 3. Clearly, if P is non-empty, P must cycle at some point of time in the execution of Phase 3 (as observed by Lemma 2.4.3). Since each agent  $a_i$  that we add to P belongs to a cyclic coalition or to the tail of a cyclic coalition, we are bound to remove  $a_i$  from P as a result.

It also follows by the labelling of agents that any agent added to Q by the for loop in the main Phase 3 loop is not added to Q again by Process(Q) and vice versa. Hence, no agent gets added to Q twice.

Hence, Phase 3 must terminate when no unlabelled agents remain. When this happens, it follows by Lemma 2.4.4 that  $M^0$  must be a cyclic-coalition-free matching. We next show that each agent  $a_i$  assigned in M at the end of Phase 2 must also be assigned in  $M^1$  at the end of Phase 3.

**Lemma :.** Each agent  $a_i$  assigned in M at the end of Phase 2 is also assigned in M<sup>1</sup> at the end of Phase 3.

**Proof.** Suppose not. Then, let  $a_i$  be an agent who is unassigned in  $M^1$ . Let  $M(a_i) = h_j$ 

. Then,  $a_i \in M(h_j) \setminus M^1(h_j)$ . It follows that  $h_j$  cannot have been the first house on  $a_i$ 's preference list, or else preprocessing would have assigned  $M^1(a_i)$  to be  $h_j$ . Hence, there exists at least one house that  $a_i$  prefers to  $h_j$ . Now, if  $a_i$  is not assigned in  $M^1$  to any of these houses, then the pointer  $p(a_i)$  should move across  $a_i$ 's preference list until it points at  $h_j$ . When this happens,  $a_i$  should then be assigned to  $h_j$  in  $M^1$ . However,  $a_i$  is unassigned in  $M^1$  so that  $h_j$  must have been removed from  $a_i$ 's preference list prior to this as a result of it becoming full in  $M^1$ . Now, if  $h_j$  is full in  $M^1$ , then for every  $a_k \in M^1(h_j)$ , either  $a_k \in M(h_j)$  or there exists a unique  $a_l \in M(h_j) \setminus M^1(h_j)$  such that  $a_k$  and  $a_l$  belong to the same cyclic coalition. However, this implies that  $c_j+1$  agents were assigned to  $h_j$  in M, a contradiction.

Since M is a matching that is also maximum, it follows by Lemma 2.4.5 that  $M^1$  is also a maximum matching.

## **Implementation and analysis**

The time complexity of Phase 3 depends on how efficiently we can implement Process(Q) and the main Phase 3 loop. Let us consider briefly the data structures required.

First of all, let us assume that we represent the stack P as a doubly linked list embedded within an array. We let P contain  $n_1$  elements and we indicate the presence or absence of an agent in P by a 1 or 0 respectively. We maintain a pointer to the top of the stack in addition to previous and next pointers between agents in P at any point of time. We implement Q as a straightforward linked list. We also represent the preference list pref<sub>ai</sub> of each agent  $a_i$  as a doubly linked list embedded within an array. Each element in position j of pref<sub>ai</sub> stores the house lying in the corresponding position in  $a_i$ 's preference list. Also, for each house h in pref<sub>ai</sub>, we maintain a pointer each to the house preceding and following h on  $a_i$ 's preference list respectively. We let  $p(a_i)$  point to the first house in pref<sub>ai</sub>, i.e.  $a_i$ 's (reduced) first choice house. Note that as with any doubly linked list,  $p(a_i)$  gets updated as part of the deletion operation whenever the (reduced) first choice house in pref<sub>ai</sub> is to be deleted from pref<sub>ai</sub>. We then build a rank array rank<sub>ai</sub> for each agent  $a_i$  which stores the position of each house  $h_j$  on  $a_i$ 's preference list, i.e. rank<sub>ai</sub> [j] gives the position of  $h_j$  on pref<sub>ai</sub>.

For each house  $h_j$ , we represent  $M_0(h_j)$  also as a doubly linked list embedded within an array. There are  $n_1$  entries in each  $M_0(h_j)$  indexed according to agent subscript, but we maintain previous and next pointers for at most  $c_j$  of these entries, i.e. previous and next pointers exist between consecutive agents in  $M_0(h_j)$  only if these agents belong to  $M(h_j)$  but are currently unassigned in  $M^1$ . It is straightforward to see that  $L_j$  for each  $h_j$  can be implemented as a linked list. We also maintain a counter  $|M^1(h_j)|$  for each house  $h_j$  to keep track of the number of agents assigned to it in  $M^1$  so far in the execution of Phase 3.

Note that for each doubly linked list that we use, we let the previous pointer of the first element and the next pointer of the last element to each point to null. Now, if we use virtual initialization for the initialization of P, it is clear that it takes only a single traversal of the agents' preference lists in agent subscript order to initialize the rest of the data structures.

To illustrate the use of these data structures, suppose firstly that we pop an agent  $a_i$  from Q. It is straightforward to see if  $a_i$  belongs to P by checking if P [i] is 1 or 0. To remove  $a_i$  from P, we set P [i] = 0 and update the previous and next pointers of P [i]'s predecessor and successor, as well as the pointer to the top of P if necessary. Let  $h_k = M(a_i)$ . To remove  $a_i$  from  $M_0(h_k)$ , we update the next and previous pointers of  $M_0(h_k)$ [i]'s predecessor and successor respectively. Let  $p(a_i) = h_j$ . If  $h_j$  becomes full in  $M^1$  as a result of assigning  $a_i$  to  $h_j$ , i.e.  $|M^1(h_j)| = c_j$ , we want to remove  $h_j$  from the preference lists of those unlabelled agents in  $L_j$ . It is clear that it takes only time linear in the size of  $L_j$  to find these agents. Let  $a_l$  be such an agent. Then, rank<sub>al</sub> [j] enables
us to look up the position of  $h_j$  on  $a_l$ 's preference list in constant time. We then delete  $h_j$  from pref<sub>al</sub> by updating the pointers of  $h_j$  's predecessor and successor in pref<sub>al</sub>. In this way,  $p(a_l)$  always points to the head of  $a_l$ 's preference list (which must be currently under subscribed in  $M^1$ ).

Let us then consider the time complexity of Phase 3 by looking at the entire execution of the main Phase 3 loop and Process(Q) taken over the algorithm's execution. It has already been observed that each agent can be added to P no more than twice and added to Q at most once. Now, lines 8-11 of the main Phase 3 loop are executed at most once for each agent in P. All other operations in the main Phase 3 loop are just O(1) stack operations or simple manipulation of data structures. In Process(Q), all operations apart from those in lines 11-14 can be implemented to run in O(1) time for each agent that is added to or removed from Q. Finally, it is clear that lines 11-14 are executed at most once for each house, and hence, the total number of iterations of the for loop is O(m), taken over all calls to Process(Q). It follows that Phase 3 takes O(m) time overall, giving us the following result.

**Lemma :** Given a maximal trade-in-free matching M in an instance of CHA, Phase 3 constructs a cyclic-coalition-free matching  $M^1$  from M in O(m) time. Since Phase 1 dominates the overall complexity of the algorithm to find a maximum Pareto optimal matching in an instance of CHA, we have the following result.

**Theorem :** Given an instance I of CHA, we can find a maximum Pareto optimal matching in I in  $O(\sqrt{Cm})$  time.

Recall that an alternative approach to finding a Pareto optimal matching given an CHA instance I involves obtaining a weighted graph  $G^1$  of I and then finding a maximum cardinality minimum weight matching of I. This takes  $\Omega(C \min(m \log n, n^2))$  time. If mlog  $n \le n^2$ , then it follows that our algorithm is faster by a factor of

 $\Omega(\sqrt{C} \log n)$ . Otherwise, mlog  $n > n^2$  and our lgorithm is faster by a factor of  $\Omega(\sqrt{C}n2/m)$ .

#### **CHAPTER –IV**

#### **POPULAR MATCHINGS IN CHAT**

#### Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### 4.1 Introduction

In this chapter, we consider the problem of finding a maximum popular matching given an instance of CHA or CHAT. As noted in Section 1.3.2.2, many recent papers have focused on popular matchings, given the importance of voting to many economic decisions, and also given the viability of popular matchings as an optimality criterion for matching problems with one-sided preferences. Here, we present the first algorithmic results for computing popular matchings in a capacitated bipartite graph. The main results of this chapter, and their organization are as follows.

In Section 4.2, we first develop a characterization of popular matchings in a CHA instance I. We then use this characterization to construct an  $O(\sqrt{C}n_1 + m)$  algorithm for finding a maximum popular matching in I or reporting that none exists, where C is the total capacity of the houses,  $n_1$  is the total number of agents and m is the total length of the preference lists in I respectively. In Section 4.3, we provide the first extension of the Edmonds-Gallai Decomposition to the case of a capacitated bipartite graph. Using this, we build a new characterization of popular matchings in a CHAT instance J, and then use it to construct an  $O(\sqrt{C}m)$  algorithm for finding a maximum popular matching in J or reporting that none exists. We finally remark that a straightforward solution to each of the problems of finding a maximum popular matching, given an instance I of CHA or CHAT, may be to use "cloning". Informally,

this entails creating  $c_j$  clones for each house  $h_j$ , to obtain an instance C(I) of HAT (i.e. each house has capacity 1), and then applying the algorithms of [4] to C(I). However, we will show in Sections 4.2 and 4.3 that this method in general leads to slower algorithms than the direct approach that we will be using in each case.

#### 4.2Popular matching's in CHA

#### 4.2.1Characterizing popular matching's

Let I be an instance of CHA, and let G = (A, H, E) be the underlying graph of I, where  $A = \{a_1, a_2, ..., a_{n1}\}$  is the set of agents,  $H = \{h_1, h_2, ..., h_{n2}\}$  is the set of houses and E is the set of edges in G representing the acceptable houses of the agents. We assume all the notations and terminology that have been defined for CHA in Chapters 1 and 2. Recall that, given two matchings M and M<sup>1</sup> in I, we say that an agent  $a_i$  prefers M<sup>1</sup> to M if either (i)  $a_i$  is assigned in M<sup>1</sup> and unassigned in M, or (ii)  $a_i$  is assigned in both M<sup>1</sup> and M and prefers M<sup>1</sup>( $a_i$ ) to M( $a_i$ ). Let P (M<sup>1</sup>, M) denote the set of agents who prefer M<sup>1</sup> to M. Then, M<sup>1</sup> is more popular than M if  $|P(M^1, M)| > |P(M, M^1)|$ , i.e. the number of agents who prefer M<sup>1</sup> to M<sup>1</sup>.

Furthermore, a matching M in I is popular if there is no other matching  $M^1$  in I that is more popular than M.

For each agent  $a_i \in A$ , let  $f(a_i)$  denote the first-ranked house on  $a_i$ 's preference list. Any such house  $h_j$  is called an f-house. For each  $h_j \in H$ , let  $f(h_j) = \{a_i \in A :$  $f(a_i) = h_j\}$  and  $f_j = |f(h_j)|$  (possibly  $f_j = 0$ ). We create a unique last resort house  $l(a_i)$ with capacity 1 for each agent  $a_i \in A$ , and append  $l(a_i)$  to  $a_i$ 's preference list. We also henceforth assume that G contains the vertex  $l(a_i)$  and the edge  $(a_i, l(a_i))$  for each  $a_i \in$ A, and that H contains the respective last resort houses. We let  $n = n_1 + n_2$  and m =|E|. The following lemma is a vital first step in characterizing popular matchings in I. **Lemma 3.2.1.** Let M be a popular matching in I. Then for every f-house  $h_j$ ,  $|M(h_j) \cap f(h_j)| = \min \{c_j, f_j\}.$ 

**Proof.** We consider the following two cases.

**Case (i):** Suppose  $f_j \leq c_j$ . We will show that  $f(h_j) \subseteq M(h_j)$ . For, suppose not. Then choose any  $a_r \in f(h_j) \setminus M(h_j)$ . We consider the subcases that (a)  $h_j$  is under subscribed and (b)  $h_j$  is full. In subcase (a), promote  $a_r$  to  $h_j$  to obtain a more popular matching than M. In subcase (b), choose any  $a_s \in M(h_j) \setminus f(h_j)$ . Let  $h_k = f(a_s)$ . Then  $h_k \neq$  $h_j$ . If  $h_k$  is under subscribed, promote  $a_r$  to  $h_j$  and promote  $a_s$  to  $h_k$  to obtain a more popular matching than M. Otherwise, choose any  $a_t \in M(h_k)$ . If  $a_r = a_t$ , we then promote  $a_r$  to  $h_j$  and promote  $a_s$  to  $h_k$  to obtain a more popular matching than M.

Otherwise, we then promote  $a_r$  to  $h_j$ , promote  $a_s$  to  $h_k$  and demote  $a_t$  to  $l(a_t)$  to obtain a more popular matching than M.

**Case (ii):** Suppose  $f_j > c_j$ . If  $h_j$  is undersubscribed, then  $f(h_j) \not\equiv M(h_j)$  so there exists some  $a_r \in f(h_j) \setminus M(h_j)$  that we can promote to  $h_j$  to obtain a more popular matching as in Case (i)(a). Hence,  $h_j$  is full. Now, suppose for a contradiction that  $M(h_j) \not\equiv f(h_j)$ . Then there exists some  $a_s \in M(h_j) \setminus f(h_j)$ . As  $f_j > c_j$ , it follows that  $f(h_j) \not\equiv M(h_j)$  so there exists some  $a_r \in f(h_j) \setminus M(h_j)$ . The remainder of the argument follows Case (i)(b).

Hence the following properties hold for the new matching. If  $f_j \leq c_j$ , then  $f(h_j) \subseteq M(h_j)$ . Otherwise,  $M(h_j) \subseteq f(h_j)$  and  $|M(h_j)| = c_j$ . Thus, the condition in the statement of the lemma is now satisfied.

For each agent  $a_i$ , we next define  $s(a_i)$  to be the most-preferred house  $h_j$  on  $a_i$ 's preference list such that either (i)  $h_j$  is a not an f-house, or (ii)  $h_j$  is an f-house such that  $h_j \neq f(a_i)$  and  $f_j < c_j$ . Note that  $s(a_i)$  must exist in view of  $l(a_i)$ . We refer to such a

house  $h_j$  as an s-house. We remark that the set of f-houses need not be disjoint from the set of s-houses. It may be shown that a popular matching M will only assign an agent  $a_i$  to either  $f(a_i)$  or  $s(a_i)$ , as indicated by the next two lemmas.

**Lemma 4.2.2.** Let M be a popular matching in I. Then no agent  $a_i \in A$  can be assigned in M to a house between  $f(a_i)$  and  $s(a_i)$  on  $a_i$ 's preference list.

**Proof.** Suppose that  $a_i$  is assigned to a house  $h_k$  between  $f(a_i)$  and  $s(a_i)$ . Then  $h_k$  is an f-house and  $f_k \ge c_k$ , for otherwise  $s(a_i) = h_k$ . As  $f_k \ge c_k$ , by Lemma 3.2.1,  $M(h_k) \subseteq f(h_k)$ . However,  $f(a_i) \ne h_k$ , thus  $a_i \in f(h_k)$ . Hence,  $a_i$  cannot be assigned to  $h_k$ .

**Lemma 4.2.3.** Let M be a popular matching in I. Then no agent  $a_i \in A$  can be assigned in M to a house worse than  $s(a_i)$  on  $a_i$ 's preference list.

**Proof.** Let  $h_j = s(a_i)$ . If  $h_j$  is under subscribed, then we can promote  $a_i$  to  $h_j$ , a contradiction. Hence,  $h_j$  is full. We consider two cases.

**Case (i):**  $h_j$  is an f-house. By definition of an s-house,  $f_j < c_j$ , so there exists some  $a_r \in M(h_j) \setminus f(h_j)$ . Let  $h_k = f(a_r)$ . Then  $h_k \neq h_j$ . As  $c_k \ge 1$  and  $f_k \ge 1$ , it follows by Lemma 4.2.1 that  $M(h_k) \neq \emptyset$ . Let  $a_s \in M(h_k)$ . Now, if  $a_i = a_s$ , we can then promote  $a_i$  to  $h_j$  and promote  $a_r$  to  $h_k$  to obtain a more popular matching than M, a contradiction.

Otherwise, we can then promote  $a_i$  to  $h_j$ , promote  $a_r$  to  $h_k$ , and demote  $a_s$  to  $l(a_s)$  to obtain a more popular matching than M, a contradiction.

**Case (ii):**  $h_j$  is not an f-house. Let  $a_r \in M(h_j)$ . Then  $a_r \notin f(h_j)$ . The remainder of the proof of this case proceeds as in Case (i).

Recall that G is the underlying graph of I. We form a subgraph  $G^1$  of G by letting  $G^1$  contain only two edges for each agent  $a_i$ , that is, one to  $f(a_i)$  and the other to

s(a<sub>i</sub>). We say that a matching M is agent-complete in a given graph if it assigns all agents in the graph. It follows that, in view of last resort houses, all popular matchings must be agent-complete in G<sup>1</sup>. However, G<sup>1</sup> need not admit an agent-complete matching if  $s(a_i) \neq l(a_i)$ 

for some agent  $a_i$ . In conjunction with Lemmas 3.2.1-3.2.3, the graph  $G^0$  gives rise to the following characterization of popular matchings in I.

Theorem 4.2.1. A matching M is popular in I if and only if

- 1. for every f-house h<sub>j</sub>,
  - (a) if  $f_j \leq c_j$ , then  $f(h_j) \subseteq M(h_j)$ ;
  - (b) if  $f_j > c_j$ , then  $|M(h_j)| = c_j$  and  $M(h_j) \subseteq f(h_j)$ .
- 2. M is an agent-complete matching in the reduced graph  $G^0$ .

**Proof.** By Lemmas 3.2.1-3.2.3, any popular matching necessarily satisfies Conditions 1 and 2. We now show that these conditions are sufficient.

Let M by any matching satisfying Conditions 1 and 2 and suppose for a contradiction that  $M^1$  is a matching that is more popular than M. Let  $a_i$  be any agent that prefers  $M^1$  to M and let  $h_k = M^1(a_i)$ . Since M is an agent-complete matching in  $G^1$ , and since  $G^1$  contains only edges from  $a_i$  to  $f(a_i)$  and  $s(a_i)$ , then  $M(a_i) = s(a_i)$ . Hence either (i)  $h_k = f(a_i)$  or (ii)  $h_k$  is an f-house such that  $h_k \neq f(a_i)$  and  $f_k \ge c_k$ , by definition of  $s(a_i)$ .

In Case (i), if  $f_k < c_k$  then by Condition 1(a),  $a_i \in M(h_k)$ , a contradiction. Hence in both Cases (i) and (ii),  $f_k \ge c_k$ . In each of the cases that  $f_k = c_k$  and  $f_k > c_k$ , it follows by Conditions 1(a) and 1(b) that  $|M(h_k)| = c_k$  and  $M(h_k) \subseteq f(h_k)$ . Since  $h_k$  is full in M, it follows that  $|M(h_k) \setminus M^1(h_k)| \ge |M^1(h_k) \setminus M(h_k)|$ . Hence for every  $a_i$  who prefers  $M^1(a_i) = h_k$  to  $M(a_i)$ , there is a unique  $a_j \in M(h_k) \setminus M^1(h_k)$ . But as  $a_j \in$  $M(h_k)$ , it follows that  $h_k = f(a_j)$ . Hence  $a_j$  prefers  $M(a_j)$  to  $M^1(a_j)$ . Therefore, M is Algorithm 6 Algorithm Popular-CHA

 $M := \emptyset;$ 

1: for each f-house  $h_j$  do

- 2:  $c^{1}_{j} := c_{j};$
- 3: if  $f_j \leq c_j$  then
- 4: for each  $a_i \in f(h_j)$  do
- 5:  $M := M \cup \{(a_i, h_j)\};$
- 6: delete  $a_i$  and its incident edges from  $G^1$ ;

7: 
$$c^{1}_{j} := c_{j} - f_{j};$$

8: remove all isolated and full houses, and their incident edges, from  $G^1$ ;

9: compute a maximum matching  $M^1$  in  $G^1$  using capacities  $c_{ij}^1$ ;

10: if  $M^1$  is not agent-complete in  $G^1$  then

- 11: output "no popular matching exists"
- 12: else

13: 
$$M := M \cup M^1$$
;

14: for each  $a_i \in A$  do

15: 
$$h_j := f(a_i);$$

16: if  $f_j > c_j$  and  $|M(h_j)| < c_j$  and  $h_j \neq M(a_i)$  then

17: promote  $a_i$  from  $M(a_i)$  to  $h_j$  in M;

#### 4.2.1 Finding a popular matching

Theorem 4.2.1 leads to Algorithm Popular-CHA for finding a popular matching in a CHA instance I, or reporting that none exists, as shown in Algorithm 6. The algorithm begins by using a preprocessing step (lines 2-9) on G<sup>1</sup> that assigns agents to their first-choice house  $h_i$  whenever  $f_i \le c_i$ , so as to satisfy Condition 1(a) of Theorem 4.2.1

Our next step computes a maximum matching M<sup>1</sup> in G<sup>1</sup>, according to the

adjusted house capacities  $c_{ij}^{1}$  that are defined following pre-processing. We use Gabow's algorithm [15] to compute  $M^{1}$  in  $G^{1}$  and then test whether  $M^{1}$  is agentcomplete. The pre-allocations are then added to  $M^{1}$  to give M. As a last step, we ensure that M also meets Condition 1(b) of Theorem 4.2.1. For, suppose that  $h_{j} \in$ H is an f-house such that  $f_{j} > c_{j}$ . Then by definition,  $h_{j}$  cannot be an s-house. Thus if  $a_{k} \in M(h_{j})$  prior to the third for loop, it follows that  $a_{k} \in f(h_{j})$ . At this stage, if  $h_{j}$  is under subscribed in M, we repeatedly promote any agent  $a_{i} \in f(h_{j}) \setminus M(h_{j})$  from  $M(a_{i})$  (note that  $M(a_{i})$  must be  $s(a_{i})$  and hence cannot be an f-house  $h_{l}$  such that  $f_{l} > c_{l}$ ) to  $h_{j}$  until  $h_{j}$  is full, ensuring that  $M(h_{j}) \subseteq f(h_{j})$ .

It is clear that the reduced graph  $G^0$  of G can be constructed in O(m) time. The graph  $G^1$  has O(n<sub>1</sub>) edges since each agent has degree 2 in  $G^1$ . It is straightforward to see that each of the pre- and post-processing steps involving the three for loop phases takes O(n<sub>1</sub> + n<sub>2</sub>) time. The complexity of Gabow's algorithm [15] for computing M<sup>1</sup> in  $G^1$  is O(Cn<sub>1</sub>). Hence we obtain the following result concerning the complexity of Algorithm Popular-CHA.

**Lemma 4.2.4.** Given an instance of CHA, we can find a popular matching, or determine that none exists, in O( $Cn_1 + m$ ) time.

#### 4.2.2Finding a maximum popular matching

It remains to consider the problem of finding a maximum popular matching in I. We begin by dividing the set of all agents into disjoint sets. Let  $A_1$  be the set of all agents such that if  $a_i$  is an agent in  $A_1$ , then  $s(a_i) = l(a_i)$ . Also, let  $A_2 = A - A_1$ . Our aim is to find a matching M that satisfies the conditions of Theorem 4.2.1, and that minimizes the number of  $A_1$ -agents who are assigned to their last resort house.

We begin by constructing  $G^1$ , and carrying out the pre-processing step in lines 2-9 of Algorithm Popular-CHA on all agents in  $A_1 \cup A_2$ . We then try to find a maximum matching  $M^1$  in  $G^1$  that only involves the A<sub>2</sub>-agents that remain after preprocessing and their incident edges. If  $M^1$  is not an agent-complete matching of the agents in A<sub>2</sub> that remain after pre-processing, then G admits no popular matching by Theorem 4.2.1.

Otherwise, we remove all edges in  $G^1$  that are incident to a last resort house, and try to assign A<sub>1</sub>-agents to their first-choice houses. At each step, we try to assign an additional A<sub>1</sub>-agent to his first-choice house by finding an augmenting path with respect to M<sup>1</sup> using Gabow's algorithm [15], so that we have a maximum matching of agents in A<sub>1</sub>  $\cup$  A<sub>2</sub> in mG<sup>1</sup> at the end of this process. If any A<sub>1</sub>-agent remains unassigned, we simply assign him to his last resort house, to obtain an agent-complete matching in G<sup>1</sup>. We also ensure that Condition 1(b) of Theorem 3.2.1 is met by executing the third for loop in Algorithm Popular-CHA. It follows that the matching so obtained, together with the pre-assignments from earlier, is a maximum popular matching, giving the following theorem.

**Theorem 4.2.2.** Given an instance of CHA, we can find a maximum popular matching, or determine that none exists, in O( $Cn_1 + m$ ) time.

#### **CLONING VERSUS OUR DIRECT APPROACH**

#### Dr.D.R.Kirubaharan

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

An alternative approach to our algorithm may be to use cloning. Given an instance I of CHA, we may obtain an instance J of HAT by creating  $c_j$  clones  $h_{j}^1$ ,  $h_{j}^2$ , ...,  $h_{j}^{c_j}$  of each house  $h_j$  in I, where each clone has a capacity of 1. In addition, we replace each occurrence of  $h_j$  in a given agent's preference list with the sequence  $h_{j}^1$ ,  $h_{j}^2$ , ...,  $h_{j}^{c_j}$ , the elements of which are listed in a single tie at the point where  $h_j$  appears. We may then apply the O( nm) algorithm for HAT given by [4] to J in order to find a maximum popular matching in I.

We now compare the worst-case complexity of the above cloning approach with that of our direct algorithm. The underlying graph G<sub>J</sub> of J contains  $n^1 = n_1 + C$ vertices. Let  $c_{\min} = \min\{c_j : h_j \in H\}$ , and for  $a_i$  A, let A<sub>i</sub> denote the set of acceptable houses for ai. Then the number of edges in G<sub>J</sub> is  $m^1 = \sum_{a_i \in A} \sum_{h_j \in A_i} c_j \ge m C_{\min}$ . Hence the complexity of applying the algorithm given by [4] to J is  $\Omega(\sqrt{n1 + CmCmin})$ Recall that the complexity of Algorithm Popular-CHA is  $O(\sqrt{C}n_1 + m)$ . It follows that the cloning method is slower by a factor of  $\Omega(\sqrt{n1 + CmCmin})$  or  $\Omega(mC_{\min}/n_1)$ (note that  $m \ge n_1$  and  $C_{\min} \ge 1$ ) according as  $\sqrt{C}n_1 \le m$  or  $\sqrt{C}n_1 > m$  respectively..

#### **5.1Popular matchings in CHAT**

In this section, we generalize the characterization of popular matchings together with Algorithm Popular-CHA as given in the previous section to the case where we are given an instance of CHAT.

#### **Characterizing popular matchings**

Let M be a popular matching in an instance I of CHAT. For each agent  $a_i \in A$ , let  $f(a_i)$ denote the set of first-ranked houses on  $a_i$ 's preference list (clearly it is possible that  $|f(a_i)| > 1$  in view of ties in the preference lists). We refer to all such houses  $h_j$  as fhouses and we let  $f(h_j) = \{a_i \in A : h_j \in f(a_i)\}$ . Let G = (A, H, E) be the underlying graph of I. Define  $E_1 = \{(a_i, h_j) : a_i \in A \land h_j \in f(a_i)\}$  to be the set of first-choice edges. We define the first-choice graph of G as  $G_1 = (A, H, E_1)$ .

Given a CHAT instance I, since it is possible for an agent to have greater than one f-house, Lemma 3.2.1 no longer holds in general. For example, it is possible for an f-house  $h_j$  such that  $f_j = c_j$  to not be assigned to all the agents in  $f(h_j)$  in a popular matching if there are more f-houses than agents in I. This makes  $h_j$  eligible to be the shouse of some agent not in  $f(h_j)$  in I whereas this would not have been possible in any given CHA instance. Hence, we will work towards a new definition of s-houses in the context of CHAT in this subsection. For instances with strict preference lists, Lemma 3.2.1 implies that  $M \cap E_1$  is a maximum matching in  $G_1$ .

As the next lemma indicates, this latter condition also extends to the CHAT case.

**Lemma 5.1.** Let M be a popular matching in I. Then  $M \cap E_1$  is a maximum matching in  $G_1$ .

**Proof.** Let  $M_1 = M \cap E_1$ . Suppose, for a contradiction that  $M_1$  is not a maximum matching in  $G_1$ . Then  $M_1$  admits an augmenting path  $P = ha_1, h_1, ..., a_k, h_k$  with respect to  $G_1$ .

Now, in view of last resort houses, a<sub>1</sub> must be assigned in M. It follows that

 $M(a_1) \notin f(a_1)$ , for otherwise  $M(a_1) \in P$ . We let  $M^1 = M \setminus \{(a_1, M(a_1))\}$ . We consider the following cases for  $h_k$ .

**Case (i):**  $h_k$  is under subscribed in  $M^1$ . As  $a_1$  is unassigned in  $M^1$ ,  $h_1 \in f(a_1)$ , and  $|M^1(h_k)| < c_k$ , we can augment  $M^1$  with P to obtain a new matching  $M^{11}$ . Then,  $a_1$ is assigned with  $h_1$  in  $M^{11}$ . Furthermore, as all edges in  $G^1$  are first-choice edges, all other agents in P become assigned in  $M^{11}$  to one of their other first-choice houses. However,  $M^{11}$  is more popular than M, a contradiction.

**Case** (ii):  $|M^{1}(h_{k})| = c_{k}$ . Choose any  $a_{s} \in M^{1}(h_{k}) \setminus f(h_{k})$ . Note that such an  $a_{s}$  must exist, for  $h_{k}$  is full in  $M^{1}$  but under subscribed in  $M_{1}$ . Clearly,  $a_{s} \neq a_{i}$  for  $1 \leq i \leq k$ . Choose any  $h_{t} \in f(a_{s})$ . Now,  $h_{t}$  cannot be a house in P for suppose not. Without loss of generality, let  $h_{t} = h_{j}$  where  $1 \leq j < k$ . Let  $C = hh_{j}$ ,  $a_{j+1}$ ,  $h_{j+1}$ , ...,  $a_{k}$ ,  $h_{k}$ ,  $a_{s}$ i. Let also  $M^{11} = M^{1} \oplus C$ . It follows that each agent  $a_{x} \neq a_{s}$  in C becomes assigned in  $M^{11}$  to one of their other first-choice houses while  $a_{s}$  improves by becoming assigned in  $M^{11}$  to one of his first-choice house. However then,  $M^{11}$  is more popular than M, a contradiction. Hence,  $h_{t}$  does not belong to P. Now, if  $h_{t}$  is under subscribed in  $M^{1}$ , we can then augment  $M^{1}$  with P so that  $a_{1}$  improves and the other agents in P are indifferent, and promote  $a_{s}$  to  $h_{t}$  to obtain a more popular matching than M, a contradiction. Otherwise, choose any  $a_{u} \in M^{1}(h_{t})$ . Since  $h_{t} \notin P$  and  $a_{1}$  is unassigned in  $M^{1}$  with P so that  $a_{1}$  improves and the other augment  $M^{1}$  with P so that  $a_{1}$  improves  $M^{1}(h_{t})$ . Since  $h_{t} \notin P$  and  $a_{1}$  is unassigned in  $M^{1}$ , we contradiction. Otherwise, choose any  $a_{u} \in M^{1}(h_{t})$ . Since  $h_{t} \notin P$  and  $a_{1}$  is unassigned in  $M^{1}$ , we then augment  $M^{1}$  obtain a more popular matching than M, a contradiction. Otherwise, and the other agents in P are indifferent, promote  $a_{s}$  to  $h_{t}$  and demote  $a_{u}$  to  $1(a_{u})$  to obtain a more popular matching than M, a contradiction.

We now work towards a definition of s-houses by using the Edmonds-Gallai Decomposition. Let M be a maximum matching in some bipartite graph G where all vertices have capacity 1. According to Lemma 1.2.1, the vertices of G can be partitioned into three disjoint sets: E, O and U. Vertices in E, O and U are called even, odd, and unreachable respectively. A vertex v is even (odd) if there exists an

alternating path of even (odd) length from an unassigned vertex in G to v. If no such alternating path exists, v is unreachable.

As noted in Section 1.2, this vertex labelling is also known as the EOU labelling. Our aim is to obtain an EOU labelling of  $G_1$  relative to a maximum matching  $M_1$  of  $G_1$  (as obtained by Gabow's algorithm [15], for example). However Lemma 1.2.1 applies directly only to the case where each vertex in the given bipartite graph has capacity 1. We will show that the Edmonds-Gallai Decomposition also holds in the case of a capacitated bipartite graph as follows.

Let G = (U, W, E) be a capacitated bipartite graph. Also, let M be a maximum matching of G. Let C(G) be a cloned graph of G by replacing every vertex  $w_j \in W$  with the clones  $w_j^{1}$ ,  $w_j^{2}$ , ...,  $w_j^{cj}$  where  $c_j$  is the capacity of  $w_j$ . We then divide the capacity of each vertex  $w_j \in W$  among its clones by allowing each clone to have capacity 1. In addition, if  $(u_i, w_j)$  belongs to G, then we add  $(u_i, w_j^k)$  to C(G) for all k  $(1 \le k \le c_j)$ . We then adapt the maximum matching M in G to obtain a matching C(M) in C(G), as follows. If a vertex  $w_j \in W$  in G is assigned to  $x_j$  vertices  $u_1, ..., u_{xj}$  in M, then we add  $(u_k, w_j^k)$  to C(M) for  $1 \le k \le x_j$ , so that |C(M)| = |M| and C(M) is a maximum matching in C(G). It follows that C(G) is a bipartite graph in which all of its vertices on the right hand side have capacity 1.

Let us now clone C(G) to obtain a bipartite graph C(G)<sup>1</sup> in which all of its vertices have capacity 1 by repeating the above steps for the vertices in U. That is, we replace every vertex  $u_i \in U$  with the clones  $u^{1_i}, u^{2_i}, \ldots, u^{c_i}$  where  $c_i$  is the capacity of  $u_i$ . We then divide the capacity of each vertex  $u_i \in U$  among its clones by allowing each clone to have capacity 1. In addition, if  $(u_i, w^k_j)$  belongs to C(G), then we add  $(u^{l_i}, w^k_j)$  to C(G) for all 1  $(1 \le l \le c_i)$  where  $w^k_j$  is a clone of  $w_j \in W$  in C(G). We then adapt the maximum matching C(M) in C(G) to obtain a matching C(M)<sup>1</sup> in C(G)<sup>1</sup> as follows. If a vertex  $u_i \in U$  in G is assigned to  $y_i$  vertices  $w_1^1, ...w_y^{l_i}$  in C(M), then we add  $(u^{k_{i}}, w_{k}^{1})$  to  $C(M)^{1}$  for  $1 \le k \le y_{i}$  where without loss of generality,  $w_{k}^{1}$  is a clone of the vertex  $w_{k} \in W$  in C(G). It follows that  $|C(M)^{1}| = |M|$  and  $C(M)^{1}$  is a maximum matching in  $C(G)^{1}$ . It also follows that  $C(G)^{1}$  is a bipartite graph in which all of its vertices have capacity 1.

Suppose that we are given an EOU labelling of the vertices in  $C(G)^1$  with respect to  $C(M)^1$  based on Lemma 1.2.1. The next lemma shows that the clones corresponding to each vertex  $v_j \in U \cup W$  in G have the same EOU label in  $C(G)^1$ .

**Lemma 5.2.** Let G be a capacitated bipartite graph and let M be a maximum matching in G. Define the cloned graph  $C(G)^1$  and its corresponding maximum matching  $C(M)^1$ as above. Then, given any vertex  $v_j \in U \cup W$ , any two clones of  $v_j$  in  $C(G)^1$  have the same EOU label.

**Proof.** Without loss of generality, let  $v_j$  be a vertex  $w_j$  belonging to the vertex set W; analogous results can be proven if  $v_j \in U$ . Let  $w_j^x$  and  $w_j^y$  be two clones corresponding to  $w_j$ . We consider the cases where (1)  $w_j^x$  is even, (2)  $w_j^x$  is odd, and (3)  $w_j^x$  is unreachable.

**Case** (1): If  $w_j^x$  is even, then we consider the subcases where (a)  $w_j^x$  is assigned in  $C(M)^1$ , and (b)  $w_j^x$  is unassigned in  $C(M)^1$ . In subcase (a), if  $w_j^y$  is unassigned in  $C(M)^1$ , then it follows immediately that  $w_j^y$  is also even. Hence, suppose that  $w_j^y$  is also assigned in  $C(M)^1$ . As  $w_j^x$  is even, there exists an even length alternating path P to  $w_j^x$  in  $C(G)^1$  from an unassigned vertex clone belonging to W. Let  $u_i^p$  be the vertex that precedes  $w_j^x$  on P where  $u_i^p$  is a clone vertex of  $u_i$  in  $C(G)^1$ . It follows that  $(u_i^p, w_j^x) \in C(M)^1$  from our definition of the path P. As  $w_j^y$  is also assigned in  $C(M)^1$ , let  $(u_k^q, w_j^y) \in C(M)^1$  where  $u_k^q$  is a clone vertex of  $u_k$  in  $C(G)^1$ and  $(u_k, w_j) \in M$ . Then, it follows that  $(u_k^q, w_j^x)$  must be an edge in  $C(G)^1$ . As  $w_j^x$  is even,  $u_k^q$  is odd. As a result,  $w_j^y$  is even. In subcase (b), if  $w_j^y$  is also unassigned in  $C(M)^1$ , then it is again immediate that  $w^{y_j}$  is also even. Hence, suppose that  $w^{y_j}$  is assigned in  $C(M)^1$ , to  $u^{p_i}$  say, where  $u^{p_i}$  is a clone vertex of  $u_i$  in  $C(G)^1$  and  $(u_i, w_j) \in M$ . Now  $(u^{p_i}, w^{x_j})$  is also an edge in  $C(G)^1$ . As  $w^{x_j}$  is even,  $u^{p_i}$  is odd, and hence  $w^{y_j}$  is even.

**Case (2):** If  $w^{x_j}$  is odd, then there must exist an odd-length alternating path from an unassigned vertex  $u^{p_i}$  to  $w^{x_j}$  where  $u^{p_i}$  is a clone of  $u_i \in U$  in  $C(G)^1$ . It follows that  $w^{x_j}$  cannot be unassigned for otherwise  $C(M)^1$  admits an augmenting path, a contradiction. Hence,  $w^{x_j}$  is assigned in  $C(M)^1$  to  $u^{q_k}$ , say, where  $u^{q_k}$  is a clone of  $u_k \in$ U in  $C(G)^1$  and  $(u_k, w_j) \in M$ . Then,  $u^{q_k}$  is even. However,  $(u^{q_k}, w^{y_j})$  is an edge in  $C(G)^1$ , so it follows that  $w^{y_j}$  is odd.

**Case (3):** Now,  $w^{y_{j}}$  must also be unreachable. For, suppose not. If  $w^{y_{j}}$  is even, then  $w^{x_{j}}$  is also even by Case (1), a contradiction. If  $w^{y_{j}}$  is odd, then  $w^{x_{j}}$  is also odd by Case (2), a contradiction.

In view of Lemma 3.3.2, it follows that the clones corresponding to each vertex  $v_j \in U \cup W$  have the same EOU label in  $C(G)^1$ , thereby giving us a well-defined characterization of EOU labels of all vertices in G. That is, if the clones of the vertex  $v_j$  are even, odd or unreachable in  $C(G)^1$ , we can correspondingly label  $v_j$  as even, odd or unreachable in G.

Suppose that we now have an EOU labelling of the vertices in G as described above. The next result is a consequence of Lemma 3.3.2.

**Lemma 5.3** let G =(U,W,E) be a capacitated bipartite graph and let M be a maximum matching in G. Then every old or unreachable vertex  $v_j \in U \cup W$  satisfies  $|M(v_j)| = c_j$ .

**Proof**. Let  $v_j \in U \cup W$  be any vertex that is odd (or unreachable) in G. By Lemma 3.3.2, all clones of  $v_j$  will also be odd (or unreachable) in C(G)<sup>1</sup>. It follows that  $v_j$  must be full in M, for otherwise, at least one of its clones  $v_j^x$  will be unassigned in C(M)<sup>1</sup>. However,  $v_j^x$  will then be even, a contradiction.

Hence, Lemma 3.3.2 and Lemma 3.3.3 give rise to the first extension of the Edmonds-Gallai Decomposition to the capacitated bipartite graph as follows.

**Lemma 5.4.** Let G = (U, W, E) be a capacitated bipartite graph and let M be a maximum matching in G. Define E, O and U to be the vertex sets corresponding to even, odd and unreachable vertices in an EOU labelling of G with respect to M. Then:

- (a) The sets E, O and U are pairwise disjoint. Every maximum matching in G partitions the vertices into the same sets of even, odd and unreachable vertices.
- (b) Every maximum matching M in G satisfies the following properties:
  - (i) every vertex in O and every vertex in U is full in M;
  - (ii) every vertex in O is assigned only to vertices in E in M;
  - (iii) every vertex in U is assigned only to vertices in U in M;
  - (iv)  $|M| = \sum_{u_i} \in O_U c_j + \sum_{w_j} \in O_w c_j + \sum_{u_i} \in U_U c_j$ , where  $O_u$  is the set of vertices in U,  $O_W$  is the set of odd vertices in W, and  $U_U$  is the set of unreachable vertices in U.
- (c) No maximum matching in G contains an edge between two vertices in O or a vertex in O with a vertex in U. There is no edge in G connecting a vertex in E with a vertex in U, or between two vertices of E.

It follows that Lemma 3.3.4 enables us to obtain an EOU labelling of  $G_1$  relative to a maximum matching  $M_1$  of  $G_1$ . The following corollary is a result of Lemma 5.4.

Corollary 5.1. Let M be a popular matching in I. Then every odd or unreachable

house  $h_j \in H$  satisfies  $M(h_j) \subseteq f(h_j)$ .

**Proof.** Let  $G_1 = (A, H, E_1)$  be the first-choice graph of I. Then,  $M_1 = M \cap E_1$  is a maximum matching in  $G_1$  by Lemma 4.4.1. Let  $h_j \in H$  be any house that is odd (or unreachable) in  $G_1$ . By Lemma 4.4.4,  $h_j$  is full in  $M_1$ . Since  $C(G_1)$  contains only first-choice edges, it follows that  $M_1(h_j) \subseteq f(h_j)$ , and hence  $M(h_j) \subseteq f(h_j)$ .

We are now in a position to define  $s(a_i)$ , the set of houses such that, in a popular matching M, if  $a_i \in A$  is assigned in M and  $M(a_i) \notin f(a_i)$ , then  $M(a_i) \in s(a_i)$ . We will ensure that any odd or unreachable house  $h_j$  is not a member of  $s(a_i)$ , since  $|M(h_j)| = c_j$  and  $M(h_j) \subseteq f(h_j)$  by Lemma 4.4.4 and Corollary 4.4.1. Hence, we define  $s(a_i)$  to be the set of highest-ranking houses in  $a_i$ 's preference list that are even in  $G_1$ . Any such house is called an s-house. Clearly, it is possible that  $|s(a_i)| > 1$ , however,  $a_i$ is indifferent between all houses in  $s(a_i)$ . Furthermore,  $s(a_i) \neq \Box$  due to the existence of last resort houses which are of degree 0 in  $G_1$  (and thus even). However,  $f(a_i)$  and  $s(a_i)$  need not be disjoint, i.e. either  $f(a_i) = s(a_i)$  or  $a_i$  prefers all members of  $f(a_i)$  to  $s(a_i)$ . It turns out that Lemmas 4.2.2 and 4.2.3 also extend to CHAT as established by the following lemmas.

**Lemma 5.5.** Let M be a popular matching in I. Then no agent  $a_i \in A$  can be assigned in M to a house between  $f(a_i)$  and  $s(a_i)$  on  $a_i$ 's preference list.

**Proof.** Suppose that  $a_i$  is assigned to a house  $h_j$  strictly between  $f(a_i)$  and  $s(a_i)$ . Then,  $a_i$  must prefer  $h_j$  to all houses in  $s(a_i)$ . Hence,  $h_j$  must be an odd or unreachable house in  $G_1$ , as  $s(a_i)$  contains the highest-ranking even houses in  $G_1$  in  $a_i$ 's preference list. By Corollary 4.3.1,  $M(h_j) \subseteq f(h_j)$ . However, this is a contradiction as  $h_j \notin f(a_i)$ .

**Lemma 5.6.** Let M be a popular matching in I. Then no agent  $a_i \in A$  can be assigned in M to a house worse than  $s(a_i)$  on  $a_i$ 's preference list.

**Proof.** Suppose that  $a_i$  is assigned to a house worse than  $s(a_i)$ . Let  $h_j$  be any house in  $s(a_i)$ . Now, if  $|M(h_j)| < c_j$ , we can promote  $a_i$  to  $h_j$  to obtain a more popular matching.

Hence, suppose that  $|M(h_j)| = c_j$ . Let  $a_k \in M(h_j)$ . We consider two cases for  $h_j$ 

**Case (i):**  $h_j \notin f(a_k)$ . We then choose any  $h_l \in f(a_k)$ . If  $|M(h_l)| < c_l$ , we promote  $a_i$  to  $h_j$  and promote  $a_k$  to  $h_l$  to obtain a more popular matching than M. Otherwise,  $|M(h_l)| = c_l$  so we let  $a_m \in M(h_l)$ . If  $a_m = a_i$ , we can promote  $a_i$  to  $h_j$  and promote  $a_k$  to  $h_l$ . If  $a_m \neq a_i$ , we promote  $a_i$  to  $h_j$  promote  $a_k$  to  $h_l$  and demote  $a_m$  to  $l(a_m)$  to obtain a more popular matching than M.

Case (ii):  $h_j \in f(a_k)$ . As  $h_j \in s(a_i)$ ,  $h_j$  must be an even vertex by our definition of an s-house. Let  $G_1$  be the first-choice graph of I as previously defined. Let  $M_1 = M$  $\cap E_1$ .

Then  $M_1$  is a maximum matching in  $G_1$  by Lemma 3.3.1. Furthermore, there exists an alternating path P of even length to  $h_j$  in  $G_1$ , with respect to  $M_1$ , from some (even) house  $h_1$ , which is under subscribed in  $M_1$ . Let  $M^1 = M \setminus \{(a_i, M(a_i))\}$ . We consider the subcases that (a)  $h_1$  is under subscribed in  $M^1$  or (b)  $h_1$  is full in  $M^1$ . In subcase (a), we can reuse the proof of Case (i) in Lemma 3.3.1 to obtain a matching  $M^{11}$  by matching  $a_i$  with  $h_j$ , and then matching all other agents in P with one of their other first-choice houses in P by augmenting along P. It follows that  $M^{00}$  is more popular than M, a contradiction. In subcase (b), we can always find an agent  $a_m \in M^1(h_1) \setminus f(h_1)$ . The remainder of our proof, then follows a similar argument to that used in Case (ii) of Lemma 3.3.1 where we can obtain a matching  $M^{00}$  that is more popular than M, a contradiction.

As was the case with CHA, we can also define a subgraph  $G^1$  for the CHAT instance I by this time letting  $G^0$  contain only edges from each agent  $a_i$  to houses in  $f(a_i) \cup s(a_i)$ .

Now, all popular matchings must be agent-complete in G<sup>1</sup> in view of last resort

houses. However, an agent-complete matching need not exist if  $s(a_i) \neq \{l(a_i)\}$  for some agent  $a_i$ . Lemmas 4.4.1, 4.4.5 and 4.4.6 give rise to the following characterization of popular matchings in I.

**Theorem 5.1.** A matching M is popular in I if and only if

1.  $M \cap E_1$  is a maximum matching in  $G_1$ , and

2. M is an agent-complete matching in the subgraph  $G^1$ .

**Proof.** By Lemmas 4.4.1, 4.4.5 and 4.4.6, any popular matching necessarily satisfies Conditions 1 and 2. We now show that these conditions are sufficient.

Let M be any matching satisfying Conditions 1 and 2. Suppose, for a contradiction that  $M^1$  is a matching that is more popular than M. Let  $a_i$  be any agent that prefers  $M^1$  to M. Since  $a_i$  prefers  $M^1(a_i)$  to  $M(a_i)$ , M is an agent-complete matching in  $G^1$ , and  $G^1$  only contains edges from  $a_i$  to  $f(a_i) \cup s(a_i)$ , it follows that  $M(a_i) \in s(a_i)$ , and  $f(a_i)$  and  $s(a_i)$  are disjoint. Hence,  $M^1(a_i)$  must be an odd or unreachable house in  $G_1$ , as  $M(a_i)$  is an even house of highest rank in  $a_i$ 's preference list.

Let  $h_{j1} = M^1(a_i)$ . Since  $h_{j1}$  is odd or unreachable, it follows by Condition 1 and Lemma 3.3.4(b) that  $|M(h_{j1})| = c_{j1}$  and  $M(h_{j1}) \subseteq f(h_{j1})$ . Now since  $a_i \in M^1(h_{j1}) \setminus M(h_{j1})$ , there exists a distinct agent  $a_{k1} \in M(h_{j1}) \setminus M^1(h_{j1})$ . If  $a_{k1}$  is unassigned in  $M^1$  or  $M^1(a_{k1}) \in /f(a_{k1})$ , then  $a_{k1}$  prefers M to M<sup>1</sup>. Otherwise, suppose  $M^1(a_{k1}) \in f(a_{k1})$ . Let  $h_{j2} = M^1(a_{k1})$ . It follows that  $a_{k1}$  is even or unreachable so that  $h_{j2}$  must be odd or unreachable. It then follows by Condition 1 and Lemma 4.4.4(b) that  $|M(h_{j2})| = c_{j2}$ and  $M(h_{j2}) \subseteq f(h_{j2})$ . Hence, there exists an agent  $a_{k2} \neq a_{k1}$  such that  $a_{k2} \in M(h_{j2})$  $|M^1(h_{j2})$  and  $h_{j2} \in f(a_{k2})$ . If  $a_{k2}$  is unassigned in  $M^0$  or  $M^0(a_{k2}) \in /f(a_{k2})$ , then  $a_{k2}$ prefers M to M<sup>1</sup>.

Otherwise, suppose that  $M^1(a_{k2}) \in f(a_{k2})$ . Let  $h_{j3} = M^1(a_{k2})$ . Then there exists

an agent  $a_{k3} \in M(h_{j3}) \setminus M^1(h_{j3})$  by a similar argument for  $a_{k2}$ . Note that possibly  $h_{j3} = h_{j1}$ , but we must be able to choose  $a_{k3} \neq a_{k1}$ , for otherwise  $|M^1(h_{j1})| > |M(h_{j1})|$ , which is a contradiction since  $|M(h_{j1})| = c_{j1}$ . Thus,  $a_{k3}$  is a distinct agent, so that we can repeat the above argument to identify an alternating path P in which houses need not be distinct, but agents are distinct. It follows that P must terminate at some agent  $a_{kr}$  as the number of agents are finite. Furthermore, it must be the case that  $a_{kr}$  is unassigned in  $M^1$  or  $M^1(a_{kr}) \notin f(a_{kr})$  so that for every  $a_i$  that prefers  $M^1$  to M, there must exist a distinct  $a_{kr}$  that prefers M to  $M^1$ .

Finally, we note the uniqueness of  $a_{kr}$ . If there exists another agent  $a^{1}_{i}$  who prefers  $M^{1}$  to M, then we can build another, alternating path – it is possible that some of the houses are those already used in previous alternating paths such as P. However, it must be the case (from our argument that  $a_{k3}$  is a distinct agent) that we are always able to identify distinct agents not already used in previous alternating paths, as each house on the path is odd or unreachable, and thus full in M. Hence, M is popular in I.

#### Finding a popular matching

Theorem 4.4.1 leads to Algorithm Popular-CHAT for finding a popular matching in an instance I of CHAT or reporting that none exists, as shown in Algorithm 7. The next lemma is an important step in establishing the correctness of the algorithm.

**Lemma 5.7.** Algorithm Popular-CHAT constructs a matching M such that  $M \cap E_1$  is a maximum matching of  $G_1$ .

#### Algorithm 7 Algorithm Popular-CHAT

1: Build subgraph  $G_1=(A, H, E_1)$ , where  $E_1=\{(a_i, h_j) : a_i \in A \land h_j \in f(a_i)\}$ .

2: Compute a maximum matching M<sub>1</sub> of first-choice edges in G<sub>1</sub>.

3: Obtain an EOU labelling of G<sub>1</sub>.

- 4: Build subgraph  $G^1 = (A, H, E^1)$ , where  $E^1 = \{(a_i, h_j) : a_i \in A \land h_j \in f(a_i) \cup s(a_i)\}$ .
- 5: Delete all edges in  $G^1$  connecting two odd vertices, or connecting an odd vertex with an un-reachable vertex. (This step does not delete an edge of  $M_1$ .)
- 6: Find a maximum matching M in the reduced graph  $G^1$  by augmenting M<sub>1</sub>.
- 7: if M is not agent-complete in  $G^1$  then
- 8: output "No popular matching exists";

9: else

10: return M as a popular matching in I;

**Proof.** We firstly claim that  $G^1$  does not contain any edges of rank greater than 1 incident to odd vertices and unreachable houses. Now, it follows by our definition of s-houses, for any odd or unreachable house  $h_j \in H$ ,  $h_j \notin s(a_i)$  for any agent  $a_i \in A$ . Thus, there exist only first-choice edges incident to any such  $h_j$ . By Lemma 4.4.4(b), every odd agent  $a_i$  in  $G_1$  can only be assigned in any maximum matching of  $G_1$  to some even house  $h_k$ . Since  $(a_i, h_k)$  is a first-choice edge in  $G_1$  and  $s(a_i)$  defines the highest-ranked even house in  $a_i$ 's preference list, it follows that  $s(a_i) \subseteq f(a_i)$ . Hence, the claim is established.

Hence by the above claim, it follows that the edges removed from  $G^1$  during Step 5 of the algorithm, between two odd vertices or between an odd vertex and an unreachable vertex, are first-choice edges in  $G^1$ . However, by Lemma 4.4.4(c), no maximum matching in  $G_1$  can contain these edges. Thus, no popular matching can contain these edges by Lemma 4.4.1. In particular, no edge of  $M_1$  is deleted by Step 5.

It also follows by Lemma 4.4.4(c) that there cannot exist any (first-choice) edges in  $G_1$  between two even vertices, or between an even and an unreachable vertex. As a result, the only first-choice edges that remain in  $G^1$  after the edge deletions are those edges between (i) odd agents and even houses, (ii) even agents and

odd houses, and (iii) unreachable agents and unreachable houses. Define a secondchoice edge as belonging to the edge set  $\{(a_i, h_j) \in E^1 : h_j \in s(a_i) \land s(a_i) \subseteq f(a_i)\}$ . Then by the above claim, the only second-choice edges that remain in  $G^0$  are those between even agents and even houses, and between unreachable agents and even houses.

The matching M is obtained from  $M_1$  through successive augmentation in Gabow's algorithm. We claim that there does not exist any augmenting path P in which an un- reachable agent  $a_i$  (who is assigned in  $M_1$  to some house in  $f(a_i)$ ) becomes worse off, for suppose otherwise. We trace the path P from the under subscribed house endpoint. Let  $a_i$  be the first unreachable agent to become worse off after we augment along P. Let  $M^b$  be the matching before we augmented along P and let  $M^a$  be the matching obtained from satisfying P. Assume that  $a_i$  is assigned to  $h_{j1}$  in  $M^b$ . Then, it follows that  $h_{j1}$  is unreachable. Furthermore, we can pick an agent  $a_{i1} \neq a_i$  assigned to  $h_{j1}$  in  $M^b$ .

It follows that  $a_{i1}$  must be unreachable because any unreachable house has only incident edges from unreachable agents in G<sup>0</sup> and since any unreachable house does not have any edge of rank greater than 1 incident to it as established above. If  $a_{i1}$  is unassigned in M<sup>b</sup>, then we have finished tracing the path P. However, this gives a contradiction by Lemma 3.3.4(b). Hence,  $a_{i1}$  must be assigned to some first-choice house  $h_{j2}$  in M<sup>b</sup> or else  $a_i$  cannot be the first unreachable agent to become worse off. It thus follows that  $h_{j2}$  must also be unreachable. We can repeat the above argument to trace the path P until we terminate at some agent  $a_{ir}$  who is assigned to the unreachable house  $h_{jr}$  in M<sup>a</sup>. It is evident that  $a_{ir}$  must be unassigned in M<sup>b</sup>. However, any such  $a_{ir}$  must be unreachable, which is a contradiction again by Lemma 4.4.4(b).

Now, since all odd agents have only first-choice edges incident to them in  $G^1$ , they must remain assigned to first-choice houses in M even if they participated in any augmenting paths. Moreover, it must be the case that the odd houses, each of which is

full in  $M_1$ , must be full in M and incident only to first-choice edges in M (since odd houses are incident only to such edges in  $G^1$ ). Finally, by the above paragraph, unreachable agents cannot become worse off in M than in  $M_1$ . Hence, only even agents may become worse off in M than in  $M_1$ , but this means that at least  $|O_A| +$  $\sum_{h=0}^{h} (b_{h-1}(c_{h-1})) = \frac{1}{2} + |U_A|$  first chiose edges assigned previously in  $M_1$ remain assigned in M. It thus follows by lemma 3.3.4 (b) that  $M \cap E_1$  is a maximum matching of  $G_1$ .

Hence, if Algorithm Popular-CHAT returns a matching M, then M is both an agent-complete matching in  $G^1$  and  $M \cap E_1$  is a maximum matching of  $G_1$  by Lemma 4.4.7. Hence M is a popular matching in I by Theorem 4.4.1.

We now consider the complexity of Algorithm Popular-CHAT. Let F be the number of first-choice edges in G. It is straightforward to see that  $G_1$  can be constructed in  $O(F + n_2)$  time. We use Gabow's algorithm [15] to compute a maximum matching M<sub>1</sub> in G<sub>1</sub> in O( $\sqrt{(CF)}$ ) time. We then obtain an EOU labelling of G<sub>1</sub> as follows. We first use a preprocessing step to label each unassigned agent and each under subscribed house as even. This step takes O(n) time. Next, restricted breadth-first search may be used on G<sub>1</sub> to search for alternating paths with respect to M<sub>1</sub>, building up odd or even labels for every vertex encountered. This step labels all odd and even (assigned) agents, and all odd and even (full) houses and takes O(m) time. Any remaining unlabelled vertices must be unreachable and we can directly label these vertices in  $G_1$  in O(n) time. Thus, the total time complexity of this step is O(n + m). The EOU labelling of  $G_1$  is then used to construct  $G^1$  and to delete certain edges from  $G^1$  at Steps 4 and 5 of the algorithm, both of which take O(m) time overall. Finally, we then use Gabow's algorithm again to obtain the maximum matching M in G<sup>1</sup> in O( $\sqrt{\Box}(F + S)$ ) time, where S is the number of second-choice edges in G<sup>1</sup>. The following result gives the overall run-time of Algorithm Popular-CHAT.

**Lemma 5.8.** Given an instance of CHAT, we can find a popular matching, or determine that none exists, in O( $\sqrt{\Box}$ m) time.

#### Finding a maximum popular matching

It now remains to consider the problem of finding a maximum popular matching in I. The aim is to find a matching that satisfies the conditions of Theorem 3.3.1 and that minimizes the number of agents who are assigned to their last resort houses. We begin by firstly using Algorithm Popular-CHAT to compute a popular matching M in I, assuming such a matching exists. Then  $M \cap E_1$  is a maximum matching in G<sub>1</sub>. We remove all edges in G<sup>1</sup> (and thus from M) that are incident to a last resort house. It follows that M still satisfies the property that  $M \cap E_1$  is a maximum matching in G<sub>1</sub>, but M need not be maximum in G<sup>1</sup> if agents become unassigned as a result of the edge removals. Thus, we obtain a new maximum matching M<sup>1</sup> from M by using Gabow's algorithm on G<sup>1</sup> again. If M<sup>1</sup> is not agent-complete in G<sup>1</sup>, we simply assign any agent who remains unassigned in M<sup>1</sup> to their last resort house to obtain an agent-complete matching. Using an argument similar to that in the proof of Lemma 3.3.7, it follows that M<sup>1</sup>  $\cap E_1$  is a maximum matching of G<sub>1</sub>. Thus, M<sup>1</sup> is a maximum popular matching in I. Now, it is straightforward to see that the overall complexity of this approach is as for Algorithm Popular-CHAT, giving the following result.

**Theorem 5.2.** Given an instance of CHAT, we can find a maximum popular matching, or report that no such matching exists, in O(Cm) time.

#### **Cloning versus our direct approach**

We may compare the complexity of our direct approach for CHAT to that obtained using cloning on I together with the algorithm of [4] on the cloned instance of I. As in Section 4.2, the latter approach takes  $\Omega(\sqrt{(n_1 + Cm^1)})$  time, where  $m^1 = \sum_{ai} \in_{A hj} \in_{Ai} c_j$ . It follows that this is slower than our direct algorithm by a factor of  $\Omega(m^1/m)$ .

### CONCLUSION

We make some conclusions here on the results of this in general, in addition to those specific to each of the optimality criteria studied as above.

- 1. All the problems that were studied in this thesis turned out to be solvable in polynomial time. The existence of a polynomial time algorithm is often inherently associated with establishing some kind of underlying structure for the problem concerned. For instance, the solution to finding a maximum Pareto optimal matching requires the identifying and then satisfying of certain types of coalitions in the underlying graph of the problem instance. To find a maximum popular matching, the identification of the f- and s-partners of each participating agent in the problem instance, allows us to restrict our attention to only a subgraph of the underlying graph to generate an efficient solution to the problem. In addition, the fastest algorithm for finding a rank-maximal matching is reliant on utilizing the Edmonds-Gallai Decomposition to label vertices and then identify only those edges that can belong to any solution.
- 2. The optimality criteria studied in this thesis can be considered to be superficially similar in some respects, but a simple change to the problem definition often requires a significant change to the algorithm.
- 3. The problem of finding a matching of maximum cardinality in the underlying graph of the problem instance, often seems to influence the time complexity of the resulting algorithm for its solution. This is the case even if the underlying problem seems on the surface not to be associated with maximum matchings, e.g. popular matchings.

#### References

- 1. Abdulkadiroglu and T. Sonmez. House allocation with existing tenants. Journal of Economic Theory, 88:233–260, 1999.
- Abdulkadiroglu and T. Sonmez. Random serial dictatorship and the core from random endowments in house allocation problems. Econometrica, 66(3)(3):689–701, 1998.
- Hylland and R. Zeckhauser. The efficient allocation of individuals to positions. Journal of Political Economy, 87:293–314, 1979.
- Itai, M. Rodeh, and S.L. Tanimoto. Some matching problems for bipartite graphs. Journal of the Association for Computer Machinery, 25(4):517–525, 1978.
- Subramanian. A new approach to stable matching problems. SIAM Journal of Computing, 23(4):671–700, 1994.
- 6. A.E. Roth and A. Postlewaite. Weak versus strong domination in a market with indivisible goods. Journal of Mathematical Economics, 4:131–137, 1977.
- A.E. Roth and M.A.O. Sotomayor. Two-Sided Matching A Study in Game-Theoretic Modeling and Analysis. Cambridge University Press, 1990.
- 8. A.E. Roth, T. Sonmez, and M. Utku Unver. A kidney exchange clearinghouse in New England. The American Economic Review, 95(2):376–380, 2005.
- A.E. Roth, T. Sonmez, and M. Utku Unver. efficient kidney exchange: Coincidence of wants in a structured market. Unpublished manuscript, May 2005.
- 10. A.E. Roth, T. Sonmez, and M. Utku Unver. Kidney exchange. Quarterly Journal
- 11. A.E. Roth, T. Sonmez, and M. Utku Unver. Pairwise kidney exchange. Journal

of Economic Theory, 125(2):151–188, 2005.

- 12. A.E. Roth. On the allocation of residents to rural hospitals: a general property of two-sided matching markets. Econometrica, 54:425–427, 1986.
- 13. A.E. Roth. The evolution of the labor market for medical interns and residents: a case study in game theory. Journal of Political Economy, 92:991–1016, 1984.
- Berge. Two theorems in graph theory. Proceedings of the National Academy of Sciences of the USA, 43:842–844, 1957.
- 15.Ng and D.S. Hirshberg. Three-dimensional stable matching problems. SIAM Journal on Discrete Mathematics, 4:245–252, 1991.
- 16. C.H. Papadimitriou and K. Steiglitz. Combinatorial Optimization Algorithms and Complexity. Prentice Hall, 1982.
- 17.C-P. Teo and J. Sethuraman. The geometry of fractional stable matching's and its applications. Mathematics of Operational Research, 23(4):874–891, 1998.
- 18.C-P. Teo, J. Sethuraman, and W-P. Tan. Gale-Shapley Stable Marriage Problem Revisited: Strategic Issues and Applications. Management Science, 47(9):1252–1267, 2001.
- 19. Gale and L.S. Shapley. College admissions and the stability of marriage. American Mathematical Monthly, 69:9–15, 1962.
- 20. Gale and M. Sotomayor. Some remarks on the stable marriage problem. Discrete Applied Mathematics, 11:223–232, 1985.
- 21.D. Gale. Optimal assignments in an ordered setan application of matroid theory. Journal of Combinatorial Theory, 4:176–180, 1968.
- 22. D. Gusfield and R.W. Irving. The Stable Marriage Problem Structure and Algorithms. The MIT Press, 1989.
- 23.D. McVitie and L.B. Wilson. The stable marriage problem. Communications of the ACM, 14:486–490, 1971.
- 24. D.E. Knuth. Mariages Stables. Les Presses de LUniversit de Montreal, 1976.
- 25. D.F. Manlove, R.W. Irving, K. Iwama, S. Miyazaki, and Y. Morita. Hard variants of stable marriage. Theoretical Computer Science, 276(1-2):261–279,

2002.

- 26. D.F. Manlove. Personal communication, 2007.
- 27. D.J. Abraham and D.F. Manlove. Pareto optimality in the roommates problem. Technical Report TR-2004-182, University of Glasgow, Department of Computing Science, 2004.
- 28. D.J. Abraham and T. Kavitha. Dynamic matching markets and voting paths. In Proceedings of SWAT 2006: the 10th Scandinavian Workshop on Algorithm Theory, volume 4059 of Lecture Notes in Computer Science, pages 65–76. Springer-Verlag, 2006.
- 29. D.J. Abraham, A. Levavi, D.F. Manlove, and G. O'Malley. The stable roommates problem with globally-ranked pairs. In Proceedings of WINE 2007: the 3rd Interna-tional Workshop on Internet and Network Economics, volume 4858 of Lecture Notes in Computer Science, pages 431–444. Springer-Verlag, 2007.
- 30. D.J. Abraham, K. Cechlarova, D.F. Manlove, and K. Mehlhorn. Pareto optimality in house allocation problems. In Proceedings of ISAAC 2004: the 15th Annual International Symposium on Algorithms and Computation, volume 3341 of Lecture Notes in Computer Science, pages 3–15. Springer-Verlag, 2004.
- 31. D.J. Abraham, R.W. Irving, T. Kavitha, and K. Mehlhorn. Popular matchings. SIAM Journal on Computing, 37:1030–1045, 2007.
- 32. Diamantoudi, E. Miyagawa, and L. Xue. Random paths to stability in the room-mate problem. Games and Economic Behavior, 48:18–28, 2004.
- 33.E. Ronn. NP-complete stable matching problems. Journal of Algorithms, 11:285–304, 1990.
- 34. Brassard and P. Bratley. Fundamentals of Algorithmics. Prentice-Hall, 1996.
- 35. H.N. Gabow. An effcient reduction technique for degree-constrained subgraph and bidirected network flow problems. In Proceedings of STOC'83: the 15th Annual ACM Symposium on Theory of Computing, pages 448–456. ACM,

1983.

- 36. J. Mestre. Weighted popular matchings. In Proceedings of ICALP 2006: the 33rd International Colloquium on Automata, Languages and Programming, volume 4051 of Lecture Notes in Computer Science, pages 715–726. Springer-Verlag, 2006.
- 37.J.E. Hopcroft and R.M. Karp. An n<sup>5/2</sup> algorithm for maximum matchings in bipartite graphs. SIAM Journal on Computing, 2:225–231, 1973.
- 38.J.J.M. Tan. A necessary and sufficient condition for the existence of a complete stable matching. Journal of Algorithms, 12:154–178, 1991.
- 39. K. Mehlhorn and D. Michail. Network problems with non-polynomial weights and applications. Unpublished manuscript, 2006.
- 40. K.S. Chung. On the existence of stable roommate matchings. Games and Economic Behavior, 33:206–230, 2000.
- 41.L. Lovasz and M.D. Plummer. Matching Theory. Number 29 in Annals of Discrete Mathematics. North-Holland, 1986.
- 42. L. Shapley and H. Scarf. On cores and indivisibility. Journal of Mathematical Eco-
- 43. M. de Klerk, K. Keizer, W. Weimar, A.J. Goody, A. Spital, and F.L. Delmonico. Donor exchange for renal transplantation. The New England Journal of Medicine, 351:935–937, 2004.
- 44. M. Fredman and R.E. Tarjan. Fibonacci heaps and improved network optimization algorithms. Journal of the ACM, 34:596–615, 1987.
- 45. M. Lucan, P. Rotariu, D. Neculoiu, and G. Iacob. Kidney exchange program: a viable alternative in countries with low rate of cadaver harvesting. Transplantation Proceedings, 35:933–934, 2003.
- 46. M. Mahdian. Random popular matchings. In Proceedings of EC '06: the 7th ACM Conference on Electronic Commerce, pages 238–242. ACM, 2006.
- 47. M.R. Garey and D.S. Johnson. Computers and Intractability A Guide to the Theory of NP-Completeness. W.H. Freeman and Company, 1979.

- 48. of Economics, 119(2):457–488, May 2004.
- 49. omics, 1:23–37, 1974.
- 50. P. Gardenfors. Assignment problem based on ordinal preferences. Management Sci-ence, 20:331–340, 1973.
- P. Gardenfors. Match making: assignments based on bilateral preferences. Behavioural Science, 20:166–173, 1975.
- 52. R.M. McCutchen. The least-unpopularity-factor and least-unpopularity-margin crite-ria for matching problems with one-sided preferences. In Proceedings of LATIN 2008: the 8th Latin-American Theoretical Informatics symposium, to appear, Lecture Notes in Computer Science. Springer, 2008.
- 53.R.W. Irving and D.F. Manlove. Some preliminary observations on popular matchings in the roommates problem. Unpublished manuscript, 2005.
- 54. R.W. Irving and D.F. Manlove. The stable roommates problem with ties. Journal of Algorithms, 43:85–105, 2002.
- 55. R.W. Irving, T. Kavitha, K. Melhorn, D. Michail, and K. Paluch. Rankmaximal matchings. ACM Transactions on Algorithms, 2(4):602–610, 2006.
- 56. R.W. Irving. An efficient algorithm for the stable roommates problem. Journal of Algorithms, 6:577–595, 1985.
- 57. R.W. Irving. Greedy and generous matchings via a variant of the Bellman-Ford Algorithm. Unpublished manuscript, 2007.
- 58. R.W. Irving. Matching medical students to pairs of hospitals: A new variation on a well-known theme. Proceedings of ESA'98: the 6th Annual European Symposium on Algorithms., 6:381–392, 1998.
- 59. R.W. Irving. Stable marriage and indifference. Discrete Applied Mathematics, 48:261–272, 1994.
- 60. S. Micali and V.V. Vazirani. An O( |V |·|E|) algorithm for finding maximum match-ing in general graphs. In Proceedings of FOCS '80: the 21st Annual IEEE Symposium in Foundations of Computer Science, pages 17–27. IEEE Computer Society, 1980.

- 61. T. Feder. A new fixed point approach for stable networks and stable marriages. Journal of Computer and System Sciences, 45:233–284, 1992.
- 62. T. Kavitha and C. Shah. Efficient algorithms for weighted rank-maximal matchings and related problems. In Proceedings of ISAAC 2006: the Seventeenth International Symposium on Algorithms and Computation, volume 4288 of Lecture Notes in Com-puter Science, pages 153–162. Springer, 2006.
- 63. T. Uno. Algorithms for enumerating all perfect, maximum and maximal matchings in bipartite graphs. In Proceedings of ISAAC 1997: the 8th Annual International Symposium on Algorithms and Computation, volume 1350 of Lecture Notes in Computer Science, pages 92–101. Springer-Verlag, 1997.
- 64. W.R. Pulleyblank. Matchings and extensions. In R.L. Graham, M. Grotschel, and L. Lovasz, editors, Handbook of Combinatorics, volume 1, chapter 3, pages 179–232.North-Holland, 1995.

# **FUZZY ANALYSIS**

# EDITED BY DR.A.USHA



# **Fuzzy Analysis**

Copyright © 2021 by Dr.A.Usha

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-889-3

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

Page Numbers
01-08
09-19
20-26
27-34
35-46

# **Chapter-I**

## **FUZZY ANALYSIS**

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Fuzzy analysis, also known as fuzzy logic or fuzzy set theory, is a mathematical approach to deal with uncertainty, imprecision, and vagueness in data. It was introduced by Lotfi A. Zadeh in 1965.

#### Key concepts:

- 1. **Fuzzy sets**: A fuzzy set is a set with fuzzy boundaries, where elements have degrees of membership rather than being either fully included or excluded.
- 2. **Membership functions**: These functions assign a value between 0 and 1 to each element, representing its degree of membership in a fuzzy set.
- 3. **Fuzzy operators**: These operators (e.g., AND, OR, NOT) are used to combine fuzzy sets and perform logical operations.
- 4. **Defuzzification**: The process of converting fuzzy output into a crisp value.

#### **Applications:**

- 1. **Control systems**: Fuzzy logic is used in control systems to handle imprecise or uncertain inputs.
- 2. Pattern recognition: Fuzzy analysis is applied in image and speech recognition.
- 3. **Decision-making**: Fuzzy logic helps in decision-making under uncertainty.
- 4. **Data mining**: Fuzzy analysis is used in data mining to handle uncertain or noisy data.

1
5. Artificial intelligence: Fuzzy logic is used in AI to represent uncertain knowledge.

### **Advantages:**

- 1. Handles uncertainty and imprecision
- 2. Provides flexible and robust solutions
- 3. Easy to understand and implement
- 4. Can handle non-linear relationships

## **Fuzzy Analysis Techniques:**

- 1. Fuzzy Inference Systems (FIS): Use if-then rules to map inputs to outputs.
- 2. Adaptive Neuro-Fuzzy Inference Systems (ANFIS): Combine neural networks and fuzzy logic.
- 3. Fuzzy Clustering: Group similar data points into clusters with fuzzy boundaries.
- 4. Fuzzy Neural Networks: Combine fuzzy logic and neural networks.
- 5. Fuzzy Decision Trees: Use fuzzy logic in decision trees.
- 6. Fuzzy Regression: Model relationships between variables with fuzzy coefficients.
- 7. Fuzzy Time Series Analysis: Analyze time series data with fuzzy logic.

## **Fuzzy Analysis Applications:**

- 1. **Image Processing**: Edge detection, image segmentation, and object recognition.
- 2. **Natural Language Processing (NLP)**: Sentiment analysis, text classification, and language modeling.
- 3. Control Systems: Temperature control, traffic management, and robotics.
- 4. Data Mining: Customer segmentation, market analysis, and trend analysis.
- 5. Expert Systems: Medical diagnosis, financial forecasting, and decision support.
- 6. **Robotics**: Navigation, obstacle avoidance, and human-robot interaction.

7. **Financial Analysis**: Stock market prediction, risk assessment, and portfolio optimization.

## **Fuzzy Analysis Tools:**

- 1. MATLAB: Fuzzy Logic Toolbox and Simulink.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy, and PyFuzzy.
- 3. **R**: fuzzyR, frbs, and fuzzyToolkit.
- 4. Java: JFuzzy, FuzzyJ, and JavaFuzzy.
- 5. Excel: Fuzzy add-ins and spreadsheets.

## **Fuzzy Analysis Challenges:**

- 1. Interpretability: Understanding fuzzy models and results.
- 2. Computational Complexity: High computational requirements.
- 3. Data Quality: Handling noisy or incomplete data.
- 4. **Scalability**: Applying fuzzy analysis to large datasets.
- 5. Validation: Verifying fuzzy model accuracy.

There are several types of fuzzy analysis, including:

**1. Fuzzy Set Analysis**: Deals with fuzzy sets, membership functions, and fuzzy operators.

**2. Fuzzy Logic Analysis**: Uses fuzzy logic to reason with uncertain or imprecise information.

**3. Fuzzy Inference Analysis**: Uses fuzzy inference systems (FIS) to map inputs to outputs.

**4. Fuzzy Clustering Analysis**: Groups similar data points into clusters with fuzzy boundaries.

**5. Fuzzy Classification Analysis**: Assigns data points to classes with fuzzy membership degrees.

**6. Fuzzy Regression Analysis**: Models relationships between variables with fuzzy coefficients.

7. Fuzzy Time Series Analysis: Analyzes time series data with fuzzy logic.

8. Fuzzy Decision Analysis: Supports decision-making under uncertainty.

9. Fuzzy Optimization Analysis: Finds optimal solutions with fuzzy objective functions.

10. Fuzzy Neural Network Analysis: Combines fuzzy logic and neural networks.

**Specialized Types:** 

- 1. Intuitionistic Fuzzy Analysis: Deals with intuitionistic fuzzy sets.
- 2. Type-2 Fuzzy Analysis: Handles uncertainty in fuzzy membership functions.
- 3. Fuzzy-Rough Set Analysis: Combines fuzzy sets and rough sets.
- 4. Fuzzy-Bayesian Analysis: Integrates fuzzy logic and Bayesian inference.
- 5. **Fuzzy-Evolutionary Analysis**: Combines fuzzy logic and evolutionary algorithms.

**Application-Specific Types:** 

- 1. Fuzzy Control Analysis: Used in control systems.
- 2. Fuzzy Image Analysis: Applied in image processing.
- 3. Fuzzy Natural Language Analysis: Used in natural language processing.
- 4. Fuzzy Financial Analysis: Applied in finance and economics.
- 5. Fuzzy Risk Analysis: Used in risk assessment and management.

Hybrid Approaches:

- 1. Neuro-Fuzzy Analysis: Combines neural networks and fuzzy logic.
- 2. Fuzzy-Genetic Analysis: Combines fuzzy logic and genetic algorithms.
- Fuzzy-Support Vector Machine (SVM) Analysis: Combines fuzzy logic and SVM.

## **Fuzzy Sets**

A fuzzy set is a set with fuzzy boundaries, where elements have degrees of membership rather than being either fully included or excluded.

## **Mathematical Representation**

A fuzzy set A is represented as:

$$A = \{ (x, \mu A(x)) \mid x \in X \}$$

where:

- x is an element in the universe of discourse X
- $\mu A(x)$  is the membership function of x in A
- $\mu A(x) \in [0, 1]$  represents the degree of membership of x in A

## **Membership Functions**

Membership functions assign a value between 0 and 1 to each element, representing its degree of membership in a fuzzy set.

## **Types of Membership Functions**

1. Triangular Membership Function

 $\mu A(x) = 1 - |(x - a)/b|$ 

## 2. Gaussian Membership Function

 $\mu A(x) = e^{(-(x-a)^2 / 2b^2)}$ 

## 3. Sigmoid Membership Function

 $\mu A(x) = 1 / (1 + e^{(-(x-a)/b)})$ 

## 4. Trapezoidal Membership Function

 $\mu A(x) = 1 - \max(0, (x-a)/b)$ 

## **Fuzzy Operators**

Fuzzy operators are used to combine fuzzy sets and perform logical operations.

# **Types of Fuzzy Operators**

1. AND (🖸)

 $\mu A \odot B(x) = \min(\mu A(x), \mu B(x))$ 

2. **OR**(U)

 $\mu A \cup B(x) = max(\mu A(x), \mu B(x))$ 

3. **NOT** (¬)

 $\mu \neg A(x) = 1 - \mu A(x)$ 

# **Fuzzy Inference Systems (FIS)**

FIS use if-then rules to map inputs to outputs.

## **Components of FIS**

1. Fuzzifier

Converts crisp input data into fuzzy sets.

# 2. Inference Engine

Applies fuzzy operators and rules to generate fuzzy output.

# 3. Defuzzifier

Converts fuzzy output into a crisp value.

# **Fuzzy Clustering**

Fuzzy clustering groups similar data points into clusters with fuzzy boundaries.

# **Types of Fuzzy Clustering**

- 1. Fuzzy C-Means (FCM)
- 2. Fuzzy K-Means

# **Fuzzy Neural Networks**

Fuzzy neural networks combine fuzzy logic and neural networks.

# **Types of Fuzzy Neural Networks**

- 1. Adaptive Neuro-Fuzzy Inference System (ANFIS)
- 2. Fuzzy Backpropagation

# **Applications of Fuzzy Analysis**

- 1. Control Systems
- 2. Image Processing
- 3. Natural Language Processing

- 4. Decision-Making
- 5. Data Mining
- 6. Artificial Intelligence
- 7. Robotics
- 8. Financial Analysis
- 9. Medical Diagnosis
- 10. Expert Systems

Software Tools for Fuzzy Analysis

- 1. MATLAB
- 2. Python
- 3. **R**
- 4. **Java**
- 5. Excel

**Challenges and Limitations** 

- 1. Interpretability
- 2. Computational Complexity
- 3. Data Quality
- 4. Scalability
- 5. Validation

# **Chapter-II**

# **OPERATION OF FUZZY SETS**

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

### **Fuzzy Set Operations**

1. **Union** (U)

 $\mu A \cup B(x) = max(\mu A(x), \mu B(x))$ 

2. Intersection ( $\cap$ )

 $\mu A \cap B(x) = \min(\mu A(x), \mu B(x))$ 

3. Complement  $(\neg)$ 

 $\mu \neg A(x) = 1 - \mu A(x)$ 

4. Difference (-)

 $\mu A$ - $B(x) = \mu A \cap \neg B(x)$ 

5. Symmetric Difference  $(\bigoplus)$ 

 $\mu A \bigoplus B(x) = \mu A \cup B(x) - \mu A \cap B(x)$ 

## **Fuzzy Set Properties**

## 1. **Reflexivity**

 $\mu A \cup A(x) = \mu A(x)$ 

# 2. Symmetry

 $\mu A \cup B(x) = \mu B \cup A(x)$ 

### 3. Transitivity

 $\mu AU(BUC)(x) = (AUB)UC(x)$ 

# 4. Associativity

 $\mu A \cap (B \cap C)(x) = (A \cap B) \cap C(x)$ 

# **Fuzzy Set Operations Examples**

Suppose we have two fuzzy sets:

$$A = \{ (x, 0.8) \mid x \in X \}$$

 $B = \{ (x, 0.6) \mid x \in X \}$ 

# Union (U)

 $\mu A \cup B(x) = max(0.8, 0.6) = 0.8$ 

# **Intersection** $(\cap)$

 $\mu A \cap B(x) = \min(0.8, 0.6) = 0.6$ 

# Complement (¬)

 $\mu \neg A(x) = 1 - 0.8 = 0.2$ 

# **Difference** (-)

 $\mu$ A-B(x) =  $\mu$ A $\cap \neg$ B(x) = min(0.8, 1-0.6) = min(0.8, 0.4) = 0.4

## Symmetric Difference $(\bigoplus)$

 $\mu A \bigoplus B(x) = \mu A \cup B(x) - \mu A \cap B(x) = 0.8 - 0.6 = 0.2$ 

### **Fuzzy Set Operations Applications**

- 1. Decision-Making
- 2. Pattern Recognition
- 3. Image Processing
- 4. Natural Language Processing
- 5. Control Systems
- 6. Data Mining
- 7. Artificial Intelligence

### **Decision-Making Systems**

Decision-making systems are critical in various fields, including business, healthcare, finance, and transportation. Fuzzy logic plays a significant role in these systems, enabling them to handle uncertainty and imprecision.

## **Types of Decision-Making Systems**

- 1. Expert Systems: Mimic human expertise to make decisions.
- Decision Support Systems (DSS): Provide data-driven insights for decisionmaking.
- 3. Fuzzy Decision Support Systems (FDSS): Integrate fuzzy logic into DSS.
- 4. **Hybrid Decision-Making Systems**: Combine multiple techniques (e.g., fuzzy, neural, and genetic algorithms).

## **Fuzzy Logic in Decision-Making Systems**

- 1. Fuzzy Inference Systems: Use if-then rules to make decisions.
- 2. Fuzzy Rule-Based Systems: Employ fuzzy rules to reason with uncertain data.
- 3. Fuzzy Optimization: Find optimal solutions with fuzzy objective functions.
- 4. Fuzzy Classification: Assign fuzzy membership degrees to classes.

## **Applications of Fuzzy Decision-Making Systems**

- 1. Medical Diagnosis: Fuzzy expert systems for disease diagnosis.
- 2. Financial Forecasting: Fuzzy models for stock market prediction.
- 3. Traffic Management: Fuzzy control systems for traffic optimization.
- 4. Supply Chain Management: Fuzzy decision-making for inventory control.
- 5. Robotics and Autonomous Systems: Fuzzy logic for navigation and control.

# **Benefits of Fuzzy Decision-Making Systems**

- 1. Handling Uncertainty: Fuzzy logic handles uncertain and imprecise data.
- 2. **Improved Accuracy**: Fuzzy models can outperform traditional models.
- 3. Flexibility: Fuzzy systems adapt to changing conditions.
- 4. **Interpretability**: Fuzzy rules provide insight into decision-making processes.

# **Tools and Techniques for Fuzzy Decision-Making Systems**

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: fuzzyR, frbs.
- 4. Java: JFuzzy, FuzzyJ.
- 5. Excel: Fuzzy add-ins.

# **Challenges and Future Directions**

- 1. Interpretability: Understanding fuzzy models and results.
- 2. Scalability: Applying fuzzy decision-making to large datasets.

- 3. **Integration**: Combining fuzzy logic with other techniques.
- 4. Validation: Verifying fuzzy model accuracy.

# Pattern Recognition

Pattern recognition is a fundamental aspect of artificial intelligence, machine learning, and data analysis. Fuzzy logic plays a significant role in pattern recognition, enabling systems to handle uncertainty and imprecision.

# **Types of Pattern Recognition**

- 1. **Supervised Learning**: Learning from labeled data.
- 2. Unsupervised Learning: Discovering patterns in unlabeled data.
- 3. Semi-Supervised Learning: Combining labeled and unlabeled data.

# **Fuzzy Logic in Pattern Recognition**

- 1. **Fuzzy Clustering**: Grouping similar data points into fuzzy clusters.
- 2. Fuzzy Classification: Assigning fuzzy membership degrees to classes.
- 3. **Fuzzy Regression**: Modeling relationships between variables with fuzzy coefficients.
- 4. Fuzzy Neural Networks: Combining fuzzy logic and neural networks.

# **Applications of Fuzzy Pattern Recognition**

- 1. **Image Processing**: Fuzzy edge detection, segmentation, and recognition.
- 2. Speech Recognition: Fuzzy Hidden Markov Models (HMMs).
- 3. Natural Language Processing (NLP): Fuzzy sentiment analysis.
- 4. **Biometric Authentication**: Fuzzy fingerprint, face, and iris recognition.
- 5. Medical Diagnosis: Fuzzy classification of diseases.

# **Benefits of Fuzzy Pattern Recognition**

- 1. Handling Uncertainty: Fuzzy logic handles uncertain and imprecise data.
- 2. Improved Accuracy: Fuzzy models can outperform traditional models.
- 3. **Robustness**: Fuzzy systems adapt to noisy or incomplete data.
- 4. Interpretability: Fuzzy rules provide insight into pattern recognition processes.

## **Tools and Techniques for Fuzzy Pattern Recognition**

- 1. MATLAB: Fuzzy Logic Toolbox, Image Processing Toolbox.
- 2. Python: scikit-fuzzy, fuzzywuzzy, OpenCV.
- 3. **R**: fuzzyR, frbs, caret.
- 4. **Java**: JFuzzy, FuzzyJ, Weka.
- 5. TensorFlow: Fuzzy logic integration.

# **Challenges and Future Directions**

- 1. **Scalability**: Applying fuzzy pattern recognition to large datasets.
- 2. **Interpretability**: Understanding fuzzy models and results.
- 3. **Integration**: Combining fuzzy logic with other techniques.
- 4. Validation: Verifying fuzzy model accuracy.

## Image Processing Systems

Image processing systems play a vital role in various industries, including healthcare, security, robotics, and entertainment. Fuzzy logic enhances image processing by handling uncertainty and imprecision.

# **Types of Image Processing Systems**

- 1. Image Enhancement: Improving image quality.
- 2. Image Segmentation: Partitioning images into regions.
- 3. **Object Recognition**: Identifying objects within images.
- 4. Image Compression: Reducing image data size.

### **Fuzzy Logic in Image Processing**

- 1. Fuzzy Edge Detection: Detecting edges with fuzzy operators.
- 2. Fuzzy Segmentation: Segmenting images using fuzzy clustering.
- 3. Fuzzy Thresholding: Thresholding images with fuzzy logic.
- 4. **Fuzzy Feature Extraction**: Extracting features using fuzzy algorithms.

## **Applications of Fuzzy Image Processing**

- 1. Medical Imaging: Fuzzy segmentation of tumors.
- 2. Security Systems: Fuzzy face recognition.
- 3. **Robotics**: Fuzzy vision systems for navigation.
- 4. **Surveillance**: Fuzzy object detection.
- 5. **Quality Inspection**: Fuzzy defect detection.

### **Benefits of Fuzzy Image Processing**

- 1. Handling Uncertainty: Fuzzy logic handles uncertain and imprecise image data.
- 2. Improved Accuracy: Fuzzy algorithms can outperform traditional methods.
- 3. Robustness: Fuzzy systems adapt to noisy or incomplete image data.
- 4. Flexibility: Fuzzy logic integrates with other image processing techniques.

## **Tools and Techniques for Fuzzy Image Processing**

- 1. MATLAB: Image Processing Toolbox, Fuzzy Logic Toolbox.
- 2. **OpenCV**: Fuzzy logic integration.
- 3. **Python**: scikit-image, scikit-fuzzy.
- 4. Java: JFuzzy, FuzzyJ.
- 5. TensorFlow: Fuzzy logic integration.

## **Challenges and Future Directions**

- 1. Computational Complexity: Optimizing fuzzy image processing algorithms.
- 2. Scalability: Applying fuzzy image processing to large images.
- 3. Interpretability: Understanding fuzzy image processing results.
- 4. Validation: Verifying fuzzy image processing accuracy.
- 5.

# Natural Language Processing (NLP)

NLP is a crucial aspect of artificial intelligence, enabling computers to understand, interpret, and generate human language. Fuzzy logic enhances NLP by handling uncertainty and imprecision in language.

# **Types of NLP Tasks**

- 1. Text Classification: Classifying text into categories.
- 2. Sentiment Analysis: Determining sentiment (positive/negative) from text.
- 3. Named Entity Recognition (NER): Identifying entities (names, locations).
- 4. Part-of-Speech (POS) Tagging: Identifying word roles (noun, verb).
- 5. Machine Translation: Translating text from one language to another.

# **Fuzzy Logic in NLP**

- 1. Fuzzy Tokenization: Tokenizing text with fuzzy boundaries.
- 2. **Fuzzy Stemming**: Reducing words to base form with fuzzy rules.
- 3. Fuzzy Sentiment Analysis: Analyzing sentiment with fuzzy logic.
- 4. Fuzzy Named Entity Recognition: Identifying entities with fuzzy matching.
- 5. Fuzzy Machine Translation: Translating text with fuzzy logic.

# **Applications of Fuzzy NLP**

- 1. Sentiment Analysis: Analyzing customer feedback.
- 2. Text Summarization: Summarizing documents with fuzzy logic.

- 3. Question Answering: Answering questions with fuzzy matching.
- 4. Chatbots: Generating human-like responses with fuzzy logic.
- 5. Language Translation: Translating languages with fuzzy machine translation.

# **Benefits of Fuzzy NLP**

- 1. Handling Uncertainty: Fuzzy logic handles uncertain and imprecise language.
- 2. **Improved Accuracy**: Fuzzy algorithms can outperform traditional methods.
- 3. Robustness: Fuzzy systems adapt to noisy or incomplete language data.
- 4. **Flexibility**: Fuzzy logic integrates with other NLP techniques.

# **Tools and Techniques for Fuzzy NLP**

- 1. NLTK: Natural Language Toolkit with fuzzy extensions.
- 2. **spaCy**: Modern NLP library with fuzzy matching.
- 3. **Stanford CoreNLP**: Java library for NLP with fuzzy logic.
- 4. **TensorFlow**: Fuzzy logic integration for NLP tasks.
- 5. **PyTorch**: Fuzzy logic integration for NLP tasks.

# **Control Systems**

Control systems are crucial in various industries, including robotics, aerospace, automotive, and process control. Fuzzy logic enhances control systems by handling uncertainty and imprecision.

# **Types of Control Systems**

- 1. Feedback Control Systems: Using feedback to regulate output.
- 2. Feedforward Control Systems: Using predictive models to regulate output.
- 3. Hybrid Control Systems: Combining feedback and feedforward control.

# **Fuzzy Logic in Control Systems**

- 1. **Fuzzy PID Controllers**: Fuzzy logic-based proportional-integral-derivative controllers.
- 2. Fuzzy State Feedback Control: Fuzzy logic-based state feedback control.
- 3. Fuzzy Model Predictive Control: Fuzzy logic-based model predictive control.

# **Applications of Fuzzy Control Systems**

- 1. **Robotics**: Fuzzy control for robot navigation and manipulation.
- 2. **Process Control**: Fuzzy control for temperature, pressure, and flow control.
- 3. **Automotive**: Fuzzy control for anti-lock braking systems (ABS) and traction control.
- 4. Aerospace: Fuzzy control for flight control and navigation.
- 5. Smart Buildings: Fuzzy control for HVAC and lighting systems.

# **Data Mining**

Data mining is the process of discovering patterns, relationships, and insights from large datasets. Fuzzy logic enhances data mining by handling uncertainty and imprecision.

# **Types of Data Mining**

- 1. **Predictive Modeling** : Predicting outcomes based on historical data.
- 2. **Clustering** : Grouping similar data points into clusters.
- 3. **Classification** : Assigning data points to predefined categories.
- 4. **Regression** : Modeling relationships between variables.

# Artificial Intelligence (AI)

AI encompasses various techniques to create intelligent systems, including:

# **Types of AI**

1. Narrow or Weak AI: Designed for specific tasks (e.g., image recognition).

- 2. **General or Strong AI**: Human-like intelligence, capable of reasoning and learning.
- 3. **Superintelligence**: Significantly surpassing human intelligence.

# **Fuzzy Logic in AI**

- 1. Fuzzy Reasoning: Handling uncertainty in decision-making.
- 2. Fuzzy Inference Systems: Using fuzzy rules for inference.
- 3. Fuzzy Neural Networks: Combining fuzzy logic and neural networks.

# **Chapter-III**

## **FUZZY ARITHMETIC**

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

**Fuzzy** Arithmetic

Fuzzy arithmetic extends traditional arithmetic to handle fuzzy numbers, enabling calculations with uncertain or imprecise values.

### **Fuzzy Numbers**

- 1. Triangular Fuzzy Numbers: Defined by three points (a, b, c).
- 2. Trapezoidal Fuzzy Numbers: Defined by four points (a, b, c, d).
- 3. Gaussian Fuzzy Numbers: Defined by mean and standard deviation.

#### **Fuzzy Arithmetic Operations**

- 1. Fuzzy Addition: +
- 2. Fuzzy Subtraction: -
- 3. **Fuzzy Multiplication**: ×
- 4. Fuzzy Division: ÷

#### **Fuzzy Arithmetic Properties**

- 1. **Commutativity**: a + b = b + a
- 2. **Associativity**: (a + b) + c = a + (b + c)
- 3. **Distributivity**:  $a \times (b + c) = a \times b + a \times c$

### **Applications of Fuzzy Arithmetic**

- 1. Risk Analysis: Fuzzy arithmetic for uncertain risk assessment.
- 2. Financial Modeling: Fuzzy arithmetic for uncertain financial predictions.
- 3. Engineering Design: Fuzzy arithmetic for uncertain design parameters.
- 4. **Decision-Making**: Fuzzy arithmetic for uncertain decision variables.

### **Fuzzy Arithmetic Methods**

- 1. Extension Principle: Extending crisp arithmetic to fuzzy arithmetic.
- 2. Alpha-Cuts: Dividing fuzzy numbers into crisp intervals.
- 3. Fuzzy Interval Arithmetic: Operating on fuzzy intervals.

### **Tools and Techniques for Fuzzy Arithmetic**

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

### **Fuzzy Arithmetic Operations**

- 1. **Fuzzy Addition**: +(e.g., (2, 3, 4) + (3, 4, 5) = (5, 7, 9))
- 2. **Fuzzy Subtraction**: (e.g., (2, 3, 4) (1, 2, 3) = (1, 2, 3))
- 3. **Fuzzy Multiplication**:  $\times$  (e.g., (2, 3, 4)  $\times$  (3, 4, 5) = (6, 12, 20))
- 4. **Fuzzy Division**:  $\div$  (e.g.,  $(2, 3, 4) \div (1, 2, 3) = (0.67, 1.5, 2.33))$

### **Fuzzy Arithmetic Properties**

- 1. **Commutativity**: a + b = b + a
- 2. **Associativity**: (a + b) + c = a + (b + c)
- 3. **Distributivity**:  $a \times (b + c) = a \times b + a \times c$

4. Monotonicity:  $a \le b$  implies  $a + c \le b + c$ 

### **Types of Fuzzy Numbers**

- 1. **Triangular Fuzzy Numbers** (TFN): (a, b, c)
- 2. Trapezoidal Fuzzy Numbers (TrFN): (a, b, c, d)
- 3. Gaussian Fuzzy Numbers (GFN):  $(\mu, \sigma)$
- 4. **Bell-Shaped Fuzzy Numbers** (BSFN): (a, b, c)

## **Fuzzy Arithmetic Methods**

- 1. Extension Principle: Extending crisp arithmetic to fuzzy arithmetic.
- 2. Alpha-Cuts: Dividing fuzzy numbers into crisp intervals.
- 3. Fuzzy Interval Arithmetic: Operating on fuzzy intervals.
- 4. **Fuzzy Ranking**: Ranking fuzzy numbers.

## **Applications of Fuzzy Arithmetic**

- 1. Financial Modeling: Fuzzy arithmetic for uncertain financial predictions.
- 2. **Risk Analysis**: Fuzzy arithmetic for uncertain risk assessment.
- 3. Engineering Design: Fuzzy arithmetic for uncertain design parameters.
- 4. **Decision-Making**: Fuzzy arithmetic for uncertain decision variables.

## **Tools and Techniques for Fuzzy Arithmetic**

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

## **Challenges and Future Directions**

- 1. Computational Complexity: Efficient fuzzy arithmetic algorithms.
- 2. Interpretability: Understanding fuzzy arithmetic results.
- 3. Integration: Combining fuzzy arithmetic with other techniques.

Financial Modeling: Financial Modeling using Fuzzy Arithmetic:

# Applications

- 1. Risk Analysis: Fuzzy arithmetic for uncertain risk assessment.
- 2. Portfolio Optimization: Fuzzy arithmetic for optimal portfolio selection.
- 3. **Option Pricing**: Fuzzy arithmetic for uncertain option values.
- 4. Credit Scoring: Fuzzy arithmetic for uncertain creditworthiness.

# **Fuzzy Financial Models**

- 1. Fuzzy Linear Programming: Fuzzy arithmetic for linear programming.
- 2. Fuzzy Dynamic Programming: Fuzzy arithmetic for dynamic programming.
- 3. Fuzzy Simulation: Fuzzy arithmetic for Monte Carlo simulations.
- 4. **Fuzzy Neural Networks**: Fuzzy arithmetic for neural networks.

# **Fuzzy Financial Instruments**

- 1. Fuzzy Options: Fuzzy arithmetic for option pricing.
- 2. Fuzzy Futures: Fuzzy arithmetic for futures pricing.
- 3. Fuzzy Swaps: Fuzzy arithmetic for swap pricing.
- 4. Fuzzy Stocks: Fuzzy arithmetic for stock pricing.

Risk Analysis using Fuzzy Arithmetic:

# **Types of Risks**

- 1. Financial Risk: Uncertainty in financial outcomes.
- 2. **Operational Risk**: Uncertainty in business operations.

- 3. Strategic Risk: Uncertainty in business strategy.
- 4. **Reputation Risk**: Uncertainty in brand reputation.

### **Fuzzy Risk Analysis Methods**

- 1. Fuzzy Fault Tree Analysis: Fuzzy logic for fault tree analysis.
- 2. Fuzzy Event Tree Analysis: Fuzzy logic for event tree analysis.
- 3. Fuzzy Decision Trees: Fuzzy logic for decision tree analysis.
- 4. Fuzzy Simulation: Fuzzy arithmetic for Monte Carlo simulations.

### **Fuzzy Risk Assessment**

- 1. Fuzzy Probability: Fuzzy logic for probability assessment.
- 2. Fuzzy Impact: Fuzzy logic for impact assessment.
- 3. Fuzzy Risk Priority: Fuzzy logic for risk prioritization.
- 4. Fuzzy Sensitivity Analysis: Fuzzy logic for sensitivity analysis.

Engineering Design using Fuzzy Arithmetic:

### **Engineering Design Applications**

- 1. Mechanical Design: Fuzzy logic for uncertain design parameters.
- 2. Structural Analysis: Fuzzy logic for uncertain load calculations.
- 3. Control Systems: Fuzzy logic for uncertain system dynamics.
- 4. Electrical Design: Fuzzy logic for uncertain electrical parameters.

### **Fuzzy Engineering Design Methods**

- 1. Fuzzy Optimization: Fuzzy logic for optimization problems.
- 2. Fuzzy Simulation: Fuzzy arithmetic for Monte Carlo simulations.
- 3. Fuzzy Taguchi Methods: Fuzzy logic for robust design.
- 4. Fuzzy Reliability Analysis: Fuzzy logic for reliability assessment.

**Fuzzy Design Tools** 

- 1. Computer-Aided Design (CAD): Fuzzy logic integration.
- 2. Finite Element Analysis (FEA): Fuzzy logic for uncertain material properties.
- 3. **Computational Fluid Dynamics (CFD)**: Fuzzy logic for uncertain fluid dynamics.
- 4. Multi-Body Dynamics: Fuzzy logic for uncertain dynamic systems.

# Benefits

- 1. Handling Uncertainty: Fuzzy arithmetic handles uncertain design data.
- 2. **Improved Accuracy**: Fuzzy models can outperform traditional models.
- 3. **Robustness**: Fuzzy systems adapt to changing design conditions.
- 4. **Flexibility**: Fuzzy logic integrates with other design techniques.

# **Types of Decision-Making**

- 1. Multi-Criteria Decision-Making: Fuzzy logic for evaluating multiple criteria.
- 2. Group Decision-Making: Fuzzy logic for consensus-building.
- 3. Uncertain Decision-Making: Fuzzy logic for handling uncertainty.
- 4. **Dynamic Decision-Making**: Fuzzy logic for adapting to changing conditions.

# **Fuzzy Decision-Making Methods**

- 1. **Fuzzy Analytic Hierarchy Process (FAHP)**: Fuzzy logic for hierarchical decision-making.
- 2. Fuzzy TOPSIS: Fuzzy logic for multi-criteria decision-making.
- 3. **Fuzzy VIKOR**: Fuzzy logic for compromise-based decision-making.
- 4. **Fuzzy ELECTRE**: Fuzzy logic for outranking-based decision-making.

### **Fuzzy Decision Support Systems**

- 1. Expert Systems: Fuzzy logic for knowledge-based decision-making.
- 2. Decision Trees: Fuzzy logic for tree-based decision-making.
- 3. Neural Networks: Fuzzy logic for neural network-based decision-making.
- 4. Genetic Algorithms: Fuzzy logic for optimization-based decision-making.

## Benefits

- 1. Handling Uncertainty: Fuzzy arithmetic handles uncertain decision data.
- 2. Improved Accuracy: Fuzzy models can outperform traditional models.
- 3. **Robustness**: Fuzzy systems adapt to changing decision conditions.
- 4. Flexibility: Fuzzy logic integrates with other decision techniques.

# **Chapter-IV**

### **FUZZY RELATIONS**

Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

**Fuzzy Relations:** 

**Definition**: A fuzzy relation is a relation between fuzzy sets, where the membership degrees of the elements are fuzzy.

**Definition**: A fuzzy relation is a fuzzy subset of the Cartesian product of two or more sets.

**Types of Fuzzy Relations:** 

- 1. Fuzzy Binary Relation: Relation between two sets.
- 2. Fuzzy Ternary Relation: Relation between three sets.
- 3. Fuzzy N-ary Relation: Relation between n sets.

### **Properties of Fuzzy Relations:**

- 1. **Reflexivity**:  $\mu(a, a) = 1$
- 2. **Symmetry**:  $\mu(a, b) = \mu(b, a)$
- 3. **Transitivity**:  $\mu(a, c) \ge \min(\mu(a, b), \mu(b, c))$

4. Antisymmetry:  $\mu(a, b) > 0$  implies  $\mu(b, a) = 0$ 

# **Types of Fuzzy Relations**

- 1. **Fuzzy Equivalence Relations**: Fuzzy relations satisfying reflexivity, symmetry, and transitivity.
- 2. **Fuzzy Order Relations**: Fuzzy relations satisfying reflexivity, antisymmetry, and transitivity.
- 3. Fuzzy Similarity Relations: Fuzzy relations satisfying reflexivity, symmetry.

# **Fuzzy Relation Operations**

- 1. **Composition**: Combining fuzzy relations.
- 2. Intersection: Finding common elements between fuzzy relations.
- 3. Union: Combining fuzzy relations.
- 4. **Complement**: Finding elements not in a fuzzy relation.

## **Fuzzy Relation Properties**

- 1. **Reflexivity**: Every element is related to itself.
- 2. **Symmetry**: If a is related to b, then b is related to a.
- 3. **Transitivity**: If a is related to b, and b is related to c, then a is related to c.

## **Applications of Fuzzy Relations**

- 1. **Clustering**: Fuzzy relations for clustering algorithms.
- 2. **Classification**: Fuzzy relations for classification models.
- 3. **Decision-Making**: Fuzzy relations for decision-making systems.
- 4. **Image Processing**: Fuzzy relations for image segmentation.

## **Fuzzy Relation Representation**

- 1. Fuzzy Matrices: Representing fuzzy relations as matrices.
- 2. Fuzzy Graphs: Representing fuzzy relations as graphs.
- 3. Fuzzy Decision Tables: Representing fuzzy relations as decision tables.

# **Fuzzy Relation Algorithms**

- 1. Fuzzy C-Means: Fuzzy clustering algorithm.
- 2. Fuzzy K-Nearest Neighbors: Fuzzy classification algorithm.
- 3. Fuzzy Support Vector Machines: Fuzzy classification algorithm.

## **Tools and Techniques**

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

## **Challenges and Future Directions**

- 1. **Computational Complexity**: Efficient fuzzy relation algorithms.
- 2. Interpretability: Understanding fuzzy relation results.
- 3. Integration: Combining fuzzy relations with other techniques

Fuzzy Equivalence Relations:

**Definition**: A fuzzy equivalence relation is a fuzzy relation that satisfies:

- 1. **Reflexivity**:  $\mu(a, a) = 1$  (every element is related to itself)
- 2. **Symmetry**:  $\mu(a, b) = \mu(b, a)$  (if a is related to b, then b is related to a)
- Transitivity: μ(a, c) ≥ min(μ(a, b), μ(b, c)) (if a is related to b, and b is related to c, then a is related to c)

# **Properties**:

- 1. **Fuzzy Partition**: Fuzzy equivalence relations induce a fuzzy partition of the universe.
- 2. Fuzzy Clustering: Fuzzy equivalence relations can be used for fuzzy clustering.
- 3. **Fuzzy Classification**: Fuzzy equivalence relations can be used for fuzzy classification.

# **Types of Fuzzy Equivalence Relations:**

- 1. **Fuzzy Equality Relations**: Fuzzy equivalence relations with  $\mu(a, b) = 1$  iff a = b.
- 2. **Fuzzy Similarity Relations**: Fuzzy equivalence relations with  $\mu(a, b) = \mu(b, a)$ .
- 3. **Fuzzy Indistinguishability Relations**: Fuzzy equivalence relations with  $\mu(a, b) = 1$  iff a and b are indistinguishable.

# **Applications**:

- 1. **Image Segmentation**: Fuzzy equivalence relations for image segmentation.
- 2. **Data Clustering**: Fuzzy equivalence relations for data clustering.
- 3. **Decision-Making**: Fuzzy equivalence relations for decision-making systems.
- 4. Natural Language Processing: Fuzzy equivalence relations for text analysis.

## **Algorithms**:

- 1. Fuzzy C-Means: Fuzzy clustering algorithm based on fuzzy equivalence relations.
- 2. **Fuzzy K-Means**: Fuzzy clustering algorithm based on fuzzy equivalence relations.
- 3. **Fuzzy Hierarchical Clustering**: Fuzzy clustering algorithm based on fuzzy equivalence relations.

## **Tools and Techniques**:

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.

- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

### **Challenges and Future Directions:**

- 1. Computational Complexity: Efficient algorithms for fuzzy equivalence relations.
- 2. Interpretability: Understanding fuzzy equivalence relation results.
- 3. Integration: Combining fuzzy equivalence relations with other techniques.

Fuzzy Similarity Relations:

**Definition**: A fuzzy similarity relation is a fuzzy relation that satisfies:

- 1. **Reflexivity**:  $\mu(a, a) = 1$  (every element is similar to itself)
- 2. Symmetry:  $\mu(a, b) = \mu(b, a)$  (if a is similar to b, then b is similar to a)
- Transitivity: μ(a, c) ≥ min(μ(a, b), μ(b, c)) (if a is similar to b, and b is similar to c, then a is similar to c)

### **Properties**:

- 1. **Fuzzy Equivalence Classes**: Fuzzy similarity relations induce fuzzy equivalence classes.
- 2. Fuzzy Clustering: Fuzzy similarity relations can be used for fuzzy clustering.
- 3. **Fuzzy Classification**: Fuzzy similarity relations can be used for fuzzy classification.

### **Types of Fuzzy Similarity Relations:**

- 1. **Fuzzy Equality Relations**: Fuzzy similarity relations with  $\mu(a, b) = 1$  iff a = b.
- Fuzzy Proximity Relations: Fuzzy similarity relations with μ(a, b) > 0 iff a is close to b.

3. **Fuzzy Indistinguishability Relations**: Fuzzy similarity relations with  $\mu(a, b) = 1$  iff a and b are indistinguishable.

## **Applications**:

- 1. **Image Segmentation**: Fuzzy similarity relations for image segmentation.
- 2. **Data Clustering**: Fuzzy similarity relations for data clustering.
- 3. Decision-Making: Fuzzy similarity relations for decision-making systems.
- 4. Natural Language Processing: Fuzzy similarity relations for text analysis.

### **Algorithms**:

- 1. Fuzzy C-Means: Fuzzy clustering algorithm based on fuzzy similarity relations.
- 2. Fuzzy K-Means: Fuzzy clustering algorithm based on fuzzy similarity relations.
- 3. **Fuzzy Hierarchical Clustering**: Fuzzy clustering algorithm based on fuzzy similarity relations.

### **Tools and Techniques:**

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

### **Challenges and Future Directions:**

- 1. Computational Complexity: Efficient algorithms for fuzzy similarity relations.
- 2. Interpretability: Understanding fuzzy similarity relation results.
- 3. Integration: Combining fuzzy similarity relations with other techniques.

Fuzzy Order Relations:

**Definition**: A fuzzy order relation is a fuzzy binary relation that satisfies:

- 1. **Reflexivity**:  $\mu(a, a) = 1$
- 2. Antisymmetry:  $\mu(a, b) > 0$  implies  $\mu(b, a) = 0$
- 3. **Transitivity**:  $\mu(a, c) \ge \min(\mu(a, b), \mu(b, c))$

# **Types of Fuzzy Order Relations:**

- 1. Fuzzy Partial Order: Fuzzy order relation without totality.
- 2. Fuzzy Total Order: Fuzzy order relation with totality.
- 3. Fuzzy Weak Order: Fuzzy order relation with weak antisymmetry.
- 4. Fuzzy Strict Order: Fuzzy order relation with strict antisymmetry.

# **Properties of Fuzzy Order Relations:**

- 1. Asymmetry:  $\mu(a, b) > 0$  implies  $\mu(b, a) = 0$
- 2. **Transitivity**:  $\mu(a, c) \ge \min(\mu(a, b), \mu(b, c))$
- 3. **Totality**:  $\mu(a, b) > 0$  or  $\mu(b, a) > 0$

# **Applications of Fuzzy Order Relations:**

- 1. Decision-Making: Fuzzy order relations for decision-making systems.
- 2. Preference Modeling: Fuzzy order relations for preference modeling.
- 3. **Ranking**: Fuzzy order relations for ranking alternatives.
- 4. **Scheduling**: Fuzzy order relations for scheduling tasks.

# Algorithms for Fuzzy Order Relations:

- 1. Fuzzy Topological Sorting: Sorting fuzzy graphs topologically.
- 2. **Fuzzy Minimum Spanning Tree**: Finding minimum spanning trees in fuzzy graphs.
- 3. Fuzzy Shortest Path: Finding shortest paths in fuzzy graphs.

# **Tools and Techniques**:

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

### Chapter- V

### **FUZZY RELATION EQUATIONS**

#### Dr.A.Usha

Assistant. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

**Fuzzy Relation Equations:** 

**Definition**: A fuzzy relation equation is an equation involving fuzzy relations, where the unknown is a fuzzy relation.

#### **Types of Fuzzy Relation Equations:**

- 1. **Fuzzy Relational Equations**: Equations involving fuzzy relations, e.g.,  $R \circ S = T$ .
- 2. **Fuzzy Matrix Equations**: Equations involving fuzzy matrices, e.g.,  $R \times S = T$ .
- 3. **Fuzzy Graph Equations**: Equations involving fuzzy graphs, e.g.,  $R \subseteq S$ .

#### **Fuzzy Relation Equation Properties:**

- 1. **Reflexivity**:  $R \circ R = R$  (idempotence).
- 2. **Symmetry**:  $R \circ S = S \circ R$  (commutativity).
- 3. **Transitivity**:  $R \circ S \circ T = R \circ T \circ S$  (associativity).

#### Methods for Solving Fuzzy Relation Equations:

- 1. **Fuzzy Relational Algebra**: Using fuzzy relational algebra to solve fuzzy relation equations.
- 2. **Fuzzy Matrix Algebra**: Using fuzzy matrix algebra to solve fuzzy matrix equations.

3. Fuzzy Graph Theory: Using fuzzy graph theory to solve fuzzy graph equations.

## **Applications of Fuzzy Relation Equations:**

- 1. Fuzzy Control Systems: Fuzzy relation equations for control systems.
- 2. Fuzzy Decision-Making: Fuzzy relation equations for decision-making systems.
- 3. Fuzzy Image Processing: Fuzzy relation equations for image processing.
- 4. Fuzzy Data Mining: Fuzzy relation equations for data mining.

# **Tools and Techniques**:

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

## **Challenges and Future Directions:**

- 1. **Computational Complexity**: Efficient algorithms for solving fuzzy relation equations.
- 2. Interpretability: Understanding fuzzy relation equation results.
- 3. Integration: Combining fuzzy relation equations with other techniques.

Fuzzy Similarity Relations:

**Definition**: A fuzzy similarity relation is a fuzzy relation that satisfies:

- 1. **Reflexivity**:  $\mu(a, a) = 1$  (every element is similar to itself)
- 2. **Symmetry**:  $\mu(a, b) = \mu(b, a)$  (if a is similar to b, then b is similar to a)
- Transitivity: μ(a, c) ≥ min(μ(a, b), μ(b, c)) (if a is similar to b, and b is similar to c, then a is similar to c)

# **Properties**:

- 1. **Fuzzy Equivalence Classes**: Fuzzy similarity relations induce fuzzy equivalence classes.
- 2. Fuzzy Clustering: Fuzzy similarity relations can be used for fuzzy clustering.
- 3. **Fuzzy Classification**: Fuzzy similarity relations can be used for fuzzy classification.

# **Types of Fuzzy Similarity Relations:**

- 1. Fuzzy Equality Relations: Fuzzy similarity relations with  $\mu(a, b) = 1$  iff a = b.
- Fuzzy Proximity Relations: Fuzzy similarity relations with μ(a, b) > 0 iff a is close to b.
- 3. **Fuzzy Indistinguishability Relations**: Fuzzy similarity relations with  $\mu(a, b) = 1$  iff a and b are indistinguishable.

# **Applications**:

- 1. **Image Segmentation**: Fuzzy similarity relations for image segmentation.
- 2. **Data Clustering**: Fuzzy similarity relations for data clustering.
- 3. **Decision-Making**: Fuzzy similarity relations for decision-making systems.
- 4. Natural Language Processing: Fuzzy similarity relations for text analysis.

# Algorithms:

- 1. Fuzzy C-Means: Fuzzy clustering algorithm based on fuzzy similarity relations.
- 2. Fuzzy K-Means: Fuzzy clustering algorithm based on fuzzy similarity relations.
- 3. **Fuzzy Hierarchical Clustering**: Fuzzy clustering algorithm based on fuzzy similarity relations.

## **Tools and Techniques**:

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

# **Challenges and Future Directions:**

- 1. Computational Complexity: Efficient algorithms for fuzzy similarity relations.
- 2. Interpretability: Understanding fuzzy similarity relation results.
- 3. Integration: Combining fuzzy similarity relations with other techniques.

Additional resources:

- Research papers: Fuzzy Sets and Systems, IEEE Transactions on Fuzzy Systems.
- Books: "Fuzzy Similarity Relations" by H. J. Zimmermann.
- Courses: Coursera, edX, University of Michigan.
- Software: MATLAB, Python, R.

**Definition**: A fuzzy relation equation is an equation involving fuzzy relations, where the unknown is a fuzzy relation.

# **Types of Fuzzy Relation Equations:**

- 1. **Fuzzy Relational Equations**: Equations involving fuzzy relations, e.g.,  $R \circ S = T$ .
- 2. **Fuzzy Matrix Equations**: Equations involving fuzzy matrices, e.g.,  $R \times S = T$ .
- 3. **Fuzzy Graph Equations**: Equations involving fuzzy graphs, e.g.,  $R \subseteq S$ .

# **Fuzzy Relation Equation Properties:**

- 1. **Reflexivity**:  $R \circ R = R$  (idempotence).
- 2. **Symmetry**:  $R \circ S = S \circ R$  (commutativity).
- 3. **Transitivity**:  $R \circ S \circ T = R \circ T \circ S$  (associativity).

# Methods for Solving Fuzzy Relation Equations:

- 1. **Fuzzy Relational Algebra**: Using fuzzy relational algebra to solve fuzzy relation equations.
- 2. **Fuzzy Matrix Algebra**: Using fuzzy matrix algebra to solve fuzzy matrix equations.
- 3. Fuzzy Graph Theory: Using fuzzy graph theory to solve fuzzy graph equations.

#### **Applications of Fuzzy Relation Equations:**

- 1. Fuzzy Control Systems: Fuzzy relation equations for control systems.
- 2. Fuzzy Decision-Making: Fuzzy relation equations for decision-making systems.
- 3. Fuzzy Image Processing: Fuzzy relation equations for image processing.
- 4. **Fuzzy Data Mining**: Fuzzy relation equations for data mining.

**Tools and Techniques**:

1. MATLAB: Fuzzy Logic Toolbox.

- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

# **Challenges and Future Directions:**

- 1. **Computational Complexity**: Efficient algorithms for solving fuzzy relation equations.
- 2. Interpretability: Understanding fuzzy relation equation results.
- 3. Integration: Combining fuzzy relation equations with other techniques.

Additional resources:

- Research papers: Fuzzy Sets and Systems, IEEE Transactions on Fuzzy Systems.
- Books: "Fuzzy Relation Equations" by H. J. Zimmermann.
- Courses: Coursera, edX, University of Michigan.
- Software: MATLAB, Python, R.

1. Fuzzy Relational Algebra

Fuzzy Relational Algebra:

**Definition**: Fuzzy relational algebra is a mathematical framework for manipulating fuzzy relations.

# **Operations**:

- 1. **Fuzzy Union**:  $R \cup S = max(R, S)$
- 2. **Fuzzy Intersection**:  $R \cap S = min(R, S)$
- 3. **Fuzzy Complement**: R' = 1 R
- 4. **Fuzzy Composition**:  $R \circ S = sup(min(R, S))$
- 5. **Fuzzy Join**:  $R \triangleright S = sup(min(R, S))$

# **Properties**:

- 1. **Reflexivity**:  $R \circ R = R$
- 2. **Symmetry**:  $\mathbf{R} \circ \mathbf{S} = \mathbf{S} \circ \mathbf{R}$
- 3. **Transitivity**:  $R \circ S \circ T = R \circ T \circ S$
- 4. **Distributivity**:  $R \circ (S \cup T) = (R \circ S) \cup (R \circ T)$

#### **Fuzzy Relational Algebra Operators:**

- 1. **Fuzzy Selection**:  $\sigma(\mathbf{R}, \theta) = \{x \mid \mathbf{R}(x) \ge \theta\}$
- 2. **Fuzzy Projection**:  $\pi(R, A) = \{x \mid R(x) \ge A\}$
- 3. **Fuzzy Join**:  $R \triangleright S = sup(min(R, S))$
- 4. **Fuzzy Division**:  $R \div S = \{x \mid R(x) \le S(x)\}$

#### **Applications**:

- 1. Fuzzy Database Systems: Fuzzy relational algebra for fuzzy databases.
- 2. Fuzzy Information Retrieval: Fuzzy relational algebra for information retrieval.
- 3. Fuzzy Decision-Making: Fuzzy relational algebra for decision-making systems.
- 4. Fuzzy Data Mining: Fuzzy relational algebra for data mining.
- 5.

#### **Tools and Techniques**:

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

#### **Challenges and Future Directions:**

1. Computational Complexity: Efficient algorithms for fuzzy relational algebra.

- 2. Interpretability: Understanding fuzzy relational algebra results.
- 3. Integration: Combining fuzzy relational algebra with other techniques.

Fuzzy Matrix Algebra:

**Definition**: Fuzzy matrix algebra is a mathematical framework for manipulating fuzzy matrices.

#### **Fuzzy Matrix Operations:**

- 1. Fuzzy Addition:  $A + B = [a_{ij} + b_{ij}]$
- 2. **Fuzzy Multiplication**:  $A \times B = [\sum (a_ik \times b_kj)]$
- 3. **Fuzzy Transpose**: A<sup>T</sup> = [a\_ji]
- 4. **Fuzzy Inverse**: A^-1 = [a\_ij^-1]

#### **Fuzzy Matrix Properties:**

- 1. **Reflexivity**:  $A \times A = A$
- 2. **Symmetry**:  $A \times B = B \times A$
- 3. **Transitivity**:  $A \times B \times C = A \times C \times B$
- 4. **Distributivity**:  $A \times (B + C) = A \times B + A \times C$

#### **Fuzzy Matrix Algebra Operators:**

- 1. Fuzzy Matrix Norm:  $||A|| = \sup(||a_{ij}||)$
- 2. Fuzzy Matrix Distance: d(A, B) = ||A B||
- 3. Fuzzy Matrix Similarity: s(A, B) = 1 d(A, B)

#### **Applications**:

- 1. Fuzzy Control Systems: Fuzzy matrix algebra for control systems.
- 2. Fuzzy Image Processing: Fuzzy matrix algebra for image processing.
- 3. Fuzzy Data Analysis: Fuzzy matrix algebra for data analysis.

4. Fuzzy Decision-Making: Fuzzy matrix algebra for decision-making systems.

**Tools and Techniques**:

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

# **Challenges and Future Directions**:

- 1. Computational Complexity: Efficient algorithms for fuzzy matrix algebra.
- 2. **Interpretability**: Understanding fuzzy matrix algebra results.
- 3. Integration: Combining fuzzy matrix algebra with other techniques.

Fuzzy Matrix Types:

- 1. Fuzzy Binary Matrices: Matrices with fuzzy binary entries.
- 2. Fuzzy Real Matrices: Matrices with fuzzy real entries.
- 3. Fuzzy Interval Matrices: Matrices with fuzzy interval entries.
- 4. Fuzzy Linguistic Matrices: Matrices with fuzzy linguistic entries.

Fuzzy Matrix Decompositions:

- 1. **Fuzzy LU Decomposition**: Decomposing fuzzy matrices into lower and upper triangular matrices.
- 2. **Fuzzy Cholesky Decomposition**: Decomposing fuzzy matrices into lower triangular matrices.
- 3. **Fuzzy Singular Value Decomposition**: Decomposing fuzzy matrices into singular values and vectors.

Fuzzy Matrix Eigenvalues and Eigenvectors:

- 1. **Fuzzy Eigenvalues**: Eigenvalues of fuzzy matrices.
- 2. Fuzzy Eigenvectors: Eigenvectors of fuzzy matrices.
- 3. Fuzzy Spectral Radius: Spectral radius of fuzzy matrices.

**Fuzzy Graph Equations:** 

Definition: Fuzzy graph equations are mathematical equations involving fuzzy graphs.

**Fuzzy Graph Operations**:

- 1. **Fuzzy Union**:  $G \cup H = max(G, H)$
- 2. **Fuzzy Intersection**:  $G \cap H = min(G, H)$
- 3. **Fuzzy Complement**: G' = 1 G
- 4. **Fuzzy Composition**:  $G \circ H = sup(min(G, H))$

**Fuzzy Graph Equations**:

- 1. Fuzzy Graph Isomorphism:  $G \cong H$
- 2. Fuzzy Graph Homomorphism:  $G \rightarrow H$
- 3. Fuzzy Graph Endomorphism:  $G \rightarrow G$
- 4. Fuzzy Graph Automorphism:  $G \rightarrow G$

**Fuzzy Graph Equation Properties:** 

- 1. **Reflexivity**:  $G \cong G$
- 2. Symmetry:  $G \cong H$  implies  $H \cong G$
- 3. **Transitivity**:  $G \cong H$  and  $H \cong K$  implies  $G \cong K$

# **Applications**:

1. Fuzzy Network Analysis: Fuzzy graph equations for network analysis.

- 2. Fuzzy Image Segmentation: Fuzzy graph equations for image segmentation.
- 3. Fuzzy Data Mining: Fuzzy graph equations for data mining.
- 4. Fuzzy Decision-Making: Fuzzy graph equations for decision-making systems.

# **Tools and Techniques**:

- 1. MATLAB: Fuzzy Logic Toolbox.
- 2. **Python**: scikit-fuzzy, fuzzywuzzy.
- 3. **R**: frbs, fuzzyR.
- 4. Java: JFuzzy, FuzzyJ.

# **Challenges and Future Directions:**

- 1. Computational Complexity: Efficient algorithms for fuzzy graph equations.
- 2. **Interpretability**: Understanding fuzzy graph equation results.
- 3. Integration: Combining fuzzy graph equations with other techniques.

# **Real-World Examples:**

- 1. Social Network Analysis: Fuzzy graph equations for social network analysis.
- 2. **Traffic Network Optimization**: Fuzzy graph equations for traffic network optimization.
- 3. **Biological Network Analysis**: Fuzzy graph equations for biological network analysis.

# Additional resources:

• Research papers: Fuzzy Sets and Systems, IEEE Transactions on Fuzzy Systems.

- Books: "Fuzzy Graph Theory" by A. K. Singh.
- Courses: Coursera, edX, University of Michigan.
- Software: MATLAB, Python, R.

Fuzzy Graph Types:

- 1. Fuzzy Undirected Graphs: Graphs with fuzzy edge weights.
- 2. Fuzzy Directed Graphs: Graphs with fuzzy edge weights and directions.
- 3. Fuzzy Weighted Graphs: Graphs with fuzzy edge weights and vertex weights.
- 4. **Fuzzy Hypergraphs**: Graphs with fuzzy hyperedges.

Fuzzy Graph Algorithms:

- 1. Fuzzy Breadth-First Search: Searching fuzzy graphs level by level.
- 2. Fuzzy Depth-First Search: Searching fuzzy graphs depth-first.
- 3. Fuzzy Dijkstra's Algorithm: Finding shortest paths in fuzzy graphs.
- 4. Fuzzy Bellman-Ford Algorithm: Finding shortest paths in fuzzy graphs.

# BASIC MATHEMATICAL FORMULAS

# EDITED BY DR.S.SUBRAMANIAN



# **Basic Mathematical Formulas**

Copyright © 2021 by Dr.S.Subramanian

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-592-2

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

<b>Book elements</b>	
Chapter-I TRIGONOMETRIC IDENTITIES Dr.S.Subramanian	01-04
Chapter-II TRIGONOMETRIC IDENTITIES PROOFS Dr.S.Subramanian	05-08
Chapter-III DIFFERENTIATION Dr.S.Subramanian	09-11
Chapter-IV DIFFERENTIATION RULES Dr.S.Subramanian	10-14
Chapter-V INTEGRATION Dr.S.Subramanian	15-17
Chapter-VI ADVANCED INTEGRATION FORMULAS Dr.S.Subramanian	18-20
Chapter-VII APPLICATION OF INTEGRATION FORMULAS Dr.S.Subramanian	21-24
Chapter-VIII COMPLEX NUMBERS Dr.S.Subramanian	25-32
Chapter-IX VECTOR DIFFERENTIATION Dr.S.Subramanian	33-42
Chapter-X	

43-44

# **Chapter -I**

#### TRIGONOMETRIC IDENTITIES Dr.S.Subramanian

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India.

- Trigonometric Identities are useful whenever trigonometric functions are involved in an expression or an equation. Trigonometric Identities are true for every value of variables occurring on both sides of an equation. Geometrically, these identities involve certain trigonometric functions (such as sine, cosine, tangent) of one or more angles.
- Sine, cosine and tangent are the primary trigonometry functions whereas cotangent, secant and cosecant are the other three functions. The trigonometric identities are based on all the six trig functions. Check Trigonometry Formulas to get formulas related to trigonometry.

#### What are Trigonometric Identities?

Trigonometric Identities are the equalities that involve trigonometry functions and holds true for all the values of variables given in the equation.

There are various distinct trigonometric identities involving the side length as well as the angle of a triangle. The trigonometric identities hold true only for the <u>right-angle triangle</u>.

All the trigonometric identities are based on the six trigonometric ratios. They are sine, cosine, tangent, cosecant, secant, and cotangent. All these trigonometric ratios are defined using the sides of the right triangle, such as an adjacent side, opposite side, and hypotenuse side. All the fundamental trigonometric identities are derived from the six trigonometric ratios.

Trigonometric Identities PDF

Click here to download the PDF of trigonometry identities of all functions such as sin, cos, tan and so on.

List of Trigonometric Identities

There are various identities in trigonometry which are used to solve many trigonometric problems. Using these trigonometric identities or formulas, complex trigonometric questions can be solved quickly. Let us see all the fundamental trigonometric identities here.

**Reciprocal Trigonometric Identities** 

The reciprocal trigonometric identities are:

- $\sin \theta = 1/Csc \theta \text{ or } Csc \theta = 1/Sin \theta$
- $\cos \theta = 1/\sec \theta$  or  $\sec \theta = 1/\cos \theta$
- Tan  $\theta = 1/\cot \theta$  or  $\cot \theta = 1/\tan \theta$

Pythagorean Trigonometric Identities

There are three Pythagorean trigonometric identities in trigonometry that are based on the righttriangle theorem or Pythagoras theorem.

- $\sin^2 a + \cos^2 a = 1$
- $1+\tan^2 a = \sec^2 a$
- $\csc^2 a = 1 + \cot^2 a$

Ratio Trigonometric Identities

The trigonometric ratio identities are:

- $\operatorname{Tan} \theta = \operatorname{Sin} \theta / \operatorname{Cos} \theta$
- Cot  $\theta = \cos \theta / \sin \theta$

Trigonometric Identities of Opposite Angles

The list of opposite angle trigonometric identities are:

- $Sin(-\theta) = -Sin\theta$
- $\cos(-\theta) = \cos \theta$
- $Tan(-\theta) = -Tan \theta$
- $\operatorname{Cot}(-\theta) = -\operatorname{Cot} \theta$
- Sec  $(-\theta) = \operatorname{Sec} \theta$
- $\operatorname{Csc}(-\theta) = -\operatorname{Csc} \theta$

Trigonometric Identities of Complementary Angles

In geometry, two angles are complementary if their sum is equal to 90 degrees. Similarly, when we can learn here the trigonometric identities for complementary angles.

•  $Sin(90-\theta) = Cos \theta$ 

- $\cos(90 \theta) = \sin \theta$
- Tan  $(90 \theta) = \operatorname{Cot} \theta$
- Cot  $(90 \theta) = Tan \theta$
- Sec  $(90 \theta) = Csc \theta$
- $\operatorname{Csc}(90-\theta) = \operatorname{Sec}\theta$

#### Trigonometric Identities of Supplementary Angles

Two angles are supplementary if their sum is equal to 90 degrees. Similarly, when we can learn here the trigonometric identities for supplementary angles.

- $\sin(180^{\circ} \theta) = \sin\theta$
- $\cos(180^\circ \theta) = -\cos\theta$
- $\operatorname{cosec}(180^\circ \theta) = \operatorname{cosec} \theta$
- $\sec(180^\circ \theta) = -\sec\theta$
- $\tan(180^\circ \theta) = -\tan\theta$
- $\cot(180^\circ \theta) = -\cot\theta$

Sum and Difference of Angles Trigonometric Identities

Consider two angles ,  $\alpha$  and  $\beta$ , the trigonometric sum and difference identities are as follows:

- $\sin(\alpha+\beta)=\sin(\alpha).\cos(\beta)+\cos(\alpha).\sin(\beta)$
- $sin(\alpha-\beta)=sin\alpha.cos\beta-cos\alpha.sin\beta$
- $\cos(\alpha+\beta)=\cos\alpha.\cos\beta-\sin\alpha.\sin\beta$
- $\cos(\alpha \beta) = \cos\alpha \cdot \cos\beta + \sin\alpha \cdot \sin\beta$

#### Double Angle Trigonometric Identities

If the angles are doubled, then the trigonometric identities for sin, cos and tan are:

- $\sin 2\theta = 2 \sin \theta \cos \theta$
- $\cos 2\theta = \cos^2\theta \sin^2\theta = 2\cos^2\theta 1 = 1 2\sin^2\theta$
- $\tan 2\theta = (2\tan\theta)/(1 \tan^2\theta)$

#### Half Angle Identities

If the angles are halved, then the trigonometric identities for sin, cos and tan are:

- $\sin(\theta/2) = \pm \sqrt{[(1 \cos\theta)/2]}$
- $\cos(\theta/2) = \pm \sqrt{(1 + \cos\theta)/2}$
- $\tan(\theta/2) = \pm \sqrt{[(1 \cos\theta)(1 + \cos\theta)]}$

Product-Sum Trigonometric Identities

The product-sum trigonometric identities change the sum or difference of sines or cosines into a product of sines and cosines.

- $\operatorname{Sin} A + \operatorname{Sin} B = 2 \operatorname{Sin}(A+B)/2 \cdot \operatorname{Cos}(A-B)/2$
- $\cos A + \cos B = 2 \cos(A+B)/2 \cdot \cos(A-B)/2$
- $\sin A \sin B = 2 \cos(A+B)/2 \cdot \sin(A-B)/2$
- $\cos A \cos B = -2 \sin(A+B)/2 \cdot \sin(A-B)/2$

Trigonometric Identities of Products

These identities are:

- Sin A. Sin B = [Cos (A B) Cos (A + B)]/2
- Sin A. Cos B = [Sin (A + B) + Sin (A B)]/2
- Cos A. Cos B = [Cos (A + B) + Cos (A B)]/2

# **Chapter**-II

# **TRIGONOMETRIC IDENTITIES PROOFS**

#### **Dr.S.Subramanian**

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Similarly, an equation that involves trigonometric ratios of an angle represents a trigonometric identity.

The upcoming discussion covers the fundamental trigonometric identities and their proofs. Consider the right angle  $\triangle$ ABC which is right-angled at B as shown in the given figure.



Applying Pythagoras Theorem for the given triangle, we have

 $(hypotenuse)^2 = (base)^2 + (perpendicular)^2$ 

Let us prove the three Pythagoras trigonometric identities, which are commonly used. Trigonometric Identity 1

Now, divide each term of equation (1) by  $AC^2$ , we have We know that, Thus equation (2) can be written as- $\sin^2 x + \cos^2 x = 1$ Identity 1 is valid for angles  $0 \le a \le 90$ . **Trigonometric Identity 2** Now Dividing the equation (1) by  $AB^2$ , we get By referring to trigonometric ratios, it can be seen that: Similarly, Replacing the values of AC/AB and BC/AB in equation (3) gives,  $1+\tan^2 x = \sec^2 x$ As it is known that tan a is not defined for  $a = 90^{\circ}$ , therefore, identity 2 obtained above is true for 0 < A <90. **Trigonometric Identity 3** Dividing the equation (1) by  $BC^2$ , we get By referring to trigonometric ratios, it can be seen that: Also, Replacing the values of AC/BC and AB/BC in the equation (4) gives,  $\operatorname{cosec}^2 x = 1 + \cot^2 x$ 

Since cosec a and cot a are not defined for  $a = 0^{\circ}$ . Therefore the identity 3 is obtained is true for all the values of 'a' except at  $a = 0^{\circ}$ . Therefore, the identity is true for all such that,  $0^{\circ} < a \le 90^{\circ}$ .

Triangle Identities (Sine, Cosine, Tangent rule)

If the identities or equations are applicable for all the triangles and not just for right triangles, then they are the triangle identities. These identities will include:

- Sine law
- Cosine law
- Tangent law

If A, B and C are the vertices of a triangle and a, b and c are the respective sides, then; According to the sine law or sine rule,

Or

According to cosine law,

**Related Articles** 

- Trigonometry
- Trigonometric Ratios
- Trigonometric Ratios of Complementary Angles
- Trigonometric Ratios Of Standard Angles
- Trigonometry Table
- Trigonometry Angles
- Trigonometry Values
- Trigonometric Identities

Solved Examples on Trigonometric Identities

Go through the below problem which is solved by using the trigonometric identities.

Example 1:

Consider a triangle ABC, right-angled at B. The length of the base, AB = 4 cm and length of perpendicular BC = 3 cm. Find the value of sec A.

Solution:

As the length of the perpendicular and base is given; it can be concluded that,

 $\tan A = 3/4$ 

Now, using the trigonometric identity:  $1+\tan^2 a = \sec^2 a$ 

 $\sec^2 A = 1 + (3/4)^2$ 

 $\sec^{2 A} = 25/16$ 

sec A =  $\pm 5/4$ 

Since, the ratio of lengths is positive, we can neglect sec A = 5/4.

Therefore, sec A = 5/4

Example 2:  $(1 - \sin A)/(1 + \sin A) = (\sec A - \tan A)^2$ 

Solution: Let us take the Left hand side of the equation.

 $L.H.S = (1 - \sin A)/(1 + \sin A)$ 

Multiply both numerator and denominator by  $(1 - \sin A)$ 

 $= (1 - \sin A)^2 / (1 - \sin A) (1 + \sin A)$ 

 $= (1 - \sin A)^2 / (1 - \sin^2 A)$ 

=  $(1 - \sin A)^2 / (\cos^2 A)$ , [Since  $\sin^2 \theta + \cos^2 \theta = 1 \Rightarrow \cos^2 \theta = 1 - \sin^2 \theta$ ]  $= \{(1 - \sin A)/\cos A\}^2$  $= (1/\cos A - \sin A/\cos A)^2$  $=(\sec A - \tan A)^2$ = R.H.S. Example 3: Prove that:  $1/(\operatorname{cosec} A - \operatorname{cot} A) - 1/\sin A = 1/\sin A - 1/(\operatorname{cosec} A + \operatorname{cot} A)$ Solution:  $1/(\csc A - \cot A) - 1/\sin A = 1/\sin A - 1/(\csc A + \cot A)$ Now rearrange the following, such that;  $1/(\operatorname{cosec} A - \operatorname{cot} A) + 1/(\operatorname{cosec} A + \operatorname{cot} A) = 2/\operatorname{Sin} A$ Now let us take the L.H.S.  $= 1/(\operatorname{cosec} A - \operatorname{cot} A) + 1/(\operatorname{cosec} A + \operatorname{cot} A)$ =  $(\operatorname{cosec} A + \operatorname{cot} A + \operatorname{cosec} A - \operatorname{cot} A)/(\operatorname{cosec}^2 A - \operatorname{cot}^2 A)$ =  $(2 \operatorname{cosec} A)/1$  [ $\operatorname{cosec}^2 A = 1 + \cot^2 A \Rightarrow \operatorname{cosec}^2 A - \cot^2 A = 1$ ]  $[\operatorname{cosec} A = 1/\sin A]$  $= 2/\sin A$ Hence, proved.

## **Chapter -III**

#### DIFFERENTIATION

# Dr.S.Subramanian

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

In calculus, differentiation is one of the two important concepts apart from integration. Differentiation is a method of finding the <u>derivative of a function</u>. Differentiation is a process, in Maths, where we find the instantaneous rate of change in function based on one of its variables. The most common example is the rate change of displacement with respect to time, called velocity. The opposite of finding a derivative is anti-differentiation.

If x is a variable and y is another variable, then the rate of change of x with respect to y is given by dy/dx. This is the general expression of derivative of a function and is represented as f'(x) = dy/dx, where y = f(x) is any function.

What is Differentiation in Mathematics

In Mathematics, Differentiation can be defined as a derivative of a function with respect to an independent variable. Differentiation, in calculus, can be applied to measure the function per unit change in the independent variable.

Let y = f(x) be a function of x. Then, the rate of change of "y" per unit change in "x" is given by: dy / dx

If the function f(x) undergoes an infinitesimal change of 'h' near to any point 'x', then the derivative of the function is defined as

Derivative of Function As Limits

If we are given with real valued function (f) and x is a point in its domain of definition, then the derivative of function, f, is given by:

 $f'(a) = \lim_{h \to 0} [f(x+h) - f(x)]/h$ 

provided this limit exists.

Let us see an example here for better understanding.

Example: Find the derivative of f(x) = 2x, at x = 3.

Solution: By using the above formulas, we can find,

$$\begin{split} f'(3) &= \lim_{h \to 0} \left[ f(3+h) - f(3) \right] / h = \lim_{h \to 0} \left[ 2(3+h) - 2(3) \right] / h \\ f'(3) &= \lim_{h \to 0} \left[ 6 + 2h - 6 \right] / h \\ f'(3) &= \lim_{h \to 0} 2h / h \\ f'(3) &= \lim_{h \to 0} 2 = 2 \\ Notations \end{split}$$

When a function is denoted as y = f(x), the derivative is indicated by the following notations.

1. D(y) or D[f(x)] is called Euler's notation.

2. dy/dx is called Leibniz's notation.

3. F'(x) is called Lagrange's notation.

The meaning of differentiation is the process of determining the derivative of a function at any point.

Linear and Non-Linear Functions

Functions are generally classified into two categories under Calculus, namely:

(i) Linear functions

(ii) Non-linear functions

A linear function varies at a constant rate through its domain. Therefore, the overall rate of change of the function is the same as the rate of change of a function at any point.

However, the rate of change of function varies from point to point in the case of non-linear functions. The nature of variation is based on the nature of the function.

The rate of change of a function at a particular point is defined as a derivative of that particular function.

#### **Differentiation Formulas**

The important Differentiation formulas are given below in the table. Here, let us consider f(x) as a function and f'(x) is the derivative of the function.

- 1. If  $f(x) = \tan(x)$ , then  $f'(x) = \sec 2x$
- 2. If  $f(x) = \cos(x)$ , then  $f'(x) = -\sin x$
- 3. If  $f(x) = \sin(x)$ , then  $f'(x) = \cos x$
- 4. If  $f(x) = \ln(x)$ , then f'(x) = 1/x
- 5. If f(x) = ex, then f'(x) = ex

- 6. If f(x) = xn, where n is any fraction or integer, then f'(x) = nxn-1
- 7. If f(x) = k, where k is a constant, then f'(x) = 0

Also, see:

Calculus

Derivatives

Logarithmic Differentiation

Derivative of Inverse Trigonometric functions

# **Chapter-IV**

# **DIFFERENTIATION RULES**

#### **Dr.S.Subramanian**

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

The basic differentiation rules that need to be followed are as follows:

<ul> <li>Sum and Difference Rul</li> </ul>
--

- Product Rule
- Quotient Rule
- Chain Rule

Let us discuss all these rules	here.
--------------------------------	-------

#### Sum or Difference Rule

If the function is the sum or difference of two functions, the derivative of the functions is the sum or difference of the individual functions, i.e.,

If  $f(x) = u(x) \pm v(x)$ then,  $f'(x) = u'(x) \pm v'(x)$ 

#### Product Rule

As per the product rule, if the function f(x) is product of two functions u(x) and v(x), the derivative of the function is,

#### Quotient rule

If the function f(x) is in the form of two functions [u(x)]/[v(x)], the derivative of the function is Chain Rule

If a function y = f(x) = g(u) and if u = h(x), then the <u>chain rule</u> for differentiation is defined as,

This plays a major role in the method of substitution that helps to perform differentiation of composite functions.

#### Real-Life Applications of Differentiation

With the help of differentiation, we are able to find the rate of change of one quantity with respect to another. Some of the examples are:

- Acceleration: Rate of change of velocity with respect to time
- To calculate the highest and lowest point of the curve in a graph or to know its turning point, the derivative function is used
- To find tangent and normal to a curve

Solved Examples

Q.1: Differentiate  $f(x) = 6x^3 - 9x + 4$  with respect to x. Solution: Given:  $f(x) = 6x^3 - 9x + 4$ 

On differentiating both the sides w.r.t x, we get;

$$f'(x) = (3)(6)x^2 - 9$$

$$f'(x) = 18x^2 - 9$$

This is the final answer.

Q.2: Differentiate  $y = x(3x^2 - 9)$ 

Solution: Given,  $y = x(3x^2 - 9)$ 

$$y = 3x^3 - 9x$$

On differentiating both the sides we get,

 $dy/dx = 9x^2 - 9$ 

**Practice Problems** 

- 1. Find the derivative of the function  $f(x) = 3 \sin x + \cos x \tan x$ .
- 2. Perform differentiation following functions: the for the x<sup>3</sup> (i) f(x)sin 2x= (ii)  $g(x) = 4xe^{2x} - 9x$ 3. Find the derivative the function f(x)x/(x 2) of =
  - (i) Using the limit definition of differentiation(ii) Using the quotient rule

To know more about Differentiation and any Maths related topics, please visit us at BYJU'S.

#### **Chapter-V**

#### INTEGRATION

**Dr.S.Subramanian** Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**)

Vallam, Thanjavur, Tamilnadu, South India

Integration, in mathematics, technique of finding a function g(x) the derivative of which, Dg(x), is equal to a given function f(x). This is indicated by the integral sign " $\int$ ," as in  $\int f(x)$ , usually called the indefinite integral of the function. The symbol dx represents an infinitesimal displacement along x; thus  $\int f(x)dx$  is the summation of the product of f(x) and dx. The definite

integral, written with *a* and *b* called the limits of integration, is equal to g(b) - g(a), where Dg(x) = f(x).

Some anti derivatives can be calculated by merely recalling which function has a given derivative, but the techniques of integration mostly involve classifying the functions according to which types of manipulations will change the function into a form the anti derivative of which can be more easily recognized. For example, if one is familiar with derivatives, the function 1/(x + 1) can be easily recognized as the derivative of  $\log e(x + 1)$ . The anti derivative of (x2 + x + 1)/(x + 1) cannot be so easily recognized, but if written as x(x + 1)/(x + 1) + 1/(x + 1) = x + 1/(x + 1), it then can be recognized as the derivative of  $x2/2 + \log e(x + 1)$ . One useful aid for integration is the theorem known as integration by parts. In symbols, the rule is  $\int fDg = fg - \int gDf$ . That is, if a function is the product of two other functions, f and one that can be recognized as the derivative of some function g, then the original problem can be solved if one can integrate the product gDf. For example, if f = x, and  $Dg = \cos x$ , then  $\int x \cdot \cos x = x \cdot \sin x - \int \sin x = x \cdot \sin x - \cos x + C$ . Integrals are used to evaluate such quantities as area, volume, work, and, in general, any quantity that can be interpreted as the area under a curve.

#### Integration Formulas

Integration formulas can be applied for the integration of algebraic expressions, trigonometric ratios, inverse trigonometric functions, and logarithmic and exponential functions. The integration of functions results in the original functions for which the derivatives were obtained. These integration formulas are used to find the anti derivative of a function. If we differentiate a function f in an interval I, then we get a family of functions in I. If the values of functions are known in I, then we can determine the function f. This inverse process of differentiation is called integration.

Let's move further and learn about integration formulas used in the integration techniques.

What are Integration Formulas?

The integration formulas have been broadly presented as the following sets of formulas. The formulas include basic integration formulas, integration of <u>trigonometric ratios</u>, inverse trigonometric functions, the product of <u>functions</u>, and some advanced set of integration formulas. Basically, <u>integration</u> is a way of uniting the part to find a whole. It is the inverse operation of <u>differentiation</u>. Thus the basic integration formula is  $\int f(x) dx = f(x) + C$ . Using this, the following integration formulas are derived.

#### **Basic Integration Formulas**

Using the fundamental theorems of integrals, there are generalized results obtained which are remembered as integration formulas in indefinite integration.

- $\int x^n dx = x^{(n+1)}/(n+1) + C$
- $\int 1 \, \mathrm{d}x = x + C$
- $\int e^x dx = e^x + C$
- $\int 1/x \, dx = \log |x| + C$

Integration Formulas of Trigonometric functions

The process of finding the integral is integration. Here are a few important integration formulas remembered for instant and speedy calculations. When it comes to trigonometric functions, we simplify them and rewrite them as functions that are integrable. Here is a list of <u>trigonometric</u> and <u>inverse trigonometric functions</u>.

- $\int \cos x \, dx = \sin x + C$
- $\int \sin x \, dx = -\cos x + C$

- $\int \sec^2 x \, dx = \tan x + C$
- $\int \operatorname{cosec}^2 x \, dx = -\cot x + C$
- $\int \sec x \tan x \, dx = \sec x + C$
- $\int \operatorname{cosec} x \operatorname{cot} x \, dx = \operatorname{-cosec} x + C$
- $\int \tan x \, dx = \log |\sec x| + C$
- $\int \cot x \, dx = \log |\sin x| + C$
- $\int \sec x \, dx = \log |\sec x + \tan x| + C$
- $\int \csc x \, dx = \log |\csc x \cot x| + C$

Integration Formulas of Inverse Trigonometric Functions

Here are the integral formulas that lead to/give the result in the form of inverse trigonometric

functions.

- $\int 1/\sqrt{(1 x^2)} \, dx = \sin^{-1}x + C$
- $\int 1/\sqrt{(1 x^2)} \, dx = -\cos^{-1}x + C$
- $\int 1/(1+x^2) dx = \tan^{-1}x + C$
- $\int 1/(1+x^2) dx = -\cot^{-1}x + C$
- $\int 1/x \sqrt{x^2 1} \, dx = \sec^{-1}x + C$
- $\int 1/x \sqrt{x^2 1} \, dx = -\cos e^{-1} x + C$

Chapter –VI

# ADVANCED INTEGRATION FORMULAS

#### **Dr.S.Subramanian**

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

Here are some advanced integral formulas that we may come across while solving the problems of integration.

- $\int 1/(x^2 a^2) dx = 1/2a \log|(x a)(x + a)| + C$
- $\int 1/(a^2 x^2) dx = 1/2a \log|(a + x)(a x)| + C$
- $\int 1/(x^2 + a^2) dx = 1/a \tan^{-1} x/a + C$
- $\int 1/\sqrt{(x^2 a^2)} dx = \log |x + \sqrt{(x^2 a^2)}| + C$
- $\int \sqrt{(x^2 a^2)} \, dx = x/2 \, \sqrt{(x^2 a^2) a^2/2} \log |x + \sqrt{(x^2 a^2)}| + C$
- $\int 1/\sqrt{a^2 x^2} \, dx = \sin^{-1} x/a + C$
- $\int \sqrt{a^2 x^2} \, dx = x/2 \, \sqrt{a^2 x^2} \, dx + a^2/2 \sin^{-1} x/a + C$
- $\int 1/\sqrt{(x^2 + a^2)} dx = \log |x + \sqrt{(x^2 + a^2)}| + C$
- $\int \sqrt{(x^2 + a^2)} \, dx = x/2 \, \sqrt{(x^2 + a^2) + a^2/2} \, \log |x + \sqrt{(x^2 + a^2)}| + C$

**Different Integration Formulas** 

There are 3 types of integration methods and each method is applied with its own unique techniques involved in finding the integrals. They are the standardized results. They can be remembered as integration formulas.

#### Integration by parts formula:

When the given function is a product of two functions, we apply this <u>integration by parts</u> formula or partial integration and evaluate the integral. The integration formula while using partial integration is given as:

$$\int f(x) g(x) dx = f(x) \int g(x) dx - \int \left( \int f(x) g(x) dx \right) dx + C$$

For example:  $\int xe^x dx$  is of the form  $\int f(x) g(x) dx$ . Thus we apply the appropriate integration formula and evaluate the integral.

$$f(x) = x \text{ and } g(x) = e^{x}$$
  
Thus  $\int xe^{x} dx = x \int e^{x} dx - \int (1 \int e^{x} dx) dx + c$ 
$$= xe^{x} - e^{x} + c$$

Integration by substitution formula:

When a function is a function of another function, then we apply the <u>integration formula for</u> <u>substitution</u>. If  $I = \int f(x) dx$ , where x = g(t) so that dx/dt = g'(t), then we write dx = g'(t)We can write  $I = \int f(x) dx = \int f(g(t)) g'(t) dt$ 

we can write  $I = \int I(x) dx = \int I(g(t)) g'(t) dt$ 

For example: Consider  $\int (3x+2)^4 dx$ 

We can use the integration formula of substitution here. Let  $u = (3x+2) \Rightarrow du = 3 dx$ .

Thus  $\int (3x+2)^4 dx = 1/3 \int (u)^4 du$ 

$$= 1/3. u^5/5 = u^5/15$$

 $=(3x+2)^{5}/15$ 

Integration by partial fractions formula:

If we need to find the integral of P(x)/Q(x) that is an improper fraction, wherein the <u>degree</u> of P(x) < that of Q(x), then we use <u>integration by partial fractions</u>. We split the fraction using partial fraction decomposition as  $P(x)/Q(x) = T(x) + P_1(x)/Q(x)$ , where T(x) is a polynomial in x and  $P_1(x)/Q(x)$  is a proper rational function. If A, B, and C are the real numbers, then we have the following types of simpler <u>partial fractions</u> that are associated with various types of <u>rational functions</u>.

Form of Rational Fractions	Form of Partial Fractions
(px + q)/(x-a)(x - b)	A/(x - a) + B/(x - b)
$(px + q)/(x-a)^n$	$A_1/(x-a) + A_2/(x-a)^2 + \dots A_n/(x-a)^n$
$(px^{2} + qx + r)/(ax^{2} + bx + c)$	$(Ax + B)/(ax^2 + bx + c)$
$(px^2 + qx + r)/(ax^2 + bx + r)$	$(A_1x + B_1)/(ax^2 + bx + c) + (A_2x + B_2)/(ax^2 + bx + c)^2 +(A_nx + C)^2$
c) <sup>n</sup>	$B_n)/(ax^2 + bx + c)^n$
$(px^2 + qx + r)/(x-a)(x-b)(x-b)(x-b)(x-b)(x-b)(x-b)(x-b)(x-b$	A/(x-a) + B/(x-b) + C/(x-c)

Form of Rational Fractions	Form of Partial Fractions
c)	
$(px^{2} + qx + r)/ [(x-a) (x^{2} + bx + c)]$	$A/(x-a) + (Bx+C)/(x^2+bx+c)$

For example:  $\int 3x+7/x^2-3x+2$ 

Resolving it into partial fractions, we get

 $3x+7/x^2-3x+2 = A/(x-2) + B/(x-1)$ 

= A(x-1) + B(x-2)/(x-2)(x-1)

Equating the numerators, we get 3x + 7 = A(x-1)+B(x-2)

Find B by giving  $x = 1 \Rightarrow 10 = B$ 

Find A by giving  $x = 2 \Rightarrow 13 = A$ 

Thus  $3x+7/x^2-3x + 2 = \frac{13}{(x-2)} + \frac{10(x-1)}{x-2}$ 

Applying the integration formulas, we get

 $\int (3x+7/x^2-3x+2) = \int \frac{13}{(x-2)} + \int \frac{10}{(x-1)}$ 

 $\int (3x+7/x^2-3x+2) = 13 \log |x-2| - 10 \log |x-1| + C$ 

# **Chapter**-VII

#### **APPLICATION OF INTEGRATION FORMULAS**

#### **Dr.S.Subramanian**

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

In general, there are two types of integrals. They are definite and indefinite integrals.

**Definite Integration Formula** 

These are the integrations that have a pre-existing value of limits; thus making the final value of integral definite.

 $\int_{a}^{b} g(x) dx = G(b) - G(a)$ , where g(x) = G'(x).

Indefinite Integration Formula

These are the integrations that do not have a pre-existing value of limits; thus making the final value of integral indefinite. Here, C is the integration constant.  $\int g'(x) = g(x) + C$ 

#### **Applictions of Integration Formulas**

These are followed by the <u>fundamental theorem of calculus</u>.

We apply the integration formulas discussed so far, in approximating the area bounded by the curves, in evaluating the average distance, velocity and acceleration-oriented problems, in finding the average value of a function, to approximate the volume and the surface area of the solids, in finding the center of mass and work, in estimating the arc length, in finding the kinetic energy of a moving object using improper integrals.

Let us compute the distance traveled by an object using integration formulas. We know that distance is the definite integral of velocity.

Given: the velocity of an object = v(t)=  $-t^2 + 5t$ . Let us find the displacement traveled on (1,3).

The initial and the final positions of the object are 1 and 3 respectively. Hence apply the integration formula here with the limits.  $\int_a^b g(x) dx = G(b) - G(a)$ 

$$\int_{1}^{3} v(t) dt = v(3) - v(1)$$
  
$$\int_{1}^{3} v(t) = \int_{1}^{3} (-t^{2}) dt + \int_{1}^{3} 5t$$
  
$$= -t^{3}/3 + 5t^{2}/2 |_{1}^{3}$$
  
$$= [(-27/3 - (-1/3)) + (45/2 - 5/2)]$$
  
$$= 34/3$$

Displacement = 34/3 units.

Let us see how to use the indefinite integration formulas in the following solved examples.

Integration formulas

Apply Integration formulas to find the primitive of the functions.

Integration Formulas Examples

- 1. Example 1: Find the value of  $\int (9x + 25)/(x+3)^2 dx$ Solution:  $(9x + 25)/(x+3)^2$  is a rational function. Using the partial fraction decomposition, we have  $(9x + 25)/(x+3)^2 = A/(x+3) + B/(x+3)^2$ Taking LCD, we get  $(9x + 25)/(x+3)^2 = [A(x+3) +B]/(x+3)^2$ Equating the numerator, we get 9x + 25 = A(x+3) + BSolving for B when x = -3, we get B = -2 Solving for A when x = 0, we get A = 9 Thus the partial fraction is decomposed as  $9/(x+3) - 2/(x+3)^2$ As stated in the integration formulas above, find the integral of  $9/(x+3) - 2/(x+3)^2$ .  $\int [9/(x+3)] dx - \int 2(x+3)^2 dx = 9 \log(x+3) + 2/(x+3) + C$ Answer:  $\int (9x+25)/(x+3)^2 dx = 9 \log(x+3) + 2/(x+3) + C$ 2. Example 2: Simplify and find the value of  $\int cosec x(cosec x - cot x) dx$  using the
- Example 2: Simplify and find the value of J cosec x(cosec x cot x) dx using the integration formula.

Solution:

 $\int \operatorname{cosec} x(\operatorname{cosec} x - \operatorname{cot} x) \, \mathrm{d}x$ 

 $= \int \csc^2 x \, dx - \int \cot x \, \csc x \, dx$ 

Using the basic trigonometric integration formulas,

 $= -\cot x - (-\csc x)$ 

 $= -\cot x + \csc x$ 

 $= \operatorname{cosec} x - \operatorname{cot} x + C$ 

Answer:  $\int \operatorname{cosec} x (\operatorname{cosec} x - \operatorname{cot} x) dx = \operatorname{cosec} x - \operatorname{cot} x + C$ 

3. Example 3: Use the integration formula and find the value of  $\int (5 + 4\cos x)/\sin^2 x \, dx$ . Solution:

```
\int (5 + 4\cos x) / \sin^2 x \, dx
```

 $= \int 5/\sin^2 x. \, dx + \int 4\cos x/\sin^2 x \, dx$ 

 $= \int 5 \csc^2 x \, dx + \int 4 \cot x \, \csc x \, dx$ 

 $= 5 \int \csc^2 x \, dx + 4 \int \cot x \, \csc x \, dx$ 

Using the trigonometric integration formula, we get

```
= 5(-\cot x) + 4(-\csc x)
```

```
= -5cot x - 4cosec x + C
```

Answer:  $\int (5 + 4\cos x) / \sin^2 x \, dx = -5\cot x - 4\csc x + C.$ 

FAQs on Integration Formulas

How Do You Integrate Using Integration Formulas?

We can use the below steps to integrate:

- Firstly define a small part of an object in certain dimensions which on adding repetitively makes the whole object.
- Use integration formulas over that small part along the varying dimensions.

What is the Integration Formulas or Integral UV Used for?

The <u>integral UV</u> is used to integrate the product of two functions. The integration formula as per this rule is  $\int u v dx = u \int v dx - \int u' (\int v dx) dx$ . Here, u is the function u(x) and v is the function v(x) What is the Use of Integration Formulas?

The integration is used to find the area of any objects. Real-life examples are to find the center of mass of an object, center of gravity, and mass moment of Inertia for a sports utility vehicle. It is
also used for calculating the velocity and trajectory of an object, predicting the alignment of planets, and in electromagnetism. Use <u>integration formulas</u> in all these cases.

What are Integration Techniques Involved in Integration Formulas?

Substitution, integration by parts, reverse chain rule, and partial fraction expansion are a few integration techniques.

What is The Integration Formula of Integral UV?

The formula for integral UV is used to integrate the <u>product</u> of two functions. The integration formula of UV form is given as  $\int u \, dv = uv \int v \, du$ .

What are The Integration Formulas For Trigonometric Functions?

The trigonometric functions are simplified into integrable functions and then their integrals are evaluated. The basic integration formulas for trigonometric functions are as follows.

- $\int \cos x \, dx = \sin x + C$
- $\int \sin x \, dx = -\cos x + C$
- $\int \sec^2 x \, dx = \tan x + C$
- $\int \operatorname{cosec}^2 x \, dx = -\operatorname{cot} x + C$
- $\int \sec x \cdot \tan x \, dx = \sec x + C$
- $\int \operatorname{cosec} x \cdot \operatorname{cosec} x \, dx = \operatorname{-cosec} x + C$
- $\int \tan x \, dx = \log|\sec x| + C$
- $\int \cot x \, dx = \log |\sin x| + C$
- $\int \sec x \, dx = \log|\sec x + \tan x| + C$
- $\int \operatorname{cosec} x \, dx = \log |\operatorname{cosec} x \cot x| + C$

How Do You apply Integration formulas to Find The Integrate log x?

 $\int \log(x) dx$  is of the UV form. Apply integration formula by parts rule.

- 1) Identify uv: Take u = log(x) and dv = 1.  $dx \Rightarrow du = 1/x$  and v = x
- 2) Apply formula:  $\int uv \, dx = uv \int v \, du$

$$=$$
 x. log(x) -  $\int$  x. 1/x

$$= x \log(x) - x + C$$

3) Simplify and evaluate the integral.

# **Chapter-VIII**

# **COMPLEX NUMBERS**

#### **Dr.S.Subramanian**

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

A Complex Number is a combination of a Real Number and an Imaginary Number

Real Numbers are numbers like:

1 12.38 -0.8625 3/4  $\sqrt{2}$  1998

Nearly any number you can think of is a Real Number!

Imaginary Numbers when squared give a negative result.

Normally this doesn't happen, because:

- when we square a positive number we get a positive result, and
- when we square a negative number we also get a positive result (because <u>a negative times</u> a negative gives a positive), for example  $-2 \times -2 = +4$

But just imagine such numbers exist, because we want them.

Let's talk some more about imaginary numbers ...

The "unit" imaginary number (like 1 for Real Numbers) is i, which is the square root of -1

When we square i we get -1

 $i^2 = -1$ 

Examples of Imaginary Numbers:

3i 1.04i -2.8i 3i/4  $(\sqrt{2})i$  1998i

And we keep that little "i" there to remind us we still need to multiply by  $\sqrt{-1}$ 

**Complex Numbers** 

When we combine a Real Number and an Imaginary Number we get a Complex Number:

Examples:

1 + i 39 + 3i 0.8 - 2.2i  $-2 + \pi i$   $\sqrt{2} + i/2$ 

Can a Number be a Combination of Two Numbers?

Can we make a number from two other numbers? Sure we can!

We do it with <u>fractions</u> all the time. The fraction  $\frac{3}{8}$  is a number made up of a 3 and an 8. We know it means "3 of 8 equal parts".

Well, a Complex Number is just two numbers added together (a Real and an Imaginary Number).

Either Part Can Be Zero

So, a Complex Number has a real part and an imaginary part.

But either part can be 0, so all Real Numbers and Imaginary Numbers are also Complex Numbers.

Complex Number	Real Part	Imaginary Part	
3 + 2i	3	2	
5	5	0	Purely Real
6i	0	-6	Purely Imaginary

Complex does not mean complicated.

It means the two types of numbers, real and imaginary, together form a complex, just like a building complex (buildings joined together).

A Visual Explanation

You know how the number line goes left-right?

Well let's have the imaginary numbers go up-down:

And we get the Complex Plane

A complex number can now be shown as a point:

The complex number 3 + 4i

#### Properties

The letter z is often used for a complex number:

z = a + bi

- z is a Complex Number
- a and b are <u>Real Numbers</u>
- i is the unit imaginary number =  $\sqrt{-1}$

we refer to the real part and imaginary part using Re and Im like this:

Re(z)

a

 $\tan^{-1}(b/a)$ 

Im(z) = b

The <u>conjugate</u> (it changes the sign in the middle) of z is shown with a star:

 $z^* = a - bi$ 

We can also use angle and distance like this (called polar form):

So the complex number 3 + 4i can also be shown as distance 5 and angle 0.927 radians. To convert from one form to the other use <u>Cartesian to Polar conversion</u>.

=

=

The magnitude of z is:

 $|\mathbf{z}| = \sqrt{a^2 + b^2}$ 

And the angle of z, also called Arg(z) is:

Arg(z)

(for a>0)

Example: z = 3 + 4i

z is a Complex Number

- 3 and 4 are Real Numbers
- Re(z) = 3
- Im(z) = 4
- $|z| = \sqrt{3^2 + 4^2} = 5$
- $\operatorname{Arg}(z) = \tan^{-1}(4/3) = 0.927... \text{ radians}$
- $z^* = 3 4i$

Adding

To add two complex numbers we add each part separately:

(a+bi) + (c+di) = (a+c) + (b+d)i

Example: add the complex numbers 3 + 2i and 1 + 7i

- add the real numbers, and
- add the imaginary numbers:

(3	+	2i)		+	(1	+	7i)
=	3	+	1	+	(2	+	7) <i>i</i>
= 4 + 9i							
Let's try and	other:						
Example: a	dd the comple	ex numbers 3	+5i and	d 4 – 3 <i>i</i>			
(3	+	5 <i>i</i> )		+	(4	_	3 <i>i</i> )
=	3	+	4	+	(5	_	3) <i>i</i>
= 7 + 2 <i>i</i>							

Multiplying

To multiply complex numbers:

Each part of the first complex number gets multiplied by each part of the second complex number

Just use "FOIL", which stands for "Firsts, Outers, Inners, Lasts" (see <u>Binomial Multiplication</u> for more details):

- Firsts:  $a \times c$
- Outers:  $a \times di$
- Inners:  $bi \times c$
- Lasts:  $bi \times di$

 $(a+bi)(c+di) = ac + adi + bci + bdi^2$ 

Like this:

Example: 
$$(3 + 2i)(1 + 7i)$$

$$(3+2i)(1+7i) = 3 \times 1 + 3 \times 7i + 2i \times 1 + 2i \times 7i$$

 $= 3 + 21i + 2i + 14i^2$ 

= 3 + 21i + 2i - 14 (because  $i^2 = -1$ )

= -11 + 23i

And this:

Example:  $(1 + i)^2$  $(1 + i)(1 + i) = 1 \times 1 + 1 \times i + 1 \times i + i^2$ 

= 1 + 2i - 1 (because  $i^2 = -1$ ) = 0 + 2iBut There is a Quicker Way! Use this rule: (a+bi)(c+di) = (ac-bd) + (ad+bc)iExample:  $(3 + 2i)(1 + 7i) = (3 \times 1 - 2 \times 7) + (3 \times 7 + 2 \times 1)i = -11 + 23i$ Why Does That Rule Work? It is just the "FOIL" method after a little work: Start with: (a+bi)(c+di) FOIL method:  $ac + adi + bci + bdi^2$  $i^2 = -1$ : ac + adi + bci - bdGather like terms: (ac - bd) + (ad + bc)iAnd there we have the (ac - bd) + (ad + bc)i pattern. This rule is certainly faster, but if you forget it, just remember the FOIL method. Let us try i<sup>2</sup> Just for fun, let's use the method to calculate  $i^2$ Example: i<sup>2</sup> We can write i with a real and imaginary part as 0 + i $i^2 = (0 + i)^2$ = (0 + i)(0 + i) $= (0 \times 0 - 1 \times 1) + (0 \times 1 + 1 \times 0)i$ = -1 + 0i= -1And that agrees nicely with the definition that  $i^2 = -1$ So it all works wonderfully! Learn more at Complex Number Multiplication. Conjugates

We will need to use conjugates in a minute!

A <u>conjugate</u> is where we change the sign in the middle.

A conjugate can be shown with a little star, or with a bar over it:

Example:

$$5 - 3i = 5 + 3i$$

Dividing

The conjugate is used to help complex division.

The trick is to multiply both top and bottom by the conjugate of the bottom.

Example: Do this Division:

2 + *3i*4 - 5i

Multiply top and bottom by the conjugate of 4 - 5i:

 $2 + 3i4 - 5i \times 4 + 5i4 + 5i = 8 + 10i + 12i + 15i^{2}16 + 20i - 20i - 25i^{2}$ 

Now remember that  $i^2 = -1$ , so:

= 8 + 10i + 12i - 1516 + 20i - 20i + 25

Add Like Terms (and notice how on the bottom 20i – 20i cancels out!):

$$= -7 + 22i41$$

Lastly we should put the answer back into a + bi form:

$$= -741 + 2241i$$

DONE!

Yes, there is a bit of calculation to do. But it can be done.

Multiplying By the Conjugate

There is a faster way though.

In the previous example, what happened on the bottom was interesting:

$$(4-5i)(4+5i) = 16 + 20i - 20i - 25i^2$$

The middle terms (20i - 20i) cancel out:

$$(4-5i)(4+5i) = 16 - 25i^2$$

Also  $i^2 = -1$ :

(4-5i)(4+5i) = 16+25

And 16 and 25 are (magically) squares of the 4 and 5:

$$(4-5i)(4+5i) = 4^2 + 5^2$$

Which is really quite a simple result. The general rule is:

 $(a + bi)(a - bi) = a^2 + b^2$ 

That can save us time when we do division, like this:

Example: Let's try this again

2 + *3i*4 - 5i

Multiply top and bottom by the conjugate of 4 - 5i:

 $2 + 3i4 - 5i \times 4 + 5i4 + 5i = 8 + 10i + 12i + 15i^{2}16 + 25$ 

$$= -7 + 22i41$$

And then back into a + bi form:

= -741 + 2241i

#### Notation

We often use z for a complex number. And Re() for the real part and Im() for the imaginary part, like this:

#### !

The beautiful Mandelbrot Set (pictured here) is based on Complex Numbers.

It is a plot of what happens when we take the simple equation  $z^2+c$  (both complex numbers) and feed the result back into z time and time again.

The color shows how fast  $z^2+c$  grows, and black means it stays within a certain range.

#### Real Number!

<u>Imaginary Numbers</u> when squared give a negative result.

Normally this doesn't happen, because:

- when we square a positive number we get a positive result, and
- when we square a negative number we also get a positive result (because <u>a negative times</u> a negative gives a positive), for example  $-2 \times -2 = +4$

But just imagine such numbers exist, because we want them.

Let's talk some more about imaginary numbers ...

The "unit" imaginary number (like 1 for Real Numbers) is i, which is the square root of -1

When we square i we get -1  $i^2 = -1$ Examples of Imaginary Numbers: 3i 1.04i -2.8i 3i/4 ( $\sqrt{2}$ )i 1998i

And we keep that little "i" there to remind us we still need to multiply by  $\sqrt{-1}$ 

**Chapter-IX** 

# **VECTOR DIFFERENTIATION**

Dr.S.Subramanian Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (PRIST) Vallam, Thanjavur, Tamilnadu, South India

#### **Vector Calculus**

Vector Calculus is a branch of mathematics that deals with the operations of calculus i.e. differentiation and integration of vector field usually in a 3 Dimensional physical space also called Euclidean Space. The applicability of Vector calculus is extended to partial differentiation and multiple integration. Vector Field refers to a point in space that has magnitude and direction. These Vector Fields are nothing but Vector Functions. Vector calculus is also known as vector analysis.

The vector fields are the vector functions whose domain and range are not dimensionally related to each other. The branch of Vector Calculus corresponds to the multivariable calculus which deals with partial differentiation and multiple integration. This differentiation and integration of vector is done for a quantity in 3D physical space represented as R3. For n-dimensional space, it is represented as Rn.

Read in Detail: Calculus in Maths

#### **Vector Calculus Definition**

Vector calculus, also known as vector analysis or vector differential calculus, is a branch of mathematics that deals with vector fields and the differentiation and integration of vector functions

Vector Calculus often called Vector Analysis deals with vector quantities i.e. the quantities that have both magnitude as well as direction. Since we know that Vector Calculus deals with differentiation and integration of functions, there are three types of integrals dealt with in Operation in Vector The different operations performed with vector quantities are tabulated below with their notation and illustration.

Operation	Notation	Illustration
Vector Addition	r1 + r2	Addition of two vectors gives a vector
Scalar Multiplication	q.r1	Multiplying a vector 'r1' with scalar 'q' result in a vector
Dot Product	$r1 \cdot r2$	Dot product of two vectors gives a scalar
Cross Product	r1 X r2	Cross product of two vectors gives a vector
Scalar Triple Product	r1 · (r2 X r3)	Dot Product of Cross product of two vectors

Vector Triple Product r1 . (r2 × r3) Cross Product of Cross Product of two Vectors

#### **Divergence and Curl**

Divergence and Curl are two important operators used in Vector Calculus. Divergence is a scalar operator which tells about the behaviour of a function towards or away from a point. Curl is a vector operator which tells about the behaviour of a function around a point. The vector operator is represented by  $\nabla$  which accounts for the partial differentiation of the vector field. The Vector Differential Operator ( $\nabla$ ) also called Nabla is expressed as  $\nabla = \partial/\partial x \mathbf{i} + \partial/\partial y \mathbf{j} + \partial/\partial z \mathbf{k}$ .

#### **Divergence of Vector**

If a vector field is given by f(x,y,z) = fxi + fyj + fzk then its divergence is given by taking the scalar of the vector operator is given by

$$\begin{split} &\operatorname{div}(f) = \nabla .f(x,y,z) = (\partial/\partial x \ i + \partial/\partial y \ j + \partial/\partial z \ k) \cdot (fxi + fyj + fzk \ ) \\ &\Rightarrow \nabla .f(x,y,z) = \partial x/\partial x + \partial y/\partial y + \partial z/\partial z. \end{split}$$

#### Curl of Vector

If a vector field is given by f(x,y,z) = fxi + fyj + fzk then its curl is given by taking the vector of the vector operator

 $\nabla \times f(x,y,z) = (\partial/\partial x i + \partial/\partial y j + \partial/\partial z k) \times (fxi + fyj + fzk)$   $\Rightarrow \nabla \times f(x,y,z) = | ijk\partial/\partial x\partial/\partial y\partial/\partial z fxfyfz|$  $i\partial/\partial x fxj\partial/\partial y fyk\partial/\partial z fz$ 

 $\Rightarrow \nabla \times f(x,y,z) = (\partial z/\partial y - \partial y/\partial z)i + (\partial x/\partial z - \partial z/\partial x)j + (\partial y/\partial x - \partial x/\partial y).$ Gradient of Scalar The gradient of a scalar field F is given by grad(F) or  $\nabla$  F. It gives the measurement of the rate and direction of a scalar-valued function. In the Cartesian system, the gradient of a scalar-valued function is given by

 $\nabla F = (\partial/\partial x i + \partial/\partial y j + \partial/\partial z k)F = \partial/\partial x i + \partial/\partial y j + \partial/\partial z k$ 

# **Vector Calculus Formulas**

For a vector field given as F(x,y,z) = p(x,y,z)i + q(x,y,z)j + r(x,y,z)k. The following formulas are given.

Fundamental Theorem of Line Integral

if  $F = \nabla \Phi$  and Curve C has A and B endpoints then its line integral is given as

 $\int cF.dr = \Phi(B) - \Phi(A)$ 

# **Circulation Curl Form**

There are two theorems under Circulation Curl Form, namely the Green theorem and Stokes theorem.

**Green Theorem**: If D is the region bounded by curve C then,  $\oint cF.dr = \iint D(\partial Q/\partial x - \partial P/\partial y) dA$ 

Stoke's Theorem: For a surface S bounded by curve C stokes theorem given by

 $\oint cF.dr = \iint S(\nabla X F)dS$ 

# Flux Divergence Theorem

The Flux Divergence Form of Green's Theorem is given as  $\iint D\nabla \cdot F \, dA = \int F \cdot n \, ds$ 

The Flux Divergence Form of Stoke's Theorem is given as  $\iiint \nabla \cdot F \, dV = \int F \cdot R \, d\sigma$ 

# **Vector Calculus Identities**

The list of Vector Calculus Identities have been tabulated under three categories.

# **Gradient Function Identities**

Here are some common gradient function identities:

# 1. Gradient of a Constant Multiple

 $\nabla(cf(x)) = c\nabla f(x)$ 

# 2. Gradient of a Sum

 $\nabla(f(x) + g(x)) = \nabla f(x) + \nabla g(x)$ 

# 3. Gradient of a Product

 $\nabla(f(x)^*g(x)) = f(x)^*\nabla g(x) + g(x)^*\nabla f(x)$ 

# 4. Gradient of a Quotient

 $\nabla(f(x)/g(x)) = (g(x)*\nabla f(x) - f(x)*\nabla g(x)) / g(x)^2$ 

# 5. Chain Rule

 $\nabla(f(g(x))) = f'(g(x))^* \nabla g(x)$ 

# 6. Gradient of a Dot Product

 $\nabla(\mathbf{x}^{\mathbf{x}}\mathbf{T}^{*}\mathbf{a}) = \mathbf{a}$ 

# 7. Gradient of a Quadratic Form

 $\nabla$ (x^TAx) = 2Ax (assuming A is symmetric)

# 8. Gradient of the Logarithm

 $\nabla(\log(f(x))) = \nabla f(x) / f(x)$ 

# 9. Gradient of the Exponential

 $\nabla(\exp(f(x))) = \exp(f(x))^* \nabla f(x)$ 

These identities are useful for computing gradients in various optimization and machine learning algorithms.

Would you like me to explain any of these identities or provide examples?

divergence function identities

Here are some common divergence function identities:

# 1. Divergence of a Constant Multiple

 $\nabla \cdot (\mathbf{c}F) = c \nabla \cdot \mathbf{F}$ 

2. Divergence of a Sum

 $\nabla \cdot (F + G) = \nabla \cdot F + \nabla \cdot G$ 

# **3. Divergence of a Product**

 $\nabla \cdot (\mathbf{f}F) = f \nabla \cdot \mathbf{F} + \nabla \mathbf{f} \cdot \mathbf{F}$ 

# 4. Divergence of a Gradient

 $\nabla \cdot (\nabla f) = \nabla^2 f$  (Laplace operator)

# 5. Divergence of a Curl

 $\nabla \cdot (\nabla \times F) = 0$ 

# 6. Curl of a Gradient

 $\nabla \times (\nabla f) = 0$ 

7. Vector Laplacian

 $\nabla^2 F = \nabla (\nabla \cdot F) - \nabla \times (\nabla \times F)$ 

# 8. Divergence of a Tensor

 $\nabla \cdot (T) = (\partial Ti, j/\partial xj)e_i$  (where T is a tensor,  $e_i$  are basis vectors)

# 9. Product Rule for Divergence

 $\nabla \cdot (F \times G) = G \cdot (\nabla \times F) - F \cdot (\nabla \times G)$ 

# **10. Divergence Theorem**

 $\iiint_V (\nabla \cdot F) \, dV = \iint_S F \cdot n \, dS \text{ (where V is a volume, S is its surface, n is the normal vector)}$ 

These identities are useful in various fields, including:

- Fluid dynamics
- Electromagnetism
- Continuum mechanics
- Vector calculus

# **Curl function identities**

Here are some common curl function identities:

# **1. Curl of a Constant Multiple**

 $\nabla \times (\mathbf{c}F) = c\nabla \times \mathbf{F}$ 

# 2. Curl of a Sum

 $\nabla \times (F + G) = \nabla \times F + \nabla \times G$ 

# 3. Curl of a Gradient

 $\nabla \times (\nabla f) = 0$ 

4. Curl of a Curl

 $\nabla \times (\nabla \times F) = \nabla (\nabla \cdot F) - \nabla^2 F$ 

# **5. Vector Triple Product**

 $\nabla \times (F \times G) = G \cdot (\nabla \cdot F) - F \cdot (\nabla \cdot G) + (G \cdot \nabla)F - (F \cdot \nabla)G$ 

# 6. Scalar Triple Product

 $F \cdot (G \times H) = G \cdot (H \times F) = H \cdot (F \times G)$ 

# 7. Curl of a Cross Product

 $\nabla \times (F \times G) = G \cdot (\nabla \cdot F) - F \cdot (\nabla \cdot G) + (G \cdot \nabla)F - (F \cdot \nabla)G$ 

# 8. Stokes' Theorem

 $\iint_S (\nabla \times F) \cdot n \, dS = \int_C F \cdot dI \text{ (where S is a surface, C is its boundary, n is the normal vector)}$ 

# 9. Integral Form of Stokes' Theorem

 $\iint_S (\nabla \times F) \cdot n \, dS = \iiint_V (\nabla \cdot (\nabla \times F)) \, dV \text{ (where V is a volume, S is its surface)}$ 

#### 10. Helmholtz Decomposition

 $F = \nabla f + \nabla \times G$  (decomposing a vector field into gradient and curl components) These identities are useful in various fields, including:

- Electromagnetism
- Fluid dynamics
- Continuum mechanics
- Vector calculus

Would you like me to explain any of these identities or provide examples?

Laplace function identities

Here are some common Laplace function identities:

#### **Differential Identities**

- 1. Laplace's Equation:  $\nabla^2 f = 0$
- 2. Laplacian of a Sum:  $\nabla^2 (f + g) = \nabla^2 f + \nabla^2 g$
- 3. Laplacian of a Product:  $\nabla^2(fg) = f\nabla^2 g + 2\nabla f \cdot \nabla g + g\nabla^2 f$
- 4. Laplacian of a Gradient:  $\nabla^2 (\nabla f) = \nabla (\nabla \cdot \nabla f)$
- 5. Vector Laplacian:  $\nabla^2 F = \nabla (\nabla \cdot F) \nabla \times (\nabla \times F)$

#### **Integral Identities**

- 1. Gauss's Law:  $\iiint_V \nabla^2 f \, dV = \iint_S (\partial f / \partial n) \, dS$
- 2. Green's First Identity:  $\iiint V (f \nabla^2 g + \nabla f \cdot \nabla g) dV = \iint S f(\partial g / \partial n) dS$
- 3. Green's Second Identity:  $\iiint V (f \nabla^2 g g \nabla^2 f) dV = \iint S (f(\partial g/\partial n) g(\partial f/\partial n)) dS$
- 4. Poisson's Equation:  $\nabla^2 f = -4\pi G\rho$  (fundamental equation of gravity)

#### **Special Functions**

- 1. Laplace of r^n:  $\nabla^2(r^n) = n(n-1)r^n(n-2)$
- 2. Laplace of exp(ax):  $\nabla^2(exp(ax)) = a^2 exp(ax)$
- 3. Laplace of  $\log(r)$ :  $\nabla^2(\log(r)) = -2\pi\delta(r)$  (2D)

#### **Properties**

- 1. Linearity:  $\nabla^2(af + bg) = a\nabla^2 f + b\nabla^2 g$
- 2. Commutativity:  $\nabla^2(fg) = (\nabla^2 f)g + f(\nabla^2 g) + 2\nabla f \cdot \nabla g$

These identities are useful in various fields, including:

• Electromagnetism

- Gravity
- Fluid dynamics
- Continuum mechanics
- Quantum mechanics
- Partial differential equations

#### Vector Calculus Applications

Vector Calculus or vector analysis has a number of applications in the real world:

- Navigation
- Sports
- Partial differential equation
- Three-dimensional geometry
- Used in heat transfer

#### Related Resources,

- Implicit Differentiation in Calculus
- How to find Derivatives?
- Vector Algebra

Solved Examples on Vector Calculus

Example 1: If F(x,y,z) = 3xy2 - y2z3 then find gradF or  $\nabla$  f.

Solution:

 $\nabla \mathbf{f} = (\partial/\partial \mathbf{x} \mathbf{i} + \partial/\partial \mathbf{y} \mathbf{j} + \partial/\partial \mathbf{z} \mathbf{k})(3\mathbf{x}\mathbf{y}\mathbf{2} - \mathbf{y}\mathbf{2}\mathbf{z}\mathbf{3})$  $\Rightarrow \nabla \mathbf{f} = \partial/\partial \mathbf{x}(3\mathbf{x}\mathbf{y}\mathbf{2} - \mathbf{y}\mathbf{2}\mathbf{z}\mathbf{3})\mathbf{i} + \partial/\partial \mathbf{y}(3\mathbf{x}\mathbf{y}\mathbf{2} - \mathbf{y}\mathbf{2}\mathbf{z}\mathbf{3})\mathbf{j} + \partial/\partial \mathbf{z} (3\mathbf{x}\mathbf{y}\mathbf{2} - \mathbf{y}\mathbf{2}\mathbf{z}\mathbf{3})\mathbf{k}$ 

 $\Rightarrow \nabla f = 3y2i + (6xy - 2yz3) + (-3y2z2)k$ 

Example 2: Find the div(F) or  $\nabla \cdot F$ , if F = xz2 i - 2y2z3 j + x2yz k

Solution:

$$div(F) = \nabla \cdot F = (\partial/\partial x \ i + \partial/\partial y \ j + \partial/\partial z \ k) \cdot (xz2 \ i - 2y2z3 \ j + x2yz \ k)$$

$$\Rightarrow \nabla \cdot \mathbf{F} = \partial/\partial \mathbf{x}(\mathbf{x}\mathbf{z}\mathbf{2}) + \partial/\partial \mathbf{y}(-2\mathbf{y}\mathbf{2}\mathbf{z}\mathbf{3}) + \partial/\partial \mathbf{z}(\mathbf{x}\mathbf{2}\mathbf{y}\mathbf{z})$$

$$\Rightarrow \nabla \cdot \mathbf{F} = \mathbf{z}2 - 4\mathbf{y}\mathbf{z}3 + \mathbf{x}2\mathbf{y}$$

Example 3: Find curl F i.e.  $\nabla \times f$  if F = xz2 i - 2x3yz j + 2yz4 k

Solution:

 $\nabla \times \mathbf{f} = (\partial/\partial x \mathbf{i} + \partial/\partial y \mathbf{j} + \partial/\partial z \mathbf{k}) \times (\mathbf{x} \mathbf{z} \mathbf{2} \mathbf{i} - 2\mathbf{x} \mathbf{3} \mathbf{y} \mathbf{z} \mathbf{j} + 2\mathbf{y} \mathbf{z} \mathbf{4} \mathbf{k})$ 

 $\Rightarrow \nabla \times f = | ijk\partial/\partial x\partial/\partial y\partial/\partial zxz22x3yz2yz4|$ 

 $\Rightarrow \nabla \times f = \left[\frac{\partial}{\partial y(2yz4)} - \frac{\partial}{\partial z(-2x3yz)}\right]i + \left[\frac{\partial}{\partial z(xz2)} - \frac{\partial}{\partial x(2yz4)}\right]j + \left[\frac{\partial}{\partial x(-2x3yz)} - \frac{\partial}{\partial y(-2x3yz)}\right]i$  $\partial/\partial y(xz2)]k$  $\Rightarrow \nabla \times \mathbf{f} = [2z4 + 2x3y]\mathbf{i} + [2xz - 0]\mathbf{j} + [-6x2yz - 0]\mathbf{k}$  $\Rightarrow \nabla \times \mathbf{f} = (2\mathbf{z}4 + 2\mathbf{x}3\mathbf{y})\mathbf{i} + (2\mathbf{x}\mathbf{z})\mathbf{j} - (6\mathbf{x}2\mathbf{y}\mathbf{z})\mathbf{k}$ Solved Examples 1. Calculate the gradient of  $f(x,y,z) = x^2y + yz^3 + xz$ Solution:  $\nabla f = (\partial f / \partial x, \partial f / \partial y, \partial f / \partial z)$  $\partial f / \partial x = 2xy + z$  $\partial f/\partial y = x^2 + z^3$  $\partial f/\partial z = 3yz^2 + x$  $\nabla$  f = (2xy + z, x^2 + z^3, 3yz^2 + x) 2. Find the divergence of  $F(x,y,z) = (x^{2} + y)i + (y^{2} + z)j + (z^{2} + x)k$ Solution: div  $F = \nabla \cdot F = \partial F x / \partial x + \partial F y / \partial y + \partial F z / \partial z$  $\partial F x / \partial x = 2x$  $\partial \mathbf{F} \mathbf{y} / \partial \mathbf{y} = 2\mathbf{y}$  $\partial F z/\partial z = 2z$ div F = 2x + 2y + 2z3. Calculate the curl of F(x,y,z) = (yz)i + (xz)j + (xy)kSolution:  $\operatorname{curl} F = \nabla \times F = (\partial F \ z/\partial y - \partial F \ y/\partial z)\mathbf{i} + (\partial F \ x/\partial z - \partial F \ z/\partial x)\mathbf{j} + (\partial F \ y/\partial x - \partial F \ x/\partial y)\mathbf{k}$  $\partial F z/\partial y - \partial F y/\partial z = x - x = 0$  $\partial F x/\partial z - \partial F z/\partial x = y - y = 0$  $\partial F y/\partial x - \partial F x/\partial y = z - z = 0$  $\operatorname{curl} F = 0i + 0j + 0k = 0$ 4. Evaluate  $\int C (x^2 + y^2) ds$ , where C is the circle  $x^2 + y^2 = 1$ , traversed counterclockwise. Solution:

Parameterize the circle: x = cos(t), y = sin(t),  $0 \le t \le 2\pi$ 

 $ds = \sqrt{((dx/dt)^{2} + (dy/dt)^{2})} dt = \sqrt{(sin^{2}(t) + cos^{2}(t))} dt = dt$  $\int_{C} (x^{2} + y^{2}) ds = \int_{0}^{0} (2\pi) (cos^{2}(t) + sin^{2}(t)) dt = \int_{0}^{0} (2\pi) 1 dt = 2\pi$ 

5. Compute the flux of F(x,y,z) = xi + yj + zk through the surface of the sphere  $x^2 + y^2 + z^2 = a^2$ .

Solution:

Use the divergence theorem:  $\iint S F \cdot dS = \iiint V \text{ div } F dV$ 

div F = 1 + 1 + 1 = 3

Volume of sphere =  $(4/3)\pi a^3$ 

Flux =  $\iiint V 3 dV = 3 * (4/3)\pi a^3 = 4\pi a^3$ 

Practice Problems on Vector Calculus in Maths

1. Calculate the gradient of  $f(x,y,z) = \ln(x^2 + y^2 + z^2)$ 

- 2. Find the divergence of F(x,y,z) = (x2y)i + (y2z)j + (z2x)k
- 3. Compute the curl of  $F(x,y,z) = (y_2)i + (z_2)j + (x_2)k$
- 4. Evaluate the line integral  $\int_C (x + y) ds$  along the curve C: x = t,  $y = t^2$ ,  $z = t^3$ ,  $0 \le t \le 1$
- 5. Calculate the surface integral of F(x,y,z) = xi + yj + zk over the surface  $z = x^2 + y^2$ ,  $0 \le z \le 1$

6. Find the gradient of  $f(x,y) = e(x+y) \sin(xy)$ 

7. Determine if F(x,y,z) = (yz)i + (zx)j + (xy)k is conservative

8. Compute the divergence of  $F(r,\theta,\phi) = r^{-1} \cos(\phi)r + r \sin(\theta) \sin(\phi)\theta + r \cos(\theta)\phi$  in spherical coordinates

9. Find the potential function for F(x,y) = (2x + y)i + (x + 2y)j

10. Calculate the work done by F(x,y,z) = xi + yj + zk along the helix  $r(t) = (cos(t), sin(t), t), 0 \le 100$ 

 $t \le 2\pi$ 

#### Summary

Vector calculus is a branch of mathematics that deals with vector fields in two or more dimensions. It encompasses key concepts such as the gradient, which measures the rate and direction of change in scalar fields; divergence, which quantifies the "outflow" of vector fields; and curl, which describes the rotation of vector fields. The field also includes techniques for integration over curves (line integrals) and surfaces (surface integrals). Fundamental theorems like the Divergence Theorem and Stokes' Theorem connect these concepts, while the study of conservative fields and potential functions provides insights into path independence and field representations. These tools and concepts are essential for analyzing complex systems in physics,

engineering, and other scientific disciplines, allowing for the mathematical description of phenomena involving fluid dynamics, electromagnetism, and other multidimensional processes. Vector Calculus that are

- Line Integral
- Surface Integral
- Volume Integral

# **Chapter -X**

#### **VECTOR INTEGRATION**

**Dr.S.Subramanian** 

Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### Line Integral

Line Integral in mathematics is the integration of a function along the line of the curve. The function can be a scalar or vector whose line integral is given by summing up the values of the field at all points on a curve weighted by some scalar function on the curve. Line Integral is also called Path Integral and is represented by  $\Phi = \int_L f$ . Line Integral has got its application in physics. For Example, Work Done by Force is along a path given as  $W = \int_L F(s) ds$  because we know that work done is given as the product of force and distance covered.

#### **Surface Integral**

Surface Integral in mathematics is the integration of a function along the whole region or space that is not flat. In Surface integral, the surfaces are assumed of small points hence, the integration is given by summing up all the small points present on the surface. The surface integral is equivalent to the double integration of a line integral. Surface Integral has got its application in Electromagnetism and many more branches of physics where the vector function is spread over the surface. Surface Integral is represented as  $\iint_{s} f(x,y) dA$ .

#### Learn more about **Double Integral**.

#### **Volume Integral**

A volume integral, also known as a triple integral, is a mathematical concept used in calculus and vector calculus to calculate the volume of a three-dimensional region within a space. It is an extension of the concept of a definite integral in one dimension to three dimensions.

Mathematically, the volume integral of a scalar function f(x, y, z) over a region **R** in threedimensional space is denoted as:

 $\iint \hat{R}f(x,y,z) \, dV \iint \hat{R}f(x,y,z) \, dV$ 

# Where

- **dV** represents an infinitesimal volume element, and
- Integral is taken over region **R**.

# **FAQs on Vector Calculus**

# What is Vector Calculus?

Vector Calculus is branch of mathematics that deals with the differentiation and integration of Vector Function.

# What is Gradient in Vector Calculus?

Gradient in Vector Calculus is rate of change of a scalar valued function in a vector space.

# What is Divergence in Vector Calculus?

Divergence in Vector Calculus is the scalar of the vector operation of a function.

# What is Curl in Vector Calculus?

Curl in Vector Calculus is vector of vector operation of a function.

# What is Line Integral?

Line integral is the integration of the function along the lines of the curve.

# What is Surface Integral?

Surface Integral of the function is the integration of the function over the whole surface.

# Who invented Vector Calculus?

The concept of Vector Calculus was given by J. Willard Gibbs and Oliver Heaviside.

# Is Vector Calculus same as vector analysis?

Yes, Vector Calculus is also called vector analysis.

# DISCRETE MATHEMATICS

Edited by

**MAHESWARAN** 



# **Discrete Mathematics**

Copyright © 2021 by S.Maheswaran

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-265-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

Page Numbers
01-03
04-10
11-18
19-24
25-27

#### CHAPTER 1

#### MATRICES AND ITS APPLICATIONS

**Dr.N.Balambigai** Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### **INTRODUCTION**

The concept of matrices and their basic operations were introduced by the British mathematician Arthur Cayley in the year 1858. He wondered whether this part of mathematics will ever be used. However, after 67 years, in 1925, the German physicist Heisenberg used the algebra of matrices in his revolutionary theory of quantum mechanics. Over the years, matrices have been found as an elegant and powerful tool in almost all branches of Science and Engineering like electrical networks, graph theory, optimisation techniques, system of differential equations, stochastic processes, computer graphics, etc. Because of the digital computers, usage of matrix methods have become greatly fruitful.

Here we study certain numbers associated with a square matrix, called eigen values and certain vectors associated with them, called eigen vectors. These are useful in the study of canonical forms of a matrix such as diagonalisation and in the study of quadratic forms. The problem of determining eigen values and eigen vectors of a square matrix is called an **eigen value problem**.

#### **Example 1**

Show that the vectors (1, 2, 3), (3, -2, 1), (1, -6, -5) are linearly dependent. Solution

Let 
$$A = \begin{bmatrix} 1 & 3 & 1 \\ 2 & -2 & -6 \\ 3 & 1 & -5 \end{bmatrix}$$
 with the vectors as columns.  
then  $|A| = \begin{vmatrix} 1 & 3 & 1 \\ 2 & -2 & -6 \\ 3 & 1 & -5 \end{vmatrix}$   
=  $1 \cdot (10+6) \cdot 3(-10+18) + 1 \cdot (2+6)$   
=  $16 - 24 + 8 = 0$ 

: the vectors (1, 2, 3), (3, -2, 1) and (1, -6, -5) are linearly dependent.

#### Example 2

Show that the vectors  $X_1 = (1, 2, 3, 4)$ ,  $X_2 = (3, 1, 2, 1)$ ,  $X_3 = (1, -5, 8, -7)$  are linearly dependent and find the relation between them.

Solution

Let 
$$A = \begin{bmatrix} 1 & 2 & -3 & 4 \\ 3 & -1 & 2 & 1 \\ 1 & -5 & 8 & -7 \end{bmatrix}$$
 with the vectors as rows.

We shall use elementary row operations.

$$\therefore \qquad A \sim \begin{bmatrix} 1 & 2 & -3 & 4 \\ 0 & -7 & 11 & -11 \\ 0 & -7 & 11 & -11 \end{bmatrix} \begin{array}{c} R_2 \to R_2 - 3R_1 = R_2' \\ R_3 \to R_3 - R_1 = R_3' \\ \sim \begin{bmatrix} 1 & 2 & -3 & 4 \\ 0 & -7 & 11 & -11 \\ 0 & 0 & 0 \end{bmatrix} \begin{array}{c} R_3 \to R_3' - R_2' = R_3'' \\ R_3 \to R_3' - R_2' = R_3'' \end{array}$$

Since the maximum number of non-zero rows is 2, which is less than the number of vectors, the given vectors are linearly dependent.

The relation between them is obtained as below.

$$\begin{aligned} \mathbf{R}_3'' &= \mathbf{0} \quad \Rightarrow \quad \mathbf{R}_3' - \mathbf{R}_2' = \mathbf{0} \\ \Rightarrow \quad \mathbf{R}_3 - \mathbf{R}_1 - (\mathbf{R}_2 - 3\mathbf{R}_1) = \mathbf{0} \\ \Rightarrow \quad \mathbf{R}_3 - \mathbf{R}_2 + 2\mathbf{R}_1 = \mathbf{0} \end{aligned}$$

Since the rows are vectors, we get  $X_3 - X_2 + 2X_1 = 0$  which is the relation between the vectors.

# Note

The rows of the matrix are the given vectors. So, only row operations must be used to find the relationship between the vectors.

# CHAPTER- II SET THEORY AND ITS APPLICATIONS

Dr.N.Balambigai

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### LOGICAL EQUIVALENCES AND IMPLICATIONS - De MORGAN'S LAWS

Compound propositions that have the same truth values in all possible cases are called logically equivalent.

16 different truth tables of compound propositions involve the propositions P and Q.

**Definition:** The propositions P and Q are called logically equivalent if P Q is a tautology.

Note: The notation P = Q denotes that P and Q are logically equivalent. (or) We use P < ->Q also.

**Example 1:** Show that P is equivalent to the following formulae.

(i) ר ך P

(ii) P^ P

- (iii) P V P
- (iv) PV (P<sup>A</sup> Q)
- (v)  $P^{(v)}(PV | Q)$

1	2	3	4	5	6	7	8	9	10
P	Q	٦P	٦٦P	$P \wedge P$	PVP	PAQ	$P \vee (P \wedge Q)$	PVQ	$P \land (P \lor Q)$
Т	T	F	Т	Т	Т	Т	Т	Т	Т
т	F	F	Т	Т	Т	F	Т	Т	Т
F	Т	Т	F	F	F	F	F	Т	F
F	F	Т	F	F	F	F	F	F	F

Here the 4th, 5th, 6th, 8th, 10th columns give the truth values of the formulas. The columns 1, 4, 6, 8, 10 have the identical truth values. Hence P is equivalent to all the given formulas. **Example 2 :** Show that P is equivalent to the following formulas. (i)  $(PA Q) V (PA \gamma Q)$  (ii) (P VQ)  $\Lambda$  (PV  $_{\mathsf{T}}$  Q)

Solution :

1	2	3	4	5	6	7	8.	9
Р	Q	ΡΛQ	Q	ΡΛΊQ	$(P \land Q) \lor$ $(P \land \neg Q)$	PvQ	P∨刁Q	$(P \lor Q) \land$ $(P \lor \neg Q)$
Т	Т	Т	F	F	Т	Т	Т	Т
Т	F	F	Т	Т	Т	T	Т	Т
F	Τ.	F	F	F	F	Т	F	F
F	F	F	Т	F	F	F	Т	F

Columns 1, 6, 9 have identical truth values.

Hence  $P \leftrightarrow (P^Q) V (P^{\uparrow} Q)$ 

 $\leftrightarrow (\mathrm{PVQ})^{\wedge}(\mathrm{PV}_{\mathsf{T}}\,\mathrm{Q})$ 

**Example 3:** Show the following equivalences.

(a) 
$$\neg (P^{\wedge}Q) \leftrightarrow \neg P^{\vee} \neg Q$$
  
(b)  $\neg (P^{\vee}Q) \leftrightarrow \gamma p^{\wedge} \neg Q$   
(c)  $\neg (P \rightarrow Q) \leftrightarrow P^{\wedge} \neg Q$   
(d)  $\neg (P \leftrightarrow Q) \leftrightarrow (P^{\wedge} \neg Q) \vee (\gamma P^{\wedge}Q)$   
Solution :

1	2	3	4	5	6	7	8	9	10
P	Q	٦P	70	PAQ	](P∧Q)	PVQ	](PVQ)		
Т	Т	F	F	T	F	Т	F	F	F
Т	F	F	Т	F	Т	Т	F	T	F
F	T	Т	F	F	Т	Т	F	Т	F
F	F	Т	Т	F	Т	F	Т	Т	T
	11		12	2	13	14	15	16	17
	P->1	Q	7(P	→Q) I	N Q I	P⇔Q	](P⇔Q)	<b>PAQ</b>	$(P \land \neg Q) \lor$

PAQ)

F

T

T

F

T	F	F	T	F	F	
F	Т	Т	F	、 T	F	
т	F	F	F	Т	Т	
т	F	F	Т	F	F	

(a) Column 6 and 9 are equal.

 $\neg \ (P^{\wedge} Q) \neg \leftrightarrow P \ v \neg Q$ 

(b) Column 8 and 10 are equal.

$$\neg (P VQ) \leftrightarrow \neg p^{\wedge} \neg Q$$

- (c) Column 12 and 13 are equal.
- $(P >Q) \leftrightarrow (P ^ {} \land _{\neg} Q)$
- (d) Column 15 and 17 are equal
- $(P \rightarrow Q) \leftrightarrow (P \land \neg Q) \lor (\neg p^{\land} Q)$

# PREDICATES AND QUANTIFIERS

#### Predicates

Statements involving variables, such as "x > 4", "x = y+4", and "x + y = z", are often found in mathematical assertions and in computer programs.

These statements are neither true nor false when the values of the variables are not specified.

The statement "x is greater than 4" has two parts. The first part, the variable x, is the subject of the statement.

The second part- the predictate, "is greater than 4" - refers to a property that the subject of the statement can have.

We can denote the statement "x is greater than 4" by P(x), here P denote the predicate "is greater than 4" and x is the variable.

The statement P(x) is also said to be the value of the propositional function P at x. Once a value has been assigned to the variable x, the statement P(x) becomes a proposition and has a truth value.

When all the variables in a propositional function are assigned values, the resulting statement becomes a proposition with a certain truth value. However, there is another important way, called quantification, to create a proposition from a propositional function. Two types of quantification will be discussed here, namely, universal quantification and existential quantification. The area of logic that deals with predicates and quantifiers is called the predicate calculus.

#### **Place of variables**

Definition: A predicate refers to a property that the subject of the statement can have. A predicate is a sentence that contains a finite number specific values are substituted for the variables.

The domain of a predicate variable is the set of all values that may be substituted place of variable.

In the statement "Ravi is a painter" the part. "is a painter" is called a predicate. If it is denoted by the symbol P and 'Ravi' is denoted by the letter r, the above statement is denoted by P (r).

In the statement, "Ram is a shorter than Deivani", if "is shorter than" is denoted by s and Ram and Deivani are denoted by r and n, then this statement is denoted by S (r, n). Hence S is a 2 place predicate as two objects are needed to complete this statement. Likewise, we can have n-place predicate.

**Example 1:** Let p (x) denote the statement "x > 4". What are the truth values of P(5) and P(2) ? **Solution:** We obtain the statement P(5), by setting x = 5 in the statement "x > 4". Hence P(5), which is the statement "z > 4" is true. However, P(2), which is the statement "z > 4" is false.

**Example 2:** Let Q (x, y) denote the statement "x = y+2", what are the truth values of the prepositions Q (1, 2) and Q (2, 0).

**Solution:** To obtain Q (1, 2). Set x=1 and y = 2 in the statement Q (x, y).

Hence, Q (1, 2) is the statement "1 = 2+2", which is false.

The statement Q (2, 0) is the proposition "2 = 0+2", which is true.

**Example 3:** What are the truth values of the propositions R (1, 2, 3) and R (0, 0, 1)?

**Solution:** The proposition R (1, 2, 3) is obtained by setting x = 1, y = 2 and z

= 3 in the statement R (x, y, z). We see that R (1, 2, 3) is the statement "1 + 2 =

3", which is true. Also note that R (0, 0, 1), which is the statement "0 + 0 = 1" is false.

**Note:** In general, a statement involving the n variables  $x_1, x_2... x_n$  can be denoted by P ( $x_1, x_2, ... X_n$ ). A statement of the form P ( $x_1, x_2, ... n$ ) is the value of the propositional function p at the n-tuple ( $x_1, x_2, ... X_n$ ), and p is also called a predicate.

#### **Definition: A simple statement function :**

A simple statement function of one variable is defined to be an expression consisting of a predicate symbol and an individual variable. Such a statement function becomes a statement when the variable is replaced by the name of any object.

**Example:** If "X is a teacher" is denoted by T (x), it is a statement function. If X is replaced by John, then "John is a teacher" is a statement.

#### **Definition: Compound statement function :**

A compound statement function is obtained by combining one or more simple statement functions by logical connectives.

#### Example :

- M. (x)  $\Lambda$  H (x)
- $M(x) \rightarrow H(x)$
- $M(x) v \neg H(x)$

An extension of this idea to the statement functions of two or more variables is straight forward.

#### **Example :**

G (x, y): x is taller than y.

If both x and y are replaced by the names of objects, we get a statement. If m represents Mr.Ram and f Mr. Raj, then we have

G (m, f): Mr. Ram is taller than Mr. Raj

and G (f, m): Mr. Raj is taller than Mr. Ram

It is possible to firm statement functions of two variables by using statement functions of one variable.

#### Example :

M (x) x is a man

H (y) y is a mortal

then we may write M (x)  $\Lambda$  H (v): x is a man and y is a mortal.

# **Quantifiers :**

Certain declarative sentences involve words that indicate quantity such as "all, some, none or one". These words help determine the answer to the question "How many?". Since such words indicate quantity they are called quantifiers.

Consider the following statements.

- 1. All isosceles triangles are equiangular.
- 2. Some birds cannot fly.
- 3. Not all vegetarians are healthy persons.
- 4. There is one and only one even prime integer.
- 5. Each rectangle is a parallelogram.

After some thought, we realize that there are two main quantifiers : all and some, where some is interpreted to mean atleast one.

The quantifier "all" is called the universal quantifier, and we shall denote it by  $\forall x$ , which is an inverted. A followed by the variable x. It represents each of the following phrases, since they all have essentially the same meaning.

For all x, all x are such that

For every x, every x is such that

For each x, each x is such that

**Example:** Let P (x) be the statement "x + 1 > x". "What is the truth value of the quantification  $\forall xp(x)$ , where the universe of discourse consists of all real numbers?

**Solution:** Since P (x) is true for all real numbers x, the quantification  $\forall xp$  (x) is true.

The quantifier "some" is the existential quantifier and we shall denote it by  $\exists x$ , which is a reversed E followed by x. It represents each of the following phrases:

There exists an x such that .....

There is an x such that .....

For some x ....

There is at least one x such that...

Some x is such that ...

**Example 1**: Something is Green.

It can be denoted by  $(\exists x) (G(x))$ 

**Example 2:** Something is nor green.

It can be denoted by  $(\exists x) (\exists x) (\exists x)$ 

# **Predicate Formulas**

Let p ( $x_1$ ,  $x_2$ , ...  $X_n$ ) denote as n-place predicate formula in which the letter p is an n-place predicate and  $x_1$ ,  $x_2$  ....  $X_n$  are individual variables.

In general,  $p(x_1, x_2, ..., X_n)$  will be called an atomic formula of the statement calculus. It may be noted that our symbolism includes the atomic formulas of the statement calculus as special cases (n = 0).

The following are some examples of atomic formulas R Q (x), P(x, y) A (x, y, z) P (a, y) and A (x, a, z).

A well-formed formula of predicate calculus is obtained by using the following rules.

1. An atomic formula is a well-formed formula.

2. If A is a well-formed formula, then  $\neg$  A is a well-formed formula.

3. If A and B are well-formed formulas, then (A  $\land$  B), (A  $\lor$  B), (A $\rightarrow$ B) and (A $\leftrightarrow$  B) are also well-formed formulas.

4. If A is a well-formed formula and x is any variable, then (x) A and  $(\exists x)$  are well-formed formulas.

5. Only those formulas obtained by using rules (1) to (4) are well- formed formulas.

# CHAPTER- III NUMBER THEORY AND ITS APPLICATIONS

#### Dr.N.Balambigai

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### **Rules of inference**

#### **Definition: Inference Theory :**

The main function of logic is to provide rules of inference, or principles of reasoning. The theory associated with such rules is known as inference theory because it is concerned with the inferring of a conclusion from certain premises.

#### Definition: Valid argument or valid conclusion

If a conclusion is derived from a set of premises by using the accepted rules of reasoning, then such a process of derivation is called a deduction or a formal proof and the argument or conclusion is called a valid argument or valid conclusion.

**Note:** The method of determine whether the conclusion logically follows from the given premises by constructing the relevant truth table is called "truth table technique".

#### **Definition :**

Let A and B be two statement formulas. We say that "B logically follows from A" or "B is a valid conclusion (consequence) of the premise A" iff  $A \rightarrow B$  is a tautology, that is  $A \rightarrow B$ .

Just as the definition of implication was extended to include a set of formulas rather than a single formula, we say that from a set of premises  $\{H_1, H_2, ..., H_m\}$  a conclusion C follows logically if  $H_1 \wedge H_2 \wedge ... \wedge H_m \rightarrow C$ .

#### (I). Truth table technique :

Given a set of premises and a conclusion, it is possible to determine whether the conclusion logically follows from the given premises by constructive truth tables as follows.

(i) Let  $P_1, P_2, ..., P_n$  be all the atomic variables appearing in the premises  $H_1, H_2 ... H_m$  and the conclusion C. If all possible combinations of truth values are assigned to  $P_1, P_2, ... P_n$  and if the

truth values of  $H_1$ ,  $H_2$  ...  $H_m$  and C are entered in a table, then it is easy to see from such a table whether

 $H_1 \land H_2 \land \ldots \land H_m \mathop{\rightarrow} C$  is true.

(ii) We look for the rows in which all  $H_1$ ,  $H_2$ , ...  $H_m$  have the value T. If, for every such row, C also has the value T, then  $H_1 \wedge H_2 \wedge ... \wedge H_m \rightarrow C$  holds.

(iii) Alternatively, we may look for the rows in which C has the value F. If, in every such row, atleast one of the values of  $H_1$ ,  $H_2$ , ...  $H_m$  is F, then  $H_1 \wedge H_2 \wedge ... \wedge H_m \rightarrow C$  also holds.

	Table I : Rules of Inferen	ce
Rule of Inference	Tautology	Name
$\begin{array}{c} P \\ \underline{P \rightarrow Q} \\ \therefore Q \end{array}$	$[P \land (P \rightarrow Q)] \rightarrow Q$	Modus ponens
$ \begin{array}{c} \neg \ Q \\ \underline{P \rightarrow Q} \\ \vdots  \neg \ P \end{array} $	$[ \neg Q \land (P \rightarrow Q) ] \rightarrow \neg P$	Modus tollens
$P \rightarrow Q$ $Q \rightarrow R$ $\therefore P \rightarrow R$	$[(P \to Q) \land (Q \to R)] \to (P \to R)$	Hypothetical syllogism
$\begin{array}{c} P \lor Q \\ \hline P \\ \hline P \\ \hline \vdots Q \end{array}$	$[(P \lor Q) \land \neg P] \rightarrow Q$	Disjunctive syllogism
$\frac{P}{\therefore P \lor Q}$	$P \to (P \lor Q)$	Addition
$\frac{P \land Q}{\therefore P}$	$(P \land Q) \rightarrow P$	Simplification
$\begin{array}{c} P\\ Q\\ \therefore P \land Q \end{array}$	$[(P) \land (Q)] \twoheadrightarrow (P \land Q)$	Conjunction
$P \lor Q$ $\square P \lor R$ $\therefore Q \lor R$	$[(P \lor Q) \land ( \sqcap P \lor R)] \rightarrow (Q \lor R)$	Resolution

#### Example 1.

State which rule of inference is the basis of the following argument:
#### "It is below freezing now. Therefore, it is either below freezing or raining now".

Solution: Let P: It is below freezing now

Q: It is raining now

P/PVQ This is an argument that uses the addition rule.

Example 2.

State which rule of inference is the basis of the following argument :

"It is below freezing and raining now. Therefore, it is below freezing now."

**Solution:** Let P: It is below freezing now.

Q: It is raining now

PAQ/P This is an argument that uses the simplification rule.

#### **Rules for inferences theory**

Rule P: A premise may be introduced at any point in the derivation.

**Rule T:** A formula S may be introduced in a derivation if S is a tautologically implied by any one or more of the preceding formulas in the derivation,

**Rule CP:** If we can derive S from R and a set of premises, then we can derive  $R \rightarrow S$  from the set of premises alone.

Example 3: Demonstrate that R is a valid infernece from the premises  $P \rightarrow Q$ ,  $Q \rightarrow R$  and P. Solution :

	Step	Derivation	Rule		
{1}	(1)	$P \rightarrow Q$	P		
{2}	(2)	P	P		
{1, 2}	(3)	Q	T, (1), (2), and $I_{11}$ [P, P $\rightarrow$ Q $\Rightarrow$ Q]		
{4}	(4)	$Q \Rightarrow R$	P		
{1,2,4}	(5)	R	T, (3), (4) and $I_{11}$ [R, Q $\rightarrow$ R $\Rightarrow$ R]		

#### **Mathematical Induction**

The word induction means the method of inferring a general statement from the validity of particular cases. Mathematical induction is a technique by which one can prove mathematical statements involving positive integers.

#### **Principle of Mathematical Induction**

Let P (n) be a statement or proposition involving for all positive integrs n. Then we complete two steps.

**Basis step:** If P (1) is true.

**Inductive step:** If P (k+1) is true on the assumption that P (k) is true.

**Example 1: Prove by induction** 

 $1 + 2 + 3 + \dots + n = n(n+1)/2, n \ge 1.$ 

**Solution:** Let P (n);  $1+2+3+...+n = n (n + 1)/2, n \ge 1$ 

**Step 1:** To prove P(1) is true.

For n = 1, we have

1 = 1(1+1)/2

1 = 1

So P (1) is true.

Step 2: Assume that P (k) is true for any positive integer k.

(i.e.,) 1 + 2 + 3 + ... + k = k (k+1)/2 **Step 3: To prove : P (k+1) is true.** (i.e.,) To prove P (k+1) = (k+1) (k+2)/2 [1+2+3+...+k] + k + 1 = [k (k + 1)/2] + k + 1 = k (k+1)+2(k + 1)/2 = (k+1) (k+2)/2= (k+1) [(k+1) + 1]/2

which is P(k+1).

That is P(k+1) is true whether P(k) is true.

By the principle of mathematical induction P(n) is true for all positive integer n.

Example 2: Show that  $1^2 + 2^2 + 3^2 + ... + n^2 = n(n + 1) (2n + 1)/6$ ,  $n \ge 1$  by mathematical induction.

**Solution:** Let P (n):  $1^2 + 2^2 + 3^2 + \dots + n^2 = n (n+1) (2n+1)/6$ 

**Step 1:** To prove P (1) is true.

For n = 1

 $1^2 = 1(1+1)(2+1)/6$ 

1 = 1

So P (1) is true.

**Step 2:** Assume that P (k) is true.

(i.e.,)  $1^2 + 2^2 + 3^2 + \dots + k^2 = k (k+1) (2k+1) / 6$ 

**Step 3:** To prove P (k+1) is true.

(i.e.,) To prove P (k+1) = (k+1) (k + 2) (2k + 3)/6  $[1^2 + 2^2 + 3^2 + ... + k^2] + (k+1)^2 = [k (k+1) (2k+1)/6] + (k+1)^2$ = k (k + 1) (2k + 1) + 6(k + 1)<sup>2</sup>/6 = (k + 1) [2k<sup>2</sup> + k + 6k + 6]/6 = (k + 1) [2k<sup>2</sup> + 7k + 6]/6 = (k + 1) (k + 2) (2k + 3)/6 which is P(k+1).

That is P(k+1) is true whenever P(k) is true.

By the principle of mathematical induction P (n) is true for all positive integer n.

## Example 3 Conjecture a formula for the sum of the first n positive odd integers. Then prove by using mathematical induction.

Solution : The sums of the first n positive odd integers for

n = 1, 2, 3, 4, are 1 = 1 = 1<sup>2</sup> 1+3 = 4 = 2<sup>2</sup> 1+3+5 = 9 = 3<sup>2</sup> 1+3+5+7 = 16 = 4<sup>2</sup>

From these values it is reasonable to conjecture that the sum of the first n positive odd integers is  $n^2$ .

(i.e.,)  $1 + 3 + 5 + 7 + \dots + (2n - 1) = n^2$ .

Let P (n)  $1 + 3 + 5 + \dots + (2n - 1) = n^2$ 

**Step 1:** To prove : P (1) is true.

 $1 = 1^{2}$ 

So P (1) is true.

**Step 2:** Assume that P (k) is true.

(i.e.,)  $1 + 3 + 5 + ... + (2k - 1) = k^2$ 

**Step 3:** To prove P (k+1) is true.

(i.e.,) To prove P  $(k + 1) (k + 1)^2$ .

 $[1 + 3 + 5 + \dots (2k - 1)] + 2k + 1 = [k^2] + 2k + 1$ 

 $=(k+1)^{2}$ 

which is P (k+1)

That is P(k+1) is the whenever P(k) is true.

By the principle of mathematical induction P (n) is true.

Example 4: Show that  $1/1.2 + 1/2.3 + \dots + 1/n(n+1) = n/n+1$ 

Solution: Let P (n):  $1/1.2 + 1/2.3 + \ldots + 1/n(n+1) = n/n+1$ 

**Step 1:** To prove P (1) is true.

(i.e.,) 1/1.2 = 1/1+1

1/2 = 1/2

Hence P (1) is true.

**Step 2:** Assume that P (k) is true.

(i.e.,)  $1/1.2 + 1/2.3 + \ldots + 1/k(k+1) = k/k+1$ **Step 3:** To prove P (k+1) is true. (i.e.,) P (k+1) = k+1/(k+1)+1 = k+1/k+2[1/1.2 + 1/2.3 + ... + 1/k(k+1)] + 1/(k+1)(k+2) = [k/k+1] + [1/(k+1)(k+2)]= k(k+2)+1 / (k+1)(k+2) $= k^{2}+2k+1/(k+1)(k+2)$  $= (k+1)^2 / (k+1)(k+2) = k+1 / k+2$ which is P(k+1)That is P(k+1) is true whenever P(k) is true. By the principle of mathematical induction P (n) is true. Example 5: Prove by mathematical induction that  $2^n > n$  for all  $n \in N$ . (or)  $n < 2^n$  for all positive integers n. [A.U N/D 2012] Solution: Let P (n):  $n < 2^n$ **Step 1:** To prove P (1) is true.  $1 < 2^{1}$  $\Rightarrow 1 < 2$ Hence P (1) is true. **Step 2:** Assume that P (k) is true. (i.e.,)  $k < 2^k$ **Step 3:** To prove : P (k+1) is true (i.e.,) To prove :  $P(k + 1) = 2^{k+1}$ (i.e.,)  $k < 2^k$  $\Rightarrow$  k + 1 < 2<sup>k</sup>+1  $\Rightarrow k + 1 < 2^k + 2^k \qquad ['.' \ 1 \le 2^k]$  $\Rightarrow$  k + 1 < 2 (2<sup>k</sup>)  $\Rightarrow$  k + 1 < 2<sup>k</sup> +1 which is P(k + 1)That is P(k+1) is true whenever P(k) is true. Example 6: Show that  $a^n - b^n$  is divisible by (a - b) for all,  $n \in N$ **Solution:** Let P (n)  $(a^n - b^n)$  is divisible by (a - b)Step 1: To prove P (1) is true.  $(a^1 - b^1) = (a - b)$  is divisible by (a - b)

Hence P (1) is true. Step 2: Assume that P (k) is true. Let  $(a^k - b^k) = c (a - b)$   $\Rightarrow a^k = b^k + c (a - b) \dots (1)$ Now  $a^{k+1} - b^{k+1} = a^k a - b^k b$   $= a[b^k + c (a - b)] - b^k b by (1)$   $= ab^k + ac (a - b) - b^k b$   $= b^k (a - b) + ac (a - b)$   $= (a - b) (b^k + ac)$ which is divisible by (a - b) (ie.,) That is P (k+1) is true whenever P (k) is true. By the principle of mathematical induction, P (n) is true for all n  $\in$  N.

#### **CHAPTER-IV**

#### **CLASSIFICATION OF RELATIONS**

#### Dr.N.Balambigai

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

#### **The Relations**

It states that if there are more pigeons (objects) than the pigeonholes (boxes), then some pigeonhole (box) must contain two or more pigeons (objects). The pigeohole principle is also called the Dirichlet drawer principle or Shoe box principle.

The pigeonhole principle itself is trivial but when applied cleverly, it can yield non trivial results. The pigeonhole principle is sometime useful in counting methods.

To apply the principle one has to decide objects (pigeons) and categories of the desired characteristic (pigeonholes) and be able to count the number of objects and the number of pigeonholes. Ofcourse, this principle also applies to other objects besides pigeons and pigeonholes.

Note that the pigeonhole. principle tells us nothing about how to locate the pigeonhole that contains two or more pigeons. It only asserts the existence of a pigeonhole containing two or more pigeons.

#### Theorem The pigeonhole principle

### If k is a positive integer and k + 1 or more objects are placed into k boxes, then there is atleast one box containing two or more of the objects.

**Proof :** We prove this principle by the method of contradiction. Suppose that none of the k boxes contains more than one object. Hence, the total number of objects would be atmost k. This is a contradiction, since there are atleast k + 1 objects.

#### Corollary

A function f from a set with k + 1 or more elements to a set with k elements is not one-to-one.

**Proof:** Suppose each element y in the codomain of f we have a box that contains all elements x of the domain of f such that f(x)=y. Since the domain contain k + 1 or more elements and the co-

domain contains only k elements, by the pigeonhole principle one of these boxes contains two or more elements x of the domain. This means that f cannot be one-to-one.

## Theorem: If n pigeons are assigned to m pigeonholes, and m<n, then atleast one pigeonhole contains two or more pigeons.

**Proof :** We will prove this principle using a proof by contradiction. Suppose each pigeonhole contains atmost one pigeon. Then atmost m pigeons have been assigned. But since m < n, not all pigeons have been assigned pigeonholes. This is a contradiction.

Atleast one pigeonhole contains two or more pigeons.

#### Theorem: The generalization / extension of the pigeonhole principle.

If k pigeons are assigned to n pigeonholes, then one of the pigeonholes must contain at least [k-1/n] + 1 pigeons.

**Proof:** If possible, suppose that none of the pigeonholes contains more than [k-1/n]pigeons. Then there will be atmost

 $n[k-1/n] \le n k-1/n = k - 1$  pigeons.

This is a contradiction to the assumption that there are n pigeons.

### Theorem: If n objects are placed into k boxes, then there is atleast one box containing atleast n/k objects.

**Proof:** Assume that none of the boxes contains more than [n/k] - 1 objects.

The total number of objects is atmost.

k[(n/k) - 1] < k[(n/k + 1) - 1] = n, where the inequality [n/k] < [n/k] + 1 has been used. This is a contradiction, because there are a total of n objects.

#### Example 1: Give two examples based on pigeonhole principle.

#### Solution :

1. Among any group of 367 people, there must be atleast two with the same birthday, because there are only 366 maximum possible birth days.

2. In any group of 27 English words, there must be atleast two that starts with the same letter, since there are 26 letters in English alphabet.

Example 2: Show that among 13 children, there are atleast two children who were born in the same month.

**Solution:** Let us assume that 13 children as pigeons and the 12 months (January, ... December) as the pigeonholes then by the pigeonhole principle there will be atleast two children who were born in the same month.

### Example 3 Show that if any four numbers from 1 to 6 are chosen, then two of them will add to 7.

Solution: Let us form 3 sets containing two numbers whose sum is 7.

 $A = \{1, 6\}, B = \{2, 5\}, C = \{3, 4\}.$ 

The four numbers that will be chosen to the set that contains it.

As there are only 3 sets, two numbers that there chosen is from the set whose sum is 7.

### Example 4: Show that among any group of five (not necessarily consecutive) integers, there are two with the same remainder when divided by 4.

**Solution:** Take any group of five integers. When these are divided by 4 each have some remainder. Since there are five integers and four possible remainders when an integer is divided by 4, the pigeonhole principle implies that given five integers, atleast two have the same remainder.

Example 5: A bag contains 12 pairs of socks (each pair is in different color). If a person draws the socks one by one at random, determine atmost how many draws are required to get atleast one pair of matched socks.

**Solution:** Let n denote the number of the draw. For  $n \le 12$ , it is possible that the socks drawn are of different colors, since there are 12 colors. For n 13, all socks cannot have different colors at least two must have the same color. Here 13 as the number of pigeons and 12 colors as 12 pigeonholes. Thus, at most 13 draws are required to have at least one pair of socks of the same color.

#### **RECURRENCE RELATIONS**

#### **Recurrence Relations**

**Definition:** Recurrence relation : (Sometimes called difference equation).

A recurrence relation for the sequence  $\{a_n\}$  is an equation that shows  $a_n$  in terms of one or more of the previous terms of the sequence  $a_0, a_1, ..., a_{n-1}$ , for all integers n with  $n \ge n_0$ , where no is a non-negative integer.

**Note:** A sequence is called a solution of a recurrence relation if its terms satisfy the recurrence relation.

21

**Example:** The Fibonancci sequence is defined by the recurrence relation  $a_r = a_{r-2} + a_{r-1}$ ,  $r \ge 2$ , with the initial conditions  $a_0 = 1$  and  $a_1 = 1$ .

Example 1: Let {A} be a sequence that satisfies the recurrence relation  $a_n = a_{n-2} + a_{n-1}$  for n = 2, 3, 4, 5, .... and suppose that  $a_0 = 3$  and  $a_1 = 5$ . What are  $a_2$  and  $a_3$ ?

**Solution :** Given:

 $\begin{array}{l} a_n = a_{n-2} + a_{n-1} \\ a_2 = a_0 + a_1 = 3{+}5 = 8 \\ a_3 = a_1{+}a_2 = 5 + 8 = 13 \end{array}$ 

Example 2: Find the first five terms of the sequence defined by each of these recurrence relations and initial conditions.

(b) 
$$a_n = a_{n-1+3} a_{n-2}, a_0 = 1, a_1 = 2$$
  
(c)  $a_n = a_{n-1} + a_{n-3}, a_0 = 1, a_1 = 2, a_2 = 0$   
Solution : (a)  $a_n = 6 a_{n-1}$ , given  $a_0 = 2$   
 $a_1 = 6 a_0 = 6$  (2) = 12  
 $a_2 = 6 a_1 = 6$  (12) = 72  
 $a_3 = 6 a_2 = 6$  (72) = 432  
 $a_4 = 6 a_3 = 6$  (432) = 2592  
(b)  $a_n = a_{n-1} + 3 a_{n-2}$  given :  $a_0 = 1, a_1 = 2$   
 $a_2 = a_1 + 3 a_0 = 2 + 3$  (1) = 5  
 $a_3 = a_2 + 3 a_1 = 5 + 3$  (2) = 11  
 $a_4 = a_3 + 3 a_2 = 11 + 3$  (5) = 26  
(c)  $a_n = a_{n-1} + a_{n-3}$ , given :  $a_0 = 1, a_1 = 2, a_2 = 0$   
 $a_3 = a_2 + a_0 = 2 + 1 = 3$   
 $a_4 = a_3 + a_1 = 3 + 2 = 5$ 

(a)  $a_n = 6 a_{n-1}$ , given  $a_0 = 2$ 

Example 3: Let an  $2^{n} + (5) (3^{n})$  for n = 0, 1, 2, ..(a) Find  $a_{0}$ ,  $a_{1}$  and  $a_{2}$ . (b) Show that  $a_{4} = 5a_{3} - 6a_{2}$ . Solution: (a) Given:  $a_{n} = 2^{n} + (5) (3^{n})$  $a_{0} = 2^{0} + (5) (3^{0}) = 1 + (5) (1) = 6$  $a_{1} = 2^{1} + (5) (3^{1}) = 2 + (5) (3) = 17$  $a_{2} = 2^{2} + (5) (3^{2}) = 4 + (5) (9) = 49$ (b) Given:  $a_{n} = 2^{n} + (5) (3^{n})$  $a_{3} = 2^{3} + (5) (3^{3}) = 8 + 5(27) = 143$  To prove:  $a_4 = 5a_3 - 6a_2$ L.H.S =  $a_4 = 2^4 + (5) (3^4) = 16 + 5(81) = 421$ R.H.S. =  $5a_3 - 6a_2$ = 5 (143) - 6 (49) = 715-294 = 421

L.H.S. = R.H.S. Hence the proof.

#### **CHAPTER-V**

#### MATHEMATICAL LOGIC

#### Dr.N.Balambigai

Assistant Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India

A proposition is a declarative sentence that is either true or false, but not both. The area of logic that deals with propositions is called the propositional calculus or propositional logic. It was first developed systematically by the Greek philosopher Aristotle more than 2300 years ago.

English mathematician George Boole discussed new propositions from those that we already have, in his book "The Laws of Thought" in 1854.

Many mathematical statements are constructed by combining one or more propositions. New propositions called compound propositions, are formed from existing propositions using logical operators.

#### PROPOSITIONS

#### Definition

A proposition (statement) is a declarative sentence that is either true or false, but not both.

#### Example

- 1. Chennai is the capital of Tamil Nadu [True]
- 2. 1+5 = 6 [True]
- 3. 2+7 = 10 [False]
- 4. Delhi is in America. [False]

**Note:** Here we will not use declarative sentences that can be simultaneously true and false, we are ruling out certain "self-contradictory statements".

#### Examples

1. This statement is false [we cannot say True or False]

- 2. Do you speak English? is a question, not a statement.
- 3. Obey orders, is a command, not a statement.

4. x + 4 = 2 is neither true nor false.

In the above sentences we cannot assign true or false.

#### Notation :

P, Q, R, S ... are used to denote propositions.

T is used to denote True proposition.

F is used to denote False proposition.

#### **Definition :**

#### Atomic statements: [Primary statements] [Simple]

Declarative sentences which cannot be further split into simpler sentences are called Atomic statements (also called primary statements or primitive statements)

#### *Example:* Rama is a boy.

The area of logic that deals with propositions is called the propositional calculus or propositional logic.

	English language usages	Logical connectives	Type of operator	Symbols
1.	and	conjunction	binary	٨
2.	or	disjunction	binary	V
3.	not	negation (or) denial	unary	☐ (or) ~
4.	if then	implication (or) conditional	binary	>
5.	if and only if	biconditional	binary	⇔
		the second se	Concerning the second se	and the second se

#### FIVE BASIC CONNECTIVES

#### Molecular [compound] [composite] statements

#### **Definition :**

Molecular (compound) (composite) statements

New statements can be formed from atomic statements through the use of connectives such as 'and', 'but', 'or' etc. The resulting statements are called molecular or compound statements.

#### Example :

Niranjan is a boy and Sita is a girl.



# TRIGONOMETRY AND SERIES

Edited by

**DR.R.ABIRAMI** 



#### **Trigonometry and Series**

Copyright © 2021 by Dr.R.Abirami

First Edition,2021

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-338-6

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### **TABLE OF CONTENTS**

Book elements	Page Numbers
<b>Chapter-I</b> <b>RECIPROCAL EQUATIONS</b> Dr.R.Abirami	01-16
<b>Chapter-II</b> SUMMATION OF SERIES Dr.R.Abirami	17-20
<b>Chapter-III</b> <b>MATRICES</b> Dr.R.Abirami	21-27
Chapter-IV EXPANSION OF TRIGONOMETRIC FUNCTIONS Dr.R.Abirami	28-33
Chapter-V HYPERBOLIC FUNCTIONS Dr.R.Abirami	34-37

#### **Chapter-I**

#### **RECIPROCAL EQUATIONS**

#### Dr.R.Abirami

Asst. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India.

Reciprocal equations have various definitions depending on the context:

#### **General Definition:**

A reciprocal equation is an equation where the unknown variable or expression appears in the denominator of a fraction.

#### **Mathematical Definition:**

Given an equation f(x) = 0, if f(x) contains a term with x in the denominator, then the equation is called a reciprocal equation.

#### **Types of Reciprocal Equations:**

1. Linear Reciprocal Equations: ax + b/x = c

#### Problem 1:

Solve:

2x + 3/x = 5

#### Solution:

Multiply by x:

 $2x^2 - 5x + 3 = 0$ 

Factor:

(2x - 3)(x - 1) = 0

Solve:

x = 3/2 or x = 1

#### **Problem 2:**

Solve:

x - 2/x = 3

#### Solution:

Multiply by x:

 $x^2 - 3x - 2 = 0$ 

Factor:

(x - 2)(x + 1) = 0

Solve:

x = 2 or x = -1

#### Problem 3:

Solve:

3x + 2/x = 7

#### Solution:

Multiply by x:

 $3x^2 - 7x + 2 = 0$ 

Use quadratic formula:

 $x = (7 \pm \sqrt{(49 - 24)}) / 6$ 

Solve:

x = 2 or x = 1/3

#### **Problem 4:**

Solve:

x/2 + 4/x = 3

#### Solution:

Multiply by 2x:

 $x^2 - 6x + 8 = 0$ 

Factor:

(x - 4)(x - 2) = 0

Solve:

x = 4 or x = 2

#### Problem 5:

Solve:

2x - 5/x = 11

#### **Solution:**

Multiply by x:

 $2x^2 - 11x - 5 = 0$ 

Use quadratic formula:

 $x = (11 \pm \sqrt{(121 + 40)}) / 4$ 

Solve:

x = 5 or x = -1/2

2. Quadratic Reciprocal Equations:  $ax^2 + bx + c/x = 0$ 

#### **Problem 1:**

#### Solve:

4x + 2/x = 11

#### Solution:

Multiply by x:

 $4x^2 - 11x + 2 = 0$ 

Factor:

(4x - 1)(x - 2) = 0

Solve:

x = 1/4 or x = 2

#### Problem 2:

Solve:

3x - 5/x = 7

#### Solution:

Multiply by x:

 $3x^2 - 7x - 5 = 0$ 

Use quadratic formula:

$$x = (7 \pm \sqrt{(49 + 60)}) / 6$$

Solve:

x = 5 or x = -1/3

#### **Problem 3:**

Solve:

2x + 1/x = 5

#### Solution:

Multiply by x:

 $2x^2 - 5x + 1 = 0$ 

Factor:

(2x - 1)(x - 1) = 0

Solve:

x = 1/2 or x = 1

#### Problem 4:

Solve:

x - 3/x = 2

#### Solution:

Multiply by x:

 $x^2 - 2x - 3 = 0$ 

#### Factor:

(x - 3)(x + 1) = 0

Solve:

x = 3 or x = -1

#### Problem 5:

Solve:

5x + 2/x = 13

#### **Solution:**

Multiply by x:

 $5x^2 - 13x + 2 = 0$ 

Use quadratic formula:

 $\mathbf{x} = (13 \pm \sqrt{169 - 40}) / 10$ 

Solve:

x = 2 or x = 1/5

 Rational Reciprocal Equations: f(x)/g(x) = 0, where f and g are polynomials Solve:

 $(x^2 + 2x - 3) / (x + 1) = 0$ 

#### Solution:

Set numerator equal to 0:

 $x^2 + 2x - 3 = 0$ 

Factor:

(x+3)(x-1) = 0

Solve:

x = -3 or x = 1

#### Problem 2:

Solve:

 $(x^3 - 2x^2 - 5x + 6) / (x - 2) = 0$ 

#### Solution:

Set numerator equal to 0:

 $x^3 - 2x^2 - 5x + 6 = 0$ 

Factor:

 $(x - 2)(x^2 - 5) = 0$ 

Solve:

x = 2 or  $x = \pm \sqrt{5}$ 

#### Problem 3:

Solve:

 $(2x^2 + 3x - 1) / (x + 2) = 0$ 

#### Solution:

Set numerator equal to 0:

 $2x^2 + 3x - 1 = 0$ 

Use quadratic formula:

$$x = (-3 \pm \sqrt{(9+8)}) / 4$$

Solve:

 $x = (-3 \pm \sqrt{17}) / 4$ 

#### Problem 4:

Solve:

 $(x^2 - 4x - 3) / (x - 3) = 0$ 

#### **Solution:**

Set numerator equal to 0:

 $x^2 - 4x - 3 = 0$ 

Factor:

(x - 3)(x + 1) = 0

Solve:

x = 3 or x = -1

#### Problem 5:

Solve:

 $(3x^2 + 2x - 5) / (2x + 1) = 0$ 

#### **Solution:**

Set numerator equal to 0:

 $3x^2 + 2x - 5 = 0$ 

Use quadratic formula:

 $x = (-2 \pm \sqrt{(4+60)}) / 6$ 

Solve:

$$x = (-2 \pm \sqrt{64}) / 6$$

#### **Trigonometric Reciprocal Equations:**

1. Involving reciprocal trigonometric functions: csc(x), sec(x), cot(x)

#### Problem 1:

Solve:

 $\csc(x) - \cot(x) = 1$ 

#### Solution:

Use trigonometric identities:

 $1/\sin(x) - \cos(x)/\sin(x) = 1$ 

Simplify:

 $1 - \cos(x) = \sin(x)$ 

Square both sides:

 $1 - 2\cos(x) + \cos^2(x) = \sin^2(x)$ 

Use Pythagorean identity:

 $1 - 2\cos(x) + \cos^2(x) = 1 - \cos^2(x)$ 

Solve:

 $\cos(\mathbf{x}) = 0$ 

 $x = \pi/2 + k\pi$ 

#### Problem 2:

Solve:

 $\sec(x) + \tan(x) = 2$ 

#### Solution:

Use trigonometric identities:

 $1/\cos(x) + \sin(x)/\cos(x) = 2$ 

Simplify:

 $1 + \sin(x) = 2\cos(x)$ 

Square both sides:

 $1 + 2\sin(x) + \sin^2(x) = 4\cos^2(x)$ 

Use Pythagorean identity:

 $1 + 2\sin(x) + 1 - \cos^2(x) = 4\cos^2(x)$ 

Solve:

 $\cos(x) = 1/2$ 

 $x = \pi/3 + 2k\pi$ 

#### Problem 3:

Solve:

 $\cot(\mathbf{x}) - \csc(\mathbf{x}) = 1$ 

#### Solution:

Use trigonometric identities:

 $\cos(x)/\sin(x) - 1/\sin(x) = 1$ 

Simplify:

 $\cos(x) - 1 = \sin(x)$ 

Square both sides:

 $\cos^2(x) - 2\cos(x) + 1 = \sin^2(x)$ 

Use Pythagorean identity:

 $\cos^2(x) - 2\cos(x) + 1 = 1 - \cos^2(x)$ 

Solve:

 $\cos(x) = 1$ 

 $x = 2k\pi$ 

#### Problem 4:

Solve:

 $\csc(x) + \cot(x) = 3$ 

#### Solution:

Use trigonometric identities:

 $1/\sin(x) + \cos(x)/\sin(x) = 3$ 

Simplify:

 $1 + \cos(x) = 3\sin(x)$ 

Square both sides:

 $1 + 2\cos(x) + \cos^2(x) = 9\sin^2(x)$ 

Use Pythagorean identity:

 $1 + 2\cos(x) + 1 - \sin^2(x) = 9\sin^2(x)$ 

Solve:

sin(x) = 2/3

 $x = \arcsin(2/3) + 2k\pi$ 

#### Problem 5:

Solve:

 $\sec(x) - \tan(x) = 2$ 

#### Solution:

Use trigonometric identities:

 $1/\cos(x) - \sin(x)/\cos(x) = 2$ 

Simplify:

 $1 - \sin(x) = 2\cos(x)$ 

Square both sides:

 $1 - 2\sin(x) + \sin^2(x) = 4\cos^2(x)$ 

Use Pythagorean identity:

 $1 - 2\sin(x) + 1 - \cos^2(x) = 4\cos^2(x)$ 

2. Involving trigonometric functions with reciprocal arguments: sin(1/x), cos(1/x)

#### Problem 1:

Solve:

sin(1/x) = 1/2

#### Solution:

Use reciprocal identity:

 $\mathbf{x} = 1/(\pi/6 + \mathbf{k}\pi)$ 

Simplify:

 $x = 6/(\pi + 6k\pi)$ 

#### Problem 2:

Solve:

 $\cos(1/x) = 0$ 

#### Solution:

Use reciprocal identity:

 $x = 1/(\pi/2 + k\pi)$ 

Simplify:

 $x = 2/(\pi + 2k\pi)$ 

#### Problem 3:

Solve:

 $\tan(1/x) = 1$ 

#### Solution:

Use reciprocal identity:

 $x = 1/(\pi/4 + k\pi)$ 

Simplify:

 $x = 4/(\pi + 4k\pi)$ 

#### **Problem 4:**

Solve:

 $\sin(1/x) + \cos(1/x) = 1$ 

#### Solution:

Square both sides:

 $\sin^2(1/x) + 2\sin(1/x)\cos(1/x) + \cos^2(1/x) = 1$ 

Use Pythagorean identity:

 $1 + 2\sin(1/x)\cos(1/x) = 1$ 

Simplify:

sin(2/x) = 0

 $x = 1/(k\pi)$ 

#### Problem 5:

Solve:

 $\cos(1/x) - \sin(1/x) = 1$ 

#### Solution:

Square both sides:

 $\cos^2(1/x) - 2\cos(1/x)\sin(1/x) + \sin^2(1/x) = 1$ 

Use Pythagorean identity:

 $1 - 2\cos(1/x)\sin(1/x) = 1$ 

Simplify:

 $\sin(2/x) = 0$ 

 $x = 1/(k\pi)$ 

#### **Algebraic Reciprocal Equations:**

1. Involving rational expressions: 1/x + 1/y = 2

#### **Problem 1:**

Solve:

1/x + 1/y = 2

#### **Solution:**

Multiply by xy:

y + x = 2xy

Rearrange:

 $\mathbf{x} + \mathbf{y} - 2\mathbf{x}\mathbf{y} = \mathbf{0}$ 

Factor:

(x - 1)(y - 1) = 1

Solve:

x = 1 or y = 1

#### Problem 2:

Solve:

1/x - 1/y = 1

#### Solution:

Multiply by xy:

 $\mathbf{y} - \mathbf{x} = \mathbf{x}\mathbf{y}$ 

Rearrange:

xy - x - y = 0

Factor:

(x - 1)(y - 1) = 1

Solve:

x = 1 or y = 1

#### **Problem 3:**

Solve:

2/x + 1/y = 3Solution: Multiply by xy: 2y + x = 3xyRearrange:  $\mathbf{x} + 2\mathbf{y} - 3\mathbf{x}\mathbf{y} = \mathbf{0}$ Factor: (x - 1)(2y - 3) = -3Solve: x = 1 or y = 3/2**Problem 4:** 

Solve:

1/x + 2/y = 4

#### **Solution:**

Multiply by xy:

y + 2x = 4xy

Rearrange:

 $\mathbf{x} + 2\mathbf{y} - 4\mathbf{x}\mathbf{y} = \mathbf{0}$ 

Factor:

(x - 2)(2y - 1) = 2

Solve:

x = 2 or y = 1/2

#### **Problem 5:**

Solve:

3/x - 2/y = 1

#### **Solution:**

Multiply by xy:

3y - 2x = xy

Rearrange:

### xy - 3y + 2x = 0Factor: (x + 2)(y - 3) = -6

#### Solve:

x = -2 or y = 3

**2.** Involving algebraic functions:  $\sqrt{(x) + 1/\sqrt{(x)}} = 3$ 

#### Problem 1:

Solve:

 $\sqrt{(x)} + 1/\sqrt{(x)} = 3$ 

#### **Solution:**

Multiply by  $\sqrt{(x)}$ :

$$\mathbf{x} + 1 = 3\sqrt{\mathbf{x}}$$

Square both sides:

 $x^2 + 2x + 1 = 9x$ 

Rearrange:

 $x^2 - 7x + 1 = 0$ 

Solve:

 $x = (7 \pm \sqrt{45}) / 2$ 

#### Problem 2:

Solve:

 $\sqrt{2x} - 1/\sqrt{2x} = 1$ 

#### Solution:

Multiply by  $\sqrt{2x}$ : 2x - 1 =  $\sqrt{2x}$ Square both sides:  $4x^2 - 4x + 1 = 2x$ 

Rearrange:

 $4x^2 - 6x + 1 = 0$ 

Solve:

 $x = (6 \pm \sqrt{32}) / 8$ 

#### **Problem 3:**

Solve:

 $\sqrt{(x)} - 2/\sqrt{(x)} = 1$ 

#### Solution:

Multiply by  $\sqrt{(x)}$ :

 $x - 2 = \sqrt{x}$ 

Square both sides:

 $x^2 - 2x + 4 = x$ 

Rearrange:

 $x^2 - 3x + 4 = 0$ 

Solve:

 $x = (3 \pm \sqrt{1})/2$ 

#### Problem 4:

Solve:

 $1/\sqrt{(x)} + \sqrt{(x)} = 4$ 

#### Solution:

Multiply by  $\sqrt{(x)}$ :

 $1 + x = 4\sqrt{x}$ 

Square both sides:

 $1 + 2x + x^2 = 16x$ 

Rearrange:

 $x^2 - 14x + 1 = 0$ 

Solve:

 $x = (14 \pm \sqrt{(192)}) / 2$ 

#### **Problem 5:**

Solve:

 $2\sqrt{(x)} - 1/\sqrt{(x)} = 3$ 

#### **Solution:**

Multiply by  $\sqrt{(x)}$ :  $2x - 1 = 3\sqrt{(x)}$ Square both sides:  $4x^2 - 4x + 1 = 9x$ Rearrange:  $4x^2 - 13x + 1 = 0$ Solve:  $x = (13 \pm \sqrt{(153)}) / 8$ 

#### **Problem 1:**

Solve:

 $\sin(x) + \cos(x) = 1$ 

#### Solution:

 $\tan(\mathbf{x}) + 1 = \sec(\mathbf{x})$ 

 $x = \pi/4$ 

#### Problem 2:

Solve:

 $\operatorname{cosec}(\mathbf{x}) - \operatorname{cot}(\mathbf{x}) = 1$ 

#### Solution:

 $\sin(x) - \cos(x) = 1$ 

 $x = 3\pi/4$ 

#### Problem 3:

Solve:

 $\sec(x) + \tan(x) = 2$ 

#### **Solution:**

 $\cos(x) + \sin(x) = 1$ 

 $x = \pi/4$ 

#### **Problem 4:**

Solve:

sin(2x) = 2sin(x)cos(x)

#### **Solution:**

 $2\tan(x) = \tan(2x)$ 

 $x = \pi/6$ 

#### Problem 5:

Solve:

 $\cos(3x) = 4\cos^3(x) - 3\cos(x)$ 

#### Solution:

 $4\cos^3(x) - 3\cos(x) = \cos(3x)$ 

 $\mathbf{x} = \mathbf{0}$ 

#### **Series Problems:**

#### Problem 6:

Find the sum:

 $\sum [\sin(n\pi/6)]$  from n=1 to  $\infty$ 

#### Solution:

 $\sum [\sin(n\pi/6)] = 1/2 + 1/2 + 0 + (-1/2) + (-1/2) + 0 + \dots$ = 1/2

#### Problem 7:

Find the sum:

 $\sum [\cos(n\pi/3)]$  from n=1 to  $\infty$ 

#### Solution:

 $\sum [\cos(n\pi/3)] = 1/2 + (-1/2) + (-1/2) + 1/2 + \dots$ = 0

#### Problem 8:

Find the sum:

 $\sum [\tan(n\pi/4)]$  from n=1 to  $\infty$ 

#### Solution:

 $\sum [\tan(n\pi/4)] = 1 + (-1) + 1 + (-1) + \dots$ = 0

#### **Chapter-II**

#### **SUMMATION OF SERIES**

#### Dr.R.Abirami

Asst. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India.

Summation of series involves finding the sum of the terms of a sequence or series.

#### **Types of Series:**

- 1. Arithmetic Series:  $a + (a+d) + (a+2d) + \dots$
- 2. Geometric Series:  $a + ar + ar^2 + ...$
- 3. Harmonic Series: 1 + 1/2 + 1/3 + ...
- 4. Power Series:  $a_0 + a_1x + a_2x^2 + ...$
- 5. Taylor Series:  $f(x) = f(a) + f'(a)(x-a) + f''(a)(x-a)^2/2! + ...$

#### **Formulas for Summation:**

- 1. Arithmetic Series: S = n/2 [2a + (n-1)d]
- 2. Geometric Series: S = a / (1 r)
- 3. Harmonic Series:  $S = ln(n) + \gamma$  (Euler's constant)
- 4. Power Series:  $S = \sum [a_n x^n]$  from n=0 to  $\infty$

#### **Examples:**

#### Problem 1:

Find the sum of the series:

 $1 + 2 + 3 + \dots + 100$ 

#### Solution:

Arithmetic Series:

$$S = n/2 [2a + (n-1)d]$$

$$= 100/2 [2(1) + (100-1)1]$$

= 5050

#### **Problem 2:**

Find the sum of the series:

2 + 4 + 8 + ... + 64

#### Solution:

Geometric Series:

$$S = a / (1 - r)$$
  
= 2 / (1 - 2)  
= 2(1 - 2<sup>10</sup>) / (1 - 2)  
= 126

#### Problem 3:

Find the sum of the series:

 $1/1 + 1/2 + 1/3 + \dots + 1/10$ 

#### Solution:

Harmonic Series:

 $S\approx ln(10)+\gamma$ 

 $\approx 2.92896825397$ 

#### Problem 1:

Find the power series representation of:

f(x) = 1 / (1 - x)

#### Solution:

Geometric Series:

 $f(x) = 1 + x + x^2 + x^3 + \dots$ 

#### Problem 2:

Find the power series representation of:

 $f(x) = e^x$ 

#### **Solution:**

Taylor Series:

 $f(x) = 1 + x + \frac{x^2}{2!} + \frac{x^3}{3!} + \dots$ 

#### Problem 3:

Find the power series representation of:
f(x) = sin(x)

#### **Solution:**

Taylor Series:

 $f(x) = x - \frac{x^3}{3!} + \frac{x^5}{5!} - \frac{x^7}{7!} + \dots$ 

#### Problem 4:

Find the power series representation of:

f(x) = 1 / (1 + x)

#### **Solution:**

Geometric Series:

 $f(x) = 1 - x + x^2 - x^3 + \dots$ 

#### Problem 5:

Find the power series representation of:

 $f(x) = \ln(1+x)$ 

#### **Solution:**

Taylor Series:

 $f(x) = x - \frac{x^2}{2} + \frac{x^3}{3} - \frac{x^4}{4} + \dots$ 

#### Problem 1:

Find the Taylor series expansion of:

 $f(x) = e^x$  around x = 0

#### **Solution:**

 $f(x) = 1 + x + \frac{x^2}{2!} + \frac{x^3}{3!} + \dots$ 

#### Problem 2:

Find the Taylor series expansion of:

f(x) = sin(x) around x = 0

#### **Solution:**

 $f(x) = x - \frac{x^3}{3!} + \frac{x^5}{5!} - \frac{x^7}{7!} + \dots$ 

#### Problem 3:

Find the Taylor series expansion of:

f(x) = cos(x) around x = 0

#### **Solution:**

 $f(x) = 1 - x^2/2! + x^4/4! - x^6/6! + \dots$ 

#### **Problem 4:**

Find the Taylor series expansion of:

f(x) = ln(x) around x = 1

#### **Solution:**

 $f(x) = (x-1) - (x-1)^2/2 + (x-1)^3/3 - (x-1)^4/4 + \dots$ 

#### **Problem 5:**

Find the Taylor series expansion of:

f(x) = 1/x around x = 1

#### Solution:

 $f(x) = 1 - (x-1) + (x-1)^2 - (x-1)^3 + \dots$ 

#### **Chapter-III**

#### MATRICES

#### Dr.R.Abirami

Asst. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India.

A matrix is a rectangular array of numbers, symbols, or expressions arranged in rows and column: Matrices are often used in mathematics and computer science to represent and manipulate data. Each element in a matrix is identified by two indices: one for the row and one for the column.

For example, a matrix A with m rows and n columns is often denoted as:

$$A = egin{bmatrix} a_{11} & a_{12} & \ldots & a_{1n} \ a_{21} & a_{22} & \ldots & a_{2n} \ dots & dots & \ddots & dots \ a_{m1} & a_{m2} & \ldots & a_{mn} \end{bmatrix}$$

Where  $a_{ij}$  represents the element in the *i*-th row and *j*-th column. Matrices can be added, subtracted, and multiplied under certain conditions, and they are foundational in various fields, including linear algebra, statistics, and machine learning.

#### Types of Matrix

1. Row Matrix: A matrix with only one row (1 x n dimensions).

$$A = \begin{bmatrix} a_1 & a_2 & \dots & a_n \end{bmatrix}$$

2. Column Matrix: A matrix with only one column (m x 1 dimensions).

$$A = egin{bmatrix} a_1 \ a_2 \ dots \ a_m \end{bmatrix}$$

3. Square Matrix: A matrix with the same number of rows and columns (n x n dimensions).

4. Diagonal Matrix: A square matrix where all off-diagonal elements are zero.

$$D = egin{bmatrix} d_1 & 0 & \dots & 0 \ 0 & d_2 & \dots & 0 \ dots & dots & \ddots & dots \ 0 & 0 & \dots & d_n \end{bmatrix}$$

5. Scalar Matrix: A diagonal matrix where all the diagonal elements are the same (a scalar).

$$S = egin{bmatrix} k & 0 & \dots & 0 \ 0 & k & \dots & 0 \ dots & dots & \ddots & dots \ dots & dots & \ddots & dots \ 0 & 0 & \dots & k \end{bmatrix}$$

6. Identity Matrix: A square matrix with ones on the diagonal and zeros elsewhere. Denoted as  $I_n$  for an n imes n matrix.

$$I = egin{bmatrix} 1 & 0 & \dots & 0 \ 0 & 1 & \dots & 0 \ dots & dots & \ddots & dots \ 0 & 0 & \dots & 1 \end{bmatrix}$$

7. Zero Matrix: A matrix where all elements are zero. It can be of any dimension.

$$O = \begin{bmatrix} 0 & 0 & \dots & 0 \\ 0 & 0 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \dots & 0 \end{bmatrix}$$

8. Symmetric Matrix: A square matrix that is equal to its transpose ( $A = A^T$ ).

$$A=egin{bmatrix} a_{11}&a_{12}\ a_{12}&a_{22} \end{bmatrix}$$

9. Skew-Symmetric Matrix: A square matrix where the transpose is equal to the negative of the original matrix ( $A = -A^T$ ).

$$A = egin{bmatrix} 0 & a_{12} \ -a_{12} & 0 \end{bmatrix}$$

10. Orthogonal Matrix: A square matrix whose rows and columns are orthogonal unit vectors, satisfying  $A^T A = I$ .

#### Definitions

- Eigenvalue (λ): A scalar that satisfies the equation Av = λv, where A is a square matrix and v is a non-zero vector (the eigenvector).
- Eigenvector (v): A non-zero vector that changes by only a scalar factor when the linear transformation is applied.

#### Steps to Find Eigenvalues and Eigenvectors

 Form the Characteristic Equation: The eigenvalues are found by solving the characteristic equation:

$$\det(A - \lambda I) = 0$$

where I is the identity matrix of the same size as A.

- 2. Calculate Eigenvalues: Expand the determinant to find the roots  $\lambda$ .
- 3. Find Eigenvectors: For each eigenvalue  $\lambda$ 
  - Substitute  $\lambda$  back into the equation  $(A \lambda I)\mathbf{v} = 0$ .

• Solve the resulting system of linear equations to find the eigenvectors v.

#### Example

Let's find the eigenvalues and eigenvectors of the matrix:

$$A = \begin{bmatrix} 4 & 2 \\ 1 & 3 \end{bmatrix}$$

Step 1: Form the Characteristic Equation

Calculate  $det(A - \lambda I)$ :

$$A-\lambda I=egin{bmatrix} 4-\lambda&2\1&3-\lambda\end{bmatrix}$$

Now calculate the determinant:

Now calculate the determinant:

$$\det(A - \lambda I) = (4 - \lambda)(3 - \lambda) - (2)(1)$$

Expanding this:

$$\lambda=(12-4\lambda-3\lambda+\lambda^2)-2=\lambda^2-7\lambda+10$$

Set the characteristic equation to zero:

$$\lambda^2 - 7\lambda + 10 = 0$$

#### Step 2: Calculate Eigenvalues

Solve the quadratic equation using the quadratic formula:

$$\lambda = \frac{-b \pm \sqrt{b^2 - 4ac}}{2a} = \frac{7 \pm \sqrt{(-7)^2 - 4 \cdot 1 \cdot 10}}{2 \cdot 1}$$
$$= \frac{7 \pm 3}{2}$$

The solutions are:

 $\lambda_1=5 \quad ext{and} \quad \lambda_2=2$ 

Step 3: Find Eigenvectors

For  $\lambda_1=5$ :

Substitute into  $(A-5I)\mathbf{v}=0$ :

$$A - 5I = \begin{bmatrix} 4 - 5 & 2 \\ 1 & 3 - 5 \end{bmatrix} = \begin{bmatrix} -1 & 2 \\ 1 & -2 \end{bmatrix}$$

Set up the equation:

$$\begin{bmatrix} -1 & 2 \\ 1 & -2 \end{bmatrix} \begin{bmatrix} x \\ y \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$

This simplifies to the equation -x+2y=0 or x=2y.

Choosing y = 1, we get:

$$\mathbf{v_1} = \begin{bmatrix} 2\\1 \end{bmatrix}$$

For  $\lambda_2 = 2$ :

Substitute into  $(A-2I)\mathbf{v}=0$ :

$$A-2I=egin{bmatrix} 4-2&2\1&3-2\end{bmatrix}=egin{bmatrix} 2&2\1&1\end{bmatrix}$$

Set up the equation:

$$\begin{bmatrix} 2 & 2 \\ 1 & 1 \end{bmatrix} \begin{bmatrix} x \\ y \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$

This simplifies to the equation 2x+2y=0 or x=-y.

This simplifies to the equation 2x + 2y = 0 or x = -y.

Choosing y = 1, we get:

$$\mathbf{v_2} = \begin{bmatrix} -1\\ 1 \end{bmatrix}$$

#### Summary

- Eigenvalues:  $\lambda_1=5, \lambda_2=2$
- Eigenvectors:

• For 
$$\lambda_1 = 5$$
:  $\mathbf{v_1} = \begin{bmatrix} 2 \\ 1 \end{bmatrix}$ 

• For 
$$\lambda_2 = 2$$
:  $\mathbf{v_2} = \begin{bmatrix} -1 \\ 1 \end{bmatrix}$ 

The **Cayley-Hamilton Theorem** states that every square matrix satisfies its own characteristic equation.

#### Example 1: 2x2 Matrix

Let's take a matrix:

$$A = \begin{bmatrix} 4 & 1 \\ 6 & 3 \end{bmatrix}$$

#### Step 1: Find the Characteristic Polynomial

The characteristic polynomial is given by:

$$p(\lambda) = \det(A - \lambda I) = \det\left( egin{bmatrix} 4 & 1 \ 6 & 3 \end{bmatrix} - \lambda egin{bmatrix} 1 & 0 \ 0 & 1 \end{bmatrix} 
ight)$$
 $= \det egin{bmatrix} 4 - \lambda & 1 \ 6 & 3 - \lambda \end{bmatrix}$ 

The determinant of this 2x2 matrix is:

$$\det \begin{bmatrix} 4-\lambda & 1\\ 6 & 3-\lambda \end{bmatrix} = (4-\lambda)(3-\lambda) - (1)(6)$$
$$= (12-7\lambda+\lambda^2) - 6 = \lambda^2 - 7\lambda + 6$$

Thus, the characteristic polynomial is:

$$p(\lambda) = \lambda^2 - 7\lambda + 6$$

#### Step 2: Apply the Cayley-Hamilton Theorem

The Cayley-Hamilton theorem states that p(A)=0. So, substitute A into the characteristic polynomial:

$$A^2 - 7A + 6I = 0$$

We need to calculate  $A^2$ , 7A, and 6I.

• First, compute  $A^2$ :

$$A^2=A imes A=egin{bmatrix}4&1\6&3\end{bmatrix} imesegin{bmatrix}4&1\6&3\end{bmatrix}=egin{bmatrix}22&7\42&21\end{bmatrix}$$

• Next, calculate 7*A*:

$$7A = 7 imes egin{bmatrix} 4 & 1 \ 6 & 3 \end{bmatrix} = egin{bmatrix} 28 & 7 \ 42 & 21 \end{bmatrix}$$

• Also, compute 6I (where I is the identity matrix):

$$6I = 6 \times \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix} = \begin{bmatrix} 6 & 0 \\ 0 & 6 \end{bmatrix}$$

Now, substitute these into the Cayley-Hamilton equation:

$$\begin{bmatrix} 22 & 7\\ 42 & 21 \end{bmatrix} - \begin{bmatrix} 28 & 7\\ 42 & 21 \end{bmatrix} + \begin{bmatrix} 6 & 0\\ 0 & 6 \end{bmatrix} = \begin{bmatrix} 0 & 0\\ 0 & 0 \end{bmatrix}$$

As expected, this gives the zero matrix, verifying that A satisfies its characteristic equation.

#### **Chapter-IV**

#### **EXPANSION OF TRIGONOMETRIC FUNCTIONS**

#### Dr.R.Abirami

Asst. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India.

Expansion of trigonometric functions involves expressing them in terms of simpler trigonometric functions, polynomials, or infinite series.

#### **Trigonometric Identities:**

- 1. Pythagorean identity:  $\sin^2(x) + \cos^2(x) = 1$
- 2. Sum and difference formulas:
- $sin(a \pm b) = sin(a)cos(b) \pm cos(a)sin(b)$
- $\cos(a \pm b) = \cos(a)\cos(b) \mp \sin(a)\sin(b)$
- 3. Double-angle formulas:
- sin(2x) = 2sin(x)cos(x)
- $\cos(2x) = 2\cos^2(x) 1$

#### **Trigonometric Series Expansions:**

- 1. Fourier Series:
- $f(x) = a_0/2 + \sum [a_n \cos(n\pi x/L) + b_n \sin(n\pi x/L)]$  from n=1 to  $\infty$
- 2. Taylor Series:
- $\sin(x) = x \frac{x^3}{3!} + \frac{x^5}{5!} \frac{x^7}{7!} + \dots$
- $\cos(x) = 1 \frac{x^2}{2!} + \frac{x^4}{4!} \frac{x^6}{6!} + \dots$

#### **Examples:**

#### **Problem 1:**

Expand sin(3x) using sum and difference formulas.

#### **Solution:**

 $\sin(3x) = \sin(2x + x) = \sin(2x)\cos(x) + \cos(2x)\sin(x)$ 

#### Problem 2:

Expand cos(2x) using double-angle formulas.

#### **Solution:**

 $\cos(2x) = 2\cos^2(x) - 1$ 

#### Problem 3:

Find the Fourier Series expansion of:

 $f(x) = x, \ 0 \le x \le 2\pi$ 

#### Solution:

 $f(x) = 2\pi/2 + \sum [(-1)^{n/n} sin(nx)]$  from n=1 to  $\infty$ 

some other concepts related to Trigonometric Series:

#### 1. Fourier Analysis:

Decomposing periodic functions into trigonometric series (Fourier Series)

#### Problem 1:

Find the Fourier Series of:

 $f(x) = x, \ 0 \le x \le 2\pi$ 

#### Solution:

 $f(x) = 2\pi/2 + \sum [(-1)^{n/n} sin(nx)]$  from n=1 to  $\infty$ 

#### Problem 2:

Find the Fourier Transform of:

 $f(x) = e^{(-|x|)}$ 

#### Solution:

 $F(\omega) = 2/(1 + \omega^2)$ 

#### Problem 3:

Find the Inverse Fourier Transform of:

 $F(\omega) = 1/(1 + \omega^2)$ 

#### Solution:

 $f(x) = e^{(-|x|)/2}$ 

#### Problem 4:

Solve the differential equation:

 $y'' + 4y = \sin(2x), y(0) = 0, y'(0) = 0$ 

using Fourier Series.

#### **Solution:**

 $y(x) = 1/3\sin(2x)$ 

#### Problem 5:

Find the Fourier Series of:

 $f(x) = x^2, \ 0 \le x \le 2\pi$ 

#### **Solution:**

 $f(x) = 4\pi^{2}/3 + \sum[(-1)^{n/n^{2}}\cos(nx)]$  from n=1 to  $\infty$ 

#### 2. Harmonic Analysis:

Studying the properties of trigonometric series and their applications

#### roblem 1:

Find the harmonic conjugate of:

 $\mathbf{u}(\mathbf{x},\mathbf{y}) = \mathbf{x^2} - \mathbf{y^2}$ 

#### **Solution:**

v(x,y) = 2xy (using Cauchy-Riemann equations)

#### Problem 2:

Prove:

|f(z)| = |u(x,y)| + i|v(x,y)| (where f(z) = u(x,y) + iv(x,y))

#### Solution:

Using triangle inequality and properties of complex numbers

#### Problem 3:

Find the Poisson Integral Formula for:

 $u(x,y) = 1/(1 - x^2 - y^2)$ 

#### Solution:

 $u(x,y) = 1/(2\pi) \int [u(\theta)/(1 - 2r \cos(\theta - \phi) + r^2)] d\theta \ (0 \le r < 1)$ 

#### **Problem 4:**

Solve the Dirichlet Problem:

 $\Delta u = 0$ ,  $u(x,0) = x^2$ , u(0,y) = u(1,y) = 0

#### Solution:

 $u(x,y) = x^2 - y^2/3$  (using separation of variables)

#### Problem 5:

Find the Fourier Series of:

 $f(x) = x^3, 0 \le x \le 2\pi$ 

#### Solution:

 $f(x) = 8\pi^{3}/6 + \sum[(-1)^{n/n^{2}}sin(nx)]$  from n=1 to  $\infty$ 

#### 3. Bessel Functions:

Solutions to differential equations involving trigonometric functions

#### Problem 1:

Solve:

 $x^2y'' + xy' + (x^2 - n^2)y = 0$ 

#### **Solution:**

y(x) = AJn(x) + BYn(x) (Bessel's Equation)

#### Problem 2:

Find:

 $Jn(x) = \int [\cos(x \sin(\theta))\cos(n\theta)] d\theta$  from 0 to  $\pi$ 

#### Solution:

Jn(x) = Bessel Function of the First Kind

#### Problem 3:

Prove:

 $Yn(x) = (Jn(x)cos(n\pi) - J-n(x))/sin(n\pi)$ 

#### Solution:

Yn(x) = Bessel Function of the Second Kind

#### Problem 4:

Solve:

 $\Delta u = 0$ ,  $u(r,\theta) = r^n \cos(n\theta)$ 

#### Solution:

 $u(r,\theta) = Jn(r)cos(n\theta)$  (Laplace's Equation in Polar Coordinates)

#### Problem 5:

#### Find:

 $\int [xJn(x)]dx$ 

#### Solution:

xJn+1(x) + C

#### 4. Legendre Polynomials:

Polynomials used to approximate functions in trigonometric series

Find:

P0(x), P1(x), P2(x), P3(x)

#### Solution:

P0(x) = 1

P1(x) = x

 $P2(x) = (3x^2 - 1)/2$ 

 $P3(x) = (5x^3 - 3x)/2$ 

#### Problem 2:

Prove:

 $\int [-1 \text{ to } 1] \Pr(x) \Pr(x) dx = 2/(2n+1)\delta nm$ 

#### Solution:

Using orthogonality property

#### Problem 3:

Solve:

 $(1 - x^2)y'' - 2xy' + n(n + 1)y = 0$ 

#### Solution:

y(x) = Pn(x) (Legendre's Equation)

#### Problem 4:

Find:

Pn(1), Pn(-1)

#### Solution:

Pn(1) = 1

 $Pn(-1) = (-1)^n$ 

**Chapter-V** 

#### HYPERBOLIC FUNCTIONS

#### Dr.R.Abirami

Asst. Professor in Mathematics Ponnaiyah Ramajayam Institute of Science and Technology (**PRIST**) Vallam, Thanjavur, Tamilnadu, South India.

#### **Key Points:**

- 1. Hyperbolic Sine (sinh):  $\sinh(x) = (e^x e^{(-x)})/2$
- 2. Hyperbolic Cosine (cosh):  $\cosh(x) = (e^x + e^{(-x)})/2$
- 3. Hyperbolic Tangent (tanh): tanh(x) = sinh(x)/cosh(x)
- 4. Hyperbolic Cotangent (coth): coth(x) = cosh(x)/sinh(x)
- 5. Hyperbolic Secant (sech): sech(x) = 1/cosh(x)
- 6. Hyperbolic Cosecant (csch): csch(x) = 1/sinh(x)

#### **Identities:**

- 1.  $\sinh(-x) = -\sinh(x)$
- 2.  $\cosh(-x) = \cosh(x)$
- 3. tanh(-x) = -tanh(x)
- 4.  $\sinh(x + y) = \sinh(x)\cosh(y) + \cosh(x)\sinh(y)$
- 5.  $\cosh(x + y) = \cosh(x)\cosh(y) + \sinh(x)\sinh(y)$

#### **Problems:**

#### Problem 1:

Find:

dh/dx (sinh(2x))

#### **Solution:**

dh/dx (sinh(2x)) = 2cosh(2x)

#### **Problem 2:**

Prove:

 $\cosh^2(x) - \sinh^2(x) = 1$ 

#### Solution:

Using definitions of cosh and sinh

#### Problem 3:

Find:

 $\int tanh(x) dx$ 

#### Solution:

 $\int \tanh(x) \, dx = \log(\cosh(x)) + C$ 

#### **Problem 4:**

Solve:

y" - y = 0

#### Solution:

y(x) = c1cosh(x) + c2sinh(x)

#### Problem 5:

Find:

 $\int \cosh^2(x) dx$ 

#### Solution:

 $\int \cosh^2(x) dx = (1/2)x + (1/2)\sinh(x)\cosh(x) + C$ 

#### 1. Inverse Hyperbolic Functions:

- arcsinh(x)
- arccosh(x)
- arctanh(x)
- arccoth(x)
- arcsech(x)
- arccsch(x)
  - 2. Hyperbolic Identities:

- $\sinh(x+y) = \sinh(x)\cosh(y) + \cosh(x)\sinh(y)$
- $\cosh(x+y) = \cosh(x)\cosh(y) + \sinh(x)\sinh(y)$

3. Derivatives:

- $d/dx(\sinh(x)) = \cosh(x)$
- $d/dx(\cosh(x)) = \sinh(x)$
- $d/dx(tanh(x)) = sech^2(x)$

4. Integrals:

- $\int \sinh(x) dx = \cosh(x) + C$
- $\int \cosh(x) dx = \sinh(x) + C$

#### **Problems:**

#### Problem 1:

Find:

d/dx (tanh(2x))

#### Solution:

 $d/dx (tanh(2x)) = 2sech^2(2x)$ 

#### Problem 2:

Prove:

 $\cosh^2(x) - \sinh^2(x) = 1$ 

#### Solution:

Using definitions of cosh and sinh

#### Problem 3:

Find:

 $\int \coth^2(x) dx$ 

#### Solution:

 $\int \coth^2(x) \, dx = x - \operatorname{csch}(x) + C$ 

#### **Problem 4:**

Solve:

y'' - 4y = 0

#### Solution:

 $y(x) = c1\cosh(2x) + c2\sinh(2x)$ 

#### Problem 5:

Find:

 $\int \operatorname{sech}^3(x) dx$ 

#### Solution:

 $\int \operatorname{sech}^3(x) dx = (1/2)\operatorname{sech}(x) \tanh(x) + C$ 

# **IMAGE PROCESSING**

### EDITED BY: T.DIVYAMANOHARI



#### IMAGE PROCESSING

#### COPYRIGHT @ 2021 BY T. DIVYAMANOHARI

ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 9789362558725

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

### TABLE OF CONTENTS

CHAPTER	CONTENT	PAGE
	INTRODUCTION	3
1	INTRODUCTION TO IMAGE PROCESSING	7
	Dr. SMITHA ELSA PETER	
2	IMAGE FUNDAMENTALS	22
	R.SARASWATHI	
3	IMAGE ENHANCEMENT	41
	P. GEETHA	
4	IMAGE RESTORATION	54
	Dr. N. PARVATHAM	
5	IMAGE COMPRESSION	72
	E.PRIYADHARSHINI	
6	IMAGE SEGMENTATION	94
	S. LILLYPET	
7	IMAGE TRANSFORMATION	109
	A. AMUDHA	
8	FEATURE EXTRACTION	134
	Dr. S. DEVI	
9	<b>OBJECT RECOGNITION AND COMPUTER VISION</b>	158
	R. SAVITHA	
10	ADVANCED TOPICS	177
	S. MAHESHWARAN	
11	APPLICATIONS	198
	Dr. A. RIJUVANA BEGUM	
12	CASE STUDIES	214
	S.LILLYPET	
	REFERENCES	233

#### CHAPTER 1 INTRODUCTION TO IMAGE PROCESSING Dr. SMITHA ELSA PETER

Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

#### **1.1. Basics of Digital Imaging**

#### 1.1.1. Definition and Scope

Image processing involves the manipulation of digital images to improve their quality or extract meaningful information. It integrates principles from computer science, electrical engineering, and applied mathematics, facilitating applications across various fields such as medical imaging, remote sensing, and computer vision.

#### 1.1.2. Digital Image Representation

Digital images consist of a grid of pixels, each representing a specific color or intensity. Images can be grayscale, where each pixel has a single intensity value, or color, where multiple channels (e.g., RGB—Red, Green, Blue) define the pixel's color.

#### 1.1.3. Image Formats and Standards

Common image formats include JPEG, PNG, and TIFF. Each format has unique features regarding compression, color depth, and file size. JPEG and PNG are popular for their balance between image quality and file size, making them widely used in digital imaging.

#### **1.2. Overview of Image Processing**

#### **1.2.1.** Applications and Importance

Image processing is crucial in various industries:

- Medical Imaging: Enhances diagnostic images, such as MRI and CT scans, for better analysis.
- **Remote Sensing**: Processes satellite and aerial images for environmental monitoring and resource management.
- **Computer Vision**: Supports applications like facial recognition, autonomous vehicles, and augmented reality.

#### **1.2.2. Image Processing Pipeline**

The typical pipeline for image processing includes:

- 1. Acquisition: Capturing images using cameras or scanners.
- 2. **Preprocessing**: Enhancing or filtering images to improve quality.
- 3. Segmentation: Identifying and isolating specific regions of interest.

#### CHAPTER 2 IMAGE FUNDAMENTALS R. SARASWATHI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

#### 1. Definition and Representation

**1.1. What is a Digital Image?** A digital image is a discrete representation of visual information, consisting of a grid of pixels. Each pixel holds data about color or intensity, making digital images crucial for applications such as photography, medical imaging, and computer vision.

#### 1.2. Pixel and Resolution

- **Pixel**: The smallest unit of a digital image, representing a single point within the grid. In grayscale images, a pixel holds a value for brightness. In color images, it contains multiple values for color channels, such as Red, Green, and Blue (RGB).
- **Resolution**: Expressed as width × height in pixels (e.g., 1920×1080), resolution determines the level of detail and sharpness of the image. Higher resolution means more pixels and clearer images.

#### **1.3.** Color Models

- **RGB** (**Red**, **Green**, **Blue**): Combines red, green, and blue light to create colors in digital images. Each pixel in an RGB image has three values corresponding to these primary colors.
- **CMYK** (**Cyan, Magenta, Yellow, Key/Black**): Used in color printing by subtracting light from white, involving cyan, magenta, yellow, and black inks.
- HSV (Hue, Saturation, Value): Represents colors by hue (the color type), saturation (color intensity), and value (brightness), useful for adjusting color and brightness in image processing.

#### 2. Image Characteristics

#### 2.1. Brightness and Contrast

- **Brightness**: Refers to the overall lightness or darkness of an image. Adjusting brightness changes the uniform intensity of all pixels.
- **Contrast**: The difference between the darkest and lightest areas. Higher contrast enhances feature distinction, while lower contrast results in a more uniform appearance.

#### 2.2. Dynamic Range

• **Dynamic Range**: The spectrum of intensity levels from the darkest to lightest parts of an image. A higher dynamic range captures more detail in shadows and highlights.

#### CHAPTER 3 IMAGE ENHANCEMENT P. GEETHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Image enhancement involves modifying digital images to improve their quality or emphasize specific features, which is essential in fields such as medical imaging, remote sensing, and digital photography. Below are key techniques used for image enhancement:

#### **1. Basic Enhancement Techniques**

#### 1.1. Brightness and Contrast Adjustment

- **Brightness**: Adjusts the overall lightness or darkness of an image uniformly. Increasing brightness lightens the image, while decreasing it darkens the image.
- **Contrast**: Enhances the distinction between the darkest and lightest areas of an image. High contrast makes features stand out more distinctly, whereas low contrast results in a more even appearance.

#### **1.2. Histogram Equalization**

- **Definition**: A method that improves an image's contrast by redistributing the intensity values across the entire range. This technique enhances visibility, particularly in images with poor contrast.
- **Application**: Ideal for correcting images with uneven lighting or low contrast to make details more discernible.

#### 2. Filtering Techniques

#### 2.1. Smoothing Filters

- **Gaussian Filter**: Applies a blur effect by averaging pixel values in a local neighborhood, reducing noise and detail. The degree of smoothing is controlled by the filter's standard deviation.
- **Median Filter**: Replaces each pixel's value with the median value of its surrounding pixels, effectively reducing salt-and-pepper noise while preserving edges.

#### 2.2. Sharpening Filters

- **Sobel Filter**: Enhances edges by calculating the gradient of pixel intensity, making transitions in intensity more pronounced and defining edges more clearly.
- Laplacian Filter: Highlights edges and fine details by applying a second-order derivative, which accentuates areas of rapid intensity change.

#### CHAPTER 4 IMAGE RESTORATION Dr. N. PARVATHAM

Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Image restoration aims to correct distortions and degradations in images to recover their original quality and details. This process is distinct from image enhancement, which focuses on improving visual appearance. Key techniques and methods in image restoration include:

#### 1. Types of Image Degradation

1.1. Noise

• **Description**: Random variations in pixel values that obscure details and degrade image clarity. Common noise types include Gaussian noise, salt-and-pepper noise, and speckle noise.

#### • Restoration Methods:

- **Median Filter**: Removes salt-and-pepper noise by replacing each pixel with the median value of its neighbors, preserving edges better.
- **Gaussian Filter**: Reduces Gaussian noise by averaging pixel values in a neighborhood, effectively smoothing the image.

#### 1.2. Blur

- **Description**: Loss of image sharpness caused by factors such as camera shake, defocus, or motion, which reduces detail and clarity.
- Restoration Methods:
  - Wiener Filter: Adapts to local image variance to mitigate blur and noise, assuming known blur functions and noise levels.
  - **Inverse Filtering**: Attempts to reverse the blurring by applying the inverse of the blur function, effective if the blur characteristics are accurately known.

#### **1.3. Distortion**

- **Description**: Geometric distortions such as warping or irregularities caused by lens imperfections or other factors, leading to misalignment or deformation of the image.
- Restoration Methods:
  - Geometric Correction: Corrects distortions using transformation techniques (e.g., affine or projective transformations) to map the distorted image to a corrected coordinate system.
  - Undistortion Algorithms: Utilize calibration data to correct lens-induced distortions.

#### CHAPTER 5 IMAGE COMPRESSION E. PRIYADHARSHINI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Image compression reduces file sizes to optimize storage, transmission, and bandwidth use while preserving acceptable quality. It falls into two main categories: lossy and lossless compression.

#### 1. Types of Image Compression

#### **1.1. Lossy Compression**

- **Definition**: Reduces file size by permanently discarding some image data, which may lead to quality loss. The goal is to achieve significant size reduction while minimizing perceptual loss.
- Techniques:
  - JPEG Compression: Utilizes Discrete Cosine Transform (DCT) to convert image data into frequency components, quantizes these components, and encodes them using Huffman coding.
  - **Transform Coding**: Converts the image into a different domain (e.g., frequency) for more efficient compression, followed by quantization and encoding.
  - **Color Subsampling**: Reduces color resolution while maintaining luminance information, commonly used in JPEG to compress chroma channels.

#### **1.2.** Lossless Compression

- **Definition**: Compresses images without any loss of information, allowing exact reconstruction of the original image. Essential for applications where quality cannot be compromised.
- Techniques:
  - **PNG Compression**: Uses lossless methods like filtering and the DEFLATE algorithm to preserve quality, suitable for images with text, sharp edges, and transparency.
  - **TIFF Compression**: Supports both lossless and lossy methods, frequently used for high-quality image storage and professional applications.
  - **Run-Length Encoding (RLE)**: Encodes sequences of identical pixels into a single value and count, effective for images with large areas of uniform color.

#### 2.1. Transform Coding

• **Definition**: Converts image data into a domain that allows for more efficient compression, followed by quantization and encoding.

#### CHAPTER 6 IMAGE SEGMENTATION S.LILLYPET

#### Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Image segmentation is a vital process in image analysis, dividing an image into distinct regions or segments, each representing different structures or objects. This technique is fundamental for extracting meaningful information across various fields, including medical imaging, object recognition, and autonomous vehicles.

#### 1. Fundamentals of Image Segmentation

#### **1.1. Definition and Purpose**

- **Definition**: Image segmentation involves partitioning an image into multiple segments or regions where each segment corresponds to specific features or objects. This segmentation simplifies the image representation, making it easier to analyze and interpret.
- **Purpose**: It isolates regions of interest, facilitating tasks such as object recognition, feature extraction, and detailed image analysis.

#### **1.2.** Types of Segmentation

- Semantic Segmentation: Each pixel is classified into predefined categories or classes (e.g., sky, road, vehicle), with a label assigned based on its class.
- **Instance Segmentation**: Identifies and differentiates between different instances of the same class (e.g., distinguishing between multiple cars).

#### 2. Segmentation Techniques

#### 2.1. Thresholding

- **Definition**: Segments an image by converting grayscale values into binary values based on a threshold. Pixels above the threshold are assigned one value (e.g., white), while those below are assigned another (e.g., black).
- Types:
  - Global Thresholding: Applies a single threshold value across the entire image.
  - Adaptive Thresholding: Uses varying threshold values for different regions based on local image characteristics, effective in varying illumination conditions.

#### 2.2. Edge Detection

• **Definition**: Detects boundaries or edges where there are significant changes in intensity. Edges help delineate different segments.

#### CHAPTER 7 IMAGE TRANSFORMATION A. AMUDHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Image transformation involves altering the spatial or intensity attributes of an image to meet specific goals, such as enhancing visualization, aligning images, or extracting key features. It is a fundamental component of image processing and analysis. Below is an overview of the primary concepts and methods used in image transformation:

#### **1.** Geometric Transformations

#### 1.1. Rotation

- **Definition**: Rotates an image around a specified point, typically its center, by a given angle. This adjustment is crucial for correcting image orientation or aligning multiple images.
- Applications: Useful for image registration, correcting camera angles, and aligning objects.

#### 1.2. Scaling

- **Definition**: Alters the size of an image, either enlarging or reducing it. This involves adjusting the image resolution and using interpolation techniques to estimate pixel values at new locations.
- Applications: Commonly employed in resizing images for different display resolutions, printing, or fitting into designated spaces.

#### 1.3. Translation

- **Definition**: Shifts an image's position within the coordinate system without modifying its content. This transformation changes the image's location but not its internal structure.
- Applications: Helpful for aligning images in sequences or correcting spatial misalignments.

#### **1.4. Affine Transformation**

- **Definition**: Maintains the geometric properties of points, straight lines, and planes while allowing for scaling, rotating, translating, and shearing.
- Applications: Often used for correcting geometric distortions and aligning images.

#### **1.5.** Perspective Transformation

- **Definition**: Adjusts an image to correct perspective distortions by mapping points from one plane to another, accounting for changes in depth and viewing angle.
- **Applications**: Essential for rectifying distortions in images captured from various viewpoints or for photogrammetry.

#### CHAPTER 8 FEATURE EXTRACTION Dr. S. DEVI

Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Feature extraction is a critical step in image analysis and computer vision, aimed at identifying and isolating key attributes or patterns from images to support various tasks like classification, recognition, and analysis. The primary goal is to convert raw image data into a set of meaningful features that simplify subsequent processing. Here's a concise overview of the key concepts and techniques involved in feature extraction:

#### **1. Basic Concepts**

#### **1.1. Definition and Purpose**

- **Definition**: Feature extraction involves isolating distinctive characteristics or patterns within an image to be used for further analysis or classification. These features provide a reduced and manageable representation of the image.
- **Purpose**: Simplifies image data, making it easier to perform tasks such as object recognition, classification, and pattern analysis.

#### **1.2. Types of Features**

- **Global Features**: Capture overall properties of the image, such as color histograms, texture patterns, and shapes.
- Local Features: Focus on specific regions or objects within the image, including edges, corners, and key points.

#### 2. Feature Extraction Techniques

#### 2.1. Color Features

- **Color Histograms**: Show the distribution of colors in an image by counting pixel occurrences for each color value. Useful for image retrieval and classification.
- **Color Moments**: Statistical measures (mean, variance, skewness) of the color distribution that offer a compact representation of color data.

#### **2.2. Texture Features**

- **Gray-Level Co-occurrence Matrix (GLCM)**: Analyzes spatial relationships between pixel values to derive features such as contrast, correlation, and entropy.
- Local Binary Patterns (LBP): Describes local texture by comparing each pixel with its neighbors, creating binary patterns that highlight texture characteristics.

#### CHAPTER 9 OBJECT RECOGNITION AND COMPUTER VISION R. SAVITHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Object recognition and computer vision are interrelated fields focused on interpreting and understanding visual information from images and videos. These fields leverage various techniques to enable machines to perceive, analyze, and make sense of visual data, crucial for applications in automation, surveillance, robotics, and more.

#### **Object Recognition**

#### **Definition and Purpose**

- **Definition**: Object recognition is the process of identifying and classifying objects within an image or video.
- **Purpose**: Essential for tasks such as automated surveillance, autonomous driving, image retrieval, and robotics, where understanding image content is key to making informed decisions and performing actions.

#### **Core Techniques**

#### 1. Feature-Based Methods

- SIFT (Scale-Invariant Feature Transform): Detects and describes local features that are invariant to changes in scale and rotation, facilitating object recognition by matching features across images.
- SURF (Speeded-Up Robust Features): A faster alternative to SIFT, designed to be robust to various transformations and noise, ensuring efficient and accurate object detection.
- ORB (Oriented FAST and Rotated BRIEF): Combines FAST keypoint detection with BRIEF descriptors, providing a fast and efficient method for object recognition while being robust to rotation.

#### 2. Template Matching

- **Definition**: Involves comparing segments of an image with a predefined template to find matches. Effective for detecting known objects with minimal variability.
- Applications: Used in quality control, object tracking, and aligning images.
- **Support Vector Machines (SVM)**: Classifies objects by finding the optimal hyperplane that separates different classes based on their features.

#### CHAPTER 10 ADVANCED TOPICS S. MAHESHWARAN

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

#### 1. Deep Learning in Computer Vision

#### 1.1. Generative Adversarial Networks (GANs)

- **Definition**: GANs consist of two neural networks, the generator and the discriminator, which are trained simultaneously. The generator creates synthetic images, while the discriminator evaluates their authenticity. The objective is for the generator to produce images so realistic that they are indistinguishable from real ones.
- **Applications**: GANs are utilized for image synthesis, style transfer, image super-resolution, and data augmentation.

#### **1.2. Neural Style Transfer**

- **Definition**: This technique employs deep learning to apply the visual style of one image (e.g., a painting) to the content of another image (e.g., a photograph). It involves extracting and blending style and content features from both images.
- **Applications**: Used for artistic image generation, creative design, and enhancing virtual reality experiences.

#### **1.3. Deep Reinforcement Learning for Vision**

- **Definition**: Integrates deep learning with reinforcement learning to enable agents to make decisions based on visual input. It involves training models to perform specific tasks through trial and error, guided by rewards and penalties.
- **Applications**: Applied in robotics, autonomous driving, and interactive AI systems to improve decision-making based on visual observations.

#### 2. Advanced Image Segmentation Techniques

#### 2.1. U-Net Architecture

- **Definition**: A specialized convolutional network designed for biomedical image segmentation. It features an encoder-decoder structure with skip connections, enabling it to capture both high-level and low-level features effectively.
- Applications: Widely used for medical image segmentation and satellite imagery analysis.
- **Definition**: An extension of Faster R-CNN that includes an additional branch for predicting segmentation masks for each detected object.

## EMBEDDED SYSTEMS AND IOT DESIGN LAB MANUAL

EDITED BY



## EMBEDDED SYSTEMS AND IOT DESIGN LAB MANUAL COPYRIGHT @ 2021 BY M.SOUNDHARYA

## ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 978-93-6255-157-3

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### **EMBEDDED SYSTEMS AND IOT DESIGN**

#### EXP NO TITLE OF THE EXPERIMENT PAGE NO **EXPERIMENTS USING 8051** PROGRAMMING ARITHMETIC AND LOGICAL OPERATIONS IN 1 14 8051 2 GENERATION OF SQUARE WAVEFORM USING 8051 22 3 PROGRAMMING USING ON – CHIP PORTS IN 8051 23 4 PROGRAMMING USING SERIAL PORTS IN 8051 25 5 DESIGN OF A DIGITAL CLOCK USING TIMERS/COUNTERS IN 27 8051 **EXPERIMENTS USING ARM** 6 INTERFACING ADC AND DAC 36 7 BLINKING OF LEDS AND LCD 41 8 INTERFACING KEYBOARD AND STEPPER MOTOR 45

#### **STUDY OF 8051 MICROCONTROLLER**



#### **Architecture of 8051 Microcontroller**

Architecture of 8051 microcontroller has following features

- 4 Kb of ROM is not much at all.
- 128Kb of RAM (including SFRs) satisfies the user's basic needs.
- 4 ports having in total of 32 input/output lines are in most cases sufficient to make all necessary connections to peripheral environment.

The whole configuration is obviously thought of as to satisfy the needs of most programmers working on development of automation devices. One of its advantages is that nothing is missing and nothing is too much. In other words, it is created exactly in accordance to the average user's taste and needs. Other advantages are RAM organization, the operation of Central Processor Unit (CPU) and ports which completely use all recourses and enable further upgrade.

#### Pin Description of 8051 Micro Controller:

Pins: 1-8 Port1: Each of these pins can be configured as an input or an output.

**Pin 9: RS**: A logic one on this pin disables the microcontroller and clears the contents of most registers. In other words, the positive voltage on this pin resets the microcontroller. By applying logic zero to this pin, the program starts execution from the beginning.
Pin 10: RXD: Serial asynchronous communication input or Serial synchronous communication output.

**Pin 11: TXD:** Serial asynchronous communication output or Serial synchronous Communication Clock Output.

Pin12: INT0: Interrupt 0 input

Pin 13: INT1: Interrupt 1 input

Pin14: T0: Counter 0 Clock Input

Pin15: T1: Counter 1 Clock Input

Pin16: WR: Write to external (additional) RAM.

Pin17: RD: Read from external RAM

**Pin 18, 19: X1, X2:** Internal oscillator input and output. A quartz crystal which specifies operating frequency is usually connected to these pins. Instead of it, miniature ceramics resonators can also be used for frequency stability. Later versions of microcontrollers operate at a frequency of 0 Hz up to over 50 Hz.

#### Pin20: GND: Ground.

**Pin 21-28: Port2:** If there is no intention to use external memory then these port pins are configured as general inputs/outputs. In case external memory is used, the higher address byte, i.e. addresses A8-A15 will appear on this port. Even though memory with capacity of 64Kb is not used, which means that not all eight port bits are used for its addressing, the rest of them are not available as inputs/outputs.

**Pin29: PSEN:** If external ROM is used for storing program then a logic zero (0) appears on itevery time the **microcontroller reads a byte from memory.** 

**Pin30: ALE:** Prior to reading from external memory, the microcontroller puts the lower address byte (A0-A7) on P0 and activates the ALE output. After receiving signal from the ALE pin, the external register (usually 74HCT373 or 74HCT375 add-on chip) memorizes the state of P0 and uses it as a memory chip address. Immediately after that, the ALU pin is returned its previous logic state and P0 is now used as a Data Bus. As seen, port data multiplexing is performed by means of only one additional (and cheap) integrated circuit. In other words, this port is used for both data and address transmission.

**Pin31: EA:** By applying logic zero to this pin, P2 and P3 are used for data and address transmission with no regard to whether there is internal memory or not. It means that even there is a program written to the microcontroller, it will not be executed. Instead, the program written to external ROM will be executed. By applying logic one to the EA pin, the microcontroller will use both memories, first internal then external (if exists).

**Pin32-39: Port0:** Similar to P2, if external memory is not used, these pins can be used as general inputs/outputs. Otherwise, P0 is configured as address output (A0-A7) when the ALE pin is driven high (1) or as data output (Data Bus) when the ALE pin is driven low (0).

**Pin40: Vcc**: +5V power supply

#### Input / Output Ports (I/O Ports):

All 8051 microcontrollers have 4 I/O ports each comprising 8 bits which can be configured as inputs or outputs. Accordingly, in total of 32 input/output pins enabling the microcontroller to be connected to peripheral devices are available for use.

Pin configuration, i.e. whether it is to be configured as an input (1) or an output (0), depends on its logic state. In order to configure a microcontroller pin as an input, it is necessary to apply a logic zero (0) to appropriate I/O port bit. In this case, voltage level on appropriate pin will be 0.

Similarly, in order to configure a microcontroller pin as an input, it is necessary to apply a logic one (1) to appropriate port. In this case, voltage level on appropriate pin will be 5V (as is the case with any TTL input). This may seem confusing but don't loose your patience. It all becomes clear after studying simple electronic circuits connected to an I/O pin.

#### **Memory Organization:**

The 8051 has two types of memory and these are Program Memory and Data Memory. Program Memory (ROM) is used to permanently save the program being executed, while Data Memory (RAM) is used for temporarily storing data and intermediate results created and used during the operation of the microcontroller. Depending on the model in use (we are still talking about the 8051 microcontroller family in general) at most a few Kb of ROM and 128 or 256 bytes of RAM is used. However All 8051 microcontrollers have a 16-bit addressing bus and are capable of addressing 64 kb memory. It is neither a mistake nor a big ambition of engineers who were working on basic core development. It is a matter of smart memory organization which makes these microcontrollers a real "programmers' goody".



#### **Special Function Registers (SFRs):**

Special Function Registers (SFRs) are a sort of control table used for running and monitoring the operation of the microcontroller. Each of these registers as well as each bit they include, has its name, address in the scope of RAM and precisely defined purpose such as timer control, interrupt control, serial communication control etc. Even though there are 128 memory locations intended to be occupied by them, the basic core, shared by all types of 8051 microcontrollers, has only 21 such registers. Rest of locations is intentionally left unoccupied in order to enable the manufacturers to further develop microcontrollers keeping them compatible with the previous versions. It also enables programs written a long time ago for microcontrollers which are out of production now to be used today.



#### Program Status Word (PSW) Register



PSW register is one of the most important SFRs. It contains several status bits that reflect the current state of the CPU. Besides, this register contains Carry bit, Auxiliary Carry, two register bank select bits, Overflow flag, parity bit and user-definable status flag.

**P** - **Parity bit.** If a number stored in the accumulator is even then this bit will be

automatically set (1), otherwise it will be cleared (0). It is mainly used during data transmit and receive via serial communication.

- Bit 1. This bit is intended to be used in the future versions of microcontrollers.

**OV Overflow** occurs when the result of an arithmetical operation is larger than 255 and cannot be stored in one register. Overflow condition causes the OV bit to be set (1). Otherwise, it will be cleared (0).

**RS0, RS1 - Register bank select bits.** These two bits are used to select one of four register banks of RAM. By setting and clearing these bits, registers R0-R7 are stored in one of four banks of RAM.

RS1	RS2	Space
		in RAM
0	0	Bank0
		00h-07h
0	1	Bank1
		08h-0Fh
1	0	Bank2
		10h-17h
1	1	Bank3
		18h-1Fh

F0 - Flag 0. This is a general-purpose bit available for use.

AC - Auxiliary Carry Flag is used for BCD operations only.

CY - Carry Flag is the (ninth) auxiliary bit used for all arithmetical operations and shift instructions.



**Data Pointer Register (DPTR):**DPTR register is not a true one because it doesn't physically exist. It consists of two separate registers: DPH (Data Pointer High) and (Data Pointer Low). For this reason it may be treated as a 16-bit register or as two independent 8-bit registers. Their 16 bits are primarly used for external memory addressing. Besides, the DPTR Register is usually used for storing data and intermediate results.

Stack Pointer (SP) Register



A value stored in the Stack Pointer points to the first free stack address and permits stack availability. Stack pushes increment the value in the Stack Pointer by 1. Likewise, stack pops decrement its value by 1. Upon any reset and power-on, the value 7 is stored in the Stack Pointer, which means that the space of RAM reserved for the stack starts at this location. If

another value is written to this register, the entire Stack is moved to the location.

P0, P1, P2, P3 - Input/Output Registers



If neither external memory nor serial communication system are used then 4 ports with in total of 32 input/output pins are available for connection to peripheral environment. Each bit within these ports affects the state and performance of appropriate pin of the microcontroller. Thus, bit logic state is reflected on appropriate pin as a voltage (0 or 5 V) and vice versa, voltage on a pin reflects the state of appropriate port bit.

As mentioned, port bit state affects performance of port pins, i.e. whether they will be

configured as inputs or outputs. If a bit is cleared (0), the appropriate pin will be configured as an output, while if it is set (1), the appropriate pin will be configured as an input. Upon reset and power-on, all port bits are set (1), which means that all appropriate pins will be configured as inputs.

#### **Counters and Timers:**

As you already know, the microcontroller oscillator uses quartz crystal for its operation. As the frequency of this oscillator is precisely defined and very stable, pulses it generates are always of the same width, which makes them ideal for time measurement. Such crystals are also used in quartz watches. In order to measure time between two events it is sufficient to count up pulses coming from this oscillator. That is exactly what the timer does. If the timer is properly programmed, the value stored in its register will be incremented (or decremented) with each coming pulse, i.e. once per each machine cycle. A single machine-cycle instruction lasts for 12 quartz oscillator periods, which means that by embedding quartz with oscillator frequency of 12MHz, a number stored in the timer register will be changed million times per second, i.e. each microsecond.

The 8051 microcontroller has 2 timers/counters called T0 and T1. As their names suggest, their main purpose is to measure time and count external events. Besides, they can be used for generating clock pulses to be used in serial communication, so called Baud Rate.

As seen in figure below, the timer T0 consists of two registers – TH0 and TL0 representing a low and a high byte of one 16-digit binary number.



Accordingly, if the content of the timer T0 is equal to 0 (T0=0) then both registers it consists of will contain 0. If the timer contains for example number 1000 (decimal), then the TH0 register (high byte) will contain the number 3, while the TL0 register (low byte) will contain decimal number 232.



Formula used to calculate values in these two registers is very simple:  $TH0 \times 256 + TL0 = T$ Matching the previous example it would be as follows:

 $3 \times 256 + 232 = 1000$ 



Since the timer T0 is virtually 16-bit register, the largest value it can store is 65 535. In case of exceeding this value, the timer will be automatically cleared and counting starts from 0. This condition is called an overflow. Two registers TMOD and TCON are closely connected to this timer and control its oper tion.

## TMOD Register (Timer Mode):

The TMOD register selects the operational mode of the timers T0 and T1. As seen in figure below, the low 4 bits (bit0 - bit3) refer to the timer 0, while the high 4 bits (bit4 - bit7) refer to the timer 1. There are 4 operational modes and each of them is described herein.



Bits of this register have the following function:

- GATE1 enables and disables Timer 1 by means of a signal brought to the INT1 pin (P3.3):
  - o **1** Timer 1 operates only if the INT1 bit is set.
  - o 0 Timer 1 operates regardless of the logic state of the INT1 bit.
- **C/T1** selects pulses to be counted up by the timer/counter 1:
  - o **1** Timer counts pulses brought to the T1 pin (P3.5).
  - **0** Timer counts pulses from internal oscillator.
- **T1M1,T1M0** These two bits select the operational mode of the Timer 1.

T1M1	T1M0	Mode	Descri
			ption
0	0	0	13-bit
			timer
0	1	1	16-bit
			timer
1	0	2	8-bit
			auto-
			reload
1	1	3	Split
			mode

- GATE0 enables and disables Timer 1 using a signal brought to the INT0 pin (P3.2):
  - o **1** Timer 0 operates only if the INT0 bit is set.
  - **0** Timer 0 operates regardless of the logic state of the INT0 bit.
- **C/T0** selects pulses to be counted up by the timer/counter 0:

- o **1** Timer counts pulses brought to the T0 pin (P3.4).
- o 0 Timer counts pulses from internal oscillator.
- **T0M1,T0M0** These two bits select the oprtaional mode of the Timer 0.

T0M1	ТОМО	Mode	Description
0	0	0	13-bit timer
0	1	1	16-bit timer
1	0	2	8-bit auto-reload
1	1	3	Split mode

### Timer Control (TCON) Register:

TCON register is also one of the registers whose bits are directly in control of timer operation.

Only 4 bits of this register are used for this purpose, while rest of them is used for interrupt control to be discussed later.



- **TF1** bit is automatically set on the Timer 1 overflow.
- **TR1** bit enables the Timer 1.
  - o **1** Timer 1 is enabled.
  - o **0** Timer 1 is disabled.
- **TF0** bit is automatically set on the Timer 0 overflow.
- **TR0** bit enables the timer 0.
  - o **1** Timer 0 is enabled.
  - **0** Timer 0 is disabled.

## Timer 1:

Timer 1 is identical to timer 0, except for mode 3 which is a hold-count mode. It means that they have the same function, their operation is controlled by the same registers TMOD and TCON and both of them can operate in one out of 4 different modes.



## **1A. ARTHMETIC OPERATIONS USING 8051**

# Aim:

To do the arithmetic operations using 8051 microprocessor

# **Apparatus required:**

8051 microprocessor kit

# Algorithm:

# Addition / Subtraction:

Step 1	:	Move 1 <sup>H</sup> data to memory
Step 2	:	Add or subtract $1^{H}$ data with $2^{nd}$ data
Step 3	:	Initialize data pointer.
Step 4	:	Move result to memory pointed by DPTR.



# **Multiplication / Division:**

Step 1	:	Get 1 <sup>H</sup> data and 2 <sup>nd</sup> data to memory
Step 2	:	Multiply or divide 1 <sup>H</sup> data with 2 <sup>nd</sup> data
Step 3	:	Initialize data pointer.
Step 4	:	Move result to memory pointed by DPTR (first port)
Step 5	:	Increment DPTR
Step 6	:	Move 2 <sup>nd</sup> part of result to register A
Step 7	:	Move result to 2 <sup>nd</sup> memory location pointer by DPTR



**Program: 8-bit Addition:** 

Memory Location	Label	Opcode	Mnemonics	Comments
4100		74 01	MOV A, #01	Moves data 1 to register A
4102		24 02	ADD A, #02	Add content of A and data 2 and store in A
4104		90 45 00	MOV DPTR,#4500	Moves data 4500 to DPTR
4107		FO	MOVX @DPTR,A	Moves control of A to location pointed DTPR
4108		80 FE	SJMP 4108	Short jump to 4108

**Execution: Addition:** 

ML	Input		
4101		ML	Output
4103		4500	

Memory Location	Label	Opcode	Mnemonics	Comments
4100		74 05	MOV A,#05	Moves data 1 to register A
4102		94 02	SUBB A,#02	Subtract data 2 from content of A and store result in A
4104		90 45 00	MOV DPTR,#4500	Moves 4500 to DPTR
4107		FO	MOVX @DPTR,A	Moves result by location by DTPR
4108		80 FE	SJMP 4109	Short jump to 4109

# **Execution: Subtraction:**

ML	Input		
4101		ML	Output
4103		4500	

# **Program: 8-bit Multiplication:**

Memory Location	Label	Opcode	Mnemonics	Comments
4100	Start	74 03	MOV A,#03	Move immediate data to accumulator
4101		75 F0 02	MOV B,#02	Move 2 <sup>nd</sup> data to B register
4105		A4	MUL AB	Get the product in A & B
4106		90 45 00	MOV DPTR, # 4500	Load data in 4500 location
4109		FO	MOVX @DPTR,A	Move A t ext RAM
410A		A3	INC DPTR	
410B		E5 F0	MOV A,B	Move 2 <sup>nd</sup> data in A
410D		FO	MOVX @DPTR,A	Same the ext RAM
410E		80 FE	SJMP 410E	Remain idle in infinite loop

**Execution: Multiplication:** 

ML	Input		
4101		Output Address	Value
4103		4500	

# **Program: 8-bit Division:**

Memory Location	Label	Opcode	Mnemonics	Comments
4100	Start	74 04	MOV A,#04	Move immediate data to accumulator
4102		75 F0 02	MOV B,#02	Move immediate to B reg.
4105		84	DIV AB	Divide content of A & B
4106		90 45 00	MOV DPTR, # 4500	Load data pointer with4500 location
4109		FO	MOVX @DPTR,A	Move A to ext RAM
410A		A3	INC DPTR	Increment data pointer
410B		E5 F0	MOV A,B	Move remainder to A
410D		FO	MOVX @DPTR,A	Move A to ext RAM
410E		80 FE	SJMP 410E	Remain idle in infinite loop

# **Execution:**

**Division:** 

ML	Input		
4101		Output Address	Value
4103		4500	

# **Result:**

Thus 8-bit addition, subtraction, multiplication and division is performed using 8051.

## **1.B LOGICAL OPERATIONS USING 8051**

## Aim:

To do the logical operations using 8051 microprocessor

## **Apparatus required:**

8051 microprocessor kit

## Algorithm:

- 1. Start the program
- 2. Get the  $1^{st}$  data.
- 3. Perform OR of  $1^{st}$  and the 2 nd data.
- 4. Store the results in the memory location.
- 5. Stop the program

# **PROGRAM:**

# LOGICAL OR OPERATION

Address	Label	Opcode	Mnemonics	Operand	Comments
5000		E4	CLR	А	Clear A register
5001		74	MOV	A, #data1	Move data1 to A
5003		54	ORL	A,#data2	Perform OR with data2
5005		90	MOV	DPTR, #4150	Initialize DPTR
5008		F0	MOVX	@DPTR,A	Store result in DPTR
5009	Here	80	SJMP	Here( 5009)	Loop is terminated

## FLOW CHART: LOGICAL AND OPERATION:



# **OUTPUT:**

	Address	Data
Input		Data 1:
		Data 2:
Output	4150H	

# PROGRAM:

# LOGICAL AND OPERATION

Address	Label	Opcode	Mnemonics	Operand	Comments
4100		74	MOV	A, #data1	Move data1 to A register
4102		74	MOV	A, #data2	Move data2 to B register
4105		72 AB	ANL	A,B	ANL A with B
4107		F0	MOV	DPTR,#4150	Move the address 4150 to DPTR
410A		FO	MOVX	@DPTR,A	Store result in DPTR address
410B	Here	80 FE	SJMP	Here (410B)	Stop the program

## **BIT MANIPULATION OPERATION:**

# Algorithm:

- Step1: Start the Program
- Step2: Get the data and move into Accumulator
- Step3: Take 1's complement for the data and move it to data
- Step4: Take 2's Complement for the got 1's complement
- Step5: The data 2 stored in the next address
- Step6: Store the result in memory
- Step7: Stop the Program

# **PROGRAM:**

Memory Location	Label	Opcode	Mnemonics	Comments
4100	Start	74, 23	MOV A,#D1	Move the data1 to A register
4102		F4	CPL A	Complement A
4103		90,45,00	MOV DPTR, #4500	Move 4500 to DPTR
4106		F0, A3	MOVX @DPTR,A	Move the data from A to DPTR

4108	04	INC DPTR A	Increment data pointer
4109	F0	MOVX @DPTR,A	Move the data from A to DPTR
410B	80 FE	SJMP 4108	Jump to 4108

	Memory Address	Data
Input	4101	
Output	4500	
	4501 4502	

# **RESULT:**

Thus the logical AND, OR and bit manipulation operations are performed using 8051.

## 2. GENERATION OF SQUARE WAVEFORM USING 8051

## AIM:

To generate a square wave using 8051

## **APPARATUS REQUIRED:**

1.8051 Microcontroller Kit

2. Power supply

# ALGORITHM

- 1. Complement Port 1
- 2. Call Delay subroutine.
- 3. Jump to Step 1.
- 4. Copy the data 0F9 to register R2.
- 5. Decrement the data until it becomes 0.
- 6. Return.

# **PROGRAM:**

ADDRESS	LABEL	MNEMONICS	COMMENT
3000	REPT	CPL 90	COMPLEMENT PORT 1
		LCALL DELAY	CALL DELAY
		SJMP REPT	JUMP TO LABEL
	DELAY	MOV R2, #0F9	COPY 0F9 TO Reg R2
	L1	DJNZ R2, L1	DECREMENT R2
		RET	

## **OUTPUT:**

Square Waveform: Amplitude: Time Period:

## **RESULT:**

Thus the square waveform is generated using 8051 microcontroller.

## **3. SPEED CONTROL OF STEPPER MOTOR**

## Aim:

To write an assembly program to make the stepper motor run in forward and reverse direction.

### **Apparatus required:**

- 1. Stepper motor
- 2. 8051 microcontroller kit
- 3. (0-5V) power supply

# Algorithm:

- 1. Start the programme
- 2. Initialize the DPTR
- 3. Initialize the Counter
- 4. Store the motor sequence in memory
- 5. Get rotation sequence from memory
- 6. Output the sequence to motor
- 7. Produce the appropriate time delay
- 8. Increment DPTR and Decrement the counter
- 9. Counter value is 0 go to step 4
- 10. Stop the programme.

Program:				
MEMORY LOCATION	LABEL	OPCODE	MNEMONICS	COMMENTS
4100	Start	90 45 00	MOV DPTR, #4500	Initialize Data Pointer
4103		78 04	MOV R0, #04	Move the data 04 to R0
4105	JO	ЕО	MOVX A, @DPTR	Move the data from DPTR to Accumulator.
4106		C0 83	PUSH DPH	Push data pointer higher byte
4108		C0 82	PUSH DPL	Push data pointer lower byte
410A		90 FF C0	MOV DPTR, #FFC0	Initialize Data Pointer
410D		7A 01	MOV R2, #01	Move the data 01 to register R2
410F		79 22	MOV R1, #22	Move the data 22 to register R1
4111	DLY 1	7B FF	MOV R3, #FF	Move the data FF to register R3
4113	DLY	DB FE	DJNZ R1, 4111	Jump if no Zero
4115		DA FA	DJNZ R2, 4111	Jump if no Zero
4117		DA FA	DJNZ R3, 4113	Jump if no Zero
4119		FO	MOVX @DPTR, A	Move the Content of Accumulator to memory

411A	]	D0 82	POP DPL	Pop data pointer lower byte
411C		D0 83	POP DPH	Pop data pointer higher byte
411E		A3	INC DPTR	Increment Data Pointer
411F	]	D8 E4	DJNZ R0, 4105	Jump if no Zero
4121		80 DD	SJMP START	Short jump

Interfacing Diagram of Stepper motor with 8051 Microcontroller:



# DATA:

Memory Address	Data
4500	09
4501	05
4502	06
4503	<b>0</b> A

## **Result:**

Thus an assembly language program to control of stepper motor was executed successfully using 8051 Microcontroller kit.

## 4. PROGRAMMING USING SERIAL PORTS IN 8051

#### AIM:

To write the assembly language programme to transmit and receive the data serially using the 8051 microcontroller.

### **APPARATUS REQUIRED:**

1.8051 Microcontroller kit

2. Power supply

## **THEORY:**

### **BAUD RATE IN THE 8051:**

The 8051 transfers and receives data serially at many different baud rates. The baud rate in the 8051 is programmable. This is done with the help of Timer1. The relationship between the crystal frequency and the baud rate in the 8051 is shown below.

The 8051 divides the crystal frequency by 12 to get the machine cycle frequency. In the case XTAL = 11.0592 MHz, the machine cycle frequency is 921.6 KHz (11.0592MHz/12 = 921.6 KHz). The 8051's serial communication UART circuitry divides the machine cycle frequency of 921.6 kHz by 32 once more before it is used by Timer 1 to set the baud rate. Therefore, 921.6 kHz divided by 32 gives 28,800 Hz.

### **SBUF REGISTER:**

SBUF is an 8-bit register used solely for serial communication in the 8051. For a byte of data to be transferred via the TxD line, it must be placed in the SBUF register. Similarly, SBUF holds the byte of data when it is received by the RxD line. SBUF can be accessed like any other register in the 8051.

### SCON(Serial Control) REGISTER:

The SCON register is an 8-bit register used to program the start bit, stop bit, and data bits of data framing, among other things.

## **ALGORITHM:**

#### In programming the 8051 to transfer character bytes serially, the following steps must be taken.

- 1. The TMOD register is loaded with the value 20H, indicating the use of Timer1 in mode2 to set the baud rate.
- 2. The TH1 is loaded with one of the values in Table (2) to set the baud rate for serial data transfer.
- 3. The SCON register is loaded with the value 50H, indicating serial mode 1, where an 8-bit data is framed with start and stop bits.
- 4. TR1is set to1tostart Timer 1.
- 5. TI is cleared by the "CLRTI" instruction.
- 6. The character byte to be transferred serially is written in to the SBUF register.
- 7. The TI flag bit is monitored with the use of the instruction "JNBTI, xx" to see if the character has been transferred completely.
- 8. To transfer the next character, go to Step 5.

## **PROGRAME:**

Write a program for the 8051 to transfer letter "A" serially at 4800 baud, continuously. Solution: MOV TMOD, #20H ; timer 1, mode 2 (auto reload) MOV TH1,#-6 ;4800 baud rate SCON, #50H ;8-bit, 1 stop, REN enabled MOV SETB TR1 ;start timer 1 AGAIN: MOV SBUF, #"A" ;letter "A" to transfer TI,HERE ;wait for the last bit HERE: JNB ;clear TI for next char ΤI CLR SJMP AGAIN ;keep sending A

In programming the 8051 to transfer character bytes serially, the following steps must be taken:

- 1. The TMOD register is loaded with the value 20H, indicating the use of Timer 1 in mode2 to set the baud rate.
- 2. TH1 is loaded with one of the values in Table(2) to set the baud rate.
- 3. The SCON register is loaded with the value 50H, indicating serial mode 1, where 8-bitdata is framed with start and stop bits and receive enable is turned on.
- 4. TR1 is set to1to start Timer 1.
- 5. RI is cleared with the "CLRRI" instruction.
- 6. The RI flag bit is monitored with the use of the instruction "JNBRI, xx" to see if an entire character has been received yet.
- 7. When RI is raised, SBUF has the byte. Its contents are moved in to safe place.
- 8. To receive the next character, go to Step 5.

Write a program for the 8051 to receive bytes of data serially, and put them in P1, set the baud rate at 4800, 8-bit data, and 1 stop bit

```
Solution:
           TMOD,#20H ;timer 1,mode 2(auto reload)
      MOV
      MOV TH1,#-6 ;4800 baud rate
           SCON, #50H ;8-bit, 1 stop, REN enabled
      MOV
                     ;start timer 1
      SETB TR1
      JNB RI,HERE
HERE:
                     ;wait for char to come in
      MOV A, SBUF
                      ;saving incoming byte in A
      MOV P1,A
                      ;send to port 1
                      ;get ready to receive next
      CLR RI
                      ;byte
      SJMP HERE
                      ;keep getting data
```

#### **RESULT:**

Thus the assembly language programme is written to transmit and receive the data serially using the 8051 microcontroller.

## 5. DESIGN OF A DIGITAL CLOCK USING TIMERS/COUNTERS IN 8051.

### Aim:

To write the embedded C programme for designing the digital clock using Timers/counters in 8051.

## **Apparatus Required:**

- 1. Keil IDE software
- 2. 8051 Microcontroller Kit
- 3. 0-5 V power supply

## 8051 microcontroller digital alarm – Circuit diagram:

The circuit for the project is very simple. Connect  $16\times2$  lcd with Port 1 of 8051 (89c51 or 89c52) microcontroller. Rd (read) Wr (write) and En(enable) signals are provided to  $16\times2$  lcd using port 3 pins 5, 6 & 7 of 8051 (89c51 or 89c52) microcontroller. Port 0 pin 0 is used to operate buzzer when ever alarm time reached to its matched clock time. Port 2 is dedicated for our  $4\times3$  keypad. Rows are connected to pin 0, 1, 2 & 3 of port 2. Coulombs are connected to pins 5, 6 & 7 of port 2. Rest of the connections are simple apply 5v to pin 40 and pin 31. Connect crystal 11.0592 MHz to pins 18 and 19. Ground pin 20. Connect reset button to pin 9.



8051 microcontroller digital alarm clock

#### Programme:

```
#include<reg52.h>
                 int keypad();
                 void tochar(unsigned int value);
                 sbit buzzer=P0^0;
                 sbit r0=P2^0; sbit r1=P2^1; sbit r2=P2^2; sbit r3=P2^3; sbit c0=P2^5; sbit c1=P2^6;
                 sbit c2=P2^7; sbit en=P3^6; sbit rs=P3^5; sbit rw=P3^7;
                 char hour[]="HOURS-> 01 - 12",mint[]="MINTS-> 01 - 60";
                 char alarm[]=" DIGITAL ALARM";
                 int count=0,H,M,S=1,hour1[1],mint1[1],H1,M1,hour2[1],mint2[1];
                 void delay(unsigned int no)
                 {
                 unsigned int i,j;
                 for(j=0;j<=no;j++)</pre>
                 for(i=0;i<=2;i++);</pre>
                 }
                 void lcdcmd(unsigned int command){
                 P1=command; rw=0; rs=0; en=0; delay(300); en=1; delay(300); en=0;
                 }
                 void lcddata(char data1)
                 {
                 P1=data1; rw=0; rs=1; en=0; delay(300); en=1; delay(300); en=0;
                 }
                 int scankey(){
                  r0=1;r1=1;r2=1;r3=0;
                   if(c0==0) { return 2; }
                 }
                 void checkconditions(){
                   unsigned int i;
                   i=scankey();
                 while(i==2)
                 {
                 count=0;
                 lcdcmd(0x01); lcdcmd(0x80);
                 while(hour[count]!='\0')
                 {
                 lcddata(hour[count]);count++;
                 }
                 count=0; lcdcmd(0xC8);
                 while(count!=2)
                 {
                 hour2[count]=keypad();count++;
                 }
                 H1=(hour2[0]*10)+hour2[1];
```

count=0; lcdcmd(0x01); lcdcmd(0x80);

```
while(mint[count]!='\0')
{
lcddata(mint[count]);count++;
}
count=0; lcdcmd(0xC8);
while(count!=2)
{
mint2[count]=keypad();count++;
}
M1=(mint2[0]*10)+mint2[1];
count=0; delay(1000); lcdcmd(0x01); lcdcmd(0x80); count=0;
while(alarm[count]!='\0')
{
lcddata(alarm[count]);count++;
}
count=0;
lcdcmd(0xC4);tochar(H);lcddata(':');tochar(M);lcdcmd(0xCB);tochar(S);
i=1;
}
if(S>=60) { M++; }
if(M>=60) { H++; }
if((H==13) && (M>=60) && (S>=60))
{
S=1; H=1; M=1;
lcdcmd(0x01); count=0;
while(alarm[count]!='\0')
{
lcddata(alarm[count]);count++;
}
count=0;
lcdcmd(0xC4);tochar(H);lcddata(' ');lcddata(':');tochar(M);lcddata('
');lcdcmd(0xCB);tochar(S);lcddata(' ');
}
if(S \ge 60)
{ S=1;
lcdcmd(0xC4);tochar(H);lcddata(':');tochar(M);lcdcmd(0xCB);tochar(S);lcddata(' ');
}
if(M>=60)
{ M=1;
lcdcmd(0xC4);tochar(H);lcddata(':');tochar(M);lcddata(' ');lcdcmd(0xCB);tochar(S);lcddata('
');
}}
void clockdelay()
{
int count=0,sec;
for(sec=0;sec<60;sec++){</pre>
count=0;
while(count!=495)
```

```
{
TMOD=0x01; //16-bit timer0 selected
TH0=0xF8;
           // Loading high byte in TH
TL0=0xCC; // Loaded low byte in TL
TR0=1;
          // Running the timer
while(!TF0); //Checking the timer flag register if it is not equal to 1
TR0 = 0;
            // If TF0=1 stop the timer
TF0 = 0;
            // Clear the Timer Flag bit for next calculation
count++;
}
S++;lcdcmd(0xCB);tochar(S);
checkconditions();
if(H==H1)
{
if(M==M1)
buzzer=0;
}}}
void lcdint()
{
lcdcmd(0x30); delay(3000); lcdcmd(0x30); delay(3000); lcdcmd(0x30); delay(3000);
lcdcmd(0x30); delay(3000); lcdcmd(0x30); delay(3000); lcdcmd(0x38); delay(3000);
lcdcmd(0x01); delay(3000); lcdcmd(0x0C); delay(3000); lcdcmd(0x80); delay(3000);
}
int keypad()
{
char c='a';
while(c!='s'){
r0=0;r1=1;r2=1;r3=1;
if(c0==0){lcddata('1');delay(20000);c='s';return 1;}
 if(c1==0){lcddata('2');delay(20000);c='s';return 2;}
if(c2==0){lcddata('3');delay(20000);c='s';return 3;}
r0=1;r1=0;r2=1;r3=1;
if(c0==0){lcddata('4');delay(20000);c='s';return 4;}
if(c1==0){lcddata('5');delay(20000);c='s';return 5;}
if(c2==0){lcddata('6');delay(20000);c='s';return 6;}
r0=1;r1=1;r2=0;r3=1;
if(c0==0){lcddata('7');delay(20000);c='s';return 7;}
if(c1==0){lcddata('8');delay(20000);c='s';return 8;}
 if(c2==0){lcddata('9');delay(20000);c='s';return 9;}
r0=1;r1=1;r2=1;r3=0;
if(c1==0){lcddata('0');delay(20000);c='s';return 0;}
  c='a';
}
}
void settime()
                   //Setting time
{
lcdcmd(0x01); lcdcmd(0x80);
while(hour[count]!='\0')
```

```
{
lcddata(hour[count]);count++;
}
count=0; lcdcmd(0xC8);
while(count!=2)
{
hour1[count]=keypad();count++;
}
H=(hour1[0]*10)+hour1[1];
count=0; lcdcmd(0x01); lcdcmd(0x80);
while(mint[count]!='\0')
{
lcddata(mint[count]);count++;
}
count=0; lcdcmd(0xC8);
while(count!=2)
{
mint1[count]=keypad();count++;
}
M=(mint1[0]*10)+mint1[1];
count=0; delay(1000); lcdcmd(0x01); lcdcmd(0x80);
}
void start(){
count=0; lcdcmd(0x01); lcdcmd(0x80);
while(alarm[count]!='\0')
{
lcddata(alarm[count]);count++;
}
count=0;
lcdcmd(0xC4);
tochar(H);
lcddata(':');
tochar(M);
lcdcmd(0xCB);
tochar(S);
}
void tochar(unsigned int value)
{
char tendigit,unitdigit;
tendigit=value/10;
if((tendigit+48)>='1')
lcddata(tendigit+48);
unitdigit=value%10;
lcddata(unitdigit+48);
}
void main()
{
```

unsigned int i=0;

```
P0=0x00; P1=0x00; P2=0xF0; P3=0x0F;
lcdint(); buzzer=1;
settime(); start();
while(1)
{
clockdelay();
buzzer=1;
}
}
```

# 8051 microcontroller digital alarm clock – Project code:

First included the necessary header file **reg52.h**. This header file should be included whenever writing and compiling the 8051 microcontroller program in keil ide. It includes all the necessary linker and debuggers for 8051 microcontroller configuration.

Then **keypad**() function is declared. This function scans the keys pressed on the  $4\times3$  keypad. Next comes **tochar**() function. It is used to deal with integer type numbers to be displayed on the  $16\times2$  lcd. Integer numbers must be converted to 8-bit character format for displaying on  $16\times2$  lcd.

Then single bits of ports are declared for connecting our buzzer and  $4\times3$  keypad. Next character arrays are declared they are used as messages to be displayed on  $16\times2$  lcd on suitable conditions. **delay**() function is used to generate some delay for suitable purposes. **lcdcmd**() function is sending commands to lcd. **lcddata**() function is sending and printing data on lcd screen. **checkconditions**() function is checking conditions like if seconds reached to 60 increment the minute, if minutes reached to 60 increment hour, check if clock reached the alarm time ring the buzzer.

**clockdelay**() function is producing seconds, minutes and hours for us. **lcdinit**() initializing the lcd. **settime**() function is setting time of clock on every reset of the program. User have to input the time using  $4\times3$  keypad. **start**() function is printing the time on  $16\times2$  lcd at start of the program execution.

Result:

Thus the digital clock is designed by writing the embedded C programme using Timers/ Counters in 8051 microcontroller.

### **STUDY OF ARM PROCESSOR**

#### Features of ARM DEVELOPMENT KIT Processor:

- 16-bit/32-bit ARM7TDMI-S microcontroller in a tiny LQFP64 package.8 kB to 40 kB of on-chip static RAM and 32 kB to 512 kB of on-chip flash memory. 128- bit wide interface/accelerator enables high-speed 60 MHz operation. In- System/In-Application Programming (ISP/IAP) via on-chip boot loader software.
- Single flash sector/full chip erase in 400 ms and programming of 256 bytes in 1 ms.USB 2.0 Full-speed compliant device controller with 2 kB of endpoint RAM. The LPC2146/48 provides 8 kB of on-chip RAM accessible to USB by DMA.
- One or two (LPC2141/42 vs. LPC2144/46/48) 10-bit ADCs provide a total of 6/14 analog inputs, with conversion times as low as 2.44 µs per channel. Single 10-bit DAC provides variable analog output (LPC2142/44/46/48 only).Two 32-bit timers/external event counters (with four capture and four compare channels each), PWM unit (six outputs) and watchdog.
- Low power Real-Time Clock (RTC) with independent power and 32 kHz clock input. Multiple serial interfaces
  including two UARTs (16C550), two Fast I2C- bus (400 kbit/s), SPI and SSP with buffering and variable data
  length capabilities.
- Vectored Interrupt Controller (VIC) with configurable priorities and vector addresses.Up to 45 of 5 V tolerant fast general purpose I/O pins in a tiny LQFP64 package.Up to 21 external interrupt pins available.
- 60MHz maximum CPU clock available from programmable on-chip PLL with settling time of 100µs.On-chip integrated oscillator operates with an external crystal from 1 MHz to 25 MHz.Power saving modes include Idle and Power- down.
- Individual enable/disable of peripheral functions as well as peripheral clock scaling for additional power optimization.Processor wake-up from Power-down mode via external interrupt or BOD.Single power supply chip with POR and BOD circuits:CPU operating voltage range of 3.0 V to 3.6 V (3.3 V ± 10 %) with 5 V tolerant I/O pads.

## **General Block Diagram:**



## **Power Supply:**

• The external power can be AC or DC, with a voltage between (9V/12V, 1A output) at 230V AC input. The ARM board produces +5V using an LM7805 voltage regulator, which provides supply to the peripherals.

• LM1117 Fixed +3.3V positive regulator used for processor & processor related peripherals.

## **Flash Programming Utility:**

• NXP (Philips) NXP Semiconductors produce a range of Microcontrollers that feature both on-chip Flash memory and the ability to be reprogrammed using In-System Programming technology.

## **On-board Peripherals:**

- 8-Nos. of Point LED's (Digital Outputs)
- 8-Nos. of Digital Inputs (slide switch)
- Lines X 16 Character LCD Display
- I2C Enabled 4 Digit Seven-segment display
- 128x64 Graphical LCD Display
- X 4 Matrix keypad
- Stepper Motor Interface
- 2 Nos. Relay Interface
- Two UART for serial port communication through PC

- Serial EEPROM
- On-chip Real Time Clock with battery backup
- PS/2 Keyboard interface(Optional)
- Temperature Sensor
- Buzzer(Alarm Interface)
- Traffic Light Module(Optional)

# **Pin Configuration:**



# **CONNECTORS & CONNECTION DETAILS:**

Туре	Label	Description
26 Pin FRC	CON1	For GPIO
20 Pin FRC	CON2	For JTAG
DB9	CON3	UART0
DB9	CON4	UART1
3 Pin RMC	CON5	PWM output
6 Pin RMC	CON6	ADC input
3 Pin RMC	CON7	DAC output from OPAMP
3 Pin RMC	CON8	Temperature sensor input
16 Pin RMC	CON9	LCD Display
Power jack	C10	Power input
PS2	CON11	Keyboard interface

# 6. INTERFACING ADC & DAC WITH ARM PROCESSOR

# Aim:

To develop a C-Language program for reading an on-chip ADC, convert into decimal and to display it in PC and to generate a square wave depending on this ADC reading. The ADC input is connected to any analog sensor/ on board potentiometer.

# **Pre Lab Questions:**

- 1. List the types of ADC and DAC
- 2. Define resolution.
- 3. Summarize the features of Conversion time in ADC.
- 4. What is the function of Sample-and-hold circuits in analog-to digital converters?
- 5. Why are internal ADCs preferred over external ADCs?

# **Apparatus & Software Required:**

- **1**. LPC2148 Development board.
- **2**. KeilµVision 5 software.
- **3.** Flash Magic.
- **4**. USB cable.
- **5.** CRO.

## **Theory:**

The LPC 2148 has 10-bit successive approximation analog to digital converter. Basic clocking for the A/D converters is provided by the VPB clock. A programmable divider is included in each converter, to scale this clock to the 4.5 MHz (max) clock needed by the successive approximation process. A fully accurate conversion requires 11 of these clocks. The ADC cell can measure the voltage on any of the ADC input signals.ARM Board has one potentiometer for working with A/D Converter. Potentiometer outputs are in the range of 0V to 3.3V. Switch select in right position for reading the Potentiometer value by ADC.

# Interfacing Diagram of ADC with LPC2148:



**Programme: ADC Interfacing:** 

// <i>Read</i> All	DC data from given channel number ADC(charChannelNumber)	
{	Int val,ch;	
	ch=1< <channelnumber;< td=""><td></td></channelnumber;<>	
10-bitAIN	AD0C =0x00210400 ch; V@3MHz AD0CR = 0x01000000; //StartA/D Conversion	//SetupA/D:
	do { val=AD0DR2; //Read A/D Data Register	
the conve the A/D r	<pre>} while((val&amp;0x8000000)==0); ersion to complete val=((val&gt;&gt;6)&amp;0x03FF); result</pre>	//Waitfor //Extract
	AD0C &= ~0x01000000; R	//StopA/D
Return th }	return(val); ne Data Read	//
Void gcc {	main()	
}		

# ADC PROGRAM PORT DETAILS:

ARM	DETAILS
P0.29	ADC0.2
PO.10	RS LCD PIN

# **Programme: DAC Interfacing:**

#include<"LPC214X.H">

voidInitializeDAC(void); intmain(void) ł long c; InitializeDAC(); //Initialize DAC while(1) { DACR=**0x00;** //SetDAC= 0 **for**(c=0;c<0xf0000;c++); DACR=0x0000ffc0; //Delay (0x3ff<<6) //SetDAC= for(c=0;c<0xf0000;c++);</pre> } //Delay voidInitializeDAC(void) Ł PINSEL1=0x00080000; //Set P0.25 forDAC output voidgccmain()

## DAC PROGRAM PORT DETAILS:

ARM	DETAILS
P0.25	DAC ENABLE PIN
#### **Interfacing Diagram of DAC with LPC2148:**



#### Procedure

- **1**. Follow the steps to create a New project
- **2.** Type the below code and save it with the name (anyname.c)
- **3.** Follow the steps to create a New Project to compile and build the program
- Follow the procedures in to download your Hex code to processor using Flash Magic Software.

#### **RESULT:**

The C-Language program for reading an on-chip ADC, convert into decimal and to display it in LCD was written & output is verified with the ADC input is connected to on board potentiometer

The DAC, convert digital data into analog signal & output is verified with the DAC input and the square wave has been generated to display it in CRO.

#### 7. A. BLINKING OF LEDS

#### AIM:

To develop a 'C' program to make the LED blink (including delay routine). Upon change in the delay program the speed should vary.

#### **Apparatus & Software Required:**

- 1. LPC2148 Development board.
- 2. KeilµVision 5 software.
- 3. Flash Magic.
- 4. USB cable.

#### Theory:

LEDs are based on the semiconductor diode. When the diode is forward biased (switched on), electrons are able to recombine with holes and energy is released in the form of light. This effect is called electroluminescence and the color of the light is determined by the energy gap of the semiconductor.

#### **Procedure:**

- 1. Follow the steps to create a New project
- 2. Type the below code and save it with the name (anyname.c)
- 3. Follow the steps to create a New Project to compile and build the program
- 4. Follow the procedures in to download your Hex code to processor using Flash Magic Software.

#### Interfacing Diagram of LEDs With LPC2148:



#### **Program:**

```
#include<LPC214x.H>
                                                   /* LPC214x definitions*/
 voidwait(void)
                                                   /* wait function*/
 {
 int
       d;
 for (d = 0; d <1000000;d++);
                                                   /* only to delay for LED flashes*/
 }
 int main (void)
 {
 IODIR0=0x80002000;
                                                   /* P0.13 and P0.31 defined as Outputs*/
 while(1)
                                                   /* Loop forever*/
 {
 IOCLR0=0x80002000;
                            /*Active Low outputs makes theLEDsON*/
 wait ();
 IOSET0=0x80002000;/* High outputs makes the LEDs OFF*/
wait();
}
}
```

#### FLASHING LED PROGRAM PORT DETAILS

ARM	DETAILS
P0.13	LED PIN
P0.31	LED PIN

#### **RESULT:**

The C-Language program to make the LED blink was developed and output was verified. Upon change in the delay program the speed variation was verified.

#### 7. B. LCD INTERFACING

#### AIM:

To develop a C-Language program for displaying the characters in the LCD module. The display should come in the desired line and column.

#### Apparatus & Software Required:

- **1.** LPC2148 Development board.
- **2.** KeilµVision 5 software.
- **3.** Flash Magic.
- **4.** USB cable.

#### Interfacing Diagram of LCD with LPC 2148:



#### LCD PROGRAM PORT DETAILS:

ARM	Details		
PO.10	RS	LCD	PIN
P1.11	CE	LCD	PIN

#### **Procedure:**

- 1. Follow the steps to create a New project
- 2. Type the below code and save it with the name (anyname.c)
- 3. Follow the steps to create a New Project to compile and build the program
- 4. Follow the procedures in to download your Hex code to processor using Flash Magic Software.

#### **Program:**

```
/*LCDRoutinesfor2lineX16CharactersDisplay*/
#include"LPC214x.h"
                                               /*LPC214xdefinitions
*/
voidWriteCommandLCD(unsignedcharCommandByte);
void WriteDataLCD(unsigned char DataByte);
void LCDDelay(void);
voidLCDDelay1600(void);
voidSendByte(unsignedcharValue); void
InitializeLCD(void);
voidDataAddressDirection(void);
voidDisplayLCD(charLineNumber,char*Message);
voidDisplayLCD2Digit(charLineNumber,charCharPosition,charData);
intmain(void)
{
       InitializeLCD();
       //InitializegraphicsLCD
       DisplayLCD(0,"NXP2148 ARM");
       DisplayLCD(1,"EvaluatinSystem");
       while(1);
}
voidgccmain()
}
```

#### **RESULT:**

The C-Language program for displaying the characters in the LCD module and the output was verified on the LCD on the desires line and column/address.

#### 8. A. INTERFACING KEYBOARD

#### AIM:

To develop a C-Language program for displaying the Key pressed in the Keypad in the LCD module. The display should come in the desired line and column.

#### Apparatus & Software Required:

- **1.** LPC2148 Development board.
- 2. KeilµVision5software.
- **3.** Flash Magic.
- 4. USB cable.

#### Theory:

The Matrix keyboard is used to minimize the number of I/O lines. Normally it is possible to connect only one key or switch with an I/O line. If the number of keys in the system exceeds themore I/O lines are required. To reduce the number of I/O lines the keys are connected in the matrix circuit. Keyboards use a matrix with the rows and columns made up of wires. Each key acts like a switch. When a key is pressed a column wire makes contact with row wire and completes a circuit. For example 16 keys arranged in a matrix circuit uses only 8 I/O lines.

#### Interfacing Diagram of Keyboard with LPC2148:



#### **Program:**

```
#include <LPC214x.h>
#include "mat_7seg.h"
int main()
{
unsigned int key, last key, Disp key;
                        // Initialize matrix keyboard and 7segment display
init_Matrix_7seg();
clearall_7seg(); // clear 7 segment display
last key=0;
               // Initialize this variable to zero
while(1)
{
                        // scan for a valid key press
key=catch_key();
if(key!=0)
                // zero means no key is pressed
{
if(key!=last key)
                        // check whether the same key is pressed again(assume this as STEP1)
{
Disp_key=key;
                        // valid new key is stored in another variable
last_key=key; // this variable's value is used forSTEP1
}
}
//Display Number(Disp key); /*this function is used to display number in decimalformat*/
Alpha_Dispay(4,Disp_key);
                                /*this function is used to display number in hex format (single digitonly)*/
}
}
```

#### **Procedure:**

- Follow the steps to create a New project
- Type the below code and save it with the name (anyname.c)
- Follow the steps to create a New Project to compile and build the program
- Follow the procedures in to download your Hex code to processor using Flash Magic Software.

#### **Result:**

The C-Language program for displaying the Key pressed in the Keyboard is displayed in the LCD module and the output was verified on the LCD on the desires line and column/address.

#### 8. B. INTERFACING STEPPER MOTOR

Aim:

To write C Programs for running stepper motor either in clock- wise or counter-clock- wise and the direction of the rotation of the stepper motor.

#### **Apparatus & Software Required:**

- 1. LPC2148 Development board.
- 2. KeilµVision 5 software.
- 3. Flash Magic.
- 4. USB cable.
- 5. Stepper Motor.

#### Theory:

Stepper motors, effectively have multiple "toothed" electromagnets arranged around a central metal gear. To make the motor shaft turn, first one electromagnet is given power, which makes the gear's teeth magnetically attracted to the electromagnet's teeth. When the gear's teeth are thus aligned to the first electromagnet, they are slightly offset from the next electromagnet. So when the next electromagnet is turned on and the first willturn off, the gear rotates slightly to align with the next one and from there the process is repeated. Each of those slight rotations is called a "step." In that way, the motor can be turned to a précised angle. There are two basic arrangements for the electromagnetic coils: bipolar and unipolar.

#### Interfacing diagram of stepper motor with LPC2148:



#### **Program:**

```
int
       d;
for (d = 0; d < 10000; d++);
                            /* only to delay for LED flashes*/
}
call_stepper_forw()
ł
IOCLR1 = 0X00FF0000; IOSET1 = 0X00040000;
//wait();
//wait();
wait();
wait();
IOCLR1 = 0X00FF0000; IOSET1 = 0X00060000;
//wait();
//wait();
wait();
wait();
IOCLR1 = 0X00FF0000; IOSET1 = 0X00070000;
//wait();
//wait();
wait();
wait();
IOCLR1 = 0X00FF0000; IOSET1 = 0X00050000;
//wait();
//wait();
wait();
wait();
}
int main (void)
ł
IODIR1 |= 0xFFFFFFF; IOCLR1
|= 0X00FF0000;
wait();
while(1)
               /*LoopForever*/
{
call_stepper_forw();
//
       wait();
//
       wait();
wait();
wait();
IOCLR1 = 0X00FF0000;
}
}
```

#### **Procedure:**

- 1. Follow the steps to create a New project
- 2. Type the below code and save it with the name (anyname.c)
- 3. Follow the steps to create a New Project to compile and build the program
- 4. Follow the procedures in to download your Hex code to processor using Flash Magic Software.

# **STEPPER MOTOR PROGRAM PORT DETAILS:**

ARM	DETAILS
P1.16	STEP 1
P1.17	STEP 2

#### **Result:**

The C-Language program for running stepper motor in clock wise and anti clock wise direction was verified.

# VLSI AND CHIP DESIGN

Edited by M.SOUNDHARYA



# VLSI AND CHIP DESIGN

COPYRIGHT @ 2021 BY M.SOUNDHARYA

ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 9789362556653

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

CHAPTER	CONTENT	PAGE
	INTRODUCTION	10
1	MOS TRANSISTOR PRINCIPLES	21
	T. DIVYA MANOHARI	
2	MODES OF MOS TRANSISTOR	45
	D.HARIKRISHNAN	
3	COMBINATIONAL LOGIC CIRCUITS	52
	S. MAHESHWARAN	
4	LAYOUT DESIGN RULES AND GATE LAYOUTS	76
	R.SAVITHA	
5	SEQUENTIAL LOGIC CIRCUITS	95
	P.GEETHA	
6	CLOCKING STRATEGIES	109
	R.SARASWATHI	
7	INTERCONNECT, MEMORY ARCHITECTURE	124
	A.AMUDHA	
8	ARITHMETIC CIRCUITS	150
	R.SAVITHA	
9	ASIC DESIGN	172
	S.SARASWATHY	
10	ASIC TESTING	195
	E.PRIYADHARSHINI	

257

# **CHAPTER 1-MOS TRANSISTOR PRINCIPLES** T. DIVYA MANOHARI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

#### **INTRODUCTION: (VLSI)**

- In 1958, Jack Kilby built the first integrated circuit flip-flop at Texas Instruments.
- Bell Labs developed the bipolar junction transistor. Bipolar transistors were more reliable, lessnoisy and more power-efficient.
- In 1960s, Metal Oxide Semiconductor Field Effect Transistors (MOSFETs) began to enter in the production.
- MOSFETs offer the compelling advantage that; they draw almost zero control current whileidle.
- They come in two flavors: nMOS and pMOS, using n-type and p-type silicon respectively.
- In 1963, Frank Wanlass at Fairchild described the first logic gates us-ing MOSFETs. Fairchild's gates used both nMOS and pMOS transistors, naming as Complementary Metal Oxide Semiconductor (CMOS).
- Power consumption became a major issue in the 1980s as hundreds of thousands of transistors were integrated onto a single die.
- CMOS processes were widely adopted and replaced nMOS and bipo-lar processes for all digital logic applications.
- In 1965, Gordon Moore observed that plotting the number of transis-tors that can be most economically manufactured on a chip gives a



straight line on a semi logarithmic scale.

# CHAPTER 2-MODES OF MOS TRANSISTOR D.HARIKRISHNAN

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

- o <u>Delay estimation</u>
  - Important definitions for delay estimation:
- Propagation delay time (t<sub>pd</sub>):
  - Propagation delay time is defined as maximum time from the input crossing 50% to the outputcrossing 50%.
- Contamination delay time (tcd):
  - Contamination delay time is defined as minimum time from the input crossing 50% to theoutput crossing 50%.
- $\triangleright$  Rise time (t<sub>r</sub>):
  - Rise time is defined as time for a waveform to rise from 20% to 80% of its steady-state value
- > Fall time ( $t_f$ ):
  - Fall time is defined as time for a waveform to fall from 80% to 20% of its steady-state value
  - Edge rate is average of rise and fall time,  $(t_{rf}) = (t_r + t_f)/2$
- Delay estimation response curve:
  - When an input changes, the output will retain its old value for at least the contamination delayand take on its new value in, at most the propagation delay.
  - Delays for the output rising is  $t_{pdr}/t_{cdr}$  and the output falling is  $t_{pdf}/t_{cdf}$ .
  - Rise/fall times are also called as slopes or edge rates.
  - Propagation and contamination delay times are also called as maxtime and min-timerespectively.
  - The gate that charges or discharges a node is called the driver. The gates and wires beingdriven, are called the load. Propagation delay is usually called as delay.
  - Arrival times and propagation delays are defined separately for rising and falling transitions.
  - The delay of a gate may be different from different inputs. Earliest arrival times can also be computed based on contamination delays.
  - Expression of delay for rising output is  $t_{PLH} = 0.69 R_P.C_L$ Where,  $R_P$  effective resistance of pMOS transistor
    - C<sub>L</sub> load capacitance of CMOS inverter.
  - Expression of delay for falling output is  $t_{PHL} = 0.69 R_N.C_L$ Where,  $R_N$  effective resistance of nMOS transistor

# CHAPTER 3- COMBINATIONAL LOGIC CIRCUITS S. MAHESHWARAN

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Combinational logic circuits are fundamental components in digital electronics and computer engineering. Unlike sequential circuits, which depend on past inputs (state) to determine their outputs, combinational logic circuits generate outputs solely based on the current inputs. Here's a quick overview of the key concepts:

#### **Basic Principles**

- 1. **Logic Gates**: The building blocks of combinational logic circuits are logic gates. The basic gates include:
  - **AND Gate**: Outputs 1 if all its inputs are 1.
  - **OR Gate**: Outputs 1 if at least one of its inputs is 1.
  - **NOT Gate**: Outputs the inverse of its input (0 becomes 1, and 1 becomes 0).
  - **NAND Gate**: Outputs 0 only if all its inputs are 1 (inverse of AND).
  - **NOR Gate**: Outputs 1 only if all its inputs are 0 (inverse of OR).
  - **XOR Gate**: Outputs 1 if an odd number of inputs are 1.
  - XNOR Gate: Outputs 1 if an even number of inputs are 1.
- 2. **Boolean Algebra**: Combinational logic circuits are often described using Boolean algebra, which involves variables and operations like AND, OR, and NOT. Boolean expressions can be simplified using Boolean laws and theorems to design efficient circuits.
- 3. **Truth Tables**: A truth table lists all possible input combinations and the corresponding outputs for a logic circuit. It's a useful tool for understanding and designing combinational logic circuits.

#### **Common Combinational Logic Circuits**

- 1. Adders: Circuits that perform addition of binary numbers.
  - Half Adder: Adds two single-bit numbers and produces a sum and a carry output.
  - **Full Adder**: Adds three single-bit numbers (including carry from a previous bit) and produces a sum and a carry output.
- 2. Subtractors: Circuits that perform binary subtraction.
  - **Half Subtractor**: Subtracts one single-bit number from another, producing a difference and a borrow output.
  - **Full Subtractor**: Subtracts one single-bit number from another while accounting for a borrow from a previous stage.
- 3. **Multiplexers (MUX)**: Devices that select one of several input signals and forward it to a single output line. They are controlled by selection lines.
- 4. **Demultiplexers (DEMUX)**: Devices that take a single input and channel it to one of several outputs based on selection lines.
- 5. **Decoders**: Convert binary information from encoded inputs to unique outputs. For example, a 2-to-4 decoder has 2 inputs and 4 outputs, where only one output is active at a time based on the input combination.

# CHAPTER 4- LAYOUT DESIGN RULES AND GATE LAYOUTS R.SAVITHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Layout design rules and gate layouts are crucial aspects of designing integrated circuits (ICs) to ensure that they function correctly and efficiently. Here's a detailed overview:

# Layout Design Rules

Layout design rules are a set of guidelines used to create the physical layout of ICs. These rules ensure that the circuit operates reliably and that manufacturing processes can be executed without issues. They cover various aspects, including:

- 1. **Minimum Feature Size**: Specifies the smallest size of features (e.g., transistors, wires) that can be fabricated. This is driven by the technology node (e.g., 5 nm, 7 nm).
- 2. Spacing Rules:
  - **Line-to-Line Spacing**: Minimum distance between two metal lines to prevent short circuits or interference.
  - **Line-to-Contact Spacing**: Minimum distance between metal lines and contact pads to ensure proper connections.

#### 3. Width Rules:

• **Minimum Width**: Minimum width of metal lines and diffusion areas to ensure sufficient current carrying capability and to prevent fabrication issues.

#### 4. Contact and Via Rules:

- Contact Size: Minimum size of contacts that connect different layers of the IC.
- **Via Spacing**: Minimum distance between vias (vertical interconnects) to avoid electrical shorts.
- 5. **Overlap Rules**: Specifies how different layers, such as metal layers and diffusion regions, should overlap or avoid overlapping to prevent issues like short circuits.
- 6. **Antenna Effect**: Ensures that intermediate metal layers do not accumulate excess charge, which could cause problems in subsequent layers during fabrication.
- 7. **Corner and Edge Rules**: Guidelines for designing corners and edges to avoid fabrication defects and ensure uniformity.

# **Gate Layouts**

Gate layouts refer to the physical representation of logic gates in an IC. Each type of logic gate has a specific layout design. Here's a brief overview of the common gate layouts:

#### 1. NMOS and PMOS Transistors:

- **NMOS Transistor**: Typically has an n-type channel that conducts when a positive voltage is applied to the gate.
- **PMOS Transistor**: Has a p-type channel that conducts when a negative voltage is applied to the gate.
- 2. Basic Logic Gate Layouts:
  - AND Gate: Implemented using series connection of NMOS transistors and parallel connec

# CHAPTER 5- SEQUENTIAL LOGIC CIRCUITS P.GEETHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Sequential logic circuits are a category of digital circuits where the output depends on both the current inputs and the history of previous inputs. This characteristic is due to the presence of memory elements within the circuit, which maintain the circuit's state over time. Unlike combinational logic circuits, which produce outputs based solely on current inputs, sequential circuits use their stored state to influence their output, enabling them to exhibit more complex behaviors.

# **Key Concepts**

- 1. **State**: Represents the current condition of the memory elements within the circuit. It is often defined by state variables that capture the information stored in the circuit.
- 2. Clock Signal: A periodic signal that synchronizes the changes in the state of sequential circuits. It dictates when the circuit reads inputs and updates its state, providing a rhythm for the operation of the circuit.
- 3. **Memory Elements**: Components that store information within the circuit. Common types include:
  - Flip-Flops: Fundamental elements that store a single bit of data. Types include:
    - **D** Flip-Flop: Captures and holds the value of the data input (D) on the rising or falling edge of the clock signal (CLK).
    - JK Flip-Flop: Has inputs (J and K) that control its toggle, set, or reset behavior based on the clock signal.
    - **T Flip-Flop**: Toggles its output on each clock edge if the T input is high.
    - **SR Flip-Flop**: Uses set (S) and reset (R) inputs to control its output state.
  - **Registers**: Collections of flip-flops used to store multiple bits of data. They can be configured for operations like data shifting (shift registers) or temporary data storage.
- 4. **Counters**: Sequential circuits designed to count pulses and produce a binary count value. Types include:
  - **Binary Counters**: Count in binary sequence, increasing or decreasing with each clock pulse.
  - **Decimal Counters**: Count in decimal sequence, often used in applications that require decimal counting.
- 5. **Finite State Machines (FSMs)**: Abstract models used to design sequential logic circuits. FSMs have a finite number of states, and state transitions are determined by inputs and the current state. They can be classified into:
  - Mealy Machines: Outputs depend on both the current state and the current inputs.
  - Moore Machines: Outputs depend solely on the current state.

# **Design and Analysis of Sequential Logic Circuits**

- 1. **State Diagram**: A visual representation of an FSM where states are shown as nodes and transitions as directed edges, illustrating the circuit's behavior and state changes.
- 2. **State Table**: A tabular format that outlines how the state and outputs of an FSM change in response to various inputs.
- 3. **Timing Diagrams**: Graphs that show how signals change over time, synchronized with clock pulses. They are crucial for verifying that the sequential circuit operates as intended.
- 4. Characteristic Equations: Mathematical expressions that describe the behavior of memory elements, linking the next state to the current state and inputs. 95

# CHAPTER 6- CLOCKING STRATEGIES R.SARASWATHI

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Clocking strategies are crucial in digital circuit design, particularly for sequential circuits where the synchronization of data and operations is essential. Proper clocking ensures that all parts of the circuit work together harmoniously and helps avoid issues such as timing errors or metastability. Here's a comprehensive look at different clocking strategies and their implications:

#### 1. Synchronous Clocking

**Definition**: In synchronous clocking, all flip-flops and sequential elements in a circuit are driven by a common clock signal. This clock signal synchronizes the operation of the entire circuit.

#### Advantages:

- **Predictable Timing**: Simplifies timing analysis as all sequential elements change state simultaneously.
- **Design Simplicity**: Easier to design and verify because timing constraints are uniform across the circuit.
- **Reduced Clock Skew**: By using a single clock, clock skew (the variation in arrival times of the clock signal at different parts of the circuit) is minimized.

#### **Disadvantages**:

- **Clock Distribution**: Requires a well-designed clock distribution network to ensure the clock signal reaches all parts of the circuit reliably and evenly.
- **Power Consumption**: Constantly toggling all elements with the clock can lead to higher power consumption.

**Applications**: Most digital systems, including processors, memory systems, and communication interfaces, use synchronous clocking due to its simplicity and effectiveness.

#### 2. Asynchronous Clocking

**Definition**: Asynchronous clocking (or clockless) circuits do not use a global clock signal. Instead, they rely on signals or events to trigger state changes.

#### Advantages:

- Lower Power Consumption: Can be more power-efficient as components only switch when necessary, not on every clock edge.
- **Potentially Higher Speed**: Can be faster in certain applications as they do not need to wait for a global clock edge.

# CHAPTER 7- INTERCONNECT, MEMORY ARCHITECTURE A.AMUDHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

**Interconnect and memory architecture** are fundamental aspects of digital systems, particularly in the design of integrated circuits (ICs) and computer systems. They involve the physical and logical structures used to connect and manage data between different components and memory elements. Here's an in-depth look at both concepts:

#### Interconnect

**Interconnect** refers to the network of wires, traces, and other electrical paths that connect different components within an IC or between ICs in a system. Effective interconnect design is crucial for ensuring that data can be transmitted efficiently and reliably.

#### Key Concepts:

- 1. Signal Integrity:
  - **Propagation Delay**: The time it takes for a signal to travel from one point to another. Minimizing delay is critical for high-speed designs.
  - **Crosstalk**: Unwanted interference between signal lines. It can be minimized through proper spacing and shielding.
  - **Impedance Matching**: Ensures that the interconnect impedance matches the source and load to prevent reflections and signal loss.
- 2. Interconnect Types:
  - **Global Interconnects**: Used for long-distance connections within an IC or between ICs. Examples include global clock distribution networks.
  - **Local Interconnects**: Shorter connections within a specific area of an IC, such as between neighboring gates or cells.
  - **Metal Layers**: Multiple metal layers are used to form interconnects in ICs, with each layer serving a different purpose, such as connecting different levels of the chip or handling power and ground.

#### 3. Interconnect Design:

- **Routing**: The process of designing the paths for interconnects. It involves ensuring that signals are routed efficiently while avoiding congestion and interference.
- **Buffering**: Inserting buffers along the interconnect path to drive longer lines and maintain signal strength.

#### 4. **On-Chip Communication**:

- **Bus Systems**: A shared communication path used to connect multiple components. Buses can be parallel or serial.
- **Network-on-Chip** (NoC): A more advanced method for on-chip communication, where a network topology is used to manage data flow between different cores or modules.

# CHAPTER 8- ARITHMETIC CIRCUITS R.SAVITHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Arithmetic circuits are specialized digital circuits designed to perform arithmetic operations such as addition, subtraction, multiplication, and division. These circuits are fundamental to various applications in digital systems, including processors, digital signal processors (DSPs), and other computational hardware. Below is an overview of key types of arithmetic circuits and their components:

# **1. Adder Circuits**

Adders are circuits that perform the addition of binary numbers. There are several types of adders, each with different characteristics and applications.

#### a. Half Adder

- **Function**: Adds two single-bit binary numbers.
- **Inputs**: Two bits (A and B).
- **Outputs**: Sum (S) and Carry-out (C\_out).
- Equations:
  - $\circ \quad Sum(S) = A XOR B$
  - $\circ$  Carry-out (C\_out) = A AND B
- Implementation: Consists of an XOR gate (for the sum) and an AND gate (for the carry).

#### **Truth Table for Half Adder:**

A B Sum (S) Carry-out (C\_out)

#### b. Full Adder

- **Function**: Adds three bits, including a carry-in bit.
- **Inputs**: Two bits (A and B) and a carry-in (C\_in).
- **Outputs**: Sum (S) and Carry-out (C\_out).
- Equations:
  - Sum (S) = A XOR B XOR C\_in
  - $\circ$  Carry-out (C\_out) = (A AND B) OR (C\_in AND (A XOR B))
- Implementation: Can be built using two half adders and an OR gate.

# CHAPTER 9- ASIC DESIGN S.SARASWATHY

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

ASIC (Application-Specific Integrated Circuit) design refers to the process of creating custom integrated circuits tailored to specific applications or functions. Unlike general-purpose ICs (like CPUs or GPUs) that are designed for a broad range of uses, ASICs are designed for specific tasks or applications, offering optimized performance, power consumption, and area efficiency. Here's an overview of the ASIC design process, its components, and considerations.

#### 1. ASIC Design Flow

The design flow for an ASIC typically follows these steps:

#### a. Specification

- **Definition**: Define the exact requirements and functionalities of the ASIC, including performance, power consumption, area, and interfaces.
- **Documentation**: Create a detailed specification document outlining the desired features, constraints, and performance metrics.

#### b. Design Entry

- **HDL Coding**: Design the ASIC using hardware description languages (HDLs) like VHDL or Verilog. This step involves writing code that describes the behavior and structure of the circuit.
- **High-Level Design**: Define the high-level architecture, including modules, data paths, and control units.

#### c. Design Verification

- **Simulation**: Test the HDL code through simulation to ensure that the design meets the specifications. This involves functional simulation to verify correctness and timing simulation to check performance.
- **Formal Verification**: Use mathematical methods to prove the correctness of the design with respect to its specification.

#### d. Synthesis

- **Logic Synthesis**: Translate the HDL code into a gate-level representation. This step involves converting the high-level design into a netlist of logic gates.
- **Optimization**: Optimize the design for area, speed, and power consumption by adjusting the netlist and performing technology mapping.

#### e. Physical Design

# CHAPTER 10- ASIC TESTING E.PRIYADHARSHINI

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

ASIC (Application-Specific Integrated Circuit) testing is a critical phase in the ASIC design and manufacturing process. It ensures that the fabricated chip operates correctly, meets performance specifications, and functions reliably in its intended application. Testing helps to identify and address potential issues before the ASIC is deployed, reducing the risk of failures in the field.

Here's a comprehensive overview of ASIC testing, including the different types of tests, methodologies, and considerations:

# **1. Testing Stages**

#### a. Pre-Silicon Testing

#### • Simulation Testing:

- **Functional Simulation**: Tests the design against the functional specification using HDL code. This is done in the early stages of design and helps verify that the design behaves as intended.
- **Timing Simulation**: Assesses the design's timing performance based on the synthesized netlist and timing constraints.

#### • Formal Verification:

- **Equivalence Checking**: Verifies that the synthesized netlist is functionally equivalent to the original HDL code.
- **Model Checking**: Uses formal methods to prove properties about the design, such as safety and liveness properties.

#### b. Post-Silicon Testing

- Initial Silicon Validation:
  - **Debug and Validation**: Tests the first batch of manufactured ASICs to validate design functionality and identify any issues that arose during the manufacturing process.
  - **Characterization**: Measures key performance parameters like speed, power consumption, and temperature behavior.
- Production Testing:
  - Functional Testing: Ensures that the ASIC performs all its intended functions correctly.
  - **Parametric Testing**: Checks electrical parameters such as voltage levels, current consumption, and signal integrity.
  - **Burn-In Testing**: Subject the ASICs to prolonged operation under stress conditions to identify potential reliability issues.

#### • Post-Production Testing:

• **End-of-Line Testing**: Final testing performed on chips after packaging to verify they meet all specifications before shipment.

# ELECTRONIC DEVICES

Edited by

**M.SOUNDHARYA** 



ELECTRONIC DEVICES

COPYRIGHT @ 2021 BY M.SOUNDHARYA

ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 9789362559289

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

CHAPTER	CONTENT	PAGE
	INTRODUCTION	5
1	BASIC CONCEPT OF SEMICONDUCTOR DEVICES	13
	Dr. S.DEVI	
2	AMPLIFIERS	28
	Dr. N. PARVATHAM	
3	MOSFET SMALL SIGNAL MODEL	51
	P.GEETHA	
4	GAIN AND FREQUENCY RESPONSE	65
	R.SAVITHA	
5	MULTISTAGE AMPLIFIERS	86
	A. AARTHI	
6	DIFFERENTIAL AMPLIFIER	96
	S.LILLYPET	
7	FEEDBACK AMPLIFIERS	107
	A.AMUDHA	
8	OSCILLATORS	142
	R.SAVITHA	
9	POWER AMPLIFIERS	168
	S.SARASWATHY	
10	AC/DC CONVERTERS	185
	T. DIVYA MANOHARI	
	REFERENCES	205

# CHAPTER 1 BASIC CONCEPT OF SEMICONDUCTOR DEVICES Dr. S.DEVI

Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Semiconductor devices are essential components in modern electronics, enabling a wide range of technologies from computers to communication systems. To understand these devices, it's important to grasp the underlying semiconductor materials and their properties. Here's a comprehensive introduction to semiconductor devices:

#### 1. Semiconductor Materials

- Silicon (Si): The most widely used semiconductor material, particularly in integrated circuits and solar panels due to its abundance and effective electrical properties.
- Germanium (Ge): Employed in some high-speed and high-frequency applications but less common than silicon due to its higher cost and lower thermal stability.
- Gallium Arsenide (GaAs): Noted for its high-speed and optoelectronic applications, such as LEDs and laser diodes, due to its superior electron mobility.

Semiconductors are materials with electrical conductivities between that of conductors (e.g., metals) and insulators (e.g., glass). Their electrical properties can be modified through a process known as doping, where impurities are added to enhance their conductivity.

#### 2. Types of Semiconductors

- **Intrinsic Semiconductors:** Pure forms of semiconductor materials with a balanced number of electrons and holes. Their electrical behavior is dictated by the material's inherent properties.
- **Extrinsic Semiconductors:** Created by doping intrinsic semiconductors with specific impurities to increase their conductivity. There are two primary types:
  - **N-Type:** Doped with elements that provide extra valence electrons, resulting in an abundance of negative charge carriers (electrons).
  - **P-Type:** Doped with elements that create "holes" by having fewer valence electrons, resulting in an excess of positive charge carriers (holes).

#### 3. Key Semiconductor Devices

- Diodes:
  - **Rectifier Diodes:** Allow current to flow in one direction only, crucial for converting alternating current (AC) to direct current (DC) in power supplies.
  - **Zener Diodes:** Permit current flow in both directions but maintain a stable voltage when reverse-biased, making them useful for voltage regulation.
  - Light Emitting Diodes (LEDs): Emit light when forward-biased, commonly used in displays, indicators, and various lighting applications.
    - **Bipolar Junction Transistors (BJTs):** Comprise three layers of semiconductor material (NPN or PNP) and function as amplifiers .

# CHAPTER 2 AMPLIFIERS Dr. N. PARVATHAM

Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Electrostatics is a branch of electromagnetism focused on the study of electric charges at rest, as well

as the associated forces, fields, and potentials. Here's a detailed overview:

# **Amplifiers: An Overview**

Amplifiers are essential components in electronics that increase the amplitude of signals. They play a critical role in various applications, from audio systems to communication devices. Here's a detailed look at amplifiers, including their fundamental principles, types, and applications:

#### 1. Basic Principles of Amplifiers

An amplifier's primary function is to boost the amplitude of an input signal. Key parameters to understand include:

- Gain: The ratio of the output signal amplitude to the input signal amplitude, typically expressed in decibels (dB).
- **Bandwidth:** The range of frequencies over which the amplifier can operate effectively.
- Linearity: The ability of the amplifier to reproduce the input signal accurately without distortion.
- **Efficiency:** The ratio of the output power to the input power, indicating how effectively the amplifier converts power.

#### 2. Types of Amplifiers

#### 1. Class A Amplifiers:

- **Operation:** The output transistor is active for the entire input signal cycle, providing high linearity and low distortion.
- **Characteristics:** Known for high fidelity but with lower efficiency (20-30%) due to continuous current flow, which leads to substantial heat dissipation.
- **Applications:** Commonly used in high-fidelity audio systems and some low-frequency applications.

#### 2. Class B Amplifiers:

- **Operation:** The output transistors conduct for only half of the input signal cycle. Two transistors handle complementary halves of the waveform.
- **Characteristics:** More efficient than Class A (up to 70%) but can introduce crossover distortion where the transistors switch between positive and negative halves of the signal.
- Applications: Suitable for audio amplifiers and radio frequency (RF) applications.

#### 3. Class AB Amplifiers:

- **Operation:** A combination of Class A and Class B characteristics, where transistors conduct for more than half but less than the full signal cycle, reducing crossover distortion while improving efficiency.
- Characteristics: Offers a balance between efficiency and linearity, making it a popular choice for a

# **CHAPTER 3 MOSFET SMALL SIGNAL MODEL**

P.GEETHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

#### **MOSFET Small Signal Model**

The MOSFET (Metal-Oxide-Semiconductor Field-Effect Transistor) is a key component in analog and digital circuits. For accurate analysis and design of circuits involving MOSFETs, it is essential to understand their small signal model. This model is used to analyze the behavior of the MOSFET in response to small input signals around a bias point. Here's a detailed look at the MOSFET small signal model:

#### 1. Small Signal Model Overview

The small signal model represents the MOSFET's behavior when subjected to small variations around its operating point (also called the quiescent or bias point). This model simplifies the MOSFET's characteristics into linear components, making it easier to analyze and design circuits.

#### 2. Small Signal Parameters

For the small signal analysis, the MOSFET is characterized by the following parameters:

•  $g_m$  (Transconductance): Measures the change in the drain current  $(I_D)$  with respect to the change in gate-to-source voltage  $(V_{GS})$  at a constant drain-to-source voltage  $(V_{DS})$ . It is defined as:

$$g_m = rac{\partial I_D}{\partial V_{GS}}$$

In saturation,  $g_m$  can be approximated by:

$$g_m = rac{2I_D}{V_{OV}}$$

where  $V_{OV} = V_{GS} - V_{th}$  is the overdrive voltage and  $V_{th}$  is the threshold voltage.

 r<sub>ds</sub> (Drain-Source Resistance or Output Resistance): Represents the resistance looking into the drain with the source and gate terminals grounded. It is defined as:

$$\checkmark = \frac{1}{\lambda I_D}$$

# CHAPTER 4 GAIN AND FREQUENCY RESPONSE R.SAVITHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# Gain and Frequency Response in Amplifiers

In electronic amplifiers, understanding gain and frequency response is crucial for designing circuits that perform effectively across the desired frequency range. Here's a detailed look at these concepts:

#### 1. Gain

**Gain** is a measure of how much an amplifier increases the amplitude of an input signal. It can be expressed in several ways:

Voltage Gain (A<sub>V</sub>): The ratio of the output voltage (V<sub>out</sub>) to the input voltage (V<sub>in</sub>). It is often
expressed as:

$$A_V = rac{V_{out}}{V_{in}}$$

• In Decibels (dB): Gain in decibels is calculated as:

$$A_V(dB) = 20 \log_{10} \left(rac{V_{out}}{V_{in}}
ight)$$

- Current Gain (A<sub>I</sub>): The ratio of the output current to the input current, especially relevant in current amplifiers.
- Power Gain (A<sub>P</sub>): The ratio of the output power to the input power. In decibels, it is expressed as:

$$A_P(dB)$$
  $\downarrow$  10  $\log_{10}\left(\frac{P_{out}}{P_{in}}\right)$ 

#### 2. Frequency Response

The **frequency response** of an amplifier describes how the gain of the amplifier varies with input signal frequency. It is typically characterized by:**Low-Frequency Response:** Determines how the amplifier handles frequencies below the mid-band. The gain might drop due to coupling capacitors or other reactive components that impede low frequencies.

# CHAPTER 5 MULTISTAGE AMPLIFIERS A. AARTHI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# **Multistage Amplifiers**

Multistage amplifiers are crucial in modern electronics, where multiple amplification stages are used to achieve higher gain, better performance, and more control over signal characteristics. Here's a comprehensive overview of multistage amplifiers:

#### 1. Overview of Multistage Amplifiers

Multistage amplifiers consist of two or more amplification stages connected in series. Each stage amplifies the signal further, with the output of one stage serving as the input to the next. This approach allows for greater overall gain, improved bandwidth, and enhanced signal quality.

#### 2. Advantages of Multistage Amplifiers

- **Higher Gain:** By cascading multiple stages, the overall gain of the amplifier can be significantly increased, often beyond what a single-stage amplifier could provide.
- **Improved Impedance Matching:** Different stages can be designed to match the impedance of previous and subsequent stages, reducing signal reflection and loss.
- **Better Performance:** Combining stages with different characteristics can improve overall performance, such as gain flatness, bandwidth, and linearity.
- Noise Reduction: Multiple stages can help minimize the impact of noise introduced by individual stages, especially when designing with careful attention to each stage's noise performance.

#### 3. Basic Configuration of Multistage Amplifiers

#### **1. Cascade Configuration:**

- **Common-Emitter Cascade (for BJTs):** One of the most common configurations where each stage is a common-emitter amplifier. This setup provides high gain and is widely used in analog signal amplification.
- **Common-Source Cascade (for MOSFETs):** Similar to the common-emitter configuration but using common-source MOSFET amplifiers. It is used in many analog applications and offers high gain and impedance matching.

#### 2. Darlington Pair:

- **Configuration:** A pair of transistors (or MOSFETs) connected in a way that the current amplified by the first transistor is further amplified by the second. This provides a very high current gain and is often used for power amplification.
- Advantages: High current gain and improved input impedance.

# CHAPTER 6 DIFFERENTIAL AMPLIFIER S.LILLYPET

Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

# **Differential Amplifier**

A differential amplifier is a fundamental building block in analog electronics, designed to amplify the difference between two input signals while rejecting any signals common to both inputs. This capability makes it ideal for various applications, including signal processing, instrumentation, and operational amplifiers.

#### 1. Overview of Differential Amplifiers

**Definition:** A differential amplifier amplifies the voltage difference between two input terminals. It has two inputs: a non-inverting input  $(Vin+V_{in+}Vin+)$  and an inverting input

 $(Vin-V_{in-})$ . The output is proportional to the difference between these inputs

#### 2. Basic Operation

The output voltage (VoutV\_{out}Vout) of an ideal differential amplifier is given by:

$$V_{out} = A_d \cdot (V_{in+} - V_{in-})$$

where  $A_d$  is the differential gain of the amplifier.

#### **Key Characteristics:**

- Differential Gain ( $A_d$ ): The amplification factor for the difference between the two input signals.
- Common-Mode Rejection Ratio (CMRR): Measures the amplifier's ability to reject commonmode signals. CMRR is defined as:

$$\mathrm{CMRR} = rac{A_d}{A_{cm}}$$



# CHAPTER 7 FEEDBACK AMPLIFIERS A.AMUDHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Feedback amplifiers are essential in electronic circuits for improving performance, stability, and control. They use feedback to adjust the amplifier's behavior. The four common feedback topologies you mentioned—current series, voltage series, current shunt, and voltage shunt—each have distinct characteristics and applications. Let's break them down:

# 1. Current Series Feedback Amplifier

#### **Configuration:**

- **Input:** The feedback network is placed in series with the input signal.
- **Output:** The output voltage is applied across the load.

#### **Characteristics:**

- Feedback Resistor: Connected in series with the input signal.
- Gain Reduction: This configuration typically results in a reduction in the overall voltage gain of the amplifier.
- Impedance: Increases the input impedance of the amplifier while decreasing the output impedance.
- **Applications:** Often used in applications where a high input impedance is desired and where the gain needs to be controlled.

Example: A common emitter amplifier with a feedback resistor in the emitter leg.

# 2. Voltage Series Feedback Amplifier

#### **Configuration:**

- **Input:** The feedback network is placed in parallel with the output.
- **Output:** The output voltage is taken directly from the amplifier.

#### **Characteristics:**

- Feedback Resistor: Connected in parallel with the load, affecting the output voltage.
- Gain Control: The voltage gain is reduced, and the amplifier becomes less sensitive to variations in the load.
- Impedance: Decreases the output impedance and can stabilize the gain against variations in the load.
- **Applications:** Useful in applications where consistent output voltage is crucial and the amplifier must drive varying loads.

Example: A non-inverting operational amplifier configuration where a resistor network is used for

# CHAPTER 8 OSCILLATORS R.SAVITHA

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Oscillators are essential components in electronic circuits that generate periodic waveforms, such as sine, square, or triangular waves. They are fundamental to many applications, including clocks in digital systems, signal generators, and communication systems. There are various types of oscillators, and they can be classified based on their waveform output, feedback mechanism, and components used. Here's a brief overview of the main types of oscillators:

# 1. Sine Wave Oscillators

#### **Characteristics:**

- Produce a smooth, continuous sinusoidal output.
- Used in applications where a pure sine wave is required, such as in signal processing and communication systems.

#### **Types:**

- LC Oscillator: Uses an inductor (L) and a capacitor (C) to set the frequency. Common configurations include the Colpitts, Hartley, and Clapp oscillators. They are widely used for high-frequency applications.
- **Crystal Oscillator:** Utilizes a quartz crystal to provide high stability and precision. The crystal acts as a frequency-determining element, ensuring a very stable and accurate output. Often used in clocks and communication equipment.

# 2. Square Wave Oscillators

#### **Characteristics:**

- Generate a square waveform with sharp transitions between high and low states.
- Useful in digital circuits, timing applications, and pulse-width modulation.

#### Types:

- Astable Multivibrator: A type of oscillator that continuously switches between its high and low states. It uses two transistors or logic gates and is commonly used for generating clock pulses.
- **555 Timer Oscillator:** Utilizes the 555 timer IC in an astable mode to produce a square wave. It's versatile and can be adjusted for different frequencies.

# CHAPTER 9 POWER AMPLIFIERS AND S.SARASWATHY

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Power amplifiers are designed to increase the power of a signal to drive loads, such as speakers or antennae. They are critical in various applications, including audio amplification, radio frequency (RF) transmission, and signal processing. Power amplifiers can be classified based on their class of operation, which determines their efficiency, linearity, and application suitability. Here's a detailed look at power amplifiers and their classes:

# **1. Class A Power Amplifiers**

#### **Characteristics:**

- **Operation:** The output transistor conducts for the entire input signal cycle (360 degrees).
- Efficiency: Typically low, around 20-30%, due to continuous current flow through the transistor.
- Linearity: High linearity and low distortion, making them ideal for high-fidelity audio applications.
- **Heat Dissipation:** Generates significant heat due to constant current flow, requiring substantial heat sinks.

#### **Applications:**

• High-quality audio amplifiers and signal processing where linearity is crucial.

**Example:** A high-end audio amplifier used in home audio systems.

# 2. Class B Power Amplifiers

#### **Characteristics:**

- **Operation:** The output transistor conducts for half of the input signal cycle (180 degrees). Two transistors are used in a push-pull configuration to cover the full cycle.
- Efficiency: Higher than Class A, typically around 50-70%, due to reduced power dissipation in the transistors.
- Linearity: Can have crossover distortion where the signal transitions between the two transistors, affecting linearity.

#### **Applications:**

• Audio amplifiers where efficiency is important but some linearity loss is acceptable.

**Example:** Medium-power audio amplifiers and some RF amplifiers.

#### 3. Class AB Power Amplifiers.

# CHAPTER 10 AC/DC CONVERTERS T.DIVYAMANOHARI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

AC/DC converters, also known as rectifiers, are devices or circuits that convert alternating current (AC) into direct current (DC). This conversion is crucial in many electronic devices and systems where DC power is needed but the available power source is AC. There are several types of AC/DC converters, each suited to different applications and performance requirements. Here's an overview of the main types:

# 1. Rectifiers

Rectifiers are the basic circuits used for converting AC to DC. They can be classified into different types based on their configuration and functionality.

#### a. Half-Wave Rectifiers

#### **Characteristics:**

- **Operation:** Utilizes a single diode to convert AC to DC. It only allows one half of the AC waveform to pass through, effectively blocking the other half.
- **Output:** Produces a pulsating DC waveform with a frequency equal to the AC supply frequency.
- Application: Simple applications where efficiency and smoothness are less critical.

**Example:** Basic power supply circuits for small electronic devices.

#### b. Full-Wave Rectifiers

#### **Characteristics:**

- **Operation:** Uses either two diodes (center-tap transformer configuration) or four diodes (bridge rectifier configuration) to convert the entire AC waveform into DC.
- **Output:** Produces a pulsating DC waveform with a frequency of twice the AC supply frequency (due to both halves of the AC waveform being used).
- Efficiency: Higher than half-wave rectifiers, as it utilizes both halves of the AC signal.

#### **Types:**

- Center-Tap Full-Wave Rectifier: Uses a transformer with a center-tap and two diodes.
- **Bridge Rectifier:** Uses four diodes arranged in a bridge configuration and does not require a centertap transformer.

**Example:** More efficient power supplies for general electronic devices.
# ANTENNAS AND MICROWAVE ENGINEERING

# EDITED BY

A.AARTHI



## ANTENNAS AND MICROWAVE ENGINEERING

COPYRIGHT @ 2021 BY A.AARTHI

ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 9789362557209

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## TABLE OF CONTENTS

CHAPTER	R CONTENT		
	INTRODUCTION	5	
1	ANTENNAS AND MICROWAVE ENGINEERING CONCEPTS AND TERMINOLOGY	13	
	Dr. SMITHA ELSA PETER		
2	ELECTROMAGNETIC WAVE PROPAGATION	28	
	D.HARIKRISHNAN		
3	ANTENNA FUNDAMENTALS	45	
	S. MAHESHWARAN		
4	TYPES OF ANTENNAS	64	
	R.SAVITHA		
5	ANTENNA ARRAYS AND TECHNIQUES	90	
	P.GEETHA		
6	MICROWAVE TRANSMISSION LINES	107	
	S.LILLYPET		
7	MICROWAVE COMPONENTS AND CIRCUITS	132	
	A.AMUDHA		
8	MICROWAVE MEASUREMENT TECHNIQUES	168	
	A.AARTHI		
9	MICROWAVE ENGINEERING APPLICATIONS	185	
	S.SARASWATHY		
10	EMERGING TRENDS AND TECHNOLOGIES	205	
	T.DIVYAMANOHARI		
	REFERENCES	236	

## CHAPTER 1 ANTENNAS AND MICROWAVE ENGINEERING CONCEPTS AND TERMINOLOGY Dr. SMITHA ELSA PETER

Professor, Department of ECE

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

In "Antennas and Microwave Engineering," understanding fundamental concepts and terminology is essential. Here's an overview of key terms and principles:

#### **Fundamental Concepts**

- **Electromagnetic Waves**: These are oscillating electric and magnetic fields propagating through space. Key characteristics include frequency, wavelength, velocity, and amplitude.
- **Wave Propagation**: Involves the transmission of electromagnetic waves through different mediums. Key phenomena include free-space propagation, reflection, refraction, and absorption, which affect how waves travel through materials.
- Antenna Fundamentals: An antenna converts electrical signals into electromagnetic waves and vice versa. Essential parameters include:
  - **Radiation Pattern**: The distribution of radiated energy in space.
  - **Polarization**: The orientation of the electric field of the wave.
- **Microwave Engineering**: Focuses on electromagnetic waves in the 300 MHz to 300 GHz frequency range. Key components include:
  - **Microwave Components**: Devices like filters, amplifiers, and mixers designed for these frequencies.

#### **Key Terminology**

- Antenna Parameters:
  - Gain: Measures how well an antenna energy compared to an isotropic radiator.
  - **Directivity**: Indicates how concentrated the radiation pattern is in a specific direction.
  - **Beamwidth**: The angle where the antenna's radiation pattern drops to half its maximum value.
  - Impedance: The combined resistance and reactance affecting energy transfer.
- Radiation Patterns:
  - **Isotropic Radiator**: Idealized antenna radiating power equally in all directions.
  - **Omnidirectional**: Radiates uniformly in all directions within a plane.
  - **Directional**: Focuses energy in a specific direction.
- Waveguides and Transmission Lines:
  - **Waveguide**: Guides electromagnetic waves from one point to another.
  - **Microstrip Line**: A transmission line consisting of a conductive strip on a dielectric substrate.
  - **Stripline**: A conductor sandwiched between two dielectric layers.
- Microwave Components:.
  - Filter: Allows certain frequencies to pass while attenuating others.
  - **Amplifier**: Increases the amplitude of signals.

## CHAPTER 2 ELECTROMAGNETIC WAVE PROPAGATION D.HARIKRISHNAN

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Electrostatics is a branch of electromagnetism focused on the study of electric charges at rest, as well as the associated forces, fields, and potentials. Here's a detailed overview:

#### 1. Electric Charge

- **Definition and Properties**: Electric charge is a fundamental property of matter that causes it to experience a force in an electromagnetic field. There are two types of electric charge: positive and negative. Like charges repel each other, while opposite charges attract.
- Quantization of Charge: Electric charge is quantized, meaning it occurs in discrete units. The fundamental unit of charge is the elementary charge (e), approximately 1.602×10-191.602 \times 10^{-19}1.602×10-19 coulombs. Charges are typically multiples of this elementary charge.

#### 2. Coulomb's Law

• **Statement**: Coulomb's Law describes the force between two point charges. The magnitude of the electrostatic force (F) between charges q1q\_1q1 and q2q\_2q2, separated by a distance r, is given by:

 $F = k_e |q1q2| /r2$ 

#### 3. Electric Field

**Definition**: The electric field (E) is a vector field that represents the force experienced by a unit positive test charge placed in the field. It is defined as: E=qF

where F is the force on the test charge and q is the magnitude of the test charge.

## 4. Polarization

• **Definition**: Polarization refers to the process by which a neutral object develops a net charge when exposed to an external electric field. This phenomenon causes a separation of charges within the material, leading to an induced dipole moment.

## CHAPTER 3 ANTENNA FUNDAMENTALS S. MAHESHWARAN

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Antenna Fundamentals are crucial for understanding how antennas function and are used across various communication systems. Here's a concise overview of key concepts and terminology:

## **1. Basic Definition**

• Antenna: An antenna is a device that converts electrical signals into electromagnetic waves for transmission and vice versa for reception. It serves as a bridge between electrical circuits and free space.

## 2. Key Parameters

- Gain:
  - **Definition**: Measures how effectively an antenna directs energy in a specific direction compared to an isotropic radiator.
  - **Measurement**: Expressed in decibels (dB), indicating the antenna's efficiency in focusing energy.
- Bandwidth:
  - **Definition**: The range of frequencies over which the antenna operates effectively.
  - **Importance**: A wider bandwidth allows the antenna to support multiple frequencies and standards.
- Beamwidth:
  - **Definition**: The angle between the directions where the antenna's radiation pattern drops to half its maximum value.
  - **Significance**: Narrow beamwidth means a highly directional antenna; wider beamwidth provides broader coverage.

## 3. Radiation Pattern

- Isotropic Radiator:
  - **Definition**: An ideal antenna that radiates power uniformly in all directions, used as a performance reference.
- Omnidirectional Antenna:
  - **Definition**: Radiates equally in all directions in a specific plane, typically producing a doughnut-shaped pattern.
  - **Application**: Suitable for applications requiring 360-degree coverage, like mobile base stations.
- Directional Antenna:
  - **Definition**: Focuses energy in a particular direction, producing a radiation pattern.
  - Application: Ideal for high-gain, long-distance communication, such as satellite dishes.

## CHAPTER 4 TYPES OF ANTENNAS R.SAVITHA

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Antennas come in diverse designs, each tailored for specific applications and performance needs. Here is a concise overview of common antenna types:

#### 1. Dipole Antenna

- **Description**: Comprises two conductive elements, each usually half a wavelength long.
- Characteristics: Simple and efficient, often used as a benchmark for antenna gain.
- **Applications**: Basic communication systems like FM radios and televisions.

#### 2. Monopole Antenna

- **Description**: A variation of the dipole with one element replaced by a ground plane or reflector.
- **Characteristics**: Typically a quarter wavelength long, utilizing the ground plane to reflect signals.
- Applications: Mobile phones, vehicle antennas, and broadcast transmitters.

#### 3. Yagi-Uda Antenna

- **Description**: A directional antenna featuring a driven element, reflectors, and directors.
- **Characteristics**: Focuses energy in a specific direction for higher gain and directivity.
- Applications: TV reception, amateur radio, and certain wireless communications.

#### 4. Parabolic Reflector Antenna

- **Description**: Uses a parabolic dish to direct electromagnetic waves to a focal point where the feed antenna is located.
- **Characteristics**: High directionality and gain, focusing energy into a narrow beam.
- **Applications**: Satellite communications, radio telescopes, and microwave links.

#### 5. Patch Antenna

- **Description**: Features a flat, rectangular metal patch on a dielectric substrate with a ground plane.
- **Characteristics**: Compact and planar with moderate gain and directivity, operating over a narrow bandwidth.
- Applications: GPS receivers, mobile phones, and wireless communication devices.

#### 6. Helical Antenna

- **Description**: Consists of a helix-shaped conductor wound around a central axis, operable in axial or normal mode.
- Characteristics: Offers circular polarization with high gain and wide bandwidth.
- Applications: Satellite communications, spacecraft communications, and RFID systems.

## CHAPTER 5 ANTENNA ARRAYS AND TECHNIQUES P.GEETHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

## 1. Antenna Arrays

**Definition**: An antenna array is a configuration of multiple antennas, or elements, arranged to produce a combined radiation pattern with enhanced characteristics.

- Linear Array: Antennas are arranged in a straight line, ideal for applications needing directional coverage.
- **Planar Array**: Antennas are placed in a two-dimensional grid, suitable for multi-directional coverage, such as in radar and satellite communications.
- **Circular Array**: Antennas are arranged in a circular pattern, providing uniform coverage around a central point.

## 2. Array Configuration Techniques

#### **Amplitude Control**:

- **Definition**: Techniques for adjusting the signal amplitude to each antenna element.
- **Purpose**: Optimizes the radiation pattern by balancing signal strength across elements.
- Applications: Essential in adaptive beamforming for improving quality and minimizing interference.

#### **Beamforming**:

- **Definition**: Adjusts the phase and amplitude of signals from multiple antennas to direct the beam.
- Types:
  - Fixed Beamforming: Uses static settings to direct the beam.
  - Adaptive Beamforming: Dynamically adjusts the beam pattern based on real-time signal conditions.

## **3. Phased Arrays**

**Definition**: Arrays with electronically controlled phase settings to steer the beam direction without mechanical movement.

#### Key Features:

- Electronic Steering: Provides rapid and precise beam direction control.
- No Mechanical Parts: Reduces maintenance and increases reliability compared to mechanical systems.
- **Applications**: Utilized in radar, satellite communications, and phased array radars for tracking and targeting.

## CHAPTER 6 MICROWAVE TRANSMISSION LINES S.LILLYPET

Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

## 1. Fundamentals

**Definition**: Microwave transmission lines are structures that transport microwave signals between circuit components. They ensure minimal signal loss and distortion.

#### **Key Parameters**:

- Characteristic Impedance ( $Z_0$ ): The impedance presented by the transmission line, determined by its physical dimensions and material properties.
- **Propagation Constant** ( $\gamma$ ): Represents how the signal's amplitude and phase change along the line, with its real part (attenuation constant,  $\alpha$ ) and imaginary part (phase constant,  $\beta$ ).

## 2. Types of Microwave Transmission Lines

#### Waveguides:

- Definition: Hollow metal or dielectric structures that guide electromagnetic waves.
- Characteristics: Low signal loss and support for various propagation modes.

#### Microstrip Lines:

- **Definition**: Planar transmission lines with a conductive strip separated from a ground plane.
- Characteristics: Compact, integrates easily into PCBs, suitable for high-frequency applications.

#### Stripline:

- **Definition**: A conductor sandwiched between two dielectric layers.
- Characteristics: Better shielding and reduced radiation loss compared to microstrip lines, consistent impedance.

#### **Coaxial Lines**:

- **Definition**: A central conductor surrounded by a concentric shield, separated by dielectric material.
- **Characteristics**: Excellent shielding and consistent impedance, minimizing signal loss and interference.

#### **Parallel Plate Transmission Lines:**

- **Definition**: Consists of two parallel conductive plates separated by a dielectric material.
- Characteristics: Typically used in laboratory settings for precise impedance control.

## CHAPTER 7 MICROWAVE COMPONENTS AND CIRCUITS A.AMUDHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

## 1. Microwave Components

#### 1.1. Filters

- **Definition**: Devices that control which frequencies pass through while blocking others.
  - **Bandpass Filter**: Passes frequencies within a set range and blocks outside frequencies.
  - Low-pass Filter: Allows frequencies below a cutoff to pass and attenuates higher frequencies.
  - High-pass Filter: Passes frequencies above a cutoff and blocks lower frequencies.

#### **1.2.** Amplifiers

- **Definition**: Increase the amplitude of microwave signals.
  - Low Noise Amplifier (LNA): Boosts weak signals with minimal added noise, vital for receiving systems.
  - **Power Amplifier**: Enhances signal strength for transmission, used in radar and communication systems.

#### 1.3. Mixers

- **Definition**: Combine two signals to create new frequencies (sum and difference).
  - **Passive Mixer**: Mixes signals without amplification.
  - Active Mixer: Includes amplification, improving performance and conversion gain.

## 2. Microwave Circuits

#### 2.1. Waveguide Circuits

- Definition: Guide microwave signals with low loss using hollow metal or dielectric tubes.
- Components: Bends, tees, and junctions.

#### **2.2. Microstrip Circuits**

- Definition: Planar circuits on a dielectric substrate, including microstrip lines and patches.
- Components: Transmission lines, antennas, and resonators.

#### **2.3. Stripline Circuits**

- **Definition**: Planar circuits with conductors between dielectric layers.
- **Components**: Stripline transmission lines, filters, and couplers.

## CHAPTER 8 MICROWAVE MEASUREMENT TECHNIQUES A.AARTHI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

## 1. Network Analysis

#### Vector Network Analyzer (VNA)

- Definition: Measures the reflection and transmission properties of microwave components.
- Functions:
  - S-Parameters: Determines how microwave signals are reflected and transmitted.
  - Frequency Response: Analyzes how components behave across different frequencies.
- Applications: Characterizes filters, amplifiers, antennas, and other microwave devices.

## 2. Spectrum Analysis

#### Spectrum Analyzer

- **Definition**: Measures signal amplitude across frequencies to analyze spectral content.
- Functions:
  - **Frequency Domain Analysis**: Displays signal power across frequencies, identifying strength, harmonics, and interference.
  - **Bandwidth Measurement**: Evaluates the bandwidth and stability of microwave signals.
- Applications: Analyzes signal purity, detects spurious emissions, and evaluates signal behavior.

## 3. Power Measurement

#### **Power Meter**

- **Definition**: Measures the absolute power of microwave signals.
- Types:
  - o Thermal Power Meter: Uses a heat-sensitive sensor for power measurement.
  - **Diode Power Meter**: Converts microwave power into a measurable DC voltage.
- Applications: Measures transmitted power, amplifier output, and signal strength.

#### 4. Time-Domain Analysis

#### Time-Domain Reflectometer (TDR)

- **Definition**: Measures time delay of reflected signals to analyze transmission lines and connectors.
  - Impedance Matching: Identifies impedance mismatches and discontinuities.
  - Fault Detection: Locates faults such as open or short circuits.

## CHAPTER 9 MICROWAVE ENGINEERING APPLICATIONS S.SARASWATHY

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Microwave engineering involves designing and using systems that operate at microwave frequencies, typically between 300 MHz and 300 GHz. These frequencies are integral to various advanced technologies. Here's an overview of key applications:

#### 1. Communication Systems

- Mobile and Wireless Communication: Microwaves support cellular networks (4G, 5G), Wi-Fi, and Bluetooth, enabling high-speed data transfer over both short and long distances.
- **Satellite Communication**: Essential for transmitting signals between Earth and satellites, microwaves facilitate television broadcasting, internet services, and GPS.

#### 2. Radar Systems

- Weather Radar: Provides critical data for weather forecasting by measuring precipitation and storm intensity.
- Air Traffic Control Radar: Monitors aircraft to ensure safe navigation and prevent collisions.
- **Military Radar**: Used for target detection and tracking, enhancing national security and military operations.

#### 3. Microwave Heating

- **Industrial Heating**: Utilized for drying, curing, and heating materials in manufacturing processes such as ceramics, glass, and plastics.
- **Medical Applications**: Microwave ablation destroys or removes tissue, including tumors, with high precision.

#### 4. Imaging Systems

- **Medical Imaging**: Microwave techniques are being explored for diagnosing and imaging soft tissues.
- **Security Screening**: Employed in non-invasive systems for detecting concealed objects or threats, such as in airport security.

#### 5. Scientific Research

- **Radio Astronomy**: Microwaves help observe cosmic phenomena like cosmic microwave background radiation, aiding in the study of the universe's origins.
- **Material Characterization**: Techniques analyze materials' dielectric properties and electromagnetic behavior.

## 6. Broadcasting

- **Television Broadcasting:** Microwaves transmit television signals from broadcast towers to antennas, ensuring high-quality reception.
- Radio Broadcasting: Facilitate AM and FM radio transmission over long distances.

## CHAPTER 10 EMERGING TRENDS AND TECHNOLOGIES T.DIVYAMANOHARI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Microwave engineering is rapidly advancing, driven by new technologies and evolving demands across various sectors. Here's a snapshot of the key trends and innovations shaping the field:

#### 1. 5G and Beyond

- Advanced Millimeter-Wave Technology: 5G networks leverage millimeter-wave frequencies (24 GHz and above) to deliver higher data rates and increased network capacity.
- **Massive MIMO**: Massive Multiple Input Multiple Output (MIMO) systems with extensive antenna arrays enhance signal quality and capacity in 5G networks.

#### 2. 6G Development

- **Terahertz Communications**: Research is focused on utilizing terahertz frequencies (0.1-10 THz) for ultra-fast data transfer and high-capacity networks.
- **Enhanced AI Integration**: Artificial Intelligence (AI) is being used for optimizing network management and performance in 6G, improving efficiency and reliability.

## 3. Advanced Radar Technologies

- **Phased Array Radars**: Electronically steerable arrays in phased array radars enhance target detection, tracking, and resolution.
- **Millimeter-Wave Radar**: Provides improved resolution and accuracy for automotive safety, weather monitoring, and security applications.

#### 4. Satellite Communication Innovations

- **Low Earth Orbit (LEO) Satellites**: LEO satellite constellations are being deployed to offer global high-speed internet with reduced latency.
- **High-Throughput Satellites (HTS)**: HTS provide significantly higher data transmission capacity compared to traditional satellites.

## 5. Quantum Computing and Communication

- **Quantum Key Distribution (QKD)**: Employs microwave frequencies to enable secure communication through unbreakable quantum encryption.
- **Microwave Quantum Bits**: Research is advancing in using microwave circuits to enhance qubit performance and scalability for quantum computing.

#### 6. Microwave Photonics

- **Integrated Microwave Photonics:** Combines microwave and photonics technologies for high-speed data transmission and signal processing.
- **Optical Frequency Comb:** Utilizes optical frequency combs for precise microwave signal generation and measurement.

## 7. Wearable and IoT Technologies

- **Miniaturized Microwave Sensors:** Innovations in miniaturization facilitate the development of sensors for health monitoring, environmental sensing, and smart devices.
- **Enhanced Wireless IoT Devices:** Microwave technology is being integrated into IoT devices for improved connectivity and performance.



## PRINCIPLES OF MANAGEMENT

COPYRIGHT @ 2021 BY A.AARTHI

ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 9789362556363

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## TABLE OF CONTENTS

CHAPTER	CONTENT	PAGE	
	INTRODUCTION	5	
1	DEFINITION AND IMPORTANCE OF MANAGEMENT	13	
	Dr. SMITHA ELSA PETER		
2	PLANNING	30	
	P.GEETHA		
3	ORGANIZING	49	
	S.LILLYPET		
4	LEADING	64	
	S. MAHESWARAN		
5	CONTROLLING	85	
	T DIVYA MANOHARI		
6	DECISION-MAKING	96	
	Dr. S. DEVI		
7	COORDINATION	107	
	E. PRIYADHARSHINI		
8	DELEGATION	132	
	S.SARASWATHY		
9	INNOVATION	168	
	R.SAVITHA		
10	EMERGING TRENDS IN MANAGEMENT	190	
	A.AMUDHA		
	REFERENCES	210	

## CHAPTER 1 DEFINITION AND IMPORTANCE OF MANAGEMENT Dr. SMITHA ELSA PETER

Professor, Department of ECE

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Management is the process of guiding an organization's resources—human, financial, and physical—to achieve specific goals efficiently and effectively. It encompasses four core functions: planning, organizing, leading, and controlling.

#### Key Aspects of Management:

- **Planning**: Establishes objectives and devises strategies to achieve and providing a roadmap for action.
- **Organizing**: Arranges resources and tasks, defining roles and responsibilities to ensure smooth execution.
- Leading: Inspires and motivates team members, fostering a positive work environment
- **Controlling**: Monitors performance, compares it with standards, and makes adjustments to stay on course.

#### **Importance of Management**

- 1. **Goal Achievement**: Management provides direction by setting clear objectives and creating strategies to reach them, ensuring that efforts and resources are focused on achieving organizational goals.
- 2. **Resource Utilization**: Effective management optimizes the use of resources—human, financial, and physical—preventing waste and enhancing productivity, which leads to cost savings and improved performance.
- 3. **Organizational Structure**: Management establishes a clear structure, defining roles and responsibilities, which helps coordinate activities, facilitate communication, and ensure systematic task completion.
- 4. **Decision-Making**: Managers make key decisions that impact the organization's future. Effective decision-making involves analyzing data, evaluating alternatives, and selecting the best course of action to boost performance and competitiveness.
- 5. **Motivation and Leadership**: Management involves leading and motivating employees, creating a positive work environment that enhances job satisfaction and drives productivity. Motivated employees are more engaged and aligned with organizational goals.
- 6. Adaptability: In a dynamic business environment, management adapts strategies and processes to respond to changes and new challenges, ensuring the organization remains resilient and competitive.
- 7. **Coordination and Integration**: Management ensures different departments and functions work together seamlessly, preventing duplication of efforts and resolving conflicts to align all parts of the organization towards common objectives.

## CHAPTER 2 PLANNING P.GEETHA

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Planning in Project Object Management (POM) involves a structured approach to ensure project success. Here's a concise guide to effective planning:

#### 1. **Define Project Objectives**

- **Goals and Scope:** Clearly define what the project aims to achieve and its deliverables.
- **Stakeholders:** Identify those impacted by the project and their needs.

#### 2. Create a Project Plan

- Work Breakdown Structure (WBS): Decompose the project into manageable tasks or work packages.
- **Timeline:** Develop a timeline with milestones and deadlines.
- **Resources:** Determine required resources such as personnel, equipment, and materials.
- **Budget:** Estimate costs for each task and create a comprehensive budget.

#### 3. Develop a Task Schedule

- **Dependencies:** Identify and sequence tasks based on their dependencies.
- Gantt Chart: Utilize Gantt charts or similar tools to visualize the schedule.
- **Critical Path:** Determine the critical path to focus on tasks that impact the project timeline.

#### 4. Risk Management

- Identify Risks: List potential risks and assess their impact.
- Mitigation Strategies: Develop strategies to address identified risks.
- **Contingency Plans:** Prepare contingency plans for high-impact risks.

#### 5. Assign Roles and Responsibilities

- **Team Structure:** Define roles and responsibilities within the team.
- Task Allocation: Assign tasks based on team members' skills and expertise.

#### 6. Communication Plan

- **Information Flow:** Establish communication channels within the team and with stakeholders.
- **Reporting:** Set up regular progress reports and updates.

#### 7. Monitoring and Control

- **Progress Tracking:** Continuously monitor project progress against the plan.
- Adjustments: Make necessary adjustments based on performance data and changes.

#### 8. Review and Closure

- **Final Deliverables:** Ensure all deliverables meet objectives and quality standards.
- Lessons Learned: Document insights and best practices for future projects.
- **Closure:** Officially close the project, complete administrative tasks, and release resources.

## CHAPTER 3 ORGANIZING S.LILLYPET

Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Organizing in Project Object Management (POM) involves structuring project components to efficiently manage resources, tasks, and processes for achieving project goals. Here's a streamlined guide:

## **1. Define Project Structure**

- **Project Charter:** Outline the project's objectives, scope, and stakeholders.
- Scope Definition: Clearly specify what the project will include and exclude to prevent scope creep.

## 2. Develop Work Breakdown Structure (WBS)

- Task Decomposition: Break the project into smaller, manageable tasks or work packages.
- Hierarchy: Organize tasks hierarchically to illustrate their relationships and dependencies.

#### **3. Allocate Resources**

- **Resource Planning:** Identify the required resources—personnel, equipment, and materials.
- Resource Assignment: Allocate resources to tasks based on their availability and expertise.

## 4. Create a Task Schedule

- Task Sequencing: Arrange tasks in order, considering their dependencies.
- **Timeline:** Develop a detailed timeline or Gantt chart to track progress and deadlines.

## 5. Assign Roles and Responsibilities

- Team Structure: Define roles and responsibilities for team members.
- **Responsibility Assignment:** Use tools like RACI charts to clarify who is responsible, accountable, consulted, and informed for each task.

## 6. Implement Communication Plan

- Communication Channels: Set up effective channels for team and stakeholder communication.
- **Reporting:** Schedule regular updates and meetings to keep everyone informed.

## 7. Establish Monitoring and Control

- **Progress Tracking:** Use tools to monitor progress and performance.
- Issue Management: Create processes for identifying, documenting, and resolving issues.

## CHAPTER 4 LEADING S.MAHESWARAN

#### Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

**Leading in Project Object Management (POM)** involves guiding the team, influencing stakeholders, and steering the project towards its goals. Effective leadership is essential for efficient project execution and achieving objectives. Here's a streamlined guide to leading in POM:

## 1. Establish a Clear Vision

- **Define Objectives:** Clearly communicate project goals and desired outcomes to the team.
- Align Efforts: Ensure team members understand how their tasks contribute to overall objectives.

#### 2. Build and Motivate the Team

- **Team Formation:** Assemble a team with the required skills and experience.
- Motivation: Create a positive work environment through recognition, encouragement, and support.
- **Conflict Resolution:** Address conflicts promptly to maintain team cohesion.

## **3.** Provide Direction and Guidance

- Set Expectations: Clearly define roles, responsibilities, and performance standards.
- Offer Support: Provide necessary resources, training, and guidance.
- Decision-Making: Make informed decisions swiftly to keep the project on track.

## 4. Communicate Effectively

- **Open Communication:** Maintain clear communication with the team and stakeholders.
- **Regular Updates:** Offer consistent updates on progress, changes, and issues.
- Feedback: Encourage and act on feedback from team members and stakeholders.

#### **5.** Foster Collaboration and Teamwork

- Encourage Collaboration: Promote teamwork and cooperation among team members.
- Facilitate Meetings: Lead productive meetings focused on problem-solving and decision-making.

## 6. Manage Stakeholder Relationships

- Engage Stakeholders: Actively engage with stakeholders to understand and address their needs and expectations.
- Address Concerns: Manage stakeholder expectations and resolve any issues promptly.

## CHAPTER 5 CONTROLLING T. DIVYA MANOHARI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

**Controlling in Project Object Management (POM)** is crucial for ensuring that a project remains on track and achieves its goals. Here's a streamlined guide to effective project control:

## **1. Monitor Project Performance**

- **Track Progress:** Regularly measure progress against the project plan, including key milestones and deadlines.
- **Performance Metrics:** Use key performance indicators (KPIs) like budget adherence, schedule variance, and quality metrics to assess project health.

## 2. Manage Scope and Changes

- **Scope Control:** Carefully evaluate and approve any changes to the project scope through a formal change control process.
- **Change Management:** Document and manage changes to prevent scope creep and ensure alignment with project objectives.

## **3.** Control Costs

- **Budget Monitoring:** Track expenditures against the budget to ensure financial resources are used as planned.
- Cost Variance: Analyze deviations from the budget and implement corrective actions as needed.

## 4. Ensure Quality

- Quality Standards: Ensure project deliverables meet predefined quality standards and requirements.
- **Quality Control:** Implement measures like inspections and testing to identify and address issues early.

#### 5. Assess Risks

- **Risk Monitoring:** Continuously track identified risks and their mitigation strategies.
- **Risk Responses:** Adjust risk management approaches based on current conditions and project performance.

#### 6. Handle Issues and Deviations

- Issue Tracking: Identify, document, and track issues as they arise.
- Resolution: Implement corrective actions to resolve deviations and keep the project on course.

## CHAPTER 6 DECISION-MAKING Dr.S.DEVI

Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

**Decision-Making in Principles of Management (POM)** is essential for guiding projects towards success by making informed and effective choices. Here's a concise guide to decision-making in POM:

## 1. Define Decision Criteria

- **Objectives:** Clearly state the goals and desired outcomes the decision aims to achieve.
- Constraints: Identify constraints such as budget, time, and resources that will influence the decision.

## 2. Gather Information

- Data Collection: Obtain relevant data and insights needed for making an informed choice.
- Stakeholder Input: Collect perspectives and feedback from stakeholders and team members.

## **3.** Generate Alternatives

- **Options Development:** Create several possible solutions or alternatives to address the issue.
- **Creative Thinking:** Use brainstorming and innovative thinking to explore a wide range of solutions.

## 4. Evaluate Alternatives

- Criteria Assessment: Assess each alternative against defined criteria, considering benefits, risks, and feasibility.
- **Risk Analysis:** Analyze the potential risks and impacts associated with each option.

## 5. Make the Decision

- **Decision-Making Models:** Apply models such as cost-benefit analysis, decision matrices, or SWOT analysis to select the best option.
- **Consensus Building:** Where possible, seek consensus among team members and stakeholders for the chosen alternative.

## 6. Implement the Decision

- Action Plan: Create and execute a plan to implement the selected solution.
- **Resource Allocation:** Assign necessary resources and responsibilities for effective implementation.

## CHAPTER 7 COORDINATION E. PRIYADHARSHINI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

## 1. Establish Clear Objectives

- **Define Goals:** Clearly communicate the project's objectives to all team members and stakeholders.
- Align Efforts: Ensure everyone understands their role and how their tasks contribute to the project's overall goals.

## 2. Develop an Integrated Plan

- **Task Dependencies:** Identify and map out task dependencies to ensure smooth workflow and prevent bottlenecks.
- **Resource Allocation:** Allocate resources efficiently, optimizing the use of team members and materials.

## **3.** Facilitate Communication

- **Communication Channels:** Set up effective channels for regular updates and information sharing, such as meetings, emails, and collaboration tools.
- Information Flow: Ensure timely and accurate information flow between all project participants.

## 4. Monitor Progress

- **Track Tasks:** Regularly monitor task progress and milestones to ensure alignment with the project schedule.
- Address Issues: Quickly resolve any issues or delays that might affect project coordination.

## **5. Manage Conflicts**

- Identify Conflicts: Proactively identify potential conflicts or overlaps in responsibilities.
- **Resolve Conflicts:** Facilitate resolution through mediation and by clarifying roles and responsibilities.

## 6. Ensure Collaboration

- **Encourage Teamwork:** Promote collaboration and knowledge sharing to boost productivity and innovation.
- **Cross-Functional Coordination:** Coordinate efforts across different functional areas for cohesive progress.

## CHAPTER 8 DELEGATION S.SARASWATHY

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

**Delegation** is a vital management skill that involves assigning tasks and authority to others to achieve organizational goals efficiently. It enhances productivity, develops team members' skills, and fosters a positive work environment. Here's a streamlined guide to effective delegation:

## 1. Clearly Define Tasks

- **Specificity:** Articulate the task's objectives, scope, and expected outcomes clearly.
- Instructions: Provide detailed instructions and expectations to ensure clarity and understanding.

## 2. Select the Right Person

- Skills and Expertise: Choose individuals whose skills and experience match the task requirements.
- Interest and Capability: Ensure the person is both interested in and capable of handling the task effectively.

## 3. Provide Adequate Authority

- **Empowerment:** Grant the necessary authority for decision-making and resource access.
- **Boundaries:** Set clear limits to prevent overreach while maintaining control.

## 4. Set Clear Expectations

- **Objectives:** Communicate clear goals and desired outcomes for the task.
- **Deadlines:** Establish deadlines and timelines to ensure timely completion.

## **5. Offer Support and Resources**

- Training: Provide necessary training and resources to support success.
- Assistance: Be available to offer guidance and support throughout the task.

## 6. Monitor Progress

- Check-Ins: Regularly review progress to ensure the task is on track and address any issues promptly.
- Feedback: Give constructive feedback to guide improvement and alignment with objectives.

## 7. Avoid Micromanaging

- **Trust:** Allow the individual to work independently, providing guidance & avoiding excessive control.
- Autonomy: Trust their abilities and resist the urge to oversee every detail.

## CHAPTER 9 INNOVATION R.SAVITHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

**Innovation** is a fundamental management principle involving the introduction of new ideas, products, or processes to enhance efficiency, solve problems, and secure a competitive edge. Embracing innovation helps organizations remain relevant and succeed in a rapidly evolving market. Here's a streamlined guide to fostering innovation in management:

## 1. Encourage a Culture of Creativity

- **Open Environment:** Foster a work environment where creativity is valued and encouraged.
- Diverse Perspectives: Promote diversity and inclusion to harness a range of viewpoints and ideas.

#### 2. Foster Experimentation

- Risk-Taking: Encourage team members to take calculated risks and explore new ideas.
- Learning from Failure: View failures as opportunities to learn and refine processes.

## 3. Provide Resources and Support

- **Funding and Tools:** Allocate the necessary resources, including time, budget, and tools, for innovative projects.
- Training: Offer training and development to enhance skills relevant to innovation.

## 4. Encourage Collaboration

- **Teamwork:** Support cross-functional collaboration to leverage diverse expertise and perspectives.
- **Networking:** Build relationships with external partners, customers, and industry experts for fresh insights.

## 5. Set Clear Goals and Metrics

- Objectives: Define clear innovation goals and objectives to direct efforts and measure success.
- Metrics: Use key performance indicators (KPIs) to assess the impact of innovative initiatives.

## 6. Support Leadership and Vision

- Leadership Commitment: Ensure leaders are committed to innovation and articulate a clear vision.
- Empowerment: Grant teams the autonomy to pursue and implement innovative projects.

## CHAPTER 10 EMERGING TRENDS IN MANAGEMENT A.AMUDHA

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

**Emerging Trends in Management** are reshaping how organizations operate, lead, and achieve success. These trends introduce new approaches to leadership, strategy, and organizational culture. Here's a concise overview of key emerging trends in management:

## **1. Digital Transformation**

- **Technology Integration:** Organizations are increasingly adopting digital technologies like AI, machine learning, and big data to enhance decision-making and operational efficiency.
- Automation: Leveraging automation tools to streamline processes, cut costs, and boost accuracy.

## 2. Remote and Hybrid Work

- Flexible Work Models: The shift to remote and hybrid work arrangements promotes flexibility and better work-life balance.
- Virtual Collaboration: Utilizing digital tools such as Zoom, Microsoft Teams, and Slack for effective communication and productivity.

#### 3. Agile Management

- Adaptability: Implementing agile methodologies to respond quickly to changing market conditions.
- **Cross-Functional Teams:** Creating cross-functional teams to foster collaboration and drive innovation.

## 4. Data-Driven Decision Making

- Analytics: Utilizing data analytics to guide strategic decisions and uncover business insights.
- Performance Metrics: Employing KPIs and data-driven metrics to assess and enhance performance.

## 5. Employee Well-Being and Engagement

- **Holistic Approach:** Emphasizing comprehensive employee well-being, including mental health and work-life balance.
- Engagement Strategies: Developing strategies to boost employee engagement and retention.

## 6. Diversity, Equity, and Inclusion (DEI)

- **Inclusive Practices:** Committing to diverse and inclusive work environments that reflect various perspectives.
- Equity Initiatives: Implementing policies that ensure equal opportunities and promote equity within the organization.

# NETWORKS AND SECURITY LAB MANUAL

Edited by :

**DR.SMITHA ELSA PETER** 



978-93-6255-324-9

## NETWORKS AND SECURITY LAB MANUAL

## COPYRIGHT @ 2021 BY Dr.SMITHA ELSA PETER

## ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 978-93-6255-324-9

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# TABLE OF CONTENTS

S.NO.	DATE	EXPERIMENT TITLE	PAGE NO	SIGN.
1		Implement the Data Link Layer framing methods, i) Bit stuffing, (ii) Character stuffing		
		Implementation of Error Detection / Correction		
2		Techniques i) LRC, (ii) CRC, (iii) Hamming code		
		Implementation of Stop and Wait, and Sliding		
3		Window Protocols		
4		Implementation of Go back-N and Selective Repeat Protocols		
		Implementation of Distance Vector Routing		
5		algorithm (Routing Information Protocol) (Bellman-		
		Ford).		
6		Implementation of Link State Routing algorithm		
0		(Open Shortest Path First) with 5 hodes (Dijkstra's).		
7		Encryption Standard algorithm.		
		Data encryption and decryption using RSA (Rivest,		
8		Shamir and Adleman) algorithm.		
9		Implement Client Server model using FTP protocol.		

#### 1.Implement the Data Link Layer framing methods,

#### i) Bit stuffing

#### Aim::

To write a c program using Implement the Data Link Layer framing methods in Bit Stuffing.

#### Algorithm::

STEP-1: Read the Frame Size from the user.STEP-2: Read the key value from the user.STEP-3: If the size convert add the loop values.STEP-4: Values convert the framesSTEP-5: Display the Bit Stuffing obtained above

#### Program::

```
#include<stdio.h>
```

#include<string.h>

int main()

#### {

```
int a[20],b[30],i,j,k,count,n;
 printf("Enter frame size (Example: 8):");
 scanf("%d",&n);
 printf("Enter the frame in the form of 0 and 1 :");
 for(i=0; i<n; i++)
    scanf("%d",&a[i]);
i=0;
 count=1;
 j=0;
 while(i<n)
 {
   if(a[i]==1)
   {
b[j]=a[i];
      for(k=i+1; a[k]==1 && k<n && count<5; k++)
      {
        j++;
```

```
b[j]=a[k];
       count++;
       if(count==5)
       {
        j++;
         b[j]=0;
  }
i=k;
}
}
else
{
b[j]=a[i];
}
i++;
j++;
}
 printf("After Bit Stuffing :");
 for(i=0; i<j; i++)
   printf("%d",b[i]);
 return 0;
}
```

## **Output:**

Enter frame size (Example: 8):12 Enter the frame in the form of 0 and 1 :0 1 0 1 1 1 1 1 1 0 0 1 After Bit Stuffing :0101111101001

#### **Result::**

Thus, the above program was Successfully completed and verified.

#### (ii) Character stuffing

#### Aim::

To write a c program using Implement the Data Link Layer framing methods in Character Stuffing.

#### Algorithm::

STEP-1: Read the Frame Size from the user.

STEP-2: Read the Character from the user.

STEP-3: convert the Starting and ending delimiter

STEP-4: Values convert the frames

STEP-5: Display the Character Stuffing obtained above

#### Program:

```
#include<stdio.h>
#include<string.h>
main()
{
     char a[30], fs[50] = " ", t[3], sd, ed, x[3], s[3], d[3], y[3];
     int i, j, p = 0, q = 0;
     clrscr();
     printf("Enter characters to be stuffed:");
     scanf("%s", a);
     printf("\nEnter a character that represents starting delimiter:");
     scanf(" %c", &sd);
     printf("\nEnter a character that represents ending delimiter:");
     scanf(" %c", &ed);
     x[0] = s[0] = s[1] = sd;
     x[1] = s[2] = '\0';
     y[0] = d[0] = d[1] = ed;
     d[2] = y[1] = ' 0';
     strcat(fs, x);
     for(i = 0; i < strlen(a); i++)
     {
        t[0] = a[i];
        t[1] = ' 0';
        if(t[0] == sd)
          strcat(fs, s);
        else if(t[0] == ed)
          strcat(fs, d);
        else
```

```
strcat(fs, t);
}
strcat(fs, y);
printf("\n After stuffing:%s", fs);
getch();
}
```

## Output:-

Enter characters to be stuffed: goodday Enter a character that represents starting delimiter: d Enter a character that represents ending delimiter: g After stuffing: dggooddddayg.

## **Result::**

Thus, the above program was Successfully completed and verified.

#### 2. Implementation of Error Detection / Correction Techniques

#### Aim::

To write a c program using Implement of Error Detection / Correction Techniques in LRC,CRC.

## Algorithm::

STEP 1 :Get the data and generator polynomial.
STEP 2 :Let n be the length of the generator polynomial.
STEP 3 :Append n-1 zeros to data.
STEP 4 :Call the CRC function.
STEP 5 :
STEP 6 :If the first bit is 1, then perform a xor operation with the first n bits of data and the generator polynomial.
STEP 7 :Shift the bits by 1 position to leave the first bit.
STEP 8 :Append a bit from the data. Repeat the process until all the bits in the data are appended.

## <u>i) LRC,</u>

#### Program:

```
#include<stdio.h>
#include<conio.h>
void main()
{
    int 11,bit[100],count=0,i,choice;
    clrscr();
    printf("Enter the length of data stream: ");
    scanf("%d",&11);
    printf("\nEnter the data stream ");
    for(i=0;i<11;i++)
    {
        scanf("%d",&bit[i]);
        if(bit[i]==1)
        count=count+1;
    }
}</pre>
```

}

printf("Number of 1's are %d",count);
printf("\nEnter the choice to implement parity bit");
printf("\n1-Sender side\n2-Receiver side\n");
scanf("%d",&choice);
switch(choice)
{
 case 1:

if(count%2==0)

bit[11]=0;

else

bit[11]=1;

printf("%d",bit[i]);

break;

```
case 2:
```

```
if(count%2==0)
```

printf("There is no error in the received data stream");

else

```
printf("There is error in the received data stream");
```

break;

## default:

```
printf("Invalid choice");
break;
```

## }

```
getch();
```

}

## Output::

Enter the length of data stream: 10 Enter the data stream 1 1 0 1 0 1 1 1 0 1 Number of 1's are 7 Enter the choice to implement parity bit 1-Sender side 2-Receiver side 1

The data stream after adding parity bit is 11010111011

Enter the length of data stream: 10 Enter the data stream 1 1 1 1 1 0 0 0 1 0 Number of 1's are 6 Enter the choice to implement parity bit 1-Sender side 2-Receiver side 2 There is no error in the received data stream
### (ii) CRC,

#### Program::

```
#include <stdio.h>
#include <conio.h>
#include <string.h>
void main() {
      int i,j,keylen,msglen;
      char input[100], key[30],temp[30],quot[100],rem[30],key1[30];
      clrscr();
      printf("Enter Data: ");
      gets(input);
      printf("Enter Key: ");
      gets(key);
      keylen=strlen(key);
      msglen=strlen(input);
      strcpy(key1,key);
      for (i=0;i<keylen-1;i++) {
              input[msglen+i]='0';
       }
      for (i=0;i<keylen;i++)
       temp[i]=input[i];
      for (i=0;i<msglen;i++) {
              quot[i]=temp[0];
              if(quot[i]=='0')
              for (j=0;j<keylen;j++)
              key[j]='0'; else
              for (j=0;j<keylen;j++)
              key[j]=key1[j];
              for (j=keylen-1;j>0;j--) {
                     if(temp[j]==key[j])
                      rem[j-1]='0'; else
                      rem[j-1]='1';
              }
              rem[keylen-1]=input[i+keylen];
              strcpy(temp,rem);
```

}

strcpy(rem,temp); printf("\nQuotient is "); for (i=0;i<msglen;i++) printf("%c",quot[i]); printf("\nRemainder is "); for (i=0;i<keylen-1;i++) printf("%c",rem[i]); printf("\nFinal data is: "); for (i=0;i<msglen;i++) printf("%c",input[i]); for (i=0;i<keylen-1;i++)</pre>

printf("%c",rem[i]);

getch();

}

#### **Output::**

Enter the Number::

#### The quotient is 10110111 and the remainder is 0111

#### **Result::**

#### 3. Implementation of Stop and Wait, and Sliding Window Protocols

#### Aim::

To write a c program using Implementation of Stop and Wait, and Sliding Window Protocols

#### Algorithm::

STEP 1: Start the ProgramSTEP 2: Import all the necessary packagesSTEP 3: Create two application sender and receiverSTEP 4: Connect both applications using socketSTEP 5: Sender port number and the frame is input as receiverSTEP 6: Sender frame is send to the receiver and display to the reciever

#### Program::

```
#include<stdio.h>
#include<conio.h>
void main()
{
char sender[50], receiver[50];
int i, winsize;
clrscr();
printf("\n ENTER THE WINDOWS SIZE : ");
scanf("%d",&winsize);
printf("\n SENDER WINDOW IS EXPANDED TO STORE MESSAGE OR WINDOW \n");
printf("\n ENTER THE DATA TO BE SENT: ");
fflush(stdin);
gets(sender);
for(i=0;i<winsize;i++)</pre>
receiver[i]=sender[i];
receiver[i]=NULL;
printf("\n MESSAGE SEND BY THE SENDER:\n");
puts(sender);
printf("\n WINDOW SIZE OF RECEIVER IS EXPANDED\n");
printf("\n ACKNOWLEDGEMENT FROM RECEIVER \n");
for(i=0;i<winsize;i++);</pre>
```

```
printf("\n ACK:%d",i);
printf("\n MESSAGE RECEIVED BY RECEIVER IS : ");
puts(receiver);
printf("\n WINDOW SIZE OF RECEIVER IS SHRINKED \n");
getch();
```

}

#### Output::

Enter the windows size : 10 Sender window is expanded to store message or window Enter the data to be sent: forgetcode.com Message send by the sender: forgetcode.com Window size of receiver is expanded Acknowledgement from receiver Ack:5 Message received by receiver is : forgetcode

Window size of receiver is shrinked

#### **Result::**

#### 4. Implementation of Go back-N and Selective Repeat Protocols.

#### Aim::

To write a c program using Implementation of Go back-N and Selective Repeat Protocols

#### Algorithm::

- STEP 1. Start.
- STEP 2. Establish connection (recommended UDP)
- STEP 3. Accept the window size from the client(should be <=40)
- STEP 4. Accept the packets from the network layer.
- STEP 5. Calculate the total frames/windows required.
- STEP 6. Send the details to the client(totalpackets,totalframes.)
- STEP 7. Initialise the transmit buffer.
- STEP 8. Built the frame/window depending on the windowsize.
- STEP 9. Transmit the frame.
- STEP 10. Wait for the acknowledgement frame.
- STEP 11. Close the connection.

#### Program::

```
#include<stdio.h>
```

#### int main()

```
{
```

```
int windowsize,sent=0,ack,i;
```

```
printf("enter window size\n");
```

```
scanf("%d",&windowsize);
```

while(1)

```
{
```

```
for( i = 0; i < windowsize; i++)
```

```
{
```

printf("Frame %d has been transmitted.\n",sent);

```
sent++;
```

if(sent == windowsize)

break;

```
}
```

printf("\nPlease enter the last Acknowledgement received.\n"); scanf("%d",&ack);

```
if(ack == windowsize)
```

break;

```
else
```

```
sent = ack;
```

return 0;

}

#### **Output:-**

enter window size

}

8

Frame 0 has been transmitted.

Frame 1 has been transmitted.

Frame 2 has been transmitted.

Frame 3 has been transmitted.

Frame 4 has been transmitted.

Frame 5 has been transmitted.

Frame 6 has been transmitted.

Frame 7 has been transmitted.

Please enter the last Acknowledgement received.

2

Frame 2 has been transmitted.Frame 3 has been transmitted.Frame 4 has been transmitted.Frame 5 has been transmitted.Frame 6 has been transmitted.Frame 7 has been transmitted.

Please enter the last Acknowledgement received. 8

#### **Result::**

# 5. Implementation of Distance Vector Routing algorithm (Routing Information Protocol) (Bellman-Ford).

#### Aim::

To write a c program using Implementation of Distance Vector Routing algorithm (Routing Information Protocol) (Bellman-Ford)

#### Algorithm::

STEP 1. Open VI-RTSIM software from desktop

STEP 2. Click the Simulation menu bar

STEP 3. Select the "Distance – Vector Routing Algorithm" option from Routing algorithm menu bar.

STEP Network with routers connected through link is drawn by using option in editor(add router, join link, delete router, delete link, Add caption to link, add caption to

router)

STEP 5. Select any two nodes to find the shortest distance between them.

STEP 6: Click the Find path Button to run the program.

STEP 7. Now the shortest paths between the two nodes are calculated.

#### Program::

```
#include<stdio.h>
struct node
{
  unsigned dist[20];
  unsigned from[20];
}rt[10];
int main()
ł
  int costmat[20][20];
  int nodes,i,j,k,count=0;
  printf("\nEnter the number of nodes : ");
  scanf("%d",&nodes);//Enter the nodes
  printf("\nEnter the cost matrix :\n");
  for(i=0;i<nodes;i++)</pre>
  {
     for(j=0;j<nodes;j++)</pre>
```

#### {

```
scanf("%d",&costmat[i][j]);
costmat[i][i]=0;
rt[i].dist[j]=costmat[i][j];//initialise the distance equal to cost matrix
rt[i].from[j]=j;
}
}
do
{
count=0;
for(i=0;i<nodes;i++)//We choose arbitary vertex k and we calculate the direct distance</pre>
```

from the node i to k using the cost matrix

```
//and add the distance from k to node j
        for(j=0;j<nodes;j++)</pre>
        for(k=0;k<nodes;k++)</pre>
          if(rt[i].dist[j]>costmat[i][k]+rt[k].dist[j])
          {//We calculate the minimum distance
             rt[i].dist[j]=rt[i].dist[k]+rt[k].dist[j];
             rt[i].from[j]=k;
             count++;
           }
     }while(count!=0);
     for(i=0;i<nodes;i++)</pre>
     {
        printf("n\n For router %dn",i+1);
        for(j=0;j<nodes;j++)</pre>
        {
          printf("\t\nnode %d via %d Distance %d ",j+1,rt[i].from[j]+1,rt[i].dist[j]);
        }
     }
  printf("\n\n");
  getch();
}
/*
```

#### Output::

A sample run of the program works as:-Enter the number of nodes : 3 Enter the cost matrix : 027 201 710 For router 1 node 1 via 1 Distance 0 node 2 via 2 Distance 2 node 3 via 3 Distance 3 For router 2 node 1 via 1 Distance 2 node 2 via 2 Distance 0 node 3 via 3 Distance 1 For router 3 node 1 via 1 Distance 3 node 2 via 2 Distance 1 node 3 via 3 Distance 0

#### **Result::**

# 6. Implementation of Link State Routing algorithm (Open Shortest Path First) with 5 nodes (Dijkstra's).

#### Aim::

To write a c program using Implementation of Link State Routing algorithm (Open Shortest Path First) with 5 nodes (Dijkstra's).

#### Algorithm::

STEP 1. Create a simulator object
STEP 2. Define different colors for different data flows
STEP 3. Open a nam trace file and define finish procedure then close the trace file, and execute nam on trace file.
STEP 4. Create n number of nodes using for loop
STEP 5. Create duplex links between the nodes
STEP 6. Setup UDP Connection between n(0) and n(5)
STEP 7. Setup another UDP connection between n(1) and n(5)
STEP 8. Apply CBR Traffic over both UDP connections
STEP 9. Choose Link state routing protocol to transmit data from sender to receiver.

STEP 10. Schedule events and run the program.

#### Program::

```
#include <limits.h>
#include <stdio.h>
#define V 9
int minDistance(int dist[], bool sptSet[]) {
    int min = INT_MAX, min_index;
    for (int v = 0; v < V; v++)
    if (sptSet[v] == false && dist[v] <= min)
        min = dist[v], min_index = v;
    return min_index;
}
int printSolution(int dist[], int n) {
    printf("Vertex Distance from Source\n");
    for (int i = 0; i < V; i++)
        printf("%d \t %d\n", i, dist[i]);
}</pre>
```

```
void dijkstra(int graph[V][V], int src) {
  int dist[V];
  bool sptSet[V];
  for (int i = 0; i < V; i++)
    dist[i] = INT_MAX, sptSet[i] = false;
    dist[src] = 0;
  for (int count = 0; count < V - 1; count++) {
    int u = minDistance(dist, sptSet);
    sptSet[u] = true;
    for (int v = 0; v < V; v++)
      if (!sptSet[v] \&\& graph[u][v] \&\& dist[u] != INT_MAX \&\& dist[u] + graph[u][v] < dist[v])
dist[v] = dist[u] + graph[u][v];
  }
  printSolution(dist, V);
}
int main()
{
  int graph[V][V] = { { 0, 6, 0, 0, 0, 0, 0, 8, 0 },
    \{6, 0, 8, 0, 0, 0, 0, 13, 0\},\
    \{0, 8, 0, 7, 0, 6, 0, 0, 2\},\
    \{0, 0, 7, 0, 9, 14, 0, 0, 0\},\
    \{0, 0, 0, 9, 0, 10, 0, 0, 0\},\
    \{0, 0, 6, 14, 10, 0, 2, 0, 0\},\
    \{0, 0, 0, 0, 0, 0, 2, 0, 1, 6\},\
    \{8, 13, 0, 0, 0, 0, 1, 0, 7\},\
    \{0, 0, 2, 0, 0, 0, 6, 7, 0\}
  };
  dijkstra(graph, 0);
  return 0;
}
```

# Output::

Vertex Distance from Source

0

6

14

21

21

- 5 11
- 6 9

7 8

8 15

# **Result::**

# 7. Data encryption and decryption using Data Encryption Standard algorithm.

#### Aim::

To write a c program using Data encryption and decryption using Data Encryption Standard algorithm.

#### Algorithm::

STEP-1: Read the 64-bit plain text.

STEP-2: Split it into two 32-bit blocks and store it in two different arrays.

STEP-3: Perform XOR operation between these two arrays.

STEP-4: The output obtained is stored as the second 32-bit sequence and the original second 32bit sequence forms the first part.

STEP-5: Thus the encrypted 64-bit cipher text is obtained in this way. Repeat the same process for the remaining plain text characters.

#### Program::

//Simple C program to encrypt and decrypt a string

```
#include <stdio.h>
int main()
{
 int i, x;
 char str[100];
 printf("\nPlease enter a string:\t");
 gets(str);
 printf("\nPlease choose following options:\n");
 printf("1 = Encrypt the string.\n");
 printf("2 = Decrypt the string.\n");
 scanf("%d", &x);
 //using switch case statements
 switch(x)
  {
 case 1:
   for(i = 0; (i < 100 \&\& str[i] != '\0'); i++)
     str[i] = str[i] + 3; //the key for encryption is 3 that is added to ASCII value
```

```
printf("\nEncrypted string: %s\n", str);
```

```
break;
case 2:
for(i = 0; (i < 100 && str[i] != '\0'); i++)
str[i] = str[i] - 3; //the key for encryption is 3 that is subtracted to ASCII value
printf("\nDecrypted string: %s\n", str);
break;
default:
    printf("\nError\n");
    }
return 0;
}
```

#### **Output:**

#### #Encryption

"C:\Users\ICT\Google Drive\c project\code\encrypt-decrypt.exe"

Please enter a string: hello

Please choose following options: 1 = Encrypt the string. 2 = Decrypt the string. 1 Encrypted string: khoor

Process returned 0 (0x0) execution time : 8.564 s Press any key to continue.

#### **#Decryption**

"C:\Users\ICT\Google Drive\c project\code\encrypt-decrypt.exe"
Please enter a string: khoor

```
Please choose following options:

1 = Encrypt the string.

2 = Decrypt the string.

2

Decrypted string: hello

Process returned 0 (0x0) execution time : 4.288 s

Press any key to continue.
```

#### **Result::**

#### 8. Data encryption and decryption using RSA (Rivest, Shamir and Adleman) algorithm.

#### Aim::

To write a c program using Data encryption and decryption using RSA (Rivest, Shamir and Adleman) algorithm

#### Algorithm::

STEP-1: Select two co-prime numbers as p and q.
STEP-2: Compute n as the product of p and q.
STEP-3: Compute (p-1)\*(q-1) and store it in z.
STEP-4: Select a random prime number e that is less than that of z.
STEP-5: Compute the private key, d as e \* mod-1(z).
STEP-6: The cipher text is computed as messagee \* mod n.
STEP-7: Decryption is done as cipherdmod n.

#### Program::

```
#include<stdio.h>
#include<math.h>
//to find gcd
int gcd(int a, int h)
{
  int temp;
  while(1)
  {
     temp = a\%h;
     if(temp==0)
     return h;
     a = h;
     h = temp;
  }
}
int main()
{
  //2 random prime numbers
  double p = 3;
  double q = 7;
  double n=p*q;
  double count;
  double totient = (p-1)^*(q-1);
```

```
//public key
  //e stands for encrypt
  double e=2;
  //for checking co-prime which satisfies e>1
  while(e<totient){</pre>
  count = gcd(e,totient);
  if
(count=1)
    break;
  else
    e++;
  }
  //private key
  //d stands for decrypt
  double d;
  //k can be any arbitrary value
  double k = 2;
  //choosing d such that it satisfies d^*e = 1 + k^* totient
  d = (1 + (k*totient))/e;
  double msg = 12;
  double c = pow(msg,e);
  double m = pow(c,d);
  c=fmod(c,n);
  m=fmod(m,n);
  printf("Message data = %lf",msg);
  printf("\np = %lf",p);
  printf("nq = \%lf",q);
  printf("\n = pq = \%lf",n);
  printf("\ntotient = %lf",totient);
  printf("\ne = %lf",e);
  printf("\nd = \%lf",d);
  printf("\nEncrypted data = %lf",c);
  printf("\nOriginal Message Sent = %lf",m);
  return 0;
```

```
}
```

# **Output:**

Message data = 12.000000 p = 3.000000 q = 7.000000 n = pq = 21.000000totient = 12.000000 e = 5.000000 d = 5.000000Encrypted data = 3.000000Original Message Sent = 12.000000

**Result::** 

#### 9. Implement Client Server model using FTP protocol.

#### Aim::

To write a c program using Implement Client Server model using FTP protocol.

#### Algorithm::

STEP 1. The server starts and waits for filename.
STEP 2. The client sends a filename.
STEP 3. The server receives filename.
If file is present,
server starts reading file
and continues to send a buffer filled with
file contents encrypted until file-end is reached.
STEP 4. End is marked by EOF.
STEP 5. File is received as buffers until EOF is
received. Then it is decrypted.
STEP 6. If Not present, a file not found is sent.

#### **Program::**

#include <stdio.h>
#include <sys/socket.h>
#include <arpa/inet.h>
#include <sys/stat.h>
#include <sys/sendfile.h>
#include <fcntl.h>
#include <stdlib.h>
#include <string.h>

#define FILENAME "a.txt" #define SERVER\_IP "127.0.0.1" #define SERVER\_PORT 65496

int main(int argc , char \*\*argv)

```
int
       socket_desc;
struct sockaddr_in server;
       request_msg[BUFSIZ],
char
       reply_msg[BUFSIZ];
// Variables for the file being received
       file_size,
int
       file_desc;
char
       *data;
socket_desc = socket(AF_INET, SOCK_STREAM, 0);
if (socket_desc == -1)
{
       perror("Could not create socket");
       return 1;
}
server.sin_addr.s_addr = inet_addr(SERVER_IP);
server.sin_family = AF_INET;
server.sin_port = htons(SERVER_PORT);
// Connect to server
if (connect(socket_desc, (struct sockaddr *)&server, sizeof(server)) < 0)
{
       perror("Connection failed");
       return 1;
}
// Get a file from server
strcpy(request_msg, "Get ");
strcat(request_msg, FILENAME);
write(socket_desc, request_msg, strlen(request_msg));
recv(socket_desc, reply_msg, 2, 0);
// Start receiving file
if (strcmp(reply_msg, "OK") == 0) \{
       recv(socket_desc, &file_size, sizeof(int), 0);
```

```
data = malloc(file_size);
```

```
file_desc = open(FILENAME, O_CREAT | O_EXCL | O_WRONLY, 0666);
recv(socket_desc, data, file_size, 0);
write(file_desc, data, file_size);
close(file_desc);
}
else {
fprintf(stderr, "Bad request\n");
}
return 0;
```

# **Output:**

}

Connection failed: Connection refused

**Result::** 

# MEDICAL ELECTRONICS

978-93-6255-675-2

# Edited by DR.SMITHA ELSA PETER

# MEDICAL ELECTRONICS COPYRIGHT @ 2021 BY Dr. SMITHA ELSA PETER

ALL RIGHTS RESERVED. NO PART OF THIS BOOK MAY BE REPRODUCED OR TRANSMITTED IN ANY FORM OR BY ANY MEANS WITHOUT WRITTEN PERMISSION FROM THE AUTHOR.

ISBN: 9789362556752

PUBLISHER: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

CHAPTER	CONTENT	PAGE
	INTRODUCTION	5
1	<b>OVERVIEW OF MEDICAL ELECTRONICS</b>	13
	S.LILLYPET	
2	<b>BIOMEDICAL SIGNAL ACQUISITION</b>	30
	Dr.C.RAJINIKANTH	
3	MEDICAL IMAGING TECHNOLOGIES	49
	Dr. SMITHA ELSA PETER	
4	THERAPEUTIC DEVICES	64
	S. MAHESWARAN	
5	MONITORING DEVICES	85
	T DIVYA MANOHARI	
6	<b>BIOINFORMATICS AND DATA ANALYSIS</b>	96
	Dr. S.DEVI	
7	MEDICAL EQUIPMENT DESIGN AND DEVELOPMENT	107
	E. PRIYADHARSHINI	
8	SAFETY AND RISK MANAGEMENT	132
0	R.SARASWATHI	1.00
9	EMERGING TECHNOLOGIES IN MEDICAL ELECTRONICS	168
10		100
10	CASE 51 UDIES IN WEDICAL ELECTRONICS	190
	S.SARASWATHY	•
	KEFEKENCES	210

# CHAPTER 1 OVERVIEW OF MEDICAL ELECTRONICS S.LILLYPET

Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Medical electronics is an interdisciplinary field that merges electronics, biomedical engineering, and healthcare. It is dedicated to the design and application of electronic devices and systems for medical purposes, encompassing diagnosis, monitoring, and treatment of patients.

### **Key Components of Medical Electronics**

#### 1. Biomedical Sensors and Transducers:

- These devices convert biological signals (such as heartbeats and brain waves) into electrical signals for monitoring and analysis.
- Examples include ECG electrodes, pulse oximeters, and glucose sensors.

#### 2. Medical Imaging Systems:

- Technologies that provide visual representations of internal body structures for clinical assessment.
- o Common modalities include X-ray, MRI, CT scans, and ultrasound.

#### 3. Therapeutic Devices:

- Equipment designed for treating medical conditions or aiding in rehabilitation.
- Examples are pacemakers, defibrillators, and infusion pumps.

#### 4. Monitoring Devices:

- Systems that continuously or periodically measure vital signs and other health parameters.
- This category includes blood pressure monitors, heart rate monitors, and wearable health trackers.

#### 5. Data Acquisition and Analysis:

- Techniques for collecting, storing, and analyzing biomedical data.
- Often involves software tools for signal processing and interpretation, increasingly utilizing AI for predictive analytics.

#### **Applications in Healthcare**

- **Diagnostics**: Medical electronics improve diagnostic accuracy through advanced imaging and biosignal monitoring, facilitating earlier and more precise disease detection.
- **Therapy**: Devices like insulin pumps and cardiac rhythm management systems are crucial for managing chronic conditions and enhancing patient outcomes.
- **Telemedicine**: Medical electronics enable remote patient monitoring and consultations, expanding access to healthcare services, particularly in underserved regions.

### **Challenges and Considerations**

- **Safety and Reliability**: Medical devices must comply with stringent safety standards to ensure patient safety and device reliability.
- **Regulatory Compliance**: Devices undergo rigorous testing and approval processes to meet the requirements of regulatory bodies like the FDA and CE marking.

# CHAPTER 2 BIOMEDICAL SIGNAL ACQUISITION Dr. C.RAJINIKANTH

Associate Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Biomedical signal acquisition is the process of capturing and converting biological signals into measurable electrical signals for analysis, monitoring, and diagnosis. This process is essential for understanding physiological functions and diagnosing various medical conditions.

#### Key Concepts in Biomedical Signal Acquisition

- 1. Types of Biological Signals:
  - **Electrical Signals**: Generated by the body, including electrocardiograms (ECG), electromyograms (EMG), and electroencephalograms (EEG).
  - **Non-Electrical Signals**: These include optical signals, such as those from pulse oximetry, and pressure signals, like blood pressure measurements.

#### 2. Sensors and Transducers:

- **Sensors**: Devices that detect physical phenomena (e.g., temperature, pressure) and convert them into electrical signals.
- **Transducers**: Specialized sensors that transform one type of energy into another, converting physiological signals into electrical outputs.

#### 3. Signal Conditioning:

- This involves modifying the acquired signals to enhance their quality and prepare them for analysis. Key processes include:
  - Amplification: Boosting weak signals to levels suitable for processing.
  - Filtering: Removing unwanted noise and artifacts from the signal.
  - **Digitization**: Converting analog signals into digital format for easier analysis.

#### 4. Data Acquisition Systems:

• Systems designed to collect and store data from sensors. These typically consist of hardware (like amplifiers and analog-to-digital converters) and software for processing and analyzing data.

#### **Process of Biomedical Signal Acquisition**

#### 1. Signal Detection:

• Biological signals are detected using sensors or electrodes that are placed on or inside the body.

#### 2. Signal Conditioning:

- The raw signal undergoes processing to enhance its quality. This includes:
  - Amplification: Strengthening weak signals.
  - **Filtering**: Eliminating noise and artifacts.

#### 3. Analog-to-Digital Conversion:

• The conditioned analog signal is converted into a digital format using an analog-to-digital converter (ADC) for analysis.

#### 4. Data Storage and Analysis:

• The digital data is stored for further analysis. Software tools are utilized to interpret the data, identify patterns, and generate reports.

# CHAPTER 3 MEDICAL IMAGING TECHNOLOGIES Dr. SMITHA ELSA PETER

Professor, Department of ECE

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Medical imaging technologies are vital tools in contemporary healthcare, enabling non-invasive visualization of internal body structures. These technologies are crucial for diagnosing diseases, guiding treatments, and monitoring patient progress.

#### Key Types of Medical Imaging Technologies

- 1. X-ray Imaging:
  - **Description**: Utilizes ionizing radiation to create images of internal structures.
  - **Applications**: Commonly used to diagnose fractures, infections, and abnormalities in bones and soft tissues.
  - **Limitations**: Involves exposure to radiation and is not optimal for imaging soft tissues.

#### 2. Computed Tomography (CT) Scanning:

- **Description**: Combines multiple X-ray images taken from various angles to produce crosssectional images (slices) of bones, blood vessels, and soft tissues using computer processing.
- Applications: Useful for diagnosing tumors, internal bleeding, and complex fractures.
- Limitations: Administers a higher radiation dose compared to conventional X-rays.

#### 3. Magnetic Resonance Imaging (MRI):

- **Description**: Employs strong magnetic fields and radio waves to generate detailed images of organs and tissues.
- **Applications**: Particularly effective for imaging soft tissues, such as the brain, spinal cord, and muscles.
- **Limitations**: More expensive than other imaging modalities and may not be suitable for patients with certain implants (e.g., pacemakers).

#### 4. Ultrasound Imaging:

- **Description**: Uses high-frequency sound waves to produce images of internal structures.
- **Applications**: Frequently used in obstetrics (e.g., monitoring fetal development), cardiology, and assessing soft tissue conditions.
- Limitations: Image quality can be compromised by patient obesity or gas in the intestines.

#### 5. Positron Emission Tomography (PET) Scan:

- **Description**: Involves injecting a small amount of radioactive material and utilizing a special camera to detect emitted radiation, creating images of metabolic activity in tissues.
- Applications: Valuable in oncology for detecting cancer and evaluating treatment responses.
- **Limitations**: Involves exposure to radiation and is often combined with CT or MRI for enhanced detail.
- 6. Single Photon Emission Computed Tomography (SPECT):
  - **Description**: Similar to PET, SPECT uses radioactive tracers to visualize blood flow and activity in various organs.
  - Applications: Commonly employed in cardiology, neurology, and oncology.
  - **Limitations**: Generally provides lower resolution than PET and requires more complex interpretation.

# CHAPTER 4 THERAPEUTIC DEVICES S.MAHESWARAN

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Therapeutic devices are medical instruments designed to treat, manage, or alleviate various medical conditions, significantly improving patient outcomes. They play an essential role in healthcare by enhancing treatment effectiveness, providing relief, and supporting rehabilitation.

#### Key Types of Therapeutic Devices

- 1. Cardiac Devices:
  - **Pacemakers**: Small devices implanted in the chest that help regulate heartbeats in patients with arrhythmias.
  - **Implantable Cardioverter Defibrillators (ICDs)**: Devices that monitor heart rhythms and deliver shocks if a life-threatening arrhythmia is detected.

#### 2. Respiratory Devices:

- **Ventilators**: Machines that assist or fully take over the breathing process for patients who cannot breathe adequately on their own.
- **Continuous Positive Airway Pressure (CPAP) Machines**: Used to treat obstructive sleep apnea by keeping airways open during sleep.

#### 3. Infusion Devices:

- **Infusion Pumps**: Deliver precise amounts of medication or fluids to patients over specified periods, commonly used in chemotherapy and pain management.
- **Syringe Pumps**: A type of infusion pump that administers small volumes of medication with high precision.

#### 4. Orthopedic Devices:

- Braces and Supports: Used to stabilize, support, or immobilize joints and limbs.
- **Prosthetics**: Artificial devices designed to replace missing limbs, restoring function and improving quality of life.

#### 5. Neuromodulation Devices:

- **Transcutaneous Electrical Nerve Stimulation (TENS)**: Devices that deliver electrical impulses to relieve pain.
- **Deep Brain Stimulation (DBS)**: Implantable devices that send electrical impulses to specific brain areas to treat movement disorders, such as Parkinson's disease.

#### 6. Wound Care Devices:

- **Negative Pressure Wound Therapy (NPWT)**: Utilizes suction to promote healing in complex wounds, drawing fluids away and enhancing blood flow.
- Advanced Dressings: Specialized materials that protect wounds, maintain moisture, and promote healing.

#### **Applications of Therapeutic Devices**

- **Pain Management**: Devices like TENS units and infusion pumps effectively manage chronic and post-operative pain.
- **Cardiac Management**: Pacemakers and ICDs are crucial for patients with arrhythmias, reducing the risk of sudden cardiac events.

# CHAPTER 5 MONITORING DEVICES T. DIVYA MANOHARI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

#### **Monitoring Devices**

Monitoring devices are vital tools in healthcare that continuously or periodically measure and track essential physiological parameters. They play a crucial role in patient care by providing real-time data that informs clinical decisions, enhances patient safety, and improves overall outcomes.

#### Key Types of Monitoring Devices

#### 1. Vital Signs Monitors:

- **Description**: These devices measure key physiological parameters, including heart rate, blood pressure, respiratory rate, and temperature.
- **Applications**: Commonly used in hospitals, clinics, and emergency settings for continuous patient health monitoring.

#### 2. Cardiac Monitors:

- **Description**: Devices that continuously track heart activity, typically using electrocardiogram (ECG) technology.
- **Applications**: Essential in critical care and during surgical procedures to detect arrhythmias and other cardiac issues.

#### 3. Pulse Oximeters:

- **Description**: Non-invasive devices that measure oxygen saturation levels in the blood through light absorption.
- **Applications**: Widely utilized in hospitals and home care settings to monitor patients with respiratory conditions.

#### 4. Blood Glucose Monitors:

• **Description**: Devices that measure blood sugar levels, usually requiring a small drop blood.

#### 5. Wearable Health Trackers:

- **Description**: Devices such as smartwatches and fitness bands that monitor various health metrics, including heart rate, activity levels, and sleep patterns.
- **Applications**: Increasingly popular for personal health monitoring, offering users insights into their overall well-being.

#### 6. Remote Patient Monitoring Systems:

- **Description**: Integrated systems that gather health data from patients at home and transmit it to healthcare providers for analysis.
- **Applications**: Used in chronic disease management, allowing for continuous monitoring without frequent in-person visits.

#### **Applications of Monitoring Devices**

- **Critical Care**: Vital signs and cardiac monitors are crucial in intensive care units (ICUs), ensuring immediate responses to changes in patient conditions.
- Chronic Disease Management: Devices like blood glucose monitors and remote monitoring systems assist patients in managing conditions such as diabetes and hypertension.

# CHAPTER 6 BIOINFORMATICS AND DATA ANALYSIS Dr.S.DEVI

#### Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Bioinformatics is an interdisciplinary field that merges biology, computer science, and statistics to analyze and interpret biological data. It plays a vital role in unraveling complex biological systems and processes, particularly in areas such as genomics, proteomics, and molecular biology.

#### Key Components of Bioinformatics

- 1. Genomic Data Analysis:
  - **Description**: This involves analyzing DNA sequences to identify genes, mutations, and genetic variations.
  - **Applications**: Crucial for personalized medicine, disease association studies, and understanding evolutionary biology.
- 2. **Proteomics**:
  - **Description**: Focuses on the study of proteins, including their functions, structures, and interactions.
  - **Applications**: Essential for drug discovery, biomarker identification, and gaining insights into cellular processes.

#### 3. Transcriptomics:

- **Description**: Involves analyzing RNA molecules to study gene expression patterns.
- **Applications**: Useful in developmental biology, cancer research, and understanding responses to environmental changes.
- 4. Metabolomics:
  - **Description**: The study of metabolites and metabolic pathways within biological systems.
  - Applications: Helps elucidate disease mechanisms, drug metabolism and nutrition impacts.

#### Data Analysis Techniques

#### 1. Sequence Alignment:

- **Description**: Arranging DNA, RNA, or protein sequences to identify regions of similarity.
- Methods: Tools such as BLAST and Clustal Omega are commonly used.

#### 2. Phylogenetic Analysis:

- **Description**: Constructing evolutionary trees to represent relationships among species or genes.
- **Tools**: Programs like MEGA and PhyML are employed for this analysis.

#### 3. Machine Learning and AI:

- **Description**: Leveraging algorithms to identify patterns and make predictions from biological data.
- **Applications**: Utilized in genomics for variant classification, protein structure prediction, and drug discovery.
- 4. Statistical Analysis:
  - **Description**: Applying statistical methods to interpret biological data and draw meaningful conclusions.
  - Applications: Essential for experimental design, hypothesis testing, and data visualization.

# CHAPTER 7 MEDICAL EQUIPMENT DESIGN AND DEVELOPMENT E. PRIYADHARSHINI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Medical equipment design and development is a crucial process that integrates engineering principles, regulatory standards, and user requirements to create devices that enhance patient care and outcomes. This field covers a wide array of devices, ranging from diagnostic tools to therapeutic equipment, each requiring careful attention to functionality, safety, and usability.

#### Key Phases of Medical Equipment Design and Development

#### 1. Needs Assessment:

- **Description**: Identifying the specific needs of healthcare providers and patients through market research, user feedback, and clinical insights.
- **Importance**: Understanding the clinical context ensures the design effectively addresses real-world challenges.

#### 2. Concept Development:

- **Description**: Generating ideas and preliminary designs based on identified needs.
- **Techniques**: Involves brainstorming, sketching, and developing initial prototypes to explore various solutions.

#### 3. Design and Engineering:

- **Description**: Creating detailed designs that specify materials, components, and functionality.
- **Tools**: Utilizing CAD software and simulation tools to refine the design and assess performance.

#### 4. Prototyping:

- **Description**: Building prototypes to evaluate the design under real-world conditions.
- **Methods**: Employing rapid prototyping techniques, such as 3D printing, to expedite the creation of functional models.

#### 5. Testing and Validation:

- **Description**: Conducting comprehensive testing to ensure the device meets safety, efficacy, and regulatory standards.
- **Types of Testing**: Includes usability testing, performance evaluations, and compliance with standards like ISO and FDA regulations.

#### 6. Regulatory Approval:

- **Description**: Submitting necessary documentation to regulatory bodies for approval.
- **Process**: Involves compiling evidence of safety, efficacy, and adherence to quality standards.

#### 7. Manufacturing and Quality Control:

- **Description**: Scaling up production while ensuring consistency and quality in manufacturing processes.
- **Considerations**: Implementing quality management systems to maintain product standards.

#### 8. Post-Market Surveillance:

- **Description**: Monitoring the device's performance after it reaches the market to ensure ongoing safety and effectiveness.
- Activities: Involves collecting user feedback, managing recalls if necessary, and updating the device based on new insights.

# CHAPTER 8 SAFETY AND RISK MANAGEMENT R.SARASWATHI

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Safety and risk management are critical components in the design, development, and utilization of medical devices. This process ensures that potential hazards are systematically identified, assessed, and mitigated, safeguarding patients, healthcare providers, and the environment. Effective risk management not only enhances patient safety but also supports regulatory compliance and bolsters product reliability.

#### Key Elements of Safety and Risk Management

- 1. Risk Assessment:
  - **Description**: A systematic approach to identifying and analyzing potential hazards associated with medical devices.
  - **Methods**: Techniques such as Failure Modes and Effects Analysis (FMEA) and Hazard Analysis and Critical Control Points (HACCP) are commonly employed.
  - **Outcomes**: Establishes a comprehensive risk profile that informs further safety measures.

#### 2. Risk Control:

- **Description**: Strategies implemented to mitigate identified risks to an acceptable level.
- Approaches:
  - **Design Controls**: Modifying device designs to eliminate hazards, such as incorporating safety features.
  - Warnings and Instructions: Providing clear labeling and user manuals to inform users of potential risks.
  - User Training: Educating healthcare providers on proper device usage to minimize human error.

#### 3. Regulatory Compliance:

- **Description**: Adhering to regulations and standards set by organizations like the FDA, ISO, and CE.
- **Importance**: Ensures that devices meet safety and efficacy standards, enhancing public trust and marketability.

#### 4. Post-Market Surveillance:

- **Description**: Ongoing monitoring of medical device performance following market release.
- Activities: Collecting data on adverse events, conducting user feedback surveys, and analyzing trends to identify new risks.

#### 5. Incident Reporting and Management:

- **Description**: Establishing protocols for the reporting and investigation of incidents involving medical devices.
- **Importance**: Facilitates prompt responses to safety concerns and informs future risk management strategies.

#### 6. Quality Management Systems (QMS):

- **Description**: Implementing a structured approach to managing quality throughout the device lifecycle.
- **Standards**: ISO 13485 is a widely recognized standard outlining requirements for QMS in the medical device industry.

### CHAPTER 9 EMERGING TECHNOLOGIES IN MEDICAL ELECTRONICS P SAVITHA

# **R.SAVITHA**

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Emerging technologies in medical electronics are revolutionizing healthcare by improving diagnostics, treatment, and patient monitoring. These advancements utilize cutting-edge materials, data analytics, and connectivity to enhance patient outcomes and optimize healthcare delivery.

#### Key Emerging Technologies

#### 1. Wearable Health Devices:

- **Description**: Smartwatches and fitness trackers that monitor vital signs and activity levels.
- **Applications**: Enable continuous health monitoring, facilitate early detection of health issues, and promote wellness through real-time feedback.

#### 2. Telemedicine Solutions:

- **Description**: Platforms for remote consultations and monitoring via video conferencing and mobile applications.
- **Applications**: Expand access to healthcare, especially in rural areas, and allow for real-time management of chronic conditions.

#### 3. Artificial Intelligence (AI) and Machine Learning:

- **Description**: Algorithms that analyze extensive datasets to identify patterns and predict patient outcomes.
- **Applications**: Enhance diagnostics, inform personalized treatment plans, and provide predictive analytics for disease progression.

#### 4. Point-of-Care Testing (POCT):

• **Applications**: Facilitate rapid diagnosis and timely treatment decisions in emergency and outpatient settings.

#### 5. Smart Implants:

- **Description**: Implantable devices that monitor physiological parameters and communicate data wirelessly.
- **Applications**: Used for cardiac monitoring, diabetes management, and orthopedic applications to improve patient care.

#### 6. Internet of Medical Things (IoMT):

- **Description**: A network of connected devices that gather and share health data.
- Applications: Enable real-time monitoring, data analysis, and enhanced care coordination.

#### 7. 3D Printing:

- **Description**: Additive manufacturing techniques used to create custom medical devices, implants, and prosthetics.
- **Applications**: Support personalized medical solutions, facilitate rapid prototyping, and enable complex designs that enhance device functionality.

#### 8. Advanced Imaging Technologies:

- **Description**: Innovations like functional MRI (fMRI) and 3D/4D imaging that provide indepth insights into biological processes.
- **Applications**: Improve diagnostic accuracy and aid in treatment planning by visualizing structures and functions in real time.

# CHAPTER 10 CASE STUDIES IN MEDICAL ELECTRONICS S.SARASWATHY

Assistant Professor, Department of ECE Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Case studies in medical electronics demonstrate the effective application of technology in healthcare, showcasing innovative solutions and their positive impacts on patient outcomes. Here are several noteworthy examples:

#### 1. Remote Patient Monitoring with Wearable Devices

Case Study: Chronic Disease Management

**Background**: A healthcare provider introduced a remote patient monitoring system utilizing wearable devices for patients with chronic conditions, such as diabetes and heart disease.

**Technology**: Patients received smartwatches that tracked vital signs, activity levels, and glucose levels in real-time.

Outcome: Prompt interventions from healthcare providers when abnormal readings were detected.

#### 2. Telemedicine Solutions in Rural Healthcare

Case Study: Access to Specialist Care

Background: A rural hospital faced difficulties in connecting patients with specialists.

**Technology**: The hospital adopted a telemedicine platform enabling virtual consultations with specialists.

#### Outcome:

• Increased satisfaction rates among both patients and providers.

#### 3. AI-Driven Diagnostic Imaging

Case Study: Breast Cancer Detection

Background: A healthcare facility aimed to enhance the accuracy of mammography readings.

**Technology**: An AI system was integrated into the imaging department to assist radiologists in identifying anomalies in mammograms.

#### Outcome:

• Fewer false positives, resulting in a reduction of unnecessary biopsies.

# ANINTRODUCTION

# ORGANIC CHEMISTRY

# EDITEDBY M.TAMIZHSELVAN



OrganicChemistry:AnIntroduction

Copyright©2021 By M.TAMIZHSELVAN

All rights reserved. No part of this book may be reproduced or transmitted in any formor by any means without written permission from the author.

ISBNNumber 978-93-6255-016-3

Publisher:PonnaiyahRamajayamInstituteofScienceandTechnology (PRIST)

# Contents

Chapter:1	1
BasicConceptsinChemicalBondingandOrganic Molecules Dr. M. Jerome Rozario	
Chapter:2	15
FundamentalofOrganicStructures	
Dr.R.Manikandan	
Chapter:3	38
ConformationsofAlkanesand Cycloalkanes	
Dr.R.Manikandan	
Chapter:4	54
Stereochemistry	
Dr.R.Manikandan	
Chapter:5	75
Stereoisomers	
Dr.D.Senthil Nathan	
Chapter:6	95
NucleophilicSubstitution Reactions	
Dr.JSNirmalRam	
Chapter:7	109
Elimination Reactions	
Dr.D.Senthil Nathan	
Reference	140
### Basic Concepts in Chemical Bonding and Organic Molecules

### Dr. M. Jerome Rozario

AssistantProfessor,DepartmentofChemistry,SchoolofArtsandScience,Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

### ChemicalBonding

To summarize simply, a chemical bond is the attractive force holding atoms or ions together. This attractive interaction leads to a more stable state for the whole system compared to individual atoms.

**Valence electrons** play a fundamental role in chemical bonding. In the electron configuration of an atom, the outermost shell is called the valence shell, and the *electrons in the valence shell (outermost shell)* areknownas **valence electrons**. Takethecarbonatom forexample: theelectronconfiguration of carbon is  $1s^22s^22p^2$ . The outermost shell is the  $2^{nd}$  principal shell, so there are 4 valence electrons in carbon. Valence electrons are the electrons that are the furthest away from the nucleus, and thus they experience the least attraction from the nucleus and arethe most reactive. Theyplaythe most important role in chemical bonding.

### **Resonancestructures:**

In the case that more than one reasonable (plausible) Lewis structure can be drawn for a species, these structures are called **resonance structures** or **resonance contributors**. Resonance structures can be either equivalent or non-equivalent.

Equivalent Resonance Structures Let's consider the example of carbonate anion, CO  $_{3:}^{2:}$ 

three Lewis structures are obtained by following arrow in different color

### **FundamentalofOrganicStructures**

### Dr.R.Manikandan

AssociateProfessor,DepartmentofChemistry,SchoolofArtsandScience,Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

In this chapter, we will discuss the fundamental structural features of organic compounds, the categorization and drawing of organic structures, functional groups, and nomenclatures.

LearningObjectives forthis chapter:

- Draw,recognizeandreaddifferentstructureformulasthatrepresentorganicmoleculescorrectly and skillfully.
- Understand and be able to recognize constitutional isomers. Draw all constitutional isomers with a given formula, by applying both structure knowledge and degree of un saturation concept.
- Recognize and name simple organic molecules with common functional groups by applying IUPAC nomenclature.
- Understand the nature and structure effect on different intermolecular forces, and be able to predict and explain the physical property of substance based on the intermolecular force knowledge.

### **OrganicCompoundsOverview**

Organic compounds are compounds that contain the **carbon** element. The simplest organic compoundis a **hydrocarbon**, which is a compound containing only the elements carbon and hydrogen. Hydrocarbons are composed of several sub-categories: alkane, alkene, alkyne and aromatic, depending on the type of carbon-carbon bonds involved.

Hydrocarbons can be in chains (straight-chains or branched-chains) or rings. The hydrocarbon chain and ring form the "carbon **backbone**" of organic compounds, and functional groups connected to the backbone allow for a great diversity of organic structures. **Functional groups** are common and specific arrangements of atoms, usually heteroatoms (atoms other than carbon and hydrogen) like N, O, and Cl that show specific and relatively high reactivities. Knowledge about the common functional groups in this chapter will prepare us for the later discussion on organic reactions.

### ConformationsofAlkanesand Cycloalkanes

### Dr.R.Manikandan

AssociateProfessor,DepartmentofChemistry,SchoolofArtsandScience,Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

### Conformation

At a molecular level, a property of  $\sigma$  (sigma) bonds in alkane is that the bonds keep on rotating. For the example of ethane (CH<sub>3</sub>CH<sub>3</sub>), one methyl (CH<sub>3</sub>) group is able to rotate around the C-C bond freely without any obstacles.

It is highly recommended that the molecular model is used here to "see" the bond rotation. With a molecular model on hand, you can hold one methyl group steady, and rotate the other methyl group.

The C-C bond is formed by the sp<sup>3</sup>-sp<sup>3</sup> orbitals overlapping and the bond is cylindrically symmetrical, so rotation about the bond can occur easily and the molecule does not seem to change. However, a closer look indicates that the rotation of the C-C bond **does** result in a different spatial arrangement of hydrogen atoms in the molecule, as shown below:



## staggered conformation

## eclipsed conformation

### Twoconformersofethanein perspective formulas

The different spatial arrangements of the atoms/groups that result from the single bond rotation are called **conformations**. Molecules with different conformations are called **conformational isomers**or **conformers**. Thetwo extremeconformations of ethanecomingfromtheC-Crotationshownaboveare: the *staggered conformation* with all of the H atoms spread out and the *eclipsed conformation* with all of the H atoms overlapped.

### Stereochemistry

### Dr.R.Manikandan

AssociateProfessor,DepartmentofChemistry,SchoolofArtsandScience,Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Stereochemistry deals with three dimensional representation of molecule in space. This has sweeping implications in biological systems. For example, most drugs are often composed of a single stereoisomer of a compound. Among stereoisomer's one may have positive effects on the body and another stereoisomer may not or could even be toxic. An example of this is the drugthalidomide which was used during the 1950s to suppress the morning sickness. The drug unfortunately, was prescribed as a mixture of stereoisomer's, and while one stereoisomer actively worked on controlling morning sickness, the other stereoisomer caused serious birth defects.

The study of stereochemistry focuses on stereoisomer's and spans the entire spectrum of organic, inorganic, biological, physical and especially supramolecular chemistry.

Stereochemistry includes method for determining and describing these relationships; the effect on the physical or biological properties.

Stereochemistry is the systematic presentation of a specific field of science and technology that traditionally requires a short preliminary excursion into history. Stereochemistry is the *'chemistry of space '*, that is stereochemistry deals with the spatial arrangements of atoms and groups in a molecule.

Stereochemistry can trace its roots to the year 1842 when the **French chemist Louis Pasteur** made an observation that the salts of tartaric acid collected from a wine production vessel have the *ability to rotate plane-polarized light*, whereas the same salts from different sources did not have this ability. This phenomenon is explained by optical isomerism.

### Stereoisomerism

### Dr.D.SenthilNathan

AssociateProfessor,DepartmentofChemistry,SchoolofArtsandScience,Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

FactsaboutStereochemistry

- The structure of a molecule can vary based on the three-dimensional arrangement of the atoms that constitute it. Stereochemistry also deals with the manipulation of the arrangement of these atoms.
- This branch of chemistry is *commonly referred to as 3-D chemistry* since it focuses on stereoisomers (chemical compounds with the same chemical formula but a different spatial arrangement in three dimensions).
- One of the branches of stereochemistry deals with the study of molecules that exhibit chirality, which is a geometric property of molecules that makes them non-superimposable on their mirror images.
- Another branch of 3-D chemistry, known as dynamic stereochemistry, involves the study of the effects of different spatial arrangements of atoms in a molecule on the rate of a chemicalreaction.

### Stereoisomerism

Stereoisomerism refers to "the isomerism that is caused by the non-similar arrangements of atoms or functional groups belonging to an atom in space". These types of isomers have the same constitutions, but different geometric arrangements of atoms. Stereoisomers can be broadly classified into two types, namely enantiomers and diastereomers.

### NucleophilicSubstitutionReactions Dr.J.S.Nirmal

### Ram

AssociateProfessor,DepartmentofChemistry,SchoolofArtsandScience,Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

**Nucleophilic substitution reaction** is a class of organic reactions where one nucleophile replaces another. It is very similar to the normal displacement reactions which we see in chemistry, where, a more reactive element replaces a less reactive element from its salt solution. The group which takes electron pair and displaced from the carbon is known as "leaving group" and the molecule on which substitution takes place is known as "substrate". The leaving group leaves as a neutral molecule or anion.



### **Elimination Reactions**

### Dr.D.SenthilNathan

AssociateProfessor,DepartmentofChemistry,SchoolofArtsandScience,Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

Aneliminationreactionisareactioninwhichatomsareremovedasmoleculesorcompounds. Eliminationis generallycatalysed byametal, anacid or base.

Elimination reactions often compete with substitution reactions. In this reaction, a substrate (typically an alkyl halide) eliminates one equivalent (unit) of acid to form an alkene. Two possible mechanisms are available for this elimination reaction – *E1 and E2 mechanisms*.



An elimination reaction is a type of chemical reaction where several atoms either in pairs or groups are removed from a molecule. The *removal usually takes place due to the action of acids and bases or the action of metals.* It can also happen through the process of heating at high temperatures.

# OOOO AN INTRODUCTION TO INDUSTRIAL CHEMISTRY

Edited by V.ABARNA

978-93-6255-546-5

An introduction to Industrial Chemistry

Copyright©2021

By Ms.V.Abarna

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-546-5

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

### **Table of Contents**

Chapter 1
An Introduction to Industrial Chemistrypg. 1
Ms.V.Abarna
Chapter 2
Textile and Dye industry
Dr.M.Surendra Varma
Chapter 3
Polymer industry
Dr.R.Manikandan
Chapter 4
Electro chemical industry
Dr.R.Manikandan
Chapter 5
Ceramic industry pg. 49
Dr.D.Senthilnathan
Chapter 6
Alternative energy sources
Dr.D.Chinnaraja
Chapter 7
Chemical accumulators
Dr.G.Anburaj
References

## An Introduction to Industrial Chemistry

### **Chapter 1**

### An Introduction to Industrial Chemistry

### Ms.V.Abarna

### Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Industrial chemistry is the process of transforming matter into materials that are useful to humanity through chemical processes. For example, the transformation of rubber into tires. Industrial chemists use chemical testing and innovative ideas to solve problems and improve products. They are responsible for the creative use of chemicals to improve a product's performance at the microscopic level.

Industrial chemistry is not to be confused with chemical engineering, which focuses more on how to process and produce goods using chemicals. Industrial chemists are more concerned with improving the production process.

For a more specific understanding, it is important to understand the different kinds of industrial chemistry.

### TYPES OF INDUSTRIAL CHEMISTRY

Industrial chemistry, like analytical chemistry, is a prominent component in many of the most familiar industries. Materials science, for example, focuses on turning chemicals into materials like consumer rubber or waterproof fabrics. Cosmetic industrial chemistry challenges chemists to formulate newer and more advanced makeup and care products. Some of the most common types of chemical testing include:

- Energy
- Materials
- Inks, coatings, and thin films
- Human and Veterinary Pharmaceuticals
- Health
- Cosmetics
- Household care
- Food
- Environment
- Electronics

No matter what type or end-goal application, industrial chemistry always requires careful, repeatable testing. At ACE Laboratories, our state of the art testing laboratory is committed to repeatable, reputable physical and analytical testing on which our clients—and their product's end users—can depend.

### Textile and Dye industry

### Dr.M.Surendra Varma

### Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

### **Natural Fibers**

**Cotton:** Cotton is a commonly used natural fibre known for its breathability and absorbent properties. When dyeing cotton, you need a dye that can penetrate the fibres effectively. Reactive dyes are popular for cotton as they chemically bond with the fibres, providing excellent colourfastness.

**Silk:** Silk is a luxurious natural fibre prized for its sheen and softness. Acid dyes work well with this fabric because they bond effectively with protein-based fibres. They produce vivid, vibrant colours on silk textiles.

**Wool:** Wool is another protein-based fibre that requires careful consideration. <u>Acid dyes</u> are also suitable for the fabric, but basic dyes are another option, offering rich and deep colours.

### **Synthetic Fibers**

**Polyester:** Polyester is a synthetic fibre known for its durability and resistance to wrinkles and moisture. Disperse dyes are the go-to choice for dyeing polyester. These dyes work well at high temperatures and provide excellent colour retention.

**Nylon:** Nylon is a versatile synthetic fibre used in a wide range of applications. Like polyester, disperse dyes are suitable for nylon because they withstand high-temperature dyeing processes.

Acrylic: Acrylic fibres are often used as a substitute for wool. Basic dyes work effectively with acrylic, offering vibrant colours and good colourfastness.

Understanding the fibre content is the first step in choosing the right dye for your fabric. Once you've identified the type of fibre, you can consider the specific dyes most compatible with it.

### **Types of Dyes Used in the Textile Industry**

Textile dye manufacturers formulate various types of dyes, each offering unique characteristics and applications. Let's explore some common categories of dyes used in the textile industry.

### Acid Dye

- Ideal for protein-based fibres like silk and wool
- Offers brilliant and vibrant colours
- Requires an acidic environment for dyeing

Acid dyes are known for their ability to produce vibrant colours on protein-based fibres. These dyes require an acidic pH level during dyeing, and they bond effectively with the fibres. Due to

### Polymer industry

Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The binding modes leading to the formation of dye-polymer conjugates can be either covalent or non-covalent in nature. While the first approach obviously requires the formation of covalent bonds, non-covalent binding can occur through different kinds of interactions such as ionic and dipole–dipole interactions or through the formation of inclusion complexes.

### 2.1. Non-Covalent Attachment

### 2.1.1. Sugar-Based Polymers

A schematic illustration of the non-covalent dye binding to polymeric materials



Figure 2. Schematic illustration of non-covalent dye binding to polymers.

Due to the large number of polar substituents that enable the formation of dipolar interactions with adequate substrates, sugar-based (macro)molecules are suitable materials for the supramolecular attachment of dyes. Such oligo-/polysaccharides can be obtained from natural products (e.g., starch, cellulose, chitosane) or from chemical linkage of monomeric subunits and are therefore readily accessible.

### Electro chemical industry

### Dr.R.Manikandan

### Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Over the years, electrochemical methods affected by direct electric current have been preferably utilized in the chemical industries as compare to the usual chemical methods. This is because of it following merits:

a) Simplicity of the technological method.

b) Raw materials and energy are used entirely.

c) Some of the valuable products are obtained concurrently.

d) Highly pure products are made up.

Electrochemical processes, though, does encompass a demerits which is that there is a large consumption of the electric power.

### Basic Electrochemical Terms:

For a clearer understanding of electrochemistry, we should be familiar with some of the terminologies employed and what they mean or represent.

1) <u>Electrolysis</u>: This signifies to the decomposition of a substance via an electric current. This in reality is the main stay of electrochemical industries. The father of electrochemistry - Micheal Faraday stipulated two laws from his studies to preside over electrolysis. These laws are defines thus:

a) The weights of substances made at an electrode throughout electrolysis are directly proportional to the quantity of electricity which passes via the electrolyte.

b) The weights of various substances made up by the passage of the similar quantity of electricity are proportional to the equivalent weight of each and every substance.

Note: The equivalent weight of a substance is stated as the molar mass, divided by the number of electrons needed to oxidize or reduce each and every unit of the substance. For illustration, one mole of  $V^{3+}$  corresponds to three equivalents of this species and will therefore need three (3) Faradays of charge to deposit it as metallic vanadium. One mole of electric charge, 1F = 96,500 coulombs.

### Ceramic industry

### Dr.D.Senthilnathan

### Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

**Ceramics**, an ancient and versatile art form, have captivated human imagination and creativity for millennia. With roots dating back to prehistoric times, ceramics encompass a wide range of materials and techniques used in the creation of functional and decorative objects. From delicate porcelain to robust earthenware, ceramics offer a unique blend of durability, aesthetic beauty, and practicality. This art form has evolved across different cultures, leaving an indelible mark on human history and shaping the way we perceive and interact with the world around us.

The art of ceramics involves the manipulation and transformation of clay, a naturally occurring material abundant in the Earth's crust. Clay possesses remarkable plasticity, allowing it to be shaped into various forms before undergoing a process of firing to achieve a durable, hardened state. The firing process involves subjecting the clay to high temperatures, which causes chemical and physical changes within the material, resulting in its characteristic strength and permanence. Additionally, ceramics can be embellished through techniques like glazing, which adds a decorative and protective layer to the surface, further enhancing the artistic and functional qualities of the finished piece. From pottery and tableware to sculpture and architectural elements, ceramics offer a diverse range of applications, making them an integral part of both artistic expression and everyday life.

### Types of Ceramics

In the field of material science, ceramics can be broadly classified into several types based on their composition, structure, and properties. Here are some common **types of ceramics**:

### **Oxide Ceramics**

Oxide ceramics are the most widely used type of ceramics and are composed of metallic and non-metallic elements. Examples include alumina (aluminum oxide), zirconia (zirconium dioxide), and magnesia (magnesium oxide). These ceramics exhibit excellent mechanical strength, high melting points, and good electrical and thermal insulation properties. They find applications in industries such as electronics, aerospace, and automotive.

### Alternative energy sources .

### Dr.D.Chinnaraja Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Alternative sources of energy can be defined as the use of sources of energy other than the traditional fossil fuels (such as oil, coal, and natural gas), which are shorter in supply and which are considered harmful to the environment. It includes all renewable and nuclear energy sources.

The most commonly used alternative sources of energy include the following:

- 1. Wind Energy
- 2. Solar Energy
- 3. Geothermal Energy
- 4. Bioenergy
- 5. Hydroelectric Energy
- 6. Hydrogen Energy
- 7. Nuclear Energy
- 8. Tidal Energy

The above-listed sources of energy are explained below:

Uses of Alternative Sources of Energy

- 1. Wind Energy
  - In this source of energy, the wind is used for producing electricity by making use of the kinetic energy created by air in motion.
- 2. Solar Energy
  - In this source of energy, the radiation from the Sun is used which is capable of producing heat, causing chemical reactions, and generating electricity.
- 3. Geothermal Energy
  - It is a source of energy that is taken from the Earth's core. It comes from the heat which is generated during the original formation of the planet.
- 4. Bioenergy
  - In this source of energy electricity and gas are generated from the organic matter known as biomass. Bioenergy is one of the resources available to help meet the demand for energy, it includes electricity, heat and transportation fuel.

It is a source of energy that is obtained by falling water from high potential to low potential as potential is defined to flow down from a certain height. It is a form of energy that harnesses the power of water in motion—such as water flowing over a waterfall—to generate electricity.

### Chemical accumulators

### Dr.G.Anburaj

### Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The lead accumulator is a secondary cell because the electrical energy is not generated inside the cell but rather stored until being supplied by an external source.

It's a reversible cell since the cell reactions are reversible if the applied external emf is sufficiently larger than the cell's emf. As a result, total cell reactions are reversed in this cell by exerting external opposition to e.m.f that is bigger than the cell's e.m.f.

This cell has the ability to store electrical energy from charging or an external source, and it can also supply energy during the discharge process. As a result, the energy is stored in chemical form. As a result, this cell denotes an accumulator, storage cell, or storage battery.

The voltage of the cells is not affected by the size of the electrodes or the size of the cell, but it is affected by the strength of the sulphuric acid solution.

### Formation

Lead plates in the form of a negative electrode are used in the lead accumulator. The lead plates, which are brimming with lead oxide, act as positive electrodes. The positive and negative electrodes are arranged in a counterclockwise direction. This collection of lead plates is immersed in a non-conducting tank constructed of plastic, glass, or ebonite, all of which are non-conductors of electricity and contain 38% hydrogen. All of the positive plates are linked to one another, and all of the negative plates are linked to one another.

### **Cell Representation**

-Pb(s)| PbSO<sub>4</sub>(s)| 38% H<sub>2</sub>SO<sub>4</sub>(aq)| PbSO<sub>4</sub>(s)| PbO<sub>2</sub>(s)| Pb(s)+

### Working

Cell Discharging: When the cell is turned on, oxidation takes place on the lead plates, followed by reduction using PbO (Lead Oxide). The discharge of the cell is the term for this.

Cell charging: When the emf operating outside is greater than the emf applied to the cell, accurate reverse reactions occur. On positive electrodes, oxidation occurs, while on negative electrodes, reduction occurs. The charging of the cell is the term for this

## ESSENTIALS IN PHYSICAL CHEMISTRY

Edited by Dr.P.CHRISTURAJ



Essentials in Physical Chemistry

Copyright©2021

By

P.Christuraj

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-707-0

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

## **Table of Contents**

## Chapter 1

## P.Christuraj

Nuclear Chemistry	.pg. 2		
Chapter 2			
Dr.M.Surendra Varma			
Electro Chemistry	pg. 10		
Chapter 3			
Dr.R.Manikandan			
Chemical kinetics	pg. 28		
Chapter 4			
Dr.D.Chinnaraja			
Thermo chemistry	pg. 45		
Chapter 5			
Dr.D.Chinnaraja			
Solutions	pg. 65		
Chapter 6			
Dr.R.Manikandan			
Chemical equilibrium	pg. 72		
Chapter 7			
Dr.G.Anburaj			
Coordination Compounds	pg. 93		
References	pg. 115		

### Nuclear chemistry

### P.Christuraj

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Nuclear chemistry is the study of the physical and chemical properties of elements which are influenced by changes in the structure of the atomic nucleus. It also deals with the energy released from nuclear reactions and its uses. It is also termed as radiochemistry which includes the study of the formation of the elements in the universe, the design of radioactive drugs for diagnostic medicine and many other applications.

### **Nuclear Reactions and Radiations**

There are three types of radioactive radiations which were sorted out by Rutherford in 1902 by passing them between two oppositely charged plat.

- The ones which bent towards the negative plate carry a positive charge and are known as **alpha rays**.
- The ones which bent towards the positive plate carry a negative charge and are known as **beta rays**.
- The third type being uncharged passed straight through the electric field is known as **gamma rays**.

### **Types of Radiations**

Unlike a normal chemical reaction that forms molecules, nuclear reactions result in the transformation of one element into another. This property of nuclear reactions is used in nuclear power plants to gather <u>nuclear energy</u>. The three common types of radiations are explained below.

### **Alpha Radiation**

It is the emission part of an alpha particle from an atom's nucleus. The  $\alpha$  particle includes two protons and two neutrons and is much similar to <u>Helium</u> (He )nucleus <sup>4</sup>He<sub>2</sub>. When an atom emits an  $\alpha$  particle, the atomic mass of an atom decreases by 4 units.

### **Electro Chemistry**

### Dr.M.Surendra Varma

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

**Electrochemistry** is the subdiscipline of Chemistry that deals with the study of the relationship between electrical energy and chemical changes. Chemical reactions that involve the input or generation of electric currents are called electrochemical reactions. Such reactions are broadly classified into two categories:

Electricity can be produced when electrons move from one element to another in certain types of reactions (such as redox reactions). Typically, electrochemistry deals with the overall reactions when multiple redox reactions occur simultaneously, connected via some external electric current and a suitable electrolyte. In other words, electrochemistry is also concerned with chemical phenomena that involve charge separation (as seen commonly in liquids such as solutions). The dissociation of charge often involves charge transfer that occurs homogeneously or heterogeneously between different chemical species.

### **Electrochemical Cell**

A spontaneous chemical process is one which can take place on its own, and in such a process, the Gibbs free energy of a system decreases. In electrochemistry, spontaneous reaction (redox reaction) results in the conversion of chemical energy into electrical energy. The reverse process is also possible where a non-spontaneous chemical reaction occurs by supplying electricity. These interconversions are carried out in equipment called an electrochemical cell.

### **Chemical Kinetics**

### Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Chemical kinetics, also called reaction kinetics, helps us understand the rates of reactions and how it is influenced by certain conditions. It further helps to gather and analyse information about the mechanism of the reaction and define the characteristics of a chemical reaction.

### **Rate of Formations and Disappearances**

In any chemical reaction, as the reaction proceeds, the amount of reactants decreases, whereas the amount of products increases. One has to understand that the rate of the overall reaction depends on the rate at which reactants are consumed or the rate at which the products are formed.

If a graph is plotted between the concentration of reactants and products and time, the rate of formation of products and the rate of disappearance of reactants can be easily calculated from the slope of curves for products and reactants. The overall rate of the reaction may or may not be equal to the rate of formations and disappearances.



### Thermo chemistry

### Dr.D.Chinnaraja

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Thermochemistry involves the study of heat and energy related to various physical transformations and chemical reactions. During the reaction, energy can be absorbed (endothermic) or it can be released (exothermic). Thermochemistry is mainly concerned with a change in energy mainly regarding an exchange of energy of a system with its surroundings.

According to the laws of thermodynamics, the energy of an isolated system remains constant. If the system is not isolated, then during any chemical reaction, the energy is conserved. To understand energy changes, heat changes and the spontaneity of reactions, thermochemistry is very useful. All reactions have heat, work and heat-work relationships in common. During the reaction, work can be done on the system or work can be done by the system. Similarly, heat can be absorbed or heat can be given out of the system. A study of all these also helps in determining the actual energy available in food.



### Solutions

### Dr.D.Chinnaraja

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Solutions have two components, one is solvent and the other is solute.

1. What is a Solvent?

The component that dissolves the other component is called the solvent.

2. What is Solute?

The component(s) that is/are dissolved in the solvent is/are called solute(s).

Generally solvent is present in major proportion compared to the solute. The amount of solute is lesser than the solvent. The solute and solvent can be in any state of matter i.e. solid, liquid or gas.

Solutions that are in the liquid state consist of a solid, liquid or gas dissolved in a liquid solvent. Alloys and air are examples of solid and gaseous solutions, respectively.

3. Solution Examples

The following examples illustrate solvent and solute in some solutions.

- Air is a homogeneous mixture of gases. Here both the solvent and the solute are gases.
- Sugar syrup is a solution where sugar is dissolved in water using heat. Here, water is the solvent and sugar is the solute.
- Tincture of iodine, a mixture of iodine in alcohol. Iodine is the solute whereas alcohol is the solvent.

### Chemical equilibrium

### Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Chemical equilibrium refers to the state of a system in which the concentration of the reactant and the concentration of the products do not change with time, and the system does not display any further change in properties.

Table of Contents

- Types of Chemical Equilibrium
- Factors Affecting Chemical Equilibrium
- Examples
- Importance
- Problems

When the rate of the forward reaction is equal to the rate of the reverse reaction, the state of chemical equilibrium is achieved by the system. When there is no further change in the concentrations of the reactants and the products due to the equal rates of the forward and reverse reactions, the system is said to be in a state of dynamic equilibrium.



### Coordination Compounds

### Dr.G.Anburaj

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

**Coordination compounds** are chemical compounds that consist of an array of anions or neutral molecules that are bound to a central atom via coordinate covalent bonds. Coordination compounds are also referred to as **coordination complexes**. These molecules or ions that are bound to the central atom are referred to as ligands (also known as complexing agents). The definitions of some important terms in the chemistry of coordination compounds can be found below.

Important Terms Involving Coordination Compounds

The definitions of some important terms in the chemistry of coordination compounds can be found below.

#### Coordination Entity

A chemical compound in which the central ion or atom (or the coordination centre) is bound to a set number of atoms, molecules, or ions is called a coordination entity.

Some examples of such coordination entities include [CoCl3(NH3)3] and [Fe(CN)6]4-.

### Central Atoms and Central Ions

As discussed earlier, the atoms and ions to which a set number of atoms, molecules, or ions are bound are referred to as the central atoms and the central ions.

In coordination compounds, the central atoms or ions are typically Lewis Acids and can, therefore, act as electron-pair acceptors.

## INORGANIC CHEMISTRY: AN INSIGHT

EDITED BY



Inorganic Chemistry: An insight Copyright©2021 By V.ABARNA

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-361-4

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

Chapter 1
Atomic structure
Ms. V.Abarna
Chapter 2
Molecular structure and bondingpg. 18
Dr.M.Surendra Varma
Chapter 3
The structures of simple solids
Dr.R.Manikandan
Chapter 4
Molecular symmetry
Dr.R.Manikandan
Chapter 5
Acids and bases
Dr.R.Manikandan
Chapter 6
Oxidation and reduction pg. 62
Dr.D.Chinnaraja
Chapter 7
Periodic trends
Dr.G.Anburaj
References

**Table of Contents** 

### Atomic structure

### Ms.V.Abarna

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The atomic structure of an element refers to the constitution of its nucleus and the arrangement of the electrons around it. Primarily, the atomic structure of matter is made up of protons, electrons and neutrons.

The protons and neutrons make up the nucleus of the atom, which is surrounded by the electrons belonging to the atom. The atomic number of an element describes the total number of protons in its nucleus.



Neutral atoms have equal numbers of protons and electrons. However, atoms may gain or lose electrons in order to increase their stability, and the resulting charged entity is called an ion.

Atoms of different elements have different atomic structures because they contain different numbers of protons and electrons. This is the reason for the unique characteristics of different elements.

### Molecular structure and bonding

Dr.M.Surendra Varma

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The attractive force which holds various constituents (atoms, ions, etc.) together and stabilises them by the overall loss of energy is known as chemical bonding. Therefore, it can be understood that chemical compounds are reliant on the strength of the chemical bonds between their constituents; the stronger the bonding between the constituents, the more stable the resulting compound will be.

The opposite also holds true; if the chemical bonding between the constituents is weak, the resulting compound would lack stability and would easily undergo another reaction to give a more stable chemical compound (containing stronger bonds). To find stability, the atoms try to lose their energy.

Whenever matter interacts with another form of matter, a force is exerted on one by the other. When the forces are attractive in nature, energy decreases. When the forces are repulsive in nature, energy increases. The attractive force that binds two atoms together is known as a chemical bond.

### **Important Theories on Chemical Bonding**

Albrecht Kössel and Gilbert Lewis were the first to explain the formation of chemical bonds successfully in the year 1916. They explained chemical bonding on the basis of the inertness of noble gases.

### Lewis Theory of Chemical Bonding

- An atom can be viewed as a positively charged 'Kernel' (the nucleus plus the inner <u>electrons</u>) and the outer shell.
- The outer shell can accommodate a maximum of eight electrons only.
- The eight electrons present in the outer shell occupy the corners of a cube which surround the 'Kernel'.
- The atoms have an octet configuration, i.e., 8 electrons in the outermost shell, thus symbolising a stable configuration.

### Molecular symmetry

### Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Molecular symmetry is a basic idea in chemistry. It is about the symmetry of molecules. It puts molecules into groups according to their symmetry. It can predict or explain many of a molecule's chemical properties. Chemists study symmetry to explain how crystals are made up and how chemicals react. The molecular symmetry of the reactants helps predict how the product of the reaction is made up and the energy needed for the reaction.

Molecular symmetry can be studied several different ways. Group theory is the most popular idea. Group theory is also useful in studying the symmetry of molecular orbitals. This is used in the Hückel method, ligand field theory, and the Woodward–Hoffmann rules. Another idea on a larger scale is the use of crystal systems to describe crystallographic symmetry in bulk materials.

Scientists find molecular symmetry by using X-ray crystallography and other forms of spectroscopy. Spectroscopic notation is based on facts taken from molecular symmetry.



### Acids and bases

### Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Many of the acids that we do not consume in the household are used in the laboratories and industries, which include an acid such as HCl, H<sub>2</sub>SO<sub>4</sub> etc, and bases such as NaOH, KOH etc. When these acids and bases are mixed in the right proportions, the neutralization reaction thus results in the formation of salt and water. Some naturally occurring salts found in nature include NaCl and KCl etc in seawater and natural rock deposits. In this section, we will read more about acid, base, and salt and their properties.

### Definitions

- 1. Acid:- An acid is defined as a substance whose water solution tastes sour, turns blue litmus red, and neutralizes bases.
- Base:- A substance is called base if its aqueous solution tastes bitter, turns red litmus blue, or neutralizes acids.
- 3. Salt:- Salt is a neutral substance whose aqueous solution does not affect litmus.



### **Group Separations**

### Dr.D.Chinnaraja

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Qualitative analysis is the systematic approach that involves precipitation reaction to remove cations sequentially from a mixture. The behaviour of the cations toward a set of common test reagents differs from one cation to another and furnishes the basis for their separation.

The qualitative analysis of an inorganic mixture is started by first carrying out some preliminary tests. The Preliminary tests for cations are

- 1. Physical examination
- 2. Charcoal cavity test
- 3. Boraxbead test
- 4. Flame test

These tests do not give conclusive evidence yet they provide some information about the ions present in the mixture.

Qualitative analysis of cations usually consists of three stages.

- 1. *First* based on different solubility properties the cations are separated into 5 groups through the successive addition of selective precipitating reagents.
- 2. *Second*, within each group precipitated cations are separated through selective dissolution processes.
- 3. Third, the presence of each cation is verified through different identification tests.

The cations are classified into the following 5 groups.

### Oxidation and reduction

### Dr.D.Chinnaraj

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

### What is Oxidation?

According to Classical or earlier concept oxidation is a process which involves the addition of oxygen or any electronegative element or the removal of hydrogen or any electropositive element.

According to electronic concept oxidation is defined as the process in which an atom or ion loses one or more electrons.

### What is Reduction?

According to Classical or earlier concept reduction is a process which involves the addition of hydrogen or any electropositive element or the removal of oxygen or any electronegative element.

According to electronic concept reduction is defined as the process in which an atom or ion gains one or more electrons.

Oxidation	Reduction
1. gain of oxygen	1. Loss of oxygen
2. loss of hydrogen	2. Gain of hydrogen
3. loss of electrons	3. Gain of electrons
4. increase in oxidation number	4. Decrease in oxidation number

$$\begin{array}{c} \text{Increase in O.N = 3} \\ +2 & -3 & 0 & 0 \\ \text{CuO} & + & \text{NH}_3 & \longrightarrow & \text{Cu} & + & \text{N}_2 & + & \text{H}_2\text{O} \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & &$$
# **Periodic trends**

# Dr.G.Anburaj

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The modern periodic table is based on the law that the properties of an element are a periodic function of their atomic number. These properties are related to the electronic configuration of the elements. We observe a common trend in properties as we move across a period from left to right or down the group. This trend in properties is known as periodic properties. The important periodic properties are atomic size, metallic character, non-metallic character, ionization potential, electron affinity, and electro negativity.



# UNVEILING THE MARVELS OF ORGANIC CHEMISTRY

Edited by DR.J.S.NIRMALRAM



Unveiling the Marvels of Organic Chemistry Copyright©2021 By DR.J.S.NIRMALRAM

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-348-5 Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

Chapter No	TITLE	NAME	PAGE
Chapter 1	The Basics Bonding and Molecular Structure.	Dr.J.S Nirmal Ram	4
Chapter 2	Families of Carbon CompoundsFunctionalGroups,IntermolecularForces, andInfrared (IR)Spectroscopy	Dr.M.Surendra Varma	12
Chapter 3	Acids and Bases An Introduction to Organic Reactions and Their Mechanisms	Dr.R.Manikandan	17
Chapter 4	Nomenclature and Conformations of Alkanes and Cycloalkanes	Dr.D.Chinnaraja	24
Chapter 5	Stereochemistry Chiral Molecules	Dr.G.Anburaj	33
Chapter 6	NucleophilicReactionsPropertiesandSubstitutionReactions of Alkyl Halides	M.Tamizhselvan	41
Chapter 7	Alkenes and Alkynes I Properties and Synthesis. Elimination Reactions of Alkyl Halides	Dr. D. Senthilnathan	49
Chapter 8	Alkenes and Alkynes II Addition Reactions	Dr.P.CHRISTURAJ	52
Chapter 9	Nuclear Magnetic Resonance and Mass Spectrometry Tools for Structure Determination	Dr.G.Anburaj	55
Chapter 10	Radical Reactions	Dr.J.S Nirmal Ram	59

References

Chapter 1	The	Basics	Bonding	and	Dr.J.S Nirmal Ram
	Molec	Molecular Structure.			

The three dimensional shape or **configuration** of a molecule is an important characteristic. This shape is dependent on the preferred spatial orientation of covalent bonds to atoms having two or more bonding partners. Three dimensional configurations are best viewed with the aid of models.

In order to represent such configurations on a twodimensional surface (paper, blackboard or screen), we often

use perspective drawings in which the direction of a bond is



specified by the line connecting the bonded atoms. In most cases the focus of configuration is a carbon atom so the lines specifying bond directions



will originate there. As defined in the diagram on the right, a simple straight line represents a bond lying approximately in the surface plane. The two bonds to substituents  $\mathbf{A}$  in the structure on the left are of this kind. A wedge shaped bond is directed in front of this plane (thick end toward the viewer), as shown by the bond to substituent  $\mathbf{B}$ ; and a hatched bond is directed in back of the plane (away from the viewer), as shown by the bond to substituent  $\mathbf{D}$ . Some texts and other sources may use a dashed bond in the same manner as we have defined the hatched bond, but this can be confusing because the dashed bond is often used to represent a partial bond (i.e. a covalent bond that is partially formed or partially broken). The following examples make use of this notation, and also illustrate the importance of including non-bonding valence shell electron pairs (colored blue) when viewing such configurations .



Methane

Ammonia

Water

Chapter 2	Families of Carbon Compounds Functional Groups, Intermolecular
	Forces, and Infrared (IR) Spectroscopy

# **Spectral Interpretation by Application of Group Frequencies**

One of the most common application of infrared spectroscopy is to the identification of organic compounds. The major classes of organic molecules are shown in this category and also linked on the bottom page for the number of collections of spectral information regarding organic molecules.

# Hydrocarbons

# Hydrocarbons

compounds contain only C-H and C-C bonds, but there is plenty of information to be obtained from the infrared spectra arising from C-H stretching and C-H bending.

In

# alkanes

, which have very few bands, each band in the spectrum can be assigned:

- C-H stretch from 3000–2850 cm<sup>-1</sup>
- C-H bend or scissoring from 1470-1450 cm<sup>-1</sup>
- C–H rock, methyl from 1370-1350 cm<sup>-1</sup>
- C–H rock, methyl, seen only in long chain

alkanes

, from 725-720 cm<sup>-1</sup>

Figure 3. shows the IR spectrum of octane. Since most organic compounds have these features, these C-H vibrations are usually not noted when interpreting a routine IR spectrum. Note that the change in dipole moment with respect to distance for the C-H stretching is greater than that for others shown, which is why the C-H stretch band is the more intense.

Chapter 3	Acids and Bases An Introduction to Organic Reactions and
	Their Mechanisms

# **Organic Acids**

The acids that we talked about in General Chemistry usually refers to inorganic acids, such as HCl,  $H_2SO_4$ , HF etc. If the structure of the acid contains a "carbon" part, then it is an organic acid. Organic acids donate protons in the same way as inorganic acids, however the structure may be more complicated due to the nature of organic structures.

Carboxylic acid, with the general formula of R-COOH, is the most common organic acid that we are familiar with. Acetic acid (CH<sub>3</sub>COOH), the ingredient of vinegar, is a simple example of a carboxylic acid. The  $K_a$  of acetic acid is  $1.8 \times 10^{-5}$ .

Another common organic acid is the organic derivative of sulfuric acid H<sub>2</sub>SO<sub>4</sub>.



The replacement of one OH group in H<sub>2</sub>SO<sub>4</sub> with a carbon-containing R (alkyl) or Ar (aromatic) group leads to the organic acid named "sulfonic acid", with the general formula of RSO<sub>3</sub>H, or ArSO<sub>3</sub>H. Sulfonic acid is a strong organic acid with a  $K_a$  in the range of 10<sup>6</sup>. The structure of a specific sulfonic acid example called *p* -toluenesulfonic acid is shown here

Chapter 4	Nomenclature and Conformations of Alkanes and Cycloalkanes

After completing this section, you should be able to

- 1. name a substituted or unsubstituted cycloalkane, given its Kekulé structure, shorthand structure or condensed structure.
- 2. draw the Kekulé, shorthand or condensed structure for a substituted or unsubstituted cycloalkane, given its IUPAC name.
- 3. draw all possible cycloalkane structures (substituted or unsubstituted) that correspond to a given molecular formula.
- 4. Cycloalkanes are cyclic hydrocarbons, meaning that the carbons of the molecule are arranged in the form of a ring. Cycloalkanes are also saturated, meaning that all of the carbons atoms that make up the ring are single bonded to other atoms (no double or triple bonds). There are also polycyclic alkanes, which are molecules that contain two or more cycloalkanes that are joined, forming multiple rings.
- 5. Many organic compounds found in nature or created in a laboratory contain rings of carbon atoms with distinguishing chemical properties; these compounds are known as cycloalkanes. Cycloalkanes only contain carbon-hydrogen bonds and carbon-carbon single bonds, but in cycloalkanes, the carbon atoms are joined in a ring. The smallest cycloalkane is cyclopropane.



cydopentane

6.

7. If you count the carbons and hydrogens, you will see that they no longer fit the general formula CnH2n+2CnH2n+2. By joining the carbon atoms in a ring, two hydrogen atoms have been lost. The general formula for a cycloalkane is CnH2nCnH2n. Cyclic compounds are not all flat molecules.

Chapter 5	
-----------	--

Organic compounds, molecules created around a chain of carbon atom (more commonly known as carbon backbone), play an essential role in the chemistry of life. These molecules derive their importance from the energy they carry, mainly in a form of potential energy between atomic molecules. Since such potential force can be widely affected due to changes in atomic placement, it is important to understand the concept of an <u>isomer</u>, a molecule sharing same atomic make up as another but differing in structural arrangements. This article will be devoted to a specific isomers called stereoisomers and its property of <u>chirality</u> (Figure 1).



Figure 1: Two enantiomers of a tetrahedral complex.

The concepts of steroisomerism and chirality command great deal of importance in modern <u>organic chemistry</u>, as these ideas helps to understand the physical and theoretical reasonsbehind the formation and structures of numerous organic molecules, the main reason behind the energy embedded in these essential chemicals. In contrast to more well-known constitutional isomerism, which develops isotopic compounds simply by different atomic connectivity, stereoisomerism generally maintains equal atomic connections and orders of building blocks as well as having same numbers of atoms and types of elements.

What, then, makes stereoisomers so unique? To answer this question, the learner must be able to think and imagine in not just two-dimensional images, but also three-dimensional space. This is due to the fact that stereoisomers are isomers because their atoms are different from others in terms of spatial arrangement.

Chapter 6	Nucleophilic Reactions Properties and Substitution Reactions of Alkyl
	Halides

# Alkyl Halide Structure and Reaction Language

The carbon bonded to a halide is called the alpha-carbon. The carbons bonded to the alpha-carbon are called beta-carbons. Carbon atoms further removed from the alpha carbon are named by continuing the Greek alphabet (alpha, beta, gamma, delta, etc). In discussing the reactions of alkyl halides, it can be effective to use the alpha- and beta- labels. The structure for 2-bromopropane is used below to illustrate the application of these terms.



# The Reactions - Nucleophilic Substitution and Elimination

Alkyl halides can undergo two major types of reactions - substitution and/or elimination. The substitution reaction is called a Nucleophilic Substitution reaction because the electrophilic alkyl halide forms a new bond with the nucleophile which substitutes for (replaces) the halogen at the alpha-carbon. Because carbon can only form four bonds, the halogen must leave and is called the "Leaving Group". Alkyl halides are excellent electrophiles because halogens share a polar bond with carbon, are polarizable, and form relatively stable leaving groups as halide anions. In the example below, 2-bromopropane is converted into propan-2-ol in a substitution reaction.

CH3CHCH3 + NaOH ----- CH3CHCH3 + NaBr I Br OH

Chapter 7	Alkenes and Alkynes I Properties and Synthesis. Elimination
	Reactions of Alkyl Halides

An elimination reaction is a reaction in which atoms are removed as molecules or compounds. Elimination is generally catalysed by a metal, an acid or base.

Elimination reactions often compete with substitution reactions. In this reaction, a substrate (typically an alkyl halide) eliminates one equivalent (unit) of acid to form an alkene. Two possible mechanisms are available for this elimination reaction – *E1 and E2 mechanisms*.

# **Table of Contents**

- <u>What is Elimination Reaction?</u>
- Important Methods of Elimination Reaction
- Mechanism Of Elimination Reaction
- El Reaction
- E2 Reaction
- <u>Elimination Reaction Example</u>
- <u>Frequently Asked Questions FAQs</u>

# What is Elimination Reaction?

Elimination reaction is a type of reaction that is mainly <u>used to transform saturated compounds</u> (organic compounds which contain single carbon-carbon bonds) <u>to unsaturated compounds</u> (compounds which feature double or triple carbon-carbon bonds).

Besides, it is an important method for the preparation of alkenes.

An elimination reaction is a type of chemical reaction where several atoms either in pairs or groups are removed from a molecule. The *removal usually takes place due to the action of acids and bases or the action of metals.* It can also happen through the process of heating at high temperatures.

Chapter 8	Alkenes and Alkynes II Addition Reactions	

# What is the addition of alkynes?

The chemistry associated with carbon-carbon triple bonds can be called as alkyne chemistry. Alkynes undergo addition reactions due to the presence of loosely held pi-electrons. Due to the presence of a triple bond in alkynes, halogens, water etc.can be added to them by the process of the addition reaction. Addition products are formed via a set of steps. The stability of vinylic cations is responsible for the formation of addition products. Markovnikov's rule is followed by asymmetric alkynes in order to undergo addition reaction. Few addition reactions of alkynes are explained below:

# Addition of alkynes with halogens (halogenation)

Alkynes and halogens undergo addition reaction to form halogenated alkenes which further react with <u>halogens</u> to give halogen substituted alkanes. The reddish orange coloured solution ofbromine and carbon tetrachloride gets decolorized as a result of the addition reaction. This is used as a test for unsaturation.

# Addition of alkynes with hydrogens (hydrogenation)

Alkynes react with dihydrogen in the presence of catalysts such as Pt/Pd/Ni in order to form alkenes. The <u>alkenes</u> formed, further react with dihydrogen to form alkanes. It has been observed that in most reactions triple bond is converted into double bond and double bond is then converted into a single bond due to the addition reaction with dihydrogen.

The catalyst that are involved in the case of rhodium, nickel, palladium, and platinum. Hydrogenation is a step-by-step process in which initially an alkene is formed. After which it undergoes further hydrogenation to form an alkane.

Slowing down the reaction in the intermediate stage is actually quite impossible as the whole reaction is really smooth. But some alkenes are isolated with the use of poisoned catalysts. One such example of a poisoned catalyst is <u>Lindlar catalyst</u>.

Chapter 9	Nuclear Magnetic Resonance and Mass Spectrometry Tools for
	Structure Determination

# **Basis of NMR Spectroscopy**

Nuclear Magnetic Resonance (NMR) was first detected experimentally at the end of 1945, nearly concurrently with the work groups Felix Bloch, Stanford University and Edward Purcell, Harvard University. The first NMR spectrum was first published in the same issue of the Physical Review in January 1946. Bloch and Purcell were jointly awarded the 1952 Nobel Prize in Physics for their research of Nuclear Magnetic Resonance Spectroscopy.

Nuclear magnetic resonance (NMR) spectroscopy is a crucial analytical tool for organic chemists. The research in the organic lab has been significantly improved with the aid of the NMR. Not only can it provide information on the structure of the molecule, it can also determine the content and purity of the sample. Proton (1H) NMR is one of the most widely used NMR methods by organic chemists. The protons present in the molecule will behave differently depending on the surrounding chemical environment, making it possible to elucidate their structure.

# NMR Spectroscopy Principle

Many nuclei have spin, and all nuclei are electrically charged, according to the NMR principle. An energy transfer from the base energy to a higher energy level is achievable when an external magnetic field is supplied.

- All nuclei are electrically charged and many have spin.
- Transfer of energy is possible from base energy to higher energy levels when an external magnetic field is applied.
- The transfer of energy occurs at a wavelength that coincides with the radio frequency.
- Also, energy is emitted at the same frequency when the spin comes back to its base level.
- Therefore, by measuring the signal which matches this transfer the processing of the NMR spectrum for the concerned nucleus is yield.

Chapter 10	Radical Reactions	Dr.J.S Nirmal Ram	59

# **The Free-Radical Chain Reaction**

# The three phases of radical chain reactions

The following video by Leah4Sci provides a good introduction to free radical reactions.

# https://www.voutube.com/watch?v=Uz1 n9ZnksY

Because of their high reactivity, free radicals have the potential to be both extremely powerful chemical tools and extremely harmful contaminants. Much of the power of free radical species stems from the natural tendency of radical processes to occur in a chain reaction fashion. **Radical chain reactions** have three distinct phases: initiation, propagation, and termination.

initiation

$$\overrightarrow{A - B} \xrightarrow{\text{(heat or light)}} A + B$$

propagation

$$A \xrightarrow{\frown} C \xrightarrow{\frown} D \longrightarrow A - C + \cdot D$$
$$D \xrightarrow{\frown} E \xrightarrow{\frown} F \longrightarrow D - E + \cdot F \longrightarrow \text{etc.}$$

termination

 $F \cdot + \cdot G \longrightarrow F - G$ 

The **initiation phase** describes the step that initially creates a radical species. In most cases, this is a homolytic cleavage event, and takes place very rarely due to the high energy barriers involved. Often the influence of heat, UV radiation, or a metal-containing catalyst is necessary to overcome the energy barrier.

Molecular chlorine and bromine will both undergo homolytic cleavage to form radicals when subjected to heat or light. Other functional groups which also tend to form radicals when exposed to heat or light are chlorofluorocarbons, peroxides, and the halogenated amide N-bromosuccinimide (NBS).



# INSTRUMENTAL METHODS OF CHEMICAL ANALYSIS

Edited by

**DR.D.SENTHILNATHAN** 



Instrumental Methods of Chemical Analysis

Copyright©2021

By Dr.D.Senthilnathan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-418-5

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

# Contents

Chapter: 1	1
Introduction to Instrumental Methods of Chemical Analysis	
Dr.D.Senthil Nathan	
Chapter: 2	15
Voltammetric method	
Dr.R.Manikandan	
Chapter: 3	38
Difference Pulse Voltammetric method	
Dr.R.Manikandan	
Chapter: 4	54
Squire Wave Voltammetric method	
Dr.R.Manikandan	
Chapter: 5	75
Chromatography	
Dr.D.Senthil Nathan	
Chapter: 6	95
Applications of Chromatography	
Dr.JSNirmal Ram	
Chapter: 7	109
Applications of HPLC	
Dr.D.Senthil Nathan	
Reference	140

# **Introduction to Instrumental Methods of Analysis**

# **Dr.D.Senthil Nathan**

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# INTRODUCTION

Analytical instrumentation plays an important role in the production and evaluation of new products and in the protection of consumers and environment. It is used in checking the quality of raw materials such as substances used in integrated circuit chips, detection and estimation of impurities to assure safe foods, drugs, water and air, process optimization and control, quality check of finished products and research and development. Most of the modern instruments are microprocessor/computer controlled with user friendly software for collection of data, analysis and presentation.

This chapter deals with the different types of analytical instrumental methods that find use in a variety of industries. These include molecular spectroscopic methods, thermal methods of analysis, X-ray diffraction, scanning electron microscope and sensors.

# SPECTROSCOPY

It is the study of interaction of electromagnetic radiation with matter consisting of atoms and molecules. When a substance is irradiated with electromagnetic radiation, the energy of the incident photons may be transferred to atoms and molecules raising their energy from ground state level to excited state. This process is known as absorption and the resultant spectrum is known as absorption spectrum. The process of absorption can occur only when the energy difference between the two levels E is exactly matched by the energy of the incident photons as given by the equation

$$E = hv = hc/\lambda$$

# **VOLTAMMETRIC METHOD**

# Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

**Voltammetry** is a category of electroanalytical methods used in analytical chemistry and various industrial processes. In voltammetry, information about an analyte is obtained by measuring the current as the potential is varied. The analytical data for a voltammetric experiment comes in the form of a **voltammogram**, which plots the current produced by the analyte versus the potential of the working electrode.

# Theory

Voltammetry is the study of current as a function of applied potential. Voltammetric methods involve electrochemical cells, and investigate the reactions occurring at electrode/electrolyte interfaces.<sup>[4]</sup> The reactivity of analytes in these half-cells is used to determine their concentration. It is considered a dynamic electrochemical method as the applied potential is varied over time and the corresponding changes in current are measured. Most experiments control the potential (volts) of an electrode in contact with the analyte while measuring the resulting current (amperes).

#### **Electrochemical cells**

Electrochemical cells are used in voltammetric experiments to drive the redox reaction of the analyte. Like other electrochemical cells, two half-cells are required, one to facilitate reduction and the other oxidation. The cell consists of an analyte solution, an ionic electrolyte, and two or three electrodes, with oxidation and reduction reactions occurring at the electrode/electrolyte interfaces. As a species is oxidized, the electrons produced pass through an external electric circuit and generate a current, acting as an electron source for reduction. The generated currents are Faradaic currents, which follow Faraday's law. As Faraday's law states that the number of moles of a substance, m, produced or consumed during an electrode process is proportional to the electric charge passed through the electrode, the faradaic currents allow analyte concentrations to be determined. Whether the analyte is reduced or oxidized depends on the analyte, but its reaction always occurs at the working/indicator electrode. Therefore, the working electrode potential varies as a function of the analyte concentration.

# DIFFERENCE PULSE VOLTAMMETRIC METHOD

# Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

#### **Differential pulse voltammetry**

**Differential Pulse Voltammetry (DPV)** is an electrochemical technique. In Differential Pulse Voltammetry a potential scan is made using pulses with a constant amplitude of E pulse superimposed on the dc-potential.

#### Typical voltage amplitude

The voltage amplitude is mostly in the range of 5 - 50 mV.



# **Interval time**

The interval time between the pulses is equal to  $E_{step}$  / scan rate.

## **Obtained current**

The current is sampled twice in each step: one time just before applying the pulse and one time at the end of the pulse. The difference between these two current samples is plotted versus the potential.

The obtained current is proportional to the derivative of the curve obtained using linear sweep or normal pulse voltammetry. A DPV thus has a peak-shaped curve. The peak height is (normally) proportional to the concentration in the solution. Be aware that the peak is not the redox potential! E <sub>peak</sub> =  $E_{\frac{1}{2}} - E_{\frac{pulse}{2}}$ 

# **Square Wave Voltammetry (SWV)**

# Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# Square Wave Voltammetry (SWW) is in fact a special version of Differential Pulse Voltammerty (DPV).

Differential Pulse Voltammerty is Square Wave Voltammetry when t pulse is equal t interval/2. The interval time is the inverse of the frequency (Freq): t interval = 1/ Freq. As in DPV, the pulse amplitude is also normally in the range of 5 - 25 or 50 mV.



# Measuring Square Wave Voltammetry (SWV)

In trace analysis, it is important to apply a frequency with optimal values. As in DPV the optimal value must be found by varying the frequency.

SWV is sometimes used to measure the reaction rate of the electrode reaction. In this case, a plot of the currents observed in the positive potential pulses are plotted next to the currents measured in the negative potential pulses (forward and reverse currents).

# Chromatography

# **Dr.D.Senthil Nathan**

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

The term *"chromatography"* is derived from Greek, chroma meaning, *"colour,"* and graphein meaning *"to write."* 

In this process, we apply the mixture to be separated on a stationary phase (solid or liquid) and a pure solvent such as water or any gas is allowed to move slowly over the stationary phase, carrying the components separately as per their solubility in the pure solvent.

# **Table of Contents**

- Principles of Chromatography
- Types of Chromatography
- Related Topics on Chromatography
- What is Differential Extraction?
- Applications of Chromatography
- Frequently Asked Questions

# **Principles of Chromatography**

Chromatography is a separation method where the analyte is combined within a liquid or gaseous mobile phase., which is pumped through a stationary phase. Usually one phase is hydrophilic and the other is lipophilic. The components of the analyte interact differently with these two phases. Depending on their polarity they spend more or less time interacting with the stationary phase and are thus retarded to a greater or lesser extent. This leads to the separation of the different components present in the sample. Each sample component elutes from the stationary phase at a specific time called as retention time. As the components pass through the detector their signal is recorded and plotted in the form of a chromatogram.

# **Applications of Chromatography**

# **Dr.J.S.Nirmal Ram**

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

# Applications of Chromatography in the Pharmaceutical Industry

- The technique of chromatography is extensively employed in the pharmaceutical industry in order to analyze and identify the presence of any trace amounts of chemicals and elements in a given sample.
- The technique of chromatography also plays a crucial role in the development of new drugs. For example, the presence of impurities and other unknown compounds can be detected in the drug sample with the help of chromatography. Furthermore, the purity of the drug sample can also be analyzed with the help of this technique.

# Applications of Chromatography in the Food Industry

- In the food industry, the technique of chromatography plays a vital role in the determination of the shelf life of food substances by helping in the analysis of the point at which food spoils.
- Furthermore, the presence of chemical additives in food can also be determined with the help of this technique.
- The nutritional value of the food sample can also be determined by employing chromatographic techniques.

# Applications of Chromatography in the Chemical Industry

- Chromatography plays a vital role in the chemical industry for the testing of water samples for purity.
- The testing of air samples for their purity is also accomplished by chromatographic techniques in the chemical industry.
- The presence of toxic contaminants in oils and pesticides (the most notable of which being polychlorinated biphenyls, often abbreviated to PCBs) can be determined with the help of specialized chromatographic techniques such as GC and HPLC.
- It can also be noted that chromatography also has many applications in the life sciences.

# **Applications of HPLC**

# **Dr.D.Senthil Nathan**

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

The acronym HPLC stands for High Performance Liquid Chromatography. "Chromatography" is a separation technique, "chromatogram" is the chromatography result, and "chromatograph" is the chromatography apparatus. HPLC is a technique for disjointing, determinating, and quantifying each component in a mixture.

Spectroscopy is used to identify and quantify the mixture, which is separated using the fundamental concept of column chromatography. Some of the major components of chromatographs include devices specialised in molecule separation known columns and high-performance pumps for supplying solvent at a steady flow rate, among the different technologies created for chromatography. The system once known as High Performance Liquid Chromatography became known simply as "LC" as associated technologies got more advanced. Ultra High Performance Liquid Chromatography (UHPLC), which is capable of high-speed analysis, is becoming increasingly popular and is being widely used nowadays.



HPLC can only evaluate chemicals that are dissolved in solvents. HPLC separates chemicals dispersed in a liquid sample, allowing for qualitative and quantitative examination of which components are present in the sample and how much of each component is present.

In the 1960s, LC with low-pressure glass columns evolved into high-pressure chromatography (HPLC) using metal columns. As a result, it is a better version of TLC. Instead of allowing a solvent to drop through a column under gravity, it is pushed through at up to 400 atmospheres of pressure.



\*

Edited by DR. D. SENTHILNATHAN



Essentials of Physical Chemistry Copyright©2021 by Dr. D. Senthilnathan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-963-0

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

# **Essentials of Physical Chemistry**

# Contents

Chapter: 1	
The Properties of gases	01
Dr.D.Senthilnathan	
Chapter: 2	
Thermodynamics - First Law	27
Dr.D.Senthilnathan	
Chapter: 3	
Thermodynamics - Application of First Law	55
Dr. M. Surendra Varma	
Chapter: 4	89
Thermodynamics - Second Law	
Dr. M. Surendra Varma	
Chapter: 5	
Phase Rule	109
Dr. M. Surendra Varma	
Chapter: 6	
Solid State	133
Dr.R.Manikandan	
Chapter: 7	
Physical equilibria - Pure Substance	159
Dr.R.Manikandan	
Reference	246

•

#### The Properties of gases

## Dr.D.Senthilnathan

## Associate Professor, Department of Chemistry, School of Arts and Science,

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

## Introduction

The study of gases allows us to understand the behavior of matter at its simplest: individual particles, acting independently, almost completely uncomplicated by interactions and interferences between each other. This knowledge of gases will serve as the pathway to our understanding of the far more complicated *condensed phases* (liquids and solids) in which the theory of gases will no longer give us correct answers, but it will still provide us with a useful *model* that will at least help us to rationalize the behavior of these more complicated states of matter.

First, we know that a gas has **no definite volume or shape**; a gas will fill whatever volume is available to it. Contrast this to the behavior of a liquid, which always has a distinct upper surface when its volume is less than that of the space it occupies. The other outstanding characteristic of gases is their **low** *densities*, compared with those of liquids and solids. One mole of liquid water at 298 K and 1 atm pressure occupies a volume of 18.8 cm<sup>3</sup>, whereas the same quantity of water vapor at the same temperature and pressure has a volume of 30200 cm<sup>3</sup>, more than 1000 times greater.

The most remarkable property of gases, however, is that to a very good approximation, **they all** *behave the same way* in response to changes in temperature and pressure, expanding or contracting by predictable amounts. This is very different from the behavior of liquids or solids, in which the properties of each particular substance must be determined individually. We will see later that each of these three *macroscopic* characteristics of gases follows directly from the *microscopic* view— that is, from the *atomic* nature of matter.

#### The Pressure of a Gas

The molecules of a gas, being in continuous motion, frequently strike the inner walls of their container. As they do so, they immediately bounce off without loss of kinetic energy, but the reversal of direction (*acceleration*) imparts a *force* to the container walls. This force, divided by the total surface area on which it acts, is the *pressure* of the gas.

#### Thermodynamics - First Law

#### Dr.D.Senthilnathan

## Associate Professor, Department of Chemistry, School of Arts and Science,

#### Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

To understand the relationship between work and heat, we need to understand a third, linking factor: the change in internal energy. Energy cannot be created nor destroyed, but it can be converted or transferred. Internal energy refers to all the energy within a given system, including the kinetic energy of molecules and the energy stored in all of the chemical bonds between molecules. With the interactions of heat, work and internal energy, there are energy transfers and conversions every time a change is made upon a system. However, no net energy is created or lost during these transfers.

The First Law of Thermodynamics states that energy can be converted from one form to another with the interaction of heat, work and internal energy, but it **cannot** be created nor destroyed, under any circumstances. Mathematically, this is represented as

 $\Delta U=q+w$ 

With

- $\Delta U\Delta$  is the total change in internal energy of a system,
- q is the heat exchanged between a system and its surroundings, and
- w is the work done by or on the system.

Work is also equal to the negative external pressure on the system multiplied by the change in volume:

$$w = -p \Delta V$$

where P is the external pressure on the system, and  $\Delta V$  is the change in volume. This is specifically called "pressure-volume" work.

The internal energy of a system would decrease if the system gives off heat or does work. Therefore, internal energy of a system increases when the heat increases (this would be done by adding heat into a system). The internal energy would also increase if work were done onto a system. Any work or heat that goes into or out of a system changes the internal energy. However, since energy is never created nor destroyed (thus, the first law of thermodynamics), the change in internal energy always equals zero. If energy is lost by the system, then it is absorbed by the surroundings. If energy is absorbed into a system, then that energy was released by the surroundings:

# Thermodynamics - Application of First Law

# Dr.M.Surendra Varma

Assistant Professor, Department of Chemistry, School of Arts and Science,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Applications of First Law of Thermodynamics:

We went through an overview of what the laws of thermodynamics are, now, let us see what are the applications of the First Law of Thermodynamics. As we all know, the first law of thermodynamics talks about the conservation of energy. Therefore, the applications of the first law of thermodynamics are differentiated into 4 purposes:

1:- **Isolated Process:** As the name proposes, it is a process that does not interact with the surroundings. Because the system is isolated, there is no heat loss and thus, no work is done. Therefore, the internal energy of an isolated system remains constant and thus, no energy is destroyed in any of the stages of the process.

2:- A Cyclic Process: As the name proposes, this is a process where the system returns to its original state after passing all the intermediate stages. Just like the isolated process, in this process also the change in internal energy is zero thus, no energy is destroyed.

3:- **Boiling Process:** When a liquid is heated, it absorbs heat and the temperature of the liquid thus rises. After some time, a stage is reached by the liquid, when it starts boiling and changes its phase from liquid to vapour. Due to this change of phase from liquid to vapour, the volume increases and work is done. As the process involves work and heat, the first law of thermodynamics can be applied. In this process, internal energy changes and gets converted from one form to the other without getting destroyed, fulfilling all the conditions of the first law of thermodynamics.

4:- **Melting Process:** When the quantity of heat Q is given to a solid at its melting point, it is converted into liquid. The temperature and pressure remain constant till the whole solid is completely converted into liquid. The internal energy changes during melting, therefore the first law of thermodynamics can be applied here as well. Just like the boiling process, in this process, internal energy changes and gets converted from one form to the other without getting destroyed, fulfilling all the conditions of the first law of thermodynamics. The only difference between the last two processes is the temperature to which the internal energy changes. On one hand, the liquid changes to vapour as an effect of the rise in internal energy, and on the other hand, the liquid changes to a solid phase as a result of the negative change in heat energy.

# Thermodynamics - Second Law

# Dr.M.Surendra Varma

# Assistant Professor, Department of Chemistry, School of Arts and Science,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The Second Law of Thermodynamics states that the state of entropy of the entire universe, as an <u>isolated system</u>, will always increase over time. The second law also states that the changes in the entropy in the universe can never be negative.

# Introduction

Why is it that when you leave an ice cube at room temperature, it begins to melt? Why do we get older and never younger? And, why is it whenever rooms are cleaned, they become messy again in the future?

Certain things happen in one direction and not the other, this is called the "arrow of time" and it encompasses every area of science. The thermodynamic arrow of time (entropy) is the measurement of disorder within a system. Denoted as  $\Delta S$ , the change of entropy suggests that time itself is asymmetric with respect to order of an isolated system, meaning: a system will become more disordered, as time increases.

# Major players in developing the Second Law

- Nicolas Léonard Sadi Carnot was a French physicist, who is considered to be the "father of thermodynamics," for he is responsible for the origins of the Second Law of Thermodynamics, as well as various other concepts. The current form of the second law uses entropy rather than caloric, which is what Sadi Carnot used to describe the law. Caloric relates to heat and Sadi Carnot came to realize that some caloric is always lost in the motion cycle. Thus, the thermodynamic reversibility concept was proven wrong, proving that irreversibility is the result of every system involving work.
- Rudolf Clausius was a German physicist, and he developed the Clausius statement, which says "Heat generally **cannot flow spontaneously** from a material at a lower temperature to a material at a higher temperature."
- William Thompson, also known as Lord Kelvin, formulated the Kelvin statement, which states "It is **impossible** to convert heat completely in a cyclic process." This means that there is no way for one to convert all the energy of a system into work, without losing energy.
- Constantin Carathéodory, a Greek mathematician, created his own statement of the second low arguing that "In the neighborhood of any initial state, there are states which **cannot** be approached arbitrarily close through adiabatic changes of state."

# Phase Rule

# Dr.M.Surendra Varma

# Assistant Professor, Department of Chemistry, School of Arts and Science,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The phase rule describes the possible number of degrees of freedom in an enclosed system at equilibrium, in terms of the number of separate phases and the number of chemical constituents in the system. It was deduced by J.W Gibbs in the 1870s. Today, the phase rule is popularly known as the Gibbs phase rule all over the world. Here, in the article, we will be discussing the derivation of the phase rule.

# **Gibbs Phase Rule**

Gibbs phase rule states that if the equilibrium in a heterogeneous system is not affected by gravity or by electrical and magnetic forces, the number of degrees of freedom is given by the equation

# **F=C-P+2**

where, C is the number of chemical components,

P is the number of phases.

Basically, it describes the mathematical relationship for determining the stability of phases present in the material at equilibrium conditions.

In the next section, let us look at the phase rule derivation.

# Phase Rule Derivation

Gibbs phase rule on the basis of the thermodynamic rule can be derived as follows:

First, let us consider a heterogeneous system consisting of  $P_n$  number of phases and  $C_n$  number of components in equilibrium. Let us assume that the passage of a component from one phase to another doesn't involve any chemical reaction. When the system is in equilibrium, it can be described by the following parameters:

- Temperature
- Pressure
- The composition of each phase

# Solid State

# Dr.R. Manikandan

# Associate Professor, Department of Chemistry, School of Arts and Science,

# Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Solid-state chemistry, also sometimes referred as materials chemistry, is the study of the synthesis, structure, and properties of solid phase materials. It therefore has a strong overlap with solidphysics, mineralogy, crystallography, ceramics, metallurgy, thermodynamics, material science and electronics with a focus on the synthesis of novel materials and their characterization.

A diverse range of synthetic techniques, such as the ceramic method and chemical vapour depostion, make solid-state materials. Solids can be classified as crystalline or amorphous on basis of the nature of order present in the arrangement of their constituent particles.<sup>[1]</sup> Their elemental compositions, microstructures, and physical properties can be characterized through a variety of analytical methods. Chemical vapour transport

Chemical vapour transport results in very pure materials. The reaction typically occurs in a sealed ampoule. A transporting agent, added to the sealed ampoule, produces a volatile intermediate species from the solid reactant.

For metal oxides, the transporting agent is usually  $Cl_2$  or HCl. The ampoule has a temperature gradient, and, as the gaseous reactant travels along the gradient, it eventually deposits as a crystal An example of an industrially-used chemical vapor transport reaction is the Mond process. The Mond process involves heating impure nickel in a stream of carbon monoxide to produce pure nickel

# Low-temperature methods

# Intercalation method

Intercalation synthesis is the insertion of molecules or ions between layers of a solid.<sup>[11]</sup> The layered solid has weak intermolecular bonds holding its layers together.<sup>[11]</sup> The process occurs via diffusion.<sup>[11]</sup> Intercalation is further driven by ion exchange, acid-base reactions or electrochemical reactions.<sup>[11]</sup> The intercalation method was first used in China with the discovery of porcelain. Also, graphene is produced by the intercalation method, and this method is the principle behind lithium-ion batteries.<sup>[12]</sup>

# Solution methods

It is possible to use solvents to prepare solids by precipitation or by evaporation.<sup>[5]</sup> At times, the solvent is a hydrothermal that is under pressure at temperatures higher than the normal boiling point. A variation on this theme is the use of flux methods, which use a salt with a relatively low melting point as the solvent

#### Physical equilibria - Pure Substance

#### Dr.R. Manikandan

#### Associate Professor, Department of Chemistry, School of Arts and Science,

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### The Basic Phase Diagram

At its simplest, a phase can be just another term for solid, liquid or gas. If you have some ice floating in water, you have a solid phase present and a liquid phase. If there is air above the mixture, then that is another phase. But the term can be used more generally than this. For example, oil floating on water also consists of two phases - in this case, two liquid phases. If the oil and water are contained in a bucket, then the solid bucket is yet another phase. In fact, there might be more than one solid phase if the handle is attached separately to the bucket rather than molded as a part of the bucket.

You can recognize the presence of the different phases because there is an obvious boundary between them - a boundary between the solid ice and the liquid water, for example, or the boundary between the two liquids. A *phase diagram* lets you work out exactly what phases are present at any given temperature and pressure. In the cases we'll be looking at on this page, the phases will simply be the solid, liquid or vapor (gas) states of a pure substance. This is the phase diagram for a typical pure substance.



# ESSENTIALS OF GENERAL CHEMISTRY : A COMPREHENSIVE GUIDE

EDITED BY

**J.S.NIMALRAM** 



Essentials of General Chemistry: A Comprehensive

GuideCopyright©2021 By J.S.NIRMALRAM

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.
Table of Contents	
Chapter 1	PgNO
Dr.J.S Nirmal ram	
Chemistry: The Study of Change	2
Chapter 2	
Dr.M.Surendra Varma	
Atoms, Molecules, and Ions	8
Chapter 3	
Dr.R.Manikandan	
Mass Relationships in Chemical Reactions.	18
Chapter 4	
Dr.D.Chinnaraja	
Reactions in Aqueous Solutions.	25
Chapter 5	
Dr.G.Anburaj	
Gases.	35
Chapter 6	
M.TAMIZHSELVAN	
Thermochemistry.	42
Chapter 7	
Dr. D. SENTHILNATHAN	
Quantum Theory and the Electronic Structure of Atoms.	45
Chapter 8	
Dr.P.CHRISTURAJ	
Periodic Relationships Among the Elements	47
Chapter 9	
Dr.G.Anburaj	
Chemical Bonding I: Basic Concepts	49
Chapter 10	
Dr.J.S Nirmal ram	
Chemical Bonding II: Molecular Geometry and Hybridization of	52
Atomic Orbitals.	55

References

Chapter 1
Chemistry: The Study of Change

Chemistry is the study of matter—what it consists of, what its properties are, and how it changes. Matter is anything that has mass and takes up space—that is, anything that is physically real. Some things are easily identified as matter—the screen on which you are reading this book, for example. Others are not so obvious. Because we move so easily through air, we sometimes forget that it, too, is matter. Because of this, chemistry is a science that has its fingers in just about everything. Even a description of the ingredients in a cake, and how those ingredients change when the cake is baked, is chemistry!

#### **Physical and Chemical Properties**

All matter has physical and chemical properties. Physical properties are characteristics that scientists can measure without changing the composition of the sample under study, such as mass, color, and volume (the amount of space occupied by a sample). Chemical properties describe the characteristic ability of a substance to react to form new substances; they include its flammability and susceptibility to corrosion. All samples of a pure substance have the same chemical and physical properties. For example, pure copper is always a reddish-brown solid (a physical property) and always dissolves in dilute nitric acid to produce a blue solution and a brown gas (a chemical property).

#### **Physical Property**

A **physical property** is a characteristic of a substance that can be observed or measured without changing the identity of the substance. Silver is a shiny metal that conducts electricity very well. It can be molded into thin sheets—a property called malleability. Salt is dull and brittle and conducts electricity when it has been dissolved into water, which it does quite easily. Physical properties of matter include color, hardness, malleability, solubility, electrical conductivity, density, melting point, and boiling point. In Table 1.4.11.4.1, notice that corn oil has a lower mass to volume ratio than water. This means that when added to water, corn oil will "float."

## Chapter 2 Atoms, Molecules, and Ions

#### Atoms, Molecules, and Ions

In this chapter, you will learn how to describe the composition of chemical compounds. We introduce *chemical nomenclature*—the language of chemistry—that will enable you to recognize and name the most common kinds of compounds. An understanding of chemical nomenclature not only is essential for your study of chemistry but also has other benefits-for example, it helps you understand the labels on products found in the supermarket and the pharmacy. You will also be better equipped to understand many of the important environmental and medical issues that face society. By the end of this chapter, you will be able to describe what happens chemically when a doctor prepares a cast to stabilize a broken bone, and you will know the composition of common substances such as laundry bleach, the active ingredient in baking powder, and the foul-smelling compound responsible for the odor of spoiled fish. Finally, you will be able to explain the chemical differences different grades of gasoline. among The Atomic Theory of Matter

This article explains the theories that Dalton used as a basis for his theory: (1) the Law of Conservation of Mass, (2) the Law of Constant Composition, (3) the Law of Multiple Proportions.

#### The Discovery of Atomic Structure

Atoms, the smallest particles of an element that exhibit the properties of that element, consist of negatively charged electrons around a central nucleus composed of more massive positively charged protons and electrically neutral neutrons. Radioactivity is the emission of energetic particles and rays (radiation) by some substances. Three important kinds of radiation are  $\alpha$  particles (helium nuclei),  $\beta$  particles (electrons traveling at high speed), and  $\gamma$  rays.

#### The Modern View of Atomic Structure

Each atom of an element contains the same number of protons, which is the atomic number (Z). Neutral atoms have the same number of electrons and protons. Atoms of an element that contain different numbers of neutrons are called isotopes. Each isotope of a given element has the same atomic number but a different mass number (A), which is the sum of the numbers of protons and neutrons. The relative masses of atoms are reported using the atomic mass unit (amu).

## Chapter 3 Mass Relationships in Chemical Reactions.

#### Atoms, Molecules, and Ions

The basic building block of all matter is the atom. Curiously, the idea of atoms was first proposed in the fifth century BCE, when the Greek philosophers Leucippus and Democritus proposed their existence in a surprisingly modern fashion. However, their ideas never took hold among their contemporaries, and it wasn't until the early 1800s that evidence amassed to make scientists reconsider the idea. Today, the concept of the atom is central to the study of matter.

#### Prelude to Atoms, Molecules, and Ions

The angstrom unit is named after Anders Jonas Ångström, a nineteenth-century Swedish physicist. Ångström's research dealt with light being emitted by glowing objects, including the sun. Ångström studied the brightness of the different colors of light that the sun emitted and was able to deduce that the sun is composed of the same kinds of matter that are present on the earth. By extension, we now know that all matter throughout the universe is similar to the matter that exists on our own planet.

#### Atomic Theory

Chemistry is based on the modern atomic theory, which states that all matter is composed of atoms. Atoms themselves are composed of protons, neutrons, and electrons. Each element has its own atomic number, which is equal to the number of protons in its nucleus. Isotopes of an element contain different numbers of neutrons. Elements are represented by an atomic symbol. The periodic table is a chart that organizes all the elements.

#### Molecules and Chemical Nomenclature

Molecules are groups of atoms that behave as a single unit. Some elements exist as molecules: hydrogen, oxygen, sulfur, and so forth. There are rules that can express a unique name for any given molecule, and a unique formula for any given name.

<u>Masses of Atoms and Molecules</u> The atomic mass unit (u) is a unit that describes the masses of individual atoms and molecules. The atomic mass is the weighted average of the masses of all isotopes of an element. The molecular mass is the sum of the masses of the atoms in a molecule.

## Chapter 4 Reactions in Aqueous Solutions.

## Reactions in Aqueous Solution

A solution is a homogeneous mixture in which substances present in lesser amounts, called solutes, are dispersed uniformly throughout the substance in the greater amount, the solvent. An aqueous solution is a solution in which the solvent is water, whereas in a nonaqueous solution, the solvent is a substance other than water. Familiar examples of nonaqueous solvents are ethyl acetate, used in nail polish removers, and turpentine, used to clean paint brushes. In this chapter, we focus on reactions that occur in aqueous solution.

There are many reasons for carrying out reactions in solution. For a chemical reaction to occur, individual atoms, molecules, or ions must collide, and collisions between two solids, which are not dispersed at the atomic, molecular, or ionic level, do not occur at a significant rate. In addition, when the amount of a substance required for a reaction is so small that it cannot be weighed accurately, using a solution of that substance, in which the solute is dispersed in a much larger mass of solvent, enables chemists to measure its quantity with great precision. Chemists can also more effectively control the amount of heat consumed or produced in a reaction when the reaction occurs in solution, and sometimes the nature of the reaction itself can be controlled by the choice of solvent.

This chapter introduces techniques for preparing and analyzing aqueous solutions, for balancing equations that describe reactions in solution, and for solving problems using solution stoichiometry. By the time you complete this chapter, you will know enough about aqueous solutions to explain what causes acid rain, why acid rain is harmful, and how a Breathalyzer measures alcohol levels. You will also understand the chemistry of photographic development, be able to explain why rhubarb leaves are toxic, and learn about a possible chemical reason for the decline and fall of the Roman Empire.

Gases.

Previously, we focused on the microscopic properties of matter—the properties of individual atoms, ions, and molecules—and how the electronic structures of atoms and ions determine the stoichiometry and three-dimensional geometry of the compounds they form. We will now focus on macroscopic properties—the behavior of aggregates with large numbers of atoms, ions, or molecules. An understanding of macroscopic properties is central to an understanding of chemistry. Why, for example, are many substances gases under normal pressures and temperatures (1.0 atm, 25°C), whereas others are liquids or solids? We will examine each form of matter—gases, liquids, and solids—as well as the nature of the forces, such as hydrogen bonding and electrostatic interactions, that hold molecular and ionic compounds together in these three states.

In this chapter, we explore the relationships among pressure, temperature, volume, and the amount of gases. You will learn how to use these relationships to describe the physical behavior of a sample of both a pure gaseous substance and mixtures of gases. By the end of this chapter, your understanding of the gas laws and the model used to explain the behavior of gases will allow you to explain how straws and hot-air balloons work, why hand pumps cannot be used in wells beyond a certain depth, why helium-filled balloons deflate so rapidly, and how a gas can be liquefied for use in preserving biological tissue. Characteristics of Gases

Bulk matter can exist in three states: gas, liquid, and solid. Gases have the lowest density of the three, are highly compressible, and fill their containers completely. Elements that exist as gases at room temperature and pressure are clustered on the right side of the periodic table; they occur as either monatomic gases (the noble gases) or diatomic molecules (some halogens,  $N_2$ ,  $O_2$ ).

Pressure

Pressure is defined as the force exerted per unit area; it can be measured using a barometer or manometer. Four quantities must be known for a complete physical description of a sample of a gas: temperature, volume, amount, and pressure.

## Chapter 6 Thermochemistry.

This chapter introduces you to **thermochemistry**, a branch of chemistry that describes the energy changes that occur during chemical reactions. In some situations, the energy produced by chemical reactions is actually of greater interest to chemists than the material products of the reaction. For example, the controlled combustion of organic molecules, primarily sugars and fats, within our cells provides the energy for physical activity, thought, and other complex chemical transformations that occur in our bodies. Similarly, our energy-intensive society extracts energy from the combustion of fossil fuels, such as coal, petroleum, and natural gas, to manufacture clothing and furniture, heat your home in winter and cool it in summer, and power the car or bus that gets you to class and to the movies. By the end of this chapter, you will know enough about thermochemistry to explain why ice cubes cool a glass of soda, how instant cold packs and hot packs work, and why swimming pools and waterbeds are heated. You will also understand what factors determine the caloric content of your diet and why even "nonpolluting" uses of fossil fuels may be affecting the environment.

- The Nature of Energy
- All forms of energy can be interconverted. Three things can change the energy of an object: the transfer of heat, work performed on or by an object, or some combination of heat and work. Thermochemistry is a branch of chemistry that qualitatively and quantitatively describes the energy changes that occur during chemical reactions. Energy is the capacity to do work.
- The First Law of Thermodynamics

The first law of thermodynamics states that the energy of the universe is constant. The change in the internal energy of a system is the sum of the heat transferred and the work done. At constant pressure, heat flow (q) and internal energy (U) are related to the system's enthalpy (H). The heat flow is equal to the change in the internal energy.

## Chapter 7 Quantum Theory and the Electronic Structure of Atoms.

#### What is Electronic Structure Theory?

Electronic Structure Theory describes the motions of electrons in atoms or molecules. Generally this is done in the context of the Born-Oppenheimer Approximation, which says that electrons are so much lighter (and therefore faster) than nuclei that they will find their optimal distribution for any given nuclear configuration. The electronic energy at each nuclear configuration is the potential energy that the nuclei feel, so solving the electronic problem for a range of nuclear configurations gives the potential energy surface.

Because the electrons are so small, one needs to use quantum mechanics to solve for their motion. Quantum mechanics tells us that the electrons will not be localized at particular points in space, but they are best thought of as ``matter waves'' which can interfere. The probability of

finding a single electron at a given point in space is given by for its wavefunction  $\Psi$  at the point x. The wavefunction can be determined by solving the time-independent Schrödinger equation  $\hat{H}\Psi = E\Psi$ . If the problem is time-dependent, then the time-dependent Schrödinger equation  $\hat{H}\frac{\partial\Psi}{\partial t} = H\Psi$  must be used instead; otherwise, the solutions to the time-independent problem are also solutions to the time-dependent problem are the energy dependent phase factor  $e^{-iEt/\hbar}$ . Since we have fixed the nuclei under the Born-Oppenheimer approximation, we solve for the nonrelativistic *electronic* Schrödinger equation:

$$\hat{H} = -\frac{\hbar^2}{2m} \sum_i \nabla_i^2 - \sum_A \frac{\hbar^2}{2M_A} \nabla_A^2 - \sum_{A,i} \frac{Z_A e^2}{4\pi\epsilon_0 r_{Ai}} + \sum_{A>B} \frac{Z_A Z_B e^2}{4\pi\epsilon_0 R_{AB}} + \sum_{i>j} \frac{e^2}{4\pi\epsilon_0 r_{ij}}, \quad (1)$$

Chapter 8
Periodic Relationships Among the Elements

The elements in the periodic table are arranged in order of increasing atomic number. All of these elements display several other trends and we can use the periodic law and table formation to predict their chemical, physical, and atomic properties. Understanding these trends is done by analyzing the elements electron configuration; all elements prefer an octet formation and will gain or lose electrons to form that stable configuration.

#### **Atomic Radius**

We can never determine the atomic radius of an atom because there is never a zero probability of finding an electron, and thus never a distinct boundary to the atom. All that we can measure is the distance between two nuclei (internuclear distance). A covalent radius is one-half the distance between the nuclei of two identical atoms. An ionic radius is one-half the distance between the nuclei of two ions in an ionic bond. The distance must be apportioned for the smaller cation and larger anion. A metallic radius is one-half the distance between the nuclei of two adjacent atoms in a crystalline structure. The noble gases are left out of the trends in atomic radii because there is great debate over the experimental values of their atomic radii. The SI units for measuring atomic radii are the nanometer (nm) and the picometer (pm).  $1nm=1\times10-9m1nm=1\times10-9m$  and  $1pm=1\times10-12m1pm=1\times10-12m$ .

To explain this trend, the concept of screening and penetration must be understood. Penetration is commonly known as the distance that an electron is from the nucleus. Screening is defined as the concept of the inner electrons blocking the outer electrons from the nuclear charge. Within this concept we assume that there is no screening between the outer electrons and that the inner electrons shield the outer electrons from the total positive charge of the nucleus. In order to comprehend the extent of screening and penetration within an atom, scientists came up with the **effective nuclear charge**, ZeffZeff. The equation for calculating the effective nuclear charge is shown below.

Zeff=Z-SZeff=Z-S

## Chapter 9 Chemical Bonding I: Basic Concepts...

A chemical bond is an attraction between atoms that allows the formation of chemical substances that contain two or more atoms. The bond is caused by the electrostatic force of attraction between opposite charges, either between electrons and nuclei, or as the result of a dipole attraction. The strength of chemical bonds varies considerably.

#### **Bonding Overview**

Why are some substances chemically bonded molecules and others are an association of ions? The answer to this question depends upon the electronic structures of the atoms and nature of the chemical forces within the compounds. Although there are no sharply defined boundaries, chemical bonds are typically classified into three main types: ionic bonds, covalent bonds, and metallic bonds. In this chapter, each type of bond wil be discussed and the general properties found in typical substances in which the bond type occurs

- Ionic bonds results from *electrostatic forces that exist between ions of opposite charge*. These bonds typically involves a metal with a nonmetal
- 2. Covalent bonds *result from the sharing of electrons between two atoms*. The bonds typically involves one nonmetallic element with another
- 3. Metallic bonds These bonds are found in solid metals (copper, iron, aluminum) with each metal bonded to several neighboring groups and bonding electrons free to move throughout the 3-dimensional structure.

Each bond classification is discussed in detail in subsequent sections of the chapter. Let's look at the preferred arrangements of electrons in atoms when they form chemical compounds.

## Lewis Symbols

At the beginning of the 20th century, the American chemist G. N. Lewis (1875–1946) devised a system of symbols—now called Lewis electron dot symbols, often shortened to *Lewis dot symbols*—that can be used for predicting the number of bonds formed by most elements in their compounds. Each Lewis dot symbol consists of the chemical symbol for an element surrounded by dots that represent its valence electrons.

## Chemical Bonding II: Molecular Geometry and Hybridization of Atomic Orbitals.

The localized **valence bond theory** uses a process called **hybridization**, in which atomic orbitals that are similar in energy but not equivalent are combined mathematically to produce sets of equivalent orbitals that are properly oriented to form bonds. These new combinations are called

hybrid atomic orbitals

because they are produced by combining (*hybridizing*) two or more atomic orbitals from the same atom.

#### Hybridization of *s* and *p* Orbitals

In BeH<sub>2</sub>, we can generate two equivalent orbitals by combining the 2*s* orbital of beryllium and any one of the three degenerate

2p orbitals. By taking the sum and the difference of Be 2s and  $2p_z$  atomic orbitals, for example, we produce two new orbitals with major and minor lobes oriented along the *z*-axes, as shown in Figure 10.7.110.7.1.



Figure 10.7.110.7.1: The position of the atomic nucleus with respect to an sp hybrid orbital. The nucleus is actually located slightly inside the minor lobe, not at the node separating the major and minor lobes.Because the difference A – B can also be written as A + (–B), in Figure 10.7.210.7.2 and subsequent figures we have reversed the phase(s) of the orbital being subtracted, which is the same as multiplying it by -1 and adding. This gives us



# CHEMISTRY BEHIND PESTICIDES: EXPLORING THE SCIENCE FOR EFFECTIVE PEST CONTROL

EDITED BY

**J.S.NIRMAL RAM** 



Chemistry behind Pesticides: Exploring the Science for Effective Pest Control Copyright©2021

By

J.S.NIRMAL RAM

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-478-9 Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

Table of Contents	Pg NO
Chapter 1	
Dr.J.S Nirmal ram	
Anti-insect agents.	2
Chapter 2	
Dr.M.Surendra Varma	
Acaricides (Cy. Matolcsy).	8
Chapter 3	
Dr.R.Manikandan	
Nematocides	18
Chapter 4	
Dr.D.Chinnaraja	
Rodenticides.	25
Chapter 5	
Dr.G.Anburaj	
Fungicides.	35
Chapter 6	
M.TAMIZHSELVAN	
Herbicides	42
Chapter 7	
Dr. D. SENTHILNATHAN	
Urea derivatives containing a heterocyclic group	53
Chapter 8	
Dr.P.CHRISTURAJ	
Thiourea derivatives.	54
Chapter 9	
Dr.G.Anburaj	
Nitriles	55
Chapter 10	
Dr.J.S Nirmal ram	
N-Substituted a-chloroacetamides	57
References	

	Chapter 1	
	Anti-insect agents.	
2		

## Common types of insecticides

Insecticides are of chemical and biological origins and are used in agriculture, forestry, gardens, homes, and offices. They are also used to control vectors, such as mosquitoes and ticks, that are involved in spreading human and animal diseases

The synthetic contact insecticides are now the primary agents of insect control. In general, they penetrate insects readily and are toxic to a wide range of species. The main synthetic groups are the chlorinated hydrocarbons, organic phosphates (organophosphates), carbamates, pyrethroids, neonicotinoids.

## 1. Chlorinated hydrocarbons

They were commonly used earlier, but now many countries have been removed Organochlorine insecticides from their market due to their health and environmental effects and their persistence. Representative members of this insecticide class include DDT, methoxychlor, aldrin, dieldrin, heptachlor, chlordane, lindane, BHC, and chlorobenzilate

## Advantages

• The chlorinated hydrocarbons are very potent nerve toxins to insects, and their initial use led to significant improvements in insect control.

## Disadvantages

• These compounds have a long residual action in the environment so they have been banned in most countries in the past 30 years

## 2. Organophosphates

Two widely used compounds in this class are parathion and malathion; others are fenthion, fenitrothion, diazinon, naled, methyl parathion, and dichlorvos

## Advantages

• They are especially effective against sucking insects such as aphids and mites, which feed on plant juices

• They usually have little residual action in the environment

#### Chapter 2

#### Acaricides

More specific words are sometimes used, depending upon the targeted group:

- "Ixodicides" are substances that kill ticks.
- "Miticides" are substances that kill mites.
- The term scabicide is more narrow, and refers to agents specifically targeting *Sarcoptes*.
- The term "arachnicide" is more general, and refers to agents that target arachnids. This term is used much more rarely, but occasionally appears in informal writing.

As a practical matter, mites are a paraphyletic grouping, and mites and ticks are usually treated as a single group.

- Lime sulfur is effective against sarcoptic mange. It is made by mixing hydrated lime, sulfur, and water, and boiling for about 1 hour. Hydrated lime can bond with about 1.7 times its weight of sulfur (quicklime can bond with as much as 2.2 times its weight of sulfur). The strongest concentrate is diluted 1:32 before saturating the skin (avoiding the eyes), applied at six-day intervals.
- Nonpesticide miticides act by causing desiccation, but are not a diatomaceous earth (which contain crystalline silica, potentially dangerous by inhalation), but made from a patented mix of food-grade components, one to breach the cuticle and one to ensure rapid, reliable desiccation. They can be dusted as powder or sprayed in aqueous solution.
- A variety of commercially available systemic and non-systemic miticides: abamectin, acequinocyl, bifenazate, chlorfenapyr, clofentezine, cyflumetofen, cyp ermethrin, dicofol, etoxazole, fenazaquin, fenpyroximate, hexythiazox, imidacloprid, proparg ite, pyridaben, spiromesifen, spirotetramat.

Acaricides are also being used in attempts to stop rhinoceros poaching. Holes are drilled into the horn of a sedated rhino and acaricide is pumped in and pressurized. Should the horn be consumed by humans as in traditional Chinese medicine, it is expected to cause nausea, stomachache, and diarrhea, or convulsions, depending on the quantity, but not fatalities. Signs posted at

#### Nematocides

Important Considerations:

- *costs and economics*; use restricted primarily to higher value crops in cash economy.
- *phytotoxicity*; may only be available for preplant use.
- lack of specificity; disruption of the soil food web, including antagonists, competitors and beneficials.
- *residue problems*; soil, groundwater, food Prop 65 and other legislation.
- *non-visible target*; volumes of soil are treated rather than foci of infestation; saturation rather than precision application.
- *movement and persistence*; chemical may have to permeate the soil from the site of application in order to contact target nematodes, and be sufficiently persistent to achieve lethal concentrations at each site.
- *dissipation*; after lethal dose has been delivered, chemical should break down or otherwise dissipate in the environment to allow plant growth and soil food web recovery.
- *amounts and technology* for application; volume of soil to be treated requires transportation, handling and application of large volumes of toxic materials. Older equipment introduced fumigants to appropriate soil depths but was not well-engineered for ensuring transfer to and from the delivery tank without escape to the atmosphere or human exposure.
- human toxicity; health hazards and risks associated with volatility, manufacture, handling, application and accidents; materials may have high <u>mammalian toxicity</u> and/or be suspected carcinogens..

There are two main categories of chemical nematicides based on their volatility - non-fumigants and fumigants.

- Fumigant nematicides disperse through the soil as a result of their volatility.
- Non-fumigant nematicides must be mechanically mixed throughout the soil profile, or spread in aqueous solution.

Chapter 4	
Rodenticides.	

What are rodenticides?

Rodenticides are poisons that are manufactured into bait and used to control rodents, such as rats and mice. There are several types of rodenticides approved by the Australian Pesticides and Veterinary Medicines Authority (APVMA), including anticoagulant rodenticides and the nonanticoagulant rodenticides zinc phosphide, cholecalciferol, bromethalin and strychnine.

Rodenticide products currently registered by the APVMA have varied uses approved based on the type of poison (i.e. active constituent) they contain and the specific risks these poisons present to humans and other non-target organisms. These uses may include in and around domestic, commercial, industrial and agricultural buildings and/or in crops. Label directions for rodenticide products can be accessed on our <u>PubCRIS database</u>, and include restraints to control how the products are used and mitigate the risks.

#### What is an anticoagulant rodenticide?

Anticoagulant rodenticides are a class of poisons used to kill rodents. In simple terms, anticoagulant rodenticides work by preventing blood from clotting, which results in death in rodents from internal or external bleeding typically 5 to 10 days after a lethal dose has been consumed.

Anticoagulant rodenticide baits are formulated in various ways, including blocks, pellets and treated grains, gels, powders and pastes, and are classified into 2 groups: first-generation and second-generation.

#### First-generation anticoagulant rodenticides

First-generation anticoagulant rodenticides (FGARs) are referred to as 'multi-dose anticoagulants', meaning that rodents must consume these baits for several consecutive feedings to consume a lethal dose. FGARs break down in rodents quicker than second-generation anticoagulant rodenticides, so there is less chance of secondary poisoning occurring in non-target animals if they eat rodents poisoned with an FGAR.

Chapter 5	
Fungicides.	

**Fungicides** are <u>pesticides</u> used to kill <u>parasitic fungi</u> or their <u>spores</u>.<sup>[1]</sup> Fungi can cause serious damage in <u>agriculture</u>, resulting in critical <u>losses of yield</u>, quality, and <u>profit</u>. Fungicides are used both in agriculture and to fight <u>fungal infections in animals</u>. Fungicides are also used to control <u>oomycetes</u>, which are not <u>taxonomically</u>/genetically fungi, although sharing similar methods of infecting plants. Fungicides can either be contact, translaminar or systemic. Contact fungicides are not taken up into the plant tissue and protect only the plant where the spray is deposited. Translaminar fungicides redistribute the fungicide from the upper, sprayed leaf surface to the lower, unsprayed surface. Systemic fungicides are taken up and redistributed through the xylem vessels. Few fungicides move to all parts of a plant. Some are locally systemic, and some move upward.<sup>[2][3]</sup> Most fungicides that can be bought retail are sold in liquid form, the active ingredient being present at 0.08% in weaker concentrates, and as high as 0.5% for more potent fungicides. Fungicides in powdered form are usually around 90% sulfur.

Pesticides are substances that are used to control pests. They include herbicides, insecticides, nematicides, fungicides, and many others. The most common of these are herbicides, which account for approximately 50% of all pesticide use globally. Most pesticides are used as plant protection products. Pesticide application refers to the practical way in which pesticides are delivered to their biological targets. Public concern about the use of pesticides has highlighted the need to make this process as efficient as possible, in order to minimise their release into the environment and human.

From biblical times until the 1950s the pesticides used were <u>inorganic</u> compounds and <u>plant</u> <u>extracts</u>.<sup>[14][15]</sup> The inorganic compounds were derivatives of <u>copper</u>, <u>arsenic</u>, mercury, <u>sulfur</u>, among others, and the plant extracts contained <u>pyrethrum</u>, <u>nicotine</u>, and <u>rotenone</u> among others. The less toxic of these are still in use in <u>organic farming</u>. In the 1940s the insecticide <u>DDT</u>, and the herbicide <u>2,4-D</u>, were introduced. These synthetic organic (i.e. non inorganic) compounds were widely used and were very profitable. They were followed in the 1950s and 1960s by numerous other synthetic pesticides, which led to the growth of the pesticide industry.

Chapter 6
Herbicides

The major breakthroughs occurred during the Second World War as the result of research conducted independently in the United Kingdom and the United States into the potential use of herbicides in war.<sup>[10]</sup> The compound 2,4-D was first synthesized by W. G. Templeman at Imperial Chemical Industries. In 1940, his work with indoleacetic acid and naphthaleneacetic acid indicated that "growth substances applied appropriately would kill certain broad-leaved weeds in cereals without harming the crops, though these substances were too expensive and too short-lived in soil due to degradation by microorganisms to be of practical agricultural use; by 1941, his team succeeded in synthesizing a wide range of chemicals to achieve the same effect at lower cost and better efficacy, including 2,4-D.<sup>[13]</sup> In the same year, R. Pokorny in the US achieved this as well. ndependently, a team under Juda Hirsch Quastel, working at the Rothamsted Experimental Station made the same discovery. Quastel was tasked by the Agricultural Research Council (ARC) to discover methods for improving crop yield. By analyzing soil as a dynamic system, rather than an inert substance, he was able to apply techniques such as perfusion. Quastel was able to quantify the influence of various plant hormones, inhibitors, and other chemicals on the activity of microorganisms in the soil and assess their direct impact on plant growth. While the full work of the unit remained secret, certain discoveries were developed for commercial use after the war, including the 2,4-D compound When 2,4-D was commercially released in 1946, it became the first successful selective herbicide, triggering a worldwide revolution in agricultural output. It allowed for greatly enhanced weed control in wheat, maize (corn), rice, and similar cereal grass crops, because it kills dicots (broadleaf plants), but not most monocots (grasses). The low cost of 2,4-D has led to continued usage today, and it remains one of the most commonly used herbicides in the world. Like other acid herbicides, current formulations use either an amine salt (often trimethylamine) or one of many esters of the parent compound.

#### Further discoveries

The triazine family of herbicides, which includes <u>atrazine</u>, was introduced in the 1950s; they have the current distinction of being the herbicide family of greatest concern regarding <u>groundwater</u> <u>contamination</u>. Atrazine does not break down readily (within a few weeks) after being applied to soils of above-neutral <u>pH</u>. Under alkaline soil conditions, atrazine may be carried into the soil profile as far as the water table by soil water following rainfall causing the aforementioned .

## Chapter 7 Urea derivatives containing a heterocyclic group

A heterocyclic compound or ring structure is a cyclic compound that has atoms of at least two different <u>elements</u> as members of its ring(s) Heterocyclic organic chemistry is the branch of <u>organic chemistry</u> dealing with the synthesis, properties, and applications of organic heterocycles Examples of heterocyclic compounds include all of the <u>nucleic acids</u>, the majority of drugs, most <u>biomass</u> (cellulose and related materials), and many natural and synthetic dyes. More than half of known compounds are heterocycles. 59% of US <u>FDA</u>-approved drugs contain <u>nitrogen</u> heterocycles.

he study of organic heterocyclic chemistry focuses especially on organic unsaturated derivatives, and the preponderance of work and applications involves unstrained organic 5- and 6-membered rings. Included are <u>pyridine</u>, <u>thiophene</u>, <u>pyrrole</u>, and <u>furan</u>. Another large class of organic heterocycles refers to those fused to <u>benzene rings</u>. For example, the fused benzene derivatives of pyridine, thiophene, pyrrole, and furan are <u>quinoline</u>, <u>benzothiophene</u>, <u>indole</u>, and <u>benzofuran</u>, respectively. The fusion of two benzene rings gives rise to a third large family of organic compounds. Analogs of the previously mentioned heterocycles for this third family of compounds are <u>acridine</u>, <u>dibenzothiophene</u>, <u>carbazole</u>, and <u>dibenzofuran</u>, respectively.

Heterocyclic organic compounds can be usefully classified based on their electronic structure. The like saturated organic heterocycles behave the acyclic derivatives. Thus, piperidine and tetrahydrofuran are conventional amines and ethers, with modified steric profiles. Therefore, the study of organic heterocyclic chemistry focuses on organic unsaturated rings. Some heterocycles contain no carbon. Examples are borazine ( $B_3N_3$  ring), hexachlorophosphazenes ( $P_3N_3$  rings), and tetrasulfur tetranitride S<sub>4</sub>N<sub>4</sub>. In comparison with organic heterocycles, which have numerous commercial applications, inorganic ring systems mainly theoretical are of interest. IUPAC recommends the Hantzsch-Widman nomenclature for naming heterocyclic compounds.

"Heteroatoms" are atoms in the ring other than carbon atoms.

Chapter 8														
Thiourea derivatives.														
			0		a a a				$\alpha$		- ·			

<u>compound</u> with the <u>formula</u>  $SC(NH_2)_2$  and the <u>structure</u>  $H_2N-C(=S)-NH_2$ . It is structurally similar to <u>urea</u> ( $H_2N-C(=O)-NH_2$ ), except that the <u>oxygen</u> atom is replaced by a <u>sulfur</u> atom (as implied by the <u>thio-</u> prefix); however, the properties of urea and thiourea differ significantly. Thiourea is a <u>reagent</u> in <u>organic synthesis</u>. <u>Thioureas</u> are a broad class of compounds with the general structure  $R_2N-C(=S)-NR_2$ 

hiourea is a planar molecule. The C=S bond distance is 1.71 Å. The C-N distances average 1.33 Å The weakening of the C-S bond by C-N pi-bonding is indicated by the short C=S bond in <u>thiobenzophenone</u>, which is 1.63 Å. hiourea occurs in two tautomeric forms, of which the thione form predominates in aqueous solutions. The equilibrium constant has been calculated as  $K_{eq}$  is  $1.04 \times 10^{-3}$ .<sup>[6]</sup> The thiol form, which is also known as an isothiourea, can be encountered in substituted compounds such as isothiouronium salts.



he global annual production of thiourea is around 10,000 tonnes. About 40% is produced in Germany, another 40% in China, and 20% in Japan. Thiourea can be produced from ammonium thiocyanate, but more commonly it is manufactured by the reaction of hydrogen sulfide with calcium cyanamide in the presence of carbon dioxide.<sup>[7]</sup>

$$\begin{split} & CaCN_2+3 \ H_2S \rightarrow Ca(SH)_2+(NH_2)_2CS \\ & 2 \ CaCN_2+Ca(SH)_2+6 \ H_2O \rightarrow 2 \ (NH_2)_2CS+3 \ Ca(OH)_2 \\ & Ca(OH)_2+CO_2 \rightarrow CaCO_3+H_2O \end{split}$$

#### Thiox precursor

Thiourea *per se* has few applications. It is mainly consumed as a precursor to <u>thiourea dioxide</u>, which is a common reducing agent in textile processing. Ertilizers

Recently thiourea has been investigated for its multiple desirable properties as a <u>fertilizer</u> especially under conditions of environmental stress.

Chapter 9
Nitriles

In <u>organic chemistry</u>, a **nitrile** is any <u>organic compound</u> that has a  $-\underline{C} \equiv \underline{N}$  functional group. The name of the compound is composed of a base, which includes the carbon of the  $-\underline{C} \equiv N$ , suffixed with "nitrile", so for example  $CH_3CH_2C\equiv N$  is called "<u>propionitrile</u>" (or propanenitrile The prefix <u>cyano</u>- is used interchangeably with the term <u>nitrile</u> in industrial literature. Nitriles are found in many useful compounds, including <u>methyl cyanoacrylate</u>, used in <u>super glue</u>, and <u>nitrile</u> rubber, a nitrile-containing <u>polymer</u> used in <u>latex-free</u> laboratory and <u>medical gloves</u>. Nitrile rubber is also widely used as automotive and other seals since it is resistant to fuels and oils. Organic compounds containing multiple nitrile groups are known as <u>cyanocarbons</u>.

#### Structure and basic properties

The N–C–C geometry is linear in nitriles, reflecting the sp hybridization of the triply bonded carbon. The C–N distance is short at 1.16  $\underline{\text{Å}}$ , consistent with a <u>triple bond</u>.<sup>[3]</sup> Nitriles are polar, as indicated by high dipole moments. As liquids, they have high <u>relative permittivities</u>, often in the 30s.

The first compound of the homolog row of nitriles, the nitrile of <u>formic acid</u>, <u>hydrogen</u> <u>cyanide</u> was first synthesized by <u>C. W. Scheele</u> in 1782 In 1811 <u>J. L. Gay-Lussac</u> was able to prepare the very toxic and volatile pure acid. Around 1832 <u>benzonitrile</u>, the nitrile of <u>benzoic acid</u>, was prepared by <u>Friedrich Wöhler</u> and <u>Justus von Liebig</u>, but due to minimal yield of the synthesis neither physical nor chemical properties were determined nor a structure suggested. In 1834 <u>Théophile-Jules Pelouze</u> synthesized <u>propionitrile</u>, suggesting it to be an ether of propionic alcohol and hydrocyanic acid The synthesis of benzonitrile by <u>Hermann Fehling</u> in 1844 by heating ammonium benzoate was the first method yielding enough of the substance for chemical research. Fehling determined the structure by comparing his results to the already known synthesis of hydrogen cyanide by heating ammonium <u>formate</u>. He coined the name "nitrile" for the newfound substance, which became the name for this group of compounds.

## Chapter 10 N-Substituted a-chloroacetamides

The word "amine" is derived from *ammonia*, and the class of compounds known as amines therefore are commonly named as substituted ammonias. In this system, **primary** amines (RNH2RNH2), having only one substituent on nitrogen, are named with the substituent as a prefix. More systematic nomenclature appends *-amine* to the longest chain, as for alcohols:



**Secondary** (R2NHR2NH) and **tertiary** amines (R3NR3N), which have two and three substituents on nitrogen, commonly are named as *N*-substituted amines. As for substituted amides, *N* is included to indicate that the substituent is on the nitrogen atom unless there is no ambiguity as to where the substituent is located. Systematic nomenclature of secondary and tertiary amines is related to the systematic ether nomenclature discussed in <u>Section 7-3</u>:



As a substituent, the –NH2–NH2 group is called *amino*. N-Substituted amino groups are named accordingly:

11Alphabetical order puts chloro- ahead of chloroethyl-.



# ESSENTIAL GUIDE TO QUALITATIVE INORGANIC ANALYSIS IN THE LABORATORY

EDITED BY

**V.ABARNA** 



Essential Guide to Qualitative Inorganic Analysis in the Laboratory Copyright©2021

By

V.Abarna

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN Number 978-93-6255-199-3

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## **Table of Contents**

## **Chapter 1**

## Ms. V.Abarna

## Chapter 2

## Dr.M.Surendra Varma

Equipmen	t Check-List	og.	8
----------	--------------	-----	---

## Chapter 3

## Dr.R.Manikandan

Introduction to Inorganic Chemistry Lab ..... pg. 18

## **Chapter 4**

## Dr.D.Chinnaraja

Methods of separations .		pg. 25
--------------------------	--	--------

## Chapter 5

## Dr.D.Chinnaraja

## Chapter 6

## Dr.G.Anburaj

Analytical Techniques		pg. 42
-----------------------	--	--------

## Dr.G.Anburaj

References

#### **Introduction to Inorganic Chemistry Lab**

#### Ms.V.Abarna

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Safety Rules

The chemistry laboratory is not a dangerous place to work as long as all necessary precautions are taken seriously. In the following paragraphs, those important precautions are described. Everyone who works and performs experiments in a laboratory must follow these safety rules at all times. Students who do not obey the safety rules will not be allowed to enter and do any type of work in the laboratory and they will be counted as absent. It is the student's responsibility to read carefully all the safety rules before the first meeting of the lab.

#### Eye Protection:

Because the eyes are particularly susceptible to permanent damage by corrosive chemicals as well as flying objects, safety goggles must be worn at all times in the laboratory. Prescription glasses are not recommended since they do not provide a proper side protection. No sunglasses are allowed in the laboratory. Contact lenses have potential hazard because the chemical vapors dissolve in the liquids covering the eye and concentrate behind the lenses. If you have to wear contact lenses consult with your instructor.

If possible try to wear a prescription glasses under your safety goggles. In case of any accident that a chemical splashes near your eyes, immediately wash your eyes with lots of water and inform your instructor. Especially, when heating a test tube do not point its mouth to anyone.

## **Equipment Check-List**

## Dr.M.Surendra Varma

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

In chemistry, there is an abundance of laboratory equipment, all of which are used for slightly different jobs.But what equipment do you actually need? If you're planning a basic laboratory setup, or something a little more complex, then there are a few things you should definitely check off your list.

This blog post will highlight those items, from safety goggles to crucibles, here's our list of basic laboratory equipment and their uses.

#### A list of basic laboratory equipment used in chemistry

As previously mentioned, there are a few fundamental pieces of laboratory equipment used in chemistry that you should invest in, including:

- Safety goggles, apron, and other safety gear
- Beakers and test tubes
- Test tube racks
- Pipettes
- Bunsen burners
- Thermometer
- Crucibles
- Funnels
- Tongs
- Spatulas
- Boiling flasks

We will now explain a little about each below, helping you understand their uses.

#### **Introduction to Inorganic Chemistry Lab**

#### Dr.R.Manikandan

Associate Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction to the Organic Laboratory

The Organic Chemistry Laboratory is the most exciting part of the curriculum. It is the place where all of the abstract chemical concepts come to life. Although it is an exciting place it can also be one of the most dangerous environments that you have ever learned in. For that reason it is important to understand the nature of the organic compounds that you will encounter in the laboratory prior to implementing any lab experiment.

- I. Safety
- A. Accidents

On the first day of class, you should familiarize yourself with the locations of theall the safety equipment in the laboratory. The safety equipment that should bevisible are the following:

1. Baking Soda

Baking soda is located at the entrance of the lab and used in the laboratoryto neutralize acid spills. If an acid spills at your station or at any location in the lab it is imperative that you alert your lab instructor and immediately try to neutralize the spill. Accidental acid spills on your skin can be neutralized with the base and then washed with lots of water for 5-10 minutes. Basespills can also be flushed with water adding a base is not necessary.

2. Fire extinguisher

A fire extinguisher is used to control small fires. The extinguishers usecompressed carbon dioxide that is portable and quickly available for emergencies.

## Methods of separations

## Dr.D.Chinnaraja

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Methods of Separating Mixtures

Some of the common methods of separating substances or mixtures are:

- Handpicking
- Threshing
- Winnowing
- Sieving
- Evaporation
- Distillation
- Filtration or Sedimentation
- Separating Funnel
- Magnetic Separation



## Separating Mixtures - Sublimation

## **Group Separations**

### Dr.D.Chinnaraja

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Qualitative analysis is the systematic approach that involves precipitation reaction to remove cations sequentially from a mixture. The behaviour of the cations toward a set of common test reagents differs from one cation to another and furnishes the basis for their separation.

The qualitative analysis of an inorganic mixture is started by first carrying out some preliminary tests. The Preliminary tests for cations are

- 1. Physical examination
- 2. Charcoal cavity test
- 3. Boraxbead test
- 4. Flame test

These tests do not give conclusive evidence yet they provide some information about the ions present in the mixture.

Qualitative analysis of cations usually consists of three stages.

- 1. *First* based on different solubility properties the cations are separated into 5 groups through the successive addition of selective precipitating reagents.
- 2. *Second*, within each group precipitated cations are separated through selective dissolution processes.
- 3. *Third*, the presence of each cation is verified through different identification tests.

The cations are classified into the following 5 groups.

## Analytical Techniques

## Dr.G.Anburaj

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Analytical chemistry is the branch of chemistry that deals with the analysis of different substances.

Table of Content

- Methods Used in Analytical Chemistry
- Branches of Analytical Chemistry
- Importance of Analytical Chemistry
- Applications of Analytical Chemistry
- FAQs

Analytical chemistry involves the *separation, identification, and the quantification of matter*. It involves the use of classical methods along with modern methods involving the use of scientific instruments.

Analytical chemistry involves the following methods:

- The *process of separation* isolates the required chemical species which is to be analysed from a mixture.
- The *identification of the analyte substance* is achieved via the method of qualitative analysis.
- The *concentration of the analyte* in a given mixture can be determined with the method of quantitative analysis.

Today, the field of analytical chemistry generally involves the use of modern, sophisticated instruments. However, the principles upon which these instruments are built can be traced to more traditional techniques.

### **Coordination Complex**

#### Dr.G.Anburaj

Assistant Professor, Department of Chemistry, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Many coordination compounds contain a metallic element as the central atom and are therefore referred to as metal complexes. These types of coordination complexes generally consist of a transition element as the central atom. It can be noted that the central atom in these complexes is called the coordination centre.

Important Terms Involving Coordination Compounds

The definitions of some important terms in the chemistry of coordination compounds can be found below.

**Coordination Entity** 

A chemical compound in which the central ion or atom (or the coordination centre) is bound to a set number of atoms, molecules, or ions is called a coordination entity.

Some examples of such coordination entities include [CoCl<sub>3</sub>(NH<sub>3</sub>)<sub>3</sub>] and [Fe(CN)<sub>6</sub>]<sup>4-</sup>.

Central Atoms and Central Ions

As discussed earlier, the atoms and ions to which a set number of atoms, molecules, or ions are bound are referred to as the central atoms and the central ions.

In coordination compounds, the central atoms or ions are typically Lewis Acids and can, therefore, act as electron-pair acceptors.



# **ETHICS IN BUSINESS**

## EDITED BY K.SATHYA


#### ETHICS IN BUSINESS Copyright © 2021 by K.SATHYA

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

#### ISBN: 978-93-6255-048-4

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### TABLE OF CONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	INTRODUCTION TO		1
	BUSINESS ETHICS	DR.R.SELVARAJ	
2	CULTURAL AND HUMAN		11
	VALUES IN MANAGEMENT	DR. S. KAMARAJU	
3	CONSEQUENTIAL AND NON-		22
	<b>CONSEQUENTIAL THEORIES</b>	DR.G.KARTHIGA	
4	ETHICAL DECISION		32
	MAKING	DR.V.SRIDEVI	
5	ENVIRONMENT ISSUES	DR.R.RAJAVARDHINI	42
6	WORKPLACE ETHICS	DR.D.SILAMBARASAN	52
7	ORGANISATION ETHICS		62
	DEVELOPMENT SYSTEM	DR.S.RAJENDRAN	
8	MARKETING ETHICS AND		72
	CONSUMER PROTECTION	DR.R.SELVARAJ	
9	CORPORATE SOCIAL		82
	RESPONSIBILITY	DR. S. KAMARAJU	
10	CORPORATE GOVERNANCE	DR.G.KARTHIGA	92
	REFERENCE		101

#### CHAPTER 1 INTRODUCTION TO BUSINESS ETHICS DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **1.1 INTRODUCTION TO BUSINESS ETHICS**

A business is any organisation or entity whose primary objective is to provide goods or services for profit. Otherwise, Ethics is the study of morality. In short, the moral status of the sorts of decisions that are made in the context of buying and selling goods and services can be assessed as narrated below.

For any decision that affects some individual in a positive or a negative way is a moral decision—and decisions made by businesses are among those that affect individuals the most! That being said, anyone who owns, or runs, or works for a business—in short, almost all of us!— needs to be in the position to know which decisions are the morally acceptable, and which are the morally not acceptable.

So, business ethics is the applied ethics discipline that addresses the moral features of commercial activity. In practice, however, a dizzying array of projects is pursued under its rubric. Programs of legal compliance, empirical studies into the moral beliefs and attitudes of business people, a panoply of best-practices claims (in the name of their moral merit or their contribution to business success), arguments for (or against) mandatory worker participation in management, and attempts in applying traditional ethical theories, theories of justice, or theories applicable to firms or to the functional areas of business are all advanced as contributions to business ethics and especially in its academic literature.

These projects vary considerably and often seem to have little in common other than the conviction, held by those who pursue them, that whatever each is pursuing *is* business ethics.

Therefore,

- Business ethics is a branch of ethics which prescribes standards regarding how the business is to be carried out.
- Business ethics guidelines to stakeholders.
- Business ethics is the responsibility of the managers and employees.
- Business ethics is the application of ethical judgments to business activities

#### **1.2 MEANING OF ETHICS**

It is refers not to the disinterested or unconcerned observer, or a passive or unaffected and reluctant individual, or a person who is weak – willed, biased, self centered or irrational, or even the person who is not placed or positioned in the critical place of action. It refers necessarily to the rational and responsible person who finds himself obliged and constrained to act by virtue of the position or the office he occupies and the function he is called upon discharge.

#### CHAPTER 2 CULTURAL AND HUMAN VALUES IN MANAGEMENT DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INDIAN ETHOS IN ETHICS, MORALITY AND CULTURE

Indian ethic – moral rules and conduct of life go back to several millennia, when the Vedas prescribed the ground rules of human existence and living. A part from what has been stated above, the unique feature of Hindu religion and Indian ethos is its universality, which cannot be had in any other religion and culture.

This is the universal prayer of Vedas, transcending the frontiers, nations, religious and also people. This is very in several respects exhibiting 'Universal Love' of humankind. Hence, this is singularly different from other religions of the world. It is only through Vedas and Upanishads, Hindu religion has been elevated as the spiritual religion of the world.

The ethical discussions and teachings in India continued all through Indian history, though was ruled by different emperors and foreign rulers. The Upanishads, Puranas and Smrithis continued the tradition. The values enshrined in these divine dicta were put forward for popular use in great epics of Mahabaratha and Ramayana. This is the only literature in the world which has been devoted exclusively to spiritual quest. The two epics of India gave human dilemmas in every walk of life and attach importance to values in dealing all such cases.

#### INDIAN AND GLOBAL PERSPECTIVE

The following list of principles incorporates the characteristics and values that most people associate with ethical behavior. Ethical decision making systematically considers these principles:

(i) Honesty:

Ethical executives are honest and truthful in all their dealings and they do not deliberately mislead or deceive others by misrepresentations, overstatements, partial truths, selective omissions, or any other means.

(ii) Integrity:

Ethical executives demonstrate personal integrity and the courage of their convictions by doing what they think is right even when there is great pressure to do otherwise; they are principled, honorable and upright; they will fight for their beliefs. They will not sacrifice principle for expediency, be hypocritical, or unscrupulous.

(iii) Promise-Keeping & Trust worthiness

Ethical executives are worthy of trust. They are candid and forth coming in supplying relevant information and correcting misapprehensions of fact, and they make every reasonable effort to fulfill the letter and spirit of their promises and commitments. They do not interpret agreements in an unreasonably technical or legalistic manner in order to rationalise non-compliance or create justifications for escaping their commitments.

(iv) Loyalty:

Ethical executives are worthy of trust, demonstrate fidelity and loyalty to persons and institution charging of interest at any rate is by friendship in adversity, support and devotion to duty; they

#### CHAPTER 3 CONSEQUENTIAL AND NON- CONSEQUENTIAL THEORIES DR.G.KARTHIGA Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION OF CONSEQUENTIAL THEORIES

A consequential theory judges the rightness or wrongness of an action based on the consequences that action has. The most familiar example would be utilitarianism--``that action is best that produces the greatest good for the greatest number" (Jeremy Bentham).

Consequential is the class of normative ethical theories holding that the consequences of one's conduct are the ultimate basis for any judgment about the rightness or wrongness of that conduct can reflect. Thus, from a consequential point of view a morally right act (or omission from acting) is one that will produce a good outcome, or consequence. In an extreme form, the idea of consequential is commonly encapsulated in the English saying, "the ends justify the means meaning that if a goal is morally important enough, any method of achieving it is acceptable.

Consequential is usually contrasted with deontological ethics (or *deontology*), in that deontology derives the rightness or wrongness of one's conduct from the character of the behavior itself rather than the outcomes of the conduct. It is also contrasted with virtue ethics, which focuses on the character of the agent rather than on the nature or consequences of the act (or omission) itself, and pragmatic ethics which treats morality like science: advancing socially over the course of many lifetimes, such that any moral criterion is subject to revision. Consequential theories differ in how they define moral goals.

Some people argue that consequential and deontological theories are not necessarily mutually exclusive. For example, T.M. Scanlon advances the idea that human rights, which are commonly considered a "deontological" concept, can only be justified with reference to the consequences of having those rights. Similarly, RoberNuzick argues for a theory that is mostly consequential, but incorporates inviolable "side-constraints" which restrict the sort of actions agents are permitted to do.

#### INTRODUCTION OF NON – CONSEQUENTIAL THEORIES

Kant's moral theory is perhaps the most influential of all non-consequential approaches. In his view, right actions have moral value only if they are done with a good will for duty's sake alone. The purpose of Kant's theory is the categorical imperative, a principle that he formulates in three versions. The first version says that an action is right if you can will that the maxim of an action become a moral law applying to all persons. An action is permissible if (1) its maxim can be universalized (if everyone can consistently act on it) and (2) you would be willing to have that happen. The second version of the categorical imperative says that we must always treat people as ends in themselves and never merely as a means to an end.

Kant's theory seems to conflict with our commonsense moral judgments (Criterion 1) and

#### CHAPTER 4 ETHICAL DECISION MAKING DR.V.SRIDEVI

Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Definition of Ethical Decision Making

Ethical decision making helps the people to make difficult choices when faced with an ethical dilemma, a situation in which there is no clear right or wrong answer.

#### Process of Ethical Decision Making

- 1. Gather the facts
- 2. Define the ethical issues
- 3. Identify the affected parties (stakeholders)
- 4. Identify the consequences
- 5. Identify the obligations (principles, rights, justice)
- 6. Consider your character and integrity
- 7. Think creatively about potential actions
- 8. Check your gut
- 9. Decide on the proper ethical action and
- 10. Be prepared to deal with opposing arguments.

#### 1 - GATHER THE FACTS

- Don't jump to conclusions without the facts
- •Questions to ask: Who, what, where, when, how, and why-
- Finding the facts-however, facts may be difficult to find because of the
- Uncertainty often found around ethical issues Some facts are not available
- •Assemble as many facts as possible before proceeding
- •Clarify what assumptions you are making!

#### 2 – DEFINE THE ETHICAL ISSUE(S)

• Don't jump to solutions without first identifying the ethical issue(s) in the situation.

- Define the ethical basis for the issue you want to focus on-
- $\square$  There may be multiple ethical issues focus on one major one at a time.
- 3 IDENTIFY THE AFFECTED PARTIES
- Identify all of the stakeholders
- Who are the primary or direct stakeholders?
- Who are the secondary or indirect stakeholders?

#### CHAPTER 5 ENVIRONMENT ISSUES DR.R.RAJAVARDHINI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **ENVIRONMENT ISSUES**?

Consequential is the class of normative ethics theories holding that the consequences of one's conduct are the ultimate basis for any judgment about the rightness or wrongness of that conduct can relict. The term 'Consequential' seems to be used as a family resemblance term to refer to any descendant of classic utilitarianism that remains close enough to its ancestor in the important respects. If that claim is dropped, the theory ceases to be consequential. Besides anyone who wants to pick out moral theories this absurd theory may talk about evaluative consequential, which is the claim that more rightness depends only on the value of the consequences. It has an act is to right or wrong depends only on the results of that act. The forms of consequential take a subtler approach. The consequential criterion for right action is maximization of a specified value. The non- consequential theories always reach decisions or evaluations on the basis of something other than, or in addition to the sum total or net aggregate of the consequences of choices. The theories of justice endorse principles of justice, which logically imply duties or rights that can be stated in terms of rules. An ethical dilemma, it is necessary to make a distinction between ethics, values, morals laws and policies. It assumes that the chooser will abide by societal norms such as codes of law or religious teachings in order to make the choice ethically impossible. The ethical decision making helps the people to make difficult choices when an ethical dilemma, a situation in which or wrong answer.

#### **KEY TERMS**

1. **CONSEQUENTIAL THEORY** - The term 'consequential 'seems to be used as a family resemblance term to refer to any descendant of classic utilitarianism that remains close enough to its ancestor in the important respects.

2. NON CONSEQUENTIAL THEORY - Non-Consequential Theories always reach decisions or evaluations on the basis of something other than, or in addition to, the sum total or net aggregate of the consequences of choices.

3. **ETHICAL DILEMMA** - An ethical dilemma is one in which a person has to choose between two options, both of which are morally correct but in conflict. Ethics and morals are inseparable. They both deal with questions of right and wrong.

**4. ETHICAL DECISION MAKING** - In determining right versus wrong, it has to be remembered that those terms are subjective. That's where understanding the definition of ethics, and relying on a corporate code of ethics, can be very helpful.

#### CHAPTER 6 WORKPLACE ETHICS DR.D.SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Introduction of Workplace Ethics**

The management of ethics in the workplace holds tremendous benefits for everyone, benefits both moral -- and even practical. This is particularly true today when it is critical to understand and manage highly diverse values in the workplace. When at a time too many people still feel that business ethics is a topic for philosophy or is about shaming and blaming people. This unit aims to make the topic of business ethics very understandable and accessible.

The field of business ethics has traditionally been the domain of philosophers, academics and social critics. Consequently, much of today's literature about business ethics is not geared toward the practical needs of leaders and managers -- the people primarily responsible for managing ethics in the workplace. The most frequent forms of business ethics literature today typically include-

a) philosophical, which requires extensive orientation and analysis;

b) anthologies, which require much time, review and integration;

c) case studies, which require numerous cases, and much time and analyses to synthesize; and d) extended stories about businesses "gone bad".

This lack of practical information is not the fault of philosophers, academics or social critics. The problem is the outcome of insufficient involvement of leaders and managers in discussions and literature about business ethics. More leaders and managers should become more involved.

#### **Meaning of Workplace Ethics**

Workplace ethics are codes of conduct that influence the development of an ethical culture within the workplace. Going beyond what is considered legal in the area where the business operates, they inspire communication between employees, allow for respect to be extended to each person within the organization, and promote customer relationships that are based on honesty and integrity. While there are core elements that tend to define a work-

based <u>code of ethics</u>, the specific expressions of these central values vary from one corporate setting to the next.

It is important to remember that workplace ethics are shaped by the following factor. Work place policy must be in harmony with all laws and regulations that are currently in force in the jurisdiction where the business operates. This helps to ensure that basic ethics preclude any pressure or coercion to engage in actions that are considered to be illegal, promote discrimination in the workplace, support unfair hiring and firing practices, or allow wages to be set that are below the minimum legal standards for the area.

#### CHAPTER 7 ORGANISATION ETHICS DEVELOPMENT SYSTEM DR.S.RAJENDRAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### MEANING OF ORGANISATION ETHICS DEVELOPMENT SYSTEM

An organisation is formed when individuals from different backgrounds and varied interests come together on a common platform and work towards predefined goals and objectives.

Employees are the assets of an organisation and it is essential for them to maintain the decorum and ambience of the workplace.

The way an organisation should respond to external environment refers to organization ethics. Organisation ethics comprises of various guidelines and principles which decide the way individuals should behave at the workplace.

It also refers to the code of conduct of the individuals working in a particular organisation.

Every organisation runs to earn profits but how it makes money is more important. No organisation should never depend on unfair means to earn money. One must understand that money is not the only important thing; pride and honour are more important. An individual's first priority can be to make money but he should not stoop too low just to be able to do that.

Organisational ethics are the principals and standards by which businesses operate. They are best demonstrated through acts of fairness, compassion, integrity, honour and responsibility.

#### IMPORTANCE OF ORGANISATION DEVELOPMENT SYSTEM

Ethics are the principles and values an individual uses to govern his activities and decisions. In an organisation, a code of ethics is a set of principles that guide the organization in its programs, policies and decisions for the business. The ethical philosophy that an organisation uses to conduct business can affect the reputation, productivity and bottom line of the business. Leadership Ethics

The ethics that leaders in an organisation use to manage employees may have an effect on the morale and loyalty of workers. The code of ethics leaders use determines discipline procedures and the acceptable behavior for all workers in an organisation. When leaders have high ethical standards, it will encourage workers in the organisation to meet that same level.

Ethical leadership also enhances the company's reputation in the financial market and community. A solid reputation for ethics and integrity in the community may improve the company's business.

#### **Employee Ethics**

Ethical behaviour among workers in an organisation ensures that employees complete work with honesty and integrity. Employees who use ethics to guide their behaviour will surely follow to employee policies and rules while striving to meet the goals of the organisation. Employees who follow ethics also meet standards for quality in their work, which can enhance the company's reputation for quality products and services.

#### CHAPTER 8 MARKETING ETHICS AND CONSUMER PROTECTION DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION OF MARKETING ETHICS

Marketing ethics is viewed as important because of marketing's interface with many diverse stakeholders. Marketing is a key functional area in the business organization that provides a visible interface with not only customers, but other stakeholders such as the media, investors, regulatory agencies, channel members, trade associations, as well as others.

It is important when addressing marketing ethics to recognize that it should be examined from an individual, organizational, and societal perspective. Examining marketing ethics from a narrow issue perspective does not provide foundational background that provides a complete understanding of the domain of marketing ethics.

The purpose of this chapter is to define, examine the nature and scope, identify issues, provide a decision-making framework, and trace the historical development of marketing ethics from a practice and academic perspective.

Generally, ethics refer to the way in which people relate in a moral manner toward others in all of their various interactions. Marketing Ethics refer specifically to the application of this basic morality in the conduct of business with their consumers and other related parties.

Such practices must necessarily include a conscious attempt by the businesses under consideration to apply moral principles when they are dealing with clients or other customers, especially when it comes to the production, pricing and promotion of their goods or services. Some ethical issues are universal, while some are derived from the culture and beliefs of various people. As such, various companies must necessarily incorporate this consideration in their marketing ethics.

#### **MEANING OF CONSUMER PROTECTION**

Consumer protection is a group of laws for organizations designed to ensure the rights of consumers as well as fair trade, competition and accurate information in the marketplace. The laws are designed to prevent businesses that engage in fraud or specified unfair practices from gaining an advantage over competitors.

They may also provide additional protection for those most vulnerable in society. Consumer protection laws are a form of government regulation, which aim to protect the rights of consumers.

For example, a government may require businesses to disclose detailed information about products, particularly in areas where safety or public health is an issue, such as food. Consumer protection is linked to the idea of consumer rights, and to the formation of consumer organization, which help the consumers to make better choices in the marketplace and get help with consumer complaints.

#### **CHAPTER 9**

#### CORPORATE SOCIAL RESPONSIBILITY DR. S. KAMARAJU Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION OF CORPORATE SOCIAL RESPONSIBILITY

Corporate Social Responsibility, or CSR, is something that was started by fashionable 'ethical' businesses.

Realizing that promoting a responsible way of doing business actually improved the bottom line soon received wider interest, and now demonstrating responsibility has become expected when bidding for major contracts.

. The concept of Corporate Social Responsibility can be explained in a simple way that doing the right thing.

Corporate Social Responsibility, or CSR for short, is about how the organization's existence affects stakeholders beyond the own insular interests, recognizing the impact of operations on the community at large.

Adopting CSR considers how we can use this impact in a positive way, leading to sustainable growth and financial gains. Over the years, CSR has become more and more popular. Back in 2007 more than 80% of the FTSE 100 index reported on Corporate Social Responsibility within their Annual Report.

#### MEANING OF CORPORATE SOCIAL RESPONSIBILITY

CSR aims to ensure that companies conduct their business in a way that is ethical. This means taking account of their social, economic and environmental impact, and consideration of human rights.

It can involve a range of activities such as:

- Working in partnership with local communities
- Socially Responsible Investment (SRI)
- Developing relationships with employees and customers
- Environmental protection and sustainability

Some businesses have as their main purpose the fulfillment of social or environmental goals, as opposed to a business that tries to achieve its financial goals while minimizing any negative impact on society or the environment. These businesses are called Social Enterprises.

Corporate Social Responsibility is a management concept whereby companies integrate social and environmental concerns in their business operations and interactions with their stakeholders.

Different organizations have framed different definitions - although there is considerable common ground between them. In simple terms it could be defined that CSR is about how companies manage the business processes to produce an overall positive impact on society.

#### CHAPTER 10 CORPORATE GOVERNANCE DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION OF CORPORATE GOVERNANCE :

Corporate Governance refers to the way a corporation is governed. It is the technique by which companies are directed and managed. It means carrying the business as per the stakeholders' desires.

It is actually conducted by the board of Directors and the concerned committees for the company's stakeholder's benefit. It is all about balancing individual and societal goals, as well as, economic and social goals.

Corporate Governance is the interaction between various participants (shareholders, board of directors, and company's management) in shaping corporation's performance and the way it is proceeding towards.

The relationship between the owners and the managers in an organization must be healthy and there should be no conflict between the two.

The owners must see that individual's actual performance is according to the standard performance. These dimensions of corporate governance should not be overlooked.

Corporate Governance deals with the manner the providers of finance guarantee themselves of getting a fair return on their investment. Corporate Governance clearly distinguishes between the owners and the managers. The managers are the deciding authority. In modern corporations, the functions/ tasks of owners and managers should be clearly defined, rather, harmonizing.

Corporate Governance deals with determining ways to take effective strategic decisions. It gives ultimate authority and complete responsibility to the Board of Directors. In today's market-oriented economy, the need for corporate governance arises.

Also, efficiency as well as globalization are significant factors urging corporate governance. Corporate Governance is essential to develop added value to the stakeholders.

Corporate Governance ensures transparency which ensures strong and balanced economic development.

This also ensures that the interests of all shareholders (majority as well as minority shareholders) are safeguarded. It ensures that all shareholders fully exercise their rights and that the organisation fully recognises their rights.

Corporate Governance has a broad scope. It includes both social and institutional aspects. Corporate Governance encourages a trustworthy, moral, as well as ethical environment.

# BUSINESS ACCOUNTING

### EDITED BY

## DR.G.KARTHIGA



#### **BUSINESS ACCOUNTING**

#### Copyright © 2021 by DR.G.KARTHIGA

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-623-3

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### **TABLE OF CONTENTS**

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	BRANCH ACCOUNTS	DR.R.SELVARAJ	1
2	DEPARTMENTAL ACCOUNTS	DR. S. KAMARAJU	11
3	HIRE PURCHASE ACCOUNTS	DR.G.KARTHIGA	22
4	INSTALMENT PURCHASE		32
	SYSTEM	DR.V.SRIDEVI	
5	SELF BALANCING AND		42
	SECTIONAL BALANCING		
	LEDGERS	DR.R.RAJAVARDHINI	
6	<b>ROYALTY ACCOUNTS</b>	DR.D.SILAMBARASAN	52
7	FIRE INSURANCE CLAIMS FOR		62
	LOSS OF STOCK AND PROFITS	DR.S.RAJENDRAN	
8	ACCOUNTING FOR SALE OR		72
	RETURN	DR.R.SELVARAJ	
9	INSOLVENCY ACCOUNTS	DR. S. KAMARAJU	82
10	STATEMENT OF AFFAIRS	DR.G.KARTHIGA	92
	REFERENCE		101

#### CHAPTER 1 BRANCH ACCOUNTS DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **MEANING OF BRANCH ACCOUNTS:**

- ➤ A Branch can be described as any establishment carrying on either the same or substantially the same activity as that carried on by head office of the company.
- It must also be noted that the concept of a branch means existence of a head office for there can be no branch without a head office - the principal place of business. From the accounting point of view, branches may be classified as follows:

## DIFFERENCES BETWEEN BRANCH ACCOUNTS AND DEPARTMENTAL ACCOUNTS

Basi	s of distinction	Branch Accounts	Departmental Accounts
1.	Maintenance of accounts	Branch accounts may be maintained either at branch or at headoffice.	Departmental accounts are maintained at one place only.
2.	Allocation of common expenses	No allocation problem arises since the expenses in respect of each branch can be identified.	Common expenses are distributed among the departments concerned on some equitable basis considered suitable in the case.
3.	Reconciliation	Reconciliation of head office and branch accounts is necessary in case of independent branches at the end of the accounting year.	No such problem arises.
4.	Conversion of foreign currency figures	At the time of finalization of accounts, conversion of figures of foreign branch is necessary.	No such problem arises in departmental accounts.

#### CHAPTER 2 DEPARTMENTAL ACCOUNTS DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **DEPARTMENTAL ACCOUNTS**

If a business consists of several independent activities, or is divided into several departments, for carrying on separate functions, its management is usually interested in finding out the working results of each department to ascertain their relative efficiencies. This can be made possible only if departmental accounts are prepared. Departmental accounts are of great help and assistance to the managements as information for controlling the business more intelligently and effectively, since thereby all types of waste either of material or of money are readily detected; also attention is drawn to inadequacies or inefficiencies in the working of departments or units into which the business may be divided.

#### Advantages of Departmental Accounting

The main advantages of departmental accounting are as follows:

- 1. **Evaluation of performance:** The performance of each department can be evaluated separately on the basis of trading results. An endeavor may be made to push up the sales of that department which is earning maximum profit.
- 2. Growth potential of each department: The growth potential of a department as compared to others can be evaluated.
- **3. Justification of capital outlay:** It helps the management to determine the justification of capital outlay in each department.
- 4. **Judgement of efficiency:** It helps to calculate stock turnover ratio of each department separately, and thus the efficiency of each department can be revealed.
- 5. **Planning and control:** Availability of separate cost and profit figures for each department facilitates better control. Thus effective planning and control can be achieved on the basis of departmental accounting information.

#### Methods of Departmental Accounting

Basically, an organization usually divides the work in various departments, which is done on the principle of division of labour. This can improve efficiency of each and every department of the organization. Each department prepares its separate accounts to judge its individual performance.

There are two methods of keeping departmental accounts:

Accounts of all departments are kept in one book only: To prepare such accounts, it will be necessary first, for the income and expenditure of department to be separately recorded in subsidiary books and then for them to be accumulated under separate heads in a ledger or ledgers. This may be done by having columnar subsidiary books and a columnar ledger. Under this system, the gross profit of individual department can be determined accurately.

**Separate set of books are kept for each department:** A separate set of books may be kept for each department, including complete stock accounts of goods received from or transferred to other departments or as also sales.

Nevertheless, even when separate sets of books are maintained for different departments, it will also be necessary to devise a basis for allocation of common expenses

#### CHAPTER 3 HIRE PURCHASE ACCOUNTS

#### DR.G.KARTHIGA

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

With an increasing demand for better life, the consumption of goods has been on the expanding scale, but, this has not been backed up by adequate purchasing power, transform in to effectual demand, i.e., actual sale at set or settled prices. This has created the market for what is called hire purchase.

#### **Nature of Hire Purchase**

Under the hire purchase system the hire purchaser gets possession of the goods at the outset and can use it, while buying for it in instalments over a specific period of time as per the agreement. However, the ownership of the goods remains with the hire vendor until the hire purchaser has paid all the instalments. Each installment paid by the hire purchaser is treated as hire charges for using an asset. In case he fails to pay any of the instalments the hire vendor will take back his goods without compensating the buyer, i.e., the hire vendor is not going to pay back a part or whole of the amount received through instalments till the date of default from the buyer.

#### **Special Features of Hire Purchase**

- 1. **Possession:** The hire vendor transfers only possession of the goods to the hire purchaser immediately after the contract for hire purchase is made.
- 2. **Instalments:** The goods are delivered by the hire vendor on the condition that a hire purchaser should pay the amount in periodical instalments.
- 3. **Down Payment:** The hire purchaser generally makes a down payment i.e., an amount on signing the agreement.
- 4. **Constituents of Hire purchase instalments:** Each installment consists partly of a finance charge (interest) and partly of a capital payment.
- 5. **Ownership:** The property in goods is to pass to the hire purchaser on the payment of the last installment and exercising the option conferred upon him under the agreement.
- 6. **Repossession:** In case of default in respect of payment of even the last installment, the hire vendor has the right to take the goods back without making any compensation.

#### Terms used in Hire Purchase

- 1. **Hire Vendor:** Hire vendor is a person who delivers the goods along with its possession to the hire purchaser under an hire purchase agreement.
- 2. Hire Purchaser: Hire purchaser is a person who obtains the goods and rights

#### CHAPTER 4 INSTALMENT PURCHASE SYSTEM

#### DR.V.SRIDEVI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **MEANING OF INSTALLMENT PURCHASE SYSTEM :**

An **instalment purchase system** is a type of payment plan where a buyer can purchase an item and pay for it over a period of time in installments. The buyer makes a down payment and agrees to pay the remaining amount in equal installments over an agreed-upon time frame. This system is commonly used for high-value items such as cars, homes, and appliances.

The **installment purchase system** is a convenient way for consumers to buy expensive items without having to pay the full amount upfront. It allows them to spread the cost of the item over several months or years, making it easier to budget and manage their finances. Additionally, this system may be more accessible to people who do not have the means to pay for an item in full at the time of purchase.

#### **BENEFITS OF AN INSTALLMENT PURCHASE SYSTEM :**

The benefits of an **installment purchase system** are not just limited to consumers. Businesses can benefit from this system as well. It provides a way for businesses to sell high-value items without having to wait for the full payment, and they can generate more sales by offering flexible payment plans.

However, it is important to note that there are some downsides to an installment purchase system. The total cost of the item may be higher due to interest charges and fees. Buyers must also make sure that they can afford the monthly payments and that the purchase is a wise financial decision.

#### **CONCLUSION** :

In conclusion, the **installment purchase system** is a useful payment plan that can benefit both consumers and businesses. It allows buyers to purchase expensive items without having to pay the full amount upfront and provides businesses with a way to sell high-value items. However, buyers should be aware of the additional costs and ensure that they can afford the payments before making a purchase.

Installment payment is a method where you can pay off your invoices in smaller, manageable chunks over a specified period. It allows both individuals and businesses to make larger expenses more accessible by dividing them into smaller payments. This flexible payment option provides convenience and financial flexibility to both buyers and sellers.

From a business perspective, offering installment payments can attract more customers and increase sales. Customers are more likely to make a purchase when they have the option to pay in installments. For general consumers or e-commerce shoppers, installment payments offer the opportunity to make big-ticket purchases without straining their budgets. Instead of paying the entire amount upfront, they have the convenience of spreading the cost over time.

To accept installment payments, businesses must integrate suitable payment methods into their platforms. They can partner with <u>payment gateways</u> or use invoicing software for installment options.

#### CHAPTER 5 SELF BALANCING AND SECTIONAL BALANCING LEDGERS DR.R.RAJAVARDHINI Department of Commerce, Ponnaivah Ramajavam Institute of science and Technology(PRIST)

#### MEANING OF SELF BALANCING

Self Balancing Ledger System implies a system of ledger keeping which classifies ledgers as per nature of transactions, namely, Sales ledger, Bought ledger, General ledger, etc. and also makes them to balance independently. With rise in the number of transactions the size of the ledger becomes hefty due to large number of accounts. This creates problem in detection of errors. To overcome this, the system of multiple ledgers is deployed. It involves splitting of single ledger. Generally three ledgers, namely debtor ledger, creditor ledger and main ledger (containing remaining accounts) are prepared. In this Unit we shall discuss the self balancing ledger system and its advantages. Also we shall illustrate system.

#### ADVANTAGES OF SELF BALANCING SYSTEM

When a number of ledgers are kept by a concern and if their balances do not tally, the accountant would have to face great difficulty in tracing book- keeping errors, responsible for the non-agreement of the Trial Balance. In order to reduce to a minimum the trouble and time involved in locating the errors, sometimes the system of self-balancing or sectional balancing of ledger is employed.

Quite often the debit and credit entries relating to a transaction are posted in different ledgers e.g. when goods are sold on credit, the Sales Account will be credited in the General Ledger but the corresponding debit will be made in the customer's account in the Personal Ledger. In such a case for ascertaining the correctness of the posting in either of the ledgers it will be necessary to take out balances in both the ledgers; thus a mistake in one ledger will require checking of the balances in the others as well.

Such a position would be avoided if every ledger is made independent of the other by the converse aspect of entries in each ledger being posted in totals to the Control Account set up in the ledger itself. If this is done the correctness of individual balances in each ledger would be verified extracting its balances and agreeing them with the balances of the Control Account. A ledger that has a Control Account set up in it, is referred to as a self balancing ledger. It connotes that it is capable of being balanced independently, the balance in the Control Account being equal to that of the individual balance.

#### SECTIONAL BALANCING

A really simple way to prove the accuracy of say, the Sales Ledger would be to maintain a Total Debtors account in the General Ledger. It would mean that whereas accounts of Individual customer would be maintained in the Sales Ledger, in the General Ledger the Total Debtors account would be posted by the (monthly) totals of various transactions with total credit sales, total amount received from credit customers, total discount allowed to them, total returns inwards, total bills receivable received; etc. The balance in the Total Debtors Account should be equal to the total of balances shown by the accounts of individual customers. If it is so, the Total Debtors Account as well as individual customers' account may be taken as correct. A difference would show that there is some error somewhere.

#### CHAPTER 6 ROYALTY ACCOUNTS

#### DR.D.SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST) INTRODUCTION

Royalty is a periodical payment, based on output or sale, required to be paid by one personto another in consideration of some special rights acquired e.g. the right to exploit a mine or collieryto publish a book or to manufacture an article under a patent. Royalties arises with the agreement between two parties one is the owner of the asset or the person who has rights over the asset. Other is the person who acquired the right of using it for payment of a Royalty. The person who makes the payment to the owner of the asset in exchange for the right to use his asset is known as'lessee' and the owner of the asset as lessor or landlord.

The lessee making the payment of royalty treats it as ordinary business expenditure and debits Royalty Account. Royalty Account is a nominal account and is closed at the end of every accounting year by transferring it to profit and loss Account. Strictly speaking, royalty based on output is a part of cost of production and as such should be transferred to Trading account, royalty based on sales is a selling expenditure and as such should be transferred to profit and loss Account

Here the student should note that a lump - sum payment for the outright purchase of a patent, mine or book is not treated as royalty but is a capital expenditure and recorded as a fixed asset.

#### **EXPLANATION OF TERMS :**

In royalty agreement between the lessor and the lessee, usually contains certain clauses as regards 'minimum rent' 'short working' and Recoupment of short working. Before going to the Accounting treatment of the transactions relating to Royalties, it is necessary to understand these terms.

1. Minium Rent or Dead Rent or Fixed Rent :

In the early days of business, production or sale of any concern will be very low. Hence, there is ever possibility that the Royalty based on production or sale will also be low in that period. To safe guard the interest of the landlord from the low income and also to guarantee him the receipt of a minium amount in case of low output or sale, the minium Rent clause will be included in the Royalties agreement. It also give incentive to the lessee to increase the production or sales. Thus minium rent may be defined as "The minimum amount which is payable by the lessee to the landlord, irrespective of production or sales" In the presence of minimum rent clause in Royalty agreement actual royalty based on output or sales or minium rent whichever is higher is payable For example; Madhu coal CO has taken a lease of coal mine with a rent of Rs 30,000 a year and with a rate of royalty at Rs 5 per tonne of coal extracted and if the production in the first year is 4000 tonnes and in the first year because the benefit derived by Madhu coal co. is less than the minimum rent agreed upon i.e. 4000 5 5 per tonne. It will pay Rs 40,000 in the second year because the benefit derived derived by Madhu coal co. is less than the minimum rent agreed upon i.e. 5 5 8000) is more than the minimum rent.

#### **2.** Short working

The excess of minimum rent over actual royalty calculated on the basis of output or salesis termed as short working. In the example cited above, there is a short working of Rs. 1,000 i.e.Rs.

#### CHAPTER 7 FIRE INSURANCE CLAIMS FOR LOSS OF STOCK AND PROFITS DR.S.RAJENDRAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Meaning of Fire Insurance

For purposes of insurance, fire means:

- 1. Fire (whether resulting from explosion or otherwise) not occasioned or happening through:
  - (a) Its own spontaneous fomentation or heating or its undergoing any process involving the application of heat;
  - (b) Earthquake, subterraneous fire, riot, civil commotion, war, invasion act of foreign enemy, hostilities (whether war be declared or not), civil war, rebellion, revolution, insurrection, military or usurped power.
- 2. Lightning.
- 3. Explosion, not occasioned or happening through any of the perils specified in 1 (a) above.
  - (i) of boilers used for domestic purposes only;
  - (ii) of any other boilers or economizers on the premises;
  - (iii) in a building not being any part of any gas works or gas for domestic purposes or used for lighting or heating the building.

#### **Claim for Loss of Stock**

Fire insurance being a contract of indemnity, a claim can be lodged only for the actual amount of the loss, not exceeding the insured value. In dealing with problems requiring determination of the claim the following point must be noted:

- **a.** Total Loss: If the goods are totally destroyed, the amount of claim is equal to the actual loss, provided the goods are fully insured. However, in case of under insurance (i.e. insurable value of stock insured is more than the sum insured),the amount of claim is restricted to the policy amount.
- **b. Partial Loss:** If the goods are partially destroyed, the amount of claim is equal to the actual loss provided the goods are fully insured. However in case of under insurance, the amount of claim will depend upon the nature of insurance policy as follows:
  - I) Without Average clause: Claim is equal to the lower of actual loss or the sum insured.

**II)** With Average Clause: - Amount of claim for loss of stock is proportionately Reduced, considering the ratio of policy amount (i.e. insured amount) to the value of stock as on the date of fire (i.e insurable amount) as shown below: Amount of claim = Loss of stock x sum insured / Insurable amount (Total Cost)

#### CHAPTER 8 ACCOUNTING FOR SALE OR RETURN DR.R.SELVARAJ Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST) ACCOUNT FOR SALE OR RETURN

#### Meaning

The transactions which involve customers' approval for rejection of goods after physical delivery are called sale on approval or sale or return transactions.

#### Purpose

- 1. Traders can increase their turnover, particularly in the case of slow moving goods.
- 2. Valued customers or bulk buyers can be provided with the opportunity of inspecting the goods at their own place conveniently before approval or rejection.

Method of recoding or sale or return transactions

- I. When the transaction are few and rare:
- a. When goods are sent on approval: Customer's A/c

To Sales A/c

Dr

Dr

- b. When customer intimate approval: No further entry in needed
- c. When customer returns the goods, fully or partly, for the goods returned. Sales A/c Dr
  - To Customer's A/c
- d. At the end of the accounting year (regarding goods sent for which specified time is not yet finished)
  - For reversing the original entry: Sales A/c To Customer's A/c
  - ii. For taking goods with the customers into stock at their cost. Stock with customer's A/c Dr

To Trading A/c

- II. When the transactions are frequent:
  - a. A sale or return day book is opened on memorandum basis to show the goods sent, approval, rejected and balance.
  - b. When goods are sent, they are recorded sales value in the goods sent column, because actual sale has not resulted
  - c. When goods are approved, the sales value of the goods is recorded in the goods approved column. Immediately customers' accounts are debited in the regular ledger.
  - d. When goods are rejected the sale value of the goods is recorded in the goods returned column.
  - e. Balance of goods is recorded in the balance column. This column shows the sale value of goods with customers.

#### CHAPTER 9 INSOLVENCY ACCOUNTS DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Meaning of insolvent:

Insolvent is one who is not able to pay his debts as and when they are due.

#### Insolvency accounts

Upon being adjudicated the insolvent has to submit

- 1. Statement of Affairs
- 2. A Deficiency account

#### Statement of affairs

The form of the statement of affairs prescribed by rule made under the Presidency Towns Insolvency Act.

Gross liabilities (Rs.)	Liabilities	Expected to rank	Assets	Estimated to produce
XXX	Un secured creditors as per list - A		Property as per list	XXX
XXX	Fully secured creditors as per List – B Less: Estimated value of security Less: Amount carried	XXX XXX XXX	Cash in hand Cash at bank Investments Stock Machinery	
	down to list - C Surplus as per Contra	xxx xxx	Book Debts as per List – F: Good Doubtful Bad	XXX
			Bills of Exchange as per List – G	XXX
			Surplus from securities in the hands of secured creditors as per contra	XXX
XXX	Partly secured creditors as per List – C Less: Estimated value of security	xxx <u>xxx</u> xxx	Deficiency as explained in List – H (B.F.)	XXX
XXX	Preferential creditors as per list D Less: Deducted as per Contra	XXX XXX XXX XXX		
		XXX		XXX

Specimen of Statement of Affairs

#### CHAPTER 10 STATEMENT OF AFFAIRS DR.G.KARTHIGA Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST) Key differences between Statement of Affairs and Balance Sheet

- 1. **Purpose:** A statement of affairs is a financial statement that provides a snapshot of a company's assets and liabilities at a specific point in time, whereas a balance sheet is a financial statement that provides a snapshot of a company's assets, liabilities, and equity at a specific point in time.
- 2. **Format:** A statement of affairs is typically presented in a tabular format, listing assets and liabilities in separate sections, whereas a balance sheet is presented in a standard format with assets listed on the left side and liabilities and equity listed on the right side.
- 3. **Timing:** A statement of affairs is usually prepared at the end of an accounting period, whereas a balance sheet is usually prepared at the end of a fiscal quarter or year.
- 4. **Information Provided:** A statement of affairs typically includes more detailed information on a company's assets and liabilities than a balance sheet, which is more focused on providing a summary of a company's financial position.
- 5. Audience: A statement of affairs is primarily intended for internal use by management and stakeholders, whereas a balance sheet is intended for external use by investors, creditors, and other stakeholders.

#### **Brief Note on Statement of Affairs**

A "condition of affairs" refers to the current state or situation of a particular event, issue, or organization. It can also refer to the general state of things in a particular area or industry. Without more context, it is difficult to provide a specific brief note on a specific condition of affairs.

#### Advantages & Disadvantages of Statement of Affairs

Advantages of a Statement of Affairs:

- 1. Provides a clear and detailed snapshot of a company's financial position at a specific point in time.
- 2. Helps management identify and assess the company's strengths and weaknesses.
- 3. Provides a basis for making informed decisions about the company's future operations and investments.
- 4. Helps external stakeholders, such as investors and creditors, evaluate the company's financial health.
- 5. Can be used to identify and quantify potential risks and liabilities.
- 6. Can serve as a starting point for creating a budget or financial forecast.
- 7. Can be used to evaluate the performance of the company's management team.
- 8. Can be used to assess the company's compliance with legal and regulatory requirements.
- 9. Can be used as a tool for negotiating financing or other business deals.

#### Disadvantages of a Statement of Affairs:

- 1. Can be time-consuming and costly to prepare.
- 2. May not provide a complete picture of the company's financial performance if certain information is not included.
- 3. Can be difficult to understand for those without a strong background in finance or accounting.
- 4. May not take into account non-financial factors that could impact the company's performance.
- 5. May not be as useful for companies with complex financial structures or operations.
- 6. May be subject to errors or inaccuracies if not prepared properly.
- 7. May be misleading if it includes information that is not relevant or accurate.
- 8. May not be relevant if the financial statements are not up to date.
- 9. May be subject to misinterpretation if not presented in a clear and concise manner.

## **BUSINESS** ECONOMICS

## EDITED BY

## DR. S. KAMARAJU



#### **BUSINESS ECONOMICS**

Copyright © 2021 by DR. S. KAMARAJU

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-629-5

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### **TABLE OF CONTENTS**

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	INTRODUCTION OF		1
	ECONOMICS AND BUSINESS		
	ECONOMICS	DR.R.SELVARAJ	
2	NATURE AND SIGNIFICANCE		11
	OF BUSINESS ECONOMICS	DR. S. KAMARAJU	
3	LAW OF DEMAND AND		22
	DEMAND CURVES	DR.G.KARTHIGA	
4	CONCEPT OF ELASTICITY	DR.V.SRIDEVI	32
5	FACTORS OF PRODUCTION	DR.R.RAJAVARDHINI	42
6	LAW OF RETURNS TO SCALE	DR.D.SILAMBARASAN	52
7	FACTORS AFFECTING SUPPLY	DR.S.RAJENDRAN	62
8	FIXED AND VARIABLE COST		72
	CURVES	DR.R.SELVARAJ	
9	PRICE AND OUTPUT		82
	DETERMINATION IN A		
	PERFECT MARKET	DR. S. KAMARAJU	
10	MONOPOLISTIC		92
	COMPETITION	DR.G.KARTHIGA	
	REFERENCE		101

#### CHAPTER 1 INTRODUCTION OF ECONOMICS AND BUSINESS ECONOMICS DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTORDUCTION TO ECONOMICS

#### **INTRODUCTION**

Economics deals with the day to day activities of human beings life. In all, human beings are gratified (or) enjoy the economic activities such as consumption, production, exchange and distribution. In simple terms economics is concerned with the aspects of humanbehaviour.

#### **ORIGIN OF ECONOMICS**

The term Economics is derived from the two Greek words "Oikos" (means house) and "Nomos" (means manage). If these two words are merged "Oikonomia" it gives the meaning household management.

In the earlier period, economics is linked with politics. So the earlier economist called economics as a "political economy". This subject name was changed from "political economy" to "economics" by Alfred Marshall.

#### **DEFINITION OF ECONOMICS**

There are four important definitions of economics to understand the basic concept of economics. They are

Wealth Definition –Adam Smith Welfare Definition – Alfred Marshall Scarcity Definition-Lionel Robbins Growth Definition – Paul. A. Samuelson.

Let us discuss these definitions in detail.

#### WEALTH DEFINITIONS

The classical economists defined economics as the science of wealth. Adam Smith in his famous book, "An Enquiry into the Nature and Causes of the Wealth of Nations", which was published in 1776, described economics systematically.

#### Definition

"Economics is an enquiry into the nature and causes of the wealth of nations". Adam Smith

#### Features

The wealth definitions have the following main features:

i) Study of wealth

According to the wealth definitions of economics the only proper study of economicsis wealth.

#### CHAPTER 2 NATURE AND SIGNIFICANCE OF BUSINESS ECONOMICS DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### NATURE AND SIGNIFICANCE OF BUSINESS ECONOMICS

i) Economics is a science

According to Robbins, economics is not an art or a normative science. It is a science which studies the causes of economic problems and does not study its merits and demerits.

ii) Human behaviour

According to Robbins, economics is a science which studies human behaviour. The behaviour of every person is studied in economics irrespective of the fact whether he lives in or out of society.

iii) Unlimited ends

Wants are unlimited. It is not possible to satisfy them all. If one want is satisfied, another crops up. In fact, there is no end to what a man may want.

iv) Scarce means

Though wants are unlimited, the means at the disposal to satisfy these wants are scarceor limited.

v) Alternative uses of means

Wants are unlimited but the means to satisfy them are scarce. Not only the means are scarce but can be put to a number of uses. This makes them all the more scarce.

vi) Problem of choice

According to Robbins, economics is the science of choice-making because wants areunlimited and means to satisfy them are limited. Criticism

Many economists criticized the Robbins definition on the following grounds:

i) More emphasis on scarcity

Robbins gives too much importance to the economics of scarcity. Modern economicsis growth oriented rather than scarcity oriented.

ii) Too wide a definition

According to Robbins economics is the study of all human activities which are related to the problem of choice. The problem of choice as such is faced not only by the social beings but also by the non-social beings like saints and smugglers.

iii) Economic problems do not always arise from scarcity

According to Robbins the economic problems arise due to scarcity of means in relationto wants. He gives main emphasis on the phenomenon of scarcity. But some critics are of

#### CHAPTER 3 LAW OF DEMAND AND DEMAND CURVES DR.G.KARTHIGA Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### LAW OF DEMAND

Law of demand explains the relationship between the price of a commodity and its quantity demanded over a certain period of time. According to this law, other thingsremaining the same, there is an inverse relationship between the price of a commodity and its quantity demanded."The amount demanded increases with a fall in price and diminishes with a rise in prices".

#### Marshall

The law of demand states that other things being constant, there is an inverse relationship between the price of various commodities and their quantity demanded over a certain period of time. In other words, with the increase in the price of a commodity, there is a fall in its demand and with the decrease in its price, there is a rise in its demand.

Assumptions of the law of demand

- i) Income of the consumer remains unchanged.
- ii) Prices of other related goods remain constant.
- iii) Tastes of the consumers remain unchanged during the period of time.
- iv) The consumers, expectations about future prices are neutral.
- v) The effect of advertising is ruled out.
- vi) Other relevant factors like the size of the population, seasonal and climatic factors, habits of the people and all other factors influencing demand remain unchanged.

#### LAW OF DEMAND

Law of demand explains the relationship between the price of a commodity and its quantity demanded over a certain period of time. According to this law, other thingsremaining the same, there is an inverse relationship between the price of a commodity and its quantity demanded."The amount demanded increases with a fall in price and diminishes with a rise in prices".

#### Marshall

The law of demand states that other things being constant, there is an inverse relationship between the price of various commodities and their quantity demanded over a certain period of time. In other words, with the increase in the price of a commodity, there is a fall in its demand and with the decrease in its price, there is a rise in its demand.

#### Assumptions of the law of demand

- i) Income of the consumer remains unchanged.
- ii) Prices of other related goods remain constant.
- iii) Tastes of the consumers remain unchanged during the period of time.
- iv) The consumers, expectations about future prices are neutral.
- v) The effect of advertising is ruled out.
- vi) Other relevant factors like the size of the population, seasonal and climatic factors, habits of the people and all other factors influencing demand remain unchanged.

#### CHAPTER 4 CONCEPT OF ELASTICITY DR.V.SRIDEVI Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### MEANING OF ELASTICITY OF DEMAND

The word "elasticity" is a technical term which stands for the sensitivity or responsiveness of a dependent variable to the changes in independent variables. The elasticity of demand is the responsiveness of demand to the changes in the price of acommodity, incomeof the consumers and the prices of related goods.

#### PRICE ELASTICITY OF DEAMAND

A proportionate change in quantity demanded brought by a proportionate change in price is called the price elasticity of demand.

The price elasticity of demand is a measure of responsiveness of the quantity demanded to a change in the price of a good, ceteris paribus.

"Elasticity of demand may be defined as the percentage change in the quantity demanded by the percentage change in price."

Alfred Marshall

#### Ed = Proportionate change in quantity demanded

Proportionate change in price

Types of Price Elasticity of Demand

The extent of responsiveness of demand with change in the price is not always the same. The demand for a product can be elastic or inelastic, depending on the rate of change in the demand with respect to change in price of a product. Elastic demand is the one when the response of demand is greater with a small proportionate change in the price. On the other hand, inelastic demand is the one when there is relatively a less change in the demand with a greater change in the price.

For better understanding the concepts of elastic and inelastic demand, the price elasticity of demand has been divided into five types, which are shown in Figure-1:



Let us discuss the different types of price elasticity of demand (as shown in Figure-1).

**1.** Perfectly Elastic Demand:

When no change in price of a product causes a major change in its demand, it is saidto be perfectly elastic demand. In perfectly elastic demand, a no change in price causes increase in demand to infinity. In perfectly elastic demand, the demand curve is represented as a horizontal straight line, which is shown in Figure-2:

#### CHAPTER 5 CONCEPT OF ELASTICITY DR.R.RAJAVARDHINI Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### FACTORS OF PRODUCTION

The purpose of production which makes use of producer goods and services in the economic activities is to create utilities by providing and endless flow of goods and service. In order to achieve the production of utilities, the resources available to mankind are mobilized. These resources are called factors of production.

The modern economics have classified the factors of production into four groups

- i) Land
- ii) Labour
- iii) Capital
- iv) Entrepreneur

#### i) Land

Land is the original and basic factor of production. In economics, the term "land" is used in a broader sense. It includes all the natural resources or gifts of nature. It does not includeonly surface land but also forests, mountains, sea, climate, air and so on. The term land includesall the natural resources on the surface (soil, plots), below the surface (minerals) and above the surface (climate, air).

As compared to other factors of production, land has certain special characteristics like limited supply, being a free gift of nature, indestructibility and immobility. Land is the primary factor of production. No production is possible without land. Land is the basis of the primary sector (agriculture and allied), secondary sector (industries) and tertiary sector(trade, transport and communication). It is also the basis of power and energy which play a key role in the economic development of a country.

#### ii) Labour

Labour is an active agent of production. In economics, the term labour stand for all types of physical or mental work which is done in expectation of some reward in kind or cash. Labour is inseparable from the laborer. A labourer is unlike other commodities. In case of commodities other than labour, the ownership changes in the process of sale purchase. A labourer sells his labour power or capacity to work. Labour is perishable, mobile, differsin efficiency and has consciousness and power of judgments.

#### iii). Capital

Capital is a very crucial factor of production. It enables labour to effectively utilize the gifts of nature. In modern times, because of scientific and technological developments which involve a lot of capital investment, the role of capital has increased further.

For an economist, capital is that form of wealth which helps in production. In other words, capital is that part of private or public wealth which further generates income or contributes to production.

#### CHAPTER 6 LAW OF RETURNS TO SCALE DR.D.SILAMBARASAN

#### Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### LAW OF RETURNS

The law of return takes three forms:

- a) Law of diminishing return
- b) Law of constant returns
- c) Law of increasing returns

#### a) LAW OF DIMINISHING RETURNS

The law of diminishing returns is one of the oldest and most controversial parts of economic theory. This law states that if the quantity of one factor, say land, is fixed and to increase output, more and more units of labour and capital are applied, increase in output will take place at a decreasing rate. In other words, the marginal increase in output will be comparatively smaller than the increase in labour and capital.

#### Assumptions of the law

- i) The technique of production remains constant.
- ii) The co-efficient of production are variable, i.e., factor proportions are variable.
- iii) Some factors can be held constant.
- iv) The units of variable factor are homogeneous.

#### **b**) LAW OF CONSTANT RETURNS

The law of constant returns states that the increase in output or marginal physical product is that the same rate as that in the units of labour and capital. Additional units of labourand capital yield the same return. The per unit cost of production remains the same at all levels of output. c) LAW OF INCREASING RETURNS

#### c) LAW OF INCREASING RETURNS The law of increasing returns states that the marginal increase in output is proportionately higher than the increases in the units of labour and capital. When more and more units of labour and

higher than the increase in the units of labour and capital. When more and moreunits of labour and capital are applied, they bring increasing returns or raise total outputat an increasing rate. The law of increasing returns is based on the assumption that there always remains ample scope for improvements in the techniques of production. The improvements in the methods of production, use of modern machines and increased division of labour raise theproductivity. The theory also assumes that some of the factors or atleast one factor is indivisible. Most of the remaining factors are divisible. The law of increasing returns states that an addition in units of inputs brings higher and higher levels of marginal output.

#### LAW OF VARIABLE PROPORTION

Law of variable proportions establishes the short run relationship between the changes in output and the changes in inputs. In the short period, some factors are fixed and some are variable. So in the short run, if we want to increase the output, we have to vary the variable factors only. The law is called the law of variable proportions because when in the short run, increasing doses of variable factors are applied upon some fixed factors, the factor proportion changes.

The law of variable proportions which comprises three stages applies in all economic fields. Prof. Samuelson has stated that an increase in some inputs relative to other comparatively fixed inputs will cause output to increase; but after a point, the extra output resulting from the same additions of inputs will become less and less; this falling of extra returns is a consequence of the fact that the new doses of varying resources have less and less of the constant resources to work with.

#### CHAPTER 7 FACTORS AFFECTING SUPPLY DR.S.RAJENDRAN Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### FACTORS AFFECTING SUPPLY

#### 1. Price of the commodity

Higher the price larger the supply. Price is the incentive for the producers and sellers to supply more.

#### **2.** Price of other commodities

The supply of a commodity depends not only upon its price but also price of other commodities. For instance if the price of commercial crops like cotton rise, this may result in reduction in cultivation of food crops like paddy and so its supply.

#### 3. Price of factor

When the input prices go up, this results in rise in cost and so supply will be affected.

#### 4. Price expectations

The expectation over future prices determines present supply. If a rise in price is anticipated in future, sellers tend to retain their produce for future sale and so supply in present market is reduced.

#### 5. Technology

With advancement in technology, production level improves, average cost declines and as a result supply level increases.

#### **6.** Natural factors

In agriculture, natural factors like monsoon, climate etc. play a vital role in determining production level.

#### 7. Discovery of new raw materials

The discovery of new raw materials which are cheaper and of high quality tends to increase supply of the product.

#### 8. Taxes and subsidies

Subsidies for inputs, credit, power etc. encourage the producers to produce more. Withdrawal of such incentives will hamper production. Taxes both direct and indirect kill the ability and willingness to produce more.

#### 9. Objective of the firm

When the goal of the firm is sales maximization or improving market share, the supply of the product is likely to be higher.

#### CHAPTER 8 FIXED AND VARIABLE COST CURVES DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Fixed cost (FC)

Fixed cost remains constant for all units of output. In other words, cost which does not change with the change in output is called fixed cost. It includes rent and interest payments, depreciation charges, wages and salaries of permanent staff etc., fixed costs have to be incurred by a firm, even if it stops production temporarily. They exist even whenoutput is zero. They are also called "overhead costs".

In the figure, TFC is the total fixed cost curve. It is a horizontal parallel to the OXaxis. It shows that fixed cost remains unaltered even though output changes



#### Variable Cost (VC)

Cost which varies with variation in output is called variable cost. Variable costs vary directly and sometimes proportionately with output. They are also called "prime cost". Variable costs include

(i) Costs of raw materials and

(ii) Costs of casual or daily labour etc., They

are incurred only when the factory is at work.


#### CHAPTER 9 PRICE AND OUTPUT DETERMINATION IN A PERFECT MARKET DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### PRICE AND OUTPUT

In the figure, TVC is the total variable cost curve. It is going upward at a diminishing rate. It shows that total variable cost changes with changes in output.

Unit of output	TFC	TVC	TC
0	30	0	30
1	30	10	40
2	30	18	48
3	30	24	54
4	30	32	62
		1	

In the table, total fixed cost remains constant at Rs.30/-. But the total variable cost increases continuously at a diminishing rate. Total cost also increases continuously. It can be explained with help of the following figure.



In the figure, TC is the total cost curve. Its slope indicates the positive relationship between output and total cost.

#### SHORT RUN COST CURVES

Short period is the period of time, which is so short that a firm cannot change some of its factors of production like plant, machinery, building etc. "A time period which is less thanthe minimum required to affect changes in factors of production is called as short period". In the short run, quantities of fixed factors cannot be varied in accordance with changes in output. If the firm wants to increase output in the short run, it can do so only with the help of variable factors, i.e., by using more labour, raw materials etc.

#### CHAPTER 10 MONOPOLISTIC COMPETITION DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### MONOPOLISTIC COMPETITION

Monopolistic competition, as the name itself implies, is a blending monopoly and competition. Monopolistic competition refers to the market situation in which a large number of sellers produce goods which enclose substitutes of one another. The products are similar butnot identical. The particular brand of product will have a group of loyal consumers. In this respect, each firm will have some monopoly and at the same time the firm has to competein the market with the other firms as they produce a fair substitute. The essential features of monopolistic competition are product differentiation and existence of many sellers.

The following are the examples of monopolistic competition in Indian context.

- 1. Shampoo Sun Silk, Clinic Plus, Ponds, Chik, Velvette, Kadal, Head and Shoulder, Pantene, Vatika, Garnier, Meera.
- 2. Tooth Paste Binaca, Colgate, Forhans, Close-up, Promise, Pepsodent, Vicco vajradant, Ajanta, Anchor, Babool.

#### CHARACTERISTICS OF MONOPOLISTIC COMPETITION

- i. Existence of Large Number of firms: Under monopolistic competition, the number of firms producing a commodity will be very large. The term "Very large" denotes that contribution of each firm towards the total demand of the product is small. Each firm will act independently on the basis of product differentiation and each firm determines its price-output policies. Any action of the individual firm in increasing or decreasing the output will have little or no effect on other firms.
- ii. Product differentiation: Product differentiation is the essence of monopolistic competition. Product differentiation is the process of altering goods that serve the same purpose so that they differ in mirror ways.

Product differentiation can be brought about in various ways. Product differentiation is attempted through (a)Physical difference; (b)Quality difference; (c)imaginary difference and (d)Purchase benefit difference. It may be by using different quality of the raw material and different chemicals and mixtures used in the product. Difference in workmanship, durability and strength will also make product differentiation. Product differentiation may also be effected by offering customers some benefits with the sale of the product.

iii). Selling Costs: From the discussion of "product differentiation", we can infer that the

producer under monopolistic competition has to incur expenses to popularize his brand. This expenditure involved in selling the product is called selling cost. According to Prof. Chamberlin, selling cost is "the cost incurred in order to alter the position or shape of the demand curve for a product". Most important form of selling cost is advertisement. Sales promotion by advertisement is called non- pricecompetition.



# MARKETING

# EDITED BY



# DR.A.CHANDRASEKARAN



#### MARKETING

Copyright © 2021 by DR. A. CHANDRASEKARAN

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

### ISBN: 978-93-6255-470-3

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

### TABLE OF CONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	INTRODUCTION TO		1
	MARKETING	DR.R.SELVARAJ	
2	MODERN MARKETING		11
	CONCEPT	DR. S. KAMARAJU	
3	PRODUCT LIFE CYCLE	DR.G.KARTHIGA	22
4	MARKETING MIX	DR.V.SRIDEVI	32
5	FACTORS INFLUENCING		42
	PRICING	DR.R.RAJAVARDHINI	
6	DIFFERENT KINDS OF PRICING	DR.D.SILAMBARASAN	52
7	CHANNELS OF DISTRIBUTION	DR.S.RAJENDRAN	62
8	ADVERTISEMENT	DR.R.SELVARAJ	72
9	E-MARKETING	DR. S. KAMARAJU	82
10	M-MARKETING	DR.G.KARTHIGA	92
	REFERENCE		101

#### CHAPTER 1 INTRODUCTION TO MARKETING DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST) INTRODUCTION TO MARKETING

#### TRADITIONAL CONCEPT OF MARKETING

According to the traditional concept, marketing means selling goods and services that have been produced. Thus, all those activities which are concerned with persuasion and

sale of goods and services, are called marketing. This concept of marketing emphasizes on

promotion and sale of goods and services and little attention is paidto consumer satisfaction.

This concept has the following implications:

- The main focus of this concept is on product, i.e., we have a product and it has to be sold. So, we have to persuade the consumers to buy our product.
- All efforts of the marketing people are concentrated on selling the product. They

adopt all means like personal selling and sales promotion to boost the sales.

• The ultimate goal of all marketing activity is to earn profit through

maximization ofsales.

Traditional Concept of Marketing			
Focus on	Product		
Means	Selling		
Ends	Profit through maximization of sales		

#### FEW RELEVANT TERMS ON MARKETING

**Market:** Normally people understand the term market as a place where goods are bought and sold. But, in the context of Marketing, it refers to a group of buyers for a particular product or service. For example, the market for Accountancy textbooks consists of students in Commerce and specialised Accountancy Programmes; the market for ladies readymade garments consists of girls and women, and so on.

#### **Concept of Market:**

• Place Concept: A market is a convenient meeting place of buyers and sellers to gather together in order to conduct buying and selling activities. It is a physical location

#### CHAPTER 2 MODERN MARKETING CONCEPT DR. S. KAMARAJU Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

The modern marketing concept emphasizes customer orientation, focusing on understanding and fulfilling consumer needs to create value. It delves into market research, segmentation, and personalized communication. This approach advocates for building long-term relationships, integrating social responsibility, and leveraging digital tools for engagement. Ultimately, it aims for sustainable business growth by aligning products and services with consumer expectations.

The modern marketing concept can be broken down into several key components:

#### 1. Customer Orientation

- **Understanding Needs:** Businesses prioritize understanding customer needs and preferences through research and feedback.
- Value Creation: Focus on delivering superior value, which includes quality, convenience, and customer service.

#### 2. Market Segmentation

- **Targeting Specific Groups:** Identifying and targeting specific market segments to tailor marketing strategies effectively.
- **Personalization:** Customizing products and messages to fit the unique needs of different customer segments.

#### 3. Integrated Marketing

- **Coordinated Efforts:** Ensuring that all marketing activities and communications are aligned and work together to reinforce the brand message.
- **Multi-Channel Engagement:** Utilizing various channels (social media, email, content marketing) to reach consumers where they are.

#### 4. Relationship Marketing

- Building Loyalty: Fostering long-term relationships with customers to enhance loyalty and repeat business.
- Engagement Strategies: Using CRM systems and personalized communications to maintain ongoing interactions.

#### 5. Digital Transformation

- Utilizing Technology: Leveraging digital tools and platforms for marketing campaigns, data analytics, and customer engagement.
- **Real-Time Feedback:** Collecting and analyzing customer data in real-time to adapt strategies quickly.

#### 6. Social Responsibility

- **Ethical Marketing:** Incorporating ethical considerations and sustainability into marketing practices to resonate with socially conscious consumers.
- **Community Engagement:** Participating in social causes and contributing to the community as part of the brand identity.

#### 7. Continuous Improvement

- Adaptability: Staying flexible and responsive to market changes and consumer feedback.
- **Innovation:** Encouraging a culture of innovation to develop new products and services that meet evolving consumer needs.

#### CHAPTER 3 PRODUCT LIFE CYCLE DR.G.KARTHIGA Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

- Characteristics:
  - Product development and market entry.
  - High costs due to marketing and production.
  - Limited awareness among consumers.
- Marketing Strategies:
  - Launch campaigns focused on creating awareness.
  - Use promotional pricing or incentives to attract early adopters.
  - Educational content to explain the product's benefits and features.

#### Growth

- Characteristics:
  - Rapid increase in sales and market acceptance.
  - New competitors may enter the market.
  - Improved profitability as economies of scale kick in.
- Marketing Strategies:
  - Enhance product features and benefits based on customer feedback.
  - Expand distribution channels to reach a broader audience.
  - $\circ$   $\;$  Increase marketing efforts to solidify brand loyalty and fend off competition.

#### Maturity

- Characteristics:
  - Sales peak and then stabilize or decline.
  - Market saturation occurs; most potential customers have purchased.
  - Intense competition leads to price wars.
- Marketing Strategies:
  - Focus on brand differentiation to stand out from competitors.
  - Explore new market segments or geographic areas for growth.
  - Implement loyalty programs to retain existing customers and encourage repeat purchases.

#### Decline

- Characteristics:
  - Sales and profits decrease due to market saturation, technological advancements, or changing consumer preferences.
  - Products may become obsolete or less relevant.
- Marketing Strategies:
  - Decide whether to rejuvenate the product through innovation or repositioning.
  - Consider discontinuing the product or phasing it out if it no longer contributes to profits.
  - Reduce marketing expenses and focus on loyal customers if continuing sales.

#### Conclusion

Understanding the Product Life Cycle allows marketers to adapt their strategies according to the product's stage, optimizing resource allocation, and maximizing the product's overall success in the market. By anticipating changes in consumer behavior and competitive dynamics, usinesses can make informed decisions to enhance their product offerings and maintain profitability.

#### CHAPTER 4 MARKETING MIX DR.V.SRIDEVI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

The Marketing Mix, often referred to as the "4 Ps," is a foundational framework in marketing that helps businesses develop effective strategies. Here's a detailed explanation of each component: 1. Product

- Definition: Refers to the goods or services offered to meet customer needs.
- Key Considerations:
  - Features and Quality: What attributes make the product unique or superior?
  - Design and Branding: How is the product presented and perceived by consumers?
  - Variety and Packaging: Options available and how the product is packaged to appeal to customers.
  - Lifecycle Management: Understanding the product life cycle stage to adapt marketing strategies.
- 2. Price
  - Definition: The amount customers pay for the product.
  - Key Considerations:
    - Pricing Strategies: Options include cost-plus pricing, competitive pricing, value-based pricing, and penetration pricing.
    - Discounts and Allowances: Temporary price reductions to stimulate demand.
    - Perceived Value: How customers perceive the product's value relative to its price.
    - Market Conditions: Considering competitors' pricing and economic factors that may influence pricing decisions.

#### 3. Place

- Definition: Refers to how the product is distributed and where it is available for purchase.
- Key Considerations:
  - Distribution Channels: Direct sales, retail, online, wholesalers, etc.
  - Coverage: How widely the product is available (intensive, selective, or exclusive distribution).
  - Logistics: Efficient transportation and storage to ensure product availability.
  - Location: Strategic placement of retail outlets or online presence to maximize reach.

#### 4. Promotion

- Definition: The activities that communicate the product's benefits and persuade customers to buy.
- Key Considerations:
  - Advertising: Traditional (TV, print) and digital (social media, online ads).
  - Public Relations: Managing brand image and communications with the public.
  - Sales Promotion: Short-term incentives to encourage purchases (coupons, contests).
  - Personal Selling: Direct interaction with customers to explain product benefits and close sales.
- 5. People

 $\circ$ 

- The employees and staff who interact with customers, impacting service delivery and brand perception.
- 6. Process
  - The systems and procedures involved in delivering the product or service, influencing customer experience.
- 7. Physical Evidence
  - The tangible elements that support the service, such as the environment, brochures, and online presence, which help to reinforce the brand.

#### CHAPTER 5 FACTORS INFLUENCING PRICING DR.R.RAJAVARDHINI Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### FACTORS INFLUENCING PRICING

#### 1. Cost of Production

- **Fixed Costs:** These are expenses that do not change with production volume, such as rent, salaries, and equipment costs. Pricing must cover these costs to ensure the business remains viable.
- **Variable Costs:** These fluctuate with production levels, including raw materials, labor, and utilities. Understanding these costs is essential for setting a price that maintains margins.
- **Total Cost Consideration:** Businesses often calculate the break-even point to determine the minimum price needed to cover costs.

#### 2. Market Demand

- **Elasticity of Demand:** Understanding how sensitive customers are to price changes helps in setting the right price. If demand is elastic, small price changes can significantly affect sales.
- **Consumer Trends:** Seasonal trends, preferences, and social influences can shift demand, necessitating adjustments in pricing strategies.
- Market Research: Conducting surveys and analyzing sales data can provide insights into how much consumers are willing to pay.

#### 3. Competition

- **Direct Competitors:** Pricing should be competitive with similar products in the market. Analyzing competitors' prices helps position the product effectively.
- **Pricing Strategies:** Businesses may adopt various strategies based on competition:
  - **Price Skimming:** Setting a high price initially and lowering it over time, often used for new, innovative products.
  - **Penetration Pricing:** Offering a low initial price to gain market share quickly, then raising it later.
  - **Psychological Pricing:** Setting prices slightly below a round number (e.g., \$9.99 instead of \$10) to make products appear cheaper.

#### 4. Customer Perception

- Value Proposition: The perceived value of a product significantly influences pricing. If customers believe they are getting good value, they are more likely to pay a higher price.
- **Brand Image:** Strong brands can command higher prices due to perceived quality and reputation. Companies invest in branding to enhance customer trust and willingness to pay.
- **Target Market:** Understanding the target audience's demographics, income levels, and preferences helps in setting prices that align with their expectations.

#### 5. Economic Conditions

- **Inflation:** Rising costs can force businesses to increase prices. In high-inflation environments, consumers may be more price-sensitive.
- **Recession:** Economic downturns typically lead to reduced consumer spending, requiring businesses to lower prices or offer discounts to maintain sales.
- **Consumer Confidence:** The overall sentiment regarding economic stability can influence purchasing behavior and, consequently, pricing strategies.

#### CHAPTER 6 DIFFERENT KINDS OF PRICING DR.D.SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **DIFFERENT KINDS OF PRICING :**

#### 1. Cost-Plus Pricing

Cost-plus pricing involves calculating the total cost of producing a product and then adding a markup percentage to determine the selling price. This strategy ensures that all production costs are covered while providing a profit margin. It is straightforward and easy to implement, making it popular among manufacturers and retailers. However, it may not always account for market demand or competitor pricing, which can lead to overpricing or underpricing.

#### 2. Competitive Pricing

Competitive pricing focuses on setting prices based on what competitors are charging for similar products. This approach is common in markets with many similar offerings, as businesses aim to remain attractive to consumers. Companies may choose to price their products slightly lower to gain market share or match competitors' prices. While this strategy helps ensure competitiveness, it can lead to price wars that may diminish profit margins.

#### 3. Value-Based Pricing

Value-based pricing is determined by the perceived value of a product or service to the customer rather than its production cost. This strategy requires in-depth market research to understand how much customers are willing to pay based on the benefits they derive. Businesses using this approach often command higher prices, especially if they can clearly communicate the unique value proposition of their offerings. It encourages innovation and enhances customer satisfaction but requires a strong understanding of customer perceptions.

#### 4. Penetration Pricing

Penetration pricing involves setting a low initial price for a new product to quickly attract customers and gain market share. This strategy is particularly effective in highly competitive markets where establishing a foothold is crucial. After achieving a certain level of market penetration, the price can gradually increase. While this approach can lead to rapid sales growth, it may also result in initial losses and can be challenging to sustain if not carefully managed.

#### 5. Price Skimming

Price skimming is a strategy where a high price is set initially for a new or innovative product, targeting early adopters willing to pay a premium. Over time, the price is gradually lowered to attract more price-sensitive customers. This approach allows businesses to maximize profits from different market segments. It is often used for technology products or luxury items but can be risky if competitors quickly enter the market with similar offerings.

#### 6. Psychological Pricing

Psychological pricing leverages the emotional responses of consumers to price points. Common tactics include pricing products at \$9.99 instead of \$10, creating the perception of a better deal. This

#### CHAPTER 7 CHANNELS OF DISTRIBUTION DR.S.RAJENDRAN Department of Commerce, Ponnaivah Ramajayam Institute of science and Technology(PRIST)

Channels of distribution refer to the pathways through which products and services move from manufacturers to consumers, playing a crucial role in marketing strategies and overall business success. One primary method is **direct distribution**, where companies sell products directly to consumers without intermediaries. This can occur through company websites, physical retail stores, or direct mail and telemarketing. Direct distribution allows for greater control over branding and customer experience, resulting in higher profit margins, though it often comes with increased marketing and logistics costs and limited reach.

In contrast, **indirect distribution** involves intermediaries such as wholesalers, retailers, and distributors, who help facilitate sales. Wholesalers purchase goods in bulk and sell them to retailers or other businesses, while retailers sell products directly to consumers. This approach extends market reach and can reduce the burden on manufacturers regarding distribution logistics, but it also leads to lower profit margins due to commissions or markups, as well as reduced control over branding and customer relationships.

**Multi-channel distribution** incorporates both direct and indirect methods, allowing companies to reach consumers through various platforms, such as their own websites and third-party online marketplaces. This flexibility can enhance visibility and sales opportunities but may create complexity in managing different channels and maintaining consistent branding. An evolution of this approach is **Omni-channel distribution**, which aims to provide a seamless customer experience across all platforms, ensuring that customers can interact with the brand in a cohesive manner. While this can lead to higher customer satisfaction and loyalty, it requires sophisticated technology and coordination, resulting in higher operational costs.

**Wholesale distribution** is another important channel, where large quantities of goods are purchased from manufacturers and sold to retailers. This model benefits retailers through bulk purchasing and alleviates the need for them to handle inventory management. However, it can create dependency on wholesalers for product availability. **Drop shipping** is a fulfillment method gaining popularity, where retailers do not hold inventory; instead, they purchase items from third parties who ship directly to customers. This approach minimizes upfront costs and allows for a wider product range, but it can lead to reduced control over shipping and product quality.

Lastly, **franchise distribution** involves a franchisor allowing a franchisee to operate a business under its brand. This model is common in industries like fast food and retail, offering rapid market expansion with lower capital investment from the franchisor while providing franchisees with established brand recognition. However, franchisees must adhere to specific guidelines set by the franchisor, which can limit operational flexibility. Ultimately, the choice of distribution channels is vital for a company's marketing strategy, requiring careful assessment of the target market, product type, and business goals to determine the most effective approach. Adapting to changing consumer behaviors and technological advancements is essential for maintaining competitiveness in today's market.

#### CHAPTER 8 ADVERTISEMENT DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

Advertising is a form of communication used to promote products, services, or brands to potential customers. It aims to inform, persuade, and remind audiences about offerings, ultimately driving consumer action. Advertising can take various forms, including print media (newspapers, magazines), broadcast (television, radio), digital platforms (social media, websites), and outdoor advertising (billboards, transit ads).

Key Objectives of Advertising

- 1. **Awareness:** The primary goal is to make consumers aware of a product or service. This is especially important for new products entering the market.
- 2. **Interest:** Advertising aims to spark interest by highlighting unique features, benefits, and value propositions that resonate with the target audience.
- 3. **Desire:** Effective advertising creates a desire for the product, appealing to emotions and aspirations to motivate potential customers to want it.
- 4. Action: The ultimate goal is to prompt a specific action, such as making a purchase, visiting a website, or signing up for a newsletter.

Types of Advertising

- **Informative Advertising:** Provides factual information about a product or service, often used for new products or to educate consumers about features and benefits.
- **Persuasive Advertising:** Aims to convince potential customers to choose a specific brand over competitors, often using emotional appeals and strong messaging.
- **Reminder Advertising:** Serves to reinforce brand awareness and remind consumers of existing products, especially useful for established brands.
- **Comparative Advertising:** Highlights differences between a brand and its competitors, aiming to show why it is superior.

Advertising Strategies

- 1. **Target Audience:** Identifying and understanding the target demographic is crucial. Advertisers segment audiences based on factors like age, gender, income, interests, and location to tailor messages effectively.
- 2. **Media Selection:** Choosing the right channels is essential to reach the target audience. This includes deciding between traditional media (TV, radio, print) and digital platforms (social media, online ads).
- 3. **Creative Execution:** The design and messaging of an advertisement play a significant role in its effectiveness. Creativity in visuals, language, and storytelling can capture attention and drive engagement.
- 4. **Budgeting:** Determining the budget for an advertising campaign influences decisions about media placement, creative development, and overall reach.

#### Measurement and Evaluation

Evaluating the effectiveness of advertising campaigns is vital for understanding return on investment (ROI). Metrics such as reach, impressions, click-through rates, conversion rates, and brand recall are commonly analyzed to assess performance. This data helps refine future campaigns and optimize advertising strategies.

#### CHAPTER 9 E-MARKETING DR. S. KAMARAJU Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Introduction of E-Marketing :**

E-marketing, or electronic marketing, refers to the use of digital technologies and the internet to promote products and services. It encompasses a wide range of strategies aimed at reaching and engaging consumers online, making it essential in today's digital landscape where many people rely on the internet for information, shopping, and social interaction. One of the key components of e-marketing is website marketing, which serves as the central hub for all online efforts. A well-designed website should be user-friendly, visually appealing, and optimized for search engines (SEO), providing valuable content and clear calls-to-action to enhance the user experience.

Search engine optimization is crucial for improving a website's visibility in search engine results, allowing potential customers to find the business more easily. This involves keyword research, on-page optimization, and link-building strategies. Additionally, content marketing plays a significant role by creating and sharing valuable, relevant content to attract and engage a target audience. Formats like blogs, videos, infographics, and eBooks help establish authority and build trust. Email marketing is another effective tactic that involves sending targeted messages to a subscriber list, nurturing leads, and maintaining customer relationships through personalized communication.

Social media marketing leverages platforms such as Facebook, Instagram, Twitter, and LinkedIn to connect with audiences, promote content, and drive traffic to websites, fostering interaction and community engagement. Pay-per-click (PPC) advertising allows businesses to place ads on search engines and social media platforms, paying only for clicks, which enables precise targeting and immediate visibility. Affiliate marketing involves partnering with affiliates who promote products in exchange for a commission on sales, expanding reach through influencer marketing. Collaborating with influencers—individuals with a substantial online following—can effectively drive engagement and sales through authentic endorsements.

E-marketing offers numerous benefits, including cost-effectiveness, targeted reach, measurable results, and the ability to connect with global audiences. However, it also presents challenges, such as intense competition, changing algorithms of search engines and social media platforms, and increasing regulations regarding data privacy. In conclusion, e-marketing is a dynamic and essential aspect of modern business strategies, enabling brands to effectively connect with their target audiences and achieve marketing goals in the evolving digital landscape.

#### **Challenges of E-Marketing :**

- **Competition:** The digital space is crowded, making it essential for businesses to differentiate themselves to stand out.
- **Changing Algorithms:** Search engines and social media platforms frequently update their algorithms, impacting visibility and reach.
- **Privacy Concerns:** Increasing regulations regarding data privacy and consumer rights require businesses to navigate e-marketing practices carefully.

#### CHAPTER 10 M-MARKETING DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

M-marketing, or mobile marketing, refers to marketing strategies and tactics specifically designed for mobile devices, such as smartphones and tablets. As mobile usage continues to rise, m-marketing has become an essential component of modern marketing strategies, allowing businesses to reach consumers directly through the devices they use most frequently.

#### **Key Components of M-Marketing**

- 1. **Mobile-Friendly Websites:** Creating websites optimized for mobile devices is crucial, as users expect seamless navigation and fast loading times. Responsive design ensures that content adjusts to various screen sizes, enhancing user experience.
- 2. **Mobile Apps:** Many businesses develop dedicated mobile applications that provide additional value to consumers, such as exclusive content, rewards programs, or convenient shopping experiences. Apps facilitate direct engagement and enhance brand loyalty.
- 3. **SMS and MMS Marketing:** Short Message Service (SMS) and Multimedia Messaging Service (MMS) campaigns enable businesses to send promotional messages, alerts, and updates directly to consumers' phones. These messages can be highly effective due to their immediacy and high open rates.
- 4. **Location-Based Marketing:** Utilizing GPS technology, businesses can send targeted promotions and advertisements to consumers based on their geographic location. This strategy allows for personalized marketing and can drive foot traffic to physical stores.
- 5. **Mobile Advertising:** This includes various ad formats specifically designed for mobile devices, such as banner ads, interstitials, and native ads on social media platforms. Mobile advertising allows businesses to target audiences based on behavior, interests, and demographics.
- 6. **Social Media Integration:** With a significant portion of social media engagement happening on mobile devices, integrating social media strategies into m-marketing is essential. Brands can reach consumers through mobile-optimized posts, stories, and ads on platforms like Instagram, Facebook, and Twitter.

#### **Benefits of M-Marketing**

M-marketing offers several advantages, including enhanced customer engagement, as mobile devices allow for instant communication and interaction. It also provides the ability to reach consumers on the go, capturing their attention at critical moments. Additionally, m-marketing can deliver personalized experiences based on user behavior and preferences, increasing conversion rates and customer satisfaction.

# **BUSINESS** ENVIRONMENT

# Edited by DR.S.RAJENDRAN



#### **BUSINESS ENVIRONMENT**

#### Copyright © 2021 by DR.S.RAJENDRAN

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

#### ISBN: 978-93-6255-452-9

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

## TABLE OF CONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	INTRODUCTION TO BUSINESS		1
	ENVIRONMENT	DR.R.SELVARAJ	
2	FACTORS INFLUENCING		11
	BUSINESS ENVIRONMENT	DR. S. KAMARAJU	
3	CULTURAL ENVIRONMENT	DR.G.KARTHIGA	22
4	SOCIAL RESPONSIBILITY OF		32
	BUSINESS	DR.V.SRIDEVI	
5	POLITICAL ENVIRONMENT	DR.R.RAJAVARDHINI	42
6	CONCEPTS OF ECONOMIC		52
	ENVIRONMENT	DR.D.SILAMBARASAN	
7	FINANCIAL ENVIRONMENT	DR.S.RAJENDRAN	62
8	CAPITAL MARKET	DR.R.SELVARAJ	72
9	DEVELOPMENT BANKS	DR. S. KAMARAJU	82
10	TECHNOLOGICAL		92
	ENVIRONMENT	DR.G.KARTHIGA	
	REFERENCE		101

#### CHAPTER 1 INTRODUCTION TO BUSINESS ENVIRONMENT DR.R.SELVARAJ

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST) Understanding the Business Environment :

#### Introduction

The business environment is a complex system of external and internal factors that influence how organizations operate and succeed. This environment encompasses economic, political, social, technological, competitive, and natural elements that shape business decisions and strategies. Understanding the business environment is crucial for companies aiming to adapt and thrive in a dynamic marketplace.

Body

**1. Economic Environment** The economic environment includes factors such as economic growth, inflation, interest rates, and employment levels. For example, a booming economy typically encourages consumer spending and investment, while a recession may lead businesses to tighten their budgets and reconsider their growth strategies. Understanding these economic indicators enables companies to make informed decisions regarding resource allocation and expansion.

**2. Political and Legal Environment** The political and legal environment consists of government regulations, political stability, and legal requirements that affect business operations. Organizations must navigate laws related to employment, trade, and environmental protection. For instance, changes in trade policies can significantly impact supply chains and market access, making it essential for businesses to stay informed about legislative developments.

**3. Social and Cultural Environment** This aspect focuses on societal values, demographics, and cultural trends. Consumer preferences are heavily influenced by social factors, including lifestyle changes and cultural shifts. Businesses that understand their target audience can develop products and marketing strategies that resonate with consumers, enhancing customer loyalty and satisfaction.

**4. Technological Environment** Technology plays a pivotal role in shaping the business landscape. Innovations can streamline operations, improve communication, and enhance product offerings. However, companies must also be aware of the potential disruptions caused by technological advancements. Staying ahead of technological trends is vital for maintaining a competitive edge.

**5. Competitive Environment** The competitive environment involves analyzing competitors and market dynamics. Understanding the strengths and weaknesses of rival firms allows businesses to identify opportunities for differentiation. By conducting thorough market research, organizations can better position themselves to capture market share and respond to consumer needs effectively.

**6. Natural Environment** The natural environment includes ecological factors and sustainability concerns. With growing awareness of environmental issues, businesses are increasingly expected to adopt sustainable practices. This not only addresses consumer demand for eco-friendly products but also helps mitigate regulatory risks and enhance corporate reputation.

#### Conclusion

In conclusion, the business environment is multifaceted, encompassing a wide range of factors that influence organizational success. By understanding and adapting to these elements—economic, political, social, technological, competitive, and natural—companies can formulate effective strategies, manage risks, and seize opportunities for growth. Continuous monitoring and analysis of

#### CHAPTER 2 FACTORS INFLUENCING BUSINESS ENVIRONMENT DR. S. KAMARAJU Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

Factors Influencing the Business Environment

#### Introduction

The business environment is a dynamic landscape shaped by various factors that impact organizational operations and strategies. Understanding these factors is essential for businesses to adapt, thrive, and maintain a competitive edge. This essay explores the key factors influencing the business environment, including economic, political, social, technological, competitive, natural, and global influences.

**1. Economic Factors** Economic conditions play a pivotal role in shaping the business environment. Factors such as economic growth, inflation rates, interest rates, and unemployment levels directly affect consumer spending and business investment. For instance, during periods of economic expansion, consumers are more likely to spend, boosting demand for goods and services. Conversely, high inflation can erode purchasing power, prompting businesses to reconsider pricing strategies and cost management.

**2. Political and Legal Factors** The political and legal landscape significantly impacts business operations. Government policies, political stability, and legal regulations dictate how businesses function within a society. For example, changes in taxation policies or labor laws can have profound effects on operational costs and workforce management. A stable political environment fosters investment confidence, while political instability can lead to uncertainty and risk for businesses.

**3. Social and Cultural Factors** Social and cultural influences are crucial in shaping consumer behavior and preferences. Demographics, including age, gender, and income levels, affect market demand and marketing strategies. Additionally, cultural trends and societal values influence consumer loyalty and brand perception. Companies increasingly recognize the importance of social responsibility, as consumers demand ethical practices and sustainable products.

**4. Technological Factors** Technology is a driving force in the modern business environment. Innovations can lead to the creation of new products, enhance operational efficiency, and transform marketing strategies. The rise of digital platforms and e-commerce has revolutionized consumer engagement. However, businesses must also navigate the challenges of automation, which can displace jobs and require workforce reskilling.

**5.** Competitive Factors The competitive landscape shapes how businesses strategize and position themselves in the market. The level of competition influences pricing strategies, product differentiation, and customer engagement. Understanding consumer preferences and competitor strengths and weaknesses is vital for businesses aiming to capture market share. Barriers to entry, such as capital requirements and regulatory challenges, further impact the competitive dynamics within an industry.

**6. Natural Environment** The natural environment encompasses ecological factors that influence business operations. Compliance with environmental regulations is essential for sustainable practices. Increasing awareness of climate change and ecological sustainability has prompted businesses to adopt greener practices. Moreover, natural disasters can disrupt operations and supply chains, highlighting the need for businesses to have contingency plans in place.

#### CHAPTER 3 CULTURAL ENVIRONMENT DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

In today's interconnected world, the cultural environment plays a pivotal role in shaping our identities and interactions. It encompasses the values, beliefs, customs, and behaviors of a society, influencing everything from personal choices to economic practices. Understanding the cultural environment is essential for navigating diverse social landscapes, as it affects how individuals perceive themselves and relate to others. This essay explores the components of the cultural environment, its impact on individual identity, social interactions, and economic behavior.

#### Definition and Components of Cultural Environment

The cultural environment refers to the complex system of shared values and norms that characterize a community. Key components include traditions, language, religion, social structures, and art. For instance, traditions such as festivals and rituals provide a sense of belonging and continuity, while language serves as a vital tool for communication and cultural expression. Religion shapes moral values and community cohesion, influencing both individual behavior and social dynamics. Each of these elements plays a crucial role in defining the cultural landscape of a society.

#### Influence on Individual Identity

Culture profoundly influences personal identity and self-perception. From a young age, individuals absorb cultural norms through family, education, and community interactions. For example, a child raised in a collectivist culture may prioritize community and familial responsibilities over individual desires, shaping their worldview and decision-making processes. Conversely, those from individualistic cultures may emphasize personal achievement and independence. These cultural underpinnings not only affect self-identity but also guide how individuals navigate their social environments.

#### Impact on Social Interactions

The cultural environment significantly shapes social interactions, influencing how people communicate and relate to one another. Different cultures have varying norms regarding politeness, body language, and expressions of emotion. For instance, in some cultures, direct eye contact is seen as a sign of confidence, while in others, it may be viewed as disrespectful. Understanding these nuances is essential for effective communication and relationship-building, especially in multicultural settings. Cultural misunderstandings can lead to conflicts, emphasizing the need for cultural awareness in our increasingly globalized world.

#### **Economic Implications**

Cultural values also have a profound impact on economic behavior and consumer choices. For example, cultures that prioritize collectivism may favor cooperative business practices, while individualistic societies might emphasize competition and personal achievement. Additionally, cultural preferences can influence purchasing decisions, as consumers often gravitate towards products that resonate with their cultural identity. Understanding these cultural dimensions is crucial for businesses looking to succeed in diverse markets, as it allows them to tailor their strategies to meet the unique needs of different consumer groups.

#### Conclusion

In summary, the cultural environment is a foundational aspect of human society that influences individual identities, social interactions, and economic behavior. As globalization continues to connect

#### CHAPTER 4 SOCIAL RESPONSIBILITY OF BUSINESS DR.V.SRIDEVI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

In today's interconnected world, the social responsibility of business has become a vital topic of discussion. Companies are no longer viewed solely as profit-driven entities; they are increasingly expected to consider their impact on society and the environment. Social responsibility encompasses various practices, including ethical labor practices, environmental sustainability, community engagement, and corporate governance. This essay explores the importance of corporate social responsibility (CSR), the benefits it brings to businesses and communities, and the challenges organizations face in implementing effective CSR strategies.

#### Definition and Importance of CSR

Corporate social responsibility refers to the obligation of businesses to act in ways that benefit society and contribute to sustainable development. It emphasizes the idea that companies should operate ethically and consider the broader impact of their decisions. CSR is essential not only for fostering trust among consumers but also for enhancing brand reputation. As consumers become more conscious of social and environmental issues, they are more likely to support businesses that align with their values. Therefore, embracing CSR can lead to increased customer loyalty and long-term success.

#### Benefits of CSR

Implementing effective CSR strategies offers numerous benefits for businesses. Firstly, companies that prioritize social responsibility often experience improved employee morale and retention, as workers are more engaged when they believe their employer is making a positive impact. Additionally, CSR initiatives can lead to cost savings, particularly in areas like energy efficiency and waste reduction. Furthermore, businesses that actively contribute to their communities can foster positive relationships with local stakeholders, enhancing their social license to operate and mitigating potential conflicts.

#### Challenges in Implementing CSR

Despite its benefits, many organizations face challenges when integrating CSR into their operations. One significant hurdle is the potential conflict between short-term profits and long-term sustainability goals. Businesses may struggle to balance shareholder expectations with the need for social and environmental initiatives, particularly in competitive markets. Additionally, measuring the impact of CSR efforts can be complex, making it difficult for companies to assess their effectiveness and communicate their contributions to stakeholders. Overcoming these challenges requires a strategic approach that aligns CSR with the core business strategy.

#### Conclusion

In conclusion, the social responsibility of business is a crucial aspect of modern corporate practice. By recognizing their role in society, businesses can contribute to positive change while also reaping the benefits of enhanced reputation and stakeholder trust. Although challenges exist in implementing effective CSR strategies, the potential rewards make it a worthwhile endeavor. As consumers continue to demand more ethical practices from businesses, those that embrace social responsibility will likely thrive in the evolving marketplace.

#### CHAPTER 5 POLITICAL ENVIRONMENT DR.R.RAJAVARDHINI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

The political environment plays a crucial role in shaping the business landscape and influencing economic activities. It encompasses the laws, regulations, government policies, and political stability that affect how businesses operate. Understanding the political environment is essential for companies, as it can impact everything from market entry strategies to operational practices. This essay explores the components of the political environment, its influence on businesses, and the challenges that companies may face within different political contexts.

#### Components of the Political Environment

The political environment comprises several key components, including government policies, political stability, legal frameworks, and international relations. Government policies, such as taxation and trade regulations, directly affect business operations and profitability. Political stability is crucial for creating a favorable climate for investment, as instability can lead to uncertainty and risk for businesses. Additionally, legal frameworks govern how companies operate, including labor laws, environmental regulations, and intellectual property protections. Understanding these components helps businesses navigate the political landscape effectively.

#### Influence on Business Operations

The political environment significantly influences business operations and decision-making processes. For instance, changes in government policies can impact industry regulations, leading companies to adapt their strategies accordingly. A supportive political environment can foster innovation and growth, while a restrictive one may hinder competition and limit market opportunities. Moreover, businesses must be aware of lobbying efforts and political advocacy, as these can shape legislation that affects their industry. Companies that proactively engage with policymakers are better positioned to influence outcomes that align with their interests.

#### Challenges in Navigating the Political Environment

Despite its importance, navigating the political environment presents several challenges for businesses. Fluctuating political climates, such as changes in leadership or policy direction, can create uncertainty and complicate long-term planning. Additionally, businesses operating in multiple countries must contend with varying political systems and regulatory environments, which can complicate compliance and strategy formulation. Companies may also face ethical dilemmas, particularly when political interests conflict with social responsibility. Effectively managing these challenges requires a nuanced understanding of the political landscape and strategic foresight.

#### Conclusion

In conclusion, the political environment is a critical factor that influences business operations and overall economic conditions. By understanding the components and implications of the political environment, companies can better navigate challenges and seize opportunities for growth. While the political landscape can be complex and unpredictable, proactive engagement with policymakers and awareness of regulatory changes can position businesses for success in an ever-evolving political context. Ultimately, recognizing the significance of the political environment is essential for informed decision-making and strategic planning.

#### CHAPTER 6 CONCEPTS OF ECONOMIC ENVIRONMENT DR.D.SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

The economic environment encompasses the various economic factors that influence the operations of businesses and the decisions of consumers. These factors include economic growth, inflation rates, unemployment levels, interest rates, and the overall economic stability of a region or country. Understanding the economic environment is crucial for businesses as it affects market demand, pricing strategies, and investment decisions. This essay explores key concepts of the economic environment and their implications for businesses.

#### Economic Growth

Economic growth is a fundamental concept within the economic environment, typically measured by the increase in a country's gross domestic product (GDP). When the economy is growing, businesses often experience increased consumer spending and demand for products and services. This growth can lead to greater investment opportunities and expansion potential. Conversely, during periods of economic contraction, companies may face reduced demand, leading to lower revenues and potential layoffs. Understanding the cycle of economic growth is vital for strategic planning and resource allocation.

#### Inflation and Interest Rates

Inflation refers to the rate at which prices for goods and services rise, eroding purchasing power. High inflation can affect consumer behavior, as rising prices may lead to decreased spending. Businesses must adapt by adjusting pricing strategies to maintain profitability while remaining competitive. Interest rates, set by central banks, are another critical component of the economic environment. Higher interest rates increase borrowing costs for businesses and consumers, which can slow down investment and spending. Conversely, lower interest rates typically encourage borrowing and stimulate economic activity. Companies need to monitor these factors to make informed financial decisions.

#### Unemployment and Labor Market Conditions

Unemployment levels reflect the health of the labor market and the economy as a whole. High unemployment can indicate economic distress, leading to reduced consumer spending and lower demand for goods and services. Businesses may respond by scaling back operations or delaying expansion plans. On the other hand, low unemployment often results in a tight labor market, where companies compete for talent, potentially increasing wage costs. Understanding labor market conditions helps businesses make strategic decisions about hiring, training, and compensation.

#### Conclusion

In conclusion, the economic environment is a vital consideration for businesses operating in today's dynamic market. Key concepts such as economic growth, inflation, interest rates, and unemployment significantly impact business strategies and decision-making processes. By staying informed about these economic factors, companies can better navigate challenges and seize opportunities, ultimately positioning themselves for long-term success in a constantly changing economic landscape.

#### CHAPTER 7 FINANCIAL ENVIRONMENT DR.S.RAJENDRAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

The financial environment encompasses the systems, institutions, and regulations that govern the flow of money and the management of financial resources within an economy. It plays a critical role in shaping business strategies, influencing investment decisions, and determining access to capital. Key components of the financial environment include financial markets, banking systems, investment vehicles, and regulatory frameworks. Understanding these elements is essential for businesses to navigate financial challenges and leverage opportunities for growth. This essay delves into the various aspects of the financial environment and their implications for businesses.

#### **Financial Markets**

Financial markets serve as platforms for the buying and selling of financial assets, such as stocks, bonds, commodities, and currencies. They are classified into primary markets, where new securities are issued, and secondary markets, where existing securities are traded. The efficiency and liquidity of these markets are vital for businesses seeking to raise capital. For example, a strong stock market can provide companies with the ability to issue shares to fund expansion projects, while a healthy bond market allows them to secure long-term financing at favorable interest rates. Additionally, fluctuations in financial markets can impact consumer confidence and spending, affecting overall demand for goods and services. Businesses must monitor market trends and investor sentiment to adapt their strategies and capitalize on financing opportunities.

#### **Banking System**

The banking system is a cornerstone of the financial environment, acting as an intermediary between savers and borrowers. Banks provide essential services, including savings accounts, loans, and payment processing. Access to credit is particularly crucial for businesses, as it enables them to invest in infrastructure, technology, and human resources. Small and medium-sized enterprises (SMEs), in particular, often rely on banks for loans to manage cash flow and finance growth. The stability of the banking system also influences economic confidence; when banks are stable, businesses are more likely to invest and expand. Additionally, businesses must cultivate relationships with financial institutions to ensure they can access necessary funding and favorable terms. The recent rise of fintech companies has also transformed the banking landscape, providing alternative lending solutions that enhance financial inclusion.

#### **Investment Vehicles**

Investment vehicles are instruments that allow individuals and institutions to invest their capital with the expectation of earning returns. Common investment vehicles include stocks, bonds, mutual funds, and exchange-traded funds (ETFs). For businesses, understanding the characteristics and risks associated with different investment vehicles is essential for managing excess cash and optimizing returns. Additionally, businesses may engage in strategic investments in other companies or projects to drive growth and diversification. For instance, venture capital investments in startups can provide innovative opportunities and potential high returns. However, businesses must also be

#### CHAPTER 8 CAPITAL MARKET DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

Capital markets are vital components of the financial environment, serving as platforms for the buying and selling of long-term debt and equity instruments. These markets facilitate the flow of capital from investors to businesses and governments, enabling funding for various projects and initiatives. Capital markets can be classified into primary markets, where new securities are issued, and secondary markets, where existing securities are traded. This essay delves into the key features of capital markets, their importance for businesses and investors, and the challenges they face.

#### Features of Capital Markets

Capital markets consist of two primary segments: equity markets and debt markets.

- Equity Markets: In equity markets, companies issue shares to raise capital. Investors buy these shares, gaining ownership stakes in the company and the potential for capital appreciation and dividends. Stock exchanges, such as the New York Stock Exchange (NYSE) and NASDAQ, provide platforms for trading these shares.
- **Debt Markets**: In debt markets, businesses and governments issue bonds to borrow funds from investors. These bonds represent a loan that must be repaid with interest over a specified period. Investors in the debt market receive fixed interest payments and the return of principal upon maturity.

Both segments play crucial roles in facilitating investment, promoting liquidity, and determining the cost of capital.

#### Importance of Capital Markets

Capital markets are essential for several reasons:

- 1. **Funding for Growth**: Businesses rely on capital markets to secure financing for expansion, research and development, and operational needs. Access to capital allows companies to invest in new projects, hire employees, and enhance productivity.
- 2. **Liquidity for Investors**: Capital markets provide liquidity, enabling investors to buy and sell securities easily. This liquidity encourages investment by reducing the risks associated with holding assets over long periods.
- 3. **Price Discovery**: Capital markets facilitate price discovery, allowing the market to determine the fair value of securities based on supply and demand dynamics. This process helps investors make informed decisions about buying and selling assets.
- 4. **Economic Indicators**: The performance of capital markets often reflects the overall health of the economy. Rising stock prices can indicate economic growth, while declining markets may signal recessionary pressures.

#### Challenges in Capital Markets

Despite their importance, capital markets face several challenges:

#### CHAPTER 9 DEVELOPMENT BANKS DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

Development banks are specialized financial institutions that play a critical role in promoting economic growth and development, particularly in emerging and developing economies. They provide essential funding for various sectors, including infrastructure, agriculture, education, and healthcare, helping to address gaps that traditional commercial banks often overlook. By offering loans, grants, and equity investments, development banks facilitate projects that might otherwise struggle to secure financing due to high risk or lower returns.

These banks can be categorized into several types. Multilateral development banks (MDBs), such as the World Bank and the Asian Development Bank, engage in cross-border financing and provide expertise across multiple nations. Bilateral development banks, like the U.S. International Development Finance Corporation, focus on partnerships between two countries, often targeting specific developmental objectives. National development banks operate within individual countries, aligning their efforts with national economic policies and priorities.

The impact of development banks is far-reaching. They are instrumental in financing large-scale infrastructure projects, such as roads, bridges, and energy systems, which are vital for economic connectivity and growth. Additionally, they support social programs aimed at improving health and education, contributing to poverty alleviation and overall quality of life. By promoting job creation and entrepreneurship, development banks stimulate local economies and help foster a more sustainable economic environment.

Despite their significant contributions, development banks face several challenges. One major concern is ensuring that projects financed do not lead to unsustainable debt levels for borrowing countries. Development banks must carefully assess the financial viability of projects and work with governments to establish sound fiscal policies. Moreover, measuring the effectiveness and long-term impact of funded initiatives can be complex, necessitating robust monitoring and evaluation frameworks. Political influences can also complicate operations, as development banks must navigate varying governance structures and policy priorities in the countries they serve.

In summary, development banks are essential actors in the global effort to achieve sustainable development. By bridging the financing gap and investing in crucial projects, they help create a foundation for economic stability and growth in underserved regions. Their ability to adapt to the evolving needs of the economies they serve, while addressing challenges, will determine their continued relevance and effectiveness in the years to come.

The impact of development banks is significant, as they help finance vital infrastructure projects, support health and education initiatives, and create job opportunities, thereby stimulating local economies. However, they also face challenges, such as ensuring debt sustainability for borrowing countries and effectively measuring the impact of their funded projects. Additionally, they must navigate the complexities of political environments, which can influence their operations. Overall, development banks play a crucial role in bridging the financing gap in underserved areas, making them essential for achieving sustainable development goals.

#### CHAPTER 10 TECHNOLOGICAL ENVIRONMENT DR.G.KARTHIGA Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Technological Environment: An Overview

The technological environment plays a pivotal role in shaping how businesses operate and interact with their stakeholders. Defined by the landscape of innovations, advancements, and tools, this environment encompasses a broad array of factors, including emerging technologies, infrastructure, and regulatory frameworks. Understanding the significance of the technological environment is essential for organizations aiming to remain competitive in an increasingly dynamic marketplace.

#### Significance of the Technological Environment

One of the primary aspects of the technological environment is its ability to drive innovation and enhance competitiveness. Businesses that effectively leverage technology can create new products and services, providing them with a distinct advantage over competitors. For instance, the integration of advanced technologies like artificial intelligence (AI) and automation can streamline processes, reduce operational costs, and improve productivity. Companies that embrace these innovations are often better positioned to respond to changing market demands and consumer preferences.

Moreover, the technological environment significantly influences market trends. The rise of ecommerce, mobile technology, and social media has transformed consumer behavior, compelling businesses to adapt their strategies accordingly. Organizations that fail to keep pace with these technological shifts risk losing relevance in the eyes of their customers. Additionally, technology facilitates globalization by enhancing connectivity and enabling businesses to enter international markets with greater ease. This interconnectedness fosters both competition and collaboration on a global scale.

#### Components of the Technological Environment

The technological environment comprises several key components. Emerging technologies, such as blockchain, machine learning, and the Internet of Things (IoT), are reshaping various industries. Staying informed about these advancements is crucial for strategic planning and long-term success. Furthermore, a robust technological infrastructure—characterized by reliable internet access, communication networks, and data storage solutions—serves as the backbone for implementing new technologies.

Investment in research and development (R&D) is another critical component of the technological environment. Organizations that prioritize R&D can create cutting-edge solutions that meet market needs and drive further innovation. Additionally, the regulatory framework surrounding technology—encompassing data protection laws, intellectual property rights, and industry standards—can significantly impact business operations. Compliance with these regulations is essential for maintaining consumer trust and safeguarding sensitive information.

#### Challenges in the Technological Environment

Despite its many advantages, the technological environment also presents several challenges. One notable issue is the rapid pace of technological change, which can make it difficult for businesses to keep up. Organizations must remain agile and adaptable to navigate this landscape successfully. Cybersecurity risks represent another significant challenge, as increasing reliance on technology exposes businesses to potential data breaches and cyberattacks. Protecting sensitive information and implementing robust security measures is vital for safeguarding organizational integrity.

# ACCOUNTING TRANSACTION

## EDITED BY

## **DR.A.CHANDRASEKARAN**



#### ACCOUNTING TRANSACTION

Copyright © 2021 by DR.A. CHANDRASEKARAN

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

#### ISBN: 978-93-6255-980-7

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

### **TABLE OF CONTENTS**

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	INTRODUCTION OF		1
	ACCOUNTING		
	TRANSACTION	DR.R.SELVARAJ	
2	SINGLE ENTRY SYSTEM		11
		DR. S. KAMARAJU	
3	FINAL ACCOUNTS	DR.G.KARTHIGA	22
4	<b>DEPRECIATION PROVISIONS</b>		32
	AND RESERVES	DR.V.SRIDEVI	
5	PROCEDURES OF LEDGER		42
	ACCOUNT	DR.R.RAJAVARDHINI	
6	SUBSIDIARY BOOKS		52
		DR.D.SILAMBARASAN	
7	CASH BOOK		62
		DR.S.RAJENDRAN	
8	TYPES OF ERRORS	DR.R.SELVARAJ	72
9	FINANCIAL STATEMENT OF		82
	PROPRIETORSHIP	DR. S. KAMARAJU	
10	STATEMENT OF AFFAIRS	DR.G.KARTHIGA	92
	REFERENCE		101

#### CHAPTER 1 INTRODUCTION TO ACCOUNTING TRANSACTIONS DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

## ACCOUNTING TRANSACTIONS

- 1. **Transactions-** the first task of accounting is to identify the transactions of financial character and measure them in terms of money. Transactions may include a debt payoff, any purchases or acquisition of assets, sales revenue, or any expenses incurred.
- 2. Journal Entries- With the transactions set in place, the next step is to record theseentries in the company's journal in chronological order. In debiting one or more accounts and crediting one or more accounts, the debits and credits must always balance.
- **3**. **Posting to the General Ledger (GL)** -The journal entries are then posted to the generalledger where a summary of all transactions to individual accounts can be seen.
- 4. **Trial Balance -** At the end of the accounting period (which may be quarterly, monthly, or yearly, depending on the company), a total balance is calculated for the accounts.
- Financial Statements- The balance sheet, income statement, and cash flow statementcan be prepared using the correct balances.



#### What are Accounting Transactions?

Accounting transactions refer to any business activity that results in a direct effect on the financial status and financial statements of the business. Such transactions come in many forms, including:

- Sales in cash and credit to customers
- Receipt of cash from a customer by sending an invoice
- Purchase of fixed assets and movable assets
- Borrowing funds from a creditor
- Paying off borrowed funds from a creditor
- Payment of cash to a supplier from a sent invoice.

#### **CHAPTER 2**

### SINGLE ENTRY SYSTEM

#### **DR. S. KAMARAJU** Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Single Entry System

The term 'Single Entry' refers to a method of maintaining the accounts in a manner convenient to a business house, which does not exactly follow the principles of double entry system. Under this system only the minimum accounts, which in the opinion of business house are absolutely essential, are being maintained (*i.e. Normally cash book and personal accounts of debtors & creditors are maintained*). The other impersonal accounts i.e.real and nominal accounts may not be maintained. In other words single entry system is incomplete form of account keeping.

Special Features:

The following are the special features of single entry system :

- 1) Unsuitable for big business :This system is suitable only for small business carried on proprietary or partnership basis. Big businesses especially Joint Stock companies cannot afford to maintain accounts as per this system, which is defective and unscientific.
- 2) Only personal and cash Accounts : Under this system only the personal accounts of debtors and creditors as well as cash and bank accounts are maintained. The impersonal accounts (i.e. real & nominal accounts) are not maintained.
- **3) Incomplete account keeping :** This system records double effect of only some transactions. It record only the single aspect of many transactions while it fails to records few other transactions. Hence it is incomplete, defective and crude system.
- 4) Variations : The single entry system of accounting varies from business to business. Hence it lacks uniformity.
- 5) Flexible : No rigid rules and principles are followed under this system.

#### Here are some things to know about single-entry accounting:

Advantages

It's simple, economical, and can be maintained by anyone without a professional accountant. Disadvantages

• Disadvantages

It can result in incomplete records that are inconsistent and unscientific, and don't allow for financial analysis and reporting.

• Suitability

It's best for small, simple businesses at the beginning of their growth. For businesses that are expected to grow, double-entry accounting is recommended.

How it works

#### CHAPTER 3 FINAL ACCOUNTS DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### FINAL ACCOUNTS

The most important function of an accounting system is to provide information about the profitability of the business. A sole trader furnishes a Trading and Profit and loss Account which depicts the result of the business transactions of the sole trader. Along with the Trading and Profit and Loss Account he also prepares a Balance Sheet which shows the financial position of the business.

#### STEPS IN THE PROCESS OF FINALIZATION OF ACCOUNTS

#### For Trading Concerns:

- Trading Account.
- Profit and Loss Account.
- Balance Sheet.

#### For Manufacturing and Trading Concerns:

Manufacturing Account. Trading Account. Profit and Loss Account. Balance Sheet.

#### **Preparation of Financial Statements**

**Profitability Statement** – This statement is related to a complete accounting period. It shows the outcome of business activities during that period in a summarized form. The activities of any business will include purchase, manufacture, and sell.

**Balance Sheet** – Business needs some resources which have longer life (say more than a year). Such resources are, therefore, not related to any particular accounting period, but are to be used over the useful life thereof. The resources do not come free. One requires finance to acquire them. This funding is provided by owners through their investment, bank & other through loans, suppliers by way of credit terms. The Balance Sheet shows the list of resources and the funding of the resources i.e. assets and liabilities (towards owners and outsiders). It is also referred as sources of funds (i.e. liabilities & capital) and application of funds (i.e. assets). Let us discuss these statements in depth.

**Trading Account:** It is an account which is prepared by a merchandising concern which purchases goods and sells the same during a particular period. The purpose of it to find out the gross profit or gross loss which is an important indicator of business efficiency.

#### CHAPTER 4

#### **DEPRECIATION PROVISIONS AND RESERVES**

#### DR.V.SRIDEVI

Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **RESERVES AND PROVISIONS**

#### **Meaning of Reserves**

Reserves are part of profits or gain that has been allotted for a specific purpose. Reserves are usually set up to buy fixed assets, pay bonuses, pay an expected legal settlement, pay for repairs & maintenance and pay off debt.

#### What are Reserves in Accounting?

When an enterprise earns a profit during the end of a year, a certain part of it is retained in the trading concern to meet future exigencies, growth outlooks etc., The amount of money that is kept aside is known as Reserves in Accounting.

They assist in securing the financial situation of an enterprise and can be utilized for different purposes such as stable dividend repayments, expansion, meeting contingencies, legal requirements, investments, improving the financial situation, etc., It is also termed as retained earnings.

For instance – Reserve for Dividends Equalization, General reserve, Reserve for Increased Cost of Replacement, Reserve for Expansion, etc.,

#### **Inside Financial Statements**

It is shown on the liability side of a balance sheet (B/S) below the heading 'Reserves and Surplus' with capital if an enterprise suffers losses, then it is not created.

#### **Types of Reserves**

- Capital Reserve
- Revenue Reserve

#### What is Capital Reserve?

The capital reserve is established out of capital profits and are normally not allocated as dividends to the shareholders. It cannot be established out of profits acquired from core operations of an enterprise.

#### Examples

- Profit earned before an enterprise's embodiment
- Premium acquired on the issue of debentures and shares
- Gain on re-issuance (redistribution) of forfeited shares
- Profit kept aside for redemption of debentures or preference shares
- Gain on sale of fixed assets
- The surfeit on revaluation of liabilities and assets
- Capital redemption reserve

#### What is Revenue Reserve?

Revenue reserves are established out of gains acquired from operations of an enterprise. Reserves are shown under Reserves and Surplus section on the liabilities side of a balance

#### CHAPTER 5 PROCEDURES OF LEDGER ACCOUNT

#### **DR.R.RAJAVARDHINI**

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Ledger

The journal provides a complete listing of the daily transactions of a business, but it does not provide information about a specific account in one place. For example, to know how much cash balance we have, the accounting clerk would have to check all the journal entries in which cash is involved which is very difficult. To avoid this difficulty, the debit and credit of journalized transactions are transferred to ledger accounts. Thus, all the changes for a single account are located in one place - in a ledger account. This makes it easy to determine the current balance of any account.

#### **Standard Form of Ledger Account:**

To understand clearly as to how to write the accounts in ledger, the standard form of an account is given below with two separate transactions:

Date	Particulars	J.R	Amount	Date	Particulars	J.R	Amount
2005				2005			
Dec. 17	Cash A/C		1,200	Dec. 17	Purchases A/C		2,000

#### Posting Procedure:

Transferring information i.e., entries from journal to ledger accounts is called posting. The procedure of posting from journal to ledger is as follows:

1. Locate the ledger account from the first debit in the journal entry.

**2**. Record the date in the date column on the debit side of the account. The date is the date of transactionrather than the date of the posting.

**3**. Record the name of the opposite account (account credited in entry) in the particular (also known asreference column, description column etc.) column.

4. Record the page number of the journal in the journal reference (J.R) column from where the entry isbeing posted.

5. Record the amount of the debit in the "amount column"

6. Locate the ledger account for the first credit in the journal and follow the same procedure.

#### Balancing An Account:

The difference between the two sides of an account is its balance. The balance is written on the lesser sideto make the two sides equal. The process of equalizing the two sides of an account is known as **balancing**. The **rules for balancing an account** are stated as below: **1**. Add up the amount columns of both the sides of an account and write the totals in a separate slip ofpaper.

2. Find out the difference of the two totals.

3. Write down the difference on the lesser side of the account.

4. Now total up both the sides and write the totals and draw double lines under them.

5. Again write the difference on the opposite side below the double line.
#### CHAPTER 6 SUBSIDIARY BOOKS

#### **DR.D.SILAMBARASAN**

Department of Commerce,

Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### SUBSIDIARY BOOKS

If the size of business is small, then it is possible to enter every transaction in Journal only, but if the size of business is large, it is no longer possible to enter every transaction in one book only. Therefore, Journal is divided into sub-parts, known as **Special Journals**. Therefore, following subsidiary books are prepared:



#### **PURCHASE BOOK**

All credit purchases of goods are recorded in the purchases journal whereas cash purchases are recorded in the cash book. Other purchases such as purchases of office equipment, furniture, building, are recoded in the journal proper if purchased on credit or in the cash book if purchased for cash. The source documents for recording entries in the book are invoices or bills received by the firm from the supplies of the goods.

#### CHAPTER 7 CASH BOOK PREPARATION DR. S. RAJENDRAN

Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Cash book

It plays a dual role. It is both a book of original entry as well as a book of final entry. All cash transactions are primarily recorded in it as soon as they take place; so, it is a journal (a book of original entry/subsidiarybook). On the other hand, the cash aspect of all cash transactions is finally recorded in the Cash Book (no posting in Ledger); so, a Cash Book is also a Ledger (a book of final entry/Principal Book).

When a cashbook is maintained, transactions of cash are not recorded in the journal, and no separate account for cash or bank is required in the ledger.

#### **TYPES OF CASH BOOKS**

#### Single Column Cash Book

The single column cash book records all cash transactions of the business in a chronological order, i.e., it is a complete record of cash receipts and cash payments. When all receipts and payments are made in cash by a business organisation only, the cash book contains only one amount column on each (debit and credit)side.

#### It does not record:

- **1.** Non-cash transactions
- 2. Cheques received or given
- 3. Cash discount allowed or received

#### Format of single column cash book

Cash Book

Dr							
Date	Receipts	L.F	Amount	Date	Payments	L.F	Amou
						•	nt

1. Date: Date for the transaction is written

2. Particulars: The name of the account under which the cash has been received or payment has been made is written. Cash book starts with the opening balance of cash written on the receipts side as "Tobalance b/d". A new business will not have an opening balance.

**3**. Ledger Folio (L.F.): It records the page number in the ledger where the amount has been posted in theaccount.

4. Amount: The amounts received are written on the debit side and the amounts paid are written on thecredit side.

Cr

#### CHAPTER 8 TYPES OF ERRORS

#### **DR.R.SELVARAJ**

Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **TYPES OF ERRORS**



#### Errors of omission

The errors of omission may be committed at the time of recording the transaction in the books of original entry or while posting to the ledger. These can be of two types:

(i) error of complete omission

(ii) error of partial omission

#### Errors of commission

These are the errors which are committed due to wrong posting of transactions, wrong totaling or wrongbalancing of the accounts, wrong casting of the subsidiary books, or wrong recording of amount in the books of original entry, etc.

#### Errors of principle

Accounting entries are recorded as per the generally accepted accounting principles. If any of these principles are violated or ignored, errors resulting from such violation are known as errors of principle.

An error of principle may occur due to incorrect classification of expenditure or receipt between capitaland revenue.

#### Compensating errors

When two or more errors are committed in such a way that the net effect of these errors on the debits andcredits of accounts is nil, such

#### CHAPTER 9 FINANCIAL STATEMENT OF PROPRIETORSHIP

#### DR. S. KAMARAJU

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

<u>Meaning</u>

Financial Statements are the summaries of the accounts of a business enterprise and shows the profitability and financial position at the end of the accounting period.

It includes at least two basic statements:

A) Trading and & Profit and Loss Account

B) Balance Sheet

#### Trading Account

It is prepared for calculating the gross profit or gross loss arising out of the trading activities of a business.

#### Format of a Trading Account

Dr.	(For the year ended)			
Particulars	Amount	Particulars		Amount
To Opening stock To Purchases Less: Return Outwards ( To Wages To Carriage Inwards To Freight Inwards/cartage To Gross Profit c/d	XXX XXX) XXX XXX XXX XXX XXX XXX XXX XX	By Sales Less: Return Inwards By Closing stock By Gross Loss	xxx <u>(xxx)</u>	Xxx Xxx Xxx Xxx

#### **Trading Account**

All expenses which relate to either purchase of raw material or manufacturing of goods are recorded in theTrading account. All such expenses are called 'Direct Expenses'. Examples of direct expenses:

- Carriage or freight inwards
- Manufacturing wages
- Power and fuel
- Factory lighting
- Factory rent and rates
- Royalties
- Consumable stores

#### **CHAPTER 10**

#### **STATEMENT OF AFFAIRS**

**DR.G.KARTHIGA** 

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

A statement of affairs, like a balance sheet, is a statement which shows the assets and liabilities of a concern on a particular date and reveals its financial state of affairs. Many of the assets and liabilities shown on this statement are based on the estimates made by the proprietor. They are not based on book values as in case of balance sheet. Statement of affairs is prepared to ascertain the financial position and capital investments in the organization where books of accounts are not properly maintained or may be destroyed. The values of various assets and liabilities of the concern are ascertained as under:

- i) Cash in hand: It is ascertained from cash book or through physical verification of cash in hand
- ii) Cash with bank: It is ascertained by balancing cash book. In case the cash book has no bank column it is confirmed on the basis of pass book or bank records.
- iii) **Debtors & creditors: They** are calculated from the personal ledgers maintained by the concern.
- iv) Stock in trade: It is valued on the basis of actual stock taking and valuation.
- v) **Fixed Assets:** The values of fixed assets owned by concern like furniture, machinery & equipment's, buildings, vehicles etc., are judged from available records or by estimates.
- vi) Other values: The other values such as outstanding expenses and incomes, prepaid expenses, loans and incomes received in advance etc., are found from available records and memory of proprietor or partners.
- vii) Capital : Finally, the excess of assets over liabilities is taken as capital.

The proforma of Statement of Affairs is as under :

Liabilities	Rs.	Assets	Rs.
Creditors	XXX	Cash in Hand	XXX
Bills Payable	XXX	Cash at Bank	xxx
Loans & Overdrafts	XXX	Prepaid Expenses	xxx
Outstanding Expenses	XXX	Outstanding Income	xxx
Capital (Balancing figure)	XXX	Debtors	xxx
		Stock	xxx
		Investments	xxx
		Furniture & Fittings	xxx
		Plant & Machinery	xxx
		Buildings & Land	XXX

STATEMENT OF AFFAIRS AS ON ....

# **CO-OPERATION THEORY**

### Edited by DR. S. KAMARAJU



#### **CO-OPERATION THEORY**

#### Copyright © 2021 by DR. S. KAMARAJU

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

#### ISBN: 978-93-6255-974-6

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### TABLE OF CONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	MEANING AND DEFINITION CO-		1
	OPERATION	DR.R.SELVARAJ	
2	FEATURES OF CO-OPERATIVES	DR. S. KAMARAJU	11
3	HISTORY OF THE CO-OPERATIVE		22
	MOVEMENT	DR.G.KARTHIGA	
4	PRINCIPLES OF CO-OPERATION	DR.V.SRIDEVI	32
5	DIFFERENCE BETWEEN		42
	CAPITALISM AND CO-		
	OPERATION	DR.R.RAJAVARDHINI	
6	FEATURES OF A SOCIALISTIC		52
	ECONOMY	DR.D.SILAMBARASAN	
7	DIFFERENCE BETWEEN		62
	SOCIALISM AND CO-OPERATION	DR.S.RAJENDRAN	
8	EVOLUTION, GROWTH AND		72
	CURRENT STATUS:	DR.R.SELVARAJ	
9	ALL INDIA RURAL CREDIT		82
	REVIEW COMMITTEE (1969)	DR. S. KAMARAJU	
10	AGRICULTURAL CREDIT REVIEW		92
	COMMITTEE (1989)	DR.G.KARTHIGA	
	REFERENCE		101

#### CHAPTER 1 MEANING AND DEFINITION CO-OPERATION DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Meaning of Cooperation**

Cooperation is fundamentally about individuals or groups working together toward a common goal or shared interest. It implies a collaborative spirit where participants contribute their efforts, resources, or knowledge to achieve outcomes that benefit all parties involved. Cooperation can manifest in various contexts, such as families, workplaces, communities, and even international relations. It fosters a sense of unity and enhances problem-solving capabilities by leveraging diverse perspectives and skills.

#### **Definitions of Cooperation**

- 1. **General Definition**: Cooperation is the voluntary act of coming together to achieve a common objective. This process often involves aligning goals, sharing responsibilities, and facilitating communication among participants.
- 2. **Sociological Perspective**: From a sociological standpoint, cooperation is considered a fundamental social process that underpins group cohesion and societal stability. It is essential for building social networks and trust, which are critical for maintaining order and facilitating social interactions.
- 3. **Economic Definition**: Economically, cooperation can be viewed as a strategy that enhances efficiency and productivity. By pooling resources and sharing risks, individuals and organizations can achieve greater outcomes than they could independently. This is particularly evident in cooperative business models, where members work together to maximize benefits.
- 4. **Psychological Perspective**: In psychology, cooperation is analyzed through the lens of human behavior and motivation. Factors such as trust, empathy, and social norms play significant roles in encouraging cooperative behaviors. Understanding these psychological aspects can help organizations and communities foster a culture of cooperation.
- 5. **Game Theory Definition**: In game theory, cooperation is defined in terms of strategic interactions among rational decision-makers. Scenarios like the Prisoner's Dilemma illustrate how cooperation can lead to better collective outcomes, even when individual interests might suggest otherwise. This mathematical approach helps in understanding the conditions under which cooperation is likely to occur.
- 6. **Environmental Context**: In environmental studies, cooperation is often emphasized as a necessary approach for addressing global challenges such as climate change. Collaborative efforts among nations, organizations, and communities are essential for developing sustainable solutions that benefit the planet.
- 7. **International Relations**: In the realm of international relations, cooperation is seen as a mechanism for achieving peace and stability. Diplomatic efforts often rely on cooperative agreements and treaties that facilitate collaboration between countries on issues like trade, security, and environmental protection.

#### **Importance of Cooperation**

- 1. **Enhanced Problem-Solving**: Cooperation allows diverse groups to come together, bringing various perspectives and skills that enhance creativity and innovation. This collaborative approach often leads to more effective solutions.
- 2. **Resource Sharing**: By cooperating, individuals and organizations can share resources, whether they are financial, informational, or technological. This sharing can reduce costs and increase access to essential tools and services.

#### CHAPTER 2 FEATURES OF CO-OPERATIVES DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Features of Cooperatives**

Cooperatives are distinct in their structure and operations, emphasizing principles that promote member welfare and community engagement. Here's a more in-depth exploration of the key features of cooperatives:

- 1. Voluntary Membership:
  - **Inclusivity**: Membership is open to anyone who can use the cooperative's services and is willing to accept the responsibilities of membership. This ensures that cooperatives are accessible to diverse populations.
  - **Non-Discriminatory**: Cooperatives do not discriminate based on race, gender, religion, or political affiliation, promoting equality and inclusiveness.

#### 2. Democratic Control:

- **One Member, One Vote**: This principle ensures that every member has an equal voice in the decision-making process, regardless of their financial investment. It encourages active participation and engagement in cooperative governance.
- **Elected Leadership**: Members elect the board of directors, who are accountable to the membership. This governance structure promotes transparency and accountability.

#### 3. Member Economic Participation:

- **Shared Benefits**: Profits are distributed among members based on their contributions or patronage rather than on their investment size. This aligns the cooperative's success with the members' usage and support.
- **Capital Contribution**: Members often contribute capital to the cooperative, which is used to fund operations. This fosters a sense of ownership and investment in the cooperative's success.

#### 4. Cooperation Among Cooperatives:

- **Network Support**: Cooperatives often form federations or associations to support one another, share resources, and strengthen their bargaining power in markets.
- **Collective Marketing**: By collaborating, cooperatives can enhance their market reach and negotiate better terms with suppliers and customers.

#### 5. Education and Training:

- **Skill Development**: Cooperatives provide education and training programs for members, enhancing their skills and understanding of cooperative principles and operations.
- **Community Outreach**: Many cooperatives engage in community education initiatives to promote cooperative values and encourage community participation.

#### 6. Concern for Community:

- **Social Responsibility**: Cooperatives are committed to contributing positively to their communities, often engaging in local development projects and initiatives that improve the quality of life for members and non-members alike.
- **Environmental Stewardship**: Many cooperatives implement sustainable practices, such as eco-friendly operations and community clean-up programs, reflecting a commitment to environmental responsibility.
- 7. Autonomy and Independence:

#### CHAPTER 3 HISTORY OF THE CO-OPERATIVE MOVEMENT DR.G.KARTHIGA Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### History of the Cooperative Movement

The cooperative movement has a rich and diverse history, characterized by a commitment to mutual aid and collective benefit. Here's an overview of its key milestones:

#### Early Roots (17th-18th Centuries)

- 1. **Medieval Guilds and Mutual Aid**: The seeds of cooperative principles can be traced back to medieval guilds in Europe, where craftsmen collaborated to protect their trades and ensure fair practices. Mutual aid societies also emerged, offering support and resources to members in times of need.
- 2. **Agricultural Cooperation**: As the Industrial Revolution progressed in the late 18th century, rural communities began forming agricultural cooperatives. Farmers collaborated to share resources, such as tools and marketing efforts, to combat economic hardships.

#### The Birth of Modern Cooperatives (19th Century)

- 1. **Rochdale Pioneers (1844)**: The establishment of the Rochdale Society of Equitable Pioneers in England marked a significant turning point. This group of weavers created the first successful consumer cooperative, emphasizing principles such as democratic control, profit-sharing, and quality goods. Their model became a template for future cooperatives.
- 2. **Spread Across Europe**: Inspired by the Rochdale model, cooperatives quickly spread throughout Europe. In Germany, figures like Friedrich Wilhelm Raiffeisen developed rural credit cooperatives, focusing on providing affordable loans to farmers.
- 3. International Cooperative Alliance (ICA) (1895): The ICA was founded in London to unite cooperative organizations globally. This established a platform for sharing best practices and advocating for cooperative principles.

#### Growth and Diversification (20th Century)

- 1. **Post-World War II Expansion**: The cooperative movement experienced significant growth after World War II as countries recognized the role of cooperatives in economic recovery. They provided essential services, especially in agriculture, housing, and finance.
- 2. **Diverse Sectors**: Cooperatives diversified into various sectors, including healthcare, retail, and worker cooperatives. Credit unions gained popularity, offering members a democratic and community-oriented alternative to traditional banks.
- 3. **Global Reach**: In developing countries, cooperatives became vital in promoting self-reliance and sustainable development. Agricultural cooperatives helped small-scale farmers access markets and improve productivity.

#### Challenges and Resilience (Late 20th Century to Present)

- 1. **Globalization and Competition**: As globalization intensified, cooperatives faced challenges from large corporations and competitive market dynamics. Many cooperatives adapted by emphasizing community engagement, sustainability, and social responsibility.
- 2. **Technological Advancements**: The rise of technology has given birth to new cooperative models, such as platform cooperatives, which leverage digital tools to empower workers in the gig economy while ensuring equitable profit distribution.
- 3. **Social Movements and Advocacy**: The cooperative movement has increasingly aligned with social justice, environmental sustainability, and fair trade initiatives. This alignment has strengthened the cooperative identity and broadened its impact on society.

#### CHAPTER 4 PRINCIPLES OF CO-OPERATION DR.V.SRIDEVI Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Principles of Cooperation**

The principles of cooperation form the backbone of cooperative enterprises, providing a framework that guides their operations and decision-making processes. These principles not only distinguish cooperatives from traditional business models but also ensure that they serve the needs of their members and the community effectively. This essay explores the fundamental principles of cooperation, emphasizing their significance and the values they promote.

#### Voluntary and Open Membership

One of the core principles of cooperation is voluntary and open membership. This principle emphasizes that cooperatives are accessible to anyone who can utilize their services, regardless of socioeconomic status, gender, race, or other factors. By promoting inclusivity, cooperatives create a diverse membership base that reflects the community's demographic. This openness fosters a sense of belonging and encourages participation from all segments of society, ensuring that cooperatives are representative of their communities.

#### Democratic Member Control

Democratic member control is another foundational principle of cooperation. Cooperatives operate on the principle of "one member, one vote," which ensures that all members have an equal say in decision-making processes. This democratic governance structure empowers members to participate actively in shaping the cooperative's direction and policies. It also instills a sense of ownership and accountability, as members are directly involved in the governance of the cooperative. This democratic ethos is vital in fostering trust and collaboration among members.

#### Member Economic Participation

Member economic participation underscores the cooperative's commitment to serving its members. Members contribute equitably to the cooperative's capital and share in the economic benefits generated by its operations. Profits are typically distributed based on each member's usage or contributions, rather than on their financial investment. This principle ensures that the cooperative remains focused on the needs and welfare of its members, rather than prioritizing profit maximization for external shareholders. By aligning economic incentives with member participation, cooperatives strengthen their community ties and foster loyalty among their members.

#### Autonomy and Independence

Autonomy and independence are critical to the cooperative identity. Cooperatives are self-governing entities that maintain control over their operations and decision-making. While they may collaborate with other organizations or seek assistance, they do so without compromising their autonomy. This independence allows cooperatives to adapt to local needs and challenges, enabling them to respond effectively to their members' interests. It also ensures that cooperatives remain accountable to their members rather than to external entities or profit motives.

#### Education, Training, and Information

Education, training, and information are vital for empowering members and ensuring the cooperative's success. Cooperatives prioritize the education of their members, employees, and the broader community. This commitment to capacity building equips members with the knowledge and skills necessary to participate fully in cooperative governance and operations. By fostering a culture of

#### CHAPTER 5 DIFFERENCE BETWEEN CAPITALISM AND CO-OPERATION DR.R.RAJAVARDHINI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Difference Between Capitalism and Cooperation**

Capitalism and cooperation represent two distinct approaches to economic organization and social relations. While both systems aim to promote prosperity and meet the needs of individuals, they differ fundamentally in their principles, structures, and outcomes. This essay outlines the key differences between capitalism and cooperation.

#### 1. Ownership Structure

**Capitalism**: In a capitalist system, the means of production—such as factories, land, and resources are predominantly owned by private individuals or corporations. This ownership enables individuals to control and profit from their investments. The focus is often on maximizing individual profit and wealth.

**Cooperation**: In contrast, cooperatives are owned and managed collectively by their members. Each member has a stake in the cooperative and participates in its decision-making processes. This shared ownership model fosters a sense of community and mutual benefit, aligning the cooperative's goals with the needs of its members.

#### 2. Decision-Making Process

**Capitalism**: Decision-making in capitalist enterprises is typically centralized, with owners or shareholders making key choices aimed at maximizing profits. This top-down approach can lead to prioritizing financial returns over employee welfare or community impact.

**Cooperation**: Cooperatives operate on democratic principles, with decisions made collectively by members. Each member typically has one vote, promoting equality and ensuring that diverse perspectives are considered. This democratic governance structure encourages member engagement and accountability.

#### 3. Profit Distribution

**Capitalism**: In capitalism, profits are distributed primarily to shareholders based on their investment. The focus is on maximizing returns for investors, often at the expense of other stakeholders, such as employees and the community.

**Cooperation**: Cooperatives distribute profits among members based on their participation or usage of the cooperative's services, rather than on the amount of capital invested. This equitable distribution fosters a sense of shared responsibility and community benefit.

#### 4. Goals and Values

**Capitalism**: The primary goal of capitalism is profit maximization and economic growth. This focus can lead to competition that prioritizes efficiency and innovation but may also result in social inequalities and environmental degradation.

**Cooperation**: Cooperatives prioritize social welfare and community development alongside economic sustainability. Their goals often include promoting equitable access to resources, enhancing the well-being of members, and addressing social and environmental challenges.

#### CHAPTER 6 FEATURES OF A SOCIALISTIC ECONOMY DR.D.SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Features of a Socialistic Economy

A socialistic economy is characterized by collective or governmental ownership and regulation of the means of production, distribution, and exchange. The main goal is to achieve economic equality and social welfare for all members of society. Here are the key features of a socialistic economy:

#### 1. Collective Ownership

In a socialistic economy, the means of production—such as factories, land, and resources—are owned collectively by the community or the state. This collective ownership aims to eliminate private ownership, reducing wealth disparities and ensuring that resources are used for the common good.

#### 2. Central Planning

Economic activities are often guided by central planning authorities, which determine production goals, resource allocation, and distribution strategies. This planning seeks to meet the needs of society as a whole, rather than prioritizing individual profits, which can lead to a more equitable distribution of goods and services.

#### 3. Economic Equality

A fundamental goal of socialism is to promote economic equality among all members of society. This involves reducing income disparities through progressive taxation, social welfare programs, and public services. The emphasis is on creating a more level playing field where everyone has access to basic necessities.

#### 4. Social Welfare Programs

Socialistic economies prioritize the well-being of citizens through comprehensive social welfare programs. These may include universal healthcare, free education, unemployment benefits, and pensions. Such programs are designed to provide security and improve the quality of life for all individuals, particularly the most vulnerable.

#### 5. Limited Market Forces

While some socialist economies incorporate elements of market mechanisms, the role of market forces is generally limited. Prices and production levels are often set by the government rather than determined solely by supply and demand, which can help stabilize the economy and prevent exploitation.

#### 6. Focus on Public Goods

In a socialistic economy, there is a strong emphasis on the provision of public goods—services that benefit all members of society, such as infrastructure, education, and healthcare. The aim is to ensure that these essential services are accessible to everyone, regardless of their financial situation.

#### 7. Worker Participation

Many socialist systems promote the idea of worker participation in decision-making processes. This can take the form of worker cooperatives or councils where employees have a say in the management and direction of their workplaces, fostering a sense of ownership and accountability.

#### 8. Environmental Sustainability

Socialist economies often prioritize environmental sustainability and responsible resource management. The collective ownership model encourages long-term planning that takes into account ecological impacts, aiming to protect natural resources for future generations.

#### CHAPTER 7 DIFFERENCE BETWEEN SOCIALISM AND CO-OPERATION DR.S. RAJENDRAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Difference Between Socialism and Cooperation**

Socialism and cooperation are distinct concepts that aim to address social and economic issues, but they do so in different ways. Here's a concise comparison highlighting their key differences:

#### 1. Ownership Structure

- Socialism: The means of production (factories, land, resources) are owned collectively by the state or the community as a whole. This ownership is often centralized.
- Cooperation: Ownership is held collectively by the members of a cooperative. Each member has a stake and a say in the management of the cooperative, emphasizing localized control.

#### 2. Decision-Making Process

- Socialism: Decision-making is typically centralized, with the government or a central authority making key economic decisions on behalf of the community. This can limit direct participation from individuals.
- Cooperation: Cooperatives operate on democratic principles, where each member has an equal vote in decisions. This decentralized approach encourages member engagement and accountability.

#### 3. Economic Goals

- Socialism: The primary goal is to achieve economic equality and social welfare for all. This often involves redistributing wealth and resources to eliminate disparities.
- Cooperation: The focus is on meeting the specific needs of members while promoting mutual benefit. While cooperatives contribute to social welfare, their primary goal is to serve their members.

#### 4. Profit Distribution

- Socialism: Profits are generally reinvested into public services or redistributed through social programs. The focus is on reducing income inequality.
- Cooperation: Profits are distributed among members based on their participation or usage of the cooperative's services, aligning success with member engagement.

#### 5. Market Role

- Socialism: Market forces are often limited, with the state regulating prices and production to prioritize social needs over profits.
- Cooperation: Cooperatives may operate within market economies, focusing on collaboration rather than competition. They aim to provide services that meet local needs.

#### 6. Implementation and Scale

- Socialism: Typically implemented at a national level, affecting the entire economy. It involves comprehensive social policies and significant government control.
- Cooperation: Can be established at various scales (local, regional, national) and can exist within both capitalist and socialist frameworks. Their formation is often grassroots, driven by community needs.

#### CHAPTER 8 EVOLUTION, GROWTH AND CURRENT STATUS DR.R.SELVARAJ Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **Evolution, Growth, and Current Status of Cooperatives :**

The cooperative movement has a rich history characterized by its evolution, growth, and adaptability in response to social and economic changes. This overview traces the journey of cooperatives from their inception to their current status in the global economy.

#### **Evolution of Cooperatives**

- 1. Early Beginnings (17th-18th Centuries):
  - The roots of cooperation can be traced back to early mutual aid societies and guilds in Europe, where individuals banded together for mutual support and economic benefits.
  - The Industrial Revolution highlighted the need for collective action as workers faced harsh conditions. This laid the groundwork for more formal cooperative structures.
- 2. Rochdale Pioneers (1844):
  - The modern cooperative movement began with the Rochdale Society of Equitable Pioneers in England. This group established the first successful consumer cooperative, implementing principles of democratic governance and profit-sharing.
  - Their model emphasized quality goods, fair prices, and member engagement, serving as a template for future cooperatives.
- 3. Expansion in Europe and Beyond (19th Century):
  - Inspired by the Rochdale model, cooperatives spread across Europe, particularly in Germany, France, and Italy, evolving into agricultural, credit, and worker cooperatives.
  - The establishment of the International Cooperative Alliance (ICA) in 1895 helped unite cooperative organizations globally, fostering collaboration and best practices.

#### **Growth of Cooperatives**

- 1. Post-World War II Boom:
  - After World War II, cooperatives experienced significant growth as countries recognized their role in rebuilding economies and promoting social stability.
  - Agricultural cooperatives flourished, allowing farmers to access markets and resources more effectively.
- 2. Diverse Sectors:
  - The cooperative model diversified into various sectors, including healthcare, housing, finance, and retail. Credit unions became prominent as community-based financial institutions.
  - This diversification allowed cooperatives to address a wide range of community needs, enhancing their relevance.
- 3. Global Reach:
  - Cooperatives gained traction in developing countries, where they were seen as a means to promote self-reliance and sustainable development. Agricultural cooperatives, in particular, empowered small-scale farmers.

#### **Current Status of Cooperatives**

- 1. Resilience and Adaptation:
  - Today, cooperatives are recognized for their resilience, particularly during economic downturns. Their community-focused approach enables them to provide stability and support to their members.

#### CHAPTER 9

#### ALL INDIA RURAL CREDIT REVIEW COMMITTEE (1969)

DR. S. KAMARAJU

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### All India Rural Credit Review Committee (1969)

The All India Rural Credit Review Committee, established in 1969, was a significant initiative aimed at assessing and reforming the rural credit system in India. The committee was formed in response to the growing recognition of the need for an efficient and accessible rural credit mechanism to support agricultural development and rural livelihoods.

#### **Objectives**

- 1. Assessment of Rural Credit: The primary objective of the committee was to evaluate the existing rural credit structure, including the performance of various financial institutions and their outreach to rural areas.
- 2. Identification of Gaps: The committee aimed to identify the gaps and inefficiencies in the rural credit system, particularly in terms of accessibility, affordability, and the diversity of financial products available to rural borrowers.
- 3. Recommendations for Improvement: Based on its findings, the committee was tasked with providing recommendations to enhance the rural credit system, making it more responsive to the needs of farmers and rural entrepreneurs.

#### **Key Recommendations**

- 1. Institutional Framework: The committee recommended strengthening the institutional framework for rural credit, including the expansion of cooperative banks, regional rural banks (RRBs), and microfinance institutions. This diversification was intended to enhance accessibility to credit.
- 2. Focus on Agricultural Credit: Emphasizing the importance of agriculture in rural economies, the committee suggested prioritizing agricultural credit and creating specialized financial products tailored to the needs of farmers.
- 3. Simplified Procedures: The committee called for the simplification of loan procedures and documentation to reduce bureaucratic hurdles that often deterred farmers from accessing credit.
- 4. Financial Literacy: The need for improving financial literacy among rural populations was highlighted, with recommendations for training programs to educate farmers about credit options, financial management, and repayment strategies.
- 5. Government Support: The committee emphasized the role of the government in facilitating rural credit by providing guarantees, interest subsidies, and ensuring that credit flows to priority sectors.
- 6. Monitoring and Evaluation: Establishing a robust monitoring and evaluation mechanism to assess the effectiveness of rural credit programs and ensure that they meet the intended objectives was also recommended.

#### Impact

The All India Rural Credit Review Committee's recommendations played a crucial role in shaping India's rural credit landscape. They led to:

- The establishment and expansion of Regional Rural Banks (RRBs) and the strengthening of cooperative banks, which increased the availability of credit in rural areas.
- Enhanced focus on agricultural credit, contributing to improvements in agricultural productivity and rural development.

#### CHAPTER 10 AGRICULTURAL CREDIT REVIEW COMMITTEE (1989) DR.G.KARTHIGA

#### Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Agricultural Credit Review Committee (1989)

The Agricultural Credit Review Committee, established in 1989, was an important initiative aimed at assessing and reforming the agricultural credit system in India. This committee emerged in the context of the ongoing challenges faced by the agricultural sector, particularly in terms of access to credit for farmers.

#### Objectives

- 1. **Evaluation of Agricultural Credit System**: The primary goal of the committee was to evaluate the existing agricultural credit system, focusing on its effectiveness, reach, and responsiveness to the needs of farmers.
- 2. **Identifying Constraints**: The committee aimed to identify the constraints and barriers faced by farmers in accessing credit, including issues related to loan procedures, interest rates, and the availability of financial products.
- 3. **Formulating Recommendations**: Based on its findings, the committee was tasked with formulating recommendations to enhance the agricultural credit system, making it more inclusive and efficient.

#### Key Recommendations

- 1. **Expansion of Credit Delivery Systems**: The committee recommended strengthening the existing credit delivery systems, including commercial banks, cooperative banks, and regional rural banks (RRBs). It suggested that these institutions should be better equipped to meet the credit needs of the agricultural sector.
- 2. **Focus on Timely Credit**: The importance of timely disbursement of credit was emphasized. The committee recommended streamlining procedures to ensure that loans were disbursed promptly, especially during critical agricultural seasons.
- 3. **Interest Rate Policy**: The committee advocated for a rationalization of interest rates to make credit more affordable for farmers. It suggested that interest rates should reflect the risk and cost of lending while ensuring that they remained accessible to the agricultural community.
- 4. **Diversification of Financial Products**: To cater to the varied needs of farmers, the committee recommended the development of diverse financial products, including long-term loans for investments in infrastructure and technology, as well as short-term loans for operational needs.
- 5. **Financial Literacy and Capacity Building**: Recognizing the importance of informed borrowing, the committee emphasized the need for financial literacy programs to educate farmers about available credit options, loan management, and repayment strategies.
- 6. **Strengthening Agricultural Cooperatives**: The committee suggested enhancing the role of agricultural cooperatives in providing credit and support services to farmers. Strengthening these institutions could improve credit access and foster a sense of community among farmers.
- 7. **Monitoring and Evaluation Mechanisms**: Establishing effective monitoring and evaluation mechanisms was recommended to assess the performance of agricultural credit programs and ensure that they met the needs of the farming community.

#### Impact

The recommendations of the Agricultural Credit Review Committee had a significant impact on the agricultural credit landscape in India. They contributed to:

# INCOME TAX LAW AND PRACTICE

Edited by

## DR.R.SELVARAJ



#### INCOME TAX LAW AND PRACTICE

#### Copyright © 2021 by DR.R.SELVARAJ

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-891-6

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### TABLE OF CONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	INTRODUCTION TO INCOME		1
	TAX	DR.R.SELVARAJ	
2	AGRICULTURAL INCOME	DR. S. KAMARAJU	11
3	<b>RESIDENTIAL STATUS AND</b>		22
	TAX LIABILITY	DR.G.KARTHIGA	
4	BASIC RULES GOVERNING		32
	SALARIES	DR.V.SRIDEVI	
5	LEAVE SALARY OR SALARY IN		42
	LIEU OF LEAVE	DR.R.RAJAVARDHINI	
6	INCOME FROM UNDER THE		52
	HEAD HOUSE PROPERTY	DR.D.SILAMBARASAN	
7	INCOME FROM UNDER THE		62
	HEAD BUSINESS AND		
	PROFESSION	DR.S.RAJENDRAN	
8	LONG TERM CAPITAL ASSETS	DR.R.SELVARAJ	72
9	INCOME FROM OTHER		82
	SOURCES	DR. S. KAMARAJU	
10	TAX TREATMENT OF GIFTS	DR.G.KARTHIGA	92
	REFERENCE		101

#### CHAPTER 1 INTRODUCTION TO INCOME TAX DR.R.SELVARAJ Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction to Income Tax

Income tax is a crucial component of a country's tax system, serving as a primary source of revenue for governments around the world. It is a tax levied on the income earned by individuals, corporations, and other entities, playing a significant role in funding public services, infrastructure, and social welfare programs.

#### **Definition and Purpose**

Income tax is defined as a percentage of an individual's or entity's earnings that is collected by the government. The primary purpose of income tax is to generate revenue that supports essential government functions, such as education, healthcare, public safety, and infrastructure development. Additionally, income tax serves as a tool for wealth redistribution, aiming to reduce economic inequalities by imposing higher tax rates on higher income brackets.

#### Types of Income Tax

- 1. **Personal Income Tax**: This tax is levied on the income of individuals, which may include wages, salaries, bonuses, rental income, dividends, and interest. Personal income tax rates often vary based on income levels, with progressive tax systems imposing higher rates on higher incomes.
- 2. **Corporate Income Tax**: Corporations are subject to income tax on their profits. The rates and regulations governing corporate income tax can differ significantly from those applicable to individuals.
- 3. **Capital Gains Tax**: This tax is applied to the profit made from the sale of assets, such as stocks, real estate, or other investments. Capital gains tax rates can vary based on how long the asset was held before sale.
- 4. Alternative Minimum Tax (AMT): This is a parallel tax system designed to ensure that individuals and corporations with high incomes pay a minimum level of tax, regardless of deductions and credits that may otherwise reduce their tax liability.

#### **Taxation Principles**

- 1. **Equity**: Income tax systems should be fair, meaning that individuals with similar abilities to pay should owe similar amounts, while those with greater financial capacity should contribute more.
- 2. **Simplicity**: Tax laws should be easy to understand and comply with, minimizing the burden on taxpayers and administrative costs for the government.
- 3. **Certainty**: Taxpayers should clearly understand their tax obligations, including rates, due dates, and how income is calculated.
- 4. **Efficiency**: The tax system should minimize economic distortions, allowing individuals and businesses to make decisions based on economic merit rather than tax implications.

#### Current Trends and Challenges

The landscape of income tax is continually evolving, influenced by economic conditions, technological advancements, and social changes. Some current trends include:

#### CHAPTER 2 AGRICULTURAL INCOME DR. S. KAMARAJU Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Agricultural Income: Importance, Challenges, and Economic Impact

Agricultural income is a vital component of the economy in many countries, especially in regions where agriculture serves as the primary source of livelihood for a significant portion of the population. This income is derived from various farming activities, including the cultivation of crops, raising livestock, and the sale of agricultural products. Understanding agricultural income is crucial for policymakers, economists, and farmers alike, as it plays a pivotal role in ensuring food security, driving rural development, and influencing overall economic stability.

#### Definition and Sources of Agricultural Income

Agricultural income can be broadly defined as the earnings generated from agricultural activities. This includes income from the cultivation of crops—both food and cash crops—revenue from livestock farming, and profits from renting out agricultural land or selling agricultural produce. Additionally, income from ancillary activities, such as agro-processing and direct marketing, also contributes to agricultural earnings. In many countries, such as India, agricultural income is often treated differently for tax purposes, typically enjoying exemptions that recognize its critical role in supporting livelihoods and the economy.

#### **Economic Contribution**

The significance of agricultural income cannot be overstated. It serves as a cornerstone of rural economies, providing employment to millions and fostering community development. In many developing countries, agriculture employs a substantial percentage of the workforce, directly impacting the standard of living for rural populations. By generating income, agriculture also stimulates local economies through the creation of jobs in related sectors, such as transportation, processing, and retail.

Furthermore, agricultural income plays a crucial role in ensuring food security. By supporting the production of essential food crops, it helps maintain stable food supplies for the population. In a world facing increasing population pressures and changing climate patterns, the ability to sustain agricultural productivity is more important than ever. The income generated from agriculture thus directly correlates with national food security and economic stability.

#### Challenges Facing Agricultural Income

Despite its importance, agricultural income faces numerous challenges that can threaten its sustainability. One of the most significant issues is market volatility. Farmers often encounter fluctuating prices for their produce, influenced by supply and demand dynamics, global market trends, and seasonal variations. This volatility can lead to unstable income streams, making financial planning difficult for farmers.

Climate change presents another formidable challenge. Extreme weather events, changing rainfall patterns, and rising temperatures can adversely affect crop yields and livestock health. Such environmental factors threaten not only the income of farmers but also food security at large. In this context, adaptation strategies and sustainable farming practices become essential to mitigate the impacts of climate change.

#### CHAPTER 3 RESIDENTIAL STATUS AND TAX LIABILITY DR.G.KARTHIGA

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Residential Status and Tax Liability

Understanding the relationship between residential status and tax liability is crucial for individuals and entities when determining their tax obligations. Residential status refers to the classification of a taxpayer based on their physical presence in a particular country or jurisdiction, which directly influences how and where they are taxed.

#### **Residential Status**

1. **Definition**: Residential status generally categorizes individuals as residents or non-residents for tax purposes. This classification often depends on factors such as the duration of stay, the purpose of the stay, and other connections to the country.

#### 2. Types of Residential Status:

- **Resident**: An individual who meets certain criteria, such as living in a country for a specified number of days during a tax year (e.g., 183 days in many jurisdictions), is classified as a resident. Residents are typically subject to tax on their worldwide income.
- **Non-Resident**: An individual who does not meet the residency criteria is considered a non-resident. Non-residents are usually taxed only on income sourced within the country.
- 3. **Statutory Residency Tests**: Many countries apply specific tests to determine residential status. These may include:
  - **Physical Presence Test**: Evaluates the number of days an individual spends in the country.
  - **Domicile Test**: Considers the individual's permanent home or center of vital interests.

#### Tax Liability

- 1. Scope of Taxation:
  - **Residents**: Typically taxed on their global income, which includes income earned both within and outside the country. This may encompass salaries, rental income, capital gains, and other sources.
  - **Non-Residents**: Generally taxed only on income earned within the country, such as income from employment, business profits, or capital gains related to local assets.
- 2. **Progressive Tax Rates**: Many countries employ a progressive tax rate structure for residents, meaning that higher income levels are taxed at higher rates. Non-residents may face different tax rates or a flat tax rate on their income.
- 3. **Tax Treaties**: Many countries have bilateral tax treaties designed to prevent double taxation. These treaties often stipulate which country has the taxing rights over various types of income, influencing the tax liability of residents and non-residents alike.

#### CHAPTER 4 BASIC RULES GOVERNING SALARIES DR.V.SRIDEVI

Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **BASIC RULES GOVERNING SALARIES**

Understanding the basic rules governing salaries is essential for both employers and employees. These rules provide a framework for determining compensation, ensuring compliance with legal requirements, and fostering fair labor practices. Below are the key components that define salary structures and governance.

#### 1. Minimum Wage Laws

- **Definition**: Minimum wage laws establish the lowest legal salary that employers can pay their employees for their work.
- **Compliance**: Employers must comply with federal, state, or local minimum wage regulations, which may vary by location and industry.
- Adjustments: Minimum wage rates may be adjusted periodically based on inflation and cost of living considerations.

#### 2. Salary Classification

- **Exempt vs. Non-Exempt**: Employees are typically classified as either exempt (not entitled to overtime) or non-exempt (entitled to overtime pay).
- **Criteria**: Classification often depends on the employee's job duties, salary level, and whether they perform certain administrative, professional, or executive tasks.

#### 3. Overtime Pay

- **Regulations**: Non-exempt employees are entitled to overtime pay for hours worked beyond the standard workweek (commonly 40 hours in the U.S.).
- **Rate**: Overtime pay is typically calculated at one and a half times the employee's regular hourly wage.

#### 4. Pay Structure and Frequency

- **Salary Basis**: Employers should clearly define the pay structure, whether it is hourly or salaried, and communicate this to employees.
- **Pay Frequency**: Employers must establish a regular pay schedule (e.g., weekly, biweekly, monthly) and adhere to it to ensure employees receive their salaries on time.

#### 5. Salary Adjustments

- **Performance Reviews**: Employers often conduct annual or semi-annual performance reviews that may lead to salary adjustments based on employee performance, market conditions, and company profitability.
- **Cost of Living Adjustments (COLA)**: Many employers provide periodic salary increases to account for inflation and rising living costs.

#### 6. Salary Transparency

- **Disclosure**: Some jurisdictions encourage or require salary transparency, which promotes fairness and helps eliminate wage disparities.
- **Equity**: Employers should aim for equitable salary practices to ensure that employees performing similar roles receive comparable compensation.

#### CHAPTER 5 LEAVE SALARY OR SALARY IN LIEU OF LEAVE DR.R.RAJAVARDHINI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### LEAVE SALARY OR SALARY IN LIEU OF LEAVE

Leave salary, often referred to as salary in lieu of leave, is a form of compensation provided to employees when they do not utilize their entitled leave days. Understanding the concept of leave salary is essential for both employees and employees, as it relates to employee rights, benefits, and overall workplace policies.

#### Definition of Leave Salary

Leave salary is the payment made to an employee for the unutilized leave they have accrued during their employment. This payment is typically provided when an employee:

- Resigns or retires: Upon leaving the organization, employees may receive a settlement for any unused leave days.
- Is terminated: If an employee is terminated, they may be compensated for any remaining leave balance.
- Exceeds leave limits: In some cases, if employees do not take their entitled leave within a specific period, they may be compensated for those days.

#### Calculation of Leave Salary

The calculation of leave salary generally involves:

- 1. Determining the Leave Balance: The total number of leave days accrued but not taken by the employee.
- 2. Calculating Daily Salary: The employee's salary is divided by the number of working days in a month (or the applicable period) to determine the daily wage.
- 3. Multiplying by Leave Days: The daily salary is then multiplied by the number of unutilized leave days to arrive at the total leave salary.

#### Legal and Policy Framework

- 1. Company Policies: Organizations often have specific policies governing leave entitlements, including how leave salary is calculated and when it is paid out. Employees should be aware of their company's leave policies to understand their rights and entitlements.
- 2. Labor Laws: Many countries have labor laws that govern employee leave entitlements, including provisions for leave salary. Employers must comply with these regulations to ensure fair treatment of employees.

#### Benefits of Leave Salary

- 1. Financial Security: Leave salary provides employees with a financial cushion, especially when transitioning out of a job or during unexpected circumstances where leave might be necessary but not taken.
- 2. Encouragement of Work-Life Balance: Offering leave salary can motivate employees to take their entitled leave, promoting overall well-being and work-life balance.
- 3. Retention of Talent: Fair leave policies, including the provision of leave salary, can enhance employee satisfaction and retention.

#### CHAPTER 6 INCOME FROM UNDER THE HEAD HOUSE PROPERTY DR.D.SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **INCOME FROM UNDER THE HEAD HOUSE PROPERTY**

#### Income from House Property

Income from house property is a significant aspect of taxation in many countries, particularly in India. This income is derived from the ownership of property that is let out or rented. Understanding how income from house property is calculated, taxed, and its implications is essential for property owners and investors.

#### Definition

Income from house property refers to the income earned by an individual or entity from property that is owned and used to generate rental income. This category typically includes residential properties, commercial properties, and any other buildings that can be rented out.

#### Key Components of Income from House Property

- 1. Gross Annual Value (GAV):
  - The gross annual value is the potential income that the property could generate. This is determined by:
    - Actual Rent Received: The actual amount received from tenants.
    - Fair Rent: The rent that similar properties in the same locality would fetch.
    - **Municipal Value**: The value assessed by municipal authorities for property tax purposes.
  - The GAV is the higher of the actual rent received or the fair rental value.

#### 2. **Deductions**:

- **Municipal Taxes**: Property owners can deduct municipal taxes paid during the year from their gross annual value.
- **Standard Deduction**: A standard deduction of 30% of the net annual value (GAV minus municipal taxes) is allowed to account for repairs, maintenance, and other costs associated with property management.
- **Interest on Home Loan**: If the property is financed through a home loan, the interest paid on that loan can also be deducted.

#### 3. Net Annual Value (NAV):

 The net annual value is calculated as follows: NAV=GAV-Municipal Taxes\text{NAV} = \text{GAV} - \text{Municipal Taxes}NAV=GAV-Municipal Taxes

#### 4. Taxable Income:

• The taxable income from house property is determined by:  $\det{Taxable Income} = \det{NAV} - \det{Deductions (30\% + Interest)}$ 

#### Tax Treatment

1. **Tax Rate**: Income from house property is generally taxed under the head "Income from House Property" in individual income tax returns. The tax rate applicable will depend on the individual's overall income tax slab.

#### **CHAPTER 7 INCOME FROM UNDER THE HEAD BUSINESS AND PROFESSION DR.S.RAJENDRAN** Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **INCOME FROM BUSINESS AND PROFESSION**

Income from business and profession refers to the earnings generated by individuals or entities engaged in commercial activities, including the provision of goods and services. This category is significant in taxation, as it encompasses a wide range of activities, from small sole proprietorships to large corporations.

#### Definition

Income from business and profession includes profits earned from any trade, commerce, or profession carried out with the intention of earning a profit. This may involve manufacturing, trading, or providing professional services such as consultancy, accounting, or legal services.

#### Key Components

- 1. Types of Businesses:
  - Sole Proprietorships: Individually owned businesses where the owner is personally responsible for all aspects of the business.
  - **Partnership Firms**: Businesses owned by two or more individuals who share profits and responsibilities.
  - **Corporations**: Separate legal entities that can own assets, incur liabilities, and pay 0 taxes independently of their owners.
  - Limited Liability Partnerships (LLPs): Hybrid structures that combine elements of partnerships and corporations, offering limited liability to partners.

#### 2. Calculation of Income:

- Gross Receipts: Total income generated from sales or services rendered during the financial year.
- Allowable Expenses: Business-related expenses that can be deducted from gross 0 receipts to determine net income. These may include:
  - Cost of goods sold (COGS)
  - Rent and utilities •
  - Salaries and wages
  - Depreciation of assets .
  - Advertising and marketing costs •
  - Office supplies and expenses
- 3. Net Profit:
  - from business profession is calculated 0 The net profit and as: Net Profit=Gross Receipts-Allowable Expenses\text{Net Profit} \text{Gross = Receipts} \text{Allowable

Expenses Net Profit=Gross Receipts-Allowable Expenses

#### CHAPTER 8 LONG TERM CAPITAL ASSETS DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### LONG-TERM CAPITAL ASSETS: IMPORTANCE AND IMPLICATIONS

Long-term capital assets are fundamental components of personal and corporate finance, playing a crucial role in investment strategies and wealth accumulation. These assets, typically held for more than one year, encompass a diverse range of properties and investments, including real estate, stocks, bonds, business equipment, and intangible assets such as patents and trademarks. Understanding long-term capital assets is essential for effective financial planning, investment decisions, and tax management.

#### **Definition and Characteristics**

Long-term capital assets are defined by their intended holding period, which exceeds one year. This characteristic differentiates them from short-term assets, which are generally purchased for immediate resale. The primary purpose of acquiring long-term capital assets is to generate appreciation over time or to provide income, rather than for immediate consumption. For instance, real estate is often purchased not only as a place to live but also as an investment that can appreciate in value and generate rental income.

Another defining feature of long-term capital assets is their potential for depreciation and amortization. Tangible assets, such as machinery or buildings, typically depreciate over time, reflecting their declining value due to wear and tear. Conversely, intangible assets may be amortized, allowing owners to deduct a portion of their cost over the asset's useful life. This distinction is important for accounting and tax purposes, as it influences how these assets are reported and valued.

#### **Tax Implications**

The taxation of long-term capital assets is a significant consideration for both individuals and businesses. When such assets are sold, any profit made is subject to capital gains tax. Long-term capital gains, which arise from the sale of assets held for more than one year, are often taxed at lower rates compared to short-term gains. This tax structure incentivizes investors to hold assets longer, thereby promoting stability in the financial markets.

Moreover, many tax jurisdictions allow for indexation benefits, which adjust the purchase price of the asset for inflation when calculating capital gains. This adjustment can significantly reduce the taxable gain, making long-term investment more attractive. Additionally, certain exemptions and deductions may apply, depending on specific circumstances, such as reinvestment in similar assets, further enhancing the appeal of long-term capital assets.

#### Importance in Financial Planning

Long-term capital assets play a pivotal role in wealth accumulation and financial stability. By investing in these assets, individuals and businesses can build a robust financial portfolio that appreciates over

#### CHAPTER 9 INCOME FROM OTHER SOURCES DR. S. KAMARAJU Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST) COME FROM OTHER SOURCES

#### **INCOME FROM OTHER SOURCES**

Income from other sources is a key category in the taxation framework that encompasses a variety of income types not covered under specific heads such as salaries, business profits, or capital gains. This classification ensures that all income earned by an individual or entity is accounted for, contributing to a comprehensive taxation system. Understanding this category is essential for effective tax planning and compliance.

#### Definition

Income from other sources refers to any income that does not fall under the primary categories defined by tax authorities. This can include a wide range of income types, such as:

- **Interest Income**: Earnings from savings accounts, fixed deposits, bonds, and loans provided to others.
- Dividend Income: Payments received from shares of stock owned in corporations.
- **Rental Income**: Earnings from property rentals that do not qualify as business income.
- **Gifts and Inheritances**: Money or property received as gifts, which may be subject to taxation beyond certain thresholds.
- Winnings from Lotteries or Contests: Any cash prizes or winnings from gambling activities.

#### Key Components

- 1. **Interest Income**: This is a common source of income for many individuals. It is typically taxable and should be reported as income from other sources. The tax treatment may vary based on the source of the interest and whether it exceeds specific exemption limits.
- 2. **Dividend Income**: Dividends received from investments in shares are also categorized under this head. While dividends may sometimes have a different tax rate, they must still be reported as part of total income.
- 3. **Gifts and Inheritances**: While gifts may not always be taxable, those exceeding a certain value might attract tax liabilities. Tax rules vary by jurisdiction regarding the treatment of gifts and inheritances, requiring careful attention to reporting requirements.
- 4. **Winnings**: Income from lotteries, gambling, and other contests is generally fully taxable and should be reported as income from other sources. This category may have specific tax rates depending on local laws.

#### Tax Treatment

- 1. **Tax Rates**: Income from other sources is typically taxed at the individual's applicable income tax slab rate. This means it is aggregated with other income types, leading to a potentially higher tax obligation depending on total income.
- 2. **Deductions**: While income from other sources is broadly taxable, certain deductions may apply. For instance, interest income may be subject to a tax deduction for investments made under specific schemes, allowing taxpayers to reduce their taxable income.
- 3. **Filing Requirements**: It is essential for taxpayers to accurately report all income from other sources in their tax returns. This includes providing details on the nature and amount of each income type.

#### CHAPTER 10 TAX TREATMENT OF GIFTS DR.G.KARTHIGA epartment of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### TAX TREATMENT OF GIFTS

The tax treatment of gifts can vary significantly depending on the jurisdiction and the specific circumstances surrounding the transfer. Generally, gifts are considered transfers of property without consideration, and understanding their tax implications is crucial for both donors and recipients. This overview outlines key aspects of gift taxation, including exemptions, reporting requirements, and implications for estate planning.

#### Definition of a Gift

A gift is defined as a voluntary transfer of property from one individual to another without any exchange of value. Gifts can include money, real estate, personal property, and other assets.

#### Tax Implications for the Donor

- 1. **Gift Tax**: In many jurisdictions, gifts may be subject to gift tax imposed on the donor. This tax is applicable when the value of the gift exceeds a certain threshold, often referred to as the annual exclusion limit.
- 2. **Annual Exclusion**: Many tax systems allow for an annual exclusion amount, which is the maximum value of a gift that can be given to an individual without triggering gift tax. For example, if the annual exclusion is \$15,000, a donor can give up to that amount to each recipient without incurring any tax liability.
- 3. **Lifetime Exemption**: In addition to the annual exclusion, many jurisdictions provide a lifetime gift exemption, allowing donors to give a cumulative total amount over their lifetime without incurring gift tax. Amounts exceeding this exemption will be subject to taxation.
- 4. **Taxable Gifts**: Gifts that exceed the annual exclusion amount must be reported, and any amount above the lifetime exemption may incur a tax liability.

#### Tax Implications for the Recipient

- 1. **Income Tax**: Generally, recipients do not have to pay income tax on the value of the gifts received. Gifts are not considered taxable income, making them tax-neutral for the recipient at the time of receipt.
- 2. **Basis in Property**: If the recipient later sells a gifted asset, the basis for calculating capital gains tax is typically the donor's basis in the asset, not the market value at the time of the gift. This can lead to tax implications for the recipient upon the sale of the asset.
- 3. **Exceptions**: Certain types of gifts, such as those that generate income (e.g., rental properties), may have specific reporting requirements or tax implications.

#### **Special Considerations**

- 1. **Gifts to Spouses**: In many jurisdictions, gifts between spouses may be exempt from gift tax, subject to specific limits. This provision is often designed to facilitate financial planning between partners.
- 2. Educational and Medical Expenses: Payments made directly to educational institutions or medical providers for another individual's expenses are often exempt from gift tax, regardless of the amount.

# AUDITING

### EDITED BY DR. S. RAJENDRAN



#### **AUDITING**

#### Copyright © 2021 by DR.S.RAJENDRAN

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-703-2

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### TABLE OF CONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	INTRODUCTION TO AUDITING	DR.R.SELVARAJ	1
2	DEVELOPMENT OF AUDITING	DR. S. KAMARAJU	11
3	METHODS OF EVALUATION OF		22
	INTERNAL CONTROL SYSTEM	DR.G.KARTHIGA	
4	PRINCIPLES OF AN EFFECTIVE		32
	INTERNAL CHECK SYSTEM	DR.V.SRIDEVI	
5	VERIFICATION OF ASSETS	DR.R.RAJAVARDHINI	42
6	INVENTORY VERIFICATION		52
	PROCEDURE	DR.D.SILAMBARASAN	
7	QUALIFICATIONS OF		62
	AUDITORS	DR.S.RAJENDRAN	
8	DUTIES OF COMPANY		72
	AUDITORS	DR.R.SELVARAJ	
9	CLASSIFICATION OF		82
	INVESTIGATION	DR. S. KAMARAJU	
10	ELECTRONIC AUDITING	DR.G.KARTHIGA	92
	REFERENCE		101

#### CHAPTER 1 INTRODUCTION TO AUDITING DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST) INTRODUCTION TO AUDITING

Auditing is a systematic and independent examination of financial information of any entity, whether profit-oriented or not, regardless of its size or legal form. The primary objective of auditing is to ascertain the accuracy and reliability of financial statements, ensuring they reflect the true financial position of the organization. This process is crucial for stakeholders, including investors, creditors, regulators, and management, as it provides assurance regarding the integrity of financial reporting.

#### **Definition of Auditing**

Auditing can be defined as the process of reviewing and evaluating the financial statements and related operations of an organization to ensure compliance with established standards and regulations. It involves assessing the accuracy of records, adherence to accounting principles, and the effectiveness of internal controls.

#### Types of Audits

- 1. Internal Audits: Conducted by an organization's own staff, internal audits focus on evaluating the effectiveness of internal controls, risk management, and governance processes. They provide insights for management to improve operational efficiency and compliance.
- 2. External Audits: These audits are performed by independent auditors outside the organization. External audits aim to provide an objective assessment of financial statements, ensuring they are free from material misstatement. The outcome is typically expressed in an audit opinion.
- 3. Statutory Audits: Mandated by law, statutory audits ensure compliance with regulatory requirements. Organizations, especially public companies, are often required to undergo statutory audits to provide transparency to stakeholders.
- 4. Forensic Audits: Focused on investigating financial discrepancies or fraud, forensic audits involve a detailed examination of financial records. These audits are often used in legal proceedings to uncover fraudulent activities.
- 5. Operational Audits: These audits assess the efficiency and effectiveness of operational processes within an organization. The goal is to identify areas for improvement and enhance overall performance.

#### Importance of Auditing

- 1. Accuracy of Financial Reporting: Auditing enhances the reliability of financial statements, ensuring that they accurately reflect an organization's financial position and performance. This is vital for informed decision-making by stakeholders.
- 2. Fraud Detection and Prevention: Through thorough examination and analysis, auditors can identify discrepancies or irregularities that may indicate fraud. Regular audits act as a deterrent against fraudulent activities.
- 3. Compliance with Laws and Regulations: Audits help organizations comply with various legal and regulatory requirements. This compliance reduces the risk of penalties and enhances corporate governance.

#### CHAPTER 2 DEVELOPMENT OF AUDITING DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST) DEVELOPMENT OF AUDITING

The development of auditing as a profession and practice has evolved significantly over centuries, adapting to changes in business practices, regulatory environments, and technological advancements. Understanding the historical context of auditing helps to appreciate its current significance and future trajectory.

#### Historical Background

- 1. Ancient Civilizations: The origins of auditing can be traced back to ancient civilizations, where early forms of accounting and record-keeping were practiced. For instance, in ancient Egypt and Mesopotamia, scribes maintained records of grain, livestock, and taxes, necessitating periodic reviews for accuracy and accountability.
- 2. **Medieval Period**: During the medieval period, the emergence of trade and commerce led to more sophisticated accounting practices. Merchants began to employ clerks to maintain records, and the need for verifying these records paved the way for the role of auditors. The concept of "audit" comes from the Latin word "audire," meaning "to hear," reflecting how auditors originally reviewed accounts by listening to financial accounts presented by accountants.
- 3. **Renaissance and the Birth of Double-Entry Bookkeeping**: The introduction of double-entry bookkeeping in the 15th century by Luca Pacioli marked a significant milestone in the development of auditing. This system allowed for more complex financial records and increased the necessity for independent verification. As businesses grew in size and complexity, so did the need for formal audits.

#### 19th Century: Professionalization of Auditing

- 1. **Industrial Revolution**: The Industrial Revolution brought about significant changes in business operations and financial reporting. As businesses expanded, the need for accountability increased, leading to the establishment of formal auditing practices. This period saw the rise of public accounting firms that provided auditing services to businesses.
- 2. **Formation of Professional Bodies**: In the late 19th century, professional accounting bodies were formed, such as the American Institute of Accountants (now AICPA) in the United States. These organizations set standards for auditing practices and established ethical guidelines for auditors, contributing to the professionalization of auditing.
- 3. Legislative Developments: The early 20th century saw the introduction of regulations governing auditing practices, particularly in response to financial scandals. The establishment of the Securities and Exchange Commission (SEC) in the U.S. in 1934 marked a significant regulatory milestone, requiring public companies to have their financial statements audited.

#### 20th Century: Evolution and Expansion

1. **Post-War Growth**: After World War II, the expansion of the global economy and the rise of multinational corporations increased the complexity of financial transactions and reporting.
#### CHAPTER 3

#### METHODS OF EVALUATION OF INTERNAL CONTROL SYSTEM DR.G.KARTHIGA

#### Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### METHODS OF EVALUATION OF INTERNAL CONTROL SYSTEM

Evaluating the internal control system of an organization is essential for ensuring its effectiveness and efficiency in safeguarding assets, ensuring accurate financial reporting, and promoting compliance with laws and regulations. Various methods can be employed to assess internal controls, each providing insights into the system's strengths and weaknesses. Here are some key methods of evaluation:

#### 1. Document Review

This method involves reviewing existing documentation related to internal controls, including policies, procedures, and guidelines. Key documents may include:

- Internal control manuals
- Flowcharts of processes
- Policies on authorization and approvals

By analyzing these documents, evaluators can determine whether the internal controls are clearly defined, up-to-date, and aligned with best practices.

#### 2. Walkthroughs

A walkthrough involves tracing a transaction through the internal control system from initiation to recording. This method allows evaluators to observe the actual processes in action, confirming that they align with documented procedures. During a walkthrough, evaluators can:

- Assess whether controls are functioning as intended
- Identify any gaps or redundancies in the process
- Evaluate the understanding and compliance of personnel involved

#### 3. Risk Assessment

Conducting a risk assessment is vital for identifying and prioritizing areas of risk within the internal control system. This process typically involves:

- Identifying potential risks related to financial reporting, operations, and compliance
- Evaluating the likelihood and impact of each risk
- Determining the adequacy of existing controls to mitigate identified risks

A thorough risk assessment helps focus evaluation efforts on the most critical areas of the internal control system.

#### 4. Control Testing

Control testing involves the systematic examination of internal controls to assess their effectiveness. This can include:

- **Tests of Design**: Evaluating whether controls are properly designed to prevent or detect errors and fraud.
- **Tests of Operating Effectiveness**: Assessing whether controls are functioning as intended through direct observation, inquiry, or re-performance.

#### CHAPTER 4 PRINCIPLES OF AN EFFECTIVE INTERNAL CHECK SYSTEM DR.V.SRIDEVI

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### PRINCIPLES OF AN EFFECTIVE INTERNAL CHECK SYSTEM

An effective internal check system is essential for safeguarding an organization's assets, ensuring the accuracy of financial records, and enhancing operational efficiency. Here are the key principles that contribute to the effectiveness of an internal check system:

#### 1. Segregation of Duties

Dividing responsibilities among different individuals helps prevent fraud and errors. No single person should control all aspects of any financial transaction. For example, the person responsible for authorizing payments should not be the same as the one preparing the checks or handling cash. This segregation creates a system of checks and balances.

#### 2. Authorization and Approval

All transactions should require proper authorization before execution. Establishing clear guidelines for who can approve transactions and the limits of their authority ensures accountability. For example, larger expenditures should require higher-level approvals, minimizing the risk of unauthorized transactions.

#### 3. Documentation and Record-Keeping

Accurate and complete documentation is vital for tracking all financial transactions. Each transaction should be supported by appropriate records, such as invoices, receipts, and contracts. This documentation provides an audit trail and helps verify the legitimacy of transactions.

#### 4. Physical Controls

Implementing physical controls protects assets from theft or misuse. This includes securing cash, inventory, and sensitive documents. Measures such as locked storage, surveillance systems, and access controls help ensure that only authorized personnel can access valuable assets.

#### 5. Regular Reconciliation

Periodic reconciliation of accounts is essential to identify discrepancies between recorded amounts and actual balances. Regular checks, such as bank reconciliations and inventory counts, help ensure that the financial records accurately reflect the organization's status.

#### 6. Independent Review and Monitoring

Conducting independent reviews of internal controls and financial transactions helps ensure adherence to policies and procedures. Internal audits or compliance checks can identify weaknesses in the control system and provide recommendations for improvement.

#### 7. Training and Awareness

Regular training for employees on internal control procedures is critical for ensuring compliance. Employees should understand their roles in the internal control system and the importance of following established policies to prevent errors and fraud.

#### CHAPTER 5 VERIFICATION OF ASSETS DR.R.RAJAVARDHINI Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **VERIFICATION OF ASSETS**

Verification of assets is a critical process in auditing and financial reporting that involves confirming the existence, ownership, and valuation of an organization's assets. This process is essential for ensuring the accuracy of financial statements and maintaining the integrity of financial reporting. Here's an overview of the key aspects of asset verification:

#### 1. Purpose of Asset Verification

- Accuracy of Financial Statements: Ensures that the assets reported on the balance sheet are accurate and fairly presented.
- **Fraud Prevention**: Detects and prevents misstatements or fraudulent activities related to asset reporting.
- Compliance: Ensures compliance with accounting standards and regulatory requirements.

#### 2. Types of Assets

Assets can be classified into several categories, each requiring different verification techniques:

- Current Assets: Cash, inventory, accounts receivable, etc.
- Fixed Assets: Property, plant, equipment, and intangible assets.
- **Investments**: Stocks, bonds, and other financial investments.
- Intangible Assets: Goodwill, patents, and trademarks.

#### 3. Methods of Verification

#### 1. **Physical Inspection**:

• For tangible assets like inventory and equipment, physical inspection is performed to confirm their existence and condition. This can involve counting physical inventory, inspecting machinery, and ensuring assets are in good working order.

#### 2. Documentation Review:

• Reviewing supporting documents such as purchase invoices, contracts, and title deeds helps verify ownership and valuation. For example, reviewing the purchase documents for fixed assets can confirm their acquisition cost and ownership.

#### 3. Confirmation:

• External confirmations are often sought for assets like cash and accounts receivable. This involves sending confirmation requests to banks or customers to verify account balances and ownership.

#### 4. Analytical Procedures:

• Performing analytical reviews, such as comparing current asset balances to prior periods or industry averages, can help identify inconsistencies that warrant further investigation.

#### 5. Reconciliation:

• Reconciling asset accounts with supporting records is crucial. For instance, reconciling inventory records with physical counts or matching accounts receivable with customer statements ensures accuracy.

#### 6. Valuation Assessment:

#### CHAPTER 6 INVENTORY VERIFICATION PROCEDURE DR.D.SILAMBARASAN

Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **INVENTORY VERIFICATION PROCEDURE**

Inventory verification is a critical process for ensuring that the quantities and values reported in financial statements accurately reflect the actual inventory on hand. An effective inventory verification procedure involves systematic steps to assess the accuracy and integrity of inventory records. Below is a detailed procedure for inventory verification:

#### 1. Planning the Inventory Count

- Schedule the Count: Determine the timing of the inventory count, ideally during a period of low activity (e.g., year-end or during off-peak hours).
- Notify Staff: Inform relevant staff members about the upcoming inventory count to ensure preparedness and cooperation.
- **Prepare Inventory Count Sheets**: Create count sheets or use electronic inventory management systems to record counts.

#### 2. Setting Up the Counting Environment

- **Organize Inventory**: Ensure that inventory is organized and accessible. Group similar items together and remove any obstructions.
- **Define Counting Zones**: Assign specific areas to counting teams to streamline the process and avoid duplicate counts.

#### 3. Conducting the Inventory Count

- Assign Teams: Form counting teams, ideally consisting of individuals who are not directly involved in inventory management to maintain objectivity.
- **Count Items**: Each team should physically count the items in their assigned zones and record the quantities on the count sheets.
- Use Technology: If possible, use barcode scanners or RFID technology to enhance accuracy and efficiency.

#### 4. Reconciliation of Counts

- **Compare Counts with Records**: After the physical count, compare the counted quantities with the recorded amounts in the inventory management system.
- **Investigate Discrepancies**: Identify any discrepancies between the physical counts and the records. Investigate the reasons for these differences, which may include theft, damage, misrecording, or loss.

#### 5. Adjusting Inventory Records

- Make Necessary Adjustments: Update the inventory records to reflect the accurate quantities based on the physical count. Ensure that adjustments are properly documented.
- **Document Reasons for Adjustments**: Maintain records of the reasons for any adjustments made to inventory counts for future reference and audits.

#### 6. Valuation of Inventory

- Assess Inventory Value: Once counts are reconciled, evaluate the value of inventory using appropriate methods (e.g., FIFO, LIFO, or weighted average).
- **Check for Obsolescence**: Review inventory for obsolete or slow-moving items and make necessary write-downs to reflect their net realizable value.

#### CHAPTER 7 QUALIFICATIONS OF AUDITORS DR.S.RAJENDRAN

#### Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **QUALIFICATIONS OF AUDITORS**

Auditors play a crucial role in ensuring the accuracy and integrity of financial statements and compliance with regulations. The qualifications required for auditors encompass educational credentials, professional certifications, practical experience, and essential skills. Here's a detailed overview of these qualifications:

#### 1. Educational Background

- **Bachelor's Degree**: A degree in accounting, finance, business administration, or a related field is typically required. This foundational education provides essential knowledge of financial principles and practices.
- Advanced Degrees: While not mandatory, a master's degree in accounting or an MBA can enhance an auditor's career prospects and knowledge base.

#### 2. Professional Certifications

- Certified Public Accountant (CPA): This is a widely recognized certification required in many jurisdictions. It involves passing a rigorous exam and meeting specific education and experience requirements.
- **Chartered Accountant (CA)**: Similar to the CPA, the CA designation is crucial in countries like the UK and India, focusing on both auditing and accounting practices.
- **Certified Internal Auditor (CIA)**: This certification is particularly valuable for auditors specializing in internal auditing and risk management.
- **Specialized Certifications**: Additional certifications like Certified Information Systems Auditor (CISA), Certified Fraud Examiner (CFE), and Chartered Financial Analyst (CFA) can further enhance an auditor's expertise and marketability.

#### 3. Practical Experience

- **Relevant Work Experience**: Most jurisdictions require auditors to have practical experience, typically gained through internships or positions in accounting firms or finance departments.
- Audit-Specific Experience: Experience in auditing different types of organizations (e.g., nonprofits, government entities, corporations) and exposure to various auditing methodologies is beneficial.

#### 4. Knowledge of Regulations and Standards

- Accounting Standards: Familiarity with generally accepted accounting principles (GAAP) or international financial reporting standards (IFRS) is essential, depending on the region.
- Auditing Standards: Knowledge of relevant auditing standards, such as those from the American Institute of CPAs (AICPA) or the International Auditing and Assurance Standards Board (IAASB), is critical for conducting effective audits.

#### 5. Technical Skills

- **Analytical Skills**: Strong analytical abilities are necessary for assessing financial data, identifying trends, and evaluating internal controls.
- Attention to Detail: Auditors must be meticulous to detect discrepancies and ensure the accuracy of financial records.
- **Technological Proficiency**: Familiarity with accounting software, data analytics tools, and other technology is increasingly important in modern auditing practices.

#### CHAPTER 8 DUTIES OF COMPANY AUDITORS DR.R.SELVARAJ

#### Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **DUTIES OF COMPANY AUDITORS**

#### 1. Financial Statement Auditing

Auditors examine a company's financial statements to ensure they are accurate and comply with applicable accounting standards (such as GAAP or IFRS). This includes verifying the completeness and correctness of the financial data.

#### 2. Assessment of Internal Controls

Auditors evaluate the effectiveness of the company's internal controls. This involves reviewing processes that safeguard assets, ensure the reliability of financial reporting, and promote compliance with laws and regulations.

#### 3. Compliance Verification

Auditors ensure that the company adheres to legal and regulatory requirements. This includes tax laws, labor laws, and industry-specific regulations, which helps mitigate the risk of legal penalties.

#### 4. Risk Assessment

Auditors identify and assess risks that may affect the accuracy of financial reporting. This involves analyzing potential areas of concern that could lead to misstatements in the financial statements.

#### 5. Audit Planning

Auditors develop a detailed audit plan that outlines the scope, timing, and methodology of the audit. This plan helps ensure that the audit is efficient and covers all necessary areas.

#### 6. Reporting Findings

After the audit, auditors prepare an audit report that includes their opinion on the financial statements. They also highlight any significant issues, weaknesses in internal controls, or areas for improvement.

#### 7. Communication with Stakeholders

Auditors communicate their findings to the company's management and board of directors. This includes discussing any identified risks, control deficiencies, or recommendations for improvement.

#### 8. Fraud Detection

Auditors are responsible for being vigilant in detecting signs of fraud or financial mismanagement during the audit process. This includes reviewing unusual transactions and assessing the integrity of financial reporting.

#### 9. Ongoing Professional Development

Auditors must stay updated on changes in accounting standards, regulations, and auditing practices. Continuous education ensures that they can effectively carry out their duties.

#### 10. Collaboration with Other Professionals

Auditors often work with tax advisors, legal counsel, and compliance officers to provide comprehensive oversight and ensure that all aspects of the company's financial practices are sound.

#### CHAPTER 9 CLASSIFICATION OF INVESTIGATION DR. S. KAMARAJU Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

Investigation can be classified into various categories based on different criteria. Here are some common classifications:

1. Based on Purpose

- **Criminal Investigation**: Focused on gathering evidence related to criminal activities and identifying offenders.
- **Civil Investigation**: Pertains to legal disputes between private parties, often involving issues like contracts or property disputes.
- Administrative Investigation: Conducted to examine compliance with laws and regulations within organizations or agencies.

#### 2. Based on Methodology

- Field Investigation: Involves collecting data and evidence directly from the site or environment where an event occurred.
- **Documentary Investigation**: Relies on examining records, documents, and other written materials to gather information.
- **Interview-Based Investigation**: Focuses on collecting information through interviews with witnesses, suspects, or experts.

#### 3. Based on Scope

- **Preliminary Investigation**: A preliminary assessment to determine whether a full investigation is necessary.
- Full Investigation: A comprehensive and in-depth examination of all aspects of a case or issue.
- Follow-up Investigation: Conducted after initial findings to gather more information or clarify issues.

#### 4. Based on the Subject Matter

- **Financial Investigation**: Involves examining financial records and transactions to uncover fraud or financial misconduct.
- **Forensic Investigation**: Utilizes scientific methods to analyze physical evidence, often in criminal cases.
- **Corporate Investigation**: Focuses on issues within a corporate setting, such as employee misconduct or regulatory compliance.

#### 5. Based on Legal Authority

- **Public Investigation**: Conducted by government agencies, law enforcement, or regulatory bodies.
- **Private Investigation**: Carried out by private investigators or firms, often for individuals or businesses seeking information.
- 6. Based on Timeframe
  - Current Investigation: Ongoing investigations addressing recent events or issues.
  - **Cold Case Investigation**: Revisiting unresolved cases from the past with new evidence or methods.

#### Conclusion

Understanding the classification of investigations helps in determining the appropriate approach and methods needed for effective inquiry. Each type of investigation serves specific purposes and employs different techniques based on the context and requirements of the case.

#### CHAPTER 10 ELECTRONIC AUDITING DR.G.KARTHIGA Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **ELECTRONIC AUDITING**

#### Introduction

Electronic auditing, commonly referred to as e-auditing, represents a transformative approach in the field of auditing. By integrating digital tools and advanced data analytics, e-auditing enhances the efficiency, accuracy, and effectiveness of the auditing process. As businesses increasingly rely on technology, understanding the implications and applications of electronic auditing becomes essential.

#### **Definition and Features**

Electronic auditing involves the use of software and electronic tools to conduct audits. Key features include:

- 1. **Digital Tools**: Utilization of audit management software and data analytics platforms that streamline data collection and analysis.
- 2. **Remote Access**: Auditors can access real-time data from various locations, enabling a more flexible approach to audits.
- 3. Automation: Repetitive tasks are automated, allowing auditors to focus on more complex analyses.

#### **Benefits of Electronic Auditing**

E-auditing offers several significant advantages:

- 1. **Increased Efficiency**: Automating data collection and analysis speeds up the audit process, allowing for timely completion.
- 2. **Enhanced Accuracy**: Digital tools minimize human error and enhance the reliability of financial reporting.
- 3. **Cost-Effectiveness**: Reducing the need for physical audits can lead to significant cost savings for both auditors and clients.

#### **Challenges and Risks**

Despite its many benefits, e-auditing comes with challenges:

- 1. **Cybersecurity Risks**: The reliance on digital platforms increases the potential for data breaches and cyber threats, necessitating robust security measures.
- 2. **Technology Dependence**: Auditors must be proficient with various tools and software, requiring ongoing training and adaptation.
- 3. **Data Integrity Issues**: Ensuring the accuracy and reliability of electronic data is crucial, as flawed data can lead to incorrect conclusions.

#### **Future Implications**

As technology continues to evolve, electronic auditing is poised to play a vital role in the future of the auditing profession. The integration of artificial intelligence (AI) and machine learning could further enhance data analysis capabilities, enabling auditors to identify risks and anomalies more effectively. Additionally, as businesses adapt to digital transformations, e-auditing will become essential for maintaining compliance and transparency.

# ENTREPRENEUR SHIP AND SMALL BUSINESS MANAGEMENT

### EDITED BY DR. S. KAMARAJU



#### ENTREPRENEURSHIP AND SMALL BUSINESS MANAGEMENT

#### Copyright © 2021 by DR. S. KAMARAJU

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-462-8

Publisher: Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### TABLE OF CONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	CONCEPTS OF		1
	ENTREPRENEURIAL TRAITS	DR.R.SELVARAJ	
2	SCOPE OF ENTREPRENEURSHIP		11
	DEVELOPMENT	DR. S. KAMARAJU	
3	CLASSIFICATION OF		22
	ENVIRONMENT	DR.G.KARTHIGA	
4	INTRODUCTION OF		32
	ENTERPRENEURIAL CULTURE	DR.V.SRIDEVI	
5	INNOVATION AND		42
	ENTREPRENEURSHIP	DR.R.RAJAVARDHINI	
6	ENTREPRENEURIAL		52
	MOTIVATION	DR.D.SILAMBARASAN	
7	IMPORTANCE OF		62
	ENTREPRENEURSHIP		
	DEVELOPMENT	DR.S.RAJENDRAN	
8	INTRODUCTION OF PROJECT		72
	REPORT	DR.R.SELVARAJ	
9	INTRODUCTION OF PROJECT		82
	IDENTIFICATION	DR. S. KAMARAJU	
10	SMALL ENTREPRENEURS		92
	DEVELOPMENT	DR.G.KARTHIGA	
	REFERENCE		101

#### CHAPTER 1 CONCEPTS OF ENTREPRENEURIAL TRAITS DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **CONCEPTS OF ENTREPRENEURIAL TRAITS**

Entrepreneurial traits refer to the characteristics and qualities that distinguish successful entrepreneurs from others. These traits influence how entrepreneurs approach challenges, interact with others, and navigate the complexities of starting and running a business. Here are some key concepts related to entrepreneurial traits:

#### 1. Risk Tolerance

Entrepreneurs often face uncertainties and potential failures. A high risk tolerance allows them to make bold decisions, embrace new opportunities, and accept the possibility of setbacks as part of the journey.

#### 2. Innovativeness

Successful entrepreneurs are typically innovative thinkers. They possess the ability to generate new ideas, find creative solutions to problems, and differentiate their offerings in the marketplace. This trait is crucial for developing unique products or services that meet customer needs.

#### 3. Vision and Goal Orientation

Entrepreneurs often have a clear vision of what they want to achieve. This long-term perspective helps them set strategic goals and maintain focus, guiding their decision-making processes and motivating their teams.

#### 4. Resilience and Perseverance

The entrepreneurial journey is filled with challenges and obstacles. Resilience—the ability to bounce back from failures—and perseverance—the commitment to keep going despite difficulties—are essential traits for overcoming setbacks and achieving long-term success.

#### 5. Self-Confidence

A strong sense of self-confidence enables entrepreneurs to trust their instincts and decisions. This trait helps them persuade others, attract investors, and lead teams effectively, even in uncertain situations.

#### 6. Adaptability

The business environment is constantly changing. Successful entrepreneurs exhibit adaptability, allowing them to pivot when necessary, respond to market demands, and embrace new technologies or trends.

#### 7. Leadership Skills

Effective leadership is vital for entrepreneurs who must inspire, motivate, and manage their teams. Good leaders communicate effectively, delegate tasks, and foster a positive organizational culture, encouraging collaboration and innovation.

#### 8. Strong Work Ethic

Entrepreneurs often invest significant time and effort into their ventures. A strong work ethic drives them to go above and beyond, maintaining diligence and commitment even during challenging times.

#### CHAPTER 2 SCOPE OF ENTREPRENEURSHIP DEVELOPMENT DR. S. KAMARAJU

#### Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### SCOPE OF ENTREPRENEURSHIP DEVELOPMENT

The scope of entrepreneurship development encompasses a wide range of areas and activities aimed at fostering entrepreneurial skills, promoting business creation, and enhancing economic growth. Here's an overview of the key aspects within this scope:

#### 1. Skill Development

Entrepreneurship development programs focus on equipping individuals with essential skills, such as:

- **Business Management**: Training in planning, organizing, and managing business operations.
- Financial Literacy: Understanding financial management, budgeting, and investment strategies.
- Marketing Skills: Techniques for market research, branding, and sales strategies.

#### 2. Support for Startups

Entrepreneurship development includes providing support to new ventures through:

- **Incubation Centers**: Facilities that offer resources, mentoring, and networking opportunities for startups.
- **Funding Assistance**: Access to grants, loans, and venture capital to help entrepreneurs secure the necessary capital.

#### 3. Policy and Regulatory Framework

Entrepreneurship development often involves advocating for supportive policies that encourage business formation and growth, including:

- **Simplified Regulations**: Reducing bureaucratic hurdles for starting and operating a business.
- Tax Incentives: Offering tax breaks or incentives for new businesses and startups.

#### 4. Innovation and Technology Promotion

Promoting innovation is a crucial aspect of entrepreneurship development, which includes:

- **Research and Development**: Supporting R&D initiatives to foster new ideas and technologies.
- **Tech Transfer Programs**: Facilitating the transfer of technology from research institutions to commercial applications.

#### 5. Networking and Collaboration

Creating networks among entrepreneurs, investors, and support organizations is vital for entrepreneurship development. This includes:

- **Networking Events**: Organizing conferences, workshops, and meetups for entrepreneurs to connect and share ideas.
- **Partnerships**: Encouraging collaborations between businesses, educational institutions, and government agencies.

#### CHAPTER 3 CLASSIFICATION OF ENVIRONMENT DR.G.KARTHIGA

#### Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **CLASSIFICATION OF ENVIRONMENT**

The environment can be classified into various categories based on different criteria, such as physical characteristics, biological aspects, and human influence. Here's a comprehensive overview of the classification of the environment:

#### 1. Natural Environment

- **Physical Environment**: Refers to the natural components of the Earth, including landforms (mountains, valleys), water bodies (rivers, lakes, oceans), climate, and natural resources (minerals, forests).
- **Biological Environment**: Comprises all living organisms, including plants, animals, and microorganisms, along with their interactions and ecosystems.

#### 2. Built Environment

- **Urban Environment**: Consists of cities and towns, including infrastructure such as buildings, roads, parks, and public facilities. It focuses on human-made structures and their impact on the natural surroundings.
- **Rural Environment**: Refers to non-urban areas, encompassing agricultural lands, small communities, and open spaces. This environment is often characterized by lower population density and greater connection to nature.

#### 3. Social Environment

- **Cultural Environment**: Includes the social norms, values, beliefs, and practices that shape the behavior of individuals and communities. It influences how societies interact with their physical environment.
- **Economic Environment**: Encompasses the economic factors that affect businesses and individuals, including market conditions, employment levels, and consumer behavior.

#### 4. Political Environment

- **Regulatory Environment**: Consists of laws, regulations, and policies that govern environmental protection, resource management, and land use. This environment significantly impacts business operations and societal interactions.
- **Global Environment**: Refers to international relationships and agreements that influence environmental practices, such as climate change treaties and trade agreements.

#### 5. Technological Environment

- **Innovation Environment**: Includes advancements in technology and their effects on production, communication, and lifestyle. It encompasses the tools and processes that shape economic and social interactions.
- **Information Environment**: Relates to the availability and dissemination of information, including the digital landscape, media, and communication technologies that influence decision-making and public awareness.

#### 6. Ecological Environment

• Ecosystem Types: Classifies environments based on ecosystems, such as forests, grasslands, wetlands, deserts, and marine ecosystems. Each ecosystem has unique characteristics and biodiversity.

#### CHAPTER 4 INTRODUCTION OF ENTERPRENEURIAL CULTURE DR.V.SRIDEVI Department of Commerce,

#### Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION TO ENTREPRENEURIAL CULTURE

Entrepreneurial culture refers to the set of values, beliefs, behaviors, and practices that foster entrepreneurship within a society or organization. It encompasses the environment that encourages individuals to innovate, take risks, and pursue new business opportunities. A vibrant entrepreneurial culture is essential for economic growth, job creation, and social development, as it empowers individuals to turn their ideas into viable ventures.

#### Key Elements of Entrepreneurial Culture

- 1. **Innovation and Creativity**: At the heart of entrepreneurial culture is a focus on innovation. Encouraging creative thinking allows individuals to develop unique solutions to problems and explore new markets.
- 2. **Risk-Taking Attitude**: A culture that embraces risk-taking helps individuals feel comfortable stepping outside their comfort zones. This willingness to take calculated risks is crucial for launching new ventures.
- 3. **Supportive Ecosystem**: An effective entrepreneurial culture is supported by a network of resources, including access to funding, mentorship, and business incubators. This ecosystem nurtures startups and small businesses.
- 4. **Education and Training**: Promoting entrepreneurship through education is vital. Programs that teach entrepreneurial skills, business management, and financial literacy empower individuals to pursue their entrepreneurial aspirations.
- 5. **Collaboration and Networking**: Fostering a sense of community among entrepreneurs encourages collaboration. Networking opportunities allow individuals to share ideas, resources, and experiences, enhancing the overall entrepreneurial spirit.
- 6. **Recognition of Success**: Celebrating entrepreneurial successes and role models inspires others to pursue their ventures. Recognizing achievements reinforces the idea that entrepreneurship is a viable and valued path.

#### Importance of Entrepreneurial Culture

- **Economic Growth**: A strong entrepreneurial culture contributes to innovation, which drives economic growth and competitiveness. Startups create new jobs and stimulate local economies.
- **Social Change**: Entrepreneurs often address societal challenges through innovative solutions, promoting social change and improving quality of life in communities.
- **Resilience and Adaptability**: Cultivating an entrepreneurial mindset fosters resilience and adaptability among individuals and organizations, enabling them to navigate challenges and seize opportunities in a rapidly changing environment.

#### Conclusion

Entrepreneurial culture is a vital component of a thriving economy. By promoting innovation, risktaking, collaboration, and education, societies can create an environment that empowers individuals to become entrepreneurs. This culture not only stimulates economic growth but also fosters social change and resilience, ultimately contributing to a dynamic and sustainable future.

#### CHAPTER 5 INNOVATION AND ENTREPRENEURSHIP DR.R.RAJAVARDHINI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INNOVATION AND ENTREPRENEURSHIP: CATALYSTS FOR ECONOMIC GROWTH AND SOCIAL CHANGE

In today's rapidly evolving world, innovation and entrepreneurship stand as fundamental pillars of economic growth and social transformation. Both concepts are intertwined, creating a dynamic ecosystem that not only generates wealth but also addresses pressing societal challenges. This essay explores the relationship between innovation and entrepreneurship, highlighting their significance, the challenges they face, and the strategies for fostering success.

#### The Essence of Innovation

Innovation is the lifeblood of progress. It encompasses the creation and implementation of new ideas, products, or processes that enhance value and solve problems. Innovations can take various forms: product innovation introduces groundbreaking goods, such as electric vehicles, while process innovation optimizes existing methods, like automation in manufacturing. Additionally, business model innovation redefines how companies operate, exemplified by subscription services that have transformed industries ranging from media to software.

The importance of innovation lies in its ability to meet evolving customer needs and adapt to changing market conditions. In a global economy characterized by rapid technological advancement, companies that fail to innovate risk obsolescence. Thus, fostering a culture of innovation is essential for organizations aiming to maintain competitive advantage and relevance.

#### The Role of Entrepreneurship

Entrepreneurship is the act of starting and managing a new business, often driven by the desire to bring innovative ideas to market. Entrepreneurs embody key traits such as risk-taking, resilience, and vision. They identify gaps in the market and are willing to invest time and resources to develop solutions. This willingness to embrace uncertainty is what differentiates entrepreneurs from traditional business leaders.

Moreover, entrepreneurship is a powerful engine for job creation. New ventures stimulate economic activity, generate employment opportunities, and enhance competition. As entrepreneurs introduce innovative products and services, they not only contribute to their own success but also catalyze broader economic growth.

#### The Synergy Between Innovation and Entrepreneurship

The relationship between innovation and entrepreneurship is symbiotic. Entrepreneurs leverage innovation to create unique value propositions that resonate with consumers. For instance, the rise of tech startups illustrates how innovative solutions can disrupt established industries, leading to enhanced consumer choices and improved services. Companies like Airbnb and Uber have redefined traditional business models, showcasing how innovation can create entirely new markets.

In addition to driving economic growth, this synergy has the potential to address pressing societal issues. Innovative entrepreneurs are increasingly focused on creating social impact, tackling challenges such as climate change, healthcare accessibility, and education. Social entrepreneurship, which combines profit motives with social objectives, exemplifies how innovation can lead to meaningful change.

#### CHAPTER 6 ENTREPRENEURIAL MOTIVATION DR.D. SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### ENTREPRENEURIAL MOTIVATION: DRIVING FORCES BEHIND SUCCESS

Entrepreneurial motivation refers to the internal and external factors that inspire individuals to initiate and sustain their own businesses. Understanding what drives entrepreneurs is crucial for fostering an environment that encourages innovation and economic growth. This essay explores the key motivators of entrepreneurial behavior, including intrinsic and extrinsic factors, and their implications for success.

#### Intrinsic Motivation

Intrinsic motivation stems from internal desires and personal satisfaction. Many entrepreneurs are driven by:

- 1. **Passion for a Idea**: A deep commitment to a particular idea or mission often ignites the entrepreneurial spirit. Entrepreneurs who are passionate about their projects are more likely to invest time and effort, as they find personal fulfillment in their work.
- 2. **Desire for Independence**: The wish for autonomy is a powerful motivator. Many entrepreneurs seek to escape traditional employment constraints, desiring control over their work environment and decision-making processes. This independence can lead to greater job satisfaction and a stronger sense of purpose.
- 3. **Challenge and Growth**: The entrepreneurial journey is fraught with challenges, and many individuals are motivated by the opportunity to push their limits. The desire for personal growth and development drives entrepreneurs to take risks and learn from failures, ultimately contributing to their success.

#### **Extrinsic Motivation**

While intrinsic factors are vital, extrinsic motivators also play a significant role in entrepreneurial motivation. These include:

- 1. **Financial Rewards**: The potential for financial gain is a primary motivator for many entrepreneurs. Successful businesses can lead to substantial income, providing the resources needed for personal and professional growth.
- 2. **Recognition and Status**: Achieving success as an entrepreneur often brings recognition and respect from peers and the community. The desire for acknowledgment can drive individuals to pursue entrepreneurial ventures, as success is frequently celebrated in society.
- 3. **Market Opportunities**: Entrepreneurs are often motivated by the identification of gaps in the market. Recognizing unmet needs or emerging trends can inspire individuals to create innovative solutions, leading to the development of new products or services.

#### The Role of Environment

The external environment significantly influences entrepreneurial motivation. Supportive ecosystems, including access to resources, mentorship, and networking opportunities, can enhance motivation. Communities that celebrate entrepreneurship and innovation often inspire individuals to take the leap into business ownership.

Additionally, socio-cultural factors play a role. Societies that value risk-taking and entrepreneurial success tend to foster higher levels of entrepreneurial activity. Conversely, cultures that stigmatize failure may deter potential entrepreneurs from pursuing their ambitions.

#### CHAPTER 7 IMPORTANCE OF ENTREPRENEURSHIP DEVELOPMENT DR.S.RAJENDRAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### IMPORTANCE OF ENTREPRENEURSHIP DEVELOPMENT

Entrepreneurship development is a critical component of economic growth and societal progress. It encompasses the process of enhancing the skills, knowledge, and mindset required to establish and run successful businesses. The significance of entrepreneurship development can be understood through its impact on economic development, job creation, innovation, and social change.

#### 1. Economic Growth

Entrepreneurship drives economic growth by creating new businesses and stimulating competition. When entrepreneurs launch ventures, they contribute to the overall economic activity, which can lead to an increase in GDP. Successful enterprises generate wealth, leading to increased consumer spending, investments, and overall economic prosperity.

#### 2. Job Creation

One of the most significant contributions of entrepreneurship development is job creation. Small and medium-sized enterprises (SMEs), often founded by entrepreneurs, are major sources of employment. As these businesses grow, they hire more employees, reducing unemployment rates and providing livelihoods for individuals and families.

#### 3. Innovation and Competitiveness

Entrepreneurs are often at the forefront of innovation, introducing new products, services, and technologies that can disrupt existing markets. This innovation fosters a competitive environment that encourages established businesses to improve and evolve. A culture of entrepreneurship promotes creativity and leads to advancements that can enhance the quality of life.

#### 4. Social Change and Community Development

Entrepreneurship development can lead to positive social change by addressing societal issues through innovative solutions. Social entrepreneurs, in particular, focus on creating ventures that generate social and environmental impact alongside profits. This dual focus can lead to improvements in education, healthcare, and sustainability within communities.

#### 5. Empowerment and Skill Development

Entrepreneurship development programs often equip individuals with essential skills such as leadership, financial literacy, and strategic thinking. These skills not only empower aspiring entrepreneurs but also contribute to personal growth and confidence. Furthermore, fostering an entrepreneurial mindset encourages resilience and adaptability, which are vital in today's rapidly changing job market.

#### 6. Diversification of the Economy

Encouraging entrepreneurship helps diversify the economy by supporting a range of industries and sectors. This diversification reduces dependence on a few key industries, making economies more resilient to market fluctuations and global challenges. A diverse entrepreneurial ecosystem fosters stability and sustainability in economic growth.

#### CHAPTER 8 INTRODUCTION OF PROJECT REPORT DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION TO THE PROJECT REPORT

The introduction of a project report serves as a critical foundation for understanding the purpose, scope, and significance of the project undertaken. This section provides a comprehensive overview of the project, outlining its objectives, methodology, and the context in which it was developed.

In an era marked by rapid technological advancements and evolving market dynamics, the importance of entrepreneurship has never been more pronounced. This project focuses on [insert specific project topic or theme], aiming to explore its implications for [specific industry, community, or target audience].

The primary objectives of this project are to:

- 1. **Identify Key Issues**: Analyze the current challenges and opportunities within [specific context].
- 2. **Investigate Best Practices**: Examine successful models and strategies employed by entrepreneurs in the field.
- 3. **Provide Recommendations**: Offer actionable insights and recommendations to enhance entrepreneurial development and effectiveness.

To achieve these objectives, a combination of qualitative and quantitative research methods was employed, including surveys, interviews, and case studies. This approach enables a well-rounded understanding of the factors influencing [insert topic], thereby facilitating informed conclusions and recommendations.

The significance of this project lies in its potential to contribute to the broader discourse on entrepreneurship and innovation. By highlighting successful strategies and identifying areas for improvement, this report aims to provide valuable insights for aspiring entrepreneurs, policymakers, and stakeholders within [specific industry or community].

In summary, this project report seeks to illuminate the vital role of [insert topic] in fostering sustainable economic growth and social development. The findings and recommendations presented herein are intended to inspire action and drive positive change within the entrepreneurial landscape.

#### Objectives of the Project

The primary objectives of this project are to:

- 1. **Identify Key Issues**: Analyze the prevailing challenges and opportunities within [specific context] that affect entrepreneurial activity.
- 2. **Investigate Best Practices**: Explore successful models and strategies employed by entrepreneurs and organizations in the field, drawing insights from both local and global examples.
- 3. **Provide Recommendations**: Develop actionable strategies and recommendations aimed at policymakers, educators, and aspiring entrepreneurs to support and enhance entrepreneurial development.

#### CHAPTER 9 INTRODUCTION OF PROJECT IDENTIFICATION DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION TO PROJECT IDENTIFICATION

Project identification is a crucial phase in the project management lifecycle, laying the groundwork for successful project planning and execution. It involves recognizing and defining potential projects that align with organizational goals, community needs, or market demands. This introductory section outlines the significance of project identification, the criteria for evaluating potential projects, and the methodologies employed in this process.

#### Background and Importance

In an increasingly competitive and rapidly changing environment, organizations must be agile in identifying projects that can drive innovation, improve efficiency, and respond to emerging challenges. Effective project identification enables organizations to prioritize initiatives that align with their strategic objectives and resource capabilities. This phase not only sets the stage for successful project outcomes but also ensures that resources are allocated to the most promising and impactful endeavors.

#### **Objectives of Project Identification**

The primary objectives of project identification include:

- 1. Alignment with Strategic Goals: Ensuring that potential projects support the organization's vision, mission, and long-term objectives.
- 2. **Needs Assessment**: Identifying gaps or opportunities in the market or community that projects can address.
- 3. **Feasibility Analysis**: Evaluating the practicality and potential return on investment of proposed projects to ensure they are viable.

#### Methodologies for Project Identification

Several methodologies can be utilized during the project identification phase, including:

- **Stakeholder Consultation**: Engaging with stakeholders to gather insights on needs and priorities, ensuring that projects are relevant and impactful.
- Market Research: Analyzing industry trends, competitive landscapes, and customer demands to identify opportunities for new projects.
- **SWOT Analysis**: Assessing the strengths, weaknesses, opportunities, and threats related to potential projects, providing a comprehensive overview of their viability.
- **Brainstorming Sessions**: Facilitating collaborative discussions among team members and stakeholders to generate innovative project ideas.

#### Significance of Effective Project Identification

Proper project identification plays a pivotal role in optimizing resource utilization and enhancing project success rates. By systematically identifying and evaluating potential projects, organizations can avoid the pitfalls of misallocation and ensure that they pursue initiatives with the highest potential for impact. Additionally, effective project identification fosters a proactive organizational culture that embraces change and innovation.

#### CHAPTER 10 SMALL ENTREPRENEURS DEVELOPMENT DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### SMALL ENTREPRENEURS DEVELOPMENT

#### Introduction

Small entrepreneurs play a vital role in driving economic growth, job creation, and innovation. These individuals often operate within local communities, bringing new ideas and services that meet specific needs. Developing small entrepreneurs is essential for fostering a vibrant economy and ensuring sustainable development. This section explores the significance of small entrepreneurs, the challenges they face, and strategies for their development.

#### Importance of Small Entrepreneurs

- 1. **Economic Contribution**: Small businesses contribute significantly to national economies. They account for a substantial portion of employment and GDP in many countries, making them crucial for economic stability and growth.
- 2. **Job Creation**: Small entrepreneurs are responsible for creating a majority of new jobs. By launching their ventures, they provide employment opportunities within their communities, helping to reduce unemployment rates.
- 3. **Innovation and Diversity**: Small businesses often bring innovative products and services to the market. Their ability to adapt quickly to changes and respond to customer needs fosters a diverse marketplace that benefits consumers.
- 4. **Community Development**: Small entrepreneurs often reinvest in their local communities, supporting local suppliers and services. This fosters a sense of community and strengthens local economies.

#### Challenges Faced by Small Entrepreneurs

- 1. Access to Capital: One of the most significant barriers for small entrepreneurs is securing funding. Traditional financing routes, such as banks, can be challenging to navigate, often leading to limited options for loans or investments.
- 2. **Market Competition**: Small businesses often face stiff competition from larger corporations with greater resources. This can make it difficult for them to establish a foothold in the market.
- 3. **Regulatory Hurdles**: Navigating complex regulations and compliance requirements can be daunting for small entrepreneurs, consuming valuable time and resources.
- 4. Lack of Support and Mentorship: Many small entrepreneurs lack access to mentorship and networks that can provide guidance and support, hindering their growth potential.

#### Strategies for Small Entrepreneurs Development

- 1. Access to Finance: Creating funding programs specifically for small businesses, such as microloans or grants, can enhance access to capital. Additionally, fostering relationships with local banks and financial institutions can provide entrepreneurs with the support they need.
- 2. **Training and Education**: Offering training programs focused on business management, marketing, and financial literacy can equip small entrepreneurs with essential skills. Workshops and seminars can provide valuable insights and foster networking opportunities.
- 3. **Business Incubators and Accelerators**: Establishing incubators and accelerators can provide small entrepreneurs with resources, mentorship, and networking opportunities to help them scale their businesses. These programs often offer structured support and guidance.

## MANAGEMENT ACCOUNTING

Edited by

**P.SAMPATHKUMAR** 



#### MANAGEMENT ACCOUNTING

#### Copyright ©2021 by P.SAMPATHKUMAR

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-777-3

Publisher:Ponnaiyah Ramajayam Institute of science and Technology (PRIST)

#### TABLEOFCONTENTS

CHAPTER	CHAPTER TITLE	TITLE OF THE AUTHOR	PAGE NO
1	TOOLS OF MANAGEMENT		1
	ACCOUNTING	DR.R.SELVARAJ	
2	SCOPE OF MANAGEMENT		11
	ACCOUNTING	DR. S. KAMARAJU	
3	SIGNIFICANCE OF BUDGETARY		22
	CONTROL	DR.G.KARTHIGA	
4	LEASE FINANCING	DR.V.SRIDEVI	32
5	PROCESS OF VALUE CHAIN		42
	ANALYSIS	DR.R.RAJAVARDHINI	
6	INTRODUCTION OF QUALITY		52
	COSTING	DR.D.SILAMBARASAN	
7	INTRODUCTION OF MARGINAL		62
	COSTING	DR.S.RAJENDRAN	
8	<b>REPORTING TO MANAGEMENT</b>	DR.R.SELVARAJ	72
9	PROCESS OF DECISION MAKING	DR. S. KAMARAJU	82
10	LIFE CYCLE COSTING:		92
	INTRODUCTION	DR.G.KARTHIGA	
	REFERENCE		101

#### CHAPTER 1 TOOLS OF MANAGEMENT ACCOUNTING DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### ESSENTIAL TOOLS OF MANAGEMENT ACCOUNTING

#### Introduction

Management accounting plays a crucial role in guiding organizational decision-making, planning, and control. Unlike financial accounting, which focuses on historical data and external reporting, management accounting emphasizes internal processes and future-oriented strategies. This essay explores the key tools and techniques utilized in management accounting, highlighting their importance in enhancing business performance.

#### Budgeting

Budgeting is one of the foundational tools of management accounting. It involves creating a financial plan that outlines expected revenues and expenditures over a specific period. Various types of budgets, such as flexible, fixed, and zero-based budgets, serve different purposes. By establishing a financial framework, budgeting facilitates resource allocation and helps organizations align their financial goals with operational activities.

#### Variance Analysis

Variance analysis complements budgeting by comparing actual financial performance against budgeted figures. This tool helps organizations identify discrepancies and understand the underlying reasons for any variances. By analyzing these differences, managers can make informed decisions to address inefficiencies and adjust strategies accordingly, fostering a culture of accountability.

#### **Cost-Volume-Profit (CVP) Analysis**

Cost-Volume-Profit analysis is instrumental in understanding the relationship between costs, sales volume, and profits. This tool allows managers to assess how changes in costs and volume affect overall profitability, aiding in critical decisions related to pricing, product lines, and market entry. By determining the break-even point, organizations can make more informed strategic choices.

#### **Activity-Based Costing (ABC)**

Activity-Based Costing allocates overhead costs based on the activities that drive those costs, providing a more accurate picture of product profitability. Unlike traditional costing methods, which may oversimplify cost allocation, ABC offers deeper insights into where resources are consumed. This enables organizations to identify non-value-added activities and focus on improving efficiency.

#### **Performance Measurement**

Performance measurement tools, such as Key Performance Indicators (KPIs) and the Balanced Scorecard, provide frameworks for assessing organizational efficiency and effectiveness. By measuring performance against strategic goals, organizations can track progress and make data-

#### CHAPTER 2 SCOPE OF MANAGEMENT ACCOUNTING DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

### SCOPE OF MANAGEMENT ACCOUNTING Introduction

Management accounting is a critical discipline that focuses on providing financial and non-financial information to managers for effective decision-making, planning, and control within an organization. Unlike financial accounting, which primarily serves external stakeholders, management accounting emphasizes internal processes and strategic objectives. This essay explores the comprehensive scope of management accounting, highlighting its various functions, tools, and areas of application.

#### **1. Planning and Budgeting**

One of the primary functions of management accounting is to assist in planning and budgeting. Management accountants develop budgets that outline expected revenues and expenditures, providing a financial roadmap for the organization. This involves forecasting future business conditions, setting financial goals, and allocating resources efficiently. Effective budgeting helps organizations anticipate challenges and seize opportunities, ensuring alignment with strategic objectives.

#### 2. Performance Measurement

Management accounting plays a vital role in performance measurement. By utilizing Key Performance Indicators (KPIs) and other metrics, management accountants assess how well an organization is achieving its goals. Performance measurement involves not only financial indicators but also operational and strategic metrics, allowing managers to track progress and identify areas for improvement. This holistic approach fosters accountability and drives continuous improvement.

#### 3. Cost Management

Cost management is a significant aspect of management accounting, involving the identification, analysis, and control of costs associated with operations. Tools such as Activity-Based Costing (ABC) and standard costing help organizations allocate costs accurately and identify non-value-added activities. By understanding cost drivers and implementing cost control measures, management accountants support profitability and enhance operational efficiency.

#### 4. Decision Support

Management accounting provides essential support for decision-making processes. Techniques such as Cost-Volume-Profit (CVP) analysis, break-even analysis, and financial modeling equip managers with the information needed to make informed choices. By analyzing various scenarios, management accountants help organizations evaluate investment opportunities, pricing strategies, and resource allocation decisions, ultimately leading to better outcomes.

#### 5. Strategic Management

#### CHAPTER 3 SIGNIFICANCE OF BUDGETARY CONTROL DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### SIGNIFICANCE OF BUDGETARY CONTROL IN MANAGEMENT

#### Introduction

Budgetary control is a vital aspect of management accounting that involves the preparation, monitoring, and analysis of budgets to ensure that organizational objectives are achieved. It serves as a framework for financial planning and control, enabling organizations to allocate resources effectively and measure performance against set targets. This essay explores the significance of budgetary control, highlighting its benefits in enhancing organizational efficiency, accountability, and strategic alignment.

#### **1. Financial Planning and Resource Allocation**

One of the primary significances of budgetary control is its role in financial planning. By preparing detailed budgets, organizations can forecast future revenues and expenditures, allowing for effective resource allocation. This process helps in identifying financial needs, prioritizing expenditures, and ensuring that resources are directed toward activities that align with organizational goals. Effective budgeting minimizes waste and enhances operational efficiency.

#### 2. Performance Measurement and Accountability

Budgetary control provides a benchmark against which actual performance can be measured. By comparing actual results with budgeted figures, management can identify variances and assess organizational performance. This process fosters accountability among managers and departments, as they are responsible for achieving budgeted targets. Regular performance reviews based on budgetary control encourage a culture of accountability and continuous improvement.

#### **3. Enhanced Decision-Making**

Budgetary control supports informed decision-making by providing relevant financial information. Managers can analyze variances and trends to understand the financial implications of their decisions. This analysis enables them to make timely adjustments, reallocate resources, or implement corrective actions as needed. By facilitating data-driven decision-making, budgetary control enhances overall organizational effectiveness.

#### 4. Risk Management

Incorporating budgetary control into organizational practices helps in identifying and mitigating financial risks. By closely monitoring budgets and analyzing variances, management can detect potential issues before they escalate. This proactive approach allows organizations to implement risk management strategies, ensuring financial stability and minimizing the impact of unforeseen events on the organization's operations.

#### **5.** Strategic Alignment

#### CHAPTER 4 LEASE FINANCING DR.V.SRIDEVI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### Introduction

Lease financing is a strategic financial arrangement that allows individuals and organizations to use assets without purchasing them outright. It involves a contractual agreement between a lessor (the owner of the asset) and a lessee (the user of the asset) that outlines the terms of use, including payment schedules and duration. This essay explores the concept of lease financing, its types, advantages, disadvantages, and its significance in business operations.

#### **Types of Lease Financing**

#### 1. **Operating Lease**:

- **Definition**: A short-term lease where the lessor retains ownership of the asset.
- **Characteristics**: Typically does not cover the asset's entire useful life, allowing the lessee to return the asset after the lease term. This type is often used for equipment that becomes obsolete quickly, such as computers and vehicles.

#### 2. Finance Lease (Capital Lease):

- **Definition**: A long-term lease that transfers substantial risks and rewards of ownership to the lessee.
- **Characteristics**: The lease term usually covers a significant portion of the asset's useful life, and the lessee may have the option to purchase the asset at the end of the lease term. This type is common for machinery and real estate.

#### 3. Sale and Leaseback:

- **Definition**: A financial arrangement where an organization sells an asset and simultaneously leases it back from the buyer.
- **Purpose**: Provides immediate capital for the seller while allowing continued use of the asset.

#### 4. Operating Lease vs. Finance Lease:

• Operating leases are generally shorter and off-balance-sheet, while finance leases are longer-term and usually recorded as liabilities on the balance sheet.

#### **Advantages of Lease Financing**

#### 1. **Preservation of Capital**:

• Leasing allows organizations to conserve cash by avoiding large upfront costs associated with purchasing assets. This enables them to allocate capital to other essential business activities.

#### 2. Flexibility:

• Lease agreements can be tailored to meet specific operational needs, including lease duration, payment schedules, and options for upgrading or purchasing the asset.

#### 3. Tax Benefits:

• Lease payments are often considered operating expenses, making them taxdeductible. This can lead to significant tax savings for businesses.

#### 4. Access to the Latest Technology:

• Leasing provides organizations with the opportunity to use the latest equipment and technology without the burden of ownership, facilitating competitiveness and innovation.

#### CHAPTER 5 PROCESS OF VALUE CHAIN ANALYSIS DR.R.RAJAVARDHINI Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### THE PROCESS OF VALUE CHAIN ANALYSIS

#### Introduction

Value chain analysis is a strategic tool that organizations utilize to identify and enhance the activities that create value for customers. Introduced by Michael Porter in his seminal work "Competitive Advantage," this analysis allows businesses to understand how various activities interrelate and contribute to competitive advantage. This essay explores the process of value chain analysis, outlining its components, steps, and significance in today's business landscape.

#### **Understanding the Value Chain**

The value chain consists of a series of activities that organizations perform to deliver a product or service. These activities are categorized into primary and support activities.

**Primary Activities** are directly related to the creation and delivery of a product or service. They include:

- **Inbound Logistics**: Involves the receiving, warehousing, and inventory management of raw materials.
- **Operations**: Refers to the processes that transform inputs into finished products.
- **Outbound Logistics**: Encompasses the distribution of the final product to customers.
- Marketing and Sales: Involves strategies to promote and sell the product.
- Service: Consists of activities that maintain and enhance the product's value after purchase.

Support Activities enhance and enable primary activities and include:

- **Procurement**: The acquisition of goods and services needed for operations.
- **Technology Development**: Encompasses research, development, and process automation.
- Human Resource Management: Involves recruitment, training, and employee development.
- Firm Infrastructure: Refers to organizational structure, planning, and management systems.

#### **Steps in the Value Chain Analysis Process**

The process of value chain analysis involves several key steps:

#### **Step 1: Identify Primary and Support Activities**

The first step is to map out all primary and support activities involved in delivering the product or service. This comprehensive understanding is crucial for assessing the entire operational process, from sourcing raw materials to delivering the finished product to customers.

#### CHAPTER 6 INTRODUCTION OF QUALITY COSTING DR.D.SILAMBARASAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION TO QUALITY COSTING

Quality costing is an essential concept in management accounting that emphasizes the economic impact of quality-related activities within an organization. This approach centers on understanding the costs associated with ensuring quality, as well as the financial implications of failing to achieve it. As businesses operate in increasingly competitive markets, the importance of quality as a differentiator becomes paramount, and effective quality costing serves as a strategic tool for continuous improvement.

The primary aim of quality costing is to quantify the costs related to quality initiatives, which are typically categorized into four main components:

- 1. **Prevention Costs**: These are the costs incurred to prevent defects and ensure that products or services meet established quality standards. Investment in prevention is often seen as a proactive approach to quality management. Key activities that fall under this category include quality training programs for employees, the implementation of process control systems, and thorough quality planning and analysis. By focusing on prevention, organizations can reduce the likelihood of defects and improve overall efficiency.
- 2. **Appraisal Costs**: Appraisal costs arise from efforts to measure and monitor quality to ensure compliance with specifications. These costs include expenses related to inspection, testing, and quality audits. While these activities may seem like an additional expense, they play a crucial role in identifying potential defects early in the production process, ultimately saving costs associated with failures and rework later on.
- 3. **Internal Failure Costs**: These costs are incurred when defects are discovered before the product reaches the customer. They include expenses related to rework, scrap materials, and the resources used to correct errors. By analyzing internal failure costs, organizations can identify inefficiencies in their processes and take corrective action to minimize waste and improve overall quality.
- 4. **External Failure Costs**: External failure costs occur when defects are identified after the product has been delivered to the customer. These costs can be particularly damaging, as they include warranty claims, product returns, and loss of reputation. The impact of external failures extends beyond immediate financial costs; they can lead to decreased customer loyalty and trust, ultimately harming long-term profitability.

Understanding these four components is crucial for organizations looking to implement effective quality costing strategies. By conducting a thorough analysis of quality costs, businesses can identify areas for improvement, optimize resource allocation, and enhance their overall quality management processes.

#### CHAPTER 7 INTRODUCTION OF MARGINAL COSTING DR.S.RAJENDRAN Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION TO MARGINAL COSTING

Marginal costing is an important managerial accounting technique that focuses on the cost of producing one additional unit of a product. It helps businesses understand the impact of variable costs on overall profitability and decision-making.

#### **Key Concepts:**

#### 1. Variable Costs vs. Fixed Costs:

- **Variable Costs**: Costs that change with the level of production, such as materials and direct labor.
- **Fixed Costs**: Costs that remain constant regardless of production levels, such as rent and salaries.

#### 2. Contribution Margin:

• The contribution margin is calculated by subtracting variable costs from sales revenue. It indicates how much revenue is available to cover fixed costs and contribute to profit.

#### 3. Break-even Analysis:

• Marginal costing is useful for break-even analysis, helping businesses determine the sales volume at which total revenues equal total costs, resulting in neither profit nor loss.

#### 4. Decision-Making:

• It aids in short-term decision-making, such as pricing, product mix, and make-or-buy decisions, by focusing on the incremental costs and benefits.

0

#### **Advantages of Marginal Costing:**

- **Simplifies Decision-Making**: Provides clear insights into the financial implications of various decisions.
- Focus on Cost Control: Encourages managers to monitor and manage variable costs effectively.
- **Performance Evaluation**: Facilitates the evaluation of different segments of the business by analyzing their contribution margins.

•

#### **Limitations of Marginal Costing:**

- **Ignores Fixed Costs**: By focusing solely on variable costs, it may lead to misleading conclusions in long-term decision-making.
- Not GAAP-Compliant: Marginal costing is not accepted under Generally Accepted Accounting Principles (GAAP), which can complicate financial reporting.

In summary, marginal costing is a valuable tool for businesses to analyze costs and make informed decisions. By understanding how costs behave with changes in production levels, managers can optimize resource allocation and enhance profitability.

#### CHAPTER 8 REPORTING TO MANAGEMENT DR.R.SELVARAJ Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### **REPORTING TO MANAGEMENT**

#### A CRITICAL TOOL FOR EFFECTIVE DECISION-MAKING

In today's fast-paced business environment, the ability to make informed decisions is paramount for organizational success. Management reporting serves as a crucial mechanism through which leaders receive vital information that influences strategic direction and operational efficiency. This essay explores the purpose, types, key components, best practices, and the impact of technology on management reporting, highlighting its significance in fostering effective decision-making.

#### Purpose of Management Reporting

The primary purpose of management reporting is to facilitate informed decision-making. By presenting comprehensive data about financial performance, operational efficiency, and market trends, management reports empower leaders to evaluate current strategies and adjust them as necessary. Furthermore, these reports play a pivotal role in performance monitoring, allowing managers to track progress against established objectives and key performance indicators (KPIs). This ongoing assessment is essential for identifying strengths and weaknesses within the organization, thereby guiding resource allocation and strategic planning.

#### Types of Management Reports

Management reports can be categorized into several types, each serving specific purposes. Financial reports, which include income statements, balance sheets, and cash flow statements, provide insights into the organization's fiscal health. These documents are essential for understanding profitability and financial stability. In contrast, operational reports focus on metrics related to production efficiency, sales performance, and inventory management, offering a detailed view of day-to-day operations.

Another critical type of report is the variance analysis report, which compares budgeted figures to actual performance. This comparison helps identify deviations, uncovering the underlying causes of variances and enabling managers to take corrective actions. Additionally, forecasting reports predict future performance based on historical data, facilitating long-term planning and budgeting efforts.

#### Key Components of Management Reports

Effective management reports share several key components that enhance their utility. An executive summary is crucial, as it distills the report's main findings and recommendations into a concise overview. This allows busy executives to grasp the essential information quickly. Moreover, clear data presentation using tables, graphs, and charts aids in visualizing complex information, making it more accessible.

Analysis and interpretation of the presented data are equally important. This section provides insights into trends, potential issues, and opportunities, enabling management to make data-driven

#### CHAPTER 9 PROCESS OF DECISION MAKING DR. S. KAMARAJU Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### THE PROCESS OF DECISION MAKING

Decision making is a critical aspect of management and leadership, influencing every level of an organization. Whether addressing everyday operational issues or strategic planning, a structured approach can enhance the quality of decisions and their outcomes. Here's a comprehensive overview of the decision-making process.

#### 1. Identifying the Problem or Opportunity

The first step in decision making is recognizing that a decision needs to be made. This could involve identifying a problem that requires a solution or an opportunity that should be seized. Clearly defining the issue at hand is crucial, as it sets the foundation for all subsequent steps.

#### 2. Gathering Information

Once the problem or opportunity is identified, the next step is to gather relevant information. This includes:

- **Data Collection**: Gather quantitative and qualitative data from various sources, such as market research, financial reports, and internal performance metrics.
- **Stakeholder Input**: Consult with team members, stakeholders, and experts to gain diverse perspectives and insights.

#### 3. Identifying Alternatives

With sufficient information at hand, the next step is to brainstorm and identify potential alternatives. This involves generating a list of possible solutions or courses of action. Encouraging creativity and open-mindedness during this phase can lead to innovative options.

#### 4. Evaluating Alternatives

Each alternative should be carefully evaluated against a set of criteria, which may include:

- **Feasibility**: Assess whether the option can be realistically implemented.
- **Cost-Benefit Analysis**: Weigh the potential benefits against the costs involved.
- **Risks**: Identify any potential risks associated with each alternative.

#### 5. Making the Decision

After evaluating the alternatives, it's time to make a decision. This involves selecting the option that best addresses the problem or opportunity, based on the evaluations conducted. It's essential to ensure that the chosen alternative aligns with the organization's goals and values.

#### 6. Implementing the Decision

Once a decision is made, it needs to be implemented effectively. This step includes:

#### CHAPTER 10 LIFE CYCLE COSTING: INTRODUCTION DR.G.KARTHIGA Department of Commerce, Ponnaiyah Ramajayam Institute of science and Technology(PRIST)

#### INTRODUCTION TO LIFE CYCLE COSTING

Life Cycle Costing (LCC) is a comprehensive approach to assessing the total costs associated with a product, project, or system throughout its entire life cycle. From conception and design to production, operation, maintenance, and eventual disposal, LCC provides a holistic view of costs, enabling better financial decision-making and resource allocation.

#### Key Concepts

- 1. **Definition**: Life Cycle Costing involves calculating all costs related to a product or project over its life span. This includes initial costs (like design and manufacturing), operational costs (such as energy and maintenance), and end-of-life costs (like disposal and recycling).
- 2. Life Cycle Phases: The life cycle of a product typically includes the following phases:
  - **Development**: Costs incurred during research, design, and prototyping.
  - **Production**: Expenses associated with manufacturing the product.
  - **Operation**: Ongoing costs during the product's use, including maintenance and energy consumption.
- 3. **Total Cost of Ownership (TCO)**: LCC is often associated with the concept of Total Cost of Ownership, which emphasizes the long-term costs of acquiring and using an asset rather than just the initial purchase price.

#### Importance of Life Cycle Costing

- 1. **Informed Decision-Making**: By considering the full range of costs, organizations can make more informed decisions about product design, purchasing, and resource allocation, ultimately leading to better financial outcomes.
- 2. **Sustainability and Environmental Impact**: LCC can highlight the environmental costs associated with a product, encouraging organizations to pursue more sustainable practices and consider the life cycle impacts of their products.
- 3. **Cost Optimization**: Understanding life cycle costs helps organizations identify areas where efficiencies can be gained, leading to reduced overall expenses and improved profitability.

#### Challenges in Life Cycle Costing

- 1. **Data Collection**: Gathering accurate data for each phase of the life cycle can be challenging and time-consuming.
- 2. **Complexity**: The process can become complex due to the need to consider various cost elements and their interdependencies.
- 3. **Uncertainty**: Predicting future costs, especially for long-lived products, involves a degree of uncertainty, which can affect decision-making.

#### Conclusion

Life Cycle Costing is a vital approach for organizations aiming to understand the total financial impact of their products or projects over time. By focusing on all associated costs, LCC empowers businesses to make strategic decisions that promote efficiency, sustainability, and long-term profitability. As industries increasingly prioritize environmental responsibility and resource optimization, LCC will continue to play a crucial role in shaping cost-effective and sustainable practices.

## **VLSI DESIGN**

Edited by

S.GOVINDASWAMY



VLSI design Copyright © 2021 by Mr.S.Govindaswamy

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-881-7

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST) Date of Publication : 17.11.2021
## **TABLE OF CONTENTS**

Introduction	03
<b>CHAPTER 1 -</b> MOS transistor principles	10
Mrs. R. Prasannadevi	
CHAPTER 2 - Combinational logic circuits	32
Mrs. R. Prasannadevi	
CHAPTER 3 - Sequential logic circuits	58
Mr. J. Vivek Raja	
<b>CHAPTER 4 -</b> Interconnect memory architecture	125
Mr.R.Elangovan	
<b>CHAPTER 5</b> – Design of Arithmetic Building Blocks	158
Mr.S.Govindaswamy	
CHAPTER 6 – contacts and Transistors	163
Mrs. R. Prasannadevi	
CHAPTER 7 – Pass transistor	179
Mrs. R. Prasannadevi	
CHAPTER 8 – VLSI Design Flow	187
Mr. J. Vivek Raja	
CHAPTER 9 – Inverter Delays	195
Mr.R.Elangovan	
CHAPTER 10 – CMOS Testing	206
Mr.S.Govindaswamy	
REFERENCES	216

## CHAPTER 1 MOS transistor principles

Mrs. R. Prasannadevi

**MOS Transistor Principles** involve understanding the operation, structure, and characteristics of Metal-Oxide-Semiconductor Field-Effect Transistors (MOSFETs). MOSFETs are a key component in modern electronics, widely used in digital circuits, analog circuits, and power electronics. Here's a detailed look at MOSFET principles:

#### 1. Basic Structure and Operation

#### **1.1 Structure**

- Source (S): The terminal through which carriers (electrons or holes) enter the transistor.
- **Drain** (**D**): The terminal through which carriers exit the transistor.
- Gate (G): The terminal that controls the flow of carriers between the source and drain.
- **Body** (**B**): The terminal connected to the substrate, often tied to the source in many applications.

The MOSFET consists of a gate electrode separated from the semiconductor channel (between the source and drain) by a thin insulating layer of silicon dioxide (SiO<sub>2</sub>). This insulation is crucial for the operation of the MOSFET, as it allows the gate voltage to control the channel without direct electrical contact.

#### **1.2 Types of MOSFETs**

- Enhancement-mode MOSFET (E-MOSFET): The channel is normally off when the gate-source voltage (V\_GS) is zero. It requires a positive (for n-channel) or negative (for p-channel) gate voltage to create a conducting channel.
- **Depletion-mode MOSFET (D-MOSFET):** The channel is normally on when V\_GS is zero. It requires a gate voltage to deplete the channel and turn off the transistor.

#### 2. Operating Principles

#### 2.1 Threshold Voltage (V\_T)

• **Definition:** The minimum gate-source voltage required to create a conducting channel between the source and drain. For an enhancement-mode n-channel MOSFET, V\_T is positive, and for a p-channel MOSFET, V\_T is negative.

#### **2.2 Operating Regions**

- **Cutoff Region:** When V\_GS is less than the threshold voltage (V\_T), the MOSFET is off, and there is no conduction between the source and drain.
- **Triode (or Linear) Region:** When V\_GS is greater than V\_T and V\_DS (drain-source voltage) is small, the MOSFET behaves like a variable resistor. The current between the drain and source is controlled by V\_GS.

#### **Key Characteristics**

- **3.1 Threshold Voltage (V\_T)** 
  - Variation: Affected by temperature, process variations, and biasing conditions.

#### 3.2 Drain-Source Voltage (V\_DS)

• Impact: Influences the operation region and current flow through the MOSFET.

## CHAPTER 2 Combinational logic circuits

Mrs. R. Prasannadevi

**Combinational Logic Circuits** are fundamental building blocks in digital electronics. Unlike sequential logic circuits, which depend on past inputs and states (memory), combinational logic circuits produce outputs solely based on the current inputs. Here's a detailed overview of combinational logic circuits:

#### 1. Basic Concepts

#### 1.1 Definition

Combinational logic circuits perform operations based on a combination of their inputs to produce a corresponding output. The output is a function of the present inputs only and not influenced by previous inputs or states.

#### 1.2 Truth Tables

A truth table is a fundamental tool used to describe the behavior of a combinational logic circuit. It lists all possible input combinations and the corresponding outputs.

#### 2. Basic Logic Gates

The fundamental building blocks of combinational logic circuits are basic logic gates. Each gate implements a basic logical operation.

- AND Gate:
  - **Operation:** Outputs 1 if all inputs are 1; otherwise, outputs 0.
  - **Symbol:** A dot  $(\cdot)$  or multiplication sign.
- OR Gate:
  - **Operation:** Outputs 1 if at least one input is 1; otherwise, outputs 0.
  - **Symbol:** A plus sign (+).
- NOT Gate (Inverter):
  - **Operation:** Outputs the inverse of the input; outputs 1 if the input is 0, and vice versa.
  - **Symbol:** A bar over the input or a prime (').
- NAND Gate:
  - **Operation:** Outputs 0 only if all inputs are 1; otherwise, outputs 1.
  - **Symbol:** An AND gate with a bubble (inversion) at the output.
- NOR Gate:
  - **Operation:** Outputs 1 only if all inputs are 0; otherwise, outputs 0.
  - Symbol: An OR gate with a bubble (inversion) at the output.
- XOR Gate (Exclusive OR):
  - **Operation:** Outputs 1 if an odd number of inputs are 1; otherwise, outputs 0.
  - **Symbol:** An XOR symbol  $(\bigoplus)$ .
- XNOR Gate (Exclusive NOR):
  - **Operation:** Outputs 1 if an even number of inputs are 1; otherwise, outputs 0.
  - **Symbol:** An XOR gate with a bubble (inversion) at the output.

#### **Combinational Logic Circuits**

#### **3.1 Simplification Techniques**

To design efficient combinational circuits, simplification techniques are used:

- Boolean Algebra: Simplifies Boolean expressions using algebraic rules.
- **Karnaugh Maps (K-Maps):** Graphical method for simplifying Boolean functions, helping to visualize and minimize logic expressions.
- **Quine-McCluskey Method:** A tabular method for minimization, often used for computer-aided design.

## CHAPTER 3 Sequential logic circuits

#### Mr. J. Vivek Raja

**Sequential Logic Circuits** are a fundamental component of digital electronics, distinguished by their reliance on both current and past input states to determine their output. Unlike combinational logic circuits, which depend only on present inputs, sequential logic circuits have memory elements that store information about past inputs. This memory capability allows sequential circuits to perform tasks such as counting, storing data, and implementing finite state machines.

#### 1. Basic Concepts

#### 1.1 Definition

Sequential logic circuits are circuits whose output depends not only on the current inputs but also on the history of inputs. This behavior is due to their ability to store state information, making them suitable for tasks requiring memory or sequential operations.

#### **1.2 Memory Elements**

The key to sequential circuits is their use of memory elements, which store the state of the circuit. Common memory elements include:

- Flip-Flops: The basic building blocks of sequential circuits, used to store binary data.
- Latches: Similar to flip-flops but typically level-sensitive rather than edge-sensitive.

#### 2. Types of Sequential Circuits

Sequential circuits can be classified into two main types:

#### 2.1 Synchronous Sequential Circuits

- **Definition:** Sequential circuits where changes in state occur at discrete times, typically synchronized with a clock signal. All state changes happen on clock edges (rising or falling).
- **Components:** Include flip-flops, registers, and counters. The state transitions are controlled by the clock signal.
- Examples:
  - **Registers:** Store multi-bit data and are used to hold intermediate data values.
  - **Counters:** Count in binary or other sequences, such as up counters, down counters, and up/down counters.
  - Finite State Machines (FSMs): Implement state transitions based on input signals and current states.

#### 2.2 Asynchronous Sequential Circuits

- **Definition:** Sequential circuits where state changes occur in response to input changes, not synchronized with a clock signal. These circuits change states whenever inputs change, making them less predictable than synchronous circuits.
- **Components:** Include latches, pulse generators, and combinational logic. Asynchronous circuits are often used in simple control applications.
- Examples:
  - Serial Adder: Adds binary numbers serially, one bit at a time.
  - Event Counters: Count events based on input changes rather than a clock signal.

#### **3. Basic Memory Elements**

**3.1 Flip-Flops** 

•

- D Flip-Flop (Data or Delay Flip-Flop):
  - **Operation:** Captures the value of the data input (D) on the rising or falling edge of the clock signal and stores it.
  - **Characteristics:** Provides a single-bit memory and is commonly used in registers and memory storage.
- JK Flip-Flop:
  - **Operation:** Allows toggling between states based on the inputs (J and K) and the clock signal. It can be configured to operate as a D flip-flop, T flip-flop, or JK flip-flop.

## CHAPTER 4 Interconnect memory architecture

Mr.R.Elangovan

**Interconnect Memory Architecture** refers to the design and organization of how memory components are connected and interact within a computer system. Efficient interconnects are crucial for achieving high performance and reliability in both general-purpose computing and specialized applications like embedded systems, high-performance computing (HPC), and data centers. Here's an overview of key concepts and components involved in interconnect memory architecture:

#### **1. Basic Concepts**

#### **1.1 Interconnect**

- **Definition:** The system of pathways and protocols that connect various components in a computer system, such as processors, memory units, and peripheral devices.
- **Purpose:** Facilitates data transfer and communication between these components, ensuring efficient and reliable operation.

#### **1.2 Memory Architecture**

- **Definition:** The organization of memory components within a system, including their layout, addressing, and access methods.
- **Components:** Includes various types of memory such as RAM (Random Access Memory), ROM (Read-Only Memory), cache memory, and storage.

#### 2. Types of Memory Interconnects

#### **2.1 Bus-Based Interconnects**

- **Definition:** A communication pathway that allows multiple components to share data.
- **Components:** Includes address lines, data lines, and control lines.
- Examples:
  - **System Bus:** Connects the CPU, memory, and peripherals. It typically includes an address bus, data bus, and control bus.
  - **Memory Bus:** Specifically connects the CPU to the main memory, allowing data to be transferred between them.

#### **2.2 Point-to-Point Interconnects**

- **Definition:** Direct connections between two components without shared communication pathways.
- Advantages: Lower latency and higher bandwidth compared to bus-based systems.
- Examples:
  - **HyperTransport:** A high-speed, point-to-point link used in AMD processors for connecting CPUs and memory.
  - Intel QuickPath Interconnect (QPI): Used for high-speed communication between processors and memory in Intel systems.

#### 2.3 Crossbar Switches

- **Definition:** A switch network that connects multiple inputs to multiple outputs in a non-blocking manner.
- Advantages: Provides high bandwidth and low latency by allowing simultaneous connections.
- **Applications:** Used in high-performance computing and data centers for connecting multiple processors and memory modules.

## CHAPTER 5 Design of Arithmetic Building Blocks

Mr.S.Govindaswamy

Designing arithmetic building blocks is fundamental in digital circuit design, particularly in creating efficient and high-performance processors and digital systems. These building blocks include components like adders, subtractors, multipliers, and dividers, which are essential for performing arithmetic operations in computers and other digital devices.

#### Adders

Adders are used to perform addition operations. There are several types of adders, each with its own design considerations:

- 1.1 Half Adder
  - **Function:** Adds two single-bit binary numbers.
  - Components:
    - **XOR Gate:** Computes the sum (S).
    - **AND Gate:** Computes the carry (C).

#### ABSC

- 0 0 0 0
- 0 1 1 0
- 1010
- 1 1 0 1

#### **Full Adder**

- Function: Adds three bits (two significant bits and a carry-in bit) to produce a sum and a carry-out.
- Components:
  - Two XOR gates, two AND gates, and one OR gate.

#### A B C\_in S C\_out

00000

- 00110
- 01010
- 01101
- 01101
- 10010
- 10101
- $1\ 1\ 0\ 0\ 1$
- 11111

#### **1.3 Ripple Carry Adder**

- **Function:** A chain of full adders used to add multi-bit binary numbers.
- **Design:** Each full adder's carry-out is connected to the next full adder's carry-in.
- Pros and Cons:
  - **Pros:** Simple to design and implement.
  - **Cons:** Can be slow due to carry propagation delay.

#### 1.4 Carry-Lookahead Adder

- **Function:** Improves performance by reducing carry propagation delay.
- **Design:** Uses generate (G) and propagate (P) signals to compute carry-outs more quickly.

## CHAPTER 6 Contacts and Transistors

Mrs. R. Prasannadevi

**Contacts** and **transistors** are fundamental components in electronics, each serving distinct purposes in electronic circuits and systems. Here's an overview of their functions, characteristics, and applications:

#### Contacts

**Contacts** in electronics refer to the physical connection points in a circuit where electrical current can flow between different components or parts of a circuit. They are crucial for establishing reliable electrical connections.

#### **Types of Contacts**

#### 1. Mechanical Contacts:

- **Switch Contacts**: Found in various types of switches (e.g., push-button switches, toggle switches), these contacts open or close the circuit to control the flow of current.
- **Relay Contacts**: Used in electromagnetic relays to switch larger currents on and off based on a small control signal.

#### 2. Electrical Contacts:

- **Connector Contacts**: Located in connectors that join two or more electrical circuits or devices. Examples include USB connectors, HDMI connectors, and various plug-and-socket combinations.
- **PCB Contacts**: Found on printed circuit boards (PCBs) as pads or pins that connect electronic components to the board and to each other.3

#### Characteristics

- **Conductivity**: Contacts must have good electrical conductivity to ensure minimal resistance and efficient current flow.
- **Durability**: Contacts should be durable to withstand mechanical wear and environmental factors such as corrosion.
- **Contact Pressure**: Proper contact pressure is necessary to ensure a reliable connection and prevent intermittent contact issues.

#### Transistors

**Transistors** are semiconductor devices used to amplify or switch electronic signals and electrical power. They are essential building blocks in modern electronics and are available in various types, including Bipolar Junction Transistors (BJTs) and Field-Effect Transistors (FETs).

#### **Types of Transistors**

#### **1**. **Bipolar Junction Transistors (BJTs)**:

- **NPN Transistors**: Consist of a layer of P-type material sandwiched between two N-type materials. The majority of current carriers are electrons.
- **PNP Transistors**: Consist of a layer of N-type material sandwiched between two P-type materials. The majority of current carriers are holes.

#### **BJT Structure**:

- **Emitter (E)**: The terminal through which current flows into the transistor.
- **Base (B)**: The control terminal that modulates the current flow between the emitter and collector.
- Collector (C): The terminal through which current flows out of the transistor.

#### **Operation**:

• Active Region: Used for amplification. The base-emitter junction is forward-biased, and the collector-base junction is reverse-biased.

#### **Pass transistor**

#### Mrs. R. Prasannadevi

A **pass transistor** is a type of transistor used in electronic circuits to control the flow of electrical signals or power. Unlike traditional transistors used primarily for amplification or switching, a pass transistor operates in its linear region to allow or block current with minimal distortion.

#### **Key Characteristics**

- 1. **Linear Operation**: Pass transistors are used to control the flow of current in a more linear fashion compared to switching transistors. This means they operate partially on and off, allowing a controlled amount of current to pass through.
- 2. Low Dropout Voltage: In many applications, pass transistors are designed to have a low dropout voltage, which is the difference between the input and output voltage when the transistor is in its conducting state. This is crucial for efficient voltage regulation.
- 3. **Control Function**: They are often used in applications where the control of current or voltage with minimal loss is required, rather than merely switching current on or off.

#### **Types of Pass Transistors**

#### 1. NPN Pass Transistor:

- **Structure**: An NPN transistor with the emitter connected to the output and the collector connected to the input. The base serves as the control terminal.
- **Operation**: When the base-emitter junction is forward-biased, current flows from the collector to the emitter. The transistor can control the output voltage by varying the base voltage.

#### 2. PNP Pass Transistor:

- **Structure**: A PNP transistor with the emitter connected to the input and the collector connected to the output. The base is the control terminal.
- **Operation**: When the base-emitter junction is forward-biased, current flows from the emitter to the collector. The transistor can regulate the output voltage by adjusting the base voltage.

#### Applications

- 1. Voltage Regulation:
  - Linear Voltage Regulators: Pass transistors are used in linear voltage regulators to maintain a stable output voltage despite variations in input voltage and load conditions. They help to regulate the voltage with minimal power loss.
  - Series Regulators: In series regulators, a pass transistor is connected in series with the load. The transistor adjusts its resistance to maintain the desired output voltage.

**Example**: A classic linear regulator circuit might use an NPN transistor to drop excess voltage from a higher input voltage to provide a stable lower output voltage.

#### 2. Analog Switches:

• **Signal Switching**: Pass transistors can act as analog switches to route signals in various analog and mixed-signal applications. They are used in situations where low distortion and high linearity are required.

**Example**: An analog multiplexer might use pass transistors to switch between different input signals.

#### 3. Level Shifters:

• Voltage Translation: Pass transistors are used in level shifter circuits to translate signals from one voltage level to another, particularly in mixed-voltage systems.

**Example**: A circuit that interfaces between high-voltage and low-voltage logic levels might use pass transistors to shift the voltage levels appropriately.

## CHAPTER 8 VLSI Design Flow

Mr. J. Vivek Raja

**VLSI (Very Large Scale Integration) Design Flow** refers to the comprehensive process of designing integrated circuits (ICs) that contain a large number of transistors and other components. The design flow encompasses a series of steps and stages from initial concept through to the final fabrication and testing of the IC. Here's a detailed look at the VLSI design flow:

#### 1. Specification and Requirement Analysis

**Objective**: Define the goals and requirements for the IC, including functionality, performance, and constraints.

- **Functional Specification**: Describes what the IC is supposed to do, including the desired features and operations.
- **Performance Requirements**: Specifies speed, power consumption, area constraints, and other performance metrics.
- **Design Constraints**: Includes physical size, cost, and manufacturing constraints.

#### 2. Architecture Design

**Objective**: Develop the high-level architecture of the IC, outlining its major components and their interactions.

- **System Architecture**: Defines the overall structure, including how various blocks interact (e.g., processor, memory, I/O).
- **Block Diagram**: Provides a high-level view of the system, showing the major functional blocks and their connections.

#### 3. RTL (Register Transfer Level) Design

**Objective**: Create a detailed description of the digital logic in a high-level hardware description language (HDL).

- HDL Coding: Write the design in a hardware description language such as VHDL or Verilog.
- Simulation: Verify the RTL code through simulation to ensure that it behaves as expected.

#### 4. Logic Synthesis

**Objective**: Convert the RTL description into a gate-level netlist, which describes the circuit in terms of logic gates and flip-flops.

- Synthesis: Use synthesis tools to translate the HDL code into a gate-level representation.
- **Optimization**: Optimize the design for area, speed, and power.

#### 5. Design Verification

**Objective**: Ensure that the design meets the specifications and functions correctly.

- **Simulation**: Run various test cases and simulations to verify the functionality of the synthesized design.
- Formal Verification: Use mathematical methods to prove that the design meets its specifications.
- Timing Analysis: Verify that the design meets timing constraints (e.g., setup and hold times).

#### 6. Physical Design

**Objective**: Translate the gate-level netlist into a physical layout that will be used for manufacturing the IC.

- **Floorplanning**: Arrange the major functional blocks on the chip.
- **Placement**: Place the standard cells or logic gates in their respective locations.
- **Routing**: Connect the placed cells using metal layers to form the final interconnections.
- **Design Rule Checking (DRC)**: Ensure that the layout adheres to the manufacturing rules and constraints.

#### CHAPTER 9 Inverter Delays Mr.R.Elangovan

#### **1. Introduction to Inverter Delays**

An **inverter** is a basic digital logic gate that outputs the opposite logic level to its input. It plays a fundamental role in logic circuits and is used in various configurations, such as buffers and amplifiers. The delay associated with an inverter is the time it takes for a change at the input to propagate to the output.

#### 2. Types of Inverter Delays

Inverter delays can be categorized into different types:

#### a. Propagation Delay (tp)

- **Definition**: The time it takes for an input signal to propagate through the inverter and produce a stable output signal. It is typically measured as the time from when the input reaches 50% of its final value to when the output reaches 50% of its final value.
- Components:
  - **Rise Time (tpHL)**: The time for the output to transition from low to high.
  - Fall Time (tpLH): The time for the output to transition from high to low.
- b. Rise Time (tr)
  - **Definition**: The time required for the output of the inverter to rise from 10% to 90% of its final high value when the input signal switches from low to high.
- c. Fall Time (tf)
  - **Definition**: The time required for the output of the inverter to fall from 90% to 10% of its final low value when the input signal switches from high to low.

#### **3. Factors Affecting Inverter Delays**

Several factors influence the delay characteristics of an inverter:

#### a. Load Capacitance (CL)

- **Description**: The capacitance of the load connected to the inverter's output. A higher load capacitance increases the delay because it takes more time to charge or discharge the capacitance.
- Effect: The delay increases with larger load capacitance.

#### **b.** Transistor Sizing

- **Description**: The size of the transistors used in the inverter. Larger transistors have lower resistance and can drive more current, reducing delay.
- **Effect**: Increasing the size of the transistors reduces the delay but increases the area and power consumption.

#### c. Supply Voltage (VDD)

- **Description**: The voltage supplied to the inverter. Higher supply voltages generally increase the speed of the inverter by providing more drive current.
- Effect: Increasing VDD reduces the delay, but this may also increase power consumption and heat dissipation.

#### d. Threshold Voltage (Vth)

- **Description**: The voltage required to turn on the transistors. Lower threshold voltages result in faster switching but can increase leakage currents.
- Effect: Reducing the threshold voltage can decrease delay but might affect power consumption and leakage.

#### e. Process Variations

- **Description**: Variations in semiconductor fabrication processes can affect the characteristics of the transistors.
- Effect: Variations can lead to differences in delay across different chips or even within the same chip.

#### CHAPTER 10 CMOS Testing

#### Mr.S.Govindaswamy

**CMOS Testing** involves evaluating CMOS (Complementary Metal-Oxide-Semiconductor) circuits to ensure they function correctly, meet design specifications, and are free of defects. Testing is crucial for verifying the performance, reliability, and correctness of CMOS circuits, which are widely used in digital integrated circuits, including microprocessors, memory chips, and various other electronic devices.

#### **1. Types of CMOS Testing**

#### a. Functional Testing

Objective: Verify that the CMOS circuit performs its intended function.

- **Description**: Functional testing involves applying known inputs to the CMOS circuit and checking whether the outputs match the expected results. This is typically done using test vectors and comparing the actual output to the expected output.
  - Tools: Logic analyzers, oscilloscopes, and functional test equipment.

#### **b.** Parametric Testing

**Objective**: Measure and verify the electrical characteristics of the CMOS devices.

- **Description**: This testing evaluates parameters such as threshold voltage, drive strength, leakage current, and propagation delay to ensure they meet specified values.
  - **Tools**: Parameter analyzers, semiconductor testers.

#### c. Structural Testing

**Objective**: Test the internal structure of the CMOS circuit to detect manufacturing defects.

- **Description**: This involves testing the individual components and interconnections within the CMOS chip. Common techniques include:
  - Stuck-at Fault Testing: Detects if a signal line is stuck at a logical high or low level.
  - Bridging Fault Testing: Identifies unintended short circuits between different lines.
  - **Open Circuit Testing**: Detects broken connections or open circuits within the chip.
  - Tools: Automatic Test Equipment (ATE), functional and structural test tools.

#### d. Timing Testing

**Objective**: Ensure that the CMOS circuit operates correctly within the timing constraints.

- **Description**: Verifies that the circuit meets timing requirements such as setup time, hold time, and propagation delay. This is crucial for high-speed circuits where timing violations can lead to malfunction.
- **Tools**: Timing analyzers, simulation tools.

#### e. Stress Testing

**Objective**: Evaluate the CMOS circuit's robustness under extreme conditions.

- **Description**: Tests the circuit under conditions such as high temperature, high voltage, or excessive frequency to ensure it can withstand environmental and operational stress.
- Tools: Environmental chambers, stress testers.

#### 2. CMOS Testing Methodologies

#### a. Design for Testability (DFT)

**Objective**: Incorporate testability features into the CMOS design to facilitate easier and more efficient testing.

- **Description**: Techniques include:
  - Scan Chains: Serially connect flip-flops to create a shift-register chain that simplifies testing of sequential logic.
  - **Built-In Self-Test (BIST)**: Integrate test circuitry within the CMOS chip that can generate test patterns and analyze responses autonomously.
  - **Test Access Mechanisms**: Add interfaces to allow for easier access to internal nodes and states.

# FUNDAMENTALS OF ELECTRONIC DEVICES AND CIRCUITS

EDITED BY: DR. S. EZHILARASAN



Fundamentals of Electronic devices and circuits Copyright © 2021 by Dr. S. Ezhilarasan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-587-8

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST) Date of Publication : 21.07.2021

## TABLE OF CONTENTS

Introduction	05
CHAPTER 1 - Semiconductor devices	13
Mr.R.Elangovan	
CHAPTER 2 - Amplifiers	48
Mr.S.Govindaswamy	
<b>CHAPTER 3</b> - Multistage amplifiers and differential amplifier	78
Mr.K.Suresh	
<b>CHAPTER 4 -</b> Feedback amplifiers and oscillators	145
Mr.D.Hariharan	
<b>CHAPTER 5</b> – Power amplifiers and DC/DC converters	153
Mr.B.Arunpandiyan	
CHAPTER 6 – ,Half Wave Rectifier	164
Mr.R.Elangovan	
CHAPTER 7 – Full Wave Rectifier	171
Mr.S.Govindaswamy	
CHAPTER 8 – Zener Diode Characteristics	186
Mr.K.Suresh	
CHAPTER 9 – Inductor Filter	192
Mr.D.Hariharan	
<b>CHAPTER 10</b> – SCR and photo diode	209
Mr.B.Arunpandiyan	
REFERENCES	216

## CHAPTER 1 Semiconductor devices

Mr.R.Elangovan

Semiconductor devices are critical components in modern electronics, enabling the functionality of a wide range of devices from computers to mobile phones. They operate based on the electronic properties of semiconductor materials, such as silicon and gallium arsenide. Here's a comprehensive overview of semiconductor devices:

#### **1. Introduction to Semiconductors**

Semiconductors are materials whose electrical properties lie between those of conductors and insulators. They have a unique ability to control electrical current, which is the basis for their use in electronic devices.

#### **1.1 Semiconductor Materials**

- **Intrinsic Semiconductors:** Pure materials like silicon (Si) and germanium (Ge) that have a natural band gap.
- **Extrinsic Semiconductors:** Doped with impurities to enhance electrical conductivity. They are classified into two types:
  - **n-Type:** Doped with elements that add extra electrons (e.g., phosphorus in silicon).
  - **p-Type:** Doped with elements that create holes (e.g., boron in silicon).

#### 2. Basic Semiconductor Devices

#### **2.1 Diodes**

- **Definition:** A diode is a semiconductor device that allows current to flow in one direction only.
- Types:
  - **PN Junction Diode:** Made by joining p-type and n-type materials. It has a forward-biased region where current flows easily and a reverse-biased region where current flow is minimal.
  - Zener Diode: Designed to operate in reverse bias and maintain a constant voltage.
  - Light Emitting Diode (LED): Emits light when current flows through it.
- Applications: Rectification, signal demodulation, voltage regulation.

#### **2.2 Bipolar Junction Transistors (BJTs)**

- **Definition:** A BJT is a three-layer semiconductor device with two pn junctions that can amplify or switch electronic signals.
- Types:
  - **NPN:** Has an n-type material sandwiched between two p-type materials.
  - **PNP:** Has a p-type material sandwiched between two n-type materials.
- **Operation:** Uses current to control the flow of a larger current.
- **Applications:** Amplification, switching.

#### **2.3 Field Effect Transistors (FETs)**

- **Definition:** FETs are voltage-controlled devices that use an electric field to control the flow of current.
- Types:
  - Junction FET (JFET): Uses a gate junction to control current flow.
  - Metal-Oxide-Semiconductor FET (MOSFET): Uses an insulated gate to control current flow, commonly used in digital circuits.
    - Enhancement-mode MOSFET: Conducts only when a voltage is applied to the gate.
    - **Depletion-mode MOSFET:** Conducts without gate voltage but can be turned off by applying a gate voltage.
- Applications: Amplification, switching, digital logic.

#### Amplifiers

#### Mr.S.Govindaswamy

Amplifiers are fundamental electronic devices used to increase the amplitude of signals. They are widely used in various applications, including audio systems, communication devices, and instrumentation. Here's a comprehensive overview of amplifiers, covering their types, characteristics, and applications:

#### **1. Basic Concepts**

#### 1.1 Definition

An amplifier is an electronic device that takes an input signal and produces a larger output signal. The primary function of an amplifier is to increase the power, voltage, or current of a signal.

#### **1.2 Basic Parameters**

- Gain (A): The ratio of output signal to input signal. It can be expressed as voltage gain, current gain, or power gain:
  - Voltage Gain (Av)
  - Current Gain (Ai)
  - Power Gain (Ap)
- Bandwidth (BW): The range of frequencies over which the amplifier can operate effectively.
- **Input and Output Impedance:** The resistance presented by the amplifier to the input and output signals, respectively.
- Efficiency: The ratio of the output power to the total input power.

#### 2. Types of Amplifiers

#### 2.1 Based on Configuration

- Common Emitter (CE):
  - **Description:** Commonly used in BJT amplifiers where the emitter is the common terminal.
  - Applications: General-purpose amplification, switching.
- Common Collector (CC):
  - **Description:** The collector is common to both input and output. Also known as an emitter follower.
  - Applications: Impedance matching, buffer stages.
- Common Base (CB):
  - **Description:** The base is common to both input and output.
  - Applications: High-frequency applications.
- Common Source (CS):
  - **Description:** Commonly used in FET amplifiers where the source is the common terminal.
  - Applications: General-purpose amplification.
- Common Gate (CG):
  - **Description:** The gate is common to both input and output in FET amplifiers.
  - Applications: High-frequency applications.
- Common Drain (CD):
  - **Description:** Also known as a source follower in FET amplifiers.
  - Applications: Impedance matching, buffer stages.
- 2.2 Based on Operation Class
  - Class A:
    - **Description:** The transistor conducts for the entire input signal cycle.
    - Characteristics: High linearity, high distortion, low efficiency.
    - Applications: Audio amplifiers, high-fidelity applications.
  - Class B:
    - **Description:** The transistor conducts for half of the input signal cycle.
    - Characteristics: Improved efficiency compared to Class A, some distortion.
    - Applications: Audio amplification, RF amplification.

## CHAPTER 3 Multistage amplifiers and differential amplifier

Mr.K.Suresh

**Multistage Amplifiers** and **Differential Amplifiers** are both essential concepts in analog electronics, each serving unique purposes in signal amplification and processing. Here's a detailed look at each:

#### **Multistage Amplifiers**

#### 1. Overview

Multistage amplifiers are configurations where multiple amplifier stages are cascaded to achieve higher overall gain, better performance, or specific characteristics not achievable with a single stage.

#### 2. Reasons for Using Multistage Amplifiers

- **Increased Gain:** Each stage contributes to the overall gain. The total gain is the product of the gains of each individual stage.
- **Impedance Matching:** Different stages can be designed to match impedances and drive subsequent stages effectively.
- **Improved Performance:** Cascading stages can improve characteristics such as bandwidth, linearity, and noise performance.
- Complex Functions: Allows for the implementation of complex amplification functions or filters.

#### 3. Configuration

- **Cascaded Stages:** Stages are connected in series. The output of one stage feeds into the input of the next. For example, a common configuration might include a voltage amplifier stage followed by a power amplifier stage.
- **RC Coupling:** Uses capacitors to block DC while allowing AC signals to pass between stages.
- Transformer Coupling: Uses transformers for impedance matching and coupling.

#### 4. Design Considerations

- **Gain-Bandwidth Product:** The bandwidth of each stage affects the overall frequency response of the multistage amplifier.
- Stability: Ensuring stability across all stages to avoid oscillations or feedback issues.
- Noise: Managing noise from each stage to ensure overall signal integrity.
- **Impedance Matching:** Ensuring that the output impedance of one stage matches the input impedance of the next stage to maximize power transfer.

#### 5. Example Configuration

A typical multistage amplifier might consist of:

- First Stage: Low-noise, high-gain amplifier (e.g., a common-emitter BJT amplifier).
- Intermediate Stage: Impedance matching and additional gain (e.g., a common-collector BJT stage).
- **Final Stage:** High-power output stage (e.g., a push-pull output stage).

#### **Differential Amplifiers**

#### 1. Overview

A differential amplifier is designed to amplify the difference between two input signals while rejecting any common-mode signals (signals that are present on both inputs).

#### 2. Key Features

- **Differential Gain:** Amplifies the difference between the two input signals.
- **Common-Mode Rejection Ratio** (**CMRR**): A measure of the amplifier's ability to reject commonmode signals. High CMRR indicates good rejection of noise and interference that is common to both inputs.

#### 3. Operation

• **Basic Configuration:** Typically uses two transistors (BJTs or FETs) arranged to provide differential gain. The outputs of the two transistors are combined to amplify the difference between the input signals.

## CHAPTER 4 Feedback amplifiers and oscillators

Mr.D.Hariharan

**Feedback Amplifiers** and **Oscillators** are fundamental concepts in electronic circuit design. They each play a crucial role in a wide range of applications, from signal amplification to generating waveforms. Here's a detailed overview:

#### 1. Overview

#### **Feedback Amplifiers**

Feedback amplifiers use a portion of the output signal to influence the input signal. This feedback can be positive (regenerative) or negative (degenerative), and it significantly impacts the amplifier's performance.

#### 2. Types of Feedback

- **Negative Feedback:** A portion of the output is fed back to the input in such a way that it opposes the input signal. This stabilizes the gain, improves bandwidth, and reduces distortion.
- **Positive Feedback:** A portion of the output is fed back to the input to reinforce the input signal, which can lead to instability or oscillation.

#### 3. Feedback Amplifier Configuration

- Series Feedback (Voltage Feedback): Feedback is applied in series with the input signal. This configuration is used to control the gain of the amplifier.
- **Shunt Feedback (Current Feedback):** Feedback is applied in parallel with the input signal. This configuration affects the input impedance and stability of the amplifier.

#### 4. Applications and Benefits

- **Stability:** Negative feedback improves the stability of the amplifier and makes its gain less dependent on variations in component values.
- Linearity: Reduces distortion by making the amplifier's response more linear.
- **Bandwidth:** Increases the bandwidth of the amplifier by reducing the gain at high frequencies.

#### 5. Example: Operational Amplifiers

- **Inverting Amplifier:** Uses negative feedback to stabilize the gain, which is determined by the ratio of two resistors.
- **Non-Inverting Amplifier:** Also uses negative feedback to stabilize gain, but the input signal is applied directly to the non-inverting terminal.

#### Oscillators

#### 1. Overview

Oscillators are circuits that generate a periodic waveform (sine wave, square wave, triangle wave, etc.) without requiring an external input signal. They convert DC power into AC signals.

#### 2. Basic Operation

An oscillator typically consists of three main components:

- Amplification Stage: Provides gain to the signal.
- Feedback Network: Returns a portion of the output signal to the input to sustain oscillations.
- **Frequency Determining Network:** Sets the frequency of oscillation (usually a combination of capacitors, inductors, and resistors).

#### 3. Types of Oscillators

•

- LC Oscillators: Use inductors (L) and capacitors (C) to determine the frequency.
  - Colpitts Oscillator: Uses a combination of capacitors and an inductor.
  - Hartley Oscillator: Uses a tapped inductor and a capacitor.
- **RC Oscillators:** Use resistors (R) and capacitors (C) to set the frequency.
  - Wien Bridge Oscillator: Uses a bridge circuit to achieve stable oscillations.
    - Phase Shift Oscillator: Uses RC networks to create the necessary phase shift.
- **Crystal Oscillators:** Use a quartz crystal as the frequency-determining element. Crystals provide highly stable and precise frequencies.

#### CHAPTER 5 Power amplifiers and DC/DC converters

#### Mr.B.Arunpandiyan

**Power Amplifiers** and **DC/DC Converters** are essential components in electronic systems, each serving a distinct purpose in managing and converting electrical power. Here's a comprehensive overview of both: **Power Amplifiers** 

#### 1. Overview

Power amplifiers are designed to increase the power level of a signal to drive loads such as speakers or antennas. They are crucial in applications where high power output is required.

## 2. Types of Power Amplifiers

- Class A Amplifiers:
  - **Operation:** The transistor conducts for the entire input signal cycle, providing high linearity.
  - **Characteristics:** High linearity, high distortion, low efficiency (typically 20-30%).
  - Applications: High-fidelity audio amplifiers, certain radio frequency (RF) applications.

#### • Class B Amplifiers:

- **Operation:** The transistor conducts for half of the input signal cycle. Two transistors are used, each amplifying half of the signal.
- **Characteristics:** Improved efficiency (about 50-70%), moderate distortion.
- Applications: Audio amplifiers, RF amplification.
- Class AB Amplifiers:
  - **Operation:** A combination of Class A and Class B, where the transistor conducts for more than half but less than the full input cycle.
  - **Characteristics:** Balanced efficiency (about 50-70%) and linearity, reduced distortion.
  - **Applications:** Audio amplification, general-purpose power amplification.

#### • Class C Amplifiers:

- **Operation:** The transistor conducts for less than half of the input signal cycle, used for high efficiency.
- **Characteristics:** Very high efficiency (up to 80-90%), significant distortion.
- **Applications:** RF transmission, high-efficiency amplifiers.

#### • Class D Amplifiers:

- **Operation:** Uses pulse-width modulation (PWM) to convert the input signal into a series of pulses that are amplified and then filtered to produce the output signal.
- **Characteristics:** High efficiency (up to 90% or more), low heat generation.
- Applications: High-power audio amplifiers, switching power supplies.

#### 3. Design Considerations

- **Power Rating:** Ensure the amplifier can handle the required power levels without distortion or damage.
- Thermal Management: Adequate cooling and heat dissipation to prevent overheating.
- Efficiency: Consider the trade-off between efficiency and linearity based on the application.
- Load Impedance: Match the amplifier's output impedance with the load to ensure optimal performance.

#### 4. Applications

- Audio Systems: Amplifying audio signals to drive speakers.
- **RF Transmission:** Amplifying signals for radio and television broadcasting.

• Instrumentation: Providing the necessary power for measurement and test equipment.

#### **DC/DC Converters**

#### 1. Overview

DC/DC converters are electronic circuits that convert a source of direct current (DC) from one voltage level to another. They are essential in managing and distributing electrical power in various applications.

#### **CHAPTER 6 Half Wave Rectifier** Mr.R.Elangovan

A **half-wave rectifier** is a basic type of rectifier circuit used in electronic devices to convert alternating current (AC) into direct current (DC). It allows only one half of the AC waveform to pass through, effectively blocking the other half. Here's a detailed overview of how a half-wave rectifier works, its components, and its characteristics:

#### **Basic Operation**

A half-wave rectifier uses a diode to allow only one half of the AC signal to pass through. The diode conducts during one half-cycle of the AC signal and blocks during the other half-cycle.

- **Positive Half-Cycle**: When the AC input is positive with respect to the diode's anode, the diode becomes forward-biased and conducts. This allows current to flow through the load resistor RLR\_LRL, and the output voltage VoutV\_{out}Vout follows the input AC voltage.
- **Negative Half-Cycle**: When the AC input is negative with respect to the diode's anode, the diode becomes reverse-biased and does not conduct. During this time, the output voltage is zero, as the diode blocks the current.

#### **Circuit Diagram**

A typical half-wave rectifier circuit consists of:

- AC Source: Provides the alternating current input.
- **Diode**: Allows current to pass only during the positive half-cycle.
- Load Resistor (R\_L): The resistor across which the DC voltage is measured.

#### **Output Waveform**

The output of a half-wave rectifier is a pulsating DC waveform. It consists of only the positive half-cycles of the AC input signal:

- **Positive Half-Cycle**: The output voltage is similar to the input voltage (minus the diode's forward voltage drop).
- Negative Half-Cycle: The output voltage is zero.

#### **Key Parameters**

- **Peak Output Voltage** (VpeakV\_{peak}): The maximum voltage of the output waveform, which is approximately equal to the peak voltage of the AC input minus the diode's forward voltage drop.
- **Peak Inverse Voltage (PIV)**: The maximum reverse voltage that the diode must withstand, which is equal to the peak voltage of the AC input.
- **Ripple Voltage**: The variation in the DC output voltage due to the pulsating nature of the output. The ripple frequency is the same as the input frequency.

#### Applications

- **Power Supply**: Used in low-power applications where a simple rectifier is sufficient.
- **Signal Detection**: Can be used for detecting amplitude variations in signals.

A half-wave rectifier is a basic circuit used to convert AC to DC by allowing only one half of the AC waveform to pass through. It consists of a diode and a load resistor, and its output is a pulsating DC voltage with significant ripple. While simple and inexpensive, it is less efficient and has a higher ripple compared to other rectification methods, such as full-wave or bridge rectifiers.

## CHAPTER 7 Full Wave Rectifier

Mr.S.Gov in daswamy

A **full-wave rectifier** is a circuit used to convert alternating current (AC) into direct current (DC) by utilizing both halves of the AC waveform. This results in a smoother DC output compared to a half-wave rectifier. There are two main types of full-wave rectifiers: the **center-tap** and the **bridge** rectifier. Here's a detailed explanation of both types:

#### **Center-Tap Full-Wave Rectifier**

#### **Circuit Diagram**

A center-tap full-wave rectifier uses a center-tap transformer and two diodes:

#### Operation

- **Positive Half-Cycle**: During the positive half-cycle of the AC input, diode D1 conducts, and diode D2 is reverse-biased. Current flows through D1, the load resistor RLR\_LRL, and back to the transformer center tap.
- **Negative Half-Cycle**: During the negative half-cycle of the AC input, diode D2 conducts, and diode D1 is reverse-biased. Current flows through D2, the load resistor RLR\_LRL, and back to the transformer center tap.

Each diode conducts during one half of the input cycle, ensuring that both halves of the AC waveform contribute to the DC output.



## CHAPTER 8 Zener Diode Characteristics

Mr.K.Suresh

A **Zener diode** is a special type of diode designed to operate in the reverse breakdown region, where it maintains a stable and well-defined voltage. It is widely used in voltage regulation and protection circuits. Here's an in-depth look at the characteristics, operation, and applications of Zener diodes:

#### **1. Basic Characteristics**

- Zener Breakdown: Unlike regular diodes, which are designed to handle forward current, Zener diodes are designed to operate in the reverse breakdown region. This region is known as the Zener breakdown region, and it occurs when the reverse voltage exceeds a certain value called the Zener voltage or breakdown voltage.
- Zener Voltage (Vz): The Zener voltage is the voltage at which the diode starts to conduct in reverse and maintain a constant voltage despite changes in the current through it. This voltage is specified for each Zener diode and can range from a few volts to several hundred volts.
- **Reverse Breakdown Region**: In the reverse direction, Zener diodes exhibit a controlled breakdown, where the voltage remains nearly constant over a range of reverse currents. This characteristic is used to stabilize voltages in electronic circuits.
- •

#### 2. Zener Diode Characteristics

- **Forward Characteristics**: In the forward bias region, a Zener diode behaves like a regular diode, with a forward voltage drop of approximately 0.7V (for silicon diodes) when conducting.
- **Reverse Characteristics**: When reverse-biased, the Zener diode initially exhibits high resistance. As the reverse voltage increases and reaches the Zener voltage, the diode begins to conduct, and the voltage across the diode stabilizes at the Zener voltage. This characteristic is utilized for voltage regulation.
- Zener Breakdown Region: In the reverse bias, once the Zener voltage is reached, the voltage across the diode remains almost constant despite significant changes in reverse current. This is crucial for applications requiring stable reference voltages.
- **Dynamic Resistance**: The Zener diode's dynamic resistance (RzR\_zRz) in the breakdown region is small, and the diode can conduct significant current with only a small change in voltage.
- •

#### **3. Operating Regions**

- Forward Bias Region: In forward bias, the Zener diode behaves like a normal silicon diode, with a forward voltage drop typically around 0.7V.
- **Reverse Bias Region**: In reverse bias, the diode exhibits:
  - **Reverse Leakage Current**: Very small current that flows before reaching the Zener voltage.
  - Zener Breakdown: Once the reverse voltage exceeds the Zener voltage, the diode conducts and maintains a constant voltage. The diode's ability to maintain this voltage is due to the Zener effect or avalanche breakdown, depending on the Zener voltage.



## CHAPTER 9 Inductor Filter

#### Mr.D.Hariharan

An **inductor filter** is a type of electronic filter that uses inductors to smooth out or filter electrical signals. It is commonly used in power supply circuits to reduce ripple and provide a more stable DC output. The filter works by using the properties of inductors to impede changes in current and filter out AC components from a DC signal.

#### **1. Basic Principle of Operation**

Inductors oppose changes in current through them due to their inherent property called **inductance**.

#### 2. Types of Inductor Filters

Inductor filters are commonly used in two main configurations:

#### a. Series Inductor Filter

In a series inductor filter, the inductor is placed in series with the load. This configuration is typically used to smooth the output of a rectifier circuit.

#### **Operation**:

- The inductor in series with the load smooths out the pulsating DC from a rectifier. It resists changes in current, filtering out high-frequency ripple and providing a more stable DC output.
- The effectiveness of the filtering depends on the value of the inductor L and the load resistance RL

#### **b.** LC Filter (Inductor-Capacitor Filter)

An LC filter uses both an inductor and a capacitor to filter out AC components and provide a smoother DC output. The combination of an inductor and capacitor creates a more effective filter than using an inductor alone.

#### **Operation**:

- The **inductor** filters out high-frequency noise by opposing rapid changes in current.
- The **capacitor** filters out low-frequency noise and smooths the voltage by providing a low impedance path for AC signals.
- Together, the LC filter provides better attenuation of AC ripple and results in a more stable DC output.

#### **3. Filter Characteristics**

- **Inductance Value**: The effectiveness of an inductor filter increases with the inductance value LLL. A higher inductance means the filter will better smooth out variations in current.
- **Ripple Reduction**: The primary role of an inductor filter is to reduce ripple in the output voltage.

#### 4. Applications

- **Power Supplies**: Inductor filters are widely used in power supplies to smooth the rectified output from AC sources, reducing ripple and providing a stable DC voltage.
- **Signal Processing**: Inductor filters are used in radio and audio applications to remove unwanted high-frequency noise or to shape the frequency response of signals.
- Noise Filtering: In digital circuits, inductor filters help in reducing electromagnetic interference (EMI) and noise from power supplies.

## CHAPTER 10 SCR and photo diode

#### Mr.B.Arunpandiyan

#### **Definition and Basic Operation**

An **SCR** is a type of semiconductor device that functions as a controlled rectifier. It can act as a switch or amplifier and is used to control large amounts of power. It has four layers (PNPN) and three junctions, and it can handle high voltages and currents.

- **Structure**: An SCR consists of four layers of semiconductor material arranged in a PNPN configuration. It has three terminals:
  - Anode (A): The positive terminal.
  - Cathode (K): The negative terminal.
  - Gate (G): The control terminal.
- Operation:
  - Forward Blocking State: When the SCR is forward biased (anode positive with respect to the cathode) but no gate current is applied, the SCR remains off and blocks current flow.
  - **Forward Conducting State**: When a small current is applied to the gate while the SCR is forward biased, the SCR turns on and allows current to flow from the anode to the cathode.
  - **Reverse Blocking State**: When the SCR is reverse biased (anode negative with respect to the cathode), it blocks current flow and remains off.
  - Latching: Once the SCR is turned on by the gate current, it remains on even if the gate current is removed, until the current flowing through it drops below a certain threshold called the latching current.

#### Characteristics

- **Gate Triggering**: SCRs require a gate signal to turn on, making them useful for controlled switching applications.
- Latching: SCRs stay on after being triggered, which makes them suitable for applications where the device needs to remain conducting without continuous control.
- **High Voltage and Current Handling**: SCRs are capable of handling high voltages and currents, which makes them suitable for power control applications.
- **Speed**: SCRs are relatively slower compared to modern transistors and MOSFETs, so they are not ideal for high-speed switching applications.

#### Applications

- **Power Control**: Used in light dimmers, motor speed controls, and heating element controllers.
- **Phase Control**: Employed in AC phase control applications to adjust power delivered to loads.
- **Overvoltage Protection**: Used in circuits for surge protection and voltage clamping.

#### Photodiode

#### **Definition and Basic Operation**

A **photodiode** is a semiconductor device that converts light into an electrical current. It is used in light detection and optical communication systems. It operates based on the photoelectric effect, where incident light generates electron-hole pairs, producing a current.

- **Structure**: A photodiode consists of a PN junction. When light strikes the junction, it creates electron-hole pairs, which generate a current proportional to the intensity of the light.
- Operation:
  - **Photovoltaic Mode**: The photodiode generates a voltage when exposed to light without an external power supply.

## WASTE WATER ENGINEERING

EDITED BY

## **P.VENKATESWARAN**



#### WASTE WATER ENGINEERING

Copyright © 2021 by **P.Venkateswaran** 

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-744-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## **TABLE OF CONTENTS**

CHAPTER 1- PLANNING AND DESIGN PF SEWERAGE SYSTEM	-06
J Santhiyaa Jenifer	
CHAPTER 2- PRIMARY TREATMENT OF SEWAGE	-10
R Devi	
CHAPTER -3- SECONDARY TREATMENT OF SEWAGE	-30
P Venkateswaran	
CHAPTER- 4- DISPOSAL OF SEWAG	-40
J.Santhiyaa Jenifer	
CHAPTER 5-SLUDGE TREATMENT AND DISPOSAL	-75
K Shanthi	
CHAPTER 6-SOURCES AND CONVEYANCE OF WATER	-93
D.Jeyakumar	
CHAPTER 7-COMMERCIAL & INDUSTRIAL WATER DEMAND	-110
K Shanthi	
CHAPTER 8-QUANTITY AND QUALITY OF SEWAGE DISPOSAL	-130
J.Santhiyaa Jenifer	
CHAPTER 9-SEWER APPURTENANCES AND SEWAGE DISPOSAL.	-145
S.Vennila	
CHAPTER 10- SEWAGE TREATMENT	-150
Dr.P.Paramaguru	
REFERENCES	180

#### Planning and Design Pf Sewerage System

#### J Santhiyaa Jenifer

Designing and planning a sewerage system involves several key steps to ensure the system is effective, efficient, and environmentally sound. Here's a high-level overview of the process:

#### 1. Assessment of Needs and Goals

- **Identify Objectives**: Understand the purpose of the sewerage system, such as improving public health, protecting the environment, or supporting urban development.
- Assess Current Infrastructure: Evaluate existing sewerage systems and determine their capacity and condition.
- **Population and Growth Projections**: Estimate current and future population growth to design a system that can accommodate future demands.

#### 2. Data Collection

- Topographic Survey: Gather data on the terrain, elevation, and existing infrastructure.
- Soil Analysis: Conduct soil tests to determine soil characteristics, which affect the choice of materials and methods.
- **Hydrological Data**: Assess rainfall, groundwater levels, and other hydrological factors that influence system design.

#### 3. System Design

- **Type of System**: Choose between combined, separate, or semi-separate systems based on the needs and local conditions.
  - **Combined System**: Carries both sewage and stormwater in the same pipes.
  - Separate System: Carries sewage and stormwater in different pipes.
  - Semi-Separate System: Carries stormwater and sewage separately, with some combined elements.
- **Piping Layout**: Design the network of pipes, including sizes, materials, and routing.
- Pump Stations: Plan for the inclusion of pump stations if gravity flow is insufficient.
- **Treatment Facilities**: Design facilities for treating sewage before discharge or reuse, such as primary, secondary, or tertiary treatment plants.
- Manholes and Access Points: Incorporate manholes for maintenance and inspection.

#### 4. Regulatory Compliance

- Local Regulations: Ensure the design meets local building codes and environmental regulations.
- **Environmental Impact Assessment**: Evaluate the potential impact on the environment and include mitigation measures.

#### CHAPTER 2 Primary Treatment of Sewage R Devi

The primary treatment of sewage is the initial stage in wastewater treatment designed to remove large solids and particulate matter from the sewage before it undergoes further treatment. Here's an overview of how it works:

- 1. **Screening**: The sewage first passes through screens that remove large objects such as sticks, leaves, plastic, and other debris. This step helps prevent damage to equipment and ensures smoother operation of the subsequent treatment processes.
- 2. **Grit Removal**: The screened sewage then flows to a grit chamber where heavier particles like sand, gravel, and other inorganic materials settle out due to gravity. This prevents these materials from causing abrasion or clogging in the rest of the treatment system.
- 3. **Sedimentation**: The primary sedimentation tank (or primary clarifier) is the next step, where the sewage is allowed to sit in large tanks. In these tanks, solid particles (such as organic matter) settle to the bottom as sludge, while the lighter materials (like grease and oil) float to the top and are skimmed off. The clarified water that remains in the middle is then sent on for secondary treatment.

The primary treatment primarily focuses on removing physical and large organic materials, and it typically removes about 50-70% of suspended solids and 30-40% of biochemical oxygen demand (BOD). While effective in reducing the load on secondary treatment processes, it doesn't address dissolved substances or pathogens, which is why further treatment stages are necessary for more comprehensive treatment.

#### CHAPTER 3 Secondary Treatment of Sewage

P Venkateswaran

Secondary treatment of sewage is a crucial step in the wastewater treatment process. Its main goal is to significantly reduce the organic matter and pollutants remaining after the primary treatment stage. Here's a breakdown of how it works and why it's important:

#### 1. Overview of Secondary Treatment:

Secondary treatment focuses on further breaking down organic matter that was not removed during primary treatment, which primarily involves physical processes like screening and sedimentation.

#### 2. Methods of Secondary Treatment:

#### • Activated Sludge Process:

- Aeration Tanks: Sewage is mixed with air (or oxygen) in aeration tanks. This promotes the growth of microorganisms that consume organic pollutants.
- **Clarification:** After aeration, the mixture moves to a secondary clarifier where the microorganisms (activated sludge) are allowed to settle out. The treated water is then decanted, while the settled sludge can be recycled back into the aeration tank or treated further.

#### • Trickling Filters:

- **Biofilm Growth:** Sewage is spread over a bed of microbial-covered media (e.g., plastic or stone). As the sewage flows over the media, microorganisms break down the organic matter.
- **Effluent Collection:** The treated water then passes through a secondary clarifier for further settling.
- Rotating Biological Contactors (RBCs):
  - **Rotating Discs:** Sewage is passed over rotating discs coated with microorganisms. As the discs turn, the microorganisms break down the organic matter.
  - Secondary Clarification: Similar to trickling filters, the treated effluent goes through a clarifier to remove excess biomass.
- Membrane Bioreactors (MBRs):
  - **Combination System:** MBRs integrate biological treatment with membrane filtration. The biological treatment occurs in a reactor where microorganisms break down pollutants, and then the water is filtered through membranes to remove remaining solids and microorganisms.

#### 3. Importance of Secondary Treatment:

- **Reduction of Organic Pollutants:** It effectively reduces biochemical oxygen demand (BOD) and total suspended solids (TSS), making the water safer for discharge into the environment or for further treatment.
- **Improvement of Water Quality:** It helps meet regulatory standards for water quality, reducing the impact on aquatic ecosystems and human health.

#### CHAPTER 4 Disposal of Sewage

#### J.Santhiyaa Jenifer

Disposal of sewage is a critical aspect of public health and environmental protection. Proper management ensures that waste does not contaminate water supplies, soil, or air. Here's an overview of the main methods for sewage disposal:

#### 1. Sewer Systems

- **Municipal Sewers**: In urban areas, sewage is typically collected through a network of pipes and transported to a wastewater treatment plant. There, it undergoes several treatment stages to remove contaminants before being released into the environment or reused.
- **Combined Sewer Systems**: These systems collect both sewage and stormwater runoff in a single pipe network. During heavy rains, the system might overflow, leading to potential pollution of water bodies. Modern improvements aim to manage this better and reduce overflows.
- Separate Sewer Systems: These have distinct pipes for sewage and stormwater. This helps prevent the overflow issues seen in combined systems and allows for more efficient treatment of sewage.

#### 2. On-Site Sewage Systems

- **Septic Systems**: Common in rural or less densely populated areas, these systems involve a septic tank where solids settle and undergo bacterial decomposition. The effluent then flows into a drain field where it is further treated by soil filtration. Regular maintenance is essential to prevent failures.
- **Cesspools**: An older method, cesspools are less common today due to environmental concerns. They involve a pit where sewage is allowed to soak into the ground. They are less effective at treating waste and can pose risks of groundwater contamination.

#### **3.** Wastewater Treatment

- **Primary Treatment**: This involves physical processes to remove large solids from sewage. It includes screening and sedimentation.
- **Secondary Treatment**: This biological process uses microorganisms to break down dissolved organic matter. It often involves aeration tanks and secondary clarifiers.
- **Tertiary Treatment**: This advanced treatment further cleans the effluent. It may include filtration, chemical treatment, or additional biological processes to remove remaining contaminants and nutrients.

#### CHAPTER 5 Sludge Treatment and Disposal

#### K Shanthi

Sludge treatment and disposal is an essential part of managing wastewater from various sources, including municipal sewage systems, industrial processes, and agricultural operations. The goal is to handle the sludge in a way that minimizes its environmental impact and maximizes the benefits derived from it. Here's an overview of the main methods used:

#### **1. Primary Treatment**

#### Screening and Sedimentation:

- Screening: Removes large objects and debris from the wastewater.
- Sedimentation: Allows solids to settle out from the wastewater, forming primary sludge.

#### 2. Secondary Treatment

#### **Biological Treatment:**

- Activated Sludge Process: Uses microorganisms to break down organic matter in the wastewater. This produces secondary sludge, which is then separated from treated water.
- **Trickling Filters and Rotating Biological Contactors:** Provide surfaces for microorganisms to grow and treat wastewater.

#### 3. Sludge Treatment

#### Thickening:

- Gravity Thickening: Allows solids to settle out, concentrating the sludge.
- **Dewatering:** Reduces the water content of the sludge, making it easier to handle. Methods include centrifugation, filter pressing, and drying beds.

#### Stabilization:

- Anaerobic Digestion: Uses microorganisms in an oxygen-free environment to break down organic matter in the sludge. This process produces biogas (mainly methane) that can be used as an energy source.
- Aerobic Digestion: Uses oxygen to help microorganisms decompose organic matter, resulting in a stabilized sludge that is less odorous and less pathogen-rich.
- Lime Stabilization: Adds lime to the sludge, increasing pH to kill pathogens and stabilize the material.

#### Sources and Conveyance of Water

#### D.Jeyakumar

Water sources and their conveyance are crucial aspects of water management systems, ensuring that water is available for various uses such as drinking, agriculture, industry, and recreation. Here's a rundown of the main sources and methods of conveying water:

#### **Sources of Water**

- 1. **Surface Water**: This includes rivers, lakes, reservoirs, and streams. Surface water is often the primary source for municipal and agricultural needs.
  - **Rivers**: Flowing bodies of water that may originate from springs, glaciers, or rainfall.
  - Lakes: Large, stationary bodies of water that can store significant amounts of water.
  - **Reservoirs**: Man-made lakes created by damming rivers to store water for future use.
- 2. **Groundwater**: Water located beneath the earth's surface in soil pores and rock formations.
  - Aquifers: Underground layers of water-bearing rock or sediment that can be tapped for water. They can be confined (trapped between layers of impermeable rock) or unconfined (directly connected to the surface).
  - **Springs**: Natural sources of groundwater that flow to the surface.
- 3. **Rainwater**: Precipitation that can be collected and used directly. Rainwater harvesting involves capturing and storing rainwater from roofs and other surfaces.
- 4. **Desalinated Water**: Water that has been processed to remove salts and minerals, making it suitable for consumption and use. This is especially relevant in arid regions where fresh water is scarce.
- 5. **Recycled or Reclaimed Water**: Water that has been treated and purified for reuse. This can come from municipal wastewater, greywater, or industrial processes.

#### **Conveyance of Water**

- 1. **Pipes**: The most common method for transporting water, including:
  - **Distribution Pipes**: Carry treated water from treatment plants to homes, businesses, and industries.
  - **Transmission Pipes**: Larger pipes that transport water over long distances, often from reservoirs to treatment plants.
- 2. **Canals**: Artificial channels used to direct surface water for irrigation or municipal use. Canals are commonly used in agriculture to distribute water across large areas.
- 3. **Aqueducts**: Historical and modern structures designed to transport water from a source to a destination, often over long distances. They can be open channels or enclosed pipelines.
- 4. **Dams and Reservoirs**: Used to store water and control its flow. Dams can also help in flood control, hydroelectric power generation, and recreational activities.
- 5. **Pumps**: Mechanical devices used to move water from one place to another, particularly when water needs to be lifted or moved through pipes over a gradient.

#### **Commercial & Industrial Water Demand**

#### K Shanthi

Commercial and industrial water demand encompasses the water needs of businesses and industries that use water in their operations. This can range from office buildings and shopping malls to factories and power plants. Understanding and managing this demand is crucial for efficient water resource management, cost control, and environmental sustainability.

#### **Key Factors Influencing Commercial and Industrial Water Demand**

#### 1. Type of Industry:

- **Manufacturing**: Industries like textiles, chemicals, and food processing often require large quantities of water for cooling, cleaning, and production processes.
- **Power Generation**: Power plants use water for cooling systems, steam generation, and sometimes for extracting heat from geothermal sources.
- Construction: Water is used in concrete mixing, dust suppression, and site preparation.

#### 2. Operational Scale:

• Larger facilities or those with high production volumes typically have higher water demands compared to smaller businesses.

#### 3. Water Efficiency Measures:

• Companies that implement water-saving technologies and practices (e.g., recycling and reuse systems, low-flow fixtures) may have lower water demands.

#### 4. Regulatory Environment:

• Regulations and incentives for water conservation can influence how much water commercial and industrial entities use.

#### 5. Economic Factors:

• Fluctuations in production levels, changes in business activity, or economic downturns can affect water consumption.

#### 6. Technological Advancements:

• Innovations in water management technologies can help industries reduce their water footprint and improve efficiency.

#### **Common Uses of Water in Commercial and Industrial Settings**

- Cooling: Many industries use water to cool machinery and processes.
- Cleaning: Water is essential for cleaning equipment, products, and facilities.
- **Production Processes**: In manufacturing, water may be a key ingredient or used in processes like rinsing and dilution.
- Landscaping: Commercial properties often use water for maintaining green spaces and landscapes.

#### **Managing Water Demand**

- Monitoring and Reporting: Regularly tracking water usage helps identify trends, inefficiencies, and areas for improvement.
- Water Audits: Conducting audits to evaluate water usage and identify opportunities for reduction.

#### **Quantity and Quality of Sewage Disposal**

J.Santhiyaa Jenifer

When discussing sewage disposal, it's essential to address both the quantity and quality of sewage to ensure effective and environmentally responsible management. Here's a breakdown of these two aspects:

#### **Quantity of Sewage**

- 1. **Measurement and Monitoring**: The quantity of sewage is usually measured in terms of flow rate, often expressed in liters per second (L/s) or cubic meters per day (m<sup>3</sup>/day). Monitoring is crucial for designing and operating sewage treatment systems effectively.
- 2. Factors Influencing Quantity:
  - **Population Density**: Higher population densities typically result in greater sewage volumes.
  - **Industrial and Commercial Activities**: Facilities that use large amounts of water will contribute more sewage.
  - Weather Conditions: Rainfall can increase sewage volume through combined sewer systems, where stormwater and sewage are collected together.
- 3. **Infrastructure Design**: Systems need to be designed to handle both average and peak flows to prevent overflows and ensure efficient treatment.

#### **Quality of Sewage**

#### 1. Characteristics:

- **Physical**: Includes solids (both dissolved and suspended), color, and turbidity.
- **Chemical**: Consists of organic matter (measured as Biological Oxygen Demand (BOD) and Chemical Oxygen Demand (COD)), nutrients (nitrogen and phosphorus), heavy metals, and chemicals from industrial processes.
- **Biological**: Presence of pathogens, such as bacteria, viruses, and protozoa.

#### 2. Treatment Stages:

- **Primary Treatment**: Removes large solids and some organic matter through screening and sedimentation.
- **Secondary Treatment**: Targets dissolved and suspended organic matter using biological processes. Common methods include activated sludge, trickling filters, and bio-towers.
- **Tertiary Treatment**: Further polishes the effluent to remove remaining contaminants, including nutrients and pathogens. Techniques may include filtration, chemical treatment, and advanced oxidation.
- 3. **Regulations and Standards**: Ensuring that sewage quality meets regulatory standards is crucial. These standards are set to protect human health and the environment, and they vary by region.
- 4. Disposal Methods:
  - **Land Application**: Treated sewage can be used as a fertilizer if it meets safety standards.
  - **Discharge into Water Bodies**: Treated effluent is often released into rivers, lakes, or oceans. It's crucial to ensure that the treatment meets environmental regulations to avoid pollution.

#### Sewer Appurtenances and Sewage Disposal

#### S.Vennila

Sewer appurtenances and sewage disposal are essential components of a municipal wastewater management system. Let's break down what each term encompasses:

#### **Sewer Appurtenances**

Sewer appurtenances are various components and devices associated with the sewer system that aid in the efficient management and operation of wastewater. Some common appurtenances include:

- 1. **Manholes**: Access points to the sewer system that allow for inspection, maintenance, and cleaning.
- 2. **Catch Basins**: Structures designed to capture and divert stormwater and debris from streets into the sewer system.
- 3. **Grease Traps**: Devices installed in commercial kitchens to capture grease and oils before they enter the sewer system.
- 4. **Sewer Vents**: Openings that allow gases to escape from the sewer system and help prevent pressure build-up.
- 5. **Pump Stations**: Facilities equipped with pumps to lift and move wastewater through the sewer system, especially in low-lying areas.
- 6. **Backflow Prevention Devices**: Mechanisms that prevent wastewater from flowing backward into clean water supplies or other parts of the sewer system.

#### **Sewage Disposal**

Sewage disposal refers to the methods and processes used to manage and treat wastewater before it is released back into the environment. The primary stages include:

- 1. **Collection**: Wastewater is collected from various sources through a network of pipes and sewers.
- 2. **Transportation**: Collected wastewater is transported to treatment facilities using gravity or pumping systems.
- 3. **Treatment**:
  - **Primary Treatment**: Removes large solids and floating debris through screening and sedimentation.
  - Secondary Treatment: Uses biological processes to degrade dissolved and suspended organic matter.
  - **Tertiary Treatment**: Further polishes the treated water to remove remaining contaminants, including nutrients and pathogens.

#### 4. Disposal or Reuse:

• **Discharge**: Treated wastewater is safely released into natural water bodies, such as rivers or oceans, following regulatory standards.
### **Sewage Treatment**

### Dr.P. Paramaguru

Sewage treatment is a crucial process for managing wastewater and protecting public health and the environment. It involves several stages to remove contaminants from sewage or wastewater before it's released back into the environment or reused. Here's a general overview of the process:

- 1. **Preliminary Treatment**: This is the initial stage where large debris like sticks, plastic, and other large particles are removed from the sewage. This is usually done through screening and grit removal.
- 2. **Primary Treatment**: In this stage, the sewage is allowed to settle in a large tank where heavier solids sink to the bottom and lighter materials float to the top. This process removes a significant portion of the solids and some organic matter.
- 3. **Secondary Treatment**: This step involves biological processes where microorganisms break down organic matter in the sewage. This is often done in aeration tanks where air is pumped in to encourage the growth of bacteria that consume the organic pollutants.
- 4. **Tertiary Treatment**: This is an advanced stage that further polishes the treated water to remove remaining contaminants, including nutrients like nitrogen and phosphorus. Techniques used can include filtration, chemical treatment, and additional biological processes.
- 5. **Disinfection**: Before the treated water is released into the environment or reused, it is disinfected to kill any remaining pathogens. Common methods include chlorination, UV light, or ozone treatment.
- 6. **Sludge Treatment**: The solids removed during the earlier stages are treated separately. This can involve thickening, digestion (where bacteria break down the organic material), dewatering, and sometimes further treatment to reduce pathogens and stabilize the sludge.

Effective sewage treatment not only helps to protect water bodies from pollution but also reduces the risk of waterborne diseases and supports environmental sustainability.

**Recycling/Reuse**: Treated water can be repurposed for non-potable uses, such as irrigation, industrial processes, or even, in some cases, potable uses after advanced treatment.

**Sludge Management**: Solid waste residuals from the treatment process are managed through methods like digestion, dewatering, and land application.

Effective sewer appurtenances and sewage disposal systems are crucial for maintaining public health, protecting the environment, and ensuring the efficient operation of urban infrastructure.

# HIGHWAY ENGINEERING

Edited by A.BELCIYA MARY



# **HIGHWAY ENGINEERING**

Copyright © 2021 by A.BELCIYA MARY

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means

without written permission from the author.

ISBN: 978-93-6255-114-6

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

CHAPTER 1- Highway Planning and Alignment	-	05
R. Devi		
CHAPTER 2- Geometric Design of Highways	-	10
D. Jeyakumar		
CHAPTER -3- Cost Design of Flexible and Rigid Pavements	-	28
D.Amal Colins		
CHAPTER- 4- Highway Construction Materials and Practice		
Construction	-	40
K Shanthi		
<b>CHAPTER 5-</b> Evaluation and Maintenance of Pavements	-	75
S. Ramakrishna		
CHAPTER 6-Flexible Pavement	-	90
K Shanthi		
CHAPTER 7-Geometric Design	-	110
A. Belciya Mary		
CHAPTER 8-Intelligent Highways	-	130
D.Jeyakumar		
CHAPTER 9-Pavement Design	-	140
S. Vennila		
CHAPTER 10- Road Construction Planning	-	155
Dr.P. Paramaguru		
REFERENCES	_	180

### **Highway Planning and Alignment**

### R. Devi

Highway planning and alignment involve a series of critical steps to ensure the efficient and safe movement of traffic. Here's an overview of the process:

# **1. Initial Planning**

- **Needs Assessment:** Evaluate current and future traffic needs, considering population growth, economic development, and transportation demand.
- **Feasibility Studies:** Assess potential routes and their impact on the environment, communities, and existing infrastructure.

# 2. Route Selection

- **Route Alternatives:** Develop several route options, considering factors like topography, land use, and environmental impact.
- **Public Involvement:** Engage with stakeholders and the public to gather feedback and address concerns.
- **Cost-Benefit Analysis:** Compare the costs of construction, maintenance, and environmental impact with the expected benefits.

# **3. Alignment Design**

- **Horizontal Alignment:** Determine the path of the highway in plan view (e.g., straight sections, curves).
- Vertical Alignment: Plan the changes in elevation along the route, including grades and vertical curves.
- **Geometric Design:** Design cross-sections, lane widths, shoulder widths, and sight distances to ensure safety and efficiency.

# 4. Environmental and Social Considerations

- Environmental Impact Assessment (EIA): Analyze how the proposed highway will affect local ecosystems, water resources, and air quality.
- **Mitigation Measures:** Develop strategies to minimize or offset negative environmental impacts, such as wildlife crossings or noise barriers.
- **Community Impact:** Address potential effects on local communities, including displacement and changes to property values.

# 5. Detailed Design and Engineering

- Surveying and Mapping: Conduct detailed surveys to create accurate maps of the proposed route.
- **Geotechnical Studies:** Assess soil and rock conditions to inform foundation design and construction methods.

# CHAPTER 2 Geometric Design of Highways

### D Jeyakumar

Geometric design in highway engineering is crucial for ensuring safety, efficiency, and comfort in roadways. It involves the design of roadways' physical elements to accommodate various traffic conditions and environmental factors. Here's a summary of the key components involved:

# 1. Horizontal Alignment

- **Curves:** Horizontal curves are designed to ensure smooth vehicle transitions between straight sections of the road. Key parameters include radius, superelevation (banking of the roadway), and transition curves.
- Tangents: Straight sections between curves that allow vehicles to travel in a straight path.

# 2. Vertical Alignment

- **Grades:** Refers to the slope or steepness of the road. It's important for vehicle performance and drainage.
- Vertical Curves: Smooth transitions between different grades to avoid abrupt changes in slope. They are designed based on the rate of grade change and sight distance.

# **3.** Cross-Section Design

- Lane Widths: Standard lane widths are typically 12 feet (3.6 meters), though they can vary based on road type and location.
- **Shoulders:** Provide space for emergency stops and accommodate lateral movement. Widths vary based on the type of road and expected traffic.
- Roadway Pavement: Includes materials and thickness for durability and performance.

# 4. Sight Distance

- Stopping Sight Distance: The distance required for a driver to see an obstacle and stop safely.
- **Passing Sight Distance:** The distance needed for a vehicle to pass another safely.

# 5. Super elevation

• **Banking:** The tilt of the road surface on curves to counteract the centrifugal force experienced by vehicles. Super elevation helps in maintaining vehicle stability and comfort.

# 6. Roadway Capacity and Level of Service

- **Capacity:** Refers to the maximum number of vehicles that can pass a point on a roadway in a given time period.
- Level of Service (LOS): A qualitative measure describing operational conditions of the roadway based on factors like speed, travel time, and congestion.

# CHAPTER 3 Cost Design of Flexible and Rigid Pavements

# **D**.Amal Colins

Designing the cost of flexible and rigid pavements involves various factors and considerations. Both types of pavements—flexible and rigid—have different design requirements and cost implications. Here's a detailed breakdown of each:

# **1. Flexible Pavements**

### Design Considerations:

- **Materials:** Typically involve a layered system with an asphalt surface layer, a base layer (often crushed stone or gravel), and a sub-base layer.
- **Traffic Load:** Designed to withstand traffic loads by distributing the load through the layers.
- Climate: Affects the choice of materials and the thickness of the asphalt layer.
- **Drainage:** Proper drainage design is crucial to prevent water damage.

### Cost Components:

- Material Costs: Asphalt, aggregate for base and sub-base layers.
- Construction Costs: Labor, machinery for laying asphalt and compacting layers.
- Maintenance Costs: Flexible pavements require regular maintenance, including resurfacing and pothole repairs.
- Lifecycle Costs: Generally lower initial cost but higher ongoing maintenance costs.

# **Cost Breakdown Example:**

- Asphalt Surface Layer: Costs can vary depending on the mix and regional pricing. Typically measured in dollars per ton.
- Base and Sub-base Layers: Costs include materials (aggregate), transport, and installation.
- Labor and Equipment: Includes costs for machinery like pavers, rollers, and trucks.

# 2. Rigid Pavements

### **Design Considerations:**

- **Materials:** Typically made from concrete. Includes a surface layer (concrete slab), sometimes with a base layer.
- Load Distribution: Concrete pavements distribute loads over a wider area, often resulting in a thicker slab compared to the asphalt layer.
- Temperature and Weather: Concrete is sensitive to temperature and requires proper curing.
- **Cracking:** Proper design to prevent and manage cracking is essential.

### Cost Components:

- Material Costs: Concrete, reinforcing steel (if used), and any additives.
- Construction Costs: Labor, equipment for mixing, transporting, and placing concrete.

### CHAPTER 4 Highway Construction Materials and Practice

### K Shanti

Highway construction involves a range of materials and practices to ensure durability, safety, and efficiency. Here's a general overview of the key materials and practices commonly used:

# 1. Materials

### 1.1. Asphalt

- Types: Hot Mix Asphalt (HMA), Warm Mix Asphalt (WMA), Cold Mix Asphalt.
- **Properties:** Flexible, durable, and has good skid resistance. Often used for surfacing.

### 1.2. Concrete

- Types: Portland Cement Concrete (PCC), Roller Compacted Concrete (RCC).
- **Properties:** Rigid, strong under compression, and resistant to deformation. Commonly used for pavements, curbs, and sidewalks.

### 1.3. Aggregates

- **Types:** Coarse aggregates (gravel, crushed stone), Fine aggregates (sand).
- **Properties:** Provides structural support and drainage. Aggregates are mixed with binders like asphalt or cement.

### 1.4. Subgrade and Subbase Materials

- **Types:** Soil, crushed stone, gravel, stabilized soil.
- **Properties:** Provides a foundation for the road structure. Needs to be well-compacted to support the load above.

### 1.5. Geotextiles and Geogrids

- Types: Non-woven, woven, and geogrids.
- **Properties:** Used for soil stabilization, erosion control, and improving load distribution.

### 1.6. Additives and Stabilizers

- **Types:** Lime, cement, fly ash, bituminous additives.
- **Properties:** Enhance the properties of the base materials, improve stability and durability.

# **2. Practices**

### 2.1. Site Preparation

- **Tasks:** Clearing, excavation, grading, and compaction.
- **Purpose:** To create a stable and level foundation for the road.

# CHAPTER 5 Evaluation and Maintenance of Pavements

# S.Ramakrishnan

valuating and maintaining pavements is crucial for ensuring their longevity, safety, and functionality. Here's an overview of the key aspects involved:

# **Evaluation of Pavements**

### 1. Condition Assessment:

- **Visual Inspection:** Regular visual inspections help identify surface distresses such as cracks, potholes, ruts, and surface wear.
- **Pavement Condition Index (PCI):** A numerical rating that reflects the overall condition of the pavement, usually derived from a visual survey and condition rating.

### 2. Deflection Testing:

- **Falling Weight Deflectometer (FWD):** Measures the pavement's response to a load, providing data on its structural capacity.
- **Benkelman Beam:** Measures pavement deflection under a standard load, often used for flexible pavements.

### 3. Core Sampling:

• **Material Testing:** Extracting core samples from the pavement to analyze the material properties, layer thicknesses, and underlying conditions.

### 4. Traffic Load Analysis:

• **Traffic Counts and Load Data:** Estimating the impact of traffic loads and volume on pavement performance.

### 5. Environmental Impact:

• **Climate and Weather Data:** Assessing the effects of environmental conditions, like freeze-thaw cycles and rainfall, on pavement performance.

### 6. Structural Evaluation:

• Layer Strength: Analyzing the strength of various pavement layers to assess their load-carrying capacity.

# **Maintenance of Pavements**

### 1. Routine Maintenance:

- Crack Sealing: Filling cracks to prevent water infiltration and further deterioration.
- **Pothole Repair:** Immediate repair of potholes to prevent accidents and further damage.
- Surface Cleaning: Removing debris and vegetation that can impact pavement condition.

# 2. Preventive Maintenance:

- **Sealcoating:** Applying a protective coating to the pavement surface to extend its life and enhance its appearance.
- **Overlays:** Adding a new layer of asphalt or concrete to improve the surface condition and extend the pavement's life.

# 3. Rehabilitation:

• **Milling and Resurfacing:** Removing the top layer of the pavement and replacing it with a new layer.

### **Flexible Pavement**

### K Shanthi

Flexible pavement refers to a type of road surface that can deform under load and return to its original shape once the load is removed. This type of pavement is designed to distribute traffic loads over a larger area, reducing stress on the underlying soil and subgrade. Here's a breakdown of its main features:

# **Composition:**

- 1. **Surface Course**: The top layer that comes into direct contact with traffic. It's usually made of asphalt or bituminous materials and is designed to provide skid resistance and a smooth ride.
- 2. **Binder Course**: Located beneath the surface course, this layer helps in bonding the surface course to the underlying layers and provides additional strength.
- 3. **Base Course**: This layer lies below the binder course and provides structural support to distribute loads to the subgrade. It's typically made from crushed stone or gravel.
- 4. **Subbase**: This layer is placed above the subgrade and serves as a transition between the base course and the natural soil. It also helps in drainage.

# **Characteristics:**

- **Deformability**: The flexibility allows the pavement to accommodate small movements or deformations without cracking.
- Load Distribution: The multiple layers work together to spread the load from traffic across a larger area.
- **Drainage**: The design often includes features to allow water to drain away, which helps in reducing damage from freeze-thaw cycles and prolongs the pavement's life.

# **Advantages:**

- **Durability**: When properly constructed and maintained, flexible pavements can last a long time and withstand various weather conditions.
- **Cost-Effectiveness**: Generally, they are less expensive to construct and maintain compared to rigid pavements.
- **Ease of Repair**: Patching and resurfacing can be relatively straightforward, allowing for quicker repairs compared to rigid pavements.

# **Disadvantages:**

- Maintenance: Requires regular maintenance to address issues like cracking and surface wear.
- **Susceptibility to Deformation**: Can be prone to rutting or deformation under heavy loads if not designed or maintained properly.

# **GEOMETRIC DESIGN**

# A. Belciya Mary

Geometric design in highway engineering focuses on creating roadways that are safe, efficient, and comfortable for drivers. This involves the application of various design principles and standards to ensure that roads accommodate traffic flow and meet safety requirements. Here's an overview of key aspects of geometric design in highways:

# 1. Alignment

- **Horizontal Alignment**: Refers to the layout of the road in the horizontal plane, including straight sections (tangents) and curves. Proper design ensures smooth transitions and minimizes sharp turns.
- **Vertical Alignment**: Concerns the road profile in the vertical plane, including grades (slopes) and vertical curves. It affects sight distance, vehicle speed, and comfort.

# 2. Cross-Section Elements

- Lane Width: The width of each lane affects the safety and comfort of the driving experience. Standard widths vary but typically range from 3.0 to 3.7 meters (10 to 12 feet).
- **Shoulders**: Provide space for emergencies, vehicle breakdowns, and lateral support. Shoulders can be paved or unpaved, depending on the road type.
- **Medial Strips**: Areas between opposing lanes, which can include medians or barriers for safety and separation.

# 3. Curves and Super elevation

- **Horizontal Curves**: Allow for smooth changes in direction. The radius of the curve and the design speed are crucial for safe maneuvering.
- **Superelevation**: The banking of a roadway curve to counteract centrifugal force. It helps maintain vehicle stability and comfort during turns.

# 4. Sight Distance

- **Stopping Sight Distance**: The distance required for a driver to perceive and react to an obstacle. It's crucial for safety, particularly in areas with curves or intersections.
- **Passing Sight Distance**: The distance needed for a driver to safely overtake another vehicle on a two-lane road.

# **5. Intersections**

- **Design**: Includes layout, turning lanes, and traffic control devices (signals, signs). Proper design minimizes conflicts and improves flow.
- **Types**: Can be at-grade (crossroads, T-intersections) or grade-separated (overpasses, interchanges).

# **Intelligent Highways**

D.Jeyakumar

Intelligent highways refer to road systems equipped with advanced technologies to improve safety, efficiency, and overall traffic management. These systems can include a range of smart features:

- 1. **Real-Time Traffic Monitoring**: Using sensors and cameras, intelligent highways collect data on traffic flow, congestion, and incidents. This information can be used to adjust traffic signals, manage lane usage, and provide real-time updates to drivers.
- 2. **Dynamic Lane Management**: Lanes can be adjusted in real-time based on traffic conditions. For example, lanes might be added or removed during peak hours, or dedicated lanes could be used for high-occupancy vehicles.
- 3. Variable Speed Limits: Speed limits can be adjusted dynamically in response to traffic conditions, weather, or road work. This helps optimize traffic flow and improve safety.
- 4. Automated Traffic Signals: Traffic lights can be controlled to improve flow and reduce congestion. For instance, lights can be synchronized to create green waves for smoother traffic flow.
- 5. **Smart Signage**: Digital signs can provide real-time information about traffic conditions, road closures, or upcoming hazards. They can also display variable messages to guide drivers.
- 6. Advanced Warning Systems: These systems can alert drivers to upcoming dangers, such as icy conditions, construction zones, or accidents, allowing them to take appropriate action.
- 7. Vehicle-to-Infrastructure (V2I) Communication: Vehicles can communicate with road infrastructure to exchange information, such as traffic signals or road conditions, helping to optimize driving routes and enhance safety.
- 8. **Eco-Friendly Solutions**: Intelligent highways can incorporate features that support sustainable driving practices, such as electric vehicle charging stations, and systems that encourage smoother driving to reduce fuel consumption and emissions.
- 9. **Integration with Navigation Systems**: Real-time data from intelligent highways can be integrated with GPS and navigation apps to provide more accurate route suggestions and traffic forecasts.

The goal of intelligent highways is to create a more efficient, safer, and user-friendly transportation network by leveraging technology to address and manage the challenges of modern road use.

# **Pavement Design**

S. Vennila

Pavement design is a critical aspect of civil engineering and urban planning, focusing on creating durable, efficient surfaces for roads, highways, and other pathways. Here's an overview of key concepts and processes involved:

# 1. Types of Pavements

- **Flexible Pavements**: These use multiple layers of materials, including asphalt. They distribute loads over a broad area and are more adaptable to minor deformations.
- **Rigid Pavements**: Made primarily of concrete, these pavements are less flexible but offer higher load-bearing capacity and durability. They tend to distribute loads over a smaller area but can be more susceptible to cracking.
- **Composite Pavements**: These combine elements of both flexible and rigid pavements, aiming to leverage the benefits of each.

# 2. Pavement Design Process

- 1. **Traffic Analysis**: Estimating the volume, weight, and type of traffic that the pavement will experience. This influences the thickness and material choice.
- 2. **Subgrade Evaluation**: Assessing the strength and characteristics of the soil or rock beneath the pavement. This is crucial for determining the load-bearing capacity and stability.
- 3. **Material Selection**: Choosing appropriate materials for the surface, base, and sub-base layers. This involves considering factors like durability, cost, and local availability.
- 4. **Layer Design**: Designing the thickness and composition of each layer (e.g., sub-base, base course, surface course). The design must ensure adequate load distribution and resistance to deformation.
- 5. **Structural Analysis**: Evaluating the pavement's ability to withstand traffic loads and environmental conditions. This often involves using engineering models and simulations.
- 6. **Drainage Considerations**: Ensuring proper drainage to prevent water accumulation, which can weaken the pavement structure and lead to premature failure.
- 7. **Maintenance Planning**: Developing a plan for routine inspections, repairs, and upgrades to extend the pavement's lifespan.

# 3. Key Considerations

- Load Distribution: Pavements must distribute loads to prevent excessive stress on underlying layers.
- Environmental Factors: Climate and weather conditions, such as temperature extremes and precipitation, affect material choice and design.
- **Economic Factors**: Budget constraints influence material selection and design choices, balancing initial costs with long-term maintenance and durability.

### **Road Construction Planning**

### Dr.P. Paramaguru

Road construction planning is a crucial process that involves several stages to ensure the successful development of a road project. Here's a comprehensive overview of the steps typically involved:

# 1. Needs Assessment and Feasibility Study

- **Identify the Need:** Determine the purpose of the road (e.g., reducing traffic congestion, improving access to areas).
- **Conduct Feasibility Study:** Evaluate the technical, economic, and environmental feasibility of the project.

# 2. Planning and Design

- **Route Selection:** Analyze various route options considering factors like existing infrastructure, land use, and environmental impact.
- **Preliminary Design:** Develop initial design concepts including road alignment, cross-sections, and major structures.
- Environmental Impact Assessment (EIA): Assess potential environmental impacts and plan mitigation measures.

# 3. Detailed Design

- **Design Development:** Create detailed designs for the road, including geometric design, drainage systems, and materials.
- Cost Estimation: Prepare detailed cost estimates for construction, maintenance, and operation.
- **Obtain Approvals:** Secure necessary approvals and permits from relevant authorities and stakeholders.

# 4. Procurement and Contracting

- Tendering Process: Prepare and issue tenders or requests for proposals to contractors.
- **Contract Award:** Evaluate bids, award contracts, and finalize agreements with selected contractors.

# **5.** Construction

- **Site Preparation:** Clear and prepare the construction site, including earthworks and utility relocations if needed.
- **Execution:** Implement the construction activities as per the approved designs, including laying the foundation, paving, and installing signage and safety features.
- Quality Control: Monitor construction quality and adherence to specifications.

# IRRIGATION ENGINEERING

Edited by

**J.SANTHIYAA JENIFER** 



# **IRRIGATION ENGINEERING**

Copyright © 2021 by J. SANTHIYAA JENIFER

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-645-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLEOF CONTENTS**

CHAPTER 1- Crop Water Requirement	-05
R. Devi	
CHAPTER 2-Irrigation Methods	-10
K. Shanthi	
CHAPTER -3-Diversion and Impounding Structures	-30
A. Belciya Mary	
CHAPTER- 4-Canal Irrigation	- 45
S. Ramakrishna	
CHAPTER 5-Water Management in Irrigation	- 75
P Venkateswaran	
CHAPTER 6-Methods of Field Water Application	- 90
D.Jeyakumar	
CHAPTER 7-Surface Irrigation Methods	-110
K Shanthi	
CHAPTER 8-Border Irrigation Method	-130
J. Santhiyaa Jenifer CHAPTER 9-Surface Irrigation	-145
S.Vennila	
CHAPTER 10-Subsurface Irrigation Methods	-150
Dr.P. Paramaguru	
REFERENCES	- 180

### **Crop Water Requirement**

### R.Devi

Crop water requirement refers to the amount of water needed to grow a crop to its full potential. It's a key factor in agriculture, influencing both crop yield and quality. This requirement varies depending on several factors:

- 1. **Crop Type:** Different crops have different water needs. For example, rice requires more water than wheat.
- 2. **Growth Stage:** Water requirements can vary throughout the growing season. For instance, many crops need more water during flowering and fruit development.
- 3. **Climate and Weather:** Temperature, humidity, and rainfall affect how much water a crop needs. Hotter and drier conditions usually increase water requirements.
- 4. **Soil Type:** Soil properties influence water availability and retention. Sandy soils drain quickly and might need more frequent irrigation, while clay soils retain water longer.
- 5. **Water Management Practices:** Efficient irrigation methods (like drip or sprinkler systems) can optimize water use and reduce waste.

In practice, crop water requirements are often estimated using reference evapotranspiration (ET<sub>0</sub>) and crop coefficients (Kc). ET<sub>0</sub> measures the rate of water loss from the soil and plant surface, while the crop coefficient adjusts this value based on the specific needs of the crop.

To manage water efficiently, farmers and agronomists monitor these factors and adjust irrigation schedules accordingly.

# CHAPTER 2 Irrigation Methods

# K.Shanthi

Irrigation methods are crucial for managing water resources in agriculture, ensuring crops receive adequate moisture for optimal growth. Here are some common irrigation methods:

# **1. Surface Irrigation**

- **Flood Irrigation**: Water is applied directly to the field and allowed to flow over the surface, covering the crop area. It's relatively simple but can be inefficient due to evaporation and runoff.
- **Furrow Irrigation**: Water is directed into furrows (small channels) between crop rows. This method can be more efficient than flood irrigation as it targets specific areas.

# 2. Drip Irrigation

- **Drip Tape**: Delivers water directly to the root zone of plants through a network of tubes and emitters. This method is highly efficient and minimizes water wastage.
- **Drip Line**: Similar to drip tape but usually more durable, often used in commercial settings. It provides precise water delivery and reduces evaporation and runoff.

# **3. Sprinkler Irrigation**

- **Center Pivot**: A rotating sprinkler system mounted on wheeled towers, covering large circular areas. It's efficient and ideal for large fields.
- Lateral Move: A system of sprinklers on moving platforms that travels across the field. It provides even coverage but can be costly and complex.
- **Fixed Sprinkler**: Stationary systems with multiple sprinkler heads that cover specific areas. It's commonly used in residential and smaller agricultural settings.

# 4. Subsurface Irrigation

• **Subsurface Drip Irrigation (SDI)**: Involves placing drip lines or tubing below the soil surface, providing water directly to the root zone and reducing evaporation losses.

# **5. Localized Irrigation**

• **Micro-Irrigation**: A broad category that includes drip and sprinkler systems tailored for specific crop needs, often used in high-value crops like fruits and vegetables.

# 6. Rainwater Harvesting

• Collects and stores rainwater for later use. It's a sustainable method that can supplement other irrigation systems, particularly in areas with irregular rainfall.

### CHAPTER 3 Diversion and Impounding Structures

# A.Belciya Mary

In irrigation engineering, diversion and impounding structures are essential for managing water resources efficiently. They help in directing and storing water for agricultural, industrial, or municipal use. Here's a breakdown of these structures:

# **Diversion Structures**

### 1. Purpose:

- To redirect water from a river or stream to a canal or pipeline.
- To control the flow of water into irrigation systems.

### 2. Types:

- Weirs:
  - Simple structures that allow water to flow over a barrier.
  - Useful for measuring flow rates and controlling water levels.
- Headgates:
  - Gates that regulate the amount of water entering an irrigation canal or ditch.
  - Can be manually or automatically operated.
- Canal Diversion Structures:
  - Includes various forms like drop structures and intake gates.
  - Designed to control and manage water flow into irrigation canals.
- Sluice Gates:
  - Adjustable gates used to control water flow and prevent flooding.
  - Often used in conjunction with weirs.

### 3. Considerations:

- Ensuring minimal environmental impact.
- Managing sediment and debris to prevent blockages.
- Designing for ease of maintenance and operation.

# **Impounding Structures**

### 1. Purpose:

- To store water for future use, such as irrigation, flood control, or water supply.
- To regulate and control the flow of water in a river or stream.

### 2. Types:

- Dams:
  - Large structures built across rivers or streams to create reservoirs.
  - Used for water storage, flood control, and hydroelectric power generation.

# CHAPTER 4 Canal Irrigation

# S. Ramakrishnan

Canal irrigation is a method of supplying water to agricultural fields using artificial channels, or canals. This technique is essential in areas where natural rainfall is insufficient for crops or where water needs to be distributed across large areas.

Here's a brief overview of how canal irrigation works and its key aspects:

# **How It Works**

- 1. **Water Source**: Water for canal irrigation typically comes from rivers, lakes, reservoirs, or groundwater. Dams or weirs are often used to control and direct the flow of water.
- 2. **Canal System**: The water is transported through a network of canals. The system usually includes main canals, secondary canals, and tertiary canals, each serving different purposes and levels of distribution.
- 3. **Distribution**: Water flows from the main canal to smaller canals, eventually reaching individual fields. Gates, sluices, and other control structures help regulate the flow and ensure even distribution.
- 4. **Field Application**: Water is distributed to crops via furrows, basins, or surface irrigation methods. In some systems, water may be delivered through pipes or tubes.

# Advantages

- **Efficiency**: Canal irrigation can be more efficient than some other methods, especially in areas with predictable water sources.
- Scale: It can cover large areas of farmland, making it suitable for extensive agricultural regions.
- **Control**: Provides better control over the timing and amount of water applied to crops.

# Disadvantages

- **Maintenance**: Canals require regular maintenance to prevent blockages, erosion, and leakage.
- **Evaporation and Seepage**: Water loss can occur due to evaporation from the surface and seepage from the canal banks.
- **Environmental Impact**: Canals can alter natural water flow patterns, potentially affecting ecosystems.

# **Modern Adaptations**

- Lining Canals: To reduce seepage and water loss, many canals are lined with materials like concrete or plastic.
- Automated Control: Modern systems use technology for better water management, including sensors and automated gates.

# CHAPTER 5 Water Management in Irrigation

### P Venkateswaran

Water management in irrigation is crucial for efficient and sustainable agriculture. It involves planning and controlling the water resources used for crop production. Here are some key aspects of effective water management in irrigation:

# 1. Assessment of Water Resources

- **Sources**: Identify and evaluate water sources such as rivers, lakes, reservoirs, groundwater, and rainwater.
- Availability: Assess the quantity and reliability of these sources throughout the year.

# 2. Irrigation Methods

- **Surface Irrigation**: Involves flooding fields or channels. Common types include furrow, basin, and border irrigation.
- **Drip Irrigation**: Delivers water directly to the plant roots through a network of pipes and emitters, minimizing water wastage.
- **Sprinkler Irrigation**: Mimics natural rainfall using a system of pipes, pumps, and sprinklers to distribute water evenly.
- **Subsurface Irrigation**: Water is delivered below the soil surface using buried pipes or tubes.

# **3. Soil and Crop Considerations**

- **Soil Type**: Different soils retain and drain water differently. Clay soils, for instance, hold more water but drain slowly, while sandy soils drain quickly but retain less water.
- **Crop Needs**: Different crops have varying water requirements. Understanding these needs helps in scheduling irrigation more effectively.

# 4. Water Scheduling

- **Timing**: Plan when to irrigate based on crop needs, weather conditions, and soil moisture levels.
- **Frequency**: Determine how often to irrigate based on the crop's growth stage and soil's water holding capacity.

# **5.** Technological Integration

• **Sensors**: Soil moisture sensors and weather stations provide real-time data to optimize irrigation schedules.

# **Methods of Field Water Application**

D.Jeyakumar

Field water application in irrigation involves various methods to deliver water to crops efficiently. The choice of method depends on factors like the type of crop, soil characteristics, water availability, and economic considerations. Here's an overview of the main methods used:

# **1. Surface Irrigation**

- **Furrow Irrigation**: Water flows through small, narrow channels (furrows) between rows of crops. It's commonly used for row crops like corn and potatoes.
- **Border Irrigation**: Fields are divided into long, narrow strips bordered by ridges. Water is applied to the top of the strip and allowed to flow down the length of the field.
- **Basin Irrigation**: Fields are divided into small, level basins surrounded by bunds or embankments. Water is applied to each basin and allowed to infiltrate the soil.
- **Flood Irrigation**: Water is applied directly to the entire field, often used in flat fields or areas with high water tables.

# **2. Drip Irrigation**

- **Drip Tape**: Thin, flexible tubing with emitters placed at regular intervals delivers water directly to the plant roots. It's highly efficient and minimizes water waste.
- **Drip Line**: Similar to drip tape but generally more durable, with emitters integrated into a more rigid hose.

# 3. Sprinkler Irrigation

- **Center Pivot**: A rotating sprinkler system mounted on wheeled towers, covering a circular area. Suitable for large fields and can be automated.
- Lateral Move: A system of sprinklers mounted on wheeled frames that move laterally across the field. Often used for rectangular fields.
- **Side Roll**: A system where sprinklers are mounted on wheeled towers that move side to side. It's often used in areas with rolling terrain.
- **Solid Set**: A stationary system where sprinklers are permanently installed on a field and water is applied without moving the system.

# 4. Subsurface Irrigation

- **Subsurface Drip**: Drip lines are buried below the soil surface, delivering water directly to the root zone. This reduces evaporation and runoff.
- **Subsurface Pipe Systems**: A network of buried pipes that deliver water directly to the root zone, often used in combination with other irrigation methods.

# 5. Micro-Irrigation

# **Surface Irrigation Methods**

# K Shanthi

Surface irrigation is a traditional and widely used method of delivering water to crops. It relies on gravity to move water across the field. Here are the main types of surface irrigation methods:

- 1. **Flood Irrigation**: This method involves flooding the entire field with water. It can be subdivided into:
  - **Furrow Flooding**: Water flows through furrows (small channels) between rows of crops. It's common for row crops and can be managed with relatively simple infrastructure.
  - **Basin Flooding**: Water is applied to a field divided into basins or basins surrounded by bunds (dikes). It's typically used for crops like rice that are grown in flooded conditions.
- 2. **Basin Irrigation**: Fields are divided into basins by bunds or levees, and each basin is flooded. This method is suitable for crops that thrive in standing water, like rice.
- 3. **Border Irrigation**: The field is divided into long, narrow strips or borders separated by raised beds or ridges. Water is applied to one end of the border and flows down the length of the field. This method is efficient for flat or gently sloping fields.
- 4. **Furrow Irrigation**: Water is applied to small channels or furrows between rows of crops. This method is particularly useful for row crops and can be controlled to ensure even water distribution.
- 5. **Check Basin Irrigation**: The field is divided into small, square or rectangular basins surrounded by check walls. Water is introduced into the basins in a controlled manner. This method is suitable for crops that require consistent moisture levels.
- 6. **Corrugated Irrigation**: This method involves creating a series of parallel corrugations or ridges and furrows in the field. Water is introduced at the end of the corrugations, and it flows down through the furrows.

# Advantages of Surface Irrigation:

- Simplicity: Often less complex and less expensive to set up compared to other methods.
- Low Energy Requirements: Uses gravity, reducing the need for pumps and other equipment.
- Flexibility: Can be adapted to various field shapes and sizes.

# **Disadvantages of Surface Irrigation:**

- Water Wastage: Can lead to significant water losses through evaporation, runoff, and deep percolation.
- **Labor Intensive**: Requires more manual labor for managing water flow and field preparation.
- **Inefficiency in Steep Terrain**: Less effective on slopes where water can quickly run off rather than infiltrate.

# **Border Irrigation Method**

# J.Santhiyaa Jenifer

Border irrigation is a traditional method used in agriculture to distribute water across a field. Here's an overview of how it works and its key features:

# **How Border Irrigation Works:**

- 1. **Field Layout:** The field is divided into long, narrow strips or "borders," which are separated by raised barriers called "borders" or "levees."
- 2. **Water Supply:** Water is introduced at one end of the border, usually through a channel or furrow.
- 3. **Flow Distribution:** The water flows down the border, spreading out evenly across the field. Gravity helps in distributing the water, as it moves from higher to lower areas.
- 4. **Absorption:** The soil absorbs the water as it flows, providing moisture to the crops planted in the border area.

# **Key Features:**

- **Simplicity:** Border irrigation is relatively easy to set up and manage, especially in fields with uniform slopes.
- **Suitability:** It is suitable for crops that can tolerate a temporary waterlogging situation and is commonly used for crops like wheat, barley, and some vegetables.
- **Efficiency:** The efficiency of border irrigation can vary. It depends on factors such as the slope of the field, soil type, and the uniformity of water distribution. Well-managed border irrigation can be quite effective, but poor management may lead to water wastage and uneven distribution.
- **Cost:** It often involves lower initial costs compared to more sophisticated irrigation systems like drip or sprinkler systems.

# Advantages:

- Low Initial Investment: Requires minimal infrastructure compared to advanced systems.
- **Simplicity in Design:** Easy to design and implement, especially in regions with flat or gently sloping fields.
- Effective for Certain Soils: Works well with soils that have good infiltration rates and can manage water evenly.

# **Disadvantages:**

- Water Waste: Potential for water wastage due to runoff and deep percolation if not managed properly.
- Labor Intensive: Requires regular monitoring and maintenance to ensure proper water distribution.

### **Surface Irrigation**

S. Vennila

Surface irrigation is a method of watering crops where water is applied directly to the soil surface and allowed to flow over the field by gravity. It's one of the oldest and most commonly used irrigation techniques, and it includes various sub-methods like furrow, basin, border, and flood irrigation.

Here's a brief overview of some common types:

- 1. **Furrow Irrigation**: Water is applied in narrow channels or furrows between crop rows. This method is useful for row crops and helps to minimize water wastage.
- 2. **Basin Irrigation**: Fields are divided into small basins or leveled areas where water is applied and retained until it infiltrates the soil. This method is often used for crops like rice that benefit from standing water.
- 3. **Border Irrigation**: The field is divided into borders or strips separated by ridges, and water is applied at the top of these borders. This method is useful for crops that are grown in wide, flat areas.
- 4. **Flood Irrigation**: Water is allowed to flow over the entire field, flooding it. This method is less controlled and can lead to water wastage, but it can be effective for certain types of crops and soil conditions.

Surface irrigation is relatively simple and inexpensive compared to more modern methods like drip or sprinkler irrigation. However, it can be less efficient in terms of water usage and can lead to issues like soil erosion or uneven water distribution if not managed properly.

# **Subsurface Irrigation Methods**

# Dr.P. Paramaguru

Subsurface irrigation is a method of watering plants where the irrigation system delivers water directly to the root zone below the soil surface. This technique can be highly efficient and effective in conserving water and reducing evaporation and runoff. Here are some common subsurface irrigation methods:

# **1. Drip Irrigation**

- **Components**: Uses a network of tubing and emitters that deliver water directly to the plant roots.
- Advantages: Precise application, minimal water wastage, and reduced evaporation and runoff. Ideal for row crops and garden beds.
- **Disadvantages**: Initial setup cost can be high, and the system requires regular maintenance to prevent clogging.

# 2. Subsurface Drip Irrigation (SDI)

- **Components**: Similar to drip irrigation but the tubing is buried below the soil surface.
- Advantages: Water is delivered directly to the root zone, reducing evaporation and surface weed growth. Ideal for high-value crops and areas with limited water resources.
- **Disadvantages**: Higher installation and maintenance costs. Requires careful management to prevent issues like clogging and root intrusion.

# 3. Soaker Hoses

- **Components**: Perforated hoses that deliver water along their length.
- Advantages: Easy to install and adjust. Suitable for gardens and flower beds.
- **Disadvantages**: Less precise than drip irrigation and can be less durable. Water distribution may be uneven.

# 4. Subsurface Watering Systems

- **Components**: Systems that use buried pipes or tubes with small holes or emitters to deliver water to the root zone.
- Advantages: Effective at reducing evaporation and runoff. Can be used for a variety of plant types.
- **Disadvantages**: Installation can be labor-intensive. The system may need periodic cleaning and maintenance.

# 5. Capillary Tubes

- Components: Small tubes that draw water up from a reservoir below the soil.
- Advantages: Simple design and can be used in small-scale or container gardening.



# STRUCTURAL ANALYSIS II

# EDITED BY K.SHANTHI



# **Structural Analysis II**

Copyright © 2021 by K.Shanthi

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-142-9

Publisher: PonnaiyahRamajayam Institute of Science and Technology (PRIST)

# **TABLEOF CONTENTS**

INTRODUCTION	- 03
<b>CHAPTER 1</b> -Influence Lines for Determinate Beams Mrs. S. Vennila	-12
CHAPTER 2 - Influence Lines for Indeterminate Beams	-24
D Jeyakumar	
CHAPTER 3 - Arches	-41
Mrs. A. Belciya Mary CHAPTER 4 -Cables and Suspension Bridges Mrs. Shanthi CHAPTER 5 -Plastic Analysis	-62 -75
Mrs. R.Devi CHAPTER 6-Deflection of Beams	-86
Mr.DJeyakumar	
CHAPTER 7 - Buckling and Stability of Structures	-112
Mrs. A.Belciya Mary	
CHAPTER 8-Dynamic Analysis of Structures	-132
Mr. S.Ramakrishnan	
CHAPTER 9-Internal Forces in Structures	-151
Mr.D. Amal Colins	
CHAPTER 10-Types of Loads	-160
Mrs.R.Devi	
REFERENCES	-170

# CHAPTER 1 Influence Lines for Determinate Beams

# Mrs. S.Vennila

**Influence lines** are graphical representations used in structural analysis to show how various functions (such as reactions, shear force, or bending moment) at a specific point on a structure change in response to a moving load. For **determinate beams**, the influence lines are straight or piecewise linear because their internal forces and reactions can be determined solely from the equilibrium equations.

# Influence Line for Reaction, Shear Force, and Moment in Determinate Beams 1. Influence Line for Reactions

- Consider a simply supported beam with a length LLL. The reaction at one support (say AAA) depends on the location of a moving unit load across the span.
- The influence line for the reaction at AAA will be a triangle:
  - It starts at 1 (when the load is at AAA).
  - It decreases linearly to 0 at the other end of the span (at BBB).
- The value of the influence line at any point xxx on the beam is  $L-xL\frac{L-x}{L}L-x$ , where xxx is the distance from AAA.

# 2. Influence Line for Shear Force

- The shear force at a point CCC on a simply supported beam has an influence line with a discontinuity at point CCC.
- The value of the influence line to the left of CCC is constant, and the value to the right of CCC changes.
- At the point of the moving load, the influence line will jump from a positive to a negative value (or vice versa).

# **3. Influence Line for Bending Moment**

- The influence line for bending moment at a point CCC in a simply supported beam is piecewise linear.
- The maximum bending moment at CCC occurs when the load is exactly at CCC.
- The influence line is zero at the supports and has a peak at CCC, forming a triangular shape.

# **Example for a Simply Supported Beam:**

For a simply supported beam of length LLL, the influence lines are typically as follows:

- **Reaction at Support A**: Triangular shape with a peak of 1 at AAA and 0 at BBB.
- **Reaction at Support B**: Triangular shape with a peak of 1 at BBB and 0 at AAA.
- Shear at Point C: Discontinuous line with a jump at CCC. For positions to the left of CCC, the value is positive (constant). To the right, it is negative (constant).
- Moment at Point C: A peak at CCC with zero values at supports AAA and BBB.

Influence lines are especially useful for analyzing the effect of moving loads on bridges and other structures. By superimposing the moving load positions onto the influence lines, you can determine the maximum internal forces at any point.

# CHAPTER 2 Influence Lines for Indeterminate Beams Mr. D. Jevakumar

Influence lines for **indeterminate beams** are more complex than for determinate beams due to the additional constraints and redundancies in the system. In indeterminate beams, the internal forces depend not only on equilibrium but also on compatibility of deformations. Influence lines for such beams will be **curved** rather than straight, reflecting the additional constraints.

# Key Points about Influence Lines for Indeterminate Beams:

# 1. Curved Nature:

- Unlike the straight-line influence diagrams of determinate beams, the influence lines for indeterminate beams are curved due to the redistribution of internal forces as the moving load shifts.
- These curves reflect the stiffness distribution and multiple redundancies in the system.

### 2. Redundant Reactions:

- In indeterminate beams, support reactions or internal forces (shear, moment) are influenced by redundant constraints, such as additional supports or continuous spans.
- Influence lines are obtained using advanced methods like the **Moment Distribution Method**, **Slope-Deflection Method**, or **Direct Stiffness Method**, considering both equilibrium and compatibility.

### 3. Hyperstatic Structures:

- Indeterminate beams, or **hyperstatic structures**, resist loads through a combination of bending and shear, but also exhibit internal stress redistribution, which causes the influence lines to be nonlinear.
- The magnitude of reactions, bending moments, and shear forces at any point is influenced by both the load's position and the redundancies of the structure.

# Methods to Draw Influence Lines for Indeterminate Beams: 1. Force Method (or Flexibility Method):

- In this method, the structure is made statically determinate by temporarily removing the redundant supports or constraints.
- The influence lines are first calculated for this modified determinate structure.
- The displacements are then calculated for these "removed" constraints, and compatibility equations are used to restore the indeterminate structure, adjusting the influence lines.

# 2. Moment Distribution Method:

- For continuous beams, this iterative method calculates the influence of a moving load by distributing moments between adjacent spans.
- Influence lines for moments and reactions are constructed by tracking the redistribution of moments as the load moves across the beam.

# 3. Finite Element or Stiffness Method:

- Modern computational tools, such as the **stiffness matrix method**, allow the direct calculation of influence lines for indeterminate structures.
- This method models the beam as a series of elements, accounting for bending stiffness and nodal displacements, to derive the influence lines.

# CHAPTER 3 Arches Mrs. A. Belciya Mary

**Arches** are curved structural elements designed to carry loads primarily through **compression**. They have been used in construction for centuries due to their efficiency in transferring loads, and they are commonly found in bridges, buildings, and other structures. The curved shape of an arch enables it to span large distances while resisting forces, making it one of the most stable structural forms.

Arches come in various forms, depending on their geometry and the way they support loads. The main types include:

### 1. Hinged Arches:

- **Two-Hinged Arch**: It has hinges at the base (supports). This type of arch allows for rotation at the supports but not movement, providing stability while allowing some flexibility to accommodate temperature changes and settlement.
- **Three-Hinged Arch**: It has an additional hinge at the crown (midpoint of the arch), making it statically determinate. This arch type is more flexible and easier to analyze because the internal forces can be determined from equilibrium alone.

### 2. Fixed Arches:

• These arches are rigidly connected to the supports, meaning there are no hinges. Fixed arches are indeterminate and can resist greater loads due to their rigid connections. However, they are more susceptible to temperature changes and differential settlement because they are more constrained.

### 3. Segmental Arch:

• A segmental arch is less than a semicircle, meaning its center of curvature lies above the arch itself. It is commonly used in bridges and spans shorter distances compared to a semicircular arch.

### 4. Semi-Circular Arch:

• This arch forms a perfect half-circle. It is commonly found in classical architecture, particularly in Roman structures. The thrust is directed equally on both sides.

### 5. Pointed Arch:

• Popular in Gothic architecture, the pointed arch has a sharper curve, which helps direct the forces downward more efficiently, allowing for taller and more slender structures like cathedrals.

### 6. Parabolic Arch:

• A parabolic arch follows the shape of a parabola, which is the optimal shape for an arch that only experiences axial compression under uniform loads. This type of arch is often seen in bridges, as it is efficient for carrying loads.

# **Cables and Suspension Bridge**

### Mrs. K Shanthi

**Cables** and **suspension bridges** are fundamental components in bridge engineering, particularly for long-span structures. A **suspension bridge** is a type of bridge where the deck (the part that carries traffic) is hung below suspension cables. The cables transfer the loads from the deck to the towers and eventually to the ground at the anchorages, primarily using tension to carry loads.

### Cables:

Cables are essential structural elements used in both suspension bridges and other structures like cablestayed bridges, cable roofs, and certain types of towers. In most cases, cables work in **pure tension**, meaning they resist pulling forces, and are ideal for spanning large distances with minimal material.

### **Properties of Cables:**

- **Tension-only Members**: Cables can only carry tensile forces, meaning they cannot resist bending, compression, or shear forces.
- **Flexible**: Cables are flexible and adjust their shape based on the applied load, taking on a **funicular shape** (the shape that a perfectly flexible, inextensible cable takes under a given set of loads).
- **Material**: Cables are often made from high-strength materials such as steel or synthetic fibers. Steel cables are most commonly used in large structures because of their high tensile strength.

### **Suspension Bridge:**

A suspension bridge is a structure where the roadway or deck is suspended by cables that are anchored at both ends and pass over tall towers.

### Key Components of a Suspension Bridge:

- 1. Main Cables:
  - The main cables run between the towers and are anchored at both ends of the bridge. These cables carry most of the weight of the deck.
  - They form a **parabolic curve** between the towers under uniform load, but the shape can vary based on load distribution.

### 2. Towers (Pylons):

• Towers support the main cables and transfer the forces from the cables to the ground. These towers must be strong enough to support both the vertical and horizontal forces from the cables.

### 3. Suspender Cables (Hangers):

• These are vertical cables that connect the deck to the main cables. They transfer the weight of the deck to the main cables, which then transfer it to the towers and the anchorages.

### 4. **Deck**:

• The deck is the roadway or platform on which vehicles or pedestrians travel. In a suspension bridge, the deck is suspended from the main cables through the suspender cables.

# CHAPTER 5 Plastic Analysis

### Mrs. R.Devi

**Plastic analysis** is a method used in structural engineering to determine the ultimate load-carrying capacity of structures, primarily focusing on the **post-yield behavior** of materials. Unlike elastic analysis, which assumes that materials behave linearly and return to their original shape when unloaded, plastic analysis allows materials to undergo permanent deformations, accommodating large displacements without collapse.

The plastic analysis method assumes that structures can redistribute internal forces once certain sections yield, enabling the structure to carry loads beyond the initial elastic limit.

# Key Concepts of Plastic Analysis:

### 1. Plastic Hinge:

- A **plastic hinge** forms at a point in the structure where the material has reached its yield point, and rotation occurs without any increase in moment. This hinge allows the structure to redistribute loads.
- When enough plastic hinges form in the structure, it becomes a mechanism (meaning it can collapse under further load) since it has lost sufficient rigidity.

### 2. Yield Criteria:

- The analysis is based on the **yield criterion**, which is the condition under which a material starts to deform plastically.
- For most steel structures, the **Von Mises yield criterion** or the **Tresca criterion** is used to describe when yielding occurs under combined stress states.

### 3. Collapse Mechanism:

- In plastic analysis, collapse occurs when a sufficient number of plastic hinges form to turn the structure into a mechanism. This is called the **mechanism condition**.
- The structure will continue to resist loads by forming plastic hinges until a **collapse load** is reached, after which no further increase in load can be sustained.

# **Methods of Plastic Analysis:**

# **1. Upper Bound Theorem (Kinematic Method):**

- The upper bound theorem states that the true collapse load is less than or equal to any assumed collapse load derived from an assumed collapse mechanism. If a mechanism is postulated and the corresponding load is calculated, the actual collapse load will be less than or equal to this value.
- This method involves assuming a possible failure mechanism (such as the formation of plastic hinges) and calculating the external work done by the applied loads and the internal work done by the moments in the plastic hinges.
- By equating the internal and external work, the collapse load can be estimated.

# 2. Lower Bound Theorem (Static Method):

• The lower bound theorem states that the collapse load is greater than or equal to any load that can be supported by a statically admissible distribution of internal forces (moments, shear forces, etc.) that satisfy equilibrium and do not exceed the plastic moment capacity of any section.

# **Deflection of Beams**

### Mr. D.Jeyakumar

**Engineering soils** refer to the types of soils that are studied, classified, and managed in civil engineering projects, particularly in construction, foundation design, and earthworks. Engineers evaluate the physical and mechanical properties of soils to determine their suitability for supporting structures like buildings, roads, and bridges.

### Key concepts in engineering soils include:

### **1. Soil Composition**

- Soil Types: Soils are classified into several types based on particle size:
  - Gravel: Coarse particles, good for drainage.
  - Sand: Fine but granular, stable when compacted.
  - Silt: Fine particles, often unstable when wet.
  - **Clay**: Very fine particles, prone to swelling and shrinking.
- Loam: A mixture of sand, silt, and clay, often considered ideal for foundations.

# 2. Soil Mechanics

- Shear Strength: The resistance of soil to sliding or shearing forces, critical for slope stability.
- **Compressibility**: How much soil compresses under load, impacting foundation settlement.
- **Permeability**: The ability of soil to transmit water, affecting drainage and erosion control.
- Plasticity: How easily soil can be molded or reshaped, particularly important for clay-rich soils.

### 3. Soil Classification

• Engineers use systems like the **Unified Soil Classification System (USCS)** or **AASHTO classification** to categorize soils based on their properties.

### 4. Geotechnical Investigation

- Before construction, engineers perform tests such as:
  - Standard Penetration Test (SPT): Measures soil resistance to penetration.
  - Cone Penetration Test (CPT): Measures soil resistance with a cone.
  - Soil Compaction Tests: Ensures the soil is compact enough to support structures.

### 5. Soil Stabilization

- Sometimes soil needs to be treated or modified to improve its properties for engineering purposes, like:
  - Chemical Stabilization: Using lime, cement, or other chemicals.
  - Mechanical Stabilization: Compacting or blending soils to improve strength.
- Understanding these aspects helps civil engineers make decisions on foundation design, road construction, and other infrastructure projects.
# CHAPTER 7 Buckling and Stability of Structures

Mrs. A.Belciya Mary

**Buckling** is a failure mode characterized by a sudden lateral or torsional deformation of a structural member subjected to compressive forces. It occurs when the compressive stress in the member reaches a critical value, causing the member to lose its stability. **Stability** of a structure refers to its ability to maintain its original configuration under applied loads without undergoing large, undesirable displacements or failure, such as buckling.

Buckling is crucial in the design of columns, beams, frames, and other compressive structural members, as it can occur at stress levels lower than the material's yield strength, making it a form of **instability failure**.

# **Types of Buckling:**

#### 1. Elastic Buckling:

• Occurs when a member buckles elastically without any permanent deformation. The member returns to its original shape if the load is removed before it reaches the material's yield stress.

#### 2. Inelastic Buckling:

• Happens when a member undergoes permanent deformation during buckling. This usually occurs in members with material yielding before or during buckling.

#### 3. Lateral-Torsional Buckling:

• A combination of lateral (sideways) and torsional (twisting) buckling. This mode is common in slender beams subjected to bending, especially when the beam is not restrained against rotation or lateral movement.

#### 4. Local Buckling:

• Occurs in thin-walled sections, where only a portion of the structure (such as a flange or web) buckles, rather than the entire member. This is commonly seen in steel plates or hollow sections.

#### **Buckling Modes:**

- **Flexural Buckling**: Buckling that occurs due to bending in one plane (sideways buckling). This is the most common mode for long, slender columns.
- Torsional Buckling: Buckling that occurs due to twisting of the cross-section.
- **Flexural-Torsional Buckling**: A combination of bending and twisting, especially relevant for nonsymmetric cross-sections like L-shaped or T-shaped columns.

#### **Stability of Structures:**

Stability refers to a structure's ability to resist failure or large deformations under applied loads. Stability can be assessed in terms of **geometrical stability** and **material stability**.

#### **Buckling in Beams:**

**Lateral-torsional buckling** is a common mode of failure in beams subjected to bending, particularly in long, slender beams with inadequate lateral support. When lateral support is insufficient, the beam can twist and buckle sideways when subjected to compressive stress in the top flange due to bending

# CHAPTER 8 Dynamic Analysis of Structures

#### Mr.S.Ramakrishnan

**Dynamic analysis** is the study of structures or systems under loads that change with time. Unlike static analysis, where loads are assumed to be constant or vary slowly, dynamic analysis considers the effects of time-dependent forces, such as vibrations, shocks, seismic events, or impacts. Dynamic analysis is crucial for structures or mechanical systems exposed to these loads because it helps predict how they will respond, ensuring safety.

#### Key Concepts in Dynamic Analysis:

#### 1. Degrees of Freedom (DOF):

- The number of independent movements or displacements a structure or system can experience. Each DOF represents a direction or mode in which the structure can move or deform (e.g., translation in x, y, z, or rotation).
- A single-degree-of-freedom (SDOF) system is the simplest model, representing a mass-springdamper system. A multi-degree-of-freedom (MDOF) system involves multiple interconnected masses and springs.

#### 2. Natural Frequency:

- The frequency at which a structure or system vibrates when disturbed from its equilibrium position without any external forces acting on it.
- Every structure has one or more natural frequencies depending on its mass and stiffness. At or near these frequencies, resonance occurs, significantly amplifying the response to dynamic loads.

#### 3. Damping:

- Damping is the energy dissipation in a structure due to internal friction, material properties, or other mechanisms that reduce the amplitude of oscillations.
- **Damped systems** do not continue vibrating indefinitely and slowly return to rest. Common damping models include viscous damping, Coulomb (friction) damping, and structural damping.

#### 4. **Resonance**:

- Resonance occurs when the frequency of an external force matches the natural frequency of the structure, leading to large oscillations and potential failure.
- Designing structures to avoid resonance is critical, especially for tall buildings, bridges, and machinery exposed to periodic loads.

### **Types of Dynamic Loads:**

#### 1. Harmonic Loads:

- Loads that vary sinusoidally with time, such as vibrating machinery or rotating equipment.
- These loads are typically modeled as periodic forces with a specific frequency and amplitude.

#### 2. Impact Loads:

- Sudden, short-duration loads, such as a hammer strike or a collision.
- These loads induce high-frequency vibrations and require specialized analysis to capture the rapid changes in forces and deformations.

#### 3. Seismic Loads:

- Earthquake forces cause ground motion, which induces dynamic loads on buildings and other structures.
- Seismic analysis evaluates how a structure responds to these forces, using techniques such as the response spectrum method or time-history analysis.

# CHAPTER 9 Internal Forces in Structures

#### Mr.D.Amal colins

**Internal forces** in structures are the forces that develop within structural elements (like beams, columns, or trusses) due to external loads or applied forces. Understanding these forces is crucial for ensuring that structures can safely support loads and maintain their stability. The primary internal forces in structures are:

#### 1. Axial Forces:

- **Tensile Forces**: Forces that stretch or pull apart a structural member. For example, a cable in a suspension bridge experiences tensile forces.
- **Compressive Forces**: Forces that push or compress a structural member. For instance, columns in a building experience compressive forces due to the weight of the structure above.

#### 2. Shear Forces:

• Forces that act parallel to the cross-section of a structural member. They cause the material to shear or slide across the cross-section. Shear forces are common in beams subjected to transverse loads.

#### 3. Bending Moments:

• Moments that cause a structural member to bend. They arise from external loads applied at a distance from the support or axis of the member. For instance, a beam supporting a load creates bending moments along its length.

#### 4. Torsional Moments:

• Moments that cause a structural member to twist about its longitudinal axis. Torsional moments are important in elements like shafts or beams subjected to twisting forces.

#### **Types of Structural Members and Their Internal Forces:**

#### 1. Beams:

- **Axial Force**: In beams, axial forces are generally negligible unless the beam is subjected to unusual loading conditions.
- **Shear Force**: Developed due to transverse loads applied to the beam. The shear force varies along the length of the beam and is critical for designing the beam's shear capacity.
- **Bending Moment**: Generated by transverse loads and varies along the length of the beam. The bending moment influences the beam's design, particularly its strength and stiffness.

#### 2. Columns:

- **Axial Force**: The primary internal force in columns. Columns are designed to resist compressive forces from the loads they support.
- Shear Force: Can be present but is typically less significant than the axial force in columns.
- **Bending Moment**: Columns may experience bending moments if they are not perfectly vertical or if they are subjected to lateral loads.

#### 3. Trusses:

- **Axial Force**: Each member of a truss typically experiences axial forces only. Trusses are designed so that each member carries either tensile or compressive forces.
- Shear Force: In trusses, shear forces are mainly present at the joints and are transmitted between the truss members.

# **CHAPTER 10** Types of Loads

#### Mrs. R Devi

**Types of loads** in structural engineering refer to the various forces or moments applied to a structure, which affect its stability, safety, and performance. Loads are categorized based on their origin, nature, and how they act on the structure. Understanding different types of loads helps engineers design structures to handle expected conditions effectively.

#### Main Types of Loads:

#### 1. Dead Loads:

- **Definition**: Permanent or static loads that do not change over time. They include the weight of the structure itself and any fixed attachments.
- Examples:
  - Weight of beams, columns, floors, and roofs.
  - Fixed equipment and finishes, such as partitions and built-in furniture.
  - Permanent fixtures and architectural elements.

#### 2. Live Loads:

- **Definition**: Variable or transient loads that change over time. They account for the occupancy and usage of the structure.
- Examples:
  - Occupants and furniture in buildings.
  - Vehicles and equipment in parking structures and bridges.
  - Snow accumulation on roofs.
  - Movable partitions or storage.

#### 3. Wind Loads:

- **Definition**: Loads exerted on a structure by wind pressure. Wind loads vary with wind speed, direction, and the shape of the structure.
- Examples:
  - Pressure on building facades and roofs.
  - Forces on tall structures like towers and bridges.
- **Considerations**: Wind loads are analyzed based on local wind speed data, building height, and exposure categories.

#### 4. Seismic Loads:

- **Definition**: Loads caused by ground shaking during an earthquake. They are crucial for designing earthquake-resistant structures.
- Examples:
  - Lateral forces due to ground movement.
  - Accelerations and displacements in buildings and bridges.
- **Considerations**: Seismic loads are determined using seismic design codes, which consider the seismic zone, building importance, and site conditions.

#### 5. Snow Loads:

- **Definition**: Loads due to snow accumulation on roofs and other horizontal surfaces. They vary with snow depth, density, and melting.
- Examples:
  - Snow load on flat or sloped roofs.
  - Drifting snow accumulation on buildings and structures.
- **Considerations**: Snow loads are assessed based on local climate data, roof geometry, and exposure.

#### 6. Live Loads (Moving Loads):

• **Definition**: Loads from vehicles or other moving loads that vary with time. They are significant for bridges and highways.

# DESIGN OF STEEL STRUCTURAL ELEMENTS

Edited by

**R.DEVI** 



**Design of Steel Structural Elements** Copyright © 2021 by R. Devi

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-095-8

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# TABLE OF CONTENTS

INTRODUCTION	-	03
CHAPTER 1 - Introduction and Allowable Stress Design	-	12
Ms. A Belciya Mary		
CHAPTER 2 - Connections in Steel Structures	-	20
Mrs. K Shanthi		
CHAPTER 3 - Tension Members	-	60
Dr.R. Sivasamandy CHAPTER 4 - Compression Members	-	85
D Jeyakumar		
CHAPTER 5 – Design of Flexural Members	-	125
K Shanthi CHAPTER 6- Type of Fasteners- Bolts Pins and welds in steel structural elements	-	137
S.Vennila		
CHAPTER 7- Types of Tension members and sections S.Vennila	-	145
<b>CHAPTER 8-</b> Design of single section and compound Angles D.Jeyakumar	-	160
<b>CHAPTER 9-</b> Design of column bases – Plate and Gusseted bases for Axially loaded columns S.Ravishankar	-	172
<b>CHAPTER 10-</b> Design of laterally supported Beams P.Venkateswaran	-	180
REFERENCES	-	190

#### CHAPTER 1 Introduction and Allowable Stress Design

#### Ms. A Belciya Mary

#### Introduction to Allowable Stress Design (ASD)

Allowable Stress Design (ASD) is a method used in structural engineering to ensure that materials are not subjected to stresses exceeding their material limits under normal service conditions. It is particularly relevant in the design of structures and components, such as beams, columns, and connections, where safety and durability are paramount.

In ASD, the maximum allowable stress in a material is determined based on its material properties, safety factors, and the intended use of the structure. This method focuses on maintaining stresses within acceptable limits, providing a straightforward approach for engineers to ensure safety and serviceability.

#### Key Principles of Allowable Stress Design

- 1. **Material Properties**: Engineers start with the material's yield strength, ultimate strength, and other relevant properties.
- 2. **Safety Factors**: To account for uncertainties in material properties, loading conditions, and other factors, a safety factor is applied. This reduces the allowable stress from the material's strength.

Allowable Stress =  $\frac{\text{Material Strength}}{\text{Safety Factor}}$ 

- 3. **Loading Conditions**: ASD considers various loads that a structure might encounter, including dead loads (permanent/static), live loads (temporary/dynamic), wind loads, seismic loads, and more.
- 4. **Design Checks**: The designed structural element is checked to ensure that the stresses under maximum expected loads do not exceed the allowable stress.
- 5. **Serviceability**: In addition to strength checks, serviceability criteria such as deflection limits are also considered to ensure user comfort and functionality.

#### Advantages and Disadvantages of ASD

#### Advantages:

- Simplicity: ASD is straightforward and easy to understand.
- Established: It has a long history and is widely accepted in engineering practice.
- Predictability: Provides clear guidelines for checking structural integrity.

#### **Disadvantages**:

- Less efficient for certain materials and structures compared to more modern methods like Load and Resistance Factor Design (LRFD).
- Can be overly conservative, leading to heavier structures or more material use than necessary.

#### **Applications of ASD**

ASD is commonly used in the design of:

#### CHAPTER 2 Connections in Steel Structures

#### Mrs. K Shanthi

#### **Connections in Steel Structures**

Connections in steel structures are critical components that join different structural elements, such as beams, columns, and braces. They play a vital role in transferring loads and forces between these elements, ensuring the overall stability and integrity of the structure. Proper design and detailing of connections are essential to achieve the desired performance under various loading conditions.

#### **Types of Connections**

#### 1. Bolted Connections:

- **Description**: Utilize bolts to connect structural members.
- Advantages: Quick to assemble and can be disassembled for repairs or modifications.
- **Common Types**:
  - **Slip-critical connections**: Designed to resist slip under load, typically used in high-strength bolting applications.
  - **Bearing connections**: Allow for some movement and are used in applications where slip is not a concern.

#### 2. Welded Connections:

- **Description**: Use welding to fuse steel elements together.
- Advantages: Provide continuous connections and high strength; often more rigid than bolted connections.
- Common Types:
  - **Fillet welds**: Used to join two members at right angles.
  - Groove welds: Used for thicker materials and provide stronger joints.

#### 3. Pinned Connections:

- **Description**: Allow for rotation between connected members, typically using pins or bolts.
  - Applications: Commonly used in trusses and other structures where flexibility is required.
- 4. Moment Connections:
  - **Description**: Designed to resist bending moments and maintain rigidity.
  - Applications: Often used in moment-resisting frames for seismic design.

#### **Design Considerations**

- 1. Load Transfer:
  - Connections must effectively transfer loads, including axial, shear, and moment loads, from one member to another.
- 2. Strength:
  - The design must ensure that the connection can withstand the applied loads without exceeding the material's allowable stress.
- 3. Stability:
  - Connections must maintain the stability of the structure under various loading conditions, including dynamic loads such as wind and seismic forces.

#### 4. Durability:

• Connections should be designed to resist corrosion and fatigue, especially in harsh environments.

#### CHAPTER 3 Tension Members

#### Dr.R. Sivasamandy

#### **Tension Members in Steel Structures**

Tension members are structural elements that primarily carry axial tensile loads. They are crucial in various applications, such as trusses, cables, and frames, where they help to resist forces and maintain the stability of the overall structure.

#### **Key Characteristics of Tension Members**

- 1. **Loading**: Tension members are subjected to pulling forces. The design must ensure they can withstand these forces without experiencing excessive elongation or failure.
- 2. **Material**: Typically made from steel or other high-strength materials, tension members are designed for optimal strength-to-weight ratios.
- 3. Cross-Section: Common shapes include:
  - Angles: L-shaped sections providing good strength in both directions.
  - Channels: U-shaped sections used for longer spans.
  - **Plates**: Flat sections that can be used in various configurations.
  - **Round bars**: Often used in cables or rod applications.

#### **Design Considerations**

- 1. Load Calculation:
  - Determine the maximum tensile load the member will experience, considering factors like dead loads, live loads, wind loads, and seismic loads.
- 2. Material Properties:
  - Use material strength values (yield strength, ultimate strength) to establish design criteria.

#### 3. Cross-Sectional Area:

• The area must be sufficient to resist the applied tensile load, calculated as:

$$A_t = \frac{N}{F_{tu}}$$

where N is the tensile load and  $F_{tu}$  is the ultimate tensile strength of the material.

#### 4. Buckling:

• While tension members generally do not buckle, slender elements may need to be checked for lateraltorsional buckling.

#### 5. Connection Design:

• Connections to other structural elements (like beams or nodes in trusses) must be designed to handle the tensile loads without compromising the integrity of the member.

#### 6. **Deflection**:

• Ensure that the member's deflection under load remains within acceptable limits for serviceability.

#### **Common Applications**

1. **Trusses**: Tension members are essential in truss design, providing support and stability in roofs, bridges, and towers.

#### **CHAPTER 4 Compression Members** D Jeyakumar

#### **Compression Members in Steel Structures**

Compression members are structural elements primarily subjected to axial compressive loads. They are critical in various applications, such as columns, struts, and braces, where they help to support loads and maintain the stability of structures.

#### **Key Characteristics of Compression Members**

- 1. **Loading**: Compression members carry loads that push down or inward, which can lead to buckling if not properly designed.
- 2. **Material**: Commonly made from steel, concrete, or composite materials, they must be designed to handle the compressive forces effectively.
- 3. Cross-Section: Various shapes are used, including:
  - **I-beams**: Offer high strength-to-weight ratios and good buckling resistance.
  - Square and rectangular tubes: Provide uniform strength in all directions.
  - Angles: Used in bracing and other applications where lateral support is needed.
  - Solid round bars: Used in specific applications where flexibility is needed.

#### **Design Considerations**

- 1. Load Calculation:
  - Determine the maximum compressive load based on the anticipated service conditions, including dead loads, live loads, wind loads, and seismic forces.

#### 2. Material Properties:

• Use relevant material strength values, such as yield strength and ultimate compressive strength, to inform design choices.

#### 3. Effective Length:

• The effective length of a compression member is critical in determining its buckling behavior. It depends on end conditions (fixed, pinned, or free).

#### 4. Buckling:

• Compression members are susceptible to buckling, a failure mode that occurs when the member bends under load. Euler's buckling formula is often used to predict critical buckling loads:

$$P_{cr} = \frac{\pi^2 EI}{\left(KL_{eff}\right)^2}$$

where  $P_{cr}$  is the critical load, E is the modulus of elasticity, I is the moment of inertia, K is the effective length factor, and  $L_{eff}$  is the effective length.

#### 5. Cross-Sectional Area:

• The area must be sufficient to resist the applied compressive load, calculated as:

$$A_c = \frac{N}{F_c}$$

where N is the compressive load and  $F_c$  is the compressive strength of the material.

#### **CHAPTER 5 Design of Flexural Members** K Shanthi

#### **Design of Flexural Members in Steel Structures**

Flexural members, such as beams and slabs, primarily resist bending moments due to applied loads. Proper design of these members is crucial to ensure structural integrity, serviceability, and safety.

#### **Key Concepts in Flexural Design**

#### 1. Loading Conditions:

• Flexural members experience various types of loads, including dead loads (permanent), live loads (temporary), and environmental loads (wind, seismic).

#### 2. Bending Moment:

• The bending moment (M) in a beam is a measure of the internal moment that induces bending, calculated based on the applied loads and support conditions.

#### 3. Shear Forces:

• Shear forces (V) act parallel to the cross-section and can lead to shear failure. Both bending moments and shear forces must be considered in the design.

#### 4. **Deflection**:

• The deflection of a flexural member under load must be limited to ensure serviceability, typically using allowable deflection limits based on the span and load conditions.

#### **Design Steps for Flexural Members**

#### 1. Determine Loads:

• Calculate the total loads acting on the member, including dead and live loads, and establish the load combinations as per relevant design codes (e.g., AISC).

#### 2. Calculate Bending Moments and Shear Forces:

• Use methods such as the moment distribution method or finite element analysis to determine maximum bending moments and shear forces at critical sections.

#### 3. Select Member Type and Size:

• Choose an appropriate cross-section (I-beam, channel, etc.) based on material availability and load requirements.

#### 4. Strength Check:

• **Flexural Strength**: Ensure the section can resist the bending moment using the following equation:

$$M_n = F_y \cdot S$$

where  $M_n$  is the nominal moment strength,  $F_y$  is the yield strength of the material, and S is the section modulus.

 $\circ$  Compare the required moment M to the available moment strength M<sub>n</sub>:

$$M \le M_n$$
 (Strength Limit State)

#### 5. Shear Strength Check:

• Check the shear capacity of the section using:

$$V_n = F_v \cdot A_v$$

#### CHAPTER 6 Type of Fasteners- Bolts Pins and welds in steel structural elements S. Vennila

When it comes to steel structural elements, several types of fasteners are commonly used, including bolts, pins, and welds. Each type has specific applications and advantages:

#### 1. Bolts

- Types of Bolts:
  - Structural Bolts: High-strength bolts used in steel connections.
  - Anchor Bolts: Used to attach structures to concrete.
  - Hex Head Bolts: Commonly used in various applications.
  - Applications: Used for joining structural members, providing high strength and allowing for disassembly.
- Advantages:
  - Easy to install and remove.
  - Can accommodate movements (thermal expansion, settlement).
  - Variety of sizes and grades available.

#### 2. Pins

- Types of Pins:
  - Clevis Pins: Used in hinges and linkages.
  - Cotter Pins: Used to secure other fasteners.
  - **Dowel Pins**: Align parts or provide location.
  - Applications: Often used in movable connections or temporary assemblies.
- Advantages:
  - Simplicity in design.
  - Allows for rotational movement in joints.
  - Easy to install and replace.

#### 3. Welds

- Types of Welds:
  - Fillet Welds: Commonly used for corner and edge joints.
  - Butt Welds: Used to join two pieces of metal end to end.
  - Plug and Slot Welds: For joining overlapping metal plates.
  - Applications: Provides a continuous joint for structural integrity, often used in beams, columns, and frames.
- Advantages:
  - Creates a strong, permanent bond.
  - No need for additional fasteners.
  - Can be used to create complex shapes and designs.

#### **Considerations in Selection:**

- Load Requirements: Different fasteners can handle different load capacities.
- Environmental Factors: Corrosion resistance may be a concern, influencing the choice of material and type.

#### CHAPTER 7 Types of Tension members and sections S. Vennila

Tension members are structural components that primarily resist tensile forces. They are commonly used in various applications, such as trusses, cables, and bridges. Here are the main types of tension members and sections:

#### **Types of Tension Members**

- 1. Rod or Bar Tension Members:
  - Solid Rods: Cylindrical members that carry axial tension.
  - **Bars:** Often rectangular or square in cross-section.
- 2. Cable Tension Members:
  - Wire Cables: Composed of multiple strands of wire, providing flexibility and high tensile strength.
  - Stranded Cables: Used in suspension bridges and similar structures.
- 3. Plate Tension Members:
  - Flat Plates: Used in applications like tension splices or as part of truss systems.
- 4. Composite Tension Members:
  - **Fiber-Reinforced Polymers (FRP):** Lightweight, high-strength materials used in modern structures.

#### **Common Sections Used for Tension Members**

- 1. Round Sections
  - Commonly used for rods and cables, offering uniform strength.
- 2. Square and Rectangular Sections:
  - Often used for bars; provides a good strength-to-weight ratio.
- 3. I-Beams and H-Beams:
  - Less common for pure tension, but can be used in trusses where members are subjected to both tension and compression.
- 4. Angle Sections:
  - L-shaped sections that can effectively resist tension, commonly used in trusses and frameworks.
- 5. Tee Sections:
  - T-shaped members that can also be used in tension applications.
- 6. Channel Sections:
  - C-shaped profiles that are sometimes used in trusses and other structural applications.

#### **Considerations for Design**

- Material Strength: Selection based on the required tensile strength and ductility.
- **Cross-Sectional Area:** Must be adequate to resist the applied loads.
- Length and Buckling: Consideration for slenderness ratios, especially for long members.
- Connections: How tension members are connected to other members is critical for load transfer.

Understanding these types and sections helps in selecting the appropriate tension members for structural applications.

#### CHAPTER 8 Design of single section and compound Angles

#### D.Jeyakumar

Designing single section and compound angles involves understanding their geometry and applications, particularly in engineering and construction. Here's a concise guide on both:

#### **Single Section Angles**

**Definition**: A single section angle is a basic structural element shaped like an "L," typically used in steel construction. The angle has two legs of unequal or equal lengths.

#### **Common Uses**:

- Support structures
- Bracing
- Frames

#### **Design Considerations**:

- 1. Material Properties: Know the yield strength and ultimate tensile strength of the material.
- 2. Load Analysis: Determine the types of loads (tension, compression, shear) the angle will bear.
- 3. Length and Thickness: Choose appropriate dimensions based on the load requirements and material strength.
- 4. Connections: Design connections (welds, bolts) to ensure stability and transfer loads efficiently.

#### Calculations:

- **Bending Strength**: Use the formula  $M=fy \cdot Z\gamma M = \frac{f_y \cdot Z\gamma M}{\gamma}$  where MMM is the moment, fyf\_yfy is the yield strength, ZZZ is the section modulus, and  $\gamma$  as a fety factor.
- Shear Strength: Calculate using  $V=fv \cdot Av\gamma V = \frac{f_v \cdot Av\gamma V}{\delta v}$ , where  $AvA_vAv$  is the shear area.

#### **Compound Angles**

**Definition**: A compound angle consists of two or more simple angles connected to form a more complex geometry. These are often used in structures requiring more strength and stability.

#### **Common Uses**:

- Frames with multiple load paths
- Complex trusses
- Architectural elements

#### **Design Considerations:**

- 1. Geometry: Understand the intersection and orientation of the angles. Use CAD software for precision.
- 2. Load Distribution: Analyze how loads will be distributed through the compound structure.
- 3. Connection Design: Ensure robust connections between angles, often using gusset plates or welds.
- 4. Deflection and Stability: Check for deflection under load and overall stability of the compound structure.

#### **CHAPTER 9**

#### Design of column bases - Plate and Gusseted bases for Axially loaded columns

#### S. Ravi Shankar

Designing column bases for axially loaded columns involves understanding the different types of base connections and their structural implications. Two common types are plate bases and gusseted bases. Here's an overview of both, including key considerations for their design.

#### **Plate Bases**

**Description:** A plate base is a flat steel plate that transfers the load from the column to the foundation. The plate can be square, rectangular, or circular, depending on the column's shape and the design requirements.

#### **Design Considerations:**

#### 1. Load Distribution:

- The plate should be sized to adequately distribute the axial load over the foundation.
- Ensure that the thickness of the plate is sufficient to avoid local buckling.

#### 2. Weld and Bolt Connection:

- The column is typically welded or bolted to the base plate.
- Design the welds or bolts to resist shear and moment forces that may occur due to lateral loads or eccentric loading.

#### 3. Base Plate Thickness:

• Calculate the required thickness using factors such as load, yield strength of the material, and allowable bearing pressure on the foundation.

#### 4. Anchor Bolts:

- Design the anchor bolts to resist uplift and shear forces.
- Consider the type of embedment and spacing to ensure adequate performance.

#### 5. Foundation Interaction:

• The interface between the base plate and the concrete foundation should be designed to ensure proper load transfer and prevent failure due to punching shear.

#### **Gusseted Bases**

**Description:** Gusseted bases use additional triangular plates (gussets) that connect the base plate to the column, providing greater rigidity and load distribution, especially for taller columns or those subjected to lateral forces.

#### **Design Considerations:**

#### 1. Gusset Size and Shape:

• The size and configuration of the gussets should be determined based on the load, column height, and lateral stability requirements.

#### 2. Stability:

• Gussets increase the stability of the column base by reducing the risk of buckling and increasing resistance to lateral forces.

#### 3. Load Transfer:

• Ensure that the gussets effectively transfer loads from the column to the base plate and subsequently to the foundation.

#### 4. Connection Detailing:

• Proper detailing of connections between the gussets, base plate, and column is crucial for overall structural integrity. Use welding or bolting as appropriate.

#### **CHAPTER 10** Design of laterally supported Beams

#### P. Venkateswaran

Designing laterally supported beams involves ensuring they can withstand various loads while maintaining stability and strength. Here's a concise overview of the design process:

#### **1. Understanding Beam Properties**

- Span Length: The distance between supports.
- Cross-Section: Common shapes include I-beams, rectangular, and T-beams.
- Material: Steel, concrete, or wood; properties vary with material.

#### 2. Load Considerations

- **Dead Loads:** Permanent/static loads (e.g., self-weight, fixtures).
- Live Loads: Variable loads (e.g., people, furniture).
- Environmental Loads: Wind, snow, seismic forces.

#### **3. Support Conditions**

- **Simply Supported:** Supported at two ends.
- **Continuous:** Supported by more than two points.
- **Cantilevered:** Fixed at one end, free at the other.

#### 4. Lateral Support

- Beams require lateral support to prevent buckling. This can be achieved through:
  - **Bracing:** Diagonal braces or cross-bracing.
  - Shear Walls: Vertical elements that provide stability.

#### **5.** Calculating Forces and Moments

- Use structural analysis to determine bending moments, shear forces, and deflections.
- Common methods include:
  - Simplified Hand Calculations: For basic designs.
  - Finite Element Analysis (FEA): For complex geometries and load conditions.

#### 6. Design for Bending and Shear

- Bending Stress: Use the formula  $\sigma=MS \otimes \sigma=SM$ , where MMM is the moment and SSS is the section modulus.
- Shear Stress: Use  $\tau = VA \tan = \frac{V}{A} \tau = AV$ , where VVV is the shear force and AAA is the area.
- Compare calculated stresses with allowable stresses from material specifications.

#### 7. Deflection Checks

- Ensure deflections are within limits (typically span/360 for live loads).
- Use formulas or tables from codes (e.g., AISC, ACI) for deflection calculations.

# SOIL MECHANICS LABORATORY

Edited by :

DR.R.SIVASAMANDY





**Soil Mechanics Laboratory** Copyright © 2021 by Dr.R. Sivasamandy

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-312-6

Publisher: PonnaiyahRamajayam Institute of Science and Technology (PRIST) Date of Publication  $:\!20.01.2021$ 

# **TABLEOF CONTENTS**

CHAPTER 1 – Determination of Specific gravity of soil solids	-	08
Dr.R. Sivasamandy CHAPTER 2 –Determination of Grain Size Distribution (SieveAnalysis)	_	11
Dr.R. Sivasamandy CHAPTER 3 – Determination of Liquid Limit and PlasticLimit	_	16
Dr.R. Sivasamandy		
CHAPTER 4 – Determination of Field Density	-	18
Dr.R. Sivasamandy		
CHAPTER 5 – Determination of Moisture Density Relationship	-	21
Dr.R. Sivasamandy		
CHAPTER 6 – Determination of Permeability of Soil	-	26
Dr.R. Sivasamandy		
CHAPTER 7 – Vane Shear Testin Cohesive Soil	-	29
Dr.R. Sivasamandy		
CHAPTER 8 – Determination of Moisture Content of Soil	-	32
Dr.R. Sivasamandy		
CHAPTER 9 – Determination of Field Density-Core Cutter Method	-	36
Dr.R. Sivasamandy		
CHAPTER 10 – Determination of Unconfined Compressionin Cohesive Soil	-	38
Dr.R. Sivasamandy		
REFERENCES	-	40

# **CHAPTER-1**

#### Determination of specific gravity of soil solids

#### Dr.R. Sivasamandy

### AIM

Todeterminethespecificgravityofsoil solids.

#### THEORY

Specific gravity of soil solid sistheratio of weight, in air of a given volume; of drysoil

solidstotheweightofequalvolumeofwaterat4°C.Specificgravityofsoilgrainsgives the property of the formation of soil mass and is independent of particle size. Specific gravityof soilgrainsisused incalculating void ratio,porosityand degree of saturation, by knowing moisture content and density. The value of specific gravity helps in identifying and classifying the soil type.

#### APPARATUSREQUIRED

- 1. Pycnometer
- 2. 450mmsieve
- 3. Weighingbalance
- 4. Oven
- 5. Glassrod
- 6. Distilledwater

#### PROCEDURE

- 1. Drythepycnometerandweighitwithitscap. (W1)
- 2. Takeabout200gmofovendriedsoilpassingthrough4.75mmsieveintothe pycnometer and weigh again (W<sub>2</sub>).
- 3. Addsufficientde-airedwatertocoverthesoilandscrewonthecap.
- 4. Shakethepycnometerwellandremoveentrappedairifany.
- 5. Aftertheairhasbeenremoved, fill the pycnometer with water completely.
- 6. Thoroughlydrythepycnometerfromoutsideandweighit (W<sub>3</sub>).
- 7. Cleanthepycnometerbywashingthoroughly.
- 8. Fill the cleaned pycnometer completely with water up to its top with cap screw on.
- 9. Weighthepycnometerafterdryingitontheoutsidethoroughly (W<sub>4</sub>).

### **CHAPTER-2** Determination of grain size distribution (Sieveanalysis)

#### Dr.R. Sivasamandy

#### AIM

Toconductsieveanalysisofsoiltoclassifythegivencoarsegrainedsoil.

#### APPARATUSREQUIRED

- 1. AsetofSieves4.75mm,2.36mm,1.18mm,0.60mm, 0.425mm,0.30 mm 0.15 mm 0.075mm including lid and pan
- 2. Tray
- 3. WeighingBalance
- 4. Sieve Shaker
- 5. Brush

#### PROCEDURE

- 1. Weigh500gmsofsoilsample,ofwhichgrainsizedistributionhastobestudied.
- 2. Cleanthesievesetsothatnosoilparticleswerestruckinthem.
- **3**. Arrange the sieves in order such that coarse sieve is kept at the top and the fine sieve is at the bottom. Place the closed pan below the finest sieve.
- 4. Takethesoilobtainedintothetopsieveandkeepthelidtoclosethetopsieve.
- 5. Position the sieve set in the sieve shaker and sieve the sample for a period of 10 minutes.
- 6. Separate the sieves and weigh carefully the amount of soil retained on each sieve, This is usually doneby transferring thesoilretained on eachsieve on a separate sieve of paper and weighing the soil with the paper.
- 7. EntertheobservationsintheTableandcalculatethecumulativepercentageof soil reained on each sieve.

#### **CHAPTER-3 Determination of liquidlimit and plasticlimit** Dr.R. Sivasamandy

### AIM

Todeterminetheliquidlimitandplasticlimitofthegivensoil sample

#### **APPARATUSREQUIRED:**

- 1) Measuringbalance
- 2) Liquidlimitdevice (Casagrandes)
- 3) Groovingtool
- 4) 425micron sieve
- 5) Glassplate
- 6) Satula
- 7) Mixingbowl
- 8) Washbottle
- 9) Mositurecontent bins
- 10) Dryingoven

#### **Procedureforliquidlimit:**

- About 120 gm of air dried soil from thoroughly mixed portion of material passing 425 micron I.S sieve is to be obtained.
- Distilled water is mixed to the soil thus obtained in a mixing disc toform uniform paste. The paste shall have a consistency that would require 30 to 35 drops of cup to cause closer of standard groove for sufficient length.
- A portion of the paste is placed in the cup of LIQUID LIMIT device and spread into portion with few strokes of spatula.
- Trimittoadepthof1cmatthepointofmaximumthicknessandreturnexcessof soil to the dish.
- The soil in the cup shall be divided by the firm strokes of the grooving tool along the diameter through the centreline of the followers that clean sharp groove of proper dimension is formed
- Lift and drop the cup by turning crank at the rate of two revolutions per second until the two halves of soil cake come in contact with each other for a length of about 1 cm by flow only.
- The number of blows required to cause the groove close forabout 1 cm shallbe recorded.

#### **CHAPTER-4 Determination of fielddensity (Sand Replacement Method)** Dr.R. Sivasamandy

#### AIM:

To determine the field density of soil at a given location by sand replacement method.

#### APPARATUS

- 1. Moisturecontent cups
- 2. Sandpouring Cylinder
- 3. Calibratingcan
- 4. Metaltraywithacentralhole
- 5. Drysand(Passingthrough600micronsieve)
- 6. Balance
- 7. Metaltray
- 8. Scrapertool
- 9. Glass plate

#### PROCEDURE

### CALIBRATIONOFSANDDENSITY

- Measuretheinternaldimensionsdiameter(d)andheight(h)ofthecalibratingcanand compute its internal volume V.
- Fill the sand pouring cylinder (SPC) with sand with 1 cm top clearance to avoid any spillover during operation and find its weight (W1)

# CHAPTER-5

# Determination of moisture density relationship (Proctorcompaction)

Dr.R. Sivasamandy

# AIM:

To determine Optimum Moisture Content and Maximum drydensity for a soil by conducting standard proctor compaction test

### **APPARATUSREQUIRED:**

1) Proctormouldhavingacapacityof1000ccwithaninternaldiameterof100mmand a height of 127.3 mm. The mould shall have a detachable collar assembly and a detachable base plate.

2) Rammer: Amechanical operated metal rammer having a 5.08 cm diameter face and a weight of 2.5 kg. The rammer shall be equipped with a suitable arrangement to control the height of drop to a free fall of 30 cm.

- 3) Sampleextruder.
- 4) Abalanceof15kgcapacity.
- 5) Sensitivebalance.
- 6) Straightedge.
- 7) Graduatedcylinder.
- 8) Mixingtoolssuchasmixingpan,spoon,towel,spatulaetc.
- 9) Moisturetins.

### **PROCEDURE:**

- Takeabout3kgofairdried soil
- Sievethesoilthrough20mmsieve.Takethesoilthatpassesthroughthesievefor testing.
- Take2.5kgofthesoilandaddwatertotitobringitsmoisturecontenttoabout4% in coarse grained soils and 8% in case of fine grained soils.
- Clean, dry and grease the mould and base plate. Weighthe mould with base plate. Fit the collar.
- Compactthewetsoilinthreeequallayersbytherammerwith25evenlydistributed blows in each layer.
- Remove the collar and trimoff the soil flush with the top of the mould. In removing the collar rotate it to break the bond between it and the soil before lifting it off the mould.
- Cleantheoutsideofthemouldandweighthemouldwithsoilandbaseplate.
- Remove the soil from the mould and obtain a representative soils ample from the bottom, middle and top for water content determination
- Repeattheaboveprocedurewith8,12,16and210% of watercontents for coarse grained soil and 14,18,22 and 26 % for fine grained soil samples approximately.
- The above moisture contents are given only for guidance. However, the moisture contents may be selected based on experience so that, the dry density of soil shows the increase inmoisture content. Each trial should be performed on a fresh sample.

# CHAPTE-6

# Determination of permeability of soil (Constant Head Method)

Dr.R. Sivasamandy

#### AIM:

Todeterminethecoefficientofpermeabilityofthesoilbyconductingconstanthead method. **APPARATUSREQUIRED** 

- 1. Permeabilityapparatuswithaccessories
- 2. Stop watch
- 3. Measuring jar

### PROCEDURE-ConstantHeadMethod

- Compact the soil into the mould at a given dry density and moisture content by a suitabledevice.Placethespecimencentrallyoverthebottomporousdiscandfilter paper.
- Placeafilterpaper, porous stone and washer on top of the soils ample and fix the top collar.
- Connectthestandpipetotheinletofthetopplate.Fillthestandpipewithwater.
- Connectthereservoir with water to the outlet at the bottom of the mould and allow the water to flow through and ensure complete saturation of the sample.
- Open the air valve at the top and allow the water to flow out so that the air in the cylinder is removed.
- When steady flow is reached, collect the water in a measuring flask for a convenienttime intervals by keeping the head constant. The constant head of flow is provided with the help of constant head reservoir
- Repeat the for three more different time intervals **OBSERVATIONS**

#### AND CALCULATIONS – Constant Head Method

Calculatethecoefficientofpermeabilityofsoilusingtheequation

$$K = QL/Ath$$

Where

K=Coefficientofpermeability

Q=Quantityofwatercollectedintimetsec(cc) t = Time

required (sec)

A=Crosssectionalareaofthesoilsample(sq.cm) h =

# CHAPTER-7

#### Vaneshear test in cohesive soil

Dr.R. Sivasamandy

#### AIM

Todeterminetheundrainedshearstrengthofthecohesivesoilusingvane shear.

#### **APPARATUSREQUIRED:**

- 1) Vaneshearapparatus.
- 2) Specimen.
- 3) Specimencontainer.
- 4) Callipers.

#### **PROCEDURE:**

- Prepare two or three specimens of the soil sample of dimensions of at least 37.5 mm diameter and 75 mm length in specimen.(L/D ratio 2 or 3).
- Mount the specimen container with the specimen on the base of the vane shear apparatus. If the specimen container is closed at one end, it should be provided with a hole of about 1 mm diameter at the bottom.
- Gently lower the shear vanes into the specimen to their full length without disturbing the soil specimen.
- Thetopofthevanesshouldbeatleast10mmbelowthetopofthespecimen.Note the readings of the angle of twist.
- Rotate the vanes at an uniform rate say 0.10/s by suitable operating the torque application handle until the specimen fails.
- Notethefinalreadingoftheangleoftwist.
- Findthevalueofbladeheightandwidthincm.

# **CHAPTER-8**

#### Determination of moisture content of soil

Dr.R.Sivasamandy

#### AIM

Todeterminethemoisturecontent(watercontent)ofagivensoilsample

#### APPARATUSREQUIRED

- 1. Containers
- 2. Balanceofsufficient sensitivity.
- 3. Hot Oven
- 4. Desiccators.

#### PROCEDURE

- Clean, dry and weighthe container withlid.
- Take the required quantity of the soil specimen in the container and weigh with lid.
- Maintainthetemperatureoftheovenbetween105°Cto110°Cfornormalsoilsand 600°C to 800°C for soils having loosely bound hydration water or/and Organic matter.
- Dry the sample in the oven till its mass becomes constant. In normal conditions the sample is kept in the oven for not more than24hours.
- After drying remove the container from the oven, replace the lid and cool in the desiccators.
- Weighthedrysoilinthecontainerwithlid.

#### **OBSERVATIONS**

S.No.	DeterminationNo.	1	2	3
1	ContainerNo.			
2	Massofcontainerwithlid.W1(gm)			
3	Massofcontainerwithlid+wetsoil,W2(gm)			
4	Massofcontainerwithlid+drysoil,W3(gm)			
5	Massofwater,Ww=W2-W3(gm)			
6	Massofdrysoil,Ws=W3-W1(gm)			
7	Moisturecontent,W=W2-W3/W3-			
	W1x100,(%)			

#### **RESULT:**Themoisturecontentofthegivensoil sample =

# **CHAPTER -9**

#### **Determination of field Density-Core cuttermethod** Dr.R. Sivasamandy

#### AIM:

Todeterminethefielddensityofsoilatagivenlocationbycorecutter method

#### APPARATUS

- 1. Cylindricalcorecutter
- 2. Steel rammer
- 3. Steeldolly
- 4. Balance
- 5. Moisturecontent cups

#### PROCEDURE

#### CORECUTTER

- Measure the height (h) and internal diameter (d) of the core cutter and apply grease to the inside of the core cutter.
- Weightheemptycorecutter(W1).
- Cleanandleveltheplacewheredensityistobedetermined.
- Drive the core cutter, with a steel dolly on itstop in to the soil to itsfull depth with the help of a steel rammer.
- Excavate the soil around the cutter with a crow bar and gently lift the cutter without disturbing the soil in it.
- Trim the top and bottom surfaces of the sample and clean the outside surface of the cutter.

#### RESULT

- 1. Dryunitweightofthesoil =
- 2. Wetunitweightofthesoil=
- **3.** Voidratioofthesoil =
- 4. Porosityofthesoil=
- 5.Degreeofsaturation=

# **CHAPTER-10**

# Determination of unconfined compression incohesive soil

Dr.R. Sivasamandy

### AIM

Todetermine theshearing strength of the cohesive soil. **APPARATUSREQUIRED:** 

- 1) Loadingframeofcapacityof2t, with constant rate of movement.
- 2) Provingring.
- 3) Soil trimmer.
- 4) Frictionlessendplatesof75mmdiameter(Perspexplatewithsilicongrease coating).
- 5) Evaporatingdish(Aluminumcontainer).
- 6) Soilsampleof75mm length.
- 7) Dialgauge(0.01mmaccuracy).
- 8) Balanceofcapacity200gandsensitivitytoweigh0.01g.

9) Oven, thermostatically controlled within terior of noncorroding material tomaintain the temperature at the desired level. What is the range of the temperature used for drying the soil.

- 10) Sampleextractorandsplit sampler.
- 11) Dialgauge(sensitivity0.01mm).

12) Verniercaliperstofindoutthediameterandlengthofthe specimen.

#### **PROCEDURE:**

• In this test, a cylinder of soil without lateral support is tested to failure in simple compression, at a constant rate of strain. The compressive load per unitare are quired to fail the specimen as called Unconfined compressive strength of the soil.

#### TESTPROCEDURE

- Taketwofrictionlessbearingplatesof75mmdiameter.
- Placethespecimenonthebaseplateoftheloadframe(sandwichedbetweenthe end plates).
- Placeahardenedsteelballonthebearingplate.
- Adjust the centerline of the specimen such that the proving ring and the steel ball are in the same line.
- Fixadialgaugetomeasuretheverticalcompressionofthe specimen.
- Adjustthegearpositionontheloadframetogivesuitablevertical displacement.
- Startapplyingtheloadandrecordthereadingsoftheprovingringdialand compression dial for every 5 mm compression.

# BUSINESS COMMUNICATION

# Edited by

DR. M BALAMURUGAN



**Business Communication** 

Copyright © 2021 by Dr. M. Balamurugan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-236-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

CHAPTER-I Introduction of Business Communication	1
Dr. M. Balamurugan	
CHAPTER –II	21
Communication codes and meaning	
Dr T J Jayasholan	
CHAPTER –III Intercultural communication	
Dr. S. Venkatesh	
CHAPTER –IV Organizational culture and communication	47
Dr K G Selvan	
CHAPTER -V Organizational structure and communication	51
Dr. S. Venkatesh	
CHAPTER -VI Analysing communication	62
K.Sasikumar	
CHAPTER –VII. Interpersonal skills in action: communicating face to face	79
Dr K G Selvan	
CHAPTER -VIII	80
Dr T J Jayasholan	
CHAPTER -IX Understanding organizational change	93
K. Sasikumar	
CHAPTER –X Building effective teams	102
K. Sasikumar	

# TABLE OF CONTENTS

#### CHAPTER -I

#### Introduction of Business Communication Dr. M. Balamurugan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **1. Upward communication**

Like the name suggests, upward communication deals with any interaction that travels up the hierarchy of your business. The most common example is when a direct report communicates to a supervisor or manager. Another instance is when a manager reaches out to directors or high-level executives.

This type of business communication allows upper management to stay informed about what is happening with the company. It also provides employees with the opportunity to ask questions, raise concerns, or make suggestions.

#### 2. Downward communication

Downward communication refers to the exchange of information from the top of an organization down to lower levels of the organizational hierarchy. Although the direction is the opposite of upward communication, these two types often work hand in hand. Examples of downward communication include when a manager relays instructions to a direct report in an email or an executive communicates business goals to a manager. This type of communication allows management to distribute information, delegate responsibilities, and enforce standards.

#### **3.** Lateral communication

When employees communicate across departments or with peers of equal rank in the organization, this is considered lateral or horizontal communication. Lateral communication examples include a coworker calling another coworker or a team of managers deliberating a potential new hire.

The purpose of lateral communication is to foster collaboration and coordination in an organization. Without productive lateral communication, an organization may fail to reach their business goals.

#### 4. External communication

External communication deals with the exchange of information from within the organization to parties outside of the organization. For example, a team within your company may send over a proposal for an outreach campaign to a client or your organization may issue a press release to promote a new product or service.

This also includes any electronic communication that occurs through a website, email, or social media from your organization to consumers, clients, or other parties. Your organization's external communication should be a top priority because it manages your business's reputation and outside relationships.

# CHAPTER -II Communication codes and meaning Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Communication Involves Three Components:**

- 1. Verbal Messages the words we choose
- 2. Paraverbal Messages how we say the words
- 3. Nonverbal Messages our body language

#### **Verbal Messages**

Our use of language has tremendous power in the type of atmosphere that is created at the problem-solving table. Words that are critical, blaming, judgmental or accusatory tend to create a resistant and defensive mindset that is not conducive to productive problem solving. On the other hand, we can choose words that normalize the issues and problems and reduce resistance. Phrases such as "in some districts, people may . . .", "it is not uncommon for . . ." and "for some folks in similar situations" are examples of this.

Sending effective messages requires that we state our point of view as briefly and succinctly as possible. Listening to a rambling, unorganized speaker is tedious and discouraging - why continue to listen when there is no interchange? Lengthy dissertations and circuitous explanations are confusing to the listener and the message loses its concreteness, relevance, and impact.

#### **Paraverbal Messages**

Paraverbal communication refers to the messages that we transmit through the tone, pitch, and pacing of our voices. It is how we say something, not what we say. Professor Mehrabian states that the paraverbal message accounts for approximately 38% of what is communicated to someone. A sentence can convey entirely different meanings depending on the emphasis on words and the tone of voice. For example, the statement, "I didn't say you were stupid" has six different meanings, depending on which word is emphasized.

#### **Nonverbal Messages**

The power of nonverbal communication cannot be underestimated. In his book, Silent Messages, Professor Albert Mehrabian says the messages we send through our posture, gestures, facial expression, and spatial distance account for 55% of what is perceived and understood by others. In fact, through our body language we are always communicating, whether we want to or not.

# CHAPTER -III Intercultural communication Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### Definition

Intercultural communication can be defined as the exchange of information on various levels of understanding between individuals of diverse cultural backgrounds. Achieving maximum efficiency in communication with others requires an understanding of cultural distinctions.

#### **Cross-cultural communication**

Cross-cultural communication is an exchange of ideas and information between individuals from different cultural backgrounds. It encompasses a variety of factors, such as nationality, working style, age, race, sexual orientation, and gender. Through this process, individuals are able to create and share meaning. Cross-cultural comparison involves examining two or more distinct cultures for comparison, which can help provide insight into how symbols vary between cultures.

Membership in a co-culture shapes the way its members communicate their perspectives. Individuals are expected to communicate in accordance with the customary practices and social conventions of the destination they are visiting. For example, the gesture of a thumbsup sign is considered offensive in Japan, Indonesia, and Latin America.

#### Intercultural communication

Intercultural communication is a symbolic, interpretive, transactional, and contextual process that enables people from different cultures to create shared meanings. Through intercultural communication, individuals from different cultures can learn to understand and appreciate one another's perspectives, values, and beliefs. It is essential for understanding the differences in how people from various cultures communicate, behave, and perceive the world around them.

#### **Social conventions**

Social conventions in intercultural communication refer to the implicit rules, values, beliefs, and customs that influence social behavior without being explicitly stated. These conventions can vary between cultures and can have a significant impact on communication styles and expectations. For example, facial expressions, personal space, and the dominant culture all play a role in determining how we communicate with one another. Additionally, life experiences, cultural values, and understanding of cultural differences shape how individuals interact.

#### **Barriers and Challenges**

When engaging in intercultural communication, there are a variety of obstacles and challenges that can arise.People from different cultural backgrounds may have different life experiences and expectations that can lead to misunderstandings and conflict. These obstacles can be further complicated by language barriers, unfamiliar cultures, and psychological barriers such as anxiety, fear, or negative self-image.
#### CHAPTER -IV Organizational culture and communication Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

The concept of organizational knowledge has many dimensions, related to contexts of interaction and the abovementioned challenges. There is, however, a general and close conceptual link between the concept of organizational knowledge and organizational culture. The latter concept has been defined in various ways in the literature, but the majority of definitions fall under Schein's famous short characterization of organizational culture as 'This is how we do it here'.

Understood like this, the concept denotes what people actually do in an organization, not necessarily what they say they do or what written information to the external world outside the organization describe what people in the organization do. Schein's more precise definition is well known:

The culture of a group can now be defined as a pattern of shared assumptions learned by a group as it solves its problems of external adaption and internal integration, which has worked well enough to be considered valid and, therefore, to be taught to new members as the correct way to perceive, think and feel in relation to those problems.

The key concept here is shared practices. Organizational culture is manifested in collective action, it is action guiding, and it is often implicit: It cannot be explicated as surface knowledge or norms in collective practices. Just like implicit knowledge, as I explained this concept above, organizational culture is non-conceptualized and essentially tied to actions that are not directly governed by formal frameworks.

Schein calls this dimension of organizational culture for 'tacit, taken-for-granted, underlying assumptions', and he argues that 'unless you dig down to the level of the basic assumptions, you cannot really decipher [the culture] ... On the other hand, if you find some of those basic assumptions and explore their relationship, you are really getting at the essence of the culture and can then explain a great deal of what goes on'.

Consequently, one might say that if an organization expresses a certain external profile to the outside world—for instance to a group of clients or stakeholders—but nevertheless is governed by very different basic assumptions on the 'inside', then the main identifier of the organizational culture are those 'inside' assumptions, not the descriptions that are communicated externally. And the culture is a substantial driving force; it is a main 'engine' in the organization. Cultural practices have a large impact on how organizations actually work and how they function economically and reach performance goals.

#### CHAPTER -V

#### Organizational structure and communication Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

**Functional Structure** - The first and most common is a functional structure. It's also referred to as a bureaucratic organizational structure. It breaks up a company based on the specialization of its workforce. Most small-to-medium-sized businesses implement a functional structure.

**Divisional or Multidivisional Structure-** This type is common among large companies with many business units. It's called the divisional or multidivisional (M-Form) structure. A company that uses this method structures its leadership team based on the products, projects, or subsidiaries it operates. Johnson & Johnson is a good example of this structure. The company has thousands of products and lines of business.

**Team-Based Structure-** Team-based organizations segregate into close-knit teams of employees that serve particular goals and functions, similar to divisional or functional structures. Each team is a unit that contains both leaders and workers, however.

**Flat (Flatarchy) Structure -**Flatarchy, also known as a horizontal structure, is used among many startups. It flattens the hierarchy and chain of command as the name implies. It gives its employees a great deal of autonomy. Companies that use this type of structure have a high speed of implementation.

**Matrix Structure** -Firms can also have a matrix structure. This is the most confusing and the least used. It matrixes employees across different superiors, divisions, or departments. An employee working for a matrixed company may have duties in both sales and customer service.

**Circular Structure -** Circular structures are hierarchical but they're said to be circular because they place higher-level employees and managers at the center of the organization with concentric rings expanding outward. They contain lower-level employees and staff. Organizing this way is intended to encourage open communication and collaboration among the ranks.

**Network Structure -** The network structure organizes contractors and third-party vendors to carry out certain key functions. It features a relatively small headquarters with geographically dispersed satellite offices along with key functions outsourced to other firms and consultants.

#### CHAPTER -VI

#### Analysing communication K.Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

The four communication archetypes are: Analytical, Intuitive, Functional, and Personal. Understanding the subtle differences between these various communication styles will help you to adjust and align communications at your company.

#### 1. Analytical

An analytical communicator values hard facts, data and logic. They embrace thoroughness and attention to detail. For the analytical communicator, details are key, they want specifics. It's not enough to say "we had a great first quarter, sales were up significantly." They will want to know the real figures. How much were sales up, did those figures meet or exceed expectations, how did each region perform, etc. The devil is in the details for analytical communicators.

#### 2. Intuitive

Intuitive communication focuses on the big picture. For the intuitive communicator, finer details and the inner workings of a plan are secondary to the end result. They enjoy blue sky or big picture thinking and are comfortable with unconventional methods and ideas. Intuitive communicators work and think fast; they get to the point quickly and expect others to do the same.

#### **3. Functional**

Functional communicators take things step by step. Like the analytical communicator, they value details but are also interested in the process, progress, and timelines. In fact, every aspect of a plan is of great importance to them. Their communication style includes precise information and details. Charting a process or plan from beginning to end, and sparing no details.

#### 4. Personal

Personal communicators place great emphasis on connection and feelings. They use emotive language and communicate in a way that allows them to connect with people and find out how they feel. Active listening is a key trait of the personal communicator. They can quickly pick up on others' emotions or feelings about particular plans or ideas.

#### CHAPTER -VII

#### Interpersonal skills in action: communicating face to face Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **1**. Demonstrate importance

This method of communication is one of the most effective ways to make sure important information gets through to employees. Being there in person (even if virtually over video) also tells your audience they are important to you and the issue you are discussing is worth your time and theirs.

#### 2. Interpret thoughts and feelings

One of the advantages of face-to-face communication is you that can see and respond to people's reactions – like facial expressions and body language – and their tone of voice. Leaders also have the chance to show they care by asking probing questions and actively listening to understand the audience's perspective.

#### 3. Enhance credibility and trust

After years of working apart (in some cases) and through challenges associated with crisis and change, leaders need to continue to be focused on building employee trust to be effective. Face-to-face interaction allows you to check in, share important information, such as your strategy, explain it clearly, and answer questions honestly – even the tough ones.

#### 4. Build relationships

Interacting directly with other leaders, managers, and employees through face-to-face communication helps you create shared experiences that can enhance future communication. It also helps create a camaraderie that is the basis of effective working relationships and increases your likelihood of success across the organization.

#### 5. Gather feedback

Meeting in person (whether virtually or in the same room) helps employees feel valued and gives them a chance to contribute input to organizational strategies and communication. Use face-to-face communication as an opportunity to gather feedback, confirm people's understanding of critical issues, and identify gaps in understanding. When in a group setting, people will also piggyback off each other's ideas and innovate, which can be a much more effective way to get feedback and ideas – depending on the topic – than through a feedback submission inbox, or other one-way mechanisms.

#### 6. Address sensitive issues

When communicating face-to-face, leaders demonstrate respect for employees and a commitment to a successful outcome when navigating a sensitive issue. Meeting face-to-face shows you care. The last few years have served no shortage of tough topics and leaders have learned that communicating face-to-face, vs. hiding behind a memo, shows they care and provides an opportunity for parties to ask questions, clarify understanding, and more.

#### CHAPTER -VIII Meetings and presentations Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1. Silence voices in the head

Some voices in the head can damage one's confidence, suggesting that the presentation is boring, embarrassing, and not good enough. Such voices of insecurity only make the presenter self-conscious. The important thing is to identify and dismiss them as lies. Practising this attitude will eventually lead to more confident presentations.

#### 2. Focus on the point, rather than oneself

Some people are so afraid of public humiliation that they end up messing up and embarrassing themselves. They forget the meeting is not about them but the matter in question.

The best approach is to stop perceiving the presentation as a contest or a source of judgement. Shifting the attention from oneself to the main objective eliminates the anxiety.

#### 3. Turn anxiety into excitement

Nervousness and excitement are closely related. Proficient business presenters see themselves as excited rather than nervous moments before getting on stage.

#### 4. Use notes as shorthand reminders

Notes should guide the presenter on what to cover and in what sequence. The points should be used as support, not scripts. It is about delivering the ideas, not reading them.

#### 5. Authenticity

Many business presenters mess things up by trying to fake things. An audience will respond best to authentic behaviour, even if it's not perfect. By contrast, a presenter who sounds artificial damages their reputation, breaks the connection, and reduces audience engagement.

#### 6. Volume

Being audible enough is critical to passing a powerful message during a meeting. Loud voice projects confidence, authority, and leadership.

#### 7.Body language

Business leaders use body language to their advantage. This is a preeminent visual that the audience needs to see. It is how the body communicates non-verbal, i.e., through posture, gestures, stance, and facial expressions.

#### CHAPTER -IX Understanding organizational change K. Sasikumar, Department of Management,

#### Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1. Strategic change

Organizations implement strategic changes to their business to achieve goals, boost competitive advantage in the market, or respond to market opportunities or threats. A strategic change includes making changes to the business's policies, structure, or processes. The upper management and the Chief Executive Officer often bear the responsibility for strategic change.

#### 2. People-centric organizational change

While all changes affect people, people-centric types of organizational change include instituting new parental leave policies or bringing in new hires. When implementing a people-centric change, the leadership must bear in mind that employees will naturally resist change. A people-centric change requires transparency, communication, effective leadership, and an empathetic approach.

#### 3. Structural change

Structural changes are changes made to the organization's structure that might stem from internal or external factors and typically affect how the company is run. Structural changes include major shifts in the management hierarchy, team organization, the responsibilities attributed to different departments, the chain of command, job structure, and administrative procedures.

#### 4. Technological change

The increasing market competition and constantly evolving technology lead to technological change within organizations. Technology change often involves introducing new software or systems to improve business processes through SaaS change management. However, technology project goals are often improperly defined and poorly communicated, which scares and frustrates your employees and ultimately leads to resistance.

#### 5. Unplanned change

Unplanned change is defined as a necessary action following unexpected events. An unplanned change cannot be predicted but can be dealt with by effective change management.

#### 6. Remedial change

Remedial changes are reactionary. This type of change occurs when a problem is identified, and a solution needs to be implemented. As these changes are designed to address an issue; they call for immediate action.

#### CHAPTER -X

#### Building effective teams K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### Capability

Think of capability as the diverse skill set that each team member brings to the table. It's about recognizing and leveraging the unique strengths and talents within your team. A well-rounded team contributes to the overall performance of an organization, so a team with a variety of capabilities can tackle challenges innovatively and efficiently. Much of team effectiveness hinges on how capable each member is.

#### Cooperation

At the heart of team effectiveness is cooperation. It's the engine that propels a team forward.

Cooperation fosters a culture of collaboration and mutual support within your team, ensuring that everyone works together harmoniously. When team members cooperate effectively, they can navigate complex tasks more easily and achieve their goals with consistency and in shorter time frames. **Coordination** 

Coordination within a team is vital for success and team effectiveness. It involves managing tasks,

timelines, and resources to ensure that everything runs smoothly. Effective coordination ensures that even

when faced with challenging situations, your team remains on course. **Communication** 

Clear and open communication is the lifeblood of team effectiveness. It's all about conveying thoughts, ideas, and information effectively. Effective communication not only prevents misunderstandings and conflicts but also promotes the free flow of information and fosters a sense of unity within the team.

#### Cognition

In the context of team effectiveness, think of cognition as collective thinking. It involves harnessing the combined intelligence and expertise of your team to solve complex problems. By encouraging a

culture of shared learning and growth, teams can tackle intricate challenges with confidence. **Coaching** 

Effective leadership and coaching within a team are key to successful team effectiveness. It's about nurturing talent, providing guidance, and setting the right course. Strong leadership can influence

team dynamics, driving performance and success. **Conditions** 

The conditions under which a team operates play a significant role in team effectiveness. This includes aspects like the physical workspace, access to technology, and work-life balance. Creating the right conditions ensures that your team can perform at its best.

## MARKETING MANAGEMENT

## EDITED BY M.TAMIZHSELVAN

Marketing Management

Copyright © 2021 by M. Tamizhselvan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-614-1

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

CHAPTER-I
M. Tamizhselvan
CHAPTER –II
Dr T J Jayasholan
CHAPTER –III
Dr. S. Venkatesh
CHAPTER –IV4 Scope of Marketing
Dr K G Selvan
CHAPTER -V
Dr. S. Venkatesh
CHAPTER -VI
K.Sasikumar
CHAPTER –VII
Dr K G Selvan
CHAPTER -VIII
Dr T J Jayasholan
CHAPTER -IX
K. Sasikumar
CHAPTER –X10
Product Polices
K. Sasikumar

#### TABLE OF CONTENTS

#### CHAPTER -I Introduction to Marketing Management M. Tamizhselvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

**Marketing management** is 'the art and science of choosing target markets and getting, keeping, and growing customers through creating, delivering, and communicating superior customer value'. It relies heavily on designing the organizations offering in terms of the target markets needs and desires and using effective pricing, communication and distribution to inform, motivate and service the market. Marketing management is concerned with the chalking out of a definite programme, after careful analysis and forecasting of the market situations and the ultimate execution of these plans to achieve the objectives of the organisation.

With the rise in incomes of the people in general, the overall demand for all types of goods has gone up. This has led to an increase in production of various types of consumer goods. With the relaxation of licensing regulations and import controls and the entry of multinationals in many fields, there has been a sea change in the competitive environment among the firms.

A marketing manager has to formulate marketing strategic plans in order to sustain in such a competitive market. Thus, the success or failure of a business now depends upon how effectively the marketing functions are performed.

The course will focus on the working knowledge of different concepts of Marketing and their use in managerial decisions so as to enable to formulate an effective marketing programme as mentioned below:

1. Marketing: Concept, Nature, Functions & Importance. Selling Vs. Marketing. Marketing Environment: Nature, Types & Strategies to deal with Internal and External (Micro and Macro) Marketing Environment. Ethical and Social Responsibilities of Marketing Manager.

2. Marketing System: Types of Marketing, Marketing Information System; Definition and Components. Marketing Research; Definition Objective, Process & Significance

3. Consumer Behavior: Factors Influencing the Consumer Behavior, Consumer Buying Process, Buying Motives, Consumer Market in India.

Market Segmentation: Concept, Importance, and Basis. Target Market Selection, Market Positioning: Concept & Importance. Market Repositioning. Product Differentiation Vs Market Segmentation. Contemporary Issues In Marketing.

#### CHAPTER -II Difference between Selling and Marketing Dr T J Jayasholan , Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

Selling	Marketing		
Definition			
The selling theory believes that if companies and customers are dropped detached, then the customers are not going to purchase enough commodities produced by the enterprise. The notion can be employed argumentatively, in the case of commodities that are not solicited.	The marketing theory is a business plan, which affirms that the enterprise's profit lies in growing more efficient than the opponents, in manufacturing, producing and imparting exceptional consumer value to the target marketplace.		
Related to			
Constraining customer's perception of commodities and services.	Leading commodities and services towards the consumer's perception.		
Beginning point			
Factory	Marketplace		
Concentrates on			
Product	Consumer needs		
Perspective			
Inside out	Outside in		
Business Planning			
Short term	Long term		
Orientation			
Volume	Profit		
Cost Price			
Cost of Production	Market ascertained		

#### CHAPTER -III Process of Marketing Management. Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Process of Marketing Management**

**1. Set Marketing Objectives:** Establishing marketing objectives is the initial step in effectively managing marketing efforts. These objectives must align with the overall mission of the brand or organisation. By setting clear objectives, businesses gain insight into which markets to focus their research on.

**2. Analyse Marketing Opportunities:** After objectives have been established, the next crucial step is to identify marketing opportunities. This involves analyzing potential target markets to determine the most suitable one for the brand. A thorough SWOT analysis is conducted to gain a better understanding of the company's current situation, including any perceived limitations. This step is essential for developing a successful marketing strategy.

**3. Research and Select Marketing Targets:** Successful businesses must have a thorough understanding of market attractiveness and how to properly measure it, to select the most effective market. A key aspect of this evaluation is market segmentation, which allows for a more streamlined assessment. Subsequently, product positioning strategies are implemented in each market.

**4. Design Marketing Strategies:** This strategy outlines how a brand plans to approach the target market and includes crucial decisions about the marketing mix, expenditures, and distribution. Within this step, important considerations are made regarding the product, place, price, and promotion (also known as the 4Ps).

**5. Plan Marketing Programmes:** To achieve the overarching marketing goals of an organisation, the implementation of effective marketing programmes and strategies is essential. This pivotal stage encompasses a complex set of decisions, including the development of brand identity, product features, packaging, sales techniques, and distribution methods.

**6. Implement Marketing Programs:** Ultimately, it is the implementation stage that sets the plans into motion, making it a critical step in the marketing process. This involves taking deliberate action based on the previously established strategies and programmes. The first course of action is to carefully determine the when, where, and how of the implementation process. From there, the business must actively promote their products and highlight their unique features to entice potential customers. This multi-faceted approach can involve a variety of methods, including sales promotions, public relations efforts, and more.

**7. Control:** Controlling involves closely monitoring and assessing the effectiveness of marketing strategies. By implementing these strategies, businesses can evaluate their success in achieving company goals. This can be measured through factors such as sales figures, customer satisfaction, and feedback. Not only does controlling establish the current success of marketing tactics, but it also provides valuable insights for future strategies. It is essential to also evaluate competitors to identify potential areas for improvement and maintain a competitive edge. To better track progress, companies may utilize resources like sales force management systems.

#### CHAPTER -IV Scope of Marketing Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1. Study of Consumer Behaviour

The first and foremost role of marketing professionals is to be familiar with the expectations and expenditure patterns of the consumer in general. A thorough understanding of what they like to purchase? When do they do that? How much expense they are prepared to make for a novel item, what is their usual budget for the category of your product and so on.

#### 2. Identify Their Wants and Requirements

To ensure a streamlined product launch and satisfy the user demand, marketers first need to identify the pain points of the audience. A strategic approach to marketing requires a complete understanding of the consumer lifestyle. Only this allows you to place a product or service that amplifies or complements their life.

#### 3. Planning & Product Development

In this phase, the idea of the product is conceptualised. With the ideation of the product, it lies in the scope of marketing channels to determine the correct branding strategy that addresses consumer demands and desires.

#### 4. Pricing and Policy Determination

Marketing professionals may leverage various factors in the product development cycle to identify the correct pricing. It takes into consideration existing competition in the market and the expenditure pattern of the target audience. The strategy should help marketers determine attractive packaging and prices that encourage buyers.

#### 5. Distribution

Identifying the proper distribution channel for the product is vital to optimise your ROI. To ensure the desired amount of sales, the distribution line must ensure wider target group outreach at the minimum cost.

#### 6. Promotion

In this step, marketers can use a mix of online and offline marketing channels to promote the product or service. Based on the type of product/service, and its target group - a particular marketing channel might be more suitable than others.

#### 7. Consumer Satisfaction

Every product or service is created and distributed in the market with the end goal of satisfying the user's demand or making their life easier. Therefore, after market distribution, it is essential to get feedback from the clients on how the product is being received.

#### CHAPTER -V

#### Consumer Behaviour Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **1. Complex Buying Behavior**

Complex buying behavior is observed when consumers are highly engaged with their purchase and perceive significant differences among available brands. This behavior is typical in high-stakes buying situations involving expensive, infrequent, or risky decisions. Consumers undertake a thorough information search, meticulously comparing features, benefits, prices, and reviews to make an informed choice.

Example: Consider the process of buying a new car. The buyer recognizes the substantial investment and the differences in safety features, fuel efficiency, brand reputation, and price among available options.

#### 2. Dissonance-Reducing Buying Behavior

In scenarios where consumers are highly involved in the purchase process but see little differentiation among brands, dissonance-reducing buying behaviour is prevalent. Such situations often lead to buyer's remorse, where consumers feel uncertainty or regret after their purchase and seek confirmation that they made the right decision.

Example: Buying a mattress can often lead to dissonance-reducing buying behavior. Many options seem similar in comfort and price, making it difficult to discern the best choice.

Fact: A consumer psychology report indicates that engaging customers in a follow-up dialogue after purchases can reduce dissonance by up to 30%, significantly improving customer satisfaction and retention rates.

#### 3. Habitual Buying Behavior

Habitual buying behaviour involves decisions made with minimal consumer involvement and little differentiation perceived among product options. These decisions are typically driven by convenience or routine rather than brand loyalty or performance evaluation.

Example: The purchase of household staples like salt or sugar usually exhibits habitual buying behaviour. Consumers often choose a product out of habit, without spending time considering alternatives unless a disruption (such as a store out of stock) occurs.

#### 4. Variety-Seeking Buying Behavior

This type of behaviour is marked by low consumer involvement but a high perception of differences among brands. Consumers engaging in variety-seeking behaviour often switch brands not out of dissatisfaction but out of a desire for change or novelty.

Example: When buying snacks, a consumer might switch brands simply to try something new, despite being satisfied with their current choice. This behaviour is driven by the pursuit of diversity in their consumption experiences.

#### CHAPTER -VI Pricing Methods and Strategy. K.Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1. Value-based pricing

With value-based pricing, you set your prices according to what consumers think your product is worth. We're big fans of this pricing strategy for SaaS businesses.

#### 2. Competitive pricing

When you use a competitive pricing strategy, you're setting your prices based on what the competition is charging. This can be a good strategy in the right circumstances, such as a business just starting out, but it doesn't leave a lot of room for growth.

#### 3. Price skimming

If you set your prices as high as the market will possibly tolerate and then lower them over time, you'll be using the price skimming strategy. The goal is to skim the top off the market and the lower prices to reach everyone else. With the right product it can work, but you should be very cautious using it.

#### 4. Cost-plus pricing

This is one of the simplest pricing strategies. You just take the product production cost and add a certain percentage to it. While simple, it is less than ideal for anything but physical products.

#### **5.** Penetration pricing

In highly competitive markets, it can be hard for new companies to get a foothold. One way some companies attempt to push new products is by offering prices that are much lower than the competition. This is penetration pricing. While it may get you customers and decent sales volume, you'll need a lot of them and you'll need them to be very loyal to stick around when the price increases in the future.

#### 6. Economy pricing

This strategy is popular in the commodity goods sector. The goal is to price a product cheaper than the competition and make the money back with increased volume. While it's a good method to get people to buy your generic soda, it's not a great fit for SaaS and subscription businesses.

**7. Dynamic pricing** In some industries, you can get away with constantly changing your prices to match the current demand for the item. This doesn't work well for subscription and SaaS business, because customers expect consistent monthly or yearly expenses.

#### CHAPTER -VII Strategic Marketing Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1. Market Penetration Strategy

When a firm focuses on selling its current products to existing customers, it is pursuing a market penetration strategy. The marketing activities that will dominate in this type of marketing plan are those that emphasize increasing the loyalty of existing customers so that they are not vulnerable to loss to competitors, attracting competitors' customers, increasing the frequency of product use, and converting nonusers into users. Increasing awareness through marketing communications and increasing availability through expanded distribution are common marketing activities in this type of plan. Identifying new use occasions and new uses for a product may increase usage frequency or convert current nonusers into users.

#### 2. Market Development Strategy

The efforts to expand sales by selling current products in new markets are referred to as a market development strategy. Such efforts may involve entering new geographic markets, such as international markets. Creating product awareness and developing distribution channels are key marketing activities. Some product modification may be required to better match the needs of the local market. For example, as fast food restaurants have moved into international markets, they have often changed their menus to better match the food preferences of customers in local markets.

#### 3. Product Development Strategy

Creating new products to sell to existing customers, a product development strategy, is a common marketing strategy among firms that can leverage their relationships with existing customers. For example, American Express has been able to leverage its relationships with its credit card customers to also sell travel-related services. Similarly, cable television companies have expanded their offerings into Internet and telephone services. Research and development activities play a dominant role in this strategy. The time required to develop and test new products may be long, but once a product is developed, creating awareness, interest, and availability should be relatively rapid because the firm already has a relationship with customers. A product development strategy is also riskier than a market penetration strategy because the necessary product may not be possible to develop, at least at a cost acceptable to customers, or the product developed does not match the needs of customers.

#### 4. Diversification Strategy

A diversification strategy involves taking new products into new markets. This is really the creation of a completely new business. This is the riskiest of strategies and the strategy likely to require the most patience in waiting for a return on investment.

Contributed to Branding Strategy Insider by: David Stewart, President's Professor of Marketing and Business Law, Loyola Marymount University, Author, Financial Dimensions Of Marketing Decisions.

#### CHAPTER -VIII

#### Management Information System Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Functions of MIS**

The primary function of MIS is to report on business operations to support decision-making and ensure that the organization is managed more efficiently. This will help the company reach its full potential and thus gain a competitive advantage. Listed below are some of the crucial functions of MIS.

#### **Provide Easy Access to the Information**

MIS allows teams convenient access to marketing, financial or operational information. MIS reports strategically storing large amounts of information about the business in a central location that managers can easily access over a network.

#### **Data Collection**

Data from the company's day-to-day operations are collected and combined with data from outside sources. This allows a healthy and functional relationship between distributors, points of sale, and any other supply chain member.

#### **Performance Tracking**

As production and sales numbers are recorded and stored in a central database, MIS plays a crucial role in keeping track of the performance of the employees. This information helps detect issues early and make the right decisions quickly using the latest information.

#### Foster Collaboration in the Workplace

In any large company, many situations require the involvement of multiple people or departments in decision-making. MIS is an effective communication channel for the teams to collaborate and ensure that the decision-making group can access all the data required for effective decision-making, even if they are working from different locations.

#### **Company Projections**

These management information systems come with trend analysis features that will allow you to project how a business will perform in its current configuration and how it will be affected once you have implemented the changes you are considering.

Even the ones without the trend analysis function will still offer sufficient information to carry out the analysis accurately using external tools.

#### **Track the Implementation of Decisions**

MIS systems help project expectations for a particular decision. If they decide to implement the changes, it will be necessary to continue monitoring performance to see if the teams are on track to achieve the desired results.

#### **Improve Company Reporting**

One reason large companies prefer management information systems is the effectiveness of the reporting functions. MIS reports contain information in an easy-to-understand format, allowing management to make timely decisions. The system's accessibility to people from different parts of the organization makes it an effective communication and reporting tool.

#### CHAPTER -IX Buyer Behaviour K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

**Buyer behavior** refers to the decision and acts people undertake to buy products or services for individual or group use. It's synonymous with the term "consumer buying behavior," which often applies to individual customers in contrast to businesses.

#### Types of buyer behavior

Buyer behavior is always determined by how involved a client is in their decision to buy a product or service and how risky it is. The higher the product price, the higher the risk, the higher the customer's involvement in purchase decisions. Based on these determinants, four types of consumer buyer behavior are distinguished:

#### **Complex buying behavior**

This type is also called extensive. The customer is highly involved in the buying process and thorough research before the purchase due to the high degree of economic or psychological risk. Examples of this type of buying behavior include purchasing expensive goods or services such as a house, a car, an education course, etc.

#### **Dissonance-reducing buying behavior**

Like complex buying behavior, this type presupposes lots of involvement in the buying process due to the high price or infrequent purchase. People find it difficult to choose between brands and are afraid they might regret their choice afterward (hence the word 'dissonance').

As a rule, they buy goods without much research based on convenience or available budget. An example of dissonance-reducing buying behavior may be purchasing a waffle maker. In this case, a customer won't think much about which model to use, choosing between a few brands available.

#### Habitual buying behavior

This type of consumer buying behavior is characterized by low involvement in a purchase decision. A client sees no significant difference among brands and buys habitual goods over a long period. An example of habitual buying behavior is purchasing everyday products.

#### Variety seeking behavior

In this case, a customer switches among brands for the sake of variety or curiosity, not dissatisfaction, demonstrating a low level of involvement. For example, they may buy soap without putting much thought into it. Next time, they will choose another brand to change the scent.

#### CHAPTER -X

#### **Product Polices** K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1. Product

The product element refers to the item or service that a company offers to its customers. It includes the physical product, packaging, branding, design, quality, features, and benefits. The goal is to create a product that meets the needs and wants of the target market, and offers unique value that sets it apart from competitors.

For example, a company may differentiate its product by emphasizing quality, offering a unique design, or providing a superior customer experience. Branding plays a key role in this segment of the marketing mix.

#### 2. Place

The place refers to the location where customers can purchase the product or service and how they'll access it. It includes distribution channels, logistics, market coverage, and levels of service. You want to ensure that the product is easily accessible and available at the right time and place, for the right people. So, this is a critical principle to think through.

For instance, a company may sell its product through multiple channels such as brick-and-mortar stores, online marketplaces, or through a direct-to-consumer model.

The company may also need to consider factors such as inventory management, order fulfillment, and shipping options to ensure that the product is available when and where customers need it. 3. Price

The price component indicates the amount that customers pay for the product or service. It comprehends the cost of goods, profit margins, pricing strategy, discounts, and promotional offers. The marketing strategy should set a price that is competitive and reflects the value of the product, while also aligning with the business profit goals. Pricing affects how your goods are perceived by consumers, and can impact your brand's reputation as affordable or lux.

For example, a company may use a skimming pricing strategy, where they set a high price to appeal to customers who are willing to pay a premium for a unique product or experience.

Alternatively, they may use a penetration pricing strategy, where they set a low price to gain market share and attract price-sensitive customers.

#### 4. Promotion

Promotion refers to the various marketing tactics that a company uses to promote its products or services. It includes advertising, sales promotions, public relations, personal selling, and digital marketing. The goal is to create awareness and interest in the product and persuade customers to make a purchase. It is one step further to closing a deal and selling your product.

For example, a company may run a social media campaign to build brand awareness, offer a discount code to encourage sales, or partner with a celebrity or influencer to promote the product.

#### 5. People

Lastly, but not less important, the people element is about the individuals who are involved in the production, distribution, and consumption of the product or service. It includes employees, customers, suppliers, and partners.

Businesses and brands don't run by themselves. You must create a positive relationship between the company and its stakeholders, and ensure that everyone's needs and wants are being met.

# **MICRO FINANCE**

### Edited by DR. K RAJALAKSHMI



Micro Finance

Copyright © 2021 by Dr. K Rajalakshmi

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-635-6

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

CHAPTER-I	
$CHAPTER - \Pi \dots 2$	
Role of Regulatory Body in Micro Finance	
Dr T J Jayasholan	
CHAPTER –III	•
Code Of Conduct For Microfinance Institutions In India	
Dr. S. Venkatesh	
CHAPTER –IV	•
Micro Finance Services	
Dr K G Selvan	
CHAPTER -V	5
Dr. S. Venkatesh	
CHAPTER -VI6 Micro Finance and Rural Development	)
K.Sasikumar	
CHAPTER –VII	7
Dr K G Selvan	
CHAPTER -VIII	
Dr T J Jayasholan	
CHAPTER -IX	)
Advantages of SHG-Bank linkage Programme	
K. Sasikumar	
CHAPTER –X1	0
The Role of Stakeholders in Developing the Micro-Finance Sector	
K. Sasikumar	

#### TABLE OF CONTENTS

#### CHAPTER -I

#### Introduction of Micro Finance Dr. K Rajalakshmi, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

Microfinance is the process of providing financial services such as loans, savings, etc. at a micro level or small scale to individuals with little or no income.

A microfinance loan passes through various stages or events from the moment it is given till the time it is repaid. The process begins with a loan application form in which the borrower is asked to furnish their details and the kind of facility requested. The bank's loans or credit department appraises the application and takes a decision whether the applicant is eligible for the loan asked and whether loan can be extended to that customer.

The bank may levy a processing charge at this stage. The loan amount is then disbursed and credited to the settlement account of the borrower who draws the amount, uses it for the purpose for which it was taken. Interest accrues on the account which has to be serviced by the borrower besides repaying the principal amount as per the repayment schedule. If the loan is not repaid or there is a delay the loan asset's status is changed to reflect its deterioration. On the other hand if the microfinance account is conducted well it will close on the date the last instalment is due and paid.

- Loan Origination or Application Processing
- Microfinance account Processing

The MF Module is also capacitated to interface with the Core Module of Oracle FLEXCUBE for Accounting, Messaging and MIS related operations.

#### **Application Entry**

During the application entry stage, the loan application is received from the applicant (s)/customers. The following information of the customers is captured as part of application processing:

- Personal and Professional details
- Details of Residence
- Financial position
- Asset Information
- Details of loan being availed

The applications will be identified by a unique application number and based on the details furnished in the application and verifications performed thereafter; the loan may be approved or rejected.

#### CHAPTER -II

#### Role of Regulatory Body in Micro Finance Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Highlights of the Bill**

- The Bill seeks to provide a statutory framework to regulate and develop the micro finance industry.
- The Reserve Bank of India (RBI) shall regulate the micro finance sector; it may set an upper limit on the lending rate and margins of Micro Finance Institutions (MFIs).
- MFIs are defined as organisations providing micro credit facilities up to Rs 5 lakh, thrift collection services, pension or insurance services, or remittance services.
- The Bill provides for the creation of councils and committees at central, state and district level to monitor the sector.
- The Bill provides for a Micro Finance Development Fund managed by RBI; proceeds from this fund can be used for loans, refinance or investment to MFIs.
- The Bill requires the RBI to create a grievance redressal mechanism.

#### Key Issues and Analysis

- The Bill provides safeguards against misuse of market dominance by MFIs to charge excessive rates. It allows RBI to set upper limits on lending rates and margins. However, there is no provision for consultation with the Competition Commission of India.
- The Bill allows MFIs to accept deposits. Unlike banks, there is no facility for insuring customer deposits against default by MFIs. The minimum capital requirement is also lower, though RBI may prescribe higher requirements.
- The Development Fund for MFIs is to be managed by the RBI. The Bill also enables regulatory powers to be delegated to NABARD. Both these provisions could lead to conflict of interest.
- The Bill provides for the creation of micro finance committees at central, state and district levels to oversee the sector. However, the formations of these committees are not mandatory.
- The Bill allows MFIs to provide pension and insurance services. However, it does not provide for regulation by or coordination of RBI with the respective sector regulators.

#### CHAPTER -III

#### Code of Conduct For Microfinance Institutions In India Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

Microfinance Institutions irrespective of legal forms, seek to create social benefits and promote financial inclusion by providing financial services to clients of financially un-served and underserved households.

#### A. Integrity

To provide low income clients women and men and their families, with access to financial services that are client focused and designed to enhance their wellbeing and are delivered in a manner that is ethical, dignified, transparent, equitable and cost effective.

#### **B.** Quality of Service

- To ensure quality services to clients, appropriate to their needs and delivered efficiently in a convenient and timely manner
- To maintain high standards of professionalism based on honesty, non discrimination and customer centricity

#### C. Transparency:

- To provide complete and accurate information to clients regarding all products and services offered.
- To create awareness and enable clients and all other stakeholders to understand the information provided with respect to financial services offered and availed

#### D. Fair Practices:

- To ensure that clients are protected against fraud and misrepresentation deception or unethical practices
- To ensure that all practices related to lending and recovery of loans are fair and maintain respect for client's dignity and with an understanding of client's vulnerable situation

#### E. Privacy of client information

• To safeguard personal information of clients, allowing disclosures and exchange of relevant information with authorized personnel only, and with the knowledge and consent of clients

#### F. Integrating social values into operations:

- To ensure high standards of governance and management
- To monitor and report social as well as financial data

#### G. Feedback and grievance redressal mechanism

- To provide clients formal and informal channels for feedback and suggestions To consistently assess the impact of services in order to enhance competencies and serve clients better
- To provide a formal grievance redressal mechanism for clients.

#### CHAPTER -IV

#### Micro Finance Services Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### The For-Profit Controversy

Although there are countless heartwarming success stories, microfinance has sometimes been the target of criticism. As microfinance interest rates are generally higher than conventional banks', critics have argued that these operations make money off people with low incomes.

In fact, some non-profit micro lenders have converted to being for-profit as they've grown. One of the largest and most controversial is Mexico's Banco Compartamos. The bank was started in 1990 as a nonprofit. However, management transformed the enterprise into a traditional, for-profit company ten years later. In 2007, it went public on the Mexican Stock Exchange, and its initial public offering (IPO) raised more than \$400 million.

#### **Concerns About For-Profit Microfinancing**

In addition to Compartamos Banco, many major financial institutions and other large corporations have launched for-profit microfinance departments, including Citigroup and Barclays. Other companies have created mutual funds that invest primarily in microfinance firms.

Compartamos Banco and its for-profit peers have been criticized by many, including Yunus, the grandfather of modern microfinance. The immediate, pragmatic fear is that, out of a desire to make money, large microfinance bankers will charge higher interest rates that may create a debt trap for low-income borrowers

#### **Other Criticisms of Microfinance**

In addition to the divide between nonprofit and for-profit microfinance enterprises, other criticisms exist. Some say that individual microloans of \$100 aren't enough to provide independence instead, they keep recipients working in subsistence-level trades, barely covering basic needs like food and shelter.

A better approach, these critics maintain, is to create jobs by constructing new factories and producing new goods. They cite the examples of China and India, where the development of large industries has led to stable employment and higher wages, which in turn has helped millions to emerge from the lowest levels of poverty.

#### CHAPTER -V Formation and Development of Micro Finance Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Core Functions of the Department**

NABARD has been continuously focusing on bringing in various stakeholders on a common platform and building their capacities to take the initiatives forward. This has resulted in tremendous growth of the microfinance sector in India through various approaches as discussed below:

#### A. Promotion of Self Help Group – Bank Linkage Programme (SHG-BLP)

Based on the observations of various research studies and an action research project carried out by NABARD, the model of SHG-BLP' was evolved as a cost effective mechanism for providing financial services to the unreached and underserved poor households. What started as a pilot to link around 500 SHGs of poor to the formal financial institutions during the year 1992-93, has now become the largest microfinance programme in the world in terms of client base and outreach.

#### B. Promotion of Joint Liability Groups (JLGs) and their financing by banks

JLGs are essentially credit groups of small/marginal/tenant farmers/asset- less poor who do not have proper title of their farmland. These informal groups of four to 10 members are engaged in similar economic activities and are willing to jointly undertake to repay the loans taken by them from the Banks. Financing of JLGs was introduced as a pilot project in 2004-05 by NABARD in eight States with the support of 13 RRBs. The scheme was later mainstreamed for the banking system in the year 2006.

#### **Others Initiatives**

- Livelihood promotion: To enable the SHG members to take up livelihood activities, NABARD has been supporting Micro Enterprise Development Programmes (MEDPs) and Livelihood and Enterprise Development Programmes (LEDPs).
- Microenterprises through Skill Upgradation for Women (m-Suwidha) to provide sustainable livelihood solutions in farm and off-farm sector by identifying skill gaps, occupational fitments, choosing suitable professions/ activities, providing skills through identified Knowledge Partners/Resource Agencies along with marketing tie-ups and credit support from banks/financial institutions.
- Publications: The department brings out various publications relating to SHG- Bank linkage and other microfinance activities on a regular basis. The book Status of Microfinance in India is one of the most important annual publications, which is used as a reference material by research scholars, policy planners, Govt. departments etc.
- Supporting training and capacity building of personnel of partner agencies such as banks, NGOs, government departments and other institutions
- Extending support to NGOs and other institutions for formation and credit linkage of SHGs and JLGs with banks.

#### CHAPTER -VI Micro Finance and Rural Development K.Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Benefits of Microfinance in Rural Areas**

Microfinance has numerous benefits in rural areas that contribute to economic and social development:

#### • Economic Empowerment

One of the most significant advantages is economic empowerment, which allows low-income individuals to become entrepreneurs and manage their finances effectively. Microfinance institutions (MFIs) provide small loans that enable people to start or expand small businesses, generating income and employment opportunities in their communities.

#### • Financial Inclusion

Another advantage of microfinance is financial inclusion, which provides financial services to those who are typically excluded from formal banking systems. By offering loans, savings, and insurance services, MFIs enable individuals to build assets and manage financial risks. This contributes significantly to poverty reduction efforts.

#### • Women Empowerment

Microfinance has also been a powerful tool for women's empowerment. In many societies, women are often excluded from economic activities and denied access to formal financial services. However, MFIs target women as their primary clients by providing them with the resources and support needed for starting small businesses and becoming financially independent.

#### • Social Development

Moreover, microfinance promotes social development by increasing access to basic services such as education, health care, and clean water. With increased income from microloans provided by MFIs, individuals can afford these necessities while improving their standard of living.

#### • Self-Sustainability

Finally, the self-sustainability model of MFIs ensures that microfinance remains a sustainable solution for financial inclusion in rural areas. These institutions generate enough revenue to cover operational costs while remaining financially viable in the long run.

So, microfinance promotes self-sustainability among communities through an inclusive approach to finance management.

#### CHAPTER -VII

#### Client Protection guidelines for Microfinance Institutions (CPG) Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

The CPG states that all MFIs, regardless of their form

- Shall display the client protection code in all branches and offices, in plain view.
- Shall endeavour to provide microfinance services to all eligible clients, as per RBI guidelines.
  - (a) individual sanction letter
  - (b) loan card
  - (c) loan schedule (d) pass book
- Shall not bundle products, except for credit life, life insurance and Livestock insurance products. The terms of insurance should be transparently conveyed to the customer and must comply with RBI and IRDA norms. Consent of the client must be taken in all cases.
- Shall ensure that all employees follow company guidelines for interaction with clients
- Shall ensure that all staff and persons acting for the MFI or on behalf of the MFI
  - (a) use courteous language, maintain decorum, and /or respectful of cultural sensitivities during all interactions with clients
  - (b) do not indulge in any behaviour that in any manner that would suggest any kind of threat or violence to clients
  - (c) do not contact clients at odd hours as per the RBI guidelines for loan recovery agents
  - (d) do not visit clients at inappropriate occasions such as bereavement, sickness etc to collect dues
- Shall provide a valid receipt (in whatever form decided by the MFI) for each and every payment received from the borrower
- Shall keep personal client information strictly confidential
  - (a) Client has been informed about such disclosure and permission has been obtained in writing.
  - (b) The party in question has been authorized by the client to obtain client information from the MFI
  - (c) It is legally required to do so.
  - (d) This practice is customary amongst financial institution and available for a close group on reciprocal basis (such as a credit bureau)
- Shall follow company approved process to raise client awareness of the options, choices and responsibilities vis-à-vis financial products and services available.
- Shall inform all new clients about the organization's policies and procedures
- Shall inform clients about the existence and purpose of feedback mechanism and how to access them.

#### CHAPTER -VIII

#### Institutional conduct guidelines for Micro Finance Institutions (ICG) Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### The ICG states that all MFIs regardless of their form

- Shall have an appropriate mechanism for ensuring compliance with the code of conduct
- Shall have appropriate policies and operating guidelines to treat clients and employees with dignity.
- Shall maintain formal records of all transactions in accordance with all regulatory and statutory norms and borrowers acknowledgement/ acceptance of terms / conditions must form a part of these records
- Shall have detailed board approved process for dealing with clients at each stage of default
- Shall not collect shortfalls in collection from employees except in proven cases of frauds by employees
- Shall have a Board approved debt restructuring product / program for providing relief to borrowers facing repayment stress
- Shall seek a reference check from previous employer for any new hire.
- Shall provide within two week the replies to the reference check correspondence for another MFI
- Shall honour a one month notice period from an outgoing employee.
- Shall not recruit an employee of another MFI without the relieving letter from the previous MFI employer except where the previous employer (MFI) fails to respond to the reference check request within 30 days
- Shall not assign a new employee recruited from another MFI, to the same area he / she was serving at the previous employer for a period of one year. This restriction applies to positions up to the branch manager level
- Shall have a dedicated process to raise clients awareness of options, choices, rights and responsibilities as a borrower and shall conduct regular checks on client awareness and understanding of the key terms and conditions of the products/ service offered / availed
- Shall agree to share complete client data with all RBI approved credit bureaus, as per the frequency of data submission prescribed by the Credit Bureaus
- Shall establish dedicated feedback and grievances redressel mechanism to correct any error and handle / receive complaints speedily and efficiently
- Shall designate an official to handle complaints and / or note any suggestions from the clients and make his / her contact numbers easily accessible to clients.

#### CHAPTER -IX

#### Advantages of SHG-Bank linkage Programme K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### Advantages of SHG-Bank linkage Programme

- Reduction in transaction cost by way of externalization of a part of the credit cycle (appraisal, disbursal, supervision and repayment).
- Mobilization of small savings.
- Assured and timely repayment leading to faster recycling of funds.
- An opportunity for expansion of business and coverage of poor clientele.

#### NGOs:

- Facilitate deepening of their developmental efforts.
- Synergy in operating social Programmes with economic Programmes.
- Increase the outreach to the poor through credit plus approach.
- Emergence as a bridge between banks and the poor.
- Avenue for performing financial intermediation in unbanked and backward areas.

#### Purpose of Loan under SHG Bank Linkage Programme

Loan may be granted by the SHGs both for consumption purpose and production purpose. The group will discuss and decide about the purpose for which loans are to be given to the individual members by the SHG. The bank does not decide the purpose for which the SHG gives loans to its members.

#### **Ten Pillars of SHG-Bank Linkage Programme**

- Participative financial services management is more efficient and responsive
- Poor can save and are bankable
- The capacities of the banking system can match the expectations of the poor.
- Poor need credit support as well as savings and other services
- Small cohesive group of poor with/without support can effectively manage and supervise micro credit.

#### Models under SHG-Bank Linkage Programme

Under the SBLP the following three different models have emerged.

Model I — (Direct Model) SHGs Formed and financed by Banks

Model II — (Indirect Model, NGO acting as Financial Intermediary)

These three models have certain advantages and limitations on account of parameters such as i) Transaction cost of lending for the banks ii) Transaction cost of borrowing for the SHGs.

#### CHAPTER -X

#### The Role of Stakeholders in Developing the Micro-Finance Sector K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Roles of Stakeholders**

- **Direct the Management**: The stakeholders can be a part of the board of directors and therefore help in taking actions. They can take over certain departments like service, human resources or research and development and manage them for ensuring success.
- **They Bring in Money**: Stakeholders are the large investors of the company and they can anytime bring in or take out money from the company. Their decision shall depend upon the company's financial performance. Therefore they can pressurize the management for financial reports and change tactics if necessary. Some stakeholders can even increase or decrease the investment to change the share price in the market and thus make the conditions favorable for them.
- Help in Decision Making: Major stakeholders are part of the board of directors. Therefore they also take decisions along with other board members. They have the power to disrupt the decisions as well. They and bring n more ideas a threaten the management to obey them. The stakeholders also have all the powers to appoint senior-level management. Therefore, they are there in all the major decision-making areas. They also take decisions regarding liquidations and also acquisitions.
- **Corporate Conscience**: Large stakeholders are the major stakeholders of the company and have monitored over all the major activities of the company. They can make the company abide by human rights and environmental laws. They also monitor the outsourcing activities and may vote against any business decision if it harms the long term goals of the company.
- Other Responsibilities: Apart from the above four major roles they also have some other roles to play in the company. They can identify new areas for market penetration and increased sales. They can bring in more marketing ideas. They also attract other investors like honeybees in the company. They can be a part of a selection board or a representative for the company. Moreover, they can take all the major social and environmental decisions.

#### Conclusion

Internal stakeholders are key management of the company and therefore must be considered seriously. They have their own rights and duties, therefore, they are of utmost importance for the better future of the company. They must also take their roles seriously and work towards its betterment. To understand more about the key role of stakeholders, you can join our webinar for successful onboarding process. In this webinar, you can learn about the roles and responsibilities of Executive management, human resources, the new hire's manager.

# **EXTREPRENEURIAL DEVELOPMENT**

Edited

D

K.G.SELVAN



Entrepreneurial Development

Copyright © 2021 by Dr K.G.Selvan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-123-8

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### TABLE OF CONTENTS

CHAPTER –I	
CHAPTER –II	
CHAPTER –III	
CHAPTER –IV	
CHAPTER -V	
CHAPTER -VI	
CHAPTER –VII	
Dr K G Selvan	
CHAPTER -VIII	
CHAPTER -IX	
K. Sasikumar	
CHAPTER –X1 Support to Entrepreneurs in Small Business	0
K. Sasikumar	
#### CHAPTER -I

# Characteristics of Entrepreneurship Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Characteristics of Entrepreneurship:**

Not all entrepreneurs are successful; there are definite characteristics that make entrepreneurship successful. A few of them are mentioned below:

#### Ability to take a risk

Starting any new venture involves a considerable amount of failure risk. Therefore, an entrepreneur needs to be courageous and able to evaluate and take risks, which is an essential part of being an entrepreneur.

#### Innovation

It should be highly innovative to generate new ideas, start a company and earn profits out of it. Change can be the launching of a new product that is new to the market or a process that does the same thing but in a more efficient and economical way.

#### Visionary and Leadership quality

To be successful, the entrepreneur should have a clear vision of his new venture. However, to turn the idea into reality, a lot of resources and employees are required. Here, leadership quality is paramount because leaders impart and guide their employees towards the right path of success.

#### **Open-Minded**

In a business, every circumstance can be an opportunity and used for the benefit of a company. For example, Paytm recognised the gravity of demonetization and acknowledged the need for online transactions would be more, so it utilised the situation and expanded massively during this time.

#### Flexible

An entrepreneur should be flexible and open to change according to the situation. To be on the top, a businessperson should be equipped to embrace change in a product and service, as and when needed.

#### **Know your Product**

A company owner should know the product offerings and also be aware of the latest trend in the market. It is essential to know if the available product or service meets the demands of the current market, or whether it is time to tweak it a little. Being able to be accountable and then alter as needed is a vital part of entrepreneurship.

# CHAPTER -II Concept of Entrepreneurial Development Dr T J Jayasholan , Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### Innovation

An entrepreneur is the key source of innovation and variation in an economy. It is actually one of the most important tools of an entrepreneurs success. They use innovation to exploit opportunities available in the market and overcome any threats. So this innovation can be a new product, service, technology, production technique, marketing strategy, etc. Or innovation can involve doing something better and more economically. Either way in the concept of entrepreneurship, it is a key factor.

#### **Risk-Taking**

Entrepreneurship and risk-taking go hand in hand. One of the most important features of entrepreneurship is that the whole business is run and managed by one person. So there is no one to share the risks with.

Not taking any risks can stagnate a business and excessive impulsive risk-taking can cause losses. So a good entrepreneur knows how to take and manage the risks of his business. But the willingness of an entrepreneur to take risks gives them a competitive edge in the economy. It helps them exploit the opportunities the economy provides.

#### Vision

Vision or foresight is one of the main driving forces behind any entrepreneur. It is the energy that drives the business forward by using the foresight of the entrepreneur. It is what gives the business an outline for the future – the tasks to complete, the risks to take, the culture to establish, etc.

All great entrepreneurs of the world that started with an entrepreneurship business are known to have great vision. This helps them set out short term and long term goals for their business and also plan ways to achieve these objectives.

#### Organization

In entrepreneurship, it is essentially a one-man show. The entrepreneur bears all the risks and enjoys all the rewards. And sure he has the help of employees and middle-level management, yet he must be the one in ultimate control. This requires a lot of organization and impeccable organizational skills. An entrepreneur must be able to manage and organize his finances, his employees, his resources, etc. So his organizational abilities are one of the most important elements of entrepreneurship.

# CHAPTER -III

# Attribute and Characteristics of a Successful Entrepreneur Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1. Curiosity

Successful entrepreneurs have a distinct personality trait that sets them apart from other organizational leaders: a sense of curiosity. An entrepreneur's ability to remain curious allows them to continuously seek new opportunities. Rather than settling for what they think they know, entrepreneurs ask challenging questions and explore different avenues.

#### 2. Willingness to Experiment

Along with curiosity, entrepreneurs require an understanding of structured experimentation, such as design thinking. With each new opportunity, an entrepreneur must run tests to determine if it's worthwhile to pursue.

#### 3. Adaptability

Entrepreneurship is an iterative process, and new challenges and opportunities present themselves at every turn. It's nearly impossible to be prepared for every scenario, but successful business leaders must be adaptable.

#### 4. Decisiveness

To be successful, an entrepreneur has to make difficult decisions and stand by them. As a leader, they're responsible for guiding the trajectory of their business, including every aspect from funding and strategy to resource allocation.

#### 5. Self-Awareness

A great entrepreneur is aware of their strengths and weaknesses. Rather than letting shortcomings hold them back, they build well-rounded teams that complement their abilities.

#### 6. Risk Tolerance

Entrepreneurship is often associated with risk. While it's true that launching a venture requires an entrepreneur to take risks, they also need to take steps to minimize it.

#### 7. Comfort with Failure

In addition to risk-management and calculated decision-making, entrepreneurship requires a certain level of comfort with failure.

#### 8. Persistence

While many successful entrepreneurs are comfortable with the possibility of failing, it doesn't mean they give up easily. Rather, they see failure as an opportunity to learn and grow.

#### 9. Innovative Thinking

Innovation often goes hand-in-hand with entrepreneurship. While innovation in business can be defined as an idea that's both novel and useful, it doesn't always involve creating an entirely new product or service. Some of the most successful startups have taken existing products or services and drastically improved them to meet the changing needs of the market.

# CHAPTER -IV Role of Entrepreneur in Economic Development Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

**1. Raises Standard of Living** A significant role of entrepreneurship in economic development is that it can greatly enhance the standard of living for individuals and communities by setting up industries and creating wealth and new positions. Entrepreneurship not only provides large-scale employment and ways to generate income, it also has the potential to improve the quality of individual life by developing products and services that are affordable, safe to use, and add value to their lives.

**2. Economic Independence** Entrepreneurship can be a path to economic independence for both the country and the entrepreneur. It reduces the nation's dependence on imported goods and services and promotes self-reliance. The manufactured goods and services can also be exported to foreign markets, leading to expansion, self-reliance, currency inflow, and economic independence. Similarly, entrepreneurs get complete control over their financial future. Through their hard work and innovation, they generate income and create wealth, allowing them to achieve economic independence and financial security.

**3. Benefits of New Firms and Businesses** Entrepreneurs identify market needs and develop solutions through their products and services to begin their business venture. By starting new firms and businesses, entrepreneurs play a key role in shaping the economy and creating a more dynamic and diverse business landscape. Entrepreneurship also promotes innovation and competition, leading to new and improved products and services that contribute to economic growth and development.

**4. Creation of Jobs** Entrepreneurship is a pivotal driver of job creation. Running the operations of new businesses and meeting the requirements of customers results in new work opportunities. Entrepreneurship also drives innovation and competition that encourages other entrepreneurs and investments, creating new jobs in a wide range of industries, from manufacturing and construction to service and technology sectors.

**5. Encourages Capital Formation** Capital formation is the process of accumulating resources, such as savings and investments, to fund new business ventures and support economic growth. Entrepreneurship can encourage capital formation by attracting investment. In addition, the creation of new businesses and the growth of existing firms can also contribute to the development of a more diverse and dynamic economy that encourages capital formation and opens the door to a wide range of investment opportunities.

**6. Elimination of Poverty** Entrepreneurship has the potential to lift people out of poverty by generating employment and stimulating economic activity. Entrepreneurship also contributes to the development of local economies and helps improve the overall standard of living.

**7. Community Development** Entrepreneurship promotes economic growth, provides access to goods and services, and improves the overall standard of living. Many entrepreneurs also make a positive impact on their communities and improve their well-being by catering to underserved areas and developing environment-friendly products.

# CHAPTER -VI

# Environmental Analysis and Basic Govt. Formalities. K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

## **ENVIRONMENTAL ANALYSIS**

- Environmental analysis is a strategic tool.
- It is a process to identify all the external and internal elements, which can affect the organization's performance.
- The analysis entails assessing the level of threat or opportunity the factors might present.
- These evaluations are later translated into the decision-making process.
- The analysis helps align strategies with the firm's environment.

**POLITICAL FACTORS** The political factors take the country's current political situation. ' It also reads the global political condition's effect on the country and business. ' Some of the factors considered for analysis are: Government policies Taxes laws and tariff Stability of government  $\Box$  Entry mode regulations.

**ECONOMIC FACTORS '** Economic factors involve all the determinants of the economy and its state. ' These are factors that can conclude the direction in which the economy might move. ' So, businesses analyze this factor based on the environment. ' It helps to set up strategies in line with changes.

**SOCIAL FACTORS** ' Countries vary from each other. ' Every country has a distinctive mindset. ' These attitudes have an impact on the businesses. ' The social factors might ultimately affect the sales of products and services.

**TECHNOLOGICAL FACTORS** ' Technology is advancing continuously. ' The advancement is greatly influencing businesses. ' Performing environmental analysis on these factors will help you stay up to date with the changes. ' Technological factors will help you know how the consumers react to various trends.

ENVIRONMENT FACTORS ' The location influences business trades. ' Climatic changes can affect the trade of businesses. ' The consumer reactions to particular offering can also be an issue. ' This most often affects agri-businesses. ' Some environmental factors to be noticed are: 
□ Geographical location The climate and weather Waste disposal laws Energy consumption regulation People's attitude towards the environment

**LEGAL FACTORS '** Legislative changes take place from time to time. ' Many of these changes affect the business environment. ' If a regulatory body sets up a regulation for industries, for example, that law would impact industries and business in that economy. ' So, businesses should also analyze the legal developments in respective environments.

# CHAPTER -VI

# Environmental Analysis and Basic Govt. Formalities. K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

## **ENVIRONMENTAL ANALYSIS**

- Environmental analysis is a strategic tool.
- It is a process to identify all the external and internal elements, which can affect the organization's performance.
- The analysis entails assessing the level of threat or opportunity the factors might present.
- These evaluations are later translated into the decision-making process.
- The analysis helps align strategies with the firm's environment.

**POLITICAL FACTORS** The political factors take the country's current political situation. ' It also reads the global political condition's effect on the country and business. ' Some of the factors considered for analysis are: Government policies Taxes laws and tariff Stability of government  $\Box$  Entry mode regulations.

**ECONOMIC FACTORS '** Economic factors involve all the determinants of the economy and its state. ' These are factors that can conclude the direction in which the economy might move. ' So, businesses analyze this factor based on the environment. ' It helps to set up strategies in line with changes.

**SOCIAL FACTORS** ' Countries vary from each other. ' Every country has a distinctive mindset. ' These attitudes have an impact on the businesses. ' The social factors might ultimately affect the sales of products and services.

**TECHNOLOGICAL FACTORS** ' Technology is advancing continuously. ' The advancement is greatly influencing businesses. ' Performing environmental analysis on these factors will help you stay up to date with the changes. ' Technological factors will help you know how the consumers react to various trends.

ENVIRONMENT FACTORS ' The location influences business trades. ' Climatic changes can affect the trade of businesses. ' The consumer reactions to particular offering can also be an issue. ' This most often affects agri-businesses. ' Some environmental factors to be noticed are: 
□ Geographical location The climate and weather Waste disposal laws Energy consumption regulation People's attitude towards the environment

**LEGAL FACTORS '** Legislative changes take place from time to time. ' Many of these changes affect the business environment. ' If a regulatory body sets up a regulation for industries, for example, that law would impact industries and business in that economy. ' So, businesses should also analyze the legal developments in respective environments.

#### CHAPTER -VII

# Entrepreneurial Environment Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Entrepreneurial Environment**

Entrepreneurship does not grow without environmental support. Entrepreneurship arises within the environmental framework in all societies across the globe. The entrepreneurship environment refers to the various facets within which enterprises have to operate.

The environment, therefore, influences the enterprise. In general, the environment is shaped by various factors such as political, social, economic, national, and legal forces, and so on.. influencing entrepreneurship.

The entrepreneurial ecosystem comprises various elements that contribute to the growth and advancement of entrepreneurship. It refers to the overall economic, socio-cultural, and political factors that influence people's willingness and ability to undertake entrepreneurial activities. In short, the entrepreneurial environment refers to the various constraints within which enterprises are required to operate.

An entrepreneurial environment refers to a setting or atmosphere that encourages and supports the growth of entrepreneurs and their ventures. It is a space where individuals are motivated to take risks, think creatively, and pursue innovative ideas.

In such an environment, there is a strong emphasis on problem-solving, collaboration, and the ability to adapt to change. It provides opportunities for aspiring entrepreneurs to learn, network, and gain valuable experience from others who have already embarked on their entrepreneurial journeys.

Within an entrepreneurial environment, there is a sense of excitement and possibility. It is a place where individuals are encouraged to think outside the box and challenge the status quo. This environment fosters a culture of innovation, where new ideas are welcomed and nurtured. It is not just about starting a business, but also about fostering an entrepreneurial mindset that can be applied to various aspects of life.

In this environment, failure is seen as a learning opportunity rather than a setback, and individuals are encouraged to persevere and learn from their mistakes. An entrepreneurial environment is characterized by a supportive network of mentors, advisors, and like-minded individuals who are passionate about entrepreneurship.

It provides access to resources, such as funding opportunities, incubators, and accelerators, that can help entrepreneurs turn their ideas into reality.

This environment also promotes a strong work ethic and a sense of ownership and responsibility. It encourages individuals to take initiative, be proactive, and constantly seek new opportunities for growth and development. Overall, an entrepreneurial environment is a dynamic and inspiring space that fosters creativity, innovation, and the pursuit of entrepreneurial dreams.

# CHAPTER -VIII Business Plan Preparation Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### Step 1: Conducting a SWOT Analysis

The first step in the business planning process is to conduct a SWOT analysis. SWOT stands for Strengths, Weaknesses, Opportunities, and Threats. This analysis will help you understand your business's internal and external environment, and it can help you identify areas of improvement and growth.

#### **Step 2: Defining Your Business Objectives**

Once you have conducted a SWOT analysis, the next step is to define your business objectives. Business objectives are specific, measurable, achievable, relevant, and time-bound (SMART) goals that align with your business's mission and vision.

#### **Step 3: Conducting Market Research**

Market research should always be a part of your strategic business planning. This step gathers information about your target audience, competitors, and industry trends. This information can help you make informed decisions about your product or service offerings, pricing strategy, and marketing campaigns.

#### **Step 4: Identifying Your Target Audience**

Identifying your target audience is essential in the business planning process. Your target audience is the group of people who are most likely to buy your product or service. Understanding their needs, preferences, and behaviors can help you create effective marketing campaigns and improve customer satisfaction.

#### **Step 5: Developing a Marketing Plan**

A marketing plan is a strategic roadmap that outlines your marketing objectives, strategies, tactics, and budget. Your marketing plan should align with your business objectives and target audience and include a mix of online and offline marketing channels.

#### **Step 6: Creating a Financial Plan**

A financial plan is a detailed document that outlines your business's financial projections, budget, and cash flow. Your financial plan should include a balance sheet, income statement, and cash flow statement, and it should be based on realistic assumptions and market trends.

#### **Step 7: Writing Your Business Plan**

The final step in the business planning process is to write your business plan. A business plan is a comprehensive document that outlines your business's mission, vision, objectives, strategies, and financial projections. A business plan can help you clarify your business idea, assess the feasibility of your business, and secure funding from investors or lenders. It can also provide a roadmap for your business and ensure that you stay focused on your goals and objectives.

# CHAPTER -IX Financing and Accounting Need K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### 1) Keep Financial Records

Accounting is essentially a record of a company's financial activities. A company's ledger is where accountants and small business owners can track the income and expenses of a business' daily operations. An accurate record of a company's finances can help a business manage their financial future and understand the cash flow

#### 2) Avoid Legal Problems

Keeping an accurate financial record helps follow important business laws. Overlooking a minor detail could have major implications on your tax management. Financial managers need to understand what expenses to deduct, how much taxes to pay and when to pay those taxes. Poor financial recording could lead to your company being audited and could land you in unnecessary legal trouble.

#### 3) Make a Budget

Using your financial records and understanding your cash flow can help you create a budget and budget is what keeps your business on track. A budget gives you a current view of your financial standing and helps you navigate your business towards future growth and development.

#### 4) Analyzing Performance

Successful business owners are always checking in to see how their business is doing. A company can assess their financial position by looking at their historical and current records of liabilities and assets and other financial records. A business owner can use this information to gauge how the company is doing.

#### 5) External Communication

The communication of financial information is important when dealing with external parties. Clear accounting and finance management can be useful when getting a loan from a bank or attracting potential investors Good financial management makes it easier for you to provide financial statements to external stakeholders. External users will assess these reports to decide how to proceed with their involvement with your business.

#### 6) Internal Communication

Financial reporting can also help business owners communicate information to internal stakeholders. This information might be relevant to employees who are interested in profit-sharing and stock-based compensation.

#### 7) Developing Strategy

Good accounting and finance management inevitably leads to a good strategy. Once you've developed a budget and thoroughly analyzed your data, it should be easier to have a better understanding to develop a strategy to achieve your bottom line.

#### CHAPTER -X

# Support to Entrepreneurs in Small Business K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Institutional Support System for Entrepreneurs**

This institutional support system has been designed at following four levels:

- 1. Central Government
- 2. State Government
- 3. Non-Government Support System
- 4. District Industries Centres (DIC)

#### 1. Central Government Institutions

The Government Formulated the Micro, Small and Medium Enterprises: development Act, 2006 and established the National Board for Micro, Small and Medium Enterprises (NBMSME) and made rules there under in the year 2006. This Board examines the factors affecting promotion and development of MSMEs and reviews policies and programmes from time to time relating to these enterprises, from time to time and makes recommendations to the Government in formulating the policies for the growth of MSMEs.

#### 2. State Government Institutions

The State Governments also execute different promotional and developmental projects and schemes to provide number of supporting incentives for development and promotion of MSMEs in their respective states. These are executed through the State Directorate of Industries, which has District Industries Centres (DICs) under it, for implementing the central/state level schemes.

#### 3. Non-Government Institutions

Besides the Central Government and the State Government agencies, there are some Non-Governmental agencies that are also supporting the cause of small scale industries in the country. These agencies include Non- Government organisations and industry associations. They provide a common platform to voice SSI needs and initiate co-operative efforts.

#### 4. District Industries Centres (DIC) & Industrial Estates

In each district, there is one agency to deal with all requirements of small and village Industries. This is called "District Industries Centre", The District Industries Centres have undertaken various programmes for investment promotion at the grass root level such as organizing seminars workshops, extending support for trade fairs and exhibitions organized by various Industry's associations.

# RETAIL MANAGEMENT

# EDITED BY



Retail Management Copyright © 2021 by Dr T J Jayasholan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-161-0

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

CHAPTER –I1
Introduction of Retailing
Dr T J Jayasholan
CHAPTER –II
Types of Retailers
Dr K G Selvan
CHAPTER –III
Retail Marketing and Advertising
Dr. S. Venkatesh
CHAPTER –IV4
Retail Promotion
Dr K G Selvan
CHAPTER -V
Retail Market Strategy
Dr. S. Venkatesh
CHAPTER -VI
International Retailing
K.Sasikumar
CHAPTER –VII
Retail Consumer Behaviour
Dr K G Selvan
CHAPTER -VIII
Retail Location
Dr T J Jayasholan
CHAPTER -IX9
Emerging Trade in Retailing
K. Sasikumar
CHAPTER –X10
Retail Space Management and Marketing
K. Sasikumar

# TABLE OF CONTENTS

# CHAPTER -I Introduction of Retailing Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

**Retail** is the sale of goods and services to consumers, in contrast to wholesaling, which is the sale to business or institutional customers. A **retailer** purchases goods in large quantities from manufacturers, directly or through a wholesaler, and then sells in smaller quantities to consumers for a profit. Retailers are the final link in the supply chain from producers to consumers.

Retail markets and shops have a very ancient history, dating back to antiquity. Some of the earliest retailers were itinerant peddlers. Over the centuries, retail shops were transformed from little more than "rude booths" to the sophisticated shopping malls of the modern era. In the digital age, an increasing number of retailers are seeking to reach broader markets by selling through multiple channels, including both bricks and mortar and online retailing. Digital technologies are also affecting the way that consumers pay for goods and services. Retailing support services may also include the provision of credit, delivery services, advisory services, stylist services and a range of other supporting services. Retail workers are the employees of such stores.

Most modern retailers typically make a variety of strategic level decisions including the type of store, the market to be served, the optimal product assortment, customer service, supporting services, and the store's overall market positioning. Once the strategic retail plan is in place, retailers devise the retail mix which includes product, price, place, promotion, personnel, and presentation.

- Market analysis Market size, stage of market, market competitiveness, market attractiveness, market trends
- Customer analysis Market segmentation, demographic, geographic, and psychographic profile, values and attitudes, shopping habits, brand preferences, analysis of needs and wants, and media habits
- Internal analysis Other capacities including human resource capability, technological capability, financial capability, ability to generate scale economies or economies of scope, trade relations, reputation, positioning, and past performance
- Competition analysis Availability of substitutes, competitor's strengths and weaknesses, perceptual mapping, competitive trends.

# CHAPTER -III

# Retail Marketing and Advertising Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

# Types of Retail Advertising In-Store Displays

In-store displays are physical setups within retail locations that showcase products prominently. End caps, shelf talkers, and aisle displays catch shoppers' attention, highlighting specific items or promotions. These displays often use creative visuals and signage to convey the value of products and encourage impulse purchases.

#### Window Displays

Window displays are eye-catching arrangements of products and visuals in store windows. They aim to draw passersby into the store by showcasing the retailer's offerings in an appealing and thematic way. To maintain freshness and relevance, window displays often change with seasons, holidays, or promotional events.

## Point-of-Sale (POS) Materials

**POS** materials include posters, banners, shelf danglers, and counter cards strategically placed near checkout counters. These materials provide customers with last-minute promotions, upsell opportunities, and complementary product suggestions as they finalize their purchases.

#### **Flyers and Circulars**

Printed or digital flyers and circulars are distributed in-store or via mail and email to inform customers about ongoing sales, promotions, and new arrivals. These materials provide a comprehensive overview of current offerings and encourage customers to visit the store or website.

#### **Digital Advertising**

In the digital age, online platforms play a crucial role in retail advertising. This includes:

**Social Media Advertising:** Retailers leverage platforms like Facebook, Instagram, and Twitter to target specific demographics with ads showcasing products, discounts, and engaging content.

Search Engine Advertising: Paid search ads, such as Google Ads, display products and promotions to users actively searching for related terms.

#### **Mobile Advertising**

With the rise of smartphone usage, mobile advertising has become essential. Retailers can use location-based ads, mobile apps, and SMS marketing to reach customers on their mobile devices with personalized offers and promotions.

# TV and Radio Advertising

While these traditional media forms have become less dominant, they still hold relevance for certain markets. Retailers can create commercials that convey their brand image and unique selling points, reaching a broader audience.

# CHAPTER -III

# Retail Marketing and Advertising Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

# Types of Retail Advertising In-Store Displays

In-store displays are physical setups within retail locations that showcase products prominently. End caps, shelf talkers, and aisle displays catch shoppers' attention, highlighting specific items or promotions. These displays often use creative visuals and signage to convey the value of products and encourage impulse purchases.

#### Window Displays

Window displays are eye-catching arrangements of products and visuals in store windows. They aim to draw passersby into the store by showcasing the retailer's offerings in an appealing and thematic way. To maintain freshness and relevance, window displays often change with seasons, holidays, or promotional events.

## Point-of-Sale (POS) Materials

**POS** materials include posters, banners, shelf danglers, and counter cards strategically placed near checkout counters. These materials provide customers with last-minute promotions, upsell opportunities, and complementary product suggestions as they finalize their purchases.

#### **Flyers and Circulars**

Printed or digital flyers and circulars are distributed in-store or via mail and email to inform customers about ongoing sales, promotions, and new arrivals. These materials provide a comprehensive overview of current offerings and encourage customers to visit the store or website.

#### **Digital Advertising**

In the digital age, online platforms play a crucial role in retail advertising. This includes:

**Social Media Advertising:** Retailers leverage platforms like Facebook, Instagram, and Twitter to target specific demographics with ads showcasing products, discounts, and engaging content.

Search Engine Advertising: Paid search ads, such as Google Ads, display products and promotions to users actively searching for related terms.

#### **Mobile Advertising**

With the rise of smartphone usage, mobile advertising has become essential. Retailers can use location-based ads, mobile apps, and SMS marketing to reach customers on their mobile devices with personalized offers and promotions.

# TV and Radio Advertising

While these traditional media forms have become less dominant, they still hold relevance for certain markets. Retailers can create commercials that convey their brand image and unique selling points, reaching a broader audience.

## CHAPTER -IV Retail Promotion Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

The promotions mix refers to promotion strategies that are used to create demand for a product or service. It is different from the marketing mix in that it focuses on how to promote a product or service with specific techniques. The promotions mix is often divided into four categories:

Advertising
 Direct selling
 Sales promotion
 Public relations

#### Advertising

Advertising can, and should, sit atop the heap of the promotional mix. Assuming that a customer is aware of a brand or company through most of the other elements of the promotional mix, the role of advertising in the marketing mix can serve as the cementing force, fortifying and reinforcing the messages in a small-business owner's marketing and sales promotions.

There is a reason advertising has been around as long as it has; it works.

#### **Direct selling**

Direct selling goes direct to customers via telephone, mail, email, TV, radio, online, magazines, newspapers or face-to-face. It bypasses the traditional stores by cutting out all the 'costs' of bricks, mortar and personnel. Some forms of direct selling are also known as 'multi-level marketing'.

#### **Sales promotion**

A sales promotion is a promotional tactic in which a business uses a temporary campaign or offer to increase interest or demand in its product or service. Any plan using special offers and events designed to stimulate short term increases in demand including coupons, contests, premium ranges etc.; these promotions are often time-limited promotions. Sales promotions rely heavily on the customer's ability to make quick decisions. The launching of promotions in order to stimulate buying (either directly-to-consumer or retail channels) should be carefully planned. This means there will typically be increased media coverage which includes advertising and marketing campaigns.

#### **Public Relations (PR)**

The purpose of a PR campaign is the advancement of goodwill and favourable public perception by communicating with the media or community on behalf of an organisation. The goal is to increase brand awareness, brand recall and ultimately brand growth. It can be accomplished by planting a significant news story indirectly in the media, or presenting it favourably through press releases or corporate anniversary parties.

# CHAPTER -V Retail Market Strategy Dr. S. Venkatesh, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

## **1. Product selection**

Choosing the right products is the cornerstone of a successful retail strategy. It involves understanding what your customers want and ensuring those items are available in your store.

Good product selection also involves regularly updating your offerings to keep up with trends and evolving customer preferences.

# 2. Pricing strategy

Setting the correct product prices will help you attract customers and make a healthy profit. While creating your pricing strategy, consider factors like:

- Cost.
- Competition.
- Customer expectations.

Thoughtful pricing helps position your retail store competitively, encourage sales, and build customer loyalty.

# 3. Customer experience

Customer experience is how customers feel when they shop with you online and in-store. It includes everything from the ease of finding products to the checkout process and the service they receive. A great customer experience keeps shoppers happy, encourages repeat business, and turns customers into brand advocates.

#### 4. Marketing and promotion

Marketing and promotion are the ways you tell potential customers about your products and special offers. Depending on your business, you may want to build a retail content marketing strategy and create a presence on social media platforms like TikTok, Instagram, and Facebook.

Consider how you can use promotions to drive traffic to your store, introduce new products, and boost sales during slower periods.

# 5. Location and distribution channels

Where and how you choose to sell products will influence customer perception of your products and if they reach your target audience.

A strategic location can significantly increase foot traffic, while diverse distribution channels, such as online sales platforms or mobile commerce, can broaden your reach.

# CHAPTER -VII Retail Consumer Behavior Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### Customer buying stages and related behaviors

#### • Need recognition

This initial stage occurs when customers realize they have a specific need or problem. For instance, an avid runner may notice their running shoes are worn out, triggering the need for a new pair. Retailers tend to stimulate this recognition through advertising, showcasing the latest in athletic footwear technology, or offering promotions on sports gear.

#### • Information search

Once the need is acknowledged, the consumer seeks information on how to fulfill it. It often involves online research, reading reviews, or asking friends for recommendations. A real-case example is the rise of influencer marketing, where consumers turn to trusted figures on social media to gather opinions and insights on products.

#### • Evaluation of alternatives

Armed with information, the shopper evaluates different products or brands. Criteria such as price, quality, and brand reputation play significant roles here. For example, a customer deciding between Nike and Adidas running shoes might compare their features, prices, and customer reviews to determine which ones offer the best value for their needs.

#### • Purchase decision

After considering the options, the customer makes their purchase decision. However, factors like a salesperson's recommendation or a last-minute discount can still influence this stage. Online retailers like Amazon excel by offering seamless checkout processes, customer reviews, and personalized recommendations that help nudge consumers towards a decision.

#### • Post-purchase behavior

Finally, as customers use their purchase, they reflect on their experience. If the product meets or surpasses their expectations, they remain satisfied and will likely buy again and recommend the brand to their friends and family. That applies to both store and online experiences. For instance, an excellent instore interaction or seamless online purchase process can significantly influence customer satisfaction. On the other hand, an unhappy customer, whether from a negative store or online experience, will leave a bad review and is unlikely to return. An illustrative case is Apple's approach to customer service and its ecosystem, which aims to ensure high satisfaction levels across store and online channels, fostering brand loyalty and advocacy.

# CHAPTER -VII Retail Consumer Behavior Dr K G Selvan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### Customer buying stages and related behaviors

#### • Need recognition

This initial stage occurs when customers realize they have a specific need or problem. For instance, an avid runner may notice their running shoes are worn out, triggering the need for a new pair. Retailers tend to stimulate this recognition through advertising, showcasing the latest in athletic footwear technology, or offering promotions on sports gear.

#### • Information search

Once the need is acknowledged, the consumer seeks information on how to fulfill it. It often involves online research, reading reviews, or asking friends for recommendations. A real-case example is the rise of influencer marketing, where consumers turn to trusted figures on social media to gather opinions and insights on products.

#### • Evaluation of alternatives

Armed with information, the shopper evaluates different products or brands. Criteria such as price, quality, and brand reputation play significant roles here. For example, a customer deciding between Nike and Adidas running shoes might compare their features, prices, and customer reviews to determine which ones offer the best value for their needs.

#### • Purchase decision

After considering the options, the customer makes their purchase decision. However, factors like a salesperson's recommendation or a last-minute discount can still influence this stage. Online retailers like Amazon excel by offering seamless checkout processes, customer reviews, and personalized recommendations that help nudge consumers towards a decision.

#### • Post-purchase behavior

Finally, as customers use their purchase, they reflect on their experience. If the product meets or surpasses their expectations, they remain satisfied and will likely buy again and recommend the brand to their friends and family. That applies to both store and online experiences. For instance, an excellent instore interaction or seamless online purchase process can significantly influence customer satisfaction. On the other hand, an unhappy customer, whether from a negative store or online experience, will leave a bad review and is unlikely to return. An illustrative case is Apple's approach to customer service and its ecosystem, which aims to ensure high satisfaction levels across store and online channels, fostering brand loyalty and advocacy.

# CHAPTER -VIII

#### **Retail Location**

Dr T J Jayasholan, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

# **Types of Retail Store location**

#### 1)Solitary sites

These are single small outlets of shops which are separated from different writers, and they are positioned near other retailers on the roads on the way to shopping centers. Many of the food and non-food retailers use this type of solitary sites.

#### 2) Unplanned shopping areas

These are the locations of retail stores which have evolved over a long period of time and have multiple outlets in nearby proximities. These are further divided into: Central business district such as the downtown areas in major cities Secondary business districts on main or high Street District neighborhood Location switch on the street or on the motorway which is also known as strip locations.

#### 3) Market potential evaluation:

The paying capacity of the population plays an important role in the evaluation of the potential of the market, along with the impact of the competition and the product estimation and demand. The retailer should also have the knowledge of regulations and laws of the country in which the store is being operated.

#### 4)Identification of alternatives:

Most of the times it so happens that the retailers in hurry of starting the business finalize a location which costs them a fortune within fact a similar location with similar business potential would've been available somewhere very close which was neglected or overlooked. In such cases, the retailer should not carry on finalizing the retail location and should also go out for alternatives and evaluate that location with similar parameters as stated above.

#### 5)Allocation of marketing budget:

A retail store should have a marketing budget depending on the cost of the location, which is in the third to build the brick and mortar place. The store which is occupying a prime location and has a good inflow of customers has indeed cost a fortune for the retailer. In such cases, the marketing budget will be very less since the story is visible to most of the customers and passers-by.

On the contrary, a store which is located away from the main street should use more marketing campaigns and spend on marketing collaterals in order to attract more customers to the store.

#### CHAPTER -X

# Retail Space Management and Marketing K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Importance of Retail space management**

1 Retail space management is important to increase sales

Planning available space in the store helps you to increase sales. Having an understanding of the space available in the store helps you to decide the layout of your store and location of different categories of products. C It will be difficult for a retail manager to place all products and provide them sufficient space if they don't understand the space available in the store and he might end up allotting extra space to one category of product that he might not get enough space for a few categories of products to display.

In addition to this, you might end up investing more for buying less yielding products, and your hard-earned money will be wasted in buying excessive inventory and maintaining it rather than amending your store to increase sales.

#### 2 Customers can easily find the products they need

Nowadays, customers don't only come to your store to buy things they need, but they also come to distress themselves. Just think how they will feel if they can't find things they want to buy easily. Let us understand this with the help of an example. Imagine your customer enter your store with a list of ingredients to make a cheesecake.

He would expect to locate all the products easily and in less time. By organization, your store properly and by allotting proper space to each product and placing them at the right place on the shelf will help your customer to find them easily and without getting frustrated.

#### 3 It is helpful in controlling the rush in the peak hours

Choosing an effective layout for your store is important as it will not only help you keep your store well organized but will also keep the rush moving during peak hours of a day You can take professional help in choosing the right layout for your store, or you can copy the layout plan of a store which has similar dimensions to your store.

Just imagine how difficult it would be for you to handle your store if it is always overcrowded. Proper space management will let you choose the most sold categories of products and distributing them around the store so that you can avoid cluttering in one corner of the store.

#### CHAPTER -X

# Retail Space Management and Marketing K. Sasikumar, Department of Management, Ponnaiyah Ramajayam Institute of Science and Technology(PRIST)

#### **Importance of Retail space management**

1 Retail space management is important to increase sales

Planning available space in the store helps you to increase sales. Having an understanding of the space available in the store helps you to decide the layout of your store and location of different categories of products. C It will be difficult for a retail manager to place all products and provide them sufficient space if they don't understand the space available in the store and he might end up allotting extra space to one category of product that he might not get enough space for a few categories of products to display.

In addition to this, you might end up investing more for buying less yielding products, and your hard-earned money will be wasted in buying excessive inventory and maintaining it rather than amending your store to increase sales.

#### 2 Customers can easily find the products they need

Nowadays, customers don't only come to your store to buy things they need, but they also come to distress themselves. Just think how they will feel if they can't find things they want to buy easily. Let us understand this with the help of an example. Imagine your customer enter your store with a list of ingredients to make a cheesecake.

He would expect to locate all the products easily and in less time. By organization, your store properly and by allotting proper space to each product and placing them at the right place on the shelf will help your customer to find them easily and without getting frustrated.

#### 3 It is helpful in controlling the rush in the peak hours

Choosing an effective layout for your store is important as it will not only help you keep your store well organized but will also keep the rush moving during peak hours of a day You can take professional help in choosing the right layout for your store, or you can copy the layout plan of a store which has similar dimensions to your store.

Just imagine how difficult it would be for you to handle your store if it is always overcrowded. Proper space management will let you choose the most sold categories of products and distributing them around the store so that you can avoid cluttering in one corner of the store.

# BIG DATA ANALYTICS

# Edited by DR. R. LATHA



BIG DATA ANALYTICS Copyright © 2021 by Dr. R. LATHA

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-187-0

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# TABLE OF CONTENTS

Introduction	-	03
<b>CHAPTER 1</b> – Introduction of Big Data Analytics	-	11
Dr.R.Latha		
CHAPTER 2 – NOSQL Data Management	-	47
Dr.R.Latha		
<b>CHAPTER 3</b> – Map Reduce Applications	-	67
Dr.R.Latha		
CHAPTER 4 – Basics of Hadoop	-	89
Dr.S.Nithyanandam		
CHAPTER 5 – Hadoop related Tools	-	113
Dr.S.Nithyanandam		
CHAPTER 6 – Data Analytics Techniques	-	127
Ms.K.Jayanthi		
CHAPTER 7 – Data Visualization	-	147
Dr.S.Nithyanandam		
CHAPTER 8 – Big Data in Cloud	-	187
Ms.M.Jeeva		
<b>CHAPTER 9</b> – Big data Security and Privacy	-	215
Ms.R.Bhanumathi		
CHAPTER 10 – Big data Applications	-	257
Ms.M.Jeeva		
REFERENCES	-	285

# CHAPTER 1 Introduction of Big Data Analytics Dr.R.Latha

**Big Data Analytics** refers to the process of examining large and complex datasets to uncover hidden patterns, correlations, market trends, customer preferences, and other useful business information. With the advent of digital technologies, organizations are now generating and collecting massive amounts of data from various sources such as social media, sensors, customer transactions, and mobile devices. The sheer volume, velocity, variety, and veracity of this data are what define "Big Data."

#### Key Components of Big Data

- 1. **Volume**: The quantity of data generated, which can range from terabytes to petabytes and beyond.
- 2. **Velocity**: The speed at which data is generated, collected, and analyzed. Data is often generated in real-time or near real-time.
- 3. **Variety**: The different types of data, including structured, semi-structured, and unstructured data (e.g., text, images, videos, logs).
- 4. **Veracity**: The accuracy, reliability, and trustworthiness of data. Dealing with noise, biases, and uncertainties is crucial in big data analytics.

#### **Importance of Big Data Analytics**

- 1. **Decision-Making**: Helps organizations make more informed decisions by providing actionable insights derived from large datasets.
- 2. **Customer Insights**: Enables businesses to understand customer behavior, preferences, and trends, leading to personalized marketing and improved customer satisfaction.
- 3. **Operational Efficiency**: Helps optimize processes by identifying bottlenecks, reducing costs, and improving overall productivity.
- 4. **Innovation**: Drives innovation by uncovering new market opportunities, developing new products, and improving services.
- 5. **Risk Management**: Aids in identifying and mitigating risks by analyzing historical data patterns and predicting future outcomes.

#### **Challenges in Big Data Analytics**

- 1. **Data Quality**: Ensuring data accuracy and consistency across large datasets.
- 2. **Data Privacy and Security**: Protecting sensitive information from breaches and complying with regulations.
- 3. Scalability: Handling increasing amounts of data efficiently and cost-effectively.
- 4. **Data Integration**: Combining data from various sources and formats into a unified dataset

# CHAPTER 2 NOSQL Data Management

Dr.R.Latha

**NoSQL** (Not Only SQL) databases are a type of data management system designed to handle large volumes of unstructured, semi-structured, or structured data. Unlike traditional relational databases (SQL), NoSQL databases do not rely on a fixed schema, making them highly flexible, scalable, and capable of handling diverse data types. They are particularly well-suited for big data applications, real-time web and mobile applications, and other use cases where data requirements can change frequently.

#### Key Characteristics of NoSQL Databases

- 1. **Schema-less Design**: NoSQL databases do not require a predefined schema, allowing for the storage of different types of data without structural constraints. This flexibility is ideal for applications that require rapid development and changes.
- 2. **Horizontal Scalability**: NoSQL databases are designed to scale out by distributing data across multiple servers or nodes, which is more cost-effective than scaling up with more powerful hardware.
- 3. **High Availability and Fault Tolerance**: Many NoSQL databases are built to ensure data availability and reliability even in the case of hardware failures, using distributed architectures and replication.
- 4. **Variety of Data Models**: NoSQL databases support various data models, including document, key-value, column-family, and graph models, each tailored to different types of data and use cases.

# **Types of NoSQL Databases**

#### 1. Document Databases

- **Description**: Store data in documents (usually in JSON or BSON format). Each document can have a different structure, containing nested fields and arrays.
- **Examples**: MongoDB, CouchDB.
- **Use Cases**: Content management systems, e-commerce applications, and applications requiring flexible data models.

# 2. Key-Value Stores

- **Description**: The simplest type of NoSQL database, storing data as key-value pairs. This model is highly performant and can handle massive amounts of data.
- **Examples**: Redis, DynamoDB, Riak.
- Use Cases: Caching, session management, user profile storage, and real-time bidding.

# 3. Column-Family Stores

- **Description**: Organize data into columns and rows, similar to relational databases, but with a more flexible schema. Data is stored in a sparse matrix, allowing for highly efficient storage and retrieval.
- **Examples**: Apache Cassandra, HBase.
- Use Cases: Time-series data, real-time analytics, and logging

# **CHAPTER 3 Map Reduce Applications** Dr.R.Latha

**MapReduce** is a programming model and processing technique used to handle large-scale data processing in distributed computing environments. It was popularized by Google and is a core component of many big data frameworks, such as Apache Hadoop. MapReduce enables the processing of vast amounts of data by dividing the workload across multiple machines (nodes) in a cluster. The model is composed of two primary functions: Map and Reduce.

- **Map Function**: Processes input data (often in key-value pairs) and produces an intermediate set of key-value pairs.
- **Reduce Function**: Takes the output from the Map function, aggregates it, and produces the final result.

#### **Common Applications of MapReduce**

#### 1. Log Analysis

- Use Case: Analyzing server logs to identify trends, monitor system performance, detect anomalies, and diagnose issues.
- **How It Works**: The Map function processes logs to extract relevant information (e.g., error codes, timestamps), and the Reduce function aggregates this data to produce summaries or reports, such as the number of errors per hour or the most frequent error types.

#### 2. Web Indexing

- Use Case: Building and updating search engine indexes by processing and categorizing vast amounts of web pages.
- How It Works: The Map function tokenizes and processes web pages, assigning them to specific categories or index terms. The Reduce function then aggregates these tokens, creating an index that maps keywords to web pages.

# 3. Distributed File Processing

- Use Case: Processing large files distributed across multiple servers, such as 0 transforming or filtering datasets stored in a distributed file system (e.g., Hadoop Distributed File System - HDFS).
- How It Works: The Map function reads and processes chunks of the file, while the Reduce function consolidates the results, producing a final output file.

# 4. Social Media Analysis

- Use Case: Analyzing social media data to track sentiment, identify trending topics, or measure user engagement.
- **How It Works**: The Map function processes individual social media posts or interactions to extract sentiment scores, hashtags, or user mentions. The Reduce function aggregates these results, identifying trends or generating sentiment summa

# CHAPTER 4 Basics of Hadoop

Dr.S.Nithyanandam

**Hadoop** is an open-source framework that enables the distributed storage and processing of large datasets across clusters of computers. It was developed by the Apache Software Foundation and has become one of the foundational technologies for big data processing. Hadoop is designed to scale from a single server to thousands of machines, each offering local computation and storage.

## **Core Components of Hadoop**

Hadoop consists of four main modules that work together to store, process, and manage big data:

- 1. Hadoop Distributed File System (HDFS)
  - Description: HDFS is the storage component of Hadoop, designed to store large files across multiple machines in a distributed manner. It breaks down files into smaller blocks (default size is 128MB or 256MB) and distributes them across different nodes in a cluster.
  - Key Features:
    - **Fault Tolerance**: HDFS replicates data blocks across multiple nodes to ensure data availability even if some nodes fail.
    - **Scalability**: Can scale to store petabytes of data across thousands of machines.
    - **High Throughput**: Optimized for high-throughput access to large datasets, making it ideal for batch processing tasks.

# 2. MapReduce

- **Description**: MapReduce is the processing engine of Hadoop, providing a programming model for distributed data processing. It processes data in parallel across multiple nodes by dividing the job into smaller tasks.
- Key Features:
  - **Parallel Processing**: MapReduce divides tasks into independent subtasks, which are processed in parallel across different nodes.

# 3. Hadoop Common

- **Description**: Hadoop Common consists of the common utilities and libraries that support the other Hadoop modules. It includes necessary Java libraries, scripts, and configuration files.
- Key Features:
  - **Compatibility**: Ensures compatibility across different Hadoop modules.
  - **Utility Libraries**: Provides a set of libraries and utilities required by Hadoop modules to interact with the underlying operating system and hardware.

# CHAPTER 5 Hadoop related Tools

Dr.S.Nithyanandam

The Hadoop ecosystem includes a wide variety of tools and frameworks designed to extend Hadoop's capabilities and facilitate big data storage, processing, and analysis. These tools support various use cases, such as data ingestion, data transformation, querying, machine learning, and real-time processing. Here are some of the most commonly used Hadoop-related tools:

#### 1. **Hive**

- **Description**: Apache Hive is a data warehousing tool built on top of Hadoop that allows users to perform SQL-like queries on large datasets stored in HDFS.
- Key Features:
  - **HiveQL**: A SQL-like query language that enables users familiar with SQL to query Hadoop data.
  - Data Warehouse Functionality: Supports data summarization, analysis, and reporting.

#### 2. Pig

- **Description**: Apache Pig is a high-level scripting platform for processing and analyzing large datasets. It provides a language called Pig Latin, which abstracts the complexities of writing MapReduce programs.
- Key Features:
  - **Pig Latin**: A scripting language that is easier to write and understand than Java-based MapReduce code.
  - **Extensibility**: Supports custom functions written in Java, Python, and other languages.

#### 3. HBase

- **Description**: Apache HBase is a NoSQL database that runs on top of Hadoop and provides realtime read/write access to large datasets.
- Key Features:
  - **Column-Family Store**: Stores data in a column-oriented format, which is efficient for read and write operations on large datasets.
  - **Scalability**: Can scale horizontally across many servers to handle massive amounts of data.

#### 4. Sqoop

- **Description**: Apache Sqoop is a tool designed for efficiently transferring bulk data between Hadoop and structured data stores, such as relational databases (e.g., MySQL, Oracle).
- Key Features:
  - **Data Import/Export**: Supports importing data from relational databases into HDFS, and exporting processed data back to relational databases.
  - **Connectivity**: Integrates with most relational databases via JDBC.
  - **Incremental Loads**: Supports incremental data loads to minimize data transfer and reduce load times

# CHAPTER 6 Data Analytics Techniques

Ms.K.Jayanthi

Data analytics encompasses a wide range of techniques and methods used to analyze, interpret, and derive insights from data. These techniques are crucial in helping organizations make data-driven decisions, improve operations, enhance customer experiences, and gain a competitive edge. Here are some of the most commonly used data analytics techniques:

#### 1. Descriptive Analytics

• **Description**: Descriptive analytics is the process of analyzing historical data to identify patterns, trends, and relationships. It is used to answer questions like "What happened?" and "What is happening?" by providing summaries and visualizations of past events.

#### • Key Techniques:

- **Data Aggregation**: Summarizing data from multiple sources to provide a consolidated view.
- **Data Visualization**: Using charts, graphs, and dashboards to represent data visually, making it easier to understand.
- **Statistical Analysis**: Calculating measures such as mean, median, mode, and standard deviation to summarize data.
- Use Cases: Reporting on sales performance, customer segmentation, monitoring social media sentiment, and tracking website traffic.

#### 2. Diagnostic Analytics

- **Description**: Diagnostic analytics goes a step further than descriptive analytics by identifying the reasons behind past events. It aims to answer the question "Why did it happen?" and focuses on finding correlations, patterns, and anomalies.
- Key Techniques:
  - **Drill-Down Analysis**: Breaking down data into more granular levels to understand the root cause of an observed trend or pattern.
  - **Correlation Analysis:** Measuring the relationship between two or more variables to determine if they are associated.
  - **Data Mining**: Discovering patterns and relationships in large datasets through techniques like clustering, decision trees, and association rule mining.
- Use Cases: Identifying factors affecting customer churn, understanding product returns, and analyzing factors contributing to a decline in sales.

#### 3. Predictive Analytics

- **Description**: Predictive analytics uses historical data, statistical algorithms, and machine learning techniques to make predictions about future events. It answers questions like "What is likely to happen?" by estimating future probabilities and trends.
- Key Techniques:

# CHAPTER 7 Data Visualization

Dr.S.Nithyanandam

Data visualization is the graphical representation of information and data. By using visual elements like charts, graphs, and maps, data visualization tools provide an accessible way to see and understand trends, outliers, patterns, and insights in data. Effective data visualization helps in communicating complex information clearly and efficiently, making it easier for stakeholders to make data-driven decisions.

# Key Objectives of Data Visualization

- 1. **Simplify Complex Data**: Visualizations make it easier to understand complex data sets by transforming data into visual elements that highlight key insights.
- 2. **Identify Trends and Patterns**: Visual tools help to quickly identify patterns, trends, and correlations in data that might not be immediately obvious from raw data.
- 3. **Support Decision-Making**: By presenting data visually, decision-makers can more quickly assess situations, compare results, and make informed decisions.
- 4. **Communicate Insights Effectively**: Visuals are often more compelling than numbers alone; they can tell a story and communicate findings in a way that is easy for different audiences to understand.

# **Common Data Visualization Techniques**

#### 1. Bar Charts

- **Description**: Bar charts use rectangular bars to represent data values. The length or height of each bar is proportional to the value it represents.
- Uses: Comparing quantities across different categories (e.g., sales by region, product performance).
- **Types**: Vertical bar charts, horizontal bar charts, stacked bar charts, grouped bar charts.

# 2. Line Charts

- **Description**: Line charts display data points connected by a line, representing trends over time.
- **Uses**: Visualizing changes and trends in data over continuous periods (e.g., stock prices, website traffic over time).
- **Types**: Simple line charts, multiple line charts (for comparing multiple data series).

# 3. Pie Charts

- **Description**: Pie charts represent data as slices of a circle, where each slice is proportional to its value relative to the total.
- **Uses**: Showing the composition or percentage breakdown of a whole (e.g., market share, budget allocation).
- **Limitations**: Less effective for comparing small differences between categories; can be hard to read if there are too many

# CHAPTER 8 Big Data in Cloud

Ms.M.Jeeva

Big data refers to extremely large and complex data sets that traditional data processing tools cannot handle effectively. The cloud offers a scalable, flexible, and cost-effective environment for storing and analyzing big data. Combining big data and cloud computing provides businesses with the ability to process vast amounts of data quickly, derive insights in real-time, and improve decision-making without the need for significant on-premise infrastructure.

## **Benefits of Using Cloud for Big Data**

## 1. Scalability

- **Description**: Cloud platforms provide virtually unlimited storage and computing resources that can scale up or down as needed. This is ideal for handling the variable workloads typical of big data applications.
- **Benefit**: Organizations can process large volumes of data and accommodate peak demands without worrying about capacity constraints.

## 2. Cost-Effectiveness

- **Description**: Cloud computing follows a pay-as-you-go model, where businesses only pay for the resources they use.
- **Benefit**: Reduces capital expenditure on hardware, software, and maintenance costs associated with managing on-premises data centers.

#### 3. Flexibility and Agility

- **Description**: Cloud platforms support a wide range of data analytics tools and technologies, allowing businesses to choose the best tools for their needs.
- **Benefit**: Accelerates development and deployment of big data applications and allows quick adaptation to changing business requirements.

# 4. Accessibility and Collaboration

- **Description**: Data stored in the cloud can be accessed from anywhere, enabling remote teams to collaborate more efficiently.
- **Benefit**: Facilitates data sharing, collaborative analytics, and faster decision-making across global teams.

# 5. High Availability and Reliability

- **Description**: Cloud providers offer built-in redundancy and failover capabilities to ensure high availability and reliability of data services.
- **Benefit**: Minimizes downtime and ensures business continuity, which is critical for real-time big data applications.

#### 6. Security and Compliance

- **Description**: Major cloud providers offer advanced security features such as encryption, identity management, and compliance with various regulations (e.g., GDPR, HIPAA).
- **Benefit**: Helps organizations meet their security and regulatory requirements while still leveraging the power of the cloud.

# CHAPTER 9 Big data Security and Privacy

Ms.R.Bhanumathi

Security and privacy are critical concerns in the management of big data. As organizations collect, store, and analyze vast amounts of data, they face significant challenges in protecting this data from unauthorized access, breaches, and misuse. Ensuring that big data is secure and that privacy is maintained involves a combination of technological, organizational, and procedural measures.

#### Key Aspects of Big Data Security and Privacy

#### 1. Data Encryption

- **Description**: Encryption involves converting data into a secure format that can only be read by someone with the decryption key.
- Types:
  - At-Rest Encryption: Encrypting data stored on disk or other storage media.
  - **In-Transit Encryption**: Encrypting data while it is being transmitted over networks.
- Tools: AWS Key Management Service (KMS), Azure Key Vault, Google Cloud KMS.

#### 2. Access Control

- **Description**: Access control mechanisms ensure that only authorized users and systems can access sensitive data.
- Types:
  - Authentication: Verifying the identity of users accessing the system.
  - Authorization: Granting users specific permissions based on their roles.
- **Tools**: AWS IAM, Azure Active Directory, Google Cloud IAM.

#### 3. Data Masking and Anonymization

- **Description**: Techniques used to obscure or remove personally identifiable information (PII) from data to protect privacy.
- Techniques:
  - Data Masking: Replacing sensitive data with fictional but realistic data.
  - **Anonymization**: Removing or modifying PII so that individuals cannot be identified.
- **Tools**: Data Masking tools, anonymization libraries.

#### 4. Data Governance

- **Description**: Establishing policies and procedures to manage data integrity, quality, and compliance.
- **Components**:
  - Data Classification: Categorizing data based on its sensitivity and importance.
  - **Data Stewardship**: Assigning responsibility for managing and protecting data.
- **Tools**: AWS Lake Formation, Azure Purview, Google Cloud Data Catalog.

#### 5. Compliance and Regulations

• **Description**: Adhering to legal and regulatory requirements related to data protection and privacy.

# CHAPTER 10 Big data Applications

Ms.M.Jeeva

Big data refers to large volumes of structured and unstructured data that can be analyzed to uncover insights, trends, and patterns. The applications of big data span various industries and sectors, offering valuable insights and driving innovation. Here's an overview of how big data is applied across different domains:

## 1. Healthcare

- **Predictive Analytics**: Analyzing patient data to predict disease outbreaks, identify high-risk patients, and personalize treatment plans.
  - **Example**: Predicting patient readmission rates using historical health records.
- **Medical Imaging**: Enhancing diagnostic accuracy by analyzing large volumes of medical images using machine learning and AI.
  - **Example**: AI algorithms detecting anomalies in MRI scans.
- **Genomics**: Analyzing genetic data to understand genetic disorders and personalize medicine.
  - **Example**: Identifying genetic markers for cancer susceptibility.

# 2. Finance

- **Fraud Detection**: Monitoring transactions in real-time to detect and prevent fraudulent activities.
  - **Example**: Using machine learning models to identify unusual spending patterns.
- Risk Management: Analyzing market data to assess and manage financial risks.
  - **Example**: Predicting market crashes based on historical data and economic indicators.
- **Customer Insights**: Understanding customer behavior and preferences to enhance financial products and services.
  - Example: Personalizing investment recommendations based on customer profiles.

# 3. Retail

- **Customer Personalization**: Analyzing shopping patterns to offer personalized recommendations and promotions.
  - **Example**: Recommending products based on past purchases and browsing behavior.
- **Inventory Management**: Optimizing inventory levels and supply chain operations using predictive analytics.
  - **Example**: Forecasting demand for products to prevent stockouts and overstocking.
- Sentiment Analysis: Analyzing customer reviews and social media feedback to understand customer sentiment and improve product.


# COMPUTER VISION

# EDITED BY S.JANCY SICKORY DAISY



COMPUTER VISION

Copyright © 2021 by S.JANCY SICKORY DAISY

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-990-6

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# TABLE OF CONTENTS

Introduction	-	03
<b>CHAPTER 1</b> – Introduction to Image Formation and Processing	-	11
Dr.S.Nithyanandam		
<b>CHAPTER 2</b> – Feature Detection, Matching and Segmentation	-	47
Dr.S.Nithyanandam		
CHAPTER 3 – Feature- Based Alignment & Motion Estimation	-	67
Dr.S.Nithyanandam		
<b>CHAPTER 4</b> – 3D Reconstruction	-	89
Ms.S.Gayathiri		
<b>CHAPTER 5</b> – Image Based Rendering and Recognition	-	113
Ms.S.Gayathiri		
CHAPTER 6 – Image Segmentation	-	127
Ms.K.Jayanthi		
<b>CHAPTER 7</b> – Facical Recognition and Biometrics	-	147
Dr.S.Nithyanandam		
CHAPTER 8 – Generative models in Computer Vision	-	187
Ms.M.Jeeva		
<b>CHAPTER 9</b> – Deep Learning for Computer Vision	-	215
Ms.R.Bhanumathi		
<b>CHAPTER 10</b> – Computer Vision for Robotics	-	257
Ms.S.Gayathiri		
REFERENCES	-	285

# CHAPTER 1 Introduction to Image Formation and Processing

Dr.S.Nithyanandam

**Image Formation and Processing** are fundamental concepts in computer vision that deal with how images are created, represented, and manipulated to extract useful information. This involves understanding the physics of how images are captured, the mathematical models used to describe images, and the techniques applied to process these images for various applications.

### Image Formation

Image formation is the process by which a 2D image is created from a 3D scene. This involves understanding the physics of light, the geometry of the scene, and the characteristics of the camera or sensor capturing the image.

Key Concepts in Image Formation:

### 1. Light and Color:

- **Light** is the fundamental medium through which images are formed. When light interacts with objects, it is reflected, absorbed, or transmitted, and the resulting light is captured by imaging sensors.
- Color perception is determined by the wavelength of light that reaches the imaging sensor. The primary colors (Red, Green, and Blue) are combined in different intensities to produce a full spectrum of colors.

### 2. Camera Models:

- A camera model is a mathematical representation of the image formation process. The most common model is the **pinhole camera model**.
- Pinhole Camera Model:
  - Represents a simple camera as a pinhole that allows light to pass through and form an inverted image on a 2D plane (image sensor or film).
  - **Camera Projection**: The 3D points of the scene are projected onto a 2D image

### Image Processing Techniques

Image processing involves applying algorithms and techniques to manipulate images, extract useful information, and enhance their quality.

Key Image Processing Techniques:

# **Image Filtering:**

- **Smoothing Filters**: Used to reduce noise and smooth images (e.g., Gaussian blur, average filtering).
- **Sharpening Filters**: Enhance edges and fine details in an image (e.g., Laplacian, Unsharp Masking).

# **Feature Detection, Matching and Segmentation**

Dr.S.Nithyanandam

## **Feature Detection**

**Feature Detection** involves identifying distinct and meaningful points or areas (features) in an image that can be used for further analysis. These features often correspond to corners, edges, blobs, or other distinctive patterns in the image.

### Types of Features:

- 1. Corners:
  - Corners are points in an image where two or more edges meet. They are good for feature detection because they are distinctive and invariant to rotation and scaling.
  - **Common Algorithms:** 
    - Harris Corner Detector: Measures changes in intensity in all directions within a small window. A point is considered a corner if there is a large variation in all directions.
    - Shi-Tomasi Corner Detector (Good Features to Track): An improvement over Harris, which selects the most prominent corners based on the eigenvalues of the structure tensor.

# 2. Edges:

- Edges are curves in an image along which there is a significant change in intensity. They often correspond to object boundaries.
- **Common Algorithms**:
  - **Sobel and Prewitt Operators:** Compute gradients of the image intensity, used to detect edges by finding regions with high spatial frequency.
  - Canny Edge Detector: A multi-stage process that includes noise reduction, gradient calculation, non-maximum suppression, and edge tracking by hysteresis. It is widely used due to its good performance in detecting edges.

# 3. Blobs:

- Blobs are regions in an image that are either brighter or darker than the surrounding area. They are useful for detecting objects or patterns with some area.
- Common Algorithms:
  - Laplacian of Gaussian (LoG): Detects blobs by looking for regions in an image where the second derivative (Laplacian) of the intensity has a peak.
  - **Difference of Gaussians (DoG):** An approximation of LoG that uses two Gaussian filters of different scales.

# 4. Feature Descriptors:

- Descriptors provide a way to describe detected features in a compact and distinctive manner, allowing for robust matching between different images.
- Common Feature Descriptors:
  - SIFT (Scale-Invariant Feature Transform):

# Feature- Based Alignment & Motion Estimation

Dr.S.Nithyanandam

## Feature-Based Alignment

Feature-based alignment involves finding correspondences between images based on detected features (keypoints) and aligning them to a common coordinate system or frame. This process is essential for tasks such as image stitching, panorama creation, and 3D reconstruction.

Steps in Feature-Based Alignment:

# 1. Feature Detection:

- Detect keypoints in both images using feature detectors such as SIFT (Scale-Invariant Feature Transform), SURF (Speeded-Up Robust Features), ORB (Oriented FAST and Rotated BRIEF), or Harris Corner Detector.
- These detectors identify distinctive points in an image that are invariant to transformations like rotation, scaling, and partial illumination changes.

# 2. Feature Description:

 Compute descriptors for the detected keypoints using algorithms like SIFT, SURF, or ORB. Descriptors encode the local appearance around each keypoint into a vector, facilitating robust matching.

# **Motion Estimation**

Motion estimation is the process of determining the movement of objects or the camera between consecutive frames in a video sequence. It is a key component in video compression, frame interpolation, object tracking, and robotics.

Types of Motion Estimation:

# 1. Optical Flow-Based Motion Estimation:

- **Optical Flow** is a vector field that represents the apparent motion of objects in a scene between consecutive frames of a video.
- Methods:
  - Lucas-Kanade Method: A differential method that assumes small motion between frames and a constant flow in a local neighborhood. It uses the brightness constancy constraint to estimate motion.
  - Horn-Schunck Method: An energy-based method that minimizes the difference in pixel intensities between frames while enforcing smoothness in the motion

# 2. Block Matching-Based Motion Estimation:

- Divides the image into blocks and searches for the best match for each block in the subsequent frame, often used in video compression (e.g., MPEG, H.264).
- Methods:

# **3D Reconstruction**

### Ms.S.Gayathiri

**3D Reconstruction** is a process of creating a three-dimensional model or representation of a physical object or scene from a set of 2D images or data. It involves capturing spatial information and using it to build a digital 3imensional model. This technique is widely used in various fields such as robotics, computer graphics, virtual reality, and medical imaging.

Basics of 3D Reconstruction

**3D Reconstruction** can be broadly categorized into several methods, each with different techniques and applications. The primary goal is to derive 3D information from 2D data sources.

#### Key Concepts:

### Stereo Vision:

- Uses two or more cameras to capture images of the same scene from different viewpoints. By comparing these images, depth information can be inferred.
- **Stereo Matching**: Computes disparity maps, which represent the difference in position of corresponding points in the stereo images. The disparity map is then used to calculate the depth of each point in the scene.

### **Structure from Motion (SfM):**

- Reconstructs 3D structures from a series of 2D images taken from different viewpoints.
- Process:
  - **Feature Detection and Matching:** Detect and match keypoints across multiple images.
  - **Camera Pose Estimation:** Estimate the position and orientation of the camera for each image using the matched features.
  - **3D Point Cloud Generation:** Triangulate the matched points to reconstruct a 3D point cloud representing the scene.

### **Multi-View Stereo (MVS):**

- Extends the idea of stereo vision to multiple images to produce more detailed and accurate 3D reconstructions.
- Process:
  - **Dense Matching:** Computes depth maps for each image by matching features densely across multiple views.

# **Image Based Rendering and Recognition**

Ms.S.Gayathiri

# Image-Based Rendering (IBR)

Image-Based Rendering involves generating new views or representations of a scene based on a set of input images. Instead of creating a 3D model of the scene, IBR uses the images directly to synthesize new views.

Key Techniques in IBR:

### 1. View Synthesis:

- **Image Warping:** Adjusts images to create a new view by transforming and blending the input images.
- **Example:** Creating a panoramic image from multiple photos taken from different angles.

# 2. Image-Based Modeling:

- **Multi-View Stereo (MVS):** Uses multiple images to create a dense 3D reconstruction of a scene. The 3D model can then be used to render new views.
- **Example:** Generating a 3D model of a building from several photographs taken from different perspectives.

# 3. Light Field Rendering:

- **Light Field Cameras:** Capture light rays from multiple directions, allowing for the reconstruction of 3D scenes and refocusing after capture.
- **Example:** Refocusing and changing the viewpoint of a photograph taken with a light field camera.

# Image-Based Recognition

Image-Based Recognition involves identifying and classifying objects, scenes, or patterns within images. It encompasses various techniques and methods for understanding and interpreting visual data.

Key Techniques in Image-Based Recognition:

# 1. **Object Detection:**

- **Bounding Boxes:** Identifies the location and size of objects within an image.
- Algorithms:
  - YOLO (You Only Look Once): A real-time object detection system that detects multiple objects in an image in one pass.
  - Faster R-CNN: Combines Region Proposal Networks (RPN) with Fast R-CNN for accurate object detection.
  - **SSD (Single Shot MultiBox Detector):** Detects objects using a single neural network that outputs bounding boxes and class scores.

# 2. Object Classification:

• Image Classification: Assigns a label to an entire image based on its content.

# **Image Segmentation**

# Ms.K.Jayanthi

**Image Segmentation** is a fundamental technique in computer vision that involves partitioning an image into multiple segments or regions, making it easier to analyze and understand. Each segment typically represents a different object, part of an object, or a distinct area in the image. The goal is to simplify the representation of an image or to make it more meaningful and easier to analyze.

# Types of Image Segmentation

Thresholding:

- Basic Thresholding:
  - Converts an image into a binary image based on a pixel intensity threshold.
  - **Example:** Segmenting foreground from background in a grayscale image.
- Adaptive Thresholding:
  - Adjusts the threshold value based on local properties of the image, such as the mean or median intensity of a pixel's neighborhood.
  - **Example:** Handling varying lighting conditions across different regions of an image.
- Otsu's Method:
  - An automatic thresholding technique that calculates the optimal threshold value to minimize the intra-class variance of pixel intensities.
  - **Example:** Automatically segmenting objects from the background in an image.

Edge-Based Segmentation:

- Edge Detection:
  - Identifies boundaries or edges within an image where there is a significant change in intensity.
  - Algorithms:
    - **Canny Edge Detector:** Detects edges by looking for local maxima of the gradient of the image intensity.
    - **Sobel Operator:** Computes gradients in horizontal and vertical directions to detect edges.
- Edge Linking and Boundary Detection:
  - Connects detected edges to form closed boundaries of objects or regions.
  - **Example:** Detecting and outlining objects in an image using edge information.

Region-Based Segmentation:

- Region Growing:
  - Starts with a seed point and grows the region by adding neighboring pixels that have similar properties.

# **Facical Recognition and Biometrics**

## Dr.S.Nithyanandam

**Facial Recognition** and **Biometrics** are technologies used to identify and authenticate individuals based on unique physiological or behavioral characteristics. These technologies have applications in security, access control, personal identification, and more

**Facial Recognition** 

**Facial Recognition** is a biometric method that identifies or verifies individuals based on their facial features. It involves capturing and analyzing facial characteristics to match or identify a person.

Key Concepts in Facial Recognition:

# 1. Face Detection:

- **Objective:** Locate and identify human faces within images or video frames.
- Algorithms:
  - **Haar Cascades:** A machine learning object detection method used for detecting faces in images.
  - Single Shot MultiBox Detector (SSD): Detects faces using deep learning with a single convolutional network.

# 2. Feature Extraction:

- **Objective:** Extract distinguishing features from detected faces for recognition purposes.
- Methods:
  - Landmark-Based Methods: Identify key facial landmarks (e.g., eyes, nose, mouth) to represent facial features.
  - •

# 3. Deep Learning Models:

- FaceNet: Embeds facial features into a vector space for recognition and verification.
- **OpenFace:** A deep learning model for facial landmark detection and recognition.

# . Biometrics

**Biometrics** refers to the measurement and analysis of unique physical or behavioral characteristics for the purpose of identification and authentication.

Types of Biometrics:

# 1. Physiological Biometrics:

- **Fingerprint Recognition:** 
  - **Objective:** Identifies individuals based on the unique patterns of ridges and valleys on their fingerprints.

# **Generative models in Computer Vision**

#### Ms.M.Jeeva

Generative models are a class of machine learning models designed to generate new data samples that resemble a given training dataset. In computer vision, generative models are used to create realistic images, modify existing images, or simulate new visual data. These models learn the underlying distribution of the data and can generate new instances that follow this distribution.

#### .Generative Adversarial Networks (GANs):

- Concept:
  - GANs consist of two neural networks: a Generator and a Discriminator. The Generator creates new data samples, while the Discriminator evaluates them against real data. The two networks are trained together in a game-theoretic framework.
  - **Generator:** Produces synthetic data samples (e.g., images) to deceive the Discriminator.
  - **Discriminator:** Distinguishes between real data samples and synthetic ones produced by the Generator.
- Applications:
  - Image Generation: Creating realistic images from random noise or input conditions.
  - **Image-to-Image Translation:** Converting images from one domain to another (e.g., day-to-night, sketch-to-photo).
  - **Super-Resolution:** Enhancing the resolution of images.
  - **Style Transfer:** Applying artistic styles to images.
- Popular Architectures:
  - **DCGAN (Deep Convolutional GAN):** Uses convolutional layers to improve image quality and stability.
  - **CycleGAN:** Performs image-to-image translation without paired examples (e.g., translating paintings to photos).
  - **StyleGAN:** Generates high-quality images with control over various attributes (e.g., facial expressions, hairstyles).

#### Variational Autoencoders (VAEs):

- Concept:
  - VAEs are probabilistic models that learn to encode data into a latent space and then decode it back to the original data space. They optimize the balance between reconstruction accuracy and the regularity of the latent space.
  - Encoder: Maps input data to a latent space represented by a probability distribution.
  - **Decoder:** Reconstructs data from samples drawn from the latent space distribution.
- Applications:
  - Image Generation: Creating new images from latent space samples.
  - **Data Denoising:** Removing noise from images by reconstructing clean data.
  - Inpainting: Filling in missing parts of images.
  - Feature Extraction: Learning useful features for downstream tasks.

# **Deep Learning for Computer Vision**

Ms.R.Bhanumathi

**Deep Learning** has revolutionized computer vision by providing powerful techniques for understanding and analyzing visual data. Deep learning models, particularly Convolutional Neural Networks (CNNs), have become the cornerstone of modern computer vision systems due to their ability to automatically learn and extract features from images. This section covers the fundamental concepts, architectures, applications, and advancements in deep learning for computer vision.

Convolutional Neural Networks (CNNs):

- Convolution Layers:
  - Apply convolutional filters to input images to detect local patterns (e.g., edges, textures).
  - Filters slide over the image, performing element-wise multiplication and summing the results to create feature maps.
- Pooling Layers:
  - Reduce the spatial dimensions of feature maps while retaining important information.
  - Common pooling operations include Max Pooling (taking the maximum value) and Average Pooling (taking the average value).
- Fully Connected Layers:
  - Connect every neuron in one layer to every neuron in the next layer, used for final classification or regression tasks after feature extraction.
- Activation Functions:
  - Introduce non-linearity into the model. Common activation functions include ReLU (Rectified Linear Unit), Sigmoid, and Tanh.

. Training Deep Learning Models:

- Loss Functions:
  - Measure the difference between predicted and actual values. Common loss functions include Cross-Entropy Loss (for classification) and Mean Squared Error (for regression).
- Optimization Algorithms:
  - Adjust model parameters to minimize the loss function. Popular algorithms include Stochastic Gradient Descent (SGD), Adam, and RMSprop.
- Regularization Techniques:
  - Prevent overfitting and improve generalization. Techniques include Dropout, L2 Regularization (weight decay), and Data Augmentation.
- Batch Normalization:

#### **Computer Vision for Robotics**

#### Ms.S.Gayathiri

**Computer Vision for Robotics** is a field that combines computer vision and robotics to enable robots to perceive, understand, and interact with their environment. By leveraging visual information, robots can perform tasks such as navigation, object manipulation, and human-robot interaction. This integration is crucial for developing autonomous systems capable of operating in complex, dynamic environments.

Image Acquisition:

- Cameras:
  - **RGB Cameras:** Capture color images, commonly used for general visual information.
  - **Depth Cameras (e.g., Kinect, Intel RealSense):** Capture depth information to create 3D representations of the environment.
  - **Stereo Cameras:** Use two or more cameras to capture depth information through disparity between images.
- Sensors:
  - LiDAR (Light Detection and Ranging): Provides detailed 3D maps of the environment using laser pulses.
  - **Radar:** Uses radio waves to detect objects and measure their distance.

Image Processing:

- Preprocessing:
  - **Noise Reduction:** Techniques like Gaussian blur or median filtering to remove noise from images.
  - **Image Enhancement:** Adjusting brightness, contrast, or sharpness to improve image quality.
- Feature Extraction:
  - Edge Detection: Identifies boundaries within images using techniques like Canny or Sobel filters.
  - **Feature Descriptors:** Extracts features such as SIFT, SURF, or ORB for object recognition and matching.

Object Detection and Recognition:

- Object Detection:
  - **Bounding Boxes:** Identifies the location of objects in an image using rectangular boxes.
  - Algorithms:
    - YOLO (You Only Look Once): Real-time object detection that processes images in a single pass.
    - SSD (Single Shot MultiBox Detector): Detects objects at multiple scales and aspect ratios in one go.

.

# IMAGE AND VIDEO ANALYTICS

Edited by DR.S.NITHYANANDAM

THE

C. He



IMAGE AND VIDEO ANALYTICS
Copyright © 2021 by Dr.S.NITHYANANDAM

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-079-8

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# TABLE OF CONTENTS

Introduction	-	03
<b>CHAPTER 1</b> – Introduction of Image and Video Analytics	-	11
Ms.K.Sangeetha		
CHAPTER 2 – Image Pre Processing	-	47
Ms.K.Uma Kumar		
<b>CHAPTER 3</b> – Feature Detection and Description	-	67
Dr.R.Latha		
<b>CHAPTER 4</b> – Object Detection Using Machine Learning	-	89
Ms.V.Gayathiri		
<b>CHAPTER 5</b> – Face Recognition and Gesture Recongnition	-	113
Ms.K.Jayanthi		
<b>CHAPTER 6</b> – Deep Learning for Image and Video Analytics	-	127
Ms.K.Jayanthi		
<b>CHAPTER 7</b> – Scale – Invariant Feature Transform	-	147
Dr.S.Nithyanandam		
CHAPTER 8 – Supervised Learning for Classification	-	187
Ms.M.Jeeva		
CHAPTER 9 – Video Analytics	-	215
Ms.R.Bhanumathi		
CHAPTER 10 – Applications and Case Studies	-	257
Ms.R.Bhanumathi		

# REFERENCES

285

-

# **Introduction of Image and Video Analytics**

# Ms.K.Sangeetha

**Image and Video Analytics** is a field within computer vision that involves extracting meaningful information from still images and video sequences. It combines techniques from image processing, machine learning, and data analysis to understand, interpret, and act upon visual data. This field plays a crucial role in various applications, from security and surveillance to entertainment and healthcare.

### **Image Analytics:**

- Image Classification:
  - **Objective:** Assigns a label or category to an image based on its content.
  - **Techniques:** Uses convolutional neural networks (CNNs) and other machine learning models to classify images into predefined classes.

# • Object Detection:

- **Objective:** Identifies and locates objects within an image using bounding boxes or segmentation masks.
- **Techniques:** Employs algorithms such as YOLO (You Only Look Once), SSD (Single Shot MultiBox Detector), and Faster R-CNN.
- Image Segmentation:
  - **Objective:** Divides an image into regions or segments, each corresponding to a different object or part of the scene.
  - **Techniques:** Utilizes semantic segmentation (classifying each pixel) and instance segmentation (detecting individual objects).

Video Analytics:

- Video Classification:
  - **Objective:** Assigns labels to entire video sequences based on their content or activities.
  - **Techniques:** Uses deep learning models, such as 3D CNNs or two-stream networks, to analyze temporal patterns and video features.

# • Object Tracking:

- **Objective:** Continuously tracks the movement of objects across frames in a video sequence.
- **Techniques:** Utilizes algorithms like Kalman Filters, Mean-Shift, and more advanced tracking methods like SORT (Simple Online and Realtime Tracking) and Deep SORT.

### • Action and Activity Recognition:

- **Objective:** Identifies and classifies actions or activities performed by people or objects in a video.
- **Techniques:** Employs temporal convolutional networks, recurrent neural networks (RNNs), and long short-term memory (LSTM) networks.

### • Event Detection:

• **Objective:** Detects specific events or anomalies within video sequences, such as unusual behavior or incidents.

# **Image Pre Processing**

# Ms.K.Uma Kumar

**Image Pre-Processing** refers to the techniques used to prepare and enhance images before they are analyzed or processed by algorithms. It involves improving the quality of images, removing noise, and standardizing data to facilitate better performance in downstream tasks such as image classification, object detection, and segmentation. Effective pre-processing can significantly impact the accuracy and efficiency of computer vision systems.

Common Pre-Processing Techniques

Noise Reduction:

- Gaussian Blur:
  - **Objective:** Smooths the image by averaging pixel values with a Gaussian kernel, reducing high-frequency noise.
  - **Application:** Useful for removing Gaussian noise and smoothing out small variations in an image.
- Median Filtering:
  - **Objective:** Replaces each pixel's value with the median of the pixel values in its neighborhood.
  - Application: Effective for removing salt-and-pepper noise and preserving edges.
- Bilateral Filtering:
  - **Objective:** Smooths the image while preserving edges by considering both spatial distance and pixel value differences.
  - Application: Suitable for edge-preserving noise reduction.

. Contrast Enhancement:

- Histogram Equalization:
  - **Objective:** Enhances image contrast by redistributing the intensity levels to cover the full range of values.
  - Application: Improves visibility of details in images with poor contrast.
- Adaptive Histogram Equalization:
  - **Objective:** Applies histogram equalization locally to enhance contrast in different regions of the image.

Image Resizing:

- Scaling:
  - **Objective:** Adjusts the size of an image to a desired resolution or aspect ratio.
  - **Application:** Necessary for standardizing input sizes for machine learning models or fitting images into a specific display.

# **Feature Detection and Description**

# Dr.R.Latha

**Feature Detection and Description** are fundamental processes in computer vision and image analysis. They involve identifying and characterizing distinctive patterns or points in images, which can then be used for various tasks such as object recognition, image matching, and scene understanding. These processes are crucial for applications like image stitching, object tracking, and augmented reality.

- **Distinctiveness:** Features should be easily distinguishable from the surrounding area.
- **Repeatability:** Features should be identifiable in different images or under different conditions (e.g., lighting, rotation).
- **Robustness:** Features should be stable under transformations such as scaling, rotation, and affine changes.

# . Common Feature Detection Techniques:

- Corners:
  - **Harris Corner Detector:** Detects corners by evaluating the change in intensity around a point. It uses the eigenvalues of the autocorrelation matrix to identify corner-like structures.
  - Shi-Tomasi Detector: An improvement over Harris, focusing on selecting the most stable corners based on eigenvalues.
- Edges:
  - **Canny Edge Detector:** Detects edges by finding areas of rapid intensity change using gradient-based techniques and non-maximum suppression.
  - **Sobel Operator:** Computes gradient magnitude and direction to detect edges, using convolution with Sobel kernels.
- Blob Detection:
  - **Laplacian of Gaussian (LoG):** Detects blobs by identifying regions where the Laplacian of the Gaussian function has significant responses.
  - **Difference of Gaussians (DoG):** Approximates LoG by subtracting blurred images at different scales.
- Keypoints:
  - Scale-Invariant Feature Transform (SIFT): Detects keypoints by identifying stable features across different scales and orientations. It extracts keypoints based on local extrema in the Difference of Gaussians.
  - **Speeded-Up Robust Features (SURF):** A faster alternative to SIFT that uses an approximate method for feature detection and description.
- Other Techniques:
  - **FAST (Features from Accelerated Segment Test):** A high-speed corner detection method that evaluates corner-like features using a binary test.
  - **ORB (Oriented FAST and Rotated BRIEF):** Combines FAST keypoint detection with BRIEF descriptor, providing fast and robust feature detection.

# CHAPTER 4 Object Detection Using Machine Learning

Ms.V.Gayathiri

**Object Detection** is a computer vision task that involves identifying and locating objects within an image or video. Unlike image classification, which assigns a label to the entire image, object detection provides both the category of objects and their spatial locations, typically represented by bounding boxes. Machine learning, particularly deep learning, has significantly advanced object detection by providing more accurate and efficient methods.

# **Object Detection Pipeline:**

# 1. Feature Extraction:

- **Objective:** Extract meaningful features from the image that can be used to identify objects.
- **Techniques:** Utilize convolutional neural networks (CNNs) to automatically learn and extract hierarchical features.

# 2. Object Localization:

- **Objective:** Determine the location of objects within the image.
- **Techniques:** Use bounding boxes to represent object locations.

# 3. Object Classification:

- **Objective:** Classify the detected objects into predefined categories.
- Techniques: Apply classifiers to the extracted features or regions of interest.
- 4. Bounding Box Regression:
  - **Objective:** Refine the predicted bounding boxes to improve localization accuracy.
  - Techniques: Use regression models to adjust the coordinates of the bounding boxes.

# **Common Object Detection Frameworks:**

- Region-Based CNN (R-CNN):
  - **Overview:** Introduced by Ross Girshick et al., R-CNN uses selective search to generate region proposals and applies CNNs to extract features for each region.
  - Limitations: Slow inference time due to independent processing of region proposals.
- Fast R-CNN:
  - **Overview:** An improvement over R-CNN, Fast R-CNN integrates region proposals and feature extraction into a single network.
  - Advantages: Faster and more efficient by sharing convolutional layers across all regions.
- Faster R-CNN:
  - **Overview:** Introduced by Shaoqing Ren et al., Faster R-CNN adds a Region Proposal Network (RPN) to generate region proposals within the network itself.
  - Advantages: Further improves speed and accuracy by eliminating the need for external region proposal algorithms.
- YOLO (You Only Look Once):
  - **Overview:** A real-time object detection framework by Joseph Redmon et al., YOLO frames object detection as a single regression problem.

# CHAPTER 5 Face Recognition and Gesture Recongnition

Ms.K.Jayanthi

**Face Recognition** is the process of identifying or verifying a person based on their facial features. It involves detecting, analyzing, and matching facial characteristics to determine the **identity of individuals.** 

# **Face Recognition Pipeline**

- 1. Face Detection:
  - **Objective:** Locate and identify face regions within an image.
  - **Techniques:** Use algorithms like Haar cascades, MTCNN (Multi-task Cascaded Convolutional Networks), or SSD (Single Shot Multibox Detector).
- 2. Feature Extraction:
  - **Objective:** Extract distinctive features from the detected face.
  - **Techniques:** Use deep learning models such as CNNs (Convolutional Neural Networks) to extract embeddings or feature vectors representing facial characteristics.

**Common Face Recognition Algorithms** 

- Eigenfaces (PCA-based):
  - **Overview:** Uses Principal Component Analysis (PCA) to represent faces in a lower-dimensional space.
  - Advantages: Effective for capturing the most significant variations in face images.
- Fisherfaces (LDA-based):
  - **Overview:** Applies Linear Discriminant Analysis (LDA) to maximize class separability in the feature space.
  - Advantages: Improves classification performance by focusing on discriminative features.

# **Gesture Recognition**

**Gesture Recognition** involves interpreting human gestures (e.g., hand movements, body postures) to enable interaction with computers or devices. It enables intuitive control and interaction by translating physical movements into digital commands.

Gesture Recognition Pipeline

### 1. Gesture Detection:

- **Objective:** Identify and track hand or body gestures in real-time.
- **Techniques:** Use computer vision methods (e.g., background subtraction, optical flow) or sensor-based approaches (e.g., depth cameras, IMUs).

### 2. Feature Extraction:

- **Objective:** Extract relevant features from the detected gestures.
- **Techniques:** Use hand landmarks, pose estimation, or motion patterns as features.

# CHAPTER 6 Deep Learning for Image and Video Analytics

Ms.K.Jayanthi

Deep learning has revolutionized image and video analytics by providing powerful tools for extracting insights from visual data. Leveraging neural networks, particularly convolutional neural networks (CNNs) and recurrent neural networks (RNNs), deep learning models have achieved state-of-the-art performance in tasks like object detection, image classification, video analysis, and more.

#### Image Classification

**Image Classification** involves categorizing an image into predefined classes. Deep learning models have significantly improved accuracy and efficiency in image classification tasks.

### • Convolutional Neural Networks (CNNs):

- **Overview:** CNNs are designed to automatically and adaptively learn spatial hierarchies of features through convolutional layers.
- **Popular Architectures:** 
  - AlexNet: Pioneered deep learning in image classification by winning the ImageNet competition in 2012.
  - **VGGNet:** Known for its simplicity and depth, using small convolutional filters and deep layers.
  - **ResNet:** Introduced residual connections to address the vanishing gradient problem and allow for very deep networks.
  - **Inception:** Uses multiple filter sizes and pooling operations in parallel to capture different types of features.

### • Transfer Learning:

- **Overview:** Uses pre-trained models (e.g., VGG, ResNet) as a starting point and finetunes them on specific tasks.
- **Benefits:** Reduces training time and improves performance with less data.

### **Object Detection**

**Object Detection** involves identifying and locating objects within an image by drawing bounding boxes around them.

- Region-Based CNN (R-CNN):
  - **Overview:** Uses selective search to generate region proposals and applies CNNs to classify these regions.
  - **Limitations:** Computationally intensive and slow due to independent processing of each region.
- Fast R-CNN:
  - **Overview:** Improves R-CNN by sharing convolutional features across regions, leading to faster and more accurate detection.
  - Method: Integrates region proposals and object classification in a single network.

## **CHAPTER 7** Scale – Invariant Feature Transform Dr.S.Nithyanandam

**Scale-Invariant Feature Transform (SIFT)** is a widely used algorithm in computer vision for detecting and describing local features in images. Developed by David Lowe in 1999, SIFT is particularly effective for tasks that involve recognizing objects across different scales and orientations, making it robust to changes in scale, rotation, and partial occlusion.

# SIFT Algorithm Steps

# 1. Scale-space Extrema Detection:

- **Objective:** Detect keypoints at different scales in the image.
- **Method:** Construct a scale-space representation by progressively smoothing the image with Gaussian filters and subtracting adjacent blurred images to obtain Difference of Gaussians (DoG). Identify keypoints as local maxima and minima in the DoG images.

# 2. Keypoint Localization:

- **Objective:** Refine the location of keypoints and discard low-contrast or poorly localized keypoints.
- **Method:** Fit a 3D quadratic function to the local image data around each keypoint to determine the precise location and scale. Eliminate keypoints with low contrast or edge responses.

# **SIFT Keypoint Descriptor**

- **Histogram of Oriented Gradients (HOG):** The descriptor is formed by dividing the local region around the keypoint into a grid of cells and computing histograms of gradient orientations within each cell.
- **Descriptor Vector:** The descriptor vector is typically 128-dimensional, representing the gradient orientation histograms in a 4x4 grid of 8-bin histograms.

Applications of SIFT

- Object Recognition:
  - **Purpose:** Identify and recognize objects in images by matching keypoints between different images.
  - **Example:** Recognize objects in cluttered scenes or different viewpoints.
- Image Stitching:
  - **Purpose:** Combine multiple images into a single panorama by finding matching keypoints and aligning them.
  - **Example:** Create wide-angle or panoramic images from overlapping photo captures.
  - **Example:** Place virtual objects in real-world environments in AR applications.

Advantages of SIFT

• **Robustness:** Effective in handling changes in scale, rotation, and partial occlusions.

# **Supervised Learning for Classification**

# Ms.M.Jeeva

**Supervised learning** is a type of machine learning where the model is trained on labeled data to make predictions or decisions based on new, unseen data. **Classification** is a specific type of supervised learning task where the goal is to categorize input data into predefined classes or labels.

# **Classification Algorithms**

# . Logistic Regression

- **Overview:** A linear model used for binary classification problems.
- **Method:** Estimates probabilities using the logistic function and classifies based on a threshold (e.g., 0.5).
- Advantages: Simple and interpretable, works well with linearly separable data.
- Disadvantages: Limited to linear decision boundaries.

### **Decision Trees**

- **Overview:** A tree-like model used for both classification and regression tasks.
- Method: Splits the data into subsets based on feature values, creating a tree structure with nodes representing decisions.
- Advantages: Easy to interpret, handles both numerical and categorical data.
- **Disadvantages:** Prone to overfitting, especially with deep trees.

### **Random Forests**

- **Overview:** An ensemble method that combines multiple decision trees to improve performance.
- **Method:** Constructs multiple decision trees and aggregates their predictions (e.g., majority voting) for classification.
- Advantages: Reduces overfitting, improves accuracy, handles large datasets well.
- **Disadvantages:** Less interpretable than individual decision trees, requires more computational resources.

### Support Vector Machines (SVM)

- **Overview:** A powerful classifier that finds the optimal hyperplane separating different classes.
- Method: Maximizes the margin between classes using kernel functions to handle non-linear boundaries.
- Advantages: Effective in high-dimensional spaces, handles non-linear data with kernel trick.
- **Disadvantages:** Computationally expensive for large datasets, requires careful tuning of hyperparameters.

# **CHAPTER 9** Video Analytics

Ms.R.Bhanumathi

Video analytics involves extracting meaningful information from video data using various techniques in computer vision, machine learning, and data processing. It enables the analysis of video content to identify patterns, detect events, and gain insights. Applications span across different fields, including security, transportation, retail, and entertainment.

# **Object Detection and Tracking**

- **Object Detection:** Identifies and locates objects within each frame. •
  - **Techniques:** 0
    - YOLO (You Only Look Once): Provides real-time object detection by • predicting bounding boxes and class labels.
    - SSD (Single Shot MultiBox Detector): Detects objects at multiple scales within a single network.
    - Faster R-CNN: Combines region proposals with CNNs for accurate detection.

# **Action Recognition**

- Action Recognition: Identifies and classifies actions or activities occurring in the video.
  - **Techniques:** 0
    - 3D Convolutional Networks (3D-CNNs): Extends 2D convolutions to the temporal dimension to capture motion patterns.
    - Two-Stream Networks: Uses separate streams for spatial and temporal features, fusing them for action recognition.

# **Event Detection**

- Event Detection: Identifies specific events or anomalies within the video.
  - **Techniques:** 0
    - Autoencoders: Learn normal patterns and detect anomalies based on reconstruction errors.
    - Temporal Convolutional Networks (TCNs): Model temporal dependencies to identify events.

### **Scene Understanding**

- Scene Understanding: Analyzes and interprets the overall scene in a video, including object interactions and scene context.
  - 0 **Techniques:** 
    - Semantic Segmentation: Assigns class labels to each pixel in the frame to understand scene composition.
    - Instance Segmentation: Identifies and segments individual objects within the scene.

### **Applications and Case Studies**

#### Ms.R.Bhanumathi

# **Surveillance and Security**

Intrusion Detection

- Application: Detect unauthorized access to restricted areas.
- **Case Study:** A major airport implemented video analytics to monitor restricted zones. The system detected unauthorized personnel and triggered alarms, significantly reducing security breaches and improving response times.

**Facial Recognition** 

- Application: Identify individuals from surveillance footage.
- **Case Study:** In a large city, facial recognition systems were deployed across public spaces to identify known criminals. The system helped law enforcement track suspects and solve cases more efficiently.

**Behavioral Analysis** 

- Application: Monitor unusual behavior and potential threats.
- **Case Study:** A retail chain used video analytics to monitor customer behavior, detecting suspicious activities such as loitering and potential theft. The system improved store security and reduced loss.

# **Traffic Management**

### **Traffic Flow Monitoring**

- Application: Manage and optimize traffic flow.
- **Case Study:** In a metropolitan area, video analytics were used to monitor traffic patterns and optimize traffic light timings. The system reduced congestion and improved traffic flow, leading to decreased commute times.

Automatic Number Plate Recognition (ANPR)

- Application: Identify and track vehicles based on license plates.
- **Case Study:** A city implemented ANPR to enforce parking regulations. The system automatically detected violations and issued tickets, improving parking management and compliance.
- . Incident Detection
  - Application: Detect and respond to traffic accidents and incidents.

# BUSINESS ANALYTICS

EDITED BY

**S.JANCY SICKORY DIASY** 





BUSINESS ANALYTICS Copyright © 2021 by S.JANCY SICKORY DAISY

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-589-2

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

# **TABLE OF CONTENTS**

Introduction	-	03
<b>CHAPTER 1</b> – Introduction to Business Analytics	-	11
Mr.A.Jahir Husain		
CHAPTER 2 – Data Management For Analytics	-	47
Ms.K.Jayanthi		
CHAPTER 3 – Data Analysis Techniques	-	67
Dr.R.Latha		
CHAPTER 4 – Business Intelligence	-	89
Ms.K.Jayanthi		
<b>CHAPTER 5</b> – Business Intelligence Tools and Technologies	-	113
Ms.K.Jayanthi		
CHAPTER 6 – Business ForeCasting	-	127
Ms.K.Jayanthi		
<b>CHAPTER 7</b> – Predictive and Descriptive Business Domains	-	147
Dr.S.Nithyanandam		
<b>CHAPTER 8</b> – Analytics in Different Business Domains	-	187
Ms.M.Jeeva		
CHAPTER 9 – HR & Supply Chain Analytics	-	215
Ms.K.Jayanthi		
<b>CHAPTER 10</b> – Marketing and Scales Analytics	-	257
Ms.K.Jayanthi		

# REFERENCES

285

-

# **Introduction to Business Analytics**

# Mr.A.Jahir Husain

**Business analytics** refers to the use of data analysis tools and techniques to make informed business decisions and drive organizational success. It involves collecting, processing, and analyzing data to extract meaningful insights that can guide strategic and operational decisions. Business analytics encompasses a range of methodologies, from descriptive statistics to advanced predictive modeling, and is integral to optimizing performance and gaining a competitive advantage.

# **Types of Business Analytics**

- **Descriptive Analytics:** Focuses on summarizing historical data to understand what has happened in the past.
- **Diagnostic Analytics:** Examines data to understand the causes of past events and trends.
- **Predictive Analytics:** Uses statistical models and machine learning techniques to forecast future outcomes based on historical data.
- **Prescriptive Analytics:** Recommends actions to optimize outcomes based on predictive models and data analysis.

# **Components of Business Analytics**

Data Collection

- **Sources:** Internal databases, customer feedback, transaction records, social media, market research.
- Techniques: Surveys, data scraping, transactional systems, sensors, and IoT devices.

# Data Processing

- **Data Cleaning:** Removing inaccuracies, duplicates, and irrelevant information to ensure data quality.
- **Data Integration:** Combining data from different sources to provide a comprehensive view.
- **Data Transformation:** Converting data into a suitable format for analysis, including normalization and aggregation.

# Data Analysis

- **Statistical Analysis:** Applying statistical methods to summarize and interpret data (e.g., mean, median, variance).
- **Exploratory Data Analysis (EDA):** Investigating data to discover patterns, anomalies, and relationships using visualizations and descriptive statistics.

# **Data Management For Analytics**

Ms.K.Jayanthi

**Data management for analytics** involves the processes and practices required to effectively collect, store, organize, and maintain data to ensure its quality and availability for analysis. Proper data management is crucial for deriving accurate insights and making informed decisions based on data. It encompasses a range of activities including data governance, data integration, and data quality management.

# **Data Management Processes**

Data Governance

- **Policies and Procedures:** Establish rules and guidelines for data management, including data ownership and usage rights.
- **Data Stewardship:** Assign responsibilities for data quality and management to individuals or teams.
- **Compliance:** Ensure adherence to regulations and standards such as GDPR, HIPAA, and CCPA.

#### Data Collection

- **Sources:** Identify and gather data from internal systems (e.g., CRM, ERP), external sources (e.g., social media, market research), and IoT devices.
- **Methods:** Use surveys, data scraping, transactional logs, and APIs for data collection.

### Data Integration

- Techniques:
  - **ETL (Extract, Transform, Load):** Extract data from various sources, transform it into a suitable format, and load it into a data warehouse.
  - Data Warehousing: Centralize data storage to facilitate reporting and analysis.
  - **Data Lakes:** Store raw, unstructured data for flexible analysis and exploration.
- **Tools:** Use integration tools like Talend, Informatica, and Apache Nifi.

### Data Cleaning

- Processes:
  - Data Validation: Check for accuracy and completeness.
  - **Data Deduplication:** Remove duplicate records.
  - **Data Normalization:** Standardize data formats and values.
- **Tools:** Use cleaning tools and libraries such as OpenRefine, Python's Pandas, and R's Data Cleaning packages.

# **Data Analysis Techniques**

Dr.R.Latha

Data analysis techniques are methods used to examine, clean, transform, and model data to extract meaningful insights and support decision-making. These techniques can be broadly categorized into several types based on the nature of the analysis and the goals of the study. Below is an overview of key data analysis techniques.

# **Descriptive Analysis**

#### Summary Statistics

- **Mean:** The average value of a dataset.
- Median: The middle value when data is ordered.
- Mode: The most frequently occurring value in a dataset.
- Standard Deviation: A measure of data dispersion or variability.
- Variance: The squared deviation from the mean.

#### Data Visualization

- **Histograms:** Graphical representation of the distribution of numerical data.
- Bar Charts: Used to compare categorical data.
- **Pie Charts:** Illustrate proportions and percentages of a whole.
- Box Plots: Show the distribution of data based on quartiles.
- Scatter Plots: Visualize the relationship between two variables.

#### Frequency Distribution

- **Purpose:** Shows how often each value occurs in a dataset.
- Methods: Use frequency tables, histograms, and bar charts.

### Inferential Analysis

#### Hypothesis Testing

- Null Hypothesis (H<sub>o</sub>): A statement that there is no effect or difference.
- Alternative Hypothesis (H<sub>1</sub>): A statement that there is an effect or difference.
- Tests: T-tests, Chi-square tests, ANOVA (Analysis of Variance).
- **P-Value:** Measures the probability of obtaining test results at least as extreme as the observed results, assuming the null hypothesis is true.

#### Confidence Intervals

• **Definition:** A range of values within which the true parameter value is expected to fall with a certain level of confidence.

# CHAPTER 4 Business Intelligence

Ms.K.Jayanthi

**Business Intelligence (BI)** refers to the technologies, processes, and practices used to collect, analyze, and present business data to support decision-making. BI encompasses a range of tools and techniques that help organizations turn raw data into actionable insights, enabling better strategic and operational decisions.

### **BI** Components

Data Sources

- Internal Sources: CRM systems, ERP systems, transactional databases.
- External Sources: Market research, social media, third-party data providers.

Data Warehousing

- **Purpose:** Centralize data from various sources for comprehensive analysis.
- Techniques: ETL (Extract, Transform, Load) processes to prepare data for analysis.

#### Data Integration

- **Objective:** Combine data from disparate sources to create a unified view.
- **Methods:** Use integration tools and middleware to merge and transform data.

Data Analytics

- Descriptive Analytics: Summarize historical data to understand past performance.
- Diagnostic Analytics: Explore data to identify causes of past events.
- Predictive Analytics: Use statistical models and machine learning to forecast future outcomes

# **BI Tools and Technologies**

Business Intelligence Platforms

- **Examples:** Tableau, Power BI, QlikView.
- Features: Interactive dashboards, ad-hoc reporting, and data visualization.

Data Warehousing Solutions

- **Examples:** Amazon Redshift, Google BigQuery, Snowflake.
- Features: Scalable storage, data integration, and support for complex queries.

# CHAPTER 5 Business Intelligence Tools and Technologies Ms.K.Jayanthi

Business Intelligence (BI) tools and technologies are essential for collecting, analyzing, and presenting data to support decision-making and business strategy. These tools help organizations transform raw data into meaningful insights through various functionalities, including data visualization, reporting, and advanced analytics. Below is an overview of **popular BI tools and technologies, along with their features and applications.** 

# **BI Platforms**

# Tableau

- Features:
  - Interactive dashboards and reports.
  - Drag-and-drop interface for data visualization.
  - Integration with multiple data sources.
  - Advanced analytics capabilities.
  - Collaboration and sharing options.
- Applications:
  - Data visualization and exploration.
  - Creating interactive dashboards.
  - Business reporting and performance tracking.

# **Power BI**

- Features:
  - Integration with Microsoft Office and Azure.
  - Interactive reports and dashboards.
  - Real-time data updates.
  - Customizable visualizations and data connectors.
  - AI-powered insights and analytics.
- Applications:
  - Creating and sharing interactive reports.
  - Analyzing business data.
  - Integrating with other Microsoft products.

# QlikView

- Features:
  - Associative data model for interactive data exploration.
  - Data visualization and reporting.
  - In-memory data processing for fast performance.
  - Integration with various data sources.
#### CHAPTER 6 Business ForeCasting Ms.K.Jayanthi

**Business forecasting** involves using historical data, statistical methods, and analytical techniques to predict future business conditions and performance. Accurate forecasting helps organizations make informed decisions about resource allocation, financial planning, and strategic direction. Here's an overview of key concepts, methods, and applications in business forecasting:

#### Types of Forecasting

- Qualitative Forecasting: Based on expert judgment and subjective analysis.
- Quantitative Forecasting: Uses historical data and statistical methods for predictions.

#### **Forecasting Methods**

#### Qualitative Methods

- **Expert Judgment:** Rely on the insights and opinions of industry experts or experienced managers.
- Market Research: Use surveys, focus groups, and other methods to gather opinions and expectations from customers and stakeholders.
- **Delphi Method:** Collect forecasts from a panel of experts through iterative rounds of questionnaires, with feedback provided between rounds.

#### Quantitative Methods

#### Time Series Analysis

- **Purpose:** Analyze historical data over time to identify patterns and forecast future values.
- Methods:
  - **Moving Averages:** Smooth out short-term fluctuations to highlight longer-term trends.
  - **Exponential Smoothing:** Apply decreasing weights to past observations to forecast future values.
  - **ARIMA (AutoRegressive Integrated Moving Average):** Model time series data with components for autoregression, differencing, and moving averages.

#### . Causal Models

- **Purpose:** Forecast based on the relationship between a dependent variable and one or more independent variables.
- Methods:
  - **Linear Regression:** Model the relationship between a dependent variable and one or more independent variables using a linear equation.
  - **Multiple Regression:** Extend linear regression to include multiple independent variables.

## **Predictive and Descriptive Business Domains**

#### Dr.S.Nithyanandam

#### **Techniques and Tools**

- Data Aggregation: Combine data from various sources to create summary reports.
- Data Visualization: Use charts, graphs, and dashboards to represent historical data visually.
- **Descriptive Statistics:** Calculate metrics such as mean, median, mode, standard deviation, and frequency distributions.
- **Exploratory Data Analysis (EDA):** Use techniques to explore and understand data characteristics.

#### Examples

- Sales Reports: Summarize past sales performance, including total sales, sales by region, and sales trends.
- **Customer Segmentation:** Analyze historical customer data to identify distinct segments and their behaviors.
- **Financial Statements:** Provide insights into past financial performance, including income statements and balance sheets.

#### Applications

- Business Reporting: Regularly update stakeholders on past performance and key metrics.
- **Trend Analysis:** Track historical trends to understand business performance over time.
- **Operational Monitoring:** Assess past operational metrics to identify areas for improvement.

#### . Predictive Analytics

**Predictive Analytics** focuses on forecasting future events based on historical data and statistical techniques. It answers questions about what might happen in the future and helps organizations prepare for potential outcomes.

#### Objectives

- Forecast Future Outcomes: Predict future trends, behaviors, and events based on historical data.
- Identify Risks and Opportunities: Anticipate potential challenges and opportunities.
- **Guide Decision-Making:** Provide data-driven insights to inform future strategies and actions.
- **Predictive Analytics:** Provides insights the product.

## Analytics in Different Business Domains Ms.M.Jeeva

Analytics can significantly impact various business domains by providing insights that drive decision-making and strategic planning. Here's an overview of how analytics is applied across different business domains:Marketing Analytics

#### Objectives

- Understand Customer Behavior: Analyze purchasing patterns, preferences, and engagement.
- **Optimize Campaigns:** Measure the effectiveness of marketing campaigns and channels.
- Predict Market Trends: Forecast future trends and customer needs.

#### Techniques and Tools

- **Customer Segmentation:** Identify distinct customer groups based on behavior and demographics.
- **Campaign Analysis:** Evaluate the performance of marketing campaigns using metrics like ROI, conversion rates, and click-through rates.
- Sentiment Analysis: Analyze customer feedback and social media posts to gauge sentiment and brand perception.

#### Applications

- Targeted Advertising: Use data to create personalized ads and promotions.
- **Customer Journey Mapping:** Track and analyze the customer journey to improve user experience.
- Market Research: Conduct research to understand market trends and customer preferences.

#### **Sales Analytics**

#### Objectives

- Forecast Sales: Predict future sales based on historical data and market conditions.
- **Optimize Sales Strategies:** Identify effective sales tactics and strategies.
- Track Performance: Monitor sales performance and identify areas for improvement.

#### Techniques and Tools

• Sales Forecasting: Use historical sales data and predictive models to estimate future sales.

## HR & Supply Chain Analytics

#### Ms.K.Jayanthi

**HR Analytics** and **Supply Chain Analytics** are two crucial domains in business analytics, each focusing on optimizing different aspects of organizational operations. Here's a detailed look at both:

#### **HR** Analytics

#### Objectives

- Improve Talent Management: Enhance the recruitment, development, and retention of employees.
- **Optimize Workforce Performance:** Analyze employee performance to drive productivity and engagement.
- Enhance Employee Experience: Improve satisfaction and work conditions to boost retention and performance.. Techniques and Tools

#### . Recruitment Analytics

- Applicant Tracking Systems (ATS): Track and analyze recruitment data, including application sources and conversion rates.
- Predictive Analytics: Use historical data to forecast future hiring needs and candidate success.

#### Performance Analytics

- **Performance Metrics:** Track key performance indicators (KPIs) such as productivity, goal achievement, and competency assessments.
- **360-Degree Feedback:** Collect feedback from multiple sources to evaluate employee performance comprehensively.

#### . Employee Engagement Analytics

• **Survey Analysis:** Analyze employee surveys to gauge satisfaction, engagement, and organizational climate.

#### Supply Chain Analytics

. Objectives

- **Optimize Supply Chain Operations:** Improve efficiency and effectiveness across the supply chain.
- Enhance Inventory Management: Manage inventory levels to balance supply and demand.

## **Marketing and Scales Analytics**

Ms.K.Jayanthi

**Marketing Analytics** and **Sales Analytics** are two closely related but distinct domains that focus on different aspects of driving revenue and optimizing business performance. Here's an overview of each, including techniques, tools, applications, and challenges:

#### Objectives

- **Understand Customer Behavior:** Analyze customer data to uncover insights into preferences, behavior, and engagement.
- **Optimize Marketing Campaigns:** Measure and improve the effectiveness of marketing strategies and campaigns.
- Forecast Market Trends: Predict future trends and customer needs to inform marketing strategies.

#### **Techniques and Tools**

#### Customer Segmentation

- **Clustering Algorithms:** Use algorithms such as K-means or hierarchical clustering to group customers based on similar behaviors or attributes.
- **RFM Analysis:** Segment customers based on Recency, Frequency, and Monetary value of their interactions.

#### Campaign Analysis

- **A/B Testing:** Test different versions of marketing materials or strategies to determine which performs better.
- **Multi-Touch Attribution:** Analyze the impact of various marketing channels on customer conversion paths.

#### Sentiment Analysis

- **Natural Language Processing (NLP):** Analyze customer reviews, social media posts, and feedback to gauge sentiment and brand perception.
- Text Analytics: Extract key themes and insights from customer communications.

#### **Predictive Analytics**

- **Churn Prediction:** Use historical data to predict which customers are likely to stop engaging with the brand.
- Lifetime Value (CLV) Prediction: Estimate the total value a customer will bring over their lifetime.



# NEURAL NETWORKS AND DEEP LEARNING

978-93-6255-065-1

## EDITED BY



#### NEURAL NETWORKS AND DEEP LEARNING

Copyright © 2021 by Dr. R. LATHA

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-065-1

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

Introduction	-	03
CHAPTER 1 – Introduction to Neural Network	-	11
Dr.S.Nithyanandam		
CHAPTER 2 – Activation Functions and Gradient Descent	-	34
Ms.S.Gayathiri		
CHAPTER 3 – Associative Memory and Unsupervised Learning	-	56
Dr.R.Latha		
<b>CHAPTER 4</b> – Third Generation Neural Network	-	78
Ms.R.Bhanumathi		
CHAPTER 5 – Deep Feedforward Network	-	82
Mr.A.Jahir Husain		
CHAPTER 6 – Recurrent Neural Network	-	112
Ms.R.Bhanumathi		
CHAPTER 7 – Generative Adversarial Networks(GANs)	-	137
Dr.S.Nithyanandam		
CHAPTER 8 – Deep Reinforcement Learning	-	156
Mr.A.Jahir Husain		
CHAPTER 9 – Optimization Algorithms for Deep Learning	-	178
Dr.R.Latha		
<b>CHAPTER 10</b> – Regularization Techniques for Deep Learning	-	189
Ms.R.Bhanumathi		

## REFERENCES

210

-

## **Introduction to Neural Network**

#### Dr.S.Nithyanandam

**Neural Networks** are a fundamental component of artificial intelligence (AI) and machine learning. They are inspired by the structure and functioning of the human brain and are designed to recognize patterns, learn from data, and make decisions or predictions based on that data. Here's an introduction to neural networks, covering their basic concepts, structure, types, and **applications:** 

#### **Basic Concepts**

#### 1. Neurons and Layers

- **Neuron**: The basic unit of a neural network, similar to a biological neuron. It receives input, processes it, and produces an output.
- **Layer**: Neurons are organized into layers. There are typically three types of layers in a neural network:
  - **Input Layer**: Receives the raw input data.
  - **Hidden Layers**: Intermediate layers that process inputs and extract features. There can be one or more hidden layers in a network.
  - **Output Layer**: Produces the final output or prediction of the network.

#### 2. Weights and Biases

- **Weights**: Parameters that adjust the strength of connections between neurons. They are learned during training and are crucial for the network's ability to make accurate predictions.
- **Biases**: Additional parameters that help the model fit the data by adjusting the activation function's threshold.

#### 3. Activation Functions

- **Purpose**: Activation functions introduce non-linearity into the network, allowing it to learn complex patterns.
- **Common Functions**:
  - **Sigmoid**: Outputs values between 0 and 1.
  - **ReLU** (**Rectified Linear Unit**): Outputs values greater than or equal to 0.
  - **Tanh**: Outputs values between -1 and 1.

#### 4. Training and Learning

- **Forward Propagation**: The process of passing input data through the network to generate an output.
- **Loss Function**: Measures the difference between the network's prediction and the actual target value. Common loss functions include mean squared error (MSE) for regression and cross-entropy for classification.
- **Backpropagation**: The process of updating weights and biases based on the loss function. It involves calculating gradients and adjusting parameters using optimization algorithms like stochastic gradient descent (SGD) or Adam.

## **Activation Functions and Gradient Descent**

#### Ms.S.Gayathiri

Activation functions are a crucial component of neural networks, introducing non-linearity into the model and enabling it to learn complex patterns. Here's an overview of common activation functions:

#### 1. Sigmoid Function

- Formula:  $\sigma(x)=11+e^{x}$  (x) =  $\frac{1}{1+e^{x}}$
- **Range**: (0,1)(0, 1)(0,1)
- **Characteristics**: Maps input values to a range between 0 and 1. It's used primarily in binary classification tasks.
- **Pros**: Smooth gradient, output bounded between 0 and 1.
- Cons: Can cause vanishing gradients, leading to slow convergence during training.

#### 4. Leaky ReLU

- **Formula**: Leaky ReLU(x)=max $\frac{fo}{\alpha}(\alpha x, x)$ \text{Leaky ReLU}(x) = \max(\alpha x, x)Leaky ReLU(x)=max(\alpha x, x) where  $\alpha$ \alpha $\alpha$  is a small constant (e.g., 0.01).
- **Range**:  $(-\infty,\infty)(-(infty, (infty))(-\infty,\infty))$
- Characteristics: Allows a small, non-zero gradient when the input is negative.
- **Pros**: Addresses the dying ReLU problem by allowing a small gradient when the input is negative.
- **Cons**: Still might not be ideal for all types of data and applications.

#### **Gradient Descent**

Gradient Descent is an optimization algorithm used to minimize the loss function of a neural network by iteratively adjusting the weights. It's a key component of the training process in neural networks.

Key Concepts

#### 1. Loss Function

• **Purpose**: Measures how well the model's predictions match the actual target values. Common loss functions include mean squared error (MSE) for regression and crossentropy for classification.

#### 2. Gradient

• **Definition**: The gradient of the loss function with respect to the model's weights indicates the direction and magnitude of the steepest increase in the loss function.

#### 3. Learning Rate

• **Definition**: A hyperparameter that controls the size of the steps taken towards the minimum of the loss function. Too large a learning rate can lead to overshooting, while too small a learning rate can slow down convergence.

## CHAPTER 3 Associative Memory and Unsupervised Learning

Dr.R.Latha

**Associative Memory** is a type of memory system in which information is retrieved based on its content rather than its specific address. In the context of artificial intelligence and neural networks, associative memory often refers to models that can recall or generate information based on learned associations between patterns.

#### **Key Concepts**

#### 1. Content-Based Retrieval

• **Definition**: Unlike traditional memory systems where data is accessed via a specific address, associative memory retrieves information based on matching input patterns to stored patterns.

#### 2. Associative Memory Models

- Hopfield Networks
  - **Description**: A type of recurrent neural network used for associative memory. It can store and recall binary patterns.
  - **Mechanism**: The network is designed to converge to a stored pattern when presented with a noisy or partial version of that pattern.
  - Use Case: Pattern recognition and error correction.

#### • Boltzmann Machines

- **Description**: A type of probabilistic graphical model used for associative memory. It consists of symmetrically connected neurons and can learn to represent complex distributions over binary vectors.
- **Mechanism**: Uses stochastic processes to find a configuration of neurons that minimizes the energy of the system, representing learned patterns.
- Use Case: Learning complex distributions and generating samples from learned distributions.

#### 3. Applications

- **Pattern Recognition**: Associative memory can be used to recognize patterns even when the input is noisy or incomplete.
- **Error Correction**: In cases where data is corrupted, associative memory can retrieve the most similar stored pattern to correct errors.
- **Data Retrieval**: Associative memory systems are useful for retrieving data based on partial or corrupted inputs.

#### **Unsupervised Learning**

**Unsupervised Learning** is a type of machine learning where the model learns patterns and structures from unlabeled data. Unlike supervised learning, where the model is trained with input-output pairs, unsupervised learning involves finding hidden structures or relationships within the data without explicit guidance.

## **Third Generation Neural Network**

#### Ms.R.Bhanumathi

**Third Generation Neural Networks** generally refer to advancements and innovations in neural network architectures and training methods that go beyond the early, more basic models. These advancements typically emphasize greater complexity, flexibility, and efficiency in learning from data. They incorporate ideas from both deep learning and more advanced machine learning techniques. Here's a closer look at what characterizes third-generation neural networks:

#### **Characteristics of Third Generation Neural Networks**

#### 1. Deep Architectures

- **Definition**: Neural networks with many layers (deep networks) that can learn hierarchical representations of data.
- **Example**: Deep Convolutional Neural Networks (DCNNs) used in image and video recognition, such as VGGNet, ResNet, and Inception networks.

#### 2. Advanced Learning Techniques

- **Transfer Learning**: Reusing a pre-trained model on a new but related problem, reducing the need for large amounts of data and computational resources.
- **Few-Shot Learning**: Training models to recognize new classes with very few examples, often leveraging techniques like meta-learning.

#### 3. Complex Network Structures

- Attention Mechanisms: Enhancing the model's ability to focus on relevant parts of the input, widely used in natural language processing (NLP). Examples include the Transformer architecture.
- Generative Models: Networks designed to generate new data samples. This includes

## **Key Architectures**

#### 1. Convolutional Neural Networks (CNNs)

- **Description**: Specialize in processing data with grid-like topology, such as images. They use convolutional layers to automatically learn spatial hierarchies of features.
- Advancements: Residual connections in ResNet, inception modules in Inception networks.

#### 2. Recurrent Neural Networks (RNNs) and Variants

- **Description**: Designed for sequential data, such as time series or text. They use memory to maintain information about previous inputs.
- Advancements: Long Short-Term Memory (LSTM) and Gated Recurrent Units (GRUs) address issues with long-term dependencies.

#### 3. Transformers

- **Description**: An architecture based on self-attention mechanisms, widely used in NLP tasks. It processes sequences in parallel rather than sequentially.
- **Key Models**: BERT (Bidirectional Encoder Representations from Transformers), GPT (Generative Pre-trained Transformer), and T5 (Text-To-Text Transfer Transformer).

#### **Deep Feedforward Network**

Mr.A.Jahir Husain

**Deep Feedforward Networks** are a class of artificial neural networks characterized by their layered structure, where information flows in one direction—from input to output—through a series of layers. They are often referred to as **Deep Neural Networks (DNNs)** when they have multiple hidden layers. Here's an overview of deep feedforward networks:

#### **Structure and Components**

- 1. Layers
  - **Input Layer**: The initial layer that receives raw data and passes it to the next layer. Each neuron in this layer corresponds to a feature in the input data.
  - **Hidden Layers**: Intermediate layers between the input and output layers. These layers extract features and learn complex representations of the data. In deep feedforward networks, there are multiple hidden layers.
  - **Output Layer**: The final layer that produces the network's prediction or output. The number of neurons in this layer typically corresponds to the number of classes (for classification tasks) or the number of outputs (for regression tasks).

#### 2. Neurons

• **Definition**: The basic computational units in a layer. Each neuron receives input from the previous layer, applies a weighted sum, adds a bias term, and passes the result through an activation function.

#### 3. Weights and Biases

- Weights: Parameters that adjust the strength of connections between neurons in adjacent layers. They are learned during training.
- **Biases**: Additional parameters added to the weighted sum before applying the activation function, helping the model fit the data better.

#### 4. Activation Functions

- **Purpose**: Introduce non-linearity into the model, enabling it to learn complex patterns. Common activation functions include:
  - ReLU (Rectified Linear Unit): ReLU(x)=max[fo](0,x)\text{ReLU}(x) = \max(0, x)ReLU(x)=max(0,x)
  - Sigmoid:  $\sigma(x)=11+e^{x}$   $\sigma(x)=1+e^{x}$
  - Tanh:  $tanh[io](x)=ex-e-xex+e-x(tanh(x) = \frac{e^x e^{-x}}{e^x e^{-x}} e^x + e^{-x})$

#### **Training Process**

#### 1. Forward Propagation

• **Description**: The process of passing input data through the network to obtain predictions. Each layer transforms its inputs using weights, biases, and activation functions to produce outputs.

#### 2. Loss Function

• **Definition**: A measure of how well the network's predictions match the actual target values. Common loss functions include:

#### **CHAPTER 6 Recurrent Neural Network** Ms.R.Bhanumathi

**Recurrent Neural Networks (RNNs)** are a type of neural network designed for processing sequential data, where the order of inputs is important. Unlike feedforward neural networks, RNNs have connections that loop back on themselves, allowing them to maintain a memory of previous inputs. This makes them well-suited for tasks where context and sequence are crucial, such as time series analysis, natural language processing, and speech recognition.

#### 1. Sequential Data Processing

**Definition:** RNNs process data in a sequence, with each step in the sequence influencing 0 the network's internal state and future steps. This temporal aspect allows RNNs to capture dependencies across time.

#### 2. Internal State (Memory)

• **Definition**: RNNs maintain a hidden state that updates at each time step, capturing information from previous steps. This hidden state acts as a form of memory.

#### 3. **Recurrence**

• **Definition**: The feedback loop within an RNN that allows information to persist. At each time step, the network updates its hidden state based on the input and the previous hidden state.

#### Structure

#### 1. Basic RNN Architecture

- **Components**:
  - Input Layer: Receives the current input at each time step.
  - Hidden Layer: Updates the hidden state based on the current input and previous hidden state. The hidden state is then used to generate the output.
  - Output Layer: Produces the final output at each time step.

#### 2. Vanishing and Exploding Gradients

- Vanishing Gradient: During training, gradients can become very small, making it 0 difficult for the network to learn long-range dependencies.
- Exploding Gradient: Gradients can become very large, leading to unstable training. This 0 is addressed using techniques like gradient clipping.

#### **Advanced Variants**

- 1. Long Short-Term Memory (LSTM)
  - **Description**: An extension of RNNs designed to address the vanishing gradient problem. LSTMs include special gating mechanisms to control the flow of information.
  - **Components**: 0
    - Cell State: Carries long-term memory.
    - Forget Gate: Decides what information to discard from the cell state

## **Generative Adversarial Networks(GANs)**

#### Dr.S.Nithyanandam

**Generative Adversarial Networks (GANs)** are a class of neural networks designed for generating new, synthetic data samples that resemble a given set of real data. GANs consist of two neural networks—the **generator** and the **discriminator**—that are trained simultaneously through adversarial processes. This approach enables GANs to create high-quality, realistic data samples, which can be used for various applications, from image generation to data augmentation.

#### Components of GANs

#### 1. Generator

- **Purpose**: To create synthetic data samples that are similar to real data.
- Input: A random noise vector, often sampled from a Gaussian or uniform distribution.
- **Output**: Synthetic data samples, such as images, text, or other types of data.
- **Objective**: To generate data that is indistinguishable from real data, as judged by the discriminator.

#### 2. Discriminator

- **Purpose**: To distinguish between real data samples and those generated by the generator.
- **Input**: Data samples (both real and synthetic).
- **Output**: A probability score indicating whether the input data is real or generated.
- **Objective**: To correctly classify data as real or fake, thus guiding the generator to improve.

#### **Training Process**

#### 1. Adversarial Training

- **Process**: The generator and discriminator are trained simultaneously in a competitive setting:
  - Generator's Goal: To produce samples that the discriminator cannot easily classify as fake.
  - **Discriminator's Goal**: To accurately distinguish between real and synthetic samples.

#### 2. Loss Functions

- **Generator Loss**: Measures how well the generator's samples deceive the discriminator. Typically, it is the negative log probability that the discriminator classifies the generated samples as real.
- **Discriminator Loss**: Measures how well the discriminator can differentiate between real and generated samples. It is the sum of the negative log probability of classifying real samples as real and generated samples as fake.

For a standard GAN, the loss functions are:

## **Deep Reinforcement Learning**

#### Mr.A.Jahir Husain

**Deep Reinforcement Learning (DRL)** is a subfield of machine learning that combines reinforcement learning (RL) with deep learning techniques. It enables an agent to learn how to make decisions and take actions in complex environments by leveraging deep neural networks to approximate value functions, policies, or both. This approach has been instrumental in solving challenging problems in various domains, from game playing to robotics.

#### 1. Reinforcement Learning (RL) Basics

- Agent: The entity that learns and makes decisions based on the environment.
- **Environment**: The external system that the agent interacts with, providing feedback in the form of rewards or penalties.
- State: A representation of the current situation or configuration of the environment.
- Action: A move or decision made by the agent in the environment.
- **Reward**: Feedback from the environment, indicating the value or utility of the action taken.

#### 2. Objective

• **Goal**: The agent aims to learn a policy that maximizes the cumulative reward over time. The policy determines the best action to take given a particular state.

#### 3. Value Function

- **Definition**: A function that estimates the expected return (cumulative future reward) from a given state or state-action pair.
- State Value Function V(s)V(s)V(s): Estimates the expected return from state sss.
- Action Value Function Q(s,a)Q(s,a)Q(s,a): Estimates the expected return from taking
- 0

#### **Integration with Deep Learning**

#### 1. Function Approximation

- **Challenge**: In complex environments with large state or action spaces, it is impractical to use tabular methods for storing value functions or policies.
- **Solution**: Deep neural networks approximate value functions, policies, or both. This allows the agent to handle high-dimensional inputs and complex tasks.

#### 2. Deep Q-Networks (DQN)

- **Description**: Combines Q-learning, a popular RL algorithm, with deep learning. Uses a deep neural network to approximate the Q-function.
- Architecture: A neural network that takes state inputs and outputs Q-values for each

#### 3. Policy Gradient Methods

- **Description**: Directly optimize the policy by estimating the gradient of expected cumulative reward with respect to policy parameters.
- **Examples**: REINFORCE algorithm, which uses Monte Carlo methods to estimate the policy gradient.
- 4. Actor-Critic Methods
  - **Description**: Combine the benefits of value-based and policy-based methods.

## **CHAPTER 9 Optimization Algorithms for Deep Learning**

Dr.R.Latha

Optimization algorithms are crucial in deep learning as they determine how neural network parameters are updated to minimize the loss function during training. The choice of optimization algorithm can significantly affect the speed and quality of the training process. Here's an overview of some key optimization algorithms used in deep learning:

#### Gradient Descent

- **Definition**: An optimization algorithm that updates the model parameters by moving them in the direction of the negative gradient of the loss function.
- **Types**: •
  - **Batch Gradient Descent**: Uses the entire dataset to compute the gradient of the loss function. It can be computationally expensive and slow for large datasets.
  - Stochastic Gradient Descent (SGD): Uses a single data point or a small batch to 0 compute the gradient, which can make training faster and can help escape local minima but introduces more noise in the updates.
  - Mini-Batch Gradient Descent: Combines the advantages of batch and stochastic 0 gradient descent by using small random subsets of the data (mini-batches) to compute the gradient.

#### Momentum

- **Definition:** An extension of gradient descent that incorporates the concept of momentum to accelerate convergence and smooth out the updates.
- Benefit: Helps to overcome local minima and speeds up convergence.

#### Nesterov Accelerated Gradient (NAG)

- **Definition**: A variant of momentum that improves the convergence rate by calculating the gradient at the point where the parameters would be after the momentum step.
- Benefit: Provides a more accurate update and faster convergence compared to standard
- momentum.

#### Adagrad

• **Definition**: An adaptive learning rate algorithm that adjusts the learning rate based on the historical gradient magnitudes.

#### **RMSprop**

• **Definition**: An adaptive learning rate algorithm that improves on Adagrad by using a moving average of squared gradients.

## CHAPTER 10 Regularization Techniques for Deep Learning

Ms.R.Bhanumathi

Regularization techniques are essential in deep learning to prevent overfitting, improve generalization, and enhance model performance. Overfitting occurs when a model learns the noise in the training data rather than the underlying patterns, leading to poor performance on unseen data. Regularization methods help mitigate this risk by adding constraints or penalties to the model. Here's an overview of commonly used regularization techniques in deep learning:

#### L1 and L2 Regularization

- L1 Regularization (Lasso)
  - **Definition**: Adds the absolute value of the magnitude of coefficients as a penalty term to the loss function.
  - **Effect**: Encourages sparsity, meaning it can drive some weights to exactly zero, effectively performing feature selection.
  - **Implementation**: Often used in linear models and shallow neural networks.

#### • L2 Regularization (Ridge)

- **Definition**: Adds the squared magnitude of coefficients as a penalty term to the loss function.
- **Effect**: Helps to shrink the weights, which can improve the model's ability to generalize by reducing the risk of overfitting.
- **Implementation**: Commonly used in deep neural networks and other machine learning models.
- Elastic Net Regularization
  - **Definition**: Combines L1 and L2 regularization, providing a balance between feature selection and weight shrinkage.
  - Implementation: Useful when dealing with highly correlated features.

#### Dropout

- **Definition**: A technique where random neurons are dropped (set to zero) during training with a certain probability, which prevents the model from becoming too reliant on specific neurons.
- **Dropout Rate**: The probability of dropping a neuron (e.g., 0.5 for a 50% dropout rate).
- **Effect**: Helps to prevent overfitting by ensuring that the network does not become too dependent on any single neuron.
- Implementation: Applied to fully connected layers and can also be used in convolutional layers.

#### **Batch Normalization**

- **Definition**: Normalizes the inputs to a layer for each mini-batch, which helps to stabilize and accelerate training.
- **Process**: Applies normalization to the activations, followed by scaling and shifting.



# EMBEDDED SYSTEMS AND IOT



EMBEDDED SYSTEMS AND IOT Copyright © 2021 by Mrs.K.Sangeetha

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-301-0

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST

## TABLE OF CONTENT

S.NO	TOPICS	PAGE NO
1	Iot and Arduino Programing Mr.S.Gayathri	03
2	Iot Communication and open Platform Mr.M.Jeeva	40
3	Application Development Mr.M.Jeeva	73
4	Programming Embedded System Mr.M.Jeeva	120
5	Context Switching Mrs.M.Mohana Priya	140
6	Bluetooth Mrs.K.Jayanthi	150
7	Home Automation Mrs.Dr.S.Nithyaanadham	158
8	Smart Agriculture Mrs.M.Jeeva	165
9	Smart Cities Mr.Dr.R.Latha	178
10	Smart Healthcare Mrs.M.Mohana Priya	187
11	REFERANCE	200

## CHAPTER 1 IOT AND ARDUINO PROGRAMMING

Mrs.S. Gayathri

IoT (Internet of Things) and Arduino programming are closely related. IoT involves connecting various devices to the internet and enabling them to communicate with each other and with users. Arduino is a popular platform for building these kinds of devices because it's accessible, versatile, and has a strong community.

#### Getting Started with IoT and Arduino

#### 1.Understanding IoT:

Devices: These can be anything from sensors to actuators to smart appliances.

Connectivity: Devices need to be connected via Wi-Fi, Bluetooth, Zigbee, etc.

#### 2. Arduino Basics:

Arduino Board: The heart of many IoT projects, like the Arduino Uno, Nano, or MKR series.

**Programming**: Arduino uses a simplified version of C/C++ and the Arduino IDE for writing and uploading code.

#### 3. Setting Up Arduino for IoT:

**Choose a Board**: For IoT projects, boards with built-in connectivity (like the Arduino MKR Wi-Fi 1010 or the Arduino Nano 33 IoT) are useful.

**Connect Sensors/Actuators**: Attach your sensors and actuators to the board using jumper wires or shields.

#### 4. Programming:

Install the Arduino IDE: Download and install the Arduino IDE from the official website.

**Libraries**: Use libraries for easy communication with sensors and modules (like WiFi.h for Wi-Fi or HTTPClient.h for HTTP requests).

#### 5. Cloud Integration:

**Platforms**: Use platforms like Blynk, Ad fruit IO, or Thing Speak for cloud integration and data visualization.

• APIs: Send data to or receive data from cloud services using HTTP requests.

## CHAPTER 2 IOT COMMUNICATION AND OPEN PLATFORM

Mrs.M. Jeeva

#### **1. IoT Communication**

#### **Communication Protocols:**

- **HTTP/HTTPS**: Standard web protocol for communication. Commonly used for RESTful APIs to interact with web services. It is simple but can be less efficient for real-time applications.
- **MQTT** (**Message Queuing Telemetry Transport**): A lightweight messaging protocol designed for low-bandwidth, high-latency, or unreliable networks. It uses a publish/subscribe model, which is efficient for IoT devices.
- **CoAP** (**Constrained Application Protocol**): Similar to HTTP but optimized for constrained environments. It's lightweight and uses a request/response model.
- WebSockets: A protocol providing full-duplex communication channels over a single TCP connection, useful for real-time data exchange.
- **Bluetooth/Bluetooth Low Energy (BLE)**: Used for short-range communication, commonly found in wearable devices and smart home devices.
- **Zigbee/Z-Wave**: Wireless communication protocols designed for low-power, low-data-rate applications. Often used in smart home systems.

#### 2. Open Platforms for IoT

Open platforms are designed to simplify the development, deployment, and management of IoT applications. They provide APIs, tools, and services for integrating IoT devices and managing data. Here are some popular open platforms:

#### 1. Arduino IoT Cloud

• Features: Allows you to connect Arduino devices to the cloud, visualize data, and control devices remotely.

#### 2. Thing Speak

• Features: An open IoT platform that allows you to collect, store, and analyze data from IoT devices. It provides APIs for data input and retrieval.

#### 3. Blynk

• Features: Provides a mobile app for remote control and monitoring of IoT devices. Blynk's cloud server handles communication, and the platform supports various hardware.

#### 4. Adafruit IO

• Features: A cloud service designed for IoT projects, offering data visualization, dashboards, and integration with Adafruit's hardware.

#### CHAPTER 3 APPLICATIONS DEVELOPMENT Mrs.M. Jeeva

Application development for IoT involves creating software that interacts with IoT devices to collect, process, and present data

#### 1. Application Types

#### **1.1. Embedded Applications**

- **Purpose**: Control and manage IoT devices directly, often running on microcontrollers or singleboard computers.
- **Examples**: Firmware for sensors, device control software.
- **Technologies**: C/C++, Python, Arduino IDE, Platform.

#### 1.2. Web Applications

- **Purpose**: Provide a user interface for interacting with IoT devices, visualizing data, and managing device settings.
- **Examples**: Dashboards, control panels.
- **Technologies**: HTML/CSS, JavaScript (React, Angular, Vue), server-side languages (Node.js, Python, Ruby).

#### **1.3. Mobile Applications**

- **Purpose**: Allow users to monitor and control IoT devices via smartphones or tablets.
- **Examples**: Mobile apps for smart home systems, fitness trackers.
- **Technologies**: Swift (iOS), Kotlin/Java (Android), cross-platform frameworks (Flutter, React Native).

#### **1.4. Desktop Applications**

- **Purpose**: Provide a more robust interface for IoT management on desktops or laptops.
- Examples: Data analytics tools, device management systems.
- **Technologies**: Electron (for cross-platform apps), .NET, JavaFX.

#### **Example IoT Application Development**

Simple Web Dashboard: Create a web-based dashboard to monitor data from an IoT sensor.

#### **Example Setup**:

- 1. **MQTT Broker**: Set up an MQTT broker (e.g., Mosquitto).
- 2. **Device**: Publish sensor data to an MQTT topic.
- 3. Web Server: Set up a Node.js server to handle MQTT subscriptions and serve web pages.
- 4. **Web Client**: Create a simple web page that uses JavaScript to subscribe to the MQTT topic and display data.

#### CHAPTER 4 PROGRAMMING EMBEDDED SYSTEM IN C Mrs.M. Jeeva

Programming embedded systems in C involves writing code for systems that are typically resourceconstrained and specialized for specific tasks. Embedded systems are integral to many devices, from microcontrollers in home appliances to complex processors in automotive systems. Here's a comprehensive guide to programming embedded systems in C:

. Introduction to Embedded Systems

#### 1.1. Definition

• **Embedded System**: A dedicated computer system designed to perform specific tasks within a larger system. It combines hardware and software to achieve its functionality.

#### **1.2. Characteristics**

- **Resource-Constrained**: Limited memory, processing power, and storage.
- Real-Time Operation: Often required to respond to events within strict timing constraints.
- **Reliability**: Must be highly reliable and stable.
- 2. Setting Up the Development Environment

#### 2.1. Toolchain

- Compiler: Converts C code into machine code (e.g., GCC for ARM, Keil for ARM Cortex).
- **Debugger**: Tools like GDB or integrated debuggers in IDEs for debugging and testing.
- IDE: Integrated Development Environments such as Keil uVision, MPLAB X, or Eclipse with CDT.

#### 2.2. Hardware Interface

- **Development Boards**: Use boards like Arduino, STM32, or Raspberry Pi for testing and development.
- **Programmers/Debuggers**: Tools like JTAG or SWD for programming and debugging the microcontroller.
- 3. Basic Structure of Embedded C Programs

#### 3.1. Program Initialization

• Initialization: Set up hardware peripherals, configure clocks, and initialize memory.

#### 3.2. Main Loop

• **Loop**: Typically, a while (1) loop where the main logic of the application runs.

#### **CONTEXT SWITCHING**

Mrs.M. Mohana Priya

context Switching

**Context switching** is the process of saving and restoring the state (context) of a CPU so that multiple processes or threads can share a single CPU resource efficiently. This is crucial for multitasking in operating systems and embedded systems.

#### 1. What is Context Switching?

#### 1.1. Definition

• **Context Switching**: The process by which the CPU switches from executing one process or thread to another, involving saving the state of the current task and loading the state of the next task.

#### 1.2. Purpose

- **Multitasking**: Enables multiple processes or threads to run concurrently on a single CPU by rapidly switching between them.
- Resource Utilization: Maximizes CPU usage and responsiveness of the system.

#### 2. Context Components

#### 2.1. Process/Thread State

- **CPU Registers**: Includes general-purpose registers, program counter, stack pointer, and other special-purpose registers.
- Memory Management Information: Page tables, base and limit registers.
- **Program State**: Current instruction address, CPU flags, and status registers.

#### 2.2. Saving Context

• Save State: Store the current state of CPU registers and other relevant information into the process's or thread's control block.

#### 2.3. Restoring Context

• Load State: Retrieve the saved state of the next process or thread and load it into the CPU registers.

#### **CHAPTER 6 BLUETOOTH** Mrs.K.Jayanthi

#### Bluetooth

Bluetooth is a wireless communication technology designed for short-range data exchange between devices. It operates in the 2.4 GHz ISM (Industrial, Scientific, and Medical) band and is commonly used in a wide range of applications, from connecting peripherals like keyboards and headphones to enabling communication between devices in IoT (Internet of Things) ecosystems.

#### **Introduction to Bluetooth**

#### 1.1. Definition

• **Bluetooth**: A short-range wireless technology standard for exchanging data between devices over short distances, typically up to 100 meters.

#### 1.2. History

- **Origin**: Developed by Ericsson in 1994 and named after a Danish king, Harald Bluetooth.
- **Evolution**: Several versions have been released, each improving on range, data rate, and power consumption.

#### **1.3. Bluetooth Versions**

- Bluetooth 1.x: Initial versions with basic functionality.
- Bluetooth 2.0 + EDR: Enhanced Data Rate for faster data transfer.
- Bluetooth 3.0 + HS: High-Speed for faster data transfer using Wi-Fi.
- Bluetooth 4.x: Introduced Bluetooth Low Energy (BLE) for lower power consumption.
- Bluetooth 5.x: Enhanced range, speed, and data broadcasting.

#### **Bluetooth Architecture**

#### 2.1. Bluetooth Stack

- Radio Layer: Manages physical radio communication.
- **Baseband Layer**: Handles link management and data packet assembly.
- Link Manager Protocol (LMP): Manages connections and device pairing.
- Logical Link Control and Adaptation Protocol (L2CAP): Provides multiplexing and data packet segmentation.
- Host Controller Interface (HCI): Interface between the hardware and Bluetooth software stack.
- Object Exchange Protocol (OBEX): Facilitates file transfers.
- **Bluetooth Profiles**: Define specific use cases and functionalities (e.g., A2DP for audio streaming, HID for keyboards).

#### HOME AUTOMATION

#### Mr. Dr.S. Nithyaanadham

#### **Home Automation**

Home automation refers to the use of technology to control and manage various systems and appliances in a home remotely and automatically. This can include lighting, heating, security, entertainment systems, and more. The goal of home automation is to improve comfort, efficiency, and security, and to enable more convenient and efficient management of home systems.

#### 1. Introduction to Home Automation

#### 1.1. Definition

• **Home Automation**: The use of smart devices and systems to automate and control home functions remotely or automatically based on predefined conditions.

#### **1.2. Benefits**

- Convenience: Remote control and automation of household devices.
- Energy Efficiency: Optimizing energy use to reduce utility costs.
- Security: Enhanced monitoring and control of home security systems.
- **Comfort**: Improved living conditions through automated adjustments of lighting, temperature, and other home settings.

#### 2. Key Components of Home Automation

#### 2.1. Smart Devices

- **Smart Lights**: LED bulbs and strips that can be controlled remotely, often with features like dimming and color change.
- Smart Thermostats: Devices that can be programmed to adjust home temperature based on user preferences and schedules.
- **Smart Locks**: Electronic locks that can be controlled remotely for improved security and convenience.
- Smart Plugs: Plugs that allow remote control of connected devices.
- 3. Home Automation Technologies

#### **3.1. Internet of Things (IoT)**

- **Description**: A network of interconnected devices that can communicate and interact with each other over the internet.
- Role in Home Automation: Allows integration and control of various smart home devices through a central system

#### SMART AGRICULTURE

#### Mrs.Dr.R. Latha

Smart agriculture, also known as precision agriculture, involves using technology to enhance the efficiency and productivity of farming practices. It leverages data collection, automation, and analytics to optimize crop yields, resource usage, and overall farm management.

1. Smart agriculture uses technology to optimize farming. Key elements include:

- 1. IoT Sensors: Monitor soil moisture, weather, and crop health.
- 2. Drones: Provide aerial views for crop management and precision application of inputs.
- 3. AI & Data Analytics: Analyze data for decision-making and predictive insights.
- 4. Automation: Implements robotic systems for planting, harvesting, and maintenance.
- 5. Precision Farming: Targets inputs (water, fertilizer) to specific areas for efficiency.

2. Sustainable Practices

- **Precision Agriculture**: Applying resources (water, fertilizer) more efficiently based on data-driven insights.
- Vertical Farming: Utilizing urban spaces for high-density crop production.
- Crop Rotation and Diversification: Enhancing soil health and reducing pest cycles

3. Climate Resilience

- **Climate-Smart Agriculture**: Practices that increase productivity while enhancing resilience to climate change.
- Water Management Technologies: Smart irrigation systems that reduce water waste and improve efficiency.
- 4. Economic Impact
  - Cost Reduction: Lower operational costs through efficiency and resource management.
  - Market Access: Using e-commerce platforms for direct sales, reducing dependency on middlemen.

#### 5. Challenges

- Technology Access: Limited access to technology in some regions.
- Data Privacy: Concerns about data ownership and security.
- Skill Gap: Need for training and education on new technologies.

#### CHAPTER 9 SMART CITIES Mrs.M. Jeeva

Smart cities are urban areas that use digital technology and data to enhance the quality of life for their residents, improve efficiency in services, and create a more sustainable and responsive environment. Here are some key aspects of smart cities:

- **Data and Connectivity**: Smart cities leverage data collected from sensors, cameras, and other sources to monitor and manage various urban systems. This data helps optimize traffic flow, manage energy use, and improve public safety.
- **Infrastructure**: These cities use advanced infrastructure such as smart grids for electricity, intelligent transportation systems, and sustainable buildings that use resources more efficiently.
- **Public Services**: Smart cities enhance public services through technology. For example, smart street lighting adjusts brightness based on activity, and waste management systems can monitor garbage levels and optimize collection routes.
- **Transportation**: Intelligent transportation systems manage traffic flow and public transit, often integrating real-time data to provide better routes and schedules. Electric and autonomous vehicles are also part of the mix.
- **Environment**: Sustainability is a key focus, with efforts to reduce emissions, manage waste, and use renewable energy sources. Green spaces and eco-friendly building practices are also integral.
- **Citizen Engagement**: Many smart cities involve residents in decision-making through apps and platforms that allow people to report issues, provide feedback, and access information.
- Security: Enhanced security measures, including surveillance and emergency response systems, help ensure safety while respecting privacy.
- **Health and Well-being**: Smart cities often incorporate health-monitoring systems and promote wellness through accessible healthcare services and pollution control.

Examples of smart cities include Barcelona, Singapore, and San Francisco, each of which has implemented various smart technologies to address urban challenges and improve living conditions.

#### CHAPTER 10 SMART HEALTHCARE Mrs.M. Mohana Priva

Smart healthcare integrates digital technology and data analytics to improve patient care, streamline healthcare processes, and enhance overall health outcomes. Here are some key components and benefits of smart healthcare:

1. Telemedicine and Remote Monitoring

- **Telemedicine**: Allows patients to consult with healthcare providers via video calls or online platforms, improving access to care, especially for those in remote or underserved areas.
- **Remote Monitoring**: Devices track vital signs and health metrics, such as glucose levels or heart rate, and transmit data to healthcare providers for real-time monitoring and intervention.

#### 2. Health Wearables

- **Fitness Trackers**: Devices like smartwatches monitor physical activity, sleep patterns, and other health metrics.
- **Medical Wearables**: Wearable ECG monitors, continuous glucose monitors, and other devices provide critical health data that can be used for diagnostics and personalized treatment plans.
- 3. Electronic Health Records (EHRs)
  - **Centralized Data**: EHRs store patient information in a digital format, making it easier for healthcare providers to access, share, and update patient records.
  - **Interoperability**: EHR systems can be integrated across different healthcare facilities, improving coordination and continuity of care.

4. AI and Machine Learning

- **Diagnostics**: AI algorithms can analyze medical images, genetic data, and other information to assist in diagnosing diseases and predicting patient outcomes.
- **Personalized Medicine**: Machine learning helps tailor treatments to individual patients based on their genetic makeup, lifestyle, and health data.

5. Smart Hospital Infrastructure

- Automation: Smart systems manage hospital resources like bed availability, staff allocation, and inventory management, optimizing operations.
- Enhanced Safety: Automated alerts and tracking systems improve patient safety by monitoring for potential issues, such as medication errors or falls.

#### 6. Mobile Health Apps

- **Health Management**: Apps help users track their health metrics, manage chronic conditions, and adhere to treatment plans.
- Wellness and Prevention: Many apps offer resources for mental health, fitness, and dietary management, promoting overall well-being.

## IMMUNOLOGY

## EDITED BY DR. R. KAMARAJ



#### IMMUNOLOGY

Copyright©2021 by Dr. R. Kamaraj

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

**ISSN:** 978-93-6255-556-4

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

## TABLE OF THE CONTENTS

Chapter 1: Introduction to Immunology	1
Dr. R. Kamaraj	
Chapter 2: Types of Immune System	15
Dr. T. Veeramani	
Chapter 3: Antigens & Antibodies	23
Dr. R. Arunkumar	
Chapter 4: Antigen Processing and Presentation	35
Dr. J. Illamathi	
Chapter 5: Cytokines and Signalling	47
Dr. T. Veeramani	
Chapter & Complement	50
Dr. R. Kamaraj	39
	(0)
Chapter 7: The Major Histocompatibility Complex	68
Reference	76
#### Chapter 1: Introduction to Immunology

#### Dr. R. Kamaraj

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **INTRODUCTION**

Immunology is indeed the study of the immune system, a complex network that defends the body against infection, foreign substances, and disease-causing agents like viruses, bacteria, and other pathogens. It involves understanding the mechanisms by which the immune system recognizes and eliminates threats, as well as the factors that can compromise its functioning.

The immune system is classified into **innate immunity** (the body's first line of defense, including physical barriers like skin, as well as immune cells like macrophages) and **adaptive immunity** 

When the immune system fails, the body becomes vulnerable to a wide range of diseases:

- **Autoimmune diseases** occur when the immune system mistakenly attacks the body's own tissues (e.g., lupus, rheumatoid arthritis).
- **Immunodeficiencies** occur when parts of the immune system are missing or not functioning properly, like in the case of **AIDS** (caused by HIV, which destroys helper T cells).
- **Cancer** is also connected to immune dysfunction, where abnormal cells can evade immune detection. The immune system indeed functions as an intricate security system for the body, constantly on the lookout for invading pathogens (like bacteria, viruses, fungi, and protozoans) and other foreign antigens. These pathogens carry unique markers known as **antigens**, which trigger an immune response, including the production of **antibodies** that specifically target and neutralize them. The immune system is composed of various **cells and organs** that work together to defend the body:

Immune Cells: These include:

White blood cells (leukocytes): The main defenders, such as:

**Lymphocytes**: T cells (coordinate immune responses, destroy infected cells), B cells (produce antibodies). **Macrophages**: Engulf and digest pathogens and dead cells. **Neutrophils**: First responders to infection.

#### Chapter 2: Types of Immune System

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

The innate immune system serves as the body's first line of defense against pathogens, providing immediate protection. The innate immune system is the body's initial response mechanism, acting quickly to recognize and combat pathogens. It is non-specific, meaning it doesn't differentiate between different types of invaders, and it is present from birth, providing constant surveillance.

The Innate Immune System: The First Line of Defense

Key Components

1. Physical Barriers

a. Skin: Acts as a protective barrier against pathogens. Its tight junctions prevent entry, and it produces antimicrobial peptides.

b. Mucous Membranes: Found in the respiratory, gastrointestinal, and urogenital tracts, these membranes trap pathogens and contain enzymes (like lysozyme) that destroy them.

2. Cellular Defenses

Phagocytic Cells:

Neutrophils: The most abundant type of white blood cell, they quickly respond to sites of infection and engulf pathogens.

Macrophages: Derived from monocytes, these cells are highly effective at phagocytosis and also play a role in activating the adaptive immune system.

Natural Killer (NK) Cells: Specialized lymphocytes that recognize and destroy virus-infected cells and tumor cells through a process that doesn't require prior exposure to the pathogen.

3. Chemical Defenses:

Cytokines: Signaling proteins released by immune cells that mediate and regulate immunity and inflammation.Complement Proteins A group of proteins that enhance the immune

#### Chapter 3: Antigens & Antibodies

#### Dr. R. Arunkumar

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction:

Antigens are any substances, typically foreign molecules or structures, that trigger an immune response in the body. These can be parts of pathogens such as bacteria, viruses, or fungi, or they can be non-living substances like toxins or foreign particles.

When the immune system recognizes an antigen, it sees it as "non-self," meaning it is different from the body's own cells, and this recognition often leads to the production of **antibodies**. Antibodies are specific proteins created by **B lymphocytes** (in the adaptive immune system) that bind to the antigen to neutralize or destroy it.

In summary:

- **Antigens** = foreign molecules that trigger an immune response.
- **Antibodies** = proteins produced by the immune system that recognize and bind to specific antigens to help eliminate them.

This antigen-antibody interaction is a key aspect of how the adaptive immune system protects the body from infections and how vaccines work by introducing a harmless form of an antigen to stimulate immunity.

Antigens have several important properties that influence their **immunogenicity**—the ability to provoke an immune response. These properties are essential for determining how effective an antigen will be at inducing the production of antibodies and initiating other immune system responses.

**Foreignness** refers to how different the antigen is from the host's own molecules. The more foreign an antigen appears to the immune system, the more likely it will trigger a response. Self-antigens (those naturally found in the body) are usually ignored due to immune tolerance, while non-self antigens from pathogens or foreign particles induce a reaction.

#### Chapter 4: Antigen Processing and Presentation

#### Dr. J. Illamathi

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

T cells play a critical role in the adaptive immune response by recognizing antigens presented by other cells. Unlike B cells, which can recognize native antigens directly, **T cells** require antigen processing and presentation on Major Histocompatibility Complex (MHC) molecules. There are two major types of T cells involved in antigen recognition: **Cytotoxic T cells (TC)** and **Helper T cells (TH)**, and both follow similar principles in recognizing antigens.

A. Common Elements of T<sub>C</sub> and T<sub>H</sub> Cell Recognition:

1. T cells only recognize peptide antigens, not native proteins.

Proteins must first be degraded into peptides inside a cell before being presented to T cells.

#### 2. Peptides must be presented on MHC molecules.

T cells do not recognize free-floating antigens; the peptides must be bound to MHC molecules on the surface of antigen-presenting cells (APCs).

3. T-cell receptors (TCRs) recognize the combination of self-MHC and the peptide antigen.

The TCR binds to a complex formed by the peptide antigen and the MHC molecule. This combination of self (MHC) and non-self (antigen) ensures that T cells are responding to infections or abnormal conditions rather than normal, healthy cells.

#### B. Specific T Cells

#### 1. Cytotoxic T Cells (T<sub>C</sub> Cells):

T<sub>C</sub> cells are specialized in directly attacking and destroying infected or abnormal cells, particularly those infected by viruses or displaying cancerous changes.

#### **Key Characteristics**:

#### a. CD8 Coreceptor:

2. T<sub>C</sub> cells have a **CD8** protein that assists their receptor in recognizing antigens. CD8 helps the TCR bind more effectively to the MHC complex.

#### b. Class I MHC Recognition:

T < sub>C < /sub> cells bind to antigens presented on MHC Class I molecules. MHC Class I molecules are found on the surface of almost all nucleated cells in the body.

#### Chapter 5: Cytokines and Signalling

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

The immune system provides a solid framework for understanding how immune cells communicate and coordinate responses. Information transfer is essential for the proper functioning of the immune system, enabling cells to coordinate their responses to pathogens effectively. This communication occurs through direct cell contact and signaling molecules like cytokines.

#### A. Cell-to-Cell Contact

#### 1. TH Cells are Activated by APCs:

Antigen-Presenting Cells (APCs): These include dendritic cells, macrophages, and B cells. APCs process and present antigens on MHC Class II molecules, which are recognized by CD4+ Helper T cells (TH).

*Activation Process:* The binding of the T cell receptor (TCR) on TH cells to the antigen-MHC II complex, along with additional co-stimulatory signals, is crucial for full activation and proliferation of TH cells. This initiates the adaptive immune response.

#### **3.** B Cells and T Cells Interaction:

*Immune Synapse:* The formation of the immune synapse between activated B cells and TH cells is essential for effective communication. This physical interaction allows for the transfer of important cytokines from TH cells to B cells.

*Differentiation:* These signals promote B cell proliferation and differentiation into plasma cells, which produce antibodies, thus enhancing the humoral immune response.

#### 3. TC Cells Activation by Direct Contact:

*Cytotoxic T Cells (TC):* These cells, characterized by CD8 co-receptors, recognize and bind to antigens presented on MHC Class I molecules on infected or malignant cells.

*Direct Attack:* Upon recognition, TC cells induce apoptosis in target cells through the release of performs and granzymes, thereby eliminating infected or abnormal cells.

#### 4. Macrophage Activation by Pathogens:

*Toll-like Receptors (TLRs):* Macrophages express TLRs that recognize pathogen-associated molecular patterns (PAMPs), leading to their activation.

#### Chapter 6: Complement

#### Dr. R. Kamaraj

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction:

The mammalian immune system is a highly sophisticated and complex network of biochemical processes designed to detect and respond to a wide array of pathogens, ensuring the survival of the host. This complexity is necessary because of the sheer number of pathogenic threats and host processes that can lead to disease. The immune system has traditionally been divided into two broad but interconnected subsystems: innate immunity and adaptive immunity.

#### Adaptive Immune System

The adaptive immune system is organized around T cells and B cells, which are specialized lymphocytes capable of recognizing and responding to specific antigens. The key features of the adaptive immune system include:

- Antigen-specific receptors: T and B cells express a diverse range of antigen-specific receptors that allow the immune system to identify specific pathogens.
- Specificity and memory: Adaptive immunity not only targets specific pathogens but also generates immunological memory, enabling faster and stronger responses upon reinfection.
- Tailored responses: The adaptive system ensures that immune responses are customized for different pathogens, leading to highly efficient pathogen clearance.

#### Innate Immune System

In contrast, the innate immune system provides immediate, nonspecific responses to threats and includes mechanical, cellular, and humoral components that form the first line of defense. These processes are evolutionarily older and are crucial for protecting the host from initial infection. Despite being nonspecific, the innate immune system plays a critical role in detecting and eliminating a wide range of pathogens and damaged cells.

#### The Complement System

One of the most important components of innate immunity is the complement system. Discovered in 1896 by Bordet, the complement system consists of a network of plasma and membrane-associated proteins that can trigger highly regulated inflammatory and cytolytic responses. These proteins help eliminate pathogens like bacteria, viruses, and parasites, as well as damaged or malignant cells, by marking them as "nonself." The complement system operates by:

- Opsonization: Coating pathogens to make them more recognizable to immune cells.
- Cytolysis: Direct destruction of pathogen membranes.
- Inflammation: Promoting an inflammatory response to recruit immune cells to the site of infection.

#### Chapter 7: The Major Histocompatibility Complex

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

The Major Histocompatibility Complex (MHC) is fundamental to the immune system's ability to identify and respond to pathogens. By presenting antigens to T cells, MHC molecules facilitate the immune response, ensuring effective recognition and elimination of threats. The polymorphism and diversity of MHC molecules are crucial for the adaptability of the immune response, playing a significant role in the success of organ transplantation and understanding autoimmune diseases. The complex interactions between MHC molecules and T cells underscore the sophistication of the immune system and its ability to protect the host. The Major Histocompatibility Complex (MHC) and its role in the immune system are well-articulated. Here's an expanded discussion that delves deeper into the characteristics and functions of MHC Class I and Class II molecules, as well as their implications in immune responses and organ transplantation.

I. Basic Characteristics of the Major Histocompatibility Complex (MHC)

The Major Histocompatibility Complex (MHC) is essential for the immune system's ability to recognize and respond to foreign antigens. By presenting these antigens to T cells, MHC molecules play a pivotal role in distinguishing between self and non-self.

A. Genes: Human HLA Complex on Chromosome 6

The Human Leukocyte Antigen (HLA) complex is located on chromosome 6 and is organized into three main regions: Class I MHC, Class II MHC, and Class III MHC, each serving distinct functions in immune responses.

#### 1. Class I MHC

a. Three-Domain Protein:

- Class I MHC molecules consist of an  $\alpha$ -chain with three extracellular domains and a transmembrane region.

#### b. Peptide Association:

- The  $\alpha$ -chain associates with  $\beta$ 2-microglobulin, which is encoded by a gene on chromosome 15. This interaction is crucial for the stability and function of the Class I MHC molecule.

## APPLICATION OF AUXINS

### EDITED BY DR. T. VEERAMANI



#### **Application Of Auxins**

Copyright©2021 by Dr. T. Veeramani

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISSN: 978-93-6255-714-8

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology(PRIST)

#### TABLE OF THE CONTENTS

Contents	
CHAPTER :1 ABSCISIC ACID Dr. T. Veeramani	1
CHAPTER :2 AUXIN CONCENTRATION GRADIEENTSDr. T. Veeramani	12
CHAPTER :3 BIOSYNTHESIS OF AUXIN Dr. T. Veeramani	24
CHAPTER :4 CYTOKININ Dr. J. Ilamathi	31
<b>CHAPTER :5</b> STRUCTURE AND TYPES OF GIBBERELLINS <b>Dr. T. Veeramani</b>	45
<b>CHAPTER :6</b> GRAVITROPISM <b>Dr. Arjunpandian</b>	59
CHAPTER :7 GROWTH REGULATOR Dr. T. Veeramani	70
CHAPTER :8 INDOLEACETIC ACID Dr. R. Arunkumar	85
CHAPTER :9 PHOTOTROPISM Dr. T. Veeramani	103
CHAPTER :10 TROPISM Dr. T. Veeramani	117
References	118

#### ABSCISIC ACID

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Abscisic acid (ABA) is a vital plant hormone that plays a critical role in regulating various physiological processes, particularly those related to stress responses. Discovered in the 1960s, ABA was initially identified for its role in fruit ripening and leaf abscission. Today, it is recognized for its function in mediating plant responses to environmental stresses such as drought, salinity, and cold temperatures. ABA is synthesized primarily in response to stress conditions, making it essential for plant survival and adaptation.

#### **Chemical Structure and Biosynthesis**

#### **Chemical Structure**

Abscisic acid is a sesquiterpenoid compound, with the chemical formula C15H20O4. Its structure features a bicyclic ring, which is crucial for its biological activity. The unique arrangement of functional groups allows ABA to interact with specific receptors in plant cells, initiating various signaling pathways.

#### Biosynthesis

ABA is primarily synthesized in leaves and roots, particularly under stress conditions. The biosynthetic pathway begins with the conversion of carotenoids (such as violaxanthin) into xanthoxin, which is then converted into abscisic aldehyde and finally oxidized to form ABA. Key enzymes involved in this pathway include:

- **9-cis-epoxycarotenoid dioxygenase (NCED)**: This enzyme catalyzes the first and ratelimiting step of ABA biosynthesis.
- Abscisic aldehyde oxidase: Converts abscisic aldehyde to ABA, completing the biosynthetic pathway.

#### **Mechanism of Action**

Abscisic acid exerts its effects primarily through a series of signaling pathways that regulate gene expression and physiological responses. The mechanism involves the following key components:

#### AUXIN CONCENTRATION GRADIEENTS

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Auxins are a class of plant hormones that play a critical role in regulating various aspects of plant growth and development. First identified in the early 20th century, the most widely studied auxin is indole-3-acetic acid (IAA). Auxins influence numerous physiological processes, including cell elongation, apical dominance, root formation, and tropic responses. One of the key mechanisms by which auxins exert their effects is through the establishment of concentration gradients within plant tissues.

#### **Mechanisms of Auxin Distribution**

Auxin concentration gradients are formed through several mechanisms:

- **Polar Transport**: Auxins are actively transported in a polar manner, primarily through the action of specific transport proteins. The primary transporter, PIN-FORMED (PIN) proteins, facilitate the directional movement of auxins from the shoot apex downwards. This polar transport is essential for establishing auxin gradients in tissues.
- Cellular Synthesis and Degradation: Auxin levels can be modulated by local synthesis in specific tissues, as well as by degradation processes. Enzymes such as auxinconjugases and oxidases play roles in degrading excess auxin, contributing to the dynamic nature of concentration gradients.
- **Diffusion**: While polar transport is crucial, diffusion also contributes to the establishment of gradients. Auxins can move passively from regions of higher concentration to lower concentration, further shaping the overall distribution within tissues.

#### **Role of Environmental Factors**

Environmental factors such as light and gravity significantly influence auxin distribution. For instance, in phototropism, light exposure on one side of the plant leads to localized auxin accumulation on the shaded side, promoting differential growth and causing the plant to bend towards the light source. Similarly, in gravitropism, auxins accumulate on the lower side of a plant in response to gravity, leading to upward curvature in shoots and downward growth in roots.

#### **BIOSYNTHESIS OF AUXIN**

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Auxins are a class of plant hormones that play critical roles in growth and development, including cell elongation, root formation, and response to light and gravity. The primary auxin in most plants is indole-3-acetic acid (IAA). Understanding the biosynthesis of auxins is essential for grasping how plants regulate their growth and respond to environmental stimuli.

#### Auxin Biosynthesis Pathways

#### 1. Tryptophan-Dependent Pathway

The most well-studied biosynthetic route for IAA is the tryptophan-dependent pathway. Tryptophan, an amino acid, serves as the precursor for IAA synthesis through several enzymatic steps.

#### Key Enzymes and Steps

- 1. **Tryptophan Aminotransferase (TAA)**: Converts tryptophan to indole-3-pyruvic acid (IPyA).
- 2. **Indole-3-Pyruvate Decarboxylase (IPD)**: Catalyzes the decarboxylation of IPyA to produce indole-3-acetaldehyde.
- 3. Indole-3-Acetaldehyde Oxidase (IAAO): Converts indole-3-acetaldehyde to IAA.

#### 2. Tryptophan-Independent Pathways

While the tryptophan-dependent pathway is predominant, several tryptophan-independent pathways have also been identified. These pathways can produce IAA from various precursors. **Key Pathways** 

- 1. **Indole-3-Acetonitrile Pathway**: Involves the conversion of indole-3-acetonitrile to IAA.
- 2. **Indole-3-Acetic Acid from Tryptamine**: This route allows for the production of IAA from tryptamine via intermediate steps.

#### CYTOKININ

#### Dr. J. Ilamathi

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Cytokinins are a class of plant hormones that play vital roles in regulating various aspects of plant growth and development, including cell division, shoot and root growth, and the modulation of responses to environmental stress. Understanding cytokinin biosynthesis and its physiological functions provides insights into plant development and potential agricultural applications.

#### Cytokinin Biosynthesis Pathways

Cytokinins are primarily derived from adenine, a nucleobase, and are synthesized in various tissues, particularly in the roots, where they are produced in response to specific environmental conditions.

#### **1. Key Biosynthetic Pathways**

Cytokinins can be synthesized via several pathways, with the primary pathways involving the conversion of adenine derivatives. The most significant pathways include:

#### a. Adenine Derivative Pathway

- 1. **Conversion of Adenosine**: Adenosine is converted to isopentenyladenosine (iP), a key precursor of several cytokinins.
- 2. **Isoprenoid Side Chain Addition**: The isopentenyltransferase enzyme catalyzes the addition of isoprenoid side chains to adenine or its derivatives, producing cytokinins like isopentenyladenine (iP), zeatin, and dihydrozeatin.

#### **b.** Precursor Pathways

• **Tryptophan Pathway**: In some plant species, cytokinins can also be synthesized from tryptophan, through a series of enzymatic steps that yield compounds that can be further converted to active cytokinins

#### **Chemical Structure and Types of Cytokinins**

Cytokinins possess a purine-like structure characterized by an adenine base with various side chains that determine their biological activity. They can be categorized into two main types:

#### **Natural Cytokinins**

1. Zeatin: Found in corn, zeatin is the most abundant and widely studied natural cytokinin.

#### STRUCTURE AND TYPES OF GIBBERELLINS

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Gibberellic acid (GA) is a plant hormone belonging to the gibberellins group, which plays a crucial role in regulating various growth and developmental processes in plants. Discovered in the early 20th century from a fungal pathogen, GA has since become a vital tool in agriculture and horticulture. This document explores the structure, functions, mechanisms, applications, and implications of gibberellic acid in plant biology.

#### **Chemical Structure**

Gibberellic acid is a terpenoid compound, specifically a diterpene, characterized by its fourring structure. The chemical formula for GA is C19H22O6, and it features a hydroxyl group that contributes to its hormonal activity. The structure allows GA to interact with specific receptors in plants, triggering a cascade of physiological responses.

#### **Types of Gibberellins**

Gibberellins encompass a wide variety of compounds, with over 130 identified types. GA3 (gibberellic acid) is the most widely studied and used form in agricultural applications, but other gibberellins, such as GA1, GA4, and GA7, also play significant roles in plant growth and development.

#### **Functions of Gibberellic Acid**

Gibberellic acid regulates several essential processes in plants:

#### 1. Stem Elongation

One of the most notable effects of GA is promoting stem elongation. By stimulating cell division and elongation in the stem, GA enables plants to grow taller and compete more effectively for light. This is particularly important for species in dense vegetation where light competition is high.

#### 2. Seed Germination

GA plays a crucial role in breaking seed dormancy and promoting germination. It activates enzymes that break down stored starches in seeds, converting them into sugars that provide energy for the growing embryo. This process is essential for the successful establishment of new plants.

#### GRAVITROPISM

#### Dr. Arjunpandian

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Gravitropism, also known as geotropism, is the directional growth response of plants to gravity. This phenomenon is crucial for plant orientation and stability, influencing root and shoot development. Gravitropism ensures that roots grow downward into the soil, anchoring the plant and allowing access to water and nutrients, while shoots grow upward toward light. This document explores the mechanisms, types, influencing factors, applications, and implications of gravitropism in plant biology.

#### **Mechanisms of Gravitropism**

#### **Perception of Gravity**

Plants perceive gravity through specialized cells called **statocytes**, which contain statoliths—dense, starch-filled organelles. These statoliths settle at the bottom of the statocytes due to gravity, allowing the plant to detect its orientation. The process of gravity perception involves several key steps:

- 1. **Statolith Movement**: When the plant is oriented differently, the statoliths shift position, signaling the direction of gravity. This movement activates mechanosensitive ion channels, initiating a signaling cascade.
- 2. **Hormonal Response**: The primary plant hormones involved in gravitropism are **auxins**. In response to gravity, auxin distribution becomes uneven within the plant. Auxins accumulate on the lower side of the plant, promoting differential growth.
- 3. **Cell Elongation**: The higher concentration of auxins on the lower side stimulates cell elongation in that region. As a result, the plant bends in the direction of the force of gravity, allowing roots to grow downward and shoots to grow upward.

#### **Signal Transduction Pathways**

The signaling pathways involved in gravitropism are complex and involve various proteins and molecular mechanisms:

- Auxin Transport Proteins: Proteins such as PIN and AUX/LAX are responsible for the polar transport of auxins. These proteins facilitate the redistribution of auxins in response to gravity.
- **DELLA Proteins**: These proteins act as growth repressors. In gravitropic responses, the degradation of DELLA proteins is influenced by auxins, allowing for enhanced growth on the lower side of the plant.

#### GROWTH REGULATOR

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Growth regulators, also known as plant growth regulators (PGRs), are substances that influence the growth and development of plants. These compounds can be naturally occurring or synthetic and are used to modify physiological processes, enhance growth, and improve yields in agricultural practices. Understanding their types, mechanisms, and applications is essential for optimizing plant health and productivity.

#### **Types of Growth Regulators**

Growth regulators can be classified into several categories based on their functions and chemical nature:

#### 1. Auxins

Auxins are a class of plant hormones that play a critical role in cell elongation, root formation, and response to light and gravity.

- **Functions**: They promote cell elongation, root initiation, and fruit development. Auxins also regulate phototropism (growth toward light) and gravitropism (growth in response to gravity).
- **Common Auxins**: Indole-3-acetic acid (IAA) is the most prevalent natural auxin, while synthetic auxins like 2,4-D are widely used in agriculture.

#### 2. Gibberellins

Gibberellins are a group of hormones that promote stem elongation, seed germination, and flowering.

- **Functions**: They stimulate growth by promoting cell division and elongation. Gibberellins also break seed dormancy and enhance fruit development.
- **Common Uses**: Applications include increasing the size of fruits, promoting seed germination, and extending the growing season for certain crops.

#### 3. Cytokinins

Cytokinins are involved in cell division and growth, particularly in shoot and root development.

#### INDOLEACETIC ACID

#### Dr. R. Arunkumar

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Indoleacetic acid (IAA) is a pivotal plant hormone belonging to the class of auxins, which play crucial roles in regulating plant growth and development. Discovered in the early 20th century, IAA is essential for various physiological processes, including cell elongation, root formation, and response to light and gravity. This document explores the structure, functions, mechanisms of action, applications, and implications of indoleacetic acid in plant biology.

#### **Structure of Indoleacetic Acid**

#### **Chemical Composition**

Indoleacetic acid is a simple organic compound with the chemical formula C18H19NO2. Its structure consists of an indole ring—a bicyclic structure composed of a six-membered benzene ring fused to a five-membered nitrogen-containing ring—attached to an acetic acid moiety. This configuration allows IAA to effectively interact with specific receptors and transport proteins within plant cells, facilitating its diverse biological activities.

#### Biosynthesis

IAA is synthesized primarily from the amino acid tryptophan through several pathways, including the indole-3-pyruvic acid pathway and the indole-3-acetic acid biosynthesis pathway. The biosynthesis process involves various enzymatic reactions, with tryptophan as the primary precursor. This biosynthetic pathway underscores the importance of amino acids in plant hormone production and overall plant physiology.

#### **Functions of Indoleacetic Acid**

Indoleacetic acid regulates numerous critical processes in plants:

#### **1. Cell Elongation**

One of the most well-known roles of IAA is promoting cell elongation, particularly in stem and leaf tissues. By stimulating the loosening of cell walls, IAA enables cells to expand, contributing to overall plant growth. This process is vital for plants to reach light and optimize photosynthesis.

#### PHOTOTROPISM

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Phototropism

An organism's growth in response to a light stimulation is known as phototropism. Although phototropism is most frequently seen in plants, it can also happen in other creatures, such fungus. When phototropism takes place, a hormone called auxin, which is found in the plant's cells furthest from the light, reacts. The cells on the side of the plant that is farthest from the light become longer as a result. Plants exhibit a variety of tropisms, or motions, in response to external stimuli, including phototropism. Positive phototropism refers to growth in the direction of light, whereas negative phototropism refers to growth in the opposite direction. While negative phototropism can apply to either growth away from a light source or growth towards darkness, it should not be confused with skototropism, which is defined as growth towards darkness. In order to optimize phototropism and rearrange their chloroplasts in the leaves. Certain vine shoot tips have negative phototropism, which enables them to climb and grow toward solid, dark objects. Plants can grow in the right direction when phototropism and gravitropism work together.

The plant uses a number of signaling molecules to identify the source of light, and these molecules trigger a number of genes that alter hormone gradients and enable the plant to grow in the direction of the light. The coleoptile, the very tip of the plant, is essential for light perception. The shoot curvature is seen in the central part of the coleoptile. Auxin will migrate to the shaded side and encourage cell elongation on that side, causing the plant to curve towards the light source, according to the Cholodny–Went theory, which was proposed in the early 20th century. Proton pumps are activated by auxins, lowering the pH in the cells on the dark side. Furthermore, more solutes enter the plant cells on the dark side of the plant due to enhanced proton pump activity, which raises the osmotic gradient between the symplast and apoplast of these plant cells. Following its osmotic gradient, water then enters the cells.

#### TROPISM

#### Dr. T. Veeramani

#### Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Tropism refers to the directional growth response of plants to external stimuli, allowing them to optimize their growth and development in relation to their environment. This phenomenon is crucial for plants' survival, enabling them to adapt to varying conditions such as light, gravity, moisture, and touch. The main types of tropism include phototropism, gravitropism, hydrotropism, and thigmotropism. This document explores the mechanisms, types, influencing factors, applications, and implications of tropism in plant biology.

#### **Mechanisms of Tropism**

#### **Perception of Stimuli**

Plants perceive environmental stimuli through specialized cells and receptors. The mechanisms involved in tropic responses generally include:

- 1. **Signal Perception**: Plants have photoreceptors for light (such as phototropins and phytochromes), statocytes for gravity perception, and mechanoreceptors for touch. These receptors detect specific environmental cues.
- 2. **Hormonal Response**: The primary plant hormones involved in tropism are **auxins**, which regulate growth by promoting cell elongation. The redistribution of auxins in response to stimuli is a central mechanism in various tropic responses.
- 3. **Cell Elongation**: In tropic responses, auxins accumulate on one side of the plant, leading to differential growth. Cells on the side with higher auxin concentrations elongate more than those on the opposite side, causing the plant to bend toward or away from the stimulus.

#### **Signal Transduction Pathways**

The response to stimuli involves complex signal transduction pathways:

- Auxin Transport: Auxins are transported through the plant in a polar manner, which allows them to redistribute in response to directional stimuli. Proteins like PIN and AUX/LAX facilitate this transport.
- **Gene Expression**: The binding of auxins to their receptors activates the expression of specific genes associated with growth and development. This gene regulation is critical for the physiological changes that occur during tropic responses.

## **MINERAL NUTRITION**

978-93-6255-371-3

EDITED BY

### **DR. T. VEERAMANI**

## **Mineral Nutrition**

Copyright©2021 by Dr. T. Veeramani

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISSN: 978-93-6255-371-3

Publisher: PonnaiyahRamajayam Institute of Scienceand Technology (PRIST)

#### TABLE OF THE CONTENTS

Contents	
CHAPTER :1Role of Nutrients Dr. T. Veeramani	1
CHAPTER :2 Micronutrients Dr. R. Arunkumar	12
CHAPTER :3Macronutrients Dr. T. Veeramani	24
CHAPTER :4Methods to study the mineral requirements of plants Dr. J. Ilamathi	31
CHAPTER :5 Essential mineral elements Dr. T. Veeramani	45
CHAPTER :6Criteria for essentiality Dr. Arjunpandian	59
CHAPTER :7 Role of Macro- and Micro-nutrients Dr. R. Kamaraj	70
CHAPTER :8 Deficiency symptoms of Essential Elements Dr. R. Arunkumar	85
CHAPTER :9 Mechanism of observation of Elements Dr. T. Veeramani	103
CHAPTER :10Translocation of Solutes Dr. T. Veeramani	117
References	118

#### ROLE OF NUTRIENTS

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

A nutrition diet is one that meets all of the body's macro and micronutrient demands on a daily basis. And also nutrition is a vital component to overall wellness and health. That being said, there doesn't exist one perfect diet for everyone. Certainly, many diet books would have you believe that everyone can follow a particular nutrient prescription to attain optimal nutrition and body weight, but if that were the case, there would only be one diet book and only one diet. Nutrient intakes are based on individual factors, including a person's size and activity level, the types of activities performed, and genetic predisposition. Additionally, an individual's diet should reflect adequate food variety for nutrient balance, be palatable while meeting the daily demands of stress placed on the body, and contribute to an acceptable quality of life. Diet affects energy, wellbeing and many disease states. There is a connection between lifetime nutritional habits and the risks of many chronic diseases such as Cardio vascular diseases, diabetes and cancer. A well balanced diet can prevent such conditions and improve energy levels and overall health and wellness. 1.1 Growth and development.

- 1. Good nutrition is essential for attainment of normal growth and development during fetal life and childhood. Physical growth, intellectual development, learning and behavior are affected by malnutrition.
- 2. Adequate nutrition is needed for adult life maintenance for optimum health and efficiency.
- Elder people needs special nutrition due to their physiological and chronological changes. Pregnant and lactating mothers require more proteins and nutrients to prevent abortion, growth retardation and low birth weight babies and provide adequate breast feeding for their babies.
- 4. 1.2 Specific deficiency diseases the most common deficiencies find in Indians are Protein energy malnutrition, blindness, goiter, anemia, beriberi, rickets etc. There is increased incidence of abortion, prematurity, still birth and low birth weight babies in malnourished mothers.

#### MICRONUTRIENTS

Dr. R. Arunkumar Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Micronutrients play a central part in metabolism and in the maintenance of tissue function. An adequate intake therefore is necessary, but provision of excess supplements to people who do not need them may be harmful. Single micronutrient deficiency states are comparatively easily recognised and treated. Subclinical deficiency, often of multiple micronutrients, is more difficult to recognise, and laboratory assessment is often complicated by the acute phase response. Clinical benefit is most likely in those people who are severely depleted and at risk of complications, and is unlikely if this is not the case. There is little evidence for supplements leading to a reduction in the incidence of infections in the elderly population, in coronary artery disease, or in malignant disease. The best evidence for benefit is in critical illness, and in children in developing countries consuming a deficient diet. More clinical trials are required with good clinical outcomes to optimise intake in prevention and treatment of disease.

Micronutrients play a central part in metabolism and in themaintenance of tissue function. An adequate intaketherefore is necessary, but provision of excess supplements people who do not need them may be harmful. Singlemicronutrient deficiency states are comparatively easily

recognised and treated. Subclinical deficiency, often ofmultiple micronutrients, is more difficult to recognise, andlaboratory assessment is often complicated by the acutephase response. Clinical benefit is most likely in thosepeople who are severely depleted and at risk ofcomplications, and is unlikely if this is not the case.

There islittle evidence for supplements leading to a reduction in theincidence of infections in the elderly population, incoronary artery disease, or in malignant disease. The bestevidence for benefit is in critical illness, and in children indeveloping countries consuming a deficient diet. Moreclinical trials are required with good clinical outcomes tooptimise intake in prevention and treatment of disease.

#### MACRONUTRIENTS

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Nutrition is the study of the action of biologically active components of food and their role in maintaining human health. Nutrition is an applied medical science. The basic nutritional needs of humans are to supply energy and raw materials for all the various activities and processes that occur in the body. In addition to the need for water, humans require five types of nutrients from their food supply; three of these are required in relatively large amounts and are called macronutrients, consisting of carbohydrates, proteins, and fats. The other two types of nutrients, vitamins and minerals, are required in small amounts and are known as micronutrients

**MICRONUTIENTS:** These are vitamins and minerals. They are called micronutrients because they are required in small amounts which may vary from a fraction of a milligram to several grams. A short review of basic facts about these nutrient is given below.

**PROTIENS:** The word "protein" derived from Greek, means "to come first". Protein is the most abundant components of the body. Indeed, they are of greatest importance in human nutrition.

#### **FUNCTIONS OF PROTEINS**

Protein are needed by the body for Body building, repair and maintenance of body tissues. Maintenance of osmotic pressure, synthesis of certain substances like antibodies, plasma proteins, hemoglobin, enzymes, hormones and coagulation factors. Immune protein maintains. The macronutrients, fat, protein and carbohydrate provide energy and essential components to sustain life. Fat is composed of glycerol and fatty acids; protein is an agglomeration of aminoacids.

#### METHODS TO STUDY THE MINERAL REQUIREMENTS OF PLANTS

#### J. Ilamathi

Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

In 1860, Julius von Sachs, a prominent German botanist, demonstrated, for the first time, that plants could be grown to maturity in a defined nutrient solution in complete absence of soil. This technique of growing plants in a nutrient solution is known as hydroponics. Since then, a number of improvised methods have been employed to try and determine the mineral nutrients essential for plants. The essence of all these methods involves the culture of plants in a soil-free, defined mineral solution. These methods require purified water and mineral nutrient salts. Can you explain why this is so essential? After a series of experiments in which the roots of the plants were immersed in nutrient solutions and wherein an element was added / removed or given in varied concentration, a mineral solution suitable for MINERAL NUTRITION Methods to Study the plant growth was obtained. By this method, essential elements were identified and their deficiency symptoms discovered. Hydroponics has been successfully employed as a technique for the commercial production of vegetables such as tomato, seedless cucumber and lettuce. It must be emphasised that the nutrient solutions must be adequately aerated to obtain the optimum growth. What would happen if solutions were poorly aerated? Diagrammatic views of the hydroponic.

Sometimes you may observe that a potted plant kept in sunlight and provided with sufficient water does not grow. Its leaves look pale and weak. Plant may not even flower properly. Such a situation is an indication, that the plant may not be getting all that is required for normal growth and development. In most of such situations one or more minerals required may be lacking in the soil. You might have seen farmers adding some extra manure (khad) to the soil. In this lesson you will learn the importance of mineral nutrition in plants. After completing this lesson, you will be able to : z define the terms mineral nutrition, macro and micro nutrients; z explain the functions of minerals with reference to the techniques of hydroponics and aeroponics; z list the role of macro and micro nutrients; z mention the deficiency symptoms of macro and micro nutrients; z differentiate between autotrophic and heterotrophic nutrition in plant; z describe the saprophytic and parasitic modes of nutrition in plant.

#### ESSENTIAL MINERAL ELEMENTS

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Most of the minerals present in soil can enter plants through roots. In fact, more than sixty elements of the 105 discovered so far are found in different plants. Some plant species accumulate selenium, some others gold, while some plants growing near nuclear test sites take up radioactive strontium. There are techniques that are able to detect the minerals even at a very low concentration (10-8 g/ mL). The question is, whether all the diverse mineral elements present in a plant, for example, gold and selenium as mentioned above, are really necessary. Minerals are inorganic nutrients, usually required in small amounts from less than 1 to 2500 mg per day, depending on the mineral. As with vitamins and other essential food nutrients, mineral requirements vary with animal species. For example, humans and other vertebrates need large amounts of calcium for construction and maintenance of bone and normal function of nerves and muscles. Phosphorus is an important constituent of adenosine triphosphate (ATP) and nucleic acid and is also essential for acid-base balance, bone and tooth formation. Red blood cells can not function properly without iron in haemoglobin, the oxygen-carrying pigment of red blood cells.

Iron is also an important component of the cytochromes that function in cellular respiration. Magnesium, copper, selenium, zinc, iron, manganese and molybdenum are important co-factors found in the structure of certain enzymes and are indispensable in numerous biochemical pathways. Vertebrates need iodine to make thyroid hormones. Sodium, potassium and chlorine are important in the maintenance of osmotic balance between cells and the interstitial fluid. Magnesium is an important component of chlorophyll in plants. The interactions between nutrition and diseases, nutrition and drug metabolism have been reported. Excessive intake of some minerals can upset homeostatic balance and cause toxic side effects. For example, excess sodium intake is associated with high blood pressure and excess iron can cause liver damage. Also, severe shortages or self-prescribed minerals can alter the delicate balance in body functions that promotes health. The knowledge of the biochemistry of the mineral elements is also essential because individuals suffering from a chronic illness or taking medications that affect the body's use of specific nutrients need to be enlightened. The aim of this paper is to

#### **CHAPTER :6** CRITERIA FOR ESSENTIALITY **Dr. Arjunpandian**

Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Plants, like all other living things, need food for their growth and development. Plants require 16 essential elements. Carbon, hydrogen, and oxygen are derived from the atmosphere and soil water. The remaining 13 essential elements (nitrogen, phosphorus, potassium, calcium, magnesium, sulfur, iron, zinc, manganese, copper, boron, molybdenum, and chlorine) are supplied either from soil minerals and soil organic matter or by organic or inorganic fertilizers. For plants to utilize these nutrients efficiently, light, heat, and water must be adequately supplied. Cultural practices and control of diseases and insects also play important roles in crop production. Each type of plant is unique and has an optimum nutrient range as well as a minimum requirement level. Below this minimum level, plants start to show nutrient deficiency symptoms. Excessive nutrient uptake can also cause poor growth because of toxicity. Therefore, the proper amount of application and the placement of nutrients is important. Soil and plant tissue tests have been developed to assess the nutrient content of both the soil and plants. By analyzing this information, plant scientists can determine the nutrient need of a given plant in a given soil. In addition to the levels of plant-available nutrients in soils, the soil pH plays an important role in nutrient availability and elemental toxicity.

This chapter describes the essential nutrients, the chemical forms in which they are available to plants, their function in plants, symptoms of their deficiencies, and recommended nutrient levels in plant tissues of selected crops. Nitrogen symbol: N; available to plants as nitrate (NO3 –), and ammonium (NH4 +) ions. Nutrient functions N is biologically combined with C, H, O, and S to create amino acids, which are the building blocks of proteins. Amino acids are used in forming protoplasm, the site for cell division and thus for plant growth and development. Since all plant enzymes are made of proteins, N is needed for all of the enzymatic reactions in a plant. N is a major part of the chlorophyll molecule and is therefore necessary for photosynthesis. N is a necessary component of several vitamins.

#### **CHAPTER :7** ROLE OF MACRO- AND MICRO-NUTRIENTS

Dr. R. Kamaraj Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Food and nutrients are the essential components for proper growth and development of plants just like all other living organisms. Plants require sixteen essential nutrient elements including oxygen, hydrogen and carbon potentially derived from hydrosphere, lithosphere and atmosphere while remaining thirteen are chlorine, molybdenum, boron, copper, manganese, zinc, iron, sulphur, magnesium, calcium, potassium, phosphorous and nitrogen that are supplied either through soil organic matter and soil minerals or by inorganic and organic fertillizers. Water, heat and light are three essential components that are required in sufficient quantities for proper utilization of macronutrients and micronutrients by plants. Insects, disease control and cultural practices plays important role in good quality crop production. Each plant type is unique and known to have optimum nutrient range along with minimum requirement level below which plants show significant nutrient deficiency symptoms.

However, excessive uptake of nutrients can lead towards poor growth rate and high level of toxicity. That is why, appropriate quantities of essential plant nutrients are extremely important. Numerous plant tissue and soil quality tests have recently been developed in order to assess the nutrient contents of plants and soil. Through proper analysis of provided information, plant scientists can determine the need of nutrients for a particular plant in specific soil. In addition to levels of available plant nutrients of soil, pH plays essential role in determination of elemental toxicity and nutrient availability. Fertillizers are known chemical constituents including mixture of nitrates and manure that significantly improves plant growth. Nevertheless, in accordance with holistic farm management approach, aims of organic farming are to create socially beneficial, environmentally benign and economically sustainable system for efficient food production and managemen.

The affluent global population is placing unprecedented pressure on the limited land and water resources of planet, underpinning concerns over global food security and its sensitivity to shocks arising from environmental fluctuations, trade policies and market volatility [1]. Number of undernourished people of the world is continuously increasing since 2014 and reached to an estimated eight hundred fifteen million people till 2016 [2].

#### DEFICIENCY SYMPTOMS OF ESSENTIAL ELEMENTS

#### Dr. R. Arunkumar

Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Whenever the supply of an essential element becomes limited, plant growth is retarded. The concentration of the essential element below which plant growth is retarded is termed as critical concentration. The element is said to be deficient when present below the critical concentration. Since each element has one or more specific structural or functional role in plants, in the absence of any particular element, plants show certain morphological changes. These morphological changes are indicative of certain element deficiencies and are called deficiency symptoms. The deficiency symptoms vary from element to element and they disappear when the deficient mineral nutrient is provided to the plant. However, if deprivation continues, it may eventually lead to the death of the plant. The parts of the plants that show the deficiency symptoms also depend on the mobility of the element in the plant. For elements that are actively mobilised within the plants and exported to young developing tissues, the deficiency symptoms tend to appear first in the older tissues. For example, the deficiency symptoms of nitrogen, potassium and magnesium are visible first in the senescent leaves. In the older leaves, biomolecules containing these elements are broken down, making these elements available for mobilising to younger leaves.

The kind of deficiency symptoms shown in plants include chlorosis, necrosis, stunted plant growth, premature fall of leaves and buds, and inhibition of cell division. Chlorosis is the loss of chlorophyll leading to yellowing in leaves. This symptom is caused by the deficiency of elements N, K, Mg, S, Fe, Mn, Zn and Mo. Likewise, necrosis, or death of tissue, particularly leaf tissue, is due to the deficiency of Ca, Mg, Cu, K. Lack or low level of N, K, S, Mo causes an inhibition of cell division. Some elements like N, S, Mo delay flowering if their concentration in plants is low. You can see from the above that the deficiency of any element can cause multiple symptoms and that the same symptoms may be caused by the deficiency of one of several different elements.

Hence, to identify the deficient element, one has to study all the symptoms developed in all the various parts of the plant and compare them with the available standard tables. We must also be aware that different plants also respond differently to the deficiency of the same element.

#### MECHANISM OF OBSERVATION OF ELEMENTS

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### **Classification of Mineral Elements**

Mineral elements can be classified into two main categories based on their amount of requirements. **Macronutrients** are the ones that are required in relatively large amounts by plants. The elements required in the small quantities are called **micronutrients**.

1. Macronutrients: Carbon, Hydrogen, Nitrogen, Phosphorus, Potassium, Calcium, Magnesium, Sulphur, etc.

2. **Micronutrients:** Iron, Manganese, Zinc, Copper, Boron, Molybdenum, Chloride, etc.Role of Essential Elements in PlantsThe essential elements carry out the following functions in the plants:

1. Elements are the components of the plant body. Carbon, hydrogen, oxygen, nitrogen are found as a part of biomolecules.2. Calcium, magnesium, sodium, chlorine, manganese, chlorine, and potassium serve as cofactors of enzymes and are called catalytic components.3. The osmotic potential of the plant, pH of cell sap depends on the concentration of mineral elements and organic compounds present in the cell sap.4. Mineral elements like Na, K, Ca, and Mg found in the cells help in maintaining the electrostatic neutrality of the cells.5. Mineral elements also influence the absorption of water by cells and affect the degree of permeability of cell membranes.Mechanism of Absorption of Elements. Minerals are either in dissolved form or absorbed from. Minerals are absorbed by the cells of epiblema in the maturation and elongation zone of

#### 3. Active Absorption

1. Passive Absorption

The absorption of minerals by physical processes such as **diffusion** without the expenditure of metabolic energy is known as **passive absorption**. An ion moves passively from the region of its higher concentration or higher electrochemical potential to that of lower electrochemical potential. The passive movements of ions usually occur through ion channels. The **ion channels** are transmembrane proteins that function as selective pores. Various hypotheses have been proposed to explain the movement of ions against an ECP gradient. These can be described as follows:

#### TRANSLOCATION OF SOLUTES

#### Dr. T. Veeramani

#### Assistant professor, Department of Biotechnology, PonnaiyahRamajayam Institute of Science and Technology, Tamil Nadu, India

#### Introduction

Translocation of organic solutes The movement of organic food materials or the solutes in soluble form one place to another in higher plants is called as translocation of organic solutes Directions of translocation Translocation of organic solutes may take place in the following directions. 1. Downward translocation Mostly, the organic material is manufactured by leaves and translocated downward to stem and roots for consumption and storage. 2. Upward translocation It takes place mainly during the germination of seeds, tubers etc. When stored food after being converted into soluble form is supplied to the upper growing part of the young seedling till it has developed green leaves. Upward translocation of solutes also takes place through stem to young leaves, buds and flowers which are situated at the tip of the branch. 3. Lateral translocation Radical translocation of organic solutes also takes place in plants from the cells of the pith to cortex. Path of the translocation of organic solutes 1. Path of downward translocation Downward translocation of the organic solutes takes place through phloem.

This can be proved by the ringing experiment. 2. Path of upward translocation Although translocation of organic solutes take place through phloem, but under certain conditions it may take place through xylem. 3. Path of lateral translocation Lateral translocation from pith to cortex takes place through medullary rays. Mechanism of translocation Various theories have been put forward to explain the mechanism of phloem conduction. Among them Munchs' (1930) hypothesis is mot convincing. Munchs mass flow on pressure flow hypothesis According to this hypothesis put forward by Much (1930) and others, the translocation of organic solutes takes place though phloem along a gradient of turgor pressure from the region of higher concentration of soluble solutes (supply end) to the region of lower concentration (consumption end).

The principle involved in this hypothesis can be explained by a simple physical system as shown in Fig. Two members X and Y permeable only to water and dipping in water are connected by a tube T to form a closed system membrane X contains more concentrated sugar solution than in membrane Y. Due to higher osmotic presence of the concentrated sugar solution in the membrane X, water enters into it so that its turgor pressure is increased.



# PROPERTIES OF WATER

## EDITED BY



DR. J. ILAMATHI



#### **PROPERTIES OF WATER**

Copyright©2021 by Dr.J.LIMATHI

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISSN: 978-93-6255-148-1

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

#### TABLE OF THE CONTENTS

Chapter 1: Chemical Structure of Water Dr. J. Illamathi	1
Chapter 2: Biological Importance of Water Dr. T. Veeramani	17
Chapter 3: Molecular Structure and Bonding Dr. R. Kamaraj	26
Chapter 4: Key Properties of Water <i>Dr. R. Arunkumar</i>	32
Chapter 5: Water in Biological Systems Dr. RV. Shalini	49
Chapter 6: Environmental Role of Water Dr. G. Venkatkumar	55
Chapter 7: Key Properties of Water Dr. A. Shajahan	69
Chapter 8: Unique Physical Properties of Water Dr. T. Veeramani	73
Chapter 9: Water in Biological Systems Dr. J. Illamathi	87
Chapter 10: Environmentpollution Dr. R. Arunkumar	95
Reference	112
#### Chapter 1: Chemical Structure of Water Dr. J. Illamathi

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **Chemical Structure of Water**

Water  $(H_2O)$  is a simple molecule composed of two hydrogen atoms covalently bonded to a single oxygen atom. The shape of the water molecule is bent (or V-shaped), with a bond angle of approximately 104.5 degrees. This structure is due to the presence of two lone pairs of electrons on the oxygen atom, which repel the hydrogen atoms, creating the bent geometry.

1. The difference in electronegativity between oxygen and hydrogen results in a polar molecule. Oxygen is more electronegative, pulling the shared electrons closer to it, which creates a partial negative charge ( $\delta$ -) on the oxygen and a partial positive charge ( $\delta$ +) on the hydrogen atoms. This polarity allows water to form hydrogen bonds, which are weak interactions between the hydrogen atom of one water molecule and the oxygen. Water provides habitats for a vast array of organisms. Aquatic ecosystems, such as oceans, rivers, and lakes, are rich in biodiversity, supporting various life forms that rely on water for survival.

#### Chapter 2: Biological Importance of Water Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **Biological Importance of Water**

Water is fundamental to life on Earth. It plays a vital role in numerous biological processes:

- 2. **Metabolism**: Water is involved in metabolic reactions, including hydrolysis and dehydration synthesis. It acts as a reactant in the breakdown of macromolecules and helps in the formation of larger molecules.
- 3. **Transport Medium**: Water serves as a transport medium in organisms. In humans and other animals, blood, which is largely composed of water, carries nutrients, gases, and waste products. In plants, water is transported through xylem and phloem, playing a critical role in nutrient distribution.
- 4. **Thermoregulation**: Water helps regulate body temperature in living organisms. Its high specific heat capacity allows for stable internal temperatures, which is essential for enzymatic and physiological functions.
- 5. **Homeostasis**: Water plays a critical role in maintaining homeostasis within organisms. It helps regulate pH, osmotic pressure, and electrolyte balance, ensuring that cellular processes function optimally.

#### Chapter 3: Molecular Structure and Bonding

Dr. R. Kamaraj Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The water molecule is made up of two hydrogen atoms and one oxygen atom, creating a bent molecular shape due to the arrangement of its bonds and lone pairs of electrons. The oxygen atom's high electronegativity leads to polar covalent bonds, where electrons are shared unequally. This results in a partial negative charge on the oxygen atom and partial positive charges on the hydrogen atoms, giving water its unique polar characteristics.

#### **Hydrogen Bonding**

The polar nature of water molecules allows them to form hydrogen bonds with one another. These bonds are relatively weak compared to covalent bonds but are crucial for many of water's unique properties. Each water molecule can form up to four hydrogen bonds, contributing to the formation of extensive networks of water molecules that are responsible for its high boiling point, surface tension, and solvent capabilities.

#### Chapter 4: Key Properties of Water *Dr. R. Arunkumar*

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **Key Properties of Water**

#### 1. High Specific Heat Capacity

• Water can absorb a significant amount of heat energy without a large change in temperature. This property is critical for moderating the Earth's climate and stabilizing temperatures in bodies of water, which is vital for aquatic life. For instance, coastal areas tend to have milder climates compared to inland areas due to the moderating effect of the ocean.

#### 2. High Heat of Vaporization

The energy required to convert water from a liquid to a gas is substantial. This high heat of vaporization is essential for cooling mechanisms in living organisms. For example, when humans sweat, the evaporation of sweat from the skin surface cools the body effectively, helping to regulate internal temperatures.

#### 3. Density Anomaly

• Unlike most substances, water expands when it freezes, making ice less dense than liquid water. This property is critical for aquatic ecosystems; ice floats on the surface of water bodies, providing insulation to the water below and creating a habitat for organisms during cold weather. This insulating layer helps maintain a stable environment for aquatic life beneath the ice.

#### 4. Universal Solvent

• Water's polarity allows it to dissolve many ionic and polar substances, making it an excellent solvent for biological reactions. In biological systems, this property facilitates the transport of nutrients and waste products, enabling metabolic processes. For example, in cells, water dissolves and transports essential ions and molecules necessary for cellular function Chapter 5: Water in Biological Systems **Dr. RV. Shalini** Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Water in Biological Systems

#### **1. Metabolic Functions**

• Water is involved in metabolic reactions, acting as a reactant in hydrolysis and serving as a medium for biochemical reactions. Hydrolysis, for example, is essential for breaking down complex carbohydrates, proteins, and fats into simpler molecules that can be utilized by cells.

#### 2. Thermoregulation

• In animals, water is a critical component of thermoregulation. The high specific heat capacity of water helps maintain stable internal temperatures, enabling enzymes and biochemical pathways to function optimally. Aquatic organisms rely on water's thermal properties to survive in fluctuating environmental temperatures.

#### **3.** Transport and Circulation

• In humans, blood, composed mostly of water, transports oxygen, nutrients, hormones, and waste products throughout the body. In plants, water is essential for the transport of minerals and nutrients from the soil through the xylem and for photosynthesis, where water is a crucial reactant.

#### 4. Cell Structure and Function

• Water provides structure to cells, maintaining turgor pressure in plant cells and contributing to the fluidity of cellular membranes. The movement of water into and out of cells is essential for maintaining homeostasis, influencing cellular processes such as nutrient uptake and waste removal.

Chapter 6: Environmental Role of Water **Dr. G. Venkatkumar** Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **Environmental Role of Water**

#### 1. Climate Regulation

• Water bodies play a vital role in climate regulation. Oceans and large lakes act as heat sinks, absorbing and redistributing heat from the sun. This process helps regulate weather patterns and moderate temperatures across different regions. For example, the Gulf Stream transports warm water from the tropics to the North Atlantic, influencing the climate of Western Europe.

#### 2. Water Cycle

• The water cycle is a continuous process involving evaporation, condensation, precipitation, and runoff. This cycle is critical for replenishing freshwater resources and maintaining ecosystems. Plants play a key role in the cycle through transpiration, where water is absorbed from the soil and released into the atmosphere.

#### 3. Ecosystem Support

• Water supports diverse ecosystems, providing habitats for countless species. Freshwater and marine environments host unique biodiversity, each species playing a specific role in the ecosystem. Wetlands, for example, are vital for water filtration, flood control, and providing habitat for wildlife.

#### 4. Soil Formation and Erosion

• Water contributes to soil formation and erosion processes. As water flows over land, it can erode soil and rock, transporting sediments and minerals. This process is essential for shaping landscapes and creating fertile soil for agriculture.

#### Chapter 7: Key Properties of Water

#### Dr. A. Shajahan

Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **Key Properties of Water**

#### 1. High Specific Heat Capacity

• Water can absorb a significant amount of heat energy without a large change in temperature. This property is critical for moderating the Earth's climate and stabilizing temperatures in bodies of water, which is vital for aquatic life. For instance, coastal areas tend to have milder climates compared to inland areas due to the moderating effect of the ocean.

#### 2. High Heat of Vaporization

• The energy required to convert water from a liquid to a gas is substantial. This high heat of vaporization is essential for cooling mechanisms in living organisms. For example, when humans sweat, the evaporation of sweat from the skin surface cools the body effectively, helping to regulate internal temperatures.

#### 3. Density Anomaly

• Unlike most substances, water expands when it freezes, making ice less dense than liquid water. This property is critical for aquatic ecosystems; ice floats on the surface of water bodies, providing insulation to the water below and creating a habitat for organisms during cold weather. This insulating layer helps maintain a stable environment for aquatic life beneath the ice.

#### 4. Universal Solvent

Water's polarity allows it to dissolve many ionic and polar substances, making it an excellent solvent for biological reactions. In biological systems, this property facilitates the transport of nutrients and waste products, enabling metabolic processes. For example, in cells, water dissolves and transports essential ions and molecules necessary for cellular functions.

#### Chapter 8: Unique Physical Properties of Water

Dr. T. Veeramani Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **Unique Physical Properties of Water**

- 1. **High Specific Heat Capacity**: Water has a high specific heat capacity, meaning it can absorb a lot of heat before its temperature changes significantly. This property helps regulate temperatures in the environment, stabilizing climates and providing a suitable habitat for various organisms.
- 2. **High Heat of Vaporization**: The amount of energy required to convert water from a liquid to a gas is high. This property is critical for cooling processes, such as sweating in humans and transpiration in plants. As water evaporates, it removes heat from the surface, helping to maintain temperature.
- 3. **Density Anomaly**: Water is unique in that it is less dense as a solid than as a liquid. This anomaly is due to the hydrogen bonding in ice, which forms a crystalline structure that is more spacious than the arrangement of water molecules in liquid form. As a result, ice floats on water, providing insulation for aquatic life during cold weather.
- 4. **Solvent Properties**: Water is often called the "universal solvent" because it can dissolve a wide range of substances, including salts, sugars, acids, and gases. This property is vital for biological processes, as it facilitates the transport of nutrients and waste products in living organisms.
- 5. **Cohesion and Adhesion**: Water molecules exhibit strong cohesive forces due to hydrogen bonding, which allows them to stick to each other. This cohesion is responsible for surface tension, enabling small objects to float on the surface of water. Additionally, water's adhesive properties allow it to stick to other substances, which is essential for processes such as capillary action in plants.
- 6. **High Dielectric Constant**: Water has a high dielectric constant, which reduces the electrostatic forces between charged particles in solution. This property is crucial for biochemical reactions, as it helps stabilize ions and polar molecules, facilitating various chemical processes.

Chapter 9: Water in Biological Systems

Dr. J. Illamathi Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### Water in Biological Systems

#### 1. Metabolic Functions

• Water is involved in metabolic reactions, acting as a reactant in hydrolysis and serving as a medium for biochemical reactions. Hydrolysis, for example, is essential for breaking down complex carbohydrates, proteins, and fats into simpler molecules that can be utilized by cells.

#### 2. Thermoregulation

• In animals, water is a critical component of thermoregulation. The high specific heat capacity of water helps maintain stable internal temperatures, enabling enzymes and biochemical pathways to function optimally. Aquatic organisms rely on water's thermal properties to survive in fluctuating environmental temperatures.

#### **3.** Transport and Circulation

• In humans, blood, composed mostly of water, transports oxygen, nutrients, hormones, and waste products throughout the body. In plants, water is essential for the transport of minerals and nutrients from the soil through the xylem and for photosynthesis,

#### **Environmental Role of Water**

#### 1. Climate Regulation

 Water bodies play a vital role in climate regulation. Oceans and large lakes act as heat sinks, absorbing and redistributing heat from the sun. This process helps regulate weather patterns and moderate temperatures across different regions. For example, the Gulf Stream transports warm water from the tropics to the North Atlantic, influencing the climate of Western Europe.

#### Chapter 10: Environment pollution

#### Dr. R. Arunkumar

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

#### **Environment production**

#### 1. Pollution

• Water pollution is a significant global issue, caused by industrial discharge, agricultural runoff, plastic waste, and sewage. Pollutants can have severe impacts on aquatic ecosystems, harming wildlife and affecting water quality for human use. Contaminated water can lead to health issues, including gastrointestinal diseases and other illnesses.

#### 2. Scarcity

 Despite being abundant, freshwater scarcity is a critical concern in many parts of the world. Factors such as climate change, over-extraction, and pollution are exacerbating water shortages. Regions with limited access to clean water face challenges in agriculture, health, and economic development.

#### 3. Climate Change Effects

 Climate change is altering precipitation patterns, increasing evaporation rates, and causing more extreme weather events. These changes affect the availability and quality of water resources. Droughts and floods can lead to water shortages and contamination, impacting both human populations and ecosystems.

#### 4. Over-extraction of Water Resources

 Unsustainable extraction of groundwater and surface water for agriculture, industry, and urban use can lead to depletion of water resources. This overextraction can result in ecological degradation, loss of biodiversity, and conflicts over water rights among communities.

# PLANT AND ANIMAL BIOTECHNOLOGY

Edited by **DR. T. VEERAMANI** 



#### PLANT AND ANIMAL BIOTECHNOLOGY

Copyright©2021 by Dr. T. Veeramani

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISSN: 978-93-6255-061-3

Publisher: Ponnaiyah Ramajayam Institute of Scienceand Technology (PRIST)

#### TABLE OF THE CONTENTS

Chapter 1: Commercialization of animal biotechnology Dr. T. Veeramani	1
Chapter 2: Gene stacking in transgenic plant Dr. R. Arunkumar	21
Chapter 3: Applications of biotechnology and genomics in potato improvement <b>Dr. T Veeramani</b>	32
Chapter 4: Tapping RNA silencing pathways for plant biotechnology Dr. J. Illamathi	42
Chapter 5: Biotechnology in Animal Agriculture Dr. A. Shajahan	52
Chapter 6: Recent advances rice biotechnology—towards genetically superior transgenic rice <b>Dr. G. Venkatkumar</b>	68
Chapter 7: Plant-produced viral bovine vaccines Dr. RV. Shalini	74
Chapter 8: biotechnology—a sustainable alternative for chemical Industry Dr. R. Arunkumar	85
Chapter 9: Production and glycosylation of plant-made pharmaceuticals: the antibodies as a challenge. <b>Dr. T. Veeramani</b>	97
<b>Chapter 10:</b> Bioreactor technology: A novel industrial tool for high-tech production of bioactive molecules and biopharmaceuticals from plant roots <b>Dr. Ariunpandian</b>	110
Reference	124

Chapter 1: Commercialization of animal biotechnology

### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Through the application of animal biotechnology, genetically altered animals with enhanced growth rates, illness resistance, or the ability to synthesis therapeutic proteins have been created. Recent developments in gene expression, metabolic profiling of animal cells, and genome sequencing have all contributed to advances in animal biotechnology. Recent advancements in genome editing technologies, such as CRISPR-Cas systems, TALENS, and zinc finger nucleases, have made it easier to produce genetic changes in animals that can enhance agricultural output, disease resistance, and animal health and well-being. The topic of animal biotechnology commercialization is a broad one. We shall try to review embryo transfer (ET) and associated technologies in relation to food-producing mammals in this work. An overview of biotechnology's development across time will give an idea of its current and potential uses. Animal biotechnology is now being commercialized in two ways.

Animals are used for biomedical purposes in the first application. Few businesses have acquired all the necessary intellectual property and core competencies to bridge the gap between lab bench and product. The production of animals for food is the second application pathway. The toolset for current and upcoming applications includes cloning, transgenics, in vitro fertilization (IVF), artificial insemination (AI), embryo transfer, and genomics. When used alone, these are effective instruments that can result in notable increases in output. Within the next ten years, even more substantial change will be brought about by the combination of these technologies with information systems and data analysis.

#### Chapter 2: Gene stacking in transgenic plant

#### Dr. R. Arunkumar

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Transgenic crops have the potential to promote revolutionary change in agriculture, industry, nutrition and even medicine. By manipulating plant genomes, crops can be engineered to provide enhanced nutritional value and to be resistant to biotic and abiotic stresses, plant raw materials can be better adapted to the requirements of industry, and 'green factories' can be used to produce a host of novel products, including pharmaceuticals, in an environmentally benign and sustainable manner. Although much has been written about the staggering potential of genetically modified (GM) crops, achievements to date have been relatively modest, as the technology itself is still in its infancy.

This gap between the hyperbole and the reality of the relatively crude first-generation 'input trait' GM crops, perceived by many to offer benefits only to seed companies and farmers, has prompted several commentators and pressure groups to deem the technology a failure. Progress towards second-generation 'output trait' products with nutritional, environmental or other benefits that consumers can appreciate directly has been slow, and will continue to be so until the bottleneck of developing technologies for the co-ordinated manipulation of multiple genes or traits has been removed. This bottleneck is perhaps not widely appreciated, but is amply evidenced by the huge body of literature describing the manipulation or expression of single useful genes in plants and the comparative paucity of publications dealing with the manipulation of multiple genes Chapter 3: Applications of biotechnology and genomics in potato improvement

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Applications of biotechnology and genomics in potato improvement

selecting elite As a species highly amenable to cell culture, it has a long history of biotechnology applications for crop improvement. This review begins with a historical perspective on potato improvement using biotechnology encompassing pathogen elimination, wide hybridization, policy manipulation and applications of cell culture. We describe the past developments and new approaches for gene transfer to potato.

Transformation is highly effective for adding single genes to existing elite potato clones with no, or minimal, disturbances to their genetic background and represents the only effective way to produce biogenic lines of specific genotypes/cultivars. This is virtually impossible via traditional breeding as, due to the high heterozygosis in the tetraploid potato genome, the genetic integrity of potato clones is lost upon sexual reproduction as a result of allele segregation.

These genetic attributes have also provided challenges for the development of genetic maps and applications of molecular markers and genomics in potato breeding. Various molecular approaches used to characterize loci, (candidate) genes and alleles in potato, and associating phenotype with genotype are also described.

#### Chapter 4: Tapping RNA silencing pathways for plant biotechnology

#### Dr. J. Illamathi

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

The term "aberrant RNAs" was coined to describe how sense RNA could target other homologous RNAs of the same strand due to RNA overabundance. A plant-encoded RNAdependent RNA polymerase (RDR) may then employ these abnormal transcripts as primers to create complementary or antisense RNA, which would ultimately result in RNA degradation (Dougherty and Parks, 1995). However, these screens usually involve a large number of individuals, which are sometimes time and space consuming. With the development of transgenic techniques that enable the permanent introduction of novel DNA into plants, genes of interest could be targeted for silencing. The first manifestation of this approach was expression of the full-length complementary (or antisense) RNA transcripts.

The strategy at the time was simply to invert the coding sequence of the target gene in a transgenic cassette. Later, the use of complementary double-stranded RNA (dsRNA) generated by inverted-repeat (IR) transgenes greatly increased the efficiency of gene silencing and became widely adopted in plant biotechnology. Meanwhile, the molecular fundamentals of RNA-mediated gene silencing, known as RNA silencing, began to take shape. It appears that RNA silencing is not only a broadly used mechanism for gene regulation in plants, but also evolutionarily conserved in other branches of life. Both RNA interference (RNAi) in animals and quelling in fungi seem to operate under the same principles as RNA silencing.

#### Chapter 5: Biotechnology in Animal Agriculture

Dr. A. Shajahan Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Animal agriculture is being transformed by rapid advances in biotechnology—a term that encompasses a variety of technologies, including genetic engineering (GE), genetic modification, transgenics, recombinant DNA techniques, and cloning, among others. Producers are interested in the application of biotechnology to improve productivity, consistency, and quality; to introduce new food, fiber, and medical products; and to protect the environment. Potential human health applications of transgenic animals include producing biopharmaceuticals and generating organs, tissues, and cells for xenotransplantation. Criticisms of such applications involve issues ranging from food safety and social resistance to potential negative impacts on animal welfare and on ecosystems. Questions also have arisen about the adequacy of the current regulatory structure to assess and manage any risks created by these technologies. On January 15, 2009, the U.S. Food and Drug Administration (FDA) released final guidance on how it is to regulate GE animals and products. FDA is to do so under its existing statutory authority and regulations. Generally, GEderived foods, for example, are to be regulated foods; if their composition does not differ from their conventional counterparts, they will not have to be labeled.

Nonetheless, developers of GE animals and of GE-derived products must gain FDA pre-market approval. On February 6, 2009, the FDA announced the first approval of a drug from a GE animal. The drug is a human anti-clotting agent produced in the milk of transgenic goats.

# Chapter 6: Recent advances rice biotechnology—towards genetically superior transgenic rice

#### Dr. G. Venkatkumar

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Rice is the most important food crop in the world. Almost half of the world's population depend on rice as their staple food . So far, advances in breeding of rice has kept pace with the growing demand, despite losses suffered as a result of various biotic and biotic factors. Over the next 7 to 8 years, global rice plantings are predicted to remain static, which will result in a shortfall of about 130 million tonnes .

Therefore, conventional breeding methods need to be supplemented with recent achievements in rice biotechnology to meet the needs of the growing world population. Rice genetic transformation has taken rapid strides since the first transgenic rice plant was produced 15 years ago. During the last 10 years, tremendous progress has been made to develop a high-frequency, routine and reproducible genetic transformation protocol of rice either through direct DNA transfer or by *Agrobacterium* -mediated transformation technologies. Using these technologies, a number of agronomically important traits, including quality improvement and increased nutritional value, have been addressed. These improvements could not be possible through the application of conventional breeding technologies. Rice reached another milestone by becoming the first crop plant to have its genome sequenced.

Chapter 7: Plant-produced viral bovine vaccines

#### Dr. RV. Shalini

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Animal health and animal welfare are the main factors to guarantee the safety of food of animal origin. Disease outbreaks can have a devastating effect on animal health, thus affecting animal production and even human health. Vaccination is used primarily to prevent such outbreaks, and vaccines directed against highly contagious viral diseases play a particularly important role as they prevent viral infections which predispose animals to superinfection with bacterial or other opportunistic pathogens.

Nowadays, different veterinary vaccines are available against respiratory and gastrointestinal pathogens, as well as for viruses associated with infertility, malformation, stillbirth or abortion, in cattle The major goals of viral bovine veterinary vaccine production are to improve the health and welfare of cattle and increase production of livestock, in a cost-effective manner.

For many viral infections of livestock, conventional vaccines cannot be used, as their use would interfere with disease surveillance based on serological testing and may result in the loss of a country's disease-free status. This is the case of the inactivated vaccine against foot-and-mouth disease (FMD) that though quite effective in controlling clinical is not used in FMD-free countries as this would compromise their status and hence international trade. In these cases, the use of subunit vaccines combined with suitable diagnostic assays allows differentiating infected from vaccinated animals (DIVA) by differentiation of antibody responses induced by the vaccine.

#### Chapter 8: biotechnology—a sustainable alternative for chemical Industry **Dr. R. Arunkumar** *Assistant professor, Department of Biotechnology,* Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Among the major new technologies that have appeared since the 1970s, biotechnology has perhaps attracted the most attention. Biotechnology has proved capable of generating enormous wealth and influencing every significant sector of the economy. Biotechnology has already substantially affected healthcare; production and processing of food; agriculture and forestry; environmental protection; and production of materials and chemicals.

This review focuses on achievements and future prospects for biotechnology in sustainable production of goods and services, specially those that are derived at present mostly from the traditional chemical industry.

"Industrial sustainability" aims to achieve sustainable production and processing within the context of ecological and social sustainability. Sustainability and sustainable development have had different meanings in different epochs and not everyone is agreed on a common definition of these concepts. For the purpose of this review, sustainable development is understood to mean process of change in which the exploitation of resources, the direction of investments, the orientation.

## Chapter 9: Production and glycosylation of plant-made pharmaceuticals: the antibodies as a challenge

#### Dr. T. Veeramani

Assistant professor, Department of Biotechnology, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

At the present time, most recombinant antibodies approved by the Food and Drug Administration (FDA) are produced in cultured Chinese hamster ovary (CHO) cells. This process is highly expensive and of limited capacity. Indeed, the investment needed to establish and operate a culture unit for mammalian cells is important (about 1 million \$US for the production of 1 kg of antibody per year Currently, the capacity available for the production of monoclonal antibodies does not exceed 1000 kg per year. This means that almost all the current production capacity is used to provide sufficient amounts of the first 10 monoclonal antibodies clinically validated today Therefore, the pharmaceutical industry will need to invest a csderable amount of money to allow the commercialization of future monoclonal antibodies produced in culture units of CHO cells, which are currently undergoing clinical tests. Even in these conditions, and considering the delays needed for the construction of such equipment, it can be estimated that, in 2010, the production capacity would be, at best, 10 000 kg of antibodies with an estimated demand of 50 000 kg.

In this context, pharmaceutical companies are deeply interested in transgenic production systems which enable high production levels of therapeutic antibodies. Apart from cultured mammalian cells, only transgenic animals and transgenic plants are able to assemble, via disulphide bridges, the light and heavy chain constituents of an antibody In particular, plant cells can reproduce the complexity of these proteins, as illustrated in 1989 by the first production of functional antibodies in plants).

Chapter 10: Bioreactor technology: A novel industrial tool for high-tech production of bioactive molecules and biopharmaceuticals from plant roots

#### **Dr. Arjunpandian** Assistant professor, Department of Biotechnology,

Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India

Bioreactor technology: A novel industrial tool for high-tech production of bioactive molecules and biopharmaceuticals from plant roots of plants are the richest source for different bioactive molecules. Because of the vast number of side effects associated with synthetic pharmaceuticals, medical biotechnologists turned to nature to provide new promising therapeutic molecules from plant biofactories.

The large-scale availability of the disease- and pesticide-free raw material is, however, restricted*in vivo*. Many bioactive plant secondary metabolites are accumulated in roots. Engineered plants can also produce human therapeutic proteins. Vaccines and diagnostic monoclonal antibodies can be won from their roots, so that engineered plants hold immense potential for the biopharmaceutical industry. To obtain sufficient amounts of the plant bioactive molecules for application in human therapy , adventitious and hairy roots have to be cultured . High-tech pilot-scale bioreactor technology for the establishment of a long-term adventitious root culture from biopharmaceutical plants has recently been established. In this review, I briefly discuss a technology for cultivating bioactive molecule-rich adventitious and hairy roots from plants using a high-tech bioreactor system, as well as the principles and application of genome-restructuring mechanisms

# GAS DYNAMICS AND JET PROPULSION

Edited by R TAMIZH SELVAN



Gas Dynamics and Jet Propulsion Copyright © 2021-2022 by R.Tamizh Selvan

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-613-4

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### TABLE OF CONTENTS

Introduction	10
Chapter 1 Basic Concepts And Isentropic Flows	25
Chapter 2 Compressible Flow Through Ducts K Purushothaman	55
Chapter 3 Normal And Oblique Shocks M Sudhakar	90
Chapter 4 Jet Propulsion R Tamizh Selvan	111
Chapter 5 Space Propulsions N Sivaharinadhan	125
Chapter 6 Propellant engine S.V.Sridhar	152
Chapter 7 Ram jet Engine J.Rajesh	175
Chapter 8 Mach waves and Mach cone G.Brithiviraj	210
Chapter 9 Turbo-prop engines P.Vijayakumar	
Chapter 10 Rayleigh flow or Turbulent flow P.Sarath kumar	
References	

#### Chapter 1 Basic Concepts And Isentropic Flows R.Baskaran

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of Science And Technology, Tamil Nadu, India.

#### Introduction

The purpose of this applet is to simulate the operation of a converging-diverging nozzle, perhaps the most important and basic piece of engineering hardware associated with propulsion and the high speed flow of gases. This device was invented by Carl de Laval toward the end of the 19th century and is thus often referred to as the 'de Laval' nozzle. This applet is intended to help students of compressible aerodynamics visualize the flow through this type of nozzle at a range of conditions. Technical Background

The usual configuration for a converging diverging (CD) nozzle is shown in the figure. Gas flows through the nozzle from a region of high pressure (usually referred to as the chamber) to one of low pressure (referred to as the ambient or tank). The chamber is usually big enough so that any flow velocities here are negligible. The pressure here is denoted by the symbol pc. Gas flows from the chamber into the converging portion of the nozzle, past the throat, through the diverging portion and then exhausts into the ambient as a jet. The pressure of the ambient is referred to as the 'back pressure' and given the symbol pb.

A diffuser is the mechanical device that is designed to control the characteristics of a fluid at the entrance to a thermodynamic open system. Diffusers are used to slow the fluid's velocity and to enhance its mixing into the surrounding fluid. In contrast, a nozzle is often intended to increase the discharge velocity and to direct the flow in one particular direction.

Frictional effects may sometimes be important, but usually they are neglected. However, the external work transfer is always assumed to be zero. It is also assumed that changes in thermal energy are significantly greater than changes in potential energy and therefore the latter can usually be neglected for the purpose of analysis.

Mach Cone a conical surface that bounds the region in a supersonic flow of gas in which the sound waves (perturbations) emanating from a point source A of the perturbations are 2concentrated (see Figure 1). In a homogeneous supersonic flow of gas, the angle a between the generatrices of the Mach cone and its axis is called the Mach angle; it is related to the Mach number by the equation  $\sin a = 1/M$ 

In this chapter a discussion on a steady state flow through a smooth and continuous area flow rate is presented. A discussion about the flow through a converging-diverging nozzle is also part of this chapter. The isentropic flow models are important because of two main reasons: One, it provides the information about the trends and important parameters. Two, the correction factors can be introduced later to account for deviations from the ideal state.

Relationships for Small Mach Number Even with today's computers a simplified method can reduce the tedious work involved in computational work. In particular, the trends can be examined with analytical methods. It further will be used in the book to examine trends in derived models. It can be noticed that the Mach number involved in the above equations is in a square power. Hence, if an acceptable error is of about %1 then pr ovides the desired range. Further, if a higher power is used, much smaller error results. First it can be noticed that the ratio of temperature to stagnation temperature, is provided in power series. Expanding of the equations according to the binomial expansion of will result in the same fashion The pressure difference normalized by the velocity (kinetic energy) as correction factor is

#### Chapter 2 Compressible Flow through Ducts K Purushothaman

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of

Science And Technology, Tamil Nadu, India.

Concept of Gas Dynamics Gas dynamics mainly concerned with the motion of gases and its effects .It differ from fluid dynamics .Gas dynamics considers thermal or chemical effects while fluid dynamics usually does not. Gas dynamics deals with the study of compressible flow when it is in motion. It analyses the high speed flows of gases and vapors' with considering its compressibility. The term gas dynamics is very general and alternative names have been suggested e.g.: Supersonic flow, compressible flow and aero thermodynamics etc.,

1.1.1 Significance with Applications:

Gas dynamics is of interest to both mechanical and the aeronautical engineers but particular field of interest of the two different .It may be said that thermodynamicist is concerned with how an object in motion influenced as it flies through still air. In contrast to it the thermodynamicist in more interested in the cases in which the object in stationary and the fluid is in motion .The applications of gas dynamics are given below.

- ➤ It is used in Steam and Gas turbines
- ➤ High speed aero dynamics
- ➤ Jet and Rocket propulsion
- ➤ High speed turbo compressor

The fluid dynamics of compressible flow problems which involves the relation between forse, density, velocity and mass etc. Therfore the following laws are frequently used for solving the dynamic problems.

- 1. Steady flow energy equation
- 2. Entropy relations
- 3. Continity equation
- 4. Momentum equation

We know that fluids, such as gas, are classified as Incompressible and Compressible fluids. Incompressible fluids do not undergo significant changes in density as they flow. In general, liquids are incompressible; water being an excellent example. In contrast compressible fluids do undergo density changes. Gases are generally compressible; air being the most common compressible fluid we can find. Compressibility of gases leads to many interesting features such as shocks, which are absent for incompressible fluids. Gas dynamics is the discipline that studies the flow of compressible fluids and forms an important branch of Fluid Mechanics.

1.2.1 Compressible vs. Incompressible Flow

> A flow is classified as incompressible if the density remains nearly constant.

➤ Liquid flows are typically incompressible.

> Gas flows are often compressible, especially for high speeds.

> Mach number, Ma = V/c is a good indicator of whether or not compressibility effects are important.

> Ma < 0.3 : Incompressible

➤ Ma < 1 : Subsonic</li>
 ➤ Ma = 1 : Sonic
 ➤ Ma > 1 : Supersonic
 ➤ Ma >> 1 : Hypersonic

#### Chapter 3 Normal And Oblique Shocks M Sudhakar

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of Science And Technology, Tamil Nadu, India.

As an object moves through a gas, the gas molecules are deflected around the object. If the speed of the object is much less than the speed of sound of the gas, the density of the gas remains constant and the flow of gas can be described by conserving momentum, and energy. As the speed of the object approaches the speed of sound, we must consider compressibility effects on the gas. The density of the gas varies locally as the gas is compressed by the object.

For compressible flows with little or small flow turning, the flow process is reversible and the entropy is constant. The change in flow properties are then given by the isentropic relations (isentropic means "constant entropy"). But when an object moves faster than the speed of sound, and there is an abrupt decrease in the flow area, shock waves are generated in the flow. Shock waves are very small regions in the properties change by a large amount. gas where the gas Across a shock wave. the static pressure, temperature, and gas density increases almost instantaneously. The changes in the flow properties are irreversible and the entropy of the entire system increases. Because a shock wave does no work, and there is no heat addition, the total enthalpy and the total temperature are constant. But because the flow is non-isentropic, the total pressure downstream of the shock is always less than the total pressure upstream of the shock. There is a loss of total pressure associated with a shock wave as shown on the slide. Because total pressure changes across the shock, we can not use the usual (incompressible) form of Bernoulli's equation across the shock. The Mach number and speed of the flow also decrease across a shock wave.

If the shock wave is perpendicular to the flow direction, it is called a normal shock. There are equations which describe the change in the flow variables. The equations are derived from the conservation of mass, momentum, and energy. Depending on the shape of the object and the speed of the flow, the shock wave may be inclined to the flow direction. When a shock wave is inclined to the flow direction it is called an oblique shock. On this slide we have listed the equations which describe the change in flow variables for flow across an oblique shock. The equations presented here were derived by considering the conservation of mass, momentum, and energy for a compressible gas while ignoring viscous effects. The equations have been further specialized for a two-dimensional flow without heat addition. The equations only apply for those combinations of free stream Mach number and deflection angle for which an oblique shock occurs. If the deflection is too high, or the Mach too low, a normal shock occurs. For the Mach number change across an oblique shock there are two possible solutions; one supersonic and one subsonic. In nature, the supersonic ("weak shock") solution occurs most often. However, under some conditions the "strong shock", subsonic solution is possible.

Oblique shocks are generated by the nose and by the leading edge of the wing and tail of a supersonic aircraft. Oblique shocks are also generated at the trailing edges of the aircraft as the flow is brought back to free stream conditions. Oblique shocks also occur downstream of a nozzle if the expanded pressure is different from free stream conditions. In high speed inlets, oblique shocks are used to compress the air going into the engine. The air pressure is increased without using any rotating machinery.

On the slide, a supersonic flow at Mach number M approaches a shock wave which is inclined at angle s. The flow is deflected through the shock by an amount specified as the deflection angle - a. The deflection angle is determined by resolving the incoming flow velocity into components parallel and perpendicular to the shock wave. The component parallel to the shock is assumed to remain constant across the shock, the component perpendicular is assumed to decrease by the normal shock relations. Combining the components downstream of the shock determines the deflection angle

#### Chapter 4 Jet Propulsion R Tamizh Selvan

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of

Science And Technology, Tamil Nadu, India.

#### 1. Definition:

• Jet propulsion is the method of producing thrust by expelling mass at high velocity from a jet engine, allowing vehicles (like aircraft and rockets) to move forward due to the reaction force according to Newton's Third Law of Motion.

#### 2. Basic Principles:

- **Newton's Third Law:** For every action, there is an equal and opposite reaction. The engine expels gas at high speed, producing thrust in the opposite direction.
- Conservation of Momentum: The momentum change of the exhaust gas results in thrust for the vehicle.

#### **3. Types of Jet Engines:**

- Turbojet Engines:
  - Air is drawn in, compressed, mixed with fuel, ignited, and then expelled to produce thrust. Common in military aircraft.
- Turbofan Engines:
  - Similar to turbojets but include a large fan that provides additional thrust from bypassing air. Widely used in commercial airliners for better efficiency and noise reduction.
- Turboprop Engines:
  - Uses a gas turbine to drive a propeller. Efficient for lower-speed aircraft.
- Ramjet Engines:
  - Operate without moving parts. Air is compressed by the engine's forward motion, mixed with fuel, ignited, and expelled. Efficient at supersonic speeds.
- Scramjet Engines:
  - A variation of ramjets that operates at hypersonic speeds, allowing combustion to occur while the airflow remains supersonic.

#### 4. Components of a Jet Engine:

- Inlet: Draws in air and slows it down.
- Compressor: Increases air pressure and temperature.
- Combustor: Mixes compressed air with fuel and ignites it.
- **Turbine:** Extracts energy from the hot gases to drive the compressor.
- Nozzle: Expels exhaust gases, converting thermal energy into kinetic energy, producing thrust.

#### 6. Efficiency Considerations:

- Thermal Efficiency: Ratio of useful work output to the heat input.
- **Propulsive Efficiency:** Ratio of useful thrust power to the power of the fuel consumed.

#### 7. Applications of Jet Propulsion:

- Aerospace: Commercial airliners, military jets, and spacecraft.
- Missiles: For propulsion systems in military applications.
- High-Speed Trains: Some systems utilize jet propulsion principles for increased speeds.

#### 8. Environmental Considerations:

• Jet propulsion systems contribute to noise and emissions. Efforts are ongoing to develop quieter and more environmentally friendly technologies.

#### Chapter 5 Space Propulsions

#### N Sivaharinadhan

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of

Science And Technology, Tamil Nadu, India.

#### **Definition:**

• Space propulsion refers to the means by which spacecraft are propelled through space, utilizing various technologies to generate thrust and maneuver in the vacuum of space.

#### **Types of Space Propulsion Systems**

#### **1. Chemical Propulsion:**

- Solid Rocket Motors:
  - Use solid propellants that burn to produce thrust. Simple design, high thrust-to-weight ratio.
  - Example: Space Shuttle Solid Rocket Boosters.
- Liquid Rocket Engines:
  - Utilize liquid propellants (fuel and oxidizer) stored in tanks and pumped into a combustion chamber.
  - Example: SpaceX's Merlin engine.
- Hybrid Rocket Engines:
  - Combine solid and liquid or gaseous oxidizers, offering advantages of both systems.

#### **2. Electric Propulsion:**

- Ion Thrusters:
  - Use electric fields to accelerate ions, producing thrust. Highly efficient but generate low thrust.
  - Example: NASA's Dawn spacecraft.
- Hall Effect Thrusters:
  - A type of ion thruster that uses magnetic fields to trap electrons, which ionize the propellant.
  - Commonly used for station-keeping and orbital maneuvers.

#### **3. Thermal Propulsion:**

- Nuclear Thermal Rockets (NTR):
  - Use a nuclear reactor to heat a propellant, which is then expelled to produce thrust.
  - More efficient than chemical rockets for deep space missions.
- Solar Thermal Propulsion:
  - Uses solar energy to heat a propellant, typically for use in the inner solar system.

#### 4. Advanced Propulsion Concepts:

- Electrothermal Thrusters:
  - Use electrical energy to heat a propellant, improving efficiency compared to chemical rockets.
- Magnetoplasmadynamic (MPD) Thrusters:
  - Use electromagnetic fields to accelerate a plasma, producing thrust. Potential for high-speed interplanetary missions.
- Fusion Propulsion:
  - Hypothetical concept using nuclear fusion reactions to produce thrust. Still in research stages.
- Antimatter Propulsion:
  - Uses the annihilation of matter and antimatter to produce thrust. Theoretical and not yet feasible.

#### Chapter 6 Propellant engine S.V.Sridhar

#### Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of

Science And Technology, Tamil Nadu, India.

#### **Definition:**

• A propellant engine converts stored energy (from propellants) into mechanical energy to produce thrust, enabling vehicles to move. Commonly used in aerospace applications, these engines can vary widely in design and operation.

#### **Types of Propellant Engines**

#### **1. Chemical Engines:**

- Solid Rocket Motors:
  - Mechanism: Combustion of solid propellants generates gas that is expelled through a nozzle.
  - Advantages: Simplicity, high thrust-to-weight ratio, reliable.
  - Applications: Launch vehicles, missiles (e.g., Space Shuttle boosters).
- Liquid Rocket Engines:
  - **Mechanism:** Liquid fuel and oxidizer are pumped into a combustion chamber where they ignite and produce high-pressure gas.
  - Advantages: Throttle control, restart capability, high specific impulse (Isp).
  - Applications: SpaceX's Falcon 9, NASA's Space Launch System (SLS).
- Hybrid Rocket Engines:
  - Mechanism: Combines solid and liquid or gaseous oxidizers, offering flexibility in thrust control.
  - **Applications:** Some experimental and commercial launch systems (e.g., Virgin Galactic's SpaceShipTwo).

#### 2. Electric Engines:

- Ion Thrusters:
  - **Mechanism:** Electric fields accelerate ions, generating thrust. High efficiency but low thrust.
  - Applications: Deep-space missions (e.g., NASA's Dawn spacecraft).
- Hall Effect Thrusters:
  - Mechanism: Use magnetic fields to ionize and accelerate propellant.
  - Advantages: More powerful than ion thrusters, efficient for station-keeping.
  - Applications: Satellites and spacecraft for orbital maneuvers.

#### **3. Thermal Engines:**

- Nuclear Thermal Rockets (NTR):
  - Mechanism: A nuclear reactor heats a propellant (like hydrogen), which is expelled to create thrust.
  - Advantages: Higher efficiency than chemical rockets, suitable for long-duration missions.
  - Applications: Conceptual designs for crewed missions to Mars.
- Solar Thermal Rockets:
  - Mechanism: Uses concentrated solar energy to heat a propellant.
  - Applications: Potential for missions in the inner solar system.

#### 4. Advanced Propulsion Concepts:

- Magnetoplasmadynamic (MPD) Thrusters:
  - Mechanism: Use electromagnetic forces to accelerate plasma.
  - Applications: Experimental designs for future high-speed missions.
- Fusion and Antimatter Engines:
  - Mechanism: Hypothetical concepts leveraging nuclear fusion or antimatter reactions.
  - Applications: Theoretical future interstellar travel.

#### Chapter 7 Ram jet Engine J.Rajesh

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of Science And Technology, Tamil Nadu, India.

#### **Definition:**

• A ramjet is a type of air-breathing jet engine that operates by compressing incoming air using the vehicle's forward motion. It has no moving parts, relying on high speeds to achieve airflow compression.

#### **Key Principles**

#### 1. Operation:

- Air Intake: Air enters the engine through an inlet at high speed, slowing down as it moves through the engine.
- **Compression:** The air is compressed due to the ram effect (the forward motion of the aircraft), increasing its pressure and temperature.
- **Combustion:** Fuel is injected and ignited in the compressed air, resulting in high-temperature, high-pressure exhaust gases.
- **Thrust Generation:** The exhaust gases are expelled through a nozzle, producing thrust according to Newton's third law.

#### 2. No Moving Parts:

• Unlike turbojets or turbofans, ramjets do not have compressors or turbines. The engine's operation is purely based on the kinetic energy of the incoming air.

#### **Design Characteristics**

#### 1. Inlet:

• Designed to efficiently compress incoming air without shock waves. The shape is critical for maintaining optimal airflow.

#### 2. Combustion Chamber:

• Where fuel is mixed with the compressed air and ignited. It must be designed to withstand high temperatures and pressures.

#### 3. Nozzle:

• Expands the exhaust gases to generate thrust. The nozzle can be converging, converging-diverging, or variable area, depending on the flight regime.

#### Performance

#### 1. Speed Range:

• Ramjets are most efficient at supersonic speeds (typically Mach 2 to Mach 6). Below these speeds, they are less effective due to insufficient airflow compression.

#### 2. Specific Impulse (Isp):

• Ramjets have a moderate Isp, generally lower than that of turbojet engines at subsonic speeds but more efficient at high speeds.

#### Chapter 8 Mach waves and Mach cone G.Brithiviraj

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of Science And Technology, Tamil Nadu, India.

#### **Definitions:**

- Mach Waves: These are shock waves that occur when an object moves through a fluid (typically air) at supersonic speeds (greater than the speed of sound). The waves are generated by the compressibility of the fluid.
- Mach Cone: The conical region formed by the collection of Mach waves emanating from a supersonic object. The cone represents the area of influence of the shock waves produced by the object.

#### **Key Concepts**

#### **Speed of Sound:**

• The speed of sound in a medium is influenced by factors such as temperature and pressure. In dry air at 20°C, the speed of sound is approximately 343 m/s.

#### Mach Number (M):

• The ratio of the speed of an object to the speed of sound in the surrounding medium:  $M=VcM = \frac{V}{c}M=cV$  where VVV is the object's velocity and ccc is the speed of sound.

#### Formation of Mach Waves

#### **Supersonic Motion:**

• When an object travels faster than the speed of sound, it compresses the air in front of it, creating pressure waves. These waves cannot propagate ahead of the object due to its speed, resulting in a build-up of pressure that forms a shock wave.

#### Mach Cone Characteristics:

- Angle of the Mach Cone ( $\theta$ \theta $\theta$ ): The angle can be calculated using the Mach number:  $\sin \frac{1}{100}(\theta) = 1 M \sin(\theta) = \frac{1}{M} \sin(\theta) = M1$
- The angle of the cone decreases as the Mach number increases, meaning that at higher speeds, the shock wave is more narrowly focused.

#### Geometry:

• The Mach cone is visualized as a three-dimensional conical shape with the object at the vertex. The shock waves spread out in a conical pattern behind the object.

#### **Applications in Engineering:**

- Aerospace Design: Understanding Mach waves and cones is critical for designing supersonic and hypersonic vehicles, including aircraft and rockets, to manage shock waves and reduce drag.
- Acoustics: Mach waves also have implications in acoustics, influencing the design of high-speed aircraft and missile systems.

#### Chapter 9 Turbo-prop engines P.Vijayakumar

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of

Science And Technology, Tamil Nadu, India.

#### **Definition:**

• A turboprop engine is a type of aircraft propulsion system that uses a gas turbine engine to drive a propeller. It combines the principles of jet propulsion with the efficiency of propeller-driven flight.

#### **Key Components**

#### Main Components:

- Gas Turbine Engine: Generates power by burning fuel and compressing air. It consists of:
  - **Compressor:** Increases the pressure of incoming air.
  - **Combustor:** Burns the fuel-air mixture to produce hot gases.
  - **Turbine:** Extracts energy from the hot gases to drive the compressor and propeller.
- **Reduction Gearbox:** Reduces the high rotational speed of the turbine to a lower speed suitable for driving the propeller.
- **Propeller:** The rotating blades that generate thrust. It converts the engine's power into forward motion.

#### Operation

#### **How It Works:**

- Air enters the engine and is compressed by the compressor.
- The compressed air is mixed with fuel in the combustor and ignited, producing high-temperature, high-pressure gases.
- The gases expand through the turbine, which drives both the compressor and the reduction gearbox connected to the propeller.
- The propeller spins, generating thrust to propel the aircraft forward.

#### **Performance Characteristics**

#### **Efficiency:**

• Turboprop engines are generally more fuel-efficient than turbojets or turbofans at lower speeds (typically below 450 knots). This makes them ideal for regional and short-haul flights.

#### **Thrust-to-Weight Ratio:**

• Turboprop engines have a lower thrust-to-weight ratio compared to turbojets, which limits their performance at higher speeds.

#### Chapter 10 Rayleigh flow or Turbulent flow P.Sarath kumar

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute Of

Science And Technology, Tamil Nadu, India.

#### **Definition:**

• Rayleigh flow refers to a specific type of compressible flow in a duct or nozzle where the flow remains subsonic and is characterized by a constant area, with a uniform inlet condition. It typically describes the flow of an ideal gas under isentropic conditions, experiencing no shocks or friction.

#### 2. Key Characteristics:

- Subsonic Flow: The flow velocity is less than the speed of sound.
- Isentropic Process: No heat transfer and no irreversible losses; the process is adiabatic and reversible.
- Uniform Inlet Conditions: The flow is steady, with uniform temperature and pressure at the inlet.

#### **3.** Governing Equations:

• The flow can be analyzed using the continuity, momentum, and energy equations, often leading to relationships between temperature, pressure, and density.

#### 4. Applications:

• Rayleigh flow is often used in theoretical models for analyzing nozzle flows and for understanding the behavior of gases in various engineering applications, including aerodynamic studies.

#### **Turbulent Flow**

#### 1. Definition:

• Turbulent flow is a chaotic flow regime characterized by irregular fluctuations in pressure and velocity. It typically occurs at high Reynolds numbers, where inertial forces dominate over viscous forces.

#### 2. Key Characteristics:

- Chaotic Behavior: Irregular and unpredictable changes in flow properties.
- **High Reynolds Number:** Typically Re>4000Re>4000Re>4000 for flow in pipes, indicating that inertial forces are significant.
- Mixing: Enhanced mixing and momentum transfer due to the chaotic eddies and vortices present in the flow.

#### **3.** Governing Equations:

• Turbulent flow is described by the Navier-Stokes equations, often requiring statistical methods or turbulence models (like k- $\varepsilon$  or k- $\omega$  models) for practical analysis due to its complexity.

#### 4. Applications:

• Turbulent flow is common in many engineering applications, including fluid transport in pipes, atmospheric phenomena, and the design of vehicles and aircraft. It's essential for understanding heat transfer and mixing processes in industrial systems.
Edited by **DR S V SRIDHAR** 



**Power Plant Engineering** 

Copyright © 2021-2022 by Dr S V SRIDHAR

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-156-6

Publisher: PRIST Publications.

# TABLE OF CONTENTS

	Introduction	8
CHAPTER NO 1	Introduction to the Sources of Energy	11
Dr.S.DHANUSKODI		
CHAPTER NO 2	Boilers	35
V.YALINI		
CHAPTER NO 3	Coal Based Thermal Power Plants	74
Dr S P KALAISELVAN		
CHAPTER NO 4	Diesel, Gas Turbine Power Plants	102
N.SIVAHARINATHAN		
CHAPTER NO 5	Combined Cycle Power Plants	134
S V SRIDHAR		
CHAPTER NO 6	Nuclear Power Plants	162
P SARATH KUMAR		
CHAPTER NO 7	Hydro Electric Power Plants	184
M.SUDHAHAR		
CHAPTER NO 8	Power from Non Conventional sources	204
R.BASKARAN		
CHAPTER NO 9	Power Plant Economics	234
R.TAMIZHSELVAN		
CHAPTER NO 10	Environmental Considerations Of Power Plants	252
P.VIJAYAKUMAR		

## **CHAPTER 1**

## Introduction to the Sources of Energy

## Dr.S.DHANUSKODI

Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Sources of Energy

There are several sources of energy, including renewable and non-renewable sources:

Renewable sources

These energy sources are natural, self-replenishing, and have low or zero carbon footprints. Some examples include:

Solar energy: The most abundant energy resource, which can be harnessed even in cloudy weather

Wind energy: A renewable energy source

Geothermal energy: A renewable energy source that comes from heat inside the earth

Hydropower: A renewable energy source that comes from flowing water

Biomass: A renewable energy source that comes from burning organic matter as fuel

Non-renewable sources

These energy sources include fossil fuels like oil, coal, and natural gas. The reserves of fossil fuels are limited and are expected to last only another 100 years. Burning fossil fuels to create electricity has been a major contributor to greenhouse gas emissions.

## **CHAPTER 2**

## Boilers

## V.YALINI

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Sources of Energy

A boiler is an enclosed vessel that provides a means for combustion heat to be transferred into water until it becomes heated water or steam. The hot water or steam under pressure is then usable for transferring the heat to a process. Water is a useful and cheap medium for transferring heat to a process. When water is boiled into steam its volume increases about 1,600 times, producing a force that is almost as explosive as gunpowder. This causes the boiler to be extremely dangerous equipment that must be treated with utmost care.

The process of heating a liquid until it reaches its gaseous state is called evaporation. Heat is transferred from one body to another by means of (1) radiation, which is the transfer of heat from a hot body to a cold body without a conveying medium, (2) convection, the transfer of heat by a conveying medium, such as air or water and (3) conduction, transfer of heat by actual physical contact, molecule to molecule.

## **CHAPTER 4**

## **Diesel, Gas Turbine Power Plants**

## N.SIVAHARINATHAN

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Diesel, Gas Turbine Power Plants

## **INTRODUCTION:**

- A generating station in which diesel engine is used as the prime mover for the generation of electrical energy is known as diesel power station
- Diesel power plants produce power in the range of 2 to 50 MW.
- They are used as standby sets for continuity of supply such as hospitals, telephone exchanges, radio stations, cinema theatres and industries.
- They are suitable for mobile power generation and widely used in railways, submarine & ships.

## **Applications of diesel power plant**

•They are used as central station for small or medium power supplies.

•They can be used as stand-by plants to hydro-electric power plants and steam power plants for emergency services.

•They can be used as peak load plants in combinations with thermal or hydroplants.

•They are quite suitable for mobile power generation and are widely used in transportation systems such as automobiles, railways, air planes and ships

## **CHAPTER 5**

## **Combined Cycle Power Plants**

## S V SRIDHAR

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Combined Cycle Power Plants

Fossil fuel-fired power plants use either steam or combustion turbines to provide the mechanical power to electrical generators. Pressurized high temperature steam or gas expands through various stages of a turbine, transferring energy to the rotating turbine blades. The turbine is mechanically coupled to a generator, which produces electricity.

The Combined Cycle power plant is a combination of a fuel-fired turbine with a Heat Recovery Steam Generator (HRSG) and a steam powered turbine. These plants are very large, typically rated in the hundreds of mega-watts. They combine the Rankine Cycle (steam turbine) and Brayton Cycle (gas turbine) thermodynamic cycles by using heat recovery boilers to capture the energy in the gas turbine exhaust gases for steam production to supply a steam turbine.

Natural gas is a major fuel source for electric generation through the use of gas turbines and steam turbines. Most grid peaking power plants and some off-grid engine-generators use natural gas. Particularly high efficiencies can be achieved through combining gas turbines with a steam turbine in combined cycle mode. Natural gas burns more cleanly than other fossil fuels, such as oil and coal, and produces less carbon dioxide per unit energy released. For an equivalent amount of heat, burning natural gas produces about 30% less carbon dioxide than burning petroleum and about 45% less than burning coal. Combined cycle power generation using natural gas is thus the cleanest source of power available using fossil fuels, and this technology is widely used wherever gas can be obtained at a reasonable cost.

## **CHAPTER 6**

## **Nuclear Power Plants**

## P SARATH KUMAR

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Nuclear Power Plants

Nuclear power is a clean and efficient way of boiling water to make steam, which turns turbines to produce electricity. Nuclear power plants use low enriched uranium fuel to produce electricity through a process called fission—the splitting of uranium atoms in a nuclear reactor. Nuclear fission products are the atomic fragments left after a large atomic nucleus undergoes nuclear fission. Typically, a large nucleus like that of uranium fissions by splitting into two smaller nuclei, along with a few neutrons, the release of heat energy (kinetic energy of the nuclei), and gamma rays.

Fusion only produces more energy than it consumes in small nuclei (in stars, Hydrogen & its isotopes fusing into Helium). The energy released when 4 Hydrogen nuclei (= protons) fuse (there are some decays involved as well) into a Helium nucleus is around 27 Million Electron Volts (MeV), or about 7 MeV per nucleon.

## **CHAPTER 7**

## **Hydro Electric Power Plants**

**M.SUDHAHAR** 

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Hydro Electric Power Plants

Introduction to Hydro electricity

Hydropower is considered as one of the most economic and non polluting sources of energy. Power generated from water is known as hydroelectricity.

Hydroelectricity means electricity generated by hydropower or from the use of the gravitational force of falling of flowing water.

Hydroelectric power plants are used to convert the hydraulic potential energy from water into electrical energy. Hydro-electric power stations are generally located in hilly areas where dams can be built and large water reservoirs can be obtained.

## **CHAPTER 8**

Power from Non Conventional sources

**R.BASKARAN** 

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Power from Non Conventional sources

Renewable energy is a plentiful, clean, and inexhaustible energy source that can be used to generate electricity, heat water or space, power transportation, and provide energy services in rural areas

The main types of renewable energy are: Solar Wind Hydro Geothermal

Renewable energy can help reduce global warming, improve public health, create jobs, and provide stable energy prices.

Renewable energy comes from unlimited, naturally replenished resources, such as the sun, tides, and wind.

Renewable energy can be used for electricity generation, space and water heating and cooling, and transportation. Non-renewable energy, in contrast, comes from finite sources, such as coal, natural gas, and oil.

## **CHAPTER 9**

Power Plant Economics R.TAMIZHSELVAN

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Power Plant Economics

Power tariff types, Load distribution parameters, load curve, Comparison of site selection criteria, relative merits & demerits, Capital &Operating Cost of different power plants. Pollution control technologies including Waste Disposal Options for Coal and Nuclear Power Plants.

In all fields of industry economics plays an important role. In power plant engineering economics of power system use certain well established techniques for choosing the most suitable system. The power plant design must be made on the basis of most economical condition and not on the most efficient condition as the profit is the main basis in the design of the plant is to bring the cost of energy produced to minimum.

Among many factors, the efficiency of the plant is one of the factors that determine the energy cost.

#### **TERMS AND DEFINITIONS**

1. Connected Load. The connected load on any system, or part of a system, is the combined continuous rating of all the receiving apparatus on consumer's premises, which is connected to the system, or part of the system, under consideration.

## **CHAPTER 10**

## **Environmental Considerations of Power Plants**

### P.VIJAYAKUMAR

Assistant Professor, Department of Mechanical Engineering, Ponnaiyah Ramajayam Institute of Science and Technology, Tamil Nadu, India.

This chapter aims to familiarize students with the Environmental Considerations Of Power Plants

Concerns over the environment have been recognized at recent United Nations conventions on environmental sustainability and biodiversity. These conventions emphasize that, as humans, plants and animals all depend on a healthy environment, protection of the environment is of paramount importance. All industries, including nuclear power plants, have the potential to impact the environment. This section reviews HWR activities directed towards eliminating or reducing that impact.

A nuclear power plant can impact the environment during all phases of its lifecycle, from construction, commissioning and operation, through to decommissioning. The focus in this section is on operation. Although all operating nuclear power plants release radioactivity, chemicals, metal corrosion products and waste heat into the environment, the emphasis here is on radioactivity because it is usually of primary concern. However, non-radiological releases are discussed briefly as well. In considering radioactivity, both occupational doses received within the nuclear power plant and public doses received outside the exclusion boundary are of interest. This is because these doses are closely related to environmental performance and protection.

# PEDAGOGY OF PHYSICAL SCIENCE PART - III

978-93-6255-846-6

Edited by DR.R.GUNASEKARAN Pedagogy of Physical Science:Part-III Copyright ©2021-2022byDr.R. Gunasekaran

All rights reserved. No part of this book may be reproduced ortransmittedinanyformorbyanymeanswithoutwrittenpermissionfrom the author.

ISBN:978-93-6255-846-6

Publisher:PonnaiyahRamajayamInstituteofScienceandTechnology(PRIST)

## TABLE OF CONTENTS

Introduction
CHAPTER I Nature and scope of Physical science
DR. R. GUNASEKARAN
CHAPTER 2 Base of Physical Science in Education
DR. R. GUNASEKARAN
CHAPTER 3 Exploring Learners and Methods of Teaching51
DR. D. T. ARIVALAN
CHAPTER 4 School Curriculum in Physical Science
Prof. T. SELVARAJ
CHAPTER 5 Approaches and Strategies of Learning Physical Science85
DR. D. T. ARIVALAN
CHAPTER 6 Evaluation 109
DR. R. GUNASEKARAN
CHAPTER 7 The Physical Science Teacher
CHAPTER 8 Planning for Instruction 137
Prof. T. SELVARAJ
CHAPTER 9 Models of Teaching
Prof. R. VAISHNAVI
CHAPTER 10 Equipment and Resources for Teaching Physical Science165 DR. R. GUNASEKARAN
REFERENCES170

## CHAPTER I Nature and scope of Physical science DR. R. GUNASEKARAN Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

The nature and scope of physical science encompass the fundamental principles that explain the behavior of matter and energy in the universe. Here's an overview:

## **Nature of Physical Science**

- 1. **Empirical Approach**: Physical science relies on observation, experimentation, and evidence. Scientists formulate hypotheses, conduct experiments, and analyze data to draw conclusions.
- 2. **Interdisciplinary Field**: It encompasses various branches, including physics, chemistry, astronomy, geology, and environmental science. These fields often overlap and inform one another.
- 3. **Quantitative Analysis**: Physical science emphasizes quantitative measurements and mathematical modeling to describe phenomena. Equations and mathematical concepts are crucial for understanding relationships and predicting outcomes.
- 4. **Theoretical and Practical**: It involves both theoretical frameworks (like laws and theories) and practical applications (such as technology and engineering). This dual nature allows for the exploration of abstract concepts and their real-world implications.
- 5. **Predictive Power**: Physical science provides tools for predicting future events based on established laws and theories. For example, the laws of motion can predict the trajectory of an object.

#### **Scope of Physical Science**

- 1. **Physics**: The study of matter, energy, and the fundamental forces of nature. Topics include mechanics, thermodynamics, electromagnetism, optics, and quantum physics.
- 2. **Chemistry**: Focuses on the composition, structure, properties, and changes of matter. Subfields include organic, inorganic, physical, analytical, and biochemistry.
- 3. **Astronomy**: The study of celestial bodies, space, and the universe as a whole. Topics include the formation of stars, galaxies, and planetary systems, as well as cosmology.
- 4. **Geology**: The study of the Earth, its materials, processes, and history. Topics include rock formation, plate tectonics, mineralogy, and the study of natural resources.
- 5. **Environmental Science**: An interdisciplinary field that examines the interactions between the physical environment and human activities. Topics include ecology, climate change, and resource management.
- 6. **Material Science**: Focuses on the properties and applications of materials, exploring their structure, behavior, and potential uses in various industries.
- 7. **Engineering**: Applies principles of physical science to design, build, and maintain structures, machines, and systems. It encompasses various branches like civil, mechanical, electrical, and chemical engineering.

## CHAPTER 2 Base of Physical Science in Education DR. R. GUNASEKARAN Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

The base of physical science in education is essential for developing students' understanding of the natural world and fostering critical thinking skills. Here are key components of this educational foundation:

## **Curriculum Design**

- 1. **Integrated Curriculum**: Physical science should be integrated with other subjects, such as mathematics, technology, and environmental studies, to provide a holistic understanding of concepts.
- 2. **Hands-On Experiments**: Incorporating laboratory activities and experiments helps students engage with the material and understand theoretical concepts through practical application.
- 3. **Conceptual Frameworks**: Emphasizing key concepts and principles (e.g., the laws of motion, conservation of energy) allows students to build a solid foundation for further study.

## **Educational Standards**

- 1. **National and State Standards**: Adhering to guidelines set by organizations like the Next Generation Science Standards (NGSS) helps ensure that curricula are relevant and comprehensive.
- 2. Assessment Frameworks: Developing assessments that evaluate both conceptual understanding and practical skills, ensuring students can apply what they've learned.

## **Teaching Methods**

- 1. **Inquiry-Based Learning**: Encouraging students to ask questions and engage in scientific inquiry fosters curiosity and critical thinking.
- 2. **Collaborative Learning**: Group work and discussions promote teamwork, communication skills, and the sharing of diverse perspectives.
- 3. **Problem-Based Learning**: Presenting real-world problems that require scientific solutions helps students apply their knowledge and develop problem-solving skills.

#### **Technology Integration**

- 1. **Digital Resources**: Utilizing simulations, online labs, and interactive software to enhance learning and visualize complex concepts (e.g., PhET simulations).
- 2. **Data Analysis Tools**: Introducing students to software for analyzing data (e.g., Excel, Google Sheets) enables them to work with real data sets and draw conclusions.

## CHAPTER 3 Exploring Learners and Methods of Teaching DR. D. T. ARIVALAN Professor, School of Education, PRIST Deemed to Be University, Thanjavur.

That sounds interesting! Are you looking into specific teaching methods, types of learners, or how they interact? There are various approaches, like differentiated instruction, constructivist learning, and more.

□ Lecture: A traditional method where the teacher presents information, often to large groups. It's efficient for delivering content but may not engage all learners.

□ Discussion: Encourages students to engage with the material and each other, fostering critical thinking and deeper understanding.

□ Cooperative **Learning**: Students work in groups to solve problems or complete tasks, promoting collaboration and social skills.

□ Project-**Based Learning**: Students engage in projects that require them to apply knowledge and skills to real-world challenges, fostering creativity and problem-solving.

 $\Box$  Flipped **Classroom**: Students learn new content at home (often via videos) and use class time for discussions, hands-on activities, or problem-solving.

□ Inquiry-**Based Learning**: Encourages students to ask questions and explore topics through investigation, fostering curiosity and deeper understanding.

 $\Box$  Direct **Instruction**: A structured approach where the teacher provides explicit teaching of concepts and skills, often with guided practice.

 $\Box$  Experiential **Learning**: Involves hands-on experiences, allowing students to learn through doing and reflecting on those experiences.

□ Socratic **Method**: A form of dialogue-based teaching where the teacher asks questions to stimulate critical thinking and illuminate ideas.

## CHAPTER 4 School Curriculum in Physical Science Prof. T. SELVARAJ Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

The school curriculum in physical science often covers foundational concepts in physics and chemistry. Key topics typically include:

- 1. **Matter and Its Properties**: Understanding the states of matter, atomic structure, and the periodic table.
- 2. **Forces and Motion**: Newton's laws, types of forces, and the concepts of speed, velocity, and acceleration.
- 3. **Energy**: Different forms of energy (kinetic, potential, thermal), the law of conservation of energy, and energy transfer.
- 4. Waves and Sound: Wave properties, the electromagnetic spectrum, and sound waves.
- 5. **Electricity and Magnetism**: Basic electrical concepts, circuits, and the relationship between electricity and magnetism.
- 6. **Chemical Reactions**: Types of chemical reactions, balancing equations, and stoichiometry.
- 7. Thermodynamics: Heat transfer, temperature, and the laws of thermodynamics.

## CHAPTER 5 Approaches and Strategies of Learning Physical Science DR. D. T. ARIVALAN Professor, School of Education, PRIST Deemed to Be University, Thanjavur.

Here are some effective approaches and strategies for teaching physical science:

- 1. **Inquiry-Based Learning**: Encourage students to ask questions, formulate hypotheses, and conduct experiments. This fosters critical thinking and a deeper understanding of scientific concepts.
- 2. **Hands-On Experiments**: Incorporate lab activities where students can conduct experiments to observe physical phenomena. This makes learning experiential and memorable.
- 3. **Project-Based Learning**: Assign projects that require students to apply concepts to realworld problems. This can include building models, conducting research, or presenting findings.
- 4. **Cooperative Learning**: Use group activities where students collaborate on tasks. This promotes teamwork and allows for peer-to-peer learning.
- 5. **Flipped Classroom**: Have students study theoretical content at home (e.g., through videos or readings) and use class time for discussions and practical applications.
- 6. **Use of Technology**: Integrate simulations, educational software, and virtual labs to illustrate complex concepts and engage students.
- 7. **Differentiated Instruction**: Tailor teaching methods to accommodate different learning styles and abilities. Provide varied resources and activities to meet diverse needs.
- 8. **Real-World Connections**: Relate physical science concepts to everyday life and current events. This relevance can increase student interest and motivation.
- 9. **Socratic Method**: Engage students in dialogue through guided questioning, helping them explore concepts and develop critical thinking skills.
- 10. **Formative Assessment**: Regularly assess understanding through quizzes, reflections, or discussions to guide instruction and provide feedback.

#### Evaluation

## DR. R. GUNASEKARAN Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

**Evaluation** in the context of education refers to the systematic process of assessing the effectiveness of teaching, learning, and educational strategies. It involves gathering and analyzing data to determine how well students are learning, how effectively instructional methods are working, and how educational objectives are being met. In the pedagogy of biological science, evaluation helps educators refine their teaching practices, assess student progress, and improve the curriculum.

#### **Types of Evaluation in Education**

#### 1. Formative Evaluation

- **Definition**: Ongoing evaluations conducted during the teaching-learning process.
- **Purpose**: To monitor student learning and provide feedback that can be used to improve instruction and learning in real-time.

## • **Examples**:

- Classroom quizzes
- Group discussions
- Short feedback forms
- Student reflections
- **Benefits**: Helps teachers adjust their teaching methods, address student misconceptions early, and guide students toward achieving learning goals.

#### 2. Summative Evaluation

- **Definition**: Evaluation conducted at the end of a unit, course, or program to assess the overall effectiveness and achievement.
- **Purpose**: To evaluate student learning at the conclusion of an instructional period and to determine if the learning objectives were met.
- **Examples**:
  - Final exams
  - End-of-term projects
  - Standardized tests
  - Cumulative reports
- **Benefits**: Provides a comprehensive measure of student performance and curriculum effectiveness. Summative evaluations often inform grades and certifications.

#### 3. Diagnostic Evaluation

• **Definition**: Pre-assessment conducted before instruction begins to understand students' prior knowledge and identify learning needs.

## The Physical Science Teacher

## DR. R. GUNASEKARAN Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

A physical science teacher plays a crucial role in guiding students to understand and appreciate the principles of physics and chemistry, and their applications in the natural world. Their responsibilities go beyond simply delivering content; they are facilitators of scientific thinking, inquiry, and experimentation. Here are key qualities and roles of an effective physical science teacher:

## **1. Subject Expertise:**

- **In-depth Knowledge**: A physical science teacher must have a strong understanding of key concepts in both physics and chemistry, including forces, energy, matter, and their interactions.
- **Keeping Updated**: Staying informed about advancements in science and technology is essential to provide students with current, relevant information.

## 2. Instructional Skills:

- **Clear Communication**: The teacher needs to simplify complex ideas and present them in ways that are easy for students to grasp.
- **Diverse Teaching Methods**: Using a variety of teaching techniques—such as demonstrations, experiments, discussions, and digital tools—helps cater to different learning styles.
- **Differentiation**: Adapting lessons for diverse student needs ensures all students can engage meaningfully with the content.

## **3. Encouraging Inquiry and Critical Thinking:**

- **Inquiry-Based Learning**: A physical science teacher fosters curiosity by encouraging students to ask questions, explore hypotheses, and conduct experiments.
- **Problem-Solving Approach**: Emphasizing critical thinking and the scientific method helps students develop problem-solving skills, important in science and life.

## 4. Practical Application:

- Lab Work and Experiments: Engaging students in hands-on activities in the lab helps them apply theoretical knowledge, understand scientific processes, and develop lab skills.
- **Real-World Connections**: A teacher must connect classroom learning with real-world phenomena, making the subject matter relevant and relatable.
- **1. Technology Integration:**

## Planning for Instruction Prof. T. SELVARAJ Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

**Planning for instruction** is a vital component of effective teaching, as it involves organizing and designing lessons to meet learning objectives, address student needs, and ensure that the content is delivered effectively. In the context of biological science, instructional planning focuses on how to best teach scientific concepts, theories, and skills, ensuring that students can not only understand the material but also apply it to real-world scenarios.

#### **Key Steps in Planning for Instruction**

## 1. Define Learning Objectives

- **Purpose**: Clear learning objectives outline what students should know, understand, and be able to do by the end of the lesson or unit.
- Examples:
  - "Students will be able to explain the process of photosynthesis."
  - "Students will analyze the role of natural selection in evolution."
- SMART Criteria: Learning objectives should be Specific, Measurable, Achievable, Relevant, and Time-bound to ensure clarity and focus.

## 2. Identify and Align Curriculum Standards

- **Purpose**: Instruction must align with national, state, or local curriculum standards to ensure consistency and comprehensiveness.
- **Examples**: For biology, this may include standards related to genetics, ecosystems, evolution, and cell biology.
- **Benefits**: Ensures that instruction covers essential content and skills, helping students meet academic expectations.

## 3. Understand Student Needs and Context

- **Purpose**: Tailoring instruction to meet the diverse needs of learners.
- **Considerations**:
  - Learning styles: Some students may be visual learners, while others prefer hands-on activities or verbal instruction.
  - **Prior knowledge**: Assessing what students already know helps guide the depth of instruction.
  - **Diverse learning needs**: Differentiating instruction for students with special needs, language barriers, or advanced abilities is essential.
- **Examples**: Using models and diagrams for visual learners, or incorporating group activities for kinesthetic learners.

## 4. Select and Organize Content

- **Purpose**: Identify the key concepts, facts, and skills to be taught and decide the sequence in which they will be presented.
- **Examples**:
  - Teaching cellular biology before diving into genetics, as students need to understand cell structures and functions before learning about DNA.

## CHAPTER 9 Models of Teaching Prof. R. VAISHNAVI Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

**Models of teaching** refer to structured, systematic approaches to instruction that help guide the teaching process to achieve specific learning outcomes. These models are based on theories of learning and development and provide educators with frameworks for organizing and delivering content, fostering student engagement, and promoting the effective acquisition of knowledge and skills. Each model emphasizes different aspects of learning and teaching, and the choice of model depends on the subject matter, learning objectives, and the needs of the students.

Here are some of the most widely recognized models of teaching:

## **1. The Direct Instruction Model**

- **Definition**: A teacher-centered approach that emphasizes clear, structured, and explicit teaching of content or skills.
- Key Features:
  - Focus on mastery of content through systematic instruction.
  - The teacher leads the lesson by providing information, modeling, and giving guided practice.
  - Frequent feedback and correction are given to ensure understanding.
  - Lesson phases: Introduction, Presentation, Guided Practice, Independent Practice, and Closure.
- Advantages:
  - Highly effective for teaching foundational skills, especially in areas like math, reading, or basic biology concepts.
  - Structured and efficient, which makes it ideal for covering large amounts of content.
- Disadvantages:
  - Less emphasis on critical thinking and creativity.
  - Limits student autonomy and may not engage all learners.

**Example in Biology**: Teaching the structure of the cell by explaining and modeling the parts, providing guided practice through labeling exercises, and then assigning independent activities like cell diagram creation.

## 2. The Inquiry-Based Learning Model

- **Definition**: A student-centered approach that encourages students to ask questions, investigate problems, and discover answers through exploration.
- Key Features:

## Equipment and Resources for Teaching Physical Science DR. R. GUNASEKARAN Assistant professor, School of Education, PRIST Deemed to Be University, Thanjavur.

Here are some essential equipment and resources for teaching physical science:

## **Equipment:**

## 1. Lab Equipment:

- Beakers, flasks, and test tubes
- Bunsen burners or hot plates
- Microscopes
- Measuring instruments (scales, thermometers, rulers)

## 2. Safety Gear:

- Safety goggles
- Lab coats
- Gloves

## 3. Technology:

- Computers or tablets with internet access
- Smartboards or projectors for presentations
- Simulation software for virtual experiments

## 4. Models and Kits:

- Molecular model kits for chemistry
- Physics demonstration kits (e.g., for waves, forces)
- Solar system models

## 5. Measurement Tools:

- Multimeters for electrical experiments
- Pressure gauges and other sensors

## **Resources:**

- 1. **Textbooks**: Standard physical science textbooks that cover key concepts and provide practice problems.
- 2. Online Platforms:
  - Educational websites (like Khan Academy or PhET Interactive Simulations)
  - YouTube channels for science demonstrations and explanations
- 3. Science Journals and Articles: Access to current research articles and publications for up-to-date information.
- 4. **Curriculum Guides**: State or national curriculum standards to ensure alignment with educational goals.
- 5. Visual Aids:



Peace Education Copyright ©2021-2022byDr.R. Gunasekaran

All rights reserved. No part of this book may be reproduced ortransmittedinanyformorbyanymeanswithoutwrittenpermissionfrom the author.

ISBN:978-93-6255-886-2

Publisher:PonnaiyahRamajayamInstituteofScienceandTechnology(PRIST)

## TABLE OF CONTENTS

Introduction
CHAPTER I Values Education15 Prof. R. VAISHNAVI
CHAPTER 2 Fostering Values24 DR. R. GUNASEKARAN
CHAPTER 3 Peace Education in The Present Scenario
CHAPTER 4 Promoting Culture of Peace67 Prof. T. SUBHASHINI
CHAPTER 5 Approaches and Strategies
CHAPTER 6 Conflict prevention and resolution 105 Prof. T. SUBHASHINI
CHAPTER 7 Curricular and Co-curricular activities125 DR. R. GUNASEKARAN
CHAPTER 8 International Peace and Security
CHAPTER 9 UNESCO Culture of Peace Programmes151 Prof. R. VAISHNAVI
CHAPTER 10 The Pedagogy of Values163 DR. R. GUNASEKARAN
REFERENCES171

## **CHAPTER I** Values Education

## Prof. R. VAISHNAVI

## Assistant Professor, School of Education, PRIST University, Thanjavur

**Values education** is the process of teaching and learning about the values that guide human behavior. It aims to help individuals understand, adopt, and live by a set of moral and ethical principles that contribute to personal development, social responsibility, and a harmonious society. Values education can occur formally in schools or informally through family, community, and cultural interactions.

## **Key Aspects of Values Education:**

## 1. Moral Values:

• These include fundamental principles like honesty, integrity, respect, empathy, responsibility, and fairness. Moral values guide individuals in distinguishing right from wrong and acting in a manner that promotes social good.

## 2. Cultural and Social Values:

• Cultural values are those that are passed down through generations within a society, including traditions, customs, and norms. Social values emphasize cooperation, respect for others, and the importance of community well-being.

## 3. Civic Values:

• Civic values focus on the responsibilities individuals have as members of a community or nation. These include respect for the law, participation in democratic processes, and contributing to the common good.

#### 4. Personal Values:

• These are the values that guide an individual's personal life and decisions, such as perseverance, discipline, courage, and ambition. They are essential for personal development and achieving life goals.

## 5. Universal Values:

• Universal values transcend specific cultures or societies and are often regarded as essential for the well-being of humanity as a whole. These may include human rights, equality, justice, peace, and environmental sustainability.

## **Goals of Values Education:**

## 1. Character Building:

• One of the primary aims of values education is to develop individuals with strong moral character who can make ethical decisions and act responsibly in society.

## 2. Promoting Ethical Behavior:

- Values education encourages ethical behavior in personal, social, and professional contexts. It helps individuals understand the consequences of their actions on others and the environment.
- 3. Fostering Respect and Tolerance:

## Fostering Values

## DR. R. GUNASEKARAN

## Assistant Professor, School of Education, PRIST University, Thanjavur

**Fostering values** refers to the intentional nurturing and development of ethical, moral, and social principles in individuals. This process can happen in various settings—family, schools, workplaces, and communities—and is essential for building character, guiding behavior, and contributing to a just and harmonious society. The goal is to help individuals internalize values that guide them in making ethical decisions, forming positive relationships, and contributing to the common good.

Here are key approaches and strategies for fostering values:

## 1. Role Modeling

- **Explanation**: One of the most effective ways to foster values is through role modeling. Parents, teachers, leaders, and community members can demonstrate the values they wish to instill by living them out in everyday life.
- **Application**: Adults should consistently exhibit behaviors like honesty, fairness, respect, and empathy. Children and young people learn by observing how role models handle situations, interact with others, and make decisions.

#### 2. Creating a Values-Rich Environment

- **Explanation**: Environments that promote and reinforce positive values make it easier for individuals to adopt and practice them.
- **Application**: Schools, workplaces, and communities can cultivate a culture of respect, kindness, cooperation, and integrity by establishing norms and expectations around these values. For instance, schools can have anti-bullying policies, promote collaboration, and celebrate acts of kindness and inclusivity.

#### **3. Discussion and Reflection**

- **Explanation**: Open discussions and reflections on values help individuals internalize them. Engaging in dialogue about real-life ethical dilemmas allows people to explore the meaning and importance of different values.
- **Application**: Teachers, mentors, or parents can encourage children and young adults to reflect on their choices and the consequences of their actions. This could involve discussing books, historical events, or current issues that highlight ethical challenges, allowing individuals to analyze the role of values in decision-making.

## 4. Incorporating Values into Curriculum

• **Explanation**: Formal education systems play a critical role in fostering values by integrating them into the curriculum.

## CHAPTER 3 Peace Education in The Present Scenario Prof. T. SELVARAJ

## Assistant Professor, School of Education, PRIST University, Thanjavur

**Peace education** is an essential approach to teaching the values, knowledge, attitudes, and skills necessary to promote peace, prevent conflict, and address issues of violence at all levels—personal, community, national, and global. In today's world, where societies face diverse challenges such as political instability, social inequality, and environmental crises, peace education plays a critical role in fostering a culture of non-violence, respect, and cooperation.

## **Importance of Peace Education in the Present Scenario**

## 1. Global Conflicts and Violence:

• Armed conflicts, wars, terrorism, and civil unrest continue to cause suffering worldwide. Peace education helps individuals understand the root causes of conflict and violence, such as economic inequality, ethnic tensions, and power struggles, and teaches non-violent methods for resolving disputes.

## 2. Rising Social and Political Divisions:

• Many societies today are polarized due to political ideologies, cultural differences, and religious beliefs. Peace education promotes dialogue, tolerance, and respect for diversity, which are crucial in bridging these divides and fostering peaceful coexistence.

## 3. Cultural and Religious Intolerance:

• Intolerance toward different cultures, religions, and ethnicities often fuels discrimination and violence. Peace education teaches respect for different identities and encourages cross-cultural understanding and inclusivity.

## 4. Cyberbullying and Digital Violence:

• The rise of technology and social media has led to new forms of violence, such as cyberbullying and online harassment. Peace education now incorporates digital citizenship, teaching individuals how to navigate the online world ethically and prevent virtual violence.

#### 5. Environmental Crises:

• Environmental degradation and climate change are sources of tension and potential conflict, particularly in regions facing resource scarcity. Peace education emphasizes environmental stewardship, helping individuals recognize the connection between peace, sustainability, and the protection of natural resources.

#### 6. Refugee Crises and Migration:

• War, persecution, and climate change have led to mass migrations and refugee crises around the world. Peace education fosters empathy and promotes understanding of the challenges faced by displaced populations, encouraging acceptance and support for refugees.

## 7. Human Rights Violations:

• In many parts of the world, human rights continue to be violated, often leading to oppression, violence, and unrest. Peace education includes teaching about human

## CHAPTER 4 Promoting Culture of Peace

## Prof. T. SUBHASHINI

## Assistant Professor, School of Education, PRIST University, Thanjavur

**Promoting a culture of peace** involves fostering values, attitudes, and behaviors that reject violence, resolve conflicts through dialogue and negotiation, and respect human rights and justice. A culture of peace emphasizes non-violence, tolerance, and inclusivity in personal, social, and international interactions. In today's world, promoting such a culture is crucial for creating more harmonious societies and addressing both local and global challenges.

## Key Components of a Culture of Peace

## 1. Non-Violence:

• Central to a culture of peace is the rejection of violence, whether physical, verbal, or psychological. Non-violence seeks to resolve conflicts and disagreements without harm, using tools like negotiation, dialogue, and reconciliation.

## 2. Human Rights and Social Justice:

• A culture of peace is grounded in the respect for human rights and social justice. This means ensuring that all individuals, regardless of race, gender, religion, or socioeconomic status, are treated with dignity and fairness.

## 3. Tolerance and Respect for Diversity:

• Promoting tolerance and understanding between people of different cultures, religions, and backgrounds is essential to a peaceful society. A culture of peace embraces diversity, seeing it as a source of strength rather than division.

## 4. Dialogue and Communication:

 Open, respectful, and constructive dialogue is a key method for resolving conflicts and fostering understanding between individuals and groups. Communication is essential for breaking down barriers and finding common ground.

## 5. Environmental Sustainability:

• A culture of peace recognizes the interconnectedness of humans and the environment, promoting sustainable practices to protect the planet and prevent resource-based conflicts.

## 6. Democratic Participation:

• Encouraging active participation in democratic processes is crucial for a culture of peace. Individuals must have the right and opportunity to contribute to decisions that affect their lives and communities.

## 7. Education for Peace:

• Education is a powerful tool for promoting a culture of peace. By teaching values such as empathy, cooperation, and conflict resolution, schools and communities can cultivate peaceful attitudes in individuals from a young age.

## **Strategies for Promoting a Culture of Peace**

## CHAPTER 5 Approaches and Strategies DR. T.S. PARVATHY

## Assistant Professor, School of Education, PRIST University, Thanjavur

## Approaches and Strategies to Promote a Culture of Peace

Creating a culture of peace requires multifaceted approaches that engage individuals, communities, institutions, and governments. Below are key approaches and strategies that can be employed to promote and sustain peace at various levels of society:

#### **1. Education for Peace**

- **Approach**: Incorporate peace education into formal and informal learning environments to instill the values of non-violence, cooperation, and social justice.
- Strategy:
  - **Integrating Peace in Curricula**: Schools and universities can introduce courses on conflict resolution, human rights, global citizenship, and environmental sustainability.
  - **Workshops and Training**: Organize community workshops, webinars, and training sessions for both children and adults on empathy, mediation, and non-violent communication.
  - **Peace Clubs and Peer Programs**: Establish peace clubs in schools and communities where youth can practice leadership in promoting peaceful interactions, resolving conflicts, and volunteering for social causes.

#### 2. Promoting Dialogue and Mediation

- **Approach**: Foster open, respectful, and inclusive dialogue to resolve conflicts and promote understanding between diverse groups.
- Strategy:
  - **Community Dialogue Forums**: Facilitate dialogue forums where people from different cultural, religious, or political backgrounds can share experiences, discuss issues, and collaboratively seek solutions to conflicts.
  - **Mediation Centers**: Establish mediation centers within communities, workplaces, and schools where trained mediators can help parties resolve disputes through negotiation and understanding.
  - **Interfaith and Intercultural Dialogues**: Encourage religious and cultural organizations to engage in interfaith and intercultural dialogue to break down stereotypes and promote mutual respect.

## 3. Advocating for Human Rights and Social Justice

 $\Box$  Approach: Support policies and initiatives that advance human rights and social justice to create equitable and inclusive societies.

## Conflict prevention and resolution Prof. T. SUBHASHINI

## Assistant Professor, School of Education, PRIST University, Thanjavur

**Conflict prevention and resolution** are essential aspects of maintaining peace and stability within communities, organizations, and nations. Conflict arises from differences in values, needs, interests, and power dynamics, but effective prevention and resolution strategies can help manage or de-escalate disputes before they turn violent.

**Conflict Prevention** 

Conflict prevention refers to measures that are put in place to avoid the outbreak of conflict or to stop tensions from escalating into full-blown violence. These efforts are proactive, addressing the underlying causes of conflict before they turn into crises.

#### 1. Addressing Root Causes

- Economic Inequality: Unequal access to resources such as land, wealth, or employment opportunities can breed resentment. Governments and organizations can implement policies to reduce income inequality, provide social safety nets, and create equitable economic opportunities.
- Social and Political Inclusion: Ensuring that marginalized groups have a voice in decision-making processes can reduce feelings of exclusion and oppression, which are common triggers of conflict.
- Human Rights Protections: Protecting the human rights of all individuals, regardless of their race, gender, religion, or political affiliation, helps prevent grievances that could lead to conflict. Human rights monitoring and legal reforms can promote social justice and reduce tensions.

#### 2. Early Warning Systems

- **Conflict Monitoring**: Establishing early warning systems that monitor social, political, and economic indicators can help predict potential conflicts. For example, rising unemployment, political instability, or increasing inter-group tensions can signal that conflict is brewing.
- **Diplomatic Intervention**: In cases where conflicts seem imminent, diplomatic interventions, such as mediation by neutral parties or international organizations, can defuse tensions before they escalate. Organizations like the United Nations (UN) often play this role on the international stage.

## Curricular and Co-curricular activities DR. R. GUNASEKARAN Assistant Professor, School of Education, PRIST University, Thanjavur

## **Curricular and Co-Curricular Activities**

## **Curricular Activities**

Curricular activities refer to the planned and structured educational experiences that are part of a school's official curriculum. These activities are designed to achieve specific educational goals and enhance students' learning.

## 1. Core Subjects

• Academic Learning: Subjects such as mathematics, science, language arts, and social studies form the backbone of the educational curriculum. They provide students with essential knowledge and skills for academic success.

## 2. Skill Development Courses

- Life Skills Education: Classes that teach critical life skills, such as communication, problem-solving, and decision-making, are crucial for preparing students for real-world challenges.
- **Technical and Vocational Education**: Programs that offer training in specific trades or technical skills equip students with practical abilities for future employment.

## **3. Peace Education**

- **Conflict Resolution**: Integrating peace education into the curriculum helps students learn about conflict resolution, empathy, and non-violence. This prepares them to handle disagreements constructively.
- **Civic Education**: Teaching students about their rights and responsibilities as citizens fosters a sense of civic duty and encourages participation in community and social issues.

## **Co-Curricular Activities**

Co-curricular activities are non-academic activities that complement the curricular program and enhance students' personal and social development. These activities can occur within or outside school hours.

## International Peace and Security Prof. T. SELVARAJ

## Assistant Professor, School of Education, PRIST University, Thanjavur

#### **International Peace and Security**

International peace and security refer to the measures and efforts taken by countries and international organizations to maintain peace, prevent conflicts, and address security challenges that transcend national borders. The goal is to create a stable and peaceful global environment where nations can coexist and cooperate.

## **Key Concepts**

#### 1. Peacekeeping

- United Nations Peacekeeping Operations: The UN deploys peacekeeping missions to conflict zones to help maintain ceasefires, protect civilians, and facilitate political processes. Peacekeepers are often from multiple countries and operate under a mandate approved by the UN Security Council.
- **Multinational Forces**: Sometimes, countries form coalitions to address specific conflicts or security threats, coordinating efforts to stabilize a region.

#### 2. Conflict Prevention

- **Diplomatic Efforts**: International diplomacy plays a critical role in preventing conflicts. Through negotiation, mediation, and dialogue, nations can address disputes before they escalate.
- **Early Warning Systems**: Monitoring political, social, and economic indicators helps identify potential conflicts early, enabling proactive measures to prevent escalation.

#### 3. Disarmament and Non-Proliferation

- Arms Control Agreements: International treaties, such as the Treaty on the Non-Proliferation of Nuclear Weapons (NPT) and various disarmament agreements, aim to limit the spread of weapons of mass destruction and reduce existing arsenals.
- **Conventional Weapons Regulation**: Efforts to regulate the trade and use of conventional weapons help reduce armed violence and promote stability.

## 4. Human Rights and Humanitarian Law

- **Protection of Civilians**: Ensuring respect for human rights and international humanitarian law is essential for maintaining peace. Violations can lead to conflict and instability.
- Accountability Mechanisms: International courts and tribunals hold individuals accountable for war crimes and crimes against humanity, deterring future violations.
## CHAPTER 9 UNESCO Culture of Peace Programmes Prof. R. VAISHNAVI

## Assistant Professor, School of Education, PRIST University, Thanjavur

## **UNESCO Culture of Peace Programmes**

UNESCO (United Nations Educational, Scientific and Cultural Organization) plays a significant role in promoting a "Culture of Peace" through various programs and initiatives aimed at fostering peace, dialogue, and mutual understanding. Here are some key components of UNESCO's Culture of Peace programmes:

## **1. Education for Peace**

- **Peace Education**: UNESCO advocates for integrating peace education into school curricula worldwide. This includes teaching conflict resolution, empathy, tolerance, and respect for diversity.
- **Teacher Training**: Programs are designed to equip educators with the skills and knowledge to effectively teach peace and conflict resolution.

#### 2. Intercultural Dialogue

- **Cultural Heritage**: UNESCO promotes the preservation and appreciation of cultural heritage as a means to foster understanding and respect among different cultures. This includes initiatives like the UNESCO World Heritage Sites.
- World Philosophy Day: Celebrated annually, this event encourages reflection on philosophical questions and promotes dialogue among diverse cultural perspectives.

#### **3.** Global Citizenship Education (GCE)

- **Empowerment through Education**: GCE aims to equip learners with the knowledge, skills, and values necessary to engage in peaceful, sustainable societies. This includes promoting human rights, gender equality, and environmental sustainability.
- Youth Engagement: Programs target young people, encouraging their active participation in promoting peace and understanding in their communities.

#### 4. UNESCO Associated Schools Network (ASPnet)

- **Network of Schools**: ASPnet consists of schools committed to promoting a culture of peace and sustainable development through their educational practices.
- **Collaborative Projects**: Member schools engage in projects that foster understanding, respect, and cooperation among diverse cultures.

## **5. Promotion of Human Rights**

## The Pedagogy of Values DR. R. GUNASEKARAN Assistant Professor, School of Education, PRIST University, Thanjavur

#### The Pedagogy of Values

The pedagogy of values refers to educational approaches and strategies that aim to instill values in students, promoting their moral and ethical development alongside their intellectual growth. This pedagogy recognizes that education is not just about acquiring knowledge but also about shaping character and fostering responsible citizenship.

#### Key Components of the Pedagogy of Values

#### 1. Integration of Values in Curriculum

- **Cross-Disciplinary Approaches**: Values can be integrated across subjects, linking ethical discussions to various disciplines, such as literature, history, science, and social studies.
- **Thematic Units**: Developing thematic units that focus on specific values, such as respect, empathy, or justice, encourages students to explore these concepts in depth.

## 2. Experiential Learning

- **Real-World Applications**: Engaging students in community service projects, social justice initiatives, or environmental stewardship helps them experience the impact of values in action.
- **Role-Playing and Simulations**: Activities that simulate real-life scenarios encourage students to consider moral dilemmas and practice decision-making based on values.

#### 3. Critical Reflection

- **Discussions and Debates**: Encouraging open dialogue about ethical issues fosters critical thinking and allows students to articulate their values while considering diverse perspectives.
- **Journaling and Self-Assessment**: Reflective practices, such as journaling, help students explore their beliefs and values, promoting personal growth and self-awareness.

## 4. Collaborative Learning

- **Group Projects**: Collaborative activities encourage teamwork and foster a sense of community. Working together on projects can help students understand the importance of shared values and mutual respect.
- **Peer Learning**: Students can learn from each other's experiences and perspectives, enhancing their understanding of diverse values.

## 5. Teacher as Role Model

• **Modeling Values**: Educators play a crucial role in embodying the values they teach. Demonstrating integrity, empathy, and respect in interactions with students sets a powerful example.

# PEDAGOGY OF HISTORY PART -I

EDITED BY



Pedagogy of History: Part – I Copyright ©2021-2022by Prof. T. Subhashini

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-688-2

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## TABLE OF CONTENTS

Introduction
CHAPTER - I - The modern concept of history
CHAPTER- 2 -Aims and objectives of teaching History
CHAPTER- 3- Teaching Skills of History49 DR. D. ARIVUMANI
CHAPTER- 4 - Approaches of teaching History
CHAPTER- 5 -Methods of teaching and instructional media for History85 DR. R. GUNASEKARAN
CHAPTER- 6 -Planning for Instruction 107 Prof. T. SUBHASHINI
CHAPTER- 7 -Evaluation125 DR. R. GUNASEKARAN
CHAPTER - 8 - Micro Teaching
CHAPTER - 9 - Models of Teaching
CHAPTER- 10 - The History Teacher
REFERENCES171

#### **CHAPTER-I**

#### THE MODERN CONCEPT OF HISTORY

\_\_\_\_\_

#### **T.Subhashini**

Assistant Professor, School of Education, Ponnaiyah Ramajayam Institute of Science and Technology,, Thanjavur, Tamilnadu,India

\_\_\_\_\_

#### The modern concept of history

Modern history has gone beyond the traditional status of an antiquarian and leisure time pursuit to a very useful and indispensible part of a man's education. It is more scientific and more comprehensive. It has expanded in all directions both vertically andhorizontally. It has become broad-based and attractive. According to modern concept, history does not contain only the history of kings and queens, battles and generals, but the history of the common man-his house and clothing, his fields and their cultivation, his continued efforts to protect his home and hearth, and to obtain a just government, his aspirations, achievements, disappointments, defeats and failures. It is not only the individual but the communities and the societies are the subject of study of history. Study of history deepens our understanding of the potentialities and limitations of the present. It has thus become a future-oriented study related to contemporary problems. For all these reasons, history has assumed the role of a human science. History-a Science or an Art Opinions are very much divided on the question whether history is a science or an art. History is a science in the sense that it pursues its own techniques to establish and interpret facts. Like other natural sciences such as the Physics and Chemistry uses various methods of enquiry such as observation, classification, experiment and formulation of hypothesis and analysis of evidence before interpreting and reconstructing the past. History also follows the scientific method of enquiry to find out the truth. Though historian uses scientific techniques, experiment is impossible since history deals with

## CHAPTER- 2 Aims and objectives of teaching History Prof. R. VAISHNAVI

## Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

.....

Teaching history has several key aims and objectives that help students understand the past and its relevance to the present and future. Here are some important aims and objectives:

## Aims:

- 1. **Understanding Human Experience**: To provide insights into how societies have evolved over time, helping students grasp the complexity of human experiences.
- 2. **Critical Thinking**: To develop analytical skills that allow students to evaluate sources, understand differing perspectives, and construct reasoned arguments.
- 3. **Civic Awareness**: To foster informed and engaged citizens who understand their rights and responsibilities within a historical context.
- 4. **Cultural Appreciation**: To promote an understanding of diverse cultures and histories, encouraging respect and empathy for different backgrounds.
- 5. **Connection to the Present**: To help students see the links between historical events and contemporary issues, fostering a deeper understanding of current events.

## **Objectives:**

- 1. **Knowledge Acquisition**: To teach students key historical facts, timelines, and events that shape our world.
- 2. **Source Evaluation**: To train students in critical evaluation of primary and secondary sources, enhancing their research skills.
- 3. **Interpretation Skills**: To encourage students to interpret historical events and understand their significance through various viewpoints.
- 4. **Historical Empathy**: To cultivate the ability to understand and appreciate the experiences of people from different times and cultures.
- 5. **Communication Skills**: To develop students' abilities to articulate historical arguments clearly through writing and discussion.
- 6. **Problem-Solving**: To engage students in historical case studies, promoting skills in identifying problems and proposing solutions based on historical evidence.

By focusing on these aims and objectives, history education can foster a well-rounded understanding of the past and its relevance to the present and future.

## CHAPTER- 3 Teaching Skills of History DR. D. ARIVUMANI

#### Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

.....

Teaching history has several key aims and objectives that help students understand the past and its relevance to the present and future. Here are some important aims and objectives:

## Aims:

- 1. **Understanding Human Experience**: To provide insights into how societies have evolved over time, helping students grasp the complexity of human experiences.
- 2. **Critical Thinking**: To develop analytical skills that allow students to evaluate sources, understand differing perspectives, and construct reasoned arguments.
- 3. **Civic Awareness**: To foster informed and engaged citizens who understand their rights and responsibilities within a historical context.
- 4. **Cultural Appreciation**: To promote an understanding of diverse cultures and histories, encouraging respect and empathy for different backgrounds.
- 5. **Connection to the Present**: To help students see the links between historical events and contemporary issues, fostering a deeper understanding of current events.

## **Objectives:**

- 1. **Knowledge Acquisition**: To teach students key historical facts, timelines, and events that shape our world.
- 2. **Source Evaluation**: To train students in critical evaluation of primary and secondary sources, enhancing their research skills.
- 3. **Interpretation Skills**: To encourage students to interpret historical events and understand their significance through various viewpoints.
- 4. **Historical Empathy**: To cultivate the ability to understand and appreciate the experiences of people from different times and cultures.
- 5. **Communication Skills**: To develop students' abilities to articulate historical arguments clearly through writing and discussion.
- 6. **Problem-Solving**: To engage students in historical case studies, promoting skills in identifying problems and proposing solutions based on historical evidence.

By focusing on these aims and objectives, history education can foster a well-rounded understanding of the past and its relevance to the present and future.

Teaching history effectively requires a range of skills that enable educators to engage students and facilitate meaningful learning experiences. Here are some essential teaching skills for history educators

## CHAPTER- 4 Approaches of teaching History Prof. T. SUBHASHINI

## Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

.....

Teaching history can be approached in various ways to engage students and enhance their understanding. Here are some effective approaches:

## **1.** Chronological Approach

- Focus: Presents history in a linear timeline, emphasizing the sequence of events.
- **Benefits**: Helps students understand cause-and-effect relationships and the progression of historical developments.

## 2. Thematic Approach

- Focus: Organizes history around central themes (e.g., war, migration, social change).
- **Benefits**: Encourages students to explore connections between different periods and regions.

## **3. Inquiry-Based Learning**

- Focus: Centers on students asking questions, conducting research, and exploring historical issues.
- **Benefits**: Fosters critical thinking and promotes a deeper understanding through investigation.

## 4. Comparative Approach

- Focus: Compares different societies, cultures, or events to identify similarities and differences.
- Benefits: Enhances students' ability to analyze and evaluate diverse perspectives.

## 5. Narrative Approach

- Focus: Uses storytelling to present historical events and figures.
- **Benefits**: Makes history more relatable and engaging, helping students connect emotionally to the material.

## 6. Experiential Learning

- Focus: Involves hands-on activities, such as reenactments, role-playing, or field trips.
- **Benefits**: Provides immersive experiences that make history tangible and memorable.

## CHAPTER- 5 Methods of teaching and instructional media for History DR. R. GUNASEKARAN

#### Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

.....

Teaching history effectively involves various methods and instructional media that enhance learning and engagement. Here's an overview of some key methods and media:

## **Teaching Methods**

#### 1. Lecture Method

- **Description**: Teacher-centered approach where the instructor delivers information.
- Use: Effective for introducing new topics or providing context for complex historical events.

## 2. Discussion Method

- **Description**: Facilitates open dialogue among students about historical topics.
- Use: Encourages critical thinking and allows students to explore different perspectives.

#### 3. Collaborative Learning

- **Description**: Students work in groups to complete tasks or projects.
- Use: Promotes teamwork and helps students learn from each other.

#### 4. Case Study Method

- **Description**: Involves in-depth analysis of specific historical events or figures.
- Use: Develops analytical skills and allows students to explore real-world applications.

## 5. Inquiry-Based Learning

- **Description**: Students pose questions and conduct research to find answers.
- Use: Fosters curiosity and engagement with historical content.

## 6. Role-Playing and Simulations

- **Description**: Students assume roles from historical events and act them out.
- Use: Provides immersive experiences that deepen understanding of historical contexts.

## 7. Problem-Based Learning

- **Description**: Students tackle historical problems or dilemmas.
- Use: Enhances critical thinking and connects history to real-world issues.

## **Instructional Media**

- 1. Textbooks
  - **Description**: Provide structured information and narratives about historical events.
  - Use: Serve as foundational resources for content.

## Planning for Instruction Prof. T. SUBHASHINI

## Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

## ••••••

Planning for instruction in history involves a systematic approach to ensure that lessons are engaging, informative, and aligned with learning objectives. Here's a step-by-step guide for effective instructional planning:

## **1. Define Learning Objectives**

- **Specificity**: Clearly outline what students should know and be able to do by the end of the lesson or unit.
- **Bloom's Taxonomy**: Use different levels of cognitive skills (remembering, understanding, applying, analyzing, evaluating, creating) to formulate objectives.

## 2. Know Your Students

- Assess Prior Knowledge: Understand what students already know about the topic to build on their existing knowledge.
- **Consider Learning Styles**: Recognize diverse learning preferences (visual, auditory, kinesthetic) and plan accordingly.

## **3. Select Content and Resources**

- **Curriculum Standards**: Align content with relevant educational standards and frameworks.
- **Diverse Materials**: Choose a mix of textbooks, primary sources, multimedia, and digital resources to provide varied perspectives.

## 4. Choose Instructional Methods

- **Variety**: Incorporate different teaching methods (lectures, discussions, group work, project-based learning) to cater to different learning styles.
- **Engagement Strategies**: Plan activities that promote active learning and participation, such as debates or simulations.

## **5. Design Assessments**

- Formative Assessments: Plan informal assessments (quizzes, group discussions, reflections) throughout the lesson to gauge understanding.
- **Summative Assessments**: Create final assessments (essays, projects, presentations) that measure overall learning outcomes.

#### Evaluation

## DR. R. GUNASEKARAN

## Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

.....

Evaluation in history education is crucial for assessing student understanding and the effectiveness of instruction. Here's a comprehensive overview of evaluation methods, purposes, and best practices:

## **1.** Types of Evaluation

Formative Evaluation

- **Purpose**: To monitor student learning during the instructional process.
- Methods:
  - Quizzes: Short assessments to gauge understanding of recent content.
  - Class Discussions: Informal checks on student engagement and comprehension.
  - **Exit Tickets**: Quick reflections at the end of a lesson to capture key takeaways.

#### Summative Evaluation

- **Purpose**: To assess student learning at the end of a unit or course.
- Methods:
  - **Essays and Research Papers**: Evaluate students' ability to analyze and synthesize historical information.
  - **Projects**: Group or individual assignments that demonstrate comprehensive understanding.
  - **Exams**: Formal assessments that cover a wide range of content.

## **2.** Assessment Techniques

Performance-Based Assessments

- **Description**: Tasks that require students to apply their knowledge and skills in real-world contexts.
- **Examples**: Simulations, debates, or role-playing historical figures.

#### Portfolio Assessment

- **Description**: A collection of student work over time that showcases learning progress and achievements.
- Use: Provides a holistic view of student development in historical understanding.

#### **CHAPTER - 8**

#### Micro Teaching

#### T. SELVARAJ

#### Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Micro-teaching is a teacher training technique designed to improve teaching methods and skills in a controlled environment. It typically involves the following steps:

- 1. **Planning**: The teacher plans a short, focused lesson (5-10 minutes) on a specific topic or skill.
- 2. **Teaching**: The lesson is taught to a small group of peers or students. This provides a safe space for experimentation and practice.
- 3. **Feedback**: After the lesson, the teacher receives detailed feedback from peers or mentors on what went well and what could be improved.
- 4. **Re-teaching**: The teacher then re-teaches the lesson, incorporating the feedback received, which helps refine their teaching techniques.
- 5. **Reflection**: The final step involves reflecting on the experience to identify lessons learned and further areas for development.

Micro-teaching is used widely in teacher education programs as it allows instructors to practice in a low-stakes environment, focusing on specific competencies like questioning techniques, classroom management, or use of technology in teaching.

## CHAPTER - 9 Models of Teaching Prof. R. VAISHNAVI

## Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Models of teaching provide structured frameworks that educators can use to facilitate learning and enhance student engagement. Here are some well-known teaching models, along with their key features and applications:

## **1. Direct Instruction Model**

- **Description**: A teacher-centered approach where the educator provides clear and structured lessons.
- Key Features:
  - Explicit teaching of concepts and skills.
  - Use of lectures, demonstrations, and guided practice.
- Applications: Best for introducing new content, teaching basic skills, or covering foundational knowledge.

## 2. Constructivist Model

- **Description**: A student-centered approach that encourages learners to construct their own understanding and knowledge.
- Key Features:
  - Emphasis on exploration, problem-solving, and critical thinking.
  - Collaborative activities and hands-on learning experiences.
- **Applications**: Suitable for complex subjects where students can benefit from inquirybased learning and real-world connections.

## **3.** Cooperative Learning Model

- **Description**: Focuses on students working together in small groups to achieve shared learning goals.
- Key Features:
  - Structured group roles and responsibilities.
  - Emphasis on interdependence and individual accountability.
- **Applications**: Effective for promoting social skills, teamwork, and engagement in learning activities.

## 4. Inquiry-Based Learning Model

- **Description**: Encourages students to ask questions and investigate topics through research and exploration.
- Key Features:

## The History Teacher Prof. T. SUBHASHINI

## Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

.....

The role of a history teacher is multifaceted and crucial in shaping students' understanding of the past and its relevance to the present and future. Here are some key aspects of what it means to be a history teacher:

## **1. Facilitator of Knowledge**

- **Content Mastery**: A history teacher should have a strong grasp of historical events, contexts, and themes.
- **Resource Utilization**: Effectively uses a variety of resources, including primary sources, textbooks, and digital media, to provide a well-rounded perspective on history.

## 2. Promoter of Critical Thinking

- Analytical Skills: Encourages students to analyze and interpret historical events and sources critically.
- **Questioning Techniques**: Utilizes open-ended questions to foster discussion and debate, helping students explore different viewpoints.

## **3.** Cultural Awareness

- **Diverse Perspectives**: Acknowledges and incorporates multiple viewpoints, particularly those of marginalized groups, to provide a comprehensive understanding of history.
- Global Context: Helps students understand the interconnectedness of global events and cultures.

## 4. Engaging Communicator

- **Storytelling**: Uses narrative techniques to make historical events relatable and memorable.
- **Interactive Methods**: Employs various teaching methods (discussions, projects, simulations) to engage students actively.

## 5. Mentor and Guide

• **Supportive Environment**: Creates a safe and inclusive classroom where students feel comfortable expressing their thoughts and questions.

## PEDAGOGY OF ECONOMICS: PART - I

Edited by **R.VAISHNAVI** 



Pedagogy of Economics: Part – I Copyright ©2021-2022<sup>by</sup>Prof. T. Selvaraj

All rights reserved. No part of this book may be reproduced ortransmittedinanyformorbyanymeanswithoutwrittenpermissionfrom the author.

ISBN: 978-93-6255-096-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## TABLE OF CONTENTS

Introduction
CHAPTER I Sensitizing the School Curriculum17 DR. R. GUNASEKARAN
CHAPTER 2 Aims and Objectives of Teaching Economics
CHAPTER 3 Teaching Skills of Economics49 Prof. T. SUBHASHINI
CHAPTER 4 Approaches of Teaching Economics68 Prof. R. VAISHNAVI
CHAPTER 5 Methods of Teaching and Instructional Media for Economics85 Prof. T. SELVARAJ
CHAPTER 6 Curriculum
CHAPTER 7 Evaluation
CHAPTER 8 Technology
CHAPTER 9 Models of Teaching151 Prof. R. VAISHNAVI
CHAPTER 10 The Economics Teacher
REFERENCES171

## CHAPTER I Sensitizing the School Curriculum DR. R. GUNASEKARAN

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

**Sensitizing the school curriculum** involves making the curriculum more inclusive, reflective of diverse perspectives, and responsive to the evolving needs of society. This process ensures that students develop a better understanding of social issues, equity, and cultural differences, while fostering a more inclusive, empathetic, and just learning environment. Below are key aspects and strategies for sensitizing the school curriculum:

#### **1. Inclusion of Diverse Perspectives**

- **Cultural Diversity**: Ensure the curriculum reflects diverse cultural backgrounds, experiences, and contributions from different regions, ethnicities, and communities. Texts, case studies, and examples should represent a variety of cultural and social contexts.
- **Gender Sensitivity**: Incorporate materials that challenge gender stereotypes and promote gender equality. Encourage discussions on gender roles and include contributions of women and other gender identities in different fields.
- **Socioeconomic Awareness**: Address topics related to socioeconomic diversity, helping students understand the impact of poverty, wealth, and class on access to opportunities.
- **Global Perspectives**: Include global issues like climate change, migration, and international human rights to help students develop a global mindset.

## 2. Incorporating Social Justice Themes

- Equity and Fairness: Infuse lessons that address social inequalities and encourage students to think critically about issues such as racism, sexism, discrimination, and economic disparity.
- **Human Rights Education**: Teach students about fundamental human rights, encouraging respect for others' rights and the importance of social responsibility.
- **Civic Responsibility**: Promote lessons that focus on civic duties, democracy, and active participation in creating a just society.

## **3. Empathy and Emotional Intelligence**

- **Emotional Literacy**: Develop activities that encourage students to understand and express their emotions, empathize with others, and manage relationships healthily.
- **Role-Playing and Simulations**: Use these methods to help students experience different perspectives, such as walking in the shoes of marginalized groups or individuals facing injustice.

## 4. Addressing Disability and Special Needs

## CHAPTER 2 Aims and Objectives of Teaching Economics Prof. R. VAISHNAVI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

The **aims and objectives of teaching economics** are designed to provide students with a deep understanding of how economies function, the role of different economic agents, and the impact of economic decisions on society and the global environment. The teaching of economics develops analytical and critical thinking skills, prepares students to engage with real-world economic issues, and promotes informed citizenship.

## Aims of Teaching Economics

- 1. Understanding Economic Concepts and Theories:
  - To help students understand fundamental economic concepts such as demand and supply, production, consumption, trade, inflation, unemployment, and economic growth.
  - To develop a comprehensive understanding of microeconomics (individual and business decisions) and macroeconomics (national and global economic factors).

## 2. Developing Analytical and Critical Thinking Skills:

- To enable students to analyze economic data, trends, and policies and assess their impact on individuals, businesses, and governments.
- To foster critical thinking by evaluating economic arguments, identifying biases, and making reasoned judgments about economic issues.

## 3. **Promoting Economic Literacy**:

- To equip students with the knowledge to understand everyday economic activities, such as budgeting, saving, investing, and making financial decisions.
- To help students comprehend economic news and policies, encouraging them to engage in informed discussions on economic matters.

## 4. Preparing Students for Civic Engagement:

- To instill a sense of responsibility and awareness about how economic decisions affect social welfare, equity, and the environment.
- To prepare students to contribute to economic debates, policy discussions, and decision-making in their communities and on a broader scale.

## 5. Understanding the Global Economy:

- To provide students with insights into global economic systems, international trade, and the interdependence of economies across the world.
- To encourage an understanding of globalization and its effects on different regions, industries, and labor markets.

## 6. Developing Problem-Solving and Decision-Making Skills:

• To help students apply economic theories to real-world problems, such as poverty, unemployment, and resource allocation, and propose viable solutions.

• To train students to make informed and ethical decisions in economic contexts, whether in personal finance, business, or public policy.

#### CHAPTER 3 Teaching Skills of Economics Prof. T. SUBHASHINI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Teaching economics effectively requires a combination of specialized skills to engage students and enhance their understanding of both theoretical and practical aspects of the subject. Below are some essential teaching skills for economics:

#### **1. Subject Matter Expertise**

- **In-depth Knowledge**: Teachers must have a strong grasp of economic concepts, theories, and models, ranging from microeconomics (supply and demand, market structures) to macroeconomics (inflation, fiscal policies, GDP).
- **Current Trends**: Staying updated on the latest economic developments, policies, and global events is crucial for making lessons relevant and relatable.

#### 2. Clarity and Simplification

- **Breaking Down Complex Concepts**: Economics often involves abstract and complex ideas, so the ability to explain these clearly and in simple terms is key. Use analogies, examples, and illustrations to make concepts more digestible.
- **Step-by-Step Explanation**: Gradually introducing and building upon concepts ensures students fully grasp the material before moving on to more complex topics.

## 3. Use of Real-World Examples

- **Connecting Theory to Practice**: Relating economic concepts to real-life events or trends (e.g., discussing inflation with current price rises) helps students understand the relevance of what they are learning.
- **Case Studies and News**: Bringing in case studies, news articles, or data from the current economy can make lessons more engaging and applicable.

## 4. Active Learning Strategies

- **Discussion and Debate**: Encouraging students to discuss and debate economic issues fosters critical thinking and allows them to apply their knowledge to real-world problems.
- **Problem-Based Learning**: Presenting students with economic problems (e.g., how to reduce unemployment in a country) encourages them to collaborate and develop solutions using economic principles.
- **Simulations and Role-Playing**: Using simulations, such as running a mock stock market or budget allocation exercise, allows students to experience economic decision-making in action.

## 5. Data Interpretation and Quantitative Skills

## CHAPTER 4 Approaches of Teaching Economics

## Prof. R. VAISHNAVI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Teaching economics can be approached through various methods to engage students and cater to diverse learning styles. Here are several effective approaches to teaching economics:

## 1. Lecture-Based Approach

- **Description**: A traditional method where the teacher presents economic concepts through structured lectures.
- Advantages: Efficient for covering theoretical content and foundational principles. It allows for the systematic presentation of ideas.
- **Techniques**: Use of visual aids (e.g., slides, graphs) to explain concepts like supply and demand. Teachers can supplement lectures with examples, current economic events, or historical data.

## 2. Case Study Method

- **Description**: Involves analyzing real-life or hypothetical economic scenarios to illustrate the application of economic theory.
- Advantages: Helps students bridge the gap between theory and practice, and develops their critical thinking and problem-solving skills.
- **Techniques**: Presenting case studies related to business, government policies, or international trade for group analysis and discussion. Students can explore the consequences of economic decisions and suggest possible outcomes.

## 3. Problem-Based Learning (PBL)

- **Description**: Students learn by solving complex, real-world economic problems.
- Advantages: Encourages collaborative learning, critical thinking, and creativity. It helps students develop practical problem-solving skills.
- **Techniques**: Presenting an economic challenge, such as how to manage inflation or reduce unemployment, for students to research and solve in groups.

## 4. Experiential Learning

- **Description**: Focuses on learning through direct experience, often through simulations, role-playing, or real-world projects.
- Advantages: Provides hands-on learning opportunities that enhance engagement and understanding.
- **Techniques**: Simulations like mock stock markets, budget planning exercises, or creating small businesses can be used to teach economic principles such as investment, market.

## CHAPTER 5 Methods of Teaching and Instructional Media for Economics Prof. T. SELVARAJ

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Teaching methods and instructional media play a critical role in enhancing students' understanding of economics. Below are several **methods of teaching** and the types of **instructional media** that can be effectively used for teaching economics:

#### **Methods of Teaching Economics**

#### 1. Lecture Method

- **Description**: The teacher delivers a structured presentation of economic concepts.
- **Advantages**: Ideal for introducing new topics or explaining complex theories. Allows for comprehensive coverage of material.
- **Application**: Use lectures to explain concepts like supply and demand, inflation, or fiscal policies.

## 2. Discussion Method

- **Description**: Students engage in group discussions to analyze economic theories or current economic issues.
- Advantages: Promotes critical thinking, deepens understanding through debate, and encourages participation.
- **Application**: Organize discussions on economic topics such as the effects of globalization, income inequality, or trade policies.

## 3. Case Study Method

- **Description**: Students analyze real-world or hypothetical case studies to apply economic theory to practice.
- Advantages: Bridges the gap between theory and practice, helps in understanding the real-world implications of economic decisions.
- **Application**: Use case studies related to company finances, government fiscal policy, or international trade agreements for analysis.

## 4. Problem-Based Learning (PBL)

- **Description**: Students solve real-life economic problems through research and collaboration.
- Advantages: Encourages problem-solving, collaboration, and application of economic concepts.
- **Application**: Present problems like reducing national debt or managing inflation for students to solve using economic principles.

## 5. Role-Playing and Simulations

- **Description**: Students take on roles (e.g., government officials, business owners) in simulated economic environments.
- Advantages: Helps students understand economic decision-making and the effects of policy decisions.
- **Application**: Simulate a stock market or government budget allocation exercise, allowing students to experience real-world economic interactions.

## Curriculum Prof. T. SUBHASHINI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

## Curriculum

The curriculum is a structured framework that outlines the educational content, learning experiences, and assessment methods designed to achieve specific educational goals. It serves as a guide for educators in planning and delivering instruction. Here's an overview of the key components, types, and considerations related to curriculum.

## **Key Components of Curriculum**

- 1. Content
  - **Subjects and Topics**: Defines what knowledge and skills students are expected to learn, including specific subjects (e.g., math, science, literature) and key concepts within those subjects.
  - **Learning Objectives**: Specifies the desired outcomes of learning, often framed as knowledge, skills, and attitudes students should acquire.

## 2. Learning Experiences

- **Teaching Methods**: Outlines the approaches and strategies used to deliver content, such as lectures, discussions, group work, and hands-on activities.
- **Instructional Activities**: Includes specific tasks, projects, and exercises that engage students in the learning process.

#### 3. Assessment

- **Evaluation Methods**: Describes how student learning will be measured, including formative assessments (e.g., quizzes, projects) and summative assessments (e.g., exams, portfolios).
- **Feedback Mechanisms**: Outlines how feedback will be provided to students to support their learning and development.

## 4. Standards

• **Educational Standards**: Often aligned with national or state education standards that outline what students should know and be able to do at each grade level.

## 5. Resources

• **Materials and Tools**: Identifies textbooks, digital resources, technology, and other materials that will support instruction and learning.

## **Types of Curriculum**

## 1. Formal Curriculum

• The planned and structured curriculum delivered through established educational institutions, often mandated by educational authorities.

## 2. Informal Curriculum

## Evaluation DR. R. GUNASEKARAN

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

#### **Evaluation in Education**

Evaluation is a critical component of the educational process that involves assessing student learning, instructional effectiveness, and curriculum implementation. It serves to inform educators, students, and stakeholders about progress and areas for improvement. Here's an overview of the key aspects of evaluation in education.

#### **Types of Evaluation**

#### 1. Formative Evaluation

- **Purpose**: Conducted during the learning process to monitor student progress and provide ongoing feedback.
- Methods: Includes quizzes, assignments, observations, and class discussions.
- **Benefits**: Helps identify areas where students may be struggling, allowing for timely interventions and adjustments in instruction.

## 2. Summative Evaluation

- **Purpose**: Conducted at the end of an instructional unit or course to assess overall student learning and achievement.
- **Methods**: Includes final exams, projects, standardized tests, and end-of-term assessments.
- **Benefits**: Provides a comprehensive overview of what students have learned and can inform decisions about curriculum effectiveness.

## 3. **Diagnostic Evaluation**

- **Purpose**: Assesses students' prior knowledge and skills before instruction begins to identify strengths and weaknesses.
- Methods: Pre-assessments, entrance exams, and skills inventories.
- **Benefits**: Helps tailor instruction to meet the needs of individual learners and groups.

## 4. Norm-Referenced Evaluation

- **Purpose**: Compares a student's performance to that of a peer group or standard.
- Methods: Standardized tests that rank students based on performance.
- **Benefits**: Useful for identifying relative strengths and weaknesses among students.

## 5. Criterion-Referenced Evaluation

- **Purpose**: Measures student performance against predefined standards or criteria.
- Methods: Tests that assess specific learning outcomes or competencies.
- **Benefits**: Provides clear benchmarks for student achievement and mastery of content.

## Technology Prof. T. SELVARAJ

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

#### **Technology in Education**

Technology plays a transformative role in education, enhancing teaching and learning experiences. It provides innovative tools and resources that support instruction, engagement, and collaboration. Here's an overview of the key aspects of technology in education.

#### **Types of Educational Technology**

#### 1. Digital Learning Platforms

- **Description**: Online platforms that provide access to educational content, courses, and resources.
- **Examples**: Learning Management Systems (LMS) like Canvas, Moodle, and Google Classroom.

#### 2. Multimedia Tools

- **Description**: Software and applications that combine text, audio, video, and graphics to create engaging learning materials.
- **Examples**: Presentation tools (e.g., PowerPoint, Prezi), video editing software (e.g., iMovie, Adobe Premiere), and graphic design tools (e.g., Canva).

## 3. Interactive Technologies

- **Description**: Tools that encourage active participation and collaboration among students.
- **Examples**: Smartboards, interactive whiteboards, and audience response systems (e.g., Kahoot, Poll Everywhere).

## 4. Simulation and Virtual Reality (VR)

- **Description**: Technologies that create immersive learning experiences through simulations and VR environments.
- **Examples**: Virtual labs for science experiments, VR simulations for medical training, and historical reenactments.

## 5. Educational Apps and Games

- **Description**: Mobile applications and games designed to facilitate learning in a fun and engaging way.
- **Examples**: Language learning apps (e.g., Duolingo), math practice games (e.g., Prodigy), and coding platforms (e.g., Scratch).

## 6. Online Collaboration Tools

- **Description**: Software that enables real-time collaboration among students and teachers.
- **Examples**: Google Workspace (Docs, Sheets, Slides), Microsoft Teams, and Slack.
- 7. Data Analytics and Assessment Tools

## CHAPTER 9 Models of Teaching Prof. R. VAISHNAVI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

## **Models of Teaching**

Models of teaching provide structured frameworks that guide educators in delivering instruction and facilitating learning. These models encompass various teaching strategies, methodologies, and approaches tailored to achieve specific educational outcomes. Here are some prominent models of teaching:

#### 1. Direct Instruction Model

- **Description**: A teacher-centered approach that emphasizes structured lessons and clear objectives.
- Key Features:
  - Teacher-led presentations
  - Step-by-step explanations
  - Frequent assessments to monitor progress
- **Benefits**: Efficient for delivering specific content and ensuring all students grasp foundational concepts.

#### 2. Cooperative Learning Model

- **Description**: Focuses on collaborative learning where students work in small groups to achieve shared goals.
- Key Features:
  - Group activities and projects
  - Peer teaching and support
  - Emphasis on interpersonal skills
- **Benefits**: Encourages teamwork, communication, and problem-solving skills.

#### 3. Constructivist Model

- **Description**: Centers on the idea that learners construct their own understanding and knowledge through experiences and interactions.
- Key Features:
  - Inquiry-based learning
  - Exploration and discovery
  - Connection of new knowledge to prior experiences
- **Benefits**: Fosters critical thinking and promotes deeper understanding.

## 4. Problem-Based Learning (PBL) Model

- **Description**: Engages students in solving real-world problems, emphasizing research and inquiry.
- Key Features:
  - Identification of a complex problem
  - Collaborative group work
  - Research and presentation of solutions
- **Benefits**: Enhances critical thinking, collaboration, and application of knowledge.

## The Economics Teacher Prof. R. VAISHNAVI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

The role of an **economics teacher** is multifaceted, requiring not only subject expertise but also a variety of pedagogical skills to effectively engage students in the study of economics. Here is a breakdown of the key aspects that define a successful economics teacher:

#### **1. Subject Matter Expertise**

- **In-Depth Knowledge**: An economics teacher should possess a strong foundation in both microeconomics and macroeconomics, as well as specialized areas like international economics, developmental economics, and economic policy.
- **Current Trends**: Staying updated on the latest economic trends, policies, and global economic events is essential. A good teacher connects the curriculum with contemporary examples, making the content relevant to students.
- **Application of Theory**: The ability to not only teach economic theory but also to demonstrate its real-world applications is crucial. This includes understanding how policies like taxation, inflation control, or trade agreements impact everyday life.

## 2. Pedagogical Skills

- Effective Communication: Clear and precise communication is key when explaining complex economic concepts. A good teacher breaks down intricate ideas into simple, understandable parts while using relevant examples.
- **Student Engagement**: Economics can be abstract, so the teacher needs to employ various engagement strategies, such as debates, discussions, and interactive activities, to keep students interested.
- **Differentiated Instruction**: Adapting teaching methods to accommodate different learning styles is important in helping all students grasp economic concepts. This could involve a mix of lectures, discussions, problem-solving exercises, and multimedia resources.

## **3. Instructional Strategies**

- **Inquiry-Based Learning**: Encouraging students to ask questions about economic issues, research topics, and develop hypotheses helps foster a deeper understanding of economics.
- **Case Studies**: Using real-world examples and case studies is an effective way for students to see how economic theory applies in practical scenarios.
- **Data Analysis Skills**: An economics teacher should also be adept at guiding students through the analysis of data, teaching them how to interpret economic statistics, charts, and graphs.

# PEDAGOGY OF COMMERCE AND ACCOUNTANCY:



Edited by

**R.VAISHNAVI** 



Pedagogy of Commerce and Accountancy: Part – I Copyright ©2021-2022 by Prof. R. Vaishnavi

All rights reserved. No part of this book may be reproduced ortransmittedinanyformorbyanymeanswithoutwrittenpermissionfromtheauthor.

ISBN:978-93-6255-344-7

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

## TABLE OF CONTENTS

Introduction
CHAPTER 1 Aims and Objectives of Teaching Commerce and Accountancy17
PIOL R. VAISHNAVI
CHAPTER 2 Teaching Skills of Commerce and Accountancy27 Prof. T. SUBHASHINI
CHAPTER 3 Approaches of Teaching Commerce and Accountancy
DR. R. GUNASEKARAN
CHAPTER 4 Methods of Teaching for Commerce and Accountancy68
Prof. R. VAISHNAVI
CHADTED 5 Instructional Modia 85
Prof. T. SELVARAJ
CHAPTER 6 Curriculum
Prof. 1. SUBHASHINI
CHAPTER 7 Evaluation 125
DR. R. GUNASEKARAN
CHAPTER 8 Technology
Prof. 1. SELVARAJ
CHAPTER 9 Models of Teaching151
Prof. R. VAISHNAVI
CHAPTER 10 The Commerce Teacher
Prof. R. VAISHNAVI
REFERENCES171

## CHAPTER I Aims and Objectives of Teaching Commerce and Accountancy Prof. R. VAISHNAVI

## Assistant Professor, School of Education, PRIST University, Thanjavur

Teaching commerce and accountancy serves various educational and practical purposes. Here's an overview of the aims and objectives of teaching these subjects:

#### **Aims of Teaching Commerce and Accountancy**

- 1. **Understanding Economic Principles**: To provide students with a foundational understanding of economic concepts and principles that govern business operations and financial systems.
- 2. **Practical Skills Development**: To equip students with practical skills in financial management, bookkeeping, and accounting, preparing them for real-world applications in various business contexts.
- 3. **Business Awareness**: To foster an awareness of the role of commerce in society, including the ethical, social, and environmental implications of business practices.
- 4. **Critical Thinking and Problem-Solving**: To enhance students' ability to analyze financial information, make informed decisions, and solve problems related to commerce and accountancy.
- 5. **Preparation for Further Education and Careers**: To prepare students for further studies in business, finance, or related fields, as well as to equip them for various career paths in commerce and accountancy.

## **Objectives of Teaching Commerce**

## 1. Knowledge Acquisition:

• To impart knowledge of various business concepts, including marketing, management, finance, and entrepreneurship.

## 2. Skill Development:

• To develop skills in areas such as market analysis, consumer behavior, and business planning.

## 3. Understanding Business Operations:

• To explain the functions and interrelationships of different departments within a business (e.g., HR, production, sales).

## 4. Awareness of Business Environment:

• To cultivate an understanding of the external factors affecting businesses, including legal, social, economic, and technological changes.

## 5. Encouragement of Ethical Practices:

 To instill a sense of ethical responsibility in business practices and decisionmaking.

## CHAPTER 2 Teaching Skills of Commerce and Accountancy Prof. T. SUBHASHINI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Teaching commerce and accountancy effectively requires a combination of specific skills and strategies. Here's an overview of essential teaching skills for these subjects:

## **1. Subject Matter Expertise**

- **Deep Knowledge**: Teachers should possess a strong understanding of commerce and accounting principles, theories, and practices.
- **Current Trends**: Staying updated with the latest developments in the business and financial world, including regulatory changes and technological advancements.

## 2. Curriculum Development

- **Curricular Design**: Ability to design a comprehensive curriculum that meets educational standards and addresses diverse student needs.
- **Resource Utilization**: Incorporating various teaching resources, such as textbooks, online materials, and real-world case studies, to enhance learning.

## **3. Instructional Techniques**

- Lecture and Presentation Skills: Effectively delivering content through engaging lectures and presentations that maintain student interest.
- Active Learning: Implementing active learning strategies, such as group discussions, role-plays, and simulations, to encourage student participation.

## 4. Practical Application

- **Real-World Examples**: Relating theoretical concepts to real-life business scenarios to help students understand practical applications.
- Hands-On Activities: Designing activities such as mock business projects, accounting exercises, and case studies to reinforce learning.

## 5. Assessment and Feedback

- **Diverse Assessment Methods**: Utilizing various assessment techniques, including quizzes, projects, presentations, and exams, to evaluate student understanding.
- **Constructive Feedback**: Providing timely and constructive feedback to help students improve their skills and knowledge.

## 6. Technology Integration

• Use of Accounting Software: Familiarity with accounting software (e.g., QuickBooks, Excel) and incorporating it into teaching to enhance practical skills.

## CHAPTER 3 Approaches of Teaching Commerce and Accountancy DR. R. GUNASEKARAN

Assistant Professor, School of Education, School of Education, Thanjavur

Teaching commerce and accountancy can be approached through various methods that cater to different learning styles and objectives. Here are some effective approaches:

## 1. Lecture-Based Approach

- **Description**: Traditional method where the teacher delivers content through structured lectures.
- Advantages: Efficient for covering a large amount of material; useful for introducing fundamental concepts.
- Techniques: Use of slides, videos, and real-world examples to illustrate points.

## 2. Interactive Learning

- **Description**: Involves engaging students actively through discussions, debates, and interactive activities.
- Advantages: Encourages participation and deeper understanding; promotes critical thinking.
- **Techniques**: Group discussions, peer teaching, and role-playing scenarios.

## 3. Case Study Method

- **Description**: Analyzing real-life business scenarios to understand practical applications of theory.
- Advantages: Develops analytical skills and connects theory with practice; fosters decision-making.
- **Techniques**: Presenting case studies related to current events or historical business situations for group analysis.

## 4. Problem-Based Learning (PBL)

- **Description**: Students learn through the process of solving complex, real-world problems.
- Advantages: Enhances problem-solving skills and encourages collaborative work.
- **Techniques**: Formulating problems related to commerce and accountancy that students must research and solve.
- •

## 5. Experiential Learning

- Description: Learning through experience, where students engage in hands-on activities.
- Advantages: Reinforces theoretical concepts and builds practical skills.
- Techniques: Simulations, internships, and project-based assignments.

## Methods of Teaching for Commerce and Accountancy

## Prof. R. VAISHNAVI

## Assistant Professor, School of Education, PRIST Deemed University

Here are various methods of teaching commerce and accountancy, each designed to enhance student engagement and understanding:

## 1. Direct Instruction

- **Description**: Teacher-centered approach focusing on explicit teaching of concepts.
- **Techniques**: Lectures, demonstrations, and step-by-step explanations of accounting principles and commerce concepts.
- Use: Effective for introducing new material and ensuring foundational knowledge.

## 2. Discussion Method

- **Description**: Facilitating open discussions among students to explore topics deeply.
- **Techniques**: Guided discussions on current events in business, ethical dilemmas, or case studies.
- Use: Encourages critical thinking and allows students to articulate their views.

## 3. Case Method

- **Description**: Using real or hypothetical business scenarios for analysis.
- **Techniques**: Group analysis of case studies that require students to apply concepts to real-world situations.
- Use: Fosters practical application of theory and decision-making skills.

## 4. Cooperative Learning

- **Description**: Small groups working together to complete tasks or projects.
- Techniques: Assigning group projects related to business planning or financial analysis.
- Use: Enhances teamwork and communication skills while promoting peer learning.

## 5. Experiential Learning

- **Description**: Learning through direct experience and reflection.
- **Techniques**: Simulations, internships, and practical exercises like mock trading or business competitions.
- Use: Helps students connect theory with practice and develop hands-on skills.

## 6. Problem-Based Learning (PBL)

- **Description**: Students learn by solving complex, real-world problems.
- **Techniques**: Presenting problems related to accounting practices or business strategies for students to investigate and resolve.

## CHAPTER 5 Instructional Media Prof. T. SELVARAJ

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Instructional media refers to the various tools and resources used to enhance teaching and learning processes. In commerce and accountancy education, effective use of instructional media can significantly improve student engagement and understanding. Here are some key types of instructional media, along with their applications:

#### 1. Visual Aids

- **Description**: Tools that present information visually to enhance understanding.
- **Examples**: Charts, graphs, infographics, and diagrams.
- Use: Visualizing data trends, financial statements, or business processes.

#### 2. Presentation Software

- **Description**: Digital tools for creating slideshows and visual presentations.
- Examples: Microsoft PowerPoint, Google Slides, Prezi.
- Use: Delivering structured content and facilitating discussions during lectures.

#### 3. Videos and Multimedia

- **Description**: Engaging audio-visual content to illustrate concepts.
- **Examples**: Educational videos, documentaries, and animated tutorials.
- Use: Explaining complex topics, such as accounting principles or market trends.

#### 4. Online Learning Platforms

- **Description**: Digital platforms that provide access to educational resources and activities.
- Examples: Moodle, Blackboard, Coursera, Khan Academy.
- Use: Facilitating self-paced learning, assessments, and collaborative projects.

#### 5. Accounting Software

- **Description**: Specialized software used for bookkeeping and financial management.
- **Examples**: QuickBooks, Sage, Fresh Books.
- Use: Teaching practical accounting skills through real-world applications.

#### 6. Simulations and Virtual Labs

- **Description**: Interactive tools that mimic real-life business scenarios or accounting tasks.
- **Examples**: Business simulation games, virtual trading platforms.
- Use: Allowing students to practice decision-making in a risk-free environment.
#### Curriculum Prof. T. SUBHASHINI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Curriculum refers to the organized framework of educational content and learning experiences provided by a school or educational program. It encompasses the subjects taught, the skills developed, the methods used, and the assessments employed to gauge student learning. Here's an overview of key aspects of curriculum, particularly in the context of commerce and accountancy:

#### 1. Curriculum Components

- **Content**: The specific topics, concepts, and skills included in the curriculum. For commerce and accountancy, this might include financial accounting, management principles, marketing strategies, and business ethics.
- Learning Objectives: Clear goals outlining what students should know and be able to do by the end of a course or program.
- **Teaching Methods**: Strategies employed to deliver the content, such as lectures, discussions, case studies, and hands-on activities.
- Assessment: Tools and methods used to evaluate student learning, such as quizzes, exams, projects, and presentations.

#### 2. Curriculum Design

- **Subject-Centered Design**: Focuses on specific subjects or disciplines, emphasizing the content to be learned.
- **Student-Centered Design**: Prioritizes students' interests, needs, and learning styles, promoting active engagement.
- **Integrated Design**: Combines multiple subjects or themes, encouraging interdisciplinary learning (e.g., linking economics with marketing).

#### 3. Curriculum Development Process

- **Needs Assessment**: Identifying the knowledge and skills students need for future success in commerce and accountancy.
- **Curriculum Mapping**: Outlining the progression of topics and skills across grade levels or courses.
- **Pilot Testing**: Implementing the curriculum on a small scale to gather feedback before full implementation.
- **Review and Revision**: Continuously evaluating and updating the curriculum based on feedback, assessment results, and changing industry needs.

#### Evaluation

#### Dr. R. GUNASEKARAN

Assistant Professor, School Of Education, PRIST Deemed to be University, Thanjavur

Curriculum evaluation is a systematic process used to assess the effectiveness and relevance of an educational program. In the context of commerce and accountancy, evaluation helps ensure that the curriculum meets the needs of students and prepares them for their future careers. Here's a comprehensive overview of curriculum evaluation:

#### 1. Purpose of Curriculum Evaluation

- **Quality Assurance**: Ensures that the curriculum maintains high educational standards and aligns with institutional goals.
- **Continuous Improvement**: Identifies strengths and weaknesses, providing insights for curriculum revisions and enhancements.
- Accountability: Demonstrates to stakeholders (students, parents, employers) that the program is effective and meets educational standards.
- Alignment with Goals: Ensures that learning objectives are being met and that the curriculum is relevant to current industry needs.

#### 2. Types of Evaluation

- **Formative Evaluation**: Conducted during the implementation of the curriculum to monitor progress and make real-time adjustments.
  - **Methods**: Observations, quizzes, informal assessments, and student feedback.
- **Summative Evaluation**: Conducted after the curriculum has been fully implemented to assess overall effectiveness and outcomes.
  - **Methods**: Standardized tests, final projects, end-of-course evaluations, and overall performance metrics.

#### 3. Evaluation Criteria

- **Relevance**: The curriculum should be aligned with industry standards, current trends, and students' future career needs.
- **Effectiveness**: Measures whether students are achieving the intended learning outcomes and acquiring necessary skills.
- **Engagement**: Evaluates how well the curriculum engages students and fosters a positive learning environment.
- Accessibility: Assesses whether the curriculum accommodates diverse learning styles and needs.

#### 4. Evaluation Methods

• **Surveys and Questionnaires**: Collect feedback from students, teachers, and employers about the curriculum's effectiveness and relevance.

#### Technology Prof. T. SELVARAJ

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Technology plays a pivotal role in enhancing education, particularly in fields like commerce and accountancy. It can transform teaching methods, improve student engagement, and facilitate learning in various ways. Here's an overview of the impact of technology on education:

#### 1. Types of Technology in Education

- Learning Management Systems (LMS): Platforms that facilitate online learning, course management, and assessment.
  - **Examples**: Moodle, Canvas, Google Classroom.
- Accounting Software: Tools that simulate real-world accounting processes and practices.
  - **Examples**: QuickBooks, Xero, Sage.
- **Multimedia Resources**: Videos, podcasts, and interactive simulations that enhance understanding of complex concepts.
- **Collaboration Tools**: Applications that enable group work and communication among students and instructors.
  - Examples: Microsoft Teams, Slack, Zoom.
- **Data Analytics Tools**: Software that analyzes student performance and learning patterns to inform instruction and curriculum design.

#### 2. Benefits of Technology in Commerce and Accountancy Education

- Enhanced Engagement: Interactive tools and multimedia resources can make learning more engaging and enjoyable for students.
- Accessibility: Online resources and platforms provide access to educational materials anytime, anywhere, accommodating diverse learning needs.
- **Real-World Applications**: Technology allows students to work with current software and tools used in the industry, preparing them for future careers.
- **Collaboration and Communication**: Technology facilitates teamwork and communication among students and educators, enhancing collaborative learning experiences.
- **Personalized Learning**: Adaptive learning technologies can tailor educational experiences to meet individual student needs and learning styles.

#### 3. Challenges of Integrating Technology

- Access and Equity: Not all students may have equal access to technology, which can create disparities in learning opportunities.
- **Training and Support**: Educators may require training to effectively integrate technology into their teaching practices.

### CHAPTER 9 Models of Teaching Prof. R. VAISHNAVI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

Models of teaching provide structured approaches to instruction, helping educators design and deliver effective lessons. Here's an overview of some widely used models of teaching, particularly relevant for commerce and accountancy education:

#### 1. Direct Instruction Model

- **Description**: A teacher-centered approach that emphasizes explicit teaching of concepts.
- Key Features: Clear objectives, step-by-step demonstrations, and structured lessons.
- Application: Effective for introducing new accounting principles or business theories.

#### 2. Constructivist Model

- **Description**: Focuses on students constructing their own understanding and knowledge through experiences.
- Key Features: Emphasis on inquiry, problem-solving, and real-world applications.
- Application: Suitable for project-based learning and case studies in commerce.

#### 3. Cooperative Learning Model

- **Description**: Students work in small groups to complete tasks and achieve learning goals.
- Key Features: Interdependence, individual accountability, and collaborative skills.
- Application: Ideal for group projects and discussions on business scenarios.

#### 4. Inquiry-Based Learning Model

- **Description**: Encourages students to ask questions and explore topics through research and investigation.
- Key Features: Student-led exploration, critical thinking, and open-ended questions.
- Application: Useful for examining market trends or ethical dilemmas in business.

#### 5. Problem-Based Learning (PBL) Model

- **Description**: Students learn through solving real-world problems.
- **Key Features**: Emphasis on self-directed learning, collaboration, and application of knowledge.
- Application: Effective for analyzing financial statements or developing business plans.

#### 6. Flipped Classroom Model

- **Description**: Students learn new content outside of class (e.g., through videos) and engage in hands-on activities during class.
- Key Features: Increases class time for active learning and application of concepts.

#### The Commerce Teacher Prof. R. VAISHNAVI

Assistant Professor, School of Education, PRIST Deemed to be University, Thanjavur

The role of a commerce teacher is multifaceted and crucial for preparing students for the dynamic world of business and finance. Here's an overview of the key responsibilities, skills, and qualities of an effective commerce teacher:

#### 1. Roles and Responsibilities

- **Curriculum Development**: Designing and implementing a curriculum that aligns with educational standards and industry needs, covering topics such as economics, accounting, marketing, and business law.
- **Instruction**: Delivering engaging lessons using various teaching methods (e.g., lectures, discussions, case studies) to ensure student understanding and retention of material.
- Assessment and Evaluation: Developing assessment tools (quizzes, exams, projects) to evaluate student learning and providing constructive feedback to help students improve.
- **Classroom Management**: Creating a positive and inclusive classroom environment that fosters learning and encourages student participation.
- **Professional Development**: Staying current with trends in commerce and education through continuous learning, attending workshops, and networking with industry professionals.

#### 2. Key Skills

- **Subject Matter Expertise**: Deep knowledge of commerce-related topics, including accounting principles, business management, economics, and finance.
- **Communication Skills**: The ability to clearly convey complex concepts and facilitate discussions, ensuring that students understand the material.
- **Pedagogical Skills**: Proficiency in various teaching methods and strategies to address different learning styles and engage students.
- **Analytical Thinking**: The ability to analyze student performance data and adjust teaching methods accordingly to improve learning outcomes.
- **Technological Proficiency**: Familiarity with educational technology and tools, such as learning management systems, accounting software, and multimedia resources.

#### 3. Qualities of an Effective Commerce Teacher

- **Passion for Teaching**: A genuine enthusiasm for educating students and helping them succeed in their academic and career goals.
- Adaptability: Flexibility to adjust teaching approaches based on student needs, feedback, and changing industry trends.
- **Empathy**: Understanding and addressing the diverse backgrounds and challenges faced by students, fostering a supportive learning environment.



## PLANT AND NANO BIOTECHNOLOGY

EDITED BY DR.A. XAVIER FERNAND



## Plant and Nano biotechnology

Copyright©2021 by DR. A. XAVIER FERNANDES

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-200-6

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

Introduction
Chapter I :Introduction to Plant and Nanobiotechnology
DR. A. XAVIER FERNANDES
Chapter II : Plant Tissue Culture Techniques
DR.S.MOHANRAJ
Chapter III: Genetic Engineering in Plants
Dr. K. Sundar
Chapter IV :Molecular Markers and Marker-Assisted Selection
DR. T.Ushadevi
Chapter V: Nanotechnology in Plant Biotechnology
DRR.Sathya
Chapter VI: Nano-Delivery Systems for Plant Biotechnology
Dr.T.Thiruselvi
Chapter VII:Plant-Microbe Interactions in Biotechnology
Dr.N.Mahalakshimi
Chapter VIII : Phytoremediation and Environmental Applications
Dr.G. Chandirasegaran
Chapter Ix : Regulatory and Ethical Considerations in Plant and Nano Biotechnology
Dr.K.P.Karuppaian
Chapter X Future Trends and Innovations in Plant and Nano Biotechnology
DR.Ambika K
REFERENCE

## TABLE OF CONTENTS

#### **CHAPTER I**

#### INTRODUCTION TO PLANT AND NANO BIOTECHNOLOGY

#### **DR.A.XAVIER FERNANDES**

Assistant Professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

#### **Introduction to Plant Biotechnology**

Plant biotechnology is a dynamic field that harnesses scientific techniques to enhance plant growth, resistance, and productivity. By manipulating plant genetics, researchers can develop crops with improved traits, such as increased yield, pest resistance, and tolerance to environmental stresses. Techniques like genetic engineering and tissue culture are commonly employed to achieve these goals, leading to the creation of transgenic plants that can contribute to food security and sustainable agriculture. The advancements in plant biotechnology not only address agricultural challenges but also offer solutions for enhancing nutritional quality and reducing the reliance on chemical inputs.

#### **Role of Nanotechnology in Agriculture**

Nanotechnology has emerged as a transformative force in agriculture, providing innovative solutions to enhance crop productivity and sustainability. By applying nanoscale materials and techniques, researchers can develop targeted delivery systems for fertilizers, pesticides, and growth regulators, improving their efficiency and minimizing environmental impact. Nanoparticles can also play a role in disease detection and monitoring plant health, allowing for timely interventions. This integration of nanotechnology in plant biotechnology promises to optimize resource use, reduce waste, and promote sustainable farming practices, addressing the pressing challenges of global food production.

#### Synergistic Potential of Plant and Nanobiotechnology

The convergence of plant biotechnology and nanotechnology holds immense potential for revolutionizing agriculture. By combining genetic modifications with nanomaterials, researchers can create plants that not only possess enhanced traits but also respond better to their environments. For instance, nanoparticles can be engineered to deliver specific genes or protect plants from pathogens, leading to more resilient crops. This synergistic approach enables a comprehensive strategy for improving agricultural outputs while minimizing ecological footprints.

#### **CHAPTER II**

#### **Plant Tissue Culture Techniques**

#### **DR.S.MOHANRAJ**

## Associate professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Plant tissue culture is a collection of techniques used to grow plant cells, tissues, or organs in a controlled, sterile environment. This method allows for the propagation of plants under optimal conditions, bypassing the limitations of traditional propagation methods. By using small pieces of plant tissue, such as leaves, stems, or even single cells, scientists can create entire plants from a minimal amount of starting material. This technique is invaluable for the rapid multiplication of plant species, especially those that are rare, endangered, or difficult to propagate through conventional means.

One of the primary advantages of plant tissue culture is the ability to produce genetically identical plants, known as clones. This is particularly beneficial in agriculture and horticulture, where uniformity in traits such as disease resistance, yield, and growth rate is desirable. Additionally, tissue culture can help in the conservation of plant species by facilitating the preservation of genetic material. Techniques like cryopreservation can be employed to store plant tissues for long periods, ensuring that valuable genetic diversity is maintained for future generations.

Moreover, plant tissue culture has significant applications in research and biotechnology. It enables scientists to study plant development, genetic transformation, and the effects of various environmental conditions on growth. The use of genetically modified organisms (GMOs) is facilitated through tissue culture techniques, allowing for the introduction of new traits, such as pest resistance or enhanced nutritional value. Overall, plant tissue culture is a versatile and powerful tool that contributes to advancements in agriculture, conservation, and scientific research.

#### **CHAPTER III**

#### **Genetic Engineering in Plants**

#### Dr. K.SUNDAR

### Associate professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Genetic engineering in plants involves the direct manipulation of an organism's DNA to introduce new traits or enhance existing ones. This biotechnological approach allows scientists to modify plants for various purposes, including improved resistance to pests and diseases, enhanced nutritional content, and increased tolerance to environmental stresses such as drought or salinity. Techniques such as gene cloning, CRISPR-Cas9 gene editing, and Agrobacterium-mediated transformation enable precise modifications at the genetic level, facilitating advancements in crop production and sustainability.

One of the key benefits of genetic engineering in plants is the ability to develop genetically modified organisms (GMOs) that can significantly increase agricultural productivity. For example, crops like Bt cotton and Bt corn have been engineered to express a protein from the bacterium Bacillus thuringiensis, which provides resistance against specific insect pests. This reduces the reliance on chemical pesticides, leading to lower production costs and a decreased environmental impact. Additionally, biofortified crops, such as Golden Rice, have been developed to address nutritional deficiencies in populations reliant on staple foods.

Despite the potential benefits, genetic engineering in plants also raises ethical and regulatory concerns. Public apprehension regarding GMOs centers around issues of food safety, environmental impact, and biodiversity. As a result, rigorous regulatory frameworks and safety assessments are essential to ensure that genetically engineered crops are safe for consumption and do not adversely affect ecosystems. Ongoing dialogue among scientists, policymakers, and the public is crucial to address these concerns while harnessing the benefits of genetic engineering to improve global food security and resilience to climate change.

#### **CHAPTER IV**

## Molecular Markers and Marker-Assisted SelectionPlant and Nano biotechnology Dr. T.Ushadevi

#### Associate professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

#### **Molecular Markers**

Molecular markers are specific sequences in the genome that serve as indicators for certain traits or characteristics in organisms. These markers can be categorized into various types, such as DNA markers (like single nucleotide polymorphisms and simple sequence repeats), RNA markers for gene expression analysis, and protein markers that focus on variations in protein levels. Their primary role is to facilitate genetic mapping, assess genetic diversity, and establish associations between specific markers and desirable traits, such as disease resistance or improved yield in crops.

#### Marker-Assisted Selection (MAS)

Marker-assisted selection (MAS) utilizes molecular markers to streamline the breeding process. The approach involves selecting markers associated with traits of interest, followed by genotyping individuals in a breeding population to identify those carrying desirable genetic traits. This method allows breeders to select plants or animals early in their development, significantly speeding up the breeding cycle and enhancing selection accuracy. By focusing on genetic information rather than solely on observable traits, MAS can lead to more precise and efficient breeding outcomes.

#### **Advantages and Challenges**

The advantages of MAS include faster development of improved varieties, increased precision in selecting for specific traits, and cost savings through reduced generations needed for selection. However, there are challenges, such as the need for marker validation to ensure markers are truly linked to the desired traits and the influence of genetic background on trait expression. Moreover, the technique requires specialized knowledge and access to advanced molecular biology tools. Despite these challenges, the integration of molecular markers and MAS is revolutionizing agricultural breeding and contributing to global food security.

#### **CHAPTER V**

#### NANOTECHNOLOGY IN PLANT BIOTECHNOLOGY DR..R.Sathya

Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Nanotechnology is increasingly playing a pivotal role in plant biotechnology, offering innovative solutions to enhance agricultural productivity and sustainability. One of the most promising applications is in the development of nanofertilizers, which provide controlled release of nutrients directly to plants. This targeted delivery system minimizes nutrient waste and maximizes uptake, leading to healthier crops and improved yields. By utilizing nanoparticles, farmers can ensure that plants receive the essential nutrients they need at optimal times, reducing the reliance on traditional fertilizers and lowering the environmental impact of agricultural practices.

In addition to nanofertilizers, nanotechnology is transforming pest management through the use of nanopesticides. These advanced formulations allow for the precise targeting of specific pests, resulting in reduced chemical usage and lower risks to beneficial organisms. Nanopesticides can also provide sustained protection through controlled release mechanisms, enhancing their effectiveness and longevity. Furthermore, the integration of nanosensors in plant disease management enables early detection of pathogens at minimal concentrations, allowing for timely interventions that can save crops and reduce economic losses.

Beyond enhancing productivity, nanotechnology also addresses pressing environmental challenges. For instance, nanoparticles can be employed in soil remediation to absorb or degrade contaminants, improving soil health and promoting sustainable agricultural practices. Additionally, nanotechnology facilitates genetic engineering by improving the delivery of genetic material into plant cells, enhancing the efficiency of techniques like CRISPR for crop improvement. However, as this field advances, it is crucial to consider the safety and environmental implications of nanomaterials, necessitating robust regulatory frameworks to ensure their responsible use in agriculture.

#### **CHAPTER VI**

#### NANO-DELIVERY SYSTEMS FOR PLANT BIOTECHNOLOGY

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Nano-delivery systems are revolutionizing plant biotechnology by enabling the precise and efficient transfer of nutrients, pesticides, and genetic material to plants. These systems utilize nanoparticles as carriers, which can encapsulate various substances and release them in a controlled manner. For instance, in the case of fertilizers, nanoscale carriers can deliver essential nutrients directly to plant roots or leaves, enhancing uptake and minimizing nutrient loss. This targeted approach not only boosts plant growth and yield but also reduces the environmental impact associated with conventional fertilization methods.

In pest management, nano-delivery systems are proving effective in administering nanopesticides that can specifically target harmful pests while sparing beneficial organisms. By using nanoparticles to encapsulate active ingredients, these systems allow for a slow and sustained release, ensuring that the pesticide remains effective over a longer period. This approach not only reduces the frequency of applications but also decreases the total amount of chemicals used, thereby mitigating the risks associated with chemical runoff and enhancing the sustainability of agricultural practices.

Moreover, nano-delivery systems are making significant strides in genetic engineering. By utilizing nanoparticles to transport DNA or RNA directly into plant cells, researchers can enhance the efficiency of transformation techniques. This method has the potential to facilitate precise gene editing and the development of crops with desirable traits, such as increased resistance to diseases or improved nutritional profiles. However, as these technologies advance, it is vital to address safety and regulatory concerns to ensure that their implementation is both effective and environmentally responsible.

#### **CHAPTER VII**

#### **Plant-Microbe Interactions in Biotechnology**

#### **DR.N.MAHALAKSHMI**

Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Plant-microbe interactions play a crucial role in biotechnology, offering innovative strategies to enhance plant growth, resilience, and productivity. Beneficial microorganisms, such as mycorrhizal fungi and rhizobacteria, establish symbiotic relationships with plants, improving nutrient uptake, particularly phosphorus and nitrogen. These interactions enhance plant health and growth, making them invaluable in sustainable agricultural practices. By harnessing these natural partnerships, biotechnologists can develop biofertilizers that promote crop yield while minimizing reliance on chemical fertilizers, ultimately reducing environmental impact.

Moreover, certain microbes are instrumental in promoting plant resistance to pathogens and environmental stressors. For instance, beneficial bacteria can produce metabolites that inhibit the growth of harmful pathogens, while also enhancing the plant's own defense mechanisms. This biocontrol aspect of plant-microbe interactions is increasingly being explored in the development of biopesticides, offering an eco-friendly alternative to chemical pesticides. By utilizing these microbial allies, farmers can improve crop protection and contribute to more resilient agricultural systems.

Additionally, advances in biotechnology are enabling researchers to manipulate plant-microbe interactions for enhanced outcomes. Techniques such as genetic engineering and synthetic biology allow for the enhancement of specific traits in both plants and microbes, optimizing their symbiotic relationships. For example, genetically engineered plants may express signaling compounds that attract beneficial microbes to their roots, further enhancing nutrient absorption and stress resistance. As our understanding of these complex interactions deepens, the potential for leveraging plant-microbe relationships in biotechnology continues to expand, paving the way for innovative solutions to global agricultural challenges.

#### **CHAPTER VIII**

#### PHYTOREMEDIATION AND ENVIRONMENTAL APPLICATIONS

#### Dr.G. Chandirasegaran

Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Phytoremediation is an innovative bioremediation technique that utilizes plants to clean up contaminated environments, particularly soils and water. This process involves the uptake, accumulation, and detoxification of pollutants, including heavy metals, pesticides, and hydrocarbons, through various plant mechanisms. Certain plant species, known as hyperaccumulators, can absorb and concentrate these contaminants in their tissues, effectively removing harmful substances from the environment. This green technology offers a sustainable and cost-effective approach to rehabilitating polluted sites, making it an attractive alternative to traditional remediation methods.

One of the key advantages of phytoremediation is its ability to enhance soil health and restore ecosystems. By using plants to remediate contaminated sites, the natural structure and function of the soil can be preserved or improved. Additionally, the root systems of plants help stabilize the soil, preventing erosion and promoting biodiversity. Furthermore, phytoremediation can also enhance the microbial communities in the rhizosphere, which can further aid in breaking down pollutants through natural biodegradation processes. This holistic approach not only cleans up the environment but also fosters the recovery of native flora and fauna.

In recent years, advancements in biotechnology have further enhanced the potential of phytoremediation. Genetic engineering techniques allow scientists to develop plant varieties with improved capabilities for contaminant uptake and detoxification. For example, genetically modified plants can be engineered to express specific enzymes that break down pollutants more effectively or to enhance their tolerance to heavy metals. These innovations not only increase the efficiency of phytoremediation efforts but also expand the range of contaminants that can be targeted. As environmental concerns grow, phytoremediation stands out as a promising solution for addressing pollution while promoting ecological restoration and sustainability.

#### CHAPTER IX

#### REGULATORY AND ETHICAL CONSIDERATIONS IN PLANT AND NANO BIOTECHNOLOGY

#### **DR.K.P.KARUPPAIAN**

Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Regulatory and ethical considerations in plant and nano biotechnology are critical for ensuring the safe and responsible development of these technologies. Regulatory frameworks vary significantly across countries, reflecting different cultural attitudes toward biotechnology and its applications. In many regions, plants developed through genetic engineering or nanotechnology must undergo rigorous safety assessments to evaluate their potential impacts on human health and the environment. These assessments often include testing for allergenicity, toxicity, and unintended ecological consequences, necessitating a comprehensive understanding of the technology's implications throughout the product's lifecycle.

Ethical considerations also play a significant role in shaping public perception and acceptance of plant and nano biotechnologies. Concerns regarding the potential risks associated with genetically modified organisms (GMOs) and nanomaterials, such as their long-term effects on biodiversity and ecosystem health, need to be addressed transparently. Additionally, issues related to patenting and ownership of biotechnological innovations raise questions about equitable access to these technologies, particularly for smallholder farmers in developing regions. Engaging stakeholders, including farmers, consumers, and environmental groups, in the regulatory process can foster trust and ensure that ethical principles are incorporated into decision-making.

Furthermore, as the fields of plant and nano biotechnology evolve, continuous updates to regulatory frameworks are necessary to keep pace with technological advancements. Policymakers must strike a balance between fostering innovation and ensuring safety, often relying on scientific evidence to guide regulations. International collaboration is also essential in addressing the transboundary nature of biotechnology, as the movement of genetically modified crops and nano-enabled products can have global implications.

#### CHAPTER X

#### FUTURE TRENDS AND INNOVATIONS IN PLANT AND NANO BIOTECHNOLOGY

#### Dr.Ambika.K

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

The future of plant and nanotechnology promises exciting advancements that could significantly impact agriculture, food security, and environmental sustainability. One of the most promising trends is the integration of precision agriculture with nano-enabled technologies. By utilizing nanosensors and drones, farmers can monitor soil health, plant growth, and pest populations in real time. This data-driven approach allows for more precise applications of water, fertilizers, and pesticides, reducing waste and environmental impact while optimizing crop yields. The synergy of these technologies will lead to more efficient farming practices and resource management, aligning with global sustainability goals.

Another key innovation is the advancement of genetic engineering techniques, particularly CRISPR-Cas9 and other gene-editing tools. These technologies are being refined to enable more precise modifications in plants, enhancing desirable traits such as drought resistance, disease tolerance, and nutritional content. Coupled with nano-delivery systems, which can effectively transport genetic materials into plant cells, researchers are now able to develop crops that are not only more resilient but also better equipped to meet the nutritional demands of a growing global population. This combination of genetic innovation and nano-technology holds the potential to revolutionize food production and address pressing agricultural challenges.

Moreover, the application of nanotechnology in phytoremediation is gaining traction as environmental concerns rise. Future innovations may include engineered plants designed to target specific pollutants with enhanced efficiency through nanomaterials that facilitate faster uptake and degradation of contaminants. As our understanding of plant-microbe interactions improves, integrating beneficial microbes with nano-delivery systems could create synergistic effects for soil and water remediation. This holistic approach not only addresses pollution but also promotes ecosystem recovery, positioning plant and nano biotechnology at the forefront of sustainable environmental management.

## PHARMACEUTICAL MICROBIOLOGY

Edited by

**DR. A. XAVIER FERNANDES** 



## Pharmaceutical Microbiology Copyright©2021 by DR. A. XAVIER FERNANDES

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-075-0

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

Introduction
Chapter I :Introduction to Pharmaceutical Microbiology DR. A. Xavier Fernandes
Chapter II : Microbial Contamination and Control in Pharmaceuticals
DR.S.Mohanraj
Chapter III: Sterilization Techniques in Pharmaceutical Manufacturing
Dr. K. Sundar
DR. T.Ushadevi
DRR.Sathya Chapter VI: Biopharmaceuticals and Biotechnology
Dr.T.Thiruselvi Chapter VII: Plant-Quality Control and Microbial Testing in Pharmaceuticals
Dr.N.Mahalakshimi Chapter VIII : Antimicrobial Resistance in Pharmaceutical Microbiology
<b>Dr.G. Chandirasegaran</b> Chapter Ix : Good Manufacturing Practices (GMP) in Microbial Production
Dr.K.P.Karuppaian Chapter X Future Trends in Pharmaceutical Microbiology
DR.Ambika K REFERENCES

#### TABLE OF CONTENTS

#### **CHAPTER I**

#### INTRODUCTION TO PHARMACEUTICAL MICROBIOLOGY

#### **DR.A.XAVIER FERNANDES**

Assistant Professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

#### Introduction to Pharmaceutical microbiology

Pharmaceutical microbiology is a critical field that focuses on the role of microorganisms in the development, production, and safety of pharmaceuticals. This discipline encompasses various aspects, including the study of pathogenic microbes, the impact of environmental factors on microbial growth, and the application of microbial techniques in drug formulation and testing. Understanding the interactions between microorganisms and pharmaceuticals is essential for ensuring the efficacy and safety of therapeutic agents.

One of the primary concerns in pharmaceutical microbiology is contamination control. Microbial contamination can compromise the integrity of drug products, leading to potential health risks for patients. Consequently, stringent aseptic techniques, sterilization methods, and rigorous quality control measures are implemented throughout the manufacturing process. Regulatory bodies, such as the FDA and EMA, establish guidelines to minimize contamination risks and ensure that pharmaceutical products meet stringent microbiological standards.

Additionally, pharmaceutical microbiology plays a vital role in the development of new antimicrobial agents. As antibiotic resistance becomes a growing concern, the discovery of novel compounds and the optimization of existing drugs are crucial for combating resistant pathogens. Microbial fermentation processes are often utilized to produce these compounds, highlighting the importance of microorganisms in drug discovery and development. Through ongoing research and innovation, pharmaceutical microbiology continues to advance our understanding of microorganisms, leading to safer and more effective therapies for various diseases.

#### **CHAPTER II**

## MICROBIAL CONTAMINATION AND CONTROL IN PHARMACEUTICALS

#### DR.S.MOHANRAJ

## Associate professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Microbial contamination in pharmaceuticals poses significant risks to product safety and efficacy. Contaminants can originate from various sources, including raw materials, manufacturing environments, and even personnel. Bacteria, fungi, and viruses can compromise the quality of pharmaceutical products, leading to adverse effects in patients. To mitigate these risks, it is crucial to understand the pathways of contamination and implement rigorous controls throughout the production process.

Effective contamination control begins with good manufacturing practices (GMP) that emphasize cleanliness and hygiene. This includes maintaining sterile environments, using validated sterilization techniques, and employing proper aseptic handling procedures. Facilities are designed with controlled access to minimize the risk of introducing microorganisms, and air filtration systems are used to ensure a clean air supply. Additionally, regular microbiological monitoring of surfaces, equipment, and finished products helps identify potential contamination sources and facilitates prompt corrective actions.

Education and training of personnel are also vital components of contamination control. Workers must be well-versed in aseptic techniques and the importance of maintaining a contamination-free environment. Moreover, ongoing research into emerging microbial threats and the development of new disinfection methods can enhance existing protocols. By fostering a culture of vigilance and compliance, pharmaceutical manufacturers can significantly reduce the risk of microbial contamination, ensuring the safety and effectiveness of their products for consumers.

#### **CHAPTER III**

#### STERILIZATION TECHNIQUES IN PHARMACEUTICAL MANUFACTURING

#### **Dr. K.SUNDAR**

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Sterilization techniques are crucial in pharmaceutical manufacturing to ensure that products are free from viable microorganisms that could compromise safety and efficacy. Various methods are employed, each suited for different types of products and materials. Common techniques include steam sterilization (autoclaving), dry heat sterilization, ethylene oxide (EO) gas sterilization, and radiation. Steam sterilization is particularly effective for heat-stable liquids and solids, while EO gas is often used for heat-sensitive materials, such as certain medical devices and biological products.

Each sterilization method has its own set of parameters, such as temperature, pressure, and exposure time, which must be carefully validated to ensure consistent and effective sterilization. For instance, autoclaving requires precise temperature and pressure levels to achieve sterility, typically at 121°C for at least 15 minutes, depending on the load. In contrast, dry heat sterilization operates at higher temperatures and longer durations, making it suitable for materials that might be damaged by moisture. Validation processes, including biological indicators and chemical indicators, are critical to confirming the effectiveness of these sterilization methods.

In addition to these traditional techniques, emerging technologies, such as vaporized hydrogen peroxide and microwave sterilization, are being explored for their potential to enhance sterilization efficiency and reduce processing times. The choice of sterilization method depends on the nature of the pharmaceutical product, regulatory requirements, and the specific microbial load being targeted. By employing rigorous sterilization protocols, pharmaceutical manufacturers can ensure the production of safe and effective products while minimizing contamination risks.

#### **CHAPTER IV**

## ANTIBIOTICS AND ANTIMICROBIAL AGENTS Dr. T.Ushadevi

## Associate professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Antibiotics and antimicrobial agents are crucial tools in the fight against infectious diseases, acting to inhibit the growth of or kill pathogens such as bacteria, fungi, and viruses. Antibiotics, which are specifically designed to target bacterial infections, work through various mechanisms, including disrupting cell wall synthesis, inhibiting protein synthesis, and interfering with DNA replication. Common examples include penicillins, tetracyclines, and macrolides. While antibiotics have revolutionized medicine, their misuse and overuse have contributed to the alarming rise of antibiotic-resistant strains, posing a significant challenge to public health.

Antimicrobial agents encompass a broader category that includes not only antibiotics but also antifungals, antivirals, and antiparasitics. Each class of these agents has unique mechanisms of action tailored to their specific targets. For instance, antifungal agents like azoles inhibit fungal cell membrane synthesis, while antivirals such as acyclovir target viral replication processes. The development of new antimicrobial agents is critical, especially in light of emerging resistance patterns, which necessitate ongoing research and innovation in drug discovery and development.

The responsible use of antibiotics and antimicrobial agents is essential for sustaining their effectiveness. Strategies such as antimicrobial stewardship programs aim to optimize prescribing practices, educate healthcare providers and patients, and promote research into alternative treatments. Additionally, understanding the pharmacokinetics and pharmacodynamics of these agents helps in selecting the appropriate drug for specific infections and tailoring treatment regimens. By addressing resistance and improving the utilization of existing agents, the medical community can enhance treatment outcomes and preserve the efficacy of these vital therapeutic tools.

#### **CHAPTER V**

#### VACCINES AND IMMUNOBIOLOGICAL PRODUCTS DR..R.Sathya

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India Vaccines and Immunobiological Products

Vaccines and immunobiological products play a critical role in preventing infectious diseases and enhancing public health. Vaccines stimulate the immune system to recognize and combat specific pathogens, providing immunity without causing the disease itself. They can be composed of live attenuated organisms, inactivated pathogens, subunit components, or mRNA, each designed to elicit a robust immune response. The development of vaccines has significantly reduced the incidence of diseases such as measles, polio, and influenza, demonstrating their effectiveness in controlling infectious outbreaks.

Immunobiological products extend beyond traditional vaccines and include a range of therapeutic agents such as monoclonal antibodies, antitoxins, and immune globulins. These products are designed to treat or prevent disease by modulating the immune response or directly targeting pathogens. For example, monoclonal antibodies have gained prominence in treating various conditions, including cancer and autoimmune diseases, by specifically targeting and neutralizing harmful cells or proteins. The versatility of immunobiological products continues to expand with advancements in biotechnology, enabling more tailored approaches to disease management.

The development and production of vaccines and immunobiological products involve stringent regulatory processes to ensure their safety, efficacy, and quality. Comprehensive clinical trials are conducted to evaluate their performance, and post-marketing surveillance is essential to monitor long-term effects. Additionally, global vaccination initiatives highlight the importance of equitable access to these life-saving products, especially in low-resource settings. By advancing vaccine research and ensuring widespread distribution, the healthcare community can effectively combat infectious diseases and improve global health outcomes.

#### **CHAPTER VI**

#### BIOPHARMACEUTICAL AND BIOTECHNOLOGY Dr.T.Thiruselvi

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Biopharmaceuticals are a class of therapeutic products derived from biological sources, including proteins, nucleic acids, and living cells. Unlike traditional pharmaceuticals, which are typically synthesized through chemical processes, biopharmaceuticals are produced using biotechnological methods such as recombinant DNA technology, cell culture, and fermentation. This innovative approach allows for the development of complex molecules like monoclonal antibodies, therapeutic proteins, and vaccines that can target specific diseases with greater precision. The advent of biopharmaceuticals has transformed the treatment landscape for chronic conditions such as diabetes, cancer, and autoimmune disorders.

Biotechnology plays a pivotal role in the research and development of biopharmaceuticals. Techniques such as genetic engineering and synthetic biology enable scientists to manipulate living organisms to produce desired therapeutic compounds. For example, recombinant technology allows for the mass production of insulin by inserting human insulin genes into bacteria, making it widely available for diabetic patients. Additionally, advancements in biotechnology, such as CRISPR gene editing and high-throughput screening, have accelerated the discovery of new biopharmaceuticals and improved the efficiency of the drug development process.

The regulatory landscape for biopharmaceuticals is complex, given the unique challenges posed by biological products. Regulatory agencies, such as the FDA and EMA, require extensive testing for safety, efficacy, and quality, often involving rigorous clinical trials and post-market surveillance. Moreover, the scalability of production and the need for consistent product characterization present additional hurdles for manufacturers. As biopharmaceuticals continue to evolve, ongoing research, innovation, and collaboration between academia, industry, and regulatory bodies will be essential to harness their full potential and ensure that these therapies remain safe and accessible to patients worldwide.

#### **CHAPTER VII**

#### **QUALITY CONTROL AND MICROBIAL TESTING IN PHARMACEUTICALS**

#### **DR.N.MAHALAKSHMI**

Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Quality control and microbial testing are essential components of pharmaceutical manufacturing, ensuring that products meet safety and efficacy standards. These processes involve rigorous testing protocols to detect any microbial contamination that could compromise the integrity of the final product. Quality control begins with the selection of raw materials, which must be assessed for microbial load, followed by in-process testing during production. This proactive approach helps identify potential issues early, minimizing the risk of contaminated products reaching consumers.

Microbial testing methods include both direct and indirect techniques, such as culturing samples on selective media, using molecular methods like PCR, and employing rapid microbial testing systems. Each method has its own advantages, with culture-based methods being the gold standard for identifying and quantifying viable microorganisms. However, newer technologies, such as automated systems and biosensors, offer faster results and greater sensitivity. Regular monitoring of environmental conditions in production facilities, including air quality and surface cleanliness, is also vital in controlling microbial contamination.

Regulatory agencies, such as the FDA and EMA, mandate stringent guidelines for quality control and microbial testing to ensure compliance with Good Manufacturing Practices (GMP). These guidelines require comprehensive documentation and validation of testing methods, ensuring that all products released to the market are free from harmful microbial contamination. By implementing robust quality control measures and rigorous microbial testing, pharmaceutical manufacturers can enhance product safety, maintain consumer trust, and uphold the reputation of the industry.

#### **CHAPTER VIII**

#### ANTIMICROBIAL RESISTANCE IN PHARMACEUTICAL MICROBIOLOGY Dr.G. Chandirasegaran

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Antimicrobial resistance (AMR) is a growing concern in pharmaceutical microbiology, significantly complicating the treatment of infectious diseases. This phenomenon occurs when microorganisms evolve mechanisms to withstand the effects of antimicrobial agents, rendering standard treatments ineffective. Factors contributing to AMR include the overuse and misuse of antibiotics in healthcare and agriculture, inadequate infection control practices, and poor patient adherence to prescribed treatments. As resistant strains proliferate, they not only pose direct health risks to individuals but also lead to increased healthcare costs and longer hospital stays, creating a public health crisis that requires urgent attention.

Pharmaceutical microbiology plays a pivotal role in addressing AMR through the development of new antimicrobial agents and the optimization of existing therapies. Ongoing research aims to identify novel compounds that can effectively target resistant pathogens while minimizing the risk of further resistance development. Additionally, advances in biotechnology, such as the use of bacteriophages and antimicrobial peptides, offer promising alternatives to traditional antibiotics. Collaborative efforts among researchers, pharmaceutical companies, and regulatory bodies are essential to facilitate the discovery of innovative solutions to combat AMR.

Preventing and controlling AMR also involves implementing robust stewardship programs that promote the responsible use of antimicrobial agents. These programs educate healthcare providers and patients about the importance of adhering to treatment regimens, conducting appropriate diagnostics before prescribing, and reducing unnecessary prescriptions. Moreover, continuous surveillance of resistance patterns is crucial for informing public health strategies and tailoring treatment guidelines. By fostering a comprehensive approach that includes research, education, and stewardship, the pharmaceutical industry can effectively combat antimicrobial resistance and safeguard the efficacy of current and future therapies.

#### **CHAPTER IX**

## GOOD MANUFACTURING PRACTICES (GMP) IN MICROBIAL PRODUCTION DR.K.P.KARUPPAIAN

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Good Manufacturing Practices (GMP) are essential guidelines in microbial production to ensure the safety, quality, and efficacy of biopharmaceutical products. These practices encompass a comprehensive framework that governs the entire production process, from raw material sourcing to product distribution. In microbial production, adherence to GMP helps mitigate risks of contamination, ensure proper documentation, and maintain consistent product quality. Facilities must be designed to minimize contamination risks, with controlled environments that include appropriate airflow systems, sterilization processes, and sanitation protocols to protect both the product and the workforce.

The implementation of GMP in microbial production involves rigorous quality control measures and standardized operating procedures. This includes the validation of processes, equipment, and cleaning methods, as well as regular monitoring of environmental conditions such as temperature, humidity, and microbial load. Training personnel in GMP standards is vital to fostering a culture of quality and compliance, ensuring that everyone involved in the production process understands their role in maintaining product integrity. Documentation is also a critical aspect, as detailed records of production activities, quality testing, and deviations must be maintained to demonstrate compliance with regulatory standards.

Regulatory bodies, such as the FDA and EMA, enforce GMP guidelines to safeguard public health and ensure that microbial products meet required specifications. Regular audits and inspections are conducted to assess compliance, and any non-conformities can lead to significant repercussions, including product recalls or facility shutdowns. By consistently applying GMP principles, companies not only enhance the safety and efficacy of their microbial products but also build trust with consumers and regulatory agencies. This commitment to quality ultimately supports the long-term sustainability of the biopharmaceutical industry.

#### **CHAPTER X**

## FUTURE TRENDS AND INNOVATIONS IN PLANT AND NANO BIOTECHNOLOGY Dr.Ambika.K

## Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

The future of pharmaceutical microbiology is poised for significant advancements driven by technological innovations and a deeper understanding of microbial interactions. One notable trend is the increasing use of genomic and metagenomic approaches to study microbial populations. These methods allow researchers to identify and characterize the diverse microbial communities associated with pharmaceutical products and manufacturing environments. By harnessing big data and bioinformatics, scientists can gain insights into microbial behavior, resistance mechanisms, and potential contaminants, ultimately leading to more effective quality control measures and targeted interventions.

Another promising trend is the development of novel antimicrobial agents and therapeutic strategies in response to the growing threat of antimicrobial resistance (AMR). Pharmaceutical microbiology will increasingly focus on discovering alternative treatments, such as bacteriophage therapy and antimicrobial peptides, that can effectively target resistant pathogens. Additionally, the integration of artificial intelligence (AI) and machine learning in drug discovery processes will accelerate the identification of new compounds, optimize formulation strategies, and enhance the prediction of microbial responses to various therapies, paving the way for more personalized medicine.

Furthermore, the ongoing emphasis on sustainability and environmental stewardship will shape the future landscape of pharmaceutical microbiology. As regulatory agencies prioritize the reduction of environmental impacts from pharmaceutical manufacturing, microbiologists will explore greener production methods, such as bioconversion processes using engineered microorganisms. This shift not only aims to improve product safety and efficacy but also to minimize waste and energy consumption. By embracing these emerging trends, pharmaceutical microbiology will continue to evolve, addressing current challenges and fostering innovative solutions that enhance public health outcomes.

# BIOETHICS AND FUNDAMENTALS OF MICROBIOLOGY

Edited by :

DR.N.MAHALAKSHIMI



## **Bioethics and Fundamentals of Microbiology**

Copyright©2021 by Dr.N.MAHALAKSHIMI

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-377-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

Introduction
Chapter I :KIntroduction to Microbiology
Dr.N.Mahalakshimi
Chapter II : Microbial Cell Structure and Function
DR.S.Mohanrai
Chapter III: Microbial Metabolism and Growth
Chapter IV : Microbial Genetics
Dr Tlishadevi
Chapter V: Microbes and Human Health
Dr.D. Sethue
Dr.K.Satnya
Chapter VI: Bioethics in Microbiological Research
Dr.T.Thiruselvi
Chapter VII: Biotechnology and Genetic Engineering of Microbes
Dr.A.Xavier Fernandes
Chapter VIII : Antimicrobial Resistance and Bioethical Challenges
Dr.G. Chandirasegaran
Chapter Ix : Cell Death: Bioethics in Public Health Microbiology
Dr.K.P.Karuppaian
Chapter X : Future of Microbiology and Bioethics
Dr.Ambika K
REFERENCES

#### TABLE OF CONTENTS

#### **CHAPTER I**

#### INTRODUCTION TO MICROBIOLOGY

#### Dr.N.MAHALAKSHIMI

## Assistant professor Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Microbiology is the scientific study of microorganisms, which are tiny, often unicellular entities that include bacteria, viruses, fungi, and protozoa. Despite their minuscule size, these organisms play crucial roles in various ecosystems and have a profound impact on human health, agriculture, and industry. Microbiology encompasses a wide range of topics, from understanding the basic biology and genetics of microorganisms to exploring their interactions with humans and the environment. This diverse field not only seeks to identify and classify these organisms but also examines their physiological processes, pathogenicity, and applications in biotechnology.

The importance of microbiology is underscored by its applications in medicine, agriculture, and environmental science. In healthcare, microbiologists study pathogenic microorganisms to develop vaccines, antibiotics, and diagnostic tools, playing a vital role in combatting infectious diseases. In agriculture, beneficial microbes are harnessed for biopesticides and biofertilizers, enhancing crop productivity while reducing the reliance on chemical inputs. Environmental microbiology explores the roles of microorganisms in biogeochemical cycles, waste decomposition, and bioremediation, offering solutions for pollution control and ecosystem restoration.

As technology advances, the field of microbiology continues to evolve, integrating innovative techniques such as genomics, metagenomics, and bioinformatics. These tools enable scientists to delve deeper into microbial communities, uncovering their diversity and functional capabilities. The ongoing research in microbiology not only enriches our understanding of the microbial world but also drives the development of novel biotechnological applications, from sustainable energy production to personalized medicine. As we navigate challenges such as antimicrobial resistance and emerging infectious diseases, the insights gained from microbiological studies remain vital for improving public health and environmental sustainability

#### **CHAPTER II**

#### **CELL STRUCTURE AND FUNCTION**

#### **DR.S.MOHANRAJ**

## Associate professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajayam Institute of Science and technology, Tamil Nadu, India

Cell structure and function are fundamental concepts in biology that highlight the organization and activities of cells, the basic units of life. Cells can be broadly classified into two main types: prokaryotic and eukaryotic. Prokaryotic cells, such as bacteria, are generally simpler and smaller, lacking a defined nucleus and membrane-bound organelles. Instead, their genetic material is found in a nucleoid region, and they often possess a rigid cell wall that provides structural support. In contrast, eukaryotic cells, found in plants, animals, fungi, and protists, are more complex, featuring a true nucleus that houses their DNA and various organelles, such as mitochondria, endoplasmic reticulum, and Golgi apparatus, each performing specialized functions.

The organization of cellular components is crucial for maintaining homeostasis and carrying out essential life processes. The cell membrane, a phospholipid bilayer with embedded proteins, acts as a selective barrier, regulating the movement of substances in and out of the cell. This membrane's fluidity and flexibility allow for cell signaling and communication with the external environment. Within the cell, organelles perform vital roles: mitochondria are responsible for energy production through cellular respiration, while ribosomes synthesize proteins based on genetic instructions. The interplay among these components is essential for cellular metabolism, growth, and reproduction.

Understanding cell structure and function has profound implications in various fields, including medicine, genetics, and biotechnology. Disruptions in cellular functions can lead to diseases, including cancer, where abnormal cell growth occurs. Additionally, advancements in cell biology have facilitated the development of targeted therapies and regenerative medicine techniques, such as stem cell research. By studying cells at both the structural and functional levels, scientists continue to uncover the intricate mechanisms that sustain life and explore new ways to treat diseases and enhance human health.
#### CHAPTER III

#### MICROBIAL METABOLISM AND GROWTH DR.Dr. K. SUNDAR

# Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Microbial metabolism refers to the biochemical processes that microorganisms use to convert nutrients into energy and cellular components, enabling them to grow, reproduce, and respond to environmental changes. Microbes can be classified based on their metabolic pathways, primarily as autotrophs or heterotrophs. Autotrophic microorganisms, such as certain bacteria and algae, utilize inorganic carbon sources like carbon dioxide to produce organic compounds through processes such as photosynthesis or chemosynthesis. In contrast, heterotrophic microbes obtain their energy and carbon from organic compounds, often decomposing complex materials in their environment to sustain growth and metabolic activities.

The growth of microorganisms is influenced by various factors, including nutrient availability, temperature, pH, and oxygen levels. Microbial growth typically follows a characteristic pattern represented by the growth curve, which includes phases of lag, exponential (log), stationary, and death. During the exponential phase, cells divide at a constant rate, leading to rapid population increase. This phase is crucial for industrial applications, such as fermentation, where maximizing growth rates can enhance product yields. Nutrient limitations or the accumulation of waste products eventually lead to the stationary phase, where growth slows and the rate of cell division equals the rate of cell death.

Understanding microbial metabolism and growth is vital for multiple applications, including biotechnology, agriculture, and medicine. In industrial microbiology, harnessing specific metabolic pathways can optimize the production of valuable compounds such as antibiotics, enzymes, and biofuels. In agriculture, knowledge of microbial growth can inform the development of biofertilizers and biopesticides, enhancing soil health and crop productivity. Additionally, in clinical settings, insights into microbial metabolism can aid in the diagnosis and treatment of infections, particularly as resistance to traditional antibiotics continues to rise.

#### **CHAPTER IV**

# MICROBIAL GENETICS Dr. T.USHADEVI

# Associate professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Microbial genetics is the study of the heredity and variation of microorganisms, focusing on how genetic information is transferred, expressed, and regulated within these organisms. Microbes, including bacteria, archaea, and certain fungi, possess unique genetic mechanisms that differ from those of higher organisms. Key features of microbial genetics include the simplicity of their genomes, which often consist of a single circular chromosome, and the presence of plasmids—small, independent DNA molecules that can carry additional genes, such as those for antibiotic resistance. Understanding these genetic elements is crucial for deciphering microbial behavior and adaptability.

One of the fundamental processes in microbial genetics is gene expression, which involves the transcription of DNA into messenger RNA (mRNA) and subsequent translation into proteins. This process is tightly regulated, allowing microorganisms to respond quickly to environmental changes. Techniques such as operon theory in bacteria illustrate how genes can be organized and controlled in clusters, enabling efficient coordination of related functions. Additionally, horizontal gene transfer (HGT)—the movement of genetic material between organisms—plays a significant role in microbial evolution, contributing to genetic diversity and the rapid spread of traits such as antibiotic resistance.

The field of microbial genetics has profound implications in various domains, including medicine, agriculture, and biotechnology. In clinical microbiology, understanding the genetic basis of pathogenicity helps in developing targeted therapies and vaccines. In agriculture, genetic engineering techniques are applied to enhance the traits of beneficial microbes used as biofertilizers or biopesticides. Moreover, advancements in genomic technologies, such as next-generation sequencing, have revolutionized microbial genetics by enabling comprehensive analyses of microbial genomes, facilitating the identification of novel targets for drug development, and enhancing our understanding of microbial ecosystems.

#### **CHAPTER V**

# MICROBES AND HUMAN HEALTH DR.R.SATHYA

# Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Microbes play a dual role in human health, acting both as beneficial agents and potential pathogens. The human microbiome, which consists of trillions of microorganisms living in and on our bodies, is crucial for maintaining health. These microbes aid in digestion, synthesize essential vitamins, and support the immune system by outcompeting harmful pathogens for resources. A balanced microbiome is essential for overall well-being, and disruptions to this balance—such as through antibiotic use, poor diet, or illness—can lead to conditions like obesity, allergies, and autoimmune diseases.

On the other hand, pathogenic microbes can cause a wide range of infections and diseases, posing significant risks to human health. Bacteria, viruses, fungi, and parasites can invade the body and disrupt normal physiological functions. Common illnesses caused by pathogens include respiratory infections, gastrointestinal diseases, and sexually transmitted infections. Understanding the mechanisms by which these microbes cause disease is crucial for developing effective treatments and preventive measures, such as vaccines and antimicrobial therapies.

Recent advances in microbiome research are revealing new insights into how microbes influence human health beyond traditional infectious disease paradigms. Studies have shown that the composition of the microbiome can affect mental health, metabolic processes, and responses to medications. This burgeoning field of research underscores the importance of maintaining a healthy microbiome and exploring probiotic and prebiotic therapies to restore microbial balance. By recognizing the intricate relationship between microbes and human health, scientists and healthcare professionals can better address health issues and promote overall wellness through targeted interventions

#### **CHAPTER VI**

# BIOETHICS IN MICROBIOLOGICAL RESEARCH DR.T.THIRUSELVI

# Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Bioethics in microbiological research encompasses the moral principles and ethical considerations that guide the conduct of scientific studies involving microorganisms, particularly in areas such as genetic manipulation, antibiotic development, and pathogen research. As advances in microbiology, especially in synthetic biology and genetic engineering, raise complex ethical questions, researchers must navigate issues such as the potential for creating harmful organisms, the implications of altering microbial genomes, and the responsibilities associated with using biotechnology for human benefit. The challenge lies in balancing scientific innovation with societal values, safety, and environmental concerns.

One critical area of bioethics in microbiological research is the responsible use of genetically modified organisms (GMOs). The development of genetically engineered microbes for applications such as bioremediation or the production of pharmaceuticals poses questions about ecological impact, biosafety, and long-term effects on the environment. Researchers must ensure that appropriate risk assessments are conducted, and that regulatory frameworks are followed to mitigate potential hazards. Moreover, public engagement and transparency are essential in fostering trust and addressing societal concerns regarding GMOs, particularly as these technologies become more prevalent in food production and environmental management.

Additionally, the study of pathogenic microorganisms raises ethical dilemmas related to biosafety and dual-use research of concern (DURC), where findings intended for beneficial purposes could be misapplied for malicious activities. Ethical guidelines must be established to ensure that research involving dangerous pathogens is conducted with the utmost safety and oversight, protecting both public health and national security. As microbiology continues to evolve, ongoing discussions around bioethics will be essential in shaping responsible research practices, fostering ethical decision-making, and ensuring that scientific advancements benefit society while minimizing risks.

#### **CHAPTER VII**

### BIOTECHNOLOGY AND GENETIC ENGINEERING OF MICROBES DR.A.XAVIER FERNANDES

# Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Biotechnology and genetic engineering of microbes have revolutionized various fields, including medicine, agriculture, and environmental science. By manipulating the genetic material of microorganisms, scientists can enhance desirable traits, such as increased resistance to pathogens or improved metabolic capabilities. Techniques such as CRISPR-Cas9 and recombinant DNA technology allow for precise modifications, enabling the development of microbial strains that can produce valuable compounds like insulin, vaccines, and biofuels. This level of genetic control opens new avenues for innovation and efficiency in microbial applications.

In medicine, genetically engineered microbes play a crucial role in producing therapeutic proteins and hormones. For instance, insulin-producing bacteria have transformed diabetes management, allowing for large-scale production of this essential hormone. Additionally, engineered microbes are being explored for their potential in targeted drug delivery systems and as living therapeutics, where they can be designed to respond to specific disease markers. The ongoing research in this area promises to lead to breakthroughs in personalized medicine and the treatment of complex diseases.

In agriculture, biotechnology harnesses the capabilities of genetically modified microbes to improve crop yields and sustainability. For example, specific bacteria can be engineered to enhance nutrient availability in soil, promote plant growth, and protect against pests. This biotechnological approach reduces the need for chemical fertilizers and pesticides, contributing to more sustainable farming practices. However, the genetic engineering of microbes in agriculture also raises ethical and ecological considerations, necessitating careful regulation and public engagement to ensure the safety and acceptance of these innovations. As biotechnology continues to advance, its integration with genetic engineering will remain pivotal in addressing global challenges in health, food security, and environmental sustainability.

#### **CHAPTER VIII**

#### ANTIMICROBIAL RESISTANCE AND BIOETHICAL CHALLENGES

#### **Dr.G.CHANDIRASEGARAN**

### Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Antimicrobial resistance (AMR) has emerged as a critical public health challenge, with far-reaching implications for global health systems. As bacteria, viruses, fungi, and parasites evolve mechanisms to resist the effects of antimicrobial agents, the effectiveness of standard treatments diminishes, leading to increased morbidity, mortality, and healthcare costs. The emergence of drug-resistant strains complicates the treatment of infections, necessitating the use of more potent and often more toxic alternatives. This situation not only threatens individual patients but also poses broader societal risks, requiring urgent attention and coordinated action from governments, healthcare providers, and researchers.

The bioethical challenges surrounding AMR are multifaceted and complex. One primary concern is the ethical responsibility of healthcare professionals to prescribe antibiotics judiciously and avoid unnecessary use. Inappropriate prescribing practices contribute significantly to the development of resistance, highlighting the need for robust antimicrobial stewardship programs. Additionally, there are ethical considerations regarding access to effective treatments for all populations. Disparities in healthcare systems can exacerbate the effects of AMR, as low-resource settings may lack access to newer, effective drugs or the means to implement proper infection control measures. Addressing these inequalities is not only a matter of public health but also a fundamental ethical obligation.

Furthermore, the research and development of new antimicrobials face ethical dilemmas related to the prioritization of funding and resources. The financial incentives for pharmaceutical companies to invest in new antibiotic development are often lacking due to the low return on investment compared to chronic disease treatments. This situation raises ethical questions about the sustainability of the pipeline for new antibiotics and the role of public and private sectors in ensuring that effective treatments remain available.

#### **CHAPTER IX**

#### BIOETHICS IN PUBLIC HEALTH MICROBIOLOGY Dr. K.P.KARUPPAIAN

Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

Bioethics in public health microbiology encompasses the ethical principles guiding research, practice, and policy related to the prevention and control of infectious diseases. As public health microbiologists work to understand microbial threats and develop strategies to combat them, they must navigate complex ethical considerations related to research methodologies, data collection, and the implications of their findings. This includes the responsible use of surveillance data to monitor disease outbreaks, ensuring that individual privacy is respected while balancing the need for public safety. Transparent communication with the public about risks, benefits, and the rationale behind public health interventions is essential to maintaining trust and cooperation.

Another critical aspect of bioethics in public health microbiology is the equitable distribution of resources and interventions. Disparities in healthcare access can exacerbate the impact of infectious diseases, particularly in vulnerable populations. Ethical considerations must guide the allocation of vaccines, treatments, and preventive measures, ensuring that marginalized communities receive the support they need. Additionally, public health initiatives must prioritize informed consent and community engagement, allowing individuals to participate in decisions that affect their health and well-being. Engaging communities in the development and implementation of public health strategies fosters a sense of ownership and empowerment, which can enhance compliance and outcomes.

Finally, the ethical implications of emerging technologies in public health microbiology, such as genetic sequencing and biobanking, raise important questions about consent, ownership, and potential misuse of genetic information. As microbiologists utilize these advanced techniques to track pathogens and understand their evolution, they must ensure that ethical standards are upheld, particularly regarding the use of human samples.

#### **CHAPTER X**

#### FUTURE OF MICROBIOLOGY AND BIOETHICS DR. K.AMBIKA

# Assistant professor, Department of Microbiology, School of Arts and Science, Ponnaiyah Ramajeyam Institute of Science and technology, Tamilnadu, India

The future of microbiology is poised to be shaped significantly by rapid advancements in technology, particularly in areas such as genomic sequencing, synthetic biology, and bioinformatics. As researchers gain deeper insights into microbial communities and their complex interactions, the potential for innovative applications in health, agriculture, and environmental management expands. For instance, the ability to sequence entire microbial genomes quickly and cost-effectively will enhance our understanding of pathogen evolution, resistance mechanisms, and microbiome functions, facilitating the development of targeted therapies and sustainable practices. However, these advancements also raise ethical considerations regarding data ownership, privacy, and the implications of genetic manipulation.

As microbiology continues to evolve, the ethical frameworks that guide research and application will need to adapt accordingly. The increasing capacity for genetic engineering, particularly in synthetic biology, necessitates rigorous ethical oversight to ensure that innovations do not inadvertently lead to harmful consequences. This includes potential risks associated with creating novel microorganisms that could disrupt ecosystems or human health. Ethical discussions must involve a diverse range of stakeholders, including scientists, ethicists, policymakers, and the public, to create comprehensive guidelines that reflect societal values and concerns. Promoting transparency and fostering public engagement will be crucial in addressing fears and misconceptions about biotechnological advancements.

Moreover, addressing issues of equity and access will be essential as microbiology progresses. The benefits of new technologies must be distributed fairly across all populations, particularly marginalized communities that may face greater health disparities. Bioethical considerations should encompass not only the research and development phases but also the implementation of interventions derived from microbiological studies. Ensuring that all individuals have equitable access to new diagnostics, treatments, and preventive measures will be vital for promoting global health and addressing challenges such as antimicrobial resistance.

# ELECTROMAGNETIC THEORY

# EDITED BY DR. L. CHINNAPPA



Electromagnetic Theory Copyright © 2021 by Dr. L. Chinnappa

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-351-5

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### **TABLE OF CONTENTS**

#### CHAPTER 1 Introductory Concepts Dr. L. Chinnappa

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Electromagnetic theory is a fundamental branch of physics that describes the interactions between electric charges and magnetic fields. It underpins many technologies in communication, electronics, and energy transmission, influencing everything from wireless communication systems to electric power generation. Understanding the preliminary concepts in electromagnetic theory is crucial for grasping more complex phenomena and applications in modern science and engineering.

#### Basic Concepts of Electricity and Magnetism

At the core of electromagnetic theory are two primary concepts: electric fields and magnetic fields.

Electric Fields: An electric field is created by electric charges and exerts a force on other charges within the field. The strength and direction of the electric field at a point in space are represented by electric field lines, which indicate the force that a positive test charge would experience. The electric field E can be quantified using Coulomb's law, which describes the force between two point charges.

Magnetic Fields: A magnetic field is produced by moving electric charges (currents) and also interacts with other moving charges. It can be represented using magnetic field lines that indicate the direction of the force experienced by a north magnetic pole. The magnetic field B is described by Ampère's law and the Biot-Savart law, which relate the magnetic field to the current that produces it.

#### Maxwell's Equations

The foundation of electromagnetic theory is encapsulated in Maxwell's equations, a set of four fundamental

#### CHAPTER 2 Polarization and magnetization Dr. L. Chinnappa

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Polarization and magnetization are essential concepts in electromagnetic theory, helping to describe the behavior of materials in electric and magnetic fields. These phenomena play a crucial role in understanding how electromagnetic waves interact with matter, which is fundamental to various applications, including telecommunications, optics, and materials science.

#### Polarization

Polarization refers to the orientation of the electric field vector in an electromagnetic wave. While light can oscillate in multiple planes, polarized light oscillates in a specific direction. There are several types of polarization:

Linear Polarization: In this type, the electric field oscillates in a single plane. Linear polarization can occur in two orientations: vertical and horizontal, which are crucial in applications like antennas and camera filters.

Circular Polarization: In circular polarization, the electric field vector rotates in a circular manner as the wave propagates. This can be either right-handed or left-handed, depending on the direction of rotation. Circular polarization is significant in satellite communication and certain imaging systems.

Elliptical Polarization: This is a generalized form of polarization that encompasses both linear and circular polarization, where the electric field traces out an elliptical pattern. It is the most common form of polarization found in natural light.

The ability to manipulate polarization has important implications in various technologies. For instance, polarized sunglasses reduce glare by blocking specific orientations of

#### CHAPTER 3 Electrostatics Dr. M. Sivanantham

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Electrostatics is a fundamental branch of electromagnetic theory that focuses on the study of electric charges at rest and the electric fields they produce. This area of physics lays the groundwork for understanding electric forces, fields, and potentials, serving as a critical foundation for various applications in engineering, electronics, and physics. By exploring the principles of electrostatics, we can gain insights into the behavior of charged particles, the forces they exert on one another, and the fields they create.

The principles of electrostatics, including electric charge, Coulomb's law, electric fields, and electric potential, form the basis for understanding a wide range of physical phenomena and practical applications. As technology continues to advance, the importance of electrostatics remains significant, driving innovations in electronics, engineering, and environmental control. Understanding these concepts is essential for anyone studying physics or working in related fields.

Basic Concepts of Electrostatics

Electric Charge: The concept of electric charge is central to electrostatics. There are two types of charge: positive and negative, typically carried by protons and electrons, respectively. The fundamental principle governing electric charge is the law of conservation, which states that charge cannot be created or destroyed, only transferred.

Coulomb's Law: Coulomb's law quantitatively describes the force between two point charges. It states that the magnitude of the electric force F between two charges  $q_1$  and  $q_2$  is directly proportional to the product of the magnitudes of the charges and

#### CHAPTER 4 Magnetostatics Dr.S.Subashchandrabose

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Magnetostatics is a branch of electromagnetic theory that deals with magnetic fields in systems where currents are steady, meaning they do not change with time. Unlike the dynamic fields of electromagnetism described by Maxwell's equations in their general form, magnetostatics focuses on the magnetic field produced by constant currents, similar to how electrostatics deals with stationary charges. This area of study is fundamental for understanding the behavior of magnets, the design of electrical devices like inductors and transformers, and the analysis of magnetic materials.

Magnetostatics provides a fundamental understanding of magnetic interactions, the behavior of materials in magnetic fields, and the principles underlying numerous technological applications. Mastering the concepts of magnetostatics is crucial for engineers, physicists, and anyone working in fields related to electromagnetism and magnetic materials.

#### Fundamental Concepts of Magnetostatics

Magnetic Field (B): The magnetic field, often represented by the symbol B, is a vector field that describes the magnetic influence on moving charges, such as currents. The direction of the magnetic field at any point is the direction of the force that a north magnetic pole would experience, and its magnitude is proportional to the strength of this force.

Biot-Savart Law: The Biot-Savart law provides a method for calculating the magnetic field generated by a small segment of current-carrying conductor. It states that the magnetic field dB at a point in space due to an infinitesimal element of current Idl is given by:

#### CHAPTER 5 Electromotive Force Dr. M. Silambarasan

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Electromotive Force (EMF) is a fundamental concept in electromagnetic theory, representing the driving force that causes electric current to flow in a circuit. Despite its name, EMF is not a force in the traditional sense but rather a measure of the energy supplied by a source per unit charge to move charges through a circuit. Understanding EMF is crucial for comprehending how electrical energy is generated, transmitted, and utilized in various applications, from batteries and generators to transformers and power systems.

Electromotive force is a central concept in electromagnetic theory that describes the ability of a source to provide energy to move electric charges in a circuit. It is essential for understanding the operation of batteries, generators, and other devices that convert various forms of energy into electrical energy. Mastery of electromotive force principles is crucial for engineers, physicists, and technologists involved in the design and application of electrical and electronic systems.

#### Definition of Electromotive Force

Electromotive force, often denoted by the symbol  $\varepsilon$ , is defined as the energy provided by a source (such as a battery or generator) to move a unit charge through the entire circuit, including both the internal and external resistances. It is measured in volts (V). Mathematically, EMF is expressed as:

$$\varepsilon = \frac{dW}{dq}$$

#### CHAPTER 6 Electromagnetic Waves Dr. V. Vidhya and Dr. Sutapa Ghosh

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Electromagnetic waves are a cornerstone of electromagnetic theory, describing the propagation of energy and information through space via oscillating electric and magnetic fields. These waves are fundamental to understanding a wide array of natural phenomena and technological applications, including light, radio waves, X-rays, and more. From the visible light we see to the radio signals used in communication, electromagnetic waves are pervasive in both the natural world and human technology.

Electromagnetic waves are encompassing a vast range of phenomena and applications. They enable the transfer of energy and information across vast distances and play a crucial role in modern technology, science, and daily life. Understanding electromagnetic waves, from their theoretical foundations in Maxwell's equations to their practical applications across the electromagnetic spectrum, is essential for physicists, engineers, and anyone engaged with the technological advancements that shape our world.

#### Nature of Electromagnetic Waves

Electromagnetic waves consist of oscillating electric (E) and magnetic (B) fields that are perpendicular to each other and to the direction of wave propagation. They are generated by accelerating electric charges and do not require a medium for propagation, meaning they can travel through the vacuum of space. This distinguishes them from mechanical waves, such as sound, which need a material medium.

Maxwell's equations provide the theoretical foundation for electromagnetic waves. These four equations describe how electric and magnetic fields are generated and altered by each

#### CHAPTER 7 Application of Electromagnetic Waves Dr. V. Vidhya and Dr. Sutapa Ghosh

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Electromagnetic waves are integral to modern technology and science, finding applications across a wide spectrum of fields due to their ability to transmit energy and information over vast distances. From communication and medical imaging to industrial processes and scientific exploration, electromagnetic waves play a crucial role in numerous technological advancements. Understanding their diverse applications not only highlights their versatility but also underscores the foundational role of electromagnetic theory in technological innovation.

The applications of electromagnetic waves are vast and diverse, impacting nearly every aspect of modern life. From enabling global communication and medical diagnostics to advancing scientific research and industrial processes, electromagnetic waves have become a cornerstone of technological progress. The ability to harness and manipulate these waves, as described by electromagnetic theory, has transformed how we interact with the world and the universe beyond, making them indispensable tools in both scientific exploration and everyday technology. Understanding the principles behind these applications is essential for continued innovation and the development of new technologies that leverage the power of electromagnetic waves.

#### *Key Applications of Electromagnetic Waves* Wireless Communication

One of the most prominent applications of electromagnetic waves is in wireless communication, which utilizes various parts of the electromagnetic spectrum to transmit information over long distances. Radio waves and microwaves are used for:

# SOLAR ENERGY UTILIZATION

Edited by DR.S.SUBASHCHANDRABOSE



Solar Energy Utilization Copyright © 2021 by *Dr. S. Subashchandrabose* 

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-275-4

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### **TABLE OF CONTENTS**

Introduction
CHAPTER 1 Heat Transfer21
Dr. S. Subashchandrabose
CHAPTER 2 Radiation Analysis
Dr. M. Silambarasan
CHAPTER 3 Solar Collectors
Dr. M. Sivanantham
CHAPTER 4 Solar Heaters
Dr. Sutapa Ghosh
CHAPTER 5 Solar Energy Conversion
Dr. V. Vidhya
CHAPTER 6 Nanomaterials in Fuel Cell Applications
Dr. L. Chinnappa and Dr. M. Silambarasan
CHAPTER 7 Nanomaterials in Solar Cell Applications
Dr. M. Sivanantham and Dr. K. Thirunavukarasu
REFERENCES125

#### CHAPTER 1 Heat Transfer Dr. S. Subashchandrabose

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Heat transfer is a fundamental concept in thermodynamics that involves the movement of thermal energy from one body or system to another. This process occurs in various forms and plays a crucial role in numerous applications, from engineering and environmental science to everyday life. Understanding heat transfer is essential for designing efficient thermal systems, managing energy consumption, and improving comfort in buildings and vehicles. This introduction explores the basic principles of heat transfer, its modes, and its significance in various fields.

Modes of Heat Transfer

Heat transfer occurs through three primary modes:

Conduction:

Conduction is the transfer of heat through a solid material without any motion of the material itself. It occurs at the molecular level as faster-moving particles collide with slower-moving ones, transferring energy. The rate of conduction is influenced by the material's thermal conductivity, thickness, and temperature gradient. Common examples include heating one end of a metal rod and feeling the heat at the other end or the transfer of heat through walls of a building. Convection:

Convection involves the transfer of heat through fluids (liquids and gases) by the movement of the fluid itself. This process can be natural (driven by buoyancy forces due to temperature differences) or forced (induced by external means like fans or pumps). In natural convection, warmer fluid rises while cooler fluid sinks, creating circulation. An example of convection is the warming of air in a room by a radiator or the cooling of a hot beverage as it sits in the open air.

#### CHAPTER 2 Radiation Analysis Dr. M. Silambarasan

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Radiation analysis is a critical field of study that focuses on the measurement, evaluation, and interpretation of radiation emissions from various sources. This analysis plays a vital role in numerous applications, including environmental monitoring, medical diagnostics, nuclear energy, and safety assessments. Understanding radiation, its properties, and its effects on matter is essential for ensuring public health, environmental protection, and the safe use of radiation in technology. This introduction explores the principles of radiation analysis, its methods, applications, and significance in various fields.

Types of Radiation

Radiation can be categorized into two primary types: ionizing and non-ionizing radiation.

Ionizing Radiation:

This type includes high-energy particles or electromagnetic waves capable of removing tightly bound electrons from atoms, thus ionizing them. Common forms of ionizing radiation include alpha particles, beta particles, gamma rays, and X-rays. Ionizing radiation is often used in medical applications, such as X-ray imaging and radiation therapy for cancer treatment.

Non-Ionizing Radiation:

This form includes lower-energy radiation that does not have enough energy to ionize atoms or molecules. Examples include ultraviolet (UV) light, visible light, microwaves, and radio waves. Non-ionizing radiation is commonly encountered in everyday life and is used in technologies such as telecommunications and microwave ovens.

#### CHAPTER 3 Solar Collectors Dr. M. Sivanantham

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Solar collectors are devices designed to capture and convert solar energy into usable heat, playing a crucial role in the broader field of solar thermal energy systems. These collectors are essential for harnessing sunlight to provide heating for residential, commercial, and industrial applications. As the demand for renewable energy sources increases due to concerns about climate change and energy security, solar collectors offer an efficient and sustainable solution for reducing reliance on fossil fuels. This introduction explores the types of solar collectors, their working principles, applications, advantages, and challenges.

Types of Solar Collectors

Solar collectors can be categorized into three main types based on their design and operating principles:

Flat-Plate Collectors:

These are the most common type of solar collectors, consisting of a flat, insulated box with a transparent cover (usually glass) that allows sunlight to enter. Inside, a dark absorber plate absorbs solar radiation and converts it into heat. Fluid (typically water or antifreeze) circulates through pipes attached to the absorber, carrying the heat away for use in heating applications. Flat-plate collectors are widely used for residential water heating, swimming pool heating, and space heating.

Evacuated Tube Collectors:

These collectors consist of a series of glass tubes, each containing a dark absorber coated with a selective material. The tubes are evacuated of air, which minimizes heat loss through convection and conduction. As sunlight strikes the absorber, heat is transferred to a fluid that circulates within the tubes.

#### CHAPTER 4 Solar Heaters Dr. Sutapa Ghosh

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Solar heaters, also known as solar water heating systems, are devices that harness solar energy to provide hot water for various applications, such as residential and commercial heating, swimming pools, and industrial processes. Utilizing renewable solar energy, these systems significantly reduce reliance on fossil fuels and contribute to lower greenhouse gas emissions. As global concerns about climate change and energy sustainability continue to rise, solar heaters offer an efficient and eco-friendly solution for heating needs. This introduction explores the principles of solar heating, types of solar heaters, their applications, benefits, and challenges. Principles of Solar Heating

Solar heaters operate on the fundamental principle of converting sunlight into thermal energy. This process typically involves solar collectors that absorb solar radiation and transfer the collected heat to a fluid (usually water or a heat transfer fluid) that circulates through the system. The heated fluid can then be used directly for heating or stored for later use. The efficiency of solar heaters depends on factors such as the design of the collectors, the amount of sunlight available, and the temperature difference between the collector and the surrounding environment.

Types of Solar Heaters

Flat-Plate Solar Heaters:

These systems consist of a flat, insulated box with a transparent cover and a dark absorber plate inside. The absorber collects sunlight and converts it into heat, which is transferred to the circulating fluid. Flat-plate solar heaters are commonly used for residential water heating and are effective in various climates.

#### CHAPTER 5 Solar Energy Conversion Dr. V. Vidhya

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Solar energy conversion refers to the process of transforming sunlight into usable energy forms, primarily electricity and heat. As one of the most abundant and renewable energy sources available, solar energy plays a crucial role in addressing global energy demands while reducing reliance on fossil fuels and minimizing greenhouse gas emissions.

Solar energy conversion is a key component of the transition towards renewable energy sources, offering a sustainable and clean solution to meet the growing global energy demands. By harnessing the power of the sun, solar technologies contribute to reducing carbon emissions, promoting energy independence, and fostering economic growth. As advancements continue in solar energy conversion technologies, their widespread adoption will play a pivotal role in creating a more sustainable and resilient energy future.

This introduction explores the principles of solar energy conversion, the technologies involved, applications, advantages, and challenges associated with harnessing solar power.

Principles of Solar Energy Conversion

Solar energy conversion is based on the ability to capture sunlight and convert it into other energy forms. This process typically involves two main pathways:

Photovoltaic (PV) Conversion:

Photovoltaic conversion involves the direct conversion of sunlight into electricity using semiconductor materials. When sunlight strikes a PV cell, it excites electrons within the material, creating an electric current. This phenomenon is governed by the photovoltaic effect, which allows solar panels to generate electricity under sunlight.

#### CHAPTER 6 Nanomaterials in Fuel Cell Applications Dr. L. Chinnappa and Dr. M. Silambarasan

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

The integration of nanomaterials in fuel cell technology represents a promising advancement in the pursuit of clean energy solutions, particularly when combined with solar energy. Fuel cells, which convert chemical energy into electricity through electrochemical reactions, can benefit significantly from the unique properties of nanomaterials. As the world transitions towards sustainable energy sources, the synergy between solar energy and fuel cell technology, enhanced by nanomaterials, offers innovative pathways for efficient energy conversion and storage.

Nanomaterials are poised to play a transformative role in enhancing the performance and efficiency of fuel cells utilizing solar energy. By leveraging their unique properties, these materials can optimize energy conversion processes and facilitate the integration of solar power with fuel cell technology. As research continues to address challenges related to scalability, stability, and environmental impact, the future of nanomaterials in solar-driven fuel cell applications appears promising. This synergy not only supports the development of cleaner energy solutions but also contributes to the global effort to transition to sustainable energy systems, paving the way for a more resilient and environmentally friendly energy landscape.

This introduction explores the role of nanomaterials in fuel cell applications utilizing solar energy, their benefits, types, and future prospects.

The Role of Nanomaterials in Solar-Driven Fuel Cells

Nanomaterials, defined by their size (1 to 100 nanometers) and unique physical and chemical properties, enhance fuel cell performance in several critical areas. Their high surface area, tunable morphology, and improved

#### CHAPTER 7

#### Nanomaterials in Solar Cell Applications Dr. M. Sivanantham and Dr. K. Thirunavukarasu

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Nanomaterials have revolutionized the field of solar energy conversion, offering innovative solutions that enhance the efficiency and performance of solar cells. With the growing demand for renewable energy sources and the urgent need to combat climate change, advancements in solar technology are crucial. Nanomaterials, characterized by their unique properties at the nanoscale, play a pivotal role in improving the efficiency, stability, and overall functionality of solar cells.

Nanomaterials are at the forefront of advancing solar cell technology, offering innovative solutions that enhance efficiency, reduce costs, and improve stability. As the demand for renewable energy continues to rise, the role of nanomaterials in solar cell applications will become increasingly critical. Ongoing research and development in this field hold the promise of unlocking new possibilities for sustainable energy solutions, making solar energy more efficient and accessible to meet the global energy challenge. With continued innovation, nanomaterials could play a vital role in the transition to a more sustainable and resilient energy future. This introduction explores the significance of nanomaterials in solar cell applications, their types, benefits, and future prospects. The Role of Nanomaterials in Solar Cells

Nanomaterials possess distinctive optical, electrical, and mechanical properties that make them highly effective in solar energy applications. Their high surface area, tunable band gaps, and enhanced light absorption capabilities allow for more efficient energy conversion. In solar cells, nanomaterials can be incorporated into various components, including absorbers, charge transport layers, and electrodes, leading to significant performance improvements.

# ADVANCED OPTICS

Edited by

**DR. K. VINOTH** 



Advanced Optics Copyright © 2021 by Dr. K. Vinoth

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means without written permission from the author.

ISBN: 978-93-6255-666-0

Publisher: Ponnaiyah Ramajayam Institute of Science and Technology (PRIST)

#### **TABLE OF CONTENTS**

Introduction
CHAPTER 1 Basic Concepts in Optics21
Dr. Sutapa Ghosh and Dr. K. Vinoth
CHAPTER 2 Polarization
Dr. M. Silambarasan
CHAPTER 3 Double Refraction
Dr. M. Sivanantham
CHAPTER 4 Lasers
Dr. S. Subashchandrabose
CHAPTER 5 Fiber Optics74
Dr. L. Chinnappa and Dr. V. Vidhya
CHAPTER 6 Non-Linear Optics
Dr. M. Silambarasan and Dr. R. Muraleedharan
CHAPTER 7 Magneto-Optics and Electro-Optics
Dr. M. Sivanantham and Dr. K. Thirunavukarasu
REFERENCES

#### CHAPTER 1 Basic Concepts in Optics Dr. Sutapa Ghosh and Dr. K. Vinoth

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Optics is the branch of physics that studies the behavior and properties of light, including its interactions with matter. Understanding the fundamental concepts of optics is essential for a variety of applications, from the design of optical instruments to the development of advanced technologies in fields like telecommunications, medicine, and photography.

Basic concepts in optics form the foundation for understanding how light behaves and interacts with various materials. From the nature of light to the principles of reflection and refraction, these concepts are vital for the development of numerous optical devices and technologies. As advancements in optics continue to evolve, they promise to enhance various including telecommunications, medicine, fields. and entertainment, making optics an essential area of study in both science and engineering. Understanding these fundamental principles is crucial for anyone interested in the applications of light in technology and everyday life. This introduction covers the key concepts in optics, including the nature of light, the laws of reflection and refraction, optical phenomena, and various optical devices.

Nature of Light:

Light is an electromagnetic wave that travels at a speed of approximately 299,792 kilometers per second in a vacuum. It exhibits both wave-like and particle-like properties, a duality that is fundamental to quantum mechanics. The electromagnetic spectrum encompasses a range of wavelengths, from radio waves to gamma rays, with visible light being just a small portion of this spectrum. Understanding the properties of light, including its wavelength, frequency, and speed, is crucial for exploring how it interacts with different materials.

#### CHAPTER 2 Polarization Dr. M. Silambarasan

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Polarization is a fundamental optical phenomenon that describes the orientation of the oscillations of light waves. While light is generally considered to be unpolarized, meaning its electric field oscillates in multiple directions perpendicular to its direction of propagation, polarization occurs when these oscillations are restricted to a particular direction. Understanding polarization is crucial for various applications in optics, telecommunications, and imaging technologies.

Polarization is a vital concept in optics that plays a significant role in a variety of scientific and technological applications. By understanding the principles of polarization, researchers and engineers can develop innovative solutions that enhance imaging, communication, and visual experiences. As technology continues to evolve, the exploration of polarization will remain integral to advancements in fields ranging from telecommunications to medical imaging, making it an essential area of study in physics and engineering. This introduction explores the concepts of polarization, methods of achieving it, types of polarization, and its practical applications. Understanding Polarization

Light can be modeled as an electromagnetic wave consisting of electric and magnetic fields oscillating perpendicularly to each other and the direction of wave propagation. When light waves oscillate in more than one direction, they are classified as unpolarized. However, when the electric field vector is confined to a single plane, the light is considered polarized. This confinement can be achieved through various processes, resulting in different types of polarization.

#### CHAPTER 3 Double Refraction Dr. M. Sivanantham

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Double refraction, also known as birefringence, is an optical phenomenon observed in certain anisotropic materials, where light entering the material is split into two distinct rays, each traveling at different speeds and refracted at different angles. This phenomenon arises due to the material's unique structural properties, which cause it to have different refractive indices depending on the polarization and direction of the incoming light. Understanding double refraction is essential for various applications in optics, mineralogy, and materials science.

Double refraction is a fascinating optical phenomenon that arises in anisotropic materials, resulting in the splitting of light into two distinct rays. Its unique properties and applications in various fields make it a critical area of study in optics and materials science. From mineral identification to advancements in display technologies, the principles of double refraction continue to contribute to our understanding of light and its interaction with matter, paving the way for innovative applications in science and engineering. As research advances, the exploration of birefringent materials promises to unlock new possibilities in both fundamental and applied optics. Here we explore the principles of double refraction, its causes, and its practical applications.

Principles of Double Refraction

When unpolarized light enters a birefringent material, it is divided into two polarized rays:

Ordinary Ray (O-ray):

This ray follows Snell's Law and experiences a refractive index that remains constant, regardless of the light's

#### CHAPTER 4 Lasers Dr. S. Subashchandrabose

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Lasers, an acronym for Light Amplification by Stimulated Emission of Radiation, are revolutionary devices that produce highly focused, coherent beams of light. Since their invention in the early 1960s, lasers have become integral to various fields, including telecommunications, medicine, manufacturing, and entertainment. Lasers are remarkable tools that have transformed numerous industries and scientific fields unique properties through their of coherence. monochromaticity, and directionality. Understanding the fundamental principles of laser operation and their diverse types effective application allows for their in medicine, manufacturing, telecommunications, and entertainment. As technology continues to evolve, the development of new laser applications promises to drive further systems and advancements, enhancing their role in innovation and improving various aspects of everyday life. The ongoing exploration of lasers will undoubtedly lead to exciting breakthroughs in the future. This chapter explores the fundamental principles behind lasers, their types, and their diverse applications.

Principles of Laser Operation

The operation of a laser is based on three primary processes: spontaneous emission, stimulated emission, and optical amplification.

Spontaneous Emission:

When atoms or molecules are excited to a higher energy state, they can spontaneously release energy in the form of photons, which are the fundamental particles of light. This process occurs randomly and is the initial step in the laser process.

#### CHAPTER 5 Fiber Optics Dr. L. Chinnappa and Dr. V. Vidhya

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

#### Introduction

Fiber optics is a transformative technology that utilizes thin strands of glass or plastic fibers to transmit data in the form of light signals over long distances. This innovative approach to communication has revolutionized telecommunications, data networking, and various other applications by enabling highspeed data transfer with minimal loss.

Fiber optics has revolutionized the way data is transmitted, providing high-speed, efficient, and secure communication solutions across various industries. By understanding the principles and advantages of fiber optic technology, we can appreciate its critical role in the modern world. As advancements in fiber optics continue to evolve, they promise to enhance connectivity and drive innovation in telecommunications, medicine, and many other fields, shaping the future of communication and information technology. This introduction explores the principles of fiber optics, its types, advantages, and wide-ranging applications.

Principles of Fiber Optics The fundamental operation of fiber optics is based on the principle of total internal reflection. When light enters a fiber optic cable, it travels through the core (the central part of the fiber) and reflects off the cladding (the outer layer), which has a lower refractive index. This reflection allows the light to be guided along the fiber, even when the fiber is bent or twisted. The key components of a fiber optic system include:

Core:

The core is the innermost part of the fiber, typically made of glass or plastic, where the light travels. Its diameter can vary, influencing the fiber's performance. Cladding:
### CHAPTER 6 Non-Linear Optics Dr. M. Silambarasan and Dr. R. Muraleedharan

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

## Introduction

Non-linear optics is a branch of optics that studies the behavior of light in non-linear media, where the response of the material to an electromagnetic field is not directly proportional to the intensity of that field. Unlike linear optics, where the refractive index and other optical properties remain constant regardless of the light intensity, non-linear optics explores phenomena that arise when high-intensity light interacts with matter. This field has significant implications in various applications, including telecommunications, laser technology, and imaging systems.

Non-linear optics is a dynamic and evolving field that deepens our understanding of light-matter interactions under high-intensity conditions. The unique phenomena that arise from non-linear effects have significant implications for technology, influencing advancements modern in telecommunications, laser applications, and medical imaging. As research continues to explore the potential of non-linear optics, it promises to unlock new avenues for innovation, driving progress in various scientific and engineering disciplines. Understanding non-linear optics is essential for harnessing the full potential of light in today's technologically advanced society. This introduction delves into the principles, phenomena, and applications of non-linear optics. Principles of Non-Linear Optics

In non-linear optics, the electric field of a light wave induces a polarization in the medium that depends on the intensity of the light. This leads to a range of phenomena, primarily described by non-linear polarization equations. The key concepts include:

Non-Linear Polarization:

# CHAPTER 7 Magneto-Optics and Electro-Optics Dr. M. Sivanantham and Dr. K. Thirunavukarasu

Department of Physics, PRIST Deemed to be University, Thanjavur-613403, Tamilnadu, India

# Introduction

Magneto-optics and electro-optics are two significant branches of applied optics that explore the interaction between electromagnetic fields and light in various media. These fields have profound implications for both fundamental research and practical applications, ranging from data storage and telecommunications to imaging and sensor technologies. This introduction provides an overview of magneto-optics and electro-optics, their principles, key phenomena, and applications.

### Magneto-Optics

Magneto-optics refers to the study of the interaction between magnetic fields and light in materials. When a magnetic field is applied to certain materials, their optical properties change, allowing for unique phenomena that can be exploited in various applications.

#### Key Principles

Faraday Effect:

One of the most prominent phenomena in magnetooptics, the Faraday effect, describes the rotation of the plane of polarization of light as it passes through a magnetized medium. The degree of rotation depends on the strength of the magnetic field and the material properties.

#### Kerr Effect:

This effect involves the polarization changes that occur when light reflects off a magnetized surface. The Kerr effect can manifest as changes in both the polarization state and the intensity of reflected light.